BELLSOUTH® / CLEC Agreement

Customer Name: ITC DeltaCom Communications, Inc.

ITC^DeltaCom 4 State Agreement	2
GT&Cs Part A	3
GT&Cs Part B	19
Attachment 1	24
Attachment 2	39
Attachment 3	110
Attachment 4	139
Attachment 5	181
Attachment 6	187
Attachment 7	211
Attachment 8	230
Attachment 9	231
Attachment 10	233
Att 10 SQM	234
Attachment 11	397
Attachment 11, Table 1	401
5_29_02-SC UNE Rate Amend	521
6/25/02 - Perf Meas Amend	553
SC Collocation Amend - 7/3/02	743
ITC DeltaCom - SC Port Amendment	748
ITC DeltaCom - KY, LA & MS Port Amend	750

Note: This page is not part of the actual signed contract/amendment, but is present for record keeping purposes only.

By and Between

BellSouth Telecommunications, Inc.

And

ITC DeltaCom Communications, Inc.

AGREEMENT

THIS AGREEMENT is made by and between BellSouth Telecommunications, Inc., ("BellSouth"), a Georgia corporation, and ITC^DeltaCom Communications, Inc., d/b/a ITC^DeltaCom, hereinafter referred to as ("ITC^DeltaCom") an Alabama corporation, and shall be deemed effective as of the date of the last signature of both Parties ("Effective Date"). This agreement may refer to either BellSouth or ITC^DeltaCom or both as a "Party" or "Parties."

WITNESSETH

WHEREAS, BellSouth is an incumbent local exchange telecommunications company authorized to provide telecommunications services in the states of Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee; and

WHEREAS, ITC^DeltaCom is a competitive local exchange telecommunications company ("CLEC") authorized to provide telecommunications services in the state(s) of Kentucky, Louisiana, Mississippi, and South Carolina; and

WHEREAS, the Parties wish to interconnect their facilities, purchase unbundled elements, and exchange traffic specifically for the purposes of fulfilling their obligations pursuant to sections 251 and 252 of the Telecommunications Act of 1996 ("the Act").

NOW THEREFORE, in consideration of the mutual agreements contained herein, BellSouth and ITC^DeltaCom agree as follows:

1. Term of the Agreement

- 1.1 The term of this Agreement shall expire on December 31, 2002.
- 1.2 The Parties agree that by no later than one hundred and eighty (180) days prior to the expiration of this Agreement, they shall commence negotiations with regard to the terms, conditions and prices of local interconnection to be effective beginning on the expiration date of this Agreement ("Subsequent Agreement").
- 1.3 If, within one hundred and thirty-five (135) days of commencing the negotiation referred to in Section 1.2, above, the Parties are unable to satisfactorily negotiate new local interconnection terms, conditions and prices, either Party may petition the Commission to establish appropriate local interconnection arrangements pursuant to 47 U.S.C. 252. The Parties agree that, in such event, they shall encourage the Commission to issue its order regarding the appropriate local interconnection arrangements no later than the expiration date of this Agreement. The Parties further agree that in the event the Commission does not issue its order prior to the expiration date of this Agreement, or if the Parties continue beyond the expiration date of this Agreement to negotiate the local

interconnection arrangements without Commission intervention, the terms, conditions and prices ultimately ordered by the Commission, or negotiated by the Parties, will be effective retroactive to the day following the expiration date of this Agreement. Until the Subsequent Agreement becomes effective, the Parties shall continue to exchange traffic pursuant to the terms and conditions of this Agreement.

- 1.4 Except as otherwise provided herein, the rates and terms of this Interconnection Agreement shall commence as of the Effective Date and shall not be applied retroactively prior to the Effective Date.
- 1.5 Notwithstanding any term or condition contained in this Interconnection Agreement, including Section 1.4 above, the Confidential Settlement Agreement by and between ITC^DeltaCom and BellSouth dated September 29, 2000, shall not be deemed to be modified or altered by anything contained herein and is intended by the Parties to remain in full force and effect. The terms this Interconnection Agreement that were agreed upon as part of such Confidential Settlement Agreement shall be applied retroactively to September 29, 2000.

2. Ordering Procedures

- 2.1 Detailed procedures for ordering and provisioning BellSouth services are set forth in BellSouth's Local Interconnection and Facility Based Ordering Guide, Resale Ordering Guide, and as set forth in Attachment 6 of this Agreement, as appropriate.
- 2.2 BellSouth has developed electronic systems for placing most resale and some UNE orders. BellSouth has also developed electronic systems for accessing data needed to place orders including valid address, available services and features, available telephone numbers, due date estimation on pre-order and calculation on firm order, and customer service records where applicable. Charges for OSS shall be as set forth in Attachment 1, Exhibit A, Attachment 2 and in Attachment 11 of this Agreement.

3. Parity

- 3.1 The services and service provisioning that BellSouth provides ITC^DeltaCom for resale will be at least equal in quality to that provided to BellSouth, or any BellSouth subsidiary, affiliate or end user. In connection with resale, BellSouth will provide ITC^DeltaCom with pre-ordering, ordering, maintenance and trouble reporting, and daily usage data functionality that will enable ITC^DeltaCom to provide equivalent levels of customer service to their local exchange customers as BellSouth provides to its own end users.
- 3.2 BellSouth shall also provide ITC^DeltaCom with unbundled network elements, and access to those elements. The quality of an unbundled network element, as

well as the quality of the access to such unbundled network element, that BellSouth provides to ITC^DeltaCom shall be at least equal in quality to that which BellSouth provides to itself or to any BellSouth subsidiary, affiliate or other CLEC. The terms and conditions pursuant to which BellSouth provides access to unbundled network elements, including but not limited to the time within which BellSouth provisions such access to unbundled network elements, shall, at a minimum, be no less favorable to ITC^DeltaCom than the terms and conditions under which BellSouth provisions such elements to itself. Consistent with all applicable rules and regulations, BellSouth shall provide ITC^DeltaCom with preordering, ordering, provisioning, maintenance and repair, and billing functionality at least equal to that which BellSouth provides for its own retail services.

4. White Pages Listings

BellSouth shall provide ITC^DeltaCom and their customers access to white pages directory listings under the following terms:

- 4.1 <u>Listings</u>. BellSouth or its agent will include ITC^DeltaCom residential and business customer listings in the appropriate White Pages (residential and business) or alphabetical directories. Directory listings will make no distinction between ITC^DeltaCom and BellSouth subscribers.
- 4.2 <u>Rates.</u> Subscriber primary listing information in the White Pages shall be provided at no charge to ITC^DeltaCom or its subscribers provided that ITC^DeltaCom provides subscriber listing information to BellSouth at no charge.
- 4.3 Procedures for Submitting ITC^DeltaCom Subscriber Information. BellSouth will provide to ITC^DeltaCom a magnetic tape or computer disk containing the proper format for submitting subscriber listings. ITC^DeltaCom will be required to provide BellSouth with directory listings and daily updates to those listings, including new, changed, and deleted listings, in an industry-accepted format. These procedures are detailed in BellSouth's Local Interconnection and Facility Based Ordering Guide.
- 4.4 <u>Unlisted and Non-Published Subscribers</u>. ITC^DeltaCom will be required to provide to BellSouth the names, addresses and telephone numbers of all ITC^DeltaCom customers that wish to be omitted from directories.
- 4.5 Inclusion of ITC^DeltaCom Customers in Directory Assistance Database.

 BellSouth will include and maintain ITC^DeltaCom subscriber listings in BellSouth's directory assistance databases at no charge. BellSouth and ITC^DeltaCom will formulate appropriate procedures regarding lead time, timeliness, format and content of listing information.
- 4.6 <u>Listing Information Confidentiality</u>. BellSouth will accord ITC^DeltaCom's directory listing information the same level of confidentiality that BellSouth accords its own directory listing information, and BellSouth shall limit access to

- ITC^DeltaCom's customer proprietary confidential directory information to those BellSouth employees who are involved in the preparation of listings.
- 4.7 Optional Listings. Additional listings and optional listings will be offered by BellSouth at tariffed rates as set forth in the General Subscriber Services Tariff.
- 4.8 <u>Delivery</u>. BellSouth or its agent shall deliver White Pages directories to ITC^DeltaCom subscribers at no charge.
- 4.9 Notwithstanding any provision(s) to the contrary, ITC^DeltaCom shall provide to BellSouth, and BellSouth shall accept, ITC^DeltaCom's Subscriber Listing Information (SLI) relating to ITC^DeltaCom's customers in the geographic area(s) covered by this Interconnection Agreement. ITC^DeltaCom authorizes BellSouth to release all such ITC^DeltaCom SLI provided to BellSouth by ITC^DeltaCom to qualifying third parties via either license agreement or BellSouth's Directory Publishers Database Service (DPDS), General Subscriber Services Tariff, Section A38.2, as the same may be amended from time to time. Such CLEC SLI shall be intermingled with BellSouth's own customer listings and listings of any other CLEC that has authorized a similar release of SLI. Where necessary, BellSouth will use good faith efforts to obtain state commission approval of any necessary modifications to Section A38.2 of its tariff to provide for release of third party directory listings, including modifications regarding listings to be released pursuant to such tariff and BellSouth's liability thereunder. BellSouth's obligation pursuant to this Section shall not arise in any particular state until the commission of such state has approved modifications to such tariff.
- 4.10 No compensation shall be paid to ITC^DeltaCom for BellSouth's receipt of ITC^DeltaCom SLI, or for the subsequent release to third parties of such SLI. In addition, to the extent BellSouth incurs costs to modify its systems to enable the release of ITC^DeltaCom's SLI, or costs on an ongoing basis to administer the release of ITC^DeltaCom's SLI, ITC^DeltaCom shall pay to BellSouth its proportionate share of the reasonable costs associated therewith.
- 4.11 BellSouth shall not be liable for the content or accuracy of any SLI provided by ITC^DeltaCom under this Agreement. Subject to the Limitation of Liability provisions of Section 6, ITC^DeltaCom shall indemnify, hold harmless and defend BellSouth from and against any damages, losses, liabilities, demands claims, suits, judgments, costs and expenses (including but not limited to reasonable attorneys' fees and expenses) arising from BellSouth's tariff obligations or otherwise and resulting from or arising out of any third party's claim of inaccurate ITC^DeltaCom listings or use of the SLI provided pursuant to this Agreement. BellSouth may forward to ITC^DeltaCom any complaints received by BellSouth relating to the accuracy or quality of ITC^DeltaCom listings.
- 4.12 Listings and subsequent updates will be released consistent with BellSouth system changes and/or update scheduling requirements.

5. Bona Fide Request/New Business Request Process for Further Unbundling

If ITC^DeltaCom is a facilities based provider or a facilities based and resale provider, this section and Attachment 9 shall apply. BellSouth shall, upon request of ITC^DeltaCom, provide to ITC^DeltaCom access to its unbundled elements at any technically feasible point for the provision of ITC^DeltaCom 's telecommunications service where such access is necessary and failure to provide access would impair the ability of ITC^DeltaCom to provide services that it seeks to offer. Any request by ITC^DeltaCom for access to an unbundled element that is not already available shall be treated as an unbundled element Bona Fide Request/New Business Request, and shall be submitted to BellSouth pursuant to the Bona Fide Request/New Business Request process set forth following.

- 5.1 Bona Fide Request/New Business Requests are to be used when ITC^DeltaCom makes a request of BellSouth to provide a new or modified network element, interconnection option. or other service option pursuant the Telecommunications Act of 1996; or to provide a new or custom capability or function to meet ITC^DeltaCom's business needs, referred to as a Business Opportunity Request (BOR). The BFR process is intended to facilitate the two way exchange of information between the requesting Party and BellSouth, necessary for accurate processing of requests in a consistent and timely fashion.
- 5.2 A Bona Fide Request/New Business Request shall be submitted in writing by ITC^DeltaCom and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a ITC^DeltaCom's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 or (ii) pursuant to the needs of the business. The request shall be sent to ITC^DeltaCom's Account Executive.
- 5.3 A product or service provided by BellSouth to another CLEC pursuant to the BFR/NBR process shall be made available to ITC^DeltaCom upon request, on the same terms and conditions.

6. <u>Liability and Indemnification</u>

- 6.1 <u>BellSouth Liability</u>. BellSouth shall take financial responsibility for its own actions in causing, or its lack of action in preventing, unbillable or uncollectible ITC^DeltaCom revenues.
- 6.2 <u>Liability for Acts or Omissions of Third Parties</u>. Neither BellSouth nor ITC^DeltaCom shall be liable for any act or omission of another telecommunications company providing a portion of the services provided under this Agreement.

6.3 <u>Limitation of Liability</u>.

- With respect to any claim or suit, whether based in contract, tort or any other theory of legal liability, by ITC^DeltaCom, any ITC^DeltaCom customer or by any other person or entity, for damages associated with any of the services provided by BellSouth pursuant to or in connection with this Agreement, including but not limited to the installation, provision, preemption, termination, maintenance, repair or restoration of service, and subject to the provisions of the remainder of this Part A, BellSouth's liability shall be limited to an amount equal to the proportionate charge for the service provided pursuant to this Agreement for the period during which the service was affected. With respect to any claim or suit, whether based in contract, tort or any other theory of legal liability, by BellSouth, any BellSouth customer or by any other person or entity, for damages associated with any of the services provided by ITC^DeltaCom pursuant to or in connection with this Agreement, including but not limited to the installation, provision, preemption, termination, maintenance, repair or restoration of service, and subject to the provisions of the remainder of this Part A, ITC^DeltaCom's liability shall be limited to an amount equal to the proportionate charge for the service provided pursuant to this Agreement for the period during which the service was affected. Notwithstanding the foregoing, claims for damages by ITC^DeltaCom, any ITC^DeltaCom customer or any other person or entity resulting from the gross negligence or willful misconduct of BellSouth and claims for damages by ITC^DeltaCom resulting from the failure of BellSouth to honor in one or more material respects any one or more of the material provisions of this Agreement shall not be subject to such limitation of liability. Likewise, claims for damages by BellSouth, any BellSouth customer or any other person or entity resulting from the gross negligence or willful misconduct of ITC^DeltaCom and claims for damages by BellSouth resulting from the failure of ITC^DeltaCom to honor in one or more material respects any one or more of the material provisions of this Agreement shall not be subject to such limitation of liability.
- 6.3.2 <u>Limitations in Tariffs</u>. Subject to the provisions of 6.3.1, a Party may, in its sole discretion, provide in its tariffs and contracts with its Customer and third parties that relate to any service, product or function provided or contemplated under this Agreement, that to the maximum extent permitted by Applicable Law, such Party shall not be liable to Customer or third Party for (i) any Loss relating to or arising out of this Agreement, whether in contract, tort or otherwise, that exceeds the amount such party would have charged that applicable person for the service, product or function that gave rise to such Loss and (ii) Consequential Damages. To the extent that a Party elects not to place in its tariffs or contracts such limitations of liability, and the other Party incurs a Loss as a result thereof, such Party shall indemnify and reimburse the other Party for that portion of the Loss that would have been limited had the first Party included in its tariffs and contracts the limitations of liability that such other Party included in its own tariffs at the time of such Loss.

- 6.3.3 Neither BellSouth nor ITC^DeltaCom shall be liable for damages to the other's terminal location, POI or other company's customers' premises resulting from the furnishing of a service, including, but not limited to, the installation and removal of equipment or associated wiring, except to the extent caused by a company's negligence or willful misconduct or by a company's failure to properly ground a local loop after disconnection.
- 6.3.4 Under no circumstance shall a Party be responsible or liable for indirect, incidental, or consequential damages, including, but not limited to, economic loss or lost business or profits, damages arising from the use or performance of equipment or software, or the loss of use of software or equipment, or accessories attached thereto, delay, error, or loss of data. In connection with this limitation of liability, each Party recognizes that the other Party may, from time to time, provide advice, make recommendations, or supply other analyses related to the Services, or facilities described in this Agreement, and, while each Party shall use diligent efforts in this regard, the Parties acknowledge and agree that this limitation of liability shall apply to provision of such advice, recommendations, and analyses.
- 6.4 Indemnification for Certain Claims. BellSouth and ITC^DeltaCom providing services, their affiliates and their parent company, shall be indemnified, defended and held harmless by each other against any claim, loss or damage arising from the receiving company's use of the services provided under this Agreement pertaining to (1) claims for libel, slander, invasion of privacy or copyright infringement arising from the content of the receiving company's own communications, or (2) any claim, loss or damage claimed by the other company's customer arising from one company's use or reliance on the other company's services, actions, duties, or obligations arising out of this Agreement; provided that in the event of a claim arising under this Section 6.4(2), to the extent any claim, loss or damage is caused by the gross negligence or willful misconduct of the providing party, the receiving Party shall have no obligation to indemnify, defend or hold harmless the providing Party hereunder, subject to the other terms of this Section 6.
- Ourse of Dealing, Or From Usages of Trade.
- 6.6 ITC^DeltaCom and BellSouth will work cooperatively to minimize fraud associated with third-number billed calls, calling card calls, or any other services related to this Agreement. The Parties fraud minimization procedures are to be

cost effective and implemented so as not to unduly burden or harm one Party as compared to the other.

7. <u>Court Ordered Requests for Call Detail Records and Other Subscriber</u> Information.

To the extent technically feasible, BellSouth maintains call detail records for ITC^DeltaCom end users for limited time periods and can respond to subpoenas and court ordered requests for information. BellSouth shall maintain such information for ITC^DeltaCom end users for the same length of time it maintains such information for its own end users.

- 7.1 ITC^DeltaCom agrees that BellSouth will respond to subpoenas and court ordered requests delivered directly to BellSouth for the purpose of providing call detail records when the targeted telephone numbers belong to ITC^DeltaCom end users. Billing for such requests will be generated by BellSouth and directed to the law enforcement agency initiating the request.
- 7.2 ITC^DeltaCom agrees that in cases where ITC^DeltaCom receives subpoenas or court requests for call detail records for targeted telephone numbers belonging to ITC^DeltaCom end users, ITC^DeltaCom will advise the law enforcement agency initiating the request to redirect the subpoena or court ordered request to BellSouth. Billing for call detail information will be generated by BellSouth and directed to the law enforcement agency initiating the request.
- 7.3 In cases where the timing of the response to the law enforcement agency prohibits ITC^DeltaCom from having the subpoena or court ordered request redirected to BellSouth by the law enforcement agency, ITC^DeltaCom will furnish the official request to BellSouth for providing the call detail information. BellSouth will provide the call detail records to ITC^DeltaCom and bill ITC^DeltaCom for the information. ITC^DeltaCom agrees to reimburse BellSouth for the call detail information provided.
- 7.4 ITC^DeltaCom will provide ITC^DeltaCom end user and/or other customer information that is available to ITC^DeltaCom in response to subpoenas and court orders for their own customer records. BellSouth will redirect subpoenas and court ordered requests ITC^DeltaCom end user and/or other customer information to ITC^DeltaCom for the purpose of providing this information to the law enforcement agency.

8. Intellectual Property Rights and Indemnification

8.1 <u>No License</u>. No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement. ITC^DeltaCom is strictly prohibited from any use, including but not limited to in sales, in marketing or advertising of telecommunications services, of any BellSouth name, service mark or trademark.

- 8.2 Ownership of Intellectual Property. Any intellectual property, which originates from or is developed by a Party shall remain in the exclusive ownership of that Party. Except for a limited license to use patents or copyrights to the extent necessary for the Parties to use any facilities or equipment (including software) or to receive any service solely as provided under this Agreement, no license in patent, copyright, trademark or trade secret, or other proprietary or intellectual property right now or hereafter owned, controlled or licensable by a Party, is granted to the other Party or shall be implied or arise by estoppel. It is the responsibility of each Party to ensure at no additional cost to the other Party that it has obtained any necessary licenses in relation to intellectual property of third Parties used in its network that may be required to enable the other Party to use any facilities or equipment (including software), to receive any service, or to perform its respective obligations under this Agreement.
- 8.3 Indemnification. The Party providing a service pursuant to this Agreement will defend the Party receiving such service or data provided as a result of such service against claims of infringement arising solely from the use by the receiving Party of such service and will indemnify the receiving Party for any damages awarded based solely on such claims in accordance with Section 6 of this Agreement.

Promptly after receipt of notice of any claim or the commencement of any action for which a Party may seek indemnification pursuant to this Section, such Party (the "Indemnified Party") shall promptly give written notice to the other Party (the "Indemnifying Party") of such claim or action, but the failure to so notify the Indemnifying Party shall not relieve the Indemnifying Party of any liability it may have to the Indemnified Party except to the extent the Indemnifying Party has actually been prejudiced thereby. The Indemnifying Party shall be obligated to assume the defense of such claim, at its own expense. The Indemnified Party shall cooperate with the Indemnifying Party's reasonable requests for assistance or Information relating to such claim, at the Indemnifying Party's expense. The Indemnified Party shall have the right to participate in the investigation and defense of such claim or action, with separate counsel chosen and paid for by the Indemnified Party.

- 8.4 <u>Claim of Infringement</u>. In the event that use of any facilities or equipment (including software), becomes, or in reasonable judgment of the Party who owns the affected network is likely to become, the subject of a claim, action, suit, or proceeding based on intellectual property infringement, then said Party shall promptly and at its sole expense, but subject to the limitations of liability set forth below:
 - 8.4.1 modify or replace the applicable facilities or equipment (including software) while maintaining form and function, or
 - 8.4.2 obtain a license sufficient to allow such use to continue.

- 8.4.3 In the event 8.4.1 or 8.4.2 are commercially unreasonable, then said Party may, terminate, upon reasonable notice under the circumstances, this contract with respect to use of, or services provided through use of, the affected facilities or equipment (including software), but solely to the extent required to avoid the infringement claim.
- 8.5 Exception to Obligations. Neither Party's obligations under this Section shall apply to the extent the infringement is caused by: (i) modification of the facilities or equipment (including software) by the indemnitee; (ii) use by the indemnitee of the facilities or equipment (including software) in combination with equipment or facilities (including software) not provided or authorized by the indemnitor provided the facilities or equipment (including software) would not be infringing if used alone; (iii) conformance to specifications of the indemnitee which would necessarily result in infringement; or (iv) continued use by the indemnitee of the affected facilities or equipment (including software) after being placed on notice to discontinue use as set forth herein.
- 8.6 <u>Exclusive Remedy</u>. The foregoing shall constitute the Parties' sole and exclusive remedies and obligations with respect to a third party claim of intellectual property infringement arising out of the conduct of business under this agreement.

9. <u>Treatment of Proprietary and Confidential Information</u>

9.1 Confidential Information. It may be necessary for BellSouth and ITC^DeltaCom to provide each other with certain confidential information, including trade secret information, including but not limited to, technical and business plans, technical information, proposals, specifications, drawings, procedures, customer account data, call detail records and like information (hereinafter collectively referred to as "Information"). All Information shall be in writing or other tangible form and clearly marked with a confidential, private or proprietary legend and that the Information will be returned to the owner within a reasonable time. The Information shall not be copied or reproduced in any form. BellSouth and ITC^DeltaCom shall receive such Information and not disclose such Information. BellSouth and ITC^DeltaCom shall protect the Information received from distribution, disclosure or dissemination to anyone except employees of BellSouth and ITC^DeltaCom with a need to know such Information and which employees agree to be bound by the terms of this Section. BellSouth and ITC^DeltaCom will use the same standard of care to protect Information received as they would use to protect their own confidential and proprietary Information. Where customer specific information or critical network information is communicated orally to designated company representatives in furtherance of this Agreement, both Parties agree that those employees shall protect such Information from disclosure to anyone except employees of BellSouth and ITC^DeltaCom with a need to know such Information.

9.2 Exception to Obligation. Notwithstanding the foregoing, there will be no obligation on BellSouth or ITC^DeltaCom to protect any portion of the Information that is: (1) made publicly available by the owner of the Information or lawfully disclosed by a Party other than BellSouth or ITC^DeltaCom; (2) lawfully obtained from any source other than the owner of the Information; or (3) previously known to the receiving Party without an obligation to keep it confidential.

10. Assignments

Any assignment by either Party to any non-affiliated entity of any right, obligation or duty, or of any other interest hereunder, in whole or in part, without the prior written consent of the other Party shall be void. A Party may assign this Agreement or any right, obligation, duty or other interest hereunder to an Affiliate company of the Party without the consent of the other Party. All obligations and duties of any Party under this Agreement shall be binding on all successors in interest and assigns of such Party. No assignment of delegation hereof shall relieve the assignor of its obligations under this Agreement in the event that the assignee fails to perform such obligations.

11. Resolution of Disputes

Except as otherwise stated in this Agreement, the Parties agree that if any dispute arises as to the interpretation of any provision of this Agreement or as to the proper implementation of this Agreement, either Party may petition the Commission for a resolution of the dispute; provided, however, that to the extent any issue disputed hereunder involves issues beyond the scope of authority or jurisdiction of the Commission, the parties may seek initial resolution of such dispute in another appropriate forum. However, each Party reserves any rights it may have to seek judicial review of any ruling made by the Commission concerning this Agreement. Each Party shall bear its own costs when seeking Commission or judicial review of any ruling concerning this Agreement.

12. <u>Limitation of Use</u>

The Parties agree that this Agreement shall not be offered by either Party in another jurisdiction as evidence of any concession or as a waiver of any position taken by the other Party in that jurisdiction or for any other purpose.

13. Taxes

13.1 Any Federal, state or local excise, license, sales, use or other taxes or tax-like charges (excluding any taxes levied on income) resulting from the performance of this Agreement shall be borne by the Party upon which the obligation for payment is imposed under applicable law, even if the obligation to collect and remit such taxes is placed upon the other party. Any such taxes shall be shown as separate items on applicable billing documents between the Parties. The Party obligated to collect and remit taxes shall do so unless the other Party

provides such Party with the required evidence of exemption. The Party to obligated to pay any such taxes may contest the same and shall be entitled to the benefit of any refund or recovery. The Party obligated to collect and remit taxes shall cooperate fully in any such contest by the other Party by providing, records, testimony, and such additional information or assistance as may reasonably be necessary to pursue the contest

14. Force Majeure

In the event performance of this Agreement, or any obligation hereunder, is either directly or indirectly prevented, restricted, or interfered with by reason of fire, flood, earthquake or like acts of God, wars, revolution, civil commotion, explosion, acts of public enemy, embargo, acts of the government in its sovereign capacity, labor difficulties, including without limitation, strikes, slowdowns, picketing, or boycotts, unavailability of equipment from vendor, changes requested by Customer, or any other circumstances beyond the reasonable control and without the fault or negligence of the Party affected, the Party affected, upon giving prompt notice to the other Party, shall be excused from such performance on a day-to-day basis to the extent of such prevention, restriction, or interference (and the other Party shall likewise be excused from performance of its obligations on a day-to-day basis until the delay, restriction or interference has ceased); provided however, that the Party so affected shall use best efforts to avoid or remove such causes of non-performance and both Parties shall proceed whenever such causes are removed or cease.

15. Year 2000 Compliance

Each party warrants that it has implemented a program the goal of which is to ensure that all software, hardware and related materials (collectively called "Systems") delivered, connected with or supplied in the furtherance of the terms and conditions specified in this Agreement (I) will record, store, process and display calendar dates falling on or after January 1, 2000, in the same manner, and with the same functionality as such software records, stores, processes and calendars dates falling on or before December 31, 1999; and (ii) shall include without limitation date data century recognition, calculations that accommodate same century and multicentury formulas and date values, and date data interface values that reflect the century.

16. Modification of Agreement

16.1 BellSouth shall make available to ITC^DeltaCom, pursuant to 47 USC § 252, the FCC rules and regulations and the Supreme Court Order in AT&T Corporation v. Iowa Utilities Board regarding such availability, any interconnection, service, or network element provided under any other agreement filed and approved pursuant to 47 USC § 252. The adopted interconnection, service, or network element and agreement shall apply to the same state as such other agreement and for the identical term of such other agreement.

- 16.2 No modification, amendment, supplement to, or waiver of the Agreement or any of its provisions shall be effective and binding upon the Parties unless it is made in writing and duly signed by the Parties.
- 16.3 Execution of this Agreement by either Party does not confirm or infer that the executing Party agrees with any decision(s) issued pursuant to the Telecommunications Act of 1996 and the consequences of those decisions on specific language in this Agreement. Neither Party waives its rights to appeal or otherwise challenge any such decision(s) and each Party reserves all of its rights to pursue any and all legal and/or equitable remedies, including appeals of any such decision(s).
- In the event that any effective legislative, regulatory, judicial or other legal action materially affects any material terms of this Agreement, or the ability of ITC^DeltaCom or BellSouth to perform any material terms of this Agreement, ITC^DeltaCom or BellSouth may, on thirty (30) days' written notice require that such terms be renegotiated, and the Parties shall renegotiate in good faith such mutually acceptable new terms as may be required. In the event that such new terms are not renegotiated within ninety (90) days after such notice, the Dispute shall be referred to the Dispute Resolution procedure set forth in Section 11.
- 16.5 If any provision of this Agreement, or the application of such provision to either Party or circumstance, shall be held invalid, the remainder of the Agreement, or the application of any such provision to the Parties or circumstances other than those to which it is held invalid, shall not be effective thereby, provided that the Parties shall attempt to reformulate such invalid provision to give effect to such portions thereof as may be valid without defeating the intent of such provision.
- 16.6 If ITC^DeltaCom changes its name or makes changes to its structure or identity due to a merger, acquisition, transfer or any other reason, it is the responsibility of ITC^DeltaCom to notify BellSouth of said change and request that an amendment to this Agreement, if necessary, be executed to reflect said change.

17. Waivers

A failure or delay of either Party to enforce any of the provisions hereof, to exercise any option which is herein provided, or to require performance of any of the provisions hereof shall in no way be construed to be a waiver of such provisions or options, and each Party, notwithstanding such failure, shall have the right thereafter to insist upon the specific performance of any and all of the provisions of this Agreement.

18. Governing Law

This Agreement shall be governed by, and construed and enforced in accordance with, the laws of the State of Georgia, without regard to its conflict of laws principles.

19. Arm's Length Negotiations

This Agreement was executed after arm's length negotiations between the undersigned Parties and reflects the conclusion of the undersigned that this Agreement is in the best interests of all Parties.

20. Notices

20.1 Every notice, consent, approval, or other communications required or contemplated by this Agreement shall be in writing and shall be delivered in person or given by postage prepaid mail, address to:

BellSouth Telecommunications, Inc.

CLEC Account Team 9th Floor 600 North 19th Street Birmingham, Alabama 35203

and

General Attorney - COU Suite 4300 675 W. Peachtree St. Atlanta, GA 30375

ITC^DeltaCom Communications, Inc

Senior Manager – Industry Relations 1530 DeltaCom Drive PO Box 787 Anniston, AL 36202

and

Director – Regulatory Affairs 4092 S. Memorial Parkway Huntsville, AL 35802

or at such other address as the intended recipient previously shall have designated by written notice to the other Party.

- 20.2 Where specifically required, notices shall be by certified or registered mail. Unless otherwise provided in this Agreement, notice by mail shall be effective on the date it is officially recorded as delivered by return receipt or equivalent, and in the absence of such record of delivery, it shall be presumed to have been delivered the fifth day, or next business day after the fifth day, after it was deposited in the mails.
- 20.3 BellSouth shall provide ITC^DeltaCom 45-day advance notice via Internet posting of price changes and of changes to the terms and conditions of services available for resale. To the extent that revisions occur between the time BellSouth notifies ITC^DeltaCom of changes under this Agreement and the time the changes are scheduled to be implemented, BellSouth will immediately notify ITC^DeltaCom of such revisions consistent with its internal notification process. ITC^DeltaCom may not hold BellSouth responsible for any cost incurred as a result of such revisions, unless such costs are incurred as a result of BellSouth's intentional misconduct. ITC^DeltaCom may not utilize any notice given under this subsection concerning a service to market resold offerings of that service in advance of BellSouth.

21. <u>Discontinuance of Service</u>

Each Party reserves the right to suspend or terminate service in the event of prohibited, unlawful or improper use of facilities or service pursuant to regulatory or legal authorities, for which it is purchasing services or in the event of nonpayment for services in accordance with Attachment 7 of this Agreement.

22. Rule of Construction

No rule of construction requiring interpretation against the drafting Party hereof shall apply in the interpretation of this Agreement.

23. <u>Headings of No Force or Effect</u>

The headings of Articles and Sections of this Agreement are for convenience of reference only, and shall in no way define, modify or restrict the meaning or interpretation of the terms or provisions of this Agreement.

24. Multiple Counterparts

This Agreement may be executed multiple counterparts, each of which shall be deemed an original, but all of which shall together constitute but one and the same document.

25. Entire Agreement

Page 16

This Agreement and its Attachments, incorporated herein by this reference, sets forth the entire understanding and supersedes prior agreements between the Parties relating to the subject matter contained herein and merges all prior discussions between them, and neither Party shall be bound by any definition, condition, provision, representation, warranty, covenant or promise other than as expressly stated in this Agreement or as is contemporaneously or subsequently set forth in writing and executed by a duly authorized officer or representative of the Party to be bound thereby.

26. Electronic Filing

26.1 For electronic filing purposes in the State of Louisiana, the CLEC Louisiana Certification Number is required and must be provided by ITC^DeltaCom prior to filing the Agreement. The CLEC Louisiana Certification Number for ITC^DeltaCom is TSP00128.

This agreement includes attachments with provisions for the following services:

Unbundled Network Elements (UNEs) Local Interconnection Resale Collocation

IN WITNESS WHEREOF, the Parties have executed this Agreement the day and year above first written.

BellSouth Telecommunications, Inc.	ITC^DeltaCom Communications, Inc.	
Signature	Signature	
Name	Name	
Title	Title	
 Date	 Date	

Definitions

"Access Service Request" or "ASR" means an industry standard form used by the Parties to add, establish, change or disconnect trunks for the purposes of interconnection.

"Act" means the Communications Act of 1934, 47 U.S.C. 151 et seq., as amended, including the Telecommunications Act of 1996, and as interpreted from time to time in the duly authorized rules and regulations of the FCC or the Commission/Board.

"Advanced Intelligent Network" or "AIN" is a Telecommunications network architecture in which call processing, call routing and network management are provided by means of centralized databases.

"Affiliate" is an entity directly owns or controls, or is under common ownership or control with, another entity. For purposes of this paragraph, the term "own" or "control" means to own an equity interest (or equivalent thereof) of more than 10 percent.

"American National Standards Institute" or "ANSI" is a standards setting, non-government organization, which develops and publishes standards for "voluntary" use in the United States.

"Automatic Number Identification" or "ANI" is a telephone number associated with the access line from which the a call originates.

"Calling Party Number" or "CPN" is a Common Channel Signaling parameter which refers to the number transmitted through the network identifying the calling party.

"Carrier Identification Code" or "CIC" means a three or four digit number assigned to an IXC that identifies that carrier's traffic.

"Centralized Message Distribution System" is a national system, that Local Exchange Carriers use to exchange Exchange Message Interface (EMI) formatted data among host companies.

"Commission" is defined as the appropriate regulatory agency in each of BellSouth's nine-state region, Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina, and Tennessee.

"Common Channel Signaling" or "CCS" means a method of exchanging call set-up and network control data over a digital signaling network fully separate from the Public Switched Network that carriers the actual call.

"Customer Local Area Signaling Services" or "CLASS" is a set of call management service features consisting os number translation such as call forwarding and caller identification, available within a the Local Access and Transport Area ("LATA").

"Digital Service - Level 0" or "DS-0" means the 64 Kbps zero-level signal in the time division multiplex hierarchy.

"Digital Service - Level 1" or "DS-1" means the 1.544 Mbps in the time division multiplex hierarchy. In the time-division multiplexing hierarchy of the telephone network, DS1 is the initial level of multiplexing.

"Digital Service - Level 3" or "DS-3" means the 44.736 Mbps in the time division multiplex hierarchy. In the time-division multiplexing hierarchy of the telephone network, DS3 is the initial level of multiplexing.

"Exchange Message Interface" is the nationally administered standard format for the exchange of data among the Exchange Carriers within the telecommunications industry.

"Exchange Access" means the offering by a LEC of services or facilities to an IXC for the purpose of the origination or termination of telephone toll services.

"Feature Group A" or "FGA" means FGA interexchange access as defined in BellSouth's FCC Tariff No. 1.

"Feature Group B" or "FGB" means FGB interexchange access as defined in BellSouth's FCC Tariff No. 1.

"Feature Group D" or "FGD" means FGD interexchange access as defined in BellSouth's FCC Tariff No. 1.

"Interconnection" is the linking of the BellSouth and ITC^DeltaCom networks for the mutual exchange of traffic as described in Attachment 3 of this Agreement.

"Point of Interconnection" or "POI" is as described in Attachment 3 of this Agreement.

"Interexchange Carrier" or "IXC" means a provider of interexchange telecommunications services.

"Local Exchange Carrier" or "LEC" is as defined in the Act.

"Local Exchange Routing Guide" or "LERG" means a Telcordia product that is sold to and used by LECs and IXCs to identify NPA-NXX routing and homing information as well as Network Element and equipment designations.

"Local Interconnection" is defined as 1) the delivery of local traffic to be terminated on each Party's local network so that end users of either Party have the ability to reach end users of the other Party without the use of any access code or substantial delay in the processing of the call; 2) the LEC unbundled network features, functions, and capabilities set forth in this Agreement; and 3) Service Provider Number Portability sometimes referred to as temporary telephone number portability to be implemented pursuant to the terms of this Agreement.

"Local Traffic" is as defined in Attachment 3 of this Agreement.

"Local Access and Transport Area" or "LATA" means one of the contiguous geographic areas established pursuant to the AT&T Consent Decree to define the permitted operating regions of the RBOCs prior to the enactment of the Telecommunications Act of 1996.

"Multiple Exchange Carrier Access Billing" or ("MECAB") means the document prepared by the Billing Committee of the Ordering and Billing Forum ("OBF"), which functions under the auspices of the Carrier Liaison Committee of the Alliance for Telecommunications Industry Solutions ("ATIS") and by Bellcore as Special Report SR-BDS-000983, Containing the recommended guidelines for the billing of Exchange Service access provided by two or more LECs and/or CLECs or by one LEC in two or more states within a single LATA.

"Network Element" defined to mean a facility or equipment used in the provision of a telecommunications service. Such term may include, but is not limited to, features, functions, and capabilities that are provided by means of such facility or equipment, including but not limited to, subscriber numbers, databases, signaling systems, and information sufficient for billing and collection or used in the transmission, routing, or other provision of a telecommunications service.

"Numbering Plan Area" or "NPA" is also sometimes referred to as an area code. This is the three digit indicator which is defined by the "A", "B", and "C" digits of each "digit" telephone number within the North American Numbering Plan ("NANP"). Each NPA contains 800 Possible NXX Codes. At present, there are two general categories of NPA, "Geographic NPAs" and "Non-Geographic

NPAS". A "Geographic NPA" is associated with a defined geographic area, and all telephone numbers bearing such NPA are associated with services provided within that Geographic area. A "Non-Geographic NPA", also known as a "Service Access Code" (SAC Code) is typically associated with a specialized telecommunications service which may be provided across multiple geographic NPA areas; 500, 800, 9100, 700, and 888 are examples of Non-Geographic NPAS.

"NXX", "NXX Code", "Central Office Code" or "CO Code" is the three digit switch entity indicator which is defined by the "D", "E", and "F" digits of a 10-digit telephone number within the North American Numbering Plan.

"Percent Of Interstate Usage" of "PIU" is defined as a factor to be applied to terminating access services minutes of use to obtain those minutes that should be rated as interstate access services minutes of use. The numerator includes all interstate "non-intermediary" minutes of use, including interstate minutes of use that are forwarded due to service provider number portability less any interstate minutes of use for Terminating Party Pays services, such as 800 Services. The denominator includes all "non-intermediary", local, interstate, intrastate, toll and access minutes of use adjusted for service provider number portability less all minutes attributable to terminating Party pays services

"Percent Local Usage" or "PLU" is defined as a factor to be applied to intrastate terminating minutes of use. The numerator shall include all "non-intermediary" local minutes of use adjusted for those minutes of use that only apply local due to Service Provider Number Portability. The denominator is the total intrastate minutes of use including local, intrastate toll, and access, adjusted for Service Provider Number Portability less intrastate terminating Party pays minutes of use.

"Rate Center" identifies the specific geographic point within an Exhange area which is associated with one or more particular NPA-NXX codes which have been assigned to a LEC (or CLEC) for the provision of Telephone Exchange Services. The Rate Center vertical and horizontal coordinates are used in the toll message rating process to measure distances between Rate Centers.

"Revenue Accounting Office ("RAO") Status Company" is a local exchange company/alternate local exchange company that has been assigned a unique RAO code. Message data exchanged among RAO status companies is grouped (i.e. packed) according to From/To/Bill RAO combinations.

"Transit Traffic Service" is as described in Attachment 3 of this Agreement.

"Wire Center" denotes a building or space within a building which serves as an aggregation point on a given carrier's network, where transmission facilities and

circuits are connected and/or switched. Wire Center can also denote a building in which one or more central offices, used for the provision of telecommunications services are located.

RESALE

1.0 Discount Rates

ITC^DeltaCom may purchase all retail Telecommunications Services provided by BellSouth. The price or wholesale discount for these Telecommunications Services shall be the retail rates reduced by the wholesale discount rate established by the appropriate state public utility commission. The wholesale discount shall be as set forth in Exhibit A, attached hereto and incorporated herein by this reference.

2.0 Definition of Terms

- 2.1 CUSTOMER OF RECORD means the entity responsible for placing application for service; requesting additions, rearrangements, maintenance or discontinuance of service; payment in full of charges incurred such as non-recurring, monthly recurring, toll, directory assistance, etc.
- 2.2 DEPOSIT means assurance provided by a customer in the form of cash, surety bond or bank letter of credit to be held by BellSouth.
- 2.3 END USER means the ultimate user of the telecommunications services.
- 2.4 END USER CUSTOMER LOCATION means the physical location of the premises where an end user makes use of the telecommunications services.
- 2.5 NEW SERVICES means functions, features or capabilities that are not currently offered by BellSouth. This includes packaging of existing services or combining a new function, feature or capability with an existing service.
- 2.6 COMPETITIVE LOCAL EXCHANGE COMPANY (CLEC) means a telephone company certificated by the public service commissions of BellSouth's franchised area to provide local exchange service within BellSouth's franchised area.
- 2.7 RESALE means an activity wherein a certificated CLEC, such as ITC^DeltaCom subscribes to the telecommunications services of BellSouth and then reoffers those telecommunications services to the public (with or without "adding value").
- 2.8 RESALE SERVICE AREA means the area, as defined in a public service commission approved certificate of operation, within which an CLEC, such as ITC^DeltaCom, may offer resold local exchange telecommunications service.

3.0 General Provisions

- 3.1 ITC^DeltaCom may resell the tariffed local exchange and toll telecommunications services of BellSouth contained in the General Subscriber Service Tariff and Private Line Service Tariff subject to the terms, and conditions specifically set forth herein. Notwithstanding the foregoing, the exclusions and limitations on services available for resale will be as set forth in Exhibit B, attached hereto and incorporated herein by this reference.
- 3.2 BellSouth shall make available telecommunications services for resale at the rates set forth in Exhibit A to this agreement and subject to the exclusions and limitations set forth in Exhibit B to this agreement.
- 4 States 10-26-01

Neither Party waives its right to appeal or otherwise challenge any decision regarding resale that resulted in the discount rates contained in Exhibit A or the exclusions and limitations contained in Exhibit B. Both Parties reserve the right to pursue any and all legal and/or equitable remedies, including appeals of any decisions. If such appeals or challenges result in changes in the discount rates or exclusions and limitations, the parties agree that appropriate modifications to this Agreement will be made promptly to make its terms consistent with the outcome of the appeal.

- 3.3 ITC^DeltaCom may purchase resale services from BellSouth for its own use in operating its business. The resale discount will apply to those services under the following conditions:
 - 3.3.1 ITC^DeltaCom must resell services to other end users.
 - 3.3.2 ITC^DeltaCom must order services through the LCSC and/or appropriate Resale Account Teams.
 - 3.3.3 ITC^DeltaCom cannot be a competitive local exchange telecommunications company for the single purpose of selling to themselves.
- 3.4 The provision of services by BellSouth to ITC^DeltaCom does not constitute a joint undertaking for the furnishing of any service.
- 3.5 ITC^DeltaCom will be the customer of record for all services purchased from BellSouth. Except as specified herein, BellSouth will take orders from, bill and expect payment from ITC^DeltaCom for all services.
- 3.6 ITC^DeltaCom will be BellSouth's single point of contact for all services purchased pursuant to this Agreement. BellSouth shall have no contact with the end user except to the extent provided for herein.
- 3.7 BellSouth will continue to bill the end user for any services that the end user specifies it wishes to receive directly from BellSouth. The Parties shall not restrict the customer's choice of using other telecommunications carriers and their services.
- 3.8 BellSouth maintains the right to serve directly any end user within the service area of ITC^DeltaCom. BellSouth will continue to directly market its own telecommunications products and services and in doing so may establish independent relationships with end users of ITC^DeltaCom.
- 3.9 Neither Party shall interfere with the right of any person or entity to obtain service directly from the other Party. Neither Party, its employees, nor its subcontractors shall make disparaging comments regarding the other Party or its services to end-users.
- 3.10 For the purpose of the resale of BellSouth's telecommunications services by ITC^DeltaCom, BellSouth will provide ITC^DeltaCom with an on line access to telephone numbers for reservation on a first come first serve basis. Such reservations of telephone numbers, on a pre-ordering basis shall be for a period of nine (9) days. ITC^DeltaCom acknowledges that there may be instances where there is a shortage of telephone numbers in a particular Common Language Location Identifier Code (CLLIC) and in such instances BellSouth may request that ITC^DeltaCom cancel its reservations of numbers. ITC^DeltaCom shall comply with such request.
- 3.11 Further, upon ITC^DeltaCom's request, and for the purpose of the resale of BellSouth's telecommunications services by ITC^DeltaCom, BellSouth will reserve up to 100 telephone numbers per CLLIC, for ITC^DeltaCom's sole use. Such telephone number reservations shall be valid for ninety (90)
- 4 States 10-26-01

days from the reservation date. ITC^DeltaCom acknowledges that there may be instances where there is a shortage of telephone numbers in a particular CLLIC and in such instances BellSouth shall use its best efforts to reserve for a ninety (90) day period a sufficient quantity of ITC^DeltaCom's reasonable need in that particular CLLIC.

- 3.12 ITC^DeltaCom may resell BellSouth services only within the specific resale service area as defined in its certificate(s) of authority as a local telecommunications carrier.
- 3.13 911- BellSouth shall provide to ITC^DeltaCom 911 emergency call routing services at parity with BellSouth.
- 3.14 Customer Service Functions-Except as otherwise provided in this Agreement, ITC^DeltaCom shall be the single point of contact for all ITC^DeltaCom end users.
- 3.15 BellSouth shall refer all questions regarding ITC^DeltaCom service or product directly to ITC^DeltaCom. BellSouth shall use its best efforts to ensure that all BellSouth representatives who receive inquiries regarding ITC^DeltaCom services do not in any way disparage or discriminate against ITC^DeltaCom or its products or services.
- 3.16 The same quality standards that BellSouth requires of its employees when contacting BellSouth end users (e.g. honesty, respect, and courtesy) shall apply when its employees are in contact with ITC^DeltaCom end users.

4.0 Restrictions on Provision of Service

- 4.1 Service is furnished subject to the condition that it will not be used for any unlawful purpose.
- 4.2 Service will be discontinued if any law enforcement agency advises that the service being used is in violation of the law.
- 4.3 BellSouth can refuse service when it has grounds to believe that service will be used in violation of the law.
- 4.4 BellSouth accepts no responsibility to any person for any unlawful act committed by ITC^DeltaCom or its end users as part of providing service to ITC^DeltaCom for purposes of resale or otherwise.
- 4.5 The characteristics and methods of operation of any circuits, facilities or equipment provided by any person or entity other than BellSouth shall not:
 - 4.5.1 Interfere with or impair service over any facilities of BellSouth, its affiliates, or its connecting and concurring carriers involved in its service;
 - 4.5.2 Cause damage to BellSouth's plant;
 - 4.5.3 Impair the privacy of any communications; or
 - 4.5.4 Create hazards to any employees or the public.
- 4.6 Current telephone numbers may normally be retained by the end user. ITC^DeltaCom has no property right to the telephone number or any other call number designation associated with services furnished by
- 4 States 10-26-01

BellSouth, and no right to the continuance of service through any particular central office. BellSouth reserves the right to change such numbers, or the central office designation associated with such numbers, or both, whenever BellSouth deems it necessary to do so in the conduct of its business any such changes will be implemented in a nondiscriminatory manner.

4.7 No patent, copyright, trademark or other proprietary right is licensed, granted or otherwise transferred by this Agreement. ITC^DeltaCom is strictly prohibited from any use, including but not limited to sales, marketing or advertising, of any BellSouth name or trademark.

5.0 BellSouth's Provision of Services to ITC^DeltaCom

- 5.1 ITC^DeltaCom agrees that its resale of BellSouth services shall be as follows:
- 5.1.1 The resale of telecommunications services shall be limited to users and uses conforming to the class of service restrictions.
- 5.1.2 Hotel and Hospital PBX service are the only telecommunications services available for resale to Hotel/Motel and Hospital end users, respectively. Similarly, Access Line Service for Customer Provided Coin Telephones is the only local service available for resale to Independent Payphone Provider (IPP) customers. Shared Tenant Service customers can only be sold those telecommunications services available in BellSouth Shared Tenant Service Tariff
- 5.1.3 ITC^DeltaCom is prohibited from furnishing both flat and measured rate service on the same business premises to the same subscribers (end users) as stated in A2 of BellSouth's Tariff except for backup service as indicated in the applicable state tariff Section A3.
- 5.2 BellSouth reserves the right to periodically audit services purchased by ITC^DeltaCom to establish authenticity of use. Such audit shall not occur more than once in a calendar year. ITC^DeltaCom shall make any and all records and data available to BellSouth or BellSouth's auditors on a reasonable basis. BellSouth shall bear the cost of said audit.
- 5.3 Resold services can only be used in the same manner as specified in BellSouth's Tariff. Resold services are subject to the same terms and conditions as are specified for such services when furnished to an individual end user of BellSouth in the appropriate section of BellSouth's Tariffs. Specific tariff features, e.g. a usage allowance per month, shall not be aggregated across multiple resold services. Resold services cannot be used to aggregate traffic from more than one end user customer except as specified in BellSouth's Tariff referring to Shared Tenant Service
- 5.4 BellSouth may provide any service or facility for which a charge is not established herein, as long as it is offered on the same terms to ITC^DeltaCom.
- 5.5 White page directory listings will be provided in accordance with Section 4 of the General Terms and Conditions and with the regulations set forth in Section A6 of the General Subscriber Service Tariff.
- 5.6 Where available to BellSouth's end users, BellSouth shall provide the following telecommunications services at a discount to allow for voice mail services:
 - Simplified Message Desk Interface Enhanced ("SMDI-E")

- Simplified Message Desk Interface ("SMDI") Message Waiting Indicator ("MWI") stutter dialtone and message waiting light feature capabilities.
- Call Forward on Busy/Don't Answer ("CF-B/DA")
- Call Forward on Busy ("CF/B")
- Call Forward Don't Answer ("CF/DA")

Further, BellSouth messaging services set forth in BellSouth's Messaging Service Information Package shall be made available for resale without the wholesale discount.

- 5.7 BellSouth's Inside Wire Maintenance Service Plan may be made available for resale at rates, terms and conditions as set forth by BellSouth and without the wholesale discount.
- 5.8 BellSouth will provide customer record information to ITC^DeltaCom provided ITC^DeltaCom has either executed a blanket agency agreement or has the appropriate Letter(s) of Authorization. BellSouth shall provide customer record information via an electronic interface and in accordance with the provisions of Attachment 6.
- 5.9 Telephone numbers transmitted via any resold service feature are intended solely for the use of the end user of the feature. Resale of this information is prohibited.

6.0 Operations Support Systems Functions

- BellSouth shall provide ITC^DeltaCom advance notice of changes to the prices, terms, and conditions for Resale in accordance with the provisions of Section 20.3 of the General Terms and Conditions. BellSouth provides electronic access to customer record information. Access is provided through the Local Exchange Navigation System (LENS), and the Telecommunications Access Gateway (TAG). Customer Record Information includes but is not limited to, customer specific information in CRIS and RSAG. ITC^DeltaCom agrees not to view, copy or otherwise obtain access to the customer record information of any customer without that customer's permission and only in accordance with applicable federal and state regulations.
- As provided in Section 3 of the General Terms and Conditions and Attachment 6, BellSouth shall provide ITC^DeltaCom, at its request, non-discriminatory access to BellSouth's OSS functions for pre-ordering, ordering, provisioning, maintenance and repair, and billing. Such OSS functions shall be equal in quality and provisioned with the same timeliness as provided by BellSouth to itself or to any Subsidiary, Affiliate or any other Telecommunications Carrier to which BellSouth provides the OSS functions.
- 6.3 Charges for use of OSS shall be as set forth in Exhibit A of this Attachment and in Attachment 11 of this Agreement.

7.0 Maintenance of Services

- 7.1 ITC^DeltaCom will adopt and adhere to the standards contained in the applicable BellSouth Work Center Interface Agreement regarding maintenance and installation of service.
- 7.2 Services resold under BellSouth's Tariffs and facilities and equipment provided by BellSouth shall be maintained by BellSouth
- 4 States 10-26-01

- 7.3 ITC^DeltaCom or its end users may not rearrange, move, disconnect, remove or attempt to repair any facilities owned by BellSouth, other than by connection or disconnection to any interface means used, except with the written consent of BellSouth.
- 7.4 ITC^DeltaCom accepts responsibility to notify BellSouth of situations that arise that may result in a service problem.
- 7.5 ITC^DeltaCom will be BellSouth's single point of contact for all repair calls on behalf of ITC^DeltaCom's end users. The parties agree to promptly provide one another with toll-free contact numbers for such purposes.
- 7.6 ITC^DeltaCom will contact the appropriate repair centers in accordance with reasonable procedures established by BellSouth
- 7.7 For all repair requests, ITC^DeltaCom accepts responsibility for adhering to BellSouth's reasonable prescreening guidelines prior to referring the trouble to BellSouth.
- 7.8 BellSouth will bill ITC^DeltaCom for handling troubles that are found not to be in BellSouth's network pursuant to its standard time and material charges. The standard time and material charges will be no more than what BellSouth charges to its retail customers for the same services.
- 7.9 BellSouth reserves the right to contact ITC^DeltaCom's customers, if deemed necessary, for maintenance purposes.
- 7.10 Facilities and/or equipment utilized by BellSouth to provide service to ITC^DeltaCom remain the property of BellSouth.

8.0 Establishment of Service

- 8.1 If ITC^DeltaCom has not already done so, after receiving certification as a local exchange company from the appropriate regulatory agency, ITC^DeltaCom will provide the appropriate Company service center the necessary documentation to enable BellSouth to establish a master account for ITC^DeltaCom. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable. BellSouth.
- 8.2 Service orders will be in a standard format designated by BellSouth.
- 8.3 BellSouth will not require end user confirmation prior to establishing service for ITC^DeltaCom's end user customer. ITC^DeltaCom must, however, be able to demonstrate end user authorization upon request.
- 8.4 ITC^DeltaCom will be the single point of contact with BellSouth for all subsequent ordering activity resulting in additions or changes to resold services except that BellSouth will accept a request directly from the end user for conversion of the end user's service from ITC^DeltaCom to BellSouth or will accept a request from another CLEC for conversion of the end user's service from ITC^DeltaCom to the other LEC. BellSouth will promptly notify ITC^DeltaCom that such a request has been processed.
- 8.5 If BellSouth determines that an unauthorized change in local service to ITC^DeltaCom has occurred, BellSouth will reestablish service with the appropriate local service provider. If BellSouth determines that
- 4 States 10-26-01

ITC^DeltaCom has initiated the unauthorized change, the unauthorized change charge described in F.C.C. Tariff No. 1, Section 13 will be assessed. Appropriate nonrecurring charges, as set forth in Section A4. of the General Subscriber Service Tariff, will also be assessed to ITC^DeltaCom. These charges shall be credited if ITC^DeltaCom provides proof of authorization or if it is determined that BellSouth or another LEC other than ITC^DeltaCom is the source of the error.

- 8.6 BellSouth shall take orders for resale from ITC^DeltaCom provided the deposit requirements of Section 1.11 of Attachment 7 to this Agreement are met.
- 8.7 The Parties will adopt and adhere to the BellSouth guidelines associated with each method of providing customer record information.

9.0 Standards of Performance

9.1 BellSouth shall provide Resale Services to ITC^DeltaCom (i) in accordance with Attachment 10 hereto and (ii) as required by the FCC or the applicable State Commission.

10. Resale of Customer Specific Arrangements

10.1 BellSouth shall make available CSAs for resale as provided in Exhibits A and B to this Attachment. In cases where ITC^DeltaCom resells an existing CSA, BellSouth will not impose any termination charges on the end user or on ITC^DeltaCom provided that ITC^DeltaCom agrees to execute a mutually acceptable assumption letter and thereafter abides by the terms of the CSA.

11. Payment And Billing Arrangements

- 11.1 If ITC^DeltaCom has not already done so, prior to submitting orders to BellSouth for local service, a master account must be established for ITC^DeltaCom. ITC^DeltaCom is required to provide the following before a master account is established: proof of PSC/PUC certification, the Application for Master Account, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable.
- 11.2 BellSouth shall bill ITC^DeltaCom on a current basis all applicable charges and credits.
- Payment of all charges will be the responsibility of ITC^DeltaCom. ITC^DeltaCom shall make payment to BellSouth for all services billed. BellSouth is not responsible for payments not received by ITC^DeltaCom from ITC^DeltaCom's customer. BellSouth will not become involved in billing disputes that may arise between ITC^DeltaCom and its customer. Payments made to BellSouth as payment on account will be credited to an accounts receivable master account and not to an end user's account.
- BellSouth will render bills each month on established bill days for each of ITC^DeltaCom's accounts.
- BellSouth will bill ITC^DeltaCom, in advance, charges for all services to be provided during the ensuing billing period except charges associated with service usage, which charges will be billed in arrears. Charges will be calculated on an individual end user account level, including, if applicable, any charges for usage or usage allowances. BellSouth will also bill all charges, including but not limited to 911 and E911 charges, and telecommunications relay charges and other taxes and fees in accordance with applicable laws, orders and regulations.

- 11.6 The payment will be due by the next bill date (i.e., same date in the following month as the bill date) and is payable in immediately available funds. Payment is considered to have been made when received by BellSouth.
- 11.7 If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment charge, as set forth in BellSouth's tariffs, shall apply.
- 11.8 If ITC^DeltaCom requests multiple billing media or additional copies of bills, BellSouth will provide these at an appropriate charge to ITC^DeltaCom.

12. Billing Disputes

- 12.1 Each Party agrees to notify the other Party upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the Bill Date on which such disputed charges appear. Resolution of the dispute is expected to occur at the first level of management resulting in a recommendation for settlement of the dispute and closure of a specific billing period. If the issues are not resolved within the allotted time frame, the following resolution procedure will begin:
 - 12.1.1 If the dispute is not resolved within sixty (60) days of the Bill Date, the dispute will be escalated to the second level of management for each of the respective Parties for resolution. If the dispute is not resolved within ninety (90) days of the Bill Date, the dispute will be escalated to the third level of management for each of the respective Parties for resolution.
 - 12.1.2 If the dispute is not resolved within one hundred and twenty (120) days of the Bill Date, the dispute will be escalated to the fourth level of management for each of the respective Parties for resolution.
 - 12.1.3 If a Party disputes a charge and does not pay such charge by the payment due date, such charges shall be subject to late payment charges as set forth in the Late Payment Charges provision of this Attachment. If a Party disputes charges and the dispute is resolved in favor of such Party, the other Party shall credit the bill of the disputing Party for the amount of the disputed charges along with any late payment charges assessed no later than the second Bill Date after the resolution of the dispute. Accordingly, if a Party disputes charges and the dispute is resolved in favor of the other Party, the disputing Party shall pay the other Party the amount of the disputed charges and any associated late payment charges assessed no later than the second bill payment due date after the resolution of the dispute. In no event, however, shall any late payment charges be assessed on any previously assessed late payment charges.
- 12.2 Upon proof of tax-exempt certification from ITC^DeltaCom, the total amount billed to ITC^DeltaCom will not include any taxes due from the end user. ITC^DeltaCom will be solely responsible for the computation, tracking, reporting and payment of all federal, state and/or local jurisdiction taxes associated with the services resold to the end user.
- As the customer of record, ITC^DeltaCom will be responsible for, and remit to the BellSouth, all charges applicable to its resold services for emergency services (E911 and 911) and Telecommunications Relay Service (TRS) as well as any other charges of a similar nature.

- 12.4 If any portion of the payment is received by BellSouth after the payment due date as set forth preceding, or if any portion of the payment is received by BellSouth in funds that are not immediately available to BellSouth, then a late payment penalty shall be due to BellSouth. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor. The late factor shall be as set forth in Section A2 of the General Subscriber Service Tariff and Section B2 of the Private Line Service Tariff.
- Any switched access charges associated with interexchange carrier access to the resold local exchange lines will be billed by, and due to, BellSouth. No additional charges are to be assessed to ITC^DeltaCom.
- 12.6 BellSouth will not perform billing and collection services for ITC^DeltaCom as a result of the execution of this Agreement. All requests for billing services should be referred to the appropriate entity or operational group within BellSouth.
- 12.7 Pursuant to 47 CFR Section 51.617, BellSouth will bill ITC^DeltaCom end user common line charges identical to the end user common line charges BellSouth bills its end users.
- 12.8 In general, BellSouth will not become involved in disputes between ITC^DeltaCom and ITC^DeltaCom's end user customers over resold services. If a dispute does arise that cannot be settled without the involvement of BellSouth, ITC^DeltaCom shall contact the designated Service Center for resolution. BellSouth will make every effort to assist in the resolution of the dispute and will work with ITC^DeltaCom to resolve the matter in as timely a manner as possible. ITC^DeltaCom may be required to submit documentation to substantiate the claim.

13. Discontinuance of Service

- 13.1 The procedures for discontinuing service to an end user are as follows:
 - 13.1.1. Where possible, BellSouth will deny service to ITC^DeltaCom's end user on behalf of, and at the request of, ITC^DeltaCom. Upon restoration of the end user's service, restoral charges will apply and will be the responsibility of ITC^DeltaCom. If within fifteen days after an end user's service has been denied no contact has been made in reference to restoring service, the end user's service will be disconnected.
 - 13.1.2. At the request of ITC^DeltaCom, BellSouth will disconnect a ITC^DeltaCom end user customer.
 - 13.1.3. All requests by ITC^DeltaCom for denial or disconnection of an end user for nonpayment must be in writing or via the electronic interface established pursuant to Attachment 6 to the Agreement.
 - 13.1.4 ITC^DeltaCom will be made solely responsible for notifying the end user of the proposed disconnection of the service.
 - 13.1.5 BellSouth will continue to process calls made to the Annoyance Call Center and will advise ITC^DeltaCom when it is determined that annoyance calls are originated from one of their end user's locations. BellSouth shall be indemnified, defended and held harmless by ITC^DeltaCom and/or the end user against any claim, loss or damage arising from providing this information to ITC^DeltaCom. It is the responsibility of ITC^DeltaCom to take the corrective action necessary with its customers who make annoying calls. Failure to do so will result in BellSouth's disconnecting the end user's service.

- 13.2 The procedures for discontinuing service to ITC^DeltaCom are as follows:
 - 3.2.1 BellSouth reserves the right to suspend or terminate service for nonpayment or in the event of prohibited, unlawful or improper use of the facilities or service, abuse of the facilities, or any other violation or noncompliance by ITC^DeltaCom of the rules and regulations of BellSouth's Tariffs.
 - 13.2.2 If payment of account is not received by the bill day in the month after the original bill day, BellSouth may provide written notice to ITC^DeltaCom, that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition BellSouth may, at the same time, give thirty days notice to the person designated by ITC^DeltaCom to receive notices of noncompliance, and discontinue the provision of existing services to ITC^DeltaCom at any time thereafter subject to state and federal regulatory requirements.
 - 13.3.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
 - 13.3.4 If BellSouth does not refuse additional applications for service or discontinue the provision of services on the date specified in the thirty day notice and ITC^DeltaCom's noncompliance continues, nothing contained herein shall preclude BellSouth's right to refuse additional applications for service and discontinue the provision of existing service without further notice.
 - 13.3.5 If payment is not received or arrangements made for payment by the date given in the written notification, ITC^DeltaCom's services will be discontinued. Upon discontinuance of service on a ITC^DeltaCom account, service to ITC^DeltaCom's end users will be denied. BellSouth will also reestablish service at the request of the end user or ITC^DeltaCom upon payment of the appropriate connection fee and subject to BellSouth's normal application procedures. ITC^DeltaCom is solely responsible for notifying the end user of the proposed disconnection of the service.

14. MODIFICATION OF AGREEMENT

Provisions for modifying the terms, rates and conditions of this Attachment are contained in Section 16 of the General Terms and Conditions to this Agreement.

EXHIBIT A Page 1

APPLICABLE DISCOUNTS

The telecommunications services available for purchase by Reseller for the purposes of resale to Reseller end users shall be available at the following discount off of the retail rate.

DISCOUNT*

22000112				
<u>STATE</u>	RESIDENCE	BUSINESS	CSAs***	
ALABAMA	16.3%	16.3%		
FLORIDA	21.83%	16.81%		
GEORGIA	20.3%	17.3%		
KENTUCKY	16.79%	15.54%		
LOUISIANA	20.72%	20.72%	9.05%	
MISSISSIPPI	15.75%	15.75%		
NORTH CAROLINA	21.5%	17.6%		
SOUTH CAROLINA	14.8%	14.8%	8.98%	
TENNESSEE**	16%	16%		

- * When ITC^DeltaCom provides Resale service in a cross boundary area (areas that are part of the local serving area of another state's exchange) the rates, regulations and discounts for the tariffing state will apply. Billing will be from the serving state.
- ** In Tennessee, if ITC^DeltaCom provides its own operator services and directory services, the discount shall be 21.56%. ITC^DeltaCom must provide written notification to BellSouth within 30 days prior to providing its own operator services and directory services to qualify for the higher discount rate of 21.56%.
- *** Unless noted in this column, the discount for Business will be the applicable discount rate for CSAs.

EXHIBIT A Page 2

OPERATIONAL SUPPORT SYSTEMS (OSS) RATES

BellSouth has developed and made available the following mechanized systems by which ITC^DeltaCom may submit LSRs electronically.

LENS	Local Exchange Navigation System
EDI	Electronic Data Interchange
TAG	Telecommunications Access Gateway

LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic ordering charge as specified in the Table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

OPERATIONAL SUPPORT SYSTEMS (OSS) RATES	Electronic Per LSR received from the CLEC by one of the OSS interactive interfaces	Manual Per LSR received from the CLEC by means other than one of the OSS interactive
		interfaces
OSS LSR Charge	\$3.50	\$19.99
USOC	SOMEC	SOMAN

Note: In addition to the OSS charges, applicable discounted service order and related discounted charges apply per the tariff.

Denial/Restoral OSS Charge

In the event ITC^DeltaCom provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.

Cancellation OSS Charge

ITC^DeltaCom will incur an OSS charge for an accepted LSR that is later canceled by ITC^DeltaCom.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

EXHIBIT A Page 3

Threshold Billing Plan

The Parties agree that **ITC^DeltaCom** will incur the mechanized rate for all LSRs, both mechanized and manual, if the percentage of mechanized LSRs to total LSRs meets or exceeds the threshold percentages shown below:

Year Ratio: Mechanized/Total LSRs 2001 90%

The threshold plan will be discontinued in 2002.

BellSouth will track the total LSR volume for each CLEC for each quarter. At the end of that time period, a Percent Electronic LSR calculation will be made for that quarter based on the LSR data tracked in the LCSC. If this percentage exceeds the threshold volume, all of that CLECs' future manual LSRs will be billed at the mechanized LSR rate. To allow time for obtaining and analyzing the data and updating the billing system, this billing change will take place on the first day of the second month following the end of the quarter (e.g. May 1 for 1Q, Aug 1 for 2Q, etc.). There will be no adjustments to the amount billed for previously billed LSRs.

Attachment 1 Page 14

Type of Service	AL		FL		GA		KY		LA		MS		NC		SC		TN	
	Resale	Discount																
1 Grandfathered Services (Note 1)	Yes	Yes																
2 Contract Service Arrangements	Yes	Yes																
3 Promotions - > 90 Days(Note 2)	Yes	Yes	Yes	Note 3														
4 Promotions - < 90 Days (Note 2)	Yes	No	Yes	No	Yes	No		No	Yes	No								
5 Lifeline/Link Up Services	Yes	Yes	Yes	Yes	Yes	Yes	Note 4	Note 4	Yes	Yes								
6 911/E911 Services	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	No	No	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
7 N11 Services	Yes	Yes	Yes	Yes	Yes	Yes	No	No	No	No	Yes	Yes	Yes	Yes	No	No	Yes	Yes
8 AdWatch SM Svc (See Note 6)	Yes	Yes																
9 MemoryCall [®] Service	Yes	No																
10 Mobile Services	Yes	No																
11 Federal Subscriber Line Charges	Yes	No																
12 Non-Recurring Charges	Yes	Yes	Yes	No														
13 End User Line Charge – Number Portability	Yes	No																
14 Public Telephone Access Service (PTAS)	Yes	Yes	Yes	No	Yes	Yes												

Applicable Notes:

- 1. **Grandfathered services** can be resold only to existing subscribers of the grandfathered service.
- 2. Where available for resale, **promotions** will be made available only to end users who would have qualified for the promotion had it been provided by BellSouth directly.
- 3. In Tennessee, long-term **promotions** (offered for more than ninety (90) days) may be obtained at one of the following rates:

⁴ States 10-26-01

Attachment 1 Page 15

- (a) the stated tariff rate, less the wholesale discount;
- (b) the promotional rate (the promotional rate offered by BellSouth will not be discounted further by the wholesale discount rate)
- 4. **Lifeline/Link Up** services may be offered only to those subscribers who meet the criteria that BellSouth currently applies to subscribers of these services as set forth in Sections A3 and A4 of the BellSouth General Subscriber Services Tariff.
- 5. Some of BellSouth's local exchange and toll telecommunications services are not available in certain central offices and areas.
- 6. AdWatchSM Service is tariffed as BellSouth[®] AIN Virtual Number Call Detail Service.

Attachment 2

Unbundled Network Elements

TABLE OF CONTENTS

1.	INTRODUCTION	4
2.	UNBUNDLED LOOPS	5
3.	INTEGRATED DIGITAL LOOP CARRIERS	11
4.	NETWORK INTERFACE DEVICE	11
5.	UNBUNDLED LOOP CONCENTRATION (ULC) SYSTEM	13
6.	SUB-LOOP ELEMENTS	14
7.	UNBUNDLED NETWORK TERMINATING WIRE (UNTW)	17
8.	UNBUNDLED NETWORK ELEMENT COMBINATIONS	20
9.	SWITCHING	27
10.	TRANSPORT, CHANNELIZATION AND DARK FIBER	31
11.	TANDEM SWITCHING	38
12.	OPERATOR SYSTEMS	40
13.	SIGNALING	46
14.	SIGNALING TRANSFER POINTS (STPS)	48
15.	SERVICE CONTROL POINTS/DATABASES	52
16.	PREORDERING LOOP MAKEUP (LMU)	59
17.	SS7 NETWORK INTERCONNECTION	62
18. ASS	AIN SELECTIVE CARRIER ROUTING FOR OPERATOR SERVICES, DIRECTORY ISTANCE AND REPAIR CENTERS	66
19.	PACKET SWITCHING CAPABILITY	67
20.	BASIC 911 AND E911	69
21.	RATES	70
EXH	IIBIT A – LIDB STORAGE AGREEMENTEXHIBIT A	
EXH	IIBIT B – CNAM DATABASE SERVICESEXHIBIT B	

ACCESS TO UNBUNDLED NETWORK ELEMENTS

1. <u>Introduction</u>

- BellSouth shall, upon request of ITC^DeltaCom, and to the extent technically feasible, provide to ITC^DeltaCom access to its unbundled network elements for the provision of ITC^DeltaCom's telecommunications service. If no rate is identified in the contract, the rate for the specific service or function will be as set forth in applicable BellSouth tariff or as negotiated by the parties upon request by either party. Services cannot be charged as unbundled network elements; for example, ordering services from the tariff to a point collocated in a Central Office shall not incur UNE local loop or cross connect charges. At ITC^DeltaCom's option, access services may be ordered to the collocation space.
- 1.2 ITC^DeltaCom may purchase unbundled network elements from BellSouth for use in any manner ITC^DeltaCom chooses to provide telecommunication services to its intended users, including recreating existing BellSouth services. With the exception of the sub-loop elements which are located outside of the central office, BellSouth shall deliver the unbundled network elements purchased by ITC^DeltaCom for combining to the designated ITC^DeltaCom collocation space. The unbundled network elements shall be provided as set forth in this Attachment.
- 1.3 BellSouth will provide the following combined unbundled network elements for purchase by ITC^DeltaCom. The rate of the following combined unbundled network elements is the sum of the individual element prices as set forth in Attachment 11. Order Coordination as defined in Section 2 of Attachment 2 of this Agreement is available for each of these combinations. Order Coordination for combinations listed below involving an SL1 loop is available only at an additional charge:
 - Loop and cross connect
 - Port and cross connect
 - Port and cross connect and vertical features
 - Port and cross connect and common transport
 - Port and cross connect and common transport and vertical features
 - Port and vertical features
 - Loop with loop channelization (inside central office)
 - Loop with loop channelization (inside central office) and LNP
 - Port and common transport
 - Loop and LNP

- 1.4 BellSouth shall comply with the requirements as set forth in the technical references within Attachment 2 to the extent that they are consistent with the greater of BellSouth's actual performance or applicable industry standards.
- In the event that any effective legislative, regulatory, judicial or other legal action modifies or redefines the "Network Elements" in a manner which materially affects the terms of this Attachment or the Network Elements and/or prices set forth herein, either Party may, on thirty (30) days written notice, require renegotiation of such terms, and the Parties shall renegotiate in good faith such new terms in accordance with such legislative, regulatory, judicial or other legal action. In the event such new terms are not renegotiated within ninety (90) days after the notice for renegotiation, either party may petition the Commission for resolution of the dispute between the Parties. Each Party reserves the right to seek judicial review of any Commission ruling concerning this Attachment.
- 1.6 Performance Measurements associated with this Attachment 2 are contained in Attachment 10.

2. <u>Unbundled Loops</u>

2.1 BellSouth agrees to offer access to unbundled loops pursuant to the following terms and conditions and at the rates set forth in this Attachment.

2.2 <u>Definition</u>

- 2.2.1 The loop is the physical medium or functional path on which a subscriber's traffic is carried from the MDF or similar terminating device in a central office up to the termination at the NID at the customer's premise. Each unbundled loop will be provisioned with a NID.
- 2.2.2 The provisioning of loops to ITC^DeltaCom will require cross-office cabling and cross-connections within the central office to connect the loop to a local switch or to other transmission equipment in collocation space. These cross-connects are a separate element and are not considered a part of the loop.
- 2.2.2.1 BellSouth Order Coordination referenced in this Attachment includes two types: "Order Coordination" and "Order Coordination Time Specific."
- 2.2.2.2 "Order Coordination" (also known as Manual Order Coordination) refers to standard BellSouth service order coordination involving SL2 and 4-wire voice loops and all digital loops. Order coordination for physical conversions will be scheduled at BellSouth's discretion during normal working hours on the committed due date and ITC^DeltaCom advised.

- 2.2.2.3 "Order Coordination Time Specific" (also known as Order Coordination Time Specific) refers to service order coordination in which ITC^DeltaCom requests a specific time for a service order conversion to take place. Loops on a single service order of 14 or more loops will be provisioned on a project basis. OC-TS is a chargeable option in addition to any applicable OC charge. ITC^DeltaCom may specify a time between 9:00 a.m. and 4:00 p.m. Monday through Friday. If ITC^DeltaCom specifies a time outside this window, or selects a time or quantity of loops that requires BellSouth technicians to work outside normal work hours, overtime charges will apply in addition to the OC-TS charges.
- 2.2.2.4 Where facilities are available, BellSouth will install unbundled loops within a 5-7 business days interval. For orders of 14 or more unbundled loops, the installation will be handled on a project basis and the intervals will be set by the BellSouth project manager for that order. Said interval will be set in a reasonable manner and in accordance with any required extra work times. Some unbundled loops require a Service Inquiry (SI) to determine if facilities are available prior to issuing the order. The interval for the SI process is separate from the installation interval. For expedite requests by ITC^DeltaCom, expedite charges will apply for intervals less than 5 days. The charges outlined in BellSouth's FCC # 1 Tariff, Section 5.1.1, will apply. If ITC^DeltaCom cancels an order for UNE services, any costs incurred by BellSouth in conjunction with the provisioning of that order will be recovered in accordance with FCC #1 Tariff, Section 5.4.
- 2.2.3 BellSouth will offer Unbundled Voice Loops (UVL) in two different service levels - Service Level One (SL1) and Service Level Two (SL2). SL1 loops will be nondesigned, and will not have test points. Order Coordination (OC) and/or engineering information/circuit make-up data will be chargeable options. Upon issuance of an order in the service order system, SL1 loops without optional Order Coordination will be activated on the due date in the same manner and time frames that BellSouth normally activates POTS-type loops for its customers; provided, however, that for loop activation in BellSouth staffed central offices, BellSouth will use its best efforts to provide an a.m. or p.m. designation only where loop requires dispatching of a BellSouth technician and where ITC^DeltaCom has specifically requested an a.m. or p.m. preference for activation on the LSR. Further, for loop activation in BellSouth central offices that are not staffed, BellSouth will use its best efforts to provide an a.m. or p.m. designation only where loop activation requires dispatching of a BellSouth technician and where ITC^DeltaCom has specifically requested a.m. or p.m. preference for activation on the LSR. SL2 loops shall have test points, will be designed with a Design Layout Record provided to ITC^DeltaCom, and will be provided with Order Coordination. The OC feature will allow ITC^DeltaCom to coordinate the installation of the loop with the disconnect of an existing customer's service and/or number portability service. In these cases, BellSouth will perform the order conversion with standard order coordination at its discretion during normal work hours.

- 2.2.4 BellSouth will also offer Unbundled Digital Loops (UDL). They will be designed (where appropriate), will be provisioned with test points (where appropriate), and will come standard with Order Coordination and a Design Layout Record (DLR).
- 2.2.5 As a chargeable option on all unbundled loops BellSouth will offer Order Coordination Time Specific (OC-TS). This will allow ITC^DeltaCom the ability to specify the time that the coordinated conversion takes place.
- 2.2.6 ITC^DeltaCom will be responsible for testing and isolating troubles on the unbundled loops. Once ITC^DeltaCom has isolated a trouble to the BellSouth provided loop, ITC^DeltaCom will issue a trouble to BellSouth on the loop. BellSouth will take the actions necessary to repair the loop if a trouble actually exists. BellSouth will repair these loops in the same time frames that BellSouth repairs similarly situated loops to its customers
- Either Party may charge the other for dispatching and testing of a trouble where the trouble was found not to be in the network of the dispatching or testing Party and the dispatching or testing Party's equipment did not cause the dispatch. Where there is a dispute as to the appropriateness of such charge, the Parties will meet and review the record of repair history and determine whether the charge was appropriate. Charges so assessed by BellSouth shall be on a time and materials basis as set forth in BellSouth's state commission approved tariffs. Charges so assessed by ITC^DeltaCom shall be on a time and materials basis as set forth in ITC^DeltaCom's state commission approved tariffs. If ITC^DeltaCom does not have state commission approved tariffs addressing such charges, then such charges shall be assessed by ITC^DeltaCom at the rates set forth in BellSouth's tariffs. If the trouble which was originally found not to be in the network of the dispatching or testing Party is later proven to be a trouble in the dispatching or testing Party's network, the dispatching or testing Party shall waive or refund any such charges.

2.2.8 <u>Ordering Process</u>

- 2.2.8.1 The ordering process for unbundled loops shall proceed in accordance with this Section 2.2.8 and Attachment 6 of this Agreement.
- 2.2.8.2 BellSouth shall exercise its best efforts in attempting to meet the conversion time ITC^DeltaCom requests through the LSR. However, unless ITC^DeltaCom's LSR specifies a time-specific conversion, in which case the conversion must commence at the time indicated in the LSR, then within forty-eight (48) to twenty-four (24) hours prior to the date and time requested for the loop conversion in ITC^DeltaCom's LSR and acknowledged in BellSouth's FOC, BellSouth may contact ITC^DeltaCom, via telephone, to finalize a scheduled conversion time (i.e., a specific time, on the date set forth in the FOC) which may be different from the conversion time ITC^DeltaCom requested in the LSR. The scheduled conversion time shall be the time at which the parties shall commence coordination

of loop installation with the disconnect and reconnect of an end user's service and any number portability update. BellSouth shall not assess any additional charges for scheduled conversion times commencing between BellSouth normal business hours as set forth in Section 4.6.1 of Attachment 6.

- 2.2.8.3 At the scheduled conversion time, BellSouth shall have a sixty (60) minute window within which it shall contact ITC^DeltaCom to begin the loop conversion process. Provided, however, that if ITC^DeltaCom requested a time-specific conversion, the conversion shall commence at the time indicated in ITC^DeltaCom's LSR and be completed consistently with timeframes for time-specific conversions.
- 2.2.8.3.1 If either Party dispatches a technician for a loop conversion and the other Party fails to complete the conversion at the scheduled time, the non-performing Party may be charged the one (1) hour additional engineering charges set forth in BellSouth's FCC No. 1 tariff, Section 13.1.
- 2.2.8.4 After the loop conversion process commences, a coordinated loop cutover, which shall include coordinated conversion of number portability, shall be completed within the following time periods:
- 2.2.8.4.1 For single loop conversions per location, the conversion shall be completed within fifteen (15) minutes;
- 2.2.8.4.2 For up to ten (10) loop conversions per location, the conversion of all loops shall be completed within sixty (60) minutes, and each individual loop conversion shall be completed within fifteen (15) minutes;
- 2.2.8.4.3 For loop conversions not exceeding thirty (30) loops per location and not determined complex or exceptionally large, the conversion of all loops shall be completed within one hundred and twenty (120) minutes. All loops above a thirty loop quantity, or ten (10) loop quantity and determined as complex (a cut that requires more operation than a single cut point), will be negotiated by ITC^DeltaCom and BellSouth prior to the due date.
- 2.2.8.4.4 BellSouth agrees that upon ITC^DeltaCom's request, for order coordinated loop cutovers involving three (3) or more lines, at least two lines will remain in service at all times during the conversion process.
- 2.2.8.5 Where facilities for requested new services do not currently exist, the installation intervals will be determined by BellSouth. ITC^DeltaCom will then be notified of the targeted due date. BellSouth shall provide ITC^DeltaCom adequate justification and an explanation of the unusual circumstances that caused BellSouth to be unable to meet these commitments.

2.3 <u>Technical Requirements</u>

4 States Attachment 2 Clean 12-17-01

- 2.3.1 To the extent available within BST's Network at a particular location, BellSouth will offer loops capable of supporting telecommunications services such as: POTS, Centrex, basic rate ISDN, analog PBX, voice grade private line, ADSL, HDSL, DS1 and digital data (up to 64 kb/s). Additional services may include digital PBXs, primary rate ISDN, xDSL, and Nx 64 kb/s. If a requested loop type is not available, then ITC^DeltaCom can use the Special Construction process to request that BellSouth place facilities or otherwise modify facilities in order to meet the ITC^DeltaCom's request.
- 2.3.1.1 The loop will support the transmission, signaling, performance and interface requirements of the services described in 2.3.1 above. It is recognized that the requirements of different services are different, and that a number of types or grades of loops are required to support these services. Services provided over the loop by ITC^DeltaCom will be consistent with industry standards and BST's TR73600.
- In some instances, ITC^DeltaCom will require access to a copper twisted pair loop unfettered by any intervening equipment (e.g., filters, load coils, range extenders, etc.) and/or limited amounts of bridge/end taps, so that ITC^DeltaCom can use the loop for a variety of services by attaching appropriate terminal equipment at the ends. ITC^DeltaCom will determine the type of service that will be provided over the loop. In some cases, ITC^DeltaCom may be required to pay additional charges for the removal of certain types of equipment. Unless and until BellSouth implements a separate charge for loop conditioning, BellSouth's Special Construction process will be used to determine the costs and feasibility of these activities. If ITC^DeltaCom requests loop conditioning as described in this Section, BellSouth will construct the loop type ordered and will maintain such loop to the characteristics and specifications of the loop type ordered.
- 2.3.1.2.1 In cases in which ITC^DeltaCom has requested that BellSouth remove equipment from the BellSouth loop, BellSouth will no longer be expected to maintain and repair the loop to the standards specified for that original loop type in the TR73600 and other standards referenced in this Agreement.
- 2.3.1.2.2. ITC^DeltaCom, in performance of its obligations pursuant to the preceding Section, shall maintain records that will reflect that pursuant to ITC^DeltaCom's request BellSouth has removed certain equipment from BellSouth provided loops and as such, the loop may not perform within the technical specifications associated with the original loop type. ITC^DeltaCom will not report to BellSouth troubles on said loops where the loops are not performing within the technical specifications of that original loop type. However, in the event that ITC^DeltaCom has requested such modifications to the loop and troubles arise on the modified loop, BellSouth will restore the loop only to maintain the technical characteristics of (1) electrical (DC) continuity, (2) balance between tip and ring,

- and (3) resistance on loops no longer than 18,000 feet. On loops longer than 18,000 feet, resistance will be maintained where technically feasible.
- In addition, ITC^DeltaCom recognizes there may be instances, where a loop modified in this manner may be subjected to normal network configuration changes that may cause the circuit characteristics to be changed and may create an outage of the service that ITC^DeltaCom has placed on the loop. If this occurs, BellSouth will work cooperatively with ITC^DeltaCom to restore the circuit to its previous modified status as quickly as possible. ITC^DeltaCom will pay the Time and Materials costs associated with BellSouth's work efforts needed to bring the loop back to its previous modified status. BellSouth will use best efforts to prevent the occurrence of such changes.
- 2.3.1.3 To the extent BellSouth converts a resold service to unbundled network elements for any telecommunications carrier, BellSouth shall make available to ITC^DeltaCom the same conversion for the same services and elements on the same terms and conditions and at the same rates, if any; provided, however that the rate for such conversion shall not exceed those rates set forth in Attachment 11 to this Agreement. The Parties agree that such rates are interim and upon establishment of a permanent rate, either through negotiation or by order of the Commission, the parties will amend this Agreement to reflect the new rate and will true up such rate retroactively back to the effective date of this Agreement.
- 2.3.1.4 BellSouth shall develop a process to identify the carrier for each unbundled loop and establish automated inter-company referral and/or call hand-off processes for an additional charge developed via the BFR process. In addition, ITC^DeltaCom may deploy DLC equipment (TR 303 compliant) in ITC^DeltaCom's collocation space or in ITC^DeltaCom's network.
- 2.3.2 The loop shall be provided to ITC^DeltaCom in accordance with the following Technical References:
 - BellSouth's TR73600, Unbundled Local Loop Technical Specification
- 2.3.2.1 Bellcore TR-NWT-000057, Functional Criteria for Digital Loop Carrier Systems, Issue 2, January 1993.
- 2.3.2.2 Bellcore TR-NWT-000393, Generic Requirements for ISDN Basic Access Digital Subscriber Lines.
- 2.3.2.3 ANSI T1.102 1993, American National Standard for Telecommunications Digital Hierarchy Electrical Interfaces.
- 2.3.2.4 ANSI T1.403 1989, American National Standard for Telecommunications Carrier to Customer Installation, DS1 Metallic Interface Specification.

2.3.2.5 ANSI T1.413 – 1998, American National Standard for Telecommunications Network and Customer Installation Interfaces – Asymmetric Digital Subscriber Line (ADSL) Metallic Interface.

3. Integrated Digital Loop Carriers

Where BellSouth uses Integrated Digital Loop Carrier (IDLCs) systems to provide the local loop and BellSouth has an alternate facility available, BellSouth will make alternative arrangements to permit DeltaCom to order a contiguous unbundled local loop. To the extent it is technically feasible, these arrangements will provide DeltaCom with the capability to serve end users at the same level BellSouth provides its customers. If no alternate facility is available, BST will utilize its Special Construction (SC) process to determine the additional costs required to provision the loop facilities. DeltaCom will then have the option of paying the one-time SC rates to place the loop facilities or DeltaCom may chose some other method of providing service to the end-user (e.g., Resale, private facilities, etc.)

4. Network Interface Device

4.1 Definition

The NID is defined as any means of interconnection of end-user customer inside wire to BellSouth's distribution plant, such as a cross-connect device used for that purpose. The NID is a single-line termination device or that portion of a multiple-line termination device required to terminate a single line or circuit at the premises. The NID features two independent chambers or divisions that separate the service provider's network from the End User's on-premises wiring. Each chamber or division contains the appropriate connection points or posts to which the service provider and the end user each make their connections. The NID provides a protective ground connection and is capable of terminating cables such as twisted pair cable.

- 4.2 BellSouth shall permit ITC^DeltaCom to connect ITC^DeltaCom's loop facilities the end-user's inside wire through the BellSouth NID or at any other technically feasible point.
- 4.3 Access to Network Interface Device (NID)
- 4.3.1 Due to the wide variety of NIDs utilized by BellSouth (based on subscriber size and environmental considerations), ITC^DeltaCom may access the end user's wire by any of the following means: BellSouth shall allow ITC^DeltaCom to connect its loops directly to BellSouth's multi-line residential NID enclosures that have additional space and are not used by BellSouth or any other telecommunications carriers to provide service to the premise. ITC^DeltaCom agrees to install

compatible protectors and test jacks and to maintain the protection system and equipment and to indemnify BellSouth pursuant to Section 6 of the General Terms and Conditions of this Agreement.

- 4.3.1.1 Where an adequate length of the end user's inside wire is present and environmental conditions permit, either Party may remove the inside wire from the other Party's NID and connect that wire to that Party's own NID; or
- 4.3.1.2 Enter the subscriber access chamber or "side" of "dual chamber" NID enclosures for the purpose of extending a connecterized or spliced jumper wire from the inside wiring through a suitable "punch-out" hole of such NID enclosures; or
- 4.3.1.3 Request BellSouth to make other rearrangements to the inside wiring terminations or terminal enclosure on a time and materials cost basis to be charged to the requesting Party (i.e., ITC^DeltaCom, its agent, the building owner or the subscriber). Such charges will be billed to the requesting Party.
- In no case shall either Party remove or disconnect the other Party's loop facilities from either Party's NIDs, enclosures, or protectors unless: (1) the applicable Commission has expressly permitted the same; (2) the disconnecting Party provides prior notice to the other Party, and (3) the Party disconnecting appropriately caps off and guards the other Party's loops. It will be ITC^DeltaCom's responsibility to ensure there is no safety hazard and will hold BellSouth harmless for any liability associated with the removal of the BellSouth loop from the BellSouth NID. In such cases, it shall be the responsibility of the disconnecting party, once the other Party's loop has been disconnected from the NID, to reconnect the disconnected loop to a nationally- recognized–testing-laboratory-listed station protector, which has been grounded as per Article 800 of the National Electrical Code. If ITC^DeltaCom does not wish to accept this responsibility, other options exist in which BellSouth installs a NID for ITC^DeltaCom as a chargeable option.
- 4.3.1.5 In no case shall either Party remove or disconnect ground wires from BellSouth's NIDs, enclosures, or protectors.
- 4.3.1.6 In no case shall either Party remove or disconnect NID modules, protectors, or terminals from BellSouth's NID enclosures.
- 4.3.1.7 Due to the wide variety of NID enclosures and outside plant environments BellSouth will work with ITC^DeltaCom to develop specific procedures to establish the most effective means of implementing this Section 4.
- 4.3.2 Technical Requirements
- 4.3.2.1 The NID shall provide an accessible point of interconnection and shall maintain a connection to ground.

- 4.3.2.2 The NID shall be capable of transferring electrical analog or digital signals between the subscriber's inside wiring and the Distribution Media and/or cross connect to ITC^DeltaCom's NID, consistent with the NID's function at the Effective Date of this Agreement.
- 4.3.2.3 Where a BellSouth NID exists, it is provided in its "as is" condition. ITC^DeltaCom may request BellSouth do additional work to the NID in accordance with Section 4.3.1.7. When ITC^DeltaCom deploys its own local loops with respect to multiple-line termination devices, ITC^DeltaCom shall specify the quantity of NIDs connections that it requires within such device.
- 4.3.1 <u>Interface Requirements</u>
- 4.3.2 The NID shall be equal to or better than all of the requirements for NIDs set forth in the following technical references:
- 4.3.3 Bellcore Technical Advisory TA-TSY-000120 "Customer Premises or Network Ground Wire";
- 4.3.4 Bellcore Generic Requirement GR-49-CORE "Generic Requirements for Outdoor Telephone Network Interface Devices";
- 4.3.5 Bellcore Technical Requirement TR-NWT-00239 "Indoor Telephone Network Interfaces";
- 4.4.5 Bellcore Technical Requirement TR-NWT-000937 "Generic Requirements for Outdoor and Indoor Building Entrance"

5. Unbundled Loop Concentration (ULC) System

- 5.1 BellSouth will provide to ITC^DeltaCom Unbundled Loop Concentration (ULC). Loop concentration systems in the central office concentrate the signals transmitted over local loops onto a digital loop carrier system. The concentration device is placed inside a BellSouth central office. BellSouth will offer ULC with a TR008 interface or a TR303 interface
- 5.2 ULC will be offered in two sizes. System A will allow up to 96 BellSouth loops to be concentrated onto multiple DS1s. The high-speed connection from the concentrator will be at the electrical DS1 level and may connect to ITC^DeltaCom at ITC^DeltaCom's collocation site. System B will allow up to 192 BellSouth loops to be concentrated onto multiple DS1s. System A may be upgraded to a System B. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). All DS1 interfaces will terminate to the CLEC's collocation space. ULC service is offered with or without concentration and with or without protection. A

Line Interface element will be required for each loop that is terminated onto the ULC system. Rates for ULC are as set forth in this Attachment 11.

6. Sub-loop Elements

- Where facilities permit and subject to applicable and effective FCC rules and orders, BellSouth shall offer access to its Unbundled Sub Loop (USL) and Unbundled Sub-loop Concentration (USLC) System. BellSouth shall provide non-discriminatory access, in accordance with 51.311 and Section 251(c) (3) of the Act, to the sub-loop. On an unbundled basis and pursuant to the following terms and conditions and the rates approved by the Commission and set forth in Attachment 11 of this Agreement.
- 6.1.1 Sub-loop components include but are not limited to the following:
- 6.1.2 Unbundled Sub-Loop Distribution;
- 6.1.3 Unbundled Sub-Loop Concentration/Multiplexing Functionality; and
- 6.1.4 Unbundled Sub-Loop Feeder.

6.2 <u>Unbundled Sub-Loop (distribution facilities)</u>

6.2.1 <u>Definition</u>

- 6.2.2 Subject to applicable and effective FCC rules and orders, the unbundled sub-loop distribution facility is dedicated transmission facility that BellSouth provides from a customer's point of demarcation to a BellSouth cross-connect device. The BellSouth cross-connect device may be located within a remote terminal (RT) or a stand-alone cross-box in the field or in the equipment room of a building. The unbundled sub-loop distribution media is a copper twisted pair that can be provisioned as a 2 Wire or 4 Wire facility. Following are the current sub-loop distribution offerings:
- 6.2.2.1 Voice grade Unbundled Sub-Loop Distribution (USL-D) is a sub-loop facility from the cross-box in the field up to and including the point of demarcation, at the end user's premises.
- Unbundled Sub-Loop distribution facilities were originally built as part of the entire voice grade loop from the BellSouth central office to the customer network interface. Therefore, the voice grade Unbundled Sub-Loop may have load coils, which are necessary for transmission of voice grade services.
- 6.2.2.3 Unbundled Copper Sub-Loop (UCSL) is a non-loaded copper facility of any length provided from the cross-box in the field up to and including the end-user's point of demarcation.

- 6.2.2.3.1 If available, this facility will not have any intervening equipment such as load coils between the end-user and the cross-box.
- 6.3 If ITC^DeltaCom requests a UCSL and a non-loaded pair is not available, ITC^DeltaCom may order Unbundled Sub-Loop Modification to remove load coils and/or bridge tap from an existing sub-loop facility. If load coils are removed from an existing sub-loop, that sub-loop will be classified as a UCSL. ITC^DeltaCom may order Loop Make-up to determine what loop modifications will be required.
- 6.3.1 Unbundled Sub-Loop distribution facilities shall support functions associated with provisioning, maintenance and testing of the Unbundled Sub-Loop. For access to Voice Grade USL-D and UCSL, ITC^DeltaCom would be required to deliver a cable to the BellSouth remote terminal or cross-box in the field to provide continuity to ITC^DeltaCom's feeder facilities. This cable would be connected, by a BellSouth technician, within the BellSouth RT/cross-box during the set-up process. ITC^DeltaCom's cable pairs can then be connected to BellSouth's USL within the BellSouth cross-box by the BellSouth technician.
- Unbundled Sub-Loop Intrabuilding Network Cable (USL-INC) (a.k.a. riser cable) is the distribution facility inside a subscribers' building or between buildings on one customer's same premises (continuous property not separated by a public street or road). USL-INC includes the facility from the cross-connect device in the building equipment room up to and including the point of demarcation, at the end user's premises. In a scenario that requires connection in a building equipment room, BellSouth will install a cross connect panel that will be installed for the purpose of accessing USL-INC pairs. The cross-connect panel will function as a single point of interconnection (SPOI) for USL-INC and will be accessible by multiple carriers as space permits. BellSouth will place cross-connect blocks in 25 pair increments for ITC^DeltaCom's use on this cross-connect panel. ITC^DeltaCom will be responsible for connecting its facilities to the 25 pair cross-connect block(s).
- 6.3.3 BellSouth will provide Unbundled Sub-Loops where possible. Through the firm order Service Inquiry (SI) process, BellSouth will determine if it is feasible to place the required facilities where ITC^DeltaCom has requested access to Unbundled Sub-Loops. If existing capacity is sufficient to meet ITC^DeltaCom's demand, then BellSouth will perform the set-up work as described in Section 6.3.4. If any work must be done to modify existing BellSouth facilities or add new facilities (other than adding the cross-connect panel in a building equipment room as noted in 6.3.4) to accommodate ITC^DeltaCom's request for Unbundled Sub-Loops, ITC^DeltaCom may request BellSouth's Special Construction (SC) process to determine additional costs required to provision the Unbundled Sub-Loops. ITC^DeltaCom will have the option of paying the SC charges to modify the BellSouth facilities.

- 6.3.4 Set-up work must be completed before ITC^DeltaCom can order sub-loop pairs. During the set-up in a BellSouth cross-connect box in the field, the BellSouth technician will perform the necessary work to splice ITC^DeltaCom's cable into the cross-connect box. For the set-up inside a building equipment room, BellSouth will perform the necessary work to install the cross-connect panel and the connecting block(s) that will be used to provide access to the requested USLs.
- Once the set-up is complete, ITC^DeltaCom will request sub-loop pairs through submission of a Local Service Request (LSR) form to the Local Carrier Service Center (LCSC). Manual Order Coordination is required with USL pair provisioning and is in addition to the USL pair rate. For expedite requests by ITC^DeltaCom for sub-loop pairs, expedite charges will apply for intervals less than 5 days.
- Unbundled Sub-Loop shall be equal to or better than each of the applicable requirements set forth in the applicable industry standard technical references.
- 6.3.7 Unbundled Sub-Loops will be provided in accordance with technical reference TR73600.

6.4 Unbundled Sub-Loop Concentration System (USLC)

- 6.4.1 Where facilities permit and where necessary to comply with an effective Commission order, BellSouth will provide ITC^DeltaCom with the ability to concentrate its sub-loops onto multiple DS1s back to the BellSouth Central Office. The DS1s will then be terminated into ITC^DeltaCom collocation space. TR-008 and TR303 interface standards are available.
- USLC, using the Lucent Series 5 equipment, will be offered in two different systems. System A will allow up to 96 of ITC^DeltaCom's sub-loops to be concentrated onto multiple DS1s. System B will allow an additional 96 of ITC^DeltaCom's sub-loops to be concentrated onto multiple DS1s. One System A may be supplemented with one System B and they both must be physically located in a single Series 5 dual channel bank. A minimum of two DS1s is required for each system (i.e., System A requires two DS1s and System B would require an additional two DS1s or four in total). The DS1 level facility that connects the RT site with the serving wire center is known as a Feeder Interface. All DS1 Feeder Interfaces will terminate to ITC^DeltaCom's collocation space within the SWC that serves the RT where ITC^DeltaCom's sub-loops are connected. USLC service is offered with or without concentration and with or without a protection DS1.
- In these scenarios ITC^DeltaCom would be required to place a cross-box, remote terminal (RT), or other similar device and deliver a cable to the BellSouth remote terminal. This cable would be connected, by a BellSouth technician, to a cross-connect panel within the BellSouth RT/cross-box and would allow

ITC^DeltaCom's sub-loops to then be placed on the ULSC and transported to their collocation space at a DS1 level.

6.5 <u>Unbundled Sub-Loop Feeder</u>

6.5.1 Definition

- 6.5.2 Unbundled Sub-Loop Feeder (USLF) provides connectivity between BellSouth's central office and its cross-box (or other access point) that serves an end user location.
- 6.5.3 USLF is intended to be utilized for voice traffic and can be configured as 2-wire voice (USLF-2W/V) or 4-wire voice (USLF-4W/V).
- USLF can also to be utilized for digital traffic and can be configured as 2-wire ISDN (USLF-2W/I); 2-wire Copper (USLF-2W/C); 4-wire Copper (USLF-4W/C) facilities: 4-wire DS0 level loop (USLF-4W/D0); or 4-wire DS1 & ISDN (USLF-4W/DI).
- 6.5.5 USLF will provide the facilities needed to provision a 2W or 4W communications pathway from the BellSouth central office to the BellSouth cross-box. This element will allow for the connection of ITC^DeltaCom's loop distribution elements onto BellSouth's feeder system.

6.6 Requirements

- 6.6.1 ITC^DeltaCom will extend its compatible cable to BellSouth's cross-box. The cable will then be connected to a panel inside the BellSouth cross-box to the requested level of feeder element. In those cases when there is no room in the BellSouth cross-box to accommodate the additional cross-connect panels mentioned above, BellSouth will utilize its Special Construction process to determine the costs to provide the sub-loop feeder element to ITC^DeltaCom. ITC^DeltaCom will then have the option of paying the special construction charges or canceling the order.
- USLF will be a designed circuit and BellSouth will provide a Design Layout Record (DLR) for this element.
- 6.6.3 BellSouth will provide USLF elements in accordance with applicable industry standards for these types of facilities. Where industry standards do not exist, BellSouth's TR73600 will be used to determine performance parameters.

7. <u>Unbundled Network Terminating Wire (UNTW)</u>

7.1 BellSouth agrees to offer its Unbundled Network Terminating Wire (UNTW) to ITC^DeltaCom pursuant to the following terms and conditions at rates as set forth in Attachment 11.

7.2 <u>Definition</u>

7.3 Subject to applicable and effective FCC rules and orders, UNTW is a dedicated transmission facility that BellSouth provides from the Wiring Closet /Garden Terminal (or other type of cross-connect point) at the point of termination of BellSouth's loop distribution facilities to the end user's point of demarcation. UNTW is the final portion of the loop owned by BellSouth.

7.4 <u>Requirements</u>

- 7.4.1 On a multi-unit premises where Provisioning Party owns the network terminating wire, and by request of Requesting Party, Provisioning Party will provide access to UNTW pairs on an Access Terminal that is suitable for use by multiple carriers at each Garden Terminal or Wiring Closet.
- 7.4.2 In new construction where possible, both Parties may at their option and with the property owner's agreement install their own Network Terminating Wire (NTW). In existing construction, the Provisioning Party shall not be required to install new or additional NTW beyond existing NTW to provision the services of the Requesting Party.
- 7.4.3 Upon notice from the Requesting Party to the Provisioning party that the Requesting Party desires access to the Provisioning Party's UNTW pairs in a multi-unit premises, representatives of both Parties will participate in a meeting at the site of the requested access. The purpose of the site visit will include discussion of the procedures for Access Terminal installation, location and addresses of the Access Terminals and to discuss an estimated completion date. Upon completion of site visit, the Requesting Party will submit a Service Inquiry (SI) to the person or organization designated by the Provisioning Party to receive the SI. The SI will initiate the work for the Provisioning Party to begin the Access Terminal installation. In multi-tenant unit (MTU) scenarios, Provisioning Party will provide access to UNTW pairs on an Access Terminal(s). By request of the Requesting Party, an Access Terminal will be installed either adjacent to each Provisioning Party's Garden Terminal or inside each Wiring Closet on the requested MTU. All the UNTW pairs served by a Garden Terminal/Wiring Closet will be made available on the Access Terminals. Requesting Party will deliver and connect its central office facilities to the UNTW pairs within the Access Terminal. Requesting Party may access any available pair on an Access Terminal unless the Provisioning Party or another service provider is using the pair to concurrently provide service. Prior to connecting Requesting Party's service on a pair previously used by Provisioning party, Requesting Party is responsible for ensuring

the end-user is no longer using Provisioning Party's service or another CLEC's service before accessing UNTW pairs.

- 7.4.4 Provisioning Party will use best efforts to complete installation of the Access Terminals within 30 business days of the receipt by the Provisioning Party of the Service Inquiry from the Requesting Party.
- 7.4.5 Requesting Party is responsible for obtaining the property owner's permission for Provisioning Party to install an Access Terminal(s) on behalf of the Requesting Party. The submission of the SI by the Requesting Party will serve as certification by the Requesting Party that such permission has been obtained.
- 7.4.6 Requesting Party will be billed for non-recurring and recurring charges for accessing UNTW pairs at the time the Requesting Party activates the pair(s). ITC^DeltaCom will report use of the UNTW pairs on a Local Service Request (LSR) form submitted to BellSouth's Local Carrier Service Center (LCSC).
- 7.4.7 Requesting Party will isolate and report repair problems to the UNE center. Requesting Party must tag the UNTW pair that requires repair. If Provisioning Party dispatches a technician on a reported trouble call and no UNTW trouble is found, Provisioning Party will charge Requesting Party for time spent on the dispatch and testing the UNTW pair(s).
- 7.4.8 If Requesting Party initiates the Access Terminal installation and the Requesting Party has not activated at least one pair on the Access Terminal installed pursuant to Requesting Party's request for an Access Terminal within 6 months of installation of the Access Terminal, Provisioning Party will bill Requesting Party a non-recurring charge equal to the actual cost of provisioning the Access Terminal.
- 7.4.9 If Provisioning Party determines that Requesting Party is using the UNTW pairs without reporting such usage to BellSouth, the following charges shall apply in addition to any fines which may be established by state commissions and any other remedies at law or in equity available to the Provisioning Party:
- 7.4.10 If Requesting Party issued a LSR to disconnect an end-user from BellSouth in order to use a UNTW pair, Requesting Party will be billed for the use of the pair back to the disconnect order date.
- 7.4.11 If Requesting Party activated a UNTW pair on which Provisioning Party was not previously providing service, Requesting Party will be billed for the use of that pair back to the date the end-user began receiving service using that pair. Upon request, Requesting Party will provide copies of its billing record to substantiate such date. If Requesting Party fails to provide such records, then Provisioning Party will bill the Requesting Party back to the date of the Access Terminal installation.

8. Unbundled Network Element Combinations

- 8.1 Unbundled Network Element Combinations shall include: 1) Enhanced Extended Links (EELs) 2) UNE Loops/Special Access Combinations 3) Loop/Port Combinations and 4) Transport Combinations.
- 8.2 For purposes of this Section, references to "Currently Combined" network elements shall mean that such network elements are in fact already combined by BellSouth in the BellSouth network to provide service to a particular end user at a particular location.

8.3 EELs

- Where facilities permit and where necessary to comply with an effective FCC and/or State Commission order, or as otherwise mutually agreed by the Parties, BellSouth shall offer access to loop and transport combinations, also known as the Enhanced Extended Link ("EEL") as defined in Section 8.3.2 below.
- 8.3.2 Subject to Section 8.3.3 below, BellSouth will provide access to the EEL in the combinations set forth in Section 8.3.4 following. This offering is intended to provide connectivity from an end user's location through that end user's SWC to ITC^DeltaCom's POP serving wire center. The channels on the circuit sufficient to meet the local usage options described in Section 8.3.5 below, must be connected to ITC^DeltaCom's switch for the purpose of provisioning telephone exchange service to ITC^DeltaCom's end-user customers. The EEL will be connected to ITC^DeltaCom's facilities in ITC^DeltaCom's collocation space at the POP SWC, or ITC^DeltaCom may purchase BellSouth's access facilities between ITC^DeltaCom's POP and ITC^DeltaCom's collocation space at the POP SWC.
- 8.3.3 BellSouth shall provide EEL combinations to ITC^DeltaCom in Kentucky, Louisiana, Mississippi, and South Carolina regardless of whether or not such EELs are Currently Combined. In all other states, BellSouth shall make available to ITC^DeltaCom those EEL combinations described in Section 8.3.4 below only to the extent such combinations are Currently Combined. Furthermore, BellSouth will make available EEL combinations to ITC^DeltaCom in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999, in the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs regardless of whether or not such EELs are Currently Combined. Except as stated above, EELs will be provided to ITC^DeltaCom only to the extent such network elements are Currently Combined.
- 8.3.4 EEL Combinations
- 8.3.4.1 DS1 Interoffice Channel + DS1 Channelization + 2-wire VG Local Loop

- 8.3.4.2 DS1 Interoffice Channel + DS1 Channelization + 4-wire VG Local Loop 8.3.4.3 DS1 Interoffice Channel + DS1 Channelization + 2-wire ISDN Local Loop 8.3.4.4 DS1 Interoffice Channel + DS1 Channelization + 4-wire 56 kbps Local Loop 8.3.4.5 DS1 Interoffice Channel + DS1 Channelization + 4-wire 64 kbps Local Loop 8.3.4.6 DS1 Interoffice Channel + DS1 Local Loop 8.3.4.7 DS3 Interoffice Channel + DS3 Local Loop 8.3.4.8 STS-1 Interoffice Channel + STS-1 Local Loop 8.3.4.9 DS3 Interoffice Channel + DS3 Channelization + DS1 Local Loop 8.3.4.10 STS-1 Interoffice Channel + DS3 Channelization + DS1 Local Loop 8.3.4.11 2-wire VG Interoffice Channel + 2-wire VG Local Loop 8.3.4.12 4-wire VG Interoffice Channel + 4-wire VG Local Loop 8.3.4.13 4-wire 56 kbps Interoffice Channel + 4-wire 56 kbps Local Loop 8.3.4.14 4-wire 64 kbps Interoffice Channel + 4-wire 64 kbps Local Loop 8.3.5 **Special Access Service Conversions** 8.3.5.1 ITC^DeltaCom may not convert special access services to combinations of loop and transport network elements, whether or not ITC^DeltaCom self-provides its entrance facilities (or obtains entrance facilities from a third party), unless ITC^DeltaCom uses the combination to provide a significant amount of local exchange service, in addition to exchange access service, to a particular customer. To the extent ITC^DeltaCom requests to convert any special access services to combinations of loop and transport network elements at UNE prices, ITC^DeltaCom shall provide to BellSouth a letter certifying that ITC^DeltaCom is providing a significant amount of local exchange service (as described in this Section) over such combinations. The certification letter shall also indicate under
- 8.3.5.1.1 ITC^DeltaCom certifies that it is the exclusive provider of an end user's local exchange service. The loop-transport combinations must terminate at ITC^DeltaCom's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport combinations to be connected to

what local usage option ITC^DeltaCom seeks to qualify for conversion of special access circuits. ITC^DeltaCom shall be deemed to be providing a significant amount of local exchange service over such combinations if one of the following

options is met:

⁴ States Attachment 2 Clean 12-17-01

BellSouth's tariffed services. Under this option, ITC^DeltaCom is the end user's only local service provider, and thus, is providing more than a significant amount of local exchange service. ITC^DeltaCom can then use the loop-transport combinations that serve the end user to carry any type of traffic, including using them to carry 100 percent interstate access traffic; or

- 8.3.5.1.2 ITC^DeltaCom certifies that it provides local exchange and exchange access service to the end user customer's premises and handles at least one third of the end user customer's local traffic measured as a percent of total end user customer local dialtone lines; and for DS1 circuits and above, at least 50 percent of the activated channels on the loop portion of the loop-transport combination have at least 5 percent local voice traffic individually, and the entire loop facility has at least 10 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the individual DS1 circuits must meet this criteria. The loop-transport combination must terminate at ITC^DeltaCom's collocation arrangement in at least one BellSouth central office. This option does not allow loop-transport combinations to be connected to BellSouth tariffed services; or
- 8.3.5.1.3 ITC^DeltaCom certifies that at least 50 percent of the activated channels on a circuit are used to provide originating and terminating local dial-tone service and at least 50 percent of the traffic on each of these local dial-tone channels is local voice traffic, and that the entire loop facility has at least 33 percent local voice traffic. When a loop-transport combination includes multiplexing, each of the individual DS1 circuits must meet this criteria. This option does not allow loop-transport combinations to be connected to BellSouth's tariffed services. Under this option, collocation is not required. ITC^DeltaCom does not need to provide a defined portion of the end user's local service, but the active channels on any loop-transport combination, and the entire facility, must carry the amount of local exchange traffic specified in this option.
- 8.3.5.2 In addition, there may be extraordinary circumstances where ITC^DeltaCom is providing a significant amount of local exchange service, but does not qualify under any of the three options set forth in Section 8.3.5.1.1, 8.3.5.1.2, 8.3.5.1.3. In such case, ITC^DeltaCom may petition the FCC for a waiver of the local usage options set forth in the June 2, 2000 Order. If a waiver is granted, the Parties shall amend this Agreement within 45 days of ITC^DeltaCom's request to the extent necessary to incorporate the terms of such waiver.
- 8.3.5.3 BellSouth may audit ITC^DeltaCom records to the extent reasonably necessary in order to verify the type of traffic being transmitted over combinations of loop and transport network elements. The audit shall be conducted by a third party independent auditor, and ITC^DeltaCom shall be given thirty days written notice of scheduled audit. Such audit shall occur no more than one time in a calendar year, unless results of an audit find noncompliance with the significant amount of local exchange service requirement. In the event of noncompliance, ITC^DeltaCom shall reimburse BellSouth for the cost of the audit. If, based on its audits,

BellSouth concludes that ITC^DeltaCom is not providing a significant amount of local exchange traffic over the combinations of loop and transport network elements, BellSouth may file a complaint with the appropriate Commission, pursuant to the dispute resolution process as set forth in the Interconnection Agreement. In the event that BellSouth prevails, BellSouth may convert such combinations of loop and transport network elements to special access services and may seek appropriate retroactive reimbursement from ITC^DeltaCom.

- 8.3.5.4 ITC^DeltaCom may convert special access circuits to combinations of loop and transport UNEs pursuant to the terms of this Section and subject to the termination provisions in the applicable special access tariffs, if any.
- 8.3.6 Rates
- 8.3.6.1 Kentucky, Louisiana, Mississippi and South Carolina
- 8.3.6.2 The non-recurring and recurring rates for the EEL Combinations of network elements set forth in 8.3.4 whether Currently Combined or new, are as set forth in Attachment 11.
- 8.3.6.3 On an interim basis, for combinations of loop and transport network elements not set forth in Section 8.3.4, where the elements are not Currently Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 8.3.6.4 To the extent that ITC^DeltaCom seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, ITC^DeltaCom, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.
- 8.3.6.5 <u>All Other States</u>
- 8.3.6.5.1 Subject to Section 8.3.2 and 8.3.3 preceding, for all other states, the non-recurring and recurring rates for the Currently Combined EEL combinations set forth in Section 8.3.4 and other Currently Combined network elements will be the sum of the recurring rates for the individual network elements plus a non recurring charge set forth in Attachment 11.
- 8.3.6.6 Multiplexing

8.3.6.6.1 Where multiplexing functionality is required in connection with loop and transport combinations, such multiplexing will be provided at the rates and on the terms set forth in this Agreement.

8.4 Other Network Element Combinations

8.4.1 In the states of South Carolina, Kentucky, Louisiana and Mississippi, BellSouth shall make available to ITC^DeltaCom, in accordance with Section 8.4.2.1 below: (1) combinations of network elements other than EELs that are Currently Combined; and (2) combinations of network elements other than EELs that are not Currently Combined but that BellSouth ordinarily combines in its network. In all other states, BellSouth shall make available to ITC^DeltaCom, in accordance with Section 8.4.2.2 below, combinations of network elements other than EELs only to the extent such combinations are Currently Combined.

8.4.2 <u>Rates</u>

- 8.4.2.1 Kentucky, Louisiana, Mississippi, and South Carolina
- 8.4.2.1.1 The non-recurring and recurring rates for Other Network Element combinations, whether Currently Combined or new, are as set forth in Attachment 11.
- 8.4.2.1.2 On an interim basis, for Other Network Element combinations where the elements are not Currently Combined but are ordinarily combined in BellSouth's network, the non-recurring and recurring charges for such UNE combinations shall be the sum of the stand-alone non-recurring and recurring charges of the network elements which make up the combination. These interim rates shall be subject to true-up based on the Commission's review of BellSouth's cost studies.
- 8.4.2.1.3 To the extent that ITC^DeltaCom seeks to obtain other combinations of network elements that BellSouth ordinarily combines in its network which have not been specifically priced by the Commission when purchased in combined form, ITC^DeltaCom, at its option, can request that such rates be determined pursuant to the Bona Fide Request/New Business Request (NBR) process set forth in this Agreement.

8.4.2.2 <u>All Other States</u>

8.4.2.2.1 For all other states, the non-recurring and recurring rates for the Other Network Element Combinations that are Currently Combined will be the sum of the recurring rates for the individual network elements plus a non recurring charge set forth in Attachment 11 of this Agreement.

8.5 UNE/Special Access Combinations

8.5.1 Additionally and notwithstanding the above, BellSouth shall make available to ITC^DeltaCom a combination of an unbundled loop and tariffed special access

⁴ States Attachment 2 Clean 12-17-01

interoffice facilities. To the extent ITC^DeltaCom will require multiplexing functionality in connection with such combination, BellSouth will provide access to multiplexing within the central office pursuant to the terms, conditions and rates set forth in its Access Services Tariffs. The tariffed special access interoffice facilities and any associated tariffed services, including but not limited to multiplexing, shall not be eligible for conversion to UNEs as described in Section 8.3.5.

- 8.5.2 Notwithstanding section 8.5.1 above, those "special access combinations" in service as of April 15, 2000 shall be eligible for conversion to UNEs.
- 8.5.3 Rates
- 8.5.3.1 The non-recurring and recurring rates for UNE/Special Access Combinations will be the sum of the unbundled network element loop rates as set forth in Attachment 11 and the interoffice transport rates and multiplexing rates as set forth in the Access Services Tariff.
- 8.6 **Port/Loop Combinations**
- 8.6.1 At ITC^DeltaCom's request, BellSouth shall provide access to combinations of port and loop network elements, as set forth in Section 8.6.3 below, that are Currently Combined in BellSouth's network except as specified in Sections 8.6.1.1 and 8.6.1.2 below.
- 8.6.1.1 BellSouth shall not provide combinations of port and loop network elements on an unbundled basis in locations where, pursuant to FCC rules, BellSouth is not required to provide circuit switching as an unbundled network element.
- In accordance with effective and applicable FCC rules, BellSouth shall not be required to provide circuit switching as an unbundled network element in density Zone 1, as defined in 47 C.F.R. 69.123 as of January 1, 1999 of the Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, MSAs to ITC^DeltaCom if ITC^DeltaCom's customer has 4 or more DS0 equivalent lines.
- 8.6.2 Combinations of port and loop network elements provide local exchange service for the origination or termination of calls. BellSouth shall make available the following loop and port combinations at the terms and at the rates set forth below:
- 8.6.2.1 In Kentucky, Louisiana, Mississippi, and South Carolina, BellSouth shall provide to ITC^DeltaCom combinations of port and loop network elements to ITC^DeltaCom on an unbundled basis regardless of whether or not such combinations are Currently Combined except in those locations where BellSouth is not required to provide circuit switching, as set forth in Section 8.6.1.2 above.

The rates for such combinations shall be the cost based rates set forth in Attachment 11.

- In all other states, BellSouth shall provide to ITC^DeltaCom combinations of port and loop network elements on an unbundled basis if such combinations are Currently Combined, except in those locations where BellSouth is not required to provide unbundled circuit switching, as forth in Sections 8.6.1.1 and 8.6.1.2 above. The rates for such combinations shall be the cost based rates set forth in Attachment 11.
- 8.6.2.3 In all states other than Kentucky, Louisiana, Mississippi, and South Carolina, except in those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 8.6.1.1 and 8.6.1.2, BellSouth shall provide to ITC^DeltaCom combinations of port and loop network elements that are not Currently Combined. The rate for such combinations shall be negotiated by the Parties.
- 8.6.2.4 In those locations where BellSouth is not required to provide unbundled circuit switching, as set forth in Sections 8.6.1.1 and 8.6.1.2, BellSouth shall provide to ITC^DeltaCom combinations of port and loop network elements whether or not such combinations are Currently Combined. The rates for Currently Combined combinations are the market based rates as set forth in Attachment 11. The rates for not Currently Combined combinations shall be negotiated by the Parties.

8.6.3 Combination Offerings

- 8.6.3.1 2-wire voice grade port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 8.6.3.2 2-wire voice grade DID port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 8.6.3.3 2-wire CENTREX port, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 8.6.3.4 2-wire ISDN Basic Rate Interface, voice grade loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.
- 8.6.3.5 2-wire ISDN Primary Rate Interface, DS1 loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

8.6.3.6 4-wire DS1 Trunk port, DS1 Loop, unbundled end office switching, unbundled end office trunk port, common transport per mile per MOU, common transport facilities termination, tandem switching, and tandem trunk port.

9. Switching

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of local and tandem switching.

9.1 **Local Switching**

- 9.1.1 BellSouth shall provide non-discriminatory access to local circuit switching capability, and local tandem switching capability, on an unbundled basis, except as set forth below in Section 9.1.3.3 to ITC^DeltaCom for the provision of a telecommunications service. BellSouth shall provide non-discriminatory access to packet switching capability on an unbundled basis to ITC^DeltaCom for the provision of a telecommunications service only in the limited circumstance described below in Section 9.4.6.
- 9.1.2 Except as otherwise provided herein, BellSouth shall not impose any restrictions on ITC^DeltaCom regarding the use of Switching Capabilities purchased from BellSouth provided such use does not result in demonstrable harm to either the BellSouth network or personnel or the use of the BellSouth network by BellSouth or any other telecommunication carrier.

9.1.3 Local Circuit Switching Capability, including Tandem Switching Capability

9.1.3.1 Definition

Local Circuit Switching Capability is defined as: (A) line-side facilities, which include, but are not limited to, the connection between a loop termination at a main distribution frame and a switch line card; (B) trunk-side facilities, which include, but are not limited to, the connection between trunk termination at a trunk-side cross-connect panel and a switch trunk card; and (C) All features, functions, and capabilities of the switch, which include, but are not limited to: (1) the basic switching function of connecting lines to lines, line to trunks, trunks to lines, and trunks to trunks, as well as the same basic capabilities made available to BellSouth's customers, such as a telephone number, white page listings, and dial tone; and (2) all other features that the switch is capable of providing, including but not limited to customer calling, customer local area signaling service features, and Centrex, as well as any technically feasible customized routing functions provided by the switch; (D) switching provided by remote switching modules.

9.1.3.1.1 Notwithstanding BellSouth's general duty to unbundle local circuit switching, BellSouth shall not be required to unbundle local circuit switching for ITC^DeltaCom when ITC^DeltaCom serves end-users with four (4) or more

voice-grade (DS-0) equivalents or lines in locations served by BellSouth's local circuit switches, which are in the following MSAs: Atlanta, GA; Miami, FL; Orlando, FL; Ft. Lauderdale, FL; Charlotte-Gastonia-Rock Hill, NC; Greensboro-Winston Salem-High Point, NC; Nashville, TN; and New Orleans, LA, and BellSouth has provided non-discriminatory cost based access to the Enhanced Extended Link (EEL) throughout Density Zone 1 as determined by NECA Tariff No. 4 as in effect on January 1, 1999.

- 9.1.3.2 In the event that ITC^DeltaCom orders local circuit switching for a single end user account name at a single physical end user location with four (4) or more two (2) wire voice-grade loops from a BellSouth central office listed in Attachment 11, BellSouth's sole recourse shall be to charge ITC^DeltaCom the market based rate in Attachment 11 for use of the local circuit switching functionality for the affected facilities.
- 9.1.3.3 A featureless port is one that has a line port, switching facilities, and an interoffice port. A featured port is a port that includes all features then capable or a number of then capable features specifically requested by ITC^DeltaCom. Any features that are not currently then capable but are technically feasible through the switch can be requested through the BFR process.
- 9.1.3.4 BellSouth will provide to ITC^DeltaCom customized routing of calls: (i) to a requested directory assistance services platform; (ii) to an operator services platform pursuant to Section 12 of Attachment 2; (iii) for ITC^DeltaCom's PIC'ed toll traffic in a two (2) PIC environment to an alternative OS/DA platform designated by ITC^DeltaCom. ITC^DeltaCom customers may use the same dialing arrangements as BellSouth customers.
- 9.1.3.5 Remote Switching Module functionality is included in Switching Capability. The switching capabilities used will be based on the line side features they support.
- 9.1.3.6 Switching Capability will also be capable of routing local, intraLATA, interLATA, and calls to international customer's preferred carrier; call features (e.g. call forwarding) and Centrex capabilities.
- 9.1.3.7 Where required to do so in order to comply with an effective Commission order, BellSouth will provide to ITC^DeltaCom purchasing local BellSouth switching and reselling BellSouth local exchange service under Attachment 1, selective routing of calls to a requested directory assistance services platform or operator services platform. ITC^DeltaCom customers may use the same dialing arrangements as BellSouth customers, but obtain a ITC^DeltaCom branded service.
- 9.1.4 Technical Requirements

9.1.4.1 The requirements set forth in this Section apply to Local Switching, but not to the Data Switching function of Local Switching. 9.1.4.2 Local Switching shall be equal to or better than the requirements for Local Switching set forth in the applicable industry standard technical references. 9.1.4.3 When applicable, BellSouth shall route calls to the appropriate trunk or lines for call origination or termination. 9.1.4.4 Subject to this section, BellSouth shall route calls on a per line or per screening class basis to (1) BellSouth platforms providing Network Elements or additional requirements (2) Operator Services platforms, (3) Directory Assistance platforms, and (4) Repair Centers. Any other routing requests by ITC^DeltaCom will be made pursuant to the Bona Fide Request/ New Business Request Process as set forth in General Terms and Conditions. 9.1.4.5 BellSouth shall provide unbranded recorded announcements and call progress tones to alert callers of call progress and disposition. 9.1.4.6 BellSouth shall activate service for an ITC^DeltaCom customer or network interconnection on any of the Local Switching interfaces. This includes provisioning changes to change a customer from BellSouth's services to ITC^DeltaCom's services without loss of switch feature functionality as defined in this Agreement. 9.1.4.7 BellSouth shall perform routine testing (e.g., Mechanized Loop Tests (MLT) and test calls such as 105, 107 and 108 type calls) and fault isolation on a mutually agreed upon schedule. 9.1.4.8 BellSouth shall repair and restore any equipment or any other maintainable component that may adversely impact Local Switching. 9.1.4.9 BellSouth shall control congestion points such as those caused by radio station call-ins, and network routing abnormalities. All traffic shall be restricted in a nondiscriminatory manner. 9.1.4.10 BellSouth shall perform manual call trace and permit customer originated call trace. 9.1.4.11 Special Services provided by BellSouth will include the following: 9.1.4.11.1 Telephone Service Prioritization; 9.1.4.11.2 Related services for handicapped; 9.1.4.11.3 Soft dial tone where required by law; and

- 9.1.4.11.4 Any other service required by law.
- 9.1.4.12 BellSouth shall provide Switching Service Point (SSP) capabilities and signaling software to interconnect the signaling links destined to the Signaling Transfer Point Switch (STPS). These capabilities shall adhere to the technical specifications set forth in the applicable industry standard technical references.
- 9.1.4.13 BellSouth shall provide interfaces to adjuncts through Telcordia (formerly BellCore) standard interfaces. These adjuncts can include, but are not limited to, the Service Circuit Node and Automatic Call Distributors.
- 9.1.4.14 BellSouth shall offer Local Switching that provides feature offerings at parity to those provided by BellSouth to itself or any other Party.
- 9.1.4.15 Upon ITC^DeltaCom's request, BellSouth shall provide performance data regarding a customer line, traffic characteristics or other measurable elements to ITC^DeltaCom. ITC^DeltaCom will pay BellSouth for all costs incurred to provide such performance data through the Business Opportunity Request process.

BellSouth shall offer to ITC^DeltaCom all AIN triggers in connection with its SMS/SCE offering which are supported by BellSouth for offering AIN-based services

- 9.1.4.16 Where capacity exists, BellSouth shall assign each ITC^DeltaCom customer line the class of service designated by ITC^DeltaCom (e.g., using line class codes or other switch specific provisioning methods), and shall route directory assistance calls from ITC^DeltaCom customers to ITC^DeltaCom directory assistance operators at ITC^DeltaCom's option.
- 9.1.4.17 Where capacity exists, BellSouth shall assign each ITC^DeltaCom customer line the class of services designated by ITC^DeltaCom (e.g., using line class codes or other switch specific provisioning methods) and shall route operator calls from ITC^DeltaCom customers to ITC^DeltaCom operators at ITC^DeltaCom's option. For example, BellSouth may translate 0- and 0+ intraLATA traffic, and route the call through appropriate trunks to an ITC^DeltaCom Operator Services Position System (OSPS). Calls from Local Switching must pass the ANI-II digits unchanged.
- 9.1.4.18 Local Switching shall be offered in accordance with the technical specifications set forth in the applicable industry standard references.
- 9.1.5 Interface Requirements BellSouth shall provide the following interfaces to loops:
- 9.1.5.1 Standard Tip/Ring interface including loopstart or groundstart, on-hook signaling (e.g., for calling number, calling name and message waiting lamp);
- 9.1.5.2 Coin phone signaling;
- 4 States Attachment 2 Clean 12-17-01

9.1.5.3 Basic Rate Interface ISDN adhering to appropriate Telcordia (formerly BellCore) Technical Requirements; 9.1.5.4 Two-wire analog interface to PBX; 9.1.5.5 Four-wire analog interface to PBX; 9.1.5.6 Four-wire DS1 interface to PBX or customer provided equipment (e.g. computers and voice response systems); 9.1.5.7 Primary Rate ISDN to PBX adhering to ANSI standards Q.931, Q.932 and appropriate Telcordia (formerly BellCore) Technical Requirements; 9.1.5.8 Switched Fractional DS1 with capabilities to configure Nx64 channels (where N = 1 to 24); and 9.1.5.9 Loops adhering to Telcordia (formerly BellCore) TR-NWT-08 and TR-NWT-303 specifications to interconnect Digital Loop Carriers. 9.1.6 BellSouth shall provide access to the following but not limited to: 9.1.6.1 SS7 Signaling Network or Multi-Frequency trunking if requested by ITC^DeltaCom; 9.1.6.2 Interface to ITC^DeltaCom operator services systems or Operator Services through appropriate trunk interconnections for the system; and 9.1.6.3 Interface to ITC^DeltaCom Directory Assistance Services through the ITC^DeltaCom switched network or to Directory Assistance Services through the appropriate trunk interconnections for the system; and 950 access or other ITC^DeltaCom required access to interexchange carriers as requested through appropriate trunk interfaces. 10. Transport, Channelization and Dark Fiber BellSouth agrees to offer access to unbundled transport and dark fiber pursuant to following terms and conditions and at the rates set forth in Attachment 11. 10.1 **Transport** 10.1.1 Interoffice transmission facility network elements include: 10.1.1.1 Dedicated transport, defined as BellSouth's transmission facilities, is dedicated to a particular customer or carrier that provides telecommunications between wire centers or switches owned by BellSouth, or between wire centers and switches

owned by BellSouth and ITC^DeltaCom.

⁴ States Attachment 2 Clean 12-17-01

- Dark Fiber transport, defined as BellSouth's optical transmission facilities without attached signal regeneration, multiplexing, aggregation or other electronics;
- 10.1.1.3 Common (Shared) transport, defined as transmission facilities shared by more than one carrier, including BellSouth, between end office switches, between end office switches and tandem switches, and between tandem switches, in BellSouth's network.
- 10.2 BellSouth shall:
- 10.2.1 Provide ITC^DeltaCom exclusive use of interoffice transmission facilities dedicated to a particular customer or carrier, or shared use of the features, functions, and capabilities of interoffice transmission facilities shared by more than one customer or carrier;
- Provide all technically feasible transmission facilities, features, functions, and capabilities that ITC^DeltaCom could use to provide telecommunications services;
- 10.2.3 Permit, to the extent technically feasible, ITC^DeltaCom to connect such interoffice facilities to equipment designated by ITC^DeltaCom, including but not limited to, ITC^DeltaCom's collocated facilities; and
- 10.2.4 Permit, to the extent technically feasible, ITC^DeltaCom to obtain the functionality provided by BellSouth's digital cross-connect systems in the same manner that BellSouth provides such functionality to interexchange carriers and itself.
- 10.3 <u>Common (Shared) Transport</u>
- 10.3.1 Definition of Common (Shared) Transport
- 10.3.1.1 Common (Shared) Transport is an interoffice transmission path between two BellSouth end-offices, BellSouth end-office and a local tandem, or between two local tandems. Where BellSouth Network Elements are connected by intra-office wiring, such wiring is provided as a part of the Network Elements and is not Common (Shared) Transport. Common (Shared) Transport consists of BellSouth inter-office transport facilities and is unbundled from local switching.
- 10.3.2 Technical Requirements of Common (Shared) Transport
- 10.3.2.1 Common (Shared) Transport provided on DS1 or VT1.5 circuits, shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for Central Office to Central Office ("CO to CO") connections in the appropriate industry standards.
- 10.3.2.2 Common (Shared) Transport provided on DS3 circuits, STS-1 circuits, and higher transmission bit rate circuits, shall, at a minimum, meet the performance,

availability, jitter, and delay requirements specified for CO to CO connections in the appropriate industry standards.							
BellSouth shall be responsible for the engineering, provisioning, and maintenance of the underlying equipment and facilities that are used to provide Common (Shared) Transport.							
At a minimum, Common (Shared) Transport shall meet all of the requirements set forth in the applicable industry standard technical references,(including but not limited to ANSI & Bellcore Standards).							
Dedicated Transport							
<u>Definitions</u>							
Dedicated Transport is defined as BellSouth transmission facilities dedicated to a particular customer or carrier that provide telecommunications between wire centers owned by BellSouth or requesting telecommunications carriers, or between switches owned by BellSouth or requesting telecommunications carriers.							
<u>Unbundled Local Channel</u>							
Unbundled Local Channel is the dedicated transmission path between ITC^DeltaCom's Point of Presence and the BellSouth Serving Wire Center's collocation.							
<u>Unbundled Interoffice Channel.</u>							
Unbundled Interoffice Channel is the dedicated transmission path that provides telecommunication between BellSouth's Serving Wire Centers' collocations.							
BellSouth shall offer Dedicated Transport in each of the following ways:							
As capacity on a shared UNE facility.							
As a circuit (e.g., DS0, DS1, DS3 and OCn) dedicated to ITC^DeltaCom. This circuit shall consist of an Unbundled Local Channel or an Unbundled Interoffice Channel or both.							
When Dedicated Transport is provided it shall include:							
Transmission equipment such as, line terminating equipment, amplifiers, and regenerators;							
Inter-office transmission facilities such as optical fiber, copper twisted pair, and coaxial cable.							

10.4.9 Rates for Dedicated Transport are listed in Attachment 11. For those states that do not contain rates in Attachment 11 the rates in the applicable State Access Tariff will apply as interim rates. When final rates are developed, these interim rates will be subject to true up to the effective date of the Agreement, and the Parties will amend the Agreement to reflect the new rates. 10.4.10 **Technical Requirements** 10.4.10.1 This Section sets forth technical requirements for all Dedicated Transport. 10.4.10.2 When BellSouth provides Dedicated Transport, the entire designated transmission service (e.g., DS0, DS1, DS3) shall be dedicated to ITC^DeltaCom designated traffic. 10.4.10.3 BellSouth shall offer Dedicated Transport in all technologies that become available including, but not limited to, (1) DS0, DS1 and DS3 transport services, and (2) SONET at available transmission bit rates. 10.4.10.4 For DS1 or VT1.5 circuits, Dedicated Transport shall, at a minimum, meet the performance, availability, iitter, and delay requirements specified for Customer Interface to Central Office ("CI to CO") connections in the appropriate industry standards. 10.4.10.5 Where applicable, for DS3, Dedicated Transport shall, at a minimum, meet the performance, availability, jitter, and delay requirements specified for CI to CO connections in the appropriate industry standards. 10.4.10.6 BellSouth shall offer the following interface transmission rates for Dedicated Transport: 10.4.10.6.1 DS0 Equivalent; 10.4.10.6.2 DS1 (Extended SuperFrame - ESF); 10.4.10.6.3 DS3 (signal must be framed); 10.4.10.6.4 SDH (Synchronous Digital Hierarchy) Standard interface rates in accordance with International Telecommunications Union (ITU) Recommendation G.707 and Plesiochronous Digital Hierarchy (PDH) rates per ITU Recommendation G.704. 10.4.10.6.5 When Dedicated Transport is provided, BellSouth shall design it according to BellSouth's network infrastructure to allow for the termination points specified by ITC^DeltaCom.

10.4.11

in the applicable industry technical references.

At a minimum, Dedicated Transport shall meet each of the requirements set forth

- 10.4.11.1 BellSouth Technical References:
- 10.4.11.2 TR-TSY-000191 Alarm Indication Signals Requirements and Objectives, Issue 1, May 1986.
- 10.4.11.3 TR 73501 LightGate[®] Service Interface and Performance Specifications, Issue D, June 1995.
- 10.4.11.4 TR 73525 MegaLink® Service, MegaLink Channel Service & MegaLink Plus Service Interface and Performance Specifications, Issue C, May 1996.
- 10.4.12 Provided that the facility is used to transport a significant amount of local exchange services ITC^DeltaCom shall be entitled to convert existing interoffice transmission facilities (i.e., special access) to the corresponding interoffice transport network element option.

10.5 **Unbundled Channelization**

10.5.1 BellSouth agrees to offer access to Unbundled Channelization when available pursuant to following terms and conditions and at the rates set forth in the Attachment. Channelization will be offered with both the high and the low speed sides to be connected to collocation.

10.5.2 Definition

- Unbundled Channelization (UC) provides the multiplexing capability that will allow a DS1 (1.544 Mbps) or DS3 (44.736 Mbps) or STS-1 Unbundled Network Element (UNE) or collocation cross-connect to be multiplexed or channelized at a BellSouth central office. This can be accomplished through the use of a standalone multiplexer or a digital cross-connect system at the discretion of BellSouth. Once UC has been installed, ITC^DeltaCom can have channels activated on an asneeded basis by having BellSouth connect lower level UNEs via Central Office Channel Interfaces (COCIs).
- 10.5.3 Channelization capabilities will be as follows:
- 10.5.3.1 DS3 Channelization System: An element that channelizes a DS3 signal into 28 DS1s/STS-1s.
- DS1 Channelization System: An element that channelizes a DS1 signal into 24 DS0s.
- 10.5.3.3 Central Office Channel Interfaces (COCI): Elements that can be activated on a channelization system.

- DS1 Central Office Channel Interface elements can be activated on a DS3 Channelization System.
- 10.5.5 Voice Grade and Digital Data Central Office Channel Interfaces can be activated on a DS1 Channelization System.
- 10.5.6 AMI and B8ZS line coding with either Super Frame (SF) and Extended Super Frame (ESF) framing formats will be supported as options
- 10.5.7 COCI will be billed on the lower level UNE order that is interfacing with the UC arrangement and will have to be compatible with those UNEs.
- 10.5.8 <u>Technical Requirements</u>
- In order to assure proper operation with BST provided central office multiplexing functionality, the customer's channelization equipment must adhere strictly to form and protocol standards. Separate standards exist for the multiplex channel bank, for voice frequency encoding, for various signaling schemes, and for subrate digital access.
- 10.5.8.2 DS0 to DS1 Channelization
- The DS1 signal must be framed utilizing the framing structure defined in ANSI T1.107, Digital Hierarchy Formats Specifications and ANSI T1.403.02, DS1 Robbed-bit Signaling State Definitions. DS0 to DS1 Channelization requirements are essential the same as defined in BellSouth Technical Reference 73525, MegaLink® Service, MegaLink® Channel Service, MegaLink® Plus Service, and MegaLink® Light Service Interface and Performance Specification.
- 10.5.8.3 DS1 to DS3 Channelization
- The DS3 signal must be framed utilizing the framing structure define in ANSI T1.107, *Digital Hierarchy Formats Specifications*. DS1 to DS3 Channelization requirements are essentially the same as defined in BellSouth Technical Reference 73501, *LightGate*[®] *Service Interface and Performance Specifications*. The asynchronous M13 multiplex format (combination of M12 and M23 formats) is specified for terminal equipment that multiplexes 28 DS1s into a DS3.
- 10.5.8.4 DS1 to STS Channelization
- 10.5.8.4.1 The STS-1 signal must be framed utilizing the framing structure define in ANSI T1.105, Synchronous Optical Network (SONET) Basic Description Including Multiplex Structure, Rates and Formats and T1.105.02, Synchronous Optical Network (SONET) Payload Mappings. DS1 to STS Channelization requirements are essentially the same as defined in BellSouth Technical Reference TR 73501, LightGate® Service Interface and Performance Specifications

10.6 **Dark Fiber**

10.6.1 <u>Definition</u>

Dark Fiber is optical transmission facilities without attached multiplexing, aggregation or other electronics that connects two points within BellSouth's network. Dark Fiber is unused strands of optical fiber. It may be strands of optical fiber existing in aerial or underground structure. No line terminating elements terminated to such strands to operationalize its transmission capabilities will be available.

10.6.3 Requirements

- BellSouth shall make available Dark Fiber where it exists in BellSouth's network and where, as a result of future building or deployment, it becomes available. If BellSouth has plans to use the fiber within a two-year period, there is no requirement to provide said fiber to ITC^DeltaCom.
- 10.6.3.2 If the requested dark fiber has any lightwave repeater equipment interspliced to it, BellSouth will remove such equipment at ITC^DeltaCom's request subject to time and materials charges.
- 10.6.3.3 ITC^DeltaCom may test the quality of the Dark Fiber to confirm its usability and performance specifications.
- BellSouth shall use its best efforts to provide to ITC^DeltaCom information regarding the location, availability and performance of Dark Fiber within ten (10) business days for a records based answer and twenty (20) business days for a field based answer, after receiving a request from ITC^DeltaCom ("Request"). Within such time period, BellSouth shall send written confirmation of availability of the Dark Fiber ("Confirmation"). From the time of the Request to forty-five (45) days after Confirmation, BellSouth shall hold such requested Dark Fiber for ITC^DeltaCom's use and may not allow any other party to use such media, including BellSouth.
- BellSouth shall use best efforts to make Dark Fiber available to ITC^DeltaCom within thirty (30) business days after it receives written confirmation from ITC^DeltaCom that the Dark Fiber previously deemed available by BellSouth is wanted for use by ITC^DeltaCom. This includes identification of appropriate connection points (e.g., Light Guide Interconnection (LGX) or splice points) to enable ITC^DeltaCom to connect or splice ITC^DeltaCom provided transmission media (e.g., optical fiber) or equipment to the Dark Fiber.
- 10.6.3.6 Dark Fiber shall meet the manufacturer's design specifications.

10.6.3.7 ITC^DeltaCom may splice and test Dark Fiber obtained from BellSouth using ITC^DeltaCom or ITC^DeltaCom designated personnel. BellSouth shall provide appropriate interfaces to allow splicing and testing of Dark Fiber. BellSouth shall provide an excess cable length of 25 feet minimum (for fiber in underground conduit) to allow the uncoiled fiber to reach from the manhole to a splicing van.

11. <u>Tandem Switching</u>

11.1 Definition

Tandem Switching is the function that establishes a communications path between two switching offices through a third switching office (the Tandem switch).

11.2 <u>Technical Requirements</u>

- Tandem Switching shall have the same capabilities or equivalent capabilities as those described in Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90. The requirements for Tandem Switching include, but are not limited to the following:
- 11.2.1.1 Tandem Switching shall provide signaling to establish a tandem connection;
- Tandem Switching will provide screening as jointly agreed to by ITC^DeltaCom and BellSouth;
- 11.2.1.3 Tandem Switching shall provide Advanced Intelligent Network triggers supporting AIN features where such routing is not available from the originating end office switch, to the extent such Tandem switch has such capability;
- Tandem Switching shall provide access to Toll Free number portability database as designated by ITC^DeltaCom;
- 11.2.1.5 Tandem Switching shall provide all trunk interconnections discussed under the "Network Interconnection" section (e.g., SS7, MF, DTMF, DialPulse, PRI-ISDN, DID, and CAMA-ANI (if appropriate for 911));
- Tandem Switching shall provide connectivity to PSAPs where 911 solutions are deployed and the tandem is used for 911; and
- 11.2.1.7 Where appropriate, Tandem Switching shall provide connectivity to transit traffic to and from other carriers.
- Tandem Switching shall accept connections (including the necessary signaling and trunking interconnections) between end offices, other tandems, IXCs, ICOs, CAPs and CLEC switches.

- Tandem Switching shall provide local tandem functionality between two end offices including two offices belonging to different CLECs (e.g., between a CLEC end office and the end office of another CLEC).
- 11.2.4 Tandem Switching shall preserve CLASS/LASS features and Caller ID as traffic is processed.
- Tandem Switching shall record billable events and send them to the area billing centers designated by ITC^DeltaCom. Tandem Switching will provide recording of all billable events as jointly agreed to by ITC^DeltaCom and BellSouth.
- Upon a reasonable request from ITC^DeltaCom, BellSouth shall perform routine testing and fault isolation on the underlying switch that is providing Tandem Switching and all its interconnections. The results and reports of the testing shall be made immediately available to ITC^DeltaCom.
- BellSouth shall maintain ITC^DeltaCom's trunks and interconnections associated with Tandem Switching at least at parity to its own trunks and interconnections.
- 11.2.8 BellSouth shall control congestion points and network abnormalities. All traffic will be restricted in a non discriminatory manner.
- Selective Call Routing through the use of line class codes is not available through the use of tandem switching. Selective Call Routing through the use of line class codes is an end office capability only. Detailed primary and overflow routing plans for all interfaces available within BellSouth switching network shall be mutually agreed to by ITC^DeltaCom and BellSouth.
- Tandem Switching shall process originating toll-free traffic received from ITC^DeltaCom local switch.
- In support of AIN triggers and features, Tandem Switching shall provide SSP capabilities when these capabilities are not available from the Local Switching Network Element, to the extent such Tandem Switch has such capability.
- 11.3 <u>Interface Requirements</u>
- 11.3.1 Tandem Switching shall provide interconnection to the E911 PSAP where the underlying Tandem is acting as the E911 Tandem.
- Tandem Switching shall interconnect, with direct trunks, to all carriers with which BellSouth interconnects.
- BellSouth shall provide all signaling necessary to provide Tandem Switching with no loss of feature functionality.

- Tandem Switching shall interconnect with ITC^DeltaCom's switch, using two-way trunks, for traffic that is transiting via BellSouth network to interLATA or intraLATA carriers. At ITC^DeltaCom's request, Tandem Switching shall record and keep records of traffic for billing.
- 11.3.5 Tandem Switching shall provide an alternate final routing pattern for ITC^DeltaCom traffic overflowing from direct end office high usage trunk groups.
- Tandem Switching shall meet or exceed (i.e., be more favorable to ITC^DeltaCom) each of the requirements for Tandem Switching set forth in the following technical references:
- Bell Communications Research TR-TSY-000540 Issue 2R2, Tandem Supplement, 6/1/90;
- 11.4.2 GR-905-CORE covering CCSNIS;
- 11.4.3 GR-1429-CORE for call management features; and GR-2863-CORE and BellCore GR-2902-CORE covering CCS AIN interconnection

12. Operator Systems

BellSouth agrees to offer access to operator systems pursuant to the terms and conditions following and at the rates set forth in Attachment 11.

12.1 Definition

Operator Systems is the Network Element that provides operator and automated call handling and billing, special services, customer telephone listings and optional call completion services. The Operator Systems, Network Element provides two types of functions: Operator Service functions and Directory Assistance Service functions, each of which are described in detail below.

12.2 Operator Service

12.2.1 Definition

Operator Service provides: (1) operator handling for call completion (for example, collect, third number billing, and calling card calls), (2) operator or automated assistance for billing after the customer has dialed the called number (for example, calling card calls); and (3) special services including but not limited to Busy Line Verification and Emergency Line Interrupt (ELI), Emergency Agency Call, Operator-assisted Directory Assistance, and Rate Quotes.

BellSouth will offer to ITC^DeltaCom Operator Call Processing Access Service BLV/BLVI. Busy Line Verification ("BLV") shall be performed when one Party's Customer requests assistance from the operator bureau of the other Party to

determine if the called line is in use. However, the operator bureau will not complete the call for the Customer initiating the BLV inquiry. Only one BLV attempt will be made per Customer operator bureau call, and a charge shall apply whether or not the called party releases the line.

Busy Line Verification Interrupt ("BLVI") shall be performed when one Party's Customer requests the operator bureau of the other Party to interrupt a telephone call in progress after BLV has occurred. The operator bureau will interrupt the busy line and inform the called party, that there is a call waiting. The operator bureau will only interrupt the call and will not complete the telephone call of the End User initiating the BLVI request. The operator bureau will make only one BLVI attempt per telephone call and the applicable charge applies whether or not the called party releases the line. Each Party's operator bureau shall accept BLV and BLVI inquiries from the operator bureau of the other Party in order to allow transparent provision of BLV/BLVI Traffic between the Parties' networks.

Each Party shall route BLV/BLVI traffic inquiries over separate direct trunks (and not the Local/IntraLATA Trunks) established between the Parties' respective operator bureaus. ITC^DeltaCom will route BLV and BLVI traffic to the BellSouth access tandem. BellSouth will route BLV and BLVI traffic to the ITC^DeltaCom access tandem. Each Party shall compensate the other Party for BLV/BLVI Traffic as set forth in Attachment 11 (Pricing Schedule) to the Agreement.

12.2.2 Requirements

- When ITC^DeltaCom requests BellSouth to provide Operator Services, the following requirements apply:
- 12.2.2.1.1 BellSouth shall complete 0+ and 0- dialed local calls.
- 12.2.2.1.2 BellSouth shall complete 0+ intraLATA toll calls.
- 12.2.2.1.3 BellSouth shall complete calls that are billed to ITC^DeltaCom customer's calling card that can be validated by BellSouth.
- 12.2.2.1.4 BellSouth shall complete person-to-person calls.
- 12.2.2.1.5 BellSouth shall complete collect calls.
- 12.2.2.1.6 BellSouth shall provide the capability for callers to bill to a third party and complete such calls.
- 12.2.2.1.7 BellSouth shall complete station-to-station calls.

- 12.2.2.1.8 BellSouth shall process emergency calls.
- 12.2.2.1.9 BellSouth shall process Busy Line Verify and Emergency Line Interrupt requests.
- 12.2.2.1.10 BellSouth shall process emergency call trace, as they do for their Customers prior to the Effective Date. Call must originate from a 911 provider.
- 12.2.2.1.11 BellSouth shall process operator-assisted directory assistance calls.
- BellSouth shall adhere to equal access requirements, providing ITC^DeltaCom local customers the same IXC access as provided to BellSouth customers.
- 12.2.2.3 BellSouth shall exercise at least the same level of fraud control in providing Operator Service to ITC^DeltaCom that BellSouth provides for its own operator service.
- 12.2.2.4 BellSouth shall perform Billed Number Screening when handling Collect, Personto-Person, and Billed-to-Third-Party calls.
- 12.2.2.5 BellSouth shall direct customer account and other similar inquiries to the customer service center designated by ITC^DeltaCom.
- BellSouth shall provide a feed of customer call records in "EMI" format to ITC^DeltaCom in accordance with ODUF standards specified in Attachment 7.

12.2.3 Interface Requirements

With respect to Operator Services for calls that originate on local switching capability provided by or on behalf of ITC^DeltaCom, the interface requirements shall conform to the then current established system interface specifications for the platform used to provide Operator Service and the interface shall conform to industry standards.

12.3 Directory Assistance Service

12.3.1 Definition

Directory Assistance Service provides local customer telephone number listings with the option to complete the call at the callers direction separate and distinct from local switching.

12.3.2 Requirements

12.3.2.1 Directory Assistance Service shall provide up to two listing requests per call. If available and if requested by ITC^DeltaCom's customer, BellSouth shall provide caller-optional directory assistance call completion service at rates contained in this Attachment to one of the provided listings, equal to that which BellSouth provides

its customers. If not available, ITC^DeltaCom may request such requirement pursuant to the Bona Fide Request Process of Attachment 9.

- 12.3.2.2 <u>Directory Assistance Service Updates</u>
- 12.3.2.2.1 BellSouth shall update customer listings changes daily. These changes include:
- 12.3.2.2.1.1 New customer connections: BellSouth will provide service to ITC^DeltaCom that is equal to the service it provides to itself and its customers;
- 12.3.2.2.1.2 Customer disconnections: BellSouth will provide service to ITC^DeltaCom that is equal to the service it provides to itself and its customers; and
- 12.3.2.2.1.3 Customer address changes: BellSouth will provide service to ITC^DeltaCom that is equal to the service it provides to itself and its customers;
- 12.3.2.3 These updates shall also be provided for non-listed and non-published numbers for use in emergencies.
- 12.4 <u>Branding for Operator Call Processing and Directory Assistance</u>
- The BellSouth Operator Systems Branding Feature provides a definable announcement to ITC^DeltaCom end users using Directory Assistance (DA)/Operator Call Processing (OCP) prior to placing them in queue or connecting them to an available operator or automated operator system. This feature allows ITC^DeltaCom to have its calls custom branded with ITC^DeltaCom name on whose behalf BellSouth is providing Directory Assistance and/or Operator Call Processing. Rates for Custom Branding, Operator Call Process and Directory Assistance are set forth in Attachment 11.
- BellSouth offers four service levels of branding to ITC^DeltaCom when ordering Directory Assistance and/or Operator Call Processing.
- 12.4.2.1 Service Level 1 BellSouth Branding
- 12.4.2.2 Service Level 2 Unbranded
- 12.4.2.3 Service Level 3 Custom Branding
- 12.4.2.4 Service Level 4 Self Branding (applicable only to ITC^DeltaCom for Resale or use with an Unbundled Port when routing to an operator service provider other than BellSouth).
- 12.4.3 For Resellers and Use with an Unbundled Port
- 12.4.3.1 BellSouth Branding is the Default Service Level.

- 12.4.3.2 Unbranding, Custom Branding, and Self Branding require ITC^DeltaCom to order selective routing for each originating BellSouth end office identified by ITC^DeltaCom. Rates for Selective Routing are set forth in Attachment 11.
- 12.4.3.3 Customer Branding and Self Branding require ITC^DeltaCom to order dedicated trunking from each BellSouth end office identified by ITC^DeltaCom, to either the BellSouth Traffic Operator Position System (TOPS) or ITC^DeltaCom Operator Service Provider. Rates for trunks are set forth in applicable BellSouth tariffs.
- 12.4.3.4 Unbranding Unbranded Directory Assistance and/or Operator Call Processing calls ride common trunk groups provisioned by BellSouth from those end offices identified by ITC^DeltaCom to the BellSouth TOPS. These calls are routed to "No Announcement."
- 12.4.4 For Facilities Based Carriers
- 12.4.4.1 All Service Levels require ITC^DeltaCom to order dedicated trunking from their end office(s) point of interface to the BellSouth TOPS Switches. Rates for trunks are set forth in applicable BellSouth tariffs.
- 12.4.4.2 Customized Branding includes charges for the recording of the branding announcement and the loading of the audio units in each TOPS Switch, IVS and NAV equipment for which ITC^DeltaCom requires service.

Directory Assistance customized branding uses:

- the recording of the name;
- the front-end loading of the Digital Recorded Announcement Machine (DRAM) in each TOPS switch.

Operator Call Processing customized branding uses:

- the recording of the name;
- the front-end loading of the DRAM in the TOPS Switch;
- the back-end loading in the audio units in the Automated Alternate Billing System (AABS) in the Interactive Voice Subsystem (IVS);
- the 0- automation loading for the audio units in the Enhanced Billing and Access Service (EBAS) in the Network Applications Vehicle (NAV).
- BellSouth will provide at ITC^DeltaCom's option, unbundled local BellSouth switching and resold BellSouth local exchange service, with selective routing of calls to a requested directory assistance services platform or operator services platform. ITC^DeltaCom customers may use the same dialing arrangements as BellSouth customers, but obtain a ITC^DeltaCom branded service.
- 12.5 Directory Assistance Database Service (DADS)

- BellSouth shall make its Directory Assistance Database Service (DADS) available solely for the expressed purpose of providing Directory Assistance type services to ITC^DeltaCom end users. The term "end user" denotes any entity which obtains Directory Assistance type services for its own use from a DADS customer. Directory Assistance type service is defined as Voice Directory Assistance (DA Operator assisted and Electronic Directory Assistance (Data System assisted)). ITC^DeltaCom agrees that Directory Assistance Database Service (DADS) will not be used for any purpose which violates federal or state laws, statutes, regulatory orders or tariffs. Except for the permitted users, ITC^DeltaCom agrees not to disclose DADS to others and shall provide due care in providing for the security and confidentiality of DADS. Further, ITC^DeltaCom authorizes the inclusion of ITC^DeltaCom Subscriber listings in the BellSouth Directory Assistance products.
- BellSouth shall provide ITC^DeltaCom initially with daily updates which reflect all listing change activity occurring since ITC^DeltaCom's most recent update via magnetic tape, and subsequently using electronic connectivity such as Network Data Mover to be developed mutually by ITC^DeltaCom and BellSouth. ITC^DeltaCom agrees to assume the costs associated with CONNECT: Direct TM connectivity, which will vary depending upon volume and mileage.
- BellSouth will require approximately one month after receiving an order to prepare the Base File. BellSouth will provide daily updates which will reflect all listing change activity occurring since ITC^DeltaCom most recent update. BellSouth shall provide updates to ITC^DeltaCom on a Business, Residence, or combined Business and Residence basis. ITC^DeltaCom agrees that the updates shall be used solely to keep the information current. Delivery of Daily Updates will commence the day after ITC^DeltaCom receives the Base File.
- BellSouth is authorized to include ITC^DeltaCom Subscriber List Information in its Directory Assistance Database Service (DADS) and its Directory Publishers Database Service (DPDS). Any other use by BellSouth of ITC^DeltaCom Subscriber List Information is not authorized and with the exception of a request for DADS or DPDS, BellSouth shall refer any request for such information to ITC^DeltaCom.
- 12.5.5 Rates for DADS are as set forth in Attachment 11 and to the extent appropriate in BellSouth General Subscriber Services Tariff A38.1.
- 12.6 Direct Access to Directory Assistance Service
- Direct Access to Directory Assistance Service (DADAS) will provide ITC^DeltaCom's directory assistance operators with the ability to search all available BellSouth's subscriber listings using the Directory Assistance search format. Subscription to DADAS will allow ITC^DeltaCom to utilize its own switch, operator workstations and optional audio subsystems.

- BellSouth will provide DADAS from its DA location. ITC^DeltaCom will access the DADAS system via a telephone company provided point of availability. ITC^DeltaCom has the responsibility of providing the physical links required to connect to the point of availability. These facilities may be purchased from the telephone company as rates and charges billed separately from the charges associated with this offering.
- A specified interface to each ITC^DeltaCom subsystem will be provided by BellSouth. Interconnection between ITC^DeltaCom system and a specified BellSouth location will be pursuant to the use of ITC^DeltaCom owned or ITC^DeltaCom leased facilities and shall be appropriate sized based upon the volume of queries being generated by ITC^DeltaCom.
- 12.6.4 The specifications for the three interfaces necessary for interconnection are available in the following documents:
- 12.6.4.1 DADAS to Subscriber Operator Position System—Northern Telecom Document CSI-2300-07; Universal Gateway/ Position Message Interface Format Specification
- DADAS to Subscriber Switch—Northern Telecom Document Q210-1 Version A107; NTDMS/CCIDAS System Application Protocol; and AT&T Document 250-900-535 Operator Services Position System Listing Service and Application Call Processing Data Link Interface Specification
- 12.6.4.3 DADAS to Audio Subsystem (Optional)—Directory One Call Control to Audio Response Unit system interface specifications are available through Northern Telecom as a licensed access protocol—Northern Telecom Document 355-004424 and Gateway/Interactive Voice subsystem Protocol Specification
- 12.6.5 Rates for DADAS are as set forth in Attachment 11 and to the extent appropriate in the BellSouth FCC Tariff No. 1.

13. Signaling

Unbundled signaling and access to BellSouth's signaling databases shall be provided pursuant to this Attachment and Attachment 3 Section 4.8 subject to compatibility testing and at the rates set forth in Attachment 11. BellSouth may provide mediated access to BellSouth signaling systems and databases. Available signaling elements include signaling links, signal transfer points and service control points. Signaling functionality will be available with both A-link and B-link connectivity.

13.1 Definition of Signaling Link Transport

Signaling Link Transport is a set of two or four dedicated 56 Kbps. transmission paths between CLEC-designated Signaling Points of Interconnection (SPOI) that provides appropriate physical diversity.

13.2	Technical Requirements
13.2.1	Signaling Link Transport shall consist of full duplex mode 56 kbps transmission paths.
13.2.2	Of the various options available, Signaling Link Transport shall perform in the following two ways:
13.2.2.1	As an "A-link" which is a connection between a switch or SCP and a home Signaling Transfer Point Switch (STP) pair; and
13.2.2.2	As a "B-link" which is a connection between two STP pairs in different company networks (e.g., between two STP pairs for two Competitive Local Exchange Carriers (CLECs)).
13.2.3	Signaling Link Transport shall consist of two or more signaling link layers as follows:
13.2.3.1	An A-link layer shall consist of two links.
13.2.3.2	A B-link layer shall consist of four links.
13.2.4	A signaling link layer shall satisfy a performance objective such that:
13.2.4.1	There shall be no more than two minutes down time per year for an A-link layer; and
13.2.4.2	There shall be negligible (less than 2 seconds) down time per year for a B-link layer.
13.2.5	A signaling link layer shall satisfy interoffice and intraoffice diversity of facilities and equipment, such that:
13.2.5.1	No single failure of facilities or equipment causes the failure of both links in an A-link layer (i.e., the links should be provided on a minimum of two separate physical paths end-to-end); and
13.2.5.2	No two concurrent failures of facilities or equipment shall cause the failure of all four links in a B-link layer (i.e., the links should be provided on a minimum of three separate physical paths end-to-end).
13.3	Interface Requirements

There shall be a DS1 (1.544 Mbps) interface at the ITC^DeltaCom-designated SPOIs. Each 56 kbps transmission path shall appear as a DS0 channel within the DS1 interface.

14. <u>Signaling Transfer Points (STPs)</u>

14.1 Definition

Signaling Transfer Points is a signaling network function that includes all of the capabilities provided by the signaling transfer point switches (STPs) and their associated signaling links which enable the exchange of SS7 messages among and between switching elements, database elements and signaling transfer point switches

14.2.1 <u>Technical Requirements</u>

- 14.2.1.1 STPs shall provide access to Network Elements connected to BellSouth SS7 network. These include:
- 14.2.1.2 BellSouth Local Switching or Tandem Switching;
- 14.2.1.3 BellSouth Service Control Points/DataBases;
- 14.2.1.4 Third-party local or tandem switching;
- 14.2.1.5 Third-party-provided STPs.
- 14.2.2 The connectivity provided by STPs shall fully support the functions of all other Network Elements connected to BellSouth SS7 network. This explicitly includes the use of BellSouth SS7 network to convey messages which neither originate nor terminate at a signaling end point directly connected to BellSouth SS7 network (*i.e.*, transient messages). When BellSouth SS7 network is used to convey transient messages, there shall be no alteration of the Integrated Services Digital Network User Part (ISDNUP) or Transaction Capabilities Application Part (TCAP) user data that constitutes the content of the message.
- 14.2.3 If a BellSouth tandem switch routes calling traffic, based on dialed or translated digits, on SS7 trunks between an ITC^DeltaCom local switch and third party local switch, BellSouth SS7 network shall convey the TCAP messages that are necessary to provide Call Management features (Automatic Callback, Automatic Recall, and Screening List Editing) between ITC^DeltaCom local STPs and the STPs that provide connectivity with the third party local switch, even if the third party local switch is not directly connected to BellSouth STPs.
- 14.2.4 STPs shall provide all functions of the MTP as defined in Bellcore ANSI Interconnection Requirements. This includes:

- 14.2.4.1 Signaling Data Link functions, as defined in Bellcore ANSI Interconnection Requirements,
- 14.2.4.2 Signaling Link functions, as defined in Bellcore ANSI Interconnection Requirements, and
- 14.2.4.3 Signaling Network Management functions, as defined in Bellcore ANSI Interconnection Requirements.
- STPs shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as defined in Bellcore ANSI Interconnection Requirements. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. In cases where the destination signaling point is a ITC^DeltaCom or third party local or tandem switching system directly connected to BellSouth SS7 network, BellSouth shall perform final GTT of messages to the destination and SCCP Subsystem Management of the destination. In all other cases, BellSouth shall perform intermediate GTT of messages to a gateway pair of STPs in an SS7 network connected with BellSouth SS7 network, and shall not perform SCCP Subsystem Management of the destination. If BellSouth performs final GTT to a ITC^DeltaCom database, then ITC^DeltaCom agrees to provide BellSouth with the Destination Point Code for the ITC^DeltaCom database.
- 14.2.6 STPs shall provide on a non-discriminatory basis all functions of the OMAP commonly provided by STPs, as specified in the reference in Section 12.4.5 of this Attachment. All OMAP functions will be on a "where available" basis and can include:
- 14.2.6.1 MTP Routing Verification Test (MRVT) and
- 14.2.6.2 SCCP Routing Verification Test (SRVT).
- In cases where the destination signaling point is a BellSouth local or tandem switching system or database, or is an ITC^DeltaCom or third party local or tandem switching system directly connected to the BellSouth SS7 network, STPs shall perform MRVT and SRVT to the destination signaling point. In all other cases, STPs shall perform MRVT and SRVT to a gateway pair of STPs in an SS7 network connected with the BellSouth SS7 network. This requirement shall be superseded by the specifications for Internetwork MRVT and SRVT if and when these become approved ANSI standards and available capabilities of BellSouth STPs, and if mutually agreed upon by ITC^DeltaCom and BellSouth.
- 14.2.8 STPs shall be on parity with BellSouth.
- 14.2.9 SS7 Advanced Intelligent Network (AIN) Access

- When technically feasible and upon request by ITC^DeltaCom, SS7 Access shall be made available in association with unbundled switching. SS7 AIN Access is the provisioning of AIN 0.1 triggers in an equipped BellSouth local switch and interconnection of the BellSouth SS7 network with the ITC^DeltaCom SS7 network to exchange TCAP queries and responses with an ITC^DeltaCom SCP.
- 14.2.9.2 SS7 AIN Access shall provide ITC^DeltaCom SCP access to BellSouth local switch in association with unbundled switching via interconnection of BellSouth SS7 and ITC^DeltaCom SS7 Networks. BellSouth shall offer SS7 access through its STPs. If BellSouth requires a mediation device on any part of its network specific to this form of access, BellSouth must route its messages in the same manner. The interconnection arrangement shall result in the BellSouth local switch recognizing the ITC^DeltaCom SCP as at least at parity with BellSouth's SCP's in terms of interfaces, performance and capabilities.

14.3 <u>Interface Requirements</u>

- 14.3.1 BellSouth shall provide the following STPs options to connect ITC^DeltaCom or ITC^DeltaCom-designated local switching systems or STPs to BellSouth SS7 network:
- 14.3.1.1 An A-link interface from ITC^DeltaCom local switching systems; and,
- 14.3.1.2 A B-link interface from ITC^DeltaCom local STPs.
- Each type of interface shall be provided by one or more sets (layers) of signaling links.
- The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling for interconnecting ITC^DeltaCom local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and ITC^DeltaCom will work jointly to establish mutually acceptable SPOIs.
- BellSouth CO shall provide intraoffice diversity between the SPOIs and BellSouth STPs, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP. BellSouth and ITC^DeltaCom will work jointly to establish mutually acceptable SPOIs.
- 14.3.5 BellSouth shall provide MTP and SCCP protocol interfaces that shall conform to all sections relevant to the MTP or SCCP in the following specifications:

- 14.3.5.1 Bellcore GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 14.3.5.2 Bellcore GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

14.3.6 <u>Message Screening</u>

- 14.3.6.1 BellSouth shall set message screening parameters so as to accept valid messages from ITC^DeltaCom local or tandem switching systems destined to any signaling point within BellSouth's SS7 network where the ITC^DeltaCom switching system has a legitimate signaling relation.
- 14.3.6.2 BellSouth shall set message screening parameters so as to pass valid messages from ITC^DeltaCom local or tandem switching systems destined to any signaling point or network accessed through BellSouth's SS7 network where the ITC^DeltaCom switching system has a legitimate signaling relation.
- 14.3.6.3 BellSouth shall set message screening parameters so as to accept and pass/send valid messages destined to and from ITC^DeltaCom from any signaling point or network interconnected through BellSouth's SS7 network where the ITC^DeltaCom SCP has a legitimate signaling relation.
- STPs shall be equal to or better than all of the requirements for STPs set forth in the following technical references:
- 14.4.1 ANSI T1.111-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP);
- 14.4.2 ANSI T1.111A-1994 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP) Supplement;
- 14.4.3 ANSI T1.112-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Signaling Connection Control Part (SCCP);
- 14.4.4 ANSI T1.115-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Monitoring and Measurements for Networks;
- 14.4.5 ANSI T1.116-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Operations, Maintenance and Administration Part (OMAP);
- 14.4.6 ANSI T1.118-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Intermediate Signaling Network Identification (ISNI);

⁴ States Attachment 2 Clean 12-17-01

- 14.4.7 Bellcore GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP); and
- 14.4.8 Bellcore GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

15. <u>Service Control Points/DataBases</u>

15.1 <u>Definition</u>

- Databases provide the functionality for storage of, access to, and manipulation of information required to offer a particular service and/or capability. Databases include, but are not limited to: Local Number Portability, Toll Free Number Database, Automatic Location Identification/Data Management System, access to Service Creation Environment and Service Management System (SCE/SMS) application databases and Directory Assistance.
- A Service Control Point (SCP) is a specific type of Database functionality deployed in a Signaling System 7 (SS7) network that executes service application logic in response to SS7 queries sent to it by a switching system also connected to the SS7 network. Service Management Systems provide operational interfaces to allow for provisioning, administration and maintenance of subscriber data and service application data stored in SCPs.

15.2 Technical Requirements for SCPs/Databases

Requirements for SCPs/Databases within this section address storage of information, access to information (e.g. signaling protocols, response times), and administration of information (e.g., provisioning, administration, and maintenance). All SCPs/Databases shall be provided to ITC^DeltaCom in accordance with the following requirements.

- BellSouth shall provide physical access to SCPs through the SS7 network and protocols with TCAP as the application layer protocol.
- BellSouth shall provide physical interconnection to databases via industry standard interfaces and protocols (e.g. SS7, ISDN and X.25).
- 15.2.3 The reliability of interconnection options shall be consistent with requirements for diversity and survivability.

15.2.4 Database Availability

Call processing databases shall have a maximum unscheduled availability of 30 minutes per year. Unavailability due to software and hardware upgrades shall be

scheduled during minimal usage periods and only be undertaken upon proper notification to providers which might be impacted. Any downtime associated with the provision of call processing related databases will impact all service providers, including BellSouth, equally.

The operational interface provided by BellSouth shall complete Database transactions (i.e., add, modify, delete) for ITC^DeltaCom customer records stored in BellSouth databases within 3 days, or sooner where BellSouth provisions its own customer records within a shorter interval.

15.3 <u>Local Number Portability Database</u>

15.3.1 Definition

The Permanent Number Portability (PNP) database supplies routing numbers for calls involving numbers that have been ported from one local service provider to another. PNP is currently being worked in industry forums. The results of these forums will dictate the industry direction of PNP. BellSouth agrees to provide access to the PNP database at rates, terms and conditions as set forth by BellSouth and in accordance with an effective FCC or Commission directive.

15.4 Line Information Database (LIDB)

15.4.1 Definition

The Line Information Database (LIDB) is a transaction-oriented database accessible through Common Channel Signaling (CCS) networks. It contains records associated with customer Line Numbers and Special Billing Numbers. LIDB accepts queries from other Network Elements and provides appropriate responses. The query originator need not be the owner of LIDB data. LIDB queries include functions such as screening billed numbers that provides the ability to accept Collect or Third Number Billing calls and validation of Telephone Line Number based non-proprietary calling cards. The interface for the LIDB functionality is the interface between BellSouth CCS network and other CCS networks. LIDB also interfaces to administrative systems.

- 15.4.1.1 BellSouth will store in its LIDB only records relating to service in the BellSouth region.
- Prior to the availability of a long-term solution for LNP, BellSouth shall enable ITC^DeltaCom to store in BellSouth's LIDB any customer, Line Number or Special Billing Number record, whether ported or not, for which the ITC^DeltaCom dedicated NPA-NXX or RAO-0/1XX Group is supported by that LIDB.
- Subsequent to the availability of a long-term solution for LNP, BellSouth shall enable ITC^DeltaCom to store in BellSouth's LIDB any customer, Line Number

or Special Billing Number record, whether ported or not, regardless of the number's dedicated NPA-NXX or RAO [NXX]-0/1XX.

15.4.2 <u>Technical Requirements</u>

BellSouth will offer to ITC^DeltaCom any additional capabilities that are developed for LIDB during the life of this Agreement.

- BellSouth shall process ITC^DeltaCom's Customer records in LIDB at least at parity with BellSouth customer records, with respect to other LIDB functions. BellSouth shall indicate to ITC^DeltaCom what additional functions (if any) are performed by LIDB in the BellSouth network.
- 15.4.2.2 Within two (2) weeks after a request by ITC^DeltaCom, BellSouth shall provide ITC^DeltaCom with a list of the customer data items which ITC^DeltaCom would have to provide in order to support each required LIDB function. The list shall indicate which data items are essential to LIDB function, and which are required only to support certain services. For each data item, the list shall show the data formats, the acceptable values of the data item and the meaning of those values.
- BellSouth shall provide LIDB systems for which operating deficiencies that would result in calls being blocked, shall not exceed 30 minutes per year.
- BellSouth shall provide LIDB systems for which operating deficiencies that would not result in calls being blocked shall not exceed 12 hours per year.
- 15.4.2.5 BellSouth shall provide LIDB systems for which the LIDB function shall be in overload no more than 12 hours per year.
- All additions, updates and deletions of ITC^DeltaCom data to the LIDB shall be solely at the direction of ITC^DeltaCom. Such direction from ITC^DeltaCom will not be required where the addition, update or deletion is necessary to perform standard fraud control measures (e.g., calling card auto-deactivation).
- BellSouth shall provide priority updates to LIDB for ITC^DeltaCom data upon ITC^DeltaCom's request (e.g., to support fraud detection), via password-protected telephone card, facsimile, or electronic mail within one hour of notice from the established BellSouth contact.
- BellSouth shall provide LIDB systems such that no more than 0.01% of ITC^DeltaCom customer records will be missing from LIDB, as measured by ITC^DeltaCom audits. BellSouth will audit ITC^DeltaCom records in LIDB against DBAS to identify record mismatches and provide this data to a designated ITC^DeltaCom contact person to resolve the status of the records and BellSouth will update system appropriately. BellSouth will refer record of mis-matches to ITC^DeltaCom within one business day of audit. Once reconciled records are received back from ITC^DeltaCom, BellSouth will update LIDB the same business day if less than 500 records are received before 1:00PM Central Time. If more

than 500 records are received, BellSouth will contact ITC^DeltaCom to negotiate a time frame for the updates, not to exceed three business days.

- BellSouth shall perform backup and recovery of all of ITC^DeltaCom's data in LIDB including sending to LIDB all changes made since the date of the most recent backup copy, in at least the same time frame BellSouth performs backup and recovery of BellSouth data in LIDB for itself. Currently, BellSouth performs backups of the LIDB for itself on a weekly basis and when a new software release is scheduled, a backup is performed prior to loading the new release.
- 15.4.2.10 BellSouth shall provide ITC^DeltaCom with LIDB reports of data which are missing or contain errors, as well as any misroute errors, within a reason time period as negotiated between ITC^DeltaCom and BellSouth.
- BellSouth shall prevent any access to or use of ITC^DeltaCom data in LIDB by BellSouth personnel that are outside of established administrative and fraud control personnel, or by any other party that is not authorized by ITC^DeltaCom in writing.
- BellSouth shall provide ITC^DeltaCom performance of the LIDB Data Screening function, which allows a LIDB to completely or partially deny specific query originators access to LIDB data owned by specific data owners, for Customer Data that is part of an NPA-NXX or RAO-0/1XX wholly or partially owned by ITC^DeltaCom at least at parity with BellSouth Customer Data. BellSouth shall obtain from ITC^DeltaCom the screening information associated with LIDB Data Screening of ITC^DeltaCom data in accordance with this requirement. BellSouth currently does not have LIDB Data Screening capabilities. When such capability is available, BellSouth shall offer it to ITC^DeltaCom under the Bona Fide Request process of Attachment 9.
- 15.4.2.13 BellSouth shall accept queries to LIDB associated with ITC^DeltaCom customer records, and shall return responses in accordance with industry standards.
- 15.4.2.14 BellSouth shall provide mean processing time at the LIDB within 0.50 seconds under normal conditions as defined in industry standards.
- 15.4.2.15 BellSouth shall provide processing time at the LIDB within 1 second for 99% of all messages under normal conditions as defined in industry standards.
- 15.4.3 <u>Interface Requirements</u>

BellSouth shall offer LIDB in accordance with the requirements of this subsection.

- 15.4.3.1 The interface to LIDB shall be in accordance with the technical references contained within.
- 15.4.3.2 The CCS interface to LIDB shall be the standard interface described herein.

15.4.3.3 The LIDB Data Base interpretation of the ANSI-TCAP messages shall comply with the technical reference herein. Global Title Translation shall be maintained in the signaling network in order to support signaling network routing to the LIDB.

15.5 <u>BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service</u>

All of the negotiated rates, terms and conditions set forth in this Section pertain to the provision of 8XX Access Ten Digit Screening Services.

- 15.5.1 BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database
- The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service database (herein known as 8XX SCP) is a SCP that contains customer record information and functionality to provide call-handling instructions for 8XX calls. The 8XX SCP IN software stores data downloaded from the national SMS and provides the routing instructions in response to queries from the SSP or tandem. The BellSouth SWA 8XX Toll Free Dialing Ten Digit Screening Service (herein know as 8XX TFD), utilizes the 8XX SCP to provide identification and routing of the 8XX calls, based on the ten digits dialed. 8XX TFD is provided with or without POTS number delivery, dialing number delivery, and other optional complex features as selected by ITC^DeltaCom. BellSouth shall provide 8XX TFD in accordance with the following:

15.5.3 <u>Technical Requirements</u>

- BellSouth shall provide ITC^DeltaCom with access to the 8XX record information located in the 8XX SCP. The 8XX SCP contains current records as received from the national SMS and will provide for routing 8XX originating calls based on the dialed ten digit 8XX number.
- The 8XX SCP is designated to receive and respond to queries using the American National Standard Specification of Signaling System Seven (SS7) protocol. The 8XX SCP shall determine the carrier identification based on all ten digits of the dialed number and route calls to the carrier, POTS number, dialing number and/or other optional feature selected by ITC^DeltaCom.
- The SCP shall also provide, at ITC^DeltaCom's option, such additional feature as described in SR-TSV-002275 (BOC Notes on BellSouth Networks, SR-TSV-002275, Issue 2, (Telcordia (formerly BellCore), April 1994)) as are available to BellSouth. These may include but are not limited to:
- 15.5.7 Network Management;
- 15.5.8 Customer Sample Collection; and
- 15.5.9 Service Maintenance.
- 15.6 Automatic Location Identification/Data Management System (ALI/DMS)

⁴ States Attachment 2 Clean 12-17-01

The ALI/DMS Database contains customer information (including name, address, telephone information, and sometimes special information from the local service provider or customer) used to determine to which Public Safety Answering Point (PSAP) to route the call. The ALI/DMS database is used to provide more routing flexibility for E911 calls than Basic 911. BellSouth shall provide the Emergency Services Database in accordance with the following:

15.6.1 <u>Technical Requirements</u>

- 15.6.1.1 BellSouth shall offer ITC^DeltaCom a data link to the ALI/DMS database or permit ITC^DeltaCom to provide its own data link to the ALI/DMS database. BellSouth shall provide error reports from the ALI/DMS database to ITC^DeltaCom immediately after ITC^DeltaCom inputs information into the ALI/DMS database. Alternately, ITC^DeltaCom may utilize BellSouth, to enter customer information into the database on a demand basis, and validate customer information on a demand basis.
- 15.6.1.2 The ALI/DMS database shall contain the following customer information:
- 15.6.1.2.1 Name:
- 15.6.1.2.2 Address;
- 15.6.1.2.3 Telephone number; and
- Other information as appropriate (e.g., whether a customer is blind or deaf or has another disability).
- When the BellSouth is responsible for administering the ALI/DMS database in its entirety, ported number NXXs entries for the ported numbers should be maintained unless ITC^DeltaCom requests otherwise and shall be updated if ITC^DeltaCom requests, provided ITC^DeltaCom supplies BellSouth with the updates.
- When Remote Call Forwarding (RCF) is used to provide number portability to the local customer and a remark or other appropriate field information is available in the database, the shadow or "forwarded-to" number and an indication that the number is ported shall be added to the customer record.
- 15.6.1.5 If BellSouth is responsible for configuring PSAP features (for cases when the PSAP or BellSouth supports an ISDN interface) it shall ensure that CLASS Automatic Recall (Call Return) is not used to call back to the ported number. Although BellSouth currently does not have ISDN interface, BellSouth agrees to comply with this requirement once ISDN interfaces are in place.
- 15.6.2 <u>Interface Requirements</u>

The interface between the E911 Switch or Tandem and the ALI/DMS database for ITC^DeltaCom customers shall meet industry standards.

15.7 <u>Directory Assistance Database</u>

BellSouth shall make its directory assistance database available to ITC^DeltaCom in order to allow ITC^DeltaCom to provide its customers with the same directory assistance telecommunications services BellSouth provides to BellSouth customers. BellSouth shall provide ITC^DeltaCom with an initial feed via magnetic tape and daily update initially via magnetic tape and subsequently via an electronic gateway to be developed mutually by ITC^DeltaCom and BellSouth of customer address and number changes. Directory Assistance Services must provide both the ported and ITC^DeltaCom telephone numbers to the extent available in BellSouth's database assigned to a customer. Privacy indicators must be properly identified to assure the non-published numbers are accurately identified.

15.8 Calling Name (CNAM) Database Service.

ITC^DeltaCom may provide to its account manager a written request to enter into a CNAM agreement with BellSouth. If ITC^DeltaCom is interested in requesting CNAM with volume and term pricing, ITC^DeltaCom must contact its account manager and specifically request a CNAM volume and term agreement.

- SCPs/Databases shall be equal to or better than all of the requirements for SCPs/Databases set forth in the following technical references:
- 15.9.1 GR-246-CORE, Bell Communications Research Specification of Signaling System Number 7, ISSUE 1 (Bellcore, December 199);
- 15.9.2 GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP). (Bellcore, March 1994);
- 15.9.3 GR-954-CORE, CCS Network Interface Specification (CCSNIS) Supporting Line Information Database (LIDB) Service 6, Issue 1, Rev. 1 (Bellcore, October 1995);
- 15.9.4 GR-1149-CORE, OSSGR Section 10: System Interfaces, Issue 1 (Bellcore, October 1995) (Replaces TR-NWT-001149);
- 15.9.5 BellCore GR-1158-CORE, OSSGR Section 22.3: Line Information Database 6, Issue (Bellcore, October 1995);
- 15.9.6 BellCore GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service (Bellcore, May 1995); and

- 15.9.7 BOC Notes on BellSouth Networks, SR-TSV-002275, ISSUE 2, (Bellcore, April 1994).
- 15.10 Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access.
- 15.10.1 BellSouth's Service Creation Environment and Service Management System (SCE/SMS) Advanced Intelligent Network (AIN) Access shall provide ITC^DeltaCom the capability that will allow ITC^DeltaCom and other third parties to create service applications in a BellSouth Service Creation Environment and deploy those applications in a BellSouth SMS to a BellSouth SCP. The third party service applications interact with AIN triggers provisioned on a BellSouth SSP.
- BellSouth's SCE/SMS AIN Access shall provide access to SCE hardware, software, testing and technical support (*e.g.*, help desk, system administrator) resources available to ITC^DeltaCom. Scheduling procedures shall provide ITC^DeltaCom equivalent priority to these resources.
- 15.10.3 BellSouth SCP shall partition and protect ITC^DeltaCom service logic and data from unauthorized access, execution or other types of compromise.
- When ITC^DeltaCom selects SCE/SMS AIN Access, BellSouth shall provide training, documentation, and technical support to enable ITC^DeltaCom to use BellSouth's SCE/SMS AIN Access to create and administer applications. Training, documentation, and technical support will address use of SCE and SMS access and administrative functions, but will not include support for the creation of a specific service application.
- When ITC^DeltaCom selects SCE/SMS AIN Access, BellSouth shall provide for a secure, controlled access environment in association with its internal use of AIN components. ITC^DeltaCom access will be provided via remote data connection (e.g., dial-in, ISDN).
- When ITC^DeltaCom selects SCE/SMS AIN Access, BellSouth shall allow ITC^DeltaCom to download data forms and/or tables to BellSouth SCP via BellSouth SMS without intervention from BellSouth (*e.g.*, service customization and customer subscription).

16. **Preordering Loop Makeup (LMU)**

- 16.1 Description of Service
- BellSouth shall make available to ITC^DeltaCom loop makeup (LMU) data for BellSouth's network facilities. This section addresses LMU as a *preordering* transaction, distinct from ITC^DeltaCom ordering any other service(s). Loop

⁴ States Attachment 2 Clean 12-17-01

Makeup Service Inquiries (LMUSI) for preordering loop makeup are likewise unique from other preordering functions with associated service inquiries (SI) as described in this Agreement.

- 16.1.1.2 BellSouth will provide ITC^DeltaCom with loop makeup information consisting of the composition of the loop material (copper/fiber); the existence, location and type of equipment on the loop, including but not limited to digital loop carrier or other remote concentration devises, feeder/distribution interfaces, bridge taps, load coils, pair-gain devices; the loop length; and the wire gauge. The LMUSI may be utilized by ITC^DeltaCom for the purpose of determining whether the loop requested is capable of supporting DSL service or other advanced data services. The determination shall be made solely by ITC^DeltaCom and BellSouth shall not be liable in any way for the performance of the advanced data services provisioned over said loop.
- 16.1.1.3 BellSouth's LMU information is provided to ITC^DeltaCom as it exists either in BellSouth's databases or in its hard copy facility records. BellSouth does not guarantee accuracy or reliability of the LMU information provided.
- 16.1.1.4 Mechanized LMU is available for limited deployment to those CLECs that have effective X-Digital Subscriber Line (xDSL) Beta Test Agreements in place with BellSouth.
- 16.1.2 <u>Submitting Loop Makeup Service Inquiries</u>
- 16.1.2.1 ITC^DeltaCom will be able to obtain LMU information by submitting a LMUSI mechanically or manually. Mechanized LMUSIs should be submitted through BellSouth's Operational Support Systems interfaces. After obtaining the resulting loop data from the mechanized LMUSI process, if ITC^DeltaCom determines that it needs further loop data information in order to make a determination of loop service capability, ITC^DeltaCom may initiate a separate manual SI for a separate nonrecurring charge as set forth in Section 2.14.3.

16.1.2.2 **Manual**

- 16.1.2.3 LMUSIs shall be submitted on the preordering manual LMUSI form by means of fax or electronic-mail to BellSouth's Complex Resale Support Group (CRSG)/Account Team utilizing the Preordering Loop Makeup Service Inquiry form. The standard service interval for the return of a Loop Makeup Manual Service Inquiry is seven business days which is the same as BellSouth's own internal guidelines. This service interval is distinct from the interval applied to the subsequent service order. Manual LMUSIs are not subject to expedite requests.
- 16.2 <u>LMUSI Types & Associated Charges</u>

- ITC^DeltaCom may request LMU information by submitting LMUSIs in accordance with the rate elements in Attachment 11.
- 16.2.1 ITC^DeltaCom will be assessed a nonrecurring charge for each facility queried as specified in the table above. Rates for all states are interim and subject to true-up pending approval of final rates by the respective State Commissions. True-ups will be retroactive to the effective date of this Agreement.
- ITC^DeltaCom may reserve facilities for up to four (4) days in connection with a LMUSI. Reserved facilities for which ITC^DeltaCom does not plan to place a UNE local service request (LSR) should be cancelled by ITC^DeltaCom. Should ITC^DeltaCom wish to cancel a reservation on a spare facility, the cancellation will require a facility reservation number (RESID/FRN).
- The reservation holding timeframe is a maximum of four days from the time that BellSouth's LMU data is returned to ITC^DeltaCom for the facility queried. During this holding time and prior to ITC^DeltaCom's placing an LSR, the reserved facilities are rendered unavailable to other customers, whether for CLEC(s) or for BellSouth. Notwithstanding the foregoing, BellSouth does not guarantee that a reservation will assure ITC^DeltaCom's ability to order the exact facility reserved.
- 16.2.4 If ITC^DeltaCom does not submit an LSR for a UNE service order on a reserved facility within the four-day reservation timeframe, the reservation of that spare facility will become invalid and the facility will be released.
- 16.2.5 Charges for preordering LMUSI are separate from any charges associated with ordering other services from BellSouth.
- 16.3 Ordering of Other UNE Services
- 16.3.1 Whenever ITC^DeltaCom has reserved a facility through BellSouth's preordering LMU service, should ITC^DeltaCom seek to place a subsequent UNE LSR on a reserved facility, ITC^DeltaCom shall provide BellSouth the RESID/FRN of the single spare facility on the appropriate UNE LSR. ITC^DeltaCom will be billed the appropriate rate element for the specific type UNE loop ordered by ITC^DeltaCom as set forth in this Attachment. ITC^DeltaCom will not be billed any additional Loop Makeup charges for the loop so ordered. Should ITC^DeltaCom choose to place a UNE LSR having previously submitted a request for *preordering LMU without a reservation*, ITC^DeltaCom will be billed the appropriate rate element for the specific UNE loop ordered as well as additional Loop Markup charges as set forth in this Attachment. Rates are provided in the UNE Rate Exhibits for Attachment 11.

Where ITC^DeltaCom submits an LSR to order facilities reserved during the LMUSI process, BellSouth will use its best efforts to assign to ITC^DeltaCom the facility reserved as indicated on the return of the LMU.

To the extent applicable, multi-facility reservations per single RESID/FRN as provided with the mechanized LMUSI process are less likely to result in the specific assignment requested by ITC^DeltaCom. For those occasions when BellSouth's assignment system cannot assign the specific facility reserved by ITC^DeltaCom during the LMU pre-ordering transaction, BellSouth will assign to ITC^DeltaCom, subject to availability, a facility that meets the BellSouth technical standards of the BellSouth type loop as ordered by ITC^DeltaCom. If the ordered loop type is not available, ITC^DeltaCom may utilize the Unbundled Loop Modification process or the Special Construction process, as applicable, to obtain the loop type ordered.

BellSouth offers LMU information for the sole purpose of allowing ITC^DeltaCom to determine whether, in CLEC's judgment, BellSouth's loops will support the specific services that ITC^DeltaCom wishes to provide over those loops. ITC^DeltaCom may choose to use equipment that it deems will enable it to provide a certain type and level of service over a particular BellSouth loop; however, such configurations may not match BellSouth's or the industry's standards and specifications for the intended type and level of service. Accordingly, ITC^DeltaCom shall be responsible for insuring that the specific loop type (ADSL, HDSL, or otherwise) ordered on the LSR matches the LMU of the facility requested. ITC^{Delta}Com bears full responsibility for being knowledgeable of BellSouth's technical standards and the specifications of BellSouth's loops. ITC^DeltaCom bears full responsibility for making the appropriate ordering decisions of matching BellSouth loops with ITC^DeltaCom's equipment for accomplishing ITC^DeltaCom's end goal for the intended service it wishes to provide its end-user(s). ITC^DeltaCom is fully responsible for any of its service configurations that may differ from BellSouth's technical standard for the loop type ordered.

17. SS7 Network Interconnection

17.1.1 Definition

SS7 Network Interconnection is the interconnection of ITC^DeltaCom local Signaling Transfer Point Switches (STP) and ITC^DeltaCom local or tandem switching systems with BellSouth STPs. This interconnection provides connectivity that enables the exchange of SS7 messages among BellSouth switching systems and databases (DBs), ITC^DeltaCom local or tandem switching systems, and other third-party switching systems directly connected to the BellSouth SS7 network.

- 17.1.2 <u>Technical Requirements</u>
- 17.1.2.1 SS7 Network Interconnection shall provide connectivity to all components of the BellSouth SS7 network. These include:
- 17.1.2.1.1 BellSouth local or tandem switching systems;
- 17.1.2.1.2 BellSouth DBs; and
- 17.1.2.1.3 Other third-party local or tandem switching systems.
- 17.1.2.2 The connectivity provided by SS7 Network Interconnection shall fully support the functions of BellSouth switching systems and DBs and ITC^DeltaCom or other third-party switching systems with A-link access to the BellSouth SS7 network.

If traffic is routed based on dialed or translated digits between an ITC^DeltaCom local switching system and a BellSouth or other third-party local switching system, either directly or via a BellSouth tandem switching system, then it is a requirement that the BellSouth SS7 network convey via SS7 Network Interconnection the TCAP messages that are necessary to provide Call Management services (Automatic Callback, Automatic Recall, and Screening List Editing) between the ITC^DeltaCom local STPs and BellSouth or other third-party local switch.

- 17.1.2.3 When the capability to route messages based on Intermediate Signaling Network Identifier (ISNI) is generally available on BellSouth STPs, the BellSouth SS7 Network shall also convey TCAP messages using SS7 Network Interconnection in similar circumstances where the BellSouth switch routes traffic based on a Carrier Identification Code (CIC).
- 17.1.2.4 SS7 Network Interconnection shall provide all functions of the MTP as specified in ANSI T1.111. This includes:
- 17.1.2.4.1 Signaling Data Link functions, as specified in ANSI T1.111.2;
- 17.1.2.4.2 Signaling Link functions, as specified in ANSI T1.111.3; and
- 17.1.2.4.3 Signaling Network Management functions, as specified in ANSI T1.111.4.
- 17.1.2.5 SS7 Network Interconnection shall provide all functions of the SCCP necessary for Class 0 (basic connectionless) service, as specified in ANSI T1.112. In particular, this includes Global Title Translation (GTT) and SCCP Management procedures, as specified in T1.112.4. Where the destination signaling point is a BellSouth switching system or DB, or is another third-party local or tandem switching system directly connected to the BellSouth SS7 network, SS7 Network Interconnection shall include final GTT of messages to the destination and SCCP Subsystem Management of the destination. Where the destination signaling point is an ITC^DeltaCom local or tandem switching system, SS7 Network Interconnection shall include intermediate GTT of messages to a gateway pair of

- ITC^DeltaCom local STPs, and shall not include SCCP Subsystem Management of the destination.
- 17.1.2.6 SS7 Network Interconnection shall provide all functions of the Integrated Services Digital Network User Part (ISDNUP), as specified in ANSI T1.113.
- 17.1.2.7 SS7 Network Interconnection shall provide all functions of the TCAP, as specified in ANSI T1.114.
- 17.1.2.8 If and when Internetwork MTP Routing Verification Test (MRVT) and SCCP Routing Verification Test (SRVT) become approved ANSI standards and available capabilities of BellSouth STPs, SS7 Network Interconnection shall provide these functions of the OMAP.
- 17.1.2.9 SS7 Network Interconnection shall be equal to or better than the following performance requirements:
- 17.1.2.9.1 MTP Performance, as specified in ANSI T1.111.6;
- 17.1.2.9.2 SCCP Performance, as specified in ANSI T1.112.5; and
- 17.1.2.9.3 ISDNUP Performance, as specified in ANSI T1.113.5.
- 17.1.3 <u>Interface Requirements</u>
- 17.1.3.1 BellSouth shall offer the following SS7 Network Interconnection options to connect ITC^DeltaCom or ITC^DeltaCom-designated local or tandem switching systems or STPs to the BellSouth SS7 network:
- 17.1.3.1.1 A-link interface from ITC^DeltaCom local or tandem switching systems; and
- 17.1.3.1.2 B-link interface from ITC^DeltaCom STPs.
- The Signaling Point of Interconnection (SPOI) for each link shall be located at a cross-connect element, such as a DSX-1, in the Central Office (CO) where the BellSouth STP is located. There shall be a DS1 or higher rate transport interface at each of the SPOIs. Each signaling link shall appear as a DS0 channel within the DS1 or higher rate interface. BellSouth shall offer higher rate DS1 signaling links for interconnecting ITC^DeltaCom local switching systems or STPs with BellSouth STPs as soon as these become approved ANSI standards and available capabilities of BellSouth STPs. BellSouth and ITC^DeltaCom will work jointly to establish mutually acceptable SPOI.
- 17.1.3.3 BellSouth CO shall provide intraoffice diversity between the SPOIs and the BellSouth STP, so that no single failure of intraoffice facilities or equipment shall cause the failure of both B-links in a layer connecting to a BellSouth STP.

- BellSouth and ITC^DeltaCom will work jointly to establish mutually acceptable SPOI.
- 17.1.3.4 The protocol interface requirements for SS7 Network Interconnection include the MTP, ISDNUP, SCCP, and TCAP. These protocol interfaces shall conform to the following specifications:
- 17.1.3.4.1 Bellcore GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 17.1.3.4.2 Bellcore GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service;
- 17.1.3.4.3 Bellcore GR-1429-CORE, CCS Network Interface Specification (CCSNIS) Supporting Call Management Services; and
- 17.1.3.4.4 Bellcore GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).
- 17.1.3.5 BellSouth shall set message screening parameters to block accept messages from ITC^DeltaCom local or tandem switching systems destined to any signaling point in the BellSouth SS7 network with which the ITC^DeltaCom switching system has a legitimate signaling relation.
- 17.1.4 SS7 Network Interconnection shall be equal to or better than all of the requirements for SS7 Network Interconnection set forth in the following technical references:
- 17.1.4.1 ANSI T1.110-1992 American National Standard Telecommunications Signaling System Number 7 (SS7) General Information;
- 17.1.4.2 ANSI T1.111-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP);
- 17.1.4.3 ANSI T1.111A-1994 American National Standard for Telecommunications Signaling System Number 7 (SS7) Message Transfer Part (MTP) Supplement;
- 17.1.4.4 ANSI T1.112-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Signaling Connection Control Part (SCCP);
- 17.1.4.5 ANSI T1.113-1995 American National Standard for Telecommunications Signaling System Number 7 (SS7) Integrated Services Digital Network (ISDN) User Part;

- 17.1.4.6 ANSI T1.114-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Transaction Capabilities Application Part (TCAP);
- 17.1.4.7 ANSI T1.115-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Monitoring and Measurements for Networks;
- 17.1.4.8 ANSI T1.116-1990 American National Standard for Telecommunications Signaling System Number 7 (SS7) Operations, Maintenance and Administration Part (OMAP);
- 17.1.4.9 ANSI T1.118-1992 American National Standard for Telecommunications Signaling System Number 7 (SS7) Intermediate Signaling Network Identification (ISNI);
- 17.1.4.10 Bellcore GR-905-CORE, Common Channel Signaling Network Interface Specification (CCSNIS) Supporting Network Interconnection, Message Transfer Part (MTP), and Integrated Services Digital Network User Part (ISDNUP);
- 17.1.4.11 Bellcore GR-954-CORE, CCS Network Interface Specification (CCSNIS) Supporting Line Information Database (LIDB) Service;
- 17.1.4.12 Bellcore GR-1428-CORE, CCS Network Interface Specification (CCSNIS) Supporting Toll Free Service;
- 17.1.4.13 Bellcore GR-1429-CORE, CCS Network Interface Specification (CCSNIS) Supporting Call Management Services; and,
- 17.1.4.14 Bellcore GR-1432-CORE, CCS Network Interface Specification (CCSNIS) Supporting Signaling Connection Control Part (SCCP) and Transaction Capabilities Application Part (TCAP).

18. <u>AIN Selective Carrier Routing for Operator Services, Directory Assistance and Repair Centers</u>

- 18.1 BellSouth will provide AIN Selective Carrier Routing at the request of ITC^DeltaCom. AIN Selective Carrier Routing will provide ITC^DeltaCom with the capability of routing operator calls, 0+ and 0- and 0+ NPA (LNPA) 555-1212 directory assistance, 1+411 directory assistance and 611 repair center calls to preselected destinations.
- 18.1.2 ITC^DeltaCom shall order AIN Selective Carrier Routing through its Account Team. AIN Selective Carrier Routing must first be established regionally and then on a per central office, per state basis.
- 18.1.3 AIN Selective Carrier Routing is not available in DMS 10 switches.
- 4 States Attachment 2 Clean 12-17-01

- 18.1.4 Where AIN Selective Carrier Routing is utilized by ITC^DeltaCom, the routing of ITC^DeltaCom's end user calls shall be pursuant to information provided by ITC^DeltaCom and stored in BellSouth's AIN Selective Carrier Routing Service Control Point database. AIN Selective Carrier Routing shall utilize a set of Line Class Codes (LCCs) unique to a basic class of service assigned on an 'as needed basis. The same LCCs will be assigned in each central office where AIN Selective Carrier Routing is established.
- 18.1.5 Upon ordering of AIN Selective Carrier Routing Regional Service, ITC^DeltaCom shall remit to BellSouth the Regional Service Order non-recurring charges set forth in Attachment 11. There shall be a non-recurring End Office Establishment Charge per office due at the addition of each central office where AIN Selective Carrier Routing will be utilized. Said non-recurring charge shall be as set forth in this Attachment 11. For each ITC^DeltaCom end user activated, there shall be a non-recurring End User Establishment charge as set forth in Attachment 11, payable to BellSouth pursuant to the terms of the General Terms and Conditions, incorporated herein by this reference. ITC^DeltaCom shall pay the AIN Selective Carrier Routing Per Query Charge set forth in Attachment 11.
- 18.1.6 This Regional Service Order non-recurring charge will be non-refundable and will be paid with 1/2 coming up-front with the submission of all fully completed required forms, including: Regional Selective Carrier Routing (SCR) Order Request-Form A, Central Office AIN Selective Carrier Routing (SCR) Order Request Form B, AIN_SCR Central Office Identification Form Form C, AIN_SCR Routing Options Selection Form Form D, and Routing Combinations Table Form E. BellSouth has 30 days to respond to the client's fully completed firm order as a Regional Service Order. With the delivery of this firm order response to the client, BellSouth considers that the delivery schedule of this service commences. The remaining 1/2 of the Regional Service Order payment must be paid when at least 90% of the Central Offices listed on the original order have been turned up for the service.
- 18.1.7 The non-recurring End Office Establishment Charge will be billed to the client following our normal monthly billing cycle for this type of order.
- 18.1.8 The non-recurring End-User Establishment Charges will be billed to the client following our normal monthly billing cycle for this type of order
- 18.1.9 Additionally, the AIN Selective Carrier Routing Per Query Charge will be billed to the client following the normal billing cycle for per query charges.
- 18.1.10 All other network components needed, for example, unbundled switching and unbundled local transport, etc, will be billed according per contracted rates.

19. **Packet Switching Capability**

19.1 Definition

- 19.1.1 Packet Switching Capability. The packet switching capability network element is defined as the basic packet switching function of routing or forwarding packets, frames, cells or other data units based on address or other routing information contained in the packets, frames, cells or other data units, and the functions that are performed by Digital Subscriber Line Access Mulitplexers, including but not limited to:
- 19.1.2 The ability to terminate copper customer loops (which includes both a low band voice channel and a high-band data channel, or solely a data channel);
- 19.1.3 The ability to forward the voice channels, if present, to a circuit switch or multiple circuit switches;
- 19.1.4 The ability to extract data units from the data channels on the loops, and
- 19.1.5 The ability to combine data units from multiple loops onto one or more trunks connecting to a packet switch or packet switches.
- 19.1.6 BellSouth shall be required to provide non-discriminatory access to unbundled packet switching capability only where each of the following conditions are satisfied:
- 19.1.6.1 BellSouth has deployed digital loop carrier systems, including but not limited to, integrated digital loop carrier or universal digital loop carrier systems; or has deployed any other system in which fiber optic facilities replace copper facilities in the distribution section (e.g., end office to remote terminal, pedestal or environmentally controlled vault);
- 19.1. 6.2 There are no spare copper loops capable of supporting the xDSL services ITC^DeltaCom seeks to offer;
- 19.1.6.3 BellSouth has not permitted ITC^DeltaCom to deploy a Digital Subscriber Line Access Multiplexer at the remote terminal, pedestal or environmentally controlled vault or other interconnection point, nor has the ITC^DeltaCom obtained a virtual collocation arrangement at these sub-loop interconnection points as defined by 47 C.F.R. § 51.319 (b); and
- 19.1.6.4 BellSouth has deployed packet switching capability for its own use.
- 19.1.7 If there is a dispute as to whether BellSouth must provide Packet Switching, such dispute will be resolved according to the dispute resolution process set forth in Section 12 of the General Terms and Conditions of this Agreement, incorporated herein by this reference.

20. Basic 911 and E911

If ITC^DeltaCom orders unbundled network elements, then ITC^DeltaCom is also responsible for providing E911 to its end users. BellSouth agrees to offer access to the 911/E911 network pursuant to the following terms and conditions set forth in Attachment 11.

20.1 Definition

Basic 911 and E911 is an additional requirement that provides a caller access to the applicable emergency service bureau by dialing a 3-digit universal telephone number (911).

20.2 Requirements

20.2.1 Basic 911 Service Provisioning.

For Basic 911 service, BellSouth will provide to ITC^DeltaCom a list consisting of each municipality that subscribes to Basic 911 service. The list will also provide, if known, the E911 conversion date for each municipality and, for network routing purposes, a ten-digit directory number representing the appropriate emergency answering position for each municipality subscribing to 911. ITC^DeltaCom will be required to arrange to accept 911 calls from its end users in municipalities that subscribe to Basic 911 service and translate the 911 call to the appropriate 10-digit directory number as stated on the list provided by BellSouth. ITC^DeltaCom will be required to route that call to BellSouth at the appropriate tandem or end office. When a municipality converts to E911 service, ITC^DeltaCom will be required to discontinue the Basic 911 procedures and being using E911 procedures.

20.2.2 E911 Service Provisioning.

For E911 service, ITC^DeltaCom will be required to install a minimum of two dedicated trunks originating from the ITC^DeltaCom serving wire center and terminating to the appropriate E911 tandem. The dedicated trunks shall be, at a minimum, DS-0 level trunks configured either as a 2-wire analog interface or as part of a digital (1.544 Mb/s) interface. Either configuration shall use CAMA-type signaling with multifrequency ("MF") pulsing that will deliver automatic number identification ("ANI") with the voice portion of the call. If the user interface is digital, MF pulses, as well as other AC signals, shall be encoded per the u-255 Law convention. ITC^DeltaCom will be required to provide BellSouth daily updates to the E911 database. ITC^DeltaCom will be required to forward 911 calls to the appropriate E911 tandem, along with ANI, based upon the current E911 end office to tandem homing arrangement as provided by BellSouth. If the E911 tandem trunks are not available, ITC^DeltaCom will be required to route the call to a designated 7-digit local number residing in the appropriate Public Service Answering Point ("PSAP"). This call will be transported over BellSouth's interoffice network and will not carry the ANI of the calling party. ITC^DeltaCom shall be responsible for providing BellSouth with complete and accurate data for submission to the 911/E911 database for the purpose of providing 911/E911 to its end users.

20.2.3 Rates.

Charges for 911/E911 service are borne by the municipality purchasing the service. BellSouth will impose no charge on ITC^DeltaCom beyond applicable charges for BellSouth trunking arrangements.

20.2.4 Basic 911 and E911 functions provided to ITC^DeltaCom shall be at least at parity with the support and services that BellSouth provides to its customers for such similar functionality.

20.2.5 Detailed Practices and Procedures.

The detailed practices and procedures contained in the E911 Local Exchange Carrier Guide For Facility-Based Providers as amended from time to time during the term of this Agreement will determine the appropriate practices and procedures for BellSouth and ITC^DeltaCom to follow in providing 911/E911 services. BellSouth shall provide ITC^DeltaCom with updates and the latest available copies of said Guides via webposting.

21. **Rates**

21.1. General Principles

All services and network elements currently provided hereunder and all new and additional services to be provided hereunder shall be priced in accordance with all applicable provisions of the Act and the rules and orders of the Federal Communications Commission and the applicable state commissions.

21.2. Unbundled Network Elements

The prices that ITC^DeltaCom shall pay to BellSouth for Unbundled Network Elements are set forth in Attachment 11.

21.3 <u>Operational Support Systems (OSS)</u>

BellSouth has developed and made available the following mechanized systems by which ITC^DeltaCom may submit LSRs electronically.

LENS Local Exchange Navigation System

EDI Electronic Data Interchange

TAG Telecommunications Access Gateway

22.3.1 LSRs submitted by means of one of these interactive interfaces will incur an OSS electronic ordering charge as specified in the table below. An individual LSR will be identified for billing purposes by its Purchase Order Number (PON). LSRs submitted by means other than one of these interactive interfaces (mail, fax, courier, etc.) will incur a manual order charge as specified in the table below:

OPERATIONAL SUPPORT SYSTEMS	AL, GA, LA, MS, NC, SC	FL, KY, TN
OSS LSR charge, per LSR received from ITC^DeltaCom by one of the OSS interactive	\$3.50	\$3.50
interfaces	SOMEC	SOMEC
Incremental charge per LSR received from	See applicable rate	\$19.99
ITC^DeltaCom by means other than one of the	element	
OSS interactive interfaces		SOMAN

22.3.2 <u>Denial/Restoral OSS Charge</u>

In the event ITC^DeltaCom provides a list of customers to be denied and restored, rather than an LSR, each location on the list will require a separate PON and, therefore will be billed as one LSR per location.

22.3.3 Cancellation OSS Charge

ITC^DeltaCom will incur an OSS charge for an accepted LSR that is later canceled by ITC^DeltaCom.

Note: Supplements or clarifications to a previously billed LSR will not incur another OSS charge.

22.3.4 Network Elements and Other Services Manual Additive

22.3.4.1 The Commissions in some states have ordered per-element manual additive non-recurring charges (NRC) for Network Elements and Other Services ordered by means other than one of the interactive interfaces. These ordered Network Elements and Other Services manual additive NRCs will apply in these states, rather than the charge per LSR. The per-element charges are listed on the Rate Tables in Attachment 11.

LOCAL INTERCONNECTION

1.0 Scope

BellSouth shall provide ITC^DeltaCom interconnection with BellSouth's network for the transmission and routing of Telephone Exchange Service traffic and Exchange Access traffic pursuant to Section 251 (c)(2) of the Act on the terms and conditions specified in this Attachment 3.

1.1 Local Traffic

Local Traffic is as defined in Section 6 of this Attachment.

1.2 Network Interconnection

At each Party's option, Interconnection in each LATA may be accomplished at any technically feasible point within the Parties' networks, including without limitation the trunk side of any local switch; trunk interconnection points for any tandem switch; central office cross connect points; out-of-band signal transfer points; and the points of access to unbundled elements. Requests to BellSouth for interconnection at other points may be made through the Bona Fide Request/New Business Request process set out in General Terms and Conditions and Attachment 9. Interconnection may be established by the Parties via (i) Collocation as provided in Attachment 4 to the Agreement; (ii) establishment of a Fiber Meet as provided in Section 1.3.3 hereof; or (iii) interconnection via purchase of facilities from either Party by the other Party.

1.2.1 ITC^DeltaCom shall establish at least one Point of Presence and Interconnection Point with BellSouth per LATA for the delivery of ITC^DeltaCom's originated local and intraLATA toll and transit traffic. ITC^DeltaCom must establish an interconnection trunk group(s) to at least one BellSouth access tandem within the LATA for the delivery of ITC^DeltaCom's originated local and intraLATA toll and transit traffic. ITC^DeltaCom may establish additional trunk groups in such LATA pursuant to this Agreement. If ITC^DeltaCom chooses to interconnect using a single trunk group within a LATA, the interconnection trunk group must be at a BellSouth Access Tandem. Furthermore, ITC^DeltaCom must establish an interconnection trunk group(s) to all BellSouth access and local tandems where ITC^DeltaCom NXXs are "homed." A "Homing" arrangement is defined by a "Final" Trunk Group between the BellSouth Tandem andITC^DeltaCom End Office switch. A "Final" Trunk Group is the last choice telecommunications path between the Tandem and End Office switch. It is ITC^DeltaCom's responsibility to enter its own NPA/NXX access and/or

- local tandem "homing" arrangements into the national Local Exchange Routing Guide (LERG).
- 1.2.2 In order for ITC^DeltaCom to home its NPA/NXX(s) on a BellSouth Tandem, ITC^DeltaCom's NPA/NXX(s) must be assigned to an Exchange Rate Center Area served by that BellSouth Tandem and as specified by BellSouth. The specified association between BellSouth Tandems and Exchange Rate Center Areas is defined in the Local Exchange Routing Guide (LERG). To the extent that as of the date hereof, existing arrangements do not comply with this section, the Parties will work cooperatively to transition existing arrangements to comply with this section.
- 1.2.3 The Interconnection Point is the point at which the originating Party delivers its originated traffic to the terminating Party. Interconnection Point(s) are available at either Access Tandems, Local Tandems, or End Offices as described in this Attachment. ITC^DeltaCom's requested Interconnection Point will be used for the receipt and delivery of transit traffic at BellSouth Access and Local Tandems. Interconnection Point(s) established at the BellSouth Local Tandem apply only to local, ISP-bound, and local originating and terminating transit traffic. Each Party is financially and operationally responsible for providing the network on its side of the Interconnection Point.
- **1.2.4** For the purposes of this Attachment , a **Point of Presence (POP)** is the physical location (a structure where the environmental, power, air conditioning, etc. specifications for a Party's terminating equipment can be met) at which a Party establishes itself for obtaining access to the other Party's network. The POP is the physical location within which the Interconnection Point(s) occur.
- **1.2.5** For the purposes of this Attachment, the Interconnection Point is further defined as the physical telecommunications interface between BellSouth and ITC^DeltaCom's interconnection functions. It establishes the technical interface and point of operational responsibility. The primary function is to serve as the terminus for the interconnection service. The Interconnection Point has the following main characteristics:
 - 1. It is a cross-connect point to allow connection, disconnection, transfer or restoration of service.
 - 2. It is a point where BellSouth and ITC^DeltaCom can verify and maintain specific performance objectives.
 - 3. It is specified according to the interface offered in the tariff or local interconnection agreement (for example: for DS1 service the FCC # 1 tariff specifies that the interface meets the technical specifications detailed in Generic Requirements GR-342-CORE, Issue 1, December 1995.)
 - 4. The Parties provide their own equipment (CPE) to interface with the DS0, DS1, DS3, STS1 and/or OCn circuits on the customer premises.

In the establishment of BellSouth's Interconnection Point, if BellSouth chooses to collocate with ITC^DeltaCom, BellSouth will pay ITC^DeltaCom collocation charges at rates no more than the BellSouth collocation rates that BellSouth charges ITC^DeltaCom.

- **1.2.6** A minimum of one Interconnection Point shall be established in each LATA in which ITC^DeltaCom originates, terminates, or exchanges local traffic or ISP-bound traffic and interconnects with BellSouth. The location of the initial Interconnection Point shall be established by mutual agreement of the Parties. In selecting the initial Interconnection Point, both Parties will act in good faith and select the point which is most efficient for both Parties. Each Party shall be responsible for engineering and maintaining the network on its side of the Interconnection Point. Establishment of an initial Interconnection Point will be initiated by written request and will be based on traffic volumes and patterns, facilities available, and other factors unique to the area. If the Parties are not able to reach mutual agreement on an initial Interconnection Point within 30 calendar days of the date of the written request, the dispute will be escalated to the next level of management. The next level of management will meet as often as necessary to resolve the dispute. If, 30 calendar days after the dispute has been escalated, the Parties are still unable to agree to an initial mutual Interconnection Point, each Party will designate the initial Interconnection Point for its originated traffic. The Parties agree that they have, at the time of entering into this agreement, existing Interconnection Points in certain LATAs. As such, the Parties agree that if either Party desires to transition an existing Interconnection Point or Points, the transition of the first Interconnection Point in a LATA will be pursuant to this Section 1.2.6. The transition of a subsequent Interconnection Point or Points in a LATA shall be pursuant to Section 1.2.6.1 below. Notwithstanding the above, no more than eight Interconnection Points regionwide shall be transitioned within the first twelve months of the effective date of this agreement and no more than sixteen Interconnection Points regionwide shall be transitioned over a period of twenty-four months.
 - 1.2.6.1 Additional Interconnection Points in a particular LATA may be established by mutual agreement of the Parties. Absent mutual agreement, in order to establish additional Interconnection Points in a LATA, the traffic between ITC^DeltaCom and BellSouth at the proposed additional Interconnection Point must exceed 8.9 million minutes of local or ISP-bound traffic per month for three consecutive months during the busy hour. Additionally, any end office to be designated as an Interconnection Point must be more than 20 miles from an existing Interconnection Point. BellSouth will not designate an Interconnection Point at a Central Office where physical or virtual collocation space or BellSouth fiber connectivity is not available.

Upon written notification from the Party requesting the establishment of an additional Interconnection Point, the receiving Party has 45 calendar days to analyze, respond to, and negotiate in good faith the establishment of and location of such Interconnection Point. If the receiving Party disagrees that the traffic and mileage thresholds set forth herein have been met, then such Party may utilize the dispute resolution procedures set forth in the General Terms and Conditions of this Agreement.

1.2.7 The Parties shall make available to each other one way and two way trunks for the reciprocal exchange of combined local, ISP-bound and intraLATA toll traffic.

Consistent with the provisions of Sections 1.2.6 and 1.2.6.1, the Parties shall institute a bill and keep compensation plan effective September 1, 2000 under which neither Party will charge the other Party recurring and nonrecurring charges associated with local interconnection trunks and facilities for the exchange of traffic other than Transit Traffic. Both Parties, as appropriate, shall be compensated for the furnishing of local interconnection trunks and facilities for the exchange of Transit Traffic.

1.2.8 Each Party agrees to use its best efforts and act in good faith to adopt reasonable procedures to facilitate the addition of appropriate local interconnection trunks and facilities for the exchange of traffic between each Party's network.

1.3 Methods of Interconnection

Each Party at its election shall have the sole right and discretion to specify any one of the following methods for interconnection at the Interconnection Point:

- a) a Fiber-Meet as set forth in Section 1.3.3 of this Attachment;
- b) a collocation facility which it maintains at the other Party's Interconnection Point wire center (i.e., physical collocation as set forth in Attachment 4 of this Agreement);
- c) a collocation facility maintained at the Interconnection Point wire center by a third party with whom the Party requesting interconnection has contracted for such purpose;or
- d) Interconnection via purchase of leased facility(ies) as set forth in Section 1.3.2 below.
- **1.3.1** The Party requesting interconnection shall provide written notice to the other Party to change from one of the interconnection methods specified above to another of the networks specified above. The parties shall negotiate in good faith the interval for changing the interconnection methods. A mutually acceptable third party contractor can be employed by the Party making the change.

1.3.2 Interconnection via Purchase of Facilities

- **1.3.2.1** In lieu of providing facilities on its side of the Interconnection Point, either Party may purchase Local Channel facilities from the Party's specified Interconnection Point to its designated serving wire center. The Parties agree that charges for such Local Channel facilities are as set forth in Attachment 11. If a nonrecurring or recurring rate is not identified in Attachment 11 for a Local Channel, the rate shall be as set forth in the appropriate BellSouth intrastate or interstate tariff for switched access services.
 - **1.3.2.1.1** Additionally, and in lieu of provising facilities on its side of the Interconnection Point, either party may purchase

Dedicated Transport facilities from its designated serving wire center to the other Party's first point of switching. The Parties agree that charges for such Dedicated Transport facilities are as set forth in Attachment 11. If a nonrecurring or recurring rate is not identified in Attachment 11 for Dedicated Transport, the rate shall be as set forth in the appropriate BellSouth intrastate or interstate tariff for switched access services.

- **1.3.2.2** For the purposes of this Attachment, Local Channel is defined as a switch transport facility between a Party's Point of Presence and its designated serving wire center.
- **1.3.2.3** For the purposes of this Attachment, Serving Wire Center is defined as the wire center owned by one Party from which the other Party would normally obtain dial tone for its Point of Presence.
- **1.3.2.4** For the purposes of this Attachment, Dedicated Transport is defined as a switch transport facility between a Party's designated serving wire center and the first point of switching on the other Party's common (shared) network.

1.3.3 Fiber Meet

"Fiber-Meet" or "Mid-Span Meet" means an Interconnection architecture method whereby the Parties physically interconnect their networks via an optical fiber interface (as opposed to an electrical interface) at a mutually agreed upon location, at which one Party's responsibility or service begins and the other Party's responsibility ends.

- 1.3.3.1 If ITC^DeltaCom elects to interconnect with BellSouth pursuant to a Fiber Meet, ITC^DeltaCom and BellSouth shall jointly engineer and operate a Synchronous Optical Network ("SONET") transmission system by which they shall interconnect their networks for the transmission and routing of Telephone Exchange Service traffic pursuant to Section 251 (c)(2) of the Act. The Parties shall work together to determine the specific SONET transmission system. However, ITC^DeltaCom's SONET transmission system must be compatible with BellSouth's equipment in the Serving Wire Center. The Parties respective equipment and software versions must be compatible with each other and the data communications channel must be turned off. BellSouth reserves the right to determine the equipment (and compatibility thereof) that it employs for service.
- **1.3.3.2** BellSouth shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the BellSouth central office within the interconnection wire center.

- **1.3.3.3** ITC^DeltaCom shall, wholly at its own expense, procure, install and maintain the agreed upon SONET equipment in the ITC^DeltaCom central office within the interconnection wire center.
- 1.3.3.4 BellSouth shall designate an Interconnection Point outside the BellSouth central office within the interconnection wire center as a Fiber Meet point, and shall make all necessary preparations to receive, and to allow and enable ITC^DeltaCom to deliver, fiber optic facilities into the Interconnection Point with sufficient spare length to reach the fusion splice point at the Interconnection Point. BellSouth shall, wholly at its own expense, procure, install and maintain the fusion splicing point in the Interconnection Point. A Common Language Location Identification ("CLLI") code will be established for each Point of Interface. The code established must be a building type code. All orders shall originate from the Interconnection Point (i.e., Interconnection Point to ITC^DeltaCom, Interconnection Point to BellSouth).
- **1.3.3.5** ITC^DeltaCom shall deliver and maintain such strands wholly at itsown expense. Upon verbal request by ITC^DeltaCom, BellSouth shall allow ITC^DeltaCom access to the Fiber Meet entry point for maintenance purposes as promptly as possible.
- **1.3.3.6** The Parties shall jointly coordinate and undertake maintenance of the SONET transmission system. Each Party shall be responsible for maintaining the components of the SONET transmission system (e.g., software upgrades).
- **1.3.3.7** Each Party will be responsible for (i) providing its own transport facilities to the Fiber Meet, (ii) the cost to build-out its facilities to such Fiber Meet, and (iii) ITC^DeltaCom can use BellSouth's dark fiber leases as ITC^DeltaCom's portion of the network between the ITC^DeltaCom POP and the Interconnection Point.
- **1.3.3.8** Neither Party shall charge the other for its portion of the Fiber Meet facility used exclusively for non-transit local traffic (i.e. the Local Channel). Charges incurred for other services including dedicated transport facilities to the Point of Interconnection if applicable will apply. Charges for Switched and Special Access Services shall be billed in accordance with the applicable Access Service tariff (i.e. the BellSouth Interstate or Intrastate Access Services Tariff).

1.4 Interconnection in Additional LATAs

1.4.1 If ITC^DeltaCom elects to offer Telephone Exchange Services in any other LATA in which BellSouth also offers Telephone Exchange Services,

ITC^DeltaCom shall provide written notice to BellSouth of the need to establish Interconnection in such additional LATA pursuant to this Agreement.

1.4.1.1The notice provided in <u>Section 1.4.1</u> shall include (i) the initial Interconnection Point ITC^DeltaCom has designated for its originated traffic in the new LATA; (ii) ITC^DeltaCom's requested Interconnection Activation Date; and (iii) a non-binding forecast of ITC^DeltaCom's trunking requirements.

1.5 Additional Interconnection in Existing LATAs

If either party wishes to establish additional or new Interconnection Point(s) in any LATA, the Requesting Party shall provide written notice thereof to the other Party. The terms and conditions of this Agreement shall apply to such Interconnection. Neither party shall unreasonably refuse to interconnect for the delivery of traffic at each Interconnection Point (e.g., Points of collocation, existing entrance facilities, or any other mutually agreeable location). The Parties shall negotiate in good faith the interval for such additional or new Interconnection Point. If and when either Party deploys a new tandem switch or rehomes end offices, the Party initiating such change shall notify all interconnectors (including LERG updates) so that affected Parties may establish proper interconnection arrangements.

2. 0 Percent Local Usage/Percent Interstate Usage

Percent Local Use. Each Party will report to the other a Percentage Local Usage ("PLU"). The application of the PLU will determine the amount of local minutes to be billed to the other party. For purposes of developing the PLU, each party shall consider every local call and every long distance call, excluding intermediary traffic. Effective on the first of January, April, July and October of each year, BellSouth and ITC^DeltaCom shall provide a positive report updating the PLU. Detailed requirements associated with PLU reporting shall be as set forth in BellSouth's Standard Percent Local Use Reporting Platform for Interconnection Purchasers, as it is amended from time to time during this Agreement. Notwithstanding the foregoing, where the terminating company has message recording technology that identifies the traffic terminated, such information, in lieu of the PLU factor, shall upon mutual agreement of the Parties be utilized to determine the appropriate local usage compensation to be paid.

<u>Percentage Interstate Usage</u>. Both Party's based on their respective tariffs for combined interstate and intrastate traffic terminated by the other Party over the same facilities, will be required to provide a projected Percentage Interstate Usage ("PIU") to the other Party. All jurisdictional report requirements, rules and regulations for Interexchange Carriers specified in the Parties' Access Services Tariff will apply. After interstate and intrastate

traffic percentages have been determined by use of PIU procedures, the PLU factor will be used for application and billing of local interconnection. Notwithstanding the foregoing, where the terminating company has message recording technology that identifies the traffic terminated, such information, in lieu of the PLU factor, shall upon mutual agreement of the Parties be utilized to determine the appropriate local usage compensation to be paid.

2.1 Audits

On thirty (30) days written notice, each party must provide the other the ability and opportunity to conduct an annual audit to ensure the proper billing of traffic. BellSouth and ITC^DeltaCom shall retain records of call detail for a minimum of nine months from which a PLU can be ascertained. The audit shall be accomplished during normal business hours at an office designated by the party being audited. Audit requests shall not be submitted more frequently than one (1) time per calendar year. Audits shall be performed by a mutually acceptable independent auditory paid for by the party requesting the audit. The PLU and/or PIU shall be adjusted based upon the audit results and shall apply to the usage for the quarter the audit was completed, to the usage for the quarter prior to the completion of the audit, and to the usage for the two quarters following the completion of the audit.

3.0 Methods of Interconnection

3.1 Scope of Traffic

This Section prescribes parameters for trunk groups to be effected over the Interconnections specified in this Attachment 3 hereof for the transmission and routing of Telephone Exchange Service Traffic, IntraLATA Toll Traffic and Exchange Access Traffic. The Parties agree for the purpose of this Attachment that Local Interconnection is as defined in Part B of the General Terms and Conditions of this Agreement.

3.2 Interconnection Trunking and Routing

- **3.2.1** Trunking Options. Each of the following trunking arrangements shall be available to either Party, unless BellSouth demonstrates, consistent with objectively verifiable engineering standards, that such arrangement is not technically feasible. BellSouth and ITC^DeltaCom shall establish interconnecting trunk groups and trunking configurations between networks including the establishment of one-way or two-way trunks in accordance with the following conditions:
- **3.2.1.1** <u>Initial Trunking Configuration</u>. At the specified Interconnection Point(s) per this Agreement, ITC^DeltaCom and BellSouth shall interconnect using one-way trunk groups. One group carries ITC^DeltaCom-originated local and intraLATA toll traffic destined for BellSouth end-users. The other group carries BellSouth-originated local and intraLATA toll traffic destined for ITC^DeltaCom end-users. A trunk group must be

established as a two-way group from the ITC^DeltaCom switch to BellSouth access tandem(s) or local tandem(s) as specified in this Agreement. This group carries intratandem transit traffic between ITC^DeltaCom and Independent Companies, Interexchange Carriers, other CLECs and other network providers with which ITC^DeltaCom desires interconnection and has the proper contractual arrangements. If MTA is ordered to the BellSouth access tandem, this group may also carry intertandem traffic between ITC^DeltaCom and Independent Companies, other CLECs, and other network providers (except Interexchange Carriers) with which ITC^DeltaCom desires interconnection and has the proper contractual arrangements as specified in this Agreement. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required.

- **3.2.1.2** Two-Way Trunking Configuration. Two-Way Trunk Interconnection establishes two-way trunk groups to carry local and intraLATA toll traffic between ITC^DeltaCom and BellSouth. ITC^DeltaCom and BellSouth must agree to the "BellSouth Two-way Trunking Rules" (described later in this attachment) in order to establish this architecture. The other parameters are similar to one-way trunk interconnection as described by the Initial Trunking architecture above. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required.
- 3.2.1.3 SuperGroup Trunking Configuration. In the Supergroup arrangement, all local, intraLATA toll and transit traffic is carried on a single two-way trunk group between the ITC^DeltaCom switch and each BellSouth access tandem within a LATA. ITC^DeltaCom and BellSouth must agree to the "BellSouth Two-way Trunking Rules" in order to establish this architecture. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required.
- **3.2.1.4** Preferred Trunking Configuration. At the specified Interconnection Point(s) per this Agreement, all ITC^DeltaCom-to-BellSouth local, two-way intraLATA toll and two-way transit traffic is carried on a single two-way trunk group between the ITC^DeltaCom switch and BellSouth access tandem(s) within a LATA. BellSouth-to-ITC^DeltaCom local and intraLATA Toll traffic is carried on one-way trunk groups terminating to the ITC^DeltaCom switch. Other trunk groups for operator services, directory assistance, emergency services and intercept may be established if required.
- **3.2.2** BellSouth Two-Way Trunking Rules
 - **3.2.2.1** ITC^DeltaCom will initiate 2-way trunk request, and BellSouth will concur. However, 2-way trunks will be jointly provisioned.
 - **3.2.2.2** The Interconnection Point will be located at a mutually agreed location or point designated by BellSouth. If an agreement cannot be reached on the location of the Interconnection Point, each company will establish its own Interconnection Point and order 1-way trunks.
 - **3.2.2.3** BellSouth and ITC^DeltaCom will jointly review the trunk forecast, as needed, on a periodic basis, or at least every six (6) months.

- **3.2.2.4** ITC^DeltaCom will order trunks using ASR process in place for Local Interconnection after the joint planning meeting takes place between BellSouth and ITC^DeltaCom.
- **3.2.2.5** BellSouth and ITC^DeltaCom must agree on Standard Traffic Engineering parameters that will be used in the engineering of the trunk groups.
- **3.2.2.6** BellSouth and ITC^DeltaCom must agree to meet and resolve service-affecting situations in a timely manner.
- **3.2.2.7** Establishing a 2-way trunk group does not preclude BellSouth or ITC^DeltaCom from adding 1-way trunk groups within the same Local Calling Area.
- **3.2.2.8** For technical reasons, 2-way trunk groups may not be ordered to a BellSouth DMS100 Local Tandem or DMS100 End Office.
- **3.2.2.9** BellSouth will be responsible for the installation and maintenance of its trunks and facilities to the mutually agreed Interconnection Point, and ITC^DeltaCom will be responsible for the installation and maintenance of its trunks and facilities to the mutually agreed to Interconnection Point.
- 3.2.3 Any ITC^DeltaCom interconnection request that deviates from the standard trunking configurations as described in this section or the BellSouth Call Transport & Termination Service For Facility Based CLECs section of the Facility Based CLEC Activation Requirements Customer Guide that affects traffic delivered toITC^DeltaCom from a BellSouth switch that requires special BellSouth switch translations and other network modifications will requireITC^DeltaCom to submit a Bona Fide Request/New Business Request via the Bona Fide Request/New Business Request Process set forth in General Terms and Conditions and in Attachment 9.
- 3.2.4 Except as otherwise provided in 1.2.7, all terms and conditions, as well as charges, both non-recurring and recurring, associated with interconnecting trunk groups between BellSouth and ITC^DeltaCom not addressed in Attachment 11 shall be as set forth in the appropriate BellSouth intrastate or interstate tariff for switched access services. ITC^DeltaCom shall be responsible for ordering and paying for any two-way trunks carrying transit traffic.
- **3.2.5** The Parties shall utilize direct end office trunking under the following conditions:
 - (1) Tandem Exhaust If a tandem through which the Parties are interconnected is unable to, or is forecasted to be unable to support additional traffic loads for any period of time, the Parties will mutually agree on an end office trunking plan that will alleviate the tandem capacity shortage and ensure completion of traffic between ITC^DeltaCom and BellSouth's subscribers.

- (2) Traffic Volume –To the extent either Party has the capability to measure the amount of traffic between a ITC^DeltaCom switching center and a BellSouth end office, either Party shall install and retain direct end office trunking sufficient to handle actual or reasonably forecasted traffic volumes, whichever is greater, between a ITC^DeltaCom switching center and a BellSouth end office where the traffic exceeds or is forecasted to exceed a single DS1 of local traffic per month. Either Party will install additional capacity between such points when overflow traffic between ITC^DeltaCom's switching center and BellSouth's end office exceeds or is forecasted to exceed a single DS1 of local traffic per month. In the case of one way trunking, additional trunking shall only be required by the Party whose trunking has achieved the preceding usage threshold.
- (3) Mutual Agreement The Parties may install direct end office trunking upon mutual agreement in the absence of the conditions (1) or (2) above and agreement will not unreasonably be withheld.
- **3.2.6** Switched Access traffic will be delivered to and by IXCs based on ITC^DeltaCom's NXX Access Tandem homing arrangement as specified by ITC^DeltaCom in the national Local Exchange Routing Guide (LERG).
- **3.2.7** All trunk groups will be provisioned as Signaling System 7 (SS7) capable where technically feasible.

3.3. Multiple Tandem Access.

- 3.3.1 Multiple Tandem Access (MTA) provides for LATA wide transport and termination of local and intraLATA toll traffic of a Party by establishing interconnection trunk group(s) at the transporting and terminating Party's access tandem with routing through multiple access tandems of the transporting and terminating Party as required. However, the Party originating such traffic must establish interconnection trunk group(s) at all access tandems of the transporting and terminating Party where the originating Party's NXXs are "homed". If the originating Party does not have NXXs homed at an access tandem of the transporting and terminating Party within a LATA and elects not to establish an interconnection trunk group(s) at such access tandem, the originating Party may order MTA in each access tandem of the transporting and terminating Party within the LATA where it does have interconnection trunk group(s) and the transporting and terminating Party will terminate traffic to end-users served through those access tandems where the originating Party does not have an Interconnection Point. MTA shall be provisioned in accordance with the transporting and terminating Party's Ordering Guidelines.
- 3.3.2 MTA does not include switched access traffic that transits the transporting and terminating Party's network to an Interexchange Carrier (IXC). Switched Access traffic will be delivered to and by IXCs based on the originating Party's NXX Access Tandem homing arrangement as specified

- by the originating Party in the national Local Exchange Routing Guide (LERG).
- **3.3.3** For local and intraLATA toll traffic originated by one Party that the other Party transports but is destined for termination by a third Party network (transit traffic), MTA is required if multiple access tandems are necessary to deliver the call to the third party network..
- **3.3.4** With MTA, for the delivery of a Party's local and ISP-bound traffic, such Party will be assessed charges as specified in Attachment 11 to this Agreement for the additional transport and tandem switching on an elemental basis in addition to the reciprocal compensation rate to which the Parties have agreed in Section 6.1.2. Notwithstanding the foregoing, in the situation of tandem exhaust at any particular tandem, where the Parties choose MTA as an alternative routing plan, the Parties will negotiate in good-faith appropriate rates, terms and conditions for MTA.
- 3.3.5 To the extent a Party does not purchase MTA in a calling area that has multiple access tandems serving the calling area as defined by the other Party, such must establish an interconnection trunk group(s) to every access tandem in the calling area in order to serve the entire calling area. To the extent a Party does not purchase MTA and provides intraLATA toll service to its customers, it may be necessary for it to establish interconnection trunk group(s) to additional access tandems of the transporting and terminating Party that serve end offices outside the local calling area.
- 3.4 Local Tandem Interconnection. This interconnection arrangement allows ITC^DeltaCom to establish interconnection trunk group(s) at BellSouth local tandems for: (1) the delivery of ITC^DeltaCom-originated local traffic transported and terminated by BellSouth to BellSouth end offices within the local calling area as defined in BellSouth's GSST, section A3 served by those BellSouth local tandems, and (2) for local transit traffic transported by BellSouth for third party network providers who have also established interconnection trunk group(s) at those BellSouth local tandems.
- 3.4.1 When a specified local calling area is served by more than one BellSouth local tandem, ITC^DeltaCom must designate a "home" local tandem for each of its assigned NPA/NXXs and establish trunk connections to such local tandems. Additionally, ITC^DeltaCom may choose to establish interconnection trunk group(s) at the BellSouth local tandems where it has no codes homing but is not required to do so. ITC^DeltaCom may deliver local traffic to a "home" BellSouth local tandem that is destined for other BellSouth or third party network provider end offices subtending other BellSouth local tandems in the same local calling area where ITC^DeltaCom does not choose to establish a interconnection trunk group(s). It is ITC^DeltaCom's responsibility to enter its own NPA/NXX local tandem homing arrangements into the Local Exchange Routing Guide (LERG) either directly or via a vendor in order for other third party network providers to determine appropriate traffic routing to ITC^DeltaCom's codes. Likewise, ITC^DeltaCom shall obtain its routing information from the LERG.

- 3.4.2 Notwithstanding establishing interconnection trunk group(s) to BellSouth's local tandems, ITC^DeltaCom must also establish interconnection trunk group(s) to BellSouth access tandems within the LATA on which ITC^DeltaCom has NPA/NXX's homed for the delivery of Interexchange Carrier Switched Access (SWA) and toll traffic, and traffic to Type 2A CMRS connections located at the access tandems. BellSouth cannot switch SWA traffic through more than one BellSouth access tandem. SWA, Type 2A CMRS or toll traffic routed to the local tandem in error will not be backhauled to the BellSouth access tandem for completion. (Type 2A CMRS interconnection is defined in BellSouth's A35 General Subscriber Services Tariff.)
- **3.4.3** BellSouth's provisioning of local tandem interconnection assumes thatITC^DeltaCom has executed the necessary local interconnection agreements with the other third party network providers subtending those local tandems as required by the Act.
- 3.5 Direct connection to an End Office shall provide the Party requesting Interconnection with access to all valid NXX codes served by that End Office. Direct end office trunking may not be unreasonably withheld. Trunking can be established to tandems or end offices or a combination as mutually agreed.
- 3.6 If a pre-existing trunk group is unable to support additional traffic loads, or consistent with standard trunk engineering practices, is reasonably forecasted as specified in accordance with Section 4.4 to be unable to support additional traffic loads, each Party shall, upon request of the other Party, provision additional trunks to expand the capacity of such pre-existing trunk group (within the timeframes specified in Attachments 6 and 10 to the Agreement).
- 3.7 ITC^DeltaCom may opt at any time to terminate to BellSouth some or all local exchange traffic and intraLATA toll traffic originating on its network, together with switched access traffic, via Feature Group B or D Switched Access services which ITC^DeltaCom may otherwise purchase from BellSouth, subject to the rates, terms and conditions specified in BellSouth's applicable switched access tariffs. At no time shall ITC^DeltaCom be required to route outbound traffic via facilities for which a full retail or end user toll charge would be assessed when parallel FG-B or FG-D routing, or routing via a different carrier exists which is capable of carrying and completing said traffic at more favorable rates. However, Switched Access Traffic will not be represented as Local Traffic.

4.0 Network Design and Management for Interconnection

4.1 <u>Network Management and Changes.</u> BellSouth shall work cooperatively with ITC^DeltaCom to install and maintain the most effective and reliable interconnected telecommunications networks, including but not limited to, the exchange of toll-free maintenance contact numbers and escalation procedures. BellSouth agrees to provide public notice of changes in the

information necessary for the transmission and routing of services using its Local Exchange facilities or networks, as well as of any other changes that would affect the interoperability of those facilities and networks.

- 4.2 Interconnection-Technical Standards. The interconnection of all networks will be based upon accepted industry/national guidelines for transmission standards and traffic blocking criteria. Interconnecting facilities shall conform, at a minimum, to the telecommunications industry standard of DS1 pursuant to Bellcore Standard No. TR-NWT-00499. Signal transfer point, Signaling System 7("SS7") connectivity is required at each interconnection point. BellSouth will provide out-of-band signaling using Common Channel Signaling Access Capability where technically and economically feasible, in accordance with the technical specifications set forth in the BellSouth Guidelines to Technical Publication, TR-TSV000905. Facilities of each Party shall provide the necessary on-hook, off-hook answer and disconnect supervision and shall hand off calling number ID when technically feasible.
- 4.3 <u>Network Management Controls</u>. Both Parties shall work cooperatively with each other to apply sound network management principles by invoking appropriate network management controls, e.g., call gapping, to alleviate or prevent network congestion.

4.4 Forecasting Requirements

- 4.4.1 The Parties shall exchange technical descriptions and forecasts of their Interconnection and traffic requirements in sufficient detail necessary to establish the interconnections required to assure traffic completion to and from all customers in their respective designated service areas. In order for BellSouth to provide as accurate reciprocal trunking forecasts as possible to ITC^DeltaCom, ITC^DeltaCom must timely inform BellSouth of any known or anticipated events that may affect BellSouth reciprocal trunking requirements. If ITC^DeltaCom refuses to provide such information, BellSouth shall provide reciprocal trunking forecasts based only on existing trunk group growth and BellSouth's annual estimated percentage of BellSouth subscriber line growth.
 - 4.4.2 Both Parties shall meet every six (6) months or at otherwise mutually agreeable intervals for the purpose of exchanging non-binding forecasts of their traffic and volume requirements for the Interconnection and Network Elements provided under this Agreement, in the form and in such detail as agreed by the Parties. The Parties agree that each forecast provided under this Section shall be deemed "Confidential Information" under Section 9 of the General Terms and Conditions Part A of the Agreement.

4.4.3 The trunk forecast should include trunk requirements for all of the interconnecting trunk groups for the current year plus the next two future years. The forecast meeting between the Parties could be a face-to-face meeting, video conference or audio conference. It could be held regionally or geographically. Ideally, these forecast meetings should be held at least semi-annually, or more often if the forecast is no longer usable. Updates to a forecast or portions thereof, should be made whenever the Party providing the forecast deems that the latest trunk requirements exceed the original quantities by 24 trunks or ten percent (10%), whichever is greater. Either Party should notify the other Party if they have measurements indicating that a trunk group is exceeding its designed call carrying capacity and is impacting other trunk groups in the network. Also, either Party should notify the other Party if they know of situations where the traffic load is expected to increase significantly and thus affect the interconnecting trunk requirements as well as the trunk requirements within the other Party's network. The Parties agree that the forecast information provided under this Section shall be deemed "Confidential Information" as described in the General Terms and Conditions Part A of the Agreement.

4.4.4 Binding forecast:

- 4.4.4.1 In addition to, and not in lieu of, non-binding forecasts, ITC^DeltaCom may at its electionprovide to BellSouth a binding forecast of the trunks that BellSouth will need to interconnect with ITC^DeltaCom in order to terminate traffic to ITC^DeltaCom. Unless otherwise agreed, a binding forecast may not be requested for an existing trunk group that is underutilized as defined in this section or for exhausted BellSouth switch locations. ITC^DeltaCom shall provide to BellSouth sufficient justification for the quantity of trunks contained within the binding forecast. The due date contained in the binding forecast shall be no less than three months from the date of the binding forecast. Once the binding forecast is submitted to BellSouth, ITC^DeltaCom agrees to make no changes to said forecast.
- **4.4.4.2** BellSouth shall provide the total amount of requested trunks from either tandem or end offices depending on trunk and facilities availability.
- 4.4.4.3 A binding forecast shall not replace the ASR process of ordering trunks and BellSouth shall order the quantity of trunks from ITC^DeltaCom set forth in the binding forecast. BellSouth shall request due dates on the trunk orders to coincide with the due dates specified in the binding forecast, and the Parties shall provision the ordered trunks by the due date.
- **4.4.4.4** To recover the cost associated with assuring that the quantity of trunk port terminations needed to meet the binding forecast are available on the agreed upon

due date, ITC^DeltaCom shall pay to BellSouth a non-recurring charge of \$305.00 for the first DS1 trunk port and a non-recurring charge \$152.50 for each additional DS1 trunk port forecasted in a trunk group (i.e. between an A to Z location or BellSouth switch location to an ITC^DeltaCom switch location)

- **4.4.4.5** If, within 180 days of the installation of the trunks, 60 percent of the capacity of the trunks is not being utilized, ITC^DeltaCom will pay BellSouth a percentage of the total monthly recurring trunk and facility charges as set forth in BellSouth's tariffs for the percentage of the trunks' capacity that is not being utilized.
- 4.4.4.6 If, within 360 days of the installation of the trunks, 85 percent of the capacity of the trunks is not being utilized, ITC^DeltaCom will pay BellSouth a percentage of the total monthly recurring trunk and facility charges as set forth in BellSouth's tariffs for the percentage of the trunks' capacity that is not being utilized.
- **4.4.4.7** If, within 405 days of the installation of the trunks, the trunks are not being utilized to 85 percent of the capacity of the trunks, the excess trunks may be disconnected by BellSouth.
- **4.4.4.8** Utilization on BellSouth reciprocal interconnection trunk groups associated with a binding forecast shall be measured monthly and shall be measured at the time consistent busy hour. The charges as a result of under-utilization as described in the preceding section shall apply monthly.
- **4.4.4.9** Except in the instance of underutilization by ITC^DeltaCom in Section 4.4.4.5, neither Party shall charge the other for nonrecurring trunk and recurring, if applicable, trunk charges associated with a binding forecast.
- **4.4.4.10** Where BellSouth installs additional facilities on an interconnection trunk group associated with a binding forecast, ITC^DeltaCom will not be subject to underutilization penalties as set for in this Section 4.4.4.
 - 4.5 Common Channel Signaling. Both Parties shall provide LEC-to-LEC Common Channel Signaling ("CCS") to each other, where available, in conjunction with all traffic in order to enable full interoperability of CLASS features and functions except for call return. All CCS signaling parameters will be provided, including automatic number identification ("ANI"), calling party number ("CPN"), originating line information ("OLI"), calling company category, charge number, etc. All privacy indicators will be honored, and each Party will cooperate with the other on the exchange of Transactional Capabilities Application Part ("TCAP") messages to facilitate full interoperability of CCS-based features between the respective networks. Where available, network signaling information such as Carrier Identification Parameter (CCS platform), at the standard tariff rates, and CIC/OZZ information (non-CCS environment) will be provided wherever such information is needed for call routing or billing. The Parties will

follow all Ordering and Billing Forum (OBF) adopted standards pertaining to CIC/OZZ codes. Where CCS is not available, in-band multi-frequency (MF) wink start E&M channel Associated signaling will be provided. Such MF arrangements will require a separate trunk group between ITC^DeltaCom's switch and one specified BellSouth switch. ITC^DeltaCom shall establish CCS interconnection with BellSouth signal transfer points (STPs) in each LATA, either directly or via an intermediary STP provider.

- **4.5.1** Where the interconnection is via B-link connections, charges for the SS7 interconnection elements are as follows: 1) Port Charge BellSouth will not bill an STP port charge nor will BellSouth agree to pay a port charge; 2) SS7 Network Usage BellSouth will bill SS7 Network Usage and will agree to pay usage billed by ITC^DeltaCom (to the extent that a flat rate surrogate charge is billed by ITC^DeltaCom, it will not exceed BellSouth's charge); 3) SS7 Link- BellSouth will bill for only two links of each quad ordered. Application of these charges in this manner reflects the reciprocal use of the two parties signaling networks.
- **4.5.2** Where the interconnection is via A-link connections, charges for the SS7 interconnection elements are as follows: 1) Port Charge-BellSouth will bill an STP port charge and does not agree to pay a termination charge at ITC^DeltaCom's end office; 2) SS7 Network Usage-BellSouth will bill for usage on its SS7 network and will not agree to pay for any usage billed by ITC^DeltaCom; 3) Link-BellSouth will bill full charges for each link in the A link pair and will not agree to pay ITC^DeltaCom for any portion of those links
- **4.5.3** Call Information. BellSouth and ITC^DeltaCom will send and receive 10 digits for local traffic. BellSouth and ITC^DeltaCom shall exchange the proper call information, *i.e.*, originated call company number and destination call company number, CIC, and OZZ, including all proper translations for routing between networks and any information necessary for billing.
- **4.5.4** Each Party is responsible for requesting Interconnection to the other Party's CCS network, where SS7 signaling on the trunk group(s) is desired. The Parties shall establish Interconnection at the STP.
- **4.5.5** Where available and upon the request of the other Party, each Party shall cooperate to ensure that its trunk groups are configured utilizing the B8ZS ESF protocol for 64 kbps clear channel transmission to allow for ISDN interoperability between the Parties' respective networks.

- 4.5.6 All originating Toll Free Service calls for which BellSouth performs the Service Switching Point (SSP) function (e.g., performs the database query) shall be delivered by ITC^DeltaCom using GR-394 format over the transit trunk group. Carrier Code "0110" and Circuit Code of "08" shall be used for all such calls. In the event ITC^DeltaCom becomes a toll free service provider, BellSouth shall deliver traffic using the GR-394 format over a trunk group designated for Toll Free Service.
- **4.5.7** All originating Toll Free Service calls for which ITC^DeltaCom performs the SSP function, if delivered to BellSouth, shall be delivered by ITC^DeltaCom using GR-394 format over the transit trunk group for calls destined to IXCs, or shall be delivered by ITC^DeltaCom using GR-317 format over the Local Interconnection Trunk Group for calls destined to end offices that directly subtend BellSouth access tandems.

5.0 Parity

Interconnection shall be equal in quality to that provided by the Parties to themselves or to any subsidiary, affiliate, or other party. Equal in quality means that interconnection facilities shall meet the same technical criteria and service standards that are used within the Parties' own networks, such as probability of blocking in peak hours and transmission standards.

- 5.1 The Parties shall provide Interconnection (i) in accordance with the requirements of this Attachment 3, (ii) in conformance with the Performance Standards listed in Attachment 10, and (iii) as required by the applicable state Commission and the FCC.
- **5.2** Local Dialing Parity

Each Party shall provide local dialing parity, meaning that each Party's customers will not have to dial any greater number of digits than the other Party's customers to complete the same call. In addition, under equivalent interconnection arrangements, ITC^DeltaCom local service customers will experience at least the same quality as BellSouth local service customers regarding post-dial delay, call completion rate and transmission quality.

6.0 Interconnection Compensation

6.1 <u>Compensation for Call Transportation and Termination for Local Traffic and Inter-Carrier Compensation for ISP-Bound Traffic</u>

- **6.1.1** Local Traffic is defined as any telephone call that originates and terminates in the same Local Access and Transport Area ("LATA") as that term is defined in 47 U.S.C. § 153(25) of the Communications Act of 1934, as amended.
- **6.1.2** Subject to the Parties agreement to the terms of Sections 1.2 through 1.2.8, 1.3.2, 3.3.3, and 6.1 through 6.1.9, the Parties will compensate each other on a mutual and reciprocal basis for the transport and termination of Local Traffic at the following rates:

1/1/01 - 12/31/01 \$.00175 per MOU 1/1/02 - 12/31/02 \$.00150 per MOU

The Parties recognize and agree that they negotiated these annual rates together as a complete rate structure to apply over the full term of this Agreement and that the Parties would not have mutually agreed to accept a single annual rate in any single year.

- Providers ("ISPs") should be considered Local Traffic for purposes of this Agreement. Dial-up Calls are defined as calls to an ISP that are dialed by using a local dialing pattern (7 or 10 digits) by the calling party to an ISP server located in the LATA (hereinafter referred to as "ISP-bound traffic"). However, without prejudice to either Party's position concerning the application of reciprocal compensation to ISP-bound traffic, the Parties agree for purposes of this Agreement only to compensate each other for the delivery of ISP-bound traffic at the same per minute of use rates set forth in Paragraph 6.1.2. It is expressly understood and agreed that this inter-carrier compensation mechanism for ISP-bound traffic is being established in consideration for: (1) the waiver and release by each Party for any and all claims for reciprocal compensation for ISP-bound traffic exchanged between the Parties prior to January 1, 2001, which is hereby given; and (2) the terms and conditions in Sections 1.2 through 1.2.8, 1.3.2, 3.3.3, and 6.1 through 6.1.9.
- **6.1.4**. The Parties recognize and agree that the FCC, courts of competent jurisdiction, or state commissions with jurisdiction over the Parties will issue subsequent decisions on ISP-bound traffic ("Subsequent Decisions"). Notwithstanding any provision in this Agreement to the contrary, the inter-carrier compensation mechanism established in Section 6.1.3 shall continue at the rates set forth in Section 6.1.2 through December 31, 2002 without regard to such Subsequent Decisions.
- 6.1.5 ITC^DeltaCom hereby waives its rights under this Agreement as well as under Section 252(i) of the 1996 Act and applicable FCC regulations to elect rates, terms, and conditions from any other approved interconnection agreement executed by BellSouth as they relate to: (a) Local Interconnection arrangements described in Section 1 of Attachment 3 to this Agreement; (b) Multiple Tandem Access described in Section 3 of Attachment 3 to this Agreement; and (c) Interconnection Compensation described in Section 6 of Attachment 3 to this Agreement. Accordingly, during the term of this Agreement, ITC^DeltaCom agrees that it will not seek to elect such rates, terms, or conditions from another interconnection agreement, regardless of when that interconnection agreement was entered into by BellSouth and regardless of when ITC^DeltaCom became aware of the existence of such agreement. Nothing herein shall

- be construed as a waiver of ITC^DeltaCom's rights to elect any rates, terms, and conditions to take effect after December 31, 2002.
- 6.1.6 The Parties recognize and agree that the compensation for the transport and termination of Local Traffic set forth in Section 6.1.2 and the inter-carrier compensation mechanism for ISP-bound traffic set forth in Section 6.1.3 are intended to allow each Party to recover costs associated with such traffic. The Parties agree to act in good faith in seeking compensation under Section 6.1.2 and Section 6.1.3 of this Agreement and to refrain from taking any action, whether directly or indirectly, for the primary purpose of generating compensation from the other Party. For example, the Parties recognize and agree that such compensation will not be billed and shall not be paid for a call placed by an end user customer, or placed on behalf of an end user customer, to establish or maintain a network connection if: (1) such call is not recognized by industry practice to constitute traffic (voice or data) which results from a telephone call; (2) the end user customer does not control the dialed number destination and content of that call; and (3) the primary purpose of that call is to generate the payment of compensation as a result of establishing or maintaining the network connection.
- 6.1.7 In Order No. PSC-00-0537-FOF-TP in Docket 990750-TP dated March 15, 2000, the Florida Public Service Commission approved a reciprocal compensation rate of \$.009 per minute of use for inclusion in the interconnection agreement between BellSouth and ITC^DeltaCom. Upon BellSouth's motion for reconsideration, the Florida Commission directed that the Parties mediate their dispute over the appropriate reciprocal compensation rate. In consideration for the resolution of this dispute, BellSouth hereby agrees to prepay compensation to ITC^DeltaCom for the Local Traffic to be transported and terminated to ITC^DeltaCom and the ISP-bound traffic to be handled by ITC^DeltaCom during 2001 and 2002 based upon ITC^DeltaCom's good-faith forecast of the amount of such traffic during this period. The compensation that BellSouth will prepay is Twenty Four Million Dollars (\$24,000,000) for 2001 and an amount not to exceed Twenty Four Million Dollars (\$24,000,000) for 2002. Such prepaid compensation is subject to the reconciliation process and the cap set forth in Sections 6.1.7.1, 6.1.7.2, and 6.1.7.3 respectively, and the change in control provision set forth in Section 6.1.7.4.
- 6.1.7.1 The prepayment of compensation pursuant to Section 6.1.7 shall not relieve ITC^DeltaCom of the obligation to render monthly invoices to BellSouth reflecting the total minutes of Local Traffic transported and terminated to ITC^DeltaCom and the minutes of ISP-bound traffic handled by ITC^DeltaCom for which compensation is being sought from BellSouth. Not later than July 30, 2001, the Parties agree to review these invoices to verify the minutes of Local Traffic transported and terminated to ITC^DeltaCom and the minutes of ISP-bound traffic handled by ITC^DeltaCom during the first six months of 2001. A similar review will be conducted as soon as practicable in December 2001 for the preceding three month-period ("December 2001 Review"). Subsequent reviews will be conducted quarterly in 2002. As part of these reviews, ITC^DeltaCom agrees to provide BellSouth upon request whatever information may be reasonably necessary to verify the minutes of use for which compensation is being sought from BellSouth. If, as a result of these reviews, the Parties determine that ITC^DeltaCom's actual minutes of use for Local Traffic and ISP-bound traffic are below

the forecasted amounts for such traffic, ITC^DeltaCom will refund promptly to BellSouth the prepaid compensation attributable to the difference between the actual and forecasted minutes of use. If, as a result of these reviews, the Parties determine that ITC^DeltaCom's actual minutes of use for Local Traffic and ISP-bound traffic exceed the forecasted amounts, BellSouth will pay promptly to ITC^DeltaCom compensation attributable to the difference between the actual and forecasted minutes of use, subject to the cap set forth in Section 6.1.7.3. Any dispute that may arise during this reconciliation process shall be handled pursuant to the dispute resolution procedures under this Agreement.

- **6.1.7.2** The specific amount of reciprocal compensation to be prepaid by BellSouth for the Local Traffic to be transported and terminated to ITC^DeltaCom and the ISP-bound traffic to be handled by ITC^DeltaCom during 2002 will be determined as a result of the December 2001 Review. As part of this review, the Parties will calculate the Prepayment Threshold by: (a) calculating the total minutes of use of Local Traffic and ISP-bound traffic originated by BellSouth for which ITC^DeltaCom is seeking compensation during September, October, and November 2001; (b) multiplying these total minutes by the rate of \$.0015 per minute of use; and (c) annualizing the results by applying a factor of four. If the Prepayment Threshold is equal to or exceeds Twenty One Million Six Hundred Thousand Dollars (\$21,600,000), BellSouth will prepay compensation to ITC^DeltaCom for the Local Traffic to be transported and terminated to ITC^DeltaCom and the ISPbound traffic to be handled by ITC^DeltaCom during 2002 in the amount of Twenty Four Million Dollars (\$24,000,000). If the Prepayment Threshold is less than Twenty One Million Six Hundred Thousand Dollars (\$21,600,000), BellSouth will prepay compensation to ITC^DeltaCom for the Local Traffic to be transported and terminated to ITC^DeltaCom and the ISP-bound traffic to be handled by ITC^DeltaCom during 2002 in the amount of the Prepayment Threshold.
- 6.1.7.3 Notwithstanding the prepayment of compensation by BellSouth and notwithstanding any provision in this Agreement to the contrary, BellSouth's total liability for compensation to ITC^DeltaCom for the transport and termination of Local Traffic and for the handling of ISP-bound traffic shall not exceed Twenty Seven Million and Five Hundred Thousand Dollars (\$27,500,000) for 2001 and Twenty Nine Million and Five Hundred Thousand Dollars (\$29,500,000) for 2002._ITC^DeltaCom agrees that once the cap has been met in a particular year that it will not seek to collect compensation from BellSouth, whether directly or indirectly, for the transport and termination of Local Traffic and for the handling of ISP-bound traffic. The Parties agree that no compensation will be due from BellSouth for resold services purchased by ITC^DeltaCom from BellSouth, even if ITC^DeltaCom purchases such services on behalf of or for the use of another entity, whether or not affiliated with ITC^DeltaCom. Further, the Parties agree that no compensation will be due from ITC^DeltaCom to BellSouth for transport and termination of Local Traffic over resold services.
- 6.1.7.4 DeltaCom hereby agrees that it will not dissolve, consolidate, reorganize, merge or sell all or substantially all of its stock or assets without the prior written consent of BellSouth, which consent shall not be unreasonably withheld. In the event BellSouth consents to a dissolution of ITC^DeltaCom, and prior to such dissolution taking effect, ITC^DeltaCom

shall reimburse BellSouth all unaccrued amounts of compensation prepaid by BellSouth pursuant to Section 6.1.7 as of the effective date of such dissolution. In the event BellSouth consents to a consolidation, reorganization or merger of ITC^DeltaCom or an acquisition of all or substantially all of the stock or assets of ITC^DeltaCom, BellSouth shall have the option to either continue pursuant to the terms and conditions of this Agreement or elect to be reimbursed by ITC^DeltaCom or its successor interest for all unaccrued amounts of compensation prepaid by BellSouth pursuant to Section 6.1.7 as of the effective date of such consolidation, reorganization, merger, or acquisition. In the event BellSouth elects to exercise its option of reimbursement as a result of consolidation, reorganization, merger or acquisition described herein, this Agreement shall be deemed amended as of the effective date of such consolidation, reorganization, merger, or acquisition to delete Section 6.1.7 in its entirety, such that: (a) the rates set forth in Section 6.1.2 for the transport and termination of Local Traffic and the delivery of ISP-bound traffic shall continue in effect for all applicable minutes of use during the applicable period of time following such consolidation, reorganization, merger, or acquisition; (b) compensation for such minutes of use shall be paid by BellSouth to such successor in interest to this Agreement based upon actual minutes of use during the applicable time period; and (c) the cap set forth in Section 6.1.7.3 shall be eliminated and of no effect. Prior to the effective date of any consolidation, reorganization, merger, or acquisition described herein, ITC^DeltaCom shall take all necessary steps to bind its successor in interest to the terms of this Section.

- 6.1.7.5 However, in the event BellSouth does not consent to a ITC^DeltaCom dissolution, consolidation, reorganization, merger or sale of all or substantially all of ITC^DeltaCom's stock or assets, ITC^DeltaCom may refund to BellSouth all unaccrued amounts prepaid to ITC^DeltaCom as described in Section 6.1.7 and proceed with such dissolution, consolidation, reorganization, merger or sale of all or substantially all of ITC^DeltaCom's stock or assets. However, in this event, this Agreement shall be deemed amended as of the effective date of such consolidation, reorganization, merger, or acquisition to delete Section 6.1.7 in its entirety, such that: (a) the rates set forth in Section 6.1.2 for the transport and termination of Local Traffic and the delivery of ISP-bound traffic shall continue in effect for all applicable minutes of use during the applicable period of time following such consolidation, reorganization, merger, or acquisition; (b) compensation for such minutes of use shall be paid by BellSouth to such successor in interest to this Agreement based upon actual minutes of use during the applicable time period; and (c) the cap set forth in Section 6.1.7.3 shall be eliminated and of no effect.
- **6.1.7.6** BellSouth may from time to time elect to apply a portion of the prepayment described in this Section against other amounts due from BellSouth to ITC^DeltaCom for services provided by ITC^DeltaCom to BellSouth between the date of this Agreement and December 31, 2002. BellSouth's election as described in this paragraph shall not affect BellSouth's obligation to pay in full all amounts when due for services provided to BellSouth by ITC^DeltaCom, including, without limitation the obligation of payment of reciprocal compensation as provided in this Section 6.1.6 and subject to the terms and conditions set forth therein.

- **6.1.8** Neither Party shall represent switched access services traffic as Local Traffic for purposes of payment of reciprocal compensation.
- **6.1.9** Local traffic is defined in Section 6.1.1 as any call that originated and terminates within the same LATA and therefore is subject to reciprocal compensation. All other traffic including transit traffic is subject to switched access charges as defined by the parties' respective tariffs. Notwithstanding the above, any mandated local calling areas shall be subject to reciprocal compensation and not access charges

7. Transmission and Routing of Exchange Access Traffic

The Parties shall jointly provide Tandem-transported Switched Exchange Access Services to Interexchange Carriers to enable such Interexchange Carriers to originate and terminate traffic from/to ITC^DeltaCom's End Users, and to originate and terminate traffic to/from BellSouth's End Users.

8.0 NXX Translations Implementation

It shall be the responsibility of each Party to program and update its switches and network systems pursuant to the local exchange routing guide (LERG) and other switched telecommunications industry guidelines to recognize and route traffic to the other Party's assigned NXX codes. Neither Party shall impose any fees or charges whatsoever on the other Party for such activities.

- **8.1** Testing and inputting of the translations in the BellSouth databases of ITC^DeltaCom's NXXs' should be the same as BellSouth's own.
- **8.2.1** Each Party will translate NXXs according to industry guidelines, including the terminating LATA in which the NXXs/rate center is located.
- **8.3** The Parties will cooperate and implement industry solutions for number conservation, e.g. number pooling.

9.0 Meet-Point Billing Arrangements

9.1 Meet-Point Billing

When BellSouth and ITC^DeltaCom provide an access service connection between an interexchange carrier ("IXC") and each other, each party will provide its own access services to the IXC on a multi-bill, multi-tariff meetpoint basis. Each party will bill its own access services rates to the IXC with the exception of the interconnection charge. The interconnection charge will be billed by the party providing the end office function. The Parties will use the Multiple Exchange Carrier Access Billing guidelines to establish meet point billing for all applicable traffic, including traffic terminating to ported numbers. 30-day billing periods will be employed for these arrangements. The recording party agrees to provide to the initial billing company, at no charge, the switched access detailed usage data within no more than sixty (60)

days after the recording date. The initial billing company will provide the switched access summary usage data to all subsequent billing companies within 10 days of rendering the initial bill to the IXC. Each company will notify the other when it is not feasible to meet these requirements so that the customers may be notified for any necessary revenue accrual associated with the significantly delayed recording or billing. As business requirements change data reporting requirements may be modified as necessary.

- 9.2 In the event that either Party fails to provide switched access detailed usage data to the other Party within 90 days after the recording date and the receiving Party is unable to bill and/or collect access revenues due to the sending Party's failure to provide such data within said time period, then the Party failing to send the data as specified herein shall be liable to the other Party in an amount equal to the unbillable or uncollectible revenues. Each Party will provide complete documentation to the other to substantiate any claim of unbillable access revenues. A negotiated settlement will be agreed upon between the Parties.
- **9.3** Each company will retain for a minimum period of sixty (60) days, access message detail sufficient to recreate any data which is lost or damaged by their company or any third party involved in processing or transporting data.
- **9.4** Each company agrees to recreate the lost or damaged data within five (5) days of notification by the other or by an authorized third party handling the data.
- **9.5** Each company also agrees to process the recreated data within five (5) days of receipt at its data processing center.
- **9.6** All claims should be filed with the other company with 120 days of the receipt of the date of the unbillable usage.
- 9.7 The Initial Billing Company shall keep records of its billing activities relating to jointly-provided Intrastate and Interstate access services in sufficient detail to permit the Subsequent Billing Company to, by formal or informal review or audit, to verify the accuracy and reasonableness of the jointly-provided access billing data provided by the Initial billing Company. Each company agrees to cooperate in such formal or informal reviews or audits and further agrees to jointly review the findings of such reviews or audits in order to resolve any differences concerning the findings thereof.
- **9.8** The billing percentages shall be calculated by the Parties according to one of the methodologies specified for such purposes in the MECAB document and filed by the Parties, if applicable, in the National Exchange Carrier Association ("NECA") FCC Tariff No. 4.
- **9.9** Switched Access Traffic shall be defined in accordance with either Party's Commision approved Access Tariffs.

10. Transit Traffic Service

- 10.1 Transit Traffic Service is defined as the tandem switching, transport and delivery by one Party of (1) Local Traffic, IntraLATA Toll Traffic and InterLATA Toll Traffic originated from the other Party and terminating to a third party carrier, and (2) Local Traffic, IntraLATA Toll Traffic and InterLATA Toll Traffic originated from a third party carrier and terminating to the other Party. The Parties shall provide Transit Traffic Service to ITC^DeltaCom in accordance with this Section 10. Rates for Transit Traffic Service for Local Traffic shall be the applicable call transport and termination charges for Local Traffic, as set forth in Attachment 11 to this Agreement. Rates for Transit Traffic Service for IntraLATA Toll Traffic and InterLATA Toll Traffic (i.e., Switched Access Traffic) shall be the applicable call transport and termination charges as set forth in the providing Party's Intrastate or Interstate switched access tariff, as filed and effective with the FCC or appropriate State Commission. Billing associated with all Transit Traffic Service shall be pursuant to MECAB procedures. Wireless Type 1 traffic shall not be treated as transit traffic from a routing or billing perspective. Wireless Type 2A traffic from a third Party carrier to ITC^DeltaCom shall not be treated as transit traffic from a routing or billing perspective until the transiting Party and the wireless carrier have the capability to meet point bill properly in accordance with MECAB guidelines. Where ITC^DeltaCom has a direct connection (via a cross connect between collocation spaces in a BellSouth central office or otherwise) to a third party carrier, ITC^DeltaCom and that third party carrier shall not utilize Transit Traffic Service from BellSouth.
- 10.2 The delivery of traffic originated by ITC^DeltaCom which transits the BellSouth network and is transported to another carrier's network is excluded from any BellSouth billing guarantees and will be delivered at the rates as set forth in Attachment 11 to this Agreement. ITC^DeltaCom is responsible for establishing any necessary agreements or the placement of any necessary valid orders with the terminating carrier for the receipt of this traffic through the BellSouth network. BellSouth will not be liable for any compensation to the terminating carrier as a result of providing Transit Traffic Service.
- **10.3** The Parties shall compensate each other for Transit Traffic Service as follows:
 - 10.3.1 For Local Traffic and IntraLATA Toll Traffic originating from ITC^DeltaCom that is delivered over the Transit Traffic Service, ITC^DeltaCom will pay to BellSouth the applicable Tandem Switching and/or Interoffice Transport charges set forth in Attachment 11 to the Agreement. Charges for services provided by the Parties to a third party carrier(s) shall be assessed on a meet point basis, consistent with the terms of Section 9 hereof.
 - **10.3.2** Except as provided in Section 10.3.3 hereafter, transit charges shall be assessed upon the originating carrier, and shall not be imposed on the terminating carrier.
 - **10.3.3** Transit charges associated with the provision of toll-free services (e.g. 800/888/877) shall be imposed upon the terminating carrier and shall not be assessed on the originating carrier.

11.0 Packet Switched Network Interconnection/Frame Relay

The Parties agree to interconnect their local data services networks for the exchange of Frame Relay Services ("FRS") traffic.

- 11.1 The following provisions will apply only to Frame Relay Service and Exchange Access Frame Relay Service in those states where traffic is being exchanged between ITC^DeltaCom and BellSouth Frame Relay Switches in the same LATA.
- 11.1.1 The Parties agree to establish two-way Frame Relay facilities between their respective Frame Relay Switches to the mutually-agreed upon Frame Relay Service point(s) of interconnection ("POIs") within the LATA. All POIs shall be within the same Frame Relay Network Serving Area as defined in Section A40 of BellSouth's General Subscriber Services Tariff.
- 11.1.2 Upon the request of either Party, such interconnection will be established where BellSouth and ITC^DeltaCom have Frame Relay Switches in the same LATA. Where there are multiple Frame Relay switches in the central office of a Party, an interconnection with any one of the switches will be considered an interconnection with all of the switches at that central office for purposes of routing packet traffic.
- **11.1.3** The Parties agree to provision local and IntraLATA Frame Relay Service and Exchange Access Frame Relay Service (both intrastate and interstate) over Frame Relay Trunks between the respective Frame Relay switches and the POIs.
- **11.1.4** The Parties agree to assess each other reciprocal charges for the facilities that each provides to the other according to the Percent Local Circuit Use ("PLCU") factor PLCU, determined as follows:
 - (i) Frame Relay framed packet data is transported within Virtual Circuits ("VC"). For the purposes of calculating the PLCU, if all the data packets transported within a VC remain within the LATA, then consistent with the local definitions in this Agreement, the traffic on that VC is local ("Local VC").
 - (ii) If the originating and terminating locations of the two way packet data traffic are not in the same LATA, the traffic on that VC is interLATA.
 - (iii) The PLCU shall be determined by dividing the total number of Local VCs, by the total number of VCs on each Frame Relay facility at the end of the reporting period. The Parties agree to renegotiate the method for determining PLCU, at either Parties'

request, and within 90 days, if either Party notifies the other that it has found that this method does not adequately represent the PLCU. (iv) If there are no VCs on a facility when it is billed, the PLCU will be zero.

- 11.1.5 BellSouth will provide the Frame Relay Trunk(s) between the Parties' respective Frame Relay Switches. The Parties will be compensated as follows- BellSouth will invoice, and ITC^DeltaCom will pay, the total non-recurring and recurring charges for the trunk facility. ITC^DeltaCom will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed charges for the trunk facility by one-half of ITC^DeltaCom's PLCU.
 - 11.1.16 Each Party will provide a Frame Relay network-to-network interface ("NNI") port to the other Party for each trunk facility provided pursuant to 11.1.5 above. Compensation for NNI ports shall be based upon the NNI rates set forth in the BellSouth F.C.C Tariff No. 1. Pursuant to that tariff, ITC^DeltaCom may select a month-to-month or term rate structure for the NNI ports BellSouth provides to ITC^DeltaCom. Whatever rate structure ITC^DeltaCom selects shall be deemed to be the same rate structure that applies to the NNI port ITC^DeltaCom provides to BellSouth. There shall be no termination liability to either party for the local portion of the NNI port as determined by the ITC^DeltaCom PLCU at the time of termination.
- **11.1.7** Compensation for the NNI ports shall be calculated as follows:
 - 11.1.7.1 For NNI ports provided by BellSouth to ITC^DeltaCom, BellSouth will invoice, and ITC^DeltaCom will pay, the total nonrecurring and recurring charges for the NNI port. ITC^DeltaCom will then invoice, and BellSouth will pay, an amount calculated by multiplying the BellSouth billed non-recurring and recurring charges for the NNI port by one-half of ITC^DeltaCom's PLCU.
 - 11.1.7.2 For NNI ports provided by ITC^DeltaCom to BellSouth, ITC^DeltaCom will invoice, and BellSouth will pay, the total non-recurring and recurring charges for the NNI port. BellSouth will then invoice, and ITC^DeltaCom will pay, an amount determined as follows: ITC^DeltaCom's combined interLATA and local usage will be calculated by subtracting one-half of ITC^DeltaCom's PLCU factor from one hundred percent. The difference will then be multiplied by the total charges initially billed by ITC^DeltaCom for the NNI port. BellSouth will then invoice, and ITC^DeltaCom will pay, this amount to BellSouth.

- 11.1.8 A Permanent Virtual Circuit ("PVC") is a logical channel from a frame relay network interface (e.g., NNI or User Network Interface) to another frame relay network interface. A PVC is created when a Data Link Channel Identifier ("DLCI") is mapped together with another DLCI. Neither Party will charge the other Party any DLCI or Committed Information Rate ("CIR") charges for the PVC from its Frame Relay switch to its own subscriber's premises.
- **11.1.9** For the PVC between the ITC^DeltaCom and BellSouth Frame Relay switches, compensation for the DLCI and CIR charges are based upon the rates in the BellSouth FCC Tariff No. 1. Compensation for PVC and CIR rate elements shall be calculated as follows:
- 11.1.10 For PVCs between the BellSouth Frame Relay switch and the ITC^DeltaCom Frame Relay switch, BellSouth will invoice, and ITC^DeltaCom will pay, the total nonrecurring and recurring DLCI and CIR charges. If the VC is a Local VC, ITC^DeltaCom will invoice and BellSouth will pay, 100% of the DLCI and CIR charges initially billed by BellSouth for that PVC. If the VC is not local, no compensation will be paid to ITC^DeltaCom for the PVC.
- **11.1.11**Each Party will compensate the other Party for any applicable Feature Change or Transfer of Service Charges as set forth in BellSouth's Tariff F.C.C. No. 1. The Parties agree to limit the sum of the CIR for the VCs on a given NNI port to not more than two times the port speed.
- 11.1.12 Except as expressly provided herein, this Agreement does not address or alter in any way either Party's provision of Exchange Access Frame Relay Service or interLATA Frame Relay Service. All charges by each Party to the other for carriage of Exchange Access Frame Relay Service or interLATA Frame Relay Service are included in the BellSouth access tariffs.
- **11.1.13** Until such time as BellSouth obtains authority to provide in-region, interLATA service, ITC^DeltaCom will identify and report its PLCU to BellSouth on a quarterly basis.
- 11.1.14 Either Party may request a review or audit of the various service components, including but not limited to a Party's determination of its PLCU, consistent with the provisions of section E2 of the BellSouth State Access Services tariffs or Section 2 of the BellSouth FCC No. 1 Tariff.
- 11.1.15 If during the term of this Agreement, BellSouth obtains authority to provide in-region, interLATA service, the Parties shall renegotiate the provisions of Section 11.1.5, 11.1.7, 11.1.8-11.1.10, and 11.1.13 to account for BellSouth's PLCU. In the event the parties are unable to reach agreement within one hundred eighty (180) days of the date BellSouth receives

- interLATA authority, the matter shall be resolved pursuant to the dispute resolution provisions set forth in this agreement.
- **11.1.16** If during the term of this Agreement, BellSouth makes available, to an affiliate or any other telecommunications carrier, Frame Relay interconnection on rates, terms and conditions different than those provided for in this Section 11, then ITC^DeltaCom shall be entitled, at its option, to replace any part of this Section 11 with such rates, terms, and conditions.

Page 1

Attachment 4

Physical Collocation

BELLSOUTH

PHYSICAL COLLOCATION

1. Scope of Attachment

- 1.1 The rates, terms, and conditions contained within this Attachment shall only apply when ITC^DeltaCom is physically collocated as a sole occupant or as a Host within a Premises location pursuant to this Attachment. BellSouth Premises include BellSouth Central Offices and Serving Wire Centers (hereinafter "Premises"). This Attachment is applicable to Premises owned or leased by BellSouth. However, if the Premises occupied by BellSouth is leased by BellSouth from a third party, special considerations and intervals may apply in addition to the terms and conditions of this Attachment.
- Right to Occupy. BellSouth shall offer to ITC^DeltaCom collocation on rates, terms, and conditions that are just, reasonable, non-discriminatory and consistent with the rules of the Federal Communications Commission ("FCC"). Subject to the rates, terms and conditions of this Attachment where space is available and it is technically feasible, BellSouth will allow ITC^DeltaCom to occupy that certain area designated by BellSouth within a BellSouth Premises, or on BellSouth property upon which the BellSouth Premises is located, of a size which is specified by ITC^DeltaCom and agreed to by BellSouth (hereinafter "Collocation Space"). The necessary rates, terms and conditions for BellSouth locations other than BellSouth Premises shall be negotiated upon request for collocation at such location(s).
- 1.2.1 Neither BellSouth nor any of BellSouth's affiliates may reserve space for future use on more preferential terms than those set forth below.
- 1.2.1.1 In all states other than Florida, the size specified by ITC^DeltaCom may contemplate a request for space sufficient to accommodate ITC^DeltaCom's growth within a two-year period.
- 1.2.1.2 In the state of Florida, the size specified by ITC^DeltaCom may contemplate a request for space sufficient to accommodate ITC^DeltaCom's growth within an eighteen (18) month period.
- 1.3 Space Allocation. BellSouth shall attempt to accommodate ITC^DeltaCom's requested preferences if any. In allocating Collocation Space, BellSouth shall not materially increase ITC^DeltaCom's cost or materially delay ITC^DeltaCom's occupation and use of the Collocation Space, shall not assign Collocation Space that will impair the quality of service or otherwise limit the service the ITC^DeltaCom wishes to offer, and shall not reduce unreasonably the total space available for physical collocation or preclude unreasonably physical collocation within the Premises. Space shall not be available for collocation if it is: (a) physically occupied by non-obsolete equipment; (b) assigned to another collocator; (c) used to provide physical access to

occupied space; (d) used to enable technicians to work on equipment located within occupied space; (e) properly reserved for future use, either by BellSouth or by another carrier; or (f) essential for the administration and proper functioning of BellSouth's Premises. BellSouth may segregate collocation space and require separate entrances in accordance with FCC rules.

- 1.4 <u>Space Reclamation.</u> In the event of space exhaust within a Central Office Premises, BellSouth may include in its documentation for the Petition for Waiver filing any unutilized space in the Central Office Premises. ITC^DeltaCom will be responsible for any justification of unutilized space within its space, if the appropriate state commission requires such justification.
- 1.5 <u>Use of Space</u>. ITC^DeltaCom shall use the Collocation Space for the purposes of installing, maintaining and operating ITC^DeltaCom's equipment (to include testing and monitoring equipment) necessary for interconnection with BellSouth services and facilities or for accessing BellSouth unbundled network elements for the provision of telecommunications services, as specifically set forth in this Attachment. The Collocation Space may be used for no other purposes except as specifically described herein or in any amendment hereto.
- 1.6 <u>Rates and Charges</u>. ITC^DeltaCom agrees to pay the rates and charges identified in Attachment 11 of this Interconnection Agreement.
- 1.7 <u>Due Dates</u>. If any due date contained in this Attachment falls on a weekend or National holiday, then the due date will be the next business day thereafter.
- 1.8 The parties agree to comply with all applicable federal, state, county, local and administrative laws, rules, ordinances, regulations and codes in the performance of their obligations hereunder.

2. Space Availability Report

- 2.1 Space Availability Report. Upon request from ITC^DeltaCom, BellSouth will provide a written report ("Space Availability Report") describing in detail the space that is available for collocation and specifying the amount of Collocation Space available at the Premises requested, the number of collocators present at the Premises, any modifications in the use of the space since the last report on the Premises requested and the measures BellSouth is taking to make additional space available for collocation arrangements. A Space Availability Report does not reserve space at the Premises.
- 2.1.1 The request from ITC^DeltaCom for a Space Availability Report must be written and must include the Premises street address, located in the Local Exchange Routing Guide and Common Language Location Identification ("CLLI") code of the Premises. CLLI code information is located in the National Exchange Carriers Association (NECA) Tariff FCC No. 4.

Page 4

2.1.2 BellSouth will respond to a request for a Space Availability Report for a particular Premise within ten (10) calendar days of receipt of such request. BellSouth will make best efforts to respond in ten (10) calendar days to such a request when the request includes from two (2) to five (5) Premises within the same state. The response time for requests of more than five (5) Premises shall be negotiated between the Parties. If BellSouth cannot meet the ten calendar day response time, BellSouth shall notify ITC^DeltaCom and inform ITC^DeltaCom of the time frame under which it can respond.

3. Collocation Options

- 3.1 <u>Cageless.</u> BellSouth shall allow ITC^DeltaCom to collocate ITC^DeltaCom's equipment and facilities without requiring the construction of a cage or similar structure. BellSouth shall allow ITC^DeltaCom to have direct access to ITC^DeltaCom's equipment and facilities. BellSouth shall make cageless collocation available in single bay increments. Except where ITC^DeltaCom's equipment requires special technical considerations (e.g., special cable racking, isolated ground plane, etc.), BellSouth shall assign cageless Collocation Space in conventional equipment rack lineups where feasible. For equipment requiring special technical considerations, ITC^DeltaCom must provide the equipment layout, including spatial dimensions for such equipment pursuant to generic requirements contained in Telcordia GR-63-Core, and shall be responsible for compliance with all special technical requirements associated with such equipment.
- 3.2 Caged. At ITC^DeltaCom's expense, ITC^DeltaCom may arrange with a Supplier certified by BellSouth ("Certified Supplier") to construct a collocation arrangement enclosure in accordance with BellSouth's guidelines and specifications prior to starting equipment installation. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard enclosure specification, ITC^DeltaCom and ITC^DeltaCom's Certified Supplier must comply with the more stringent local building code requirements. ITC^DeltaCom's Certified Supplier shall be responsible for filing and receiving any and all necessary permits and/or licenses for such construction. BellSouth shall cooperate with ITC^DeltaCom and provide, at ITC^DeltaCom's expense, the documentation, including existing building architectural drawings, enclosure drawings, and specifications required and necessary for ITC^DeltaCom to obtain the zoning, permits and/or other licenses. ITC^DeltaCom's Certified Supplier shall bill ITC^DeltaCom directly for all work performed for ITC^DeltaCom pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the ITC^DeltaCom's Certified Supplier. ITC^DeltaCom must provide the local BellSouth building contact with two Access Keys used to enter the locked enclosure. Except in case of emergency, BellSouth will not access ITC^DeltaCom's locked enclosure prior to notifying

ITC^DeltaCom. Upon request, BellSouth shall construct the enclosure for ITC^DeltaCom.

- 3.2.1 BellSouth may elect to review ITC^DeltaCom's plans and specifications prior to allowing construction to start to ensure compliance with BellSouth's guidelines and specifications. Notification to ITC^DeltaCom indicating BellSouth's desire to execute this review will be provided in BellSouth's response to the Initial Application, if ITC^DeltaCom has indicated their desire to construct their own enclosure. If ITC^DeltaCom's Initial Application does not indicate their desire to construct their own enclosure, but their subsequent firm order does indicate their desire to construct their own enclosure, then notification to review will be given within ten (10) calendar days after the Firm Order date. . BellSouth shall complete its review within fifteen (15) calendar days after the receipt of the plans and specifications. Regardless of whether or not BellSouth elects to review ITC^DeltaCom's plans and specifications, BellSouth reserves the right to inspect the enclosure after construction to make sure it is constructed according to the submitted plans and specifications and/or BellSouth's guidelines and specifications, as applicable. BellSouth shall require ITC^DeltaCom to remove or correct within seven (7) calendar days at ITC^DeltaCom's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth guidelines and specifications.
- Shared (Subleased) Caged Collocation. ITC^DeltaCom may allow other telecommunications carriers to share ITC^DeltaCom's caged collocation arrangement pursuant to terms and conditions agreed to by ITC^DeltaCom ("Host") and other telecommunications carriers ("Guests") and pursuant to this section, except where the BellSouth Premises is located within a leased space and BellSouth is prohibited by said lease from offering such an option. ITC^DeltaCom shall notify BellSouth in writing upon execution of any agreement between the Host and its Guest within ten (10) calendar days of its execution and prior to any Firm Order. Further, such notice shall include the name of the Guest(s) and the term of the agreement, and shall contain a certification by ITC^DeltaCom that said agreement imposes upon the Guest(s) the same terms and conditions for Collocation Space as set forth in this Attachment between BellSouth and ITC^DeltaCom.
- 3.3.1 ITC^DeltaCom, as the Host shall be the sole interface and responsible Party to BellSouth for the assessment and billing of rates and charges contained within this Attachment and for the purposes of ensuring that the safety and security requirements of this Attachment are fully complied with by the Guest, its employees and agents. BellSouth shall provide ITC^DeltaCom with a proration of the costs of the collocation space based on the number of collocators and the space used by each. In all states other than Florida, and in addition to the foregoing, ITC^DeltaCom shall be the responsible party to BellSouth for the purpose of submitting Applications for initial and additional equipment placement of Guest. In Florida the Guest may directly submit initial and additional equipment placement applications using the Host's access carrier name abbreviation (ACNA). A separate Guest application shall require the

assessment of an Initial or Subsequent Application Fee, as set forth in Attachment 11. Notwithstanding the foregoing, Guest may arrange directly with BellSouth for the provision of the interconnecting facilities between BellSouth and Guest and for the provision of the services and access to unbundled network elements.

- 3.3.2 ITC^DeltaCom shall indemnify and hold harmless BellSouth from any and all claims, actions, causes of action, of whatever kind or nature arising out of the presence of ITC^DeltaCom's Guests in the Collocation Space except to the extent caused by BellSouth's sole negligence, gross negligence, or willful misconduct.
- Adjacent Collocation. Subject to technical feasibility and space availability, BellSouth will permit adjacent collocation arrangements ("Adjacent Arrangement") on the Premises' property where physical collocation space within the Premises is legitimately exhausted, where the Adjacent Arrangement does not interfere with access to existing or planned structures or facilities on the Premises property. The Adjacent Arrangement shall be constructed or procured by ITC^DeltaCom and in conformance with BellSouth's design and construction specifications. Further, ITC^DeltaCom shall construct, procure, maintain and operate said Adjacent Arrangement(s) pursuant to all of the rates, terms and conditions set forth in this Attachment.
- 3.4.1 Should ITC^DeltaCom elect such option, ITC^DeltaCom must arrange with a Certified Supplier to construct an Adjacent Arrangement structure in accordance with BellSouth's guidelines and specifications. BellSouth will provide guidelines and specifications upon request. Where local building codes require enclosure specifications more stringent than BellSouth's standard specification, ITC^DeltaCom and ITC^DeltaCom's Certified Supplier must comply with the more stringent local building code requirements. ITC^DeltaCom's Certified Supplier shall be responsible for filing and receiving any and all necessary zoning, permits and/or licenses for such construction. ITC^DeltaCom's Certified Supplier shall bill ITC^DeltaCom directly for all work performed for ITC^DeltaCom pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by ITC^DeltaCom's Certified Supplier. ITC^DeltaCom must provide the local BellSouth building contact with two cards, keys or other access device used to enter the locked enclosure. Except in cases of emergency, BellSouth shall not access ITC^DeltaCom's locked enclosure prior to notifying ITC^DeltaCom.
- 3.4.2 ITC^DeltaCom must submit its plans and specifications to BellSouth with its Firm Order. BellSouth shall review ITC^DeltaCom's plans and specifications prior to construction of an Adjacent Arrangement(s) to ensure compliance with BellSouth's guidelines and specifications. BellSouth shall complete its review within fifteen (15) calendar days after receipt of plans and specifications. BellSouth will have the right to inspect the Adjacent Arrangement during and after construction to make sure it is constructed according to the submitted plans and specifications. BellSouth shall require ITC^DeltaCom to remove or correct within seven (7) calendar days at ITC^DeltaCom's expense any structure that does not meet these plans and specifications or, where applicable, BellSouth's guidelines and specifications.

- 3.4.3 ITC^DeltaCom shall provide a concrete pad, the structure housing the arrangement, heating/ventilation/air conditioning ("HVAC"), lighting, and all facilities that connect the structure (i.e. racking, conduits, etc.) to the BellSouth point of demarcation. At ITC^DeltaCom's option, and where the local authority having jurisdiction permits, BellSouth shall provide an AC power source and access to physical collocation services and facilities subject to the same nondiscriminatory requirements as applicable to any other physical collocation arrangement. In Louisiana, BellSouth will provide DC power to Adjacent Collocation sites where technically feasible, as that term has been defined by the FCC. ITC^DeltaCom's Certified Supplier shall be responsible, at ITC^DeltaCom's expense, for filing and receiving any and all necessary zoning, permits and/or licenses for such arrangement. BellSouth shall allow Shared (Subleased) Caged Collocation within an Adjacent Arrangement pursuant to the terms and conditions set forth herein.
- 3.5 Co-carrier cross-connect (CCXC). The primary purpose of collocating CLEC equipment is to interconnect with BellSouth's network or access BellSouth's unbundled network elements for the provision of telecommunications services. In addition to, and not in lieu of, obtaining said interconnection, ITC^DeltaCom may directly connect to other Interconnectors within the designated premises (including to its other virtual or physical collocated arrangements) through facilities owned by ITC^DeltaCom. Such Interconnectors Collocation Agreement must contain co-carrier cross connect language.
- 3.5.1 Such connections to other carriers may be made using either optical or electrical facilities. ITC^DeltaCom may deploy such optical or electrical connections directly between its own facilities and the facilities of other CLEC(s) without being routed through BellSouth equipment. ITC^DeltaCom may not self provision CCXC on any BellSouth distribution frame, Pot Bay, DSX or LGX. ITC^DeltaCom is responsible for ensuring the integrity of the signal.

ITC^DeltaCom shall be responsible for obtaining authorization from the other CLEC(s) involved. ITC^DeltaCom must use a BellSouth Certified Supplier to place the CCXC. There will be a recurring charge per linear foot of common cable support structure used. ITC^DeltaCom-provisioned CCXC shall utilize common cable support structure. In the case of two contiguous collocation arrangements, ITC^DeltaCom may have the option of constructing its own dedicated support structure. In cases where ITC^DeltaCom's equipment and the equipment of the other interconnector are located in contiguous Collocation Spaces, ITC^DeltaCom will have the option to deploy co-carrier cross connects between the sets of equipment. Cable support charges shall be assessed per linear foot of support structure used. Within BellSouth Premises, at ITC^DeltaCom's request, BellSouth will permit ITC^DeltaCom and other such CLECs to construct their own cross-connect facilities, and to connect to other physical CLECs using copper (or ABAM or coaxial as appropriate) or optical facilities between collocated equipment located within the same BellSouth Premises.

3.5.2 ITC^DeltaCom may order CCXC in its initial Application. In the Application, ITC^DeltaCom must include the type of cross connect facilities to be used, how those facilities are to be routed, and authorization from all other CLECs involved. If ITC^DeltaCom, or ITC^DeltaCom's Guest(s) in a shared arrangement, desires to order CCXC after the Bona Fide Firm Order, then ITC^DeltaCom must submit to BellSouth a complete Subsequent Application containing the same CCXC information required in an initial Application. If ITC^DeltaCom submits a Subsequent Application for CCXC only, then such fee as set forth in Attachment 11 shall apply. However, if ITC^DeltaCom request other modifications (e.g., HVAC, increase in Power, etc.) to its Collocation Space in addition to its request for CCXC then a "Subsequent Application" fee, as described in Section 6.3 of this Attachment, shall apply

4. Occupancy

- 4.1 Occupancy. BellSouth will notify ITC^DeltaCom in writing that the Collocation Space is ready for occupancy ("Space Ready Date"). ITC^DeltaCom will schedule and complete an acceptance walkthrough of each Collocation Space with BellSouth within fifteen (15) days of BellSouth's notifying ITC^DeltaCom that the collocation space is ready for occupancy. In the event that ITC^DeltaCom fails to complete an acceptance walkthrough within this fifteen (15) day interval, the Collocation Space shall be deemed accepted by ITC^DeltaCom and billing will commence on the sixteenth day after BellSouth releases the collocation space. ITC^DeltaCom must notify BellSouth in writing that collocation equipment installation is complete and is operational with BellSouth's network. BellSouth may, at its option, not accept orders for cross connects until receipt of such notice. For purposes of this paragraph, ITC^DeltaCom's telecommunications equipment will be deemed operational when cross-connected to BellSouth's network for the purpose of service provision.
- 4.2 <u>Termination of Occupancy</u>. In addition to any other provisions addressing termination of occupancy in this Attachment, ITC^DeltaCom may terminate occupancy in a particular Collocation Space by submitting a Subsequent Application requesting termination of occupancy. A Subsequent Application Fee will not apply for termination of occupancy. BellSouth may terminate ITC^DeltaCom's right to occupy the Collocation Space in the event ITC^DeltaCom fails to comply with any provision of this Agreement.
- 4.2.1 Upon termination of occupancy, ITC^DeltaCom at its expense shall remove its equipment and other property from the Collocation Space. ITC^DeltaCom shall have thirty (30) calendar days from the termination date to complete such removal, including the removal of all equipment and facilities of ITC^DeltaCom's Guests, unless ITC^DeltaCom's Guest has assumed responsibility for the collocation space housing the Guest's equipment and executed the documentation required by BellSouth prior to such removal date. ITC^DeltaCom shall continue payment of monthly fees to BellSouth until such date as ITC^DeltaCom, and if applicable ITC^DeltaCom's Guest, has fully vacated the Collocation Space and the Space Relinquish Form has been

accepted by BellSouth.. Should ITC^DeltaCom or ITC^DeltaCom's Guest fail to vacate the Collocation Space within thirty (30) calendar days from the termination date, BellSouth shall have the right to remove the equipment and other property of ITC^DeltaCom or ITC^DeltaCom's Guest at ITC^DeltaCom's expense and with no liability for damage or injury to ITC^DeltaCom or ITC^DeltaCom's Guest's property unless caused by the gross negligence or intentional misconduct of BellSouth. Upon termination of ITC^DeltaCom's right to occupy Collocation Space, ITC^DeltaCom shall surrender such Collocation Space to BellSouth in the same condition as when first occupied by ITC^DeltaCom except for ordinary wear and tear, unless otherwise agreed to by the Parties. ITC^DeltaCom or ITC^DeltaCom's BellSouth Certified Supplier shall be responsible for updating and making any necessary changes to BellSouth's records as required by BellSouth's guidelines and specifications including but not limited to Central Office Record Drawings and ERMA Records. ITC^DeltaCom shall be responsible for the cost of removing any enclosure, together with all support structures (e.g., racking, conduits, power cables, etc.), at the termination of occupancy and restoring the grounds to their original condition.

5. <u>Use of Collocation Space</u>

- Equipment Type. BellSouth permits the collocation of any type of equipment necessary for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services, as the term "necessary" is defined by FCC 47 C.F.R. Section 51.323 (b). The primary purpose and function of any equipment collocated in a Premises must be for interconnection to BellSouth's network or for access to BellSouth's unbundled network elements in the provision of telecommunications services.
- 5.1.1 Examples of equipment that would not be considered necessary include but are not limited to: Traditional circuit switching equipment, equipment used exclusively for call-related databases, computer servers used exclusively for providing information services, operations support system (OSS) equipment used to support CLEC network operations, equipment that generates customer orders, manages trouble tickets or inventory, or stores customer records in centralized databases, etc. BellSouth will determine upon receipt of an application if the requested equipment is necessary based on the criteria established by the FCC. Multifunctional equipment placed on BellSouth's Premises must not place any greater relative burden on BellSouth's property than comparable single-function equipment. BellSouth reserves the right to permit collocation of any equipment on a nondiscriminatory basis.
- 5.1.2 Such equipment must at a minimum meet the following Bellcore (Telcordia) Network Equipment Building Systems (NEBS) General Equipment Requirements: Criteria Level 1 requirements as outlined in the Bellcore (Telcordia) Special Report SR-3580, Issue 1; equipment design spatial requirements per GR-63-CORE, Section 2; thermal heat dissipation per GR-063-CORE, Section 4, Criteria 77-79; acoustic noise per GR-063-CORE, Section 4, Criterion 128, and National Electric Code standards. Except

where otherwise required by a Commission, BellSouth shall comply with the applicable FCC rules relating to denial of collocation based on ITC^DeltaCom's failure to comply with this section.

- 5.1.3 ITC^DeltaCom shall not request more DS0, DS1, DS3 and optical terminations for a collocation arrangement than the total port or termination capacity of the equipment physically installed in the arrangement. The total capacity of the equipment collocated in the arrangement will include equipment contained in the application in question as well as equipment already placed in the arrangement. If full network termination capacity of the equipment being installed is not requested in the application, additional network terminations for the installed equipment will require the submission of another application. In the event that ITC^DeltaCom submits an application for terminations that exceed the total capacity of the collocated equipment, ITC^DeltaCom will be informed of the discrepancy and will be required to submit a revision to the application.
- 5.2 ITC^DeltaCom shall not use the Collocation Space for marketing purposes nor shall it place any identifying signs or markings outside the Collocation Space or on the grounds of the Premises.
- 5.3 ITC^DeltaCom shall place a plaque or other identification affixed to ITC^DeltaCom's equipment necessary to identify ITC^DeltaCom's equipment, including a list of emergency contacts with telephone numbers.
- 5.4 Entrance Facilities. ITC^DeltaCom may elect to place ITC^DeltaCom-owned or ITC^DeltaCom-leased fiber entrance facilities into the Collocation Space. BellSouth will designate the point of interconnection in close proximity to the Premises building housing the Collocation Space, such as an entrance manhole or a cable vault, which are physically accessible by both Parties. ITC^DeltaCom will provide and place fiber cable at the point of entrance of sufficient length to be pulled through conduit and into the splice location. ITC^DeltaCom will provide and install a sufficient length of fire retardant riser cable, to which the entrance cable will be spliced by BellSouth, which will extend from the splice location to ITC^DeltaCom's equipment in the Collocation Space. In the event ITC^DeltaCom utilizes a non-metallic, riser-type entrance facility, a splice will not be required. ITC^DeltaCom must contact BellSouth for instructions prior to placing the entrance facility cable in the manhole. ITC^DeltaCom is responsible for maintenance of the entrance facilities. At ITC^DeltaCom's option BellSouth will accommodate where technically feasible a microwave entrance facility pursuant to separately negotiated terms and conditions. In the case of adjacent collocation, unless BellSouth determines that limited space is available for the entrance facilities, copper facilities may be used between the adjacent collocation arrangement and the central office demarcation point.
- 5.4.1 <u>Dual Entrance</u>. BellSouth will provide at least two interconnection points at each Premise where there are at least two such interconnection points available and where

capacity exists. Upon receipt of a request for physical collocation under this Attachment, BellSouth shall provide ITC^DeltaCom with information regarding BellSouth's capacity to accommodate dual entrance facilities. If conduit in the serving manhole(s) is available and is not reserved for another purpose for utilization within 12 months of the receipt of an application for collocation, BellSouth will make the requested conduit space available for installing a second entrance facility to ITC^DeltaCom's arrangement. The location of the serving manhole(s) will be determined at the sole discretion of BellSouth. Where dual entrance is not available due to lack of capacity, BellSouth will so state in the Application Response.

- Shared Use. ITC^DeltaCom may utilize spare capacity on an existing interconnector entrance facility for the purpose of providing an entrance facility to ITC^DeltaCom's collocation arrangement within the same BellSouth Premises. BellSouth shall allow the splice, provided that the fiber is non-working fiber. ITC^DeltaCom must arrange with BellSouth for BellSouth to splice the ITC^DeltaCom provided riser cable to the spare capacity on the entrance facility. The rates set forth in Attachment 11 will apply. If ITC^DeltaCom desires to allow another CLEC to use its entrance facilities, additional rates, terms and conditions will apply and shall be negotiated between the parties.
- 5.5 Demarcation Point. BellSouth will designate the point(s) of demarcation between ITC^DeltaCom's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For 2-wire and 4-wire connections to BellSouth's network, the demarcation point shall be a common block on the BellSouth designated conventional distributing frame (CDF). ITC^DeltaCom shall be responsible for providing, and a supplier certified by BellSouth ("Certified Supplier") shall be responsible for installing and properly labeling/stenciling, the common block, and necessary cabling pursuant to Section 6. For all other terminations BellSouth shall designate a demarcation point on a per arrangement basis. ITC^DeltaCom or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.6, following, and may self-provision crossconnects that may be required within the Collocation Space to activate service requests. At ITC^DeltaCom's option and expense, a Point of Termination ("POT") bay or frame may be placed in the Collocation Space, but will not serve as the demarcation point. ITC^DeltaCom must make arrangements with a Certified Supplier for such placement.
- In Tennessee, BellSouth will designate the point(s) of demarcation between ITC^DeltaCom's equipment and/or network and BellSouth's network. Each Party will be responsible for maintenance and operation of all equipment/facilities on its side of the demarcation point. For connections to BellSouth's network, the demarcation point shall be an ITC^DeltaCom provided Point of Termination Bay (POT Bay) in a common area within the Premises. ITC^DeltaCom shall be responsible for providing, and a supplier certified by BellSouth ("ITC^DeltaCom's Certified Supplier") shall be

responsible for installing and properly labeling, the POT Bay as well as the necessary cabling between ITC^DeltaCom's collocation space and the demarcation point. ITC^DeltaCom or its agent must perform all required maintenance to equipment/facilities on its side of the demarcation point, pursuant to Section 5.6, following, and may self-provision cross-connects that may be required within the Collocation Space to activate service requests. BellSouth will negotiate alternative rates, terms and conditions related to the demarcation point in Tennessee in the event that ITC^DeltaCom desires to avoid the use of an intermediary device as contemplated by the Tennessee Regulatory Authority.

- 5.6 ITC^DeltaCom's Equipment and Facilities. ITC^DeltaCom, or if required by this Attachment, ITC^DeltaCom's Certified Supplier, is solely responsible for the design, engineering, installation, testing, provisioning, performance, monitoring, maintenance and repair of the equipment and facilities used by ITC^DeltaCom which must be performed in compliance with all applicable BellSouth policies and guidelines. Such equipment and facilities may include but are not limited to cable(s), equipment, and point of termination connections. ITC^DeltaCom and its selected Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564.
- BellSouth's Access to Collocation Space. From time to time BellSouth may require access to the Collocation Space. BellSouth retains the right to access such space for the purpose of making BellSouth equipment and building modifications (e.g., running, altering or removing racking, ducts, electrical wiring, HVAC, and cables). BellSouth will give notice to ITC^DeltaCom at least 48 hours before access to the Collocation Space is required. ITC^DeltaCom may elect to be present whenever BellSouth performs work in the Collocation Space. The Parties agree that ITC^DeltaCom will not bear any of the expense associated with this work.
- 5.8 Access. Pursuant to Section 11, ITC^DeltaCom shall have access to the Collocation Space twenty-four (24) hours a day, seven (7) days a week. ITC^DeltaCom agrees to provide the name and social security number or date of birth or driver's license number of each employee, contractor, or agents of ITC^DeltaCom or ITC^DeltaCom's Guests provided with access keys or devices ("Access Keys") prior to the issuance of said Access Keys. Key acknowledgement forms must be signed by ITC^DeltaCom and returned to BellSouth Access Management within 15 calendar days of ITC^DeltaCom's receipt. Failure to return properly acknowledged forms will result in the holding of subsequent requests until acknowledgements are current. Access Keys shall not be duplicated under any circumstances. ITC^DeltaCom agrees to be responsible for all Access Keys and for the return of all said Access Keys in the possession of ITC^DeltaCom employees, contractors, Guests, or agents after termination of the employment relationship, contractual obligation with ITC^DeltaCom or upon the termination of this Attachment or the termination of occupancy of an individual collocation arrangement.

- 5.8.1 BellSouth will permit one accompanied site visit to ITC^DeltaCom's designated collocation arrangement location after receipt of the Bona Fide Firm Order without charge to ITC^DeltaCom. ITC^DeltaCom must submit to BellSouth the completed Access Control Request Form for all employees or agents requiring access to the BellSouth Premises a minimum of 30 calendar days prior to the date ITC^DeltaCom desires access to the Collocation Space. In order to permit reasonable access during construction of the Collocation Space, ITC^DeltaCom may submit such a request at any time subsequent to BellSouth's receipt of the Bona Fide Firm Order. In the event ITC^DeltaCom desires access to the Collocation Space after submitting such a request but prior to access being approved, in addition to the first accompanied free visit, BellSouth shall permit ITC^DeltaCom to access the Collocation Space accompanied by a security escort at ITC^DeltaCom's expense. ITC^DeltaCom must request escorted access at least three (3) business days prior to the date such access is desired.
- Lost or Stolen Access Keys. ITC^DeltaCom shall notify BellSouth in writing within 24 hours of becoming aware in the case of lost or stolen Access Keys. Should it become necessary for BellSouth to re-key buildings or deactivate a card as a result of a lost Access Key(s) or for failure to return an Access Key(s), ITC^DeltaCom shall pay for all reasonable costs associated with the re-keying or deactivating the card.
- 5.10 Interference or Impairment. Notwithstanding any other provisions of this Attachment, ITC^DeltaCom shall not use any product or service provided under this Agreement, any other service related thereto or used in combination therewith, or place or use any equipment or facilities in any manner that 1) significantly degrades, interferes with or impairs service provided by BellSouth or by any other entity or any person's use of its telecommunications service; 2) endangers or damages the equipment, facilities or other property of BellSouth or of any other entity or person; 3) compromises the privacy of any communications; or 4) creates an unreasonable risk of injury or death to any individual or to the public. If BellSouth reasonably determines that any equipment or facilities of ITC^DeltaCom violates the provisions of this paragraph, BellSouth shall give written notice to ITC^DeltaCom, which notice shall direct ITC^DeltaCom to cure the violation within forty-eight (48) hours of ITC^DeltaCom's actual receipt of written notice or, at a minimum, to commence curative measures within 24 hours and to exercise reasonable diligence to complete such measures as soon as possible thereafter. After receipt of the notice, the Parties agree to consult immediately and, if necessary, to inspect the arrangement.
- 5.10.1 Except in the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services, if ITC^DeltaCom fails to take curative action within 48 hours or if the violation is of a character which poses an immediate and substantial threat of damage to property, injury or death to any person, or any other significant degradation, interference or impairment of BellSouth's or another entity's service, then and only in that event BellSouth may take such action as it deems appropriate to correct the violation, including without limitation the interruption of electrical power to

ITC^DeltaCom's equipment. BellSouth will endeavor, but is not required, to provide notice to ITC^DeltaCom prior to taking such action and shall have no liability to ITC^DeltaCom for any damages arising from such action, except to the extent that such action by BellSouth constitutes willful misconduct.

- 5.10.2 For purposes of this Section, the term significantly degrade shall mean an action that noticeably impairs a service from a user's perspective. In the case of the deployment of an advanced service which significantly degrades the performance of other advanced services or traditional voice band services and ITC^DeltaCom fails to take curative action within 48 hours then BellSouth will establish before the relevant Commission that the technology deployment is causing the significant degradation. Any claims of network harm presented to ITC^DeltaCom or, if subsequently necessary, the relevant Commission must be supported with specific and verifiable information. Where BellSouth demonstrates that a deployed technology is significantly degrading the performance of other advanced services or traditional voice band services, ITC^DeltaCom shall discontinue deployment of that technology and migrate its customers to technologies that will not significantly degrade the performance of other such services. Where the only degraded service itself is a known disturber, and the newly deployed technology satisfies at least one of the criteria for a presumption that is acceptable for deployment under section 47 C.F.R. 51.230, the degraded service shall not prevail against the newly-deployed technology.
- 5.11 Personalty and its Removal. Facilities and equipment placed by ITC^DeltaCom in the Collocation Space shall not become a part of the Collocation Space, even if nailed, screwed or otherwise fastened to the Collocation Space, but shall retain their status as personal property and may be removed by ITC^DeltaCom at any time. Any damage caused to the Collocation Space by ITC^DeltaCom's employees, agents or representatives during the removal of such property shall be promptly repaired by ITC^DeltaCom at its expense.
- Alterations. In no case shall ITC^DeltaCom or any person acting on behalf of ITC^DeltaCom make any rearrangement, modification, improvement, addition, or other alteration which could affect in any way space, power, HVAC, and/or safety considerations to the Collocation Space or the BellSouth Premises without the written consent of BellSouth, which consent shall not be unreasonably withheld. The cost of any such specialized alterations shall be paid by ITC^DeltaCom. Any such material rearrangement, modification, improvement, addition, or other alteration shall require a Subsequent Application and Subsequent Application Fee.
- Janitorial Service. ITC^DeltaCom shall be responsible for the general upkeep of the Collocation Space. ITC^DeltaCom shall arrange directly with a BellSouth Certified Supplier for janitorial services applicable to Caged Collocation Space. BellSouth shall provide a list of such suppliers on a site-specific basis upon request.

6. Ordering and Preparation of Collocation Space

- Should any state or federal regulatory agency impose procedures or intervals applicable to ITC^DeltaCom that are different from procedures or intervals set forth in this section, whether now in effect or that become effective after execution of this Agreement, those procedures or intervals shall supersede the requirements set forth herein for that jurisdiction for all applications submitted for the first time after the effective date thereof.
- Initial Application. For ITC^DeltaCom or ITC^DeltaCom's Guest(s) initial equipment placement, ITC^DeltaCom shall submit to BellSouth a Physical Expanded Interconnection Application Document ("Application"). The Application is Bona Fide when it is complete and accurate, meaning that all required fields on the application are completed with the appropriate type of information. An application fee will apply.
- 6.3 <u>Subsequent Application.</u> In the event ITC^DeltaCom or ITC^DeltaCom's Guest(s) desires to modify the use of the Collocation Space after Bona Fide Firm Order, ITC^DeltaCom shall complete an Application detailing all information regarding the modification to the Collocation Space ("Subsequent Application"). BellSouth shall determine what modifications, if any, to the Premises are required to accommodate the change requested by ITC^DeltaCom in the Application. Such necessary modifications to the Premises may include, but are not limited to, floor loading changes, changes necessary to meet HVAC requirements, changes to power plant requirements, equipment additions, etc.
- 6.3.1 Subsequent Application Fee. The application fee paid by ITC^DeltaCom for its request to modify the use of the Collocation Space shall be dependent upon the level of assessment needed for the modification requested. Where the Subsequent Application does not require assessment for provisioning or construction work by BellSouth, no Subsequent Application fee will be required. The fee for a Subsequent Application where the modification requested has limited effect (e.g., requires limited assessment and no capital expenditure by BellSouth) shall be the Subsequent Application Fee as set forth in Attachment 11. If the modification requires capital expenditure assessment, a full Application Fee shall apply. The Subsequent Application is Bona Fide when it is complete and accurate, meaning that all required fields on the Application are completed with the appropriate type of information.
- 6.4 Space Preferences. If ITC^DeltaCom has previously requested and received a Space Availability Report for the Premises, ITC^DeltaCom may submit up to three (3) space preferences on their application identifying specific space identification numbers as referenced on the Space Availability Report. In the event that BellSouth can not accommodate the ITC^DeltaCom's preference(s), ITC^DeltaCom may elect to accept the space allocated by BellSouth or may cancel its application and submit another application requesting additional preferences, which will be treated as a new application and an application fee will apply.

- 6.5 <u>Space Availability Notification.</u>
- Unless otherwise specified, BellSouth will respond to an application within ten (10) calendar days as to whether space is available or not available within a BellSouth Premises. BellSouth will also respond as to whether the Application is Bona Fide and if it is not Bona Fide the items necessary to cause the Application to become Bona Fide. If the amount of space requested is not available, BellSouth will notify ITC^DeltaCom of the amount of space that is available and no Application Fee shall apply. When BellSouth's response includes an amount of space less than that requested by ITC^DeltaCom, or differently configured, ITC^DeltaCom must resubmit its Application to reflect the actual space available.
- BellSouth will respond to a Florida Application within fifteen (15) calendar days as to whether space is available or not available within a BellSouth Premises. BellSouth will also respond as to whether the Application is Bona Fide and if it is not Bona Fide the items necessary to cause the Application to become Bona Fide. If a lesser amount of space than requested is available, BellSouth will provide an Application Response for the amount of space that is available and an Application Fee will be assessed. When BellSouth's Application Response includes an amount of space less than that requested by ITC^DeltaCom or differently configured, ITC^DeltaCom must amend its Application to reflect the actual space available prior to submitting Bona Fide Firm Order.
- BellSouth will respond to a Louisiana Application within ten (10) calendar days for space availability for one (1) to ten (10) Applications; fifteen (15) calendar days for eleven (11) to twenty (20) Applications; and for more than twenty (20) Applications, it is increased by five (5) calendar days for every five additional Applications received within five (5) business days. If the amount of space requested is not available, BellSouth will notify ITC^DeltaCom of the amount of space that is available and no Application Fee shall apply. When BellSouth's response includes an amount of space less than that requested by ITC^DeltaCom or differently configured, ITC^DeltaCom must resubmit its Application to reflect the actual space available. BellSouth will also respond as to whether the Application is Bona Fide and if it is not Bona Fide the items necessary to cause the Application to become Bona Fide.
- 6.6 <u>Denial of Application</u>. If BellSouth notifies ITC^DeltaCom that no space is available ("Denial of Application"), BellSouth will not assess an Application Fee. After notifying ITC^DeltaCom that BellSouth has no available space in the requested Premises, BellSouth will allow ITC^DeltaCom, upon request, to tour the entire Premises within ten (10) calendar days of such Denial of Application. In order to schedule said tour within ten (10) calendar days, the request for a tour of the Premises must be received by BellSouth within five (5) calendar days of the Denial of Application.
- 6.7 <u>Filing of Petition for Waiver</u>. Upon Denial of Application BellSouth will timely file a petition with the Commission pursuant to 47 U.S.C. § 251(c)(6). BellSouth shall

provide to the Commission any information requested by that Commission. Such information shall include which space, if any, BellSouth or any of BellSouth's affiliates have reserved for future use and a detailed description of the specific future uses for which the space has been reserved. Subject to an appropriate nondisclosure agreement or provision, BellSouth shall permit ITC^DeltaCom to inspect any floor plans or diagrams that BellSouth provides to the Commission.

- Maiting List. On a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. BellSouth will notify the telecommunications carriers on the waiting list that can be accommodated by the amount of space that becomes available according to the position of the telecommunications carriers on said waiting list.
- 6.8.1 In Florida, on a first-come, first-served basis governed by the date of receipt of an Application or Letter of Intent, BellSouth will maintain a waiting list of requesting carriers who have either received a Denial of Application or, where it is publicly known that the Premises is out of space, have submitted a Letter of Intent to collocate. Sixty (60) days prior to space becoming available, if known, BellSouth will notify the Florida PSC and the telecommunications carriers on the waiting list by mail when space becomes available according to the position of telecommunications carrier on said waiting list. If not known sixty (60) days in advance, BellSouth shall notify the Florida PSC and the telecommunications carriers on the waiting list within two days of the determination that space is available. A CLEC that, upon denial of physical collocation, requests virtual collocation shall be automatically placed on the waiting list.
- When space becomes available, ITC^DeltaCom must submit an updated, complete, and correct Application to BellSouth within 30 calendar days of such notification. If ITC^DeltaCom has originally requested caged collocation space and cageless collocation space becomes available, ITC^DeltaCom may refuse such space and notify BellSouth in writing within that time that ITC^DeltaCom wants to maintain its place on the waiting list without accepting such space. ITC^DeltaCom may accept an amount of space less than its original request by submitting an Application as set forth above, and upon request, may maintain its position on the waiting list for the remaining space that was initially requested. If ITC^DeltaCom does not submit such an Application or notify BellSouth in writing as described above, BellSouth will offer such space to the next CLEC on the waiting list and remove ITC^DeltaCom from the waiting list. Upon request, BellSouth will advise ITC^DeltaCom as to its position on the list.
- 6.9 <u>Public Notification</u>. BellSouth will maintain on its Interconnection Services website a notification document that will indicate all Central Offices that are without available space. BellSouth shall update such document within ten (10) calendar days of the date BellSouth becomes aware that there is insufficient space to accommodate physical

collocation. BellSouth will also post a document on its Interconnection Services website that contains a general notice where space has become available in a Central Office previously on the space exhaust list.

- 6.10 <u>Application Response.</u>
- 6.10.1 In Alabama, Kentucky and North Carolina, when space has been determined to be available, BellSouth will provide a written response ("Application Response") within twenty-three (23) business days of the receipt of a Bona Fide Application, which will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.
- 6.10.2 In South Carolina and Mississippi, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8. When multiple applications are submitted in a state within a fifteen (15) calendar day window, BellSouth will respond to the Bona Fide Applications as soon as possible, but no later than the following: within thirty (30) calendar days for Bona Fide Applications one (1) to five (5); within thirty-six (36) calendar days for Bona Fide Applications six (6) to ten (10); within forty-two (42) calendar days for Bona Fide Applications eleven (11) to fifteen (15). Response intervals for multiple Bona Fide Applications submitted within the same timeframe for the same state in excess of fifteen (15) must be negotiated. All negotiations shall consider the total volume from all requests from telecommunications companies for collocation.
- 6.10.3 In Tennessee, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.
- In Florida, within fifteen (15) calendar days of receipt of a Bona Fide Application, when space has been determined to be available or when a lesser amount of space than that requested is available, then with respect to the space available, BellSouth will provide a written response ("Application Response") including sufficient information to enable ITC^DeltaCom to place a Firm Order. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8. When ITC^DeltaCom submits ten (10) or more Applications within ten (10) calendar days, the initial fifteen (15) day response period will increase by ten (10) days for every additional ten (10) Applications or fraction thereof.
- 6.10.5 In Georgia, when space has been determined to be available for caged or cageless arrangements, BellSouth will provide a written response ("Application Response")

 Version 4Q01: 12/01/01 4 States

within twenty (20) calendar days of receipt of a Bona Fide Application. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.

6.10.6 In Louisiana, when space has been determined to be available, BellSouth will provide a written response ("Application Response") within thirty (30) calendar days for one (1) to ten (10) Applications; thirty-five (35) calendar days for eleven (11) to twenty (20) Applications; and for requests of more than twenty (20) Application it is increased by five (5) calendar days for every five (5) Applications received within five (5) business days. The Application Response will include, at a minimum, the configuration of the space, the Cable Installation Fee, Cable Records Fee, and the space preparation fees, as described in Section 8.

6.11 <u>Application Modifications</u>.

6.11.1 If a modification or revision is made to any information in the Bona Fide Application prior to Bona Fide Firm Order, with the exception of modifications to Customer Information, Contact Information or Billing Contact Information, either at the request of ITC^DeltaCom or necessitated by technical considerations, said Application shall be considered a new Application and shall be handled as a new Application with respect to response and provisioning intervals and BellSouth may charge ITC^DeltaCom an application fee. Where the Application Modification does not require assessment for provisioning or construction work by BellSouth, no application fee will be required. The fee for an Application Modification where the modification requested has limited effect (e.g., requires limited assessment and no capital expenditure by BellSouth) shall be the Subsequent Application Fee as set forth in Attachment 11. Major changes such as requesting additional space or adding equipment may require ITC^DeltaCom to submit the Application with an Application Fee.

6.12 Bona Fide Firm Order.

- 6.12.1 In Alabama, Kentucky, North Carolina, and Tennessee, ITC^DeltaCom shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Physical Expanded Interconnection Firm Order document ("Firm Order") to BellSouth. A Firm Order shall be considered Bona Fide when ITC^DeltaCom has completed the Application/Inquiry process described in Section 6, preceding, and has submitted the Firm Order document indicating acceptance of the Application Response provided by BellSouth. The Bona Fide Firm Order must be received by BellSouth no later than five (5) business days after BellSouth's Application Response to ITC^DeltaCom's Bona Fide Application.
- Except as otherwise provided, in all States that have ordered provisioning intervals but not addressed Firm Order intervals, the following shall apply._ITC^DeltaCom shall indicate its intent to proceed with equipment installation in a BellSouth Premises by submitting a Firm Order to BellSouth. The Bona Fide Firm Order must be received

- by BellSouth no later than thirty (30) calendar days after BellSouth's Application Response to ITC^DeltaCom's Bona Fide Application or the Application will expire.
- BellSouth will establish a firm order date based upon the date BellSouth is in receipt of a Bona Fide Firm Order. BellSouth will acknowledge the receipt of ITC^DeltaCom's Bona Fide Firm Order within seven (7) calendar days of receipt indicating that the Bona Fide Firm Order has been received. A BellSouth response to a Bona Fide Firm Order will include a Firm Order Confirmation containing the firm order date. No revisions will be made to a Bona Fide Firm Order.

7. <u>Construction and Provisioning</u>

- 7.1 <u>Construction and Provisioning Intervals</u>
- 7.1.1 In Alabama (Caged Only), Kentucky, and North Carolina, BellSouth will complete construction for collocation arrangements within seventy-six (76) business days from receipt of an Application or as agreed to by the Parties. Under extraordinary conditions, BellSouth will complete construction for collocation arrangements within ninety-one (91) business days. Examples of extraordinary conditions include, but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. In the event ITC^DeltaCom submits a forecast as described in the following section three (3) months or more prior to the application date, the above intervals shall apply. In the event ITC^DeltaCom submits such a forecast between two (2) months and three (3) months prior to the application date, the above intervals may be extended by one (1) additional month. In the event ITC^DeltaCom submits such a forecast less than two (2) months prior to the application date, the above intervals may be extended by sixty (60) calendar days. BellSouth will attempt to meet standard intervals for unforecasted requests and any interval adjustments will be discussed with ITC^DeltaCom at the time the application is received. Raw space, which is space lacking the necessary infrastructure to provide collocation space including but not limited to HVAC, Power, etc.), conversion time frames fall outside the normal intervals and are negotiated on an individual case basis. Additionally, installations to existing collocation arrangements for line sharing or line splitting, which include adding cable, adding cable and splitter, and adding a splitter, will be forty five (45) business days from receipt of an Application.
- 7.1.1.1 To be considered a timely and accurate forecast, ITC^DeltaCom must submit to BellSouth the CLEC Forecast Form, as set forth in exhibit B attached hereto, containing the following information: Central Office/Serving Wire Center CLLI, number of Caged square feet and/or Cageless bays, number of DS0, DS1, DS3 frame terminations, number of fused amps and planned application date.

- 7.1.2 In Alabama (Cageless), BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a Bona Fide Firm Order and ninety (90) calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.3 In Florida, BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of ninety (90) calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. For changes to collocation space after initial space completion ("Augmentation"), BellSouth will complete construction for collocation arrangements as soon as possible and within a maximum of forty-five (45) calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. If BellSouth does not believe that construction will be completed within the relevant time frame and BellSouth and ITC^DeltaCom cannot agree upon a completion date, within forty-five (45) calendar days of receipt of the Bona Fide Firm Order for an initial request, and within thirty (30) calendar days for Augmentations, BellSouth may seek an extension from the Florida PSC.
- 7.1.4 In Georgia, BellSouth will complete construction for caged collocation arrangements under ordinary conditions as soon as possible and within a maximum of ninety (90) calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. BellSouth will complete construction for cageless collocation arrangements under ordinary conditions as soon as possible and within a maximum of sixty (60) calendar days from receipt of a Bona Fide Firm Order and ninety (90) calendar days for extraordinary conditions or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Extraordinary conditions are defined to include but are not limited to major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.5 In Louisiana, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of ninety (90) calendar days for caged and sixty (60) calendar days for cageless from receipt of a Bona Fide Firm Order for an initial request, and within sixty (60) calendar days for an

Augmentation, or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within one hundred twenty (120) calendar days for caged and ninety (90) calendar days for cageless from the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.

- 7.1.6 In Mississippi, excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction for collocation arrangements under ordinary conditions as soon as possible and within a maximum of ninety (90) calendar days from receipt of a Bona Fide Firm Order or as agreed to by the Parties. Ordinary conditions are defined as space available with only minor changes to support systems required, such as but not limited to, HVAC, cabling and the power plant(s). Excluding the time interval required to secure the appropriate government licenses and permits, BellSouth will complete construction of all other Collocation Space ("extraordinary conditions") within one hundred twenty (120) calendar days of the receipt of a Bona Fide Firm Order. Examples of extraordinary conditions include but are not limited to, extended license or permitting intervals; major BellSouth equipment rearrangement or addition; power plant addition or upgrade; major mechanical addition or upgrade; major upgrade for ADA compliance; environmental hazard or hazardous materials abatement; and arrangements for which equipment shipping intervals are extraordinary in length. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.7 In South Carolina, BellSouth will complete the construction and provisioning activities for cageless and caged collocation arrangements as soon as possible, but no later than ninety (90) calendar days from receipt of a bona fide firm order. The Parties may mutually agree to renegotiate an alternative provisioning interval or BellSouth may seek a waiver from this interval from the Commission.
- 7.1.8 In Tennessee, BellSouth will complete construction for collocation arrangements under Ordinary Conditions as follows: (i) for caged collocation arrangements, within a maximum of 90 calendar days from receipt of an Bona Fide Firm Order, or as agreed to by the Parties; (ii) for cageless collocation arrangements, within 30 calendar days from receipt of a Bona Fide Firm Order when there is conditioned space and ITC^DeltaCom installs the bays/racks. In no event shall the provisioning interval for cageless collocation exceed 90 calendar days from the receipt of a Bona Fide Firm Order, or as agreed to by the parties. Under extraordinary conditions, BellSouth may

elect to renegotiate an alternative provisioning interval with ITC^DeltaCom or seek a waiver from this interval from the Commission. For the purpose of defining conditioned space as referenced in the TRA order setting intervals for cageless collocation in Tennessee, conditioned space is defined as follows: i) floor space must be available; ii) floor space must be equipped with adequate air conditioning to accommodate equipment listed on application; iii) Cable racking, any fiber duct, riser cable support structure and power cable support structure must be in place to support equipment listed on the application; and iv) power plant capacity at BDFB or main power board must be available. If LGX or DGX equipment is requested on the application and adequate existing capacity is not available then conditioned is considered unavailable. If BellSouth is required by the application to place power cabling, conditioned space is considered unavailable.

- Joint Planning. Joint planning between BellSouth and ITC^DeltaCom will commence within a maximum of twenty (20) calendar days from BellSouth's receipt of a Bona Fide Firm Order. BellSouth will provide the preliminary design of the Collocation Space and the equipment configuration requirements as reflected in the Bona Fide Application and affirmed in the Bona Fide Firm Order. The Collocation Space completion time period will be provided to ITC^DeltaCom during joint planning.
- 7.3 <u>Permits</u>. Each Party or its agents will diligently pursue filing for the permits required for the scope of work to be performed by that Party or its agents within ten (10) calendar days of the completion of finalized construction designs and specifications.
- Acceptance Walk Through. ITC^DeltaCom will schedule and complete an acceptance walkthrough of each Collocation Space with BellSouth within fifteen (15) days of BellSouth's notifying ITC^DeltaCom that the collocation space is ready for occupancy. In the event that ITC^DeltaCom fails to complete an acceptance walkthrough within this fifteen (15) day interval, the Collocation Space shall be deemed accepted by ITC^DeltaCom. BellSouth will correct any deviations to ITC^DeltaCom's original or jointly amended requirements within seven (7) calendar days after the walk through, unless the Parties jointly agree upon a different time frame.
- Use of BellSouth Certified Supplier. ITC^DeltaCom shall select a supplier which has been approved as a BellSouth Certified Supplier to perform all engineering and installation work. ITC^DeltaCom and ITC^DeltaCom's BellSouth Certified Supplier must follow and comply with all BellSouth requirements outlined in BellSouth's TR 73503, TR 73519, TR 73572, and TR 73564. In some cases, ITC^DeltaCom must select separate BellSouth Certified Suppliers for transmission equipment, switching equipment and power equipment. BellSouth shall provide ITC^DeltaCom with a list of BellSouth Certified Suppliers upon request. The BellSouth Certified Supplier(s) shall be responsible for installing ITC^DeltaCom's equipment and components, extending power cabling to the BellSouth power distribution frame, performing operational tests after installation is complete, and notifying BellSouth's equipment engineers and ITC^DeltaCom upon successful completion of installation, etc. The

BellSouth Certified Supplier shall bill ITC^DeltaCom directly for all work performed for ITC^DeltaCom pursuant to this Attachment and BellSouth shall have no liability for nor responsibility to pay such charges imposed by the BellSouth Certified Supplier. BellSouth shall consider certifying ITC^DeltaCom or any supplier proposed by ITC^DeltaCom. All work performed by or for ITC^DeltaCom shall conform to generally accepted industry guidelines and standards.

- Alarm and Monitoring. BellSouth shall place environmental alarms in the Premises for the protection of BellSouth equipment and facilities. ITC^DeltaCom shall be responsible for placement, monitoring and removal of environmental and equipment alarms used to service ITC^DeltaCom's Collocation Space. Upon request, BellSouth will provide ITC^DeltaCom with applicable tariffed service(s) to facilitate remote monitoring of collocated equipment by ITC^DeltaCom. Both Parties shall use best efforts to notify the other of any verified environmental condition known to that Party.
- 7.7 Virtual to Physical Collocation Relocation. In the event physical collocation space was previously denied at a location due to technical reasons or space limitations, and physical collocation space has subsequently become available, ITC^DeltaCom may relocate its virtual collocation arrangements to physical collocation arrangements and pay the appropriate fees for physical collocation and for the rearrangement or reconfiguration of services terminated in the virtual collocation arrangement, as outlined in the appropriate BellSouth tariffs. In the event that BellSouth knows when additional space for physical collocation may become available at the location requested by ITC^DeltaCom, such information will be provided to ITC^DeltaCom in BellSouth's written denial of physical collocation. To the extent that (i) physical Collocation Space becomes available to ITC^DeltaCom within 180 calendar days of BellSouth's written denial of ITC^DeltaCom's request for physical collocation, (ii) BellSouth had knowledge that the space was going to become available, and (iii) ITC^DeltaCom was not informed in the written denial that physical Collocation Space would become available within such 180 calendar days, then ITC^DeltaCom may relocate its virtual collocation arrangement to a physical collocation arrangement and will receive a credit for any nonrecurring charges previously paid for such virtual collocation. ITC^DeltaCom must arrange with a BellSouth Certified Supplier for the relocation of equipment from its virtual Collocation Space to its physical Collocation Space and will bear the cost of such relocation.
- Virtual to Physical Conversion (In Place). Virtual collocation arrangements may be converted to "in-place" physical arrangements if the potential conversion meets the following four criteria: 1) there is no change in the amount of equipment or the configuration of the equipment that was in the virtual collocation arrangement; 2) the conversion of the virtual collocation arrangement will not cause the equipment or the results of that conversion to be located in a space that BellSouth has reserved for its own future needs; 3) the converted arrangement does not limit BellSouth's ability to secure its own equipment and facilities due to the location of the virtual collocation arrangement; and 4) any changes to the arrangement can be accommodated by existing power, HVAC, and other requirements. The application fee for the conversion from

- virtual to in-place, physical collocation is as set forth in Attachment 11. Unless otherwise specified, BellSouth will complete virtual to in-place physical collocation conversions within sixty (60) calendar days.
- 7.8.1 In Florida, for Virtual to Physical conversions in place that require no physical changes, the only applicable charges shall cover the administrative billing and engineering records updates.
- 7.8.2 In Tennessee, BellSouth will complete Virtual to Physical conversions in place within thirty (30) calendar days.
- 7.9 <u>Cancellation</u>. If, at anytime prior to space acceptance, ITC^DeltaCom cancels its order for the Collocation Space(s) ("Cancellation"), BellSouth will bill the applicable non-recurring rate for any and all work processes for which work has begun. In Georgia, if ITC^DeltaCom cancels its order for Collocation Space at any time prior to space acceptance, BellSouth will bill ITC^DeltaCom for all costs incurred prior to the date of Cancellation and for any costs incurred as a direct result of the Cancellation, not to exceed the total amount that would have been due had the order not been cancelled.
- 7.10 <u>Licenses.</u> ITC^DeltaCom, at its own expense, will be solely responsible for obtaining from governmental authorities, and any other appropriate agency, entity, or person, all rights, privileges, and licenses necessary or required to operate as a provider of telecommunications services to the public or to occupy the Collocation Space.
- 7.11 <u>Environmental Compliance.</u> The Parties agree to utilize and adhere to the Environmental Hazard Guidelines identified as Exhibit A attached hereto.

8. Rates and Charges

- 8.1 BellSouth shall assess an Application Fee via a service order, which shall be issued at the time BellSouth responds that space is available pursuant to Section 2. Payment of said Application Fee will be due as dictated by ITC^DeltaCom's current billing cycle and is non-refundable.
- 8.1.1 In Tennessee the applicable Application Fee is the Planning Fee for both Applications and Subsequent Applications placed by ITC^DeltaCom.
- 8.2 <u>Space Preparation</u>
- 8.2.1 Recurring Charges. The recurring charges for space preparation begin on the date ITC^DeltaCom executes the written document accepting the collocation space pursuant to section 4 or on the date ITC^DeltaCom first occupies collocation space, whichever is first. If ITC^DeltaCom fails to schedule and complete an acceptance walk through within fifteen (15) days after BellSouth releases the space for occupancy,

- BellSouth shall begin billing ITC^DeltaCom for recurring charges as of the sixteenth day after BellSouth releases the collocation space.
- Space preparation fees consist of a nonrecurring charge for Firm Order Processing and monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless collocation and per cage for caged collocation.

 ITC^DeltaCom shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event ITC^DeltaCom opts for cageless space, the space preparation fees will be assessed based on the total floor space dedicated to ITC^DeltaCom as prescribed in this Section 8.
- 8.2.3 Space Preparation Fee (Florida). Space preparation fees include a nonrecurring charge for Firm Order Processing and monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot, and Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation. ITC^DeltaCom shall remit payment of the nonrecurring Firm Order Processing Fee coincident with submission of a Bona Fide Firm Order. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event ITC^DeltaCom opts for cageless space, space preparation fees will be assessed based on the total floor space dedicated to ITC^DeltaCom as prescribed in this Section 8.
- 8.2.4 <u>Space Preparation Fee (Georgia)</u>. In Georgia, the Space Preparation Fee is a one time fee, assessed per arrangement, per location. It recovers a portion of costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, power, building and support systems. This is a set fee of \$100 per square foot as established by the Georgia Public Service Commission Order in Docket No. 7016 U. In the event ITC^DeltaCom opts for non-enclosed space, the space preparation fee will be assessed based on the total floor space dedicated to ITC^DeltaCom as prescribed in Section 8 and will be billed based upon ITC^DeltaCom's first billing cycle after Firm Order.
- 8.2.5 <u>Space Preparation Fee (North Carolina)</u>. In North Carolina, space preparation fees consist of monthly recurring charges for Central Office Modifications, assessed per arrangement, per square foot; Common Systems Modifications, assessed per arrangement, per square foot for cageless and per cage for caged collocation; and Power, assessed per the nominal –48V DC ampere requirements specified by ITC^DeltaCom on the Bona Fide Application. The charges recover the costs associated with preparing the Collocation Space, which includes survey, engineering of the Collocation Space, design and modification costs for network, building and support systems. In the event ITC^DeltaCom opts for cageless space, the space

- preparation fees will be assessed based on the total floor space dedicated to ITC^DeltaCom as described in this Section 8.
- 8.3 <u>Cable Installation</u>. Cable Installation Fee(s) are assessed per entrance cable placed.
- 8.4 Floor Space. The Floor Space Charge includes reasonable charges for lighting, HVAC, and other allocated expenses associated with maintenance of the Premises but does not recover any power-related costs incurred by BellSouth. When the Collocation Space is enclosed, ITC^DeltaCom shall pay floor space charges based upon the number of square feet so enclosed. When the Collocation Space is not enclosed, ITC^DeltaCom shall pay floor space charges based upon the following floor space calculation: $[(depth \ of \ the \ equipment \ lineup \ in \ which \ the \ rack \ is \ placed) + (0.5)$ x maintenance aisle depth) + (0.5 x wiring aisle depth)] X (width of rack and spacers). For purposes of this calculation, the depth of the equipment lineup shall consider the footprint of equipment racks plus any equipment overhang. BellSouth will assign unenclosed Collocation Space in conventional equipment rack lineups where feasible. In the event ITC^DeltaCom's collocated equipment requires special cable racking, isolated grounding or other treatment which prevents placement within conventional equipment rack lineups, ITC^DeltaCom shall be required to request an amount of floor space sufficient to accommodate the total equipment arrangement.
- 8.4.1 The recurring charges for floor space begin on the date ITC^DeltaCom executes the written document accepting the collocation space pursuant to section 4 or on the date ITC^DeltaCom first occupies collocation space, whichever is first. If ITC^DeltaCom fails to schedule and complete an acceptance walk through within fifteen (15) days after BellSouth releases the space for occupancy, BellSouth shall begin billing ITC^DeltaCom for recurring charges as of the sixteenth day after BellSouth releases the collocation space.
- 8.5 <u>Power</u>. BellSouth shall make available –48 Volt (-48V) DC power for ITC^DeltaCom's Collocation Space at a BellSouth Power Board or BellSouth Battery Distribution Fuse Bay ("BDFB") at ITC^DeltaCom's option within the Premises.
- 8.5.1 Recurring charges for -48V DC power will be assessed per ampere per month based upon the BellSouth Certified Supplier engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable rack to ITC^DeltaCom's equipment or space enclosure. Recurring power charges begin on the Space Ready Date, or on the date ITC^DeltaCom first occupies the Collocation Space, whichever is sooner. When obtaining power from a BDFB, fuses and power cables (A&B) must be engineered (sized), and installed by ITC^DeltaCom's BellSouth Certified Supplier. When obtaining power from a BellSouth power board, power cables (A&B) must be engineered (sized), and installed by ITC^DeltaCom's BellSouth Certified power Supplier. ITC^DeltaCom is responsible for contracting with a BellSouth Certified Supplier for power distribution feeder cable runs from a BellSouth BDFB or power board to ITC^DeltaCom's equipment. Determination of the BellSouth BDFB or BellSouth power board as the

power source will be made at BellSouth's sole, but reasonable, discretion. The BellSouth Certified Supplier contracted by ITC^DeltaCom must provide BellSouth a copy of the engineering power specification prior to the day on which ITC^DeltaCom's equipment becomes operational. BellSouth will provide the common power feeder cable support structure between the BellSouth BDFB or power board and ITC^DeltaCom's arrangement area. ITC^DeltaCom shall contract with a BellSouth Certified Supplier who will be responsible for the following: dedicated power cable support structure within ITC^DeltaCom's arrangement, power cable feeds, and terminations of cable. Any terminations at a BellSouth power board must be performed by a BellSouth Certified power Supplier. ITC^DeltaCom shall comply with all applicable National Electric Code (NEC), BellSouth TR73503, Telcordia (Bellcore) and ANSI Standards regarding power cabling.

- 8.5.2 If BellSouth has not previously invested in power plant capacity for collocation at a specific site, ITC^DeltaCom has the option to add its own dedicated power plant; provided, however, that such work shall be performed by a BellSouth Certified Supplier who shall comply with BellSouth's guidelines and specifications. Where the addition of ITC^DeltaCom's dedicated power plant results in construction of a new power plant room, upon termination of ITC^DeltaCom's right to occupy collocation space at such site, ITC^DeltaCom shall have the right to remove its equipment from the power plant room, but shall otherwise leave the room intact.
- 8.5.3 If ITC^DeltaCom elects to install its own DC Power Plant, BellSouth shall provide AC power to feed ITC^DeltaCom's DC Power Plant. Charges for AC power will be assessed per breaker ampere per month. Rates include the provision of commercial and standby AC power. When obtaining power from a BellSouth service panel, protection devices and power cables must be engineered (sized), and installed by ITC^DeltaCom's BellSouth Certified Supplier except that BellSouth shall engineer and install protection devices and power cables for Adjacent Collocation. ITC^DeltaCom's BellSouth Certified Supplier must also provide a copy of the engineering power specification prior to the equipment becoming operational. Charges for AC power shall be assessed pursuant to the rates specified in Attachment 11. AC power voltage and phase ratings shall be determined on a per location basis. At ITC^DeltaCom's option, ITC^DeltaCom may arrange for AC power in an Adjacent Collocation arrangement from a retail provider of electrical power.
- 8.5.4 In Tennessee, Recurring charges for -48V DC power consumption will be assessed per ampere per month based upon the engineered and installed power feed fused ampere capacity. Rates include redundant feeder fuse positions (A&B) and common cable rack to ITC^DeltaCom's equipment or space enclosure. ITC^DeltaCom shall contract with a Certified Supplier who will be responsible for the following: dedicated power cable support structure within ITC^DeltaCom's arrangement and terminations of cable within the collocation space.

- 8.5.5 In Tennessee, Non recurring charges for –48V DC power distribution will be based on the common power feeder cable support structure between the BellSouth BDFB and ITC^DeltaCom's arrangement area.
- 8.5.6 In Louisiana, ITC^DeltaCom has the option to purchase power directly from an electric utility company. Under such an option, ITC^DeltaCom is responsible for contracting with the electric utility company for their own power feed and meter, and is financially responsible for purchasing all equipment necessary to accomplish the arrangement, including inverters, batteries, power boards, bus bars, BDFBs, backup power supplies and cabling. The actual work to install this arrangement must be performed by a certified vendor hired by ITC^DeltaCom. ITC^DeltaCom must comply with all applicable safety codes, including the National Electric Safety Codes, in installing this power arrangement. Any floor space, cable racking, etc utilized by ITC^DeltaCom in provisioning said power will be billed on an ICB basis. BellSouth shall waive any application fee or charges that would otherwise be due if ITC^DeltaCom decides to reconfigure any existing collocation power arrangement so as to purchase power directly from an electric utility.

If ITC^DeltaCom elects to reconfigure its existing collocation power arrangement so as to purchase smaller increments of power from BellSouth's BDFB rather than directly from Bellsouth's main power board, BellSouth shall waive any application fee or charges that would otherwise be due. BellSouth will respond to such application within 7 calendar days from the date of application. [Order U-22252 Page 4 of 6]

- 8.6 <u>Security Escort.</u> A security escort will be required whenever ITC^DeltaCom or its approved agent desires access to the entrance manhole or must have access to the Premises after the one accompanied site visit allowed pursuant to Section 5 prior to completing BellSouth's Security Training requirements. Rates for a security escort are as set forth in Attachment 11 beginning with the scheduled escort time. BellSouth will wait for one-half (1/2) hour after the scheduled time for such an escort and ITC^DeltaCom shall pay for such half-hour charges in the event ITC^DeltaCom fails to show up.
- 8.7 <u>Cable Record charges.</u> These charges apply for work required to build cable records in BellSouth systems. The VG/DS0 per cable record charge is for a maximum of 3600 records. The Fiber cable record charge is for a maximum of 99 records.
- 8.8 Other. If no rate is identified in the contract, the rate for the specific service or function will be negotiated by the Parties upon request by either Party. Payment of all other charges under this Attachment shall be due thirty (30) calendar days after receipt of the bill (payment due date). ITC^DeltaCom will pay a late payment charge of the lessor of one and one half percent or the legal interest rate assessed monthly on any balance which remains unpaid after the payment due date.

9. <u>Insurance</u>

- 9.1 ITC^DeltaCom shall, at its sole cost and expense, procure, maintain, and keep in force insurance as specified in this Section 9 and underwritten by insurance companies licensed to do business in the states applicable under this Attachment and having a Best's Insurance Rating of A-.
- 9.2 ITC^DeltaCom shall maintain the following specific coverage:
- 9.2.1 Commercial General Liability coverage in the amount of ten million dollars (\$10,000,000.00) or a combination of Commercial General Liability and Excess/Umbrella coverage totaling not less than ten million dollars (\$10,000,000.00). BellSouth shall be named as an Additional Insured on the Commercial General Liability policy as specified herein.
- 9.2.2 Statutory Workers Compensation coverage and Employers Liability coverage in the amount of one hundred thousand dollars (\$100,000.00) each accident, one hundred thousand dollars (\$100,000.00) each employee by disease, and five hundred thousand dollars (\$500,000.00) policy limit by disease.
- 9.2.3 All Risk Property coverage on a full replacement cost basis insuring all of ITC^DeltaCom's real and personal property situated on or within BellSouth's Central Office location(s).
- 9.2.4 ITC^DeltaCom may elect to purchase business interruption and contingent business interruption insurance, having been advised that BellSouth assumes no liability for loss of profit or revenues should an interruption of service occur.
- 9.3 The limits set forth in Section 9.2 above may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days notice to ITC^DeltaCom to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- All policies purchased by ITC^DeltaCom shall be deemed to be primary and not contributing to or in excess of any similar coverage purchased by BellSouth. All insurance must be in effect on or before the date equipment is delivered to BellSouth's Premises and shall remain in effect for the term of this Attachment or until all ITC^DeltaCom's property has been removed from BellSouth's Premises, whichever period is longer. If ITC^DeltaCom fails to maintain required coverage, BellSouth may pay the premiums thereon and seek reimbursement of same from ITC^DeltaCom.
- 9.5 ITC^DeltaCom shall submit certificates of insurance reflecting the coverage required pursuant to this Section a minimum of ten (10) business days prior to the commencement of any work in the Collocation Space. Failure to meet this interval may result in construction and equipment installation delays. ITC^DeltaCom shall arrange for BellSouth to receive thirty (30) business days' advance notice of cancellation from ITC^DeltaCom's insurance company. ITC^DeltaCom shall forward

Page 31

a certificate of insurance and notice of cancellation/non-renewal to BellSouth at the following address:

BellSouth Telecommunications, Inc. Attn.: Risk Management Coordinator 17H53 BellSouth Center 675 W. Peachtree Street Atlanta, Georgia 30375

- 9.6 ITC^DeltaCom must conform to recommendations made by BellSouth's fire insurance company to the extent BellSouth has agreed to, or shall hereafter agree to, such recommendations.
- 9.7 Self-Insurance. If ITC^DeltaCom's net worth exceeds five hundred million dollars (\$500,000,000), ITC^DeltaCom may elect to request self-insurance status in lieu of obtaining any of the insurance required in Sections 9.2.1 and 9.2.2. ITC^DeltaCom shall provide audited financial statements to BellSouth thirty (30) days prior to the commencement of any work in the Collocation Space. BellSouth shall then review such audited financial statements and respond in writing to ITC^DeltaCom in the event that self-insurance status is not granted to ITC^DeltaCom. If BellSouth approves ITC^DeltaCom for self-insurance, ITC^DeltaCom shall annually furnish to BellSouth, and keep current, evidence of such net worth that is attested to by one of ITC^DeltaCom's corporate officers. The ability to self-insure shall continue so long as the ITC^DeltaCom meets all of the requirements of this Section. If the ITC^DeltaCom subsequently no longer satisfies this Section, ITC^DeltaCom is required to purchase insurance as indicated by Sections 9.2.1 and 9.2.2.
- 9.8 The net worth requirements set forth in Section 9.7 may be increased by BellSouth from time to time during the term of this Attachment upon thirty (30) days' notice to ITC^DeltaCom to at least such minimum limits as shall then be customary with respect to comparable occupancy of BellSouth structures.
- 9.9 Failure to comply with the provisions of this Section will be deemed a material breach of this Attachment.

10. Mechanics Liens

10.1 If any mechanics lien or other liens shall be filed against property of either Party (BellSouth or ITC^DeltaCom), or any improvement thereon by reason of or arising out of any labor or materials furnished or alleged to have been furnished or to be furnished to or for the other Party or by reason of any changes, or additions to said property made at the request or under the direction of the other Party, the other Party directing or requesting those changes shall, within thirty (30) business days after receipt of written notice from the Party against whose property said lien has been filed, either pay such lien or cause the same to be bonded off the affected property in the manner provided by law. The Party causing said lien to be placed against the property

of the other shall also defend, at its sole cost and expense, on behalf of the other, any action, suit or proceeding which may be brought for the enforcement of such liens and shall pay any damage and discharge any judgment entered thereon.

11. Inspections

BellSouth may conduct an inspection of ITC^DeltaCom's equipment and facilities in the Collocation Space(s) prior to the activation of facilities between ITC^DeltaCom's equipment and equipment of BellSouth. BellSouth may conduct an inspection if ITC^DeltaCom adds equipment and may otherwise conduct routine inspections at reasonable intervals mutually agreed upon by the Parties. BellSouth shall provide ITC^DeltaCom with a minimum of forty-eight (48) hours or two (2) business days, whichever is greater, advance notice of all such inspections. All costs of such inspection shall be borne by BellSouth.

12. Security and Safety Requirements

- Unless otherwise specified, ITC^DeltaCom will be required, at its own expense, to conduct a statewide investigation of criminal history records for each ITC^DeltaCom employee hired in the past five years being considered for work on the BellSouth Premises, for the states/counties where the ITC^DeltaCom employee has worked and lived for the past five years. Where state law does not permit statewide collection or reporting, an investigation of the applicable counties is acceptable. ITC^DeltaCom shall not be required to perform this investigation if an affiliated company of ITC^DeltaCom has performed an investigation of the ITC^DeltaCom employee seeking access, if such investigation meets the criteria set forth above. This requirement will not apply if ITC^DeltaCom has performed a pre-employment statewide investigation of criminal history records of the ITC^DeltaCom employee for the states/counties where the ITC^DeltaCom employee has worked and lived for the past five years or, where state law does not permit a statewide investigation, an investigation of the applicable counties.
- 12.2 ITC^DeltaCom will be required to administer to their personnel assigned to the BellSouth Premises security training either provided by BellSouth, or meeting criteria defined by BellSouth.
- ITC^DeltaCom shall provide its employees and agents with picture identification, which must be worn, and visible at all times while in the Collocation Space or other areas in or around the Premises. The photo identification card shall bear, at a minimum, the employee's name and photo, and the ITC^DeltaCom's name. BellSouth reserves the right to remove from its premises any employee of ITC^DeltaCom not possessing identification issued by ITC^DeltaCom or who has violated any of BellSouth's policies as outlined in the CLEC Security Training documents. ITC^DeltaCom shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises. ITC^DeltaCom shall be solely

- responsible for ensuring that any Guest of ITC^DeltaCom is in compliance with all subsections of this Section 12.
- 12.4 ITC^DeltaCom shall not assign to the BellSouth Premises any personnel with records of felony criminal convictions. ITC^DeltaCom shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions, except for misdemeanor traffic violations, without advising BellSouth of the nature and gravity of the offense(s). BellSouth reserves the right to refuse building access to any ITC^DeltaCom personnel who have been identified to have misdemeanor criminal convictions. Notwithstanding the foregoing, in the event that ITC^DeltaCom chooses not to advise BellSouth of the nature and gravity of any misdemeanor conviction, ITC^DeltaCom may, in the alternative, certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions (other than misdemeanor traffic violations).
- 12.4.1 ITC^DeltaCom shall not knowingly assign to the BellSouth Premises any individual who was a former employee of BellSouth and whose employment with BellSouth was terminated for a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.4.2 ITC^DeltaCom shall not knowingly assign to the BellSouth Premises any individual who was a former supplier of BellSouth and whose access to a BellSouth Premises was revoked due to commission of a criminal offense whether or not BellSouth sought prosecution of the individual for the criminal offense.
- 12.5 For each ITC^DeltaCom employee or agent hired by ITC^DeltaComwithin five years of being considered for work on the BellSouth Premises, who requires access to a BellSouth Premises pursuant to this agreement, ITC^DeltaCom shall furnish BellSouth, prior to an employee or agent gaining such access, a certification that the aforementioned background check and security training were completed. The certification will contain a statement that no felony convictions were found and certifying that the security training was completed by the employee. If the employee's criminal history includes misdemeanor convictions, ITC^DeltaCom will disclose the nature of the convictions to BellSouth at that time. In the alternative, ITC^DeltaCom may certify to BellSouth that it shall not assign to the BellSouth Premises any personnel with records of misdemeanor convictions other than misdemeanor traffic violations.
- 12.5.1 For all other ITC^DeltaCom employees requiring access to a BellSouth Premises pursuant to this Attachment, ITC^DeltaCom shall furnish BellSouth, prior to an employee gaining such access, a certification that the employee is not subject to the requirements of Section 12.5 above and that security training was completed by the employee.
- 12.6 At BellSouth's request, ITC^DeltaCom shall promptly remove from BellSouth's Premises any employee of ITC^DeltaCom BellSouth does not wish to grant access to

its premises 1) pursuant to any investigation conducted by BellSouth or 2) prior to the initiation of an investigation if an employee of ITC^DeltaCom is found interfering with the property or personnel of BellSouth or another CLEC, provided that an investigation shall promptly be commenced by BellSouth.

- 12.6.1 Notification to BellSouth. BellSouth reserves the right to interview ITC^DeltaCom's employees, agents or contractors in the event of wrongdoing in or around BellSouth's property or involving BellSouth's or another CLEC's property or personnel, provided that BellSouth shall provide 24 hours notice (or such shorter notice as may be agreed by the parties as reasonable under the circumstances) to ITC^DeltaCom's Security contact of such interview. ITC^DeltaCom and its contractors shall reasonably cooperate with BellSouth's investigation into allegations of wrongdoing or criminal conduct committed by, witnessed by or involving ITC^DeltaCom's employees, agents, or contractors. Additionally, BellSouth reserves the right to bill ITC^DeltaCom for all reasonable costs associated with investigations involving its employees, agents, or contractors if it is established and mutually agreed in good faith that ITC^DeltaCom's employees, agents, or contractors are responsible for the alleged act. BellSouth shall bill ITC^DeltaCom for BellSouth property which is stolen or damaged where an investigation determines the culpability of ITC^DeltaCom employees, agents, or contractors and ITC^DeltaCom agrees, in good faith, with the results of such investigation. ITC^DeltaCom shall notify BellSouth in writing immediately in the event that ITC^DeltaCom discovers one ofits employees already working on the BellSouth central office premises is a possible security risk. BellSouth reserves the right to permanently but lawfully remove from its premises any employee of ITC^DeltaCom identified as posing a security risk to BellSouth or any other CLEC, or having violated BellSouth policies set forth in the BellSouth CLEC Security Training. ITC^DeltaCom shall hold BellSouth harmless for any damages resulting from such removal of its personnel from BellSouth premises.
- 12.7 <u>Use of Supplies</u>. Unauthorized use of telecommunications equipment or supplies by either Party, whether or not used routinely to provide telephone service (e.g. plug-in cards,) will be strictly prohibited and handled appropriately. Costs associated with such unauthorized use may be charged to the offending Party, as may be all associated investigative costs.
- 12.8 <u>Use of Official Lines</u>. Except for non-toll calls necessary in the performance of their work, neither Party shall use the telephones of the other Party on the BellSouth Premises. Charges for unauthorized telephone calls may be charged to the offending Party, as may be all associated investigative costs.
- 12.9 <u>Accountability</u>. Full compliance with the Security requirements of this section shall in no way limit the accountability of either Party to the other for the improper actions of its employees.

13. Destruction of Collocation Space

13.1 In the event a Collocation Space is wholly or partially damaged by fire, windstorm, tornado, flood or by similar causes to such an extent as to be rendered wholly unsuitable for ITC^DeltaCom's permitted use hereunder, then either Party may elect within ten (10) business days after such damage, to terminate occupancy of the damaged Collocation Space, and if either Party shall so elect, by giving the other written notice of termination, both Parties shall stand released of and from further liability under the terms hereof. If the Collocation Space shall suffer only minor damage and shall not be rendered wholly unsuitable for ITC^DeltaCom's permitted use, or is damaged and the option to terminate is not exercised by either Party, BellSouth covenants and agrees to proceed promptly without expense to ITC^DeltaCom, except for improvements not the property of BellSouth, to repair the damage. BellSouth shall have a reasonable time within which to rebuild or make any repairs, and such rebuilding and repairing shall be subject to delays caused by storms, shortages of labor and materials, government regulations, strikes, walkouts, and causes beyond the control of BellSouth, which causes shall not be construed as limiting factors, but as exemplary only. ITC^DeltaCom may, at its own expense, accelerate the rebuild of its collocated space and equipment provided however that a BellSouth Certified Supplier is used and the necessary space preparation has been completed. Rebuild of equipment must be performed by a BellSouth Certified Supplier. If ITC^DeltaCom's acceleration of the project increases the cost of the project, then those additional charges will be incurred by ITC^DeltaCom. Where allowed and where practical, ITC^DeltaCom may erect a temporary facility while BellSouth rebuilds or makes repairs. In all cases where the Collocation Space shall be rebuilt or repaired, ITC^DeltaCom shall be entitled to an equitable abatement of rent and other charges, depending upon the unsuitability of the Collocation Space for ITC^DeltaCom's permitted use, until such Collocation Space is fully repaired and restored and ITC^DeltaCom's equipment installed therein (but in no event later than thirty (30) business days after the Collocation Space is fully repaired and restored). Where ITC^DeltaCom has placed an Adjacent Arrangement pursuant to Section 3, ITC^DeltaCom shall have the sole responsibility to repair or replace said Adjacent Arrangement provided herein. Pursuant to this section, BellSouth will restore the associated services to the Adjacent Arrangement.

14. Eminent Domain

14.1 If the whole of a Collocation Space or Adjacent Arrangement shall be taken by any public authority under the power of eminent domain, then this Attachment shall terminate with respect to such Collocation Space or Adjacent Arrangement as of the day possession shall be taken by such public authority and rent and other charges for the Collocation Space or Adjacent Arrangement shall be paid up to that day with proportionate refund by BellSouth of such rent and charges as may have been paid in advance for a period subsequent to the date of the taking. If any part of the Collocation Space or Adjacent Arrangement shall be taken under eminent domain,

Page 36

BellSouth and ITC^DeltaCom shall each have the right to terminate this Attachment with respect to such Collocation Space or Adjacent Arrangement and declare the same null and void, by written notice of such intention to the other Party within ten (10) business days after such taking.

15. <u>Nonexclusivity</u>

15.1 ITC^DeltaCom understands that this Attachment is not exclusive and that BellSouth may enter into similar agreements with other Parties. Assignment of space pursuant to all such agreements shall be determined by space availability and made on a first come, first served basis

ENVIRONMENTAL AND SAFETY PRINCIPLES

The following principles provide basic guidance on environmental and safety issues when applying for and establishing Physical Collocation arrangements.

1. GENERAL PRINCIPLES

- 1.1 Compliance with Applicable Law. BellSouth and ITC^DeltaCom agree to comply with applicable federal, state, and local environmental and safety laws and regulations including U.S. Environmental Protection Agency (USEPA) regulations issued under the Clean Air Act (CAA), Clean Water Act (CWA), Resource Conservation and Recovery Act (RCRA), Comprehensive Environmental Response, Compensation and Liability Act (CERCLA), Superfund Amendments and Reauthorization Act (SARA), the Toxic Substances Control Act (TSCA), and OSHA regulations issued under the Occupational Safety and Health Act of 1970, as amended and NFPA and National Electrical Codes (NEC) and the NESC ("Applicable Laws"). Each Party shall notify the other if compliance inspections are conducted by regulatory agencies and/or citations are issued that relate to any aspect of this Attachment.
- 1.2 Notice. BellSouth and ITC^DeltaCom shall provide notice to the other, including Material Safety Data Sheets (MSDSs), of known and recognized physical hazards or Hazardous Chemicals existing on site or brought on site. Each Party is required to provide specific notice for known potential Imminent Danger conditions. ITC^DeltaCom should contact 1-800-743-6737 for BellSouth MSDS sheets.
- 1.3 Practices/Procedures. BellSouth may make available additional environmental control procedures for ITC^DeltaCom to follow when working at a BellSouth Premises (See Section 2, below). These practices/procedures will represent the regular work practices required to be followed by the employees and contractors of BellSouth for environmental protection. ITC^DeltaCom will require its contractors, agents and others accessing the BellSouth Premises to comply with these practices. Section 2 lists the Environmental categories where BST practices should be followed by ITC^DeltaCom when operating in the BellSouth Premises.
- 1.4 <u>Environmental and Safety Inspections</u>. BellSouth reserves the right to inspect the ITC^DeltaCom space with proper notification. BellSouth reserves the right to stop any ITC^DeltaCom work operation that imposes Imminent Danger to the environment, employees or other persons in the area or Facility.
- 1.5 <u>Hazardous Materials Brought On Site</u>. Any hazardous materials brought into, used, stored or abandoned at the BellSouth Premises by ITC^DeltaCom are owned by ITC^DeltaCom. ITC^DeltaCom will indemnify BellSouth for claims, lawsuits or damages to persons or property caused by these materials. Without prior written

BellSouth approval, no substantial new safety or environmental hazards can be created by ITC^DeltaCom or different hazardous materials used by ITC^DeltaCom at BellSouth Facility. ITC^DeltaCom must demonstrate adequate emergency response capabilities for its materials used or remaining at the BellSouth Facility.

- 1.6 <u>Spills and Releases</u>. When contamination is discovered at a BellSouth Premises, the Party discovering the condition must notify BellSouth. All Spills or Releases of regulated materials will immediately be reported by ITC^DeltaCom to BellSouth.
- Coordinated Environmental Plans and Permits. BellSouth and ITC^DeltaCom will coordinate plans, permits or information required to be submitted to government agencies, such as emergency response plans, spill prevention control and countermeasures (SPCC) plans and community reporting. If fees are associated with filing, BellSouth and ITC^DeltaCom will develop a cost sharing procedure. If BellSouth's permit or EPA identification number must be used, ITC^DeltaCom must comply with all of BellSouth's permit conditions and environmental processes, including environmental "best management practices (BMP)" (see Section 2, below) and/or selection of BST disposition vendors and disposal sites.
- 1.8 Environmental and Safety Indemnification. BellSouth and ITC^DeltaCom shall indemnify, defend and hold harmless the other Party from and against any claims (including, without limitation, third-party claims for personal injury or death or real or personal property damage), judgments, damages, (including direct and indirect damages, and punitive damages), penalties, fines, forfeitures, costs, liabilities, interest and losses arising in connection with the violation or alleged violation of any Applicable Law or contractual obligation or the presence or alleged presence of contamination arising out of the acts or omissions of the indemnifying Party, its agents, contractors, or employees concerning its operations at the Facility.

2. CATEGORIES FOR CONSIDERATION OF ENVIRONMENTAL ISSUES

- When performing functions that fall under the following Environmental categories on BellSouth's Premises, ITC^DeltaCom agrees to comply with the applicable sections of the current issue of BellSouth's Environmental and Safety Methods and Procedures (M&Ps), incorporated herein by this reference. ITC^DeltaCom further agrees to cooperate with BellSouth to ensure that ITC^DeltaCom's employees, agents, and/or subcontractors are knowledgeable of and satisfy those provisions of BellSouth's Environmental M&Ps which apply to the specific Environmental function being performed by ITC^DeltaCom, its employees, agents and/or subcontractors.
- 2.2 The most current version of reference documentation must be requested from BellSouth.

ENVIRONMENTAL CATEGORIES	ENVIRONMENTAL ISSUES	ADDRESSED BY THE FOLLOWING DOCUMENTATION			
Disposal of hazardous material or other regulated material (e.g., batteries, fluorescent tubes, solvents & cleaning materials)	Compliance with all applicable local, state, & federal laws and regulations Pollution liability insurance	Std T&C 450 Fact Sheet Series 17000 Std T&C 660-3			
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)			
Emergency response	Hazmat/waste release/spill fire safety emergency	Fact Sheet Series 1700 Building Emergency Operations Plan (EOP) (specific to and located on Premises)			
Contract labor/outsourcing for services with environmental implications	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450			
to be performed on BellSouth Premises (e.g., disposition of hazardous material/waste;	Performance of services in accordance with BST's environmental M&Ps	Std T&C 450-B (Contact E/S for copy of appropriate E/S M&Ps.)			
maintenance of storage tanks)	Insurance	Std T&C 660			
Transportation of hazardous material	Compliance with all applicable local, state, & federal laws and regulations	Std T&C 450 Fact Sheet Series 17000			
	Pollution liability insurance	Std T&C 660-3			
	EVET approval of contractor	Approved Environmental Vendor List (Contact E/S Management)			
Maintenance/operations work which may produce a waste	Compliance with all application local, state, & federal laws and regulations	Std T&C 450			
Other maintenance work	Protection of BST employees and equipment	29CFR 1910.147 (OSHA Standard) 29CFR 1910 Subpart O (OSHA Standard)			

		6
Janitorial services	All waste removal and disposal must conform to all applicable federal, state and local regulations	P&SM Manager - Procurement Fact Sheet Series 17000
	All Hazardous Material and Waste Asbestos notification and protection of employees and equipment	GU-BTEN-001BT, Chapter 3 BSP 010-170-001BS (Hazcom)
Manhole cleaning	Compliance with all applicable local, state, & federal laws and regulations Pollution liability insurance EVET approval of contractor	Std T&C 450 Fact Sheet 14050 BSP 620-145-011PR Issue A, August 1996 Std T&C 660-3 Approved Environmental Vendor List (Contact E/S Management)
Removing or disturbing building materials that may contain asbestos	Asbestos work practices	GU-BTEN-001BT, Chapter 3

3. **DEFINITIONS**

<u>Generator</u>. Under RCRA, the person whose act produces a Hazardous Waste, as defined in 40 CFR 261, or whose act first causes a Hazardous Waste to become subject to regulation. The Generator is legally responsible for the proper management and disposal of Hazardous Wastes in accordance with regulations.

<u>Hazardous Chemical</u>. As defined in the U.S. Occupational Safety and Health (OSHA) hazard communication standard (29 CFR 1910.1200), any chemical which is a health hazard or physical hazard.

Hazardous Waste. As defined in section 1004 of RCRA.

<u>Imminent Danger</u>. Any conditions or practices at a facility which are such that a danger exists which could reasonably be expected to cause immediate death or serious harm to people or immediate significant damage to the environment or natural resources.

Spill or Release. As defined in Section 101 of CERCLA.

4. ACRONYMS

E/S – Environmental/Safety

EVET - Environmental Vendor Evaluation Team

 $\underline{DEC/LDEC} \text{ - Department Environmental Coordinator/Local Department Environmental Coordinator}$

<u>GU-BTEN-001BT</u> - BellSouth Environmental Methods and Procedures

NESC - National Electrical Safety Codes

P&SM - Property & Services Management

Std. T&C - Standard Terms & Conditions

THREE MONTH CLEC FORECAST

CLEC NAME DATE	
----------------	--

STATE	Central Office/City	CAG ED Sq. Ft.	CAGELESS # Bays		FRAME TERMINATI ONS	CLEC Provided BDFB Amps Load	BDFB	Heat Dissipation BTU/Hour	# chearing	Proposed Applicatio n Date	NOTES
			Standard Bays*	Non- Standar d Bays**							

^{*}Standard bays are defined as racks, bays or cabinets, including equipment and cable, with measurements equal to or less than the following: Width - 26", Depth - 25". The standard height for all collocated equipment bays in BellSouth is 7'0".

Notes: Forecast information will be used for no other purpose than collocation planning.

Forecast with application dates greater than 3 months from the date of submission will not guarantee the reservation of space in the office requested.

^{**} Any forecast for non-standard cageless bays must include an attachment describing the quantity and width and depth measurements.

ACCESS TO NUMBERS and NUMBER PORTABILITY

1.0 Non-Discriminatory Access to Telephone Numbers

Nothing in this Agreement shall be construed to limit or otherwise adversely affect in any manner either Party's right to employ, or to request and be assigned, any Central Office (NXX) Codes pursuant to the Central Office Code Assignment Guidelines, as may be amended from time to time, or to establish by Tariff or otherwise, Rate Center and Rating Points corresponding to such NXX Codes. BellSouth will provide number portability to ITC^DeltaCom and their customers with minimum impairment of functionality, quality, reliability and convenience.

- 1.1 During the term of this Agreement, the Parties shall contact Lockheed Martin for the assignment of numbering resources. In order to be assigned a Central Office Code, the Parties will be required to complete the Central Office Code (NXX) Assignment Request and Confirmation Form (Code Request Form) in accordance with Industry Numbering Committee's Central Office Code (NXX) Assignment Guidelines (INC 95-0407-008).
- 1.2 It shall be the responsibility of each Party to program and update its own switches and network systems in accordance with the Local Exchange Routing Guide ("LERG") in order to recognize and route traffic to the other Party's assigned NXX Codes at all times.
- 2.0 Local Number Portability
- 2.1 The Parties shall provide Local Number Portability ("LNP") on a reciprocal basis to each other to the extent technically feasible, and in accordance with the applicable rules and regulations as prescribed from time to time by the FCC and/or the Commission.
- 2.2 Local Number Portability
 - 2.2.1 Deployment of LNP. LNP allows End Users to keep their existing Telephone Line Numbers ("TLNs") when switching LECs. The Parties shall implement and deploy the Location Routing Number ("LRN") solution for permanent LNP in accordance with orders, rulings and policies regarding LNP issued by the North American Numbering Committee ("NANC"), the FCC and applicable state commissions, including without limitation, the FCC prescribed permanent LNP geographic deployment schedules.
 - 2.2.2 Description of LNP. LNP uses the industry standard LRN that assigns a unique 10-digit number to each Wire Center. To support LNP, LRN data is stored, and LNP is provisioned on Advanced Intelligent Network ("AIN") elements that replace the dialed TLN with the LRN so that LNP calls can be routed to the proper Wire Center for connection to the dialed party. To obtain the LRN data and properly provision LNP services, carriers must be connected to independently

- operated Regional Number Portability Administration Centers ("NPACs"), which will manage LNP services and provide LNP call routing data to carriers.
- 2.2.3 Once LNP is implemented, either Party will withdraw its Interim Number Portability ("INP") offerings, subject to (i) provision of reasonable advance notice to the other Party; and (ii) coordination to allow the seamless and transparent conversion of INP customers to LNP. Once LNP is implemented in an end office pursuant to FCC or state Commission orders, rules or regulations, with advance written notice, either Party must withdraw its INP offerings. The transition from existing INP arrangements to LNP shall occur within 120 days from the date LNP is implemented in the end office serving the telephone number. Neither Party shall charge the other Party for conversion from INP to LNP. The Parties shall comply with any INP/LNP transition processes established by the FCC and State Commissions and appropriate industry number portability work groups.
- 2.2.4 Notwithstanding the foregoing, the Parties acknowledge that the FCC has determined once LNP has been deployed pursuant to the FCC's orders, rules and regulations, that all local exchange carriers (LECs) have the duty to provide LNP. Therefore, either Party, at any time, may seek appropriate legal or regulatory relief concerning the transition from INP to LNP or other related issues.
- 2.2.5 <u>Charges</u>. The Parties agree to compensate each other for providing LNP in accordance with pertinent rules, orders and charges adopted or approved by the FCC, and effective Tariffs filed in accordance with such FCC requirements. Such charges shall be itemized and clearly designated as "LNP charges."
- 2.2.6 The Parties will exchange SS7 TCAP messages as required for the implementation of Customer Local Area Signaling Services (CLASS) or other features available.

2.3. <u>Interim Number Portability</u>

2.3.1 <u>Service Provider Number Portability</u>

- 2.3.1.1 <u>Definition</u>. Until an industry-wide permanent solution can be achieved, BellSouth shall provide Service Provider Number Portability ("SPNP"). SPNP is an interim service arrangement whereby an end user who switches subscription of his local exchange service from BellSouth to ITC^DeltaCom, or vice versa, is permitted to retain the use of his existing assigned telephone number, provided that the end user remains at the same location for his local exchange service. or changes locations and service providers but stays within the same serving wire center of his existing number.
- 2.3.1.2 <u>Methods of Providing Number Portability</u>. SPNP is available through either remote call forwarding or direct inward dialing trunks, at the election of ITC^DeltaCom. Remote call forwarding (SPNP-RCF) is an existing switch-based BellSouth service that redirects calls within the telephone network. Direct inward

- dialing trunks (SPNP-DID) allow calls to be routed over a dedicated facility to the ITC^DeltaCom switch that serves the subscriber. SS7 Signaling is required for the provision of either of these services.
- 2.3.2 SPNP-DID is available from BellSouth on a per DS0, DS1, or DS3 basis. Where SPNP-DID is technically feasible and is provided on a DS1 or a DS3 basis, the applicable channelization rates are those specified in Attachment 11 hereto, incorporated herein by this reference. SPNP is available only for basic local exchange service.
- 2.4 SPNP is available only where ITC^DeltaCom or BellSouth is currently providing, or will begin providing concurrent with provision of SPNP, basic local exchange service to the affected end user. SPNP for a particular telephone number is available only from the central office originally providing local exchange service to the end user. SPNP for a particular assigned telephone number will be disconnected when any end user, Commission, BellSouth, or ITC^DeltaCom initiated activity (*e.g.*, a change in exchange boundaries) would normally result in a telephone number change had the end user retained his initial local exchange service.
 - 2.4.1 SPNP-RCF, as contemplated by this Agreement, is a telecommunications service whereby a call dialed to an SPNP-RCF equipped telephone number is automatically forwarded to an assigned seven- or ten- digit telephone number within the local calling area as defined in BellSouth's General Subscriber Services Tariff. The forwarded-to number shall be specified by ITC^DeltaCom or BellSouth, as appropriate. The forwarding company will provide identification of the originating telephone number, via SS7 signaling, to the receiving Party. Identification of the originating telephone number to the SPNP-RCF end user cannot be guaranteed, however. SPNP-RCF provides a single call path for the forwarding of no more than one simultaneous call to the receiving Party's specified forwarded-to number. Additional call paths for the forwarding of multiple simultaneous calls are available on a per path basis at separate rates in addition to the rates for SPNP-RCF.
 - 2.4.2 SPNP-DID service, as contemplated by this Agreement, provides trunk side access to end office switches for direct inward dialing to the other company's premises equipment from the telecommunications network to lines associated with the other company's switching equipment and must be provided on all trunks in a group arranged for inward service. A SPNP-DID trunk termination charge, provided with SS7 Signaling only, applies for each trunk voice grade equivalent. In addition, direct facilities are required from the end office where a ported number resides to the end office serving the ported end user customer. The rates for a switched local channel and switched dedicated transport apply as contained in Attachment 11 hereto. Transport mileage will be calculated as the airline distance between the end office where the number is ported and the Point of Interface ("POI") using the V&H coordinate method. SPNP-DID must be established with a minimum configuration of two channels and one unassigned telephone number

per switch, per arrangement for control purposes. Transport facilities arranged for SPNP-DID may not be mixed with any other type of trunk group, with no outgoing calls placed over said facilities. SPNP-DID will be provided only where such facilities are available and where the switching equipment of the ordering company is properly equipped. Where SPNP-DID service is required from more than one wire center or from separate trunk groups within the same wire center, such service provided from each wire center or each trunk group within the same wire center shall be considered a separate service. Only customer-dialed sent-paid calls will be completed to the first number of a SPNP-DID number group; however, there are no restrictions on calls completed to other numbers of a SPNP-DID number group. Interface group arrangements provided for terminating the switched transport at the Party's terminal location are as set forth in of BellSouth's Intrastate Access Services Tariff, § E6.1.3.A as amended from time to time.

- 2.4.3 The calling Party shall be responsible for payment of the applicable charges for sent-paid calls to the SPNP number. For collect, third-Party, or other operatorassisted non-sent paid calls to the ported telephone number, BellSouth or ITC^DeltaCom shall be responsible for the payment of charges under the same terms and conditions for which the end user would have been liable for those charges. Either company may request that the other block collect and third company non-sent paid calls to the SPNP-assigned telephone number. company does not request blocking, the other company will provide itemized local usage data for the billing of non-sent paid calls on the monthly bill of usage charges provided at the individual end user account level. The detail will include itemization of all billable usage. Each company shall have the option of receiving this usage data on a daily basis via a data file transfer arrangement. arrangement will utilize the existing industry uniform standard, known as EMI standards, for exchange of billing data. Files of usage data will be created daily for the optional service. Usage originated and recorded in the sending BellSouth RAO will be provided in unrated or rated format, depending on processing system. ITC^DeltaCom usage originated elsewhere and delivered via CMDS to the sending BellSouth RAO shall be provided in rated format.
- 2.4.4 Each company shall be responsible for obtaining authorization from the end user for the handling of the disconnection of the end user's service, the provision of new local service and the provision of SPNP services. Each company shall be responsible for coordinating the provision of service with the other to assure that its switch is capable of accepting SPNP ported traffic. Each company shall be responsible for providing equipment and facilities that are compatible with the other's service parameters, interfaces, equipment and facilities and shall be required to provide sufficient terminating facilities and services at the terminating end of an SPNP call to adequately handle all traffic to that location and shall be solely responsible to ensure that its facilities, equipment and services do not interfere with or impair any facility, equipment, or service of the other company or any of its end users. In the event that either company determines in its reasonable judgment that the other company will likely impair or is impairing, or interfering

- with any equipment, facility or service or any of its end users, that company may either refuse to provide SPNP service or may terminate SPNP service to the other Party after providing appropriate notice.
- 2.4.5 Each company shall be responsible for providing an appropriate intercept announcement service for any telephone numbers subscribed to SPNP services for which it is not presently providing local exchange service or terminating to an end user. Where either company chooses to disconnect or terminate any SPNP service, that company shall be responsible for designating the preferred standard type of announcement to be provided.
- 2.4.6 Each company shall be the other company's single point of contact for all repair calls on behalf of each company's end user. Each company reserves the right to contact the other company's customers if deemed necessary for maintenance purposes.
- 2.4.7 Neither company shall be responsible for adverse effects on any service, facility or equipment from the use of SPNP services. End-to-end transmission characteristics may vary depending on the distance and routing necessary to complete calls over SPNP facilities and the fact that another carrier is involved in the provisioning of service. Therefore, end-to-end transmission characteristics cannot be specified by either company for such calls. Neither company shall be responsible to the other if any necessary change in protection criteria or in any of the facilities, operation, or procedures of either renders any facilities provided by the other company obsolete or renders necessary modification of the other company's equipment.
- 2.4.8 For terminating IXC traffic ported to either Party which requires use of either Party's 's tandem switching, the tandem provider will bill the IXC tandem switching and its portion of the transport, and the other Party will bill the IXC local switching, the carrier common line, the interconnection charge, and its portion of the transport. If the tandem provider is unable to provide the necessary access records to permit the other company to bill the IXC directly for terminating access to ported numbers, then the tandem provider will bill the IXC full terminating switched access charges at the tandem provider's rate, keep the, tandem switching and its portion of transport, and remit the local switching, the interconnection charge, the other Party's portion of transport at the tandem provider's rate, CCL revenues to the other Party. If an intraLATA toll call is delivered, the delivering Party will pay terminating access rates to the other Party. This subsection does not apply in cases where SPNP-DID is utilized for number portability.
- 2.4.9 If, through an effective order, the Federal Communications Commission ("FCC") issues regulations pursuant to 47 U.S.C. § 251 to require number portability different than that provided pursuant to this section, the Parties will comply with any such order.
- 2.4.10 Charges for INP shall be as specified in Attachment 11.

2.5 INP Requirements

- 2.5.1 Either Party shall notify the other of any technical or capacity limitations that would prevent use of a requested INP implementation in a particular End Office or Wire Center.
- 2.5.2 Either Party shall pass all Calling Party Number ("CPN") or Automatic Number Identification ("ANI") information to and from the ported number, whenever technically feasible.
- 2.5.3 BellSouth and ITC^DeltaCom shall cooperate in resolving all service calls involving the other Party's service, to avoid unnecessary service outages.

2.6 LNP Cutover Procedures.

- 2.6.1 BellSouth will ensure that the disconnect order is completed for all ported numbers once the NPAC notification of ITC^DeltaCom's Activate Subscription Version has been received by BellSouth. If BellSouth receives such notice by 12:00 noon, it will complete the disconnect the same business day. If BellSouth receives such notice after 12:00 noon, it will complete the disconnect the next business day.
- 2.6.2 For an LNP Coordinated Cutover Environment (where the loop is being purchased by ITC^DeltaCom as an unbundled Network Element at the time of LNP implementation), BellSouth shall use best efforts to update switch translations, where necessary, within fifteen (15) minutes after receiving the activate message from NPAC.

For an LNP Non-Coordinated Cutover Environment (where the Loop is supplied by ITC^DeltaCom) BellSouth shall use its best efforts to update switch translations where necessary, within fifteen (15) minutes after receiving the activate message from NPAC.

2.7 Number Portability Through NXX Migration

2.7.1 Where either Party has activated an entire NXX for a single End User, or activated 25% or more of the numbers available in an NXX for a single End User with the remaining numbers in that NXX either reserved for future use or otherwise unused, if such End User chooses to receive service from the other Party, the other Party shall cooperate with the second Party to have the entire NXX reassigned in the Local Exchange Routing Guide ("LERG") (and associated industry databases, routing tables, etc.) to an End Office operated by the second Party. Such transfer will be accomplished with appropriate coordination between the Parties and subject to appropriate industry lead-times for movements of NXX's from one switch to another.

ORDERING AND PROVISIONING

1. Ordering and Provisioning

- 1.1 BellSouth shall provide ordering and provisioning services to ITC^DeltaCom that are equal to the ordering and provisioning services BellSouth provides to itself, any affiliates or subsidiaries or any other CLEC where technically feasible and shall provide reasonable assistance to ITC^DeltaCom as necessary for ITC^DeltaCom to understand how to implement and use all of the OSS functions available to it. BellSouth shall provide information, assistance and access to training at rates as may be specified by BellSouth as necessary to provide ITC^DeltaCom with nondiscriminatory access to BellSouth's OSS. BellSouth shall make available one free seat per year for each OSS system (e.g. LENS, TAG, EDI, and TAFI). Additional training shall be available at rates specified by BellSouth. BellSouth shall provide ITC^DeltaCom with nondiscriminatory access to its Operations Support Systems ("OSS") as necessary to access pre-ordering information, place orders, and obtain maintenance and repair, of both Resale Services and Unbundled Network Elements ("UNEs"). Detailed guidelines for ordering and preordering are set forth in the Ordering Guide for manual ordering and the Local Exchange Ordering Guide for electronic ordering. Except where otherwise required by Commission order, where practicable, BellSouth will notify ITC^DeltaCom of changes to ordering and preordering interfaces and business rules via the appropriate BellSouth web site thirty days prior to such changes. Where thirty (30) days advance notice is not practicable, BellSouth will use its best efforts to provide such notification via the appropriate web site within one (1) day of BellSouth's decision to implement the changes or as soon as possible. In addition, BellSouth will use its best efforts, through the account team assigned to ITC^DeltaCom and upon ITC^DeltaCom's request, to provide such notices via e-mail to the address specified by ITC^DeltaCom.
- 1.2 All changes implemented by the Change Control Process ("CCP") shall be followed by the Parties. Upon request of ITC^DeltaCom for electronic access to the pre-ordering, ordering/provisioning, maintenance/repair and billing functions described herein, BellSouth shall make available to ITC^DeltaCom the following interfaces, without limitation: (i) for ordering, an electronic interface utilizing the Electronic Data Interchange ("EDI") protocol, consistent with the most recent industry standards for such systems established by the Ordering and Billing Forum ("OBF") and the Alliance for Telecommunications Industry Solutions ("ATIS") as determined by the Electronic Interface Change Control Process ("EICCP"); (ii) for pre-ordering and ordering, a -human-to-machine interface known as the Local Exchange Navigation System ("LENS"), and the machine-to-machine interface known as Telecommunications Access Gateway ("TAG"); (iii) facsimile-based and e-mail-based interfaces; (iv) BellSouth's Trouble Analysis and Facilitation Interface ("TAFI"), T I /M I machine-to-machine interface, and Electronic Communication Trouble Administration ("ECTA") interface for maintenance and repair; or (v) any other mutually agreeable method. Each such interface shall be made available to support the ordering of both Resale services and

UNEs (provided that as of the date hereof LENS does not support UNE ordering), and shall be upgraded as necessary to ensure that ITC^DeltaCom is provided access to OSS functions at parity to that provided by BellSouth to itself, its Affiliates or any other Telecommunications Carrier.

- 1.3 Access shall be offered immediately for pre-ordering capability integratable with ordering capability, and for associated maintenance/repair and billing functions. Any such interface shall remain consistent with ATIS, Telecommunications Industry Forum ("TIF"), and the most current industry guideline(s) as determined by the EICCP. Notwithstanding the forgoing, in the event that an industry standard interface is developed by the appropriate industry forum, and is generally accepted for implementation by the industry, then upon agreement by the EICCP, BellSouth shall implement such interface and make it available to ITC^DeltaCom on a timely basis.
- 1.4 When BellSouth is unable to test and implement OSS interfaces for multiple Telecommunications Carriers simultaneously, BellSouth shall engage in such testing and implementation on a "first come-first served" basis.
- 1.5 ITC^DeltaCom may utilize BellSouth electronic interfaces for the purpose of establishing and maintaining Resale services, UNEs, and future uses as they are made available by BellSouth.
- 1.6 When utilizing such OSS functions, the Parties shall at all times adhere to all FCC requirements relating to confidentiality of End-Users' Customer Proprietary Network Information ("CPNI") and in accordance with the terms of the Blanket Letter of Authorization provided to each Party.
- 1.7 BellSouth and ITC^DeltaCom shall jointly establish interface contingency and disaster recovery plans for the pre-order, ordering, provisioning, repair and maintenance of Resale Services and UNEs.
- 1.8 The electronic interfaces described herein shall be utilized for, but not limited to, transferring and receiving orders, Firm Order Confirmations ("FOCs"), completion notices, other electronic error notices, and service jeopardies.
- 1.9 Industry standards bodies and forums (OBF and ECIC) regularly produce updates and new releases to specifications and documentation related to electronic access to OSS functions. Except as otherwise specified in the Agreement, the Parties agree that systems utilized for access to OSS shall be compliant with the most current policies and/or guidelines of OBF and ECIC, as determined by the EICCP.
 - 1.9.1 Neither Party waives its right to participate in, or advocate any position in connection with deliberations of OBF, ATIS-TIF or other industry standards organizations to establish and conform standards for electronic interfaces for pre-ordering, ordering, provisioning, and maintenance and repair. ITC^DeltaCom

and BellSouth shall be individually responsible for evaluating the risk of developing their respective systems in advance of standards and shall support their own system modifications as necessary to comply with new requirements.

- 1.10 In areas where BellSouth does not provide an electronic interface for the pre-order and ordering processes, BellSouth and ITC^DeltaCom shall develop manual work around processes until such time as the transactions can be electronically transmitted. ITC^DeltaCom shall transmit preorder and ordering requests to the Local Carrier Service Center ("LCSC") via facsimile where electronic interfaces are not available or are not functioning.
- 1.11 BellSouth shall provide ITC^DeltaCom personnel with all relevant manuals or other publications, information concerning ordering codes and field identifiers, and information concerning other business rules or practices necessary to ensure nondiscriminatory access to OSS, including all updates, on a timely basis via an electronic means as mutually agreed by the Parties.
- 1.12 BellSouth shall deploy the necessary systems and personnel to provide sufficient access to each of the necessary OSS functions.
- 1.13 BellSouth shall provide ITC^DeltaCom with the technical specifications necessary to instruct ITC^DeltaCom on how to modify or design its systems in a manner that shall enable it to communicate with BellSouth's legacy systems and any interfaces utilized by BellSouth for such access.
- 1.14 BellSouth shall provide ITC^DeltaCom with all of the information necessary to format and process its electronic requests so that these requests flow through the interfaces, the transmission links, and into the legacy systems as quickly and efficiently as possible.
- 1.15 BellSouth shall disclose to ITC^DeltaCom any 'business rules,' including information concerning the ordering codes, that BellSouth uses which ITC^DeltaCom needs to place orders through the system efficiently via BellSouth's Interconnection Web Site in downloadable common spaced value format. Such ordering codes include universal service ordering codes ("USOCs") and field identifiers ("FIDs") used to identify the different services and features used in offering Telecommunications Services to Customers. Throughout the term of this Agreement, the following information will be available on BellSouth's Web Site:
 - 1.15.1 USOC Code. Alphanumeric code that is utilized to provision BellSouth products and services;
 - 1.15.2 USOC Description . English description of each USOC Code;
 - 1.15.3 State Code. State where the USOC is available;

- 1.15.4 Service Type Indicator. Designates whether the USOC is available at the order level: 1.15.5 Line Indicator. Designates whether the USOC is available at the line level: 1.15.6 Feature Charge Code. Designates whether the USOC is available at the feature level: 1.15.7 Resellable Code. Designates whether the USOC is available for resale: 1.15.8 Bus/Res Indicator. Designates whether the USOC is available for Business or Residential: 1.15.9 MRC. Designates appropriate Monthly Recurring Charges for the USOC; 1.15.10 NRC. Designates appropriate Non-Recurring Charges for the USOC; 1.15.11 FIDs. A list of all valid FIDs (File Identifiers) associated with the USOC; and
- 1.16 BellSouth shall ensure that its OSS are designed to accommodate both current demand and projected demand of ITC^DeltaCom and other CLECs in the aggregate for access to OSS functions.
- 1.17 For those OSS functions, if any, that have no retail analogue, BellSouth shall provide access to ITC^DeltaCom that offers ITC^DeltaCom a meaningful opportunity to compete. The specific performance measurements for OSS functions are specified in Attachment 10.
- 1.18 BellSouth shall provide access to OSS functions necessary to order both individual UNEs and those combinations of Network Elements as set forth in Attachment 2.
- 1.19 BellSouth OSS functions for ordering and provisioning shall be able to handle reasonable fluctuations in service orders by competing carriers as well as reasonably foreseeable general increases in ordering volumes.
- 1.20 The Ordering Guide and the Local Exchange Ordering Guide, and associated training and carrier consultation, shall support both Resale services and UNEs.
- 1.21 BellSouth shall provide ITC^DeltaCom notification of disconnects, updated and delivered once daily, via an electronic process known as OUTPLOC.

2.0 Change Management

- 2.1 BellSouth reserves the right to modify or discontinue the use of any OSS interface or version of such interface on the following terms:
 - 2.1.1 With respect to national standard electronic interfaces, upon the release of a new version of such interfaces, BellSouth shall maintain the current national standard version and the previous national standard version.
 - 2.1.2 With respect to discontinuation of electronic interfaces, BellSouth shall provide ITC^DeltaCom with ninety (90) days advance notice of such discontinuation consistent with applicable state and FCC requirements.
 - 2.1.3 With respect to changes or modifications to electronic interfaces other than as specified in 2.1.1 above, BellSouth shall provide prior notice of such changes and modifications and shall use its best efforts to provide ITC^DeltaCom with ninety (90) days advance notice of such changes or modifications. When necessary, the Parties shall work cooperatively to develop a temporary work around solution and to implement such changes and modifications to the electronic interfaces.

3.0 Pre-Ordering Interfaces and Functions

- 3.1 Definition. Pre-ordering is defined as the exchange of information between ITC^DeltaCom and BellSouth relating to current or proposed products, services or UNEs utilized by End Users. Pre-ordering includes the activities undertaken by ITC^DeltaCom to gather and verify information necessary to formulate an accurate order for End Users. As provided hereafter, pre-ordering functions include, without limitation: telephone number selection; street address validation; services and features availability; due date selection; and Customer Service Record ("CSR") information. BellSouth shall provide OSS to ITC^DeltaCom necessary to access such pre-order functions that is at parity with that provided by BellSouth to itself, its Affiliates, or any other Telecommunications Carrier.
- 3.2 Interfaces. BellSouth shall make available the following interfaces to ITC^DeltaCom for access to pre-order functions: LENS and TAG (as provided in Section 1.2 above). LENS shall be available for resale services and TAG shall be available in connection with both Resale services and UNEs.
- 3.3 The Parties acknowledge that ordering requirements necessitate the use of current, real time pre-order information to accurately build service orders. Each pre-order interface shall be available twenty-four (24) hours a day, seven (7) days a week, less reasonable periods required for regular maintenance and scheduled down-time. BellSouth shall use its best efforts to perform maintenance and schedule down-time during evening hours and on weekends.

- 3.4 Preordering Functions. Pre-ordering functions for Resale Services and UNEs shall include, but not be limited to:
 - 3.4.1 Obtain End User information, including, but not limited to, customer name, billing address and residence or business address, billing telephone numbers, billed telephone numbers, features and services available in the End Office where the End User is provisioned and directory listing and delivery information; BellSouth agrees to provide accurate customer service record information that is updated on a daily basis. When a customer selects ITC^DeltaCom as its primary local carrier, BellSouth will, within 24 hours after the conversion date, update the customer service record.
 - 3.4.2 Identify prices for all features and services to which the End User subscribes where required by Commission order.
 - 3.4.3 ITC^DeltaCom may electronically assign a telephone number to its End User. Telephone numbers may be reserved for up to 30 days via LENS and up to 365 days via TAG.
 - 3.4.4 Provide service availability dates (i.e., actual due dates);
 - 3.4.5 Provide information regarding the dispatch/installation schedule, if applicable;
 - 3.4.6 Provide PIC options for intraLATA toll and interLATA toll:
 - 3.4.7 Perform address verification;
 - 3.4.8 Channel Facility Assignment ("CFA"), Network Channel ("NC"), and Network Channel Interface ("NCI") data.

4.0 Ordering/Provisioning Interfaces and Functions

- 4.1 For generation of Resale service orders, ordering flows shall be available via such electronic interfaces for each of the following ordering functions: Conversion ("as is" or "with changes"); Change (features, listings, long distance, etc); New Connect; Disconnect; From and To (change of premises with same service).
 - 4.1.1 BellSouth shall accept any requests from ITC^DeltaCom to disconnect the service of an existing ITC^DeltaCom end user. BellSouth will not require end user confirmation prior to disconnection of the end user's service. If ITC^DeltaCom rescinds such disconnect order or issues a reconnect order within 24 hours of submission of the

- disconnect order, BellSouth shall use its best efforts to reconnect service within 24 hours.
- 4.2 BellSouth shall provide ITC^DeltaCom with a FOC for each Resale and UNE order. As of the date of this Agreement, the FOC includes purchase order number, telephone number, Local Service Request Number, the due date and Service Order number. Any changes to information included in the FOC shall be as determined by the EICCP.
- 4.3 BellSouth shall provision Resale Services and UNEs as prescribed in ITC^DeltaCom's service order requests. Access to status on such electronic orders of Resale services and UNEs shall be provided via the electronic interfaces utilized by ITC^DeltaCom. Status on manual orders shall be provided as mutually agreed by the Parties.
- 4.4 Order Status shall allow ITC^DeltaCom to check service order status, including Due Dates and Customer and Facility Due Date-Jeopardies.
- 4.5 BellSouth shall provide notice of a lack of facilities availability at parity (in terms of means and timing) to that BellSouth provides to itself, its Affiliates, or any other Telecommunications Carrier.

4.6 General Ordering/Provisioning Requirements

- 4.6.1 BellSouth shall provide a single point of contact ("SPOC") for the provisioning of Resale Services (LCSC) and provisioning of UNEs (UNE center) ordered by ITC^DeltaCom. For services and UNEs available electronically, preordering and ordering shall be available via an electronic interface seven (7) days a week, 24 hours a day less reasonable periods for maintenance and scheduled downtime. During provisioning of services to ITC^DeltaCom, support personnel will be available until the migration of the end user is complete. Provisioning services (LCSC and UNE Center) shall be provided during the same business hours that BellSouth provisions services to its own end users. All other ITC^DeltaCom requests for provision and installation services are considered outside of the normal hours of operation and may be performed subject to the application of additional charges.
- 4.6.2 BellSouth shall provide access to assistance for technical issues such as connectivity and passwords related to LENS, TAG and TAFI, and to the "EDI Central Group" for technical problems with EDI. Assistance will be available by telephone during normal business hours and through other contacts on nights, weekends and holidays.
- 4.6.3 BellSouth shall provide the following to ITC^DeltaCom:
 - 4.6.3.1Circuit Layout Record Card and Design Layout Records ("DLRs") for designed unbundled Network Elements;

- 4.6.3.2Upon request of ITC^DeltaCom, advance information on the details and requirements for planning and implementation of NPA splits.
- 4.6.3.3 BellSouth shall provide MSAG to ITC^DeltaCom, as a facilities based carrier, and shall provide updates to MSAG on a quarterly basis. MSAG and the updates thereto shall be provided at no charge.
- 4.6.4 BellSouth and ITC^DeltaCom shall work cooperatively to develop methods and procedures between BellSouth's LCSC and ITC^DeltaCom's corresponding Work Center(s) regarding common systems and work center interfaces.
- 4.6.5 BellSouth and ITC^DeltaCom shall establish mutually acceptable methods and procedures at no charge to the other for handling all misdirected calls from ITC^DeltaCom End Users. All misdirected calls to BellSouth from ITC^DeltaCom End Users shall be given a recording (or a live statement) directing them to call an ITC^DeltaCom designated 800 number. ITC^DeltaCom, on a reciprocal basis, shall refer all misdirected calls that ITC^DeltaCom receives from BellSouth End Users to a BellSouth-designated number.
- 4.6.6 BellSouth shall provide order format specifications to ITC^DeltaCom for all available services, features, and functions and for ancillary data that is necessary to provision these services. Business Rules and EDI Mapping for format and data requirements shall be consistent with industry guidelines and standards.
- 4.6.7 The Parties shall provide a generic intercept referral message that includes any new telephone number of an End User for the same period of time that the Party's provide such service to their end users. The intercept message shall be similar in format to the intercept referral message currently provided by BellSouth for its own End Users. Each Party shall provide this referral service at no charge to the other Party.
- 4.6.8 BellSouth shall perform all pre-testing necessary to insure the services ordered meet the specifications outlined in the technical service description provided by BellSouth for the service being ordered.
- 4.6.9 Any written "leave behind" materials that BellSouth technicians provide to ITC^DeltaCom End Users shall be non-branded materials that do not identify the work being performed as being by BellSouth. These materials shall include, without limitation, non-branded forms for the Customer and non-branded "not at home" cards.

- 4.6.10 If an ITC^DeltaCom End User requests a change of service at the time of installation, BellSouth technicians shall direct them to contact ITC^DeltaCom directly and provide a toll-free (8xx) number supplied by ITC^DeltaCom. When a BellSouth employee visits the premise of an ITC^DeltaCom End User, the BellSouth employee shall inform the Customer that he or she is there acting on behalf of ITC^DeltaCom.
- 4.6.11 BellSouth shall provide telephone and/or facsimile notification of any charges associated with any construction required for a given service, and obtain ITC^DeltaCom's approval prior to commencing construction under an ITC^DeltaCom order for such service or those charges shall be waived.
- 4.6.12 Each Party shall train and direct its employees who have contact with End Users of the other Party, including but not limited to those employees involved in the process of provisioning, maintenance or repair, not to disparage the other Party or its services in any way to the other Party's End Users. Nor shall either Party use these calls to End Users as a basis for internal referrals or to solicit customers to market services. Both Parties shall respond with accurate information in answering customer questions.
- 4.6.13 When ITC^DeltaCom places an LSR, ITC^DeltaCom shall specify a requested Due Date, and BellSouth shall assign a Due Date based on the applicable intervals. In the event ITC^DeltaCom's requested date is less than the standard interval, ITC^DeltaCom shall contact BellSouth by telephone and the Parties shall negotiate an expedited Due Date. This situation shall be considered an expedited order. In the event the negotiated Due Date assigned by BellSouth is within the standard interval, the order will not be considered expedited. BellSouth shall not complete the order prior to the Due Date unless authorized by ITC^DeltaCom. BellSouth shall use its best efforts to notify ITC^DeltaCom of any known jeopardies prior to the scheduled conversion. BellSouth shall also promptly notify ITC^DeltaCom of the revised installation Due Date if known at the time of the jeopardy notice. If ITC^DeltaCom requests that an order be expedited, BellSouth shall notify ITC^DeltaCom of the status of the order within the expedited interval (i) by the end of the same Business Day when such expedite requests are made prior to noon; or (ii) by noon the following Business Day otherwise.
- 4.6.14 ITC^DeltaCom and BellSouth shall agree to escalation procedures and contacts for resolving issues related to ordering and provisioning procedures or to the processing of individual orders, subject ultimately to the dispute resolution provisions of this Agreement. BellSouth shall use its best efforts to notify ITC^DeltaCom of any modifications to these contacts within ten business (10) days of any such modifications.

- 4.6.15 BellSouth shall transmit to ITC^DeltaCom a FOC or, in the alternative, notification of the lack of available facilities within time periods specified herein after BellSouth's receipt of a complete and correct LSR from ITC^DeltaCom, provided, however, that an LSR for complex services requiring a service inquiry shall be deemed received for these purposes only after completion of the service inquiry. The FOC shall contain a due date which shall be established on a nondiscriminatory basis with respect to installation dates for comparable orders at such time. If ITC^DeltaCom uses EDI, or any other electronic interface for the submission of the LSR, the FOC or notification shall be posted by BellSouth in EDI or in the electronic interface used within 24 hours of receipt of the LSR. If ITC^DeltaCom does not use an electronic interface, or, an electronic interface is not available for the service or UNE being ordered, BellSouth shall transmit the FOC or notification by facsimile to a toll-free number provided by ITC^DeltaCom within 48 hours of BellSouth's receipt of the LSR. When ITC^DeltaCom submits a complete and correct LSR for INP and an associated unbundled Loop simultaneously, BellSouth shall likewise issue a FOC for both the Loop and the INP simultaneously.
- 4.6.16 BellSouth shall notify ITC^DeltaCom via electronic interface, of Rejections/Errors contained in any of the data element(s) fields contained on any ITC^DeltaCom electronic Service Request. If the electronic interface is unavailable or malfunctioning, BellSouth shall notify ITC^DeltaCom by telephone, facsimile, or email as mutually agreed to by the Parties, of such Rejections and Errors.
- 4.6.17 ITC^DeltaCom shall specify on each LSR its Desired Due Date (DDD) for completion of that particular order. BellSouth shall not complete the order prior to DDD unless early turn-up is needed for testing purposes. BellSouth shall notify ITC^DeltaCom if the DDD cannot be met. BellSouth shall exercise best efforts to meet the DDD for Network Element requests.
- 4.6.18 <u>Use of Facilities</u>. When a customer of a ITC^DeltaCom elects to discontinue service and transfer service to another local exchange carrier, including BellSouth, BellSouth shall have the right to reuse the facilities provided to CLEC by BellSouth for retail or resale service, unbundled loop and/or unbundled port for that customer. In addition, BellSouth may disconnect and reuse facilities when the facility is in a denied state and BellSouth has received an order to establish new service or transfer of service from a customer or a customer's CLEC at the same address served by the denied facility. In the event of such reuse of facilities, BellSouth and ITC^DeltaCom shall coordinate such transition to avoid any service outage to the end users.
- 4.6.18.1 Upon receipt of a service order, BellSouth will do the following:
 - 4.6.18.1.1 Process disconnect and reconnect orders to provision the service which shall be due dated using current interval guidelines.

- 4.6.18.1.2 Reuse the serving facility for the retail, resale service, or unbundled network element at the same location.
- 4.6.18.1.3 Notify ITC^DeltaCom subsequent to the disconnect order being completed.
- 4.6.19 BellSouth shall provision UNEs with the same timeliness that the same or similar facilities are provisioned to BellSouth's Affiliates, or other Persons to whom BellSouth directly provides such facilities.
- 4.6.20 When available, BellSouth shall provide ITC^DeltaCom with the ability to have BellSouth end offices AIN triggers initiated via an electronic service order from ITC^DeltaCom.
- 4.6.21 ITC^DeltaCom may order from BellSouth multiple individual UNEs on a single order without the need for ITC^DeltaCom to send an order for each such UNE, if such UNEs are (i) for a single type of service, (ii) for a single location and (iii) for the same account.
- 4.6.22 BellSouth shall recognize ITC^DeltaCom as the Customer of Record for all UNEs ordered by ITC^DeltaCom and shall send all notices, invoices and pertinent Customer information directly to ITC^DeltaCom.
- 4.6.23 Within two (2) hours of ITC^DeltaCom's request, BellSouth shall use its best efforts to perform cooperative testing with ITC^DeltaCom (including trouble shooting to isolate any problems) to test UNEs purchased by ITC^DeltaCom in order to identify any performance problems.
- 4.6.24 Order Flow Through. Order Flow Through is defined as the process whereby ITC^DeltaCom's orders are transmitted electronically through the gateway and accepted into BellSouth's back office order systems without manual intervention.
- 4.6.25 BellSouth will provide ITC^DeltaCom with one line equipped for Remote Call Forwarding in certain BellSouth switches as set forth in this Section for ITC^DeltaCom's use solely to conduct verification testing of routing and translations instructions in BellSouth's switches. Such verification is intended to confirm that BellSouth has implemented proper routing and translations instructions in the BellSouth switch so that calls from end users served by such BellSouth switch will be properly routed to NXX codes assigned to ITC^DeltaCom's switch. The Remote Call Forwarding line will be equipped with one call path. Provision of such lines will generally coincide with ITC^DeltaCom's installation of new switches where ITC^DeltaCom assigns new NXX codes to such switches. In such cases, the Remote Call Forwarding lines will be provided to ITC^DeltaCom in sufficient time for ITC^DeltaCom to perform its verification testing prior to NXX code activation. Thereafter, ITC^DeltaCom may use the Remote Call Forwarding lines to perform NXX code verification

testing. Remote Call Forwarding lines will also be provided to ITC^DeltaCom as needed for verification testing relating to existing switches. ITC^DeltaCom will pay the same non-recurring and recurring rates for these Remote Call Forwarding lines as ITC^DeltaCom pays for Remote Call Forwarding used in conjunction with Interim Number Portability as set forth in Attachment 11 of this Agreement.

4.7 **UNE Conversions**

- 4.7.1 BellSouth agrees to coordinate with ITC^DeltaCom at least forty-eight hours prior to the due date a scheduled conversion date and time.
- 4.7.2 Twenty-four (24) to forty-eight (48) hours in advance of a loop cutover, BellSouth will conduct testing to ensure that dial tone from ITC^DeltaCom is available for requested loops. If dial tone is not available from ITC^DeltaCom, BellSouth will so notify ITC^DeltaCom.
- 4.7.3If ITC^DeltaCom requests or approves that a BellSouth technician perform services in excess of those necessary for the conversion of "live" Telephone Exchange Services to UNEs, BellSouth may charge ITC^DeltaCom for any additional reasonable labor charges to perform such services.
- 4.7.4 Notwithstanding any other provision hereof, the performance/intervals for installation of unbundled Loops should not exceed the interval for reestablishing service for the customer with BellSouth.

5.0 <u>Maintenance/Repair Interfaces and Functions</u>

- 5.1 BellSouth shall make available electronic interfaces to ITC^DeltaCom for maintenance, trouble reporting, and repair, including initiation of trouble tickets, updates/changes, status checking, scheduling maintenance appointments, and cancellation, for both Resale services and UNEs. Ongoing maintenance practices on unbundled loops shall equal the practices employed by BellSouth for facilities used to provide retail services. BellSouth will use its best efforts to ensure that the mean time to repair unbundled loops shall be equivalent to the mean time to repair reported by BellSouth for its retail customers.
- 5.2 BellSouth's maintenance systems and databases will allow ITC^DeltaCom maintenance personnel and customer service representatives to perform the following functions for ITC^DeltaCom Customers: (i) enter a new customer trouble ticket into the BellSouth maintenance system for an ITC^DeltaCom Customer; (ii) retrieve and track current status on all ITC^DeltaCom Customer repair tickets; (iii) receive "estimated time to repair" ("ETTR") on a real-time basis; (iv) perform where appropriate an electronic test

- at the time of ticket entry and provide test results to ITC^DeltaCom; and (v) electronic notification when trouble is cleared.
- 5.3 If an electronic interface is not available, BellSouth agrees that ITC^DeltaCom may transmit repair calls to BellSouth's repair bureau and request dispatching a BellSouth technician to an ITC^DeltaCom customer's premises by telephone. BellSouth agrees to provide the status upon ITC^DeltaCom's request, in an expedient manner. The speed of answer time for ITC^DeltaCom shall be equal to that for BellSouth.
- BellSouth shall prioritize ITC^DeltaCom end users for purposes of repair in the same manner and within the same time frames that BellSouth prioritizes its own end users for repair as described in this Section. In disaster situations, BellSouth follows the FCC's Emergency Preparedness Restoration Guidelines (010-400-002 BT). In non-disaster situations, BellSouth shall expedite, at no additional charge, the repair of ITC^DeltaCom's UNEs and services where ITC^DeltaCom reports to BellSouth that the UNE or service serves an emergency facility, an end user requiring access to emergency facilities via telecommunications services or another high priority end user. ITC^DeltaCom agrees to submit expedite reports pursuant to this Section only in the circumstances described herein, and any such report for prioritization shall be made in good faith. The Parties shall comply with the Disaster Recovery Plan as set forth in Exhibit A of this Attachment.
- 5.5 BellSouth agrees to advise ITC^DeltaCom of any central office failure or other major service interruptions that are known at the time of an inquiry or trouble report.
- 5.6 BellSouth agrees to provide, via electronic interface, an Estimated Time to Repair ("ETTR") on all trouble reports submitted electronically, an appointment time or a commitment time, as appropriate.
- 5.7 The Parties shall insure that all technicians and representatives are properly trained and that they follow such procedures in all their communications with End Users. At a minimum, the aforementioned procedures shall assume that: (1) BellSouth technicians shall provide repair service that is at least equal in quality to that provided to BellSouth customers or any other entity; (2) maintenance and repair shall take place based on a prioritization schedule devised by mutual agreement of the parties; (3) Customers shall be restored to service based on the priority system devised by mutual agreement of the parties on a non-discriminatory basis; and (4) ITC^DeltaCom may prioritize repair scheduling of its own customers through an escalation procedure.
- 5.8 The BellSouth repair bureau including the Electronic Interface, shall be on-line and operational twenty-four (24) hours per day, seven (7) days per week except for scheduled electronic interface downtime.
- 5.9 Service centers shall be established by both Parties to handle service issues, escalations, and resolution of billing issues and other administrative problems.

- 5.10 The Parties agree to adopt a process for the efficient management of misdirected service calls.
- 5.11 BellSouth shall perform Mechanized Unbundled Loop Tests ("Quick Test") at the request of ITC^DeltaCom while ITC^DeltaCom is on line.
- 5.12 BellSouth shall close all trouble reports with ITC^DeltaCom, within 24 hours of resolution of the trouble. ITC^DeltaCom shall close all trouble reports with the End User. BellSouth's outside technicians shall clear troubles to the network interface and provide callback from the fault location to ITC^DeltaCom.
- 5.13 BellSouth shall not undertake any work at an End User's request for which ITC^DeltaCom would be charged without obtaining the prior approval of ITC^DeltaCom. This includes authorizations by ITC^DeltaCom if a dispatch is required to the customer premises as well as verification of actual work completed.
- 5.14 All Auto/Subscriber Line Tests ("ALT/SLT") tests performed on ITC^DeltaCom customers that result in a failure shall be reported to ITC^DeltaCom.
- 5.15 ITC^DeltaCom shall coordinate dispatches to the customer premise. This includes redispatches for customer not-at-home.
- 5.16 BellSouth shall respond to ITC^DeltaCom customer alarms consistent with how and when they respond to alarms for their own customers. BellSouth shall ensure that all applicable alarm systems that support ITC^DeltaCom customers are operational and the supporting databases are accurate so that equipment that is in alarm will be promptly identified.
- 5.17 BellSouth will use best efforts to notify ITC^DeltaCom, of any scheduled maintenance activity performed by BellSouth that may be service affecting to ITC^DeltaCom local customers (i.e., cable throws, power tests, etc.).
- 5.18 The Parties agree to establish a special emergency escalation procedure for use in situations involving customer out-of-service situations.
- 5.19 In facility and power outage situations, BellSouth agrees to provide UNEs leased by ITC^DeltaCom the same priority for maintenance and restoral as similar elements used by BellSouth for itself or its Affiliates.
- 5.20 BellSouth shall notify ITC^DeltaCom at parity with its own retail units in the event any repair person is unable to be present for, or anticipates missing, a scheduled repair opportunity.

BELLSOUTH DISASTER RECOVERY PLANNING

for

CLECS

<u>CONTENTS</u>	PAGE
1.0 Purpose	3
2.0 Single Point of Contact.	3
3.0 Identifying the Problem	3
3.1 Site Control	4
3.2 Environmental Concerns	5
4.0 The Emergency Control Center (ECC)	6
5.0 Recovery	6
Procedures	
5.1 CLEC Outage	7
5.2 BellSouth Outage	7
5.3 Combined Outage (CLEC and BellSouth Equipment)	10
6.0 T1 Identification Procedures	10
7.0 Acronyms	11

1.0 PURPOSE

In the unlikely event of a disaster occurring that affects BellSouth's long-term ability to deliver traffic to a Competitive Local Exchange Carrier (CLEC), general procedures have been developed to hasten the recovery process. Since each location is different and could be affected by an assortment of potential problems, a detailed recovery plan is impractical. However, in the process of reviewing recovery activities for specific locations, some basic procedures emerge that appear to be common in most cases.

These general procedures should apply to any disaster that affects the delivery of traffic for an extended time period. Each CLEC will be given the same parity consideration during an outage and service will be restored as quickly as possible.

This document will cover the basic recovery procedures that would apply to every CLEC.

2.0 SINGLE POINT OF CONTACT

When a problem is experienced, regardless of the severity, the BellSouth Network Management Center (NMC) will observe traffic anomalies and begin monitoring the situation. Controls will be appropriately applied to insure the sanity of BellSouth's network; and, in the event that a switch or facility node is lost, the NMC will attempt to circumvent the failure using available reroutes.

BellSouth's NMC will remain in control of the restoration efforts until the problem has been identified as being a long-term outage. At that time, the NMC will contact BellSouth's Emergency Control Center (ECC) and relinquish control of the recovery efforts. Even though the ECC may take charge of the situation, the NMC will continue to monitor the circumstances and restore traffic as soon as damaged network elements are revitalized.

The telephone number for the BellSouth Network Management Center in Atlanta, as published in Telcordia's National Network Management Directory, is 404-321-2516. The telephone number for ITC^DeltaCom's Switch Control Center is 888-430-2722.

3.0 IDENTIFYING THE PROBLEM

During the early stages of problem detection, the NMC will be able to tell which CLECs are affected by the catastrophe. Further analysis and/or first hand observation will determine if the disaster has affected CLEC equipment only; BellSouth equipment only or a combination. The initial restoration activity will be largely determined by the equipment that is affected.

Once the nature of the disaster is determined and after verifying the cause of the problem, the NMC will initiate reroutes and/or transfers that are jointly agreed upon by the affected CLECs' Network Management Center and the BellSouth NMC. The type and percentage of controls used will depend upon available network capacity. Controls necessary to stabilize the situation will be invoked and the NMC will attempt to re-establish as much traffic as possible.

For long term outages, recovery efforts will be coordinated by the Emergency Control Center (ECC). Traffic controls will continue to be applied by the NMC until facilities are re-established. As equipment is made available for service, the ECC will instruct the NMC to begin removing the controls and allow traffic to resume.

3.1 SITE CONTROL

In the total loss of building use scenario, what likely exists will be a smoking pile of rubble. This rubble will contain many components which could be dangerous. It could also contain any personnel on the premises at the time of the disaster. For these reasons, the local fire marshal with the assistance of the police will control the site until the building is no longer a threat to surrounding properties and the companies have secured the site from the general public.

During this time, the majority owner of the building should be arranging for a demolition contractor to mobilize to the site with the primary objective of reaching the cable entrance facility for a damage assessment. The results of this assessment would then dictate immediate plans for restoration, both short term and permanent.

In a less catastrophic event, i.e., the building is still standing and the cable entrance facility is usable, the situation is more complex. The site will initially be controlled by local authorities until the threat to adjacent property has diminished. Once the site is returned to the control of the companies, the following events should occur.

An initial assessment of the main building infrastructure systems (mechanical, electrical, fire & life safety, elevators, and others) will establish building needs. Once these needs are determined, the majority owner should lead the building restoration efforts. There may be situations where the site will not be totally restored within the confines of the building. The companies must individually determine their needs and jointly assess the cost of permanent restoration to determine the overall plan of action.

Multiple restoration trailers from each company will result in the need for designated space and installation order. This layout and control is required to maximize the amount of restoration equipment that can be placed at the site, and the priority of placements.

Care must be taken in this planning to insure other restoration efforts have logistical access to the building. Major components of telephone and building equipment will need to be removed and replaced. A priority for this equipment must also be jointly established to facilitate overall site restoration. (Example: If the AC switchgear has sustained damage, this would be of the highest priority in order to regain power, lighting, and HVAC throughout the building.)

If the site will not accommodate the required restoration equipment, the companies would then need to quickly arrange with local authorities for street closures, rights of way or other possible options available.

3.2 ENVIRONMENTAL CONCERNS

In the worse case scenario, many environmental concerns must be addressed. Along with the police and fire marshal, the state environmental protection department will be on site to monitor the situation.

Items to be concerned with in a large central office building could include:

- 1. Emergency engine fuel supply. Damage to the standby equipment and the fuel handling equipment could have created "spill" conditions that have to be handled within state and federal regulations.
- 2. Asbestos containing materials that may be spread throughout the wreckage. Asbestos could be in many components of building, electrical, mechanical, outside plant distribution, and telephone systems.
- 3. Lead and acid. These materials could be present in potentially large quantities depending upon the extent of damage to the power room.
- 4. Mercury and other regulated compounds resident in telephone equipment.
- 5. Other compounds produced by the fire or heat.

Once a total loss event occurs at a large site, local authorities will control immediate clean up (water placed on the wreckage by the fire department) and site access.

At some point, the companies will become involved with local authorities in the overall planning associated with site clean up and restoration. Depending on the clean up approach taken, delays in the restoration of several hours to several days may occur.

In a less severe disaster, items listed above are more defined and can be addressed individually depending on the damage.

In each case, the majority owner should coordinate building and environmental restoration as well as maintain proper planning and site control.

4.0 THE EMERGENCY CONTROL CENTER (ECC)

The ECC is located in the Colonnade Building in Birmingham, Alabama. During an emergency, the ECC staff will convene a group of pre-selected experts to inventory the damage and initiate corrective actions. These experts have regional access to BellSouth's personnel and equipment and will assume control of the restoration activity anywhere in the nine state area.

In the past, the ECC has been involve with restoration activities resulting from hurricanes, ice storms and floods. They have demonstrated their capabilities during these calamities as well as during outages caused by human error or equipment failures. This group has an excellent record of restoring service as quickly as possible.

During a major disaster, the ECC may move emergency equipment to the affected location, direct recovery efforts of local personnel and coordinate service restoration activities with the CLECs. The ECC will attempt to restore service as quickly as possible using whatever means are available; leaving permanent solutions, such as the replacement of damaged buildings or equipment, for local personnel to administer.

Part of the ECC's responsibility, after temporary equipment is in place, is to support the NMC efforts to return service to the CLECs. Once service has been restored, the ECC will return control of the network to normal operational organizations. Any long-term changes required after service is restored will be made in an orderly fashion and will be conducted as normal activity.

5.0 RECOVERY PROCEDURES

The nature and severity of any disaster will influence the recovery procedures. One crucial factor in determining how BellSouth will proceed with restoration is whether or not BellSouth's equipment is incapacitated. Regardless of who's equipment is out of service, BellSouth will move as quickly as possible to aid with service recovery; however, the approach that will be taken may differ depending upon the location of the problem.

5.1 CLEC OUTAGE

For a problem limited to one CLEC (or a building with multiple CLECs), BellSouth has several options available for restoring service quickly. For those CLECs that have agreements with other CLECs, BellSouth can immediately start directing traffic to a provisional CLEC for completion. This alternative is dependent upon BellSouth having concurrence from the affected CLECs.

Whether or not the affected CLECs have requested a traffic transfer to another CLEC will not impact BellSouth's resolve to re-establish traffic to the original destination as quickly as possible.

5.2 BELLSOUTH OUTAGE

Because BellSouth's equipment has varying degrees of impact on the service provided to the CLECs, restoring service from damaged BellSouth equipment is different. The outage will probably impact a number of Carriers simultaneously. However, the ECC will be able to initiate immediate actions to correct the problem.

A disaster involving any of BellSouth's equipment locations could impact the CLECs, some more than others. A disaster at a Central Office (CO) would only impact the delivery of traffic to and from that one location, but the incident could affect many Carriers. If the Central Office is a Serving Wire Center (SWC), then traffic from the entire area to those Carriers served from that switch would also be impacted. If the switch functions as an Access Tandem, or there is a tandem in the building, traffic from every CO to every CLEC could be interrupted. A disaster that destroys a facility hub could disrupt various traffic flows, even though the switching equipment may be unaffected.

The NMC would be the first group to observe a problem involving BellSouth's equipment. Shortly after a disaster, the NMC will begin applying controls and finding reroutes for the completion of as much traffic as possible. These reroutes may involve delivering traffic to alternate Carriers upon receiving approval from affected carriers and notification of the CLECs involved. In some cases, changes in translations will be required. If the outage is caused by the destruction of equipment, then the ECC will assume control of the restoration.

5.2.1 Loss of a Central Office

When BellSouth loses a Central Office, the ECC will

- a) place specialists and emergency equipment on notice;
- b) inventory the damage to determine what equipment and/or functions are lost;
- c) move containerized emergency equipment and facility equipment to the stricken area, if necessary;
- d) begin reconnecting service for Hospitals, Police and other emergency agency customers of CLECs and BellSouth in a nondiscriminatory manner in accordance with NSEP-TSP guidelines; and
- e) begin restoring service to CLECs and other customers.

5.2.2 Loss of a Central Office with Serving Wire Center Functions

The loss of a Central Office that also serves as a Serving Wire Center (SWC), will be restored as described in section 5.2.1.

5.2.3 Loss of a Central Office with Tandem Functions

When BellSouth loses a Central Office building that serves as an Access Tandem and as a SWC, the ECC will

- a) place specialists and emergency equipment on notice;
- b) inventory the damage to determine what equipment and/or functions are lost;
- c) move containerized emergency equipment and facility equipment to the stricken area, if necessary;

- d) begin reconnecting service for Hospitals, Police and other emergency agency customers of CLECs and BellSouth in a nondiscriminatory manner in accordance with NSEP-TSP guidelines; and
- e) redirect as much traffic as possible to the alternate access tandem (if available) for delivery to those CLECs utilizing a different location as a SWC;
- f) begin aggregating traffic to a location near the damaged building. From this location, begin reestablishing trunk groups to the CLECs for the delivery of traffic normally found on the direct trunk groups. (This aggregation point may be the alternate access tandem location or another CO on a primary facility route.)
- g) begin restoring service to CLECs and other customers.

5.2.4 Loss of a Facility Hub

In the event that BellSouth loses a facility hub, the recovery process is much the same as above. Once the NMC has observed the problem and administered the appropriate controls, the ECC will assume authority for the repairs. The recovery effort will include

- a) placing specialists and emergency equipment on notice;
- b) inventorying the damage to determine what equipment and/or functions are lost;
- c) moving containerized emergency equipment to the stricken area, if necessary;
- d) reconnecting service for Hospitals, Police and other emergency agency customers of CLECs and BellSouth in a nondiscriminatory manner in accordance with NSEP-TSP guidelines; and
- e) restoring service to CLECs and other customers. If necessary, BellSouth will aggregate the traffic at another location and build temporary facilities. This alternative would be viable for a location that is destroyed and building repairs are required.

5.3 COMBINED OUTAGE (CLEC AND BELLSOUTH EQUIPMENT)

In some instances, a disaster may impact BellSouth's equipment as well as the CLECs'. This situation will be handled in much the same way as described in section 5.2.3. Since BellSouth and the CLECs will be utilizing temporary equipment, close coordination will be required.

6.0 T1 IDENTIFICATION PROCEDURES

Exhibit A of Attachment 6 Page 9

During the restoration of service after a disaster, BellSouth may be forced to aggregate traffic for delivery to an CLEC. During this process, T1 traffic may be consolidated onto DS3s and may become unidentifiable to the Carrier. Because resources will be limited, BellSouth may be forced to "package" this traffic entirely differently then normally received by the CLECs. Therefore, a method for identifying the T1 traffic on the DS3s and providing the information to the Carriers is required.

7.0 ACRONYMS

CO - Central Office (BellSouth)

DS3 - Facility that carries 28 T1s (672 circuits)

ECC - Emergency Control Center (BellSouth)

CLEC - Competitive Local Exchange Carrier

NMC - Network Management Center

SWC - Serving Wire Center (BellSouth switch)

T1 - Facility that carries 24 circuits

BILLING AND BILLING ACCURACY CERTIFICATION

1. Payment and Billing Arrangements

- Billing. Currently, BellSouth provides billing through the Carrier Access Billing System (CABS) and through the Customer Records Information System (CRIS) depending on the particular service(s) that ITC^DeltaCom requests. BellSouth will bill and record in accordance with this agreement those charges ITC^DeltaCom incurs as a result of ITC^DeltaCom purchasing from BellSouth Network Elements, Combinations, and Local Services, as set forth in this agreement. BellSouth will format all bills in CBOS Standard or CLUB/EDI format, depending on the type of service ordered. BellSouth's bills to ITC^DeltaCom for unbundled network elements and resold services purchased by ITC^DeltaCom shall include the item, quantity and price of such purchased services. For those services where standards have not yet been developed, BellSouth's billing format will change as necessary when standards are finalized by the Ordering and Billing Forum (OBF).
 - 1.1.1 If ITC^DeltaCom requests multiple billing media or additional copies of bills BellSouth will provide these at a reasonable cost.
- Master Account. If ITC^DeltaCom has not already done so, after receiving certification as a local exchange company from the appropriate regulatory agency, ITC^DeltaCom will provide the appropriate BellSouth service center the necessary documentation to enable BellSouth to establish a master account for resold services. Such documentation shall include the Application for Master Account, proof of authority to provide telecommunications services, an Operating Company Number ("OCN") assigned by the National Exchange Carriers Association ("NECA") and a tax exemption certificate, if applicable. To the extent applicable, BellSouth shall maintain an account under which ITC^DeltaCom shall bill BellSouth for services provided pursuant to this Agreement.
- 1.3 Payment Responsibility. Payment of all charges will be the responsibility of ITC^DeltaCom or BellSouth as applicable. ITC^DeltaCom and BellSouth shall make payment to each other for all services billed. Neither Party shall be responsible for payments not received by the other Party's customers. Neither Party shall become involved in billing disputes that may arise between the other Party and its customers. Payments made by either Party as payment on account shall be credited to an accounts receivable master account and not to an end user's account.
- 1.4 <u>Payment Due</u>. The payment shall be due by the next bill date (i.e., same date in the following month as the bill date) ("Payment Due Date") and is

payable in immediately available funds. Payment is considered to have been made when received.

- 1.4.1 If the payment due date falls on a Sunday or on a Holiday which is observed on a Monday, the payment due date shall be the first non-Holiday day following such Sunday or Holiday. If the payment due date falls on a Saturday or on a Holiday which is observed on Tuesday, Wednesday, Thursday, or Friday, the payment due date shall be the last non-Holiday day preceding such Saturday or Holiday. If payment is not received by the payment due date, a late payment penalty, as set forth in Section 1.7, below, shall apply.
- 1.5 <u>Tax Exemption</u>. Upon proof of tax exempt certification, the total amount billed shall not include any taxes due from the end user. The Retail Service provider shall be solely responsible for the computation, tracking, reporting and payment of all federal, state and/or local jurisdiction taxes associated with the services resold to the end user.
- 1.6 <u>Miscellaneous</u>. As the customer of record for resold services, ITC^DeltaCom shall be responsible for, and remit to BellSouth, all charges applicable to its resold services for emergency services (E911 and 911) and Telecommunications Relay Service (TRS) as well as any other charges of a similar nature.
- 1.7 <u>Late Payment</u>. If any portion of the payment is received by the billing party after the payment due date as set forth preceding, or if any portion of the payment is received in funds that are not immediately available to the billing party, then a late payment penalty shall be due to the billing party. The late payment penalty shall be the portion of the payment not received by the payment due date times a late factor. The late factor shall be as set forth in Section A2 of BellSouth's General Subscriber Service Tariff, Section B2 of BellSouth's Private Line Service Tariff or Section E2 of the Intrastate Access Tariff or the applicable ITC^DeltaCom tariff as appropriate.
- 1.8 Access Charges for Resold Services. Any switched access charges associated with interexchange carrier access to the resold local exchange lines will be billed by, and due to, BellSouth. No additional charges are to be assessed to ITC^DeltaCom.
- 1.9 End User Common Line Charge for Resold Services. Pursuant to 47 CFR Section 51.617, BellSouth will bill ITC^DeltaCom end user common line charges identical to the end user common line charges BellSouth bills its end users.

- 1.10 <u>Discontinuing Service</u>. The procedures for discontinuing service to ITC^DeltaCom or BellSouth are as follows:
 - 1.10.1 Each party reserves the right to suspend or terminate service for nonpayment in accordance with applicable state and federal regulations.
 - 1.10.2 If payment of account is not received by the bill day in the month after the original bill day, the billing Party may provide written notice to the other Party that additional applications for service will be refused and that any pending orders for service will not be completed if payment is not received by the fifteenth day following the date of the notice. In addition the billing party may, at the same time, give thirty days notice to the person designated by the other party to receive notices of noncompliance, to discontinue the provision of existing services at any time thereafter.
 - 1.10.3 In the case of such discontinuance, all billed charges, as well as applicable termination charges, shall become due.
 - 1.10.4 If the billing party does not discontinue the provision of the services involved on the date specified in the thirty days notice and the other Party's noncompliance continues, nothing contained herein shall preclude the billing party's right to discontinue the provision of the services without further notice.
 - 1.10.5 If payment is not received or satisfactory arrangements made for payment by the date given in the written notification, the billed party's services may be discontinued. Upon discontinuance of service on the billed party's account, service to the billed party's end users will be denied. The billing party will reestablish service at the request of the end user or the other Party upon payment of the appropriate connection fee and subject to the billing party's 's normal application procedures. The billed party is solely responsible for notifying the end user of the proposed service disconnection.
 - 1.10.6 If within fifteen days after an end user's service has been denied no contact has been made in reference to restoring service, the end user's service shall be disconnected.
- 1.11 <u>Deposit Policy</u>. Either Party may be required to provide information regarding credit worthiness. If either Party repeatedly fails to pay undisputed billed charges by the Payment due date or its financial condition deteriorates materially, such party may be required to provide information regarding credit worthiness. Based on the results of the credit

analysis, the Party providing services reserves the right to secure the account with a suitable form of security deposit. Such security deposit may take the form of an irrevocable Letter of Credit or in its sole discretion some other form of security acceptable to the Party providing service. Any such security deposit shall in no way release the customer from its obligation to make complete and timely payments of its bill. Such security may be required prior to the inauguration of service. If, in the sole opinion of the Party providing service, circumstances so warrant and/or gross monthly billing has increased beyond the level initially used to determine the level of security, the Party providing service reserves the right to request additional security. Interest on a security deposit shall accrue and be refunded in accordance with the terms in the appropriate tariff of the Party providing service.

2. <u>Billing and Billing Accuracy Certification</u>

- 2.1 At the option of ITC^DeltaCom, BellSouth and ITC^DeltaCom shall mutually agree upon a billing quality assurance program for all billing elements covered in this Agreement that shall eliminate the need for post-billing reconciliation. Appropriate terms for access to any BellSouth documents, systems, records, and procedures for the recording and billing of charges shall be part of that program.
- 2.2 As part of the billing quality assurance program, BellSouth and ITC^DeltaCom will develop standards, measurements, and performance requirements for a local billing measurements process. On a regular basis the billing party will provide the other party with mutually agreed upon performance measurement data that substantiates the accuracy, reliability, and integrity of the billing process for local billing. In return, each party shall pay all bills received from the other party in full by the payment due date.
- 2.3 Local billing discrepancies will be addressed in an orderly manner via a mutually agreed upon billing exemption process.
 - 2.3.1 Each party agrees to notify the other Party upon identifying a billing discrepancy. The Parties shall endeavor to resolve any billing discrepancy within sixty (60) calendar days of the notification date. A mutually agreed upon escalation process shall be established for resolving local billing discrepancies as part of the billing quality assurance program.
 - 2.3.2 Closure of a specific billing period shall occur by joint agreement of the Parties whereby the Parties agree that such billing period is closed to any further analysis and financial transactions except those resulting from regulatory mandates. Closure will take place

within a mutually agreed upon time interval from the Bill Date. The month being closed represents those charges that were billed or should have been billed by the designated Bill Date.

3. <u>Billing Disputes</u>

- 3.1 Where the parties have not agreed upon a billing quality assurance program, billing disputes shall be handled pursuant to the terms of this section. Provided, that nothing herein shall preclude either party from filing complaints, at any time, in accordance with the dispute resolution provisions included in the General Terms and Conditions to the Agreement.
- 3.2 Each Party agrees to notify the other Party upon the discovery of a billing dispute. In the event of a billing dispute, the Parties will endeavor to resolve the dispute within sixty (60) calendar days of the Bill Date on which such disputed charges appear. Resolution of the dispute is expected to occur at the first level of management resulting in a recommendation for settlement of the dispute and closure of a specific billing period. If the issues are not resolved within the allotted time frame, the following resolution procedure will begin.
- 3.3 If the dispute is not resolved within sixty (60) days of the Bill Date, the dispute will be escalated to the second level of management for each of the respective Parties for resolution. If the dispute is not resolved within ninety (90) days of the Bill Date, the dispute will be escalated to the third level of management for each of the respective Parties for resolution.
 - If the dispute is not resolved within one hundred and twenty (120) days of the Bill Date, the dispute will be escalated to the fourth level of management for each of the respective Parties for resolution.
- 3.4 If a Party disputes a charge and does not pay such charge by the payment due date, such charges shall be subject to late payment charges as set forth in the Late Payment Charges provision of this Attachment. If a Party disputes charges and the dispute is resolved in favor of such Party, the other Party shall credit the bill of the disputing Party for the amount of the disputed charges along with any late payment charges assessed no later than the second Bill Date after the resolution of the dispute. Accordingly, if a Party disputes charges and the dispute is resolved in favor of the other Party, the disputing Party shall pay the other Party the amount of the disputed charges and any associated late payment charges assessed no later than the second bill payment due date after the resolution of the dispute. In no event, however, shall any late payment charges be assessed on any previously assessed late payment charges.

4. RAO Hosting

- 4.1 RAO Hosting, Credit Card and Third Number Settlement System (CATS) and Non-Intercompany Settlement System (NICS) services provided to ITC^DeltaCom by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this Agreement, including such revisions as may be made from time to time by BellSouth.
- 4.2 ITC^DeltaCom shall furnish all relevant information required by BellSouth for the provision of RAO Hosting, CATS and NICS.
- 4.3 Applicable compensation amounts will be billed by BellSouth to ITC^DeltaCom on a monthly basis in arrears. Amounts due from one Party to the other (excluding adjustments) are payable within thirty (30) days of receipt of the billing statement.
- 4.4 ITC^DeltaCom must have its own unique RAO code. Requests for establishment of RAO status where BellSouth is the selected CMDS interfacing host, require written notification from ITC^DeltaCom to the BellSouth RAO Hosting coordinator at least eight (8) weeks prior to the proposed effective date. The proposed effective date will be mutually agreed upon between the Parties with consideration given to time necessary for the completion of required BellCore functions. BellSouth will request the assignment of an RAO code from its connecting contractor on behalf of ITC^DeltaCom and will coordinate all associated conversion activities.
- 4.5 BellSouth will receive messages from ITC^DeltaCom that are to be processed by BellSouth, another LEC or CLEC in the BellSouth region or a LEC outside the BellSouth region.
- 4.6 BellSouth will perform invoice sequence checking, standard EMI format editing, and balancing of message data with the EMI trailer record counts on all data received from ITC^DeltaCom.
- 4.7 All data received from ITC^DeltaCom that is to be processed or billed by another LEC or CLEC within the BellSouth region will be distributed to that LEC or CLEC in accordance with the agreement(s) which may be in effect between BellSouth and the involved LEC or CLEC.
- 4.8 All data received from ITC^DeltaCom that is to be placed on the CMDS network for distribution outside the BellSouth region will be handled in accordance with the agreement(s) which may be in effect between BellSouth and its connecting contractor.

- 4.9 BellSouth will receive messages from the CMDS network that are destined to be processed by ITC^DeltaCom and will forward them to ITC^DeltaCom on a daily basis.
- 4.10 Transmission of message data between BellSouth and ITC^ DeltaCom will be via CONNECT_Direct.
- 4.11 All messages and related data exchanged between BellSouth and ITC^DeltaCom will be formatted in accordance with accepted industry standards for EMI formatted records and packed between appropriate EMI header and trailer records, also in accordance with accepted industry standards.
- 4.12 ITC^DeltaCom will ensure that the recorded message detail necessary to recreate files provided to BellSouth will be maintained for back-up purposes for a period of three (3) calendar months beyond the related message dates.
- 4.13 Should it become necessary for ITC^DeltaCom to send data to BellSouth more than sixty (60) days past the message date(s), ITC^DeltaCom will notify BellSouth in advance of the transmission of the data. If there will be impacts outside the BellSouth region, BellSouth will work with its connecting contractor and ITC^DeltaCom to notify all affected Parties.
- In the event that data to be exchanged between the two Parties should become lost or destroyed, both Parties shall work together to determine the source of the problem. Once the cause of the problem has been jointly determined and the responsible Party (BellSouth or ITC^DeltaCom) identified and agreed to, the company responsible for creating the data (BellSouth or ITC^DeltaCom) shall make every effort to have the affected data restored and retransmitted. If the data cannot be retrieved, the responsible Party will be liable to the other Party for any resulting lost revenue. Lost revenue may be a combination of revenues that could not be billed to the end users and associated access revenues. Both Parties will work together to estimate the revenue amount based upon a reasonable estimate of three to twelve months of prior usage. The resulting estimated revenue loss will be paid by the responsible Party to the other Party within three (3) calendar months of the date of problem resolution, or as mutually agreed upon by the Parties. If access usage data is not processed and delivered by either Party in a timely manner such that the other Party is unable to bill the IXC, the responsible Party shall be liable for the amount of lost revenue. The Parties agree that the term "timely manner" as used herein shall be defined in accordance with OBF guidelines. Until such time as OBF addresses this issue, the term "timely manner" shall be reasonably determined on a case by case basis.

- 4.15 Should an error be detected by the EMI format edits performed by BellSouth on data received from ITC^DeltaCom, the entire pack containing the affected data will not be processed by BellSouth. BellSouth will notify ITC^DeltaCom of the error condition. ITC^DeltaCom will correct the error(s) and will resend the entire pack to BellSouth for processing. In the event that an out-of-sequence condition occurs on subsequent packs, ITC^DeltaCom will resend these packs to BellSouth after the pack containing the error has been successfully reprocessed by BellSouth. Both Parties agree to provide the other Party notification of any discovered errors within 7 business days of the discovery.
- 4.16 In association with message distribution service, BellSouth will provide ITC^DeltaCom with associated intercompany settlements reports (CATS and NICS) as appropriate.
- 4.17 Other than as specified in Section 4.14 and 4.15 above, in no case shall either Party be liable to the other for any direct or consequential damages incurred as a result of the obligations set out in this agreement.

4.18 RAO Compensation

- 4.18.1 Rates for message distribution service provided by BellSouth for ITC^DeltaCom are as set forth in Attachment 11 of this Agreement.
- 4.18.2 Rates for data transmission associated with message distribution service are as set forth in Attachment 11 of this Agreement.
- 4.18.3 Data circuits (private line or dial-up) will be required between BellSouth and ITC^DeltaCom for the purpose of data transmission. Where a dedicated line is required, ITC^DeltaCom will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. ITC^DeltaCom will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to ITC^DeltaCom. Additionally, all message toll charges associated with the use of the dial-up circuit by ITC^DeltaCom will be the responsibility of ITC^DeltaCom. Associated equipment on the BellSouth end, including a

modem, will be negotiated on a case by case basis between the Parties.

4.18.4 All equipment, including modems and software, that is required on the ITC^DeltaCom end for the purpose of data transmission will be the responsibility of ITC^DeltaCom.

4.19 <u>Intercompany Settlements Messages</u>

- 4.19.1 This Section addresses the settlement of revenues associated with traffic originated from or billed by ITC^DeltaCom as a facilities based provider of local exchange telecommunications services outside the BellSouth region. Only traffic that originates in one company's operating territory and bills in another company's operating territory is included. Traffic that originates and bills within the same company's operating territory will be settled on a local basis between ITC^DeltaCom and the involved company(ies), unless that company is participating in NICS.
- 4.19.2 Both traffic that originates outside the BellSouth region by ITC^DeltaCom and is billed within the BellSouth region, and traffic that originates within the BellSouth region and is billed outside the BellSouth region by ITC^DeltaCom, is covered by this Agreement (CATS). Also covered is traffic that either is originated by or billed by ITC^DeltaCom, involves a company other than ITC^DeltaCom, qualifies for inclusion in the CATS settlement, and is not originated or billed within the BellSouth region (NICS).
- 4.19.3 Revenues associated with calls originated and billed within the BellSouth region will be settled via BellCore's, its successor or assign, NICS system.
- 4.19.4 BellSouth shall receive the monthly NICS reports from BellCore, its successor or assign, on behalf of ITC^DeltaCom. BellSouth will distribute copies of these reports to ITC^DeltaCom on a monthly basis.
- 4.19.5 BellSouth shall receive the monthly Credit Card and Third Number Settlement System (CATS) reports from BellCore, its successor or assign, on behalf of ITC^DeltaCom. BellSouth will distribute copies of these reports to ITC^DeltaCom on a monthly basis.
- 4.19.6 BellSouth shall collect the revenue earned by ITC^DeltaCom from the operating company in whose territory the messages are billed (CATS), less a per message billing and collection fee of five cents

(\$0.05), on behalf of ITC^DeltaCom. BellSouth will remit the revenue billed by ITC^DeltaCom to the operating company in whose territory the messages originated, less a per message billing and collection fee of five cents (\$0.05), on behalf of ITC^DeltaCom. These two amounts will be netted together by BellSouth and the resulting charge or credit issued to ITC^DeltaCom via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

4.19.7 BellSouth shall collect the revenue earned by ITC^DeltaCom within the BellSouth territory from another CLEC also within the BellSouth territory (NICS) where the messages are billed, less a per message billing and collection fee of five cents (\$0.05), on behalf of ITC^DeltaCom. BellSouth will remit the revenue billed by ITC^DeltaCom within the BellSouth region to the CLEC also within the BellSouth region, where the messages originated, less a per message billing and collection fee of five cents (\$0.05). These two amounts will be netted together by BellSouth and the resulting charge or credit issued to ITC^DeltaCom via a monthly Carrier Access Billing System (CABS) miscellaneous bill.

BellSouth and ITC^DeltaCom agree that monthly netted amounts of less than fifty dollars (\$50.00) shall not be settled.

5. Optional Daily Usage File

- 5.1 Upon written request from ITC^DeltaCom, BellSouth shall provide the Optional Daily Usage File (ODUF) service to ITC^DeltaCom pursuant to the terms and conditions set forth in this section.
- 5.2 ITC^DeltaCom shall furnish all relevant information required by BellSouth for the provision of the Optional Daily Usage File.
- 5.3 The Optional Daily Usage Feed will contain billable messages that were carried over the BellSouth Network and processed in the BellSouth Billing System, but billed to an ITC^DeltaCom customer.
 - Charges for delivery of the Optional Daily Usage File will appear on ITC^DeltaCom's monthly bills. The charges are as set forth in Attachment 11 of this Agreement.
- 5.4 The Optional Daily Usage Feed will contain both rated and unrated messages. All messages will be in the standard Alliance for Telecommunications Industry Solutions (ATIS) EMI record format.

- 5.5 Messages that error in the billing system of ITC^DeltaCom will be the responsibility of ITC^DeltaCom. If, however, ITC^DeltaCom should encounter significant volumes of errored messages that prevent processing by ITC^DeltaCom within its systems, BellSouth will work with ITC^DeltaCom to determine the source of the errors and the appropriate resolution.
- 5.6 The following specifications shall apply to the Optional Daily Usage Feed.

5.6.1 USAGE TO BE TRANSMITTED

- 5.6.1.1 The following messages recorded by BellSouth will be transmitted to ITC^DeltaCom:
- message recording for per use/per activation type services (examples: Three Way Calling, Verify, Interrupt, Call Return, ETC.)
- measured billable Local
- Directory Assistance messages
- intraLATA Toll
- WATS & 800 Service
- -N11
- information service provider messages
- OPS services messages
- OPS messages attempted calls (UNE only)
- Credit /cancel records
- Usage for Voice Mail
 - 5.6.1.2 Rated Incollects (originated in BellSouth and from other companies) can also be on Optional Daily Usage File. Rated Incollects will be intermingled with BellSouth recorded rated and unrated usage. Rated Incollects will not be packed separately.
 - 5.6.1.3 BellSouth will perform duplicate record checks on records processed to Optional Daily Usage File. Any duplicate

- messages detected will be deleted and not sent to ITC^DeltaCom.
- 5.6.1.4 In the event that ITC^DeltaCom detects a duplicate on Optional Daily Usage File they receive from BellSouth, ITC^DeltaCom will drop the duplicate message (ITC^DeltaCom will not return the duplicate to BellSouth).

5.6.2 PHYSICAL FILE CHARACTERISTICS

- 5.6.2.1 The Optional Daily Usage File will be distributed to ITC^DeltaCom via an agreed medium CONNECT: Direct being the preferred transport method. The Daily Usage Feed will be a variable block format (2476) with an LRECL of 2472. The data on the Daily Usage Feed will be in a non-compacted EMI format (175 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be a maximum of one dataset per workday per OCN.
- 5.6.2.2 Data circuits (private line or dial-up) may be required between BellSouth and ITC^DeltaCom for the purpose of data transmission. Where a dedicated line is required, ITC^DeltaCom will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. ITC^DeltaCom will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to ITC^DeltaCom. Additionally, all message toll charges associated with the use of the dial circuit by ITC^DeltaCom will be the responsibility of ITC^DeltaCom. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including and software, that is required ITC^DeltaCom end for the purpose of data transmission will be the responsibility of ITC^DeltaCom.

5.6.3 PACKING SPECIFICATIONS

- 5.6.3.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 5.6.3.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to ITC^DeltaCom which BellSouth RAO that is sending the message. BellSouth and ITC^DeltaCom will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by ITC^DeltaCom and resend the data as appropriate.

The data will be packed using ATIS EMI records.

5.6.4 PACK REJECTION

5.6.4.1 ITC^DeltaCom shall notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). Standard ATIS EMI Error Codes will be used. ITC^DeltaCom will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and retransmitted to ITC^DeltaCom by BellSouth.

5.6.5 CONTROL DATA

ITC^DeltaCom will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate ITC^DeltaCom received the pack and the acceptance or rejection of the pack. Pack Status Code(s) will be populated using standard ATIS EMI error codes for packs that were rejected by ITC^DeltaCom for reasons stated in the above section.

5.6.6 TESTING

5.6.6.1 Upon request from ITC^DeltaCom BellSouth shall send test files to ITC^DeltaCom for the Optional Daily Usage File. The parties agree to review and discuss the file's content and/or format. For testing of usage results, BellSouth shall request that ITC^DeltaCom set up a production (LIVE) file. The live test may consist of ITC^DeltaCom's employees making test calls for the types of services ITC^DeltaCom requests on the Optional Daily Usage File. These test calls are logged by ITC^DeltaCom, and the logs are provided to BellSouth. These logs will be used to verify the files. Testing will be completed within 30 calendar days from the date on which the initial test file was sent.

6. Access Daily Usage File

6.1. SCOPE OF AGREEMENT

6.1.1 This agreement shall apply to the service of the Access Daily Usage File (ADUF) as provided by BellSouth to ITC^DeltaCom upon written request from ITC^DeltaCom. The specifications, terms and conditions for the provisions of this service are outlined in Exhibit A of this Agreement.

6.2. DEFINITIONS

- 6.2.1 <u>Compensation</u> is the amount of money due from ITC^DeltaCom to BellSouth for services provided under this Agreement.
- 6.2.2 Access Daily Usage File (ADUF) is the compilation of interstate access messages associated with an unbundled port in standard Exchange Message Interface (EMI) format exchanged from BellSouth to ITC^DeltaCom.
- 6.2.3 Exchange Message Interface is the nationally administered standard format for the exchange of data within the telecommunications industry.
- 6.2.4 <u>Message Distribution</u> is routing determination and subsequent delivery of message data from one company to another.

6.3. RESPONSIBILITIES OF THE PARTIES

- 6.3. 1 ADUF service provided to ITC^DeltaCom by BellSouth will be in accordance with the methods and practices regularly adopted and applied by BellSouth to its own operations during the term of this agreement, including such revisions as may be made from time to time by BellSouth.
- 6.3. 2 ITC^DeltaCom shall furnish in a timely manner all relevant information required by BellSouth for the provision of the ADUF.

6.4. COMPENSATION ARRANGEMENTS

6.4.1 Applicable compensation amounts will be billed by BellSouth to ITC^DeltaCom on a monthly basis in arrears. Amounts due from ITC^DeltaCom to BellSouth (excluding adjustments) are payable within 30 days of the date of the billing statement. Rates for ADUF are as set forth in Exhibit A.

6.5. ASSOCIATED EXHIBIT

6.5. 1 Listed below is the exhibit associated with this Agreement, incorporated herein by this reference.

Exhibit A Access Daily Usage File (ADUF)

6.5. 2 From time to time by written agreement of the parties, new exhibits may be substituted for the attached Exhibit A, superseding and canceling the Exhibit(s) then in effect.

Exhibit A Access Daily Usage File

1. SCOPE OF EXHIBIT

1.1 Upon request from ITC^DeltaCom, BellSouth will provide the Access Daily Usage File service to ITC^DeltaCom pursuant to the rates, terms and conditions set forth in this exhibit.

2. <u>GENERAL INFORMATION</u>

- 2.1 ITC^DeltaCom shall furnish the following information required by BellSouth for the provision of the Access Daily Usage File: (1) ITC^DeltaCom's preference for either transmitted files or magnetic tapes, (2) ITC^DeltaCom's transmission sites, (3) the data set names to be used, (4) the software to be executed on receipt of the files, and (5) other LAN-to-LAN connectivity information.
- 2.2 The Access Daily Usage File will contain access records associated with an unbundled port that ITC^DeltaCom has purchased from BellSouth. Charges for the Access Daily Usage File will be as follows:

\$0.004 per message - Message Distribution \$0.001 per message - Data Transmission (CONNECT:Direct) \$54.95 per magnetic tape

Charges for delivery of the Access Daily Usage Feed will appear on ITC^DeltaCom's monthly bills.

- 2.3 All messages provided with the Access Daily Usage File will be in the standard Bellcore EMI record format.
- 2.4 Messages that error in the billing system of ITC^DeltaCom will be the responsibility of ITC^DeltaCom. If, however, ITC^DeltaCom should encounter significant volumes of errored messages that prevent processing by ITC^DeltaCom within its systems, BellSouth will work with ITC^DeltaCom to determine the source of the errors and the appropriate resolution.

3. USAGE TO BE TRANSMITTED

- 3.1 The following messages recorded by BellSouth will be transmitted to ITC^DeltaCom:
 - Interstate access records associated with an unbundled port

- Intrastate access records associated with an unbundled port
- Undetermined jurisdiction access records associated with an unbundled port
- 3.2 When ITC^DeltaCom purchases Network Element ports from BellSouth and calls are made using these ports, BellSouth will handle the calls as follows:

Originating from Network Element and carried by Interexchange Carrier:

BellSouth will bill network element to CLEC and send access record to the CLEC via ADUF

Originating from network element and carried by BellSouth (ITC^DeltaCom is BellSouth's toll customer):

BellSouth will bill resale toll rates to ITC^DeltaCom and send toll record for the end user toll billing purposes via ODUF (Optional Daily Usage File). Access record will be sent to ITC^DeltaCom via ADUF.

Terminating on network element and carried by Interexchange Carrier:

BellSouth will bill network element to ITC^DeltaCom and send access record to ITC^DeltaCom.

Terminating on network element and carried by BellSouth:

BellSouth will bill network element to ITC^DeltaCom and send access record to ITC^DeltaCom.

- 3.3 BellSouth will perform duplicate record checks on records processed to the Access Daily Usage File. Any duplicate messages detected will be dropped and not sent to ITC^DeltaCom.
- 3.4 In the event that ITC^DeltaCom detects a duplicate on the Access Daily Usage File they receive from BellSouth, ITC^DeltaCom will drop the duplicate message (ITC^DeltaCom shall not return the duplicate to BellSouth).

4. FILE CHARACTERISTICS AND TRANSMISSION METHOD

- 4.1 The Access Daily Usage Feed will be distributed to ITC^DeltaCom via an agreed upon medium with CONNECT:Direct being the preferred transport method. The Access Daily Usage Feed will be a fixed block format (2476) with an LRECL of 2472. The data on the Access Daily Usage Feed will be in a non-compacted EMI format (210 byte format plus modules). It will be created on a daily basis (Monday through Friday except holidays). Details such as dataset name and delivery schedule will be addressed during negotiations of the distribution medium. There will be one dataset per workday per OCN.
- 4.2 Data circuits (private line or dial-up) may be required between BellSouth and ITC^DeltaCom for the purpose of data transmission. Where a dedicated line is required, ITC^DeltaCom will be responsible for ordering the circuit, overseeing its installation and coordinating the installation with BellSouth. ITC^DeltaCom will also be responsible for any charges associated with this line. Equipment required on the BellSouth end to attach the line to the mainframe computer and to transmit successfully ongoing will be negotiated on a case by case basis. Where a dial-up facility is required, dial circuits will be installed in the BellSouth data center by BellSouth and the associated charges assessed to ITC^DeltaCom. Additionally, all message toll charges associated with the use of the dial circuit by ITC^DeltaCom will be the responsibility of ITC^DeltaCom. Associated equipment on the BellSouth end, including a modem, will be negotiated on a case by case basis between the parties. All equipment, including modems and software, that is required on ITC^DeltaCom's end for the purpose of data transmission will be the responsibility of ITC^DeltaCom.

5. PACKING SPECIFICATIONS

- 5.1 A pack will contain a minimum of one message record or a maximum of 99,999 message records plus a pack header record and a pack trailer record. One transmission can contain a maximum of 99 packs and a minimum of one pack.
- 5.2 The OCN, From RAO, and Invoice Number will control the invoice sequencing. The From RAO will be used to identify to ITC^DeltaCom which BellSouth RAO that is sending the message. BellSouth and ITC^DeltaCom will use the invoice sequencing to control data exchange. BellSouth will be notified of sequence failures identified by ITC^DeltaCom and resend the data as appropriate.

The data will be packed using EMI records.

6. PACK REJECTION

6.1 ITC^DeltaCom will notify BellSouth within one business day of rejected packs (via the mutually agreed medium). Packs could be rejected because of pack sequencing discrepancies or a critical edit failure on the Pack Header or Pack Trailer records (i.e. out-of-balance condition on grand totals, invalid data populated). ITC^DeltaCom will not be required to return the actual rejected data to BellSouth. Rejected packs will be corrected and/or retransmitted to ITC^DeltaCom by BellSouth.

7. CONTROL DATA

7.1 ITC^DeltaCom will send one confirmation record per pack that is received from BellSouth. This confirmation record will indicate ITC^DeltaCom received the pack and the acceptance or rejection of the pack. Pack status Code(s) will be populated using standard Bellcore EMI codes for packs that were rejected by ITC^DeltaCom for reasons stated in the above section.

8. <u>TESTING</u>

8.1 Upon request from ITC^DeltaCom, BellSouth shall send test file(s) to ITC^DeltaCom for the Access Daily Usage File. The parties agree to review and discuss the file's content or format.

Attachment 8

Rights-of-Way, Conduits and Pole Attachments

BellSouth agrees to provide ITC^DeltaCom, pursuant to 47 U.S.C. § 224, as amended by the Act, nondiscriminatory access to any pole, duct, conduit, or right-of-way owned or controlled by BellSouth pursuant to terms and conditions that are subsequently negotiated with BellSouth's Competitive Structure Provisioning Center.

BONA FIDE REQUEST/NEW BUSINESS REQUEST PROCESS

- 1.0 The Parties agree that ITC^DeltaCom is entitled to order any network element, interconnection option, service option or other Resale service required to be made available by the Telecommunications Act of 1996, FCC requirements or State Commission requirements. Procedures applicable to requesting the addition of such facilities or service options are specified in this Attachment 9 and referenced in Section 5 of the General Terms and Conditions.
- 2.0 Bona Fide Requests ("BFR") are to be used when ITC^DeltaCom makes a request of BellSouth to provide a new or modified network element, interconnection option, or other service option pursuant to the Act; or to provide a new or custom capability or function to meet ITC^DeltaCom's business needs; that were not previously included in the Agreement. The BFR process is intended to facilitate the two way exchange of information between the requesting Party and BellSouth, necessary for accurate processing of requests in a consistent and timely fashion.
- 3.0 A Bona Fide Request shall be submitted in writing by ITC^DeltaCom and shall specifically identify the required service date, technical requirements, space requirements and/or such specifications that clearly define the request such that BellSouth has sufficient information to analyze and prepare a response. Such a request also shall include a ITC^DeltaCom's designation of the request as being (i) pursuant to the Telecommunications Act of 1996 or (ii) pursuant to the needs of the business. The request shall be sent to ITC^DeltaCom's Account Executive.
- 4.0 ITC^DeltaCom may cancel a BFR at any time. If ITC^DeltaCom cancels the request more than thirty (30) business days after submitting it, ITC^DeltaCom shall pay BellSouth's reasonable and demonstrable costs of processing and/or implementing the BFR up to the date of cancellation. If ITC^DeltaCom does not cancel a BFR, ITC^DeltaCom shall pay BellSouth's reasonable and demonstrable costs of processing and implementing the request.
- 5.0 Within twenty (20) Business Days of its receipt of a BFR from ITC^DeltaCom, BellSouth shall respond via email or overnight delivery to ITC^DeltaCom by providing a preliminary analysis of such Interconnection, Network Element, or requested level of quality thereof that is the subject of the BFR. The preliminary analysis shall confirm that BellSouth will either offer access to the Interconnection, Network Element, or requested level of quality or provide an explanation of why it is not

technically feasible and/or why the request does not qualify as an Interconnection, Network Element, or requested level of quality that is required to be provided under the Act. If BellSouth determines that the Interconnection, Network Element, or requested level of quality that is the subject of the BFR is technically feasible, BellSouth shall provide to ITC^DeltaCom a high level cost estimate and detailed development price quote.

- 6.0 Within twenty (20) business days after receipt and acceptance of preliminary business analysis, BellSouth shall propose a firm price and a detailed implementation plan via email or overnight delivery.
- 7.0 Within twenty (20) business days after its receipt of the BFR price quote and implementation plan from BellSouth, ITC^DeltaCom must either confirm its order for such Interconnection or Network Element or, if it believes such quote is not consistent with the requirements of the Communications Act, may seek Commission arbitration of its request or file a complaint with the FCC, as appropriate. Any such arbitration applicable to Network Elements and/or Interconnection shall be conducted in accordance with standards prescribed in Section 252 of the Act.
- 8.0 All prices shall be consistent with the pricing principles of the Act, the FCC and the Supreme Court's ruling of January 25, 1999 upholding the FCC pricing rules.
- 9.0 If either Party to a BFR believes that the other Party is not requesting, negotiating or processing the Bona Fide Request in good faith, or disputes a determination, or price or cost quote, such Party may seek state Commission or FCC resolution of the dispute as appropriate.
- 10.0 Upon agreement to the terms of a BFR, an amendment to the Agreement may be required. Pursuant to Sections 5.3 and 16 of the General Terms and Conditions, ITC^DeltaCom may adopt through another Company's Agreement any BFR provided to any other carrier.

Attachment 10 Performance Measurements

BellSouth Service Quality Measurement Plan (SQM)

Tennessee Performance Metrics

Measurement Descriptions Version 0.01

Issue Date: March 12, 2001

Introduction

The BellSouth Service Quality Measurement Plan (SQM) describes in detail the measurements produced to evaluate the quality of service delivered to BellSouth's customers both wholesale and retail. The SQM was developed to respond to the requirements of the Communications Act of 1996 Section 251 (96 Act) which required BellSouth to provide non-discriminatory access to Competitive Local Exchange Carriers (CLEC)¹ and its Retail Customers. The reports produced by the SQM provide regulators, CLECs and BellSouth the information necessary to monitor the delivery of non-discriminatory access.

This plan results from the many divergent forces evolving from the 96 Act. The 96 Act, the Georgia Public Service Commission (GPSC) Order (orders of 12/30/97 and 1/12/01 in Docket 7892-U), LCUG 1-7.0, the FCC's NPRM (CC Docket 98-56 RM9101 04/17/98), the Louisiana Public Service Commission (LPSC) Order (Docket U-22252 Subdocket C 04/19/98), numerous arbitration cases, LPSC sponsored collaborative workshops (10/98-02/00), and proceedings in Alabama, Mississippi, and North Carolina have and continue to influence the SQM.

The SQM and the reports flowing from it must change to reflect the dynamic requirements of the industry. New measurements are added as new products, systems, and processes are developed and fielded. New products and services are added as the markets for them develop and the processes stabilize. The measurements are also changed to reflect changes in systems, correct errors, and respond to both 3rd Party audit requirements.

This document is intended for use by someone with knowledge of telecommunications industry, information technologies and a functional knowledge of the subject areas covered by the BellSouth Performance Measurements and the reports that flow from them.

Once it is approved, the most current copy of this document can be found on the web at URL: https://pmap.bellsouth.com in the Help folder.

^{1.} Alternative Local Exchange Companies (ALEC) and Competitive Local Providers (CLP) are referred to as Competitive Local Exchange Carriers (CLEC) in this document.



Contents

	Operations Support Systems (OSS)	
OSS-1:	Average Response Time and Response Interval (Pre-Ordering/Ordering)	1-
OSS-2:	Interface Availability (Pre-Ordering/Ordering)	1-
OSS-3:	Interface Availability (Maintenance & Repair)	1-
OSS-4:	Response Interval (Maintenance & Repair)	- 1-1
PO-1:	Loop Makeup - Response Time - Manual	- 1-1
PO-2:	Loop Make Up - Response Time - Electronic	- 1-1:
Section 2:	Ordering	
O-1:	Acknowledgement Message Timeliness	2-
O-2:	Acknowledgement Message Completeness	2-
O-3:	Percent Flow-Through Service Requests (Summary)	2-
O-4:	Percent Flow-Through Service Requests (Detail)	2-
O-5:	Flow-Through Error Analysis	
O-6:	CLEC LSR Information	
O-7:	Percent Rejected Service Requests	- 2-19
O-8:	Reject Interval	- 2-2
O-9:	Firm Order Confirmation Timeliness	
O-10:	Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual	- 2-2
O-11:	Firm Order Confirmation and Reject Response Completeness	- 2-2
O-12:	Speed of Answer in Ordering Center	- 2-3
O-13:	LNP-Percent Rejected Service Requests	- 2-3
O-14:	LNP-Reject Interval Distribution & Average Reject Interval	- 2-3
O-15:	LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval	- 2-3
Section 3:	Provisioning	
P-1:	Mean Held Order Interval & Distribution Intervals	3-
P-2:	Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices	3-
P-3:	Percent Missed Installation Appointments	3-
P-4:	Average Completion Interval (OCI) & Order Completion Interval Distribution	3-
P-5:	Average Completion Notice Interval	- 3-1
P-6:	Coordinated Customer Conversions Interval	
P-6A:	Coordinated Customer Conversions – Hot Cut Timeliness % Within Interval and Average Interval	- 3-1
P-6B:	Coordinated Customer Conversions – Average Recovery Time	
P-6C:	Coordinated Customer Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order	- 3-2
P-7:	Cooperative Acceptance Testing - % of xDSL Loops Tested	
P-8:	% Provisioning Troubles within 30 days of Service Order Completion	- 3-2
P-9:	Total Service Order Cycle Time (TSOCT)	- 3-2
P-10:	LNP-Percent Missed Installation Appointments	
P-11:	LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution	- 3-3
P-12:	LNP-Total Service Order Cycle Time (TSOCT)	- 3-3
Section 4:	Maintenance & Repair	
M&R-1:	Missed Repair Appointments	4-
M&R-2:	Customer Trouble Report Rate	4-



M&R-3:	Maintenance Average Duration	4-:
	Percent Repeat Troubles within 30 Days	
	Out of Service (OOS) > 24 Hours Average Answer Time – Repair Centers	
M&R-0:	Mean Time To Notify CLEC of Network Outages	4-1
MXX-/.	Mean Time to Notify Clec of Network Ottages	4-12
Section 5:	Billing	
B-1:	Invoice Accuracy	5-
B2:	Mean Time to Deliver Invoices	
B3:	Usage Data Delivery Accuracy	5-5
B4:	Usage Data Delivery Completeness	
B5:	Usage Data Delivery Timeliness	
B6:	Mean Time to Deliver Usage	5-1
B7:	Recurring Charge Completeness	5-13
B8:	Non-Recurring Charge Completeness	5-14
Section 6:	Operator Services And Directory Assistance	
OS-1:	Speed to Answer Performance/Average Speed to Answer - Toll	6-:
OS-2:	Speed to Answer Performance/Percent Answered with "X" Seconds – Toll	6-3
DA-1:	Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA)	
DA-2:	Speed to Answer Performance/Percent Answered within "X" Seconds – Directory Assistance (DA)	6-:
Section 7:	Database Update Information	
D-1:	Average Database Update Interval	7
D-1. D-2:	Percent Database Update Accuracy	
D-2: D-3:	Percent NXXs and LRNs Loaded by the LERG Effective Date	7-
D-3.	Telective Page 3 and ERNS Educed by the EERS Effective Page 4	
Section 8:		
E-1:	Timeliness	
E-2:	Accuracy	
E-3:	Mean Interval	8-4
Section 9:	Trunk Group Performance	
TGP-1:	Trunk Group Performance-Aggregate	9
TGP-2:	Trunk Group Performance-CLEC Specific	9-j
Castian 16). Collegation	
	Collocation Collocation Average Response Time	10
C-1: C-2:	Collocation Average Response Time	
C-2. C-3:	Collocation Percent of Due Dates Missed	
C 3.	Consequent of Due Dues Missed	10.
	: Change Management	
CM-1:	Timeliness of Change Management Notices	11-1
CM-2:	Change Management Notice Average Delay Days	11-3
CM-3:	Timeliness of Documents Associated with Change	11-4
CM-4:	Change Management Documentation Average Delay Days Notification of CLEC Interface Outages	11-5
CM-5:	Notification of CLEC Interface Outages	11-
Appendix	A: Reporting Scope	A -1
A-1:	Standard Service Groupings	A-1
A-2:	Standard Service Order Activities	A-1
A nn au -1:	D. Classam of Amonyms and Tames	n 1
Appenaix	B: Glossary of Acronyms and Terms	R-I



Contents

Appendix	C: BellSouth Audit Policy	- (Z -1
C-1:	BellSouth's Internal Audit Policy	-	C-1
C-2:	BellSouth's External Audit Policy	_	C-



Section 1: Operations Support Systems (OSS)

OSS-1: Average Response Time and Response Interval (Pre-Ordering/ Ordering)

Definition

Average response time and response intervals are the average times and number of requests responded to within certain intervals for accessing legacy data associated with appointment scheduling, service & feature availability, address verification, request for Telephone numbers (TNs), and Customer Service Records (CSRs).

Exclusions

None

Business Rules

The average response time for retrieving pre-order/order information from a given legacy system is determined by summing the response times for all requests submitted to the legacy systems during the reporting period and dividing by the total number of legacy system requests for that month.

The response interval starts when the client application (LENS or TAG for CLECs and RNS or ROS for BellSouth) submits a request to the legacy system and ends when the appropriate response is returned to the client application. The number of accesses to the legacy systems during the reporting period which take less than 2.3 seconds, the number of accesses which take more than 6 seconds, and the number which are less than or equal to 6.3 seconds are also captured.

Calculation

Response Time = (a - b)

- a = Date & Time of Legacy Response
- b = Date & Time of Legacy Request

Average Response Time = $c \div d$

- c = Sum of Response Times
- d = Number of Legacy Requests During the Reporting Period

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month Legacy Contract (per reporting dimension) Response Interval Regional Scope 	 Report month Legacy Contract (per reporting dimension) Response Interval Regional Scope

OSS-1: Average Response Time and Response Interval (Pre-Ordering/Ordering)

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation **SQM** Analog/Benchmark • RSAG – Address (Regional Street Address Guide-Address) – • Parity + 4 seconds. stores street address information used to validate customer addresses. CLECs and BellSouth query this legacy system. • RSAG – TN (Regional Street Address Guide-Telephone number) – contains information about facilities available and telephone numbers working at a given address. CLECs and BellSouth query this legacy system. ATLAS (Application for Telephone Number Load Administration and Selection) – acts as a warehouse for storing telephone numbers that are available for assignment by the system. It enables CLECs and BellSouth service reps to select and reserve telephone numbers. CLECs and BellSouth query this legacy system. • **COFFI** (Central Office Feature File Interface) – stores information about product and service offerings and availability. CLECs query this legacy system. • **DSAP** (DOE Support Application) – provides due date information. CLECs and BellSouth query this legacy system. • HAL/CRIS (Hands-Off Assignment Logic/Customer Record Information System) – a system used to access the Business Office Customer Record Information System (BOCRIS). It allows BellSouth servers, including LENS, access to legacy systems. CLECs query this legacy system. • P/SIMS (Product/Services Inventory Management system) – provides information on capacity, tariffs, inventory and service availability. CLECs query this legacy system. **OASIS** (Obtain Available Services Information Systems) – Information on feature and rate availability. BellSouth queries this legacy system.

Table 1: Legacy System Access Times For RNS

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. Sec.	# of Calls
RSAG	RSAG-TN	Address	x	X	Х	x	х
RSAG	RSAG-ADDR	Address	х	X	Х	x	х
ATLAS	ATLAS-TN	TN	х	X	Х	х	Х
DSAP	DSAP-DDI	Schedule	X	X	X	х	Х
CRIS	CRSACCTS	CSR	X	X	X	х	Х
OASIS	OASISBSN	Feature/Service	X	X	X	X	X
OASIS	OASISCAR	Feature/Service	X	X	X	х	X
OASIS	OASISLPC	Feature/Service	X	X	X	х	X
OASIS	OASISMTN	Feature/Service	X	X	X	X	X
OASIS	OASISBIG	Feature/Service	X	X	X	х	Х



Table 2: Legacy System Access Times For R0S

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	X	X	X
RSAG	RSAG-ADDR	Address	X	X	х	X	X
ATLAS	ATLAS-TN	TN	X	X	Х	X	X
DSAP	DSAP-DDI	Schedule	X	X	X	X	X
CRIS	CRSOCSR	CSR	X	X	X	X	X
OASIS	OASISBIG	Feature/Service	X	X	X	X	X

Table 3: Legacy System Access Times For LENS

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	X	X	X
RSAG	RSAG-ADDR	Address	X	X	X	X	X
ATLAS	ATLAS-TN	TN	X	X	X	X	X
DSAP	DSAP-DDI	Schedule	x	X	X	X	X
HAL	HAL/CRIS	CSR	x	X	X	X	X
COFFI	COFFI/USOC	Feature/Service	х	X	Х	х	X
P/SIMS	PSIMS/ORB	Feature/Service	X	X	X	X	X

Table 4: Legacy System Access Times For TAG

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	х	х	х
RSAG	RSAG-ADDR	Address	X	X	х	Х	Х
ATLAS	ATLAS-TN	TN	X	X	Х	Х	Х
ATLAS	ATLAS-MLH	TN	X	X	X	X	х
ATLAS	ATLAS-DID	TN	X	X	X	X	х
DSAP	DSAP-DDI	Schedule	X	X	Х	Х	Х
CRIS	CRSEINIT	CSR	X	X	х	X	х
CRIS	CRSECSR	CSR	X	X	Х	X	х

SEEM Measure

SEEM Measure					
Yes	Tier I				
	Tier II	X			

Note: CLEC specific data is not available in this measure. Queries of this sort do not have company specific signatures.

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
 RSAG – Address (Regional Street Address Guide-Address) – stores street address information used to validate customer addresses. CLECs and BellSouth query this legacy system. RSAG – TN (Regional Street Address Guide-Telephone number) – contains information about facilities available and telephone numbers working at a given address. CLECs and BellSouth query this legacy system. ATLAS (Application for Telephone Number Load Administration and Selection) – acts as a warehouse for storing telephone numbers that are available for assignment by the system. It enables CLECs and BellSouth service reps to select and reserve telephone numbers. CLECs and BellSouth query this legacy system. COFFI (Central Office Feature File Interface) – stores information about product and service offerings and availability. CLECs query this legacy system. DSAP (DOE Support Application) – provides due date information. CLECs and BellSouth query this legacy system. HAL/CRIS (Hands-Off Assignment Logic/Customer Record Information System) – a system used to access the Business Office Customer Record Information System (BOCRIS). It allows BellSouth servers, including LENS, access to legacy systems. CLECs query this legacy system. P/SIMS (Product/Services Inventory Management system) – provides information on capacity, tariffs, inventory and service availability. CLECs query this legacy system. OASIS (Obtain Available Services Information Systems) – Information on feature and rate availability. BellSouth queries this legacy system. 	Percent Response Received within 6.3 seconds: > 95%

SEEM OSS Legacy Systems

System	BellSouth	CLEC					
	Telephone Number/Address						
RSAG	RNS, ROS	TAG, LENS					
Atlas	RNS,ROS	TAG. LENS					
DSAP	RNS, ROS	TAG, LENS					
	CSR Data						
CRSACCTS	RNS						
CRSOCSR	ROS						
HAL/CRIS		LENS					
CRSE INIT		TAG					
CRSOCSR		TAG					
	Service/Feature Availability						
OASISBSN	RNS						
OASISCAR	RNS						
OASISLPC	RNS						



SystemBellSouthCLECOASISMTNRNSOASISBIGRNS, ROSCOFFI/USOCLENSPSIMS/ORBLENS



OSS-2: Interface Availability (Pre-Ordering)Ordering)

Definition

Percent of time OSS interface is functionally available compared to scheduled availability. Availability percentages for CLEC interface systems and for all Legacy systems accessed by them are captured. ("Functional Availability" is the amount of time in hours during the reporting period that the legacy systems are available to users. The planned System Scheduled Availability is the time in hours per day that the legacy system is scheduled to be available.)

Scheduled availability is posted on the ICS Operations internet site: (www.interconnection.bellsouth.com/oss/osshour.html)

Exclusions

None

Business Rules

This measurement captures the availability percentages for the BellSouth systems, which are used by CLECs during Pre-Ordering functions. Comparing the percentages to BellSouth results allows conclusions as to whether an equal opportunity exists for the CLEC to deliver a comparable customer experience.

Note: Only full outages are used in the calculation of Application Availability.

A full outage is incurred when any of the following circumstances exist:

- The application or system is down.
- The application or system is inaccessible, for any reason, by the customers who normally access the application or system.
- More than one work center cannot access the application or system for any reason.
- When only one work center accesses an application or system and 40% or more of the clients in that work center cannot access the application.
- When 40% of the functions the clients normally perform or 40% of the functionality that is normally provided by an application or system is unavailable.

Calculation

Interface Availability (Pre-Ordering/Ordering) = $(a \div b) \times 100$

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month Legacy Contract Type (per reporting dimension) Regional Scope	Report month
Hours of Downtime	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Regional Level	• ≥ 99.5%



OSS Interface Availability

OSS Interface	Applicable to	% Availability
EDI	CLEC	X
HAL	CLEC	X
LENS	CLEC	X
LEO Mainframe	CLEC	X
LEO UNIX	CLEC	X
LESOG	CLEC	X
PSIMS	CLEC	X
TAG	CLEC	X
ATLAS/COFFI	CLEC/BellSouth	X
BOCRIS	CLEC/BellSouth	X
DSAP	CLEC/BellSouth	X
RSAG	CLEC/BellSouth	X
SOCS	CLEC/BellSouth	X
SONGS	CLEC/BellSouth	X
RNS	BellSouth	Under Development
ROS	BellSouth	Under Development

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• ≥ 99.5%



SEEM OSS Interface Availability

OSS Interface	Applicable to	% Availability
EDI	CLEC	X
HAL	CLEC	X
LENS	CLEC	X
LEO Mainframe	CLEC	X
LEO UNIX	CLEC	X
LESOG	CLEC	X
PSIMS	CLEC	X
TAG	CLEC	X

OSS-3: Interface Availability (Maintenance & Repair)

OSS-3: Interface Availability (Maintenance & Repair)

Definition

This measures the percentage of time the OSS Interface is functionally available compared to scheduled availability. Availability percentage for the CLEC and BellSouth interface systems and for the legacy systems accessed by them are captured.

Exclusions

None

Business Rules

This measure is designed to compare the OSS availability versus scheduled availability of BellSouth's legacy systems.

Note: Only full outages are used in the calculation of Application Availability. A full outage is incurred when any of the following circumstances exists:

- The application or system is down.
- The application or system is inaccessible, for any reason, by the customers who normally access the application or system.
- More than one work center cannot access the application or system for any reason.
- When only one work center accesses an application or system and 40% or more of the clients in that work center cannot access the application.
- When 40% of the functions the clients normally perform or 40% of the functionality that is normally provided by an application or system is unavailable.

Calculation

OSS Interface Availability (a ÷ b) X 100

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Availability of CLEC TAFI Availability of LMOS HOST, MARCH, SOCS, CRIS, PREDICTOR, LNP and OSPCM ECTA 	Availability of BellSouth TAFI Availability of LMOS HOST, MARCH, SOCS, CRIS, PREDICTOR, LNP and OSPCM

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Regional Level	• ≥ 99.5%



OSS Interface Availability (M&R)

OSS Interface	% Availability
BellSouth TAFI	x
CLEC TAFI	x
CLEC ECTA	x
BellSouth & CLEC	X
CRIS	x
LMOS HOST	x
LNP	x
MARCH	x
OSPCM	x
PREDICTOR	x
SOCS	x

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• ≥ 99.5%

OSS Interface Availability (M&R)

OSS Interface	% Availability
CLEC TAFI	х
CLEC ECTA	x

OSS-4: Response Interval (Maintenance & Repair)

OSS-4: Response Interval (Maintenance & Repair)

Definition

The response intervals are determined by subtracting the time a request is received on the BellSouth side of the interface from the time the response is received from the legacy system. Percentages of requests falling into each interval category are reported, along with the actual number of requests falling into those categories.

Exclusions

None

Business Rules

This measure is designed to monitor the time required for the CLEC and BellSouth interface system to obtain from BellSouth's legacy systems the information required to handle maintenance and repair functions. The clock starts on the date and time when the request is received on the BellSouth side of the interface and the clock stops when the response has been transmitted through that same point to the requester.

Note: The OSS Response Interval BellSouth Total Report is a combination of BellSouth Residence and Business Total.

Calculation

OSS Response Interval = (a - b)

- a = Query Response Date and Time
- b = Query Request Date and Time

Percent Response Interval (per category) = $(c \div d) \times 100$

- c = Number of Response Intervals in category "X"
- d = Number of Queries Submitted in the Reporting Period

where, "X" is
$$\leq 4$$
, $> 4 \leq 10$, ≤ 10 , > 10 , or > 30 seconds.

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
CLEC Transaction Intervals	BellSouth Business and Residential Transactions Intervals

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:	
Regional Level	• Parity	



Legacy System Access Times for M&R

BellSouth &		Count				
System	System CLEC	<u><</u> 4	> 4 <u><</u> 10	<u><</u> 10	> 10	> 30
CRIS	x	X	X	X	X	X
DLETH	х	X	X	X	X	X
DLR	X	X	X	X	х	X
LMOS	X	X	X	X	х	X
LMOSupd	X	X	X	X	х	X
LNP	X	X	X	X	х	X
MARCH	X	X	X	X	х	x
OSPCM	X	X	X	X	х	x
Predictor	X	X	X	X	х	x
SOCS	x	X	X	X	X	X
NIW	X	X	X	X	х	x

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



PO-1: Loop Makeup - Response Time - Manual

Definition

This report measures the average interval and percent within the interval from the submission of a Manual Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- · Inquiries, which are submitted electronically.
- Designated Holidays are excluded from the interval calculation.
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation.
- · Canceled Inquiries.

Business Rules

The CLEC Manual Loop Makeup Service Inquiry (LMUSI) process includes inquiries submitted via mail or FAX to BellSouth's Complex Resale Support Group (CRSG).

This measurement combines three intervals:

- From receipt of the Service Inquiry for Loop Makeup to hand off to the Service Advocacy Center (SAC) for "Look-up."
- From SAC start date to SAC complete date.
- From SAC complete date to date the Complex Resale Support Group (CRSG) distributes loop makeup information back to the CLEC.

The "Receive Date" is defined as the date the Manual LMUSI is received by the CRSG. It is counted as day Zero. LMU "Return Date" is defined as the date the LMU information is sent back to the CLEC from BellSouth. The interval calculation is reset to Zero when a CLEC initiated change occurs on the Manual LMU request.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = $(c \div d)$

- c = Sum of all Response Intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = $(e \div f) \times 100$

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- · CLEC Aggregate
- CLEC Specific
- · Geographic Scope
 - State
 - Region
- · Interval for manual LMUs:
 - 0-1 day
- >1-2 days
- >2-3 days
- 0 < 3 days
- >3 6 days



- \geq 6 10 days
- > 10 days
- · Average Interval in days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	
Total Number of Inquiries	
SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Loops	Benchmark • 95% in 3 Business Days

SEEM Measure

SEEM Measure			
Yes	Tier I		
	Tier II	X	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• Loops	Benchmark • 95% in 3 Business Days



Tennessee Performance Metrics

PO-2: Loop Make Up - Response Time - Electronic

Definition

This report measures the average interval and the percent within the interval from the electronic submission of a Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- · Manually submitted inquiries.
- Designated Holidays are excluded from the interval calculation.
- · Canceled Requests.

Business Rules

The response interval starts when the CLEC's Mechanized Loop Makeup Service Inquiry (LMUSI) is submitted electronically through the Operational Support Systems interface, LENS, TAG or RoboTAG. It ends when BellSouth's Loop Facility Assignment and Control System (LFACS) responds electronically to the CLEC with the requested Loop Makeup data via LENS, TAG or RoboTAG Interfaces.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC. EDI is not a pre-ordering system, and, therefore, is not applicable in this measure.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = $(c \div d)$

- c = Sum of all response intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = $(e \div f) \times 100$

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- · CLEC Aggregate
- CLEC Specific
- · Geographic Scope
 - State
 - Region
- Interval for electronic LMUs:
 - 0-1 minute
 - >1-5 minutes
 - $0 \le 5$ minutes
- > 5 8 minutes
- > 8 15 minutes
- > 15 minutes
- Average Interval in minutes



Tennessee Performance Metrics

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report MonthLegacy Contract	Not Applicable
Response IntervalRegional Scope	

SQM Disaggregation - Analog/Benchmark

	SQM LEVEL of Disaggregation	Retail Analog/Benchmark
•	Loop	Benchmark • 90% in 5 Minutes (Reassess after 6 months - new system)

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• Loop	• 90% in 5 Minutes (Reassess after 6 months - new system)



Section 2: Ordering

O-1: Acknowledgement Message Timeliness

Definition

This measurement provides the response interval from the time an LSR is electronically submitted via EDI or TAG until an acknowledgement notice is sent by the system.

Exclusions

None

Business Rules

The process includes EDI & TAG system functional acknowledgements for all Local Service Requests (LSRs) which are electronically submitted by the CLEC. The start time is the receipt time of the LSR at BellSouth's side of the interface (gateway). The end time is when the acknowledgement is transmitted by BellSouth at BellSouth's side of the interface (gateway). If more than one CLEC uses the same ordering center, an Acknowledgement Message will be returned to the "Aggregator", however, BellSouth will not be able to determine which specific CLEC this message represented.

Calculation

Response Interval = (a - b)

- a = Date and Time Acknowledgement Notices returned to CLEC
- b = Date and Time LSRs electronically submitted by the CLEC via EDI or TAG respectively

Average Response Interval = $(c \div d)$

- c = Sum of all Response Intervals
- d = Total number of electronically submitted LSRs received, from CLECs via EDI or TAG respectively, in the Reporting Period.

Reporting Structure

- · CLEC Aggregate
- CLEC Specific
- · Geographic Scope
 - State
 - Region
- · Electronically Submitted LSRs
 - $0 \leq 10$ minutes
- $> 10 \leq 20$ minutes
- $> 20 \le 30$ minutes
- $0 \leq 30$ minutes
- $> 30 \leq 45$ minutes
- $> 45 \leq 60$ minutes
- $> 60 \le 120$ minutes
- > 120 minutes
- Average interval for electronically submitted LSRs in minutes

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report monthRecord of functional acknowledgements	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• EDI	EDI – 90% within 30 minutes (6 months – 95% within 30 minutes)
• TAG	• TAG – 95% within 30 minutes

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	EDI – 90% within 30 minutes (6 months – 95% within 30 minutes)
• TAG	• TAG – 95% within 30 minutes

(A) **BELL**SOUTH

O-2: Acknowledgement Message Completeness

Definition

This measurement provides the percent of LSRs received via EDI or TAG, which are acknowledged electronically.

Exclusions

Manually submitted LSRs

Business Rules

EDI and TAG send Functional Acknowledgements for all LSRs, which are electronically submitted by a CLEC. If more than one CLEC uses the same ordering center, an Acknowledgement Message will be returned to the "Aggregator", however, BellSouth will not be able to determine which specific CLEC this message represented. The Acknowledgement Message is returned prior to the determination of whether the LSR will be partially mechanized or fully mechanized.

Calculation

Acknowledgement Completeness = $(a \div b) \times 100$

- a = Total number of Functional Acknowledgements returned in the reporting period for LSRs electronically submitted by EDI or TAG respectively
- b = Total number of electronically submitted LSRs received in the reporting period by EDI or TAG respectively

Report Structure

- CLEC Aggregate
- · CLEC Specific
- Geographic Scope
 - State
- Region

Note: Acknowledgement message is generated before the system recognizes whether this message (LSR) will be partially or fully mechanized.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report monthRecord of functional acknowledgements	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• EDI • TAG	Benchmark: 100%

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	Benchmark: 100%
• TAG	



O-3: Percent Flow-Through Service Requests (Summary)

Definition

The percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual intervention.

Exclusions

- · Fatal Rejects
- · Auto Clarification
- · Manual Fallout
- · CLEC System Fallout

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and two types of service: Resale, and Unbundled Network Elements (UNE). The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

- Complex*
- Special pricing plans
- 3. Some Partial migrations
- 4. New telephone number not yet posted to BOCRIS
- 5. Pending order review required
- CSR inaccuracies such as invalid or missing CSR data in CRIS
- 7. Expedites (requested by the CLEC)

- Denials-restore and conversion, or disconnect and conversion orders
- Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- 12. Transfer of calls option for the CLEC end users
- 13. Directory Listings (Indentions and Captions)

* for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

Calculation

Percent Flow Through = $a \div [b - (c + d + e + f)] \times 100$

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c =the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status.

Percent Achieved Flow Through = $a \div [b-(c+d+e)] \times 100$

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued.
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c =the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

- · CLEC Aggregate
 - Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance:	
Report month	Report month	
 Total number of LSRs received, by interface, by CLEC 	Total number of errors by type	
- TAG	- BellSouth system error	
- EDI		
- LENS		
Total number of errors by type, by CLEC		
- Fatal rejects		
- Auto clarification		
- CLEC caused system fallout		
Total number of errors by error code		
Total fallout for manual processing		

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	Benchmark: 90%
• UNE	Benchmark: 85%
• LNP	Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	• Benchmark: 90%
• UNE	• Benchmark: 85%
• LNP	• Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."



O-4: Percent Flow-Through Service Requests (Detail)

Definition

A detailed list, by CLEC, of the percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual or human intervention.

Exclusions

- · Fatal Rejects
- · Auto Clarification
- · Manual Fallout
- · CLEC System Fallout

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and three types of service: Resale, and Unbundled Network Elements (UNE) and specials. The CLEC mechanized ordering process does not include LSRs, which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions:

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

- 1. Complex*
- 2. Special pricing plans
- 3. Some Partial migrations
- 4. New telephone number not yet posted to BOCRIS
- 5. Pending order review required
- CSR inaccuracies such as invalid or missing CSR data in CRIS

- Denials-restore and conversion, or disconnect and conversion orders
- Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- 12. Transfer of calls option for the CLEC end users
- 13. Directory Listings (Indentions and Captions)

- 7. Expedites (requested by the CLEC)
- * for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

(A) **BELL**SOUTH

Calculation

Percent Flow Through = $a \div [b - (c + d + e + f)] \times 100$

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c =the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status.

Percent Achieved Flow Through = $a \div [b-(c+d+e)] \times 100$

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued.
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

Provides the flow through percentage for each CLEC (by alias designation) submitting LSRs through the CLEC mechanized ordering process. The report provides the following:

- CLEC (by alias designation)
- Number of fatal rejects
- · Mechanized interface used
- · Total mechanized LSRs
- · Total manual fallout
- Number of auto clarifications returned to CLEC
- · Number of validated LSRs
- · Number of BellSouth caused fallout
- · Number of CLEC caused fallout
- · Number of Service Orders Issued
- · Base calculation
- · CLEC error excluded calculation

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month Total number of LSRs received, by interface, by CLEC TAG EDI LENS Total number of errors by type, by CLEC Fatal rejects Auto clarification CLEC errors Total number of errors by error code Total fallout for manual processing 	Report month Total number of errors by type BellSouth system error

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	Benchmark: 90%
• UNE	Benchmark: 85%

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
• LNP	• Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELL**SOUTH

O-5: Flow-Through Error Analysis

Definition

An analysis of each error type (by error code) that was experienced by the LSRs that did not flow through or reached a status for a FOC to be issued.

Exclusions

Each Error Analysis is error code specific, therefore exclusions are not applicable.

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

Total for each error type.

Report Structure

Provides an analysis of each error type (by error code). The report is in descending order by count of each error code and provides the following:

- Error Type (by error code)
- · Count of each error type
- · Percent of each error type
- · Cumulative percent
- · Error Description
- · CLEC Caused Count of each error code
- · Percent of aggregate by CLEC caused count
- · Percent of CLEC caused count
- BellSouth Caused Count of each error code
- · Percent of aggregate by BellSouth caused count
- · Percent of BellSouth by BellSouth caused count.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month Total number of LSRs received Total number of errors by type (by error code) CLEC caused error 	 Report month Total number of errors by type (by error code) BellSouth system error

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• NA	• NA

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELL**SOUTH

O-6: CLEC LSR Information

Definition

A list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period.

Exclusions

- · Fatal Rejects
- · LSRs submitted manually

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

NA

Report Structure

Provides a list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period with an explanation of the columns and content. This report is available on a CLEC specific basis. The report provides the following for each LSR.

- CC
- PON
- Ver
- Timestamp
- Type
- Err #
- · Note or Error Description

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report month Record of LSRs received by CC, PON and Ver Record of Timestamp, Type, Err # and Note or Error Description for each LSR by CC, PON and Ver 	NA

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• NA	• NA

SEEM Measure

		SEEM Me	easure
No	Tier I		
	Tier II		



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLSC

Table 1: LSR Flow-Through Matrix

Product	F/Т³	Complex Service ⁴	Complex Order	Planned Fallout For Manual Handling ¹	Edi	Tag ²	Lens	Comments
2 wire analog DID trunk port	No	UNE	Yes	NA	Z	z	Z	
2 wire analog port	Yes	UNE	No	No	Y	¥	Z	
2 wire ISDN digital line side port	No	UNE	Yes	NA	Z	z	Z	
2 wire ISDN digital loop	Yes	UNE	Yes	No	Y	Y	Z	
3 Way Calling	Yes	No	No	No	Y	¥	Y	
4 wire analog voice grade loop	Yes	UNE	Yes	No	Y	Y	Z	
4 wire DS0 & PRI digital loop	No	UNE	Yes	NA	Z	z	Z	
4 wire DS1 & PRI digital loop	No	UNE	Yes	NA	z	z	Z	
4 wire ISDN DSI digital trunk ports	No	UNE	Yes	NA	z	z	Z	
Accupulse	No	Yes	Yes	NA	Z	z	Z	
ADSL	Yes	UNE	No	oN	Y	Y	Z	
Area Plus	Yes	No	No	oN	Y	Y	Y	
Basic Rate ISDN	No	Yes	Yes	Yes	Y	Y	Z	
Call Block	Yes	No	No	No	Y	Y	Y	
Call Forwarding-Variable	Yes	No	No	No	Y	Y	Y	
Call Return	Yes	No	No	oN	Y	Y	Y	
Call Selector	Yes	No	No	No	Y	Y	Y	
Call Tracing	Yes	No	No	No	Y	Y	Y	
Call Waiting	Yes	No	No	No	Y	Y	Y	
Call Waiting Deluxe	Yes	No	No	No	Y	Y	Y	
Caller ID	Yes	No	No	No	Y	Y	Y	
CENTREX	No	Yes	Yes	NA	Z	Z	Z	

BELLSC

Table 1: LSR Flow-Through Matrix

Product	F/T ³	Complex Service ⁴	Complex Order	Planned Fallout For Manual Handling ¹	Edi	Tag ²	Lens	Comments
DID WITH PBX ACT W	No	Yes	Yes	səA	Y	N	Y	
DID ACT W	No	Yes	Yes	Yes	Y	Z	Y	
Digital Data Transport	No	UNE	Yes	NA	z	Z	z	
Directory Listing Indentions	No	No	No	səA	Y	Y	Y	
Directory Listings Captions	No	No	Yes	Yes	Y	Y	Y	
Directory Listings (simple)	Yes	No	No	No	Y	Y	Y	
DS3	No	UNE	Yes	NA	Z	N	N	
DS1 Loop	Yes	UNE	Yes	oN	Y	Y	N	
DSO Loop	Yes	UNE	Yes	oN	Y	Y	N	
Enhanced Caller ID	Yes	No	No	oN	Y	Y	Y	
ESSX	No	Yes	Yes	NA	N	N	N	
Flat Rate/Business	Yes	No	No	oN	Y	Y	Y	
Flat Rate/Residence	Yes	No	No	No	Y	Y	Y	
FLEXSERV	No	Yes	Yes	NA	z	Z	Z	
Frame Relay	No	Yes	Yes	NA	Z	N	N	
FX	No	Yes	Yes	NA	Z	N	N	
Ga. Community Calling	Yes	No	No	oN	Y	Y	Y	
HDSL	Yes	UNE	No	oN	Y	Y	Z	
Hunting MLH	No	C/S	C/S	SeY	Y	Y	Z	
Hunting Series Completion	No	C/S	C/S	oN	Y	Y	Y	
INP to LNP Conversions	No	UNE	Yes	SeY	Y	Y	Z	
LightGate	No	Yes	Yes	NA	Z	Z	N	

BELLSC

Table 1: LSR Flow-Through Matrix

Product	F/T ³	Complex Service ⁴	Complex Order	Planned Fallout For Manual Handling ¹	Edi	Tag ²	Lens	Comments
Local Number Portability	Yes	UNE	Yes	No	Y	Y	Z	
LNP with Complex Listing	No	UNE	Yes	Yes	Y	Y	Z	
LNP with Partial Migration	oN	UNE	Yes	Yes	Y	Y	Z	
LNP with Complex Services	No	UNE	Yes	Yes	Y	Y	z	
Loop+INP	Yes	UNE	No	No	Y	Y	z	
Loop+LNP	Yes	UNE	No	No	Y	Y	Z	
Measured Rate/Bus.	Yes	No	No	No	Y	Y	Y	
Measured Rate/Res.	Yes	No	No	No	Y	Y	Y	
Megalink	No	Yes	Yes	NA	z	z	Z	
Megalink-T1	No	Yes	Yes	NA	Z	Z	Z	
Memory Call	Yes	No	No	No	Y	Y	Y	
Memory Call Ans. Svc.	Yes	No	No	No	Y	Y	Y	
Multiserv	oN	Yes	Yes	NA	z	z	z	
Native Mode LAN Interconnection (NMLI)	No	Yes	Yes	NA	z	z	Z	
Off-Prem Stations	No	Yes	Yes	NA	Z	z	z	
Optional Calling Plan	Yes	oN	No	oN	Ā	Y	Y	
Package/Complete Choice and area plus	Yes	oN	No	No	Ā	Y	Y	
Pathlink Primary Rate ISDN	No	Yes	Yes	NA	N	N	N	
Pay Phone Provider	No	No	No	NA	N	N	N	
PBX Standalone ACT A,C, D	No	Yes	Yes	Yes	Y	Y	N	
PBX Trunks	No	Yes	Yes	Yes	Y	Y	Z	

(A) **BELL**SC

Table 1: LSR Flow-Through Matrix

Product	F/T ³	Complex Service ⁴	Complex Order	Planned Fallout For Manual Handling ¹	Edi	Tag ²	Lens	Comments
Port/Loop Combo	Yes	UNE	No	No	Y	Y	Y	
Port/Loop PBX	No	No	No	Yes	Y	Y	N	
Preferred Call Forward	Yes	No	No	No	Y	Y	Y	
RCF Basic	Yes	No	No	No	Y	Y	Y	
Remote Access to CF	Yes	No	No	No	Y	Y	Y	
Repeat Dialing	Yes	No	No	No	Y	Y	Y	
Ringmaster	Yes	No	oN	No	Y	Y	Y	
Smartpath	oN	Yes	SəK	NA	Z	Z	N	
SmartRING	oN	Yes	səX	NA	Z	Z	N	
Speed Calling	Yes	No	oN	No	Y	Y	Y	
Synchronet	oN	Yes	sə _X	Yes	Y	Y	N	
Tie Lines	oN	Yes	səA	NA	Z	Z	N	
Touchtone	SeX	No	oN	No	Y	Y	Y	
Unbundled Loop-Analog 2W, SL1, SL2	Yes	UNE	No	No	Y	Y	Y	
WATS	oN	Yes	Yes	NA	Z	Z	Z	
xDSL Extended LOOP	No	UNE	Yes	NA	Z	Z	N	

Note 1: Planned Fallout for Manual Handling denotes those services that are electronically submitted and are not intended to flow through due to the complexity of the service.

Note²: The TAG column includes those LSRs submitted via Robo TAG.

cial pricing plans, denials – restore and conversion or disconnect and conversion both required, partial migrations (although conversions-as-is flow through), class of service invalid in Note³: For all services that indicate 'No' for flow-through, the following reasons, in addition to errors or complex services, also prompt manual handling: Expedites from CLECs, specertain states with some TOS – e.g. government, or cannot be changed when changing main TN on Cactivity, low volume – e.g. activity type T=move, pending order review required, more than 25 business lines, CSR inaccuracies such as invalid or missing CSR data in CRIS, Directory listing indentions and captions, transfer of calls option for CLEC end user new TN not yet posted to BOCRIS. Many are unique to the CLEC environment.

Note⁴: Services with C/S in the Complex Service and/or the Complex Order columns can be either complex or simple

(A) BELLSOUTH

O-7: Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) received which are rejected due to error or omission. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

Service Requests canceled by the CLEC prior to being rejected/clarified.

Business Rules

Fully Mechanized: An LSR is considered "rejected" when it is submitted electronically but does not pass LEO edit checks in the ordering systems (EDI, LENS, TAG, LEO, LESOG) and is returned to the CLEC without manual intervention. There are two types of "Rejects" in the Mechanized category:

A Fatal Reject occurs when a CLEC attempts to electronically submit an LSR but required fields are either not populated or incorrectly populated and the request is returned to the CLEC before it is considered a valid LSR.

An Auto Clarification occurs when a valid LSR is electronically submitted but rejected from LESOG because it does not pass further edit checks for order accuracy.

Partially Mechanized: A valid LSR, which is electronically submitted (via EDI, LENS, TAG) but cannot be processed electronically and "falls out" for manual handling. It is then put into "clarification" and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs electronically submitted by the CLEC.

Non-Mechanized: LSRs which are faxed or mailed to the LCSC for processing and "clarified" (rejected) back to the CLEC by the BellSouth service representative.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Interconnection Purchasing Center (IPC). Trunk data is reported as a separate category.

Calculation

Percent Rejected Service Requests = $(a \div b) \times 100$

- a = Total Number of Rejected Service Requests in the reporting period
- b = Total Number of Service Requests Received in the reporting period

Report Structure

- · Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- CLEC Aggregate
- · Geographic Scope
 - State
 - Region
- · Product Specific percent Rejected
- Total percent Rejected

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	
Total number of LSRs	
Total number of Rejects	
State and Region	
Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Mechanized, Partially Mechanized and Non-Mechanized	Diagnostic
Resale - Residence	
Resale - Business	
• Resale – Design (Special)	
• Resale PBX	
Resale Centrex	
Resale ISDN	
LNP Standalone	
2W Analog Loop Design	
2W Analog Loop Non-Design	
• UNE Digital Loop < DS1	
 UNE Digital Loop ≥ DS1 	
 UNE Loop + Port Combinations 	
Switch Ports	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
Local Interoffice Transport	
Local Interconnection Trunks	

SEEM Measure

	SEEM Me	easure
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLSOUTH

O-8: Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by CLEC prior to being rejected/clarified.
- Designated Holidays are excluded from the interval calculation.
- LSRs which are identified and classified as "Projects"
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups - Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

Business Rules

Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is rejected (date and time stamp or reject in EDI, TAG or LENS). Auto Clarifications are considered in the Fully Mechanized category.

Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until it falls out for manual handling. The stop time on partially mechanized LSRs is when the LCSC Service Representative clarifies the LSR back to the CLEC via LENS, EDI, or TAG.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.

Non-Mechanized: The elapsed time from receipt of a valid LSR (date and time stamp of FAX or date and time mailed LSR is received in the LCSC) until notice of the reject (clarification) is returned to the CLEC via LON.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Interconnection Purchasing Center (IPC). Trunk data is reported as a separate category.

Calculation

Reject Interval = (a - b)

- a = Date and Time of Service Request Rejection
- b = Date and Time of Service Request Receipt

Average Reject Interval = $(c \div d)$

- c = Sum of all Reject Intervals
- d = Number of Service Requests Rejected in Reporting Period

Report Structure

- · CLEC Specific
- CLEC Aggregate
- · Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- Geographic Scope

Tennessee Performance Metrics

- State
- Region
- · Mechanized:
- $0 \leq 4 \text{ minutes}$
- $> 4 \leq 8 \text{ minutes}$
- >8 \leq 12 minutes
- > 12 < 60 minutes
- $0 \leq 1 \text{ hour}$
- $> 1 \leq 4 \text{ hours}$
- $> 4 \leq 8 \text{ hours}$
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- $> 20 \le 24 \text{ hours}$
- > 24 hours
- · Partially Mechanized:
 - $0 \leq 1 \text{ hour}$
 - $> 1 \leq 4 \text{ hours}$
 - > 4 \leq 8 hours
 - $> 8 \le 10 \text{ hours}$
 - $0 \leq 10 \text{ hours}$
 - $> 10 \le 18 \text{ hours}$
 - $0 \leq 18 \text{ hours}$
 - $> 18 \le 24 \text{ hours}$
 - > 24 hours
- · Non-mechanized:
- $0 \leq 1 \text{ hour}$
- $> 1 \leq 4 \text{ hours}$
- > 4 \leq 8 hours
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- $> 20 \le 24 \text{ hours}$
- $0 \leq 24 \text{ hours}$
- > 24 hours
- Trunks:
 - < 4 days
 - $> 4 \le 8 \text{ days}$
 - $> 8 \le 12 \text{ days}$
 - $> 12 \le 14 \text{ days}$
 - $> 14 \le 20 \text{ days}$
 - > 20 days
- · Average Interval for mechanized reports in hours, non-mechanized and Trunk reports in days.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	
Reject Interval	
Total Number of LSRs	
Total number of Rejects	
State and Region	
Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
 Resale – Residence Resale – Business Resale – Design (Special) Resale PBX Resale Centrex Resale ISDN LNP Standalone 2W Analog Loop Design 2W Analog Loop Non-Design UNE Digital Loop < DS1 UNE Digital Loop > DS1 UNE Loop + Port Combinations Switch Ports UNE xDSL (ADSL, HDSL, UCL) Line Sharing Local Interoffice Transport 	 Mechanized: - 97% within 1Hour Partially Mechanized: - 85% within 18 Hours in 3 Months - 85% within 10 Hours in 6 Months Non-Mechanized: - 85% within 24 Hours
Local Interconnection Trunks	Trunks: 85% within 4 Days

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized -	• 97% ≤ 1 hour

(A) **BELL**SOUTH

O-9: Firm Order Confirmation Timeliness

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of valid LSR to distribution of a Firm Order Confirmation.

Exclusions

- Service Requests canceled by CLEC prior to being rejected/clarified.
- Designated Holidays are excluded from the interval calculation.
- LSRs which are identified and classified as "Projects" (under development)
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM
From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

Business Rules

- Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is processed, appropriate service orders are generated and a Firm Order Confirmation is returned to the CLEC via EDI, LENS or TAG.
- Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS, or TAG) which falls out for manual handling until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is returned to the CLEC via EDI, LENS, or TAG.
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.
- Non-Mechanized: The elapsed time from receipt of a valid paper LSR (date and time stamp of FAX or date and time paper LSRs received in LCSC) until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is sent to the CLEC via LON.
- Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Interconnection Purchasing Center (IPC). Trunk data is reported as a separate category.

Calculation

Firm Order Confirmation Time = (a - b)

- a = Date and Time of Firm Order Confirmation
- b = Date and Time of Service Request Receipt

Firm Order Confirmation Timeliness = $(c \div d)$

- c = Sum of all Firm Order Confirmation Times
- d = Number of Service Requests Confirmed in Reporting Period

Report Structure

- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
 - CLEC Specific
 - CLEC Aggregate
- · Geographic Scope



- State
- Region
- · Fully Mechanized:
- $0 \leq 15 \text{ minutes}$
- $> 15 \leq 30 \text{ minutes}$
- $> 30 \le 45 \text{ minutes}$
- > 45 < 60 minutes
- $> 60 \le 90 \text{ minutes}$
- $> 90 \le 120 \text{ minutes}$
- $> 120 \le 180 \text{ minutes}$
- $0 \leq 3 \text{ hours}$
- $> 3 \le 6$ hours
- $> 6 \le 12 \text{ hours}$
- $> 12 \le 24 \text{ hours}$
- $> 24 \le 48 \text{ hours}$
- > 48 hours
- Partially Mechanized:
- $0 \leq 4 \text{ hours}$
- > 4 \leq 8 hours
- $> 8 \le 10 \text{ hours}$
- $0 \leq 10 \text{ hours}$
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- $> 18 \le 24 \text{ hours}$
- > 24 \leq 48 hours
- > 48 hours
- Non-mechanized:
- $0 \leq 4 \text{ hours}$
- > 4 \leq 8 hours
- $> 8 \leq 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- > 20 \leq 24 hours
- > 24 \leq 36 hours
- $0 \leq 36 \text{ hours}$
- $> 36 \le 48 \text{ hours}$
- > 48 hours
- Trunks:
 - $0 \leq 5 \text{ days}$
- $> 5 \le 10 \text{ days}$
- $0 \le 10 \text{ days}$
- $> 10 \le 15 \text{ days}$
- $> 15 \le 20 \text{ days}$
- > 20 days
- · Average Interval in Days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	
Interval for FOC	
Total number of LSRs	
State and Region	
Total Number of ASRs (Trunks)	

BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
 Resale – Residence Resale – Business Resale – Design (Special) Resale PBX Resale Centrex Resale ISDN LNP Standalone 	 Mechanized: - 95% within 3 Hours Partially Mechanized: 85% within 18 Hours in 3 Months 85% within 10 Hours in 6 Months Non-Mechanized: 85% within 36 hours
 2W Analog Loop Design 2W Analog Loop Non-Design UNE Digital Loop < DS1 UNE Digital Loop ≥ DS1 UNE Loop + Port Combinations Switch Ports UNE xDSL (ADSL, HDSL, UCL) Line Sharing Local Interoffice Transport 	
Local Interconnection Trunks	Trunks: - 95% within 10 days

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% within 3 hours
Partially Mechanized	85% within 18 Hours in 3 Months 85% within 10 Hours in 6 Months
Non-Mechanized	85% within 36 hours
• IC Trunks	• 95% within 10 days

🕮 **BELL**SOUTH

O-10: Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual¹

Definition

This report measures the interval and the percent within the interval from the submission of a Service Inquiry (SI) with Firm Order LSR to the distribution of a Firm Order Confirmation (FOC).

Exclusions

- Designated Holidays are excluded from the interval calculation.
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation of the Service Inquiry.
- Canceled Requests
- · Electronically Submitted Requests

Business Rules

This measurement combines four intervals:

- 1. From receipt of Service Inquiry with LSR to hand off to the Service Advocacy Center (SAC) for Loop 'Look-up'.
- 2. From SAC start date to SAC complete date.
- 3. From SAC complete date to the Complex Resale Support Group (CRSG) complete date with hand off to LCSC.
- 4. From receipt of SI/LSR in the LCSC to Firm Order Confirmation.

Calculation

FOC Timeliness Interval = (a - b)

- a = Date and Time Firm Order Confirmation (FOC) for SI with LSR returned to CLEC
- b = Date and Time SI with LSR received

Average Interval = $(c \div d)$

- c = Sum of all FOC Timeliness Intervals
- d = Total number of SIs with LSRs received in the reporting period

Percent Within Interval = $(e \div f) \times 100$

- e = Total number of Service Inquiries with LSRs received by the CRSG to distribution of FOC by the Local Carrier Service Center (LCSC)
- f = Total number of Service Inquiries with LSRs received in the reporting period

Report Structure

- · CLEC Aggregate
- CLEC Specific
- · Geographic Scope
 - State
 - Region
- Intervals
- $0 \leq 3$ days
- > 3 < 5 days
- $0 \le 5 \text{ days}$
- $> 5 \le 7$ days
- $> 7 \le 10 \text{ days}$
- $> 10 \le 15 \text{ days}$
- · Average Interval measured in days

1. See O-9 for FOC Timeliness



Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	
Total Number of Requests	
SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
xDSL (includes UNE unbundled ADSL, HDSL and UNE	• 95% Returned within 5 Business days
Unbundled Copper Loops)	
Unbundled Interoffice Transport	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

🕮 **BELL**SOUTH

O-11: Firm Order Confirmation and Reject Response Completeness

Definition

A response is expected from BellSouth for every Local Service Request transaction (version). More than one response or differing responses per transaction is not expected. Firm Order Confirmation and Reject Response Completeness is the corresponding number of Local Service Requests received to the combination of Firm Order Confirmation and Reject Responses.

Exclusions

- Service Requests canceled by the CLEC prior to FOC or Rejected/Clarified.
- · Non-Mechanized LSRs

Business Rules

Mechanized – The number of FOCs or Auto Clarifications sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG).

Partially Mechanized – The number of FOCs or Rejects sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG), which fall out for manual handling by the LCSC personnel.

Total Mechanized - The number of the combination of Fully Mechanized and Partially Mechanized LSRs

Note: Manual (Non-Mechanized) LSRs have no version control by the very nature of the manual process, therefore, non-mechanized LSRs are not captured by this report.

For CLEC Results:

Firm Order Confirmation and Reject Response Completeness is determined in two dimensions:

Percent responses is determined by computing the number of Firm Order Confirmations and Rejects transmitted by BellSouth and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Percent of multiple responses is determined by computing the number of Local Service Request unique versions receiving more than one Firm Order Confirmation, Reject or the combination of the two and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Calculation

Single FOC/Reject Response Expected

Firm Order Confirmation / Reject Response Completeness = $(a \div b) \times 100$

- a = Total Number of Service Requests for which a Firm Order Confirmation or Reject is Sent
- b = Total Number of Service Requests Received in the Report Period

Multiple or Differing FOC / Reject Responses Not Expected

Response Completeness = $[(a + b) \div c] \times 100$

- a = Total Number of Firm Order Confirmations Per LSR Version
- b = Total Number of Reject Responses Per LSR Version
- c = Total Number of Service Requests (All Versions) Received in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- · State and Region
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Specific

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	Not Applicable
Reject interval	
Total number of LSRs	
 Total number of rejects 	
Total number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	• 95% Returned
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
LNP Standalone	
2W Analog Loop Design	
• 2W Analog Loop Non – Design	
• UNE Digital Loop < DS1	
 UNE Digital Loop ≥ DS1 	
UNE Loop and Port Combinations	
Switch Ports	
• UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
Local Interoffice Transport	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% Returned

BELLSOUTH



Definition

Measures the average time a customer is in queue.

Exclusions

None

Business Rules

The clock starts when the appropriate option is selected (i.e., 1 for Resale Consumer, 2 for Resale Multiline, and 3 for UNE-LNP, etc.) and the call enters the queue for that particular group in the LCSC. The clock stops when a BellSouth service representative in the LCSC answers the call. The speed of answer is determined by measuring and accumulating the elapsed time from the entry of a CLEC call into the BellSouth automatic call distributor (ACD) until a service representative in BellSouth's Local Carrier Service Center (LCSC) answers the CLEC call.

Calculation

Speed of Answer in Ordering Center = $(a \div b)$

- a = Total seconds in queue
- b = Total number of calls answered in the Reporting Period

Report Structure

Aggregate

- CLEC Local Carrier Service Center
- · BellSouth
- Business Service Center
- Residence Service Center

Note: Combination of Residence Service Center and Business Service Center data under development

Data Retained

elating to BellSouth Performance
d tracking through BellSouth Retail center support

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Aggregate CLEC – Local Carrier Service Center BellSouth Business Service Center Residence Service Center	Diagnostic

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

O-12: Speed of Answer in Ordering Center

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

🕮 **BELL**SOUTH

O-13: LNP-Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) which are rejected due to error or omission. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are excluded.

Exclusions

- Service Requests canceled by the CLEC
- Fatal Rejects
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A Fatal Reject occurs when a CLEC attempts to electronically submit an LSR (via EDI or TAG) but required fields are not populated correctly and the request is returned to the CLEC.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An Auto Clarification is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which is electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

LNP-Percent Rejected Service Requests = $(a \div b) \times 100$

- a = Number of Service Requests Rejected in the Reporting Period
- b = Number of Service Requests Received in the Reporting Period

Report Structure

- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Not Applicable	Not Applicable

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• LNP	Diagnostic
UNE Loop w/LNP	

BELLSOUTH®

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

🕮 **BELL**SOUTH

O-14: LNP-Reject Interval Distribution & Average Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are excluded.

Exclusions

- Service Requests canceled by the CLEC
- Fatal Rejects
- · Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules

The Reject interval is determined for each rejected LSR processed during the reporting period. The Reject interval is the elapsed time from when BellSouth receives LSR until that LSR is rejected back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of rejected LSRs to produce the reject interval distribution.

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A Fatal Reject occurs when a CLEC attempts to electronically submit an LSR but required fields are not populated correctly and the request is returned to the CLEC.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An Auto Clarification is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

Reject Interval = (a - b)

- a = Date & Time of Service Request Rejection
- b = Date & Time of Service Request Receipt

Average Reject Interval = $(c \div d)$

- c = Sum of all Reject Intervals
- d = Total Number of Service Requests Rejected in Reporting Period

Reject Interval Distribution = $(e \div f) \times 100$

- e = Service Requests Rejected in reported interval
- f = Total Number of Service Requests Rejected in Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- CLEC Specific
- CLEC Aggregate
- State, Region

(A) **BELLSOUTH** *

- · Fully Mechanized:
 - $0 \leq 4 \text{ minutes}$
- > 4 \leq 8 minutes
- $> 8 \le 12 \text{ minutes}$
- $> 12 \leq 60 \text{ minutes}$
- $0 \leq 1$ hour
- > 1 < 4 hours
- $> 4 \le 8 \text{ hours}$
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- $> 20 \le 24 \text{ hours}$
- > 24 hours
- · Partially Mechanized:
 - $0 \leq 1$ hour
- $> 1 \leq 4 \text{ hours}$
- > 4 \leq 8 hours
- $> 8 \le 10 \text{ hours}$
- 0 < 10 hours
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- $> 18 \le 24 \text{ hours}$
- > 24 hours
- · Non-Mechanized:
 - 0 < 1 hour
- $> 1 \leq 4$ hours
- > 4 \leq 8 hours
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- > 20 < 24 hours
- $0 \leq 24 \text{ hours}$
- > 24 hours
- · Average Interval in Days

Data Retained

	Relating to CLEC Experience	Relating to BellSouth Experience
Under	Development	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• LNP	Mechanized: 97% within 1Hour
UNE Loop with LNP	Partially Mechanized: 85% within 18 Hours
	Non-Mechanized: 85% within 24 Hours

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

🕮 **BELL**SOUTH

O-15: LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of a valid LSR to distribution of a firm order confirmation.

Exclusions

- Rejected LSRs (Clarifications or Fatal Rejects)
- Order Activities of BellSouth or the CLEC associated with interval or administrative use of local services (Record Orders, Test Orders, etc.) where identifiable.

Business Rules

The Firm Order Confirmation interval is determined for each confirmed LSR processed during the reporting period. The Firm Order Confirmation interval is the elapsed time from when BellSouth receives an LSR until that LSR is confirmed back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed to produce the Firm Order Confirmation timeliness interval distribution.

- Mechanized: The elapsed time from receipt of a valid LSR until the LSR is processed and appropriate service orders are generated in SOCS without manual intervention.
- Partially Mechanized: The elapsed time from receipt of an electronically submitted LSR which falls for manual handling by the LCSC personnel until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation system (SONGS).
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized FOCs.
- Non-Mechanized: (Under Development) A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

Reject Interval = (a - b)

- a = Date & Time of Firm Order Confirmation
- b = Date & Time of Service Request Receipt)

Average Reject Interval = $(c \div d)$

- c = Sum of all Reject Intervals
- d = Total Number of Service Requests Confirmed in Reporting Period

FOC Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Service Requests Confirmed in interval
- f = Total Service Requests Confirmed in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- CLEC Specific
- · CLEC Aggregate
- · State and Region
- · Fully Mechanized:
- $0 \leq 15$ minutes
- $> 15 \le 30 \text{ minutes}$
- $> 30 \le 45 \text{ minutes}$
- > 45 \leq 60 minutes $> 60 - \le 90 \text{ minutes}$
- $> 90 \le 120 \text{ minutes}$
- $> 120 \le 180$ minutes
- $0 \leq 3$ hours
- $> 3 \le 6$ hours



Tennessee Performance Metrics

- > 6 < 12 hours
- $> 12 \le 24 \text{ hours}$
- $> 24 \le 48 \text{ hours}$
- > 48 hours
- Partially Mechanized:
 - $0 \leq 4 \text{ hours}$
 - > 4 < 8 hours
 - $> 8 \le 10 \text{ hours}$
 - $0 \leq 18 \text{ hours}$
 - $> 10 \le 18 \text{ hours}$
 - $> 18 \le 24 \text{ hours}$
 - $> 24 \le 48 \text{ hours}$
 - > 48 hours
- · Non-Mechanized:
 - $0 \leq 4 \text{ hours}$
- $> 4 \le 8 \text{ hours}$
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- $> 20 \le 24 \text{ hours}$
- $> 24 \le 36 \text{ hours}$
- $0 \leq 36 \text{ hours}$
- $> 36 \le 48 \text{ hours}$
- > 48 hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Total Number of LSRs	
Total Number of FOCs	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
LNP UNE Loop with LNP	 Mechanized: 95% within 3 Hours Partially Mechanized: 85% within 18 hours (10 hrs. after 6 months) Non-Mechanized: 85% within 36 hours

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 3: Provisioning

P-1: Mean Held Order Interval & Distribution Intervals

Definition

When delays occur in completing CLEC orders, the average period that CLEC orders are held for BellSouth reasons, pending a delayed completion, should be no worse for the CLEC when compared to BellSouth delayed orders. Calculation of the interval is the total days orders are held and pending but not completed that have passed the currently committed due date; divided by the total number of held orders. This report is based on orders still pending, held and past their committed due date at the close of the reporting period. The distribution interval is based on the number of orders held and pending but not completed over 15 and 90 days. (Orders reported in the >90 day interval are also included in the >15 day interval.)

Exclusions

- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D) & From (F) orders
- Orders with appointment code of 'A' for Rural orders.

Business Rules

Mean Held Order Interval: This metric is computed at the close of each report period. The held order interval is established by first identifying all orders, at the close of the reporting interval, that both have not been reported as completed in SOCS and have passed the currently committed due date for the order. For each such order, the number of calendar days between the earliest committed due date on which BellSouth had a company missed appointment and the close of the reporting period is established and represents the held order interval for that particular order. The held order interval is accumulated by the standard groupings, unless otherwise noted, and the reason for the order being held. The total number of days accumulated in a category is then divided by the number of held orders within the same category to produce the mean held order interval. The interval is by calendar days with no exclusions for Holidays or Sundays.

CLEC Specific reporting is by type of held order (facilities, equipment, other), total number of orders held, and the total and average days.

Held Order Distribution Interval: This measure provides data to report total days held and identifies these in categories of >15 days and >90 days. (Orders counted in >90 days are also included in >15 days).

Calculation

Mean Held Order Interval = $a \div b$

- a = Sum of held-over-days for all Past Due Orders Held for the reporting period
- b = Number of Past Due Orders Held and Pending But Not Completed and past the committed due date

Held Order Distribution Interval (for each interval) = $(c \div d) \times 100$

- c = # of Orders Held for ≥ 15 days or # of Orders Held for ≥ 90 days
- d = Total # of Past Due Orders Held and Pending But Not Completed)

Report Structure

- CLEC Specific
- · CLEC Aggregate
- BellSouth Aggregate
- Circuit Breakout < 10, ≥ 10 (except trunks)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month CLEC Order Number and PON (PON) Order Submission Date (TICKET_ID) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Hold Reason Total line/circuit count Geographic Scope 	 Report month BellSouth Order Number Order Submission Date Committed Due Date Service Type Hold Reason Total line/circuit count Geographic Scope
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop-Non-Design	Retail Residence and Business (POTS - Excluding Switch- Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
UNE ISDN (Includes UDC)	Retail ISDN - BRI
UNE Line Sharing	ADSL provided to Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



Tennessee Performance Metrics

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-2: Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices

Definition

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC.

The interval is from the date/time the notice is released to the CLEC/BellSouth systems until 5pm on the commitment date of the order. The Percent of Orders is the percentage of orders given jeopardy notices for facility delay in the count of orders confirmed in the report period.

Exclusions

- · Orders held for CLEC end user reasons
- Disconnect (D) & From (F) orders

Business Rules

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC. The number of committed orders in a report period is the number of orders that have a due date in the reporting period. Jeopardy notices for interconnection trunks results are usually zero as these trunks seldom experience facility delays. The Committed due date is considered the Confirmed due date.

Calculation

Jeopardy Interval = a - b

- a = Date and Time of Jeopardy Notice
- b = Date and Time of Scheduled Due Date on Service Order

Average Jeopardy Interval = $c \div d$

- c = Sum of all jeopardy intervals
- d = Number of Orders Notified of Jeopardy in Reporting Period

Percent of Orders Given Jeopardy Notice = $(e \div f) \times 100$

- e = Number of Orders Given Jeopardy Notices in Reporting Period
- f = Number of Orders Confirmed (due) in Reporting Period)

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Mechanized Orders
- · Non-Mechanized Orders

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month CLEC Order Number and PON Date and Time Jeopardy Notice sent Committed Due Date Service Type 	 Report month BellSouth Order Number Date and Time Jeopardy Notice sent Committed Due Date Service Type
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark:
% Orders Given Jeopardy Notice	
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
• LNP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business (POTS Excluding Switch- Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	• Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
UNE ISDN (Includes UDC)	Retail ISDN BRI
UNE Line Sharing	ADSL provided to Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail
Average Jeopardy Notice Interval (Electronic Only)	• 95% ≥ 48 Hours

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-3: Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that the CLEC can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for Total misses and End User Misses.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders Test Orders, etc.)
- Disconnect (D) & From (F) orders
- End User Misses on Interconnection Trunks

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of orders with completion dates in the reporting period that are past the original committed due date. Missed Appointments caused by end-user reasons will be included and reported separately. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date. Which means there cannot be a cutoff time for commitments, as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation

Percent Missed Installation Appointments = $(a \div b) \times 100$

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Report in Categories of <10 lines/circuits ≥ 10 lines/circuits (except trunks)
- · Dispatch/No Dispatch

Report Explanation: The difference between End User MA and Total MA is the result of BellSouth caused misses. Here, Total MA is the total percent of orders missed either by BellSouth or CLEC end user. The End User MA represents the percentage of orders missed by the CLEC or their end user.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
CLEC Order Number and PON (PON)	BellSouth Order Number
Committed Due Date (DD)	Committed Due Date (DD)
Completion Date (CMPLTN DD)	Completion Date (CMPLTN DD)
Status Type	Status Type
Status Notice Date	Status Notice Date
Standard Order Activity	Standard Order Activity
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business (POTS Excluding Switch- Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence and Business
UNE Switch ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
UNE ISDN (Includes UDC)	Retail ISDN - BRI
UNE Line Sharing	ADSL provided to Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
• UNE Loop + Port Combinations	Retail Residence and Business
• UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

A BELLSOUTH

P-4: Average Completion Interval (OCI) & Order Completion Interval Distribution

Definition

The "average completion interval" measure monitors the interval of time it takes BellSouth to provide service for the CLEC or its own customers. The "Order Completion Interval Distribution" provides the percentages of orders completed within certain time periods. This report measures how well BellSouth meets the interval offered to customers on service orders.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D&F) orders (Except "D" orders associated with LNP Standalone)
- "L" Appointment coded orders (where the customer has requested a later than offered interval)
- · End user-caused misses

Business Rules

The actual completion interval is determined for each order processed during the reporting period. The completion interval is the elapsed time from when BellSouth issues a FOC or SOCS date time stamp receipt of an order from the CLEC to BellSouth's actual order completion date. The clock starts when a valid order number is assigned by SOCS and stops when the technician or system completes the order in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33-day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on the same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

The interval breakout for UNE and Design is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, $\ge 30 = 30$ and greater.

Calculation

Completion Interval = (a - b)

- a = Completion Date
- b = FOC/SOCS date time-stamp (application date)

Average Completion Interval = $(c \div d)$

- c = Sum of all Completion Intervals
- d = Count of Orders Completed in Reporting Period

Order Completion Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Service Orders Completed in "X" days
- f = Total Service Orders Completed in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Residence & Business reported in day intervals = 0,1,3,4,5,5+
- UNE and Design reported in day intervals =0-5,5-10,10-15,15-20,20-25,25-30,> 30
- All Levels are reported <10 line/circuits; ≥ 10 line/circuits (except trunks)
- ISDN Orders included in Non-Design

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month CLEC Company Name Order Number (PON) Application Date & Time Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope 	 Report month BellSouth Order Number Order Submission Date & Time Order Completion Date & Time Service Type Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch + 2 days
2W Analog Loop Non-Design	Retail Residence and Business (POTS Excluding Switch-Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence and Business
UNE Switch ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	7 Days w/o conditioning
UNE xDSL (HDSL, ADSL and UCL)	14 Days with conditioning
UNE ISDN (Includes UDC)	Retail ISDN BRI
UNE Line Sharing	ADSL provided to Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

P-4: Average Completion Interval (OCI) & Order Completion Interval Distribution

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	7 Days w/o conditioning
• UNE xDSL	14 Days with conditioning
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

P-5: Average Completion Notice Interval

P-5: Average Completion Notice Interval

Definitions

The Completion Notice Interval is the elapsed time between the BellSouth reported completion of work and the issuance of a valid completion notice to the CLEC.

Exclusions

- · Cancelled Service Orders
- · Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D&F orders (Exception: "D" orders associated with LNP Standalone)

Business Rules

Measurement on interval of completion date and time entered by a field technician on dispatched orders, and 5PM start time on the due date for non-dispatched orders; to the release of a notice to the CLEC/BellSouth of the completion status. The field technician notifies the CLEC the work was complete and then he/she enters the completion time stamp information in his/her computer. This information switches through to the SOCS systems either completing the order or rejecting the order to the Work Management Center (WMC). If the completion is rejected, it is manually corrected and then completed by the WMC. The notice is returned on each individual order.

The start time for all orders is the completion stamp either by the field technician or the 5PM due date stamp; the end time for mechanized orders is the time stamp the notice was transmitted to the CLEC interface (LENS, EDI, OR TAG). For non-mechanized orders the end timestamp will be timestamp of order update to C-SOTS system.

Calculation

Completion Notice Interval = (a - b)

- a = Date and Time of Notice of Completion
- b = Date and Time of Work Completion

Average Completion Notice Interval = $c \div d$

- c = Sum of all Completion Notice Intervals
- d = Number of Orders with Notice of Completion in Reporting Period

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Mechanized Orders
- · Non-Mechanized Orders
- Reporting intervals in Hours; 0,1-2,2-4,4-8,8-12,12-24, ≥ 24 plus Overall Average Hour Interval (The categories are inclusive of these time intervals: 0-1 = 0.99; 1-2 = 1-1.99; 2-4 = 2-3.99, etc.)
- Reported in categories of <10 line / circuits; ≥ 10 line/circuits (except trunks)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
CLEC Order Number (so_nbr)	BellSouth Order Number (so_nbr)
 Work Completion Date (cmpltn_dt) 	Work Completion Date (cmpltn_dt)
Work Completion Time	Work Completion Time
Completion Notice Availability Date	Completion Notice Availability Date
Completion Notice Availability Time	Completion Notice Availability Time
Service Type	Service Type
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	NOTE: Code in parentheses is the corresponding header found in the raw data file.

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non-Design	Retail Residence and Business (POTS Excluding Switch- Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence and Business
UNE Switch ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence and Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
UNE ISDN (Includes UDC)	Retail ISDN BRI
UNE Line Sharing	ADSL provided to Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



Tennessee Performance Metrics

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-6: Coordinated Customer Conversions Interval

Definition

This report measures the average time it takes BellSouth to disconnect an unbundled loop from the BellSouth switch and cross connect it to a CLEC equipment. This measurement applies to service orders with and without LNP, and where the CLEC has requested BellSouth to provide a coordinated cut over.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement.
- Delays due to CLEC following disconnection of the unbundled loop
- Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested.

Business Rules

Where the service order includes LNP, the interval includes the total time for the cut over including the translation time to place the line back in service on the ported line. The interval is calculated for the entire cut over time for the service order and then divided by items worked in that time to give the average per-item interval for each service order.

Calculation

Coordinated Customer Conversions Interval = (a - b)

- a = Completion Date and Time for Cross Connection of a Coordinated Unbundled Loop
- b = Disconnection Date and Time of an Coordinated Unbundled Loop

Percent Coordinated Customer Conversions (for each interval) = $(c \div d) \times 100$

- c = Total number of Coordinated Customer Conversions for each interval
- d = Total Number of Unbundled Loop with Coordinated Conversions (items) for the reporting period

Report Structure

- · CLEC Specific
- CLEC Aggregate
- The interval breakout is 0 < 5 = 0-4.99, 5 < 15 = 5-14.99, $\ge 15 = 15$ and greater, plus Overall Average Interval.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog Exists
CLEC Order Number	
Committed Due Date (DD)	
• Service Type (CLASS_SVC_DESC)	
Cut over Start Time	
Cut over Completion time	
 Portability start and completion times (INP orders) 	
Total Conversions (Items)	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Unbundled Loops with INP	• 95% ≤ 15 minutes
Unbundled Loops with LNP	

P-6: Coordinated Customer Conversions Interval

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Unbundled Loops	• 95% ≤ 15 minutes



P-6A: Coordinated Customer Conversions – Hot Cut Timeliness % Within Interval and Average Interval

Definition

This category measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. It measures the percentage of orders where the cut begins within 15 minutes of the requested start time of the order and the average interval.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement.
- Delays caused by the CLEC
- Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested.
- All unbundled loops on multiple loop orders after the first loop.

Business Rules

This report measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. The cut is considered on time if it starts 15 minutes before or after the requested start time. Using the scheduled time and the actual cut over start time, the measurement will calculate the percent within interval and the average interval. If a cut involves multiple lines, the cut will be considered "on time" if the first line is cut within the interval. \leq 15 minutes includes intervals that began 15:00 minutes or less before the scheduled cut time and cuts that began 15 minutes or less after the scheduled cut time; >15 minutes, \leq 30 minutes includes cuts within 15:00 – 30:00 minutes either prior to or after the scheduled cut time; >30 minutes includes cuts greater than 30:00 minutes either prior to or after the scheduled cut time. If IDLC is involved, a four hour window applies to the start time. (8 A.M. to Noon or 1 P.M. to 5 P.M.) This only applies if BellSouth notifies the CLEC by 10:30 A.M. on the day before the due date that the service is on IDLC.

A Hot Cut is considered complete when one of the following occurs:

- 1. BellSouth performs the hot cut, notifies the CLEC by telephone.
- 2. BellSouth performs the hot cut and attempts to notify the CLEC by telephone, but receives no answer and leaves a phone message.

Calculation

% within Interval = $(a \div b) \times 100$

- a = Total Number of Coordinated Unbundled Loop Orders for the interval
- b = Total Number of Coordinated Unbundled Loop Orders for the reporting period

Interval = (c - d)

- c = Scheduled Time for Cross Connection of a Coordinated Unbundled Loop Order
- d = Actual Start Date and Time of a Coordinated Unbundled Loop Order

Average Interval = $(e \div f)$

- · Sum of all Intervals
- Total Number of Coordinated Unbundled Loop Orders for the reporting period.

Report Structure

- · CLEC Specific
- · CLEC Aggregate

Reported in intervals of early, on time and late cuts %≤ 15 minutes; %>15 minutes, ≤30 minutes; %>30 minutes, plus Overall Average Interval

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog exists
CLEC Order Number (so_nbr)	
Committed Due Date (DD)	
Service Type (CLASS SVC DESC)	
Cut over Scheduled Start Time	
Cut over Actual Start Time	
Total Conversions Orders	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
 Product Reporting Level SL1 Time Specific SL1 Non-Time Specific SL2 Time Specific SL2 Non-Time Specific 	• 95% Within + or – 15 minutes of Scheduled Start Time
- SL1 IDLC - SL2 IDLC	• 95% within 4-hour window

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
- UNE Loops	• 95% Within + or – 15 minutes of Scheduled Start time
- SL1 IDLC	95% within 4-hour window

P-6B: Coordinated Customer Conversions – Average Recovery Time

Definition

Measures the time between notification and resolution by BellSouth of a service outage found that can be isolated to the BellSouth side of the network. The time between notification and resolution by BellSouth must be measured to ensure that CLEC customers do not experience unjustifiable lengthy service outages during a Coordinated Customer Conversion. This report measures outages associated with Coordinated Customer Conversions prior to service order completion.

Exclusions

- Cut overs where service outages are due to CLEC caused reasons
- Cut overs where service outages are due to end-user caused reasons

Business Rules

Measures the outage duration time related to Coordinated Customer Conversions from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The duration time is defined as the time from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The interval is calculated on the total outage time for the circuits divided by the total number of outages restored during the report period to give the average outage duration.

Calculation

Recovery Time = (a - b)

- a = Date & Time That Trouble is Closed by CLEC
- b = Date & Time Initial Trouble is Opened with BellSouth

Average Recovery Time = $(c \div d)$

- c = Sum of all the Recovery Times
- d = Number of Troubles Referred to the BellSouth

Report Structure

- · CLEC Specific
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	• None
CLEC Company Name	
CLEC Order Number (so nbr)	
• Committed Due Date (DD)	
Service Type (CLASS_SVC_DESC)	
 CLEC Acceptance Conflict (CLEC_CONFLICT) under 	
development	
CLEC Conflict Resolved (CLEC_RESOVE) under	
development	
 CLEC Conflict MFC (CLEC_CONFLICT_MFC) under 	
development	
Total Conversion Orders	
Note: Code in parentheses is the corresponding header found in the raw data file.	

Tennessee Performance Metrics

P-6B: Coordinated Customer Conversions – Average Recovery Time

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
 Unbundled Loops with INP Unbundled Loops with LNP	Diagnostic

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-6C: Coordinated Customer Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order

Definition

The Percent Provisioning Troubles received within 7 days of a completed service order associated with a Coordinated Customer Conversion (CCC) measures the quality and accuracy of Coordinated Customer Conversion Activities.

Exclusions

- Any order canceled by the CLEC
- · Troubles caused by Customer Provided Equipment

Business Rules

Measures the quality and accuracy of completed service orders associated with Coordinated Customer Conversions. The first trouble report received on a circuit ID within 7 days following a service order completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed Coordinated Customer Conversion service orders and following 7 days after the completion of the service order for a trouble report issue date.

Calculation

% Provisioning Troubles within 7 days of service order completion = $(a \div b) \times 100$

- a = The sum of all CCC Circuits with a trouble within 7 days following service order(s) completion
- b = The total number of CCC service order circuits completed in the previous report calendar month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · Dispatch/Non-Dispatch

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog exists
• CLEC Order Number (so_nbr)	
• PON	
Order Submission Date (TICKET ID)	
Order Submission Time (TICKET ID)	
Status Type	
Status Notice Date	
Standard Order Activity	
Geographic Scope	
Total conversion circuits	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
UNE Loop Design	• ≤ 5%
UNE Loop Non-Design	
 Dispatch/Non-Dispatch 	

P-6C: Coordinated Customer Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• UNE Loops	• ≤ 5%

P-7: Cooperative Acceptance Testing - % of xDSL Loops Tested

Definition

The loop will be considered cooperatively tested when the BellSouth technician places a call to the CLEC representative to initiate cooperative testing and jointly performs the tests with the CLEC.

Exclusions

- Testing failures due to CLEC (incorrect contact number, CLEC not ready, etc.)
- xDSL lines with no request for cooperative testing

Business Rules

When a BellSouth technician finishes delivering an order for an xDSL loop where the CLEC order calls for cooperative testing at the customer's premise, the BellSouth technician is to call a toll free number to the CLEC testing center. The BellSouth technician and the CLEC representative at the center then test the line. As an example of the type of testing performed, the testing center may ask the technician to put a short on the line so that the center can run a test to see if it can identify the short.

Calculation

Cooperative Acceptance Testing - % of xDSL Loops Tested = $(a \div b) \times 100$

- a = Total number of successful xDSL cooperative tests for xDSL lines where cooperative testing was requested in the reporting period
- b = Total Number of xDSL line tests requested by the CLEC and scheduled in the reporting period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- Type of Loop tested

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report Month CLEC Company Name (OCN) CLEC Order Number (so_nbr) and PON (PON) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Acceptance Testing Completed (ACCEPT_TESTING) under development Acceptance Testing Declined (ACCEPT_TESTING) under development Total xDSL Orders Note: Code in parentheses is the corresponding header 	No BellSouth analog exists
Total xDSL Orders	

SQM LEVEL of Disaggregation:	Retail Analog/Benchmark:
• UNE xDSL	• 95% of Lines Tested
- ADSL	
- HDSL	
- UCL	
- OTHER	

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation:	SEEM Analog/Benchmark:
• UNE xDSL	• 95% of Lines Tested

P-8: % Provisioning Troubles within 30 days of Service Order Completion

Definition

Percent Provisioning Troubles within 30 days of Service Order Completion measures the quality and accuracy of Service order activities.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D & F orders
- Trouble reports caused and closed out to Customer Provided Equipment (CPE)

Business Rules

Measures the quality and accuracy of completed orders. The first trouble report from a service order after completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed service orders and following 30 days after completion of the service order for a trouble report issue date.

D & F orders are excluded as there is no subsequent activity following a disconnect.

Note: Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

% Provisioning Troubles within 30 days of Service Order Activity = $(a \div b) \times 100$

- a = Trouble reports on all completed orders 30 days following service order(s) completion
- b = All Service Orders completed in the previous report calendar month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- BellSouth Aggregate
- Reported in categories of <10 line/circuits; ≥ 10 line/circuits (except trunks)
- Dispatch / No Dispatch (except trunks)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
CLEC Order Number and PON	BellSouth Order Number
Order Submission Date (TICKET_ID)	Order Submission Date
 Order Submission Time (TICKET_ID) 	Order Submission Time
Status Type	Status Type
Status Notice Date	Status Notice Date
Standard Order Activity	Standard Order Activity
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	Retail Residence



Tennessee Performance Metrics

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop Non -Design	Retail Residence and Business (POTS - Excluding Switch- Based Orders)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
UNE ISDN (Includes UDC)	Retail ISDN BRI
UNE Line Sharing	ADSL provided to Retail
UNE Switch ports	Retail Residence and Business (POTS)
UNE Loop + Port Combinations	Retail Residence and Business
UNE Combo Other	Retail Residence, Business and Design Dispatch
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
• UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

P-9: Total Service Order Cycle Time (TSOCT)

Definition

This report measures the total service order cycle time from receipt of a valid service order request to the return of a completion notice to the CLEC Interface.

Exclusions

- · Canceled Service Orders
- · Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D (Disconnect Except "D" orders associated with LNP Standalone.) and F (From) orders. (From is disconnect side of a move order when the customer moves to a new address).
- "L" Appointment coded orders (where the customer has requested a later than offered interval)
- Orders with CLEC/Subscriber caused delays or CLEC/Subscriber requested due date changes.

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI). Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = $(c \div d)$

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Total Number of Service Requests Completed in "X" minutes/hours
- f = Total Number of Service Requests Received in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- BellSouth Aggregate
- Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of <10 line/circuits; > 10 line/circuits (except trunks)
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, > 30 Days. The interval breakout is: 0-5=0-4.99, 5-10=5-9.99, 10-15=10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, $\ge 30 = 30$ and greater.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report Month Interval for FOC CLEC Company Name (OCN) Order Number (PON) Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file 	 Report Month BellSouth Order Number Order Submission Date & Time Order Completion Date & Time Service Type Geographic Scope

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	Diagnostic
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
2W Analog Loop Design	
2W Analog Loop Non-Design	
UNE Switch ports	
• UNE Digital Loops < DS1	
• UNE Digital Loops ≥ DS1	
• UNE Loop + Port Combinations	
UNE Combo Other	
• UNE xDSL (HDSL, ADSL and UCL)	
• UNE ISDN	
UNE Line Sharing	
 Local Transport (Unbundled Interoffice Trans port) 	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



P-10: LNP-Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that CLECs can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for total misses and End User Misses.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates. Missed Appointments caused by end-user reasons will be included and reported in a separate category. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date, which means there cannot be a cutoff time for commitments as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation

LNP Percent Missed Installation Appointments = $(a \div b) \times 100$

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Geographic Scope
 - State/Region
- Report in Categories of <10 lines/circuits ≥ 10 lines/circuits (except trunks)
- · Dispatch/No Dispatch

Report explanation: Total Missed Appointments is the total percent of orders missed either by BellSouth or the CLEC end user. End User MA represents the percentage of orders missed by the CLEC end user. The difference between End User Missed Appointments and Total Missed Appointments is the result of BellSouth caused misses.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report month CLEC Order Number and PON (PON) Committed Due Date (DD) Completion Date (CMPLTN DD) Status Type Status Notice Date Standard Order Activity Geographic Scope 	Not Applicable
Note: Code in parentheses is the corresponding header found in the raw data file.	



Tennessee Performance Metrics

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
• LNP	Retail Residence & Business (POTS)

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• LNP	Retail Residence & Business (POTS)

(4) **BELLSO**UTH®

P-11: LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution

Definition

Disconnect Timeliness is defined as the interval between the time ESI Number Manager receives the valid 'Number Ported' message from NPAC (signifying the CLEC 'Activate') until the time the Disconnect is completed in the Central Office switch. This interval effectively measures BellSouth responsiveness by isolating it from impacts that are caused by CLEC related activities.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable.

Business Rules

The Disconnect Timeliness interval is determined for each number ported associated with a disconnect service order processed on an LSR during the reporting period. The Disconnect Timeliness interval is the elapsed time from when BellSouth receives a valid 'Number Ported' message in ESI Number Manager (signifying the CLEC 'Activate') for each telephone number ported until each number on the service order is disconnected in the Central Office switch. Elapsed time for each ported number is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the total number of selected telephone numbers disconnected in the reporting period.

Calculation

Disconnect Timeliness Interval = (a - b)

- a = Completion Date and Time in Central Office switch for each number on disconnect order
- b = Valid 'Number Ported' message received date & time

Average Disconnect Timeliness Interval = $(c \div d)$

- c = Sum of all Disconnect Timeliness Intervals
- d = Total Number of disconnected numbers completed in reporting period

Disconnect Timeliness Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Disconnected numbers completed in "X" days
- f = Total disconnect numbers completed in reporting period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Geographic Scope
- State, Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Order Number	Not Applicable
Telephone Number / Circuit Number	
Committed Due Date	
Receipt Date / Time (ESI Number Manager)	
Date/Time of Recent Change Notice	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation:	SQM Retail Analog/Benchmark:
• LNP	• 95% within 24 hours

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• LNP	• 95% within 24 hours

🕮 **BELL**SOUTH

P-12: LNP-Total Service Order Cycle Time (TSOCT)

Definition

Total Service Order Cycle Time measures the interval from receipt of a valid service order request to the completion of the final service order associated with that service request.

Exclusions

- · Canceled Service Orders
- · Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable
- "L" appointment coded orders (indicating the customer has requested a later than offered interval)
- "S" missed appointment coded orders (indicating subscriber missed appointments), except for "SP" codes (indicating subscriber prior due date requested). This would include "S" codes assigned to subsequent due date changes.

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI). Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = $(c \div d)$

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Total Number of Service Orders Completed in "X" minutes/hours
- f = Total Number of Service Orders Received in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of <10 line/circuits; > 10 line/circuits (except trunks)
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, > 30 Days. The interval breakout is: 0-5=0-4.99, 5-10=5-9.99, 10-15=10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, > 30 = 30 and greater.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report Month Interval for FOC CLEC Company Name (OCN) Order Number (PON) Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope 	Not Applicable
Note: Code in parentheses is the corresponding header found in the raw data file	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• LNP	Diagnostic

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 4: Maintenance & Repair

M&R-1: Missed Repair Appointments

Definition

The percent of trouble reports not cleared by the committed date and time.

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

The negotiated commitment date and time is established when the repair report is received. The cleared time is the date and time that BellSouth personnel clear the trouble and closes the trouble report in his/her Computer Access Terminal (CAT) or workstation. If this is after the Commitment time, the report is flagged as a "Missed Commitment" or a missed repair appointment. When the data for this measure is collected for BellSouth and a CLEC, it can be used to compare the percentage of the time repair appointments are missed due to BellSouth reasons. (No access reports are not part of this measure because they are not a missed appointment.)

Note: Appointment intervals vary with force availability in the POTS environment. Specials and Trunk intervals are standard interval appointments of no greater than 24 hours. Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

Percentage of Missed Repair Appointments = $(a \div b) \times 100$

- a = Count of Customer Troubles Not Cleared by the Quoted Commitment Date and Time
- b = Total Trouble reports closed in Reporting Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
CLEC Company Name	BellSouth Company Code
Submission Date & Time (TICKET_ID)	Submission Date & Time
Completion Date (CMPLTN_DT)	Completion Date
• Service Type (CLASS_SVC_DESC)	Service Type
 Disposition and Cause (CAUSE_CD & CAUSE_DESC) 	Disposition and Cause (Non-Design /Non-Special Only)
Geographic Scope	Trouble Code (Design and Trunking Services)
Note : Code in parentheses is the corresponding header found in the raw data file.	Geographic Scope



SQM Disaggregation - Retail Analog/Benchmark

SQM Level of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of switch- based feature troubles
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-2: Customer Trouble Report Rate

Definition

Initial and repeated customer direct or referred troubles reported within a calendar month per 100 lines/circuits in service.

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.
- LMOS Code 7 (Test OK), Code 8 (Found OK In), Code 9 (Found OK Out)
- WFA No Trouble Found (NTF)

Business Rules

Customer Trouble Report Rate is computed by accumulating the number of maintenance initial and repeated trouble reports during the reporting period. The resulting number of trouble reports are divided by the total "number of service" lines, ports or combination that exist for the CLECs and BellSouth respectively at the end of the report month.

Calculation

Customer Trouble Report Rate = (a ÷ b) X 100

- a = Count of Initial and Repeated Trouble Reports closed in the Current Period
- b = Number of Service Access Lines in service at End of the Report Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) # Service Access Lines in Service at the end of period Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file.	 Report month BellSouth Company Code Ticket Submission Date & Time Ticket Completion Date Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) # Service Access Lines in Service at the end of period Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of switch- based feature troubles)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-3: Maintenance Average Duration

M&R-3: Maintenance Average Duration

Definition

The Average duration of Customer Trouble Reports from the receipt of the Customer Trouble Report to the time the trouble report is cleared.

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

For Average Duration the clock starts on the date and time of the receipt of a correct repair request. The clock stops on the date and time the service is restored and the BellSouth or CLEC customer is notified (when the technician completes the trouble ticket on his/her CAT or work systems).

Calculation

Maintenance Duration = (a - b)

- a = Date and Time of Service Restoration
- b = Date and Time Trouble Ticket was Opened

Average Maintenance Duration = $(c \div d)$

- c = Total of all maintenance durations in the reporting period
- d = Total Closed Troubles in the reporting period

Report Structure

- Dispatch / Non-Dispatch
- · CLEC Specific
- CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience:	Relating to BellSouth Performance:
Report month Total Tickets (LINE_NBR) CLEC Company Name Ticket Submission Data & Time (TICKET_ID)	 Report month Total Tickets BellSouth Company Code Ticket Submission Date
 Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) Geographic Scope 	 Ticket Submission Date Ticket Submission Time Ticket Completion Date Ticket Completion Time Total Duration Time
Note: Code in parentheses is the corresponding header found in the raw data file.	 Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail business



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of switch- based feature troubles)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail



M&R-4: Percent Repeat Troubles within 30 Days

Definition

Closed trouble reports on the same line/circuit as a previous trouble report received within 30 calendar days as a percent of total troubles closed reported

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

Includes Customer trouble reports received within 30 days of an original Customer trouble report

Calculation

Percent Repeat Troubles within 30 Days = $(a \div b) \times 100$

- a = Count of closed Customer Troubles where more than one trouble report was logged for the same service line within a continuous 30 days
- b = Total Trouble Reports Closed in Reporting Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report month Total Tickets (LINE_NBR) CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Total and Percent Repeat Trouble Reports within 30 Days (TOT_REPEAT) Service Type Disposition and Cause (CAUSE_CD & CAUSE_DESC) Geographic Scope 	 Report month Total Tickets BellSouth Company Code Ticket Submission Date Ticket Submission Time Ticket Completion Date Ticket Completion Time Total and Percent Repeat Trouble Reports within 30 Days Service Type Disposition and Cause (Non-Design /Non-Special Only)
Note : Code in parentheses is the corresponding header found in the raw data file.	Trouble Code (Design and Trunking Services) Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of switch- based feature troubles)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL provided to Retail
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-5: Out of Service (OOS) > 24 Hours

Definition

For Out of Service Troubles (no dial tone, cannot be called or cannot call out) the percentage of Total OOS Troubles cleared in excess of 24 hours. (All design services are considered to be out of service).

Exclusions

- Trouble Reports canceled at the CLEC request
- BellSouth Trouble Reports associated with administrative service
- Customer Provided Equipment (CPE) Troubles or CLEC Equipment Troubles.

Business Rules

Customer Trouble reports that are out of service and cleared in excess of 24 hours. The clock begins when the trouble report is created in LMOS/WFA and the trouble is counted if the elapsed time exceeds 24 hours.

Calculation

Out of Service (OOS) > 24 hours = $(a \div b) \times 100$

- a = Total Cleared Troubles OOS > 24 Hours
- b = Total OOS Troubles in Reporting Period

Report Structure

- Dispatch / Non Dispatch
- CLEC Specific
- BellSouth Aggregate
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Total Tickets	Total Tickets
CLEC Company Name	BellSouth Company Code
 Ticket Submission Date & Time (TICKET_ID) 	Ticket Submission Date
 Ticket Completion Date (CMPLTN_DT 	Ticket Submission time
 Percentage of Customer Troubles out of 	Ticket Completion Date
• Service > 24 Hours (OOS>24_FLAG)	Ticket Completion Time
 Service type (CLASS_SVC_DESC) 	• Percent of Customer Troubles out of Service > 24 Hours
 Disposition and Cause (CAUSE CD & CAUSE-DESC) 	Service type
Geographic Scope	Disposition and Cause (Non-Design/Non-Special only)
Note: Code in parentheses is the corresponding header found in the raw data file.	Trouble Code (Design and Trunking Services)Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of switch- based feature troubles)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL provided to Retail
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-6: Average Answer Time – Repair Centers

M&R-6: Average Answer Time - Repair Centers

Definition

This measures the average time a customer is in queue when calling a BellSouth Repair Center.

Exclusions

None

Business Rules

The clock starts when a CLEC Representative or BellSouth customer makes a choice on the Repair Center's menu and is put in queue for the next repair attendant. The clock stops when the repair attendant answers the call (abandoned calls are not included).

Note: The Total Column is a combined BellSouth Residence and Business number.

Calculation

Answer Time for BellSouth Repair Centers = (a - b)

- a = Time BellSouth Repair Attendant Answers Call
- b = Time of entry into queue after ACD Selection

Average Answer Time for BellSouth Repair Centers = $(c \div d)$

- c = Sum of all Answer Times
- d = Total number of calls by reporting period

Report Structure

- CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
CLEC Average Answer Time	BellSouth Average Answer Time

SQM Disaggregation - Analog / Benchmark

SQM Level of Disaggregation	Retail Analog / Benchmark
Region. CLEC/BellSouth Service Centers and BellSouth Repair Centers are regional.	For CLEC, Average Answer Times in UNE Center and BRMC are comparable to the Average Answer Times in the BellSouth Repair Centers.

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-7: Mean Time To Notify CLEC of Network Outages

M&R-7: Mean Time To Notify CLEC of Network Outages

Definition

BellSouth will inform the CLEC of any Network outages (key customer accounts)

Exclusions

None

Business Rules

The time it takes for the BellSouth Network Reliability Center (NRC) to notify the CLEC and BellSouth of a customer impacting network incident in equipment that may be utilized by the CLEC. When the BellSouth NRC becomes aware of a network incident, the CLEC and BellSouth will be notified electronically. The notification time for each outage will be measured in minutes and divided by the number of outages for the reporting period. The CLECs will be notified the same way and at the same time as BellSouth Retail. These are broadcast messages. It is up to those receiving the message to determine if they have customers affected by the incident.

Calculation

Time to Notify CLEC = (a - b)

- a = Date and Time BellSouth Notified CLEC
- b = Date and time BellSouth detected network incident

Mean Time to Notify CLEC = $(c \div d)$

- c = Sum of all Times to Notify CLEC
- d = Count of Network Incidents

Report Structure

- · BellSouth Aggregate
- · CLEC Aggregate
- · CLEC Specific

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Major Network events	Major Network events
Date/Time of Incident	Date/Time of Incident
Date/Time of Notification	Date/Time of Notification

SQM Disaggregation - Analog / Benchmark

SQM Level of Disaggregation	Retail Analog / Benchmark
BellSouth AggregateCLEC AggregateCLEC Specific	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 5: Billing

B-1: Invoice Accuracy

Definition

This measure provides the percentage of accuracy of the billing invoices rendered to CLECs during the current month.

Exclusions

- Adjustments not related to billing errors (e.g., credits for service outage, special promotion credits, adjustments to satisfy the customer)
- · Test Accounts

Business Rules

The accuracy of billing invoices delivered by BellSouth to the CLEC must enable them to provide a degree of billing accuracy comparative to BellSouth bills rendered to retail customers of BellSouth. CLECs request adjustments on bills determined to be incorrect. The BellSouth Billing verification process includes manually analyzing a sample of local bills from each bill period. The bill verification process draws from a mix of different customer billing options and types of service. An end-to-end auditing process is performed for new products and services. Internal measurements and controls are maintained on all billing processes.

Calculation

Invoice Accuracy = $[(a - b) \div a] \times 100$

- a = Absolute Value of Total Billed Revenues during current month
- b = Absolute Value of Billing Related Adjustments during current month

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Invaries Time	Report month Patril Type
• Invoice Type - UNE	Retail Type CRIS
- Resale	- CABS
- Interconnection	Total Billed Revenue
Total Billed Revenue	Billing Related Adjustments
Billing Related Adjustments	



SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product / Invoice Type	CLEC Invoice Accuracy is comparable to BellSouth Invoice
- Resale	Accuracy
- UNE	
- Interconnection	

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC StateBellSouth State	Parity with Retail



B2: Mean Time to Deliver Invoices

Definition

Bill Distribution is calculated as follows: CRIS BILLS-The number of workdays is reported for CRIS bills. This is calculated by counting the Bill Period date as the first work day. Weekends and holidays are excluded when counting workdays. J/N Bills are counted in the CRIS work day category for the purposes of the measurement since their billing account number (Q account) is provided from the CRIS system.

CABS BILLS-The number of calendar days is reported for CABS bills. This is calculated by counting the day following the Bill Period date as the first calendar day. Weekends and holidays are included when counting the calendar days.

Exclusions

Any invoices rejected due to formatting or content errors.

Business Rules

This report measures the mean interval for timeliness of billing records delivered to CLECs in an agreed upon format. CRIS-based invoices are measured in business days, and CABS-based invoices in calendar days.

Calculation

Invoice Timeliness = (a - b)

- a = Invoice Transmission Date
- b = Close Date of Scheduled Bill Cycle

Mean Time To Deliver Invoices = $(c \div d)$

- c = Sum of all Invoice Timeliness intervals
- d = Count of Invoices Transmitted in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
Invoice Type	Invoice Type
- UNE	- CRIS
- Resale	- CABS
- Interconnection	Invoice Transmission Count
Invoice Transmission Count	Date of Scheduled Bill Close
Date of Scheduled Bill Close	



SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product / Invoice Type Resale UNE Interconnection	 CRIS-based invoices will be released for delivery within six (6) business days. CABS-based invoices will be released for delivery within eight (8) calendar days. CLEC Average Delivery Intervals for both CRIS and CABS Invoices are comparable to BellSouth Average delivery for both systems.

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• CLEC State - CRIS	Parity with Retail
- CABS - BellSouth Region	

BELLSOUTH

B3: Usage Data Delivery Accuracy

Definition

This measurement captures the percentage of recorded usage that is delivered error free and in an acceptable format to the appropriate Competitive Local Exchange Carrier (CLEC). These percentages will provide the necessary data for use as a comparative measurement for BellSouth performance. This measurement captures Data Delivery Accuracy rather than the accuracy of the individual usage recording.

Exclusions

None

Business Rules

The accuracy of the data delivery of usage records delivered by BellSouth to the CLEC must enable them to provide a degree of accuracy comparative to BellSouth bills rendered to their retail customers. If errors are detected in the delivery process, they are investigated, evaluated and documented. Errors are corrected and the data retransmitted to the CLEC.

Calculation

Usage Data Delivery Accuracy = $(a - b) \div a \times 100$

- a = Total number of usage data packs sent during current month
- b = Total number of usage data packs requiring retransmission during current month

Report Structure

- CLEC Specific
- CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Record Type BellSouth Recorded Non-BellSouth Recorded 	Report month Record Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	CLEC Usage Data Delivery Accuracy is comparable to BellSouth Usage Data Delivery Accuracy

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC StateBellSouth Region	Parity with Retail

(A) **BELL**SOUTH

B4: Usage Data Delivery Completeness

Definition

This measurement provides percentage of complete and accurately recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is processed and transmitted to the CLEC within thirty (30) days of the message recording date. A parity measure is also provided showing completeness of BellSouth messages processed and transmitted via CMDS. BellSouth delivers its own retail usage from recording location to billing location via CMDS as well as delivering billing data to other companies. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of these measurements is to demonstrate the level of quality of usage data delivered to the appropriate CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Completeness = $(a \div b) \times 100$

- a = Total number of Recorded usage records delivered during current month that are within thirty (30) days of the message recording
- b = Total number of Recorded usage records delivered during the current month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Record Type BellSouth Recorded Non-BellSouth Recorded	Report month Record Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	CLEC Usage Data Delivery Completeness is comparable to BellSouth Usage Data Delivery Completeness

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLSOUTH

B5: Usage Data Delivery Timeliness

Definition

This measurement provides a percentage of recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is delivered to the appropriate CLEC within six (6) calendar days from the receipt of the initial recording. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the level of timeliness for processing and transmission of usage data delivered to the appropriate CLEC. The usage data will be mechanically transmitted or mailed to the CLEC data processing center once daily. The Timeliness interval of usage recorded by other companies is measured from the date BellSouth receives the records to the date BellSouth distributes to the CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Timeliness Current month = $(a \div b) \times 100$

- a = Total number of usage records sent within six (6) calendar days from initial recording/receipt
- b = Total number of usage records sent

Report Structure

- · CLEC Aggregate
- CLEC Specific
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Record Type BellSouth Recorded Non-BellSouth Recorded 	Report Monthly Record Type

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Region	CLEC Usage Data Delivery Timeliness is comparable to BellSouth Usage Data Delivery Timeliness

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLSOUTH

B6: Mean Time to Deliver Usage

Definition

This measurement provides the average time it takes to deliver Usage Records to a CLEC. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the average number of days it takes BellSouth to deliver Usage data to the appropriate CLEC. Usage data is mechanically transmitted or mailed to the CLEC data processing center once daily. Method of delivery is at the option of the CLEC.

Calculation

Mean Time to Deliver Usage = $(a \times b) \div c$

- a = Volume of Records Delivered
- b = Estimated number of days to deliver
- c = Total Record Volume Delivered

Note: Any usage record falling in the 30+ day interval will be added using an average figure of 31.5 days.

Report Structure

- · CLEC Aggregate
- · CLEC Specific
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Record Type BellSouth Recorded	Report Monthly Record Type
- Non-BellSouth Recorded	

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	Mean Time to Deliver Usage to CLEC is comparable to Mean Time to Deliver Usage to BellSouth

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLSOUTH

B7: Recurring Charge Completeness

Definition

This measure captures percentage of fractional recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Recurring Charge Completeness = $(a \div b) \times 100$

- a = Count of fractional recurring charges that are on the correct bill¹
- b = Total count of fractional recurring charges that are on the correct bill

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	Report month
Invoice type	Retail Analog
Total recurring charges billed	Total recurring charges billed
Total billed on time	Total billed on time

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product/Invoice Type	
Resale	• Parity
• UNE	Benchmark 90%
Interconnection	Benchmark 90%

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

¹Correct bill = next available bill

(A) **BELL**SOUTH

B8: Non-Recurring Charge Completeness

Definition

This measure captures percentage of non-recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the non-recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Non-Recurring Charge Completeness = $(a \div b) \times 100$

- a = Count of non-recurring charges that are on the correct bill¹
- b = Total count of non-recurring charges that are on the correct bill

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	Report month
Invoice type	Retail Analog
Total non-recurring charges billed	Total non-recurring charges billed
Total billed on time	Total billed on time

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
Product/Invoice Type	
Resale	• Parity
• UNE	Benchmark 90%
Interconnection	Benchmark 90%

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

¹Correct bill = next available bill



Section 6: Operator Services And Directory Assistance

OS-1: Speed to Answer Performance/Average Speed to Answer - Toll

Definition

Measurement of the average time in seconds calls wait before answered by a toll operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer - Toll = $a \div b$

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- · Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- · Average Speed of Answer

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design



SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



OS-2: Speed to Answer Performance/Percent Answered with "X" Seconds – Toll

Definition

Measurement of the percent of toll calls that are answered in less than ten seconds

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for toll is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- · Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	Retail Analog/Benchmark:
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



DA-1: Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA)

Definition

Measurement of the average time in seconds calls wait before answered by a DA operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA) = $a \div b$

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (DA)
- Average Speed of Answer

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregat	ion Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds – Directory Assistance (DA)

Definition

Measurement of the percent of DA calls that are answered in less than twelve seconds.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for DA is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- · Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.
- Month
- Call Type (DA)
- · Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 7: Database Update Information

D-1: Average Database Update Interval

Definition

This report measures the interval from receipt of the database change request to the completion of the update to the database for Line Information Database (LIDB), Directory Assistance and Directory Listings.

Exclusions

- Updates Canceled by the CLEC
- Initial update when supplemented by CLEC
- · BellSouth updates associated with internal or administrative use of local services.

Business Rules

The interval for this measure begins with the date and time stamp when a service order is completed and the completion notice is released to all systems to be updated with the order information including Directory Assistance, Directory Listings, and Line Information Database (LIDB). The end time stamp is the date and time of completion of updates to the system.

For BellSouth Results:

The BellSouth computation is identical to that for the CLEC with the clarifications noted below.

Other Clarifications and Qualification:

- For LIDB, the elapsed time for a BellSouth update is measured from the point in time when the BellSouth file maintenance process makes the LIDB update information available until the date and time reported by BellSouth that database updates are completed.
- Results for the CLECs are captured and reported at the update level by Reporting Dimension (see below).
- The Completion Date is the date upon which BellSouth issues the Update Completion Notice to the CLEC.
- If the CLEC initiates a supplement to the originally submitted update and the supplement reflects changes in customer requirements (rather than responding to BellSouth initiated changes), then the update submission date and time will be the date and time of BellSouth receipt of a syntactically correct update supplement. Update activities responding to BellSouth initiated changes will not result in changes to the update submission date and time used for the purposes of computing the update completion interval.
- Elapsed time is measured in hours and hundredths of hours rounded to the nearest tenth of an hour.
- Because this should be a highly automated process, the accumulation of elapsed time continues through off-schedule, weekends and holidays; however, scheduled maintenance windows are excluded.

Calculation

Update Interval = (a - b)

- a = Completion Date & Time of Database Update
- b = Submission Date and Time of Database Change

Average Update Interval = $(c \div d)$

- c = Sum of all Update Intervals
- d = Total Number of Updates Completed During Reporting Period



Report Structure

- CLEC Specific (Under development)
- CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
• (Under Development)	(Under Development)

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation:	Retail Analog/Benchmark:
Database Type • LIDB	Parity by Design
Directory Listings	
Directory Assistance	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



D-2: Percent Database Update Accuracy

Definition

This report measures the accuracy of database updates by BellSouth for Line Information Database (LIDB) Directory Assistance and Directory Listings using a statistically valid sample of LSRs/Orders in a manual review. This manual review is not conducted on BellSouth Retail Orders.

Exclusions

- Updates canceled by the CLEC
- Initial update when supplemented by CLEC
- · CLEC orders that had CLEC errors
- BellSouth updates associated with internal or administrative use of local services.

Business Rules

For each update completed during the reporting period, the original update that the CLEC sent to BellSouth is compared to the database following completion of the update by BellSouth. An update is "completed without error" if the database completely and accurately reflects the activity specified on the original and supplemental update (e.g., orders) submitted by the CLEC. Each database (e.g., LIDB, Directory Assistance and Directory Listings) should be separately tracked and reported.

A statistically valid sample of CLEC Orders will be pulled each month. The sample will be used to test the accuracy of the database update process. This is a manual process.

Calculation

Percent Update Accuracy = $(a \div b) \times 100$

- a = Number of Updates Completed Without Error
- b = Number Updates Completed

Report Structure

- · CLEC Aggregate
- CLEC Specific (not available in this report)
- BellSouth Aggregate (not available in this report)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number (so_nbr) and PON (PON) Local Service Request (LSR) Order Submission Date Number of Orders Reviewed 	Not Applicable
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark:
Database Type	• 95% Accurate
• LIDB	
Directory Database	



SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date

Definition

Measurement of the percent of NXX(s) and Location Routing Numbers LRN(s) loaded and tested in new end office and/or tandem switches by the Local Exchange Routing Guide (LERG) effective date when facilities are in place. BellSouth has a single provisioning process for both NXX(s) and LRN(s). In this measure BellSouth will identify whether or not a particular NXX has been flagged as LNP capable (set triggers for dips) by the LERG effective date.

An LRN is assigned by the owner of the switch and is placed into the software translations for every switch to be used as an administrative pointer to route NXX(s) in LNP capable switches. The LRN is a result of Local Number Porting and is housed in a national database provided by the Number Portability Administration Center (NPAC). The switch owner is responsible for notifying NPAC and requesting the effective date that will be reflected in the LERG. The national database downloads routing tables into BellSouth's Service Control Point (SCP) regional databases, which are queried by switches when routing ported numbers.

The basic NXX routing process includes the addition of all NXX(s) in the response translations. This addition to response translations is what supports LRN routing. Routing instructions for all NXX(s), including LRN(s), are received from the Advance Routing & Trunking System (ARTS) and all routing, including response, is established based on the information contained in the Translation Work Instructions (TWINs) document.

Exclusions

- · Activation requests where the CLEC's interconnection arrangements and facilities are not in place by the LERG effective date.
- · Expedite requests

Business Rules

Data for the initial NXX(s) and LRN(s) in a local calling area will be based on the LERG effective date or completion of the initial interconnection trunk group(s), whichever is longer. Data for additional NXX(s) in the local calling area will be based on the LERG effective date. The LERG effective date is loaded into the system at the request of the CLEC. It is contingent upon the CLEC to engineer, order, and install interconnection arrangements and facilities prior to that date.

The total Count of NXX(s) and LRN(s) that were scheduled to be loaded and those that were loaded by the LERG effective date in BellSouth switches will be captured in the Work Force Administration -Dispatch In database.

Calculation

Percent NXXs/LRNs Loaded and Tested Prior to the LERG Effective Date = $(a \div b) \times 100$

- a = Count of NXXs and LRNs loaded by the LERG effective date
- b = Total NXXs and LRNs to be scheduled and loaded by the LERG effective date

Report Structure

- CLEC Specific
- · CLEC Aggregate
- BellSouth (Not Applicable)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Company Name	Not Applicable
 Company Code 	
• NPA/NXX	
LERG Effective Date	
Loaded Date	



SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Geographic scope Region	100% by LERG effective date

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 8: E911

E-1: Timeliness

Definition

Measures the percent of batch orders for E911 database updates (to CLEC resale and BellSouth retail records) processed successfully within a 24-hour period.

Exclusions

- · Any resale order canceled by a CLEC
- · Facilities-based CLEC orders

Business Rules

The 24-hour processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing batch orders extracted from the BellSouth Service Order Control System (SOCS). Processing stops when SCC loads the individual records to the E911 database. The E911 database includes updates to the Automatic Location Identification (ALI) database. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Timeliness = $(a \div b) \times 100$

- a = Number of batch orders processed within 24 hours
- b = Total number of batch orders submitted

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



E-2: Accuracy

Definition

Measures the percent of E911 telephone number (TN) record updates (to CLEC resale and BellSouth retail records) processed successfully for E911 (including the Automatic Location Identification (ALI) database).

Exclusions

- Any resale order canceled by a CLEC
- · Facilities-based CLEC orders

Business Rules

Accuracy is based on the number of records processed without error at the conclusion of the processing cycle. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing telephone number (TN) records extracted from BellSouth's Service Order Control System (SOCS). The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Accuracy = $(a \div b) \times 100$

- a = Number of record individual updates processed with no errors
- b = Total number of individual record updates

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark	
Not Applicable	Not Applicable	



E-3: Mean Interval

Definition

Measures the mean interval processing of E911 batch orders (to update CLEC resale and BellSouth retail records) including processing against the Automatic Location Identification (ALI) database.

Exclusions

- · Any resale order canceled by a CLEC
- · Facilities-based CLEC orders

Business Rules

The processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Data is posted is 4-hour increments up to and beyond 24 hours. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Interval = (a - b)

- a = Date and time of batch order completion
- b = Date and time of batch order submission

E911 Mean Interval = $(c \div d)$

- c = Sum of all E911 Intervals
- d = Number of batch orders completed

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- · Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation		Retail Analog/Benchmark	
	• None	Parity by Design	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark	
Not Applicable	Not Applicable	



Section 9: Trunk Group Performance

TGP-1: Trunk Group Performance-Aggregate

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches.
- · Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

		Point A	Point B
Categor	y 1:	BellSouth End Office	BellSouth Access Tandem
Categor	ry 3:	BellSouth End Office	CLEC Switch
Categor	ry 4:	BellSouth Local Tandem	CLEC Switch
Categor	ry 5:	BellSouth Access Tandem	CLEC Switch
Categor	y 10:	BellSouth End Office	BellSouth Local Tandem
Categor	ry 16:	BellSouth Tandem	BellSouth Tandem

BellSouth Affecting Categories:

	Point A	Point B
Category 9:	BellSouth End Office	BellSouth End Office



Calculation

Monthly Average Blocking:

- For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.
- The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:

- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- · CLEC Aggregate
- · BellSouth Aggregate
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience	
Report Month	Report Month	
Total Trunk Groups	Total Trunk Groups	
Number of Trunk Groups by CLEC	Aggregate Hourly blocking per trunk group	
Hourly blocking per trunk group	Hourly usage per trunk group	
Hourly usage per trunk group	Hourly call attempts per trunk group	
Hourly call attempts per trunk group		

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
CLEC aggregate BellSouth aggregate	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BellSouth

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark:	
CLEC aggregate BellSouth aggregate	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1,3,4,5,10,16 for CLECs and 9 for BellSouth	

TGP-2: Trunk Group Performance-CLEC Specific

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- · Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- · Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches.
- Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

• This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

	Point A	Point B
Category 1:	BellSouth End Office	BellSouth Access Tandem
Category 3:	BellSouth End Office	CLEC Switch
Category 4:	BellSouth Local Tandem	CLEC Switch
Category 5:	BellSouth Access Tandem	CLEC Switch
Category 10:	BellSouth End Office	BellSouth Local Tandem
Category 16:	BellSouth Tandem	BellSouth Tandem

BellSouth Affecting Categories:

	Point A	Point B
Category 9:	BellSouth End Office	BellSouth End Office

Calculation:

Monthly Average Blocking:

- For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.
- The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:



- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- CLEC Specific
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Total Trunk Groups	Total Trunk Groups
Number of Trunk Groups by CLEC	Aggregate Hourly blocking per trunk group
Hourly blocking per trunk group	Hourly usage per trunk group
Hourly usage per trunk group	Hourly call attempts per trunk group
Hourly call attempts per trunk group	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
CLEC trunk group	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BellSouth

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark:
CLEC trunk group BellSouth trunk group	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BellSouth



Section 10: Collocation

C-1: Collocation Average Response Time

Definition

Measures the average time (counted in calendar days) from the receipt of a complete and accurate collocation application (including receipt of application fee if required) to the date BellSouth returns a response electronically or in writing. Within 10 calendar days after having received a bona fide application for physical collocation, BellSouth must respond as to whether space is available or not.

Exclusions

Any application canceled by the CLEC

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate collocation application accompanied by the appropriate application fee if required. The clock stops on the date that BellSouth returns a response. The clock will restart upon receipt of changes to the original application request.

Calculation

Response Time = (a - b)

- a = Request Response Date
- b = Request Submission Date

Average Response Time = $(c \div d)$

- c = Sum of all Response Times
- d = Count of Responses Returned within Reporting Period

Report Structure

- · Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

Level of Disaggregation	Retail Analog/Benchmark
• State	Virtual - 20 Calendar Days
Virtual-Initial	Physical Caged - 30 Calendar Days
Virtual-Augment	Physical Cageless - 30 Calendar Days
Physical Caged-Initial	
Physical Caged-Augment	
Physical-Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure			
No	Tier I		
	Tier II		

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELL**SOUTH

C-2: Collocation Average Arrangement Time

Definition

Measures the average time (counted in calendar days) from receipt of a complete and accurate Bona Fide firm order (including receipt of appropriate fee if required) to the date BellSouth completes the collocation arrangement and notifies the CLEC.

Exclusions

Any Bona Fide firm order canceled by the CLEC

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate Bone Fide firm order accompanied by the appropriate fee. The clock stops on the date that BellSouth completes the collocation arrangement and notifies the CLEC.

Calculation

Arrangement Time = (a - b)

- a = Date Collocation Arrangement is Complete
- b = Date Order for Collocation Arrangement Submitted

Average Arrangement Time = $(c \div d)$

- c = Sum of all Arrangement Times
- d = Total Number of Collocation Arrangements Completed during Reporting Period.

Report Structure

- Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

SQM Disaggregation - Retail Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• State	Virtual - 50 Calendar Days (Ordinary)
Virtual-Initial	Virtual - 75 Calendar Days (Extraordinary)
Virtual-Augment	Physical Caged - 90 Calendar Days (Ordinary)
Physical Caged-Initial	Physical Caged - 130 Calendar Days (Extraordinary)
Physical Caged-Augment	Physical Cageless - 90 Calendar Days (Ordinary)
Physical Cageless-Initial	 Physical Cageless - 130 Calendar Days (Extraordinary)
Physical Cageless-Augment	

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



SEEM Disaggregation	SEEM Analog/Benchmark:
Not Applicable	Not Applicable

BELLSOUTH

C-3: Collocation Percent of Due Dates Missed

Definition

Measures the percent of missed due dates for both virtual and physical collocation arrangements.

Exclusions

Any Bona Fide firm order canceled by the CLEC

Business Rules

Percent Due Dates Missed is the percent of total collocation arrangements which BellSouth is unable to complete by end of the BellSouth committed due date. The clock starts on the date that BellSouth receives a complete and accurate Bona Fide firm order accompanied by the appropriate fee if required. The arrangement is considered a missed due date if it is not completed on or before the committed due date.

Calculation

% of Due Dates Missed = $(a \div b) \times 100$

- a = Number of Completed Orders that were not completed within BellSouth Committed Due Date during Reporting Period
- b = Number of Orders Completed in Reporting Period

Report Structure

- Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• State	• ≥ 95% on time
Virtual-Initial	
• Virtual-Augment	
Physical Caged-Initial	
Physical Caged-Augment	
Physical Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
All Collocation Arrangements	• $\geq 95\%$ on time.



Section 11: Change Management

CM-1: Timeliness of Change Management Notices

Definition

Measures whether CLECs receive required software release notices on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system software vendor changes. For example: a patch to fix a software problem.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process (CCP)

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Timeliness of Change Management Notices = $(a \div b) \times 100$

- a = Total number of Change Management Notifications Sent Within Required Time frames
- b = Total Number of Change Management Notifications Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
• Region	• 95% ≥ 30 days of Release

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X



SEEM Disaggregation	SEEM Analog/Benchmark
• Region	• 95% ≥ 30 days of Release

CM-2: Change Management Notice Average Delay Days

Definition

Measures the average delay days for change management system release notices sent outside the time frame set forth in the Change

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system vendor
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification due date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Change Management Notice Delay Days = (a - b)

- a = Date Notice Sent
- b = Date Notice Due

Change Management Notice Average Delay Days = $(c \div d)$

- c = Sum of all Change Management Notice Delay Days
- d = Total Number of Notices Sent Late

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- · Notice Date
- · Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	Retail Analog/Benchmark:
• Region	• 90% ≤ 8 Days

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

CM-3: Timeliness of Documents Associated with Change

Definition

Measures whether CLECs received requirements or business rule documentation on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process.

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Timeliness of Documents Associated with Change = $(a \div b) \times 100$

- a = Change Management Documentation Sent Within Required Time frames after Notices
- b = Total Number of Change Management Documentation Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	 95% ≥ 30 days if new features coding is required 95% > 5 days for documentation defects, corrections or
	clarifications

SEEM Measure

SEEM Measure		
Yes	Tier I	
	Tier II	X

SEEM Disaggregation	SEEM Analog/Benchmark
• Region	• $95\% \ge 30$ days of the change



CM-4: Change Management Documentation Average Delay Days

Definition

Measures the average delay days for requirements or business rule documentation sent outside the time frames set forth in the Change

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process.

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Change Management Documentation Delay Days = (a - b)

- a = Date Documentation Provided
- b = Date Documentation Due

Change Management Documentation Average Delay Days = $(c \div d)$

- c = Sum of all CM Documentation Delay Days
- d = Total Change Management Documents Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- · Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
Region	• 90% ≤ 8 Days

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	

CM-4: Change Management Documentation Average Delay Days

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

CM-5: Notification of CLEC Interface Outages

Definition

Measures the time it takes BellSouth to notify the CLEC of an outage of an interface.

Exclusions

None

Business Rules

This measure is designed to notify the CLEC of interface outages within 15 minutes of BellSouth's verification that an outage has taken place. This metric will be expressed as a percentage.

Calculation

Notification of CLEC Interface Outages = $(a \div b) \times 100$

- a = Number of Interface Outages where CLECS are notified within 15 minutes
- b = Total Number of Interface Outages

Report Structure

· CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Number of Interface Outages Number of Notifications ≤ 15 minutes 	Not Applicable

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• By interface type for all interfaces accessed by CLECs	• 97% in 15 Minutes

Interface	Applicable to
EDI	CLEC
CSOTS	CLEC
LENS	CLEC
TAG	CLEC
ECTA	CLEC
TAFI	CLEC/BellSouth

SEEM Measure

SEEM Measure		
No	Tier I	
	Tier II	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Appendix A: Reporting Scope

A-1: Standard Service Groupings

See individual reports in the body of the SQM.

A-2: Standard Service Order Activities

These are the generic BellSouth/CLEC service order activities which are included in the Pre-Ordering, Ordering, and Provisioning sections of this document. It is not meant to indicate specific reporting categories.

Service Order Activity Types

- Service Migrations Without Changes
- · Service Migrations With Changes
- Move and Change Activities
- Service Disconnects (Unless noted otherwise)
- · New Service Installations

Pre-Ordering Query Types

- Address
- Telephone Number
- · Appointment Scheduling
- Customer Service Record
- · Feature Availability
- · Service Inquiry

Maintenance Query Types:

TAFI - TAFI queries the systems below

- CRIS
- March
- Predictor
- LMOS
- DLR
- DLETHLMOSupd
- LNP
- NIW
- OSPCM
- SOCS

Report Levels

- CLEC RESH
- CLEC State
- · CLEC Region
- · Aggregate CLEC State



- Aggregate CLEC Region
- BellSouth State
- BellSouth Region



Appendix B: Glossary of Acronyms and Terms

Symbols used in calculations

- Σ A mathematical symbol representing the sum of a series of values following the symbol.
- A mathematical operator representing subtraction.
- + A mathematical operator representing addition.
- ÷ A mathematical operator representing division.
- < A mathematical symbol that indicates the metric on the left of the symbol is less than the metric on the right.
- ≤ A mathematical symbol that indicates the metric on the left of the symbol is less than or equal to the metric on the right.
- > A mathematical symbol that indicates the metric on the left of the symbol is greater than the metric on the right.
- > A mathematical symbol that indicates the metric on the left of the symbol is greater than or equal to the metric on the right.
- () Parentheses, used to group mathematical operations which are completed before operations outside the parentheses.

Α

ACD: Automatic Call Distributor - A service that provides status monitoring of agents in a call center and routes high volume incoming telephone calls to available agents while collecting management information on both callers and attendants.

Aggregate: Sum total of all items in like category, e.g. CLEC aggregate equals the sum total of all CLECs' data for a given reporting level

ALEC: Alternative Local Exchange Company = FL CLEC

ADSL: Asymmetrical Digital Subscriber Line

ASR: Access Service Request - A request for access service terminating delivery of carrier traffic into a Local Exchange Carrier's network.

ATLAS: Application for Telephone Number Load Administration System - The BellSouth Operations System used to administer the pool of available telephone numbers and to reserve selected numbers from the pool for use on pending service requests/service orders.

ATLASTN: ATLAS software contract for Telephone Number.

Auto Clarification: The number of LSRs that were electronically rejected from LESOG and electronically returned to the CLEC for correction.

В

BFR: Bona Fied Request



BILLING: The process and functions by which billing data is collected and by which account information is processed in order to render accurate and timely billing.

BOCRIS: Business Office Customer Record Information System (Front-end to the CRIS database.)

BRI: Basic Rate ISDN

BRC: Business Repair Center – The BellSouth Business Systems trouble receipt center which serves large business and CLEC customers.

BellSouth: BellSouth Telecommunications, Inc.

C

CABS: Carrier Access Billing System

CCC: Coordinated Customer Conversions

CCP: Change Control Process

Centrex: A business telephone service, offered by local exchange carriers, which is similar to a Private Branch Exchange (PBX) but the switching equipment is located in the telephone company Central Office (CO).

CKTID: A unique identifier for elements combined in a service configuration

CLEC: Competitive Local Exchange Carrier

CLP: Competitive Local Provider = NC CLEC

CM: Change Management

CMDS: Centralized Message Distribution System - Telcordia administered national system used to transfer specially formatted messages among companies.

COFFI: Central Office Feature File Interface - Provides information about USOCs and class of service. COFFI is a part of DOE/SONGS. It indicates all services available to a customer.

CRIS: Customer Record Information System - The BellSouth proprietary corporate database and billing system for non-access customers and services.

CRSACCTS: CRIS software contract for CSR information

CRSG: Complex Resale Support Group

C-SOTS: CLEC Service Order Tracking System

CSR: Customer Service Record

CTTG: Common Transport Trunk Group - Final trunk groups between BellSouth & Independent end offices and the BellSouth access tandems.

D

DA: Directory Assistance



DESIGN: Design Service is defined as any Special or Plain Old Telephone Service Order which requires BellSouth Design Engineering Activities.

DISPOSITION & CAUSE: Types of trouble conditions, e.g. No Trouble Found, Central Office Equipment, Customer Premises Equipment, etc.

DLETH: Display Lengthy Trouble History - A history report that gives all activity on a line record for trouble reports in LMOS.

DLR: Detail Line Record - All the basic information maintained on a line record in LMOS, e.g. name, address, facilities, features etc.

DS-0: The worldwide standard speed for one digital voice signal (64000 bps).

DS-1: 24 DS-0s (1.544Mb/sec., i.e. carrier systems)

DOE: Direct Order Entry System - An internal BellSouth service order entry system used by BellSouth Service Representatives to input business service orders in BellSouth format.

DSAP: DOE (Direct Order Entry) Support Application - The BellSouth Operations System which assists a Service Representative or similar carrier agent in negotiating service provisioning commitments for non-designed services and Unbundled Network Elements.

DSAPDDI: DSAP software contract for schedule information.

DSL: Digital Subscriber Line

DUI: Database Update Information

Ε

E911: Provides callers access to the applicable emergency services bureau by dialing a 3-digit universal telephone number.

EDI: Electronic Data Interchange - The computer-to-computer exchange of inter and/or intra-company business documents in a public standard format.

ESSX: BellSouth Centrex Service

F

Fatal Reject: The number of LSRs that were electronically rejected from LEO, which checks to see of the LSR has all the required fields correctly populated.

Flow-Through: In the context of this document, LSRs submitted electronically via the CLEC mechanized ordering process that flow through to the BellSouth OSS without manual or human intervention.

FOC: Firm Order Confirmation - A notification returned to the CLEC confirming that the LSR has been received and accepted, including the specified commitment date.

FX: Foreign Exchange



G

Н

HAL: "Hands Off" Assignment Logic - Front end access and error resolution logic used in interfacing BellSouth Operations Systems such as ATLAS, BOCRIS, LMOS, PSIMS, RSAG and SOCS.

HALCRIS: HAL software contract for CSR information

HDSL: High Density Subscriber Loop/Line

ı

ILEC: Incumbent Local Exchange Company

INP: Interim Number Portability

ISDN: Integrated Services Digital Network

IPC: Interconnection Purchasing Center

L

LAN: Local Area Network

LAUTO: The automatic processor in the LNP Gateway that validates LSRs and issues service orders.

LCSC: Local Carrier Service Center - The BellSouth center which is dedicated to handling CLEC LSRs, ASRs, and Preordering transactions along with associated expedite requests and escalations.

Legacy System: Term used to refer to BellSouth Operations Support Systems (see OSS)

LENS: Local Exchange Negotiation System - The BellSouth LAN/web server/OS application developed to provide both preordering and ordering electronic interface functions for CLECs.

LEO: Local Exchange Ordering - A BellSouth system which accepts the output of EDI, applies edit and formatting checks, and reformats the Local Service Requests in BellSouth Service Order format.

LERG: Local Exchange Routing Guide

LESOG: Local Exchange Service Order Generator - A BellSouth system which accepts the service order output of LEO and enters the Service Order into the Service Order Control System using terminal emulation technology.

LFACS: Loop Facilities Assessment and Control System

LIDB: Line Information Database

LMOS: Loop Maintenance Operations System - A BellSouth Operations System that stores the assignment and selected account information for use by downstream OSS and BellSouth personnel during provisioning and maintenance activities.

LMOS HOST: LMOS host computer



LMOSupd: LMOS updates

LMU: Loop Make-up

LMUS: Loop Make-up Service Inquiry

LNP: Local Number Portability - In the context of this document, the capability for a subscriber to retain his current telephone number as he transfers to a different local service provider.

LOOPS: Transmission paths from the central office to the customer premises.

LRN: Location Routing Number

LSR: Local Service Request – A request for local resale service or unbundled network elements from a CLEC.

M

Maintenance & Repair: The process and function by which trouble reports are passed to BellSouth and by which the related service problems are resolved.

MARCH: BellSouth Operations System which accepts service orders, interprets the coding contained in the service order image, and constructs the specific switching system Recent Change command messages for input into end office switches.

Ν

NBR: New Business Request

NC: "No Circuits" - All circuits busy announcement.

NIW: Network Information Warehouse

NMLI: Native Mode LAN Interconnection

NPA: Numbering Plan Area

NXX: The "exchange" portion of a telephone number.

0

OASIS: Obtain Availability Services Information System - A BellSouth front-end processor, which acts as an interface between COFFI and RNS. This system takes the USOCs in COFFI and translates them to English for display in RNS.

OASISBSN: OASIS software contract for feature/service

OASISCAR: OASIS software contract for feature/service

OASISLPC: OASIS software contract for feature/service

OASISMTN: OASIS software contract for feature/service

OASISNET: OASIS software contract for feature/service

OASISOCP: OASIS software contract for feature/service



ORDERING: The process and functions by which resale services or unbundled network elements are ordered from Bell-South as well as the process by which an LSR or ASR is placed with BellSouth.

OSPCM: Outside Plant Contract Management System - Provides Scheduling Information.

OSS: Operations Support System - A support system or database which is used to mechanize the flow or performance of work. The term is used to refer to the overall system consisting of hardware complex, computer operating system(s), and application which is used to provide the support functions.

OUT OF SERVICE: Customer has no dial tone and cannot call out.

P

PMAP: Performance Measurement Analysis Platform

PON: Purchase Order Number

POTS: Plain Old Telephone Service

PREDICTOR: The BellSouth Operations system which is used to administer proactive maintenance and rehabilitation activities on outside plant facilities, provide access to selected work groups (e.g. RRC & BRC) to Mechanized Loop Testing and switching system I/O ports, and provide certain information regarding the attributes and capabilities of outside plant facilities.

Preordering: The process and functions by which vital information is obtained, verified, or validated prior to placing a service request.

PRI: Primary Rate ISDN

Provisioning: The process and functions by which necessary work is performed to activate a service requested via an LSR or ASR and to initiate the proper billing and accounting functions.

PSIMS: Product/Service Inventory Management System - A BellSouth database Operations System which contains availability information on switching system features and capabilities and on BellSouth service availability. This database is used to verify the availability of a feature or service in an NXX prior to making a commitment to the customer.

PSIMSORB: PSIMS software contract for feature/service.

Q

R

RNS: Regional Negotiation System - An internal BellSouth service order entry system used by BellSouth Consumer Services to input service orders in BellSouth format.

ROS: Regional Ordering System

RRC: Residence Repair Center - The BellSouth Consumer Services trouble receipt center which serves residential customers.

RSAG: Regional Street Address Guide - The BellSouth database, which contains street addresses validated to be accurate with state and local governments.

RSAGADDR: RSAG software contract for address search.



RSAGTN: RSAG software contract for telephone number search.

S

SAC: Service Advocacy Center

SEEM: Self Effectuating Enforcement Mechanism

SOCS: Service Order Control System - The BellSouth Operations System which routes service order images among Bell-South drop points and BellSouth Operations Systems during the service provisioning process.

SOIR: Service Order Interface Record - any change effecting activity to a customer account by service order that impacts 911/E911

SONGS: Service Order Negotiation and Generation System.

T

TAFI: Trouble Analysis Facilitation Interface - The BellSouth Operations System that supports trouble receipt center personnel in taking and handling customer trouble reports.

TAG: Telecommunications Access Gateway – TAG was designed to provide an electronic interface, or machine-to-machine interface for the bi-directional flow of information between BellSouth's OSSs and participating CLECs.

TN: Telephone Number

Total Manual Fallout: The number of LSRs which are entered electronically but require manual entering into a service order generator.

U

UNE: Unbundled Network Element

UCL: Unbundled Copper Link

USOC: Universal Service Order Code

V

W

WATS: Wide Area Telephone Service

WFA: Work Force Administration

WMC: Work Management Center

WTN: Working Telephone Number.

X

Appendix B: - Glossary of Acronyms and Terms

Υ

Ζ



Appendix C: BellSouth Audit Policy

C-1: BellSouth's Internal Audit Policy

BellSouth's internal efforts to make certain that the reports produced by the PMAP platform are of the highest accuracy has been formalized into a Performance Measurements Quality Assurance Plan (PMQAP) that documents and augments existing quality assurance processes integral to the production and validation of Performance Measurements data.

The plan consists of three sections:

- 1. Change Control addresses the quality assurance steps involved in the introduction of new measurements and changes to existing measurements.
- 2. Production addresses the quality assurance steps used to create monthly SQM reports.
- 3. Monthly Validation addresses the quality assurance steps used to ensure accurate posting of monthly results.

The BellSouth PMQAP will ensure that BellSouth effectively and consistently provides accurate performance measurements data for the activities included in the SQM. The BellSouth Internal Audit department will audit this plan and its quality assurance steps annually, beginning in 4Q01.

C-2: BellSouth's External Audit Policy

BellSouth currently provides many CLECs with audit rights as a part of their individual interconnection agreements. BellSouth has developed a proposed Audit Plan for use by the parties to an audit. If requested by a Public Service Commission or by a CLEC exercising contractual audit rights, BellSouth will agree to undergo a comprehensive audit of the current year aggregate level reports for both BellSouth and the CLECs for each of the next five (5) years (2001 - 2005), to be conducted by an independent third party auditor. The results of audits will be made available to all the parties subject to proper safeguards to protect proprietary information. Requested audits include the following specifications:

- 1. The cost shall be borne 50% by BellSouth and 50% by the CLECs.
- The independent third party auditor shall be selected with input from BellSouth, the PSC, if applicable, and the CLEC(s).
- 3. BellSouth, the PSC and the CLECs shall jointly determine the scope of the audit.

These comprehensive audits are intended to provide the basis for the PSCs and CLECs to determine that the SQM and PMAP produce accurate data that reflects each States Order for performance measurements. Once this has been verified by an initial audit, the BellSouth PMQAP will provide the basis for future audits.

PRICING

1. **General Principles**

All services currently provided hereunder (including resold Local Services, Network Elements and Ancillary Functions) and all new and additional services to be provided hereunder shall be priced in accordance with all applicable provisions of the Act and the rules and orders of the Federal Communications Commission and South Carolina Public Service Commission.

2. Local Service Resale

The rates that ITC^DeltaCom shall pay to BellSouth for resold Local Services shall be BellSouth's Retail Rates less the applicable discount. The discount that will apply to all Telecommunications Services available for resale are as set forth in Attachment 1.

3. <u>Unbundled Network Elements</u>

The prices that ITC^DeltaCom shall pay to BellSouth for Unbundled Network Elements are set forth in Table 1 as ordered and approveded by the applicable state Commission.

4. <u>Compensation For Local Interconnection (Call Transport and Termination)</u>

Unless otherwise specified in this Agreemeth the prices that ITC^DeltaCom and BellSouth shall pay each other for the termination of local calls are set forth in Table 1.

5. **Ancillary Functions**

- 5.1 Collocation The rates, terms and conditions for Physical Collocation are as set forth in Attachment 4 of this Agreement. Rates, terms, and conditions for Virtual Collocation are as set forth in Section 20 of BellSouth Telecommunications, Inc.'s Interstate Access Tariff, FCC No. 1.
- Poles, Ducts and Conduits BellSouth shall provide access to poles, conduits and ducts at rates that are consistent with 47 U.S.C. Section 224(d). CLEC may file a complaint with the appropriate regulatory authority if it believes the rates provided by BellSouth are not consistent with 47 U.S.C. Section 224(d).

6. <u>Local Number Portability</u>

The prices for interim number portability are set forth in Table 1.

7. Recorded Usage Data

The prices for recorded usage data are set forth in Table 1.

8. **Electronic Interfaces**

The costs associated with implementing electronic interfaces should be shared equitably among all parties who benefit from those interfaces. The Party requesting a special arrangement for data access should pay the reasonable and demonstrable costs for providing the access. However, if other Parties request the same or similar access and benefit from the development, these other Parties should share the cost, and CLEC would then be refunded on a proportionate share of the costs.

9. Operational Support Systems (OSS) Rates

The parties agree that Electronic Interface (EI) costs and manual work done by the LCSC will be recovered on a "per LSR' basis, with an individual LSR identified by its Purchase Order Number (PON). ITC^DeltaCom will be assessed either the manual or mechanized charge for most accepted LSRs submitted to BellSouth. Manually submitted UNE LSRs will not incur the manual LSR charge in states that have a separate UNE manual additive. ITC^DeltaCom will be charged the manual rate for most LSRs submitted by mail, courier, fax, etc. ITC^DeltaCom will be charged the mechanized rate for LSRs submitted over any of the mechanized systems (e.g. LENS, EDI, EDI-PC, and TAG).

- A. Bill a single mechanized CLEC EI charge for each resale LSR delivered over an electronic interface. This charge recovers the development and expense costs associated with the CLEC EIs that are allocated to resale LSR volumes, as well as the manual processing associated with mechanized requests that "fall out" in the LCSC for manual handling.
- B. Bill the same mechanized CLEC El charge for each UNE LSR delivered over an electronic interface.
- C. Bill a single manual LSR charge for each resale LSR delivered manually that reflects the costs associated with the manual processing of those LSRs in the LCSC.
- D. Bill the same manual LSR charge for each manually submitted UNE LSR in those states that do not have a per element UNE non-recurring manual additive.
- E. Establish a transitional plan to bill the mechanized LSR charge for manual LSRs for CLECs who submit a significant proportion of their total LSR volume on a mechanized basis. This volume threshold will

increase each year and be eliminated in 2002. This arrangement may be superceded by BellSouth with an LSR-specific process that would apply the mechanized LSR rate to only those manual LSRs which cannot be submitted over a mechanized system.

The regional average pricing plan establishes averaged prices that are the same regardless of:

- A. CLEC EI system used
- B. Action being requested on the LSR (order, change, deny, restore, cancel, disconnect, etc.)
- C. Number of supplements or clarifications received
- D. Number of service orders result from the LSR

Some CLECs presently provide lists of customers to be denied and restored, rather than individual LSRs. However, since each location on the list must have a separate PON, they will be billed as separate manual LSRs.

ITC^DeltaCom will be charged for an accepted LSR that is later canceled by ITC^DeltaCom.

At the present time, six states (AL, GA, LA, MS, NC, SC) have a manual NRC additive per element for UNEs. This manual additive supercedes the manual LSR charge for manual UNE LSRs. Until the other three states adopt this methodology, BellSouth proposes that the manual LSR charge apply for manual UNE LSRs in those states.

UNEs

OPERATIONAL SUPPORT	AL, GA, LA, MS,	FL, KY, TN
SYSTEMS	NC, SC	
OSS Order charge, per LSR received		\$3.50/SOMEC
from the CLEC by one of the OSS		
interactive interfaces/USOC		
Incremental charge per LSR received		\$19.99/SOMAN
from the CLEC by means other than	element	
one of the OSS interactive		
interfaces/USOC		

In addition to OSS charges, applicable service order and related charges apply per the tariff.

The Parties agree that ITC^DeltaCom will incur the mechanized rate for all LSRs, both mechanized and manual, if the percentage of mechanized LSRs to total LSRs exceeds the threshold percentages shown below:

Year Ratio: Mechanized/Total LSRs

2001 90%

The threshold plan will be discontinued in 2002.

BellSouth will track the total LSR volume for ITC^DeltaCom for each quarter. At the end of that time period, a Percent Electronic LSR calculation will be made for that quarter based on the LSR data tracked in the LCSC. If this percentage exceeds the threshold volume, all of that ITC^DeltaComs' future manual LSRs will be billed at the mechanized LSR rate. To allow time for obtaining and analyzing the data and updating the billing system, this billing change will take place on the first day of the second month following the end of the quarter (e.g. May 1 for 1Q, Aug 1 for 2Q, etc.). There will be no adjustments to the amount billed for previously billed LSRs.

DRY	NOTES	ELEMENT	Interim Zone	BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manua Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge Manual St Order vs Electronic-I Add'I
							Nonrec	urring		onnect						+
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
						neo	11100	7001	11154	7001	COME	COMPA	COMPA	OO.III.PAY	COMPA	- COMPA
ht	ttp://www.ii	shown in the sections for stand-alone loops or loops as part of a combination refers to nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm	Geographically	y Deaveraged UNI	E Zones. To	o view Geograpl	hically Deavera	ged UNE Zon	e Designatio	ons by Centr	al Office, re	fer to Interne	et Website:			
DLED	EXCHANG	GE ACCESS LOOP														
-	MIDE AN	ALOG VOICE GRADE LOOP														+
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1	1	UEANL	UEAL2	13.54	70.44	44.05	46.93	10.4		19.99				+
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2	2	UEANL	UEAL2	19.73	70.44	44.05	46.93	10.4		19.99				+
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3	3	UEANL	UEAL2	28.27	70.44	44.05	46.93	10.4		19.99				1
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	1	UEPSR, UEPSE	UEALS	13.54	70.44	44.05	46.93	10.4		19.99				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	2	UEPSR, UEPSE		19.73	70.44	44.05	46.93	10.4		19.99				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3	3	UEPSR, UEPSE	UEALS	28.27	70.44	44.05	46.93	10.4		19.99				
		Engineering Information Document (EI)		UEANL			28.76	28.76								
		Manual Order Coordination for UVL-SL1s (per loop)*		UEANL	UEAMC		16.31	16.31								
		Wandar Order Goordination for GVE GETS (per 190p)		OLTUL	OL7 WIO		10.01	10.01								+
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *		UEANL	OCOSL		36.18	36.18								-
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1	1	UEA	UEAL2	17.27	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start		OL, (OL, LL		200.70					10.00				
		Signaling - Zone 2	2	UEA	UEAL2	32.32	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3	3	UEA	UEAL2	55.78	236.75	177.1				19.99				
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		36.18									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -														
		Zone 1	1	UEA	UEAR2	17.27	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -														
_		Zone 2	2	UEA	UEAR2	32.32	236.75	177.1				19.99				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling - Zone 3	3	UEA	UEAR2	55.78	236.75	177.1				19.99				
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		36.18									
4-		ALOG VOICE GRADE LOOP														
		4-Wire Analog Voice Grade Loop - Zone 1	1	UEA	UEAL4	20.92	457.14	348.83				19.99				
		4-Wire Analog Voice Grade Loop - Zone 2 4-Wire Analog Voice Grade Loop - Zone 3	3	UEA UEA	UEAL4	39.14 67.57	457.14 457.14	348.83 348.83				19.99				
		·	3			67.57		348.83				19.99				1
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		36.18									
2.		N DIGITAL GRADE LOOP	4	UDN	U1L2X	22.00	E44.00	431.61			1	19.99	-		1	+
		2-Wire ISDN Digital Grade Loop - Zone 1 2-Wire ISDN Digital Grade Loop - Zone 2	2	UDN	U1L2X U1L2X	23.66 44.28	541.28 541.28	431.61			-	19.99	1			+
-		2-Wire ISDN Digital Grade Loop - Zone 2 2-Wire ISDN Digital Grade Loop - Zone 3	3	UDN	U1L2X	76.42	541.28	431.61				19.99				+
		Order Coordination For Specified Conversion Time (per LSR)	J	UDN	OCOSL	70.42	36.18	401.01				10.00				
2.	-WIDE IIni	versal Digital Channel (UDC) COMPATIBLE LOOP		ODIT	COCCE		50.10									1
- 2		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1	1	UDC	UDC2X	25.73	233.47	158.51	105.49	20.48	1	19.99	1		1	+
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2	2	UDC	UDC2X	34.83	233.47	158.51	105.49	20.48		19.99				1
		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3	3	UDC	UDC2X	45.56	233.47	158.51	105.49	20.48		19.99				
2.		YMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP [2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE														
$\perp \! \! \! \! \! \perp$		LOOP														
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation			1 T											1
_		- Zone 1	1	UAL	UAL2X	8.79	713.5	609.44				19.99	1		1	+
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation		1141	1101.007	40.40	740.5	000 11				40.00				1
-		Zone 2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation	2	UAL	UAL2X	16.46	713.5	609.44			-	19.99	1			+
		- Zone 3	3	UAL	UAL2X	28.4	713.5	609.44				19.99				
-		ZONG O	3	UAL	UNLZA	20.4	710.0	003.44			1	10.00	 		1	+
		Order Coordination for Specified Conversion Time (per LSR)		UAL	OCOSL		36.18									
_		2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -														T
		Zone 1	1	UAL	UAL2W	8.79	205.25	129.42	100.89	15.88		19.99	<u> </u>		<u> </u>	Ш.
		Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2	2	UAL	UAL2W	16.46	205.25	129.42	100.89	15.88		19.99				
		2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -					200.20	.20.72				. 5.55				†
		Zone 3	3	UAL	UAL2W	28.4	205.25	129.42	100.89	15.88	1	19.99		1		1

RY	NOTES	ELEMENT	Interim Zo	Zone	BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
NI .	NOTES	CLUMENT	meriii Z	John	503	0300						Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Il Order vs. Electronic-Disc	Increment Charge - Manual Sv Order vs c Electronic-D Add'I
				-				Nonrec	urring		ecurring						+
				-+			Rec	First	Add'l	Disc First	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Order Constitution for Constitut Conversion Time (no. 1 CD)		\neg	LIAL	000001	Rec		Addi	FIFSt	Addi	SOMEC	SOMAN	SUMAN	SUMAN	SOMAN	SOMAN
		Order Coordination for Specified Conversion Time (per LSR)		士	UAL	OCOSL		36.18									
2		SH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP															
		2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP															
		2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		-													+
		- Zone 1		1	UHL	UHL2X	6.29	713.5	609.44				19.99				
		2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		2			44.70	740.5	000 44				40.00				
		- Zone 2 2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		2	UHL	UHL2X	11.78	713.5	609.44				19.99			+	-
		- Zone 3		3	UHL	UHL2X	20.33	713.5	609.44				19.99				
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		36.18									+
		2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL2W	6.29	222.58	146.75	100.89	15.88		19.99				
		2 Wire Unbundled HDSL Loop without manual service inquiry and facility			O. I.E	O. ILL.	0.20	222.00	1.10.70	100.00	10.00		10.00			+	+
		reservation - Zone 2		2	UHL	UHL2W	11.78	222.58	146.75	100.89	15.88		19.99				
		2 Wire Unbundled HDSL Loop without manual service inquiry and facility		3			00.00	000.50	440.75	400.00	45.00		40.00				
-		reservation - Zone 3		3	UHL	UHL2W	20.33	222.58	146.75	100.89	15.88		19.99			+	-
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		36.18									
																	1
4		SH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP															
		4 Wire Unbundled HDSL Loop including manual service inquiry and facility reservation - Zone 1		4	UHL	UHL4X	7.68	748.93	646.17				19.99				
		4-Wire Unbundled HDSL Loop including manual service inquiry and facility		'	UNL	UHL4X	7.00	746.93	040.17				19.99			+	+
		reservation - Zone 2	1 :	2	UHL	UHL4X	14.38	748.93	646.17				19.99				
		4-Wire Unbundled HDSL Loop including manual service inquiry and facility															
		reservation - Zone 3		3	UHL	UHL4X	24.82	748.93	646.17				19.99				
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		36.18									
		4-Wire Unbundled HDSL Loop without manual service inquiry and facility			O. I.E	00002		00.10								+	1
		reservation - Zone 1		1	UHL	UHL4W	7.68	279.79	203.96	109.64	20.64		19.99				
		4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 2		2	UHL	LILI 4\A/	14 20	270.70	202.06	100.64	20.64		10.00				
		4-Wire Unbundled HDSL Loop without manual service inquiry and facility			UNL	UHL4W	14.38	279.79	203.96	109.64	20.64		19.99				+
		reservation - Zone 3		3	UHL	UHL4W	24.82	279.79	203.96	109.64	20.64		19.99				
		0.1.0				00001		00.40									
		Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		36.18								+	+
4	-WIRE DS	1 DIGITAL LOOP		_													1
		4-Wire DS1 Digital Loop - Zone 1		1	USL	USLXX	50.26	849.8	523.27				19.99				
		4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	94.06	849.8	523.27				19.99				
-		4-Wire DS1 Digital Loop - Zone 3	-	3	USL	USLXX	162.34	849.8	523.27				19.99			+	+
		Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		36.18									
4		2, 56 OR 64 KBPS DIGITAL GRADE LOOP		_	LIDI	LIDI 40	05.00	050.00	470.00	440.05	07.05		40.00				+
		4 Wire Unbundled Digital 19.2 Kbps 4 Wire Unbundled Digital 19.2 Kbps		2	UDL UDL	UDL19 UDL19	35.92 40.32	250.99 250.99	176.03 176.03	116.85 116.85	27.85 27.85		19.99 19.99			+	+
		4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	37.9	250.99	176.03	116.85	27.85		19.99			+	+
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	35.92	250.99	176.03	116.85	27.85		19.99			+	+
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	40.32	250.99	176.03	116.85	27.85		19.99				
		4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	37.9	250.99	176.03	116.85	27.85		19.99				
		Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		36.18									
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	35.92	250.99	176.03	116.85	27.85		19.99			+	+
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2	UDL	UDL64	40.32	250.99	176.03	116.85	27.85		19.99				
		4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3	UDL	UDL64	37.9	250.99	176.03	116.85	27.85		19.99				+
		Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		36.18									
⇉		The second of the second						23.10									=
	WID= · · ·	L II. LOOPPER LOOP		_ ∓				1									\perp
2	-WIKE Uni	bundled COPPER LOOP		\rightarrow				+		1	1	1				+	+
		2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1		1	UCL	UCLPB	14.94	283.77	164.04	120.6	22.45		19.99				
		2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility			JUL		17.34	200.11	104.04	120.0	22.40	1	13.33			+	+
		reservation - Zone 2		2	UCL	UCLPB	15.15	283.77	164.04	120.6	22.45		19.99				
		2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility		3			15.73	283.77	164.04	120.6	22.45		19.99				1
+												1		1		1	1
		reservation - Zone 3		3	UCL	UCLPB	15.73	283.11	104.04	120.0	22.43		10.00				+

CATEGORY	NOTES	ELEMENT	Interim Zone	BCS	USOC		RA ⁻	TES (\$)			1		OSS R	ATES (\$)		
								(*/			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
							Nonrecur	rring		curring						
						Rec	First	Add'I	Disc First	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility				Rec	FIFSt	Add'I	FIRST	Add 1	SOMEC	SUMAN	SUMAN	SOMAN	SUMAN	SUMAN
		reservation - Zone 1	1	UCL	UCLPW	14.94	203.39	127.56	100.89	15.88		19.99				
		2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility														
		reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	2	UCL	UCLPW	15.15	203.39	127.56	100.89	15.88		19.99				-
		reservation - Zone 3	3	UCL	UCLPW	15.73	203.39	127.56	100.89	15.88		19.99				
		Order Coordination for Unbundled Copper Loops (per loop)														
		2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility		UCL	UCLMC		16.31	16.31								
		reservation - Zone 1	1	UCL	UCL2L	36.19	270.38	150.65	120.6	22.45		19.99				
		2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility														
		reservation - Zone 2	2	UCL	UCL2L	49.31	270.38	150.65	120.6	22.45		19.99				
		2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3	3	UCL	UCL2L	80.78	270.38	150.65	120.6	22.45		19.99				
		Order Coordination for Unbundled Copper Loops (per loop)	ŭ	002	OOLLL	00.70	270.00	100.00	120.0	LL: 10		10.00				
				UCL	UCLMC		16.31	16.31								
		2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 1		UCL	UCL2W	36.19	190	114.17	100.89	15.88		19.99				
		2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		UCL	JULZVV	30.18	130	114.17	100.09	13.00		13.33				
		reservation - Zone 2	2	UCL	UCL2W	49.31	190	114.17	100.89	15.88		19.99				
		2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility	3	UCL	LICI 3/44	90.79	190	114.17	100.00	15.88		10.00				
		reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)	3	UCL	UCL2W UCLMC	80.78	16.31	16.31	100.89	15.66		19.99				+
		The state of the s														
		O Wise Heb and and Connections - New Designed 7 4	I 1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06		19.99				
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1 2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	1 2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06		19.99				-
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	I 3	UEQ	UEQ2X	20.22	44.69	22.4	25.65	7.06		19.99				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loop)		UEQ	USBMC		16.31	16.31								
		Engineering Information Document Loop Testing - Basic 1st Half Hour		UEQ UEQ	URET1		28.76 78.92	28.76 78.92								
		Loop Testing - Basic Additional Half Hour		UEQ	URETA		23.33	23.33								
		4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1 4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 2 4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop) 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 1 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 3 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 3 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 1 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2 4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)	1 2 3 1 1 2 3 3	UCL UCL UCL UCL UCL UCL UCL UCL	UCL4S UCL4S UCL4S UCLMC UCLMC UCL4W UCL4W UCL4W UCL4W UCL4L UCL4L UCL4L	25.26 23 19.08 25.26 23 19.08 61.02 55.74 88.97	332.2 332.2 332.2 16.31 251.82 251.82 251.82 16.31 318.81 318.81 318.81	212.46 212.46 16.31 175.99 175.99 16.31 199.07 199.07	130.27 130.27 130.27 109.64 109.64 130.27 130.27	27.51 27.51 27.51 20.64 20.64 20.64 27.51 27.51		19.99 19.99 19.99 19.99 19.99 19.99 19.99				
		4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 1	1	UCL	UCL4O	61.02	238.42	162.6	109.64	20.64	1	19.99				
		4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility														
		reservation - Zone 2	2	UCL	UCL40	55.74	238.42	162.6	109.64	20.64		19.99				
		4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 3	3	UCL	UCL4O	88.97	238.42	162.6	109.64	20.64		19.99				
		Order Coordination for Unbundled Copper Loops (per loop)	3	UCL	UCLMC	00.31	16.31	16.31	103.04	20.04		13.33				
				1				-								
LOOP MOD	IFICATION		 	1							1	+				+
200. 11100		Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or		UAL, UHL, UCL,												
		equal to 18k ft		UEQ, ULS	ULM2L		65.2	65.2				1				
		Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to		UCL, ULS	ULM2G		341.64	341.64				1				1
		18K ft		UHL, UCL	ULM4L		65.2	65.2								
		Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k														
1		ft	1 1	UCL	ULM4G		341.64	341.64		l			1	1	l	

	TES ELEMENT Inte	erim Zone	BCS	USOC		RAT	TES (\$)					OSS R	ATES (\$)		
										Svc Order Submitted	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svo Order vs. Electronic-Di
										per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
						Nonrecur	ring		curring						
					Rec	First	Add'l	Disco	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			UAL, UHL, UCL,		Rec	rirst	Add I	riist	Add I	SUMEC	SUMAN	SOMAN	SOMAN	SUMAN	SUMAN
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop		UEQ, UEF, ULS	ULMBT		65.24	65.24								
B-LOOPS															
Cub I	and Distribution														
Sub-L	Loop Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up		UEANL	USBSA		600.03	600.03				19.99				
	Sub-Loop - Per Cross Box Location - CELC Feeder Facility Set-Op	i	UEANL	USBSB		45.28	45.28				19.99				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	l	UEANL	USBSC		379.89	379.89				19.99				
	City I and Des Dividies Equipment Dance Des 25 Dais Board Cat III		LIFANII	USBSD		111.55	111.55				40.00				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	l I 1	UEANL UEANL	USBN2	9.03	131.64	61.93	90.83	13.44		19.99 19.99				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2		UEANL	USBN2		131.64	61.93	90.83	13.44		19.99				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	I 2 I 3	UEANL	USBN2	12.25 16.71	131.64	61.93	90.83	13.44		19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBMC		36.18	36.18								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1	1	UEANL	USBN4 USBN4	10.18	158.12	88.41	99.1	18.08		19.99				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3	3	UEANL UEANL	USBN4 USBN4	9.44 13.38	158.12 158.12	88.41 88.41	99.1 99.1	18.08 18.08		19.99 19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	3	UEANL	USBMC	10.00	36.18	36.18	33. I	10.00		13.33				
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	ı	UEANL	USBR2	3.23	106.06	36.35	90.83	13.44		19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBMC		36.18	36.18								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	l	UEANL	USBR4	6.29	118.54	48.84	99.1	18.08		19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEANL	USBMC		36.18	36.18								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	1 1	UEF	UCS2X UCS2X	8.01 9.18	131.64 131.64	61.93 61.93	90.83 90.83	13.44 13.44		19.99 19.99				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2 2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		UEF	UCS2X	11.02	131.64	61.93	90.83	13.44		19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEF	USBMC	11.02	36.18	36.18	00.00	10.11		10.00				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	l 1	UEF	UCS4X	10.65	158.12	88.41	99.1	18.08		19.99				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2		UEF	UCS4X	9.71	158.12	88.41	99.1	18.08		19.99				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	1 3	UEF	UCS4X	8.45	158.12	88.41	99.1	18.08		19.99				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UEF	USBMC		36.18	36.18								
Sub-L	Loop Feeder														
			UEA,												
			UDN,UCL,UDL,UD	USBF											
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up		C	W		600.03									
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up		C UEA,	W		600.03									
			C UEA, UDN,UCL,UDL,UD	W			45.28								
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location . per DS1 termination		C UEA, UDN,UCL,UDL,UD C	W USBFX		45.28	45.28 11.32								
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1	1	C UEA, UDN,UCL,UDL,UD C USL UEA	USBFX USBFZ USBFA	10.36	45.28 527.98 184.97	11.32 111.91	108.76	26.76		19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2	2	C UEA, UDN,UCL,UDL,UD C USL UEA	USBFX USBFZ USBFA USBFA	13.62	45.28 527.98 184.97 184.97	11.32 111.91 111.91	108.76	26.76		19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3	1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA UEA UEA	USBFX USBFZ USBFA USBFA USBFA		45.28 527.98 184.97 184.97 184.97	11.32 111.91								
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR	2 3	C UEA, UDN,UCL,UDL,UD C USL UEA UEA UEA UEA	W USBFX USBFZ USBFA USBFA USBFA USBFA	13.62 19.69	45.28 527.98 184.97 184.97 184.97 36.18	11.32 111.91 111.91 111.91	108.76 108.76	26.76 26.76		19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1	3	C UEA, UDN,UCL,UDL,UC C USL UEA UEA UEA UEA UEA UEA UEA	USBFX USBFZ USBFA USBFA USBFA USBFA USBFA USBFA USBFB	13.62 19.69 10.36	45.28 527.98 184.97 184.97 184.97 36.18 184.97	11.32 111.91 111.91 111.91	108.76 108.76	26.76 26.76		19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR	2 3	C UEA, UDN,UCL,UDL,UC USL UEA	W USBFX USBFZ USBFA USBFA USBFA USBFA USBFB USBFB USBFB	13.62 19.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97	11.32 111.91 111.91 111.91	108.76 108.76	26.76 26.76		19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR	1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFZ USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB	13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 36.18	11.32 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1	1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFZ USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFB USBFB USBFB	13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1	1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFZ USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB	13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 36.18	11.32 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2	1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFZ USBFA USBFA USBFA OCOSL USBFB USBFB USBFB USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1	1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFZ USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFB USBFB USBFB	13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR	1 2 3	C UEA, UDN,UCL,UDL,UD C USL UEA	USBFX USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76		19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1	1 2 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USI. UEA	USBFX USBFZ USBFZ USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2	1 2 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	USBFX USBFA USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC USBFC USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1	1 2 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USI. UEA	USBFX USBFZ USBFZ USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91	108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2	1 2 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UC C USL UEA	USBFX USBFA USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC USBFC USBFC USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR	1 2 3 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UDL C USL UEA	W USBFX USBF2 USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFD USBFD USBFD	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 30.69 30.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2	1 2 3 3 1 2 3 3 1 2 3 3	C UEA, UDN,UCL,UDL,UD C USL UEA	W USBFX USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 30.69 30.69 30.69 30.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR	1 2 3 3 1 2 3 1 2 3	C UEA, UDN,UCL,UDL,UDL C USL UEA	W USBFX USBF2 USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFD USBFD USBFD	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 30.69 30.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	1 2 3 3 1 2 3 3 1 2 3 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 30.69 30.69 30.69 30.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	1 2 3 3 4 1 3 4 1	C UEA, UDN,UCL,UDL,UDL C USL UEA	W USBFX USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFB USBFC USBFC USBFC USBFC USBFC USBFC USBFC USBFC USBFD OCOSL	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12 22.9	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56 213.56 36.18	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 113.6 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, Per LSR	1 2 3 3 1 2 3 3 1 2 3 3	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFA USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFC	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 30.69 30.69 30.69 30.69	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	1 1 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 2 3 3 1 1 2 1 1 1 1	C UEA, UDN,UCL,UDL,UD C USL UEA	W USBFX USBFZ USBFA USBFA USBFB USBFB USBFC USBFC USBFD USBFB USBFE USBF	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12 22.9 30.69 36.12 22.9	45.28 527.98 184.97 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 113.6 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64 122.64 122.64	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2	1 1 2 3 3 3 4 1 2 2 3 3 4 1 2	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFC USBFF USBFF	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12 22.9 30.69 36.12 22.9	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64 122.64 122.64 111.02	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64 26.01 26.01		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 3 Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 3 Order Coordination For Specified Conversion Time, Per LSR	1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 3 3 3 1 1 2 3 3 3 1 1 2 3 3 3 1 1 2 3 3 3 1 1 2 3 3 3 1 1 2 3 3 3 3	C UEA, UDN,UCL,UDL,UDL C USL UEA	W USBFX USBFZ USBFA USBFB USBFB USBFC USBFC USBFD USBFF USBF	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12 22.9 30.69 36.12 22.9 23.67 23.67 29.9	45.28 527.98 184.97 184.97 184.97 36.18 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 113.6 138.6 138.6 138.6 138.6 138.6 138.6 138.6 138.6 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64 122.64 122.64 111.02 111.02	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64 32.64 33.64 33.64 33.64 33.64		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up USL Feeder DS1 Set-up at DSX location, per DS1 termination Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Order Coordination for Specified Time Conversion, per LSR Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2 Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 2	1 1 2 3 3 3 4 1 2 2 3 3 4 1 2	C UEA, UDN,UCL,UDL,UC C USL UEA	W USBFX USBFA USBFA USBFA USBFB USBFB USBFB USBFB USBFC USBFF USBFF	13.62 19.69 10.36 13.62 19.69 10.36 13.62 19.69 30.69 36.12 22.9 30.69 36.12 22.9	45.28 527.98 184.97 184.97 184.97 36.18 184.97 184.97 184.97 184.97 184.97 184.97 184.97 36.18 213.56 213.56 213.56 213.56 213.56 213.56	11.32 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 111.91 138.6 138.6 138.6 138.6	108.76 108.76 108.76 108.76 108.76 108.76 108.76 108.76 122.64 122.64 122.64 122.64 122.64 111.02	26.76 26.76 26.76 26.76 26.76 26.76 26.76 26.76 33.64 33.64 33.64 33.64 33.64 26.01 26.01		19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99 19.99				

CATEGORY	NOTES	ELEMENT	Interim Zone	BCS	USOC		R	ATES (\$)					OSS R	ATES (\$)		
CATEGORY	NOTES	CLIMICIY	Zone	500	0300			(#)			Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manua Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc
											per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
							Nonrec	urring		ecurring						-
										onnect	201150					
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1	1	USL	USBEG	75.1	First 202.14	Add'I 127.18	First 122.64	Add'I 33.64	SOMEC	SOMAN 19.99	SOMAN	SOMAN	SOMAN	SOMAN
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1	2	USL	USBFG	104.53	202.14	127.18	122.64	33.64		19.99				
		Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3	3	USL	USBFG	152.36	202.14	127.18	122.64	33.64		19.99				
		Oribundied Sub-Loop Feeder Loop, 4-Wife DST - Zone S	3	USL	USBFG	132.30	202.14	127.10	122.04	33.04		19.99				
		Order Coordination For Specified Conversion Time, Per LSR		USL	OCOSL		36.18									
		Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1	1	UCL	USBFH	8.29	167.62	92.66	106.42	21.41		19.99				
		Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2	2	UCL	USBFH	7.3	167.62	92.66	106.42	21.41		19.99				
		Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3	3	UCL	USBFH	6.03	167.62	92.66	106.42	21.41		19.99				
		Oribunated Gub Edop'i Coder Edop, 2 Write Gopper Edop "Zone o	- 0	OOL	CODITI	0.00	107.02	32.00	100.42	21.71		10.00				+
		Order Coordination For Specified Conversion Time, per LSR		UCL	OCOSL		36.18									
		Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1	1	UCL	USBFJ	16.55	202.05	127.09	115.43	26.43		19.99				1
		Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2	2	UCL	USBFJ	15.35	202.05	127.09	115.43	26.43		19.99				
		Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3	3	UCL	USBFJ	12.52	202.05	127.09	115.43	26.43		19.99				
		Cub 200p 1 00d01 1 01 1 11110 Coppor 200p 2010 C		002	002.0	12.02	202.00	127.00	110.10	20.10		10.00				
		Order Coordination For Specified Conversion Time, per LSR		UCL	OCOSL		36.18									
		Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	1	UDL	USBFN	27.38	202.14	127.18	122.64	33.64	1	19.99			1	
		Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	2	UDL	USBFN	33.41	202.14	127.18	122.64	33.64		19.99		İ		
		Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	3	UDL	USBFN	24.47	202.14	127.18	122.64	33.64	1	19.99			İ	
		Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1	1	UDL	USBFO	27.38	202.14	127.18	122.64	33.64	1	19.99			1	
		Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2	2	UDL	USBFO	33.41	202.14	127.18	122.64	33.64	1	19.99			1	
		Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3	3	UDL	USBFO	24.47	202.14	127.18	122.64	33.64		19.99				
				-				1			1				İ	
		Order Coordination For Specified Time Conversion, per LSR		UDL	OCOSL		36.18									
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1	1	UDL	USBFP	27.38	202.14	127.18	122.64	33.64	1	19.99			1	
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2	2	UDL	USBFP	33.41	202.14	127.18	122.64	33.64		19.99				
		Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3	3	UDL	USBFP	24.47	202.14	127.18	122.64	33.64		19.99				
		1 2 2														
		Order Coordination For Specified Conversion Time, per LSR		UDL	OCOSL		36.18									
	Unbundled	Sub-Loop Modification														
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per														
		2-W PR		UEF	ULM2X		355.83	12.27				19.99				
		Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-														
		W PR		UEF	ULM4X		355.83	12.27				19.99				
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per			-											
		PR unloaded		UEF	ULM4T		560.74	14.3				19.99				
					-											
	Unbundled	Network Terminating Wire (UNTW)														1
		Unbundled Network Terminating Wire (UNTW) per Pair		UENTW	UENPP	0.64	62.83	62.83				19.99				
		3 · · · · (- · · · · / p · · · · · · · · · · · · · ·														
	Network In	terface Device (NID)														
		Network Interface Device (NID) - 1-2 lines		UENTW	UND12		89.66	57.24				19.99				
		Network Interface Device (NID) - 1-6 lines		UENTW	UND16		129.24	99.52				19.99				1
		Network Interface Device Cross Connect - 2 W		UENTW	UNDC2		11.78	11.78				19.99				1
		Network Interface Device Cross Connect - 4W		UENTW	UNDC4		11.78	11.78				19.99				1
																1
BUNDLE	D LOOP CO	DICENTRATION														
		Unbundled Loop Concentration - System A (TR008)		ULC	UCT8A	522.17	651.04	651.04		1		19.99			1	t
		Unbundled Loop Concentration - System B (TR008)		ULC	UCT8B	63.59	271.27	271.27		1	1	19.99			1	
		Unbundled Loop Concentration - System A (TR303)		ULC	UCT3A	567.21	651.04	651.04				19.99				t
		Unbundled Loop Concentration - System B (TR303)		ULC	UCT3B	107.16	271.27	271.27		1	1	19.99			1	
		\						1						İ		1
		Unbundled Loop Concentration - DS1 Loop Interface Card		ULC	UCTCO	6.04	126.61	92.17	33.46	9.37		19.99			1	
		Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)		UDN	ULCC1	9.59	21.08	20.96	10.75	10.68	1	19.99			İ	
		Unbundled Loop Concentration - UDC Loop Interface (Brite Card)		UDC	ULCCU	9.59	21.08	20.96	10.75	10.68	1	19.99			İ	
		Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop								1					1	
		Interface (POTS Card)		UEA	ULCC2	2.4	21.08	20.96	10.75	10.68		19.99				
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface			12222					1	1				İ	
		(SPOTS Card)		UEA	ULCCR	14.26	21.08	20.96	10.75	10.68		19.99				
		Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card)		UEA	ULCC4	8.51	21.08	20.96	10.75	10.68	1	19.99			1	
		Unbundled Loop Concentration - TEST CIRCUIT Card		ULC	UCTTC	41.58	21.08	20.96	10.75	10.68		19.99		İ		
		Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface		UDL	ULCC7	12.6	21.08	20.96	10.75	10.68	1	19.99			1	
		Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface		UDL	ULCC5	12.6	21.08	20.96	10.75	10.68		19.99				
		Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interface		UDL	ULCC6	12.6	21.08	20.96	10.75	10.68		19.99		İ		
		4						1								
BUNDLE	D SUB-LOC	OP CONCENTRATION (OUTSIDE CO)														
								İ						İ		
										İ	1				İ	
E OTHER	R. PROVISIO	ONING ONLY - NO RATE								İ	1				İ	
		NID - Dispatch and Service Order for NID installation		UENTW	UNDBX		1									
								İ						İ		
		UNTW Circuit Id Establishment, Provisioning Only - No Rate		UENTW	UENCE						1					
	1	2 2 2 to Education and T. Torrico and Orliny Two Natio	I	0211111	CLITOL		1	1	1	1	1	1	1		1	

CATEGORY	NOTES ELEMENT	Interim	Zone	BCS	USOC		RA	ATES (\$)					OSS R/	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svo Order vs. Electronic-Di Add'I
							Nonrecu	urring	Nonre	curring	per Lok	Lor	Electronic-1st	Electronic-Add i	150	Add I
										onnect						
				UEANL.UEF.UEQ.		Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Contract Name, Provisioning Only - No Rate			UENTW	UNECN									i		
				UAL,UCL,UDC,UD										1		
	Unbundled Contact Name, Provisioning Only - no rate			L,UDN,UEA,UHL,U	UNECN	0	0							i		
	Orbanded Contact Ferrio, Frovisioning Only Thorate					Ü							1			+
				UEA,UDN,UCL,UD		_										
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate			C UEA,USL,UCL,UD	USBFQ	0	0								-	
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			L	USBFR	0	0							i		
	Historia I DOM I are a construction of the con			1101	00005	•								i		
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0	0									
	Unbundled DS1 Loop - Expanded Superframe Format option - no rate			USL	CCOEF	0	0							i		
	Y UNBUNDLED LOCAL LOOP TE: 4 month minimum billing period														-	
140	High Capacity Unbundled Local Loop - DS3 - Per Mile per month	1	+	UE3	1L5ND	11.53							 			
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	379.72	903.34	528.05	238.2	166.62		19.99				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per month High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month	-	-	UDLSX UDLSX	1L5ND UDLS1	11.53 394.76	903.34	528.05	238.2	166.62		19.99	+			-
				ODLOX	JULUI	334.70	303.34	320.03	200.2	100.02		10.00				<u> </u>
LOOP MAKE-U	P				1000											
	Loop Makeup - Preordering Without Reservation, per working or spare facility queried (Manual).			UMK	UMKL W		47.98	47.98						i		
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		50.88	50.88								
	Loop MakeupWith or Without Reservation, per working or spare facility queried															
	(Mechanized)			UMK	PSUMK		0.6746	0.6746					-			
LINE SHARING													+			
	Line Sharing Splitter, per System 96 Line Capacity	!		ULS	ULSDA	203.33	377.71	0	357.29	0		0				
	Line Sharing Splitter, per System 24 Line Capacity Line Sharing Splitte, Per System, 8 Line Capacity	i		ULS	ULSDB ULSD8	50.83 16.94	377.71 377.71	0	357.29 357.29	0		0	+			
	Line Sharing - per Line Activation	i		ULS	ULSDC	0.61	37.02	21.2	20.1	9.87		19.99				
	Line Sharing - per Subsequent Activity per Line Rearrangement	ı		ULS	ULSDS		32.78	16.38				19.99			ļ	
													 			-
	Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8															
	lines (16 pair)	ı		ULS	ULSDG		57.72		11.43						<u> </u>	
UNBUNDLED T	RANSPORT															
co	MMON TRANSPORT (Shared) Common Transport - Per Mile, Per MOU					0.0000049									ļ	
	Common Transport - Fer Mile, Fer MOU Common Transport - Facilities Termination Per MOU					0.00049							+			+
NO	TE: INTEROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS	53 = one r	month,	DS3 and above four	months										-	
INT	EROFFICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per			U1TVX	1L5XX	0.0118								<u> </u>		
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility Termination per month			U1TVX	U1TV2	29.51	81.07	54.84	33.36	13.75		19.99		i		
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per					20.01	01.01	0 1.0 1	00.00	10.70		10.00				
	Mile per month			U1TVX	1L5XX	0.0118										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility Termination per month			U1TVX	U1TR2	29.51	81.07	54.84	33.36	13.75		19.99		i I		
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per						01.01	0	00.00							
	month	ļ	1	U1TVX	1L5XX	0.0118							1			
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility Termination per month			U1TVX	U1TV4	26.22	81.1	54.84	33.36	13.75		19.99		i I		
 	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month		-	U1TDX U1TDX	1L5XX U1TD5	0.0118 21.26	81.11	54.84	33.36	13.75	1	19.99	+		-	-
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0118							 			
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month	1		U1TDX	U1TD6	21.26	81.11	54.84	33.36	13.75		19.99				
l	EROFFICE CHANNEL - DEDICATED TRANSPORT - DS1		-										+		 	+
jat-		1	_	1	41.53/3/	0.0407		1	1		1	1	+		+	+
INT	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			U1TD1	1L5XX	0.2407								·		
INT	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			U1TD1 U1TD1	U1TF1	97.38	178.59	163.67	32.59	28.79		19.99		-		
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			U1TD1 U1TD1	U1TF1	97.38	178.59	163.67	32.59	28.79		19.99				

CATEGORY	NOTES	ELEMENT In	nterim Zone	BCS	USOC		R	ATES (\$)					OSS R	ATES (\$)		
CALEGORY	NOTES	LELIMAN II	2016	803	0300			(#)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'l
							Nonrec	urring	Nonr	ecurring	per Lon	LOK	Liecti offic-1st	Lie Ci Onic-Add	100	Addi
										onnect						
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month		U1TD3	U1TF3	Rec 1191.53	First 557.69	Add'I 325.62	First 120	Add'I 116.54	SOMEC	19.99	SOMAN	SOMAN	SOMAN	SOMAN
	INTEROFFI	ICE CHANNEL - DEDICATED TRANSPORT- STS-1														
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month		U1TS1	1L5XX	5.1										
		Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month		U1TS1	U1TFS	1165.53	557.69	325.62	120	116.54		19.99				
	LOCAL CH	IANNEL - DEDICATED TRANSPORT														
	NOTE: LOC	CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one mon	th. DS3 and a	bove=four months	5											
		Local Channel - Dedicated - 2-Wire Voice Grade Per Month	,	ULCVX	ULDV2	18.81	386.33	66.35	73.04	6.37		19.99				
		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month		ULCVX	ULDR2	18.81	386.33	66.35	73.04	6.37		19.99				
		Local Channel - Dedicated - 4-Wire Voice Grade per month		UNCVX	ULDV4	20.12	387.2	67.22	73.98	7.31		19.99				
		Local Channel - Dedicated - DS1 per month - Zone 1	1	ULDD1	ULDF1	44.63	355.06	307.53	44.24	30.42		19.99				
		Local Channel - Dedicated - DS1 per month - Zone 2	2	ULDD1	ULDF1	40.74	355.06	307.53	44.24	30.42		19.99				
		Local Channel - Dedicated - DS1 per month - Zone 3	3	ULDD1	ULDF1	42.95	355.06	307.53	44.24	30.42	1	19.99	1			-
		Local Channel - Dedicated - DS3 - Per Mile per month Local Channel - Dedicated - DS3 - Facility Termination per month		ULDD3 ULDD3	1L5NC ULDF3	8.98 583.57	903.34	528.05	238.2	166.62	1	19.99	+			
		Local Channel - Dedicated - DSS - Pacinty Termination per month		ULDS1	1L5NC	8.98	303.54	525.05	200.2	100.02	 	13.33	1			—
		Local Channel - Dedicated - STS-1 - Facility Termination per month		ULDS1	ULDFS	550.34	903.34	528.05	238.2	166.62		19.99				
MULTIPLEX	ERS							L								
		Channelization - DS1 to DS0 Channel System		UXTD1	MQ1	139.65	182.14	125.19	21	19.52	1	19.99	1			<u> </u>
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		UDL	1D1DD	1.63	13.16	9.43								
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month		UDN UEA	1D1VG	3.5 0.7676	13.16 13.16	9.43 9.43								1
		Voice Grade COCI - DS1 to DS0 Channel System - per month DS3 to DS1 Channel System per month		UXTD3	MQ3	194.82	356.4	188	66.3	63.44		19.99				
		STS1 to DS1 Channel System per month		UXTS1	MQ3	194.82	330.4	100	00.5	03.44		19.99				
		DS3 Interface Unit (DS1 COCI) used with Loop per month		USL	UC1D1	14.53	13.16	9.43				10.00				
DARK FIBE	R															
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -														
		Local Channel		UDF	1L5DC	48										
		NRC Dark Fiber - Local Channel Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		UDF	UDFC4		1278.61	275.82	632.07	394.05		19.99				
		Interoffice Channel		UDF	1L5DF	31.51										
		NRC Dark Fiber - Interoffice Channel		UDF	UDF14	31.31	1278.61	275.82	632.07	394.05		19.99				
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		<u>, </u>												
		Local Loop .		UDF	1L5DL	48										
		NRC Dark Fiber - Local Loop		UDF	UDFL4		1278.61	275.82	632.07	394.05		19.99				<u> </u>
TRANSPOR	T OTHER															
	Optional Fe	eatures & Functions:														
		Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel		UNC1X	CCOEF		184.91	23.82	1.99	0.78		19.99				
		Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel		UNC1X	CCOSF		184.91	23.82	1.99	0.78		19.99				
8XX ACCES		IT SCREENING														
		8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved		OHD	N8R1X		10.05	1.19		1		19.99	1			
		8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS		OHD OHD	N8FTX		30.59 30.59	3.22 3.22		1	1	19.99 19.99	1			
		8XX Access Ten Digit Screening, Per 8XX No. Established With POTS 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number		OHD	N8FCX		6.97	3.22		1		19.99	1			
		8XX Access Ten Digit Screening, Customized Alea of Service Fel 6XX Number 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR		310	1.07 07		5.57	5.45			T	15.55				
		Requested Per 8XX No.		OHD	N8FMX		8.16	4.67	1		1	19.99	1			
		8XX Access Ten Digit Screening, Change Charge Per Request		OHD	N8FAX		11.24	1.19				19.99				
		8XX Access Ten Digit Screening, Call Handling and Destination Features		OHD	N8FDX		6.97					19.99				
		INXX Access Top Digit Screening w/ RXX No. Delivery, per guery		OHD		0.001										
		8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per query					1		1	1	1	1				
		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with		OHD		0.0011										
		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD		0.0011										
		BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query 8XX Access Ten Digit Screening, w/ POTS No. Delivery, per query		OHD OHD		0.0011 0.001										
		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query														
		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query 8XX Access Ten Digit Screening, w/ POTS No. Delivery, per query 8XX Access Ten Digit Screening w/ POTS No. Delivery, with Optional Complex Features, per query		OHD		0.001										
LINE INFOR		8XX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query 8XX Access Ten Digit Screening, w/ POTS No. Delivery, per query 8XX Access Ten Digit Screening w/ POTS No. Delivery, with Optional Complex Features, per query		OHD		0.001										
LINE INFOR		BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD		0.001 0.0011 0.00006										
LINE INFOR		BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD OQT OQU	NDDDV	0.001	407.6					10.00				
LINE INFOR		BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD	NRPBX	0.001 0.0011 0.00006	107.6					19.99				
	MATION DA	BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD OQT OQU	NRPBX	0.001 0.0011 0.00006	107.6					19.99				
	MATION DA	BXX Access Ten Digit Screening w/BXX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD OHD OQT OQU OQT, OQU	NRPBX PT8SX	0.001 0.0011 0.00006 0.00938	107.6					19.99				
LINE INFOR	MATION DA	BXX Access Ten Digit Screening w/8XX No. Delivery for 8XX Numbers, with Optional Complex Features, per query		OHD OHD OQT OQU OQT, OQU		0.001 0.0011 0.00006 0.00938		354.95								

Part	CATEGORY	NOTES	ELEMENT	Interim	Zone	BCS	USOC		R4	ATES (\$)					OSS R	ATES (\$)		
Cold Special Contraction Power As Statistical Indian Ind											Nonre	ecurring	Submitted Elec	Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
COCK Spring Coverage, Reg and 1944 all about soon as Dates COCK Street 1.4 and 1955 1.																		
COCK Securing Labor, Pre-IEGA Message Cock								Rec					SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
COUNT Service Sample S			CCS7 Signaling Connection, Per link (B link) (also known as D link)			1DB	TPP++	16.31	354.95	354.95	174.08	174.08		19.99				
Part Part			CCS7 Signaling Usage, Per ISUP Message				CTUEC							40.00				-
Pot TTM Clark			CCS7 Signaling Osage Surrogate, per link per LATA CCS7 Signaling Point Code, per Originating Point Code Establishment or Change.			IDB	51056	329.96						19.99				
COUNT Service Processor						1DB	CCAPO		40	40				19.99				
### SERVICE																		
CALISO MARIE (CAM) SERVICE ONM or DE Cheers, For Chery OXY OXY OXY OXY OXY OXY OXY OXY OXY OXY			Per Stp Affected			1DB	CCAPD		8	8				19.99				
CALISO MARIE (CAM) SERVICE ONM or DE Cheers, For Chery OXY OXY OXY OXY OXY OXY OXY OXY OXY OXY	E044 0EBV	105																
CHANTER DR. Downer, Part Change Cha	E911 SERVI	ICE																
CANAS TOR Schowler, Fee Charty COY																		
CAMPAIN Non-Distance New CAMPAIN Non-Distance New Non-Distance N	CALLING N	AME (CNAN																
COMM Non-Displace Covery, NRC, applies when using the Character Basser User																		
Interface (CHII)			CNAM for Non DB Owners, Per Query			OQV		0.01										
Interface (CHID)																		
Interface (CHID)			CNAM (Non-Databs Owner) NRC applies when using the Character Rased User															
LIP OUER 'SERVICE OPERATOR SERVICES AND DIRECTORY ASSISTANCE OPERATOR SERVICES AND DIRECTORY ASSISTANCE OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING Impact Operator (PA Assistance Carl - Using BST I/DB 12						OOV	СDDCH		595	595				19 99				
### OPERATOR SERVICES AND DIRECTORY ASSISTANCE OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR CALL PROCES			manage (error)			041	JUDUIT		555	555				10.00				
### OPERATOR SERVICES AND DIRECTORY ASSISTANCE OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR SERVICES OPERATOR CALL PROCESSING OPERATOR CALL PROCES																		
OPERATOR SERVICES AND DIRECTORY ASSISTANCE																		
### OPERATOR CALL PROCESSING Dear Call Processing Oper Provised, Per Min - Using BST LDB 1,24	LNP QUERY	SERVICE																
### OPERATOR CALL PROCESSING Dear Call Processing Oper Provised, Per Min - Using BST LDB 1,24							1				-		1	-				+
### OPERATOR CALL PROCESSING Dear Call Processing Oper Provised, Per Min - Using BST LDB 1,24		OPERATOR	S SERVICES AND DIRECTORY ASSISTANCE															
Control all Processing - Open Provided Per Man - Using ST LUBS 1.2		0. 2.00.	OLIVIOLO VIIID DIRECTORI VIOCIOTATION															
Coper_Call Processing-Open_Provided, Per Min Using Foreign LIDB	OPERATOR	CALL PRO	CESSING															
Oper_Call Processing - FlyAuthoristed, per Call - Using SET LIDS 0.2			Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.2										
Oper_Call Processing - Puly Automated, per Call - Using Foreign LIDB			Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB															
NWARD OPERATOR SERVICES Investor Operator Services - Verification Per Call Investor Operator Services - Verification and Emergency Interrupt - Per Call BRANDING - OPERATOR CALL PROCESSING Recording of Custom Branded CN Announcement CBAOS TO00 TO			Oper, Call Processing - Fully Automated, per Call - Using BST LIDB					0.2										
Inward Operator Servicies - Verification, Per Call 1 1 1 1 1 1 1 1 1			Oper. Call Flocessing - Fully Automateu, per Call - Osing Foreign LIDB					0.2										
Inward Operator Servicies - Verification, Per Call 1 1 1 1 1 1 1 1 1	INWARD OF	PERATOR S	ERVICES															
BRANDING - OPERATOR CALL PROCESSING Recording of Custom Branded OA Arrouncement CBAOS 77000 119.99 19.99			Inward Operator Services - Verification, Per Call					1										
Recording of Custom Branded OA Announcement CBAOS 7000 7000 19.99			Inward Operator Services - Verification and Emergency Interrupt - Per Call					1.95										
Recording of Custom Branded OA Announcement CBAOS 7000 7000 19.99	DD 4 NDING	0050476	D OALL PROGESSING															
Loading of Custom Branded OA Announcement per shelf/NAV	BRANDING						CDAOC		7000	7000				40.00				
DIRECTORY ASSISTANCE ACCESS SERVICE Directory Assistance Access Service Cals, Charge Per Call Directory Assistance Call Completion Access Service (DACC). Directory Assistance Call Completion Access Service (DACC). Directory Assistance Call Completion Access Service (DACC). Directory Transport Call Completion Access Service (DACC). Directory Transport Call Completion Access Service (DACC). Directory Transport Call Call Completion Access Service (DACC). Directory Transport Call Call Call Call Call Call Call Cal			Leading of Custom Branded OA Announcement per shelf/NAV												10.00	10.00		
DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)			Loading of Custom Branded OA Affiliating the first per shell/NAV				CDAOL		300	300				13.33	13.33	15.55		+
DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)	DIRECTORY	Y ASSISTAN	ICE SERVICES															
DIRECTORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)		DIRECTOR	Y ASSISTANCE ACCESS SERVICE															
Directory Assistance Call Completion Access Service (DAČC), Per Call Attempt 0.1			Directory Assistance Access Service Calls, Charge Per Call					0.275										
Directory Assistance Call Completion Access Service (DAČC), Per Call Attempt 0.1																		
UNBRANDING DIRECTORY TRANSPORT Directory Transport - Local Channel DS1 Directory Transport - DS1 Level Interoffice Per Mile 0.45 0.45 0.55.05 298.18 231.18 19.99 0.45		DIRECTOR						0.4										
DiRECTORY TRANSPORT Directory Transport - Local Channel DS1 S6.32 637.46 546.94 19.99			Directory Assistance Call Completion Access Service (DACC), Per Call Attempt				1	0.1		1	1		1	1				+
DIRECTORY TRANSPORT DIRECTORY Transport - Local Channel DS1 Directory Transport - Local Channel DS1 Directory Transport - DS1 Level Interoffice Per Facility Termination 0.45		UNBRANDI	NG								1		1	1				
Directory Transport - Dis1 Level Interoffice Per Mile 0.45 0.50											1		1	1				
Directory Transport - DS1 Level Interoffice Per Mile 0.45 Directory Transport - DS1 Level Interoffice Per Facility Termination 55.05 298.18 231.18 19.99			Directory Transport - Local Channel DS1		_			36.32	637.46	546.94	<u></u>			19.99				
Switched Common Transport Per DA Access Service Per Call			Directory Transport - DS1 Level Interoffice Per Mile			-		0.45			1			<u> </u>				<u> </u>
Switched Common Transport Per DA Access Service Per Call Per Mile Access Tandem Switching Per DA Access Service Per Call Access Tandem Switching Per DA Access Service Per Call Directory Transport - Installation NRC, Per Trunk or Signaling Connection Directory Transport - Installation NRC, Per Trunk or Signaling Connection DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service (Darge Per Listing Directory Assistance Data Base Service, per month DBSOF 150 BRANDING - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch Selective Routing Per Unique Line Class Code Per Request Per Switch VIRTUAL COLLOCATION VIRTUAL Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 2-wire Cross Connects (loop) or Line Splitting I UEPSR, UEPSB VE1LS 0.31 54.21 51.07 19.99 Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting Vertical Collocation - 2 Wire Cross Connects (loop) or Line Splitting								55.05	298.18	231.18	1		1	19.99				
Access Tandem Switching Per DA Access Service Per Call Directory Transport - Installation NRC, Per Trunk or Signaling Connection Directory Assistance Data Base Service (DADS) Directory Assistance Data Base Service (Date Per Listing Directory Assistance Data Base Service, per month Directory Assistance Data Base Service, per month Discord Assistance Data Base Service, per month DBSOF 150 BRANDING - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch AMT CBADA 3000 3000 SELECTIVE ROUTING Selective Routing Per Unique Line Class Code Per Request Per Switch USRCR 229.65 229.65 19.99 19.99 VIRTUAL COLLOCATION Virtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 2-wire Cross Connects (loop) for Line Splitting I UEPSR, UEPSB VE1LS 0.31 54.21 51.07 19.99 VEAC2 0.31 54.21 51.07 19.99			Switched Common Transport Per DA Access Service Per Call Per Mile					0.000175										
Directory Transport - Installation NRC, Per Trunk or Signaling Connection DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listing Directory Assistance Data Base Service, per month DBSOF 150 BRANDING - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch SELECTIVE ROUTING Selective Routing Per Unique Line Class Code Per Request Per Switch VIRTUAL COLLOCATION VIRTUAL COLLOCATION VIrtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 2-wire Cross Connects (loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Solution Selection - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connects (Loop) for Line Splitting VIRTUAL Collocation - 2-wire Cross Connect			Access Tandem Switching Per DA Access Service Per Call										1					
DIRECTORY ASSISTANCE DATA BASE SERVICE (DADS) Directory Assistance Data Base Service Charge Per Listing Directory Assistance Data Base Service (Darge Per Listing Directory Assistance Data Base Service, per month Directory Assistance Data Base Service, per month DBSOF 150 BRANDING - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch AMT CBADA 3000 3000 Loading of Custom Branded Announcement per DRAM Card/Switch AMT CBADC 690 690 SELECTIVE ROUTING Selective Routing Per Unique Line Class Code Per Request Per Switch USRCR 229.65 229.65 19.99 19.99 VIRTUAL COLLOCATION Virtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting VIRTUAL COLLOCATION UEAC2 0.31 54.21 51.07 19.99			Directory Transport - Installation NRC, Per Trunk or Signaling Connection					5.000.00	501.98	13.32				19.99				1
Directory Assistance Data Base Service, Charge Per Listing Directory Assistance Data Base Service, per month DBSOF 150																		
Directory Assistance Data Base Service, per month DBSOF 150		DIRECTOR							-									
BRANDING - DIRECTORY ASSISTANCE Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch SELECTIVE ROUTING Selective Routing Per Unique Line Class Code Per Request Per Switch VIRTUAL COLLOCATION Virtual Collocation - 2-wire Cross Connects (loop)			Directory Assistance Data Base Service Charge Per Listing				DDCCC	0.04		1	1		1	1				
Custom Branding Announcement, per Recording to be used with the provision of DA Loading of Custom Branded Announcement per DRAM Card/Switch AMT CBADA 3000 3000 BELECTIVE ROUTING Selective Routing Per Unique Line Class Code Per Request Per Switch USRCR 229.65 229.65 19.99 19.99 VIRTUAL COLLOCATION ueanl,uea,udn,udc, usl,uhl,ucl,ueq UEAC2 0.31 54.21 51.07 19.99 Virtual Collocation - 2-wire Cross Connects (loop) for Line Splitting UEAC2 19.99 19.99 UEAC2 19.99 Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splitting UEAC2 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 Virtual Collocation - 2-wire Cross Connects (Loop) for Line Splitting UEAC2 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99 VIRTUAL COLLOCATION 19.99	BDANDING						DRSOL	150			-		1	-			-	
Loading of Custom Branded Announcement per DRAM Card/Switch	PUNICHANG	- DINECTO				ΔΜΤ	CBADA		3000	3000	1		1	1				+
SELECTIVE ROUTING			Loading of Custom Branded Announcement per DRAM Card/Switch										1					†
Selective Routing Per Unique Line Class Code Per Request Per Switch USRCR 229.65 229.65 19.99 19.99			2. 2 Section Standard Fundamental per Strain Galdownton			, 2411	55,100		550	550	1		1	1				
VIRTUAL COLLOCATION	SELECTIVE	ROUTING																
VIRTUAL COLLOCATION																		
Ueanl,uea,udn,udc, UEAC2 0.31 54.21 51.07 19.99			Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		229.65	229.65			1		19.99	19.99		<u> </u>
Ueanl,uea,udn,udc, UEAC2 U. UEAC2 U. U. U. U. U. U. U. U. U. U. U. U. U.	VIDTUAL OF	011004710	NI								1		1	1				
Virtual Collocation - 2-wire Cross Connects (loop) ual,uhl,ucl,ueq UEAC2 0.31 54.21 51.07 19.99 Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting I UEPSR, UEPSB VE1LS 0.31 54.21 51.07 19.99	VIRTUAL CO	ULLUCATIO	IN .			nearl nea udo udo												
Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting I UEPSR, UEPSB VE1LS 0.31 54.21 51.07 19.99			Virtual Collocation - 2-wire Cross Connects (loon)				UFAC2	0.31	54.21	51.07				19 99				
Virtual Collocation - 2-wire Cross Connects (port) VE1R2 0.31 54.21 51.07 19.99			Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting	1		UEPSR, UEPSB		0.31	54.21	51.07				19.99				
			Virtual Collocation - 2-wire Cross Connects (port)			, , , , ,			54.21	51.07								

CATEGORY	NOTES	ELEMENT Interim	Zone	BCS	USOC		R.	ATES (\$)					OSS R	ATES (\$)		
	NUTES	CELIMENT INCOME.	Zone	503	0300			RILO (ψ)					033 K	ATES (\$)	Incremental	Incremental
															Charge -	Charge -
											Svc Order	Svc Order	Incremental	Incremental	Manual Svc	Manual Svc
											Submitted Elec	Submitted Manually per	Charge - Manual Svc Order vs.	Charge - Manual Svc Order vs.	Order vs. Electronic-Disc	Order vs. Electronic-Dis
											per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'I
							Nonrec	urring	Nonr	curring						
										onnect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Virtual Collocation - 4-wire Cross Connects (loop)		uea,uhl,ucl,udl	UEAC4	0.62	54.23	50.96				19.99				
		Virtual Collocation - 4-wire Cross Connects (port)		01.0	VE1R4	0.62	54.23	50.96				19.99				
		Virtual Collocation - 2-Fiber Cross Connects		CLO	CNC2F	15.64	41.56	29.82					19.99	19.99	19.99	19.99
		Virtual Collocation - 4-Fiber Cross Connects Virtual Collocatin - DS1 Cross Connects		CLO USL,ULC,CLO	CNC4F CNC1X	28.11	50.53 44.07	38.78	12.76	11.53			19.99	19.99	19.99	19.99
		virtual Collocatin - DST Cross Connects		USL,ULC,CLU	CNCTA	1.5	44.07	31.86	12.76	11.53						
AIN CELEC	IVE CARRI	ER ROUTING														
AIN SELEC	IVE CARRI	Regional Service Establishment		SRC	SRCEC		391788					19.99				
		End Office Establishment		SRC	SRCEO		320.53	320.53				19.99				
		Line/Port NRC, per end user		SRC	SRCLP		2.06	2.06				19.99				
		Query NRC, per query		SRC		0.000448										
AIN - BELLS	OUTH AIN	SMS ACCESS SERVICE														
AIN - BELLS	OUTH AIN	TOOLKIT SERVICE														
								1								
ODUF/EDOL	JF/ADUF/CI	MDS														
	ACCESS D	AILY USAGE FILE (ADUF)														
		ADUF: Message Processing, per message				0.004										
		ADUF: Data Transmission (CONNECT:DIRECT), per message				0.001										
	ENHANCED	OPTIONAL DAILY USAGE FILE (EODUF)														
		EODUF: Message Processing, per message				0.004										
	OPTIONAL	DAILY USAGE FILE (ODUF)														
		ODUF: Recording, per message				0.0008611										
		ODUF: Message Processing, per message				0.0032357										
		ODUF: Message Processing, per Magnetic Tape provisioned ODUF: Data Transmission (CONNECT:DIRECT), per message				55.68										
		ODUF: Data Transmission (CONNECT:DIRECT), per message				0.0000365										
		(0.0000303										
						0.0000303										
ENHANCED		D LINK (EELs)				0.0000303										
	EXTENDED	D LINK (EELs)														
	EXTENDED NOTE: New	D LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL:	; Ft. La	uderdale, FLI; Nash	nville, TN; N											
	EXTENDED NOTE: New NOTE: Chai	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhili, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e	except S	Switch As Is Charge	∍.	New Orleans, LA										
	NOTE: New NOTE: Char NOTE: In all	D LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL iothe-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a	except S are conv	Switch As Is Charge verted to UNE rates	e. s. A Switch	New Orleans, LA		tly combined	facilities cor	overted to Ut	NEs.(Non-red	curring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhili, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e	except S are conv	Switch As Is Charge verted to UNE rates	e. s. A Switch	New Orleans, LA		tly combined	facilities cor	overted to Ut	IEs.(Non-red	curring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Otte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a porgia, the EEL network elements apply to ordinarily combined network elements per the GA P:	except S are conv SC orde	Switch As Is Charge verted to UNE rates	e. s. A Switch	New Orleans, LA		tly combined	facilities cor	overted to Ut	IEs.(Non-red	curring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	D LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL iothe-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a	except S are conv SC orde	Switch As Is Charge verted to UNE rates	e. s. A Switch	New Orleans, LA		tly combined	facilities cor	overted to Ut	NEs.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1	except S are conv SC orde	Switch As Is Charge verted to UNE rates	e. s. A Switch	New Orleans, LA		tly combined	facilities cor	werted to Ut	Es.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Otte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a porgia, the EEL network elements apply to ordinarily combined network elements per the GA P:	except S are conv SC orde	Switch As Is Charge verted to UNE rates er.(No Switch As Is	e. s. A Switch Charge.)	New Orleans, LA;		tly combined	facilities cor	verted to Ut	Es.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Mie	except S are conv SC orde	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX	e. s. A Switch Charge.)	New Orleans, LA; As Is Charge ap		tly combined	facilities cor	verted to Ut	Es.(Non-rec	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1	except S are conv SC orde	Switch As Is Charge verted to UNE rates er.(No Switch As Is	e. s. A Switch Charge.)	New Orleans, LA;		tly combined	facilities cor	verted to Ut	IEs.(Non-rec	curring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Mi	except Sare converse	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX	c. A Switch Charge.) UEAL2	New Orleans, LA; As Is Charge ap 17.27 32.32		tty combined	facilities cor	werted to Ut	Es.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Michte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a aorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3	except S are conv SC orde	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX UNCVX	e. s. A Switch Charge.) UEAL2 UEAL2	As Is Charge ap 17.27 32.32 55.78		tly combined	facilities cor	verted to Ut	IEs.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Ottler-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Cop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Interoffice Transport - Dedicated - DS1 combination - Per Mile per month	except Sare converse	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX	c. A Switch Charge.) UEAL2	New Orleans, LA; As Is Charge ap 17.27 32.32		tly combined	facilities cor	verted to Ut	IEs.(Non-rec	curring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channellization System Per Month	except Sare converse	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX	e. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX	As is Charge ap 17.27 32.32 55.78 0.2407		ttly combined	facilities cor	werted to Ut	Es.(Non-red	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Interest of State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Interest of States, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month	except Sare converse	witch As Is Charge rerted to UNE rates er.(No Switch As Is UNCVX UNCVX UNCVX UNC1X UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	As Is Charge and 17.27 32.32 55.78 0.2407 97.38		tly combined	facilities cor	everted to Ul	Es.(Non-rec	urring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miche-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 John 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Esch Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	except Sare converse	witch As Is Charge verted to UNE rates or.(No Switch As Is UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676		tty combined	facilities con	overted to Ul	Es.(Non-rec	urring rates	do not apply.	>		
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL tolter-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade CCC1 - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport Combination - Zone 1	except Sare converse	Switch As Is Charge verted to UNE rates er.(No Switch As Is UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38		tly combined	facilities con	werted to UI	Es.(Non-rec	urring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; lotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	except Sare conviscond	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX	e. S. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2	As Is Charge and 17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27		tty combined	facilities con	overted to UI	Es.(Non-rec	urring rates	do not apply.)		
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Ottor-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport Dedicated - DS1 combination - Facility Termination per DS1 Channellazion System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1	except Sare converse	witch As Is Charge verted to UNE rates or.(No Switch As Is UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676		tty combined	facilities cor	overted to Ul	Es.(Non-rec	urring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Ideastonia-Rockhill, NC: Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	except Sare convision of the convision o	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCTX UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	a. S. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27		tty combined	facilities con	everted to Ut	Es.(Non-rec	curring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Ottoe-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2	except Sare conviscond	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX	a. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78		tty combined	facilities cor	iverted to UI	IES.(Non-rec	urring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Ideastonia-Rockhill, NC: Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	except Sare convision of the convision o	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCTX UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	a. S. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27		tty combined	facilities con	overted to UN	Es.(Non-rec	curring rates	do not apply.			
	NOTE: New NOTE: Char NOTE: In all NOTE: In Gr	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL office-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pictor of the Carlot of th	except Sare convision of the convision o	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX	a. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 UTF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Ottor-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch - As-Is Charge	except Sare converse	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX	a. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78		tty combined	facilities con	overted to Uf	IEs.(Non-rec	urring rates	do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; lotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Month Seach Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Nonrecurring Currently Combined Nonrecurring Currently Combined Nonrecurring Currently Combined Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	except Sare converse	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX	a. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 UTF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Sol Canadia COCI - DS1 To Ds0 Interface - Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL)	except Sare converse	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	a. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TH MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				IEs.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Oliter-Gastoniar-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COC1 - DS1 to DS0 Channel System combination - per month Norrecurring Currently Combined Network Elements Switch - As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 EACH Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COC1 - DS1 to DS0 Channel System combination - per month	except Sare come SC order	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX	a. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 UTF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL office-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Proceedings of the EEL network elements apply to ordinarily combined network elements per the GA Proceedings of the EEL network elements apply to ordinarily combined network elements per the GA Proceedings of the EEL network elements apply to ordinarily combined network elements per the GA Proceedings of the EEL network elements per the GA Proceedings of the EEL network elements per the GA Proceedings of the EEL network elements per the GA Proceedings of the EEL network elements per the GA Proceedings of the EEL network elements per the GA Proceedings of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements of the EEL network elements	except S are converted to the converted	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	a. s. A Switch Charge.] UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UTATE1 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				IES.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miami, FL; Miche-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month Voice Grade COC1 - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COC1 - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch - As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	except Sare come SC order	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	a. s. A Switch Charge.) UEAL2 UEAL2 UEAL2 1L5XX U1TH MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2 UDAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miche-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a corgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Month S1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	are commission of the commissi	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	a. a. Switch. Charge.) UEAL2 UEAL2 UEAL2 11.5XX U1TF1 MQ1 1D1VG UEAL2 11.5XX UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				IEs.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miche-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	except S are converted to the converted	witch As Is Charge yerled to UNE rates or. (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCTX UNCTX UNCTX UNCTX UNCTX UNCYX UNCYX UNCVX	3. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 11.5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 UEAL4	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.2676 20.92 39.14 67.57	plies to curren				IEs.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; Miche-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Part Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To DS0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport Combination - Dedicated - DS1 combination - Per Mile Per Month	are commission of the commissi	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX	a. a. Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 11.5XX U1TF1 MQ1 1D1VG UEAL2 1D4L2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4 UEAL4	17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 First 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination 1- Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 - Facility Termination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 - Facility Termination - Per Mile Per Month	are commission of the commissi	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCVX	a. s. A Switch Charge.] UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL3 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4	32.32 55.78 0.2407 97.38 17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				IEs.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL; lotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Month Seach Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Nonrecurring Curre	are commission of the commissi	witch As Is Charge yerred to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNCVX UNC1X UNCVX UNCVX UNC1X UNCVX UNC1X UNCVX UNCVX UNC1X UNCVX UNCVX UNCVX UNCVX UNC1X UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X	a. A Switch Charge.) UEAL2 UEAL2 UEAL2 11.5XX U1TF1 MQ1 1D1VG UEAL2 1D4L2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4	17.27 32.32 55.78 0.2407 97.38 139.65 17.27 32.32 55.78 0.7676 17.27 32.32 55.78 0.7676 47.27 32.32 55.78 0.7676 20.92 39.14 67.57 0.2407 97.38 139.65	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL totte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA Pi CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Per Mile per month DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Norrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month Channelization - Channel System DS1 to DS0 combination - Per month	are commission of the commissi	witch As Is Charge verted to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCVX	a. s. A Switch Charge.] UEAL2 UEAL2 UEAL2 1L5XX U1TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL3 UEAL4 UEAL4 UEAL4 UEAL4 UEAL4	32.32 55.78 0.2407 97.38 17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 17.27 32.32 55.78 0.7676	plies to curren				IES.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Other-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Tone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combina	are commission of the commissi	witch As Is Charge yerled to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCVX	3. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 11.5XX U11TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL3	As Is Charge and 17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 20.92 39.14 67.57 0.2407 97.38 139.65 0.7676	plies to curren				Es.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	LINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL office-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements apply to ordinarily combined network elements per the GA Pt CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month interoffice Transport - Dedicated - DS1 combination - Per Mile per month Voice Grade COC1 - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Society of Schannel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CC GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 - Pacility Termination Per Month Channel System DS1 to DS0 combination - Per month Additional 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 - Pacility Termination Per Month Channel System DS1 to DS0 combination - Per month Additional 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transp	are commission of the commissi	witch As Is Charge yerred to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNCVX UNC1X UNCVX UNC1X UNCVX UNCVX UNC1X UNCVX UNC1X UNCVX UNCVX UNC1X UNCVX UNCVX UNCVX UNCVX UNC1X UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X	a. A Switch Charge.) UEAL2 UEAL2 UEAL2 11.5XX U1TF1 MQ1 1D1VG UEAL2 1D4L2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL4	17.27 32.32 55.78 0.2407 97.38 139.65 17.27 32.32 55.78 0.7676 17.27 32.32 55.78 0.7676 47.27 32.32 55.78 0.7676 20.92 39.14 67.57 0.2407 97.38 139.65	plies to curren				IES.(Non-rec		do not apply.			
	EXTENDED NOTE: NoTE: Chan NOTE: Chan NOTE: In all NOTE: In G	DLINK (EELs) EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL; Miami, FL Other-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use all rates below e states, EEL network elements shown below also apply to currently combined facilities which a sorgia, the EEL network elements sapply to ordinarily combined network elements per the GA P: CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Tone 1 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 2 First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per DS1 Channelization System Per Month Voice Grade COCI - DS1 To Ds0 Interface - Per Month Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 1 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge CE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combina	are commission of the commissi	witch As Is Charge yerled to UNE rates or (No Switch As Is UNCVX UNCVX UNCVX UNCVX UNC1X UNC1X UNC1X UNC1X UNC1X UNCVX	3. A Switch Charge.) UEAL2 UEAL2 UEAL2 UEAL2 11.5XX U11TF1 MQ1 1D1VG UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL2 UEAL3	As Is Charge and 17.27 32.32 55.78 0.2407 97.38 139.65 0.7676 20.92 39.14 67.57 0.2407 97.38 139.65 0.7676	plies to curren				IES.(Non-rec		do not apply.			

RY	NOTES	ELEMENT	Interim Zone	BCS	USOC		R	ATES (\$)					OSS R	ATES (\$)		
	NOTES	CLEMENT	Internii Zone	BC3	0300			ATES (\$)			Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental I Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Dis
_					+		1	1		l .	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
+			 		+		Nonrec	urring		ecurring onnect	-					+
-						Rec	First	Add'l	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
\neg		Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport				1100	1	7.001	11154	7001	COMEO	COMPA	COMPA	COMPA	COMPA	
		Combination - Zone 3	3	UNCVX	UEAL4	67.57										
		Normalia Constitution Internal Florida Control Acts Observed		LINGAY	1110000		44.40	44.40	40.04	40.04		40.00				
-+		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99				+
4-	WIRE 56 H	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	PORT (EEL)													
		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination) (
_		 Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination 	1	UNCDX	UDL56	35.92										
		First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2	2	UNCDX	UDL56	40.32										
-		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		UNCDA	ODLOG	40.32										+
		- Zone 3	3	UNCDX	UDL56	37.9										
_		Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		UNC1X	1L5XX	0.2407										
		Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per Month		UNC1X	U1TF1	97.38						19.99				
-		Channelization - Channel System DS1 to DS0 combination Per Month		UNC1X	MQ1	139.65						19.99				+
-		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		UNCDX	1D1DD	1.63										+
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport														
		Combination - Zone 1	1	UNCDX	UDL56	35.92			1		1	19.99				1
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 2	2	UNCDX	UDL56	40.32						19.99				
-		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	2	UNCDX	UDLS6	40.32						19.99				+
		Combination - Zone 3	3	UNCDX	UDL56	37.9						19.99				1
		OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-														
		64kbs)		UNCDX	1D1DD	1.63										
		Normalia Constitution Internal Florida Control Acts Observed		LINGAY	1110000		44.40	44.40	40.04	40.04		40.00				
-	WIDE 64 k	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	DORT (EEL)	UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99				+
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	PORT (EEL)													+
		- Zone 1	1	UNCDX	UDL64	35.92										
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination														
_		- Zone 2	2	UNCDX	UDL64	40.32										
		First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3	3	UNCDX	UDL64	37.9										
+		Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	3	UNC1X	1L5XX	0.2407										+
-		Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per		ONOTA	TEOAA	0.2401										+
		Month		UNC1X	U1TF1	97.38										
		Channelization - Channel System DS1 to DS0 combination Per Month		UNC1X	MQ1	139.65										
		OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		UNCDX	1D1DD	1.63	0	0								
-		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		UNCDX	טטוטו	1.03	0	0								+
		Combination - Zone 1	1	UNCDX	UDL64	35.92										
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport														
		Combination - Zone 2	2	UNCDX	UDL64	40.32										
		Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport	3	UNCDX	UDL64	27.0										
+		Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-	3	UNCDX	UDL64	37.9	1					1				+
		64kbs)		UNCDX	1D1DD	1.63										1
		,														
\perp		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	\longrightarrow	UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99				1
1.	WIRE DS1	I DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	PT (FFL)		+ +		1				1					+
	THINE DOT	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1	1	UNC1X	USLXX	50.26	1	-	 	 	1					+
+		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2	2	UNC1X	USLXX	94.06										1
		4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3	3	UNC1X	USLXX	162.34										
+		Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		UNC1X	1L5XX	0.2407	1				1					+
		Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month		UNC1X	U1TF1	97.38										
+		INOTIGE .	 	UNCIA	ULIFI	ər.30	1	-	 	 	1					+
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	<u> </u>	UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99			<u> </u>	<u> </u>
						-	1						ļ	ļ		
4-		DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR		LINGAY	HOLVE	50.00	1		1	1		1				1
+		First DS1Loop in DS3 Interoffice Transport Combination - Zone 1	1 2	UNC1X	USLXX	50.26			-	-						+
+		First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3	3	UNC1X UNC1X	USLXX	94.06 162.34	1	-	 	 	1					+
		Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month		UNC3X	1L5XX	5.1	1		†	1						1
-		Interoffice Transport - Dedicated - DS3 - Facility Termination per month		UNC3X	U1TF3	1191.53										
Ⅎ			1 7 7	UNC3X	MQ3	194.82										4
		DS3 to DS1 Channel System combination per month						11	1	1	1	1	1	11	1	1
		DS3 Interface Unit (DS1 COCI) combination per month		UNC1X	UC1D1	14.53							-			
#		DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1	1 2	UNC1X	USLXX	50.26										
		DS3 Interface Unit (DS1 COCI) combination per month	1 2 3													

RY NOTES	ELEMENT Interim	Zone	BCS	USOC		R	ATES (\$)					OSS R	ATES (\$)		
										Svc Order Submitted	Svc Order Submitted	Incremental Charge - Manual	Incremental	Incremental Charge - Manual Svc Order vs.	Charge Manual St
										Elec per LSR	Manually per LSR	Svc Order vs. Electronic-1st	Svc Order vs.	Electronic-Disc	Electronic-l
						Nonrec	urring	Nonre	curring	perLSK	LSK	Electronic-1st	Electronic-Add I	1St	Add'l
								Disc	onnect						
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC3X	UNCCC		11.19	11.19	13.91	13.91		19.99				
	GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL) WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	1	UNCVX	UEAL2	17.27										+
	WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2	2	UNCVX	UEAL2	32.32										+
2-W	VireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3	3	UNCVX	UEAL2	55.78										
Inte	eroffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month eroffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility		UNCVX	1L5XX	0.0118										+
	mination per month		UNCVX	U1TV2	29.51						19.99				
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC		11.19	11.19	13.91	13.91		19.99				+
4-WIRE VOICE	GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPORT (EEL)														1
	VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1	1	UNCVX	UEAL4	20.92										
4-W	VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2 VireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3	2	UNCVX	UEAL4 UEAL4	39.14 67.57	 	-			1					+
	eroffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month	3	UNCVX	1L5XX	0.0118	 	 								+
Inte	eroffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility														1
Terr	mination per month		UNCVX	U1TV4	26.22		1			1					-
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC		11.19	11.19	13.91	13.91		19.99				
															1
	EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)		UNC3X	1L5ND	11.53										-
High	h Capacity Unbundled Local Loop - DS3 combination - Per Mile per month h Capacity Unbundled Local Loop - DS3 combination - Facility Termination per		UNC3X	ILSIND	11.53										+
mor	nth		UNC3X	UE3PX	379.72										
	eroffice Transport - Dedicated - DS3 - Per Mile per month eroffice Transport - Dedicated - DS3 combination - Facility Termination per per		UNC3X	1L5XX	5.1										-
mor			UNC3X	U1TF3	1191.53										
															1
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC3X	UNCCC		11.19	11.19	13.91	13.91		19.99				-
STS1 DIGITAL I	EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EEL)														+
	h Capacity Unbundled Local Loop - STS1 combination - Per Mile per month h Capacity Unbundled Local Loop - STS1 combination - Facility Termination per		UNCSX	1L5ND	11.53										*
			LINICOV	LIDI C4	204.70										
mor Inte	ntri eroffice Transport - Dedicated - STS1 combination - Per Mile per month		UNCSX	UDLS1 1L5XX	394.76 5.1										+
	eroffice Transport - Dedicated - STS1 combination - Facility Termination per														1
mor	nth		UNCSX	U1TFS	1165.53										
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCSX	UNCCC		11.19	11.19	13.91	13.91		19.99				
			0.1007	0.1000			11110	10.01	10.01		10.00				
	XTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)		LINIONIY	1141.07	20.00										
	st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1 st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2	1 2	UNCNX	U1L2X U1L2X	23.66 44.28	1	+								+
	st 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3	3	UNCNX	U1L2X	76.42										1
	eroffice Transport - Dedicated - DS1 combination - Per Mile		UNC1X	1L5XX	0.2407										
Inte	eroffice Transport - Dedicated - DS1 combintion - Facility Termination per month annelization - Channel System DS1 to DS0 combination - per month		UNC1X UNC1X	U1TF1 MQ1	97.38 139.65										+
	rire ISDN COCI (BRITE) - DS1 to DS0 Combination - per month		UNCNX	UC1CA	3.5										1
	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone														1
1	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone	1	UNCNX	U1L2X	23.66										+
2	unional 2-wire IDSN 200p in Same DS finteronice Transport Combination - Zone	2	UNCNX	U1L2X	44.28										
Add	ditional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone														1
3	vire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month	3	UNCNX	U1L2X UC1CA	76.42										-
2-W	Mie ISBN COCI (BKTE) - BST to BSO Channel System combinitatori per montin		UNCINA	UCTCA	3.5										+
Non	nrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99				
4 WIDE DO4 DI	GITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPORT (EEL)			+		-	1								┼
	st DS1 Loop in STS1 Interoffice Transport Combination - Zone 1	1	UNC1X	USLXX	50.26		+			+					+
Firs	st DS1 Loop in STS1 Interoffice Transport Combination - Zone 2	2	UNC1X	USLXX	94.06										
	st DS1 Loop in STS1 Interoffice Transport Combination - Zone 3	3	UNC1X	USLXX	162.34	ļ									
Inte	eroffice Transport - Dedicated - STS1 combination - Per Mile Per Month eroffice Transport - Dedicated - STS1 combination - Facility Termination		UNCSX	1L5XX U1TFS	5.1 1165.53	1	+								+
STS	S1 to DS1 Channel System conbination per month		UNCSX	MQ3	194.82										t
DS3	3 Interface Unit (DS1 COCI) combination per month		UNC1X	UC1D1	14.53										
Add	ditional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 ditional DS1Loop in STS1 Interoffice Transport Combination - Zone 2	1 2	UNC1X UNC1X	USLXX	50.26 94.06	-	+			-					+
Add	ditional DS1Loop in STS1 Interoffice Transport Combination - Zone 3	3	UNC1X	USLXX	162.34	1									+
	3 Interface Unit (DS1 COCI) combination per month		UNC1X	UC1D1	14.53				1				1		T

ATEGORY	NOTES	ELEMENT	Interim	Zone	BCS	USOC		RA	TES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increment Charge Manual S Order votel Electronic- Add'I
								Nonrecu	ırring		ecurring						
				+'			Rec	First	Add'l	Disc First	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCSX	UNCCC	Rec	11.19	11.19	13.91	13.91	SOMEC	19.99	SOMAN	SOMAN	SUMAN	SOMA
4	4-WIRE 56	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (E 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1	EL)	-	UNCDX	UDL56	35.92										
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	40.32										
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	37.9										
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0118										
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility			UNCDX	U1TD5	21.26										
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99				
	4-WIRE 64	l KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (B	FI)														
	4-WIIL 04	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	35.92										
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	35.92 40.32										
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	37.9										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile		$\perp =$	UNCDX	1L5XX	0.0118										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility	-		UNCDX	U1TD6	21.26						1	1			—
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		1 '	UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99				
-+		Informeduring Currently Combined Inetwork Elements Switch -As-is Charge	-	+'	ONCDX	UNCCC		11.19	11.19	13.91	13.91	+	19.99	1			
ITIONAI	L NETWOR	K ELEMENTS		-				1					1	1			
1				1				1					†	1			
,	When used	as a part of a currently combined facility, the non-recurrng charges do not a	pply, but	a Swit	ch As Is charge	does apply.											
١	When used	I as ordinarilty combined network elements in Georgia, the non-recurring char	ges appl	y and th	ne Switch As Is	Charge does	not.										
				'													
- 1	Nonrecurrii	ng Currently Combined Network Elements "Switch As Is" Charge (One applies 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is"	to each o	combina	ation)												
		2/4-Wife v6 interoffice Channel used in a COMBINATION - Switch as is			UNCVX	UNCCC		11.19	11.19	13.91	13.91		19.99				
		Conversion Charge 56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is"		+	UNCVA	UNCCC		11.19	11.19	13.91	13.91		19.99				
		Conversion Charge			UNCDX	UNCCC		11.19	11.19	13.91	13.91		19.99				
		DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion															
		Charge			UNC1X	UNCCC		11.19	11.19	13.91	13.91		19.99				
		DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion															
		Charge		'	UNC3X	UNCCC		11.19	11.19	13.91	13.91		19.99				
		STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is"			UNCSX	UNCCC		44.40	44.40	13.91	40.04		19.99				
		Conversion Charge		+	UNCSX	UNCCC		11.19	11.19	13.91	13.91		19.99				
-	NOTE: Loc	l al Channel - Dedicated Transport - minimum billing period - Below DS3=one mo	onth DS:	3 and a	hove=four mon	iths											
		3	,														
		RT SYSTEMS															
1	NOTE: (1) E	lectronic Service Order: CLEC-1 should contact its contract negotiator if it prefers t	he state:	specific	electronic service	ce ordering cha	arges as orde	red by the State (Commissions								
1	NOTE: (1) C	Continued: The electronic service ordering charge currently contained in this rate ext	nibit is the	3 BellSo	uth regional elec	tronic service	ordering char	ge									
		Concluded: CLEC-1 may elect either the state specific Commission ordered rates for			ervice ordering o	harges, or CLI	EC-1 may ele	ct the regional ele	ectronic serv	ce ordering	charge.						
١	NOTE: (2)	Manual Service Order charge: disconnect, in the state of Florida, to be billed on a pe	er LSR ba	ısis													
		Flatharia OCC Charac and CD autoritied via DCTa OCC interview		 '		COME							1				
		Electronic OSS Charge, per LSR, submitted via BSTs OSS interactive interfaces (Regional)		1 '		SOME		3.5									
		(Regional)		+-		C		3.5									
-			to Genar	aphicall	Deaveraged III	NE Zones. To	view Geogra	phically Deaverage	ed UNF Zon	e Designati	ons by Centr	al Office, ret	fer to Interne	et Website:			
Т	The "Zone"	snown in the sections for stand-alone loops or loops as part of a compination refers	3"					,	,		,	,		=			
		shown in the sections for stand-alone loops or loops as part of a combination refers nterconnection.bellsouth.com/become a clec/html/interconnection.htm															
h	http://www.i	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm															
h	http://www.i																
UNDLED	http://www.i	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS)		\equiv													
UNDLED	D LOCAL E	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS) Ports															
UNDLED	D LOCAL E	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS)	ures will	need to	be ordered us	ing retail USO	Cs										
UNDLED E	D LOCAL E Exchange NOTE: Alth	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features.	ures will I	need to	be ordered us	ing retail USO	Cs										
UNDLED E	D LOCAL E Exchange NOTE: Alth	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, t	ures will	need to				24 98	24 98				19 90				
UNDLED E	D LOCAL E Exchange NOTE: Alth	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features.	ures will i	need to	be ordered us	ing retail USO	Cs 2.61	24.98	24.98				19.99				
BUNDLED E	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features are considered for the Port Rate 2-Wire Analog Line Port-Res.	ures will	need to				24.98	24.98				19.99				
BUNDLED E	D LOCAL E Exchange NOTE: Alth	nterconnection.bellsouth.com/become_a_clec/html/interconnection.htm XCHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, t	ures will I	need to	UEPSR	UEPRL	2.61										
BUNDLED E	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) ACHANGE SWITCHING(PORTS) Ports Ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the desire	ures will	need to	UEPSR	UEPRL	2.61										
BUNDLED E	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired feature in G	ures will I	need to	UEPSR UEPSR UEPSR	UEPRL UEPRC UEPRO	2.61 2.61 2.61	24.98 24.98	24.98 24.98				19.99				
BUNDLED E	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired feature in GA & TN, the desired features in GA & TN, the desired feature in GA & TN, the desired features in GA & TN, the d	ures will I	need to	UEPSR UEPSR UEPSR UEPSR	UEPRC UEPRO UEPRM	2.61 2.61 2.61 2.61	24.98 24.98 24.98	24.98 24.98 24.98				19.99 19.99				
BUNDLED E	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired features in GA & TN, the desired feature in G	ures will i	need to	UEPSR UEPSR UEPSR	UEPRL UEPRC UEPRO	2.61 2.61 2.61	24.98 24.98	24.98 24.98				19.99				
BUNDLED	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the	ures will	need to	UEPSR UEPSR UEPSR UEPSR UEPSR	UEPRC UEPRO UEPRM UEPAP	2.61 2.61 2.61 2.61 2.61	24.98 24.98 24.98 24.98	24.98 24.98 24.98 24.98				19.99 19.99				
BUNDLED E N 2	D LOCAL E Exchange NOTE: Alth	ACHANGE SWITCHING(PORTS) Ports ough the Port Rate includes all available features in GA & TN, the desired features in GA & TN, the	ures will i	need to	UEPSR UEPSR UEPSR UEPSR	UEPRC UEPRO UEPRM	2.61 2.61 2.61 2.61	24.98 24.98 24.98	24.98 24.98 24.98				19.99 19.99				

CATEGORY	NOTES	ELEMENT I	nterim Zone	BCS	USOC			ATES (\$)								
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. c Electronic-Disc Add'l
, i							Nonrec	urring		curring						
										onnect						_
		All Available Vertical Features		UEPSR	UEPVF	Rec 3.39	First ()	Add'I	First	Add'l	SOMEC	19.99	SOMAN	SOMAN	SOMAN	SOMAN
		All Available vertical Features		OLF SIX	OLI VI	3.33	U	U				13.33				+
	2-WIRE VO	CE GRADE LINE PORT RATES (BUS)														†
		Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus		UEPSB	UEPBL	2.61	37.55	37.55				19.99				
		Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with														
		Caller+E484 ID - Bus.		UEPSB	UEPBC	2.61	37.55	37.55				19.99				
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus. Exchange Ports - 2-Wire VG unbundled KY extended local dialing parity Port with		UEPSB	UEPBO	2.61	37.55	37.55				19.99				
		Caller ID - Bus.		UEPSB	UEPBM	2.61	37.78	37.78				19.99				
		Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus		UEPSB	UEPB1	2.61	37.55	37.55				19.99				
						_										
	FEATURES	Subsequent Activity		UEPSB	USASC	0	C	0								
	FEATURES	All Available Vertical Features		UEPSB	UEPVF	3.39		0				19.99				+
	EXCHANGE	PORT RATES (DID & PBX)		OLF 3B	OLI VI	3.39		0				13.33				+
		Exchange Ports - 2-Wire DID Port		UEPEX	UEPP2	10.97	238.69	37.49	119.4	7.5		19.99				1
		Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability Exchange Ports - 2-Wire ISDN Port (See Notes below.)		UEPDD	UEPDD	83.28	404.18	191.44	144.71	4.9		19.99				
		Exchange Ports - 2-Wire ISDN Port (See Notes below.)		UEPTX UEPSX	U1PMA	15.02	145.59	106.01	95.93	21.55		19.99				ļ
		All Features Offered		UEPTX UEPSX		3.39	0	0	1 1 11 0		1					
	NOTE: Iran	nsmission/usage charges associated with POTS circuit switched usage will also apply to ess to B Channel or D Channel Packet capabilities will be available only through BFR/N	o circuit switc	ned voice and/or cil	Cuit switch	ed data transmis	ssion by B-Cna	inneis associa	ated with 2-v	vire ISDN po	Nam Dualas	Dominat I				+
	NOTE: ACC	Exchange Ports - 2-Wire ISDN Port Channel Profiles	ew business		U1UMA	0	0	0	the Bona Fi	de Requesi/i	New Busines	s Request i	rocess.			+
		Exchange Ports - 4-Wire ISDN DS1 Port		UEPEX	UEPEX	113.21	407.77	203.18	157.84	39.98		19.99				+
		2-Wire VG Unbundled 2-Way PBX Trunk - Res		UEPSE	UEPRD	2.61	36.47	36.47				19.99				
		2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus		UEPSP	UEPPC	2.61	36.47	36.47				19.99			1	
		2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus		UEPSP	UEPPO	2.61	36.47	36.47				19.99				
		2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus		UEPSP	UEPP1	2.61	36.47	36.47				19.99				1
		2-Wire Analog Long Distance Terminal PBX Trunk - Bus		UEPSP	UEPLD	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled PBX LD Terminal Ports		UEPSP	UEPLD	2.61	36.47	36.47				19.99				
		2-Wire Vice Unbundled 2-Way PBX Usage Port		UEPSP	UEPXA	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		UEPSP	UEPXB	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled PBX LD DDD Terminals Port		UEPSP	UEPXC	2.61	36.47	36.47				19.99				-
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		UEPSP	UEPXD	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port		UEPSP	UEPXE	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port Without														
		LUD		UEPSP	UEPXF	2.61	36.47	36.47				19.99				
		2 Miles Vains Habrardad DDV Kanturdar LUD Assa Calling Dart		UEPSP	UEPXG	0.04	36.47	36.47				19 99				
		2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Callling Port		UEPSP	UEPXH	2.61 2.61	36.47	36.47				19.99				+
		2-Wire Voice Unbundled 2-Way PBX Kentucky Area Callling Port Without LUD		UEPSP	UEPXJ	2.61	36.47	36.47				19.99				+
		2-Wire Voice Unbundled 2-Way PBX Kentucky Area Callling Port Without LUD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling														
		Port		UEPSP	UEPXL	2.61	36.47	36.47				19.99				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room		UEPSP	UEPXM	2.61	36.47	36.47				19.99				+
		2-vvire voice unbundled 1-vvay Outgoing PBX Hotel/Hospital Discount Room Calling Port		UEPSP	UEPXO	2.61	36.47	36.47				19.99				
+		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		UEPSP	UEPXS	2.61	36.47	36.47				19.99				+
																1
		Subsequent Activity		UEPSP	USASC	0	0	0	<u></u>		<u> </u>		<u></u>			
ı	FEATURES															
		All Available Vertical Features		UEPSP UEPSE	UEPVF	3.39	0	0				19.99				
		PORT RATES (COIN)				0.04	40.74	40.74				40.00				_
		Exchange Ports - Coin Port		-	+-+	3.04	40.71	40.71				19.99			-	
	Local Swite	hing Features offered with Port		 	+						1					+
		nsmission/usage charges associated with POTS circuit switched usage will also apply to	circuit switc	hed voice and/or cir	cuit switch	ed data transmis	ssion by B-Cha	innels associa	ated with 2-v	vire ISDN no	rts.					+
	NOTE: Acc	ess to B Channel or D Channel Packet capabilities will be available only through BFR/N Exchange port - 4-wire ISDN trunk port -all available features included										s Request I	rocess.			
WISHING LL	D LOCAL S	WITCHING, PORT USAGE														
INDUNDLEL				ļ												
		Switching (Port Usage)				0.002562										
										1	1	1		1	1	
		End Office Switching Function, Per MOU			+	0.002302									+	+
E		End Office Switching Function, Per MOU witching (Port Usage) (Local or Access Tandem)				0.002302										

			1		1 1		1	1								
											Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manua Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Dis	sc E
											per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	+
							Nonrec	urring		ecurring connect						+
						Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	+
																Ť
Common Tra																
	Common Transport - Per Mile, Per MOU					0.0000049										+
	Common Transport - Facilities Termination Per MOU					0.000426										+
D PORT/LOC	OP COMBINATIONS - COST BASED RATES															+
																Ť
Cost Based F	Rates are applied where BellSouth is required by FCC and/or State Commission rule to	o provide l	Unbund	led Local Switch	hing or Switch	n Ports.										
Features sha	all apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the sar	ne manner	r as the	y are applied to	the Stand-Ale	one Unbundled	Port section of	this Rate Ex	xhibit.	D=#// === C:						+
	nd Tandem Switching Usage and Common Transport Usage rates in the Port section of															+
For Georgia a	and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently Co ombos in GA, TN and all other states, the nonrecurring charges shall be those identifie	mbined an	nd Not C	Jurrently Combi	ned Combos	and the first ar	nd additional Po	rt nonrecurrii	ng charges a	apply to Not	Currently Co	mbined Con	nbos. For Curr	ently		
Combined Co	ombos in GA, TN and all other states, the nonrecurring charges shall be those identified	u in the No	onrecum	ring - Currently	Combined sec	ctions.	ı						1 1			+
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															Ť
	, ,															I
UNE Port/Lo	pop Combination Rates		<u>.</u> .T													Ţ
	2-Wire VG Loop/Port Combo - Zone 1		2			16.15	1						1			+
+ + + + + + + + + + + + + + + + + + + +	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		3			22.34 30.88	1	1	+	1	+	+	+			+
			-			55.56										+
UNE Loop R																I
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.54	1	1								4
	2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX UEPRX	UEPLX	19.73 28.27	1	-	1	1	1	1	+		-	+
	2 This voice didde book (DET) Zone o		-	OLI IXX	OLI LA	20.21							1			+
	e Grade Line Port Rates (Res)							İ								Ť
	2-Wire voice unbundled port - residence			UEPRX	UEPRL	2.61		ļ					19.99	19.99		Ţ
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	2.61						19.99				
	2-wire voice unburidied port with Caller ID - res			UEPRA	UEPRC	2.01						19.99				+
2	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	2.61						19.99				
	2-Wire voice Grade unbundled Kentucky extended local dialing parity port with															T
	Caller ID - res 2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX UEPRX	UEPRM UEPAP	2.61						19.99 19.99				+
	2-Wife voice unburidies res, low usage line port with caller ID (Low)			UEFKA	UEFAF	2.01						19.99				+
																İ
FEATURES																4
	All Features Offered			UEPRX	UEPVF	3.39	0	0				19.99				+
LOCAL NUM	MBER PORTABILITY															+
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										T
																4
NONRECUR	RING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPRX	USAC2		10	10				19.99				+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			OLITA	UUAUZ		10	10	1	1		13.33				+
	change			UEPRX	USACC		10	10				19.99				
ADDITIONAL	I NDC-						1	1								4
	L NRCs 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPRX	USAS2	0	0	0	1			1	1			+
				5L. 1(/	55,102			, i								t
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															I
LINE Destr	pop Combination Rates		_						+	-		+				+
	2-Wire VG Loop/Port Combo - Zone 1	+	1			16.15	1	1	1	1	1	1	1			+
	2-Wire VG Loop/Port Combo - Zone 1		2			22.34										+
	2-Wire VG Loop/Port Combo - Zone 3		3			30.88										I
UNELSS	1244-2						1	1								+
UNE Loop R	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.54	1		1			1	1			+
1	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	19.73							1			+
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	28.27										I
2 Wire Veie	e Grade Line Port (Bus)						1		1	1		1				+
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	2.61					+	19.99	1			+
	·					2.01							1			\dagger
2	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	2.61	1					19.99				1
	2 Wire voice unbundled port outgoing only bus			UEPBX	UEPBO	2.61						19.99				
1	2-Wire voice unbundled port outgoing only - bus 2-Wire voice Grade unbundled Kentucky extended local dialing parity port with			UEPBX	UEPBO	2.01	1	1				19.99				+
	Caller ID - bus			UEPBX	UEPBM	2.61			1			19.99				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	2.61						19.99		_		I

	ELEMENT	Interim	Zone	BCS	USOC		RA	TES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremondare Charge Manual Order Electroni Add
							Nonrecu	ırring		curring						
										onnect						
100414	NUMBER PORTABILITY					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
LOCAL N	Local Number Portability (1 per port)		-	UEPBX	LNPCX	0.35										
+	Local Number Fortability (1 per port)			UEFBA	LINECX	0.33										
FEATURE	ES															
	All Features Offered			UEPBX	UEPVF	3.39	0	0				19.99				
NONREC	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		10	10				19.99				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			HEDDY	110100		40	40								
+	change			UEPBX	USACC		10	10								
ADDITIO	NAL NRCs															
ADDITION	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2							19.99				
+	2 vine voice Grade Edopreme i on Gombination Gabacquent rictivity			OLI DX	CONOZ							10.00				
2-WIRE V	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															
UNE Port	t/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1	-		16.15							1			
	2-Wire VG Loop/Port Combo - Zone 2		2			22.34										
	2-Wire VG Loop/Port Combo - Zone 3		3			30.88										
UNE Loop	n Pates		+				+				-		1			-
UNE LOO	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	13.54										1
+	2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEPRG	UEPLX	19.73										
1	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	28.27										
1	, , , , , , , , , , , , , , , , , , , ,															
2-Wire Vo	oice Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	2.61							19.99	19.99		
100111	NUMBER PORTABILITY															
LOCAL N	Local Number Portability (1 per port)		-	UEPRG	LNPCP	3.5										-
+	Local Number Fortability (1 per port)			UEFRG	LINECE	3.3										
FEATURE	FS															
	All Features Offered			UEPRG	UEPVF	3.39	0	0				19.99				
NONREC	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-															
 	IS			UEPRG	USAC2		10	10				19.99				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with			UEPRG	USACC		10	10				10.00				
+	Change			UEPRG	USACC		10	10				19.99				1
ADDITIO	NAL NRCs															
7.55	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0	0	0								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64				19.99				
2-WIRE V	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	t/Loop Combination Rates					10.15										
	// // // // // // // // // // // // //		1 2			16.15										
	t/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		1 2 3			22.34										
	// // // // // // // // // // // // //		1 2 3			16.15 22.34 30.88										
UNE Port	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3					22.34										
	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Pp Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1			UEPPX	UEPLX	22.34										
UNE Port	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Pates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		1 2	UEPPX	UEPLX	22.34 30.88 13.54 19.73										
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Pp Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1		1			22.34 30.88										
UNE Port	t/U-op Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 p Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		1 2	UEPPX	UEPLX	22.34 30.88 13.54 19.73										
UNE Port	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Pates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2		1 2	UEPPX	UEPLX	22.34 30.88 13.54 19.73										
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 P Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 Voice Grade Line Port Rates (BUS - PBX)		1 2	UEPPX UEPPX	UEPLX UEPLX	22.34 30.88 13.54 19.73 28.27						10 00				
UNE Port	t/U-op Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 p Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		1 2	UEPPX	UEPLX	22.34 30.88 13.54 19.73						19.99				
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 P Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 Poice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		1 2	UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC	22.34 30.88 13.54 19.73 28.27						19.99				
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 PRATES 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 Oice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC	22.34 30.88 13.54 19.73 28.27 2.61 2.61										
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 P Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 Poice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		1 2	UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC	22.34 30.88 13.54 19.73 28.27						19.99				
UNE Port	t/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 pp Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 oice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPXA	22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 2.61						19.99 19.99 19.99 19.99				
UNE Port	VLoop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 DP Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 Oice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled PBX LD Terminal Ports		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD	22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61						19.99 19.99 19.99				
UNE Port	t/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 p Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 foice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPXA UEPXB	22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 2.61 2.61						19.99 19.99 19.99 19.99 19.99				
UNE Port	t/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 pp Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 oice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPXA	22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 2.61						19.99 19.99 19.99 19.99				
UNE Port	t/Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 p Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 foice Grade Line Port Rates (BUS - PBX) Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		1 2	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPLX UEPLX UEPPC UEPPO UEPP1 UEPLD UEPXA UEPXB	22.34 30.88 13.54 19.73 28.27 2.61 2.61 2.61 2.61 2.61 2.61						19.99 19.99 19.99 19.99 19.99				

NOTES	ELEMENT Interim	Zone BCS	USOC			ATES (\$)			1		00011	ATES (\$)	Incremental	Т
									Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manua Svc Order vs.	Charge - Manual Svc Order vs. Electronic-Disc	c E
						1,		<u> </u>	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	+
					Nonrec	curring		ecurring						+
				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	t
	2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD	UEPPX	UEPXF	2.61							19.99	19.99		
		LIEBBY												
	2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port 2-Wire Voice Unbundled PBX Kentucky Premium Calling Port	UEPPX UEPPX	UEPXG UEPXH	2.61 2.61							19.99 19.99	19.99 19.99		+
	2-Wire Voice Unbundled PBX Rendicky Fremium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD	UEPPX	UEPXJ	2.61							19.99	19.99		+
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling													T
	Port	UEPPX	UEPXL	2.61						19.99				
	2 Wise Vales Habitatland 2 Wast BBY Hatal/Hassital Faces Bases Calling Both	UEPPX	UEPXM	2.61						19.99				
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room	UEPPA	UEPAINI	2.01						19.99				+
	Calling Port	UEPPX	UEPXO	2.61						19.99				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	UEPPX	UEPXS	2.61							19.99	19.99		I
														+
	MBER PORTABILITY	UEPPX	LNPCP	3.15										+
	Local Number Portability (1 per port)	UEFPX	LINEOP	3.13		+		+		+	1			+
FEATURES														T
	All Features Offered	UEPPX	UEPVF	3.39	0	0				19.99				Ι
NONDECLI	DDING CHARGES (AIDC+), CHRRENTI V COMPINED					1		1		1				+
NUNKECU	RRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-					1		+		+				+
	ls	UEPPX	USAC2		10	10		1		19.99				1
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with													Т
	Change	UEPPX	USACC		10	10		1		19.99				+
ADDITIONA	AL NRCs													+
7155111010	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity	UEPPX	USAS2	0	0	0								t
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group				14.64	14.64				19.99				
														+
2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT													+
UNE Port/L	.oop Combination Rates													+
	2-Wire VG Coin Port/Loop Combo – Zone 1			16.15										T
	2-Wire VG Coin Port/Loop Combo – Zone 2			22.64										Ι
UNE Loop	2-Wire VG Coin Port/Loop Combo – Zone 3			31.09										+
ONE LOOP	Rates													+
	2-Wire Voice Grade Loop (SL1) - Zone 1	UEPCO	UEPLX	13.54										T
	2-Wire Voice Grade Loop (SL1) - Zone 2	UEPCO	UEPLX	19.73										Ι
	2-Wire Voice Grade Loop (SL1) - Zone 3	UEPCO	UEPLX	28.27										+
2-Wire Voic	ce Grade Line Ports (COIN)													+
	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,													+
	MS)	UEPCO	UEPRF	2.91						19.99	19.99			
	2-Wire Coin 2-Way with Operator Screening (AL, KY)	UEPCO	UEPRE	2.91						19.99	9			Ļ
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD	UEPCO	UEPRA	2.91						19.99				1
	(AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY)	UEPCO	UEPKA	2.91		+		1		19.99				+
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &													T
	Local (AL, KY, LA, MS)	UEPCO	UEPCD	2.91		1				19.99				1
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,	UEPCO	UEPRN	2.91		1		1		19.99				+
	2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS) 2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD	UEPCO	UEPRJ	2.91		+				19.99	1			+
	(AL, KY, LA, MS)	UEPCO	UEPRH	2.91		1		1		19.99				1
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and													T
	Local (AL, KY, LA, MS)	UEPCO	UEPCN	2.91		1				19.99				1
	2-Wire 2-Way Smartline with 900/976 (all states except LA)	UEPCO UEPCO	UEPCK	2.91 2.91			-			19.99 19.99	-		-	+
ADDITIONA	2-Wire Coin Outward Smartline with 900/976 (all states except LA) AL UNE COIN PORT/LOOP (RC)	UEFCU	UEFCR	2.91		+		+		19.99	1			+
														+
	UNE Coin Port/Loop Combo Usage (Flat Rate)	UEPCO	URECU	2.57	0	0								
LOCAL ATT	MBER PORTABILITY													4
LOCAL NU	MBER PORTABILITY Local Number Portability (1 per port)	UEPCO	LNPCX	0.35				-		-				+
	Local Number Fortability (1 per port)	UEFCU	LINEUX	0.33							 			+
FEATURES														+
	All Features Offered	UEPCO	UEPVF	3.39	(0 0				19.99)			I
NONE	DDING CHARGES CHERENTLY COMPINED													#
	RRING CHARGES - CURRENTLY COMBINED	UEPCO	USAC2		10	10		1		19.99	1			+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with	UEPCO	USAC2		10	10		+		19.99				+
	change	UEPCO	USACC		10	10	1			19.99	1			

NOTES		Interim Zone		USOC			ATES (\$)			Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	sc E
										per LSR	LSR	Electronic-1st	Electronic-Add'l	l 1st	3 E
						Nonrec	urring	Nonre	ecurring onnect						+
ADDITIONA	II NIPCo				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	+
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		UEPCO	USAS2		0	0				19.99				$^{+}$
2 WIDE VO	ICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT													-	+
Z-WIKE VO	ICE GRADE LOOF- BUS ONLY - WITH 2-WIKE DID TRONK FORT														+
	oop Combination Rates														1
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2	2			28.72 34.9									-	+
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3	3			45.9										t
UNE Loop F	Potes														+
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1	1	UEPPX	UECD1	17.78						19.99				+
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2	2	UEPPX	UECD1	23.96						19.99				#
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3	3	UEPPX	UECD1	34.96						19.99			-	+
UNE Port R															Ť
	Exchange Ports - 2-Wire DID Port		UEPPX	UEPD1	10.94	1					19.99			<u> </u>	Ŧ
NONRECUR	RRING CHARGES - CURRENTLY COMBINED														t
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth														T
	Allowable Changes		UEPPX	USA1C		14.62	3.73				19.99			-	+
ADDITIONA															t
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk		UEPPX	USAS1		53.58	53.58				19.99				+
Telephone	Number/Trunk Group Establisment Charges														$^{+}$
	DID Trunk Termination (One Per Port)		UEPPX	NDT	0	0	0				19.99				Ŧ
	Additional DID Numbers for each Group of 20 DID Numbers DID Numbers, Non- consecutive DID Numbers , Per Number		UEPPX UEPPX	ND4 ND5	0	0	0				19.99 19.99				+
	Reserve Non-Consecutive DID numbers		UEPPX	ND6	0	0	0				19.99				T
	Reserve DID Numbers		UEPPX	NDV	0	0	0				19.99				+
LOCAL NU	MBER PORTABILITY														+
	Local Number Portability (1 per port)		UEPPX	LNPCP	3.15										4
2-WIRE ISD	ON DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT														1
UNE Port/Lo	oop Combination Rates														+
			UEPPB												T
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1	1	UEPPR UEPPB		35.4									-	+
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2	2	UEPPR		44.09										
	OW ICOM Digital Conde Lang/OW ICOM Digital Line Cide Dark LINE Zang 2	3	UEPPB		55.05										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3	3	UEPPR		55.35										+
UNE Loop F	Rates														+
	2-Wire ISDN Digital Grade Loop - UNE Zone 1	1		PR USL2X	22.41						19.99				╧
	2-Wire ISDN Digital Grade Loop - UNE Zone 2	2	UEPPB UEPPR	USL2X	31.1						19.99				
		_													T
	2-Wire ISDN Digital Grade Loop - UNE Zone 3	3	UEPPB UEP	PR USL2X	42.36						19.99				+
UNE Port R	tate Exchange Port - 2-Wire ISDN Line Side Port		UEPPB UEPI	DD LIEDDD	12.99						19.99				+
	Exchange Fort - 2-Wile ISDN Line Side Fort		OEFFB OEF	FK UEFFB	12.99						19.99				+
NONRECUR	RRING CHARGES - CURRENTLY COMBINED														I
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination - Conversion		UEPPB UEPI	PR USACB	0	77.04	54.04				19.99				
ADDITIONA															#
														<u> </u>	\pm
LOCAL NU	MBER PORTABILITY					<u> </u>				1	1			<u> </u>	#
	Local Number Portability (1 per port)		UEPPB UEP	PR LNPCX	0.35	0	0								
B-CHANNE	L USER PROFILE ACCESS:														+
_ 0															†
	CVS/CSD (DMS/5ESS)			PR U1UCA	0	0	0							_	4
	CVS (EWSD) CSD		UEPPB UEP	PR U1UCC	0	0	0							 	+
			JEN J OLF	5 . 000					l	+	+	+		+	+

Y	NOTES	ELEMENT	nterim Zone	ВС	cs	USOC		RA	ATES (\$)					OSS R	ATES (\$)		
	NOTES	ECCURATI II	Zono		~	0000						Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	Increment Charge Manual St Order vs c Electronic-I Add'I
								Nonrec	curring		ecurring						
_				+							connect	SOMEC	SOMAN		SOMAN	SOMAN	
-		CVS/CSD (DMS/5ESS)	-+	LIEPPR	UEPPR	LITUCD	Rec 0	First 0	Add'I O	First	Add'l	SOMEC	SUMAN	SOMAN	SOMAN	SUMAN	SOMAN
		010/005 (Emoio200)		102.11	<u> </u>	0.005		- ŭ	- ŭ			1					1
		CVS (EWSD)		UEPPB	UEPPR	U1UCE	0	0	0								
		CSD		UEPPB	UEPPR	U1UCF	0	0	0								
U	SER TERM	MINAL PROFILE	_	+		-						+					+
		User Terminal Profile (EWSD only)		LIEPPR	UEPPR	LI1LIMΔ	0	0	0								
			-	02.12		0.101111										1	1
VI	ERTICAL I	FEATURES															
		All Vertical Features - One per Channel B User Profile		UEPPB	UEPPR	UEPVF	3.39	0	0								
		CE CHANNEL MILEAGE					7.00										
IN				+								+					+
		Interoffice Channel mileage each, including first mile and facilities termination		UEPPB	UEPPR	M1GNC M1GN	26.98	142.31	56.21			+	19.99			+	
		Interoffice Channel mileage each, additional mile		UEPPB	UEPPR	M	0.0301	0	0				0				
4-	-WIRE DS1	1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT	_									+					-
UI	NE Port/L	oop Combination Rates															
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1	1 2	UEF	PPP		219.25 248.36					-					+
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3	3		PPP		299.47					+				+	+
		TW BOT Bigital Edopi-tw Tobit Bot Bigital Hallict Sit. Site 2016 9		- 02.	''		200.41					1				<u> </u>	1
UI	NE Loop F											T					
		4-Wire DS1 Digital Loop - UNE Zone 1	1	UEF		USL4P	106.04						19.99				
		4-Wire DS1 Digital Loop - UNE Zone 2	2	UEF	PP	USL4P	135.15						19.99				
		4-Wire DS1 Digital Loop - UNE Zone 3	3	UEF	'PP	USL4P	186.15					+	19.99			+	+
UI	NE Port R																1
		Exchange Ports - 4-Wire ISDN DS1 Port		UEF	PP	UEPPP	113.21						19.99				
N	ONRECUR	RRING CHARGES - CURRENTLY COMBINED		+		 						+				+	+
	ONICEOUN	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -		+								+				+	+
		Conversion -Switch-as-is		UEF	PPP	USACP	0	238.22	157.17				19.99				
Α	DDITIONA	L NRCs		+								+					+
		4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos															
		within Std Allowance		UEF	PP	PR7TF		0.9804					19.99				
		4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All															
-		States except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos		UEF	'PP	PR7TO		23.02	23.02			+	19.99			+	+
		Above Std Allowance		UEF	PP	PR7ZT		46.05	46.05				19.99				
																	_
L	OCAL NUM	MBER PORTABILITY Local Number Portability (1 per port)		LIE	PPP	LNPCN	1.75					+				-	+
_		Educat Number Fortability (1 per port)		OLI		LINI CIN	1.75					+				+	+
IN	TERFACE	(Provsioning Only)															
		Voice/Data		UEF		PR71V	0	0	0			T					
		Digital Data		UEF		PR71D	0	0	0								
_		Inward Data		UEF	PPP	PR71E	0	0	0								
NI.	ow or Add	litional "B" Channel	-+	+		+-+		+	1			+				+	+
1.40		New or Additional - Voice/Data B Channel	-+	UEF	PP	PR7BV	0	29.06				+	19.99			+	+
		New or Additional - Digital Data B Channel		UEF		PR7BF	0	29.06				1	19.99			†	1
T		New or Additional Inward Data B Channel		UEF	PPP	PR7BD	0	29.06				1	19.99				
I		New or Additional Useage Sensitive Voice Data B Channel		UEF		PR7BS	0	29.06					19.99				
\perp		New or Additional Useage Sensitive Digital Data B Channel		UEF	PP_	PR7BU	0	29.06	1				19.99				1
_	ALL TYPE	ie e		+		\longmapsto		+				+				+	-
C.		Inward	-+	UEF	DPP	PR7C1	0	0	0		1	+				+	+
+		Outward	-+	UEF		PR7C0	0	0	0			+				+	+
\dashv		Two-way		UEF		PR7CC	0	0	0			+					†
																	L
	teroffice C	hannel Mileage					-	1									
In							EE E		1 224 22	0		1	19.99	1	i .	1	1
In		Fixed Each Including First Mile Each Airline-Fractional Additional Mile		UEF)DD	1LN1A 1LN1B	55.5 0.45	298.18	231.23	U		+	13.33			+	+

N	NOTES	ELEMENT Interim	Zone	BCS	USOC		R	ATES (\$)			T		OSS R	ATES (\$)		
	.5.23	THE THE THE THE THE THE THE THE THE THE		200	2300						Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental	Electronic-Disc	Increme Charg Manual Order sc Electronic Add
							Nonrec	urring		ecurring						
										onnect						
4 14	UDE DC4	DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
4-VV	IRE DS1	I DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT									-					+
LINIE	- Dort/Lo	pop Combination Rates			+						+					+
OIVE	FOIVE	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1	1	UEPDC		189.32					+	19.99			+	+
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	2	UEPDC		218.43					+	19.99			+	+
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	3	UEPDC		269.54					-	19.99			1	1
UNE	E Loop R															
		4-Wire DS1 Digital Loop - UNE Zone 1	1	UEPDC	USLDC	106.04						19.99				
_		4-Wire DS1 Digital Loop - UNE Zone 2	2	UEPDC UEPDC	USLDC	186.15				23.33		19.99				
-		4-Wire DS1 Digital Loop - UNE Zone 3	3	UEPDC	USLDC	186.15					+	19.99				+
LINE	E Port Ra	ato.			+						+					+
OIVE		4-Wire DDITS Digital Trunk Port		UEPDC	UDD1T	83.28					+	19.99			+	+
		4-Wile DD113 Digital Hulik Folt		OLF DC	ODDII	05.20					+	19.99			+	+
ION	NRECUR	RING CHARGES - CURRENTLY COMBINED									+					1
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is		UEPDC	USAC4		261.15	134.08			-	19.99			1	1
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with									T					T
		DS1 Changes		UEPDC	USAWA		261.15	134.08				19.99				
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with			USAW											
		Change - Trunk		UEPDC	В		261.15	134.08				19.99				
	DITICAL	I NDC-			+				-	1	+	-	-		+	+
ADI	DITIONAL	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel	+						-		+	-	-		+	+-
		Activation/Chan - 2-Way Trunk		UEPDC	UDTTA		28.96	28.96				19.99				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel		UEFDC	UDITA		20.90	20.90			+	19.99			+	+
		Activation/Chan - 1-Way Outward Trunk		UEPDC	UDTTB		28.96	28.96				19.99				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan														
		Inward Trunk w/out DID		UEPDC	UDTTC		28.96	28.96				19.99				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -														
		Inward Trunk with DID		UEPDC	UDTTD		28.96	28.96				19.99				
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-														
DID		Way DID w User Trans ZERO SUBSTITUTION		UEPDC	UDTTE		28.96	28.96				19.99			+	+
DIP	ULAR 8	ZERO SUBSTITUTION									+					+
		B8ZS -Superframe Format		UEPDC	CCOSF		0	730				19.99				
-		BOZO Oupernanie i omat		OLI DO	00001			700			1	13.33				+
		B8ZS - Extended Superframe Format		UEPDC	CCOEF		0	730				19.99				
Alte	ernate Ma	ark Inversion														
		AMI -Superframe Format		UEPDC	MCOSF		0	0							↓	
		AMI - Extended SuperFrame Format		UEPDC	MCOP		0	0								
-	-	Aivii - Extended SuperFrame Format		UEPDC	- 0		U	U			+					+
\vdash			1		1 1		1	1			+		+		+	+-
Tole	enhone !	Number/Trunk Group Establisment Charges			+ +		1			†	+		1		+	+
1 CIC		Telephone Number for 2-Way Trunk Group		UEPDC	UDTGX	0				1	+				+	+-
		Telephone Number for 1-Way Outward Trunk Group		UEPDC	UDTGY	0						19.99				1
		Telephone Number for 1-Way Inward Trunk Group Without DID		UEPDC	UDTGZ	0						19.99				
		DID Numbers for each Group of 20 DID Numbers	lacksquare	UEPDC	ND4	0						19.99				1
4		DID Numbers, Non- consecutive DID Numbers , Per Number	1	UEPDC	ND5	0	-	-		1		19.99	1			+
+-		Reserve DID Numbers	+	UEPDC UEPDC	ND6 NDV	0	0	0	-		+	19.99 19.99	-		+	+-
+		IVESEINE DID MILIDEI2	1	UEPDC	INDV	U	U	U			+	19.99	+		+	+-
Dec	licated F	DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-Wire DDIT	IS True	k Port	+ +		1			†	+		1		+	+-
500		Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)		UEPDC	1LNO1	55.05	298.18	231.23	0	0	+	19.99			+	+-
L		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles		UEPDC	1LNOA	0.45	0	0								I
		Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)		UEPDC	1LNO2	0	0	0								
		Interoffice Channel Mileage - Additional rate per mile - 9-25 miles		UEPDC	1LNOB	0.45	0	0			4					4
1		Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)	1	UEPDC	1LNO3	0	0	0	0	-	+		1			+
+		Interoffice Channel Mileage - Additional rate per mile - 25+ miles	1	UEPDC	1LNOC	0.45	0	0		1	+		1		+	+-
+		Local Number Portability, per DS0 Activated Central Office Termininating Point	1	UEPDC UEPDC	LNPCP	3.15 0	0	0	0		+		1		+	+
 		Contrat Onto Terriminating Form		ULFDU	UIG	U	1			†	+		1		+	+-
t					1 1		1	1		1	+		1		†	+
4-W	/IRE DS1	LOOP WITH CHANNELIZATION WITH PORT			1 1						+				†	+
		DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations									1				T	1
	h Syster	m can have up to 24 combinations of rates depending on type and number of ports us	sed													1
Eac	- 1															1
Eac		non.									1					
	DS1 Lo	Ю														
		4-Wire DS1 Loop - UNE Zone 1 4-Wire DS1 Loop - UNE Zone 2	1 2	UEPMG UEPMG	USLDC 1		0	0								

	NOTES	ELEMENT	Interim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incr Ch Man On Electr
+										ecurring	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	,
							NOI	recurring		connect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	S
		4-Wire DS1 Loop - UNE Zone 3	3	UEPMG	LISLIDO	186.15	0	0	riiot	Addi	JOINEC	SOMA	SOMPH	JOHN	JOHIAN	
		4 WHE BOT LOOP ONE ZONE O	J	OLI WIO	OOLDO	100.10	-	-								
LIK	NE DOO C	hannelization Capacities (D4 Channel Bank Configurations)														
Ui		24 DSO Channel Capacity - 1 per DS1		UEPMG	\/I IN 40 4	136.99	0	0		1						
-						273.98	0	0								
+		48 DSO Channel Capacity - 1 per 2 DS1s		UEPMG			0	0		-						
		96 DSO Channel Capacity -1per 4 DS1s		UEPMG		547.96	· ·	0								
		144 DS0 Channel Capacity - 1 per 6 DS1s		UEPMG		821.94	0	0								
		192 DS0 Channel Capacity -1 per 8 DS1s		UEPMG		1095.92	0	0								
		240 DS0 Channel Capacity - 1 per 10 DS1s		UEPMG	VUM20		0	0								
		288 DS0 Channel Capacity - 1 per 12 DS1s		UEPMG	VUM28	1643.88	0	0								
		384 DS0 Channel Capacity - 1 per 16 DS1s		UEPMG	VUM38	2191.84	0	0								
		480 DS0 Channel Capacity - 1 per 20 DS1s		UEPMG	VUM40	2739.8	0	0								
		576 DS0 Channel Capacity -1 per 24 DS1s		UEPMG		3287.76	0	0								
		672 DS0 Channel Capacity - 1 per 28 DS1s		UEPMG		3835.72	0	0								
-		072 D30 Chaillel Capacity - 1 per 20 D318		OLI WIG	V OIVIO7	3033.72	- 0	0								
No	on-Poor	ring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with F	ort - Conversio	n Chargo Based o	on a Synt	om	-	-	1	1	1		1			H
						CIII	-		-	1	+			1		\vdash
		System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24					-		-	-				1		H
Mı		this configuration functioning as one are considered Add'l after the minimum				_										<u> </u>
1		NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes		UEPMG	USAC4	0	301.05	16.72				19.99				
		ditions at End User Locations Where 4-Wire DS1 Loop with Channelization with	Port Combinat	tion Currently Exis	sts and											
Ne		urrently Combined) In Georgia & Tennessee Only														
Ι		NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature														
		Activation - New GA & TN Only		UEPMG	VUMD4	0	716.36	468.2	149.3	17.71		19.99				
Bij	ipolar 8 Ze	ero Substitution														
	•															
		Clear Channel Capability Format, superframe - Subsequent Activity Only		UEPMG	CCOSF	0	0	730				19.99				
		Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		UEPMG	CCOEF	0	0	730				19.99				
ΔH		ark Inversion (AMI)		OL: MO	0002.							10.00				
	iterriate ivi	ark inversion (Aim)														
		Superframe Format		UEPMG	MCOSF	0	0	0								
		Extended Superframe Format		UEPMG	MCOPC		0	0								H
		Extended Supername i offiat		OLI WO	WCOI C	10	-	0		1						
г.,		Ports Associated with 4-Wire DS1 Loop with Channelization with Port														
	xchange F															H
	xunanye r	roits														
				UEPPX	UEPCX	1 66	0	0		0		19.99				
				UEFFA		1.00		U	U	U						
		Line Side Combination Channelized PBX Trunk Port - Business					U					10.00				
				LIEDDY		4.00										
		Line Side Outward Channelized PBX Trunk Port - Business		UEPPX	UEPOX		0	0	0	0		19.99				
				UEPPX UEPPX			0	0	0	0						
		Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEPOX UEP1X	1.66	0	0	0	0		19.99 19.99				
		Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPOX	1.66	0 0	0 0	0 0	0		19.99				
		Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID		UEPPX	UEPOX UEP1X UEPDM	1.66	0 0	0 0	0 0	0		19.99 19.99				
	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration		UEPPX UEPPX	UEPOX UEP1X UEPDM	1.66	0 0	0 0 0	0 0 0	0 0		19.99 19.99 19.99				
	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port		UEPPX	UEPOX UEP1X UEPDM	1.66	0 0 0	0 0 0	0 0 0 4.17	0 0 0 4.15		19.99 19.99				
	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM	1.66 10.97 0.77	0 0 0	0 0 0 13.41	0 0 0 4.17			19.99 19.99 19.99				
	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		UEPPX UEPPX	UEPOX UEP1X UEPDM	1.66		0 0 0 13.41 19.68	0 0 0 4.17 59.05	0 0 0 4.15		19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank		UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM	1.66 10.97 0.77	0 0 0 0 25.4 78.15					19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service		UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U	1.66 10.97 0.77						19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tixtations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port)		UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U	1.66 10.97 0.77						19.99 19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEPTX UEPDM 1PQW M 1PQW U NDT ND4	1.66 10.97 0.77	78.15 0					19.99 19.99 19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5	1.66 10.97 0.77	78.15 0 0					19.99 19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers - groups of 20 - Valid all States		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6	1.66 10.97 0.77	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Reserve DID Numbers		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5	1.66 10.97 0.77	78.15 0 0					19.99 19.99 19.99 19.99 19.99				
Fe	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Port Portability		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV	1.66 10.97 0.77 0.77 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe Te	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers - green Valid States Non-Consecutive DID Numbers - green Valid States Reserve DID Numbers - groups of 20 - Valid States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Ser Portability - 1 per port		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6	1.66 10.97 0.77 0.77 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe Te Lo	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Per Portability Local Number Portability - 1 per port - Vertical and Optional		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV	1.66 10.97 0.77 0.77 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe Te Lo	eature Act	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers - green Valid States Non-Consecutive DID Numbers - green Valid States Reserve DID Numbers - groups of 20 - Valid States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Ser Portability - 1 per port		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV	1.66 10.97 0.77 0.77 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe Te Lo	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Portability Local Number Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99				
Fe Te Lo	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Per Portability Local Number Portability - 1 per port - Vertical and Optional		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Fe Te	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Portability Local Number Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Te Lo	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Ser Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Te Lo	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Portability Local Number Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Te Lo	eature Act elephone ocal Numb	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Ser Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 0	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Fe Lo	elephone ocal Numb EATURES ocal Switc	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve DID Numbers Der Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEPTX UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 3.15	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Te Lo	elephone ocal Numb EATURES ocal Switc PORT LO	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Numberf Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve DID Numbers Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES		UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	UEPOX UEPTX UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 3.15	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Fe Lo	elephone ocal Numb EATURES ocal Switc PORT LO	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Number/ Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Ser Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES	switch ports per	UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX FCC and/or State (UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND5 ND6 NDV LNPCP UEPVF	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 3.15	78.15 0 0 0					19.99 19.99 19.99 19.99 19.99 19.99				
Fe Lo	elephone ocal Numb EATURES ocal Switc PORT LO larket Rate nese scena	Line Side Outward Channelized PBX Trunk Port - Business Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port tivations - Unbundled Loop Concentration Feature (Service) Activation for each Line Side Port Terminated in D4 Bank Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank Numberf Group Establishment Charges for DID Service DID Trunk Termination (1 per Port) DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve Non-Consecutive DID Numbers Reserve DID Numbers Portability Local Number Portability - 1 per port - Vertical and Optional hing Features Offered with Line Side Ports Only All Features Available OP COMBINATIONS - MARKET RATES	switch ports per	UEPPX UEPPX	UEPOX UEP1X UEPDM 1PQW M 1PQW U NDT ND4 ND6 ND0 ND6 NDV LNPCP UEPVF	1.66 10.97 0.77 0.77 0 0 0 0 0 0 0 3.15 3.39	78.15 0 0 0 0 0 0 0	19.68	59.05			19.99 19.99 19.99 19.99 19.99 19.99				

Version 2Q01: 08/30/01

KY

NOTES	ELEMENT	interim	Zone	BCS	USOC		K/	ATES (\$)		1			055 K	ATES (\$)		_
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manua Svc Order vs. Electronic-Add'	Incremental Charge - Manual Svc I Order vs. Electronic-Dis	=
							Nonrec	urring	Nonre	curring	per con	Lon	Licon onio 15t	Electronic Fada I		_
										onnect						_
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	_
The Market F	Rate for unbundled ports includes all available features in all states.															_
End Office ar	nd Tandem Switching Usage and Common Transport Usage rates in the Port ser	ction of this ra	ate exhil	oit shall apply to a	all combination	ns of loop/port	network elemen	ts except for	UNE Coin F	ort/Loop Co	mbinations	which have	a flat rate usag	ge charge (U	SOC: UREC	C
For Not Curre	ently Combined scenarios where Market Rates apply, the Nonrecurring charges	are listed in t	the First	and Additional N	IRC columns for	or each Port U	SOC. For Curre	ntly Combine	ed scenarios	, the Nonrec	urring charg	es are liste	d in the NRC -	Currently Co	mbined sect	ti
Additional NR	RCs may apply also and are categorized accordingly.							,						,		
																_
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
	•															_
UNE Port/Lo	oop Combination Rates															_
2	2-Wire VG Loop/Port Combo - Zone 1		1			27.54										_
2	2-Wire VG Loop/Port Combo - Zone 2		2			33.73										
2	2-Wire VG Loop/Port Combo - Zone 3		3			42.27										
																_
UNE Loop R							1						1			_
1	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.54					1	1	1		1	_
- 1	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	19.73	1						+			_
-	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	28.27	1						+			_
2 Wire Vel	Crade Line Bort (Bos)		1		+-+						1	1	+		-	-
	e Grade Line Port (Res)		+	UEPRX	UEPRL	14	00	00			1	19.99	+		-	_
	2-Wire voice unbundled port - residence	-	1	UEPKX	UEPKL	14	90	90			1	19.99	+		1	-
	2-Wire voice unbundled port with Caller ID - res	1		UEPRX	UEPRC	14	90	90				19.99	1		1	
- 1	2 11110 10100 unbundieu port with Galler 10 - 165	_	+	OLFIX	OLFIC	14	30	30				13.33	+		 	-
5	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	14	90	90				19.99	1			
l.	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	14	90	90				19.99	1			-
																_
LOCAL NUM	MBER PORTABILITY															_
l.	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										_
																Τ
FEATURES																
,	All Features Offered			UEPRX	UEPVF	0	0	0								
ADDITIONAL																_
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX	USAS2		0	0								_
																_
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															_
LINE Desta	O and the office Body															_
	pop Combination Rates		-		+	27.54	-	1			1	1	+		-	_
- 1	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		2		+-+	27.54 33.73					1	1	+		-	-
	2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	-	3		+	42.27		1			1	1	+		1	-
	2-Wile VG Loop/Fort Combo - Zone 3		3			42.21										-
UNE Loop R	Patos		+-				 				1	1	+		+	-
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.54										-
	2-Wire Voice Grade Loop (SL1) - Zone 2	1	2	UEPBX	UEPLX	19.73	1						1			-
- 15	2-Wire Voice Grade Loop (SL1) - Zone 3	1	3	UEPBX	UEPLX	28.27	1						1			-
ľ			Ť	221 0/1												-
2-Wire Voice	e Grade Line Port (Bus)															Τ
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	14	90	90				19.99				Ξ
															1	Ī
1	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	14	90	90				19.99	1			
T						· <u></u>									1	
	2-Wire voice unbundled port outgoing only - bus		1	UEPBX	UEPBO	14	90	90			1	19.99	1		1	_
	IDED DODTA DILITY		1		\perp								1			_
	MBER PORTABILITY		1	LIEBBY	LNDOV	0.05	1				1	1	+			_
I	Local Number Portability (1 per port)		1	UEPBX	LNPCX	0.35	1						+			_
EEATURES			1		+-+						1	1	+		-	_
FEATURES			+		+		-	1			1	1	+		-	_
NONDECTION	RING CHARGES - CURRENTLY COMBINED	-	1		+			1			1	1	+		1	-
NONKECUKI	INING CHANGES - CONNENTET COMIDINED		+		+		1				1	1	+		1	_
ADDITIONAL	NRCs	_	+										+		 	-
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent	-	1	UEPBX	USAS2		0	0			1	1	+		1	-
	1410 2 1110 Voice Grade Loop/Line Fort Combination - Subsequent		1 -	OLFDA	UUAUZ		, , , , , , , , , , , , , , , , , , ,	U			1	1	 		1	-
2-WIRE VOIC	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)		+				1						1			-
													1			-
UNE Port/I	pop Combination Rates															-
	2-Wire VG Loop/Port Combo - Zone 1	1	1			27.54	1						1			-
	2-Wire VG Loop/Port Combo - Zone 2		2			33.73										-
	2-Wire VG Loop/Port Combo - Zone 3		3			42.27		1								Τ
																_
UNE Loop R	lates								1							Ī
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	13.54	1		1		1	1	+		1	-

	NOTES	ELEMENT	Interim	20/16	BCS	USOC		KA	ATES (\$)		1	1		U33 K	ATES (\$)	I bear	_
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	•
					-			Nonrecu	urring	Nonre	ecurring						I
				\perp							onnect						_
-		0.14/ 1/ 0			LIEDDO	LIEDLY	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	4
-		2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRG UEPRG	UEPLX	19.73 28.27									+	+
+	-	2-wire voice Grade Loop (SL1) - Zone 3		-3	UEPRG	UEPLA	28.21									+	+
2-1	Niro Voice	e Grade Line Port Rates (RES - PBX)		+-+					-							+	+
2-1	WIIE VOICE	e Grade Line Fort Nates (NEG - FBX)		+-+					-							+	+
		2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	14	90	90				19.99				
1		E TITO TO ORDANAROU SOMBINATORE TRAY I BY THAIRT ON THOS		1	OE: NO	OE: NO			- 00				10.00				T
LO	CAL NUN	IBER PORTABILITY		1													T
		Local Number Portability (1 per port)			UEPRG	LNPCP	3.15										T
FE	ATURES																
NO	NRECUR	RING CHARGES - CURRENTLY COMBINED															
				\perp													_
AD	DITIONAL			\perp													_
		2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-						_	_			1					- [
-		Nonrecurring		+		++		0	0			1	40.00	-	-		4
+		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group		+		++		14.64	14.64	1		-	19.99				+
2-1	NIRE VOY	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)		+		+-+		1				1	1	1			+
2-1	L VOIC	OF SUBSE FOOL MILITA-MINE FINE LOLI (DOS - LDV)		+-+		+		1			1		1	1			+
UN	IF Port/Lo	pop Combination Rates		+ - +		+-+		+	-			+	+	<u> </u>		+	+
1		2-Wire VG Loop/Port Combo - Zone 1		1		+	27.54	1		1	1		1				$^{+}$
t		2-Wire VG Loop/Port Combo - Zone 2		2	-	+	33.73						1				$^{+}$
T		2-Wire VG Loop/Port Combo - Zone 3		3			42.27										T
																	T
UN	IE Loop R																T
		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	13.54										П
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	19.73										
_		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	28.27									_	4
	M' M-1-	On the Live Board Board (BUID BRW)		+-+													4
2-V	vire voice	e Grade Line Port Rates (BUS - PBX)		++													+
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	14	90	90				19.99				
+-		Line Side Oriburidied Combination 2-vvay FBX Trunk Fort - Bus		+-+	UEFFX	UEFFC	14	90	90				19.99			+	+
		Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14	90	90				19.99				
		Line Side Unbundled Incoming PBX Trunk Port - Bus		+	UEPPX	UEPP1	14	90	90				19.99			+	+
1		2-Wire Voice Unbundled PBX LD Terminal Ports		+	UEPPX	UEPLD	14	90	90				19.99			1	+
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14	90	90				19.99				T
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14	90	90				19.99				T
																	П
		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14	90	90				19.99				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		\perp	UEPPX	UEPXD	14	90	90				19.99				_
_		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port		\perp	UEPPX	UEPXE	14	90	90				19.99			_	4
+		2-Wire Voice Unbundled 2-Way PBX Kentucky Room Area Calling Port without LUD		+	UEPPX	UEPXF	14	90	90			1	19.99	-	-		4
		O Mire Veice Helevadled DDV Keetisela LLID Assa Cellina Det			HEDDY	LIEDYC	4.4	00	00			1	40.00				
+-		2-Wire Voice Unbundled PBX Kentucky LUD Area Calling Port		+	UEPPX	UEPXG	14	90	90		1	+	19.99	-	-	+	4
+		2-Wire Voice Unbundled PBX Kentucky Premium Calling Port 2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port without LUD		+	UEPPX UEPPX	UEPXH	14 14	90	90 90	1		-	19.99 19.99				+
+		2-Wire Voice Unbundled 2-Way Kentucky Area Calling Port Without LOD 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		+-+	UEPPA	UEPAJ	14	90	90		1		19.99	1			+
		Port			UEPPX	UEPXL	14	90	90			1	19.99				- [
H		1 01.		+	- OLF FA	OLI AL		30	30	1	1		13.33				$^{+}$
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPX	UEPXM	14	90	90			1	19.99				- [
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room		+ +				1 20	30			1	. 5.00	1		†	7
		Calling Port			UEPPX	UEPXO	14	90	90			1	19.99				- [
L		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14	90	90				19.99				J
																	J
LO		IBER PORTABILITY		┸													
	l	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15										_
L				4		\bot				1	1	1	1				4
FE	ATURES			4		\bot				1	1	1	1				4
	NDES::-	DINO OLIA POEG. OLIPPENTI VIOMBINITE		+		++		1		1	1		1	1			4
NO	NRECUR	RING CHARGES - CURRENTLY COMBINED		4		+						-					4
	DITIC	NDO.		+		++						1		-	-		4
AD	DITIONAL			4	(JESSA /	110.00						-					4
+		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent		+	UEPPX	USAS2		0	0			1		1	-		4
1		2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-										1					
		Nonrecurring		+		++		0	0			1	40.00	1	-		4
1		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64	1	I	1	19.99			+	4
Ł																	

ATEGORY	NOTES	ELEMENT Interim	Zone	BCS	USOC		R.	ATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manua Svc Order vs. Electronic-Add'l	Electronic-Disc	Incremen Charge Manual S Order v Electronic- Add'I
							Nonrec	curring		ecurring						
						_		Add'I	Disc	connect	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	LINE Dort/I	Loop Combination Rates				Rec	First	Addi	FIFSt	Add'l	SOMEC	SUMAN	SOMAN	SUMAN	SOMAN	SUMA
	ONE POIL	2-Wire VG Coin Port/Loop Combo – Zone 1				27.54										_
		2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2				33.73					_					-
		2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3				42.27										
		2-Wile VG Colli Poli/Loop Collido – Zolle 3	_			42.21								-		
	UNE Loop	Rates														t
		2-Wire Voice Grade Loop (SL1) - Zone 1		UEPCO	UEPLX	13.54										1
		2-Wire Voice Grade Loop (SL1) - Zone 2		UEPCO	UEPLX	19.73										
		2-Wire Voice Grade Loop (SL1) - Zone 3		UEPCO	UEPLX	28.27										
	2-Wire Voi	ice Grade Line Port Rates (Coin)														
		2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,														
		MS)		UEPCO	UEPRF	14	90	90				19.99				
		2-Wire Coin 2-Way with Operator Screening (AL, KY)		UEPCO	UEPRE	14	90	90				19.99				t
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD														1
		(AL, KY, LA, MS, SC)		UEPCO	UEPRA	14	90	90				19.99				
		2-Wire Coin 2-Way with Operator Screening and 011 Blocking (KY)		UEPCO	UEPKA	14	90	90					33.67	7.88		1
		2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &						7.7								
		Local (AL, KY, LA, MS)		UEPCO	UEPCD	14	90	90				19.99				
		2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,		UEPCO	UEPRN	14	90	90				19.99				1
		2-Wire Coin Outward with Operator Screening and 011Blocking (GA, KY, MS)		UEPCO	UEPRJ	14	90	90				19.99				
		2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD														
		(AL, KY, LA, MS)		UEPCO	UEPRH	14	90	90				19.99				
		2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, &														1
		Local (AL, KY, LA, MS)		UEPCO	UEPCN	14	90	90				19.99				
	LOCAL NU	JMBER PORTABILITY														
		Local Number Portability (1 per port)		UEPCO	LNPCX	0.35										
	NONRECU	RRING CHARGES - CURRENTLY COMBINED														
	ADDITION															
		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent		UEPCO	USAS2		0	0								1

EGORY	NOTES	ELEMENT Interim Zon	e BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
										Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increment Charge - Manual Sv Order vs. Electronic-D Add'I
				_		Nonrec	urring		ourring onnect						
					Rec	First	Add'I	First	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					Nec	THO	Addi	riist	Addi	JOINEO	JOHNA	SOMA	SOMA	SOMPA	JOHNA
AL INT	ERCONNEC	TION (CALL TRANSPORT AND TERMINATION)													
		Per MOU Rate for Local and ISP-bound Traffic (1/1/01-12/31/01)			\$0.0017500										
		Per MOU Rate for Local and ISP-bound Traffic (1/1/02-12/31/02)			\$0.0015000										
	TANDEM S	WITCHING													
		Tandem Switching Function Per MOU	OHD		\$0.0007555										
		Multiple Tandem Switching, per MOU (applies to intial tandem only)	OHD		\$0.0007555										
		Tandem Intermediary Charge, per MOU*	OHD		\$0.001096										
	* This charg	e is applicable only to transit traffic and is applied in addition to applicable switching													
	TRUNK CHA	ADOL													
		Installation Trunk Side Service - per DS0	OHD	TPP++		334.09bk	57.12bk								
		Dedicated End Office Trunk Port Service-per DS0**	OHD	TDE0P	\$0.00	334.03DK	37.12bk								
		Dedicated End Office Trunk Port Service-per DS0* Dedicated End Office Trunk Port Service-per DS1**	0H1 OH1MS	TDE1P	\$0.00										
			S OITINO	.JEII	ψ0.00										—
		Dedicated Tandem Trunk Port Service-per DS0**	OHD	TDW0P	\$0.00										
		Dedicated Tandem Trunk Port Service-per DS1**	OH1 OH1MS	TDW1P	\$0.00										
	** This rate	element is recovered on a per MOU basis and is included in the End Office Switching and Tandem S	Switching, per MOU i	rate elemer	nts										
AL INT	ERCONNEC	TION (TRANSPORT)													
	COMMONIT	RANSPORT (Shared)													
	COMMON	Common Transport - Per Mile, Per MOU	OHD		\$0.0000031										
		Common Transport - Facilities Termination Per MOU	OHD		\$0.000757										
	INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE													
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per	OHL, OHM	1L5NF	0.0118bk										
		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility													
		Termination per month	OHL, OHM	1L5NF	29.51bk	81.1bk	54.84bk	33.36bk	13.75bk						
	INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - 56/64 KBPS													
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month	OHL, OHM	1L5NK	0.0118bk										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month	OHL, OHM	1L5NK	21.26bk	81.11bk	54.84bk	33.36bk	13.75bk						
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month	OHL, OHM	1L5NK	0.0118bk										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month	OHL, OHM	1L5NK	21.26bk	81.11bk	54.84bk	33.36bk	13.75bk						
	INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - DS1													
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month	OH1 OH1MS	1L5NL	0.2407bk										
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month	OH1 OH1MS	1L5NL	97.38bk	178.59bk	163.67bk	32.59bk	28.79kb						
		CE CHANNEL - DEDICATED TRANSPORT- DS3													
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month	OH3 OH3MS	1L5NM	5.1bk										
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month	OH3 OH3MS	1L5NM	1191.53bk	557.69bk	325.62bk	120bk	116.54bk						
	LOCAL CH	ANNEL - DEDICATED TRANSPORT													
		Local Channel - Dedicated - 2-Wire Voice Grade per month	OHL OHM	TEFV2	18.81bk	386.33bk	66.35bk	73.04bk	6.37bk						
		Local Channel - Dedicated - 2-Wire Voice Grade per month	OHL OHM	TEFV4	20.12bk	387.2bk	67.22bk	73.98bk	7.31bk						+
		Local Channel - Dedicated - TVIII Voice Grade per month	OH1	TEFHG	44.63bk	355.06bk	307.53bk	44.24bk	30.42bk						
		Local Channel - Dedicated - DS3 Facility Termination per month	OH3	TEFHJ	583.57bk	903.34bk	528.05bk	238.2bk	166.62bk						
		ERCONNECTION MID-SPAN MEET													
	NOTE: IT AC	cess service ride Mid-Span Meet, one-half the tariffed service Local Channel rate is applical Local Channel - Dedicated - DS1 per month	OH1MS	TEFHG	\$0.00	\$0.00				 					1
		Local Channel - Dedicated - DS1 per month Local Channel - Dedicated - DS3 per month	OH1MS OH3MS	TEFHU	\$0.00	\$0.00				1					
		Local Grianner Dealoated - Doo per month	OI IOIVIO	ILIII	φυ.υυ	φυ.υυ				1					
	MULTIPLEX	(ERS													
		Channelization - DS1 to DS0 Channel System	OH1 OH1MS	SATN1	139.65bk	182.14bk	125.19bk	21bk	19.52bk						
		DS3 to DS1 Channel System per month	OH3 OH3MS	SATNS	194.82bk	356.4bk	188bk	66.3bk	63.44bk						
		DS3 Interface Unit (DS1 COCI) per month	OH1 OH1MS	SATCO	14.43bk	13.16bk	9.43bk								
	Notos: 411	no rate is identified in the contract the rates torrest and according to the contract the rates to the contract the rates to the contract to the contract the rates to the contract to the con	tion will be an and for	th in											
		no rate is identified in the contract, the rates, terms, and conditions for the specific service or func-													1
		ellSouth tariff; 2) "bk" beside a rate indicates that the parties have agreed to bill and keep for the el ns in Attachment 3.	ement pursuant to th	e terms					1						
	and act dist.														

Version 2Q01: 08/30/01

CATEGORY	NOTES	ELEMENT	Interim	Zone	BCS	USOC		RA*	TES (\$)					OSS R	ATES (\$)		
SAI ESSAI	NOTES								. = 0 (\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc
								Nonrecu	rring	Nonre	curring	perLSK	LSK	Electronic-1st	Electronic-Add I	1st	Add'l
								Nonrecu	rring		onnect						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	COLLOCAT	TION															
PHYSICAL COL																	<u> </u>
		Physical Collocation - Application Fee - Initial			CLO	PE1BA		3,761.00	3,761.00								
		Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		3,135.00	3,135.00								
		Physical Collocation - Application Fee - Subsequent for Co-Carrier Cross Connect			CLO	PE1DT		584.20	584.20								
		Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		1,202.00	1,202.00								
		Physical Collocation - Space Preparation - C.O. Modification per square ft.			CLO	PE1SK	2.38										
		Physical Collocation - Space Preparation - Common Systems Modification per square ft Cageless			CLO	PE1SL	3.30										
		Physical Collocation - Space Preparation - Common Systems Modification per Cage			CLO	PE1SM	112.11										
		Physical Collocation - Cable Installation			CLO	PE1BD		1,755.00	1,755.00								
		Physical Collocation - Floor Space per Sq. Ft.	_	-	CLO	PE1PJ	8.20										
		Physical Collocation - Cable Support Structure	-	+	CLO	PE1PM	20.14										ļ
		Physical Collocation - Power per Fused Amp		_	CLO	PE1PL PE1FB	8.77 5.58										1
		Physical Collocation - 120V, Single Phase Standby Power Rate	-	+	CLO												-
	 	Physical Collocation - 240V, Single Phase Standby Power Rate		 	CLO	PE1FD PF1FF	11.16 16.74		 	 	 	-	 				
		Physical Collocation - 120V, Three Phase Standby Power Rate Physical Collocation - 277V, Three Phase Standby Power Rate		+	CLO	PE1FG	38.65			l	l	 	l				
	-	i nysicai conocation - 277V, Tillee Filase Standby Fower Nate	 '-	+	UEANL,UEA,UDN,UDC,	FEIFG	38.00			l	l	1	l				
		Physical Collocation - 2-Wire Cross-Connects		1	UAL,UHL,UCL,UEQ	PE1P2	0.037	33.67	31.78	1	l		1				
		Physical Collocation - 4-Wire Cross-Connects			CLO	PE1P4	0.075	33.66	31.70								
			1		CLO,UEANL,UEQ,WDS					1	l		1				
		Physical Collocation - DS1 Cross-Connects		ļ	1L,WDS1S	PE1P1	1.51	52.97	39.90	 			 				
		Physical Collocation - DS3 Cross-Connects	-	1	CLO	PE1P3	19.15	52.04	38.62	-			-				-
		Physical Collocation - 2-Fiber Cross-Connect	-	-	CLO	PE1F2	3.80	52.04	38.63								-
		Physical Collocation - 4-Fiber Cross-Connect	-	+	CLO	PE1F4	6.75	64.59	51.18								ļ
		Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.	-	_	CLO	PE1BW PE1CW	189.85 18.62										1
		Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.	-	+			78.11										
		Physical Collocation - Security Access System - Security System per Central Office		+	CLO CLO	PE1AX PE1A1	78.11 0.059	55.59	55.59	l							ļ
		Physical Collocation - Security Access System - New Access Card Activation, per Card		+	CLO	PETAT	0.059	33.39	55.59								-
		Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Card			CLO	PE1AA		15.59	15.59								
		Physical Collocation - Security Access System - Replace Lost or Stolen Card, per Card			CLO	PE1AR		45.58	45.58								
		Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		26.20	26.20								
		Physical Collocation - Security Access - Key, Replace Lost or Stolen Key, per Key			CLO	PE1AL		26.20	26.20								
		Physical Collocation - Space Availability Report per premises			CLO	PE1SR		2,151.00	2,151.00								
					UEANL,UEA,UDN,UDC,												
		POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			UAL,UHL,UCL,UEQ,CL O	PE1PE	0.06										
		TOT Bay Arrangements prior to or 1/35 - 2-44 the Gross-Contrect, per Gross-Contrect		1	UEANL,UEA,UDN,UDC,	TEHE	0.00			l	l		l				···
					UAL, UHL, UCL, UEQ, CL												
		POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			0	PE1PF	0.15										
					UEANL,UEA,UDN,UDC, UAL,UHL,UCL,UEQ,CL												
		POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			O,WDS1L,WDS1S,	PE1PG	0.58										
		To the bay Arrangements prior to or 1735 - Do thorosa-connect, per cross-connect		_	UEANI UEA UDN UDC	TEILO	0.30										
					UAL, UHL, UCL, UEQ, CL												
		POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect			0	PE1PH	4.51										
					UEANL,UEA,UDN,UDC,												
		POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			UAL,UHL,UCL,UEQ,CL O	PE1B2	38.79										
		1 Or Day Fill angent and prior to or 1788 - 2-1 lost or observourinest, per cross-conflect	_	 	UEANL,UEA,UDN,UDC,	I LIDZ	30.79				l	1					1
				1	UAL,UHL,UCL,UEQ,CL					1	l		1				
		POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect		1	0	PE1B4	52.31			l			l				
		Collocation Cable Records - per request		1	CLO	PE1CR		1,709.00	1,166.00								ļ
	I	Collocation Cable Records - VG/DS0 Cable, per cable record		-	CLO	PR1CD		923.83	923.83		l		ļ				
			1	-	CLO	PE1CO		18.03	18.03	ļ	l	1	-				1
		Collocation Cable Records - VG/DS0 Cable, per each 100 pair	_			PE1C1		8.44	8.44		ļ	-					
—		Collocation Cable Records - DS1, per T1TIE		1	CLO				29.54		ı		1		1		
		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		29.54									
		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - Fiber Cable, per 99 fiber records			CLO CLO	PE1C3 PE1CB		279.05	279.05								ļ
\vdash		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour			CLO CLO CLO,CLORS	PE1C3 PE1CB PE1BT		279.05 33.86	279.05 21.46								
		Collocation Cable Records - DS1, per T1TE Collocation Cable Records - DS3, per T3TE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Owntime, per Half Hour			CLO CLO CLO,CLORS CLO,CLORS	PE1C3 PE1CB PE1BT PE1OT		279.05 33.86 44.10	279.05 21.46 27.72								
		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Premium, per Half Hour			CLO CLO CLO,CLORS CLO,CLORS CLO,CLORS CLO,CLORS	PE1C3 PE1CB PE1BT PE1OT PE1PT	0.003	279.05 33.86	279.05 21.46								
		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS2, per T3TIE Collocation Cable Records - DS2, per T3TIE Collocation - Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Reside, per Half Hour Physical Collocation - Security Escort - Portrium, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft.			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES	0.003	279.05 33.86 44.10	279.05 21.46 27.72								
		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Premium, per Half Hour			CLO CLO CLO,CLORS CLO,CLORS CLO,CLORS CLO,CLORS	PE1C3 PE1CB PE1BT PE1OT PE1PT	0.003 0.0045	279.05 33.86 44.10	279.05 21.46 27.72								
ADJACENT CO		Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS2, per TSTBE Collocation Cable Records - Size per TSTBE Collocation - Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Perfiner, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft.			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES		279.05 33.86 44.10	279.05 21.46 27.72								
ADJACENT CO		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Certime, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per line ft.			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES PE1DS	0.0045	279.05 33.86 44.10	279.05 21.46 27.72								
ADJACENT CO		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS2, per T3TIE Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Premium, per Half Hour Physical Collocation - Security Escort - Premium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per lin. ft. Adjacent Collocation - Space Charge per Sq. Ft.			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES PE1DS PE1JA	0.0045	279.05 33.86 44.10	279.05 21.46 27.72								
ADJACENT CO		Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS3, per TSTIE Collocation Cable Records - DS3, per TSTIE Collocation Cable Records - DS4, per 199 fiber records Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per lin. ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLO CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO CLO CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC	0.0045 0.02 6.01	279.05 33.86 44.10 54.35	279.05 21.46 27.72 33.97								
ADJACENT CO		Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS3, per TSTIE Collocation Cable Records - DS3, per TSTIE Collocation Cable Records - DS2, per TSTIE Physical Collocation - Security Escort - Basic, per Half Hour Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Permium, per Half Hour Physical Collocation - Socurity Escort - Permium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per lin. ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2	0.0045 0.02 6.01 0.04	279.05 33.86 44.10 54.35 33.67	279.05 21.46 27.72 33.97								
ADJACENT CO		Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS2, per T3TIE Collocation Cable Records - DS2, per T3TIE Collocation Cable Records - Step Cable, per 99 fiber records Physical Collocation - Security Escort - Passic, per Half Hour Physical Collocation - Security Escort - Permirun, per Half Hour Physical Collocation - Security Escort - Permirun, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per lin. ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Wire Cross-Connects Adjacent Collocation - Wire Cross-Connects			CLO CLO,CLORS CLO,CLORS CLO,CLORS CLO,CLORS CLO CLO CLO CLO CLO CLO CLO CLO CLO CLO	PE1C3 PE1CB PE1BT PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4	0.0045 0.02 6.01 0.04 0.08	279.05 33.86 44.10 54.35 33.67	279.05 21.46 27.72 33.97 31.78 31.70								
ADJACENT CO		Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS2, per TSTIE Collocation Cable Records - DS2, per TSTIE Collocation Cable Records - Size per SIZE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Collocation - Security Escort - Persime, per Half Hour Physical Collocation - Security Escort - Persime, per Half Hour Physical Collocation - Co-Crairer Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Crairer Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Crairer Cross Connects - Copper/Coax Cable Support Structure, per lin. ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft. Adjacent Collocation - 4-Wirre Cross-Connects Adjacent Collocation - 4-Wirre Cross-Connects Adjacent Collocation - 5ST Cross-Connects			CLO CLO,CLORS CLO,CLORS CLO,CLORS CLO CLO CLO CLO CLO CLO CLO CLO CLO UEA,UH_,UDL,UCL,CL O USL,CLO	PE1C3 PE1C8 PE1BT PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4 PE1P1	0.0045 0.02 6.01 0.04 0.08 1.51	279.05 33.86 44.10 54.35 33.67 33.66 52.97	279.05 21.46 27.72 33.97 31.78 31.70 39.90								
ADJACENT CO		Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS3, per TSITE Collocation Cable Records - DS3, per TSITE Collocation Cable Records - DS3, per TSITE Collocation Cable Records - DS4, per TSITE Physical Collocation - Security Escort - Parisin, per Half Hour Physical Collocation - Security Escort - Parisin, per Half Hour Physical Collocation - Security Escort - Parisin, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per line ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Biscricia Facility Charge per Linear Ft. Adjacent Collocation - Sya Cross-Connects Adjacent Collocation - Sya Cross-Connects Adjacent Collocation - Sya Cross-Connects Adjacent Collocation - Sya Cross-Connects			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO CLO CLO CLO CLO CLO CLO CLO	PE1C3 PE1C8 PE1BT PE1DT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4 PE1P1 PE1P3	0.0045 0.02 6.01 0.04 0.08 1.51 19.15	279.05 33.86 44.10 54.35 33.67 33.67 33.67 52.97 52.04	279.05 21.46 27.72 33.97 31.78 31.70 39.90 38.62								
ADJACENT CO	LLOCATION	Colocation Cable Records - DS1, per TITIE Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS2, per TS1E Collocation Cable Records - Size per SIZE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Colocation - Security Escort - Persiman Per Half Hour Physical Colocation - Security Escort - Persiman per Half Hour Physical Colocation - Socurity Escort - Persiman per Half Hour Physical Colocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Colocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft. Adjacent Colocation - Size Charge per Sq. Ft. Adjacent Colocation - Electrical Facility Charge per Linear Ft. Adjacent Colocation - 4-Wire Cross-Connects Adjacent Colocation - Size Consection - Size Conse			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO CLO CLO CLO CLO CLO UEA, UHL, UDL, UCL, CL CLO CLO CLO CLO CLO CLO CLO CLO CLO C	PE1C3 PE1C8 PE1B7 PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4 PE1P1 PE1P3 PE1P3 PE1F2	0.0045 0.02 6.01 0.04 0.08 1.51 19.15 3.80	279.05 33.86 44.10 54.35 33.67 33.66 52.97 52.04	279.05 21.46 27.72 33.97 31.78 31.70 39.90 38.62 38.63								
ADJACENT CO	LLOCATION	Collocation Cable Records - DS1, per T1TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS3, per T3TIE Collocation Cable Records - DS3, per T3TIE Physical Collocation - Security Escort - Passic, per Half Hour Physical Collocation - Security Escort - Persium, per Half Hour Physical Collocation - Security Escort - Persium, per Half Hour Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per line ft. Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft. Adjacent Collocation - St Wire Cross-Connects Adjacent Collocation - St Wire Cross-Connects Adjacent Collocation - St St Cross-Connects Adjacent Collocation - St St Cross-Connects Adjacent Collocation - Fiber Cross-Connects Adjacent Collocation - Fiber Cross-Connects Adjacent Collocation - Fiber Cross-Connects Adjacent Collocation - Fiber Cross-Connects			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO CLO CLO CLO CLO CLO CLO CLO	PE1C3 PE1C8 PE1B7 PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4 PE1P1 PE1P1 PE1P2 PE1P4 PE1P1 PE1P5	0.0045 0.02 6.01 0.04 0.08 1.51 19.15	279.05 33.86 44.10 54.35 33.67 33.66 52.97 52.04 64.59	279.05 21.46 27.72 33.97 31.78 31.70 39.90 38.62								
ADJACENT CO	DLLOCATION	Colocation Cable Records - DS1, per TITIE Collocation Cable Records - DS1, per TITIE Collocation Cable Records - DS2, per TS1E Collocation Cable Records - Size per SIZE Collocation Cable Records - Fiber Cable, per 99 fiber records Physical Colocation - Security Escort - Persiman Per Half Hour Physical Colocation - Security Escort - Persiman per Half Hour Physical Colocation - Socurity Escort - Persiman per Half Hour Physical Colocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft. Physical Colocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft. Adjacent Colocation - Size Charge per Sq. Ft. Adjacent Colocation - Electrical Facility Charge per Linear Ft. Adjacent Colocation - 4-Wire Cross-Connects Adjacent Colocation - Size Consection - Size Conse			CLO CLO, CLORS CLO, CLORS CLO, CLORS CLO CLO CLO CLO CLO CLO CLO CLO UEA, UHL, UDL, UCL, CL CLO CLO CLO CLO CLO CLO CLO CLO CLO C	PE1C3 PE1C8 PE1B7 PE1OT PE1PT PE1ES PE1DS PE1JA PE1JC PE1P2 PE1P4 PE1P1 PE1P3 PE1P3 PE1F2	0.0045 0.02 6.01 0.04 0.08 1.51 19.15 3.80	279.05 33.86 44.10 54.35 33.67 33.66 52.97 52.04	279.05 21.46 27.72 33.97 31.78 31.70 39.90 38.62 38.63								

CATEGORY	NOTES	ELEMENT	Interim	Zone	BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs.		Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
								Nonrec	urring		curring						
							Rec	First	Add'l	Disc First	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FD	11.16	FIISL	Add I	riist	Add I	SOMEC	SOMAN	SOMAN	SOMAN	SOWAN	SOMAN
		Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FE	16.74										
		Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FG	38.65										
PHYSICAL CO	LLOCATION IN	I THE REMOTE SITE															
		Physical Collocation in the Remote Site - Application Fee *			CLO	PE1RA		\$868.91	\$868.91								
		Cabinet Space in the Remote Site per Bay/ Rack *			CLO	PE1RB	\$224.41										
		Physical Collocation in the Remote Site - Security Access - Key *			CLO	PE1RD		\$26.60	\$26.60								
		Physical Collocation in the Remote Site - Space Availability Report per Premises Requested *			CLO	PE1SR		\$231.82	\$231.82								
		Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested *			CLO	PE1RE		\$75.13	\$75.13								
	* Interim rates	which are subject to true-up.															
	NOTE: If Secu	rity Escort and/or Add'l Engineering Fees become necessary for remote site collocation, the Parties will negotiate	e appropri	ate rates	3.												

ATEGORY	NOTES	ELEMENT	Interim	Zone	BCS	USOC		R/	ATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st			Increment Charge - Manual Sv Order vs Electronic-D Add'I
								Nonrec	urring	Nonre	curring						
										Disc	onnect						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NUMBER PO	ORTABILITY			+												
	. to																
RVICE P	ROVIDER NU	JMBER PORTABILITY (RIPH)															
	Mate. If an	rate is identified in the contract, the rate for the specific service or function will le			aliantia DallCa	4b 40 siff a 2 00											
		and CLEC will each bear their own costs of providing remote call forwarding as															-
	i) Delicouli	and occo will each ocal their own costs of providing remote call forwarding as	arrinterimmu	T P	ortability option	. (111)											-
DUF/EDOL	JF/ADUF/CN	IDS															
		AILY USAGE FILE (ADUF)															
		ADUF: Message Processing, per message				N/A	\$0.004										
		ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	\$0.001										
	OPTIONAL I	DAILY USAGE FILE (ODUF)			+												
		ODUF: Recording, per message		1		N/A	\$0.0008611										
		ODUF: Message Processing, per message		1		N/A	\$0.0032357										
		ODUF: Message Processing, per Magnetic Tape provisioned		1		N/A	\$55.68										
		ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	\$0.0000365										
	CENTRAL IZ	ED MESSAGE DISTRIBUTION SERVICE (CMDS)		-	-												₩
		CMDS: Message Processing, per message	-	+		N/A	\$0.004		1	-		1	1				
		CMDS: Nessage Processing, per message CMDS: Data Transmission (CONNECT:DIRECT), per message			+	N/A	\$0.004					1	1				<u> </u>

CATEC	GORY RATE ELEMENTS	Interim	Zone	e BCS	USOC		I	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	t Increment -al Charge - Manual r Svc Order vsElectronic- Disc Add'l
										curring			000 0	TEO (A)		
						Rec	Nonrec First	urring Add'l	First	nnect Add'l	SOMEC	SOMAN	SOMAN	ATES (\$) SOMAN	COMAN	SOMAN
							FIRST	Add I	FIFST	Addi	SUMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
																+
																+
	The "Zone" shown in the sections for stand-alone loops or loops as part of a combination	refers 1	to Ge	ographically Deaver	aged UNI	Zones. To	riew Geograph	ically Deaverag	ged UNE Z	one Designa	ations by C	Central Offic	e, refer to Int	ernet Websi	te:	*
	http://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm	1									1	1	T	ı		
	IAL SUPPORT SYSTEMS							the decay		Ti					- Co 1 Co. 41 C	
	NOTE: (1) Electronic Service Order: CLEC-1 should contact its contract negotiator if it pr															s rate
	exhibit is the BellSouth regional electronic service ordering charge. CLEC-1 may elect eit															
	NOTE: (2) Any element that can be ordered electronically will be billed according to the S elements that cannot be ordered electronically at present per the BBR-LO, the listed SOM															
	charge, SOMAN, will be applied to a CLECs bill when it submits an LSR to BellSouth.	EC rate	in this	s category reflects t	ne charg	e that would i	be billed to a C	LEC once elec	tronic ora	ering capabi	littles come	e on-line for	tnat elemen	. Otherwise	, the manu	ai ordering
	Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces		T													$\overline{}$
	(Regional)				SOMEC		3.50									
UNBUNDLE	D EXCHANGE ACCESS LOOP		1													
	2-WIRE ANALOG VOICE GRADE LOOP		1													1
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.90	36.54	16.87				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	23.33	36.54	16.87	-			15.20				
	2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	48.43	36.54	16.87				15.20				
	Loop Testing - Basic 1st Half Hour			UEANL	URET1		33.17	33.17								
	Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.28	19.28								
	Engineering Information Document (EI)			UEANL			13.04	13.04								
	Manual Order Coordination for UVL-SL1s (per loop)*			UEANL	UEAMC		7.92	7.92								
	Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *			UEANL	OCOSL		17.56	17.56								
	2-WIRE Unbundled COPPER LOOP		1	UEQ	UEQ2X	12.40	05.07	45.00				45.00				
	2-Wire Unbundled Copper Loop - Non-Designed Zone 1 2 Wire Unbundled Copper Loop - Non-Designed - Zone 2		2		UEQ2X	14.32	35.27 35.27	15.60 15.60				15.20 15.20				
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	- 1	3	UEQ	UEQ2X	16.87	35.27	15.60				15.20				+
	Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loop)		3	UEQ	USBMC	10.07	7.92	7.92				13.20				+
	Engineering Information Document			UEQ	CODINO		13.04	13.04								+
	Loop Testing - Basic 1st Half Hour			UEQ	URET1		33.17	33.17								+
	Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.28	19.28								
UNBUNDLE	D EXCHANGE ACCESS LOOP															
	2-WIRE ANALOG VOICE GRADE LOOP															
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	- 1	1	UEPSR UEPSB	UEALS	12.90	36.54	16.87	0.00	0.00		15.20				
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	- 1		UEPSR UEPSB	UEABS	12.90	36.54	16.87	0.00	0.00		15.20				
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	!	2	UEPSR UEPSB	UEALS	23.33	36.54	16.87	0.00	0.00		15.20				
	2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	- !	_		UEABS	23.33	36.54	16.87	0.00	0.00		15.20				
	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3 2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3	- 1	3		UEALS UEABS	48.43 48.43	36.54 36.54	16.87 16.87	0.00	0.00		15.20 15.20				+
UNRUNDI FI	D EXCHANGE ACCESS LOOP			OLF SIX OLF SB	ULADO	40.43	30.34	10.07	0.00	0.00		13.20				+
	2-WIRE ANALOG VOICE GRADE LOOP		+													+
	CLEC to CLEC Conversion Charge without outside dispatch (UVL-SL1)		1	UEANL	UREWO		36.54	16.87				15.20				+
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start		1													
\perp	Signaling - Zone 1		1	UEA	UEAL2	14.93	102.10	65.72								
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start			1.15.4	LIEALG	05.65	400 : 0	05.70				45.00				
\vdash	Signaling - Zone 2	-	2	UEA	UEAL2	25.35	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	50.46	102.10	65.72				15.20				
	Order Coordination for Specified Conversion Time (per LSR)		3	UFA	OCOSL	30.40	17.56	05.72				13.20				+
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -			OLA	OOOOL		17.50									+
	Zone 1		1	UEA	UEAR2	14.93	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -		1													
	Zone 2		2	UEA	UEAR2	25.35	102.10	65.72				15.20				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -															
	Zone 3		3		UEAR2	50.46	102.10	65.72				15.20				
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		17.56	20.00				45.00				
\vdash	CLEC to CLEC Conversion Charge without outside dispatch 4-WIRE ANALOG VOICE GRADE LOOP		+	UEA	UREWO		102.10	38.22				15.20			-	+
	4-WIRE ANALOG VOICE GRADE LOOP 4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	30.81	127.40	91.02				15.20				+
—	4-Wire Analog Voice Grade Loop - Zone 1 4-Wire Analog Voice Grade Loop - Zone 2	-	2	UEA	UEAL4	38.32	127.40	91.02				15.20				+
 	4-Wire Analog Voice Grade Loop - Zone 2 4-Wire Analog Voice Grade Loop - Zone 3		3	UEA	UEAL4	60.39	127.40	91.02				15.20				+
	Order Coordination for Specified Conversion Time (per LSR)		"	UEA	OCOSL	00.09	17.56	31.02				10.20				+
1:	2-WIRE ISDN DIGITAL GRADE LOOP		1	1												+
	2-Wire ISDN Digital Grade Loop - Zone 1				U1L2X	22.09	113.34	76.96				15.20				
	2-Wire ISDN Digital Grade Loop - Zone 2			UDN	U1L2X	35.28	113.34	76.96				15.20				
1	2-Wire ISDN Digital Grade Loop - Zone 3		3		U1L2X	65.18	113.34	76.96				15.20				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		17.56			· · · · · · · · · · · · · · · · · · ·	1	1			1	

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge Manual Svc Order vs. Electronic- Disc 1st	-al Charge - Manual
						_				curring						
						Rec	Nonrec			onnect	001150	001111		ATES (\$)	001111	001111
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
C	CLEC to CLEC Conversion Charge without outside dispatch			UDN	UREWO		113.34	33.04				15.20				
	ersal Digital Channel (UDC) COMPATIBLE LOOP															
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1			UDC	UDC2X	22.09	113.34	76.96				15.20				<u> </u>
2	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2 2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3		2		UDC2X	35.28 65.18	113.34	76.96 76.96				15.20				
	CLEC to CLEC Conversion Charge without outside dispatch		3	UDC	UDC2X UREWO	65.18	113.34 113.34	33.04				15.20 15.20				-
	MMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP			ODO	OKEWO		110.04	33.04				10.20				
2	Wire Unbundled ADSL Loop including manual service inquiry & facility reservation															
-	Zone 1		1	UAL	UAL2X	12.29	117.08	68.36				15.20				
	Wire Unbundled ADSL Loop including manual service inquiry & facility reservation		2	UAL	UAL2X	14.00	117.08	00.00				45.00				
- 2	Zone 2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation			UAL	UALZX	14.09	117.08	68.36				15.20				-
	Zone 3	1	3	UAL	UAL2X	15.75	117.08	68.36			1	15.20				
C	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		17.56									
2	Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -				T											
	Zone 1		1	UAL	UAL2W	12.29	92.83	56.02			1	15.20				<u> </u>
	Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2		2	UAL	UAL2W	14.09	92.83	56.02				15.20				
	Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -			O/IL	ONLEVV	14.03	32.00	30.02				10.20				
z	Zone 3		3	UAL	UAL2W	15.75	92.83	56.02				15.20				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		17.56									
	CLEC to CLEC Conversion Charge without outside dispatch			UAL	UREWO		92.83	29.29				15.20				
	H BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP Wire Unbundled HDSL Loop including manual service inquiry & facility reservation															
	Zone 1		1	UHL	UHL2X	9.79	125.50	76.77				15.20				
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation			OFIL	OFILEX	3.73	120.00	70.77				10.20				
-	Zone 2		2	UHL	UHL2X	11.52	125.50	76.77				15.20				
	Wire Unbundled HDSL Loop including manual service inquiry & facility reservation															
	Zone 3			UHL	UHL2X	12.74	125.50	76.77				15.20				
2	Order Coordination for Specified Conversion Time (per LSR) Wire Unbundled HDSL Loop without manual service inquiry and facility			UHL	OCOSL		17.56									-
re	eservation - Zone 1		1	UHL	UHL2W	9.79	101.24	64.43				15.20				
2	Wire Unbundled HDSL Loop without manual service inquiry and facility															
re	eservation - Zone 2		2	UHL	UHL2W	11.52	101.24	64.43				15.20				
	Wire Unbundled HDSL Loop without manual service inquiry and facility					40.74	404.04	04.40				45.00				
	eservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)			UHL UHL	UHL2W OCOSL	12.74	101.24 17.56	64.43				15.20				-
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO		101.24	29.29				15.20				-
	HIBIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP			0.12	U.V.E.V.G		101.21	20.20				10.20				
4	Wire Unbundled HDSL Loop including manual service inquiry and facility															
re	eservation - Zone 1		1	UHL	UHL4X	16.24	153.26	104.54				15.20				
	I-Wire Unbundled HDSL Loop including manual service inquiry and facility eservation - Zone 2		2	UHL	UHL4X	16.65	153.26	104.54				15.20				
	I-Wire Unbundled HDSL Loop including manual service inquiry and facility			UNL	UHL4X	10.05	153.26	104.54				15.20				-
	eservation - Zone 3		3	UHL	UHL4X	17.34	153.26	104.54				15.20				
C	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		17.56									
	I-Wire Unbundled HDSL Loop without manual service inquiry and facility															
	eservation - Zone 1 I-Wire Unbundled HDSL Loop without manual service inquiry and facility		1	UHL	UHL4W	16.24	129.00	92.20				15.20				
	eservation - Zone 2		2	UHL	UHL4W	16.65	129.00	92.20				15.20				
	I-Wire Unbundled HDSL Loop without manual service inquiry and facility		_	0.12	0.12	10.00	120.00	02.20				10.20				
	eservation - Zone 3			UHL	UHL4W	17.34	129.00	92.20				15.20				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		17.56									
4 WIDE DOL	CLEC to CLEC Conversion Charge without outside dispatch DIGITAL LOOP			UHL	UREWO		101.24	29.29			1	15.20				<u> </u>
	I-Wire DS1 Digital Loop - Zone 1	-	1	USL	USLXX	85.70	245.16	152.98			1	15.20				1
4	I-Wire DS1 Digital Loop - Zone 1	 		USL	USLXX	194.96	245.16	152.98		1	+	15.20				
	I-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	491.94	245.16	152.98				15.20				
C	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		17.56									
	CLEC to CLEC Conversion Charge without outside dispatch			USL	UREWO		130.07	39.99				15.20				
	, 56 OR 64 KBPS DIGITAL GRADE LOOP										1					1
	Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	30.99	121.86	85.48			1	15.20				
	Wire Unbundled Digital 19.2 Kbps Wire Unbundled Digital 19.2 Kbps		2		UDL19 UDL19	36.78 38.92	121.86 121.86	85.48 85.48				15.20 15.20				-
	Wire Unbundled Digital 19.2 Kbps Wire Unbundled Digital Loop 56 Kbps - Zone 1	-	3	UDL	UDL19	30.99	121.86	85.48 85.48				15.20				
	Wire Unbundled Digital Loop 56 Kbps - Zone 1	-		UDL	UDL56	36.78	121.86	85.48				15.20				-
										1	1					

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		1	RATES(\$)			Svc Order Submitte d Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Manual Svc Order vs. Electronic	Increment -al Charge Manual r Svc Orde vsElectronic Disc Add
										curring						
						Rec	Nonrec			onnect		T		ATES (\$)	T	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																+
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		17.56									-
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1		UDL64	30.99	121.86	85.48				15.20				+
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2		2		UDL64	36.78	121.86	85.48				15.20				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3			UDL	UDL64	38.92	121.86	85.48				15.20				1
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		17.56									
	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO		121.86	38.63				15.20				
	bundled COPPER LOOP															
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility															
	reservation - Zone 1		1	UCL	UCLPB	12.29	116.18	67.46				15.20				
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		2	UCL	UCLPB	44.00	440.40	67.46				45.00				
	reservation - Zone 2 2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility			UCL	UCLPB	14.09	116.18	67.46				15.20				+
	reservation - Zone 3		3	UCL	UCLPB	15.75	116.18	67.46				15.20				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	10.73	7.92	7.92			1	10.20			1	+
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility										1	1			1	1
	reservation - Zone 1		1	UCL	UCLPW	12.29	91.92	55.12				15.20				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility															
	reservation - Zone 2		2	UCL	UCLPW	14.09	91.92	55.12				15.20				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility		3	UCL	UCLPW	15.75	91.92	55.12				15.20				
	reservation - Zone 3			UCL	UCLPW	15.75	7.92	7.92				15.20				+
	Order Coordination for Unbundled Copper Loops (per loop) 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility			UCL	UCLIVIC		7.92	1.92								+
	reservation - Zone 1		1	UCL	UCL2L	17.21	116.18	67.46				15.20				
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			002	COLLE		110.10	07.10				10.20				-
	reservation - Zone 2		2	UCL	UCL2L	24.98	116.18	67.46				15.20				
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 3			UCL	UCL2L	39.57	116.18	67.46				15.20				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		7.92	7.92								
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		1	UCL	UCL2W	17.21	91.92	55.40				15.20				
	reservation - Zone 1 2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		- 1	UCL	UCLZVV	17.21	91.92	55.12				15.20				+
	reservation - Zone 2		2	UCL	UCL2W	24.98	91.92	55.12				15.20				
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		-	002	COLLIN	21.00	01.02	00.12				10.20				-
	reservation - Zone 3		3	UCL	UCL2W	39.57	91.92	55.12				15.20				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		7.92	7.92								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		91.92	31.37				15.20				
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-ND)			UEQ	UREWO		36.53	16.16				15.20				
	PPER LOOP															
	4-Wire Copper Loop/Short - including manual service inquiry and facility reservation - Zone 1		1	UCL	UCL4S	22.27	139.69	90.96				15.20				
	4-Wire Copper Loop/Short - including manual service inquiry and facility		'	UCL	UCL43	22.21	139.09	90.96				15.20				+
	reservation - Zone 2		2	UCL	UCL4S	18.95	139.69	90.96				15.20				
	4-Wire Copper Loop/Short - including manual service inquiry and facility															
	reservation - Zone 3		3	UCL	UCL4S	10.99	139.69	90.96				15.20				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		7.92	7.92								
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -	-														
	Zone 1		1	UCL	UCL4W	22.27	115.43	78.63				15.20				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation	-	2	ucı	LICL AW	40.05	445.40	70.00				45.00				
	Zone 2 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -		2	UCL	UCL4W	18.95	115.43	78.63				15.20				_
	Zone 3		3	UCL	UCL4W	10.99	115.43	78.63				15.20				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	10.55	7.92	7.92				10.20				+
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															1
	reservation - Zone 1		1	UCL	UCL4L	26.17	139.69	90.96				15.20				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 2		2	UCL	UCL4L	28.47	139.69	90.96				15.20				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		3	UCL	UCL4L	62.93	139.69	90.96				15.20				
	reservation - Zone 3	1		UCL	UCL4L UCLMC	6∠.93	7.92	7.92			1	15.20				+
	Order Coordination for Unbundled Copper Loops (per loop) 4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility	+	 	UUL	UCLIVIC		1.92	1.92	-		1	1			1	+
	reservation - Zone 1		1	UCL	UCL4O	26.17	115.43	78.63				15.20				1
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility				302.0	20	7.00	. 0.00			1	.0.20				+
	reservation - Zone 2	<u> Ш</u>	2	UCL	UCL4O	28.47	115.43	78.63		<u></u>	<u> </u>	15.20			<u> </u>	
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility															
					UCL4O	62.93	115.43	78.63	1	1	1	15.20	1	1	1	1
	reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)			UCL UCL	UCLMC	02.93	7.92	7.92				.0.20				+

CATEGORY	TEGORY RATE ELEMENTS			BCS	usoc		F	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svo Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charge Manual Svc Orde vs. Electronic
						_				curring				(4)		
						Rec	Nonrec			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
LOOP MODIFICATIO	N															
LOOP MODIFICATIO	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or			UAL, UHL, UCL,												
	equal to 18k ft			UEQ, ULS	ULM2L		0.00	0.00								
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft			UCL, ULS	ULM2G		0.00	0.00								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to			002, 020	OLIVILO		0.00	0.00								
	18K ft			UHL, UCL	ULM4L		0.00	0.00								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k															
	ft			UCL	ULM4G		0.00	0.00								
				UAL, UHL, UCL,												
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UEQ, UEF, ULS	ULMBT		12.15	12.15								
SUB-LOOPS																
Sub-Loc	p Distribution															
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up	I		UEANL	USBSA		144.09	144.09				15.20				
	Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	- 1		UEANL	USBSB		10.99	10.99				15.20				
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	- 1		UEANL	USBSC		86.16	86.16				15.20				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up			UEANL	USBSD		27.13	27.13				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	ı		UEANL	USBN2	7.57	63.89	30.06				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	- 1		UEANL	USBN2	12.75	63.89	30.06				15.20				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	ı	3	UEANL	USBN2	21.45	63.89	30.06				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1			UEANL	USBN4	11.76	76.75	42.92				15.20				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2			UEANL	USBN4	16.84	76.75	42.92				15.20				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN4	19.27	76.75	42.92				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	- 1		UEANL	USBR2	2.91	51.48	17.65				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)			UEANL	USBR4	6.58	57.54	23.71				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		7.92	7.92								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	-!-	1	UEF	UCS2X	6.26	63.89	30.06				15.20				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2			UEF	UCS2X	10.07	63.89	30.06				15.20				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı		UEF	UCS2X	12.70	63.89	30.06				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		7.92	7.92								
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	-!-		UEF	UCS4X	8.03	76.75	42.92				15.20				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	!		UEF	UCS4X	10.71	76.75	42.92				15.20				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	ı	3	UEF	UCS4X	6.08	76.75	42.92				15.20				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		7.92	7.92								
Unbund	ed Sub-Loop Modification															
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per			ucc			0.00	0.00				45.00				
	2-W PR Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-			UEF	ULM2X		0.00	0.00				15.20				
	W PR	1		UEF	ULM4X		0.00	0.00				15.20				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per			UEF	ULIVI4X		0.00	0.00				13.20				
	PR unloaded			UEF	ULM4T		224.55	4.29				15.20				
Unhund	ed Network Terminating Wire (UNTW)			OLI	OLIVIA I		224.00	4.23			 	13.20				
Onbana	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3454	14.72	14.72				15.20				
Network	Interface Device (NID)				OE. 11 1	0.0404	17.12	17.72				10.20				
, ioinom	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		42.26	27.83				15.20				
	Network Interface Device (NID) - 1-6 lines			UENTW	UND16		62.86	48.43				15.20				
	Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		5.73	5.73				15.20				
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.73	5.73				15.20				
SUB-LOOPS																
	p Feeder															
				UEA,												
				UDN,UCL,UDL,UI												
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up	<u> </u>	L	С	USBFW		144.09							<u> </u>		<u> </u>
				UEA,										1		
				UDN,UCL,UDL,UI												
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up			С	USBFX		10.99	10.99								
	USL Feeder DS1 Set-up at DSX location, per DS1 termination			USL	USBFZ		568.98	11.30								
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1			UEA	USBFA	8.71	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2			UEA	USBFA	13.64	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3			UEA	USBFA	30.21	89.81	54.35				15.20				
	Order Coordination for Specified Conversion Time, per LSR			UEA	OCOSL		17.56									
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1			UEA	USBFB	8.71	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2			UEA	USBFB	13.64	89.81	54.35				15.20		1		
1	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3	1 -	3	UEA	USBFB	30.21	89.81	54.35	l		1	15.20		1		

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	al Charge Manual Svc Order vs. Electronic	Manual
						B				curring			000 0	ATEO (A)		ļ
					_	Rec	Nonrec First	urring Add'l	First	nnect Add'l	SOMEC	SOMAN		ATES (\$) SOMAN	SOMAN	SOMAN
							FIISL	Auu i	FIISL	Auui	JUNEC	SUMAN	SOWAN	SOWAN	SUMAN	SOWAN
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		17.56									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1			UEA	USBFC	8.71	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2		2	UEA	USBFC	13.64	89.81	54.35				15.20				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade -		_													1
	Zone 3			UEA	USBFC	30.21	89.81	54.35				15.20				
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL USBFD	21.44	17.56 103.69	67.31				15 20				\vdash
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2		2	UEA	USBFD	24.66	103.69	67.31				15.20 15.20				\vdash
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 3			UEA	USBFD	42.84	103.69	67.31				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL	42.04	17.56	07.51				13.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1			UEA	USBFE	21.44	103.69	67.31				15.20				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	1	2	UEA	USBFE	24.66	103.69	67.31			1	15.20			İ	
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3		3	UEA	USBFE	42.84	103.69	67.31				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		17.56									
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1	L		UDN	USBFF	15.44	102.58	66.20				15.20				\Box
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2			UDN	USBFF	23.32	102.58	66.20				15.20				<u> </u>
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3		USBFF	44.57	102.58	66.20				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			UDN	OCOSL		17.56									
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	15.44	102.58	66.20				15.20				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	23.32 44.57	102.58 102.58	66.20				15.20				oxdot
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible) Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			UDC USL	USBFS	55.38	98.15	66.20 61.77				15.20 15.20				\vdash
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2			USL	USBFG	167.83	98.15	61.77				15.20				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3			USL	USBFG	469.87	98.15	61.77				15.20				
	Order Coordination For Specified Conversion Time, Per LSR			USL	OCOSL	100.07	17.56	0				10.20				
	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1			UCL	USBFH	6.96	81.36	44.98				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2		2	UCL	USBFH	4.97	81.36	44.98				15.20				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3		3	UCL	USBFH	3.99	81.36	44.98				15.20				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL		17.56									
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1	UCL	USBFJ	15.68	98.07	61.69				15.20				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2		USBFJ	9.68	98.07	61.69				15.20				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3			UCL	USBFJ	6.39	98.07	61.69				15.20				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL	00.01	17.56	04.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN USBFN	22.61 22.87	98.15 98.15	61.77 61.77				15.20				oxdot
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			UDL	USBFN	24.25	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 19.2 Robs Digital Grade Loop - Zone 1		1		USBFO	22.61	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2		2		USBFO	22.87	98.15	61.77				15.20				\vdash
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3		3		USBFO	24.25	98.15	61.77				15.20				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		17.56									
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1		1	UDL	USBFP	22.61	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2			UDL	USBFP	22.87	98.15	61.77				15.20				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3	ļ		UDL	USBFP	24.25	98.15	61.77			1	15.20				<u> </u>
	Order Coordination For Specified Conversion Time, per LSR			UDL	OCOSL		17.56									
SUB-LOOPS	Fooder	ļ			_						1	1				
Sub-Loop	Sub Loop Feeder - DS3 - Per Mile Per Month	-		UE3	1L5SL	17.00					1	-				\vdash
	Sub Loop Feeder - DS3 - Per Mile Per Month Sub Loop Feeder - DS3 - Facility Termination Per Month	1		UE3 UE3	USBF1	368.44	3.381.00	406.56			1	15.20			1	
	Sub Loop Feeder - STS-1 - Per Mile Per Month	 		UDLSX	1L5SL	17.00	3,301.00	400.00			1	15.20				
	Sub Loop Feeder - STS-1 - Fer Wille Fer World? Sub Loop Feeder - STS-1 - Facility Termination Per Month	1		UDLSX	USBF7	395.92	3,381.00	406.56			1	15.20			1	
	Sub Loop Feeder – OC-3 – Per Mile Per Month	 		UDLO3	1L5SL	12.90	5,551.50	.00.00				.0.20				
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per Month			UDLO3	USBF5	60.45										
	Sub Loop Feeder - OC-3 - Facility Termination Per Month			UDLO3	USBF2	594.77	3,381.00	406.56				15.20				
	Sub Loop Feeder - OC-12 - Per Mile Per Month			UDL12	1L5SL	15.87		_							1	
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per Month			UDL12	USBF6	683.03										
	Sub Loop Feeder - OC-12 - Facility Termination Per Month			UDL12	USBF3	1,922.00	3,381.00	406.56				15.20				<u></u> '
	Sub Loop Feeder - OC-48 - Per Mile Per Month	ļ		UDL48	1L5SL	52.07					1	1			1	<u> </u>
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month	-		UDL48	USBF9	341.64	0.500.00	4								
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	ļ		UDL48	USBF4	1,663.00	3,566.00	406.56			1	15.20				
UNBUNDLED LOOP C	Sub Loop Feeder - OC-12 Interface On OC-48	-		UDL48	USBF8	385.45	787.24	406.56			1	15.20				\vdash
ONDONDED LOOP C	Unbundled Loop Concentration - System A (TR008)	 		ULC	UCT8A	374.26	316.00	316.00			1	15.20				
	Unbundled Loop Concentration - System A (TR008)	 		ULC	UCT8B	53.40	131.67	131.67			+	15.20				
	Unbundled Loop Concentration - System B (TR000)	1		ULC	UCT3A	412.08	316.00	316.00			1	15.20				
	Unbundled Loop Concentration - System X (TR303)	1		ULC	UCT3B	89.98	131.67	131.67		1	1	15.20	1		1	\vdash

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ſ	RATES(\$)			Svc Order Submitte d Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	vs. Electronic-	
									Nonre	curring						
						Rec	Nonrec			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
														_		_
	Unbundled Loop Concentration - DS1 Loop Interface Card			ULC	UCTCO	5.12	61.46	44.74				15.20		<u> </u>		
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)			UDN	ULCC1	8.12	10.23	10.18				15.20				
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop			UDC	ULCCU	8.12	10.23	10.18				15.20		-		ļ
	Unburidied Loop Concentration2 write voice-Loop Start or Ground Start Loop Interface (POTS Card)			UEA	ULCC2	2.03	10.23	10.18				15.20				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface			UEA	ULUUZ	2.03	10.23	10.16				13.20		-		
	(SPOTS Card)			UEA	ULCCR	12.07	10.23	10.18				15.20				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card)			UEA	ULCC4	7.20	10.23	10.18				15.20		†	+	
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	35.19	10.23	10.18				15.20		-	+	
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface			UDL	ULCC7	10.67	10.23	10.18				15.20		<u> </u>	+	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface			UDL	ULCC5	10.67	10.23	10.18				15.20		<u> </u>	+	
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface		 	UDL	ULCC6	10.67	10.23	10.18			+	15.20			+	
UNE OTHER, PROVISION		1	t —		32300	10.07	10.23	10.10			1	10.20		 	+	
	NID - Dispatch and Service Order for NID installation	1	1	UENTW	UNDBX						1				+	
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		 	UENTW	UENCE						 			 	+	
	ONTO OF CUIT IN ESTABLISHMENT, 1 TOVISIONING OTHY THE TRACE			UEANL, UEF, UEQ,	OLIVOL									<u> </u>	+	
	Unbundled Contract Name, Provisioning Only - No Rate	1	1	UENTW	UNECN						1					
UNE OTHER, PROVISION	NING ONLY - NO RATE			OZ.TTT	OTTE OTT											
5 <u>5</u> 55410101				UAL,UCL,UDC,UD							†				 	
		1		L,UDN,UEA,UHL,U												
	Unbundled Contact Name, Provisioning Only - no rate	1	1	LC	UNECN	0.00	0.00				1					
	,,,			UEA,UDN,UCL,UD			5.50									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate	1		С	USBFQ	0.00	0.00									
				UEA,USL,UCL,UD												
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			L	USBFR	0.00	0.00									
	Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
	Unbundled DS1 Loop - Expanded Superframe Format option - no rate			USL	CCOEF	0.00	0.00									
HIGH CAPACITY UNBUN	IDLED LOCAL LOOP															
	onth minimum billing period															
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	10.04										ļ
	High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	362.34	438.46	256.30				15.20				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per month			UDLSX	1L5ND	10.04		-								
	High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	374.56	438.46	256.30				15.20				
LOOP MAKE-UP																
	Loop Makeup - Preordering Without Reservation, per working or spare facility															
	queried (Manual).			UMK	UMKLW		23.29	23.29								
	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		24.70	24.70								
	Loop MakeupWith or Without Reservation, per working or spare facility queried															
	(Mechanized)			UMK	PSUMK		0.19	0.19								
HIGH FREQUENCY SPEC	CIRUM	1														
	-CENTRAL OFFICE BASED	 	-													
	Line Sharing Splitter, per System 96 Line Capacity			ULS	ULSDA	187.17	183.33	0.00	0.00	0.00		0.00				
	Line Sharing Splitter, per System 24 Line Capacity	1		ULS	ULSDB	46.79	183.33	0.00	0.00	0.00		0.00				
	Line Sharing Splitter, Per System, 8 Line Capacity			ULS	ULSD8	15.59	183.33	0.00	0.00	0.00		0.00				
	Line Sharing-DLEC Owned Splitter in CO-CFA activaton-deactivation (per LSOD)	1		ULS	ULSDG		83.98		0.00		1					
	ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LIN	E SHARI														
	Line Sharing - per Line Activation	<u> </u>		ULS	ULSDC	0.61	17.97	10.29	0.00	0.00		15.20				
	Line Sharing - per Subsequent Activity per Line Rearrangement	1 !	-	ULS	ULSDS		15.91	7.95				15.20				
	Line Splitting - per line activation DLEC owned splitter	1		UEPSR UEPSB	UREOS	0.61										
	Line Splitting - per line activation BST owned - physical		L	UEPSR UEPSB	UREBP	0.642	17.97	10.29			1					
	Line Splitting - per line activation BST owned - virtual		L	UEPSR UEPSB	UREBV	0.64	17.97	10.29			1					
UNBUNDLED TRANSPO		1	L								1					
	CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE		-													
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per	1		U1TVX	1L5XX	0.013										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility															
	Termination per month	1	-	U1TVX	U1TV2	22.60	39.36	26.62				15.20		 		
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per	1		LUTON	41.500	0.010										
	Mile per month			U1TVX	1L5XX	0.013								_		_
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility	1		LIATORY	LIATES	00.00	00.00	00.00	2.22			45.00				
	Termination per month Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per	1	-	U1TVX	U1TR2	22.60	39.36	26.62	0.00	0.00		15.20		 		
		1		LUTON	41.577	0.040										
	month Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility	1	-	U1TVX	1L5XX	0.013										
		1		U1TVX	U1TV4	19.81	39.36	26.62				45.00				
	Termination per month Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month	1	-	U1 I VX U1TDX	1L5XX	0.013	39.36	26.62			-	15.20		 	+	+
			+				20.07	20.00			-	45.00		 	+	
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month	ч	ĺ	U1TDX	U1TD5	15.61	39.37	26.62		l	1	15.20		1	1	

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		ŀ	RATES(\$)	N.		d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charge - Manual Svc Order vs.
						Rec	Nonrec	urring		curring nnect			OSS P	ATES (\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.013										
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month	l .		U1TDX	U1TD6	15.61	39.37	26.62	0.00	0.00		15.20				
INTERO	FICE CHANNEL - DEDICATED TRANSPORT - DS1			LIATEA	41.57/7	0.0050										
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			U1TD1 U1TD1	1L5XX U1TF1	0.2652 70.47	86.69	79.44				15.20				
INTERO	FICE CHANNEL - DEDICATED TRANSPORT- DS3			OTIDI	01111	70.47	00.09	73.44				13.20				
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX	6.04										
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	850.45	270.69	158.05				15.20				
INTERO	FICE CHANNEL - DEDICATED TRANSPORT- STS-1															
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TS1	1L5XX	6.04										
	Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month			U1TS1	U1TFS	830.19	270.69	158.05				15.20				
	CHANNEL - DEDICATED TRANSPORT															
NOTE: L	OCAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=	one mon	tn, DS			40.00	107.51	20.01				45.00				
	Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month	-		ULDVX ULDVX	ULDV2 ULDR2	18.32 18.32	187.51 187.51	32.21 32.21	0.00	0.00		15.20				
	Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 4-Wire Voice Grade per month	1		UNDVX	ULDK2	18.32	187.51	32.21	0.00	0.00		15.20 15.20				
	Local Channel - Dedicated - 4-Wife Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1		1	ULDD1	ULDF1	39.18	172.34	149.27				15.20				
	Local Channel - Dedicated - DS1 per month - Zone 2		2	ULDD1	ULDF1	121.58	172.34	149.27				15.20				
	Local Channel - Dedicated - DS1 per month - Zone 3			ULDD1	ULDF1	70.02	172.34	149.27				15.20				
	Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3	1L5NC	7.82										
	Local Channel - Dedicated - DS3 - Facility Termination per month			ULDD3	ULDF3	469.44	438.46	256.30				15.20				
	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	7.82										
	Local Channel - Dedicated - STS-1 - Facility Termination per month			ULDS1	ULDFS	457.22	438.46	256.30				15.20				
MULTIPLEXERS	01			LIVEDA	1104	405.00	00.44	00.70				45.00				
	Channelization - DS1 to DS0 Channel System OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UXTD1 UDL	MQ1 1D1DD	105.09 1.38	88.41 6.39	60.76 4.58				15.20 15.20				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System - per month			UDN	UC1CA	2.96	6.39	4.58				15.20				
	Voice Grade COCI - DS1 to DS0 Channel System - per month			UEA	1D1VG	0.6497	6.39	4.58				15.20				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	201.48	172.99	91.25				15.20				
	STS1 to DS1 Channel System per month			UXTS1	MQ3	201.48	172.99	91.25				15.20				
	DS3 Interface Unit (DS1 COCI) used with Loop per month			USL	UC1D1	11.78	6.39	4.58				15.20				
DARK FIBER																
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -															
	Local Channel			UDF	1L5DC	52.23	000.00	100.00				45.00				
	NRC Dark Fiber - Local Channel Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -			UDF	UDFC4		620.60	133.88				15.20				
	Interoffice Channel			UDF	1L5DF	25.28										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14	20.20	620.60	133.88				15.20				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -			0.5.	OD: 11		020.00	100.00				10.20				
	Local Loop			UDF	1L5DL	52.23										
	NRC Dark Fiber - Local Loop			UDF	UDFL4		620.60	133.88				15.20				
TRANSPORT OTHER																
Optional	Features & Functions: Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel	1		UNC1X	CCOEF		184.65	23.70				15.20				
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel			UNC1X UNC1X	CCOSF		184.65	23.70				15.20				
8XX ACCESS TEN D	ICIT SCREENING			UNCIA	CCOSF		104.00	23.70				15.20				
OAA ACCESS TEND	8XX Access Ten Digit Screening, Per Call			OHD		0.0006387										
	8XX Access Ten Digit Screening, Per Calif	İ		OHD	N8R1X		2.51	0.43				15.20				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS	1		OHD			5.77	0.78				15.20				
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS			OHD	N8FTX		5.77	0.78				15.20				
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number			OHD	N8FCX		2.51	1.26				15.20				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR															
	Requested Per 8XX No.	1		OHD	N8FMX		2.93	1.68				15.20				
	8XX Access Ten Digit Screening, Change Charge Per Request 8XX Access Ten Digit Screening, Call Handling and Destination Features	 		OHD OHD	N8FAX N8FDX		2.93 2.51	0.43				15.20 15.20				
	8XX Access Ten Digit Screening, Call Handling and Destination Features 8XX Access Ten Digit Screening, w/ 8XX No. Delivery, per query	 		OHD	INOLDY	0.0006387	2.51				-	15.20				
	8XX Access Ten Digit Screening, w/ 6XX No. Delivery, per query	1		OHD		0.0006387										
LINE INFORMATION	DATA BASE ACCESS (LIDB)	1				2.0000007										
	LIDB Common Transport Per Query			OQT		0.0000221										
	LIDB Validation Per Query			OQU		0.0135077										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		33.33					15.20				
SIGNALING (CCS7)								-								
	CCS7 Signaling Termination, Per STP Port			UDB	PT8SX	147.60										
	CCS7 Signaling Usage, Per TCAP Message	ļ		UDB		0.000064										
	CCS7 Signaling Connection, Per link (A link)	-		UDB	TPP++	15.77	34.50	0150				15.20				
	CCS7 Signaling Connection, Per link (B link) (also known as D link)			UDB	TPP++	15.77	34.50	34.50			l	15.20				

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ŗ	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st		al Charge Manual Svc Order vs. Electronic	Increment -al Charge - Manual Svc Order vsElectronic- Disc Add'l
							_				curring						
							Rec	Nonrec			onnect				ATES (\$)		
			1	-		_		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		CCS7 Signaling Usage, Per ISUP Message			UDB	071150	0.000016										
		CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	732.10										
		CCS7 Signaling Point Code, per Originating Point Code Establishment or Change,															
		per STP affected			UDB	CCAPO		28.17	28.17				15.20				
		CCS7 Signaling Point Code, per Destination Point Code Establishment or Change,			LIDD	00400		00.47	00.47				45.00				
		Per Stp Affected			UDB	CCAPD		28.17	28.17				15.20				
E911 SERV	ICE																
		Local Channel - Dedicated - 2-wr Voice Grade - Zone 1					18.32	187.51	32.21				15.20				
		Local Channel - Dedicated - 2-wr Voice Grade - Zone 2					18.32	187.51	32.21				15.20				
		Local Channel - Dedicated - 2-wr Voice Grade - Zone 3					18.32	187.51	32.21				15.20				L
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile					0.013										
		Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination					22.60	79.61	36.08				15.20				
		Local Channel - Dedicated - DS1 - Zone 1					39.18	172.34	149.27				15.20				
		Local Channel - Dedicated - DS1 - Zone 2					121.58	172.34	149.27				15.20				<u> </u>
		Local Channel - Dedicated - DS1 - Zone 3					70.02	172.34	149.27				15.20				
		Interoffice Transport - Dedicated - DS1 Per Mile					0.2652										
		Interoffice Transport - Dedicated - DS1 Per Facility Termination		lacksquare			70.47	147.07	111.75				15.20				
CALLING N	IAME (CNAM																
		CNAM for DB Owners, Per Query			OQV		0.0010217										
		CNAM for Non DB Owners, Per Query			OQV		0.0010217										
		CNAM For DB Owners - Service Establishment			OQV			22.29					15.20				
		CNAM For Non DB Owners - Service Establishment			OQV			22.29					15.20				
		CNAM For DB Owners - Service Provisioning With Point Code Establishment			OQV			962.22	711.64				15.20				
		CNAM For Non DB Owners - Service Provisioning With Point Code Establishment			OQV			332.43	238.05				15.20				
LNP Query	Service	, , , , , , , , , , , , , , , , , , ,															1
		LNP Charge Per query			OQV		0.0008559										
		LNP Service Establishment Manual						12.16					15.20				1
		LNP Service Provisioning with Point Code Establishment						576.33	294.43				15.20				
OPERATOR	R CALL PRO																
		Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
		Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
		Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										
		Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
INWARD OF	PERATOR S						0.20										
		Inward Operator Services - Verification, Per Minute					1.15										
		Inward Operator Services - Verification and Emergency Interrupt - Per Minute					1.15										+
BRANDING		PR CALL PROCESSING					1.13										+
DIVANDINO		Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.20				+
						CBAOL		500.00	500.00				15.20				+
	Unbranding	Loading of Custom Branded OA Announcement per shelf/NAV g via OLNS for UNEP CLEC				CBACL		500.00	500.00				15.20				+
								1 200 00	1 200 00				15 20				+
DIRECTOR	V ASSISTAN	Loading of OA per OCN (Regional) ICE SERVICES	-	-				1,200.00	1,200.00			1	15.20	1			+
		Y ASSISTANCE ACCESS SERVICE	1	1			+				1	1		1		1	+
-			1	1			0.25				1	1		1		1	+
-	DIDECTOR	Directory Assistance Access Service Calls, Charge Per Call Y ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)	1	-			0.25					1		1			-
_			-	-			0.10					1		1			+
		Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.10										-
		Y TRANSPORT	1	-			0.0000			-	1	1	-		l	1	+
-		SWA Common transport per Directory Assistance Access Service Call	1	1			0.0003										
		SWA Common Transport per Directory Assistance Access Service Call Mile	1	-			0.00004							1			
		Access Tandem Switching per Directory Assistance Access Service Call		1			0.00055										
		Directory Assistance Interconnection per Directory Assistance Access Service Call	1	-			0.00							1			
		DS3 to DS1 Multiplexer per DA Access Service Call	1	1			0.00018										
		ICE SERVICES															
		Y ASSISTANCE DATA BASE SERVICE (DADS)	1	1													
		Directory Assistance Data Base Service Charge Per Listing					0.04										
		Directory Assistance Data Base Service, per month				DBSOF	150.00										
		RY ASSISTANCE															
	Facility Bas	sed CLEC															
		Recording and Provisioning of DA Custom Branded Announcement			AMT	CBADA		6,000.00	6,000.00								
		Loading of Custom Branded Announcement per DRAM Card/Switch			AMT	CBADC		1,170.00	1,170.00								
	UNEP CLE	3		╙													
		Recording of DA Custom Branded Announcement					İ	3,000.00	3,000.00								
		Loading of DA Custom Branded Announcement per DRAM Card/Switch per OCN						1,170.00	1,170.00								
			1								i e	1					
	Unbranding	g via OLNS for UNEP CLEC															
	Unbranding	Loading of DA per OCN (1 OCN per Order)						420.00	420.00								

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		F	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.		al Charge Manual Svc Order vs.
				I		_		_		curring						
			<u> </u>	-		Rec	Nonrec			nnect				ATES (\$)	_	
			<u> </u>	-			First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			<u> </u>													
SELECTIVE ROUTIN	10		₩		_											
SELECTIVE ROUTIN	Selective Routing Per Unique Line Class Code Per Request Per Switch		+-	<u> </u>	USRCR		82.25	82.25				15.20	_	+	+	
VIRTUAL COLLOCA			+-		USKCK		02.20	02.20				15.20	 			
VIKTUAL COLLOCA	Virtual Collocation - Application Cost		+-	CLO	FAF		1,770.40						 			
	Virtual Collocation - Cable Installation Cost, per cable			CLO	ESPCX		841.54					-	-	+	+	
	Virtual Collocation - Floor Space, per sq. ft.			CLO	ESPVX	3.20	041.04						<u> </u>	+	+	
	Virtual Collocation - Power, per breaker amp			CLO	ESPAX	8.32									+	
	Virtual Collocation - Cable Support Structure, per entrance cable			CLO	ESPSX	16.02								1		
				ueanl,uea,udn,udc											+	
	Virtual Collocation - 2-wire Cross Connects (loop)			ual,uhl,ucl,ueq	UEAC2	0.0296	11.94	11.46				15.20				
	Virtual Collocation - 4-wire Cross Connects (loop)			uea,uhl,ucl,udl	UEAC4	0.0591	12.04	11.53				15.20				
	Virtual Collocation - 2-Fiber Cross Connects			CLO	CNC2F	2.65	20.29	14.76				15.20				
	Virtual Collocation - 4-Fiber Cross Connects			CLO	CNC4F	5.31	24.81	19.29				15.20				
	Virtual Collocatin - DS1 Cross Connects			USL,ULC,CLO	CNC1X	1.04	21.39	15.47				15.20				
	Virtual Collocatin - DS3 Cross Connects			USL,ULC,CLO	CND3X	13.21	20.28	14.76				15.20				
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	PE1ES	0.0024										
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	PE1DS	0.0036										
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,per cable			AMTFS			534.79									
	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support			AMTEC			504.70									
	Structure, per cable		+	AMTFS	CDTDV		534.79	10.10					-			
	Virtual Collocatin - Security Escort - Basic, per half hour			CLO	SPTBX		16.44 21.41	10.42					-			
	Virtual Collocatin - Security Escort - Overtime, per half hour Virtual Collocatin - Security Escort - Premium, per half hour			CLO CLO	SPTPX		26.38	13.45 16.49					_	+	+	
	Virtual Collocatin - Security Escott - Fremium, per half hour			CLO	CTRLX		27.12	10.49							+	
	Virtual Collocatin - Maintenance in CO - Dasic, per half hour			CLO	SPTOM		35.42	13.45						-	+	
	Virtual Collocatin - Maintenance in CO - Premium per half hour			CLO	SPTPM		43.72	16.49					<u> </u>	+	+	
VIRTUAL COLLOCA			\vdash	1			10.72	10.10							+	
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire Analog - Res			UEPSR	VE1R2	0.0296	11.94	11.46				15.20			+	
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade Res			UEPRX	PE1R2	0.0296	11.94	11.46				15.20				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0296	11.94	11.46				15.20				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX															
	Trunk - Res			UEPSE	VE1R2	0.0296	11.94	11.46				15.20				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			UEPSB	VE1R2	0.0296	11.94	11.46				15.20				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN		-	UEPSX	VE1R2	0.0296	11.94	11.46				15.20	<u> </u>			
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN		-	UEPTX	VE1R2	0.0296	11.94	11.46				15.20	<u> </u>			
	Virtual Collocation 4-Wire Cross Connect, Exchange Port DDITS 4-Wire DS1		\vdash	UEPDD UEPEX	VE1R4 VE1R4	0.0591	12.04	11.53				15.20				
VIRTUAL COLLOCA	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1		+-	UEPEX	VE IR4	0.0591	12.04	11.53				15.20	_	+	+	
AIN SELECTIVE CAI	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting	I		UEPSR, UEPSB	VE1LS	0.0296	11.94	11.46	0.00	0.00		15.20				
AII OLLLO IIVE CAI	Regional Service Establishment		+-	UEBIB	SRCEC		100,209.33				1	15.20	 		+	
	End Office Establishment		+-	UEBIB	SRCEO		164.29	164.29			+	15.20	<u> </u>	 	+	
	Query NRC, per query		\vdash	UEBIB	UNOLO	0.0030293	104.23	104.23				10.20	<u> </u>	+	+	
AIN - BELLSOUTH A	AIN SMS ACCESS SERVICE		+-	1	+-+	2.0000200							1		†	
	AIN SMS Access Service - Service Establishment, Per State, Initial Setup		\vdash	A1N	CAMSE		38.30	38.30			1	15.20				
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.60	7.60				15.20			 	
	AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		7.60	7.60				15.20				
	AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		33.99	33.99				15.20				
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		41.39	41.39				15.20				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0022										
	AIN SMS Access Service - Session, Per Minute					0.5795										
	AIN SMS Access Service - Company Performed Session, Per Minute		<u> </u>			0.8104							<u> </u>			
AIN - BELLSOUTH A	AIN TOOLKIT SERVICE		<u> </u>	!												
1 1	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup		<u> </u>	CAM	BAPSC		38.30	38.30			1	15.20				
—	AIN Toolkit Service - Training Session, Per Customer	-	 		BAPVX		4,175.10	4,175.10			1	15.20				
				I .	BAPTT		7.60	7.60				15.20				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt		+												1	1
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		7.60	7.60				15.20			+	-
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook															
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		7.60	7.60				15.20				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook															

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ı	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	al Charge - Manual Svc Order vs. Electronic-	Increment al Charge - Manual Svc Order vs. Electronic- Disc Add'l
							_				curring						I.
							Rec	Nonrec			nnect				ATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																	
		AIN Toolkit Service - Query Charge, Per Query					0.0536446										
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node,					0.0000110										
		Per Query					0.006569										
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100															
		Kilobytes AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription			CAM	BAPMS	0.06 10.90	7.60	7.60				15.20				-
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription			CAM	BAPLS	2.80	8.41	8.41				15.20				+
		AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			CAM	BAPDS	8.20	7.60	7.60				15.20				
		AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service			CAM	BAPES	0.09	8.41	8.41				15.20				
ENHANCED		D LINK (EELs)															
	NOTE: New	EELs available in State of Georgia, density zone 1 of following SMAs: Orlando	, FL; Mia	mi, FL	Ft. Lauderdale	, FLI; Nashv	ille, TN; New	Orleans, LA;									
		rlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use a															
		Il states, EEL network elements shown below also apply to currently combined					s. A Switch As	s is Charge ap	plies to currer	ntly combin	ed facilities	converted	to UNEs.(No	on-recurring	rates do not	apply.)	
		A, TN, KY, LA & MS, the EEL network elements apply to ordinarily combined no ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO			SA notiwe only	is charge.)											
		First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1	()		UNCVX	UEAL2	14.93	94.21	45.09				15.20				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination -															
		Zone 2		2	UNCVX	UEAL2	25.35	94.21	45.09				15.20				
		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination -		_									,				
		Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile per month			UNCVX UNC1X	UEAL2 1L5XX	50.46 0.2652	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile per month Interoffice Transport - Dedicated - DS1 combination - Facility Termination per			UNC1X UNC1X	U1TF1	70.47	143.58	103.88				15.20				
		DS1 Channelization System Per Month			UNC1X	MQ1	105.09	59.97	12.96				15.20				
		Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.6497	5.91	4.26				10.20				
		Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport															
		Combination - Zone 1		1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	25.35	94.21	45.09				15.20				
		Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport			ONCVA	ULALZ	20.00	34.21	45.05				13.20				
		Combination - Zone 3		3	UNCVX	UEAL2	50.46	94.21	45.09				15.20				
		Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.6497	5.91	4.26								
	4 14/105 1/0	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	DT (EEL)		UNC1X	UNCCC		5.43	5.43				15.20				
	4-WIRE VO	ICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	KI (EEL)														
		Zone 1		1	UNCVX	UEAL4	30.81	94.21	45.09				15.20				
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	-														
		Zone 2		2	UNCVX	UEAL4	38.32	94.21	45.09				15.20				
		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	-	3	1.10.00		00.00	04.04	45.00				45.00				
		Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		-	UNCVX UNC1X	UEAL4 1L5XX	60.39 0.2652	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - DS1 - Combination - Fer Mile Fer Month	1		UNC1X	U1TF1	70.47	143.58	103.88				15.20				
		Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	105.09	59.97	12.96								
		Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.6497	5.91	4.26								
l		Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport		1	LINGVA	LIE AL 4	20.01	04.04	45.00				45.00				
		Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	1	-	UNCVX	UEAL4	30.81	94.21	45.09				15.20				
		Combination - Zone 2		2	UNCVX	UEAL4	38.32	94.21	45.09				15.20				
		Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport															
	1	Combination - Zone 3	ļ		UNCVX	UEAL4	60.39	94.21	45.09				15.20				ļ
		Voice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	-		UNCVX UNC1X	1D1VG UNCCC	0.6497	5.91 5.43	4.26 5.43				15.20				
		KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	PORT (F		UNCIA	UNCCC		5.45	5.45				13.20				
	. WIIL 30	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination															
		- Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
		First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	1			UDI 55	00 ==	04.51	45.00				45.60				
		Zone 2 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	-	2	UNCDX	UDL56	36.78	94.21	45.09				15.20				
		- Zone 3		3	UNCDX	UDL56	38.92	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		-	UNC1X	1L5XX	0.2652										
		Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per			UNC1X	U1TF1	70.47	143.58	103.88				15.20				
		Channelization - Channel System DS1 to DS0 combination Per Month	 		UNC1X	MQ1	105.09	59.97	12.96								<u> </u>
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs) Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport	1		UNCDX	1D1DD	1.38	5.91	4.26								
l		Combination - Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
		Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport															
I	1	Combination - Zone 2		2	UNCDX	UDL56	36.78	94.21	45.09				15.20		l		1

CATEGORY	RATE ELEMENTS	Interim Zone	BCS	USOC		ı	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental I	I Charge - Manual Svc Order vs.	Manual Svc Order vs. Electronic	-al Charge Manual Svc Orde vs.
					Rec	Nonrec	urring		curring onnect			OSS RAT	res (\$)		
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN			SOMAN	SOMAN
	Additional A Miss SCIChas Disital Conda Language DCA Interesting Transport														
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport Combination - Zone 3	3	UNCDX	UDL56	38.92	94.21	45.09				15.20				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-	3	UNCDA	UDLS6	36.92	94.21	45.09				13.20	-			+
	64kbs)		UNCDX	1D1DD	1.38	5.91	4.26								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		5.43	5.43				15.20				
4-WIRE 64	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	PORT (EEL)													
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	1	UNCDX	UDL64	30.99	94.21	45.09				15.20				
	- Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		UNCDA	UDL64	30.99	94.21	45.09				15.20	 			
	- Zone 2	2	UNCDX	UDL64	36.78	94.21	45.09				15.20				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination														1
	- Zone 3	3	UNCDX	UDL64	38.92	94.21	45.09				15.20				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		UNC1X	1L5XX	0.2652							\perp			
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per		UNC1X	U1TF1	70.47	143.58	103.88				15.20				1
	Channelization - Channel System DS1 to DS0 combination Per Month	-	UNC1X	MQ1	105.09	59.97	12.96					++			1
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-64kbs)		UNCDX	1D1DD	1.38	5.91	4.26								
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		SINODA	טטוטו	1.30	0.81	4.20			1		+			+
	Combination - Zone 1	1	UNCDX	UDL64	30.99	94.21	45.09				15.20				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport														
	Combination - Zone 2	2	UNCDX	UDL64	36.78	94.21	45.09				15.20				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport														
	Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-	3	UNCDX	UDL64	38.92	94.21	45.09				15.20				-
	64kbs)		UNCDX	1D1DD	1.38	5.91	4.26								
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC	1.00	5.43	5.43				15.20				+
	1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	RT (EEL)	0.10174	0.1000		0.10	0.10				10.20				†
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1	1	UNC1X	USLXX	85.70	169.22	100.89				15.20				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		UNC1X	USLXX	194.96	169.22	100.89				15.20				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3	3	UNC1X	USLXX	491.94	169.22	100.89				15.20				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		UNC1X	1L5XX	0.2652										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X UNC1X	U1TF1 UNCCC	70.47	143.58 5.43	103.88 5.43				15.20 15.20	-			-
	Nonrecurring Currently Combined Network Elements Switch -As-is Charge 1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR	T (EEL)	UNCIX	UNCCC		5.43	5.43				15.20				+
4-WIKE D3	First DS1Loop in DS3 Interoffice Transport Combination - Zone 1		UNC1X	USLXX	85.70	169.22	100.89				15.20	+			+
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		UNC1X	USLXX	194.96	169.22	100.89				15.20				t
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3		UNC1X	USLXX	491.94	169.22	100.89				15.20				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month		UNC3X	1L5XX	6.04										
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month		UNC3X	U1TF3	850.45	296.68	121.16				15.20				
	DS3 to DS1 Channel System combination per month		UNC3X	MQ3	201.48	107.05	48.07								
	DS3 Interface Unit (DS1 COCI) combination per month		UNC1X	UC1D1	11.78	5.91	4.26				15.20	-			+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2	2	UNC1X UNC1X	USLXX	85.70 194.96	169.22 169.22	100.89 100.89				15.20				+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3		UNC1X	USLXX	491.94	169.22	100.89				15.20				+
	DS3 Interface Unit (DS1 COCI) combination per month	, i	UNC1X	UC1D1	11.78	5.91	4.26				10.20				†
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC3X	UNCCC		5.43	5.43				15.20				
2-WIRE VO	CE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPO	RT (EEL)													
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	1	UNCVX	UEAL2	14.93	94.21	45.09				15.20				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		UNCVX	UEAL2	25.35	94.21	45.09			1	15.20				1
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3	3	UNCVX	UEAL2	50.46	94.21	45.09				15.20	+		 	+
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2- Wire Voice Grade combination - Facility		UNCVX	1L5XX	0.013							++			+
	Termination per month		UNCVX	U1TV2	22.60	72.60	41.75				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC		5.43	5.43				15.20				1
4-WIRE VO	CE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPO														
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		UNCVX	UEAL4	30.81	94.21	45.09				15.20				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		UNCVX	UEAL4	38.32	94.21	45.09				15.20				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3	3	UNCVX	UEAL4	60.39	94.21	45.09				15.20				
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility	 	UNCVX	1L5XX	0.013			-				+			+
	Termination per month		UNCVX	U1TV4	19.81	72.60	41 75				15.20				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNCVX	UNCCC	13.01	5.43	5.43				15.20	+ +			1
	L EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)			1 223		20	20								
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month		UNC3X	1L5ND	10.04										
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per														1
1	month		UNC3X	UE3PX	362.34	188.45	125.51								1

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ı	RATES(\$)			Submitte d Elec	Submitted	Charge - Manual Svc Order vs.	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charge - Manual Svc Order vs.
											curring						
							Rec	Nonrec			onnect				TES (\$)	T =	Γ
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	6.04										
		Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per					0.0.										
		month			UNC3X	U1TF3	850.45	296.68	121.16				15.20				
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EE			UNC3X	UNCCC		5.43	5.43				15.20			-	
	STST DIGIT	High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month	L)	-	UNCSX	1L5ND	10.04									<u> </u>	
		High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per		1	UNCOX	TESIND	10.04									 	
		month			UNCSX	UDLS1	374.56	188.45	125.51								ı l
		Interoffice Transport - Dedicated - STS1 combination - Per Mile per month			UNCSX	1L5XX	6.04										
		Interoffice Transport - Dedicated - STS1 combination - Facility Termination per				==											ı l
		month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCSX	U1TFS UNCCC	830.19	296.68 5.43	121.16 5.43				15.20 15.20				
-		N EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)			UNUOA	UNCCC		5.43	5.43			 	15.20			-	
	_ ****C 10D	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	22.09	94.21	45.09				15.20				
		First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2		2	UNCNX	U1L2X	35.28	94.21	45.09				15.20				
		First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3		3	UNCNX	U1L2X	65.18	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.2652										
		Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1 MQ1	70.47	143.58 59.97	103.88				15.20			-	
		Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNC1X UNCNX	UC1CA	105.09 2.96	5.91	12.96 4.26							-	
		Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone		1	ONCIVA	OCTOR	2.30	3.91	4.20							 	
		1		1	UNCNX	U1L2X	22.09	94.21	45.09				15.20				ı l
		Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone															
		2		2	UNCNX	U1L2X	35.28	94.21	45.09				15.20				
		Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone		3	UNCNX	U1L2X	65.18	94.21	45.09				15.20				ı l
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month		3	UNCNX	UC1CA	2.96	5.91	43.09				13.20				
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC	2.30	5.43	5.43				15.20				
	4-WIRE DS	1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPO	RT (EEL														
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1			UNC1X	USLXX	85.70	169.22	100.89				15.20				
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2			UNC1X	USLXX	194.96	169.22	100.89				15.20				
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X UNCSX	USLXX 1L5XX	491.94	169.22	100.89				15.20			ļ	
		Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	6.04 830.19	296.68	121.16				15.20				
		STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	201.48	107.05	48.07				10.20				
		DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	11.78	5.91	4.26								
		Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	85.70	169.22	100.89				15.20				
		Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	194.96	169.22	100.89				15.20				
		Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	491.94	169.22	100.89				15.20			-	
		DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X UNCSX	UC1D1 UNCCC	11.78	5.91 5.43	4.26 5.43				15.20				
	4-WIRE 56	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (E	FI)		UNCOA	UNCCC		5.45	5.45				13.20				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	30.99	94.21	45.09				15.20				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	36.78	94.21	45.09				15.20				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	38.92	94.21	45.09				15.20				\vdash
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.013	70.00	44				45.60				
-		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX UNCDX	U1TD5 UNCCC	15.61	72.60 5.43	41.75 5.43			-	15.20 15.20			 	\vdash
	4-WIRF 64	INONIECUTING CUITENTLY COMBINED NETWORK EIEMENTS SWITCH -AS-IS CHARGE KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (E	FI)		OINCDA	UNCCC		5.43	5.43				15.20			 	
<u> </u>		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1	,	1	UNCDX	UDL64	30.99	94.21	45.09			1	15.20			 	
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2			UNCDX	UDL64	36.78	94.21	45.09				15.20				
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL64	38.92	94.21	45.09				15.20				
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile			UNCDX	1L5XX	0.013										\Box
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility			UNCDX	U1TD6	15.61	72.60	41.75			1	15.20			<u> </u>	\vdash
ADDITION		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge K ELEMENTS	-		UNCDX	UNCCC	+	5.43	5.43			-	15.20				
ADDITION		NELEMENTS I as a part of a currently combined facility, the non-recurrng charges do not a	oply, but	a Swit	ch As Is charge d	loes appl	v.					—				 	
		as ordinarilty combined network elements in Georgia, the non-recurring charge															
	Access to I	DCS - Customer Reconfiguration (FlexServ)															
	Node (Sync	hroNet)															\Box
	Nonrecurrir	ng Currently Combined Network Elements "Switch As Is" Charge (One applies to 2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is"	to each o	combin	ation)												
		2/4-Wife VG Interoffice Channel used in a "COMBINATION - "Switch as is" Conversion Charge			UNCVX	UNCCC		5.43	5.43				15.20				
		56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is"	-		00 1/1	311000		0.40	0.40			+	15.20				
1	1	Conversion Charge			UNCDX	UNCCC		5.43	5.43				15.20				ı

LA

Version 4Q01:12/01/01

Description Characterised in a COMBINATION - Switch & bi* Conversion NaCLX UNCCC 5.43 5.45	e - Manual Manual Svc Svc Order Svc Order S vs. vs. vs. nic- Electronic- Electronic-E	ncremental Charge - Manual Svc Order vs.	Svc Order I Submitted Manually	Svc Order Submitte d Elec per LSR			RATES(\$)	ļ		usoc	BCS	terim Zo	CATEGORY RATE ELEMENTS
Piest Add Piest Add SOMC SOMA SOMA													
SS Fremitics Charmel used in a COMBINATION - Sintin As in Conversion Discussi	S RATES (\$)	OSS RA					urring		Rec				
Description Description	N SOMAN SOMAN	SOMAN	SOMAN	SOMEC	Add'l	First	Add'l	First					
Charge Description Descr													
Disarge													
Diss Internifice Charmel used in a COMBINATION - "Sentir As is" UMCSX UMCCC 5.43 5.43 15.20													DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion
Chargo			15.20				5.43	5.43		UNCCC	UNC1X		Charge
STS* Interoffice or Local Loop used in a COMBINATION ** Switch is bit Local Commission Million Bridge Part of State (1997)													DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion
STS* Interoffice or Local Loop used in a COMBINATION ** Switch is bit Local Commission Million Bridge Part of State (1997)			15.20				5.43	5.43		UNCCC	UNC3X		Charge
Convention Changer Checked Transport - minimum billing parted - Below DSS-com month, DSS and above-efforment - months													STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is"
NOTE Local Channel - Decicated Transport - Inhimum billing period - Belov DS3-one month, DS3 and above-four months			15.20				5.43	5.43		UNCCC	UNCSX		
											above=four month	h. DS3 ar	
Sichange Ports													
NOTE: Although the Port Rate Includes all available features in GA, KY, LA a TN, the desired features will need to be ordered using retail USOCs													
2-WINE VOICE GRADE LIMP FORT RATES (RES)									rotail USOCe	od ueina	ill need to be order	foaturos	
Exchange Posts - 2-Wire Avalage Line Port Res. LEPBR LEPR									retail 03003	eu using	iii need to be order	reatures	
Exchange Ports - 2-Wire Analog Line Fort with Caler ID - Res. LEPSR LEPSR LEPSR 2,231 2,21 15,20			15.20				2 21	2 24	1.50	HEDDI	LIEPSR	_	
Exchange Ports - 2-Wire Nabig Line Port outgoing only - Res. UEPSR	+												
Exchange Ports - 2-Wire VG urbunded LA estended Scal daing partly Port with Calter ID - Res				-									Exchange Ports - 2-Wire Analog Line Port with Caller ID - Kes.
Casler D. Res. UEPSR UEPSA 1,52 2,31 2,21 15,20	\longrightarrow		15.20				2.21	2.31	1.52	UEPRO	UEPSK		Exchange Ports - 2-wire Analog Line Port outgoing only - Res.
Exchange Ports - 2-Wire VG unbundled cousians Area Plus with Caller ID - Res (RIQL) Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID LEPSR LEPAG 1.52 2.31 2.21 15.20													
RRUL_	+		15.20				2.21	2.31	1.52	UEPAS	UEPSR		
Exchange Potts - 2-Wire VG urbanded res, low usage line port with Caller ID UEPSR UEPSR UESSG 0.00 0.00 0.00 0.00 15.20 15										l			
Subsequent Activity													
FEATURES			15.20				2.21	2.31	1.52	UEPAP	UEPSR		Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID
All Available Vertical Features UEPSR UEPVF 0.00 0.00 0.00 0.00 15.20							0.00	0.00	0.00	USASC	UEPSR		
Exchange Ports - 2-Wire No Exchange Ports - 2-Wire National Caller ID - Bus UEPSB UEPBL 1.52 2.31 2.21 15.20													FEATURES
Exchange Ports - 2-Wire VG unbunded Line Port without Caller ID - Bus UEPSB UEPSB UEPS 1.52 2.31 2.21 15.20			15.20				0.00	0.00	0.00	UEPVF	UEPSR		All Available Vertical Features
Exchange Ports - 2-Wire VG unbunded Line Port without Caller ID - Bus UEPSB UEPSB UEPS 1.52 2.31 2.21 15.20													2-WIRE VOICE GRADE LINE PORT RATES (BUS)
Exchange Ports - 2-Wire VG unbunded Line Port with unbunded port with UEPSB UEPBC 1.52 2.31 2.21 15.20			15.20				2.21	2.31	1.52	UEPBL	UEPSB		
Caleri-E484 ID - Bus UEPS UEPS UEPS UEPS 1.52 2.31 2.21 15.20													Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with
Exchange Ports - 2-Wire Vot bunded Law Assended local daing parity Port with Caller ID - Bus. UEPSB UEPAK 1.52 2.31 2.21 15.20			15 20				2 21	2 31	1.52	LIEPBC	LIEPSB		
Exchange Ports - 2-Wire VG urbunded Lexended Local dialng parity Port with Caller ID - Bus.													
Caller ID - Bus. Lehrange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus UEPSB UE			13.20				2.21	2.51	1.52	OLI DO	OLI SD		Exchange Ports - 2-Wire VG upbundled I A extended local dialing parity Port with
Exhange Ports - 2-Wire Vo inbunded concerning only port with Caller UEPSB UEPSB UEPAA 1.52 2.31 2.21 15.20			15 20				2.21	2 21	1 50	LIEDAY	LIEDED		Collect ID. Pro
Exchange Ports - 2-Wire Visc ubunded Louisiana Bus Area Caling Port with Caller UEPSB UEPAB UEPSB UEPAB UEPSB USASC 0.00 0.0													
ID- Bus (BUC)			15.20				2.21	2.31	1.52	UEPBI	UEPSB		
Subsequent Activity			45.00				2.24	0.04	4.50	LIEDAA	LIEDOD		
FEATURES All Available Vertical Features UEPSB UEPV 0.00 0.00 0.00 0.00 15.20			15.20										
All Available Vertical Features UEPSB UEPVE 0.00 0.00 0.00 0.00 15.20							0.00	0.00	0.00	USASC	UEPSB		
EXCHANGE PORT RATES (DID & PBX)													
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus UEPSP UEPPD 1.52 30.37 14.42 15.20			15.20				0.00	0.00	0.00	UEPVF	UEPSB		
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus													
2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus UEPSP UEPP0 1.52 30.37 14.42 15.20			15.20				14.42	30.37	1.52		UEPSE		2-Wire VG Unbundled 2-Way PBX Trunk - Res
2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			15.20				14.42	30.37	1.52	UEPPC	UEPSP		2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus
2-Wire Naiog Long Distance Terminal PBX Trunk - Bus			15.20				14.42	30.37	1.52	UEPPO	UEPSP		2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus
2-Wire Naiog Long Distance Terminal PBX Trunk - Bus										UEPP1	UEPSP		
2-Wire Voice Unbundled PBX LD Terminal Ports UEPSP UEPLD 1.52 30.37 14.42 15.20 15.20 2-Wire Voice Unbundled PBX LD Terminal Ports UEPSP UEPLD 1.52 30.37 14.42 15.20 15													
2-Wire Voice Unbundled PBX LD Terminal Ports UEPSP UEPLD 1.52 30.37 14.42 15.20													
2-Wire Voice Unbundled 2-Way PBX Lbage Port UEPSP UEPXA 1.52 30.37 14.42 15.20									1.52			-+	
2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports	- - - - - - - - - - 				l								
2-Wire Voice Unbundled PBX LD DDD Terminals Port	+												
2-Wire Voice Unbundled PBX LD Terminal Switchboard Port UEPSP UEPXD 1.52 30.37 14.42 15.20													
2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling Port 2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling Port UEPSP UEPXL 1.52 30.37 14.42 15.20 15.20 15.20 2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling Port UEPSP UEPXP 1.52 30.37 14.42 15.20 15.20 15.20 16.20 17.20 17.20 18.20 19.20					 								
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Callling Port UEPSP UEPXK 1.52 30.37 14.42 15.20			15.20				14.42	30.37	1.52	UEPXD	UEPSP		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Callling Port UEPSP UEPXK 1.52 30.37 14.42 15.20													
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Port UEPSP UEPXK 1.52 30.37 14.42 15.20				1									
2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Callling Port UEPSP UEPXK 1.52 30.37 14.42 15.20				1	1						1		
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPSP UEPXL 1.52 30.37 14.42 15.20			15.20				14.42	30.37	1.52	UEPXE	UEPSP		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port UEPSP UEPXL 1.52 30.37 14.42 15.20			15.20				14.42			UEPXK	UEPSP		
Port UEPSP UEPXL 1.52 30.37 14.42 15.20													
2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port UEPSP UEPXM 1.52 30.37 14.42 15.20			15.20	1			14.42	30.37	1.52	UEPXL	UEPSP		
2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room UEPSP UEPXO 1.52 30.37 14.42 15.20													2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port
Calling Port UEPSP UEPXO 1.52 30.37 14.42 15.20			.0.20		1			00.01			1		
2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling			15.20				14 42	30.37	1.52	LIEPXO	LIEPSP		
Port UEPSP UEPXP 1.52 30.37 14.42 15.20			.0.20				2	55.51	52	32. 7.0			2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling
2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port UEPSP UEPXS 1.52 30.37 14.42			15.20	1	1		14 42	30.37	1.52	LIEPXP	LIEPSP		
			10.20										
Subsequent Activity UEPSP USASC 0.00 0.00 0.00	+		 		 								
Subsequent Activity UEFSP USASC 0.00 0.00 0.00	+		l .		 		0.00	0.00	0.00	JOHOU	OLF OF		
			4	-						LIES: E	LIEDOD L'ESSE		
All Available Vertical Features UEPSP UEPSE UEPVF 0.00 0.00 0.00 15.20	\rightarrow		15.20				0.00	0.00	0.00	UEPVF	UEPSP UEPSE		
EXCHANGE PORT RATES (COIN)										1			
Exchange Ports - Coin Port 1.52 2.31 2.21 15.20 NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to circuit switched voice and/or circuit switched data transmission by B-Channels associated with 2-wire ISDN ports.													

TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		R.	ATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charge Manual Svc Orde vs. Electronic
						_		_		curring						
						Rec	Nonrecu		Disco					ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
NOTE: Ac	cess to B Channel or D Channel Packet capabilities will be available only throu	ah BFR/N	New Bu	siness Request Pr	rocess. F	Rates for the	packet capabilit	es will be de	termined vi	a the Bona	Fide Reque	st/New Bus	siness Reque	est Process.		
LED LOCAL E	XCHANGE SWITCHING(PORTS)															
	E PORT RATES (DID & PBX)															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.29	115.85	18.20				15.20				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	68.47	196.18	92.92				15.20				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	10.07	70.76	51.46				15.20				
NOTE: Tro	All Features Offered Insmission/usage charges associated with POTS circuit switched usage will also associated with POTS circuit switched with POTS circuit	o opphy	to oiro	UEPTX UEPSX	UEPVF	0.00	0.00	0.00	annels asse	oiotod with	2 wire ISDI	Inerte				
	cess to B Channel or D Channel Packet capabilities will be available only throu												inace Pague	et Process		
NOTE. AC	Exchange Ports - 2-Wire ISDN Port Channel Profiles	gii bi ivi	NEW DO		U1UMA	0.00	0.00	0.00	terriffica vi	a tile bolla	i ide Reque	SUIVEW DUS	iness iteque	St Flocess.		
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	94.82	197.92	98.62				15.20				
LED LOCAL S	WITCHING. PORT USAGE															
	Switching (Port Usage)	1	1													
	End Office Switching Function, Per MOU					0.001868										
	End Office Trunk Port - Shared, Per MOU					0.00018										
Tandem S	witching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001067										
	Tandem Trunk Port - Shared, Per MOU	-	ļ			0.000222					-					
Common 1	Common Transport - Per Mile, Per MOU					0.0000000										
	Common Transport - Per Mile, Per MOU Common Transport - Facilities Termination Per MOU					0.0000032										
ED PORT/I C	DOP COMBINATIONS - COST BASED RATES					0.0003746										
	d Rates are applied where BellSouth is required by FCC and/or State Commiss	ion rule t	o prov	ide Unbundled Loc	cal Switch	ning or Switch	Ports									
Features s	hall apply to the Unbundled Port/Loop Combination - Cost Based Rate section	in the sa	me ma	anner as they are a	applied to	the Stand-Al	one Unbundled	Port section	of this Rate	Exhibit.						
End Office	and Tandem Switching Usage and Common Transport Usage rates in the Port	section	of this	rate exhibit shall a	apply to a	II combinatio	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo	op Combir	nations.			
For Georgi	and Tandem Switching Usage and Common Transport Usage rates in the Portia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an	section	of this	rate exhibit shall a s listed apply to Cu	apply to a	II combination	ns of loop/port r	etwork elen	nents excep	t for UNE C	oin Port/Lo	op Combir	nations. ecurring char	rges apply to	Not Curren	itly
For Georgi	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an	section d Loop c	of this harges	rate exhibit shall a s listed apply to Cu	apply to a	II combination	ns of loop/port r	etwork elen	nents excep	t for UNE C	oin Port/Lo d additiona	op Combir I Port nonre	ations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO	and Tandem Switching Usage and Common Transport Usage rates in the Port ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an IDCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) .oop Combination Rates	section d Loop c	of this harges	rate exhibit shall a s listed apply to Cu	apply to a	II combination	ns of loop/port r	etwork elen	nents excep	t for UNE C	coin Port/Lo d additiona	op Combir I Port nonre	nations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) .oop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1	section d Loop c	harges	rate exhibit shall a s listed apply to Cu	apply to a	ombined and	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo	op Combir I Port nonre	nations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an IOCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) cop Combination Rates 2-Wire VG LoopPort Combo - Zone 1 2-Wire VG LoopPort Combo - Zone 2	section d Loop c	harges 1 2	rate exhibit shall a s listed apply to Cu	apply to a	ombined and 13.13 23.75	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo	op Combir I Port nonre	ations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an incce GRADE LOOP WITH 2-WIRE LINE PORT (RES) .oop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3	section d Loop c	harges	rate exhibit shall as listed apply to Cu	apply to a	ombined and	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo d additiona	op Combir I Port nonre	ations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates	section d Loop c	1 2 3	s listed apply to Cu	apply to a	13.13 23.75 49.62	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo d additiona	op Combir I Port nonre	ations. ecurring char	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates [2-Wire Voice Grade Loop (SL1) - Zone 1	d Loop c	1 2 3	s listed apply to Cu	apply to a	13.13 23.75 49.62	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo d additiona	op Combir I Port nonre	ations.	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an incide GRADE LOOP WITH 2-WIRE LINE PORT (RES) .oop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	section d Loop c	1 2 3 1 2 2 2	s listed apply to Cu UEPRX UEPRX	UEPLX UEPLX	13.13 23.75 49.62 11.77 22.39	ns of loop/port r	etwork elen	nents excep	t for UNE C	coin Port/Lod additiona	op Combir	ations.	ges apply to	Not Curren	tly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) cop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	section d Loop c	1 2 3 1 2 2 2	s listed apply to Cu	apply to a	13.13 23.75 49.62	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo	op Combir	ations.	ges apply to	Not Curren	tly
For Georgi 2-WIRE VO UNE Port/L UNE Loop	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX	13.13 23.75 49.62 11.77 22.39 48.26	ns of loop/port r Not Currently C	network elen ombined Co	nents excep	t for UNE C	Coin Port/Lo	l Port nonre	ations.	rges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) cop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2	section d Loop c	1 2 3 1 2 2 2	s listed apply to Cu UEPRX UEPRX	UEPLX UEPLX	13.13 23.75 49.62 11.77 22.39	ns of loop/port r	etwork elen	nents excep	t for UNE C	Coin Port/Lo	15.20	ations.	rges apply to	o Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L UNE Loop	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX	13.13 23.75 49.62 11.77 22.39 48.26	ns of loop/port in Not Currently C	network elemombined Co	nents excep	t for UNE C	Coin Port/Lo	15.20	ations.	ges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) cop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade Lone Grade Loop (SL1) - Zone 1	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36	ns of loop/port r Not Currently C	19.08 19.08 19.08	nents excep	t for UNE C	Coin Port/Lo	15.20 15.20	iations.	ges apply to	Not Curren	itly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled port outgoing only - res 2-Wire voice Grade unbundled Dort outgoing only - res 2-Wire voice Grade unbundled Dort outgoing only - res	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36	ns of loop/port r Not Currently C	19.08 19.08 19.08	nents excep	t for UNE C	Coin Port/Lo	15.20 15.20 15.20	iations.	rges apply to	Not Curren	etly
For Georgi 2-WIRE VO UNE Port/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) DOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port viith Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPC UEPRC UEPRO UEPAS UEPAS	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08	nents excep	t for UNE C	Coin Port/Lcd	15.20 15.20 15.20 15.20 15.20	iations.	rges apply to	Not Curren	etty
For Georgi 2-WIRE VO UNE Port/L UNE Loop 2-Wire Void	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Volice Grade Loop (SL1) - Zone 1 2-Wire Volice Grade Loop (SL1) - Zone 2 2-Wire Volice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire volice unbundled port - residence 2-Wire volice unbundled port with Caller ID - res 2-Wire volice Grade Line Port distribution of the Vire Volice unbundled port with Caller ID - res 2-Wire volice unbundled port outgoing only - res 2-Wire volice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire volice unbundled Audisiana Area Plus with Caller ID - res (RUL) 2-Wire volice unbundled Audisiana Area Plus with Caller ID - res (RUL)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPRO	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36	ns of loop/port r Not Currently C	19.08 19.08 19.08	nents excep	t for UNE C	Coin Port/Lod	15.20 15.20 15.20	ations.	rges apply to) Not Curren	ettly
For Georgi 2-WIRE VO UNE Port/L UNE Loop	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller IV res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAS UEPAG UEPAP	13.13 13.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	o Not Curren	ttly
For Georgic 2-WIRE VO UNE Port/L UNE Loop 2-Wire Void	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) cop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire WG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade Lone Middled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled sers, low usage line port with Caller ID (LUM)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPC UEPRC UEPRO UEPAS UEPAS	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/L d additiona	15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	o Not Curren	ttly
For Georgi 2-Wire VO UNE Port/L UNE Loop 2-Wire Voi	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade Lone Hondled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPRO UEPAG UEPAG UEPAF	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led	15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	ttly
FOR GEORGING UNE PORT/L UNE LOOP 2-Wire Voic FEATURES LOCAL NU	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire WG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop transience 2-Wire voice unbundled port residence 2-Wire voice unbundled port vift Caller ID - res 2-Wire voice Grade Unbundled port vift Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port)	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAS UEPAG UEPAP	13.13 13.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	o Not Curren	tly
FOR GEORGING UNE PORT/L UNE LOOP 2-Wire Voic FEATURES LOCAL NU	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCS) - CURRENTLY COMBINED	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPRO UEPAG UEPAG UEPAF	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	tty
For Georgia 2-WIRE VO UNE Port/L UNE Loop 2-Wire Void FEATURES LOCAL NU	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port oresidence 2-Wire voice unbundled port ortifolia port ID - res 2-Wire voice unbundled port ortifolia actended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRO UEPRO UEPRO UEPAS UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	tty
For Georgi 2-WIRE VO UNE POT/L UNE Loop 2-Wire Voi FEATURES LOCAL NU NONRECU	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates Rates 2-Wire Volice Grade Loop (SL1) - Zone 1 2-Wire Volice Grade Loop (SL1) - Zone 1 2-Wire Volice Grade Loop (SL1) - Zone 2 2-Wire Volice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire volice unbundled port - residence 2-Wire volice unbundled port vith Caller ID - res 2-Wire volice unbundled port outgoing only - res 2-Wire volice Grade Loup (SL1) - Zone 2 2-Wire volice unbundled port outgoing only - res 2-Wire volice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire volice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire volice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire volice unbundled Louisiana Pres Plus with Caller ID (LUM) 3 All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCS) - CURRENTLY COMBINED 2-Wire Volice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Volice Grade Loop / Line Port Combination - Conversion - Switch with change	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAS UEPAS UEPAS UEPAS UEPAS UEPAS	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	tly
For Georgic 2-WIRE VO UNE Port/L UNE Loop 2-Wire Void	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH Z-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice Grade Loop (SL1) - Zone 2 2-Wire voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Sers, low usage line port with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (RRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 13.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	atty
FOR GEORGING UNE POTUL UNE LOOP 2-Wire Void FEATURES LOCAL NU NONRECUI	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Volice Grade Loop (SL1) - Zone 1 2-Wire Volice Grade Loop (SL1) - Zone 1 2-Wire Volice Grade Loop (SL1) - Zone 3 2-Wire Volice Grade Loop (SL1) - Zone 3 2-Wire volice Grade Loop (SL1) - Zone 3 2-Wire volice unbundled port residence 2-Wire volice unbundled port with Caller ID - res 2-Wire volice unbundled port outgoing only - res 2-Wire volice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire volice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire volice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire volice unbundled Eduisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCS) - CURRENTLY COMBINED 2-Wire Volice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Volice Grade Loop / Line Port Combination - Conversion - Switch with change L NRCS 2-Wire Volice Grade Loop/Line Port Combination - Subsequent Activity	section d Loop c	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRO UEPRO UEPRO UEPAS UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 19.08	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	tty
For Georgic 2-WIRE VO UNE PORVI UNE Loop 2-Wire Voi FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates 2-Wire Vice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBEER PORTABILITY Local Number Portability (1 per port) RING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change AL NRCs 1-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity NCE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	section of	1 2 3 1 2 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 13.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	tty
For Georgic 2-WIRE VO UNE PORVI UNE PORVI UNE LOOP 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port or residence 2-Wire voice unbundled port or residence 2-Wire voice unbundled port or with Caller ID - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled se res, low usage line port with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	section of d Loop c	1 1 2 3 3 1 1 2 3 3 1 1 2 1 1 1 1 1 1 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 0.00	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	utty
Er Georgi 2-Wire Voi UNE POT/L UNE Loop 2-Wire Voi FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VOI 2-WIRE VOI	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCS) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	section of Loop c	1 1 2 3 3 1 2 2 3 3 1 1 2 1 1 1 1 1 1 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty
For Georgi 2-WIRE VOI UNE PORVI UNE Loop 2-Wire Voi FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH Z-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port viith Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade Loop (SL1) - Zone 2 2-Wire voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Sers, low usage line port with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (MRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change 1-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change 1-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity 1-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity 1-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity 1-Wire WG Loop Port Combo - Zone 1 2-Wire WG Loop Port Combo - Zone 2	section of d Loop c	1 1 2 3 3 1 1 2 3 3 1 1 1 2 1 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 2 1 1 1 2 1 1 1 2 1 1 1 2 1 1 1 2 1 1 1 1 2 1 1 1 1 2 1 1 1 1 1 2 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 0.00 0.35	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to	Not Curren	utty
For Georgi 2-WIRE VO UNE PORT/L UNE LOOP 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORT/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port or residence 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled Dort or utgoing only - res 2-Wire voice unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Fort or the Caller ID - res (RUL) 2-Wire voice unbundled Fort Orden Vire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change 1-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2	section of d Loop c	1 1 2 3 3 1 2 2 3 3 1 1 2 1 1 1 1 1 1 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAG UEPAP UEPAG UEPAP UEPAC UEACC	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		utty
For Georgi 2-WIRE VOI UNE PORVI UNE Loop 2-Wire Voi FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an DicE GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates 2-Wire Vice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RING CHARGES (NRCS) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity NCC GRADE LOOP WITH 2-WIRE LINE PORT (BUS) .oop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates	section of Loop c	1 2 3 3 1 2 2 3 3 3 1 2 2 3 3 3 1 2 2 3 3 3 1 2 2 3 3 3 1 2 2 3 3 3 1 2 3 3 3 1 2 3 3 3 1 2 3 3 3 3	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPAS UEPAG UEPAG UEPAC USAC2	13.13 13.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty
FOR Georgi 2-WIRE VO UNE PORT/L UNE LOOP 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORT/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH Z-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Orea Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Orea Plus with Caller ID - res All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with Change 1-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	section of d Loop c	1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 2 3 3 1 1 1 2 2 3 3 1 1 1 2 2 3 3 1 1 1 1	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRO UEPAS UEPAS UEPAS UEPAP UEPAS UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAS	13.13 23.75 49.62 11.77 12.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36 1.3	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		utty
FOR Georgi 2-WIRE VOI UNE PORT/L UNE LOOP 2-Wire Voi FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORT/L	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH 2-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice Grade Loop (SL1) - Zone 3 ce Grade Line Port Rates (Res) 2-Wire voice unbundled port vitin Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice Grade unbundled Louisiana extended local dialing parity port with Caller ID - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Sers, low usage line port with Caller ID (LUM) All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCS) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity IOCE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) cop Combination Rates 2-Wire VG Loop/Port Combo - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2	section of d Loop c	1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 3	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPAS UEPAG UEPAG UEPAC USAC2	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36 1.3	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty
FOR Georgia 2-WIRE VO UNE PORVL UNE LOOP 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORVL UNE LOOP	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH Z-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire WG Loop/Port Combo - Zone 1 2-Wire WG Loop/Port Combo - Zone 2 2-Wire WG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port - residence 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Orea Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Orea Plus with Caller ID - res All Features Offered MBER PORTABILITY Local Number Portability (1 per port) RRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with Change 1-Wire Voice Grade Loop / Line Port Combination - Subsequent Activity ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	section of Loop c	1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 3	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPC UEPRC UEPAG UEPAG UEPAG UEPAC UEPLX UEPLX	13.13 23.75 49.62 11.77 12.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36 1.3	38.85 38.85 38.85 38.85 38.85 0.00	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty
FOR Georgia 2-WIRE VO UNE PORT/L UNE Loop 2-Wire Void 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORT/L UNE Loop	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an Dice GRADE LOOP WITH 2-WIRE LINE PORT (RES) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates Rates Rates	section of d Loop c	1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 3	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPLX UEPRC UEPRC UEPRG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAG UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC UEPAC USAC2	13.13 13.75 49.62 11.77 12.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36 1.3	38.85 38.85 38.85 38.85 38.85 38.85 38.85 38.85 38.85	19.08 19.08 19.08 19.08 19.08 19.08 0.00 0.10 0.10	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty
FOR Georgia 2-WIRE VO UNE PORT/L UNE Loop 2-Wire Void 2-Wire Void FEATURES LOCAL NU NONRECUI ADDITIONA 2-WIRE VO UNE PORT/L UNE Loop	ia, Kentucky, Louisiana, Mississippi and Tennessee, the recurring UNE Port an INCE GRADE LOOP WITH Z-WIRE LINE PORT (RES) LOOP Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port or residence 2-Wire voice unbundled port or residence 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled port or utgoing only - res 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL) 2-Wire voice unbundled Port or Vire Voice Unbundled Coulsiana Or Vire Voice Unbundled Coulsiana Or Vire Voice Unbundled Sers, low usage line port with Caller ID - res (RUL) 2-Wire voice unbundled Fort Or Vire Voice Voice Vire Voice Vire Voice Vire Voice Vire Voice Voice Vire Voice Voi	section of Loop of	1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 1 1 2 2 3 3 3 3	UEPRX UEPRX	UEPLX UEPLX UEPLX UEPLX UEPCX USAC2 USAC2 USAC2 USAC2 USAC2	13.13 23.75 49.62 11.77 22.39 48.26 1.36 1.36 1.36 1.36 1.36 1.36 1.36 1.3	38.85 38.85 38.85 38.85 38.85 0.00 0.10 0.10	19.08 19.08 19.08 19.08 19.08 19.08 19.08 0.00	nents excep	t for UNE C	Oin Port/Led additiona	15.20 15.20 15.20 15.20 15.20 15.20 15.20	ations.	rges apply to		tty

TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge Manual Svc Order vs. Electronic Disc 1st	-al Charg Manua Svc Ord vs. -Electror
										curring						
						Rec	Nonrec			onnect		T		ATES (\$)	T	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	2-Wire voice Grade unbundled Louisiana extended local dialing parity port with															
	Caller ID - bus			UEPBX	UEPAX	1.36	38.85	19.08				15.20				
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.36	38.85	19.08				15.20				
	2-Wire voice unbuilded incoming only port with Caller ID - Bus 2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BUC)			UEPBX	UEPAA	1.36	38.85	19.08				15.20				
	MBER PORTABILITY			OLI DX	OLI 701	1.00	00.00	13.00				10.20				
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATURES	Educativation Fortability (1 per port)			OLI DX	LIVIOX	0.00										
	All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00				15.20				
	RRING CHARGES (NRCs) - CURRENTLY COMBINED			OLI DX	OLI VI	0.00	0.00	0.00				10.20				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		0.10	0.10				15.20				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			OL: DX	00/102		0.10	0.10				10.20				
	change			UEPBX	USACC		0.10	0.10								
ADDITIONA							55	20					İ	İ		
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2								31.92	7.32		
2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)												202	02		
	oop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			13.13										
	2-Wire VG Loop/Port Combo - Zone 2		2			23.75										
	2-Wire VG Loop/Port Combo - Zone 3		3			49.62										
UNE Loop F	Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEPRG	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEPRG	UEPLX	48.26										
	e Grade Line Port Rates (RES - PBX)															
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.36	66.91	31.29				15.20				
	MBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								
FEATURES	, , , , ,															
	All Features Offered			UEPRG	UEPVF	0.00	0.00	0.00				15.20				
NONRECUR	RRING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-			UEPRG	USAC2		7.68	1.85				15.20				
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with															
	Change			UEPRG	USACC		7.68	1.85					31.92	7.32		
ADDITIONA																
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00					31.92	7.32		
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.11	7.11					19.99	19.99	19.99	19
	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	oop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			13.13										
	2-Wire VG Loop/Port Combo - Zone 2		2			23.75										
	2-Wire VG Loop/Port Combo - Zone 3		3			49.62										
UNE Loop F																
	2-Wire Voice Grade Loop (SL 1) - Zone 1			UEPPX	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEPPX	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	48.26										
	e Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.36	66.91	31.29				15.20				
	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.36	66.91	31.29				15.20				
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling Port			UEPPX	UEPL2	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.36	66.91	31.29				15.20				ļ
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.36	66.91	31.29				15.20			1	1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.36	66.91	31.29			1	15.20			1	1
1	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	1.36	66.91	31.29				15.20				-
	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Port			UEPPX	UEPXK	1.36	66.91	31.29				15.20				
1	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling			LIEDDY	LIEDY	4.00	00.01	04.00				45.00				
1	Port 2 Wire Vaice Unbundled 2 Way PRV Hetal/Hassital Economy Room Calling Port			UEPPX UEPPX	UEPXL	1.36 1.36	66.91 66.91	31.29 31.29			1	15.20 15.20			1	1
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPA	UEPAM	1.36	16.90	31.29			1	15.20				1-
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room Calling Port			UEPPX	UEPXO	1.36	66.91	31.29				15.20				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling			OLFFA	UEPAU	1.30	16.00	31.29				15.20				1
	Port			UEPPX	UEPXP	1.36	66.91	31.29				15.20				
						1.50	00.31		1		1	10.20	1		1	1
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	1.36	66.91	31.29				15.20	31.92	7.32		

CATEG	ORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.		-al Charge Manual
							B				curring			000 0	ATEO (A)		
							Rec	Nonrec First	urring Add'l	First	onnect Add'l	SOMEC	SOMAN	COMAN	ATES (\$) SOMAN	SOMAN	SOMAN
								FIRST	Addi	FIRST	Addi	SUMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
																+	-
		Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00							1	
F	EATURES																
		All Features Offered			UEPPX	UEPVF	0.00	0.00	0.00				15.20				
N		RRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-			UEPPX	USAC2		7.68	1.85				15.20				
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPPX	USACC		7.68	1.85					31.92	7.32		
Δ	ADDITIONA				OLITA	OOACC		7.00	1.03					31.32	7.52	+	
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00					31.92	7.32	+	
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.11	7.11					19.99	19.99	19.99	19.99
2	-WIRE VO	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															
U		oop Combination Rates							-								
		2-Wire VG Coin Port/Loop Combo – Zone 1		1			13.13										
		2-Wire VG Coin Port/Loop Combo – Zone 2		2			23.75					1					
	JNE Loop F	2-Wire VG Coin Port/Loop Combo – Zone 3		3			49.62					1				+	
U		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	11.77									+	
		2-Wire Voice Grade Loop (SL1) - Zone 2			UEPCO	UEPLX	22.39									1	
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	48.26										
2	-Wire Voic	ce Grade Line Ports (COIN)															
		2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,															
		MS)			UEPCO	UEPRF	1.36	38.85	19.08				15.20				ļ
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD			UEPCO	UEPRA	4.00	20.05	19.08				15.20				
		(AL, KY, LA, MS) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)			UEPCO	UEPRB	1.36 1.36	38.85 38.85	19.08				15.20				-
		2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, &			OLI OO	OLITE	1.00	50.05	13.00				10.20			+	
		Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.36	38.85	19.08				15.20				
		2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,			UEPCO	UEPRN	1.36	38.85	19.08				15.20				
		2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)			UEPCO	UEPLA	1.36	38.85	19.08				15.20				
		2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD			LIEBOO	HEDDII	4.00	00.05	40.00				45.00				
		(AL, KY, LA, MS) 2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and			UEPCO	UEPRH	1.36	38.85	19.08				15.20			-	-
		Local (AL, KY, LA, MS)			UEPCO	UEPCN	1.36	38.85	19.08				15.20				
		2-Wire Coin 2-Way Smartline with 900/976 (Louisiana only)			UEPCO	UEPNA	1.36	38.85	19.08				15.20			1	
		2-Wire Coin Outward Smartline with 900/976 (Louisiana only)			UEPCO	UEPCB	1.36	38.85	19.08				15.20				
Α		L UNE COIN PORT/LOOP (RC)															
		UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	1.81	0.00	0.00								
L		MBER PORTABILITY			LIEBOO	LLIBOX											ļ
-		Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	EATURES	RRING CHARGES - CURRENTLY COMBINED															-
		2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPCO	USAC2		0.10	0.10				15.20			+	
		2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			02. 00	00/102		0.10	0.10				10.20			+	
		change			UEPCO	USACC		0.10	0.10					31.92	7.32		
А	ADDITIONA																
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00					31.92	7.32		
		OP COMBINATIONS - COST BASED RATES															
		ICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT oop Combination Rates		-												+	
U		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			23.20					1				 	
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			33.62									t	
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			58.73										
U	JNE Loop F																
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	14.93						15.20			4	
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2			UEPPX	UECD1	25.35					-	15.20				
	JNE Port R	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	50.46					1	15.20			+	
U		Exchange Ports - 2-Wire DID Port			UEPPX	UEPD1	8.27	217.95	83.92				15.20			 	
N		RRING CHARGES - CURRENTLY COMBINED				02101	0.27	217.55	00.02				10.20				
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is			UEPPX	USAC1		7.10	1.81				15.20				
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth															
		Allowable Changes			UEPPX	USA1C		7.10	1.81				15.20				<u> </u>
A	DDITIONA				HEDDY	11040:		00.51	00 01				45.00				
_	alanhara	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges			UEPPX	USAS1		26.01	26.01				15.20			-	-
		HUITIDET I TUTIK GLOUP ESTADIISHIEH GHALYES	1	i .		1 1				1	1	1	1	1	1	1	1

ATEGORY	RATE ELEMENTS	Interim	Zone	В	cs usoc		ļ	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	
									Nonre	curring						
						Rec	Nonrec	urring		onnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
,	Additional DID Numbers for each Group of 20 DID Numbers			UEPPX	ND4	0.00	0.00	0.00				15.20				
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPPX	ND5	0.00	0.00	0.00				15.20				
	Reserve Non-Consecutive DID numbers			UEPPX	ND6	0.00	0.00	0.00				15.20				
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.20				
LOCAL NUM	MBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00								
	N DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT															
UNE Port/Lo	pop Combination Rates															
				UEPPB												
1	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPR		27.48										
				UEPPB												
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2			UEPPR		40.34										
				UEPPB												
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPR		70.99										
UNE Loop R	Rates															
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR USL2X	19.09						15.20				
				UEPPB	[
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPR	USL2X	31.95						15.20				
					[
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR USL2X	62.60						15.20				
UNE Port Ra	ate															
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR UEPPB	8.39	184.10	128.42				15.20				
	RING CHARGES - CURRENTLY COMBINED															
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -															
	Conversion			UEPPB	UEPPR USACB	0.00	37.40	26.23				15.20				
ADDITIONAL																
LOCAL NUM	IBER PORTABILITY															
_	Local Number Portability (1 per port)			UEPPB	UEPPR LNPCX	0.35	0.00	0.00								
B-CHANNEL	. USER PROFILE ACCESS:															
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR U1UCC	0.00	0.00	0.00								
B-CHANNEL	. AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)															
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR U1UCD	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR U1UCE	0.00	0.00	0.00								
	000															
	CSD			UEPPB	UEPPR U1UCF	0.00	0.00	0.00								
USER TERM	MINAL PROFILE										1					
	User Terminal Profile (EWSD only)			UEPPB	UEPPR U1UMA	0.00	0.00	0.00								
VERTICAL F	-EATURES															
	All \/astiset Factores One and Observed Billion Box (its			LIEDOD	HEDDE HER T	0.00	0.00	0.00				45.00				
	All Vertical Features - One per Channel B User Profile			UEPPB	UEPPR UEPVF	0.00	0.00	0.00				15.20				
INTEROFFIC	CE CHANNEL MILEAGE															
	Interoffice Channel mileage each, including first mile and facilities termination				UEPPR M1GNC	22.613	39.36	26.62				15.20				
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR M1GNN	0.013	0.00	0.00				15.20				
	DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															
	pop Combination Rates		_	LIEDOS		400 00										
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP		180.52								-		
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2			UEPPP		289.78								-		
	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP		586.76								-		
UNE Loop R	Kates			LIEDOS								/=		-		
	4-Wire DS1 Digital Loop - UNE Zone 1			UEPPP	USL4P	85.70						15.20				
4	4-Wire DS1 Digital Loop - UNE Zone 2			UEPPP	USL4P	194.96						15.20				
	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	491.94					1	15.20				
UNE Port Ra																
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	94.82	443.08	251.60				15.20				
NONRECUR	RING CHARGES - CURRENTLY COMBINED															
4	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -															
	Conversion -Switch-as-is			UEPPP	USACP	0.00	115.63	76.29				15.20				
	LNRCs															
ADDITIONAL																
4	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos within Std Allowance			UEPPP	PR7TF		0.48					15.20				

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC	,	ı	RATES(\$)				Submitted Manually		Manual	al Charge Manual Svc Order vs.	al Char Manu Svc Or vs. Electro
									Nonrecu	ırrina						
						Rec	Nonrec	urring	Discon				OSS R	ATES (\$)		
					+ +	1100	First	Add'l	First	Add'l	COMEC	SOMAN	COMAN	SOMAN	SOMAN	COM
							FIRST	Addi	FIFST	Add I	SUMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUIVIA
4-Wire [DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All															
States e	except NC)			UEPPP	PR7TO		11.18	11.18				15.20				
4-Wire I	DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos			OL: III	11010		111.10	11110				10.20				
				UEPPP	PR7ZT		00.05	00.05				45.00				
	Std Allowance			UEPPP	PR/ZI		22.35	22.35				15.20				
LOCAL NUMBER PO																
Local Nu	lumber Portability (1 per port)			UEPPP	LNPCN	1.75										
INTERFACE (Provs	sioning Only)															
Voice/Da				UEPPP	PR71V	0.00	0.00	0.00								
Digital D				UEPPP	PR71D	0.00	0.00	0.00			ļ					
Inward D		1		UEPPP	PR71E	0.00	0.00	0.00			1			1	1	1
New or Additional "		1									1	1				1
	Additional - Voice/Data B Channel	t e		UEPPP	PR7BV	0.00	14.11					15.20			1	
		 	\vdash	UEPPP			14.11	.	+		1			-	-	1
	Additional - Digital Data B Channel		_		PR7BF	0.00						15.20			ļ	
	Additional Inward Data B Channel	<u></u>		UEPPP	PR7BD	0.00	14.11					15.20				
New or A	Additional Useage Sensitive Voice Data B Channel	1		UEPPP	PR7BS	0.00	14.11					15.20				
	Additional Useage Sensitive Digital Data B Channel	1		UEPPP	PR7BU	0.00	14.11				1	15.20				
CALL TYPES	230ago Conomito Digital Data D Ollamoi	+			50	0.00	17.11					10.20			 	1
		-	-	LIEDDD	DD=0:						-			-	-	-
Inward				UEPPP	PR7C1	0.00	0.00	0.00								1
Outward	d	1		UEPPP	PR7C0	0.00	0.00	0.00			1			1	1	1
Two-way	ay			UEPPP	PR7CC	0.00	0.00	0.00								
Interoffice Channel																
				UEPPP	1LN1A	70.7532	86.69	79.44				15.20				-
	ach Including First Mile						86.69	79.44				15.20				
	irline-Fractional Additional Mile			UEPPP	1LN1B	0.2652										
4-WIRE DS1 DIGITA	AL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UNE Port/Loop Con	mhination Rates															
AW DOL	1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		-1	UEPDC	+	154.17						15.20				
4W D31	1 Digital Loop/4W DDITS Trulk Folt - ONE Zone 1				-											
	1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2			UEPDC		263.43						15.20				
4W DS1	1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		560.41						15.20				
UNE Loop Rates																
	DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	85.70						15.20				
4-Wile L	DOT Digital Loop - ONE Zone 1			UEPDC	USLDC	194.96						15.20				_
4-Wire L	DS1 Digital Loop - UNE Zone 2															
	DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	491.94						15.20				
UNE Port Rate																
	DDITS Digital Trunk Port			UEPDC	UDD1T	68.47	441.34	245.90				15.20				
NONDECLIDRING C	HARGES - CURRENTLY COMBINED			OLI DO	CDD11	00.47	771.07	240.00				10.20				-
	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is			UEPDC	USAC4		125.75	65.08				15.20				
	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with															
DS1 Cha		1		UEPDC	USAWA		125.75	65.08			1	15.20		1	1	1
4-Wire I	DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with			· ·	1						l					
	- Trunk	1		UEPDC	USAWB		125.75	65.08				15.20		1	1	1
change	:- ITUIK	-		UEFDC	OSAWB		125.75	80.08			1	15.20		-	-	-
ADDITIONAL NRCs																1
4-Wire [DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel	1									1			1	1	1
Activation	on/Chan - 2-Way Trunk			UEPDC	UDTTA		14.06	14.06				15.20			1	1
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel															
	on/Chan - 1-Way Outward Trunk	1		UEPDC	UDTTB		14.06	14.06				15.20		1	1	1
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan		1	JE1 DO	30110		14.00	14.00	+		1	10.20			1	1
4-vvire L	DO I LOOP / 4-WIRE DOITO THURK FOR - SUBSQUE CHARRIER ACTIVATION/CRAN			LIEDDO	LIDTTO		4400	4400				45.00			1	1
inward I	Trunk w/out DID			UEPDC	UDTTC		14.06	14.06				15.20			ļ	1
	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -	1												1	1	1
	Trunk with DID	1		UEPDC	UDTTD		14.06	14.06			1	15.20		1	1	1
4-Wire D	DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-															
	D w User Trans	1		UEPDC	UDTTE		14.06	14.06				15.20		1	1	1
BIPOLAR 8 ZERO S		 	1		35 / IL		17.00	17.00	-		l	10.20				t
		-			000			***			1			-	-	-
B8ZS -S	Superframe Format			UEPDC	CCOSF		0.00	605.00				15.20				
	Extended Superframe Format		Щ Т	UEPDC	CCOEF		0.00	605.00			<u> </u>	15.20				
Alternate Mark Inve																
	perframe Format	t e		UEPDC	MCOSF		0.00	0.00							1	1
	xtended SuperFrame Format	 	1	UEPDC	MCOPO		0.00	0.00	-		l					t
		-		UEFDC	IVICUPU		0.00	0.00			1			-	-	-
	r/Trunk Group Establisment Charges		<u> </u>									1				
	one Number for 2-Way Trunk Group	1		UEPDC	UDTGX	0.00					1	15.20				
	one Number for 1-Way Outward Trunk Group	l		UEPDC	UDTGY	0.00					1	15.20		1	1	1
Tallet	Aliantes for A Was laward Trust Coop Without DID	 	1	UEPDC				<u> </u>			 			 	+	+
	one Number for 1-Way Inward Trunk Group Without DID	1			UDTGZ	0.00						15.20				1
	mbers for each Group of 20 DID Numbers		<u> </u>	UEPDC	ND4	0.00						15.20				Ш.
DID Nun	mbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00					1	15.20				
	e Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00			l	15.20				
	e DID Numbers	 		UEPDC	NDV	0.00	0.00	0.00			 	15.20		 	+	+
D															1	

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ŀ	RATES(\$)			Svc Order Submitte d Elec per LSR	Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	-al Char Manu Svc Or vs. -Electro
										curring				•		
						Rec	Nonrec			nnect	001150	001111		ATES (\$)	0011411	2014
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
																+
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	70.47	86.69	79.44				15.20			+	+
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			UEPDC	1LNOA	0.2652	0.00	0.00				10.20				_
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.2652	0.00	0.00								1
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.2652	0.00	0.00								
	Local Number Portability, per DS0 Activated			UEPDC	LNPCP	3.15	0.00	0.00	0.00							
	Central Office Termininating Point			UEPDC	CTG	0.00										
	1 LOOP WITH CHANNELIZATION WITH PORT															+
	1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations	f norto uc	204												-	+
UNE DS1 L	em can have up to 24 combinations of rates depending on type and number o	i poits de	ocu									1			+	+
ONE DOT L	4-Wire DS1 Loop - UNE Zone 1	+	1	UEPMG	USLDC	85.70	0.00	0.00			 	15.20				+
	4-Wire DS1 Loop - UNE Zone 2			UEPMG	USLDC	194.96	0.00	0.00				15.20				1
	4-Wire DS1 Loop - UNE Zone 3	1		UEPMG	USLDC	491.94	0.00	0.00			†	15.20			<u> </u>	†
	Channelization Capacities (D4 Channel Bank Configurations)															
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	97.35	0.00	0.00				15.20				
	48 DSO Channel Capacity - 1 per 2 DS1s			UEPMG	VUM48	194.70	0.00	0.00				15.20				
	96 DSO Channel Capacity -1per 4 DS1s			UEPMG	VUM96	389.40	0.00	0.00				15.20				
	144 DS0 Channel Capacity - 1 per 6 DS1s			UEPMG	VUM14	584.10	0.00	0.00				15.20				
	192 DS0 Channel Capacity -1 per 8 DS1s			UEPMG	VUM19	778.80	0.00	0.00				15.20				
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	973.50	0.00	0.00				15.20				
	288 DS0 Channel Capacity - 1 per 12 DS1s			UEPMG UEPMG	VUM28 VUM38	1,168.20 1,557.60	0.00	0.00				15.20 15.20				+
	384 DS0 Channel Capacity - 1 per 16 DS1s 480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,947.00	0.00	0.00				15.20				+
	576 DS0 Channel Capacity - 1 per 24 DS1s			UEPMG	VUM57	2,336.40	0.00	0.00				15.20				+
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,725.80	0.00	0.00				15.20			+	+-
	ring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with I	Port - Cor					0.00								+	1
	System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24															
Multiples of	f this configuration functioning as one are considered Add'l after the minimun	n system	config	uration is coun	ted.											
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes			UEPMG	USAC4	0.00	146.13	8.12				15.20				
System Ad	ditions at End User Locations Where 4-Wire DS1 Loop with Channelization wit	h Port Co	mbina	tion Currently E	xists and											
New (Not C	urrently Combined) In GA, KY, LA, MS & TN Only															
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - New GA, LA, KY, MS, &TN Only			UEPMG	VUMD4	0.00	715.54	467.54				15.20				
	ero Substitution			OLI WG	VOIVID4	0.00	713.34	407.54				13.20			+	+
Dipolal 0 2	Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0.00	0.00	605.00				15.20			+	+
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only	,		UEPMG	CCOEF	0.00	0.00	605.00				15.20				+
Alternate M	Mark Inversion (AMI)															
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								
	Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	Ports Associated with 4-Wire DS1 Loop with Channelization with Port							-								
Exchange I				LIEBBY /	r age = -							ļ				1
_	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.52	0.00	0.00	0.00	0.00		15.20				+
-	Line Side Outward Channelized PBX Trunk Port - Business	1		UEPPX	UEPOX	1.52	0.00	0.00	0.00	0.00		15.20				+
	Line Side Inward Only Channelized PBX Trunk Port without DID 2-Wire Trunk Side Unbundled Channelized DID Trunk Port	1		UEPPX UEPPX	UEP1X UEPDM	1.52 8.29	0.00	0.00	0.00	0.00		15.20 15.20	1			+
	tivations - Unbundled Loop Concentration	 		OLITA	UEFDINI	0.29	0.00	0.00	0.00	0.00	 	15.20				+
i cature At	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank	_		UEPPX	1PQWM	0.6497	25.36	13.40			 	15.20			+	+-
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.6497	78.05	18.40				15.20				†
Telephone	Number/ Group Establishment Charges for DID Service	1														1
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00				15.20				
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				15.20				
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0.00	0.00	0.00				15.20				4
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00			1	15.20				
	Reserve DID Numbers	-		UEPPX	NDV	0.00	0.00	0.00			-	15.20				+
	ber Portability	1		LIEDDY	LNDCD	2.15	0.00	0.00			1	1				+
	Local Number Portability - 1 per port - Vertical and Optional	1		UEPPX	LNPCP	3.15	0.00	0.00			-		1		-	+
	bing Features Offered with Line Side Ports Only	1									1		1			+
LUCAI OWITC	All Features Available	+		UEPPX	UEPVF	0.00	0.00	0.00			 	15.20				+-
	y an a contained a wallable	+		OLI I A	OLF VI	0.00	0.00	0.00				13.20			 	+
	OP COMBINATIONS - MARKET RATES											1			+	+
DLED PORT LO	OP COMBINATIONS - MARKET RATES es shall apply where BellSouth is not required to provide unbundled local swit	ching or	switch	ports per FCC	and/or State	Commission	rules.							l		1
DLED PORT LO Market Rate	IOP COMBINATIONS - MARKET RATES es shall apply where BellSouth is not required to provide unbundled local swit narios include:	ching or	switch	ports per FCC	and/or State	Commission	rules.									+

regory	RATE ELEMENTS	Interim	Zone	BCS	USOC		ı	RATES(\$)	Nonro	ourring.	Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge Manual Svc Order vs. Electronic Disc 1st	-al Char Manu Svc O vs. -Electro
						_				curring						
						Rec	Nonrec			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
2. Unbundle	ed port/loop combinations that are Currently Combined or Not Currently Con	bined in 2	Zone 1	of the Top 8 M	ISAS in Bells	outh's region	for end users	with 4 or mor	e DS0 equiv	valent lines.						
The Top 8 N	MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atla	nta): LA (I	New Or	leans): NC (Gre	eensboro-W	inston Salem-	Highpoint/Cha	rlotte-Gastoni	a-Rock Hill)	: TN (Nashy	ille).					
BellSouth c	urrently is developing the billing capability to mechanically bill the recurring	and non-re	ocurrin	n Market Rate	e in this sact	ion except fo	r nonrecurring	charges for n	ot currently	combined	in ΔI FI I	NC and SC	In the interio	m where Bel	South cann	not hill
	Rate for unbundled ports includes all available features in all states.	1101111	- Courtin	g market rate.	3 111 11113 300	ion except to	nomecuming	charges for i	lot currently	Combined		dia oo.	III the intern	WHICH DO	locutii cuiii	TOT DIII
Fred Office	and Tandem Switching Usage and Common Transport Usage rates in the Po		-6 4h:-		all annhita					4 4 I INIT C	ain Dantil .			barra a flat		
End Onice a	and randem switching osage and Common transport osage rates in the Po	r section	01 11115	the Electrical St	all apply to	an combinatio	iis or loop/port	network elem	ients excep	i ioi one c	OIII FOIUL	oop combi	iations willer	l'idve a ilat	NDO O	
	rrently Combined scenarios where Market Rates apply, the Nonrecurring chair	ges are il	stea in	the First and A	Additional Ni	C columns to	or each Port US	OC. For Curi	rently Comp	inea scena	rios, the No	onrecurring	cnarges are	listed in the	NRC - Curr	entiy
	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
	pop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
1	2-Wire VG Loop/Port Combo - Zone 3	1	3			62.26						İ	1		İ	
UNE Loop R			T -			52.20										
	2-Wire Voice Grade Loop (SL1) - Zone 1	-	1	UEPRX	UEPLX	11.77				l		1	l		1	+-
					UEPLX				1			1	-	1	+	+
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPRX		22.39			ļ	l		!			1	+
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	48.26						ļ			ļ	1
	e Grade Line Port (Res)														1	1
	2-Wire voice unbundled port - residence		<u></u>	UEPRX	UEPRL	14.00	90.00	90.00	L			<u> </u>	31.92	7.32	<u> </u>	
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	14.00	90.00	90.00		l			31.92	7.32	1	
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	14.00	90.00	90.00					31.92	7.32		
	2-Wire voice Grade unbundled Louisiana extended local dialing parity port with															
	Caller ID - res			UEPRX	UEPAS	14.00	90.00	90.00					31.92	7.32		
	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (RUL)			UEPRX	UEPAG	14.00	90.00	90.00					31.92	7.32		
	2-Wire voice unbundled Louisiana Area Plus with Caller ID - res (AC7)			UEPRX	UEPAH	14.00	90.00	90.00					31.92	7.32		-
		_														+-
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)	_		UEPRX	UEPAP	14.00	90.00	90.00					31.92	7.32		
	MBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
FEATURES																
	All Features Offered			UEPRX	UEPVF	0.00	0.00	0.00								
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is			UEPRX	USAC2		41.50	41.50					31.92	7.32		
	2-Wire Voice Grade Loop / Line Port Combination - Switch with change			UEPRX	USACC		41.50	41.50								
ADDITIONAL																
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPRX	USAS2		0.00	0.00					31.92	7.32		-
	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	+		OLFIX	00/102		0.00	0.00					31.32	1.52		
		_														+
UNE PORT/LC	pop Combination Rates		-			05.77										
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
	2-Wire VG Loop/Port Combo - Zone 3		3			62.26										1
UNE Loop R	Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	11.77						1			1	
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPBX	UEPLX	22.39						1			1	
	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPBX	UEPLX	48.26						l			1	1
2-Wire Voice	e Grade Line Port (Bus)	+			JEI EX	70.20						†			1	+
		+	1	UEPBX	UEPBL	14.00	90.00	90.00	 	 		1	31.92	7.32	1	+
	2-Wire voice unbundled port without Caller ID - bus		-						1			1			1	+
	2-Wire voice unbundled port with Caller + E484 ID - bus	+	1	UEPBX	UEPBC	14.00	90.00	90.00				1	31.92	7.32	1	+
1	2-Wire voice unbundled port outgoing only - bus	1	L	UEPBX	UEPBO	14.00	90.00	90.00	1			1	31.92	7.32	ļ	1
	2-Wire voice Grade unbundled Louisiana extended local dialing parity port with		1													
	Caller ID - bus			UEPBX	UEPAX	14.00	90.00	90.00					31.92	7.32		
	2-Wire voice unbundled Louisiana Bus Area Calling Port with Caller ID (BUC)			UEPBX	UEPAA	14.00							31.92	7.32		ш¯
LOCAL NUM	MBER PORTABILITY									l					1	
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATURES	* * * * * * * * * * * * * * * * * * * *											1			1	
	RRING CHARGES - CURRENTLY COMBINED											l			1	1
	2-Wire Voice Grade Loop / Line Port Combination - Switch-as-is	+	t	UEPBX	USAC2		41.50	41.50				1	31.92	7.32	1	1 -
	2-Wire Voice Grade Loop / Line Port Combination - Switch with change		 	UEPBX	USACC		41.50	41.50	1			 	51.52	1.52	 	+-
		+	1	OLFDA	USACC	 	41.00	41.00	 	 		1	1	1	1	+
ADDITIONAL		+	1	LIEDDY	110400		0.00	0.00				1	04.00	7.00	1	+
	NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent		-	UEPBX	USAS2		0.00	0.00	ļ	 		1	31.92	7.32	1	1
2-WIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)														1	1
UNE Port/Lo	oop Combination Rates	1	<u></u>			<u></u>			<u> </u>							
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77						1			1	
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
	2-Wire VG Loop/Port Combo - Zone 3		3			62.26						l			1	†
1 1	Parae		-		_	02.20			1			 	 		 	+
LINE Loop P		+	1	UEPRG	UEPLX	11.77			 	 		1	1	1	1	+
UNE Loop R									0	i e	0	0			1	1
UNE Loop R	2-Wire Voice Grade Loop (SL1) - Zone 1															_
UNE Loop R	2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		2	UEPRG UEPRG	UEPLX UEPLX	22.39 48.26										

regory	RATE ELEMENTS	Interim	Zone	BCS	usoc		F	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	Increment al Charge Manual Svc Order vs. Electronic- Disc 1st	al Charg Manua Svc Ord vs. Electron
						Rec	Nonrec	urrina		curring onnect			OSS P	ATES (\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAI
	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res MBER PORTABILITY			UEPRG	UEPRD	14.00	90.00	90.00					31.92	7.32		
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15										
FEATURES				OLI IIO	LIVI OI	0.10										
	RRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPRG	USAC2		41.50	41.50					31.92	7.32		
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change			UEPRG	USACC		41.50	41.50								
ADDITIONA																
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring						0.00	0.00								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					19.99	19.99	19.99	19
	ICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)						14.04	14.04					19.99	19.99	19.99	- 13
	oop Combination Rates															-
	2-Wire VG Loop/Port Combo - Zone 1		1			25.77										
	2-Wire VG Loop/Port Combo - Zone 2		2			36.39										
	2-Wire VG Loop/Port Combo - Zone 3		3			62.26										
UNE Loop F																
	2-Wire Voice Grade Loop (SL1) - Zone 1			UEPPX	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPPX	UEPLX	22.39										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	48.26										
	te Grade Line Port Rates (BUS - PBX)															
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX UEPPX	UEPPC	14.00	90.00	90.00					31.92	7.32		-
	Line Side Unbundled Outward PBX Trunk Port - Bus Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPPO UEPP1	14.00 14.00	90.00 90.00	90.00					31.92 31.92	7.32 7.32		-
	2-Wire Voice Unbundled 2-Way Combination PBX Louisiana Calling Port			UEPPX	UEPL2	14.00	90.00	90.00					31.92	7.32		-
	2-Wire Voice Unbundled 2-Way Combination P BX Louisiana Calling P off			UEPPX	UEPLD	14.00	90.00	90.00					31.92	7.32		ł
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled 2-Way PBX Louisiana Local Optional Calling Port			UEPPX	UEPXK	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPX	UEPXM	14.00	90.00	90.00					31.92	7.32		-
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room			OLITA	OLF XIVI	14.00	30.00	30.00					31.32	1.52		ł
	Calling Port			UEPPX	UEPXO	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Louisiana Local Discount Calling															
	Port			UEPPX	UEPXP	14.00	90.00	90.00					31.92	7.32		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPPX	UEPXS	14.00	90.00	90.00					31.92	7.32		
	MBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15										ļ
FEATURES	RRING CHARGES - CURRENTLY COMBINED															-
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is			UEPPX	USAC2		41.50	41.50					31.92	7.32		-
	2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change			UEPPX	USACC		41.50	41.50					01.02	7.02		
ADDITIONA																
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPPX	USAS2		0.00	0.00					31.92	7.32		
	2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-															
	Nonrecurring						0.00	0.00								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					19.99	19.99	19.99	1:
	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															-
UNE PORT/L	oop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1	-	1			25.77					-	-				├
	2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2		2			36.39					1	-				\vdash
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			62.26					-					
UNE Loop F			3			32.20										
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	11.77										
	2-Wire Voice Grade Loop (SL1) - Zone 2			UEPCO	UEPLX	22.39					1					
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	48.26										
2-Wire Voic	e Grade Line Port Rates (Coin)															
	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,					,										
	MS) 3 Wire Coin 3 Way with Operator Screening and Blocking 011 000/076 1 DDD			UEPCO	UEPRF	14.00	90.00	90.00					31.92	7.32		<u> </u>
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS, SC)			UEPCO	UEPRA	14.00	90.00	90.00					24.00	7.00		
	[(AL, NI, LA, IVIO, OU)	1	1	UEPCO	UEPRA	14.00 14.00	90.00	90.00					31.92 31.92	7.32 7.32		1

TEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ı	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Manual	Increment al Charge Manual Svc Order vs. Electronic Disc 1st	al Charge Manual Svc Orde vs. Electronic
									Nonre	curring					•	
						Rec	Nonrec	urring	Disco	onnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, 8															
	Local (AL, KY, LA, MS)			UEPCO	UEPCD	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,			UEPCO	UEPRN	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward with Operator Screening and 011 Blocking (LA)			UEPCO	UEPLA	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD															
	(AL, KY, LA, MS)			UEPCO	UEPRH	14.00	90.00	90.00					31.92	7.32		
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCN	14.00	90.00	90.00					31.92	7.32		
	IBER PORTABILITY			UEPCU	UEPCN	14.00	90.00	90.00					31.92	1.32		
LOCAL NUM	MBER PURTABILITY			LIEBOO	LNDOV	0.05										
LIGNIDEGUIDI	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
	RING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination - Switch-As-Is	1		UEPCO	USAC2		41.50	41.50					31.92	7.32		
2	2-Wire Voice Grade Loop/ Line Port Combination - Switch with Change	1		UEPCO	USACC		41.50	41.50								
ADDITIONAL		1														
	2-Wire Voice Grade Loop/ Line Port Combination - Subsequent			UEPCO	USAS2		0.00	0.00					31.92	7.32		
LED CENTREX	PORT/LOOP COMBINATIONS															
UNBUNDLED	D PORT/LOOP COMBINATIONS - COST BASED RATES															
UNE-P CENT	TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)															
	.oop/2-Wire Voice Grade Port (Centrex) Combo															
	pop Combination Rates (Non-Design)															
2	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design			UEP91		13.13										
2	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP91		23.75										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP91		49.62										
	pop Combination Rates (Design)															
2	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP91		16.29										
2	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP91		26.71										
2	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP91		48.26										
UNE Loop Ra	late															
2	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP91	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP91	UECS1	22.39										
2	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP91	UECS1	48.26										
2	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP91	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP91	UECS2	25.35										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP91	UECS2	50.46										
UNE Ports																
	Except North Carolina and Sout Carolina)															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP91	UEPYA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP91	UEPYB	1.36	28.85	18.08				15.20				
+	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP91	UEPYH	1.36	38.85	19.08				15.20				
+	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local	1		UEP91	UEPYM	1.36	104.41	67.93				15.20			1	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic	1		02101	OLI TIVI	1.30	104.41	07.33	 		1	10.20				
	Local Area			UEP91	UEPYZ	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local	1		UEP91	UEPY9	1.36	38.85	19.08				15.20			1	
	2-Wire Voice Grade Port Terminated in 60 Service Term - Basic Local Area	1		UEP91	UEPY2	1.36	28.85	19.08	 		1	15.20				
	MS, & TN Only				OL. 12	50	20.00					.0.20				
	2-Wire Voice Grade Port (Centrex)	1		UEP91	UEPQA	1.36	38.85	19.08	 		1	15.20				
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)	1		UEP91	UEPQB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller 10)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	1		UEP91	UEPQH	1.36	104.41	67.93	l		1	15.20			1	
+ + +	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term	1	-	UEP91	UEPQIVI	13.60	104.41	67.93			+	15.20			 	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term 2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ2	13.60	38.85	19.08			1					
	z-voire voice chage Portleminaled in on Medalink of edulvalent			UEP91	UEPQ9	1.36	38.85	19.08			-	15.20 15.20			-	
2				UEPSI	UEPQ2	1.36	38.85	19.08			-	15.20			-	
2	2-Wire Voice Grade Port Terminated on 800 Service Term					0.8577										
2 2 Local Switch	2-Wire Voice Grade Port Terminated on 800 Service Term hing			LIEBA I												
2 2 Local Switch	2-Wire Voice Grade Port Terminated on 800 Service Term http://doi.org/10.1001/2009/2009/2009/2009/2009/2009/2009/			UEP91	URECS	0.0011									1	i .
Local Switch Local Number	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port er Portability															
Local Switch Local Number	2-Wire Voice Grade Port Terminated on 800 Service Term http://doi.org/10.1001/2009/2009/2009/2009/2009/2009/2009/			UEP91 UEP91	LNPCC	0.35										
Local Number	2-Wire Voice Grade Port Terminated on 800 Service Term hitter Centrex Intercom Funtionality, per port per Portability Local Number Portability (1 per port)			UEP91	LNPCC	0.35										
Local Switch Local Number Local Number Local Number Local Number	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port per Portability Local Number Portability (1 per port) All Standard Features Offered, per port			UEP91 UEP91	LNPCC	0.35						4-6:				
Local Switch Local Number Local Number Features	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port her Portability Local Number Portability (1 per port) All Standard Features Offered, per port All Select Features Offered, per port			UEP91 UEP91 UEP91	LNPCC UEPVF UEPVS	0.35 0.00 0.00	412.25					15.20				
Local Switch Local Numbe Local Numbe Features A A	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port per Portability Local Number Portability (1 per port) All Standard Features Offered, per port			UEP91 UEP91	LNPCC	0.35	412.25					15.20				
Local Switch Local Number Local Number Features A NARS	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port per Portability Local Number Portability (1 per port) All Standard Features Offered, per port All Select Features Offered, per port All Centrex Control Features Offered, per port			UEP91 UEP91 UEP91 UEP91	LNPCC UEPVF UEPVS UEPVC	0.35 0.00 0.00 0.00						15.20				
Local Switch Local Numbe Local Numbe Features A NARS	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port her Portability Local Number Portability (1 per port) All Standard Features Offered, per port All Select Features Offered, per port All Centrex Control Features Offered, per port Unbundled Network Access Register - Combination			UEP91 UEP91 UEP91 UEP91 UEP91	UEPVF UEPVS UEPVC UARCX	0.35 0.00 0.00 0.00 0.00	0.00	0.00				15.20				
Local Switch Local Numbe Local Numbe Features A NARS	2-Wire Voice Grade Port Terminated on 800 Service Term Contrex Intercom Funtionality, per port Portability			UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	UEPVF UEPVS UEPVC UARCX UAR1X	0.35 0.00 0.00 0.00 0.00 0.00	0.00	0.00				15.20				
Local Switch Local Numbe Features A NARS L L L L L L L L L L L L L	2-Wire Voice Grade Port Terminated on 800 Service Term hing Centrex Intercom Funtionality, per port her Portability Local Number Portability (1 per port) All Standard Features Offered, per port All Select Features Offered, per port All Centrex Control Features Offered, per port Unbundled Network Access Register - Combination			UEP91 UEP91 UEP91 UEP91 UEP91	UEPVF UEPVS UEPVC UARCX	0.35 0.00 0.00 0.00 0.00	0.00					15.20				

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		ı	RATES(\$)			Svc Order Submitte d Elec per LSR	Manually	Order vs.	I Charge - Manual Svc Order vs.	vs. Electronic	
									Nonre	curring				•	1	-
						Rec	Nonrec			onnect		1		ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Trunk Side Terminations, each			UEP91	CENA6	8.29	115.85	18.20				15.20				
Interoffice	Channel Mileage - 2-Wire			OLIBI	CLIVAU	0.23	113.03	10.20				13.20				+
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	MIGBC	22.60	39.36	26.62				15.20				+
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	MIGBM	0.13	00.00	20.02				10.20				+
Feature Ac	ctivations (DS0) Centrex Loops on Channelized DS1 Service															
D4 Channe	el Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP91	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP91	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV 1PQWQ	0.6497						15.20				+
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91 UEP91	1PQWQ	0.6497 0.6497						15.20 15.20				+
	rring Charges (NRC) Associated with UNE-P Centrex			OLIBI	II QWA	0.0437						13.20				+
	Conversion - Currently Combined Switch-As-Is with allowed changes, per port			UEP91	USAC2		0.10	0.10				15.20				+
	Conversion of Existing Centrex Common Block			UEP91	USACN	0.00	36.66	16.10								
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	680.40					15.20				
	Secondary Block, per Block			UEP91	M2CC1	0.00	79.31					15.20				
	NAR Establishment Charge, Per Occasion			UEP91	URECA	0.00	73.93					15.20				
	NTREX - 5ESS (Valid in All States)															
2-Wire VG	Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE Port/L	.oop Combination Rates (Non-Design)			UEP95		10.10										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95 UEP95		13.13 23.75										+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design			UEP95		49.62										+
UNF Port/I	Loop Combination Rates (Design)		3	UEF93		49.02										+
0.12 . 0.02	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP95		16.29										+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP95		26.71										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP95		51.82										
UNE Loop	Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1			UEP95	UECS1	11.77						15.20				
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP95	UECS1	22.39						15.20				
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	48.26										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		3	UEP95	UECS2	25.35	102.10	65.72				15.20				-
UNE Port R	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	50.46	102.10	65.72				15.20				+
All States	Nate															+
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.36	38.85	19.08				15.20				+
	2-Wire Voice Grade Port (Centrex) Basic Eccentrical 2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.36	38.85	19.08				15.20				+
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.36	38.85	19.08			1	15.20				1
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local	1		UEP95	UEPYM	1.36	104.41	67.93	İ		1	15.20				1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic															1
	Local Area			UEP95	UEPYZ	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local			UEP95	UEPY9	1.36	38.85	19.08				15.20				1
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.36	38.85	19.08	1		1	15.20				1
	A, MS, SC, & TN Only	1		LIEDOS	HEDC:	10.00	00.0-	10.00				45.00				1
	2-Wire Voice Grade Port (Centrex)	1		UEP95	UEPQA	13.60	38.85	19.08	1	-	1	15.20			-	+
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex with Caller ID)1	1		UEP95 UEP95	UEPQB UEPQH	1.36 1.36	38.85 38.85	19.08 19.08			1	15.20 15.20				+
	2-Wire Voice Grade Port (Centrex with Caller 10)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	1		UEP95	UEPQH	1.36	104.41	67.93	1	 	1	15.20			 	+
	2-Wire Voice Grade Port (Centrex Horri din Serving Wire Center)2 2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term	1		UEP95	UEPQZ	1.36	104.41	67.93	1	-	1	15.20			-	+
+	2-Wire Voice Grade Port, Bill Serving Wire Certier - 500 Service Terminated in on Megalink or equivalent	1		UEP95	UEPQ9	1.36	38.85	19.08	1		1	15.20				
	2-Wire Voice Grade Port Terminated on 800 Service Term	1		UEP95	UEPQ2	1.36	38.85	19.08				15.20				
Local Switch	ching															
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.8577						15.20				
	ber Portability						_	-								
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Features	100 100 100 100 100 100 100 100 100 100								1		1					1
	All Standard Features Offered, per port			UEP95	UEPVF	0.00			1		1	15.20				1
	All Select Features Offered, per port			UEP95	UEPVS	0.00	412.25					15.20				
	All Centrex Control Features Offered, per port	1		UEP95	UEPVC	0.00			l	1	1	15.20	-	1		+
NARS																

TEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	vs. Electronic-	-al Charge Manual Svc Orde vs.
						_		_		curring						
						Rec	Nonrec			nnect		T		ATES (\$)	T	
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
														_	↓	
	Ush a Radian and Assaula Basinian da Radian			LIEDOS	UAR1X	0.00	0.00	0.00							+	
	Unbundled Network Access Register - Indial			UEP95 UEP95	UARTX	0.00	0.00	0.00							 	
	Unbundled Network Access Register - Outdial			UEP95	UARUX	0.00	0.00	0.00							+	
	ous Terminations															
2-Wire Trur				UEP95	CEND6	8.29	115.85	18.20				15.20				
4 14/2 B1-1	Trunk Side Terminations, each			UEP95	CEND6	8.29	115.85	18.20				15.20			+	
	ital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP95	M1HD1	68.47	196.18	92.92	4.90			15.20				
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.06					15.20			 	
	Channel Mileage - 2-Wire													.		
	Interoffice Channel Facilities Termination			UEP95	MIGBC	22.60	39.36	26.62				15.20		_	↓	
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.013								.		
	stivations (DS0) Centrex Loops on Channelized DS1 Service	1			\rightarrow									 		
U4 Channe	el Bank Feature Activations	1		LIEDOS	4001110	0.046=						45.00			+	+
1	Feature Activation on D-4 Channel Bank Centrex Loop Slot	1		UEP95	1PQWS	0.6497						15.20			4	
1	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.6497						15.20			4	1
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.6497						15.20			4	1
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center	1		UEP95	1PQWP	0.6497					1	15.20			4	1
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.06497						15.20				
	ring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP95	USAC2		0.10	0.10				15.20				
	Conversion of Existing Centrex Common Block, each			UEP95	USACN		36.66	16.10								
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	680.40					15.20				
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	73.93					15.20				
	ITREX - DMS100 (Valid in All States)															
	Loop/2-Wire Voice Grade Port (Centrex) Combo															
	.oop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9D		13.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9D		23.75										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9D		49.62										
UNE Port/L	.oop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP9D		16.29										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		26.71										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9D		51.82										
UNE Loop I	Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	22.39					1			1		1
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP9D	UECS1	48.26									I	
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP9D	UECS2	14.93									1	
	2-Wire Voice Grade Loop (SL 2) - Zone 2			UEP9D	UECS2	25.35									1	
	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP9D	UECS2	50.46									1	
UNE Port R															1	
ALL STATE															1	
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.36	38.85	19.08				15.20			1	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9D	UEPYB	1.36	38.85	19.08				15.20			1	
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area			UEP9D	UEPYC	1.36	38.85	19.08				15.20			T	
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.36	38.85	19.08				15.20		1	1	
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.36	38.85	19.08				15.20			1	
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.36	38.85	19.08				15.20			1	1
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	1.36	38.85	19.08				15.20		1	1	—
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.36	38.85	19.08				15.20			1	1
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	1.36	38.85	19.08				15.20		1	1	—
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	1.36	38.85	19.08				15.20		1		—
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.36	38.85	19.08				15.20			1	†
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area	1		UEP9D	UEPYH	1.36	38.85	19.08			<u> </u>	15.20				t
+	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic	1		021 00	OL1 111	1.30	55.55	15.00			 	10.20			+	+
	Local Area			UEP9D	UEPYW	1.36	38.85	19.08			1	15.20				
+	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.36	38.85	19.08			 	15.20		 	+	+
	2-Wire Voice Grade Port (Centrex/msg wtg Lamp Indication))3 Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local			UEP9D	UEPYM	1.36	104.41	67.93			 	15.20		 	+	+
	2-Wire Voice Grade Port (Centrex from dill Serving Wire Center) 2 Basic Local Area	1		UEP9D	UEPYO	1.36	104.41	67.93			1	15.20		 	+	+
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSE1)2, 3 Basic Local Area 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D UEP9D	UEPYO	1.36	104.41	67.93			-	15.20		 	+	+
		1			UEPYP						-			 	+	+
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D		1.36	104.41	67.93	ļ			15.20			4	+
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area	1	1	UEP9D	UEPYR	1.36	104.41	67.93			1	15.20		1		1

CATEGORY	RATE ELEMENTS	Interim Zone	BCS	USOC		ı	RATES(\$)			Submitte d Elec	Svc Order Submitted	Incremental Charge - Manual Svc Order vs. Electronic-	Manual Svc Order vs. Electronic-	al Charge Manual Svc Order vs. Electronic	-al Charge - Manual Svc Order vs.
					Rec	Nonrec	urrina	Nonre Disco	curring onnect			OSS RA	ATES (\$)		
						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area		UEP9D	UEPYS	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area		UEP9D	UEPY4	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area		UEP9D	UEPY5	1.36	104.41	67.93				15.20				

ATEGORY	RATE ELEMENTS	Interim Zone	BCS	USOC			RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental I (Charge - I Manual Svc Order vs. Electronic- Ele	Charge - : Manual vc Order : vs. ectronic-	Manual Svc Order vs.	-al Charg Manua Svc Ord vs. -Electron
					_				curring						
					Rec	Nonred			onnect		T	OSS RATE			
						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN S	SOMAN	SOMAN	SOMA
	VI - 0 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1									1					
	re Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area		UEP9D	UEPY6	1.36	104.41	67.93			1	15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area		UEP9D	UEPY7	1.36	104.41	67.93				15.20				+
	re Voice Grade Port, Diff Serving Wire Center - 800 Service Term		UEP9D	UEPYZ	1.36	104.41	67.93				15.20				
	re Voice Grade Port terminated in on Megalink or equivalent Basic Local Area		UEP9D	UEPY9	1.36	38.85	19.08			1	15.20				
	re Voice Grade Port Terminated on 800 Service Term Basic Local Area		UEP9D	UEPY2	1.36	38.85	19.08				15.20				
AL, KY, LA, MS,			LIEDOD	UEDOA	4.00	00.05	40.00				45.00				
	re Voice Grade Port (Centrex)		UEP9D	UEPQA	1.36	38.85	19.08			1	15.20				-
	re Voice Grade Port (Centrex 800 termination)		UEP9D	UEPQB	1.36	38.85	19.08			1	15.20				-
	re Voice Grade Port (Centrex / EBS-PSET)3		UEP9D	UEPQC	1.36	38.85	19.08			1	15.20				-
	re Voice Grade Port (Centrex / EBS-M5009)3		UEP9D	UEPQD	1.36	38.85	19.08				15.20				1
	re Voice Grade Port (Centrex / EBS-M5209)3		UEP9D	UEPQE	1.36	38.85	19.08				15.20				╀
	re Voice Grade Port (Centrex / EBS-M5112)3		UEP9D	UEPQF	1.36	38.85	19.08				15.20	-			
	re Voice Grade Port (Centrex / EBS-M5312)3		UEP9D	UEPQG	1.36	38.85	19.08			1	15.20				-
	re Voice Grade Port (Centrex / EBS-M5008)3		UEP9D	UEPQT	1.36	38.85	19.08				15.20				
	re Voice Grade Port (Centrex / EBS-M5208)3		UEP9D	UEPQU	1.36	38.85	19.08				15.20				ļ
	re Voice Grade Port (Centrex / EBS-M5216)3		UEP9D	UEPQV	1.36	38.85	19.08				15.20				
	re Voice Grade Port (Centrex / EBS-M5316)3		UEP9D	UEPQ3	1.36	38.85	19.08				15.20				
	re Voice Grade Port (Centrex with Caller ID)		UEP9D	UEPQH	1.36	38.85	19.08				15.20				
	re Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3		UEP9D	UEPQW	1.36	38.85	19.08				15.20				1
	re Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3		UEP9D	UEPQJ	1.36	38.85	19.08				15.20				
	re Voice Grade Port (Centrex from diff Serving Wire Center) 2		UEP9D	UEPQM	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3		UEP9D	UEPQO	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3		UEP9D	UEPQP	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3		UEP9D	UEPQQ	1.36	104.41					15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3		UEP9D	UEPQR	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3		UEP9D	UEPQS	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3		UEP9D	UEPQ4	1.36	104.41	67.93				15.20				ļ
	re Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3		UEP9D	UEPQ5	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3		UEP9D	UEPQ6	1.36	104.41	67.93				15.20				
	re Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3		UEP9D	UEPQ7	1.36	104.41	67.93				15.20				
	re Voice Grade Port, Diff Serving Wire Center - 800 Service Term		UEP9D	UEPQZ	1.36	104.41	67.93				15.20				
	re Voice Grade Port terminated in on Megalink or equivalent		UEP9D	UEPQ9	1.36	38.85	19.08				15.20				
	re Voice Grade Port Terminated on 800 Service Term		UEP9D	UEPQ2	1.36	38.85	19.08				15.20				
Local Switching															
	rex Intercom Funtionality, per port		UEP9D	URECS	0.8577										
Local Number Po															
	Number Portability (1 per port)		UEP9D	LNPCC	0.35										
Features						· · · · · · · · · · · · · · · · · · ·									
	andard Features Offered, per port		UEP9D	UEPVF	0.00						15.20				
All Se	elect Features Offered, per port		UEP9D	UEPVS	0.00	412.25					15.20				
All Ce	entrex Control Features Offered, per port		UEP9D	UEPVC	0.00						15.20				
NARS															
	ndled Network Access Register - Combination		UEP9D	UARCX	0.00	0.00	0.00								
Unbur	ndled Network Access Register - Inward		UEP9D	UAR1X	0.00	0.00	0.00								
	ndled Network Access Register - Outdial		UEP9D	UAROX	0.00	0.00	0.00								
Miscellaneous Te															
2-Wire Trunk Sid															
	Side Terminations, each		UEP9D	CEND6	8.29	115.85	18.20				15.20				
4-Wire Digital (1.															
	Circuit Terminations, each		UEP9D	M1HD1	68.47	196.18	98.62				15.20				
	Channels Activiated per Channel		UEP9D	M1HDO	0.00	14.06					15.20				
Interoffice Chann	nel Mileage - 2-Wire														
Interd	office Channel Facilities Termination		UEP9D	MIGBC	22.60	39.36	26.62			1	15.20				
	office Channel mileage, per mile or fraction of mile		UEP9D	MIGBM	0.013			1	1		1				
Feature Activation	ons (DS0) Centrex Loops on Channelized DS1 Service														
	k Feature Activations														
	ure Activation on D-4 Channel Bank Centrex Loop Slot		UEP9D	1PQWS	0.6497						15.20				
	ure Activation on D-4 Channel Bank FX line Side Loop Slot		UEP9D	1PQW6	0.6497						15.20				

TEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ŗ	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st		vs.	-al Charg Manua Svc Ord vs. -Electror
										curring						
						Rec	Nonrec	urring	Disco	onnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
																1
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.6497						15.20			1	1
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.6497					1	15.20			+	+
		+				0.6497									+	+
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV						-	15.20			+	+
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.6497						15.20				
	ring Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP9D	USAC2		0.10	0.10				15.20				
	Conversion of existing Centrex Common Block, each			UEP9D	USACN		36.66	16.10								1
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	680.40					15.20			1	1
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	680.40				1	15.20			+	+
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	73.93				1	15.20			+	+
LINE DOEN	TREY ENOD (ALICE ALL EL LOCASION			UEF9D	UNECA	0.00	13.83				-	15.20			+	+
	NTREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)	-							l		1	1	1	1	4	1
	Loop/2-Wire Voice Grade Port (Centrex) Combo	 										ļ			4	4
	oop Combination Rates (Non-Design)														1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design			UEP9E		13.13			<u></u>	<u> </u>		<u></u>	L	L	1	<u> Ш</u>
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP9E		23.75				1					1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP9E		49.62										1
	oop Combination Rates (Design)	1										İ			1	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP9E		16.29					1				+	+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	+		UEP9E		26.71									+	+
											-				+	+-
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9E		51.82										
UNE Loop F																
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	22.39										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	48.26										
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP9E	UECS2	14.93										1
	2-Wire Voice Grade Loop (SL 2) - Zone 2			UEP9E	UECS2	25.35									1	†
	2-Wire Voice Grade Loop (SL 2) - Zone 3			UEP9E	UECS2	50.46					1				+	+
UNE Port R			3	UEF9E	UECSZ	30.46									+	+
											-				+	+
	, LA, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9E	UEPYA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP9E	UEPYH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local			UEP9E	UEPYM	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic															1
	Local Area			UEP9E	UEPYZ	1.36	104.41	67.93				15.20				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local			UEP9E	UEPY9	1.36	38.85	19.08				15.20			1	1
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP9E	UEPY2	1.36	38.85	19.08			1	15.20			+	+
		+		OLI OL	OLI 12	1.00	50.05	13.00				10.20			+	+
	, MS, & TN Only	-	-	LIEDOE	LIEDO4	4.00	20.05	40.00		1	-	45.00			+	+
	2-Wire Voice Grade Port (Centrex)	1	-	UEP9E	UEPQA	1.36	38.85	19.08	-		1	15.20	 	 	+	+
	2-Wire Voice Grade Port (Centrex 800 termination)	-		UEP9E	UEPQB	1.36	38.85	19.08				15.20				+
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.36	104.41	67.93				15.20			1	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term	1		UEP9E	UEPQZ	1.36	104.41	67.93		<u> </u>		15.20			1	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.36	38.85	19.08				15.20			1	1
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.36	38.85	19.08				15.20				
Local Switch																1
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.8577									1	
	ber Portability	1			5.1.250	0.00.7						1	 	 	+	+
	Local Number Portability (1 per port)	+		UEP9E	LNPCC	0.35			 		1	t	 		+	+
Features	Local Hambel Foliability (1 per port)	1		OLF 3L	LINFOC	0.33				 	+	1			+	+
	All Chandred Fastings Offered and and	+	-	LIEDOE	LIED) (E	0.00				1	1	45.00			+	+-
	All Standard Features Offered, per port	-		UEP9E	UEPVF	0.00				ļ		15.20				+
	All Select Features Offered, per port	1		UEP9E	UEPVS	0.00	412.25					15.20			1	
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	0.00						15.20			1	1
NARS			L												1	
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								1
	Unbundled Network Access Register - Indial	1		UEP9E	UAR1X	0.00	0.00	0.00				İ			1	
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00				1				1
	ous Terminations	1			S, 11.5/	0.00	0.00	0.00				1	 	 	+	1
2-Wire Trur		1				+				 	+	1			+	+
		+	-	LIEDOE	OFNE	0.00	445.05	10.00		1	-	45.00			+	+
	Trunk Side Terminations, each	 		UEP9E	CEND6	8.29	115.85	18.20			-	15.20			4	4
	ital (1.544 Megabits)														1	
	DS1 Circuit Terminations, each			UEP9E	M1HD1	68.47	196.18	92.92				15.20			1	1
	DS0 Channel Activated Per Channel	1		UEP9E	M1HDO	0.00	14.06	-				15.20				
	Channel Mileage - 2-Wire										+	1	1	1	+	

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ı	RATES(\$)						Manual Svc Order vs.	al Charge - Manual Svc Order vs.	al Charg Manua Svc Ord vs. Electror
									Nonre	curring					•	
						Rec	Nonrec	urring	Disco				OSS R	ATES (\$)		
						1100	First	Add'l	First	Add'l	COMEC	SOMAN	COMAN	SOMAN	SOMAN	COMA
							FIRST	Addi	FIIST	Add I	SOMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMA
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.013										
Feature A	Activations (DS0) Centrex Loops on Channelized DS1 Service															
	nel Bank Feature Activations															
D4 Onami	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.6497						15.20				
																+
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9E	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9E	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.6497						15.20				
Non Press		+	_	JL. JL	11 2117	0.0431					1	10.20		 		
Non-Recu	urring Charges (NRC) Associated with UNE-P Centrex	1	-	LIEDOE	110400		0 10	0.45			1	45.60		1	1	
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port	1	1	UEP9E	USAC2		0.10	0.10			1	15.20		ļ	1	4
	Conversion of Existing Centrex Common Block, each	1		UEP9E	USACN		36.66	16.10			1			1	1	
	New Centrex Standard Common Block			UEP9E	M1ACS	0.00	680.40					15.20				1
	New Centrex Customized Common Block			UEP9E	M1ACC	0.00	680.40					15.20				
	NAR Establishment Charge, Per Occasion			UEP9E	URECA	0.00	73.93					15.20		1		t t
LINE D.CE	ENTREX - DCO - Valid in AL, KY, LA, MS, & TN)			JL. JL	SILOA	0.00	70.33				+	10.20		†	1	1
											1					
	G Loop/2-Wire Voice Grade Port (Centrex) Combo															
UNE Port/	/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP93		13.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP93		23.75										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design			UEP93		49.62										-
LINE Dest	//Loop Combination Rates (Design)		J	OLI 33	_	45.02										+
UNE PORT	/Loop Combination Rates (Design)										1					
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design			UEP93		16.29										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP93		26.71										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP93		51.82										
UNE Loop																
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP93	UECS1	11.77										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEP93	UECS1	22.36										+
																-
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEP93	UECS1	48.26										
	2-Wire Voice Grade Loop (SL 2) - Zone 1			UEP93	UECS2	14.93										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	25.35										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	50.46										
UNE Port			Ť													_
					_											+
AL, NI, L	.A, MS, & TN only										1					
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP93	UEPYB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP93	UEPYH	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local			UEP93	UEPYM	1.36	104.41	67.93				15.20		1		
+	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic				OC1 11VI	1.00		37.33			1	10.20		t		+
				UEP93	UEPYZ	1.36	104.41	67.93			1	15.20		1		1
	Local Area	1	1								1			 	1	1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local			UEP93	UEPY9	1.36	38.85	19.08			1	15.20		1		1
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP93	UEPY2	1.36	38.85	19.08			1	15.20				
	2-Wire Voice Grade Port (Centrex)	1	1 7	UEP93	UEPQA	1.36	38.85	19.08			1	15.20		1		1
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP93	UEPQB	1.36	38.85	19.08				15.20				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.36	38.85	19.08				15.20		1		
	2-Wire Voice Grade Port (Centrex with Caller 15)1 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2	1	1	UEP93	UEPQM	1.36	104.41	67.93			1	15.20		1	1	
-		1	1								1			 	1	1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP93	UEPQZ	1.36	104.41	67.93			1	15.20		1		1
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.36	38.85	19.08				15.20		ļ		
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.36	38.85	19.08				15.20				
Local Swi	ritching															1
	Centrex Intercom Funtionality, per port			UEP93	URECS	0.8577										
Local Nur	mber Portability	1			2200	2.0011					1			1		t —
Local Nul				UEP93	LNCCC	0.35					+	 		†	1	+
F	Local Number Portability (1 per port)	1	1	OLFBO	LINCCC	0.35					1	_		 	1	1
Features		1	1	LIEBAA							1			ļ	1	1
	All Standard Features Offered, per port			UEP93	UEPVF	0.00					1	15.20		1		
	All Centrex Control Features Offered, per port	1	1 7	UEP93	UEPVC	0.00		T			1	15.20		1		1
NARS														1	1	1
	Unbundled Network Access Register - Combination	1		UEP93	UARCX	0.00	0.00	0.00			1	1		1		T
				UEP93	UAR1X						+	 		†	1	+
_	Unbundled Network Access Register - Indial	1	1			0.00	0.00	0.00			1			 		₩
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00						ļ		
	neous Terminations													<u> </u>		
2-Wire Tru	runk Side															1
	Trunk Side Terminations, each	1		UEP93	CEND6	8.27	115.85	18.20				15.20				

CATEGORY	RATE ELEMENTS	Interim Zone	BCS	USOC		F	RATES(\$)	N			Svc Order Submitted Manually	Incremental Charge - Manual Svc	Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	-al Charge - Manual Svc Order vs.
					Rec	Nonrec	urring		curring onnect			OSS B	ATES (\$)		
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
				+		LIISI	Auu i	FIISL	Auu I	SOWIEC	JUNIAN	JUNIAN	JUNIAN	JUNAN	SUMAN
				+						1					+
4-Wire Dio	gital (1.544 Megabits)														
	DS1 Circuit Terminations, each		UEP93	M1HD1	68.47	196.18	92.92				15.20				
	DS0 Channels Activated, Per Channel		UEP93	M1HDO	0.00	14.01					15.20				
Interoffice	Channel Mileage - 2-Wire														
	Interoffice Channel Facilities Termination		UEP93	MIGBC	22.60	39.36	26.62				15.20				
	Interoffice Channel mileage, per mile or fraction of mile		UEP93	MIGBM	0.013										
	ctivations (DS0) Centrex Loops on Channelized DS1 Service														
D4 Chann	el Bank Feature Activations														
	Feature Activation on D-4 Channel Bank Centrex Loop Slot		UEP93	1PQWS	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot		UEP93	1PQW6	0.6497						15.20				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot		UEP93	1PQW7	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center		UEP93	1PQWP	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot		UEP93	1PQWV	0.6497						15.20				
	Feature Activation on D-4 Channel Bank Tie Line/Trunk Loop Slot		UEP93	1PQWQ	0.6497						15.20				
	Feature Activation on D-4 Channel Bank WATS Loop Slot		UEP93	1PQWA	0.6497						15.20				
Non-Recu	rring Charges (NRC) Associated with UNE-P Centrex														
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port		UEP93	USAC2		0.10	0.10				15.20				ļ
	Conversion of Existing Centrex Common Block, each		UEP93	USACN		36.66	16.10				15.20				ļ
	New Centrex Standard Common Block		UEP93	M1ACS	0.00	680.40					15.20				-
	New Centrex Customized Common Block		UEP93	M1ACC	0.00	680.40					15.20				ļ
No. 4 B	NAR Establishment Charge, Per Occasion		UEP93	URECA	0.00	73.93					15.20				
	equired Port for Centrex Control in 1AESS, 5ESS & EWSD			-											
	Requires Interoffice Channel Mileage			-											
Note 3 - R	equires Specific Customer Premises Equipment														

EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	RATES(\$)			Svc Order Submitte d Elec per LSR	Svc Order Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge Manual Svc Order vs. Electronic Disc 1st	-al Charg Manua Svc Ord vs. -Electror
						Rec	Nonrec	urrina		curring			OSS B	ATES (\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
							11130	Auu	11131	Auu	COMILO	COMPLE	COMPAN	COMPAR	COMPAN	COMA
LOCAL INT	FERCONNECTION															
	CTION (CALL TRANSPORT AND TERMINATION)															
	beside a rate indicates that the Parties have agreed to bill and keep for that e	lement p	ursuai	nt to the terms a	nd condition	ons in Attachn	nent 3.									
	Per MOU Rate for Local and ISP-bound Traffic (1/1/01-12/31/01)					\$0.0017500										
	Per MOU Rate for Local and ISP-bound Traffic (1/1/02-12/31/02)					\$0.0015000										
																1
TANDEM S				0.10												1
	Tandem Switching Function Per MOU			OHD		0.0005507										
TRUNK CH	Multiple Tandem Switching, per MOU (applies to intial tandem only)			OHD		0.0005507										-
	Installation Trunk Side Service - per DS0			OHD	TPP++		334.94bk	56.98bk								
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00	334.94DK	30.90DK								1
	Dedicated End Office Trunk Port Service-per DS0 Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P	0.00										
	Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P	0.00										
	Dedicated Tandem Trunk Port Service-per DS1**			OH1 OH1MS	TDW1P											
	e element is recovered on a per MOU basis and is included in the End Office Sw	itching a														
	TRANSPORT (Shared)			idom omkomig	po:oo :	ato olomonto										
	Common Transport - Per Mile, Per MOU			OHD		0.0000032										
	Common Transport - Facilities Termination Per MOU			OHD		0.0003748										
ITERCONNEC	CTION (TRANSPORT)															
INTEROFFI	ICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per			OHL, OHM	1L5NF	0.013bk										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility															
	Termination per month			OHL, OHM	1L5NF	22.6bk	26.62bk									
	ICE CHANNEL - DEDICATED TRANSPORT - 56/64 KBPS															
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			OHL, OHM	1L5NK	0.013bk	00.0011									
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month			OHL, OHM OHL, OHM	1L5NK	15.61bk 0.013bk	26.62bk									
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			OHL, OHM	1L5NK		00.0051									
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month	-		Onl, Univi	1L5NK	15.61bk	26.62bk				1					1
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			OH1, OH1MS	1L5NL	0.2652bk										1
	Interoffice Channel - Dedicated Channel - DS1 - Fer Mile per month			OH1, OH1MS	1L5NL	70.47bk	79.44bk									+
	ICE CHANNEL - DEDICATED TRANSPORT- DS3			OITI, OITINIS	ILJINL	70.47bk	73.4400									
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			OH3, OH3MS	1L5NM	6.04bk										
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			OH3, OH3MS	1L5NM	850.45bk	158.05bk									1
	IANNEL - DEDICATED TRANSPORT			,	1.20.00											
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	18.32bk	187.51bk	32.21bk								
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	19.41bk	187.94bk	32.63bk								
	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	39.18bk	172.34bk	149.27bk								
	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	469.44bk	438.46bk	256.3bk								
	FERCONNECTION MID-SPAN MEET															
	ccess service ride Mid-Span Meet, one-half the tariffed service Local Channel r	ate is ap	plicabl	e.												
MULTIPLE																
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	105.09bk	88.41bk	60.76bk			1					1
	DS3 to DS1 Channel System per month			OH3, OH3MS	SATNS	201.48bk	172.99bk	91.25bk								
	DS3 Interface Unit (DS1 COCI) per month	1	1	OH1, OH1MS	SATCO	11.78bk	6.39bk	4.58bk		1	1	1			1	1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc		ŗ	RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	vs. Electronic-	-al Charge - Manual
										curring						
						Rec	Nonrec			nnect				ATES (\$)		,
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
															 '	
																-
PHYSICAL COLLOCA	TION														 	
PHISICAL COLLOCA	Physical Collocation - Application Fee - Initial			CLO	PE1BA		1,837.24									
	Physical Collocation - Application Fee - Bubsequent			CLO	PE1CA		1,533.41								ļ	
	Physical Collocation - Application Fee - Subsequent for Co-Carrier Cross Connect			CLO	PE1DT		583.30	583.30							+	
	Physical Collocation - Space Preparation - Firm Order Processing	_		CLO	PE1SJ		583.33	000.00							+	
	Physical Collocation - Space Preparation - C.O. Modification per square ft.	-		CLO	PE1SK	2.31	000.00						*****			
	Physical Collocation - Space Preparation - Common Systems Modification per			OLO	LIOK	2.01										\vdash
	square ft Cageless			CLO	PE1SL	2.70										
	Physical Collocation - Space Preparation - Common Systems Modification per														1	
	Cage			CLO	PE1SM	91.60										
	Physical Collocation - Cable Installation			CLO	PE1BD		841.54	841.54								
	Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	5.30										
	Physical Collocation - Cable Support Structure			CLO	PE1PM	18.31									1	
	Physical Collocation - Power per Fused Amp	ı		CLO	PE1PL	8.32										
	Physical Collocation - 120V, Single Phase Standby Power Rate	I		CLO	PE1FB	5.45			I							
	Physical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	10.92									1	
	Physical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	16.37										
	Physical Collocation - 277V, Three Phase Standby Power Rate			CLO	PE1FG	37.80									1	
				UEANL,UEA,UDN,												
				UDC,UAL,UHL,UC												
	Physical Collocation - 2-Wire Cross-Connects			L,UEQ	PE1P2	0.0318	11.94	11.46								
	Physical Collocation - 4-Wire Cross-Connects			CLO	PE1P4	0.0636	12.04	11.53								
				CLO,UEANL,UEQ,												
	Physical Collocation - DS1 Cross-Connects			WDS1L,WDS1S	PE1P1	1.04	21.39	15.47								
	Physical Collocation - DS3 Cross-Connects			CLO	PE1P3	13.21	20.28	14.76								ļ
	Physical Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.62	20.28	14.76								
	Physical Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	4.65	24.81	19.29							<u> </u>	
				0.0												
	Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.	_		CLO	PE1BW	184.50										
	Physical Callegation Walded Wise Come Addll 50 Co. Et			CLO	PE1C W	40.40										
	Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.	-		CLO CLO		18.10										
	Physical Collocation - Security System Per Central Office Per Assignable Sq. Ft. Physical Collocation - Security Access System - New Access Card Activation, per	-	ļ	CLO	PE1AX	0.0224										
	Card			CLO	PE1A1	0.0579	27.50									
	Physical Collocation-Security Access System-Administrative Change, existing	-	ļ	CLO	FEIAI	0.0379	27.50								ļ	
	Access Card, per Card			CLO	PE1AA		7.74	7.74								
	Physical Collocation - Security Access System - Replace Lost or Stolen Card, per			OLO	ILIAA		7.74	7.74								
	Card			CLO	PE1AR		22.64	22.64								
	Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		13.01	13.01							1	t
	Physical Collocation - Security Access - Key, Replace Lost or Stolen Key, per Key			CLO	PE1AL		13.01	13.01							 	
	Physical Collocation - Space Availability Report per premises			CLO	PE1SR		1,044.07	1,044.07							 	
	, , , , , , , , , , , , , , , , , , ,	<u> </u>		UEANL, UEA, UDN,											+	
				UDC,UAL,UHL,UC												
	POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			L,UEQ,CLO	PE1PE	0.079										
				UEANL,UEA,UDN,												
				UDC,UAL,UHL,UC											'	
	POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			L,UEQ,CLO	PE1PF	0.158]	J
				UEANL, UEA, UDN,												
				UDC,UAL,UHL,UC												
				L,UEQ,CLO,WDS											'	
	POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			1L,WDS1S,	PE1PG	1.12										
				UEANL, UEA, UDN,												
	DOT Dev. Assessments exists 6/4/00 DOC Co			UDC,UAL,UHL,UC	DEAD	0.05								l	1 '	
	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect	1	ļ	L,UEQ,CLO UEANL,UEA,UDN,	PE1PH	9.95			ļ		<u> </u>	ļ			 '	
				UDC,UAL,UHL,UC										l	1 '	
	BOT Boy Arrangements prior to 6/4/00 00 5/1 C C				DE4DO	22.00								l	1 '	
	POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect		 	L,UEQ,CLO UEANL,UEA,UDN,	PE1B2	33.96			l	l	1	-				
		1	1	UDC,UAL,UHL,UC					1	ĺ					1 '	
	POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			L,UEQ,CLO	PE1B4	45.80								l	1 '	
	Collocation Cable Records - per request	-		CLO	PE1B4 PE1CR	45.80 10.97			l	l	<u> </u>				+	ł
	Collocation Cable Records - per request Collocation Cable Records - VG/DS0 Cable, per cable record	_		CLO	PR1CD	5.29			-			—		l		
1 1		-	 	CLO	PE1CO	0.08			 	 	1	 		l	+	
												1				1
	Collocation Cable Records - VG/DS0 Cable, per each 100 pair														1	
	Collocation Cable Records - VG/DS0 Cable, per each 100 pair Collocation Cable Records - DS1, per T3TIE Collocation Cable Records - DS3, per T3TIE			CLO CLO	PE1C1 PE1C3	0.04 0.13										

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		ı	RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	-al Charge - Manual Svc Order vs.
						Rec	Nonrec	urring		curring onnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT		16.44	10.42								
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE10T		21.41	13.45								
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		26.38	16.49								Í
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,															Ì
	per linear ft.			CLO	PE1ES	0.0024										ļ
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support			01.0	DE 4 DO	0.0000										Ì
	Structure, per lin. ft.			CLO	PE1DS	0.0036										
ADJACENT COLLOC	ATION															
/ CO	Adjacent Collocation - Space Charge per Sq. Ft.			CLO	PE1JA	0.06										1
	Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLO	PE1JC	5.61										1
	Adjacent Collocation - 2-Wire Cross-Connects			CLO	PE1P2	0.02	11.94	11.46								
				UEA,UHL,UDL,UC												
	Adjacent Collocation - 4-Wire Cross-Connects			L,CLO	PE1P4	0.05	12.04	11.53								1
	Adjacent Collocation - DS1 Cross-Connects			USL,CLO	PE1P1	0.96	21.39	15.47								
	Adjacent Collocation - DS3 Cross-Connects			CLO	PE1P3	13.01	20.28	14.76								
	Adjacent Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.20	20.28	14.76								
	Adjacent Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	4.21	24.81	19.29								
	Adjacent Collocation - Application Fee			CLO	PE1JB		1543.20									
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker															
	Amp			CLO	PE1FB	5.45										L
	Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker			0.0												1
	Amp Adiacent Collocation - 120V. Three Phase Standby Power Rate per AC Breaker			CLO	PE1FD	10.92										
i l	Amp			CLO	PE1FE	16.37										1
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker			CLO	FEIFE	10.37										├
	Amp			CLO	PE1FG	37.80										
PHYSICAL COLLOCA	TION IN THE REMOTE SITE - ADJACENT															
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	rates which are subject to true-up.															
NOTE: If S	Security Escort and/or Add'l Engineering Fees become necessary for remote sit	e colloca	tion, th	ne Parties will neg	otiate app	ropriate rates										

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)				Submitted Manually	Incrementa Charge - Manual Svo Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	Manual Svc Order vs.
						_				curring				.=== (4)		
						Rec	Nonred			onnect	201150	001111		ATES (\$)	001111	001111
		-	-				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		-	-							-						
INTEDIM SERVICE PROVIDE	ER NUMBER PORTABILITY - RCF															
	F. per number ported (Business Line)				TNPBL	2.91	0.25	0.25			3.50	15.20				
	F. per number ported (Basiness Line)				TNPRL	2.91	0.25	0.25			3.50					
	F. Per Additional Path				TIMETAL	1.24	0.23	0.23			3.30	13.20				
	ment that can be ordered electronically will be billed according to the SC	MEC rate I	isted. Ple	ase refer to Bel	South's I	Business Rul	es for Local O	rdering (BBR-L	O) to dete	rmine if a p	roduct can	be ordered	electronical	y. For those	elements	hat cannot
	ctronically at present per the BBR-LO, the listed SOMEC rate reflects the															
CLEC's bill whe	en it submits an LSR to BellSouth.	_											_	-		
INTERIM SERVICE PROVIDE	ER NUMBER PORTABILITY - DID															
DID	per number ported (Residence)				TNPDR		0.42	0.42			3.50	15.20				
DID	per number ported (Business)				TNPDB		0.42	0.42			3.50	15.20				
DID), per trunk termination, Initial				TNPT2	68.47	185.13	68.79			3.50	15.20				
SERVICE PROVIDER NUMB	BER PORTABILITY (RIPH)															
RIP	PH, Functionality, Per Rearrangement						19.24	19.24			3.50	15.20				
RIP	PH, Per Number Ported					1.62	0.19	0.19			3.50	15.20				
RIP	PH, Functionality, Per Central Ofc						79.67	79.67			3.50	15.20				
Note: If no rate	e is identified in the contract, the rate for the specific service or function	will be as	set forth in	n applicable Be	ISouth ta	riff or as neg	otiated by the	Parties upon r	equest by	either Party	1					
	·					Ĭ										

															Increment
												Incremental			
										Svc		Charge -			
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS USO			RATES(\$)					Manual Svc		Svc Order	Svc Order
												Order vs.	vs.	vs.	vs.
												Electronic-			
										per LSR	per LSR	1st	Add'l	Disc 1st	Disc Add'l
									curring						
					Rec		curring		nnect				ATES (\$)		
						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/CMI															
ACC	ESS DAILY USAGE FILE (ADUF)														
	ADUF: Message Processing, per message			N/A	.001825										
	ADUF: Data Transmission (CONNECT:DIRECT), per message			N/A	.00012147										
OPTI	ONAL DAILY USAGE FILE (ODUF)														
	ODUF: Recording, per message			N/A	0.0000117										
	ODUF: Message Processing, per message			N/A	.002446										
	ODUF: Message Processing, per Magnetic Tape provisioned			N/A	35.54										
	ODUF: Data Transmission (CONNECT:DIRECT), per message			N/A	.00010122										
CEN	TRALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)														
	CMDS: Message Processing, per message			N/A	0.004									1	
	CMDS: Data Transmission (CONNECT:DIRECT), per message			N/A	0.001										
	nnced Optional DUF (message processing, per message)				.229779									1	
Note	s: If no rate is identified in the contract, the rate for the specific service or function	n will be as	set fo	rth in applicable BellSout	h tariff or as ne	gotiated by th	e Parties upor	n request by	either Part	у.					

CATE	EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svo Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Manual Svc Order vs. Electronic-	- al Charge Manual r Svc Order vs.
											curring						
							Rec	Nonred			nnect				ATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																+	+
	The "7eee"						- 7 T-	:	hiaallu Daawaa	and UNIT	Zawa Daaiss		Camtral Offi				
DEDATIO		shown in the sections for stand-alone loops or loops as part of a combination	n reters t	to Geo	graphically Deaver	aged UN	E Zones. 10	view Geograp	nically Deaver	aged UNE	zone Desigr	lations by t	Central Offic	e, refer to in	iternet webs	ite:	
PERAIIC		Electronic Service Order: CLEC-1 should contact its contract negotiator if it pr	ofore the	ctato	enacific alactronic	corvico	ordering cha	race se ordor	nd by the State	Commiss	ione Thora	loctronic e	arvice orde	ring charge	currently cor	stained in th	nie rato
		Any element that can be ordered electronically will be billed according to the S															
		Manual Service Order Charge, Disconnect Only (MS)	CINECIE	le note	u in una category.	SOMAN	refer to belie	1.97		ocai oraci	ng (DDIC-EC) to actern	inic ii a pro	duct can be	Drucicu cic	tromcany. 1	TOT THOSE
		Electronic OSS Charge, per LSR, submitted via BST's OSS interactive interfaces				OOW UV		1.57								+	+
		(Regional)				SOMEC		3.50									
JNBUNDLI		GE ACCESS LOOP															
		ALOG VOICE GRADE LOOP															T
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1		1	UEANL	UEAL2	12.03	37.92	17.55	23.48	5.25		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL	UEAL2	16.87	37.92	17.55	23.48	5.25		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	25.68	37.92	17.55	23.48	5.25		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 1-Zone 4		4	UEANL	UEAL2	43.85	37.92	17.55	23.48	5.25		15.75				
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		34.36									
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.97									
		Engineering Information Document (EI)			UEANL			13.51	13.51								
		Manual Order Coordination for UVL-SL1s (per loop)*			UEANL	UEAMC		50.29	50.29								
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *			UEANL	OCOSL		45.27	45.27								
		bundled COPPER LOOP														↓	┷
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1	!	1	UEQ	UEQ2X	11.01	36.53	16.16	22.66	4.42		15.75				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	!		UEQ	UEQ2X	11.51	36.53	16.16	22.66	4.42		15.75				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3	!		UEQ	UEQ2X	11.57	36.53	16.16	22.66	4.42		15.75				
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 4		4	UEQ	UEQ2X	13.10	36.53	16.16	22.66	4.42		15.75				
		Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loop)			UEQ	USBMC		45.27	45.27							+	+
		Engineering Information Document			UEQ UEQ	URET1		13.51 34.36	13.51							+	+
		Loop Testing - Basic 1st Half Hour														+	+
INDUNDU		Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.97									+
INDUNDL		ALOG VOICE GRADE LOOP															+
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1	1	1	UEPSR UEPSB	UEALS	12.03	37.92	17.55	23.48	5.25			25.52	11.34	16.06	16.06
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 1	i		UEPSR UEPSB	UEABS	12.03	37.92	17.55	23.48	5.25			25.52	11.34		
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	i	2	UEPSR UEPSB	UEALS,	16.87	37.92	17.55	23.48	5.25			25.52	11.34		
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	i i	-	UEPSR UEPSB	UEABS	16.87	37.92	17.55	23.48	5.25			25.52	11.34		
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3	i	3	UEPSR UEPSB	UEALS,	25.68	37.92	17.55	23.48	5.25			25.52	11.34		
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3	i		UEPSR UEPSB	UEABS	25.68	37.92	17.55	23.48	5.25			25.52	11.34		16.06
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 4	i	4	UEPSR UEPSB	UEALS,	43.85	37.92	17.55	23.48	5.25			25.52	11.34	16.06	
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 4	ı	1	UEPSR UEPSB	UEABS	43.85	37.92	17.55	23.48	5.25			25.52	11.34		
<u>INBU</u> NDLI	ED EXCHANG	GE ACCESS LOOP		L													
	2-WIRE ANA	ALOG VOICE GRADE LOOP				Ľ											
		CLEC to CLEC Conversion Charge without outside dispatch (UVL-SL1)			UEANL	UREWO		37.92	17.55				15.75				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start				1											
		Signaling - Zone 1		1	UEA	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start															
		Signaling - Zone 2	ļ	2	UEA	UEAL2	18.75	105.96	68.28	52.82	10.37	-	15.75				+
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start	1	3	UEA	UEAL2	21.05	105.96	66.28	52.62	10.37	1	15./5			+	+
		Signaling - Zone 4		4	UEA	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75			1	1
		Order Coordination for Specified Conversion Time (per LSR)		7	UEA	OCOSL	40.72	18.19	00.20	02.02	10.07		10.70			+	+
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -			OZ/T	00002		10.10								+	1
		Zone 1		1	UEA	UEAR2	13.89	105.96	68.28	52.82	10.37		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -															1
		Zone 2		2	UEA	UEAR2	18.75	105.96	68.28	52.82	10.37		15.75				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -			l	l											
		Zone 3		3	UEA	UEAR2	27.55	105.96	68.28	52.82	10.37		15.75				4
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -				UE 454		,		F						1	1
		Zone 4		4	UEA	UEAR2	45.72	105.96	68.28	52.82	10.37	-	15.75		1	 	
		Order Coordination for Specified Conversion Time (per LSR)	ļ	-	UEA	OCOSL		18.19				-					+
		CLEC to CLEC Conversion Charge without outside dispatch	ļ	-	UEA	UREWO		105.96	38.21			-	15.75				+
		ALOG VOICE GRADE LOOP	-	1	LIFA	LIEAL 1	07.47	400.07	04.50	00.00	4404	1	45.75		1	+	+
		4-Wire Analog Voice Grade Loop - Zone 1 4-Wire Analog Voice Grade Loop - Zone 2	-		UEA	UEAL4	27.47	132.27	94.59	60.68	14.64	1	15.75		1	+	+
			1	2	UEA	UEAL4	38.26	132.27	94.59	60.68	14.64	1	15.75	l	1		
		4-Wire Analog Voice Grade Loop - Zone 3			UEA	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually		I Charge - Manual	al Charge - Manual Svc Order vs. Electronic-	Increment - al Charge - Manual - Svc Order - vs Electronic- Disc Add'I
						Dee	Manne			curring			000 D	ATEC (C)		
						Rec	First	urring Add'l	First	nnect Add'l	COMEC	SOMAN	SOMAN	ATES (\$) SOMAN	SOMAN	SOMAN
							11131	Auu i	11151	Auu	JOIVILO	JOWAN	JONAN	JONAN	JOWAN	JONAN
																†
Orde	er Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.19									
	GITAL GRADE LOOP															
2-Wi	ire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	21.01	117.61	79.92	52.82	10.37		15.75				
2-Wi	ire ISDN Digital Grade Loop - Zone 2		2	UDN	U1L2X	27.59	117.61	79.92	52.82	10.37		15.75				
	ire ISDN Digital Grade Loop - Zone 3 ire ISDN Digital Grade Loop - Zone 4			UDN	U1L2X U1L2X	37.34 59.18	117.61 117.61	79.92 79.92	52.82 52.82	10.37 10.37		15.75 15.75				
	er Coordination For Specified Conversion Time (per LSR)		4	UDN	OCOSL	59.18	18.19	79.92	52.82	10.37		15.75				+
	C to CLEC Conversion Charge without outside dispatch		1	UDN	UREWO)	117.61	33.03				15.75				+
	al Digital Channel (UDC) COMPATIBLE LOOP							00.00								
2-Wi	'ire Universal Digital Channel (UDC) Compatible Loop - Zone 1			UDC	UDC2X	21.01	117.61	79.92	52.82	10.37		15.75				
	ire Universal Digital Channel (UDC) Compatible Loop - Zone 2			UDC	UDC2X	27.59	117.61	79.92	52.82	10.37		15.75				
	ire Universal Digital Channel (UDC) Compatible Loop - Zone 3			UDC	UDC2X	37.34	117.61	79.92	52.82	10.37		15.75				
	fire Universal Digital Channel (UDC) Compatible Loop - Zone 4		4	UDC	UDC2X		117.61	79.92	52.82	10.37		15.75				-
	C to CLEC Conversion Charge without outside dispatch * IETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP			UDC	UREWO	,	117.61	33.03				15.75				+
	ire Unbundled ADSL Loop including manual service inquiry & facility reservation															+
- Zon			1	UAL	UAL2X	11.11	121.27	70.81	50.38	7.93		15.75				
2 Wi	ire Unbundled ADSL Loop including manual service inquiry & facility reservation															
	ne 2		2	UAL	UAL2X	11.47	121.27	70.81	50.38	7.93		15.75				
	ire Unbundled ADSL Loop including manual service inquiry & facility reservation ne 3		3	UAL	UAL2X	11.74	121 27	70.81	E0 20	7.93		15.75				
	rie 3 ire Unbundled ADSL Loop including manual service inquiry & facility reservation		3	UAL	UALZX	11.74	121.27	70.81	50.38	7.93		15.75				+
	ne 4		4	UAL	UAL2X	12.69	121.27	70.81	50.38	7.93		15.75				
Orde	er Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.19									
	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton -															
Zone			1	UAL	UAL2W	11.11	96.15	58.03	50.38	7.93		15.75				
Zone	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton -		2	UAL	UAL2W	11.47	96.15	58.03	50.38	7.93		15.75				
	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton -		-	O/ IL	ONLEVV	11.47	30.10	30.00	50.50	7.55		10.70				+
Zone	e 3		3	UAL	UAL2W	11.74	96.15	58.03	50.38	7.93		15.75				
	ire Unbundled ADSL Loop without manual service inquiry & facility reservaton -															
Zone			4	UAL	UAL2W	12.69	96.15 18.19	58.03	50.38	7.93		15.75				+
	er Coordination for Specified Conversion Time (per LSR) C to CLEC Conversion Charge without outside dispatch			UAL	OCOSL UREWO)	96.15	29.28				15.75				+
2-WIRE HIGH BIT	T RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP			O/ IL	OILLIVE	í	30.10	25.20				10.70				+
	ire Unbundled HDSL Loop including manual service inquiry & facility reservation															
	ne 1		1	UHL	UHL2X	8.75	129.98	79.52	50.38	7.93		15.75				
2 Wi	ire Unbundled HDSL Loop including manual service inquiry & facility reservation		2	UHL	UHL2X	0.00	400.00	79.52	50.00	7.93		15.75				
	rie 2 ire Unbundled HDSL Loop including manual service inquiry & facility reservation			UHL	UHLZX	9.22	129.98	79.52	50.38	7.93		15.75				+
	ne 3		3	UHL	UHL2X	9.87	129.98	79.52	50.38	7.93		15.75				
	ire Unbundled HDSL Loop including manual service inquiry & facility reservation															
- Zon			4	UHL	UHL2X	10.46	129.98	79.52	50.38	7.93		15.75				
Orde	er Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.19									
	ire Unbundled HDSL Loop without manual service inquiry and facility ervation - Zone 1		1	UHL	UHL2W	8.75	104.86	66.74	50.38	7.93		15.75				
	ire Unbundled HDSL Loop without manual service inquiry and facility		<u>'</u>	OFF	OI ILZVV	0.73	104.00	00.74	30.30	1.55		13.73				+
reser	ervation - Zone 2		2	UHL	UHL2W	9.22	104.86	66.74	50.38	7.93		15.75				
	ire Unbundled HDSL Loop without manual service inquiry and facility															
	ervation - Zone 3		3	UHL	UHL2W	9.87	104.86	66.74	50.38	7.93		15.75				
	ire Unbundled HDSL Loop without manual service inquiry and facility ervation - Zone 4		4	UHL	UHL2W	10.46	104.86	66.74	50.38	7.93		15.75				
	er Coordination for Specified Conversion Time (per LSR)		7	UHL	OCOSL	10.40	18.19	00.74	50.50	7.55		10.70				+
	C to CLEC Conversion Charge without outside dispatch			UHL	UREWO)	104.86	29.28				15.75				1
	T RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP															
	ire Unbundled HDSL Loop including manual service inquiry and facility		1		LILII AV	40.70	450 74	100.00	E0 70	40.00		45.75				
	ervation - Zone 1 Fire Unbundled HDSL Loop including manual service inquiry and facility	-	-	UHL	UHL4X	13.78	158.74	108.28	56.72	10.68		15.75				+
	ervation - Zone 2		2	UHL	UHL4X	13.43	158.74	108.28	56.72	10.68		15.75				
	ire Unbundled HDSL Loop including manual service inquiry and facility															
	ervation - Zone 3		3	UHL	UHL4X	15.59	158.74	108.28	56.72	10.68		15.75				1
	lire Unbundled HDSL Loop including manual service inquiry and facility ervation - Zone 4		4	UHL	UHL4X	14.46	158.74	108.28	56.72	10.68		15.75				
	er Coordination for Specified Conversion Time (per LSR)		4	UHL	OCOSL	14.40	158.74	106.28	30.72	10.08		15.75				+
	ire Unbundled HDSL Loop without manual service inquiry and facility						.0.10									1
	ervation - Zone 1	1	1	UHL	UHL4W	13.78	133.62	95.50	56.72	10.68		15.75				

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	vs. Electronic-	al Charge Manual Svc Order vs.
						_			Nonred					(4)		
						Rec	Nonrec		Disco		SOMEC	SOMAN		ATES (\$) SOMAN	SOMAN	SOMAN
							First	Add'l	First	Add'l	SOWIEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
			l												+	
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility														1	
	reservation - Zone 2		2	UHL	UHL4W	13.43	133.62	95.50	56.72	10.68		15.75				
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility															
	reservation - Zone 3 4-Wire Unbundled HDSL Loop without manual service inquiry and facility		3	UHL	UHL4W	15.59	133.62	95.50	56.72	10.68		15.75				
	reservation - Zone 4		4	UHL	UHL4W	14.46	133.62	95.50	56.72	10.68		15.75				
	Order Coordination for Specified Conversion Time (per LSR)		7	UHL	OCOSL	14.40	18.19	33.30	30.72	10.00		10.70			+	
	CLEC to CLEC Conversion Charge without outside dispatch			UHL	UREWO	1	104.86	29.28				15.75				
	1 DIGITAL LOOP															
	4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
	4-Wire DS1 Digital Loop - Zone 2	1	2		USLXX	129.38	253.93	158.45	46.10	12.07		15.75				
	4-Wire DS1 Digital Loop - Zone 3			USL	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				+
	4-Wire DS1 Digital Loop - Zone 4 Order Coordination for Specified Conversion Time (per LSP)	-		USL	USLXX	458.46	253.93 18.19	158.45	46.10	12.07	-	15.75			+	+
	Order Coordination for Specified Conversion Time (per LSR) CLEC to CLEC Conversion Charge without outside dispatch	1		USL	UREWO		130.03	39.98				15.75			+	+
	2, 56 OR 64 KBPS DIGITAL GRADE LOOP			302	OILLAND		100.00	55.50				10.70			\vdash	-
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	27.44	126.53	88.85	60.68	14.64		15.75			T	1
	4 Wire Unbundled Digital 19.2 Kbps		2		UDL19	34.55	126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital 19.2 Kbps		3		UDL19	40.76	126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	32.25	126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3			UDL UDL	UDL56 UDL56	40.76	126.53 126.53	88.85	60.68	14.64		15.75				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 4 Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL	32.25	126.53	88.85	60.68	14.64		15.75			+	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1			UDL	UDL64	27.44	126.53	88.85	60.68	14.64		15.75			+	+
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	34.55	126.53	88.85	60.68	14.64		15.75			+	
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3		3		UDL64	40.76	126.53	88.85	60.68	14.64		15.75				+
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 4			UDL	UDL64	32.25	126.53	88.85	60.68	14.64		15.75				
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		18.19									
	CLEC to CLEC Conversion Charge without outside dispatch			UDL	UREWO	1	126.53	38.62				15.75				
	oundled COPPER LOOP															
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1		1	UCL	UCLPB	11.11	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		'	OCL	OCLID	11.11	120.54	03.07	30.30	1.55		13.73			+	+
	reservation - Zone 2		2	UCL	UCLPB	11.47	120.34	69.87	50.38	7.93		15.75				
	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility														1	
	reservation - Zone 3		3	UCL	UCLPB	11.74	120.34	69.87	50.38	7.93		15.75				
	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility				LIOL DD	40.00	400.04	00.07	50.00	7.00		45.75				
	reservation - Zone 4 Order Coordination for Unbundled Copper Loops (per loop)		4	UCL	UCLPB UCLMC	12.69	120.34 8.20	69.87 8.20	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility			UCL	UCLIVIC		6.20	0.20							+	+
	reservation - Zone 1		1	UCL	UCLPW	11.11	95.21	57.09	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility															
	reservation - Zone 2	1	2	UCL	UCLPW	11.47	95.21	57.09	50.38	7.93	1	15.75				+
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility		3	UCL	UCLPW	11.74	05.04	57.09	50.38	7.93		15.75			1	1
	reservation - Zone 3 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility		3	UCL	UCLPW	11.74	95.21	57.09	50.38	7.93		15.75			+	
	reservation - Zone 4		4	UCL	UCLPW	12.69	95.21	57.09	50.38	7.93		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20							1	+
	2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility														1	
	reservation - Zone 1		1	UCL	UCL2L	29.29	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility		_	LICI	1101.01	40.40	400.01	00.0-	F0 00	7.00		45.7-				
	reservation - Zone 2 2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	-	2	UCL	UCL2L	43.46	120.34	69.87	50.38	7.93	-	15.75			+	+
	reservation - Zone 3		3	UCL	UCL2L	64.44	120.34	69.87	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	1	Ť		JOLLE	5 4	120.04	55.57	00.00			.0.70			—	1
	reservation - Zone 4	<u> </u>	4	UCL	UCL2L	87.60	120.34	69.87	50.38	7.93	<u> </u>	15.75	<u> </u>			
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20		-						
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
	reservation - Zone 1	-	1	UCL	UCL2W	29.29	95.21	57.09	50.38	7.93		15.75				1
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 2		2	UCL	UCL2W	43.46	95.21	57.09	50.38	7.93		15.75				
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility	-	-	UUL	UCLZVV	43.40	90.ZI	57.09	30.38	1.93		15.75			+	+
	reservation - Zone 3		3	UCL	UCL2W	64.44	95.21	57.09	50.38	7.93		15.75			1	1
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
	reservation - Zone 4	1	4	UCL	UCL2W	87.60	95.21	57.09	50.38	7.93	1	15.75	1	1	1	1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	vs. Electronic-	
										curring						
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Unbundled Copper Loops (per loop)		1	UCL	UCLMC		8.20	8.20							+	+
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		95.21	31.36				15.75				-
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-ND)			UEQ	UREWO		36.53	16.16				15.75				
4-WIRE CO	PPER LOOP															
	4-Wire Copper Loop/Short - including manual service inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4S	17.30	144.68	94.22	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - including manual service inquiry and facility		2	UCL	UCL4S	18.84	144.68	94.22	56.72	40.00		15.75				
	reservation - Zone 2 4-Wire Copper Loop/Short - including manual service inquiry and facility			UCL	UCL45	10.04	144.08	94.22	50.72	10.68		15.75				
	reservation - Zone 3		3	UCL	UCL4S	21.33	144.68	94.22	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - including manual service inquiry and facility		Ĭ	002	00210	21.00	111.00	O I.LL	00.12	10.00		10.70				-
	reservation - Zone 4		4	UCL	UCL4S	21.33	144.68	94.22	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -															
	Zone 1		1	UCL	UCL4W	17.30	119.56	81.44	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation - Zone 2		2	UCL	LICL AW	40.04	440.50	04.44	FC 70	40.00		45.75				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -			UCL	UCL4W	18.84	119.56	81.44	56.72	10.68		15.75				
	Zone 3		3	UCL	UCL4W	21.33	119.56	81.44	56.72	10.68		15.75				
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -			002	OOLIII	21.00	110.00	01.11	00.72	10.00		10.70				
	Zone 4		4	UCL	UCL4W	21.33	119.56	81.44	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4L	54.72	144.68	94.22	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL4L	97.47	144.68	94.22	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility			UCL	UCL4L	97.47	144.68	94.22	56.72	10.08		15.75				
	reservation - Zone 3		3	UCL	UCL4L	106.06	144.68	94.22	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 4		4	UCL	UCL4L	106.06	144.68	94.22	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4O	54.72	119.56	81.44	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility reservation - Zone 2		2	UCL	UCL4O	97.47	119.56	81.44	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility			UCL	UCL4U	91.41	119.50	01.44	30.72	10.00		15.75				
	reservation - Zone 3		3	UCL	UCL4O	106.06	119.56	81.44	56.72	10.68		15.75				
	4-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
	reservation - Zone 4		4	UCL	UCL4O	106.06	119.56	81.44	56.72	10.68		15.75				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.20	8.20								
	CLEC to CLEC Conversion Charge without outside dispatch (UCL-Des)			UCL	UREWO		95.21	31.36				15.75				
LOOP MODIFICATION	Linbundled Lean Medification, Removal of Land Calls, 2 Wiss asis land the con-	-	<u> </u>	UAL, UHL, UCL,												
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft			UEQ, ULS	ULM2L		32.57	32.57								
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft			UCL, ULS	ULM2G		171.49	171.49							+	
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to			OOL, OLO	OLIVIZO		171.43	171.43								1
	18K ft			UHL, UCL	ULM4L		32.57	32.57								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k															
	ft			UCL	ULM4G		171.49	171.49								
				UAL, UHL, UCL,												
CUD LOODS	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop	1		UEQ, UEF, ULS	ULMBT		32.59	32.59								
SUB-LOOPS Sub-Loop E	Distribution	-			1						1					
3ub-100b F	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up	1	 	UEANL	USBSA		259.69				1	15.75			+	
	Sub-Loop - Per Cross Box Location - CEEC Feeder Facility Set-Op Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up	i		UEANL	USBSB		239.69	+				15.75				†
	Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up	i		UEANL	USBSC		178.47					15.75				
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up	ı		UEANL	USBSD		56.39					15.75				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1			UEANL	USBN2	7.15	66.18	31.14	45.36	6.71		15.75				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	ı		UEANL	USBN2	9.51	66.18	31.14	45.36	6.71		15.75				
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN2	12.45	66.18	31.14	45.36	6.71	1	15.75			<u> </u>	
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 4	<u> </u>	4	UEANL	USBN2	18.26	66.18	31.14	45.36	6.71	1	15.75				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	1	-	UEANL	USBMC	7.00	45.27	45.27	E4.07	0.05	1	45.75			-	
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2	-	2	UEANL UEANL	USBN4 USBN4	7.30 13.92	79.49 79.49	44.45 44.45	51.27 51.27	9.35 9.35		15.75 15.75				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2 Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3	-	3	UEANL	USBN4	16.73	79.49	44.45 44.45	51.27	9.35		15.75			+	+
		1									 			-	+	+
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 4		4	UEANL	USBN4	16.73	79.49	44.45	51.27	9.35		15.75				

CATE	EGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	vs. Electronic-	
					1					Nonre							
							Rec	Nonrec		Disco					ATES (\$)		
				oxdot				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
				igwdown													
		Sub-Loop 2-Wire Intrabuilding Network Cable (INC)		+	UEANL	USBR2	2.29	53.32	18.28	45.36	6.71		15.75				+
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair		+-		USBMC	2.29	45.27	45.27	40.00	0.71		13.73				
		Sub-Loop 4-Wire Intrabuilding Network Cable (INC)		\vdash		USBR4	4.40	59.60	24.55	51.27	9.35		15.75				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair		\vdash		USBMC		45.27	45.27							·	1
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		1		UCS2X	6.06	66.18	31.14	45.36	6.71		15.75				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	1			UCS2X	7.09	66.18	31.14	45.36	6.71		15.75				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	1			UCS2X	8.16	66.18	31.14	45.36	6.71		15.75				
		2 Wire Copper Unbundled Sub-Loop Distribution - Zone 4				UCS2X	9.90	66.18	31.14	45.36	6.71		15.75				<u> </u>
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair				USBMC	5.40	45.27	45.27	54.07	0.05		45.75				
-		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1 4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2				UCS4X UCS4X	5.10 9.11	79.49 79.49	44.45 44.45	51.27 51.27	9.35 9.35		15.75				
		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	i			UCS4X	14.00	79.49	44.45	51.27	9.35		15.75 15.75				+
-		4 Wire Copper Unbundled Sub-Loop Distribution - Zone 4				UCS4X	14.00	79.49	44.45	51.27	9.35		15.75				
		Order Coordination for Unbundled Sub-Loops, per sub-loop pair		H		USBMC	1 1.00	45.27	45.27	01.21	0.00		10.10				
	Unbundled	Sub-Loop Modification	t														
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per 2-W PR			UEF	ULM2X		176.80	5.13				15.75				
		Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		176.80	5.13				15.75				
	Unbundlad	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per PR unloaded Network Terminating Wire (UNTW)		<u> </u>	UEF	ULM4T		279.81	6.15				15.75				ļ!
	Unbunalea	Unbundled Network Terminating Wire (UNTW) per Pair	-	-	UENTW	UENPP	0.34	30.55					15.75	-			+
	Network Int	erface Device (NID)		+-	UEINTW	UEINFF	0.34	30.33					13.73				+
	Network int	Network Interface Device (NID) - 1-2 lines		+-	UENTW	UND12		43.84	28.90				15.75				
		Network Interface Device (NID) - 1-6 lines		\vdash		UND16		65.30	50.36				15.75				
		Network Interface Device Cross Connect - 2 W		\vdash		UNDC2		5.94	5.94				15.75			·	1
		Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.94	5.94				15.75				
SUB-LOOP																	
	Sub-Loop I	Feeder		oxdot	LIEA												
					UEA, UDN,UCL,UDL,UD											,	1
		USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up			IC	USBFW		259.69					15.75				
		, , , , , , , , , , , , , , , , , , , ,			UEA,												
					UDN,UCL,UDL,UD												
		USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up			С	USBFX		22.77	22.77				15.75				
		USL Feeder DS1 Set-up at DSX location, per DS1 termination				USBFZ		534.46	11.30				15.75				<u> </u>
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1				USBFA	7.98	93.23	56.50	54.45	13.51		15.75				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3				USBFA USBFA	10.39 16.11	93.23 93.23	56.50 56.50	54.45 54.45	13.51 13.51		15.75 15.75				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start Loop, Voice Grade - Zone		-3	UEA	USBFA	10.11	93.23	36.30	34.45	13.31		15.75				+
		4		4	UEA	USBFA	28.37	93.23	56.50	54.45	13.51		15.75				
	1	Order Coordination for Specified Conversion Time, per LSR	1			OCOSL		18.19					1				
		Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1			UEA	USBFB	7.98	93.23	56.50	54.45	13.51		15.75				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2				USBFB	10.39	93.23	56.50	54.45	13.51		15.75				
		Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3				USBFB	16.11	93.23	56.50	54.45	13.51		15.75			L	 '
		Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 4	├			USBFB	28.37	93.23	56.50	54.45	13.51		15.75	ļ			<u> </u>
		Order Coordination for Specified Time Conversion, per LSR	₽			OCOSL	7.00	18.19	50.50		40.51		45	ļ			 '
-		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2				USBFC USBFC	7.98 10.39	93.23 93.23	56.50 56.50	54.45 54.45	13.51 13.51		15.75 15.75				+
		Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3				USBFC	16.11	93.23	56.50	54.45	13.51		15.75				+
-	+	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 3 Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 4				USBFC	28.37	93.23	56.50	54.45	13.51		15.75				+
		Order Coordination For Specified Conversion Time, per LSR	†	H		OCOSL	20.01	18.19	33.50	00	10.01		.0.70				
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1	t	1		USBFD	21.69	107.71	70.03	63.68	17.64		15.75				
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2				USBFD	26.06	107.71	70.03	63.68	17.64		15.75				
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3		3	UEA	USBFD	34.77	107.71	70.03	63.68	17.64		15.75				
		Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 4				USBFD	34.77	107.71	70.03	63.68	17.64		15.75				
		Order Coordination For Specified Conversion Time, Per LSR	 			OCOSL		18.19						ļ		L	
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1	├			USBFE	21.69	107.71	70.03	63.68	17.64		15.75				<u> </u>
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	—			USBFE	26.06	107.71	70.03	63.68	17.64		15.75			 '	 '
		Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3	├ ──			USBFE	34.77	107.71	70.03	63.68	17.64		15.75			 	+
	1	Sub-Loop Feeder - Per 4-Wire Analog Voice Grade Loop-Start Loop - Zone 4 Order Coordination For Specified Conversion Time, Per LSR	 	4		USBFE OCOSL	34.77	107.71 18.19	70.03	63.68	17.64		15.75	 	 		+'
	1		+	+								l				ļ	+
		IUnbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1 1 1	'UDN -	USBFF	14 60	106.46	68 78	55 58	131 13		15 /5				
		Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1 Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2	-		UDN UDN	USBFF	14.60 18.78	106.46 106.46	68.78 68.78	55.58 55.58	131.13 131.13		15.75 15.75				

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	al Charge - Manual Svc Order vs. Electronic-	Manual
						_		_		curring						
						Rec	Nonred			onnect	001150	0011411		ATES (\$)	001111	001111
		-					First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
																_
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 4		4 (IDN	USBFF	41.41	106.46	68.78	55.58	131.13		15.75				
	Order Coordination For Specified Conversion Time, Per LSR			JDN	OCOSL		18.19	00.70	00.00	101110		10.10				1
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			JDC	USBFS	14.60	106.46	68.78	55.58	131.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2 (USBFS	18.78	106.46	68.78	55.58	131.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		3 l	JDC	USBFS	25.47	106.46	68.78	55.58	131.13		15.75				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		4 l	JDC	USBFS	41.41	106.46	68.78	55.58	131.13		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1		1 l		USBFG	55.19	101.97	64.29	63.68	17.64		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2 l		USBFG	100.03	101.97	64.29	63.68	17.64		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3 (USBFG	183.66	101.97	64.29	63.68	17.64		15.75				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 4		4 l		USBFG	430.04	101.97	64.29	63.68	17.64		15.75				
	Order Coordination For Specified Conversion Time, Per LSR	+		JSL	OCOSL	5.00	18.19	40.50	50.44	40.70		45.75			1	
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 1	+	1 l		USBFH	5.88 5.21	84.27 84.27	46.59	53.14 53.14	10.70		15.75 15.75			1	
 	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2	1	2 l	JCL ICI	USBFH	4.40	84.27	46.59 46.59	53.14	10.70 10.70		15.75				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3	+	4 1		USBFH	3.63	84.27	46.59	53.14	10.70		15.75	 			
 	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 4 Order Coordination For Specified Conversion Time, per LSR	1		JCL JCL	OCOSL	3.03	18.19	40.59	55.14	10.70		10.75			1	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1		1 1		USBFJ	13.49	101.58	63.90	59.71	13.67		15.75				
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2		2 1		USBFJ	10.96	101.58	63.90	59.71	13.67		15.75				1
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3		3 (USBFJ	8.59	101.58	63.90	59.71	13.67		15.75				1
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 4		4 (USBFJ	8.59	101.58	63.90	59.71	13.67		15.75				
	Order Coordination For Specified Conversion Time, per LSR			JCL	OCOSL	0.00	18.19									
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1 (JDL	USBFN	22.89	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			JDL	USBFN	25.11	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			JDL	USBFN	30.84	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop			JDL	USBFN	41.05	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1			JDL	USBFO	22.89	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2		2 1		USBFO	25.11	101.97	64.29	63.68	17.64		15.75				_
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3		3 (USBFO	30.84	101.97	64.29	63.68	17.64		15.75				_
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 4		4 (USBFO	41.05	101.97	64.29	63.68	17.64		15.75				-
	Order Coordination For Specified Time Conversion, per LSR	+	1 (JDL	USBFP	22.89	18.19 101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1 Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2		2 (USBFP	25.11	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2 Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3 1		USBFP	30.84	101.97	64.29	63.68	17.64		15.75				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 4		4 (USBFP	41.05	101.97	64.29	63.68	17.64		15.75				
	Order Coordination For Specified Conversion Time, per LSR			JDL	OCOSL		18.19		00.00							
SUB-LOOPS																
Sub-Loc	op Feeder															
UNBUNDLED LOOP	CONCENTRATION															
	Unbundled Loop Concentration - System A (TR008)			JLC	UCT8A		327.30	327.30				15.75				
	Unbundled Loop Concentration - System B (TR008)			JLC	UCT8B	47.56	136.37	136.37				15.75				
	Unbundled Loop Concentration - System A (TR303)	1		JLC	UCT3A	397.35	327.30	327.30				15.75			1	
	Unbundled Loop Concentration - System B (TR303)	1		JLC	UCT3B	80.15	136.37	136.37	,			15.75				
	Unbundled Loop Concentration - DS1 Loop Interface Card	1		JLC	UCTCO	4.52 7.17	63.65	46.34	17.31	4.85		15.75				1
	Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)			JDN JDC	ULCC1	7.17	10.60	10.54	5.56	5.53		15.75 15.75				
	Unbundled Loop Concentration - UDC Loop Interface (Brite Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop	+	1	JDC	ULCCU	7.17	10.60	10.54	5.56	5.53		15.75	1			
	Interface (POTS Card)			JEA	ULCC2	1.80	10.60	10.54	5.56	5.53		15.75				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface	1		<i></i> , ·	32002	1.00	10.00	10.04	0.00	5.55		10.75			1	t
	(SPOTS Card)	1	l lu	JEA	ULCCR	10.66	10.60	10.54	5.56	5.53		15.75				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card)		l	JEA	ULCC4	6.36	10.60	10.54	5.56	5.53		15.75				
	Unbundled Loop Concentration - TEST CIRCUIT Card		l	JLC	UCTTC	31.07	10.60	10.54	5.56	5.53		15.75				
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface			JDL	ULCC7	9.42	10.60	10.54	5.56	5.53		15.75				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface			JDL	ULCC5	9.42	10.60	10.54	5.56	5.53		15.75				<u> </u>
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interface	1		JDL	ULCC6	9.42	10.60	10.54	5.56	5.53		15.75			1	
UNE OTHER, PROV	ISIONING ONLY - NO RATE	1	 	IENTA/	LINDDY							1				
	NID - Dispatch and Service Order for NID installation	1		JENTW JENTW	UNDBX							-				
	UNTW Circuit Id Establishment, Provisioning Only - No Rate			JEANL,UEF,UEQ												_
	Unbundled Contract Name, Provisioning Only - No Rate			JENTW	UNECN											
UNE OTHER, PROV	ISIONING ONLY - NO RATE	1			3.1.2014											
,	1000	1		JAL,UCL,UDC,UE												
				_,UDN,UEA,UHL,U	J											
	Unbundled Contact Name, Provisioning Only - no rate		l l	_C	UNECN	0.00	0.00									<u> </u>
		1	T	JEA,UDN,UCL,UE												
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate			ن	USBFQ	0.00	0.00			L		1	l	L	<u> </u>	1

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		R	ATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	al Charge - Manual Svc Order vs. Electronic-	al Charge Manual Svc Order vs.
										Nonrecu							
							Rec	Nonrecu		Disconi					ATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					UEA,USL,UCL,UD												
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate			L	USBFR	0.00	0.00									
		Unbundled DS1 Loop - Superframe Format Option - no rate			USL	CCOSF	0.00	0.00									
		Unbundled DS1 Loop - Expanded Superframe Format option - no rate			USL	CCOEF	0.00	0.00									
CH CARA		NDLED LOCAL LOOP			002	OOOL.	0.00	0.00									
		onth minimum billing period															
					UE3	1L5ND	11.20										
		High Capacity Unbundled Local Loop - DS3 - Per Mile per month		-													
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per month			UE3	UE3PX	326.15	454.13	265.47	123.23	86.19		15.75				
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per month			UDLSX	1L5ND	11.20										
		High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month			UDLSX	UDLS1	338.55	454.13	265.47	123.23	86.19		15.75				
OOP MAK																	
	1	Loop Makeup - Preordering Without Reservation, per working or spare facility	1	1								1	1				
	1	queried (Manual).		1	UMK	UMKLW		24.12	24.12				1				
	1	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).			UMK	UMKLP		25.58	25.58				1				
		Loop MakeupWith or Without Reservation, per working or spare facility queried		t			-	20.00	20.00				1				
	1	(Mechanized)		1	UMK	PSUMK		0.6652	0.6652				1				
GH FREO	UENCY SPE			1	t	. COIVIN	-	0.0002	0.0002				1				
		CENTRAL OFFICE BASED	 	1		 							1				
			— —	1	111.0	LII CDA	400.07	189.89	0.00	470.44	0.00		0.00				
		Line Sharing Splitter, per System 96 Line Capacity	!		ULS	ULSDA	186.67			178.41							
		Line Sharing Splitter, per System 24 Line Capacity	ı		ULS	ULSDB	46.67	189.89	0.00	178.41	0.00		0.00				
		Line Sharing Splitter, Per System, 8 Line Capacity	I		ULS	ULSD8	15.55	189.89	0.00	178.41	0.00		0.00				
	END USER	ORDERING-CENTRAL OFFICE BASED-HIGH FREQUENCY SPECTRUM AKA LIN	E SHARII	NG													
		Line Sharing - per Line Activation	- 1		ULS	ULSDC	0.61	18.62	10.66	10.04	4.93			25.52	11.34	16.06	16.06
		Line Sharing - per Subsequent Activity per Line Rearrangement	1		ULS	ULSDS		16.48	8.24					25.52	11.34		
		Line Splitting - per line activation DLEC owned splitter	t i		UEPSR UEPSB	UREOS	0.61	10.10	0.2.					20.02	11.01		
		Line Splitting - per line activation BST owned - physical	i		UEPSR UEPSB	UREBP	0.639	18.62	10.66	10.04	4.93						
			i														
		Line Splitting - per line activation BST owned - virtual	- 1		UEPSR UEPSB	UREBV	0.637	18.62	10.66	10.04	4.93						
IBUNDLE	D TRANSPO	DRI															
		CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE															
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per			U1TVX	1L5XX	0.0098										
,		Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility															
		Termination per month			U1TVX	U1TV2	22.52	40.77	27.57	17.26	7.11		15.75				
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per															
		Mile per month			U1TVX	1L5XX	0.0098										
		Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility															
		Termination per month			U1TVX	U1TR2	22.52	40.77	27.57	17.26	7.11		15.75				
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per			011177	O. IIIL	LL.OL		27.07	20			10.10				
		month			U1TVX	1L5XX	0.0098										
					UTTVA	ILSAA	0.0096										
		Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility			U1TVX	U1TV4	40.70	40.77	07.57	47.00	7.44		45.75				
		Termination per month	 	1			19.79	40.77	27.57	17.26	7.11	<u> </u>	15.75				
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			U1TDX	1L5XX	0.0098						ļ				
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month			U1TDX	U1TD5	15.68	40.77	27.57	17.26	7.11		15.75				
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month			U1TDX	1L5XX	0.0098										
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month			U1TDX	U1TD6	15.68	40.77	27.57	17.26	7.11		15.75				
		CE CHANNEL - DEDICATED TRANSPORT - DS1															
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			U1TD1	1L5XX	0.201						1				
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			U1TD1	U1TF1	57.33	89.79	82.28	16.86	14.90		15.75				
			-	t			000	30 3	02.20	. 0.00			.0.70				
					l .		4.76						1			 	
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3			LI4TD3							1				1	
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX		200.27	400.70	CO 00	20.00						
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3 U1TD3	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3 - Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT- STS-1			U1TD3	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT- STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TD3 U1TS1	U1TF3 1L5XX	641.90 4.76										
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month			U1TD3	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT- DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT- STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month			U1TD3 U1TS1	U1TF3 1L5XX	641.90 4.76										
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month	one mor		U1TD3 U1TS1 U1TS1	U1TF3 1L5XX U1TFS	641.90 4.76										
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=	one mor		U1TD3 U1TS1 U1TS1 3 and above=four	U1TF3 1L5XX U1TFS months	641.90 4.76 644.21	280.37	163.70	62.08	60.29		15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month	one mor		U1TD3 U1TS1 U1TS1 U1TS1 3 and above=four ULDVX	U1TF3 1L5XX U1TFS months ULDV2	641.90 4.76 644.21 14.91	280.37	163.70	62.08	60.29		15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Pacility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month	one mor		U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX	U1TF3 1L5XX U1TFS months ULDV2 ULDR2	4.76 644.21 14.91 14.91	280.37 194.22 194.22	163.70 33.36 33.36	62.08 37.79 37.79	3.30 3.30		15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 4-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 4-Wire Voice Grade per month	one mor	nth, DS	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX UNDVX	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDV4	641.90 4.76 644.21 14.91 14.91 15.99	280.37 194.22 194.22 194.66	163.70 33.36 33.36 33.80	62.08 37.79 37.79 38.27	3.30 3.30 3.78		15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per month Local Channel - Dedicated - STS - Per Month Local Channel - Dedicated - STS - Per Month Local Channel - Dedicated - STS - Per Month - Per Month Local Channel - Dedicated - STS - Per Month - Per Month Local Channel - Dedicated - STS - Per Month - Per Month Local Channel - Dedicated - STS - Per Month -	one mor	nth, DS	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX UUNDVX ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1	641.90 4.76 644.21 14.91 14.91 15.99 36.83	280.37 194.22 194.22 194.66 178.50	33.36 33.36 33.80 154.61	37.79 37.79 38.27 22.89	3.30 3.30 3.78 15.74		15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 3-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2	one mor	nth, DS	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX UNDVX ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1 ULDF1	4.76 644.21 14.91 14.91 15.99 36.83 35.99	280.37 194.22 194.22 194.66 178.50 178.50	163.70 33.36 33.36 33.80 154.61 154.61	37.79 37.79 38.27 22.89 22.89	3.30 3.30 3.78 15.74 15.74		15.75 15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 4-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3	one mor	1 2 3	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX ULDVX ULDD1 ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1 ULDF1 ULDF1	4.76 644.21 14.91 14.91 15.99 36.83 35.99 221.63	280.37 194.22 194.22 194.66 178.50 178.50	33.36 33.36 33.80 154.61 154.61	37.79 37.79 38.27 22.89 22.89 22.89	3.30 3.30 3.78 15.74 15.74		15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per month Local Channel - Dedicated - 2-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 4	one mor	1 2 3	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX ULDVX UNDVX ULDD1 ULDD1 ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1	14.91 14.91 15.99 36.83 35.99 221.63	280.37 194.22 194.22 194.66 178.50 178.50	163.70 33.36 33.36 33.80 154.61 154.61	37.79 37.79 38.27 22.89 22.89	3.30 3.30 3.78 15.74 15.74		15.75 15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month Local Channel - Dedicated - 4-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3	one mor	1 2 3	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX ULDVX ULDD1 ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1 ULDF1 ULDF1	4.76 644.21 14.91 14.91 15.99 36.83 35.99 221.63	280.37 194.22 194.22 194.66 178.50 178.50	33.36 33.36 33.80 154.61 154.61	37.79 37.79 38.27 22.89 22.89 22.89	3.30 3.30 3.78 15.74 15.74		15.75 15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT - AL CHANNEL DEDICATED TRANSPORT minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 4-Wire Voice Grade Per month Local Channel - Dedicated - 4-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 4 Local Channel - Dedicated - DS1 per month - Zone 4 Local Channel - Dedicated - DS3 - Per Mile per month	one mor	1 2 3	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX ULDVX UNDVX ULDD1 ULDD1 ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1 1LDF1 1L5NC	14.91 14.91 15.99 36.83 35.99 221.63	280.37 194.22 194.22 194.66 178.50 178.50 178.50	33.36 33.36 33.80 154.61 154.61 154.61	37.79 37.79 37.79 38.27 22.89 22.89 22.89 22.89	3.30 3.30 3.78 15.74 15.74 15.74		15.75 15.75 15.75 15.75 15.75 15.75 15.75				
	INTEROFFIC	CE CHANNEL - DEDICATED TRANSPORT - DS3 Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month CE CHANNEL - DEDICATED TRANSPORT - STS-1 Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month ANNEL - DEDICATED TRANSPORT AL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3= Local Channel - Dedicated - 2-Wire Voice Grade Per Month Local Channel - Dedicated - 2-Wire Voice Grade Per month Local Channel - Dedicated - 2-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1 Local Channel - Dedicated - DS1 per month - Zone 2 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 3 Local Channel - Dedicated - DS1 per month - Zone 4	one mor	1 2 3	U1TD3 U1TS1 U1TS1 3 and above=four ULDVX ULDVX ULDVX ULDD1 ULDD1 ULDD1 ULDD1 ULDD1 ULDD1 ULDD1 ULDD1	U1TF3 1L5XX U1TFS months ULDV2 ULDR2 ULDR2 ULDV4 ULDF1 ULDF1 ULDF1 ULDF1	4.76 644.21 14.91 14.91 15.99 36.83 35.99 221.63 9.66	280.37 194.22 194.22 194.66 178.50 178.50	33.36 33.36 33.80 154.61 154.61	37.79 37.79 38.27 22.89 22.89 22.89	3.30 3.30 3.78 15.74 15.74		15.75 15.75 15.75 15.75 15.75 15.75				

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Manually	Incrementa Charge - Manual Svo Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	Increment - al Charge - Manual - Svc Order vs Electronic- Disc Add'I
						Dan	Names			curring			000 0	ATEC (6)		
						Rec	Nonred First	Add'l	First	onnect Add'l	COMEC	SOMAN		ATES (\$) SOMAN	COMAN	SOMAN
							FIISL	Auu i	FIISL	Auu i	SOMEC	SUMAN	SOWAN	SOWAN	JUNAN	SOWAN
																1
MULTIPLEXERS																
	Channelization - DS1 to DS0 Channel System			XTD1	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			DL	1D1DD	1.22	6.62	4.74				15.75				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month			DN	UC1CA	2.62	6.62	4.74				15.75				_
	Voice Grade COCI - DS1 to DS0 Channel System - per month	-		EA	1D1VG	0.5737	6.62	4.74	04.00	00.00		15.75				
	DS3 to DS1 Channel System per month	-		XTD3	MQ3	170.63	179.17	94.52	34.30	32.82		15.75				
	STS1 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) used with Loop per month			XTS1 SL	MQ3 UC1D1	170.63 12.96	179.17 6.62	94.52 4.74	34.30	32.82		15.75 15.75				+
DARK FIBER	DOS III. (DO I COOI) used with Loop per month	1		OL.	OCIDI	12.50	0.02	4.74				13.73				+
DARKTIDEK	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -															+
	Local Channel		L	DF	1L5DC	59.95										
	NRC Dark Fiber - Local Channel			DF	UDFC4		642.79	138.67	326.97	203.85		15.75				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -															
	Interoffice Channel	1		DF	1L5DF	28.27		,								4
	NRC Dark Fiber - Interoffice Channel	1	L	DF	UDF14		642.79	138.67	326.97	203.85		15.75				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -	1	1 .	DF	1L5DL	59.95										
 	Local Loop NRC Dark Fiber - Local Loop	1		DF	UDFL4	39.95	642.79	138.67	326.97	203.85		15.75				+
TRANSPORT OTHE	FR		-	DI	ODI L4		042.73	130.07	320.31	203.03		13.73				+
	al Features & Functions:															1
	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel		L	NC1X	CCOEF		184.60	23.78	1.96	0.76		15.75				1
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel			NC1X	CCOSF		184.60	23.78	1.96	0.76		15.75				1
8XX ACCESS TEN	DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call 8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved			HD		0.0006216										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved			HD	N8R1X		2.60	0.44				15.75				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS			HD			5.97	0.81	4.60	0.54		15.75				_
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS	-		HD	N8FTX		5.97	0.81	4.60	0.54		15.75				
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number 8XX Access Ten Digit Screening, Multiple InterLATA CXR Routing Per CXR			HD	N8FCX		2.60	1.30				15.75				+
	Requested Per 8XX No.		C	HD	N8FMX		3.04	1.74				15.75				
	8XX Access Ten Digit Screening, Change Charge Per Request			HD	N8FAX		3.04	0.44				15.75				+
	8XX Access Ten Digit Screening, Call Handling and Destination Features			HD	N8FDX		2.60					15.75				1
LINE INFORMATION	N DATA BASE ACCESS (LIDB)															
	LIDB Common Transport Per Query			QT		0.0000197										
	LIDB Validation Per Query			QU		0.0137053										
	LIDB Originating Point Code Establishment or Change		C	QT, OQU	NRPBX		34.52	34.52	42.33	42.33		15.75				
SIGNALING (CCS7)			 	IDD.	DTOOY	100.01										_
	CCS7 Signaling Termination, Per STP Port			DB	PT8SX	132.21										+
	CCS7 Signaling Usage, Per TCAP Message			DB DB	TPP++	0.0000597 16.55	35.74	35.74	16.53	16.53		15.75				+
	CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D link)			DB	TPP++	16.55	35.74	35.74	16.53	16.53		15.75				+
	CCS7 Signaling Usage, Per ISUP Message			DB		0.0000149	33.14	33.14	10.00	10.00		10.70				+
	CCS7 Signaling Usage Surrogate, per link per LATA			DB	STU56	683.55										1
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change,	1			2.230											1
	per STP affected	1	L	DB	CCAPO		29.18	29.18	35.78	35.78		15.75				
E911 SERVICE																
	Local Channel - Dedicated - 2-wr Voice Grade					14.91	194.22	33.36	37.79	3.30		15.75				
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile					0.0098										
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility Termination	-	-			22.52	40.77	27.57	17.26	7.11		15.75				
	Local Channel - Dedicated - DS1 - Zone 1	1	-			36.83 35.99	178.50 178.50	154.61 154.61	22.89 22.89	15.74 15.74		15.75 15.75				+
	Local Channel - Dedicated - DS1 - Zone 2 Local Channel - Dedicated - DS1 - Zone 3	1	+ +			221.63	178.50	154.61	22.89	15.74		15.75				+
	Local Channel - Dedicated - DS1 - Zone 3 Local Channel - Dedicated - DS1 - Zone 4	1				221.63	178.50	154.61	22.89	15.74		15.75				†
	Interoffice Transport - Dedicated - DS1 Per Mile					0.2010	. 7 0.00	754.01	22.03	10.74		10.75				1
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					57.33	89.79	82.28	16.86	14.90		15.75				
												15.75				
CALLING NAME (CI																
	CNAM for DB Owners, Per Query	1		QV		0.0010231										4
	CNAM for Non DB Owners, Per Query			QV		0.0010231			61.5							
	CNAM For DB Owners - Service Establishment	1		QV			23.09	23.09	21.23	21.23		15.75				₩
	CNAM For Non DB Owners - Service Establishment	1		QV VOV			23.09	23.09	21.23	21.23		15.75			-	+
	CNAM For DB Owners - Service Provisioning With Point Code Establishment CNAM For Non DB Owners - Service Provisioning With Point Code Establishment	1		QV QV	_		996.62 344.32	737.08 246.56	270.49 276.85	198.89 198.89		15.75 15.75				+
LNP Query Service		1	1	ru, v		+	344.32	∠40.56	2/0.05	190.89		15.75				+
LIVE WHEN DEIVICE	LNP Charge Per query	+		QV		0.0008477										+
	LNP Service Establishment Manual	+				J.0000T11	12.59	12.59	11.58	11.58	+	15.75		+	+	+

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	USOC		F	ATES(\$)			Submitte d Elec	Submitted	Charge - Manual Svc Order vs.	I Charge - Manual	Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic
										Nonrec							
							Rec	Nonrecu		Discor					ATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		LNP Service Provisioning with Point Code Establishment						596.94	304.96	270.49	198.89		15.75				
OPERATOR	CALL PRO							000.01									
		Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB					1.20										
		Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB					1.24										
		Oper. Call Processing - Fully Automated, per Call - Using BST LIDB					0.20										
11.11.4.DD O	DED ATOR O	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB					0.20										
INWARD OI	PERATOR S	Inward Operator Services - Verification, Per Minute	-				1.15										
		Inward Operator Services - Verification and Emergency Interrupt - Per Minute	1				1.15										
BRANDING		DR CALL PROCESSING					1.10										
		Recording of Custom Branded OA Announcement				CBAOS		7,000.00	7,000.00				15.75				
		Loading of Custom Branded OA Announcement per shelf/NAV				CBAOL		500.00	500.00				15.75				
		g via OLNS for UNEP CLEC															
		Loading of OA per OCN (Regional)						1,200.00	1,200.00				15.75				
		NCE SERVICES															
		Y ASSISTANCE ACCESS SERVICE Directory Assistance Access Service Calls, Charge Per Call	-				0.271744										
		Y ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)					0.271744										
		Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.10										
		Y TRANSPORT					01.10										
		SWA Common transport per Directory Assistance Access Service Call					0.000178										
		SWA Common Transport per Directory Assistance Access Service Call Mile					0.000017										
		Access Tandem Switching per Directory Assistance Access Service Call					0.000287										
		Directory Assistance Interconnection per Directory Assistance Access Service Cal	I.				0.00										
DIDECTOR		DS3 to DS1 Multiplexer per DA Access Service Call NCE SERVICES					0.00018										
		Y ASSISTANCE DATA BASE SERVICE (DADS)	-														
		Directory Assistance Data Base Service Charge Per Listing	1				0.04										
		Directory Assistance Data Base Service, per month				DBSOF	150.00										
BRANDING	- DIRECTOR	RY ASSISTANCE															
	Facility Bas																
		Recording and Provisioning of DA Custom Branded Announcement			AMT	CBADA		6,000.00	6,000.00								
		Loading of Custom Branded Announcement per DRAM Card/Switch			AMT	CBADC		1,170.00	1,170.00								
	UNEP CLE	Recording of DA Custom Branded Announcement						3,000.00	3,000.00								
		Loading of DA Custom Branded Announcement Loading of DA Custom Branded Announcement per DRAM Card/Switch per OCN						1,170.00	1,170.00								
	Unbranding	g via OLNS for UNEP CLEC						1,170.00	1,170.00								
	,	Loading of DA per OCN (1 OCN per Order)						420.00	420.00								
		Loading of DA per Switch per OCN						16.00	16.00								
SELECTIVE	ROUTING																
		Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		85.19	85.19	14.19	14.19		15.75				
VIRTUAL C	OLLOCATIO	Virtual Collocation - Application Cost			CLO	EAF		1,212.25		0.51							
		Virtual Collocation - Cable Installation Cost, per cable			CLO CLO	ESPCX		926.27		22.62							
		Virtual Collocation - Floor Space, per sq. ft.			CLO	ESPVX	5.74	320.21		22.02							
		Virtual Collocation - Power, per breaker amp			CLO	ESPAX	7.33										
		Virtual Collocation - Cable Support Structure, per entrance cable			CLO	ESPSX	15.24										
					ueanl,uea,udn,udc,				,	,							
ı	1	Virtual Collocation - 2-wire Cross Connects (loop) Virtual Collocation - 4-wire Cross Connects (loop)			ual,uhl,ucl,ueq	UEAC2	0.0268	12.37	11.87	6.04	5.45		15.75				
			1		uea,uhl,ucl,udl CLO	UEAC4 CNC2F	0.0536 2.91	12.47 21.01	11.94 15.29	6.59 7.61	5.91 6.10		15.75 15.75				
						UNUZE				10.01	8.50		15.75				
		Virtual Collocation - 2-Fiber Cross Connects				CNC4F	5.82	25.70									
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects			CLO	CNC4F CNC1X	5.82 1.14	25.70 22.16	19.97 16.02				15.75				
		Virtual Collocation - 2-Fiber Cross Connects			CLO				19.97 16.02 15.29	6.60 7.61	5.97 6.10		15.75 15.75				
		Wirtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocatin - DS1 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocatin - Co-Carrier Cross Connects - Fiber Cable Support Structure,			CLO USL,ULC,CLO USL,ULC,CLO	CNC1X CND3X	1.14 14.49	22.16	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocatin - DS1 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			CLO USL,ULC,CLO	CNC1X	1.14	22.16	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocation - DSI Cross Connects Virtual Collocation - DSS Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support			CLO USL,ULC,CLO USL,ULC,CLO AMTFS	CNC1X CND3X PE1ES	1.14 14.49 0.0025	22.16	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocatin - DS1 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear from Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear from Condition - Co-Carrier Cross Connects - Copper/Coax Cable Support			CLO USL,ULC,CLO USL,ULC,CLO	CNC1X CND3X	1.14 14.49	22.16	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocation - DSI Cross Connects Virtual Collocation - DSS Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support			CLO USL,ULC,CLO USL,ULC,CLO AMTFS AMTFS	CNC1X CND3X PE1ES	1.14 14.49 0.0025	22.16 21.01	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocation - DS1 Cross Connects Virtual Collocation - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft			CLO USL,ULC,CLO USL,ULC,CLO AMTFS	CNC1X CND3X PE1ES	1.14 14.49 0.0025	22.16	16.02	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocation - DS1 Cross Connects Virtual Collocation - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable	Г		CLO USL,ULC,CLO USL,ULC,CLO AMTFS AMTFS AMTFS AMTFS	CNC1X CND3X PE1ES PE1DS	1.14 14.49 0.0025	22.16 21.01 534.65	16.02 15.29	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocatin - DS1 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocatin - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable Virtual Collocation - Security Escort - Basic, per half hour	r		CLO USL,ULC,CLO USL,ULC,CLO AMTFS AMTFS AMTFS AMTFS CLO	CNC1X CND3X PE1ES PE1DS	1.14 14.49 0.0025	22.16 21.01 534.65 534.65 17.02	16.02 15.29	6.60	5.97						
		Virtual Collocation - 2-Fiber Cross Connects Virtual Collocation - 4-Fiber Cross Connects Virtual Collocation - DS1 Cross Connects Virtual Collocation - DS3 Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per cable			CLO USL,ULC,CLO USL,ULC,CLO AMTFS AMTFS AMTFS AMTFS CLO CLO	CNC1X CND3X PE1ES PE1DS	1.14 14.49 0.0025	22.16 21.01 534.65	16.02 15.29	6.60	5.97						

CATEGOR	Y RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			d Elec	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge - Manual Svc Order vs. Electronic-	Increment - al Charge Manual - Svc Order - vs Electronic Disc Add'I
						Do.	Name			curring			000 B	ATEC (A)		
						Rec		curring Add'l		nnect Add'l	COMEC	SOMAN	SOMAN	ATES (\$) SOMAN	SOMAN	SOMAN
							First	Add I	First	Add I	SUMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
																+
	Virtual Collocatin - Maintenance in CO - Overtime, per half hour		-	CLO	SPTOM		36.69	13.94								+
	Virtual Collocatin - Maintenance in CO - Premium per half hour			CLO	SPTPM		45.28	17.08								+
VIRTUAL COLLO																+
THE COLE	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-Wire Analog - Res		ı	JEPSR	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				+
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade Res			JEPRX	PE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				+
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Line Side PBX															
	Trunk - Bus		t	JEPSP	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Voice Grade PBX															
	Trunk - Res			JEPSE	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire Analog Bus			JEPSB	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 2-Wire Cross Connect, Exchnage Port 2-Wire ISDN			JEPSX	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				$oldsymbol{\perp}$
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			JEPTX	VE1R2	0.0268	12.37	11.87	6.04	5.45		15.75				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port DDITS 4-Wire DS1			JEPDD	VE1R4	0.0536	12.47	11.94	6.59	5.91		15.75				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			JEPEX	VE1R4	0.0536	12.47	11.94	6.59	5.91		15.75				
VIRTUAL COLLO			₩.	IEDOD LIEDO-	\/E · · · o	0.000										
A IN OF LEGET (T	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting	I		JEPSR, UEPSB	VE1LS	0.0268	12.37	11.87	6.04	5.45			19.99	19.99	19.99	19.99
AIN SELECTIVE	CARRIER ROUTING		1 .	200	00050		101 005 :-		0.040.51			45				+
	Regional Service Establishment			RC	SRCEC		101,685.12		8,640.51			15.75				
	End Office Establishment			SRC	SRCEO		167.49	167.49	1.71	1.71		15.75				
	Query NRC, per query I'H AIN SMS ACCESS SERVICE			SRC		0.0030502										
AIN - BELLSOUI	AIN SMS Access Service - Service Establishment, Per State, Initial Setup		Н.	1N	CAMSE		39.67	39.67	40.92	40.92		15.75				
				A1N	CAMDP			7.87	9.14	9.14						
	AIN SMS Access Service - Port Connection - Dial/Shared Access AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		7.87 7.87	7.87	9.14	9.14		15.75 15.75				+
	AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code			A1N	CAMAU		35.21	35.21	27.21	27.21		15.75				+
	AIN SMS Access Service - Oser Identification Codes - Per Oser ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		42.13	42.13	11.78	11.78		15.75				+
	AIN SMS Access Service - Security Card, Fer User 15 Code, militar of Replacement AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)			ATIN	CAIVING	0.0021	42.13	42.13	11.70	11.70		15.75				+
	AIN SMS Access Service - Storage, Per Offic (100 Kilobytes) AIN SMS Access Service - Session, Per Minute					0.5649										+
	AIN SMS Access Service - Company Performed Session, Per Minute					0.8393										$\overline{}$
AIN - BELL SOUT	TH AIN TOOLKIT SERVICE					0.0000										$\overline{}$
7 BEEEGG.	AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup			CAM	BAPSC		39.67	39.67	40.92	40.92		15.75				+
	AIN Toolkit Service - Training Session, Per Customer				BAPVX		4,226.54	4,226.54				15.75				+
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt				BAPTT		7.87	7.87	9.14	9.14		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		7.87	7.87	9.14	9.14		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook															
	Immediate				BAPTM		7.87	7.87	9.14	9.14		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				BAPTO		34.67	34.67	14.44	14.44		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		34.67	34.67	14.44	14.44		15.75				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code				BAPTF		34.67	34.67	14.44	14.44		15.75				
	AIN Toolkit Service - Query Charge, Per Query					0.0535577										\perp
	AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node,															1
	Per Query		+-+			0.0063509										
	AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100	l				0.00										
—	Kilobytes AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription		 	CAM	BAPMS	0.06 11.11	7.87	7.87	F.F.4	F.F.4		45.75				+
—	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription	-		CAM	BAPLS	11.11 2.71	7.87 8.71	7.87 8.71	5.54	5.54		15.75 15.75				+
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription			CAM	BAPDS	8.48	7.87	7.87	5.54	5.54		15.75				+
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service			CAM	BAPES	0.09	8.71	7.87 8.71	5.54	5.54		15.75				+
ENHANCED EXT	ENDED LINK (EELs)		+ +	27 WVI	DAI LO	0.09	0.71	0.71				13.73				+
	E: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando	FI · Mia	ami FI:	Et Lauderdale	FI I: Nach	ille TN: Now	Orleans I A									+
	E: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-High Point, NC. Use a						Oncano, LA,									+
	E: In all states, EEL network elements shown below also apply to currently combined						s Is Charge a	oplies to curre	ently combin	ned facilities	converted	to UNEs (N	on-recurring	rates do no	t apply.)	+
NOT	E: In GA, TN, KY, LA & MS, the EEL network elements apply to ordinarily combined ne	twork e	lements	(No Switch As Is	Charge \		•gc u		,				roourring			\vdash
2-WI	E: In GA, TN, KY, LA & MS, the EEL network elements apply to ordinarily combined ne IRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO	RT (EEL	.)	,	301)											+
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1	_,		JNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination -															
	Zone 2		2 l	JNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75				<u> </u>
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination -															
	Zone 3			JNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 4			JNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile per month			JNC1X	1L5XX	0.1813										$oldsymbol{oldsymbol{\square}}$
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per			JNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				\perp
	DS1 Channelization System Per Month			JNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				\perp
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			JNCVX	1D1VG	0.5737	6.62	4.74								
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1 Interoffice Transport		1 , 1.	1100.07			,		F0.0-							1
	Combination - Zone 1	L	1 I	JNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75		L	L	

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	vs. Electronic-	
									Nonrec	curring						
						Rec	Nonrec	urring	Disco	nnect			OSS R	ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport															
	Combination - Zone 2		2	UNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport															
	Combination - Zone 3		3	UNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 4		4	UNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
			4						52.82	10.37					+	+
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX UNC1X	1D1VG UNCCC	0.5737	6.62 5.63	4.74 5.63	7.20	7.20		15.75 15.75			-	-
4 WIDE VC	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO	DT /EEL		UNCIX	UNCCC		5.63	5.63	7.20	7.20		15.75				+
4-WIRE VC	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	KI (EEL)												+	+
	Zone 1		1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	-	'	ONCVA	ULAL4	21.41	132.21	34.33	00.00	14.04		13.73			+	+
	Zone 2	1	2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	-	T-		JE, 12 T	33.20	.02.27	055	30.00	04		.0.70			1	
	Zone 3	1	3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination		T -			,,,,,,										
	Zone 4	1	4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1813									1	1
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75			1	1
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.5737	6.62	4.74				15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport														1	1
	Combination - Zone 1		1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport															
	Combination - Zone 2		2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport															
	Combination - Zone 3		3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport															
	Combination - Zone 4		4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				
	Voice Grade COCI - DS1 to DS0 Channel System combination - per month			UNCVX	1D1VG	0.5737	6.62	4.74				15.75				_
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		<u> </u>	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				_
4-WIRE 56	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	PORT (E	EL)													
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination			LINCDY	LIDL CC	27.44	400.50	00.05	00.00	44.04		45.75				
	- Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		1	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	Zone 2	1	2	UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination			UNCDX	UDLS6	34.55	120.53	88.83	80.08	14.04		15.75				+
	- Zone 3		3	UNCDX	UDL56	40.76	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		J	ONCDA	ODLOG	40.70	120.00	00.00	00.00	14.04		13.73			+	+
	- Zone 4		4	UNCDX	UDL56	32.25	126.53	88.85	60.68	14.64		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month		7	UNC1X	1L5XX	0.1813	120.00	00.00	00.00	14.04		15.75			+	+
	Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	1.22	6.62	4.74	. 3.01			15.75			1	†
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport				,55		0.02	7				.00			1	†
	Combination - Zone 1	1	1	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport															
	Combination - Zone 2		2	UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64	<u> </u>	15.75	<u> </u>		<u> </u>	
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport															
	Combination - Zone 3		3	UNCDX	UDL56	40.76	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport			-												
	Combination - Zone 4		4	UNCDX	UDL56	32.25	126.53	88.85	60.68	14.64		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-															
	64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				
4-WIRE 64	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS		EL)													+
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1	1	1	UNCDX	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				1
	- Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		-	ONCDA	UDL04	21.44	120.53	66.65	00.08	14.04		15.75		-	+	+
	- Zone 2	1	2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	1	-	014007	JDL04	34.00	120.03	00.00	00.00	14.04		10.75			+	
	- Zone 3	1	3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination	1		2.100/1	33204	10.70	.20.00	00.00	55.00	17.04		10.75			1	
		1	4	UNCDX	UND64	32.25	126.53	88.85	60.68	14.64		15.75				
	- Zone 4														+	+
					1L5XX	0.1813										
	- Zone 4 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per			UNC1X UNC1X	1L5XX U1TF1	0.1813 51.72	89.79	82.28	16.86	14.90		15.75				

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Manual Svc Order vs. Electronic	- al Charge Manual Svc Orde vs.
						_				curring						
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-															+
	64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport			UNCDX	טטוטו	1.22	0.02	4.74				15.75				+
	Combination - Zone 1		1	UNCDX	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		-	UNCDA	UDL04	21.44	120.55	00.00	00.00	14.04		13.73				+
	Combination - Zone 2		2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		-	OHODA	ODEO!	01.00	120.00	00.00	00.00			10.10				1
	Combination - Zone 3		3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport															+
	Combination - Zone 4		4	UNCDX	UDL64	32.25	126.53	88.85	60.68	14.64		15.75				
	OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-									-						1
	64kbs)			UNCDX	1D1DD	1.22	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				
	I DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	RT (EEL)														1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2		2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				1
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 4			UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.1813										1
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				1
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				1
	I DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR	T (EEL)														1
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				+
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		2	UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				†
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 3			UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				+
	First DS1Loop in DS3 Interoffice Transport Combination - Zone 4		4	UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				1
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	4.29	200.00	100.10	10.10	12.01		10.10				+
	Interoffice Transport - Dedicated - DS3 - Facility Termination per month			UNC3X	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				+
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3	107.85	179.17	94.52	34.30	32.82		15.75				+
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	12.96	6.62	4.74	04.00	02.02		15.75				+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2			UNC1X	USLXX	129.38	253.93	158.45	46.10	12.07		15.75				+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3			UNC1X	USLXX	206.74	253.93	158.45	46.10	12.07		15.75				+
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 4			UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75			-	+
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	12.96	6.62	4.74	40.10	12.07		15.75			-	+
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC3X	UNCCC	12.30	5.63	5.63	7.20	7.20		15.75			-	+
	CE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPO	DT /EEI		UNUSA	UNCCC		5.05	3.03	7.20	7.20		13.73			-	+
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	INT (LLL	1	UNCVX	UEAL2	13.89	105.96	68.28	52.82	10.37		15.75				+
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	18.75	105.96	68.28	52.82	10.37		15.75				+
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	27.55	105.96	68.28	52.82	10.37		15.75				+
	A.1.2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination -		3	UNCVA	UEALZ	27.55	105.90	00.20	52.62	10.37		13.73				+
	Zone 4		4	UNCVX	UEAL2	45.72	105.96	68.28	52.82	10.37		15.75				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month	!	-	UNCVX	1L5XX	0.00088	103.90	00.20	JZ.UZ	10.37	 	13.73			-	+
_	Interoffice Transport - Dedicated - 2-Wire Voice Grade combination - Facility	!	 	OHOVA	ILUAA	0.00000		1			 				-	+
	Termination per month			UNCVX	U1TV2	20.32	40.77	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	!	1	UNCVX	UNCCC	20.02	5.63	5.63	7.20	7.11	 	15.75				1
	CE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE INTEROFFICE TRANSPO	RT (FFI	1	0.10 171	0.1000		0.00	0.00	7.20	7.20		10.10				+
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	27.47	132.27	94.59	60.68	14.64		15.75				+
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	38.26	132.27	94.59	60.68	14.64		15.75				+
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				+
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 4		4	UNCVX	UEAL4	50.03	132.27	94.59	60.68	14.64		15.75				+
	Interoffice Transport - Dedicated - 4-wire VG combination - Per Mile Per Month		-	UNCVX	1L5XX	0.00088	102.21	34.03	00.00	14.04		10.70				+
	Interoffice Transport - Dedicated - 4- Wire Voice Grade combination - Facility			ONOVA	TEOXIX	0.00000										+
	Termination per month			UNCVX	U1TV4	17.86	40.77	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCVX	UNCCC		5.63	5.63	7.20	7.20		15.75				1
	L EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)		1		3.1300		0.00	0.00	1.20	1.20	 	10.75				+
	High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month			UNC3X	1L5ND	11.20										+
	High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per	!	1	2.100/1	, 20140	11.20		1			 	†				1
	month			UNC3X	UE3PX	252.17	454.13	265.47	123.23	86.19		15.75				
	Interoffice Transport - Dedicated - DS3 - Per Mile per month	!	 	UNC3X	1L5XX	4.29	404.10	200.47	120.20	00.13	1	10.75				+
	Interoffice Transport - Dedicated - DS3 - Fer Mile per month Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per	!	 	OITOOA	ILUAA	4.29		1			 				-	+
	month		1	UNC3X	U1TF3	641.90	280.37	163.70	62.08	60.29		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		†	UNC3X	UNCCC	3 71.50	5.63	5.63	7.20	7.20		15.75				1
	AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EE	1)	t			+	5.00	3.30	0	20	1	.0.70		 	1	+
		-/	1	UNCSX	1L5ND	11.20					 	 	 	-	-	+

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)				Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st		al Charge - Manual Svc Order vs. Electronic-	Increment al Charge - Manual Svc Order vs. Electronic- Disc Add'l
									Nonred	curring					•	
						Rec	Nonrec		Disco					TES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per															-
	month			UNCSX	UDLS1	264.35	454.13	265.47	123.23	86.19		15.75				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile per month			UNCSX	1L5XX	4.29	101.10	200.11	120.20	00.10		10.70				-
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination per															
	month			UNCSX	U1TFS	644.21	280.37	163.70	62.08	60.29		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCSX	UNCCC		5.63	5.63	7.20	7.20		15.75				
	N EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)															
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1	-	1 2	UNCNX	U1L2X	21.01	117.61	79.92	52.82	10.37		15.75				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3			UNCNX	U1L2X U1L2X	27.59 37.34	117.61 117.61	79.92 79.92	52.82 52.82	10.37 10.37		15.75 15.75				
	First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 4		4	UNCNX	U1L2X	59.18	117.61	79.92	52.82	10.37		15.75				+
	Interoffice Transport - Dedicated - DS1 combination - Per Mile			UNC1X	1L5XX	0.1813		70.02	02.02	10.01		10.70				-
	Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	51.72	89.79	82.28	16.86	14.90		15.75				
	Channelization - Channel System DS1 to DS0 combination - per month			UNC1X	MQ1	102.85	91.57	62.94	10.87	10.10		15.75		_		
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNCNX	UC1CA	2.62	6.62	4.74				15.75				
ŀ	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone	1		LINIONIV	U1L2X	24.04	117.61	70.00	50.00	10.37		45.75				
	1 Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone		2	UNCNX		21.01		79.92	52.82			15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone		3	UNCNX	U1L2X	27.59	117.61	79.92	52.82	10.37		15.75				
	Additional 2-wire ISDN Loop in same DS1Interoffice Transport Combination - Zone		3	UNCNX	U1L2X	37.34	117.61	79.92	52.82	10.37		15.75				
ľ	Additional 2-wire 13DN Loop in same D3 finteronice Transport Combination - Zone 4		4	UNCNX	U1L2X	59.18	117.61	79.92	52.82	10.37		15.75				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month			UNCNX	UC1CA	2.62	6.62	4.74				15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75				
	DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPO	ORT (EE	L)													
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X UNC1X	USLXX	129.38 206.74	253.93 253.93	158.45 158.45	46.10 46.10	12.07 12.07		15.75 15.75				
	First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 First DS1 Loop in STS1 Interoffice Transport Combination - Zone 4	1	4	UNC1X UNC1X	USLXX	458.46	253.93	158.45	46.10	12.07		15.75				
	Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month		-	UNCSX	1L5XX	4.29	233.93	130.43	40.10	12.01		13.73				+
	Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	644.21	280.37	163.70	62.08	60.29		15.75				
,	STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	107.63	179.17	94.52	34.30	32.82		15.75				
	DS3 Interface Unit (DS1 COCI) combination per month			UNC1X	UC1D1	12.96	6.62	4.74				15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X	USLXX	79.08	253.93	158.45	46.10	12.07		15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2	-	3	UNC1X	USLXX	129.38 206.74	253.93	158.45	46.10 46.10	12.07		15.75				
	Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 4	1		UNC1X UNC1X	USLXX	458.46	253.93 253.93	158.45 158.45	46.10	12.07 12.07		15.75 15.75				
	DS3 Interface Unit (DS1 COCI) combination per month		-	UNC1X	UC1D1	12.96	6.62	4.74	40.10	12.01		15.75				+
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	1		UNCSX	UNCCC	50	5.63	5.63	7.20	7.20		15.75				
4-WIRE 56 P	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (EEL)														
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	27.44	126.53	88.85	60.68	14.64		15.75				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2	1		UNCDX	UDL56	34.55	126.53	88.85	60.68	14.64		15.75				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 4	1	3	UNCDX UNCDX	UDL56 UDL56	40.76 32.25	126.53 126.53	88.85 88.85	60.68 60.68	14.64 14.64		15.75				
	4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 4 Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile	1	4	UNCDX	1L5XX	0.00088	120.53	88.85	80.00	14.64		15.75				
	Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility		1	UNCDX	U1TD5	14.14	40.78	27.57	17.26	7.11		15.75				+
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX	UNCCC	14.14	5.63	5.63	7.20	7.20		15.75				1
4-WIRE 64 P	KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (EEL)														
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL64	27.44	126.53	88.85	60.68	14.64		15.75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	34.55	126.53	88.85	60.68	14.64		15.75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3	-	3	UNCDX	UDL64	40.76	126.53	88.85	60.68	14.64		15.75				
	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 4 Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile	1	4	UNCDX	UDL64 1L5XX	32.25 0.00088	126.53	88.85	60.68	14.64		15.75				+
	Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility	1		UNCDX	U1TD6	14.14	40.78	27.57	17.26	7.11		15.75				
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	1		UNCDX	UNCCC	14.14	5.63	5.63	7.20	7.11		15.75				†
DDITIONAL NETWORK				-			2.20		5	0						1
	as a part of a currently combined facility, the non-recurrng charges do not a															
When used	as ordinarilty combined network elements in Georgia, the non-recurring char	ges app	y and t	he Switch As Is C	harge doe	es not.										4
Nonrecurrin	g Currently Combined Network Elements "Switch As Is" Charge (One applies	to each	combin	ation)												
	2/4-Wire VG Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion Charge	1		UNCVX	UNCCC		5.63	5.63	7.20	7.20		15.75				
	56/64 kbps Interoffice Channel used in a COMBINATION - "Switch As Is"	1		0140 4 \	UINCCC		5.03	0.03	1.20	1.20		10.75				
	Conversion Charge			UNCDX	UNCCC		5.63	5.63	7.20	7.20		15.75				
1 1	DS1 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion	1	l	UNC1X	UNCCC		5.63	5.63	7.20	7.20		15.75	1		1	1

CATE	GORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Submitted	Charge - Manual Svc Order vs.		vs. Electronic-	al Charge - Manual Svc Order vs.
							B	Managa			curring			000 B	ATEC (6)		
							Rec	Nonrec First	urring Add'l	First	onnect Add'l	COMEC	SOMAN		ATES (\$) SOMAN	SOMAN	SOMAN
								FIRST	Add I	FIIST	Add I	SUMEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMAN
																-	+
		DS3 Interoffice Channel used in a COMBINATION - "Switch As Is" Conversion															1
		Charge			UNC3X	UNCCC	ı	5.63	5.63	7.20	7.20		15.75				
		STS1 Interoffice or Local Loop used in a COMBINATION - "Switch As Is"															
		Conversion Charge			UNCSX	UNCCC		5.63	5.63	7.20	7.20		15.75				
		al Channel - Dedicated Transport - minimum billing period - Below DS3=one mo	onth, DS	3 and a	above=four months	s											
UNBUNDLE		KCHANGE SWITCHING(PORTS)															
	Exchange F			L									<u> </u>			_	_
		ough the Port Rate includes all available features in GA, KY, LA & TN, the desir	red featu	ires wil	I need to be ordere	ed using	retail USOCs							 		ļ	
		CE GRADE LINE PORT RATES (RES)			UEPSR	UEPRL	1.41	2.39	2.29	1.42	1.33		15.75			-	
		Exchange Ports - 2-Wire Analog Line Port- Res. Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.41	2.39	2.29	1.42	1.33		15.75	 		_	+
		Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res. Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.			UEPSR	UEPRO	1.41	2.39	2.29	1.42	1.33		15.75	 		 	+
		Exchange Ports - 2-Wire VG unbundled MS extended local dialing parity Port with			CE. OIL	SEI INS	1.71	2.55	2.23	1.42	1.33		10.73			 	+
1		Caller ID - Res.			UEPSR	UEPAT	1.41	2.39	2.29	1.42	1.33		15.75				
		Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID			UEPSR	UEPAP	1.41	2.39	2.29	1.42	1.33		15.75				1
		Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00						1		
	FEATURES																
		All Available Vertical Features			UEPSR	UEPVF	2.56	0.00	0.00				15.75				
		CE GRADE LINE PORT RATES (BUS)															
		Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus			UEPSB	UEPBL	1.41	2.39	2.29	1.42	1.33		15.75				
		Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with											l!				
		Caller+E484 ID - Bus.			UEPSB	UEPBC	1.41	2.39	2.29	1.42	1.33		15.75			<u> </u>	_
		Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.41	2.39	2.29	1.42	1.33		15.75				
		Exchange Ports - 2-Wire VG unbundled MS extended local dialing parity Port with Caller ID - Bus.			UEPSB	UEPAY	1.41	2.39	2.29	1.42	4.00		45.75				
-		Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus			UEPSB	UEPB1	1.41	2.39	2.29	1.42	1.33 1.33		15.75 15.75		+	_	+
		Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00	1.42	1.33		15.75			 	+
	FEATURES				OLF 3B	OOAGC	0.00	0.00	0.00							-	+
		All Available Vertical Features			UEPSB	UEPVF	2.56	0.00	0.00				15.75			1	+
		PORT RATES (DID & PBX)						0.00	0.00								+
		2-Wire VG Unbundled 2-Way PBX Trunk - Res			UEPSE	UEPRD	1.41	31.45	14.93	14.38	0.92		15.75				1
		2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus			UEPSP	UEPPC	1.41	31.45	14.93	14.38	0.92		15.75				1
		2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus			UEPSP	UEPPO	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus			UEPSP	UEPP1	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Analog Long Distance Terminal PBX Trunk - Bus			UEPSP	UEPLD	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Vice Unbundled 2-Way PBX Usage Port			UEPSP	UEPXA	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPSP	UEPXB	1.41	31.45	14.93	14.38	0.92		15.75				
<u> </u>		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPSP	UEPXC	1.41	31.45	14.93	14.38	0.92		15.75	—		 	+
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	-		UEPSP UEPSP	UEPXD	1.41	31.45 31.45	14.93	14.38	0.92		15.75	 	 		+
<u> </u>		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling			UEFOP	UEPAE	1.41	31.45	14.93	14.38	0.92		15.75			 	+
1		Port	l		UEPSP	UEPXL	1.41	31.45	14.93	14.38	0.92		15.75	1			
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPSP	UEPXM		31.45	14.93	14.38	0.92		15.75			 	+
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room						31.13		. 1.00	0.02		.00		1		+
		Calling Port			UEPSP	UEPXO	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Port			UEPSP	UEPXQ	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Port			UEPSP	UEPXR	1.41	31.45	14.93	14.38	0.92		15.75				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.41	31.45	14.93	14.38	0.92		15.75				
		Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00				<u> </u>				
	FEATURES					L											
		All Available Vertical Features			UEPSP UEPSE	UEPVF	2.56	0.00	0.00				15.75			<u> </u>	
		PORT RATES (COIN)				 		0.00	0.00		1.00		45.75	—		 	+
<u> </u>	NOTE: Tra-	Exchange Ports - Coin Port nsmission/usage charges associated with POTS circuit switched usage will als	o apple	to circ	uit ewitched voice	and/ar -:	1.41	2.39	2.29	1.42	1.33	2-wire ICC	15.75	 		 	+
	NOTE: MAR	ismission/usage charges associated with POTS circuit switched usage will als ess to B Channel or D Channel Packet capabilities will be available only throug	nh BEDA	Now D.	isings Rounce D	ALIU/OF CI	Rates for the n	uata transMi acket canabi	ilitiae will ba da	arminad s	ia the Bono	Fide Page	net/New P	eineer Dam	lest Process		+
UNRUNDI		ess to 6 Channel of 6 Channel Packet capabilities will be available only throug KCHANGE SWITCHING(PORTS)	JII DER/I	TOW DU	omess nequest Pr	JUE33. F	tates for the pa	uonei vapabi	iii.rea wiii be de	.commed \	ia uie Duild	i iue nequ	COUNTRY DU	onicoo Requ	CSCF TOCESS	-	+
		PORT RATES (DID & PBX)											\vdash			 	+
		Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.25	120.00	18.85	61.77	3.88		15.75		1	1.97	+
		Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability			UEPDD	UEPDD	58.41	203.19	96.25	74.86	2.54		15.75		1	1.97	
		Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.69	73.19	53.30	47.90	10.76		15.75			1.97	
		All Features Offered				UEPVF	2.56	0.00	0.00				15.75			1.97	
				to oiro								2 wire ICC					1
	NOTE: Tran	nsmission/usage charges associated with POTS circuit switched usage will als															
	NOTE: Tran	ess to B Channel or D Channel Packet capabilities will be available only through			ısiness Request Pr	rocess. F	Rates for the page	acket capabi	ilities will be de					siness Requ	est Process	<u> </u>	
	NOTE: Tran				ısiness Request Pr									siness Requ	iest Process	1.97	

CATEGO	DRY RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	- al Cha Mani r Svc O vs. - Electro
									Nonre	curring						
						Rec	Nonrec	curring		onnect			OSS R	ATES (\$)		
						1100	First	Add'I	First	Add'I	SOMEC	SOMAN		SOMAN	SOMAN	SOM
							11131	Auu i	1 11 5 1	Auu	SOMEC	SOWAIN	JONAN	JOWAN	JOINAIN	301
			-												+	+-
															+	4
UNDLED	LOCAL SWITCHING, PORT USAGE															
Er	nd Office Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0010269										
	End Office Trunk Port - Shared, Per MOU					0.000161										
Ta	andem Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001723										T
	Tandem Trunk Port - Shared, Per MOU					0.0001828										1
Co	ommon Transport														+	+
	Common Transport - Per Mile, Per MOU					0.0000026									+	+
	Common transport - Fer Miles, Fer Mood														+	+
	Common Transport - Facilities Termination Per MOU					0.0004541										
	PORT/LOOP COMBINATIONS - COST BASED RATES															4
Co	ost Based Rates are applied where BellSouth is required by FCC and/or State Commis	sion rule t	o provi	ide Unbundle	d Local Switc	hing or Switc	n Ports.									
Fe	eatures shall apply to the Unbundled Port/Loop Combination - Cost Based Rate section	n in the sa	ame ma	inner as they	are applied to	o the Stand-A	lone Unbundle	ed Port section	of this Ra	te Exhibit.						
Er	nd Office and Tandem Switching Usage and Common Transport Usage rates in the Po	rt section	of this	rate exhibit :	hall apply to	all combination	ns of loop/po	rt network elem	ents exce	pt for UNE	Coin Port/L	.oop Combi	nations.			
Fc	or Georgia, Kentucky, Louisiana, MIssissippi and Tennessee, the recurring UNE Port a	nd Loop c	harges	listed apply	to Currently C	Combined and	Not Currently	Combined Cor	nbos. Th	e the first ar	nd addition	al Port non	recurring cha	arges apply t	to Not Curre	ently
2-1	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)													3 117	T	T
	NE Port/Loop Combination Rates	1														+
UI.	2-Wire VG Loop/Port Combo - Zone 1	+	1			12.22									+	+
		+	2									 		1	+	+
	2-Wire VG Loop/Port Combo - Zone 2	+				17.13						1			+	+-
	2-Wire VG Loop/Port Combo - Zone 3		3			26.26										
	2-Wire VG Loop/Port Combo - Zone 4		4			44.91										
UN	NE Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1			UEPRX	UEPLX	10.98										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	15.91										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	25.04										
	2-Wire Voice Grade Loop (SL1) - Zone 4			UEPRX	UEPLX	43.68									+	+
2-1	Wire Voice Grade Line Port Rates (Res)			OLI TOX	OL: EX	10.00									+	+
				UEPRX	UEPRL	1.23	40.31	19.84	24.90	6.58		15.75			+	+
	2-Wire voice unbundled port - residence		-												+	+
	2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire voice unbundled port outgoing only - res			UEPRX	UEPRO	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire voice Grade unbundled Mississippi extended local dialing parity port with															
	Caller ID - res			UEPRX	UEPAT	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire voice unbundles res, low usage line port with Caller ID (LUM)			UEPRX	UEPAP	1.23	40.31	19.84	24.90	6.58		15.75				
FE	EATURES															
	All Features Offered			UEPRX	UEPVF	2.56	0.00	0.00				15.75				
LC	OCAL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35									+	#
NC	Local Number Portability (1 per port) ONRECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLI TOX	2.4. 0/4	0.00									+	+
1.00	2 Wire Voice Crede Loop / Line Both Combination Conversion Switch as is			UEPRX	USAC2		0.0988	0.0988				15.75			+	+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with		-	UEFRA	USACZ		0.0900	0.0900				13.73			+	+
	change			LIEDDY	110400		0.0000	0.0000				45 75				
	change		-	UEPRX	USACC		0.0988	0.0988				15.75			4	+-
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent				1											
	Database Update	1					0.00	0.00				15.75			4	+
ΑI	DDITIONAL NRCs	1													1	4
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPRX	USAS2	0.00	0.00	0.00				15.75				
2-1	WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
UN	NE Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			12.22										
	2-Wire VG Loop/Port Combo - Zone 2		2			17.13										
	2-Wire VG Loop/Port Combo - Zone 3		3			26.26									+	+
LIN	NE Loop Rates	1				20.20										+
J.		+	4	UEPBX	UEPLX	10.98									+	+
	2-Wire Voice Grade Loop (SL1) - Zone 1	+										 		1	+	+
	2-Wire Voice Grade Loop (SL1) - Zone 2	+		UEPBX	UEPLX	15.91						1		1	+	+-
	2-Wire Voice Grade Loop (SL1) - Zone 3			UEPBX	UEPLX	25.04						ļ			4	4
	2-Wire Voice Grade Loop (SL1) - Zone 4	1	4	UEPBX	UEPLX	43.68						1			4	4_
2-	Wire Voice Grade Line Port (Bus)														1	
	2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.23	40.31	19.84	24.90	6.58		15.75			1	
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.23	40.31	19.84	24.90	6.58	1	15.75				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.23	40.31	19.84	24.90	6.58		15.75				Т
	Wire voice unbundled port outgoing only - bus Wire voice Grade unbundled Mississippi extended local dialing parity port with	1			,_, 50					2.30						1
	Caller ID - bus			UEPBX	UEPAY	1.23	40.31	19.84	24.90	6.58		15.75				
-		+		UEPBX	UPEB1	1.23	40.31	19.84	24.90	6.58		15.75			+	+
	2-Wire voice unbundled incoming only port with Caller ID - Bus OCAL NUMBER PORTABILITY	+	-	ULFDA	UFEBI	1.23	40.31	19.04	24.90	0.58		10.75	-	1	+	+
	UCAL NUMBER FUR I ABILII I	- 1								-	1	ļ		-	+	+-
LC																
	Local Number Portability (1 per port) EATURES			UEPBX	LNPCX	0.35										+

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc	,		RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Increment al Charge Manual Svc Order vs. Electronic Disc 1st	Al Charge Manual Svc Ord vs. Electror
										curring						
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
NONDECLID	RRING CHARGES (NRCs) - CURRENTLY COMBINED															+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPBX	USAC2		0.0988	0.0988				15.75				+
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			02. 27.	CONOL		0.0000	0.0000				10.70				
	change			UEPBX	USACC		0.0988	0.0988								
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent															
	Database Update						0.00	0.00				15.75				
ADDITIONA	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2							15.75				+
	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			UEPBA	USASZ							15.75				+
	pop Combination Rates															+
	2-Wire VG Loop/Port Combo - Zone 1		1			12.22										+
	2-Wire VG Loop/Port Combo - Zone 2		2			17.13										1
	2-Wire VG Loop/Port Combo - Zone 3		3			26.26										
	2-Wire VG Loop/Port Combo - Zone 4		4	•		44.91	_			-				_		
UNE Loop F					1											1
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	10.98										4
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	15.91										+
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPRG	UEPLX	25.04										-
	2-Wire Voice Grade Loop (SL 1) - Zone 4 e Grade Line Port Rates (RES - PBX)		4	UEPRG	UEPLX	43.68										+
Z-VVIIE VOIC	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	1.23	69.37	32.48	37.86	6.17		15.75				+
LOCAL NUM	MBER PORTABILITY			02.7.0	OLI IID	1.20	00.07	02.10	07.00	0.17		10.70				+
	Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00								1
FEATURES																
	All Features Offered			UEPRG	UEPVF	2.56	0.00	0.00				15.75				
	RING CHARGES (NRCs) - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-			UEPRG	USAC2		7.96	1.91				15.75				_
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with Change			UEPRG	USACC		7.96	1.91				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent			UEFRG	USACC		7.90	1.91				13.73				+
	Database Update						0.00	0.00				15.75				
ADDITIONA	L NRCs															
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.75				
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.36	7.36				15.75				
	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
	pop Combination Rates		_			40.00										
	2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		2			12.22 17.13										+
	2-Wire VG Loop/Port Combo - Zone 3		3			26.26										+
	2-Wire VG Loop/Port Combo - Zone 4		4			44.91										+
UNE Loop F						11.01										+
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 2			UEPPX	UEPLX	15.91	_			-						
	2-Wire Voice Grade Loop (SL 1) - Zone 3			UEPPX	UEPLX	25.04										1
0.14/2	2-Wire Voice Grade Loop (SL 1) - Zone 4	ļ	4	UEPPX	UEPLX	43.68										4
2-Wire Voic	e Grade Line Port Rates (BUS - PBX)			UEPPX	UEPPC	4.00	69.37	32.48	37.86	C 47		45.75				+
	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.23	69.37	32.48	37.86	6.17 6.17		15.75 15.75				+
	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.23	69.37	32.48	37.86	6.17		15.75				+
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.23	69.37	32.48	37.86	6.17		15.75				+
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.23	69.37	32.48	37.86	6.17		15.75				1
	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port	1		UEPPX	UEPXD	1.23	69.37	32.48	37.86	6.17		15.75				1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling			UEPPX	UEPXE	1.23	69.37	32.48	37.86	6.17		15.75			1	1
	2-vvire voice unbundled 2-vvay PBX Hotel/Hospital Economy Administrative Calling Port	'l		UEPPX	UEPXL	1.23	69.37	32.48	37.86	6.17		15.75				1
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port	1		UEPPX	UEPXM	1.23	69.37	32.48	37.86	6.17		15.75				+
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room				OEI AIVI	1.20	00.01	32.40	37.00	0.17		10.75			1	+
	Calling Port	<u></u>		UEPPX	UEPXO	1.23	69.37	32.48	37.86	6.17		15.75			<u> </u>	<u> </u>
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Economy Calling Port			UEPPX	UEPXQ	1.23	69.37	32.48	37.86	6.17		15.75				
	2-Wire Voice Unbundled 2-Way PBX Mississippi Local Optional Calling Port			UEPPX	UEPXR	1.23	69.37	32.48	37.86	6.17		15.75				1
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	1		UEPPX	UEPXS	1.23	69.37	32.48	37.86	6.17		15.75				1
	MBER PORTABILITY	1		UEPPX	LNPCP	0.45	0.00	0.00			1				1	+
	Local Number Portability (1 per port)	0		LIEPPX		3.15	0.00	0.00	1		1	i .	ı		1	1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)	Name		d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	Manual Svc Order vs. Electronic-	
						D	Manna			curring			000 D	ATEC (C)		
						Rec	Nonred			nnect	COMEC	SOMAN		ATES (\$)	COMAN	COMAN
					-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
															+	+
	All Features Offered			UEPPX	UEPVF	2.56	0.00	0.00				15.75			+	+
NONDECLIE	RRING CHARGES (NRCs) - CURRENTLY COMBINED			OLITA	OLF VI	2.50	0.00	0.00				13.73			+	+
NONKECO	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As-			UEPPX	USAC2		7.96	1.91				15.75			+	+
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch with			UEFFX	USACZ		7.90	1.91				15.75				+
	Change			UEPPX	USACC		7.96	1.91				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent			OL: 17	00,100		7.00	1.01				10.70				-
	Database Update						0.00	0.00				15.75				
ADDITIONA															1	
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPPX	USAS2	0.00	0.00	0.00				15.75			+	+
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						7.36	7.36				15.75			1	
2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT														1	
	oop Combination Rates		1												1	1
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			12.22		j							1	
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			17.13		l							1	
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			26.26										
	2-Wire VG Coin Port/Loop Combo – Zone 4		4			44.91		j							1	
UNE Loop I																
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	10.98										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	15.91										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	25.04										
	2-Wire Voice Grade Loop (SL1) - Zone 4		4	UEPCO	UEPLX	43.68										
2-Wire Voice	ce Grade Line Ports (COIN)															
	2-Wire Coin 2-Way without Operator Screening and without Blocking (AL, KY, LA,															
	MS)			UEPCO	UEPRF	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way without Operator Screening and without Blocking; with Dialing															
	Parity (Note 3) (MS)			UEPCO	UEPMC	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD															
	(AL, KY, LA, MS)			UEPCO	UEPRA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-W with Operator Screening and Blocking: 011, 900/976, 1+DDD;															
	with Dialing Parity (MS)			UEPCO	UEPMA	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (AL, LA, MS)			UEPCO	UEPRB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity			LIEBOO	LIEDME	4.00	40.04	40.04	04.00	0.50		45.75				
	(MS)			UEPCO	UEPMB	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin 2-Way with Operator Screening & Blocking: 900/976, 1+DDD, 011+, & Local (AL, KY, LA, MS)			UEPCO	UEPCD	1.23	40.31	19.84	24.90	6.58		15.75				
				DEFCO	UEFCD	1.23	40.31	19.04	24.90	0.36		15.75				+
	2-Wire Coin 2-W Operator Screening: 900 Block: 900/976, 1+DDD, 011+, Local; with Dialing Parity (MS)			UEPCO	UEPCJ	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, LA,			UEPCO	UEPRN	1.23	40.31	19.84	24.90	6.58		15.75			+	+
	2-Wire Coin Outward without Blocking and without Operator Screening (KY, EA,			OLI CO	OLITIN	1.23	40.51	13.04	24.50	0.56		13.73			+	+
	Dailing Parity (MS)			UEPCO	UEPME	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (GA, KY, MS)			UEPCO	UEPRJ	1.23	40.31	19.84	24.90	6.58		15.75				-
	2-Wire Coin Outward with Operator Screening and 011 Blocking; with Dialing Parity			OLI OO	OLI INO	1.20	40.01	13.04	24.50	0.00		10.70			+	+
	(MS)			UEPCO	UEPMD	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD								50	2.00		12.70			1	1
	(AL, KY, LA, MS)		1	UEPCO	UEPRH	1.23	40.31	19.84	24.90	6.58		15.75				1
	2-Wire Coin Outward Operator Screening & Blocking: 900/976, 1+DDD, 011+, and															
	Local (AL, KY, LA, MS)		1	UEPCO	UEPCN	1.23	40.31	19.84	24.90	6.58		15.75				1
	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, and Local; with															
	Dialing Parity (MS)			UEPCO	UEPCS	1.23	40.31	19.84	24.90	6.58		15.75				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.23	40.31	19.84	24.90	6.58						
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.23	40.31	19.84	24.90	6.58						
ADDITIONA	AL UNE COIN PORT/LOOP (RC)															
	UNE Coin Port/Loop Combo Usage (Flat Rate)			UEPCO	URECU	4.62	0.00	0.00								
	MBER PORTABILITY															
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
FEATURES																
	RRING CHARGES - CURRENTLY COMBINED															
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPCO	USAC2		0.0988	0.0988				15.75				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with		1													1
	change			UEPCO	USACC		0.0988	0.0988				15.75				
ADDITIONA																
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0.00	0.00				15.75				
	OP COMBINATIONS - COST BASED RATES															
2-WIRE VO	ICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT															
	.oop Combination Rates	1								l	1					1
UNE Port/L	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			21.32										

CATEGO	RY RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Svc Order Submitte d Elec per LSR	Submitted Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs.	al Charge - Manual Svc Order vs. Electronic-	Increment - al Charge - Manual Svc Order vs Electronic- Disc Add'l
										curring						
						Rec		curring		onnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			34.98										
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 4		4			53.15										+
UN	E Loop Rates					00.10										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1 UE		UECD											
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2 UE		UECD											
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3 UE		UECD											
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 4		4 UE	PPX	UECD	45.72										
UN	E Port Rate	ļ		DDV	LIEDD	7.40	005.00	07.40	444.50	44.05		45.75			4.07	
NO	Exchange Ports - 2-Wire DID Port		UE	PPX	UEPD	7.43	225.96	87.13	114.59	14.25		15.75			1.97	
NO	PARECURRING CHARGES - CURRENTLY COMBINED 2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is		LIE	PPX	USAC*		7.35	1.88				15.75			1.97	-
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth		0.		00/10		7.00	1.00				10.70			1.07	
	Allowable Changes		UE	PPX	USA10	:	7.35	1.88				15.75			1.97	
AD	DITIONAL NRCs															
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk		UE	PPX	USAS′		26.94	26.94				15.75			1.97	
Te	lephone Number/Trunk Group Establisment Charges															
	DID Trunk Termination (One Per Port)			PPX	NDT	0.00	0.00	0.00				15.75			1.97	
	Additional DID Numbers for each Group of 20 DID Numbers			PPX	ND4	0.00	0.00	0.00				15.75			1.97	
	DID Numbers, Non- consecutive DID Numbers , Per Number			PPX	ND5	0.00	0.00	0.00				15.75 15.75			1.97 1.97	
	Reserve Non-Consecutive DID numbers Reserve DID Numbers			PPX	ND6 NDV	0.00	0.00	0.00				15.75			1.97	
LO	CAL NUMBER PORTABILITY		0.2	117	INDV	0.00	0.00	0.00				10.70			1.57	-
	Local Number Portability (1 per port) VIRE ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT		UE	PPX	LNPC	3.15	0.00	0.00								
	E Port/Loop Combination Rates															
				PPB												
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1			PPR		28.59										
	OW IODN District One Is I are fold IODN District Inc. Of Is Day of INF. 7 and O			PPB		05.00										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2 UE	PPR PPB		35.00										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3			PPR		45.18										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 4		4			67.61										
UN	E Loop Rates															
	2-Wire ISDN Digital Grade Loop - UNE Zone 1		1 UE	PPB UEF	PR USL2>	18.26						15.75			1.97	
	O Marc IODNI District Out In Land IINE 7 and O			PPB	1101.0	04.07						45.75			4.07	
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2 UE	PPR	USL2>	24.67						15.75			1.97	
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3 UE	PPB UEF	PR USL2	34.85						15.75			1.97	
	2-Wire ISDN Digital Grade Loop - UNE Zone 4	1	4 UE	PPB UEF	PR USL2>	57.28					ļ	15.75			1.97	
UN	E Port Rate	ļ		PPB UEP	DD LIEDDI	10.00	190.80	422.22	100.72	24.42		45.75			1.97	
NO	Exchange Port - 2-Wire ISDN Line Side Port NRECURRING CHARGES - CURRENTLY COMBINED	1	UE	TED UEP	IN DEPPI	10.33	190.80	133.22	100.72	21.13	1	15.75			1.97	+
NO	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -	1														+
	Conversion		UE	PPB UEP	PR USACI	0.00	38.73	27.17				15.75			1.97	
AD	DITIONAL NRCs															
LO	CAL NUMBER PORTABILITY															
	Level North and Device 1771 (Accessed)			.DDD 11EE	DD L NDO		0.00	0.00								
В.	Local Number Portability (1 per port)		UE	PPB UEF	PR LNPC	0.35	0.00	0.00								
D-(CHANNEL USER PROFILE ACCESS:															
	CVS/CSD (DMS/5ESS)		UE	PPB UEF	PR U1UC	0.00	0.00	0.00								
	CVS (EWSD)				PR U1UCE		0.00	0.00								
	CSD			PPB UEP	PR U1UC	0.00	0.00	0.00								
B-0	CHANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)															
	CVS/CSD (DMS/5ESS)		UE	PPB UEP	PR U1UCI	0.00	0.00	0.00								
	CVC (EWICD)			PPB UEF	PR U1UCE	0.00	0.00	0.00								
	CVS (EWSD)	1	UE	rrb UEF	-KUTUCI	0.00	0.00	0.00								+
	CSD		LIE	PPB UEF	PR U1UC	0.00	0.00	0.00								
US	ER TERMINAL PROFILE			- OLI	5.001	5.50	0.00	5.50								
	User Terminal Profile (EWSD only)		UE	PPB UEF	PR U1UM	0.00	0.00	0.00								
VE	RTICAL FEATURES		\vdash													
	All Vertical Features - One per Channel B User Profile		116	PPB UEF	PR UEPVI	2.56	0.00	0.00				15.75			1.97	
	pair vortical i catules - One per channel b Oset Fiolile	1	I DE	UEP	I INJUEF VI	2.00	0.00	0.00	1	1	1	10.75		1	1.97	

INTEROFFICE			Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Charge - Manual Svc Order vs. Electronic- 1st	vs.	Manual Svc Order vs.	vs. Electron
									Nonred	urring			•		•	
						Rec	Nonred	urrina	Disco				OSS RA	ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
			_				1 11 31	Auu	1 11 31	Auu	COMILO	COMPAR	COMPLET	COMPAR	COMPAR	COMA
			-													
																
	teroffice Channel mileage each, including first mile and facilities termination			UEPPB UEPPR		22.5298	40.77	27.57	17.26	7.11		15.75			1.97	
Int	teroffice Channel mileage each, additional mile			UEPPB UEPPR	M1GNM	0.0098	0.00	0.00								
4-WIRE DS1 D	DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															
UNE Port/Loor	p Combination Rates															
4V	W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP		155.43										
41/	W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2		2	UEPPP		205.74										
41/	W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		3	UEPPP		283.10										
	W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 4		4	UEPPP		534.81										
UNE Loop Rat																
	-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	79.08					<u></u>	15.75	1		1.97	
4-1	-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP	USL4P	129.38						15.75			1.97	
4-1	Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	206.74						15.75			1.97	
4.3	-Wire DS1 Digital Loop - UNE Zone 4		4	UEPPP	USL4P	458.46					l	15.75			1.97	
UNE Port Rate			+		30271	.50.40					1	10.73			1.57	
			1	LIEDDD	HEDDE	70.05	450.00	200 50	107.75	20.70	1	45.75	1	 	4.07	-
	xchange Ports - 4-Wire ISDN DS1 Port	-	1	UEPPP	UEPPP	76.35	458.93	260.59	127.75	32.76	 	15.75			1.97	
NONRECURRI	ING CHARGES - CURRENTLY COMBINED										ļ					
4-\	-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -															
Co	onversion -Switch-as-is			UEPPP	USACP	0.00	119.76	79.01				15.75			1.97	
ADDITIONAL N	NRCs															
4-1	-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos															
	ithin Std Allowance			UEPPP	PR7TF		0.49					15.75			1.97	
4-1	Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All			OLI I I			0.10					10.10			1.07	
	tates except NC)			UEPPP	PR7TO		11.58	11.58				15.75			1.97	
4.7	-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos			UEFFF	FK/10		11.30	11.30				15.75			1.97	
				LIEDDD	DD777		00.45	00.45				45.75			4.07	
	bove Std Allowance			UEPPP	PR7ZT		23.15	23.15				15.75			1.97	
	BER PORTABILITY															
	ocal Number Portability (1 per port)			UEPPP	LNPCN	1.75										
INTERFACE (F	Provsioning Only)															
Vo	pice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	igital Data			UEPPP	PR71D	0.00	0.00	0.00								
	ward Data			UEPPP	PR71E	0.00	0.00	0.00								
	onal "B" Channel			OLI I I		0.00	0.00	0.00								
			-	LIEDDD	DD7D1/	0.00	44.04					45.75			4.07	
	ew or Additional - Voice/Data B Channel		-	UEPPP	PR7BV	0.00	14.61					15.75			1.97	
Ne Ne	ew or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.61					15.75			1.97	
	ew or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.61					15.75			1.97	
Ne	ew or Additional Useage Sensitive Voice Data B Channel			UEPPP	PR7BS	0.00	14.61					15.75			1.97	
Ne	ew or Additional Useage Sensitive Digital Data B Channel			UEPPP	PR7BU	0.00	14.61					15.75			1.97	
CALL TYPES																
	ward			UEPPP	PR7C1	0.00	0.00	0.00								
	utward		1	UEPPP	PR7C0	0.00	0.00	0.00			 	 				
		-	1	UEPPP	PR7CC	0.00	0.00				1	1		 		\vdash
	wo-way	-	1	ULFFF	rK/UU	0.00	0.00	0.00			 	 	-		1	₩
Interoffice Cha	annei mileage		1								ļ				1	
	ixed Each Including First Mile			UEPPP	1LN1A	57.53	89.79	82.28	16.66	14.90		15.75			1.97	
	ach Airline-Fractional Additional Mile		<u></u>	UEPPP	1LN1B	0.20								<u> </u>		<u></u>
4-WIRE DS1 D	DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	1	1								1					1
	p Combination Rates															
	W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		131.78						15.75			1.97	
41/	W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		2	UEPDC		182.07					 	15.75			1.97	
		-				259.44					1			 	1.97	
	W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC							!	15.75		├		!
	W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 4		4	UEPDC		511.15					1	15.75		1	1.97	L
UNE Loop Rat											ļ					
4-1	-Wire DS1 Digital Loop - UNE Zone 1		1	UEPDC	USLDC	79.08					<u> </u>	15.75	<u> </u>	1	1.97	<u></u>
4-1	-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC	USLDC	129.38						15.75			1.97	
	-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	206.74						15.75			1.97	
	-Wire DS1 Digital Loop - UNE Zone 4		4	UEPDC	USLDC	458.46					İ	15.75			1.97	1
UNE Port Rate	A		+		55255	.50.40					†	10.73			1.57	
	-Wire DDITS Digital Trunk Port		 	UEPDC	UDD1T	52.70	457.12	254.70	120.96	14.61	1	15.75			1.97	
			1	ULFDU	ווטטט	32.70	407.12	∠54.70	120.90	14.01	1	15.75	1		1.97	
	ING CHARGES - CURRENTLY COMBINED		1								ļ				L	└
	-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is			UEPDC	USAC4		130.24	67.41			ļ	15.75			1.97	
	Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with		1									1				1
	S1 Changes			UEPDC	USAWA		130.24	67.41			<u> </u>	15.75	<u> </u>	1	1.97	<u></u>
4-1	-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with															1
CF	hange - Trunk	1	1	UEPDC	USAWB		130.24	67.41			1	15.75			1.97	1

MS

Version 4Q01:12/01/01

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charge Manual Svc Order vs. Electronic
									Nonre	curring						
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - NRC - Subsequent Channel															
				UEPDC	UDTTA		44.50	44.50				45.75			4.07	l
	Activation/Chan - 2-Way Trunk 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel			UEPDC	UDITA		14.56	14.56				15.75			1.97	
	Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		14.56	14.56				15.75			1.97	l
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan			OLI DO	ODTID		14.50	14.50				13.73			1.51	
	Inward Trunk w/out DID			UEPDC	UDTTC		14.56	14.56				15.75			1.97	l
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -															
	Inward Trunk with DID			UEPDC	UDTTD		14.56	14.56				15.75			1.97	l
	4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-															ı
	Way DID w User Trans			UEPDC	UDTTE		14.56	14.56				15.75			1.97	I
	ZERO SUBSTITUTION															L
	B8ZS -Superframe Format			UEPDC	CCOSF		0.00	600.00				15.75			1.97	
	B8ZS - Extended Superframe Format			UEPDC	CCOEF		0.00	600.00				15.75			1.97	
	lark Inversion															
	AMI -Superframe Format		1	UEPDC	MCOSF		0.00	0.00								
	AMI - Extended SuperFrame Format		1	UEPDC	МСОРО		0.00	0.00								
Telephone	Number/Trunk Group Establisment Charges															
	Telephone Number for 2-Way Trunk Group		1	UEPDC	UDTGX	0.00						15.75			1.97	
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0.00						15.75			1.97	
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0.00						15.75			1.97	
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0.00						15.75			1.97	
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0.00						15.75			1.97	-
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0.00	0.00	0.00				15.75			1.97	
	Reserve DID Numbers	l	<u> </u>	UEPDC	NDV	0.00	0.00	0.00				15.75			1.97	
Dedicated I	DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-V	Nire DDI	TS Tru													
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	57.33	89.79	82.28	16.86	14.90		15.75			1.97	
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles		-	UEPDC	1LNOA	0.20	0.00	0.00								
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)			UEPDC	1LNO2	0.00	0.00	0.00								—
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles		-	UEPDC	1LNOB	0.20	0.00		0.00							
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)			UEPDC	1LNO3	0.00	0.00	0.00	0.00							—
	Interoffice Channel Mileage - Additional rate per mile - 25+ miles			UEPDC	1LNOC	0.20	0.00	0.00	0.00							—
	Local Number Portability, per DS0 Activated		-	UEPDC	LNPCP	3.15	0.00	0.00	0.00							1
	Central Office Termininating Point 1 LOOP WITH CHANNELIZATION WITH PORT			UEPDC	CTG	0.00										
	DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations		-													
	em can have up to 24 combinations of rates depending on type and number of	norte u	and													
UNE DS1 Lo	on	ports u	Seu													
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	79.08	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	129.38	0.00	0.00								
	4-Wire DS1 Loop - UNE Zone 3	1		UEPMG	USLDC	206.74	0.00	0.00								<u> </u>
	4-Wire DS1 Loop - UNE Zone 4			UEPMG	USLDC	458.46	0.00	0.00				15.75			1.97	i
	Channelization Capacities (D4 Channel Bank Configurations)		-		30200	.50.40	0.00	0.00				10.70			1.51	i
	24 DSO Channel Capacity - 1 per DS1		1	UEPMG	VUM24	95.06	0.00	0.00				15.75			1.97	i
	48 DSO Channel Capacity - 1 per 2 DS1s		1	UEPMG	VUM48	190.12	0.00					15.75			1.97	
	96 DSO Channel Capacity -1per 4 DS1s		1	UEPMG	VUM96	380.24	0.00					15.75			1.97	
	144 DS0 Channel Capacity - 1 per 6 DS1s		1	UEPMG	VUM14	570.36	0.00	0.00				15.75			1.97	
	192 DS0 Channel Capacity -1 per 8 DS1s		1	UEPMG	VUM19	760.48	0.00					15.75			1.97	
	240 DS0 Channel Capacity - 1 per 10 DS1s		1	UEPMG	VUM20	950.60	0.00	0.00				15.75			1.97	
	288 DS0 Channel Capacity - 1 per 12 DS1s		1	UEPMG	VUM28	1,140.72	0.00	0.00				15.75			1.97	i
	384 DS0 Channel Capacity - 1 per 16 DS1s		1	UEPMG	VUM38	1,520.96	0.00	0.00				15.75			1.97	i
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	1,901.20	0.00	0.00				15.75			1.97	
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2,281.44	0.00	0.00				15.75			1.97	
	672 DS0 Channel Capacity - 1 per 28 DS1s			UEPMG	VUM67	2,661.68	0.00	0.00				15.75			1.97	
	ring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with F	ort - Co	nversio	on Charge Based o												
	System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24															
Multiples of	f this configuration functioning as one are considered Add'l after the minimum	system	config	uration is counted												
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes			UEPMG	USAC4	0.00	151.35	8.41				15.75	-		1.97	
	ditions at End User Locations Where 4-Wire DS1 Loop with Channelization wit	h Port C	ombina	tion Currently Exis	ts and											
New (Not C	urrently Combined) In GA, KY, LA, MS & TN Only															
	1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Fea Activation - New															
	GA, LA, KY, MS, &TN Only			UEPMG	VUMD4	0.00	715.15	327.39	148.05	17.56		15.75			1.97	
	ero Substitution															
	Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0.00	0.00					15.75			1.97	
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	600.00				15.75			1.97	
	ark Inversion (AMI)		1													
	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								

CATEG	GORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			Submitte d Elec	Submitted	Charge - Manual Svc	Incrementa I Charge - Manual Svc Order vs. Electronic- Add'I	Manual Svc Order vs.	al Charge Manual Svc Orde vs. Electronic
										Nonre	curring						
							Rec	Nonrec	urring	Disco	onnect				ATES (\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Extended Superframe Format			UEPMG	MCOPO	0.00	0.00	0.00								
	Exchange l	Ports Associated with 4-Wire DS1 Loop with Channelization with Port															
	Exchange I																
	_xonango .	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.23	0.00	0.00	0.00	0.00		15.75			1.97	
		Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.23	0.00	0.00	0.00	0.00		15.75			1.97	
		Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.23	0.00	0.00	0.00	0.00		15.75			1.97	
		2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	7.40	0.00	0.00	0.00	0.00		15.75			1.97	
	A				UEPPA	UEPDINI	7.40	0.00	0.00	0.00	0.00		15.75			1.97	
	reature Ac	tivations - Unbundled Loop Concentration		-	LIEBBY (
		Feature (Service) Activation for each Line Side Port Terminated in D4 Bank	1		UEPPX	1PQWM	0.61	25.36	13.39	4.29	4.26		15.75			1.97	
		Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.61	78.03	18.39	60.66	11.85		15.75			1.97	
		Number/ Group Establishment Charges for DID Service															
		DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00				15.75			1.97	
		DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0.00	0.00	0.00				15.75			1.97	
	-	Non-Consecutive DID Numbers - per number		1	UEPPX	ND5	0.00	0.00	0.00				15.75	-		1.97	
		Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00				15.75			1.97	
		Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00				15.75			1.97	
		ber Portability							0.00								
		Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
		i - Vertical and Optional	+	<u> </u>	OLITA	LIVICI	3.13	0.00	0.00								
		ching Features Offered with Line Side Ports Only			LIEBBY												
		All Features Available			UEPPX	UEPVF	2.56	0.00	0.00				15.75			1.97	
		OP COMBINATIONS - MARKET RATES															
		es shall apply where BellSouth is not required to provide unbundled local swit	ching or	switch	ports per FCC an	d/or State	Commission	rules.									
		narios include:															
		led port/loop combinations that are Not Currently Combined in Alabama, Flori	da, North	Caroli	na and South Care	olina.											
	1. Unbund						outh's region	for end user	s with 4 or mor	e DS0 eau	ivalent lines.						
	1. Unbund 2. Unbund	led port/loop combinations that are Currently Combined or Not Currently Com	bined in	Zone 1	of the Top 8 MSA	S in BellS	outh's region	for end user	s with 4 or mor	e DS0 equ	ivalent lines.): TN (Nashv	ille).					
	1. Unbund 2. Unbund The Top 8 f	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar	bined in ta); LA (Zone 1 New Or	of the Top 8 MSA leans); NC (Green	S in BellS sboro-Wii	nston Salem-H	lighpoint/Cha	arlotte-Gastoni	a-Rock Hill); TN (Nashv	ille).	NC and SC.	In the interi	m where Bel	South cann	ot bill
- - - -	1. Unbundl 2. Unbundl The Top 8 I BellSouth c	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a	bined in ta); LA (Zone 1 New Or	of the Top 8 MSA leans); NC (Green	S in BellS sboro-Wii	nston Salem-H	lighpoint/Cha	arlotte-Gastoni	a-Rock Hill); TN (Nashv	ille).	NC and SC.	. In the interi	m where Bel	South cann	ot bill
	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states.	bined in hta); LA (and non-r	Zone 1 New Or ecurrir	of the Top 8 MSA leans); NC (Green g Market Rates in	S in BellS sboro-Wir this secti	nston Salem-H on except for	lighpoint/Cha nonrecurring	rlotte-Gastoni charges for n	a-Rock Hill not current); TN (Nashv ly combined	ille). in AL, FL,					ot bill
, , , , , , , , , , , , , , , , , , ,	1. Unbund 2. Unbund The Top 8 I BellSouth o The Market End Office	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office For Not Cui	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rrently Combined scenarios where Market Rates apply, the Nonrecurring char-	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
i I I NBUNDLE	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office For Not Cur O CENTREX	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: Ft. (Orlando, Ft. Lauderdale, Miami); GA (Atlar zurrently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rrently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS)	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office For Not Cui D CENTREX UNBUNDLE	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por reently Combined scenarios where Market Rates apply, the Nonrecurring char; PORT/LOOP COMBINATIONS - COST BASED RATES	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office For Not Cui D CENTREX UNBUNDLE UNE-P CEN	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar uurrently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chart PORTLOOP COMBINATIONS ED PORT/LOOP COMBINATIONS TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office O CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar uurrently is developing the billing capability to mechanically bill the recurring at Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS). DORT/LOOP COMBINATIONS COST BASED RATES ITREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only). Loop/2-Wire Voice Grade Port (Centrex) Combo	bined in hta); LA (ind non-r	Zone 1 New Or ecurring of this	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth of The Market End Office For Not Cui UNBUNDLE UNBUNDLE UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/Lo	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por trently Combined scenarios where Marker Rates apply, the Nonrecurring chantly PORT/LOOP COMBINATIONS - COST BASED RATES ITREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combooop Combination Rates (Non-Design)	bined in hta); LA (ind non-r	Zone 1 New Or recurrir of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add	S in BellS sboro-Win this secti apply to a	nston Salem-Hon except for III combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth of The Market End Office For Not Cui UNBUNDLE UNBUNDLE UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/Lo	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar uurrently is developing the billing capability to mechanically bill the recurring at Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS). DORT/LOOP COMBINATIONS COST BASED RATES ITREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only). Loop/2-Wire Voice Grade Port (Centrex) Combo	bined in hta); LA (ind non-r	Zone 1 New Or recurrir of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall	S in BellS sboro-Win this secti apply to a	nston Salem-F on except for Il combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth of The Market End Office For Not Cui UNBUNDLE UNBUNDLE UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/Lo	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar surrently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chan (FORTILOOP COMBINATIONS DEPORTILOOP COMBINATIONS TOPORTILOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop)/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this isted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add	S in BellS sboro-Win this secti apply to a	nston Salem-Hon except for III combination	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth of The Market End Office For Not Cui UNBUNDLE UNBUNDLE UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/Lo	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this isted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add	S in BellS sboro-Win this secti apply to a	nston Salem- on except for Il combination C columns for 12.22 17.13	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I NBUNDLEC	1. Unbundl 2. Unbundl The Top 8 I BellSouth c The Market End Office For Not Cui O CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/Le	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por trently Combined scenarios where Market Rates apply, the Nonrecurring chantly PORT/LOOP COMBINATIONS DEPORT/LOOP COMBINATIONS COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	bined in hta); LA (ind non-r	Zone 1 New Or recurring of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	nston Salem- on except for Il combination C columns for 12.22 17.13 26.26	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE I	1. Unbundl 2. Unbundl 2. Unbundl The Top 8 I BellSouth c Fhe Market End Office For Not Cur D CENTREX UNBUNDLE UNE-P CEN 2-Wire VG L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar uurrently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chan (FORTILOOP COMBINATIONS) DPORTILOOP COMBINATIONS DPORTILOOP COMBINATIONS - COST BASED RATES ITREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this isted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add	S in BellS sboro-Win this secti apply to a	nston Salem- on except for Il combination C columns for 12.22 17.13	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE I	1. Unbundi 2. Unbundi 2. Unbundi The Top 8 I BellSouth c The Market End Office For Not Cui 0 CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por reently Combined scenarios where Market Rates apply, the Nonrecurring char. (PORTILOOP COMBINATIONS DOT COMBINATIONS DOT COMBINATIONS FOR THE STRES (Valid in AL, FL, GA, KY, LA, MS, & TN only). Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design). 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 0 Combination Rates (Design)	bined in hta); LA (ind non-r	Zone 1 New Or recurring of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	nston Salem- on except for Il combination C columns for 12.22 17.13 26.26 44.91	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE I	1. Unbundi 2. Unbundi 2. Unbundi The Top 8 I BellSouth c The Market End Office For Not Cui 0 CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states, and Tandem Switching Usage and Common Transport Usage rates in the Por rrently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS : OST BASED RATES DPORT/LOOP COMBINATIONS : OST BASED RATES DPORT/LOOP COMBINATIONS : OST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) . Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design [2-Wire VG Lo	bined in hta); LA (ind non-r	Zone 1 New Or recurrir of this isted in 1 2 3 4	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	nston Salem- on except for Il combination C columns for 12.22 17.13 26.26 44.91	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE I	1. Unbundi 2. Unbundi 2. Unbundi The Top 8 I BellSouth c The Market End Office For Not Cui 0 CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only). Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Non-Port Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design Occubination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this isted in 1 2 3 4 1 2	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	12.22 17.13 26.26 44.91 19.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
NBUNDLE I	1. Unbundi 2. Unbundi 2. Unbundi The Top 8 I BellSouth c The Market End Office For Not Cui 0 CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring charge (PORT/LOOP COMBINATIONS - COST BASED RATES DEPORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo opp Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this sted in 1 2 3 4 1 2 3	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	12.22 17.13 26.26 44.91 19.98 28.78	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
INBUNDLEC	I. Unbundi I. Unbundi The Top 8 I BellSouth of The Market Fon Not Cui CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS). D PORT/LOOP COMBINATIONS - COST BASED RATES ITREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, &TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo op Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this sted in 1 2 3 4 1 2 3	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	12.22 17.13 26.26 44.91 19.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
INBUNDLEC	1. Unbundi 2. Unbundi 2. Unbundi The Top 8 I BellSouth c The Market End Office For Not Cui 0 CENTREX UNBUNDLE UNE-P CEN 2-Wire VG I UNE Port/L UNE Port/L	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS). D PORT/LOOP COMBINATIONS - COST BASED RATES ITREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, &TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo op Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design op Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this sted in 1 2 3 4 1 2 3	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Wit this secti apply to a itional NR	12.22 17.13 26.26 44.91 19.98 28.78	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
INBUNDLEC	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring char (PORTILOOP COMBINATIONS DOTTO COMBINATIONS DOTTO COMBINATIONS COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire	bined in hta); LA (ind non-r	Zone 1 New Or ecurrir of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Win this secti apply to a	12.22 17.13 26.26 44.91 15.12 19.98 28.78 46.95	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
INBUNDLEC	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por Trently Combined scenarios where Market Rates apply, the Nonrecurring chant (PORT/LOOP COMBINATIONS) B PORT/LOOP COMBINATIONS D PORT/LOOP COMBINATIONS D PORT/LOOP COMBINATIONS OB PORT/LOOP COMBINATIONS OB PORT/LOOP COMBINATIONS OB PORT/LOOP COMBINATIONS OB PORT/LOOP COMBINATIONS - COST BASED RATES ITREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1	bined in hta); LA (I and non-r	Zone 1 New Or ecurrir of this sted in 1 2 3 4 1 2 3 4	of the Top 8 MSA leans); NC (green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Wiit this secti apply to a itional NR	12.22 17.13 26.26 44.91 15.12 19.98 46.95	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
INBUNDLEC	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS) DPORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (I and non-r	Zone 1 New Or ecurrir of this sted in	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91 UEP91	S in BellS sboro-Wint this secti apply to a litional NR	12.22 17.13 26.26 44.91 15.12 19.98 28.78 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I I I I I I I I I I I I I I I I I	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por reently Combined scenarios where Market Rates apply, the Nonrecurring charge (PORTILOOP COMBINATIONS DOTORTILOOP COMBINATIONS DOTORTILOOP COMBINATIONS DOTORTILOOP COMBINATIONS COST BASED RATES TIRES - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	bined in hta); LA (I and non-r	Zone 1 New Or ecurrir of this isted in 1 2 3 4 1 1 2 3 4 1 1 2 3 4	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS sboro-Wint this secti apply to a itional NR	12.22 17.13 26.26 44.91 15.12 19.98 10.98 15.91 25.04	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I I I I I I I I I I I I I I I I I	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS). D PORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, &TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG Loop/2-Wire VG	bined in hta); LA (I and non-r	Zone 1 New Or Proceedings of this state in 1 1 2 3 3 4 4 1 1 2	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS1 UECS1 UECS1 UECS1 UECS1 UECS1 UECS1	12.22 17.13 16.26 19.28 19.98 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring char (PORTILOOP COMBINATIONS DOTORTILOOP COMBINATIONS DOTORTILOOP COMBINATIONS OF THE STREX SET ALESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire Voice Grade Loop (S. 1) - Zone 2 (2-Wire Voice Grade Loop (S. 1) - Zone 2 (2-Wire Voice Grade Loop (S. 1) - Zone 2 (2-Wire Voice Grade Loop (S. 1) - Zone 3 (2-Wire Voice Grade Loop (S. 1) - Zone 4 (2-Wire Voice Grade Loop (S. 1) - Zone 4 (2-Wire Voice Grade Loop (S. 1) - Zone 4 (2-Wire Voice Grade Loop (S. 1) - Zone 4 (2-Wire Voice Grade Loop (S. 1) - Zone 1	bined in hta); LA (I and non-r	Zone 1 New Or recurring of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 1 2 1 3 3 4 4 1 1 1 2 1 3 3 4 4 1 1 1 1 2 1 3 3 4 4 1 1 1 1 2 1 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS sboro-Winths section this section apply to a sitional NR UECS1 UECS1 UECS1 UECS1 UECS1 UECS1 UECS1 UECS1	12.22 17.13 26.26 44.91 15.12 19.98 28.78 46.95 10.98 15.91 25.04 43.68	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rerently Combined scenarios where Market Rates apply, the Nonrecurring charge (PORT/LOOP COMBINATIONS - COST BASED RATES DEPORT/LOOP COMBINATIONS - COST BASED RATES DEPORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this steed in 1 2 3 4 4 1 1 2 2 3 4 4 1 1 2 2 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 2 2 3 3 3 4 4 1 1 1 2 2 3 3 3 4 4 1 1 1 1 2 2 3 3 3 4 4 1 1 1 1 2 2 3 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	12.22 17.13 15.12 19.98 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS) D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 3	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS siboro-Winths section apply to a itional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	12.22 17.13 15.04 19.98 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	I. Unbundid. I. Unbundid. Ine Top 8 I BellSouth of The Market Lend of The Market Lend Office For Office For Office OF The Market Lend OF Lend	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rerently Combined scenarios where Market Rates apply, the Nonrecurring charge (PORT/LOOP COMBINATIONS - COST BASED RATES DEPORT/LOOP COMBINATIONS - COST BASED RATES DEPORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only)	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS1 UECS2 UECS2 UECS2	12.22 17.13 15.12 19.98 10.98 11.98 11.98 11.98 11.98 11.98 11.99 11.98 11.99	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I I I I I I I I I I I I I I I I I	I. Unbundi I. Unbundi I. Unbundi The Top 8 I BellSouth o BellSouth o The Market End Office For Not Cur O CENTREX JUNEUNDLE JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE PORT/L JUNE LOOP F	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS) D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 3	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS siboro-Winths section apply to a itional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	12.22 17.13 15.04 19.98 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hill not current nents exce); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS) D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 3	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS sboro-Withis section this section apply to a sitional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2	12.22 17.13 15.04 19.98 10.98	lighpoint/Cha nonrecurring as of loop/po	arlotte-Gastoni g charges for n rt network elen	a-Rock Hilling); TN (Nashv ly combined pt for UNE 0	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS - COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only). Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Non-Gerian Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 1 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS siboro-Winths section apply to a itional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2	12.22 17.13 26.26 44.91 15.12 19.98 28.78 46.95 10.98 15.91 25.04 43.68 13.89 18.75 27.55 45.72	ilighpoint/Chi-	rlotte-Gastoni g charges for n t network elen SOC. For Cur	a-Rock Hilling); TN (Nashvhy combined pt for UNE C bined scena	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage NRC - Curre	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por reently Combined scenarios where Market Rates apply, the Nonrecurring charge (PORT/LOOP COMBINATIONS DOTO COMBINATIONS DOTO COMBINATIONS DOTO COMBINATIONS COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire Voice Grade Loop (St. 1) - Zone 1 2-Wire Voice Grade Loop (St. 1) - Zone 2 2-Wire Voice Grade Loop (St. 1) - Zone 2 2-Wire Voice Grade Loop (St. 1) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 3 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 3 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Port (Centrex) Basic Local Area	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECYA	12.22 17.13 12.22 17.13 26.26 44.91 15.12 19.98 46.95 10.98 15.91 25.04 43.68 13.89 13.89 13.89 14.75 27.55 45.72	ilighpoint/Chi-di	rlotte-Gastoni g charges for n rt network elen SOC. For Cur	a-Rock Hilliam not current nents excer rently Com); TK (Nashvhy y combined pt for UNE C bined scena	ille). in AL, FL, Coin Port/L	oop Combi	nations whic	h have a flat	rate usage NRC - Curre	
IBUNDLEC	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS) D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (Grade Loop (Grade Loop (Grade Loop (Grade Loop (Grade Loop (bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS shoro-Winths section apply to a itional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS	12.22 17.13 16.25 19.98 19.98 10.98	Highpoint/Chi- nonrecurring ss of loop/por r each Port U	rlotte-Gastoni j charges for n rt network elen SOC. For Curr	a-Rock Hilliams and Communication of Current sexcerently Communication of	j; TK (Nashvhy combined by combined pt for UNE C bined scena	ille). in AL, FL, Coin Port/L	oop Combi onrecurring	nations whic	h have a flat	rate usage NRC - Curre	
IBUNDLEC	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a: Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring char (PORTILOOP COMBINATIONS DOPORTILOOP COMBINATIONS DOPORTILOOP COMBINATIONS TOST BASED RATES TIREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Rate 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Loop (SL 2) - Zone 4 2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex) Basic Local Area 2-Wire Voice Grade Port (Centrex) Basic Local Area	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS2	12.22 17.13 16.26 19.98 19.98 19.98 10.88 10.88	ilighpoint/Chi- nonrecurring is of loop/por reach Port U	19.84	a-Rock Hilliam and contract the); TN (Nashvhy combined pt for UNE C bined scena c c c c c c c c c c c c c c c c c c c	ille). in AL, FL, Coin Port/L	15.75 15.75	nations whic	h have a flat	1.97 1.97	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	Ied port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chart (PORT/LOOP COMBINATIONS). D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo op Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design opp Combination Rates (Bosign) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design opp Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Opp Combination Rates (Design) 2-Wire Voice Grade Loop (St. 1) - Zone 1 2-Wire Voice Grade Loop (St. 1) - Zone 1 2-Wire Voice Grade Loop (St. 1) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 1 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Loop (St. 2) - Zone 2 2-Wire Voice Grade Port (Centrex) Bots Local Area 2-W	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS shoro-Winths section apply to a itional NR UECS1 UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UECS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS2 UECYS	12.22 17.13 16.25 19.98 19.98 10.98	Highpoint/Chi- nonrecurring ss of loop/por r each Port U	rlotte-Gastoni j charges for n rt network elen SOC. For Curr	a-Rock Hilliams and Communication of Current sexcerently Communication of	j; TK (Nashvhy combined by combined pt for UNE C bined scena	ille). in AL, FL, Coin Port/L	oop Combi onrecurring	nations whic	h have a flat	rate usage NRC - Curre	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS DEPORT/LOOP COMBINATIONS DEPORT/LOOP COMBINATIONS COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Non-Design Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design oop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Port (Centrex MO Lond Area 2-Wire Voice Grade Port (Centrex Mo Lond Area 2-Wire Voice Grade Port (Centrex Wire Caller ID)1Basic Local Area 2-Wire Voice Grade Port (Centrex Wire Caller ID)1Basic Local Area 2-Wire Voice Grade Port (Centrex Wire Caller ID)1Basic Local Ferm - Basic Local Port (Centrex Wire Cente	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UECS2 UEPYA UEPYM UEPYM UEPYM	12.22 17.13 26.26 44.91 15.12 19.98 46.95 10.98 15.91 25.04 43.68 13.89 18.75 27.55 45.72	40.31 40.31 108.35	19.84 19.84 70.57	a-Rock Hilliams and contract the sex ceremity Communication (Communication) and the sex ceremity (Communication) and the sex); TN (Nashvhy combined pt for UNE C bined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scena for the combined scenarios for t	ille). in AL, FL, Coin Port/L	15.75 15.75 15.75	nations whic	h have a flat	1.97 1.97	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a: Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring char. PORTILOOP COMBINATIONS DOPTON COMBINATIONS DOPTON COMBINATIONS COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire Voice Grade Loop (St. 1) - Zone 1 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 2) - Zone 3 (2-Wire Voice Grade Loop (St. 2) - Zone 3 (2-Wire Voice Grade Loop (St. 2) - Zone 2 (2-Wire Voice Grade Loop (St. 2) - Zone 2 (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Wire Voice Grade Port (Centrex) Basic Local Area	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS sboro-With section this section apply to a itional NR lead of the section of the sect	12.22 17.13 12.22 17.13 26.26 44.91 15.12 19.98 46.95 10.98 13.89 13.89 13.89 14.75 27.55 45.72 1.23 1.23 1.23 1.23	40.31 40.31 108.35	19.84 19.84 70.57	a-Rock Hilling and Communication of Current when the except and the communication of Commun	p; TN (Nashvhy combined pt for UNE C bined scena for UNE C bined s	ille). in AL, FL, Coin Port/L	15.75 15.75 15.75 15.75	nations whic	h have a flat	1.97 1.97 1.97	
I I I I I I I I I I I I I I I I I I I	I. Unbundid. I. Unbundid. In Top 8 I BellSouth or BellSou	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring chark (PORT/LOOP COMBINATIONS). D PORT/LOOP COMBINATIONS - COST BASED RATES TREX - 1AESS - (Valid in AL,FL,GA,KY,LA,MS,&TN only) Loop/2-Wire Voice Grade Port (Centrex) Combo oop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design Rate 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 2 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Port (Centrex Not remination)Basic Local Area 2-Wire Voice Grade Port (Centrex Wirth Caller ID)1Basic Local Area 2-Wire Voice Grade Port (Centrex Wirth	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	UECS1 UECS1 UECS2 UECS2 UECS2 UEPYA UEPYA UEPYM UEPYM UEPYZ UEPYZ UEPYZ	12.22 17.13 26.26 44.91 15.12 19.98 28.78 46.95 10.98 15.91 25.04 43.68 13.89 18.75 27.55 45.72	40.31 40.31 108.35 108.35	19.84 19.84 70.57 19.84	a-Rock Hilling and contract the sex cerently Communication (Communication) and contract the sex cerently Communication (Communication) and contract the sex cerently Communication (Communication) and contract the sex cere	0.7 TN (Nashvhy combined pt for UNE C bined scena c c c c c c c c c c c c c c c c c c c	ille). in AL, FL, Coin Port/L	15.75 15.75 15.75	nations whic	h have a flat	1.97 1.97 1.97	
IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII	I. Unbundil I. Unbundil The Top 8 I BellSouth or Top 8 I BellSouth or Top 8 I BellSouth or Top 8 I BellSouth or Curl O CENTREX UNBUNDLE UNBE P CEN UNBE P OT 1 UNE P OT 1	led port/loop combinations that are Currently Combined or Not Currently Com MSAs in BellSouth's region are: FL (Orlando, FL Lauderdale, Miami); GA (Atlar currently is developing the billing capability to mechanically bill the recurring a: Rate for unbundled ports includes all available features in all states. and Tandem Switching Usage and Common Transport Usage rates in the Por rently Combined scenarios where Market Rates apply, the Nonrecurring char. PORTILOOP COMBINATIONS DOPTON COMBINATIONS DOPTON COMBINATIONS COST BASED RATES TIREX - 1AESS - (Valid in AL, FL, GA, KY, LA, MS, & TN only) Loop/2-Wire Voice Grade Port (Centrex) Comboop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design (2-Wire Voice Grade Loop (St. 1) - Zone 1 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 1) - Zone 2 (2-Wire Voice Grade Loop (St. 2) - Zone 3 (2-Wire Voice Grade Loop (St. 2) - Zone 3 (2-Wire Voice Grade Loop (St. 2) - Zone 2 (2-Wire Voice Grade Loop (St. 2) - Zone 2 (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Basic Local Area (2-Wire Voice Grade Port (Centrex) Wire Voice Grade Port (Centrex) Basic Local Area	bined in hta); LA (I and non-r	Zone 1 New Or recurring the control of this isted in 1 2 3 4 4 1 1 2 3 3 4 4 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 2 2 3 3 4 4 1 1 1 1 2 2 3 3 4 4 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	of the Top 8 MSA leans); NC (Green g Market Rates in rate exhibit shall the First and Add UEP91	S in BellS sboro-With section this section apply to a itional NR lead of the section of the sect	12.22 17.13 12.22 17.13 26.26 44.91 15.12 19.98 46.95 10.98 13.89 13.89 13.89 14.75 27.55 45.72 1.23 1.23 1.23 1.23	40.31 40.31 108.35	19.84 19.84 70.57	a-Rock Hilling and a record to the current of the c	p; TN (Nashvhy combined pt for UNE C bined scena for UNE C bined s	ille). in AL, FL, Coin Port/L	15.75 15.75 15.75 15.75	nations whic	h have a flat	1.97 1.97 1.97	

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charg Manua Svc Ord vs. Electron
									Nonred	urrina	per ==::	, p		1		1 - 1 - 1 - 1 - 1
						Rec	Nonrec	urring	Disco				OSS R	ATES (\$)		
					+ -	1100	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
					_		FIISL	Auu i	FIISL	Auu i	SOWIEC	SUMAN	SUMAN	SUMAN	SUMAN	SUMA
	2-Wire Voice Grade Port (Centrex)			UEP91	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP91	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP91	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP91	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
			-	UEP91	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term															+
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP91	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP91	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
Local Switch	hing															
	Centrex Intercom Funtionality, per port			UEP91	URECS	0.7947										
	per Portability			OLI UI	UNLOU	0.7017										+
				LIEDO4	LNDOO	0.05										+
	Local Number Portability (1 per port)	-		UEP91	LNPCC	0.35						1		l		
Features												1				
	All Standard Features Offered, per port		<u> </u>	UEP91	UEPVF	2.56						15.75			1.97	
	All Select Features Offered, per port	1		UEP91	UEPVS	0.00	404.98					15.75			1.97	1
	All Centrex Control Features Offered, per port			UEP91	UEPVC	2.56						15.75			1.97	
NARS	2 2	1			021 VO	2.00						10.75			1.57	
	Liphyadlad Naturak Access Register, Combination	1	1	LIED01	UARCX	0.00	0.00	0.00			1	1	1		1	+
	Unbundled Network Access Register - Combination			UEP91		0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP91	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP91	UAROX	0.00	0.00	0.00								
Miscellaneo	ous Terminations															
2-Wire Trun																1
	Trunk Side Terminations, each			UEP91	CENA6	8.25	120.00	18.85	61.77	3.88		15.75			1.97	+
				OLF 91	CLIVAU	0.23	120.00	10.00	01.77	3.00		13.73			1.31	-
interoffice C	Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination - Voice Grade			UEP91	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75			1.97	
	Interoffice Channel mileage, per mile or fraction of mile			UEP91	MIGBM	0.0098										
Feature Act	tivations (DS0) Centrex Loops on Channelized DS1 Service															
	I Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP91	1PQWS	0.57										
			-													+
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP91	1PQW6	0.57										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP91	1PQW7	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP91	1PQWP	0.57										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP91	1PQWV	0.57										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP91	1PQWQ	0.57										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP91	1PQWA	0.57										—
				OLI 91	II QWA	0.57										-
	ring Charges (NRC) Associated with UNE-P Centrex															
	Conversion - Currently Combined Switch-As-Is with allowed changes, per port			UEP91	USAC2		0.10	0.10				15.75			1.97	
	Conversion of Existing Centrex Common Block			UEP91	USACN		37.97	16.68								
	New Centrex Standard Common Block			UEP91	M1ACS	0.00	666.32					15.75			1.97	
	New Centrex Customized Common Block			UEP91	M1ACC	0.00	666.32					15.75			1.97	
	Secondary Block, per Block			UEP91	M2CC1	0.00	77.91					15.75			1.97	
	NAR Establishment Charge, Per Occasion	1	1	UEP91	URECA	0.00	72.63				1		1		1.97	
		-	-	OLIBI	UNEUA	0.00	12.03					15.75			1.97	+
	TREX - 5ESS (Valid in All States)	1	1								1	ļ				1
	Loop/2-Wire Voice Grade Port (Centrex) Combo											1				
	oop Combination Rates (Non-Design)		<u> </u>													
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	1	UEP95		12.22										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design			UEP95		17.13						1				
	2-Wire VG Loop/2-Wire Voice Grade Fort (Centrex)Fort Combo - Non-Design	1		UEP95		26.26						t				
		-			+ -						-	1				+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	-	4	UEP95		44.91						1				+
	oop Combination Rates (Design)											ļ				1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP95		15.12		<u> </u>								
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design	1	2	UEP95		19.98										1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP95		28.78						1				1
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design			UEP95		46.95					l	1				
UNE Loop R		 	7	J_1 JJ	1 -	+0.55					1	1		 		
		1	-	LIEDOE	LIEGO	40.00					1	 	 		 	+
	2-Wire Voice Grade Loop (SL 1) - Zone 1	1	1	UEP95	UECS1	10.98					1	ļ				1
	2-Wire Voice Grade Loop (SL 1) - Zone 2	1	2	UEP95	UECS1	15.91						1				1
	2-Wire Voice Grade Loop (SL 1) - Zone 3	1	3	UEP95	UECS1	25.04										1
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP95	UECS1	43.68						1				
	2-Wire Voice Grade Loop (SL 2) - Zone 1	1	1	UEP95	UECS2	13.89						1	 		 	†
		-	2								1	1				+
	2-Wire Voice Grade Loop (SL 2) - Zone 2	1	_	UEP95	UECS2	18.75					1	ļ				4
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	27.55						1				
	2-Wire Voice Grade Loop (SL 2) - Zone 4	<u> </u>	4	UEP95	UECS2	45.72		l			<u> </u>	L		<u></u>		
UNE Port Ra	ate															
		1	1								1	1				t
All States																

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.		al Charge Manual Svc Orde vs.
						_			Nonre							
						Rec		urring		nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local			UEP95	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic															
	Local Area			UEP95	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local			UEP95	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	1
	MS, SC, & TN Only															1
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	1
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	+
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75		+	1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated in on Wegalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75		 	1.97	
FL & GA On				UEP95	UEPQZ	1.23	40.31	19.84	24.90	6.58					1.97	
												15.75			1.97	-
Local Switc				LEDAS										_		
	Centrex Intercom Funtionality, per port			UEP95	URECS	0.7947								_		
	per Portability													<u> </u>		
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Features																
	All Standard Features Offered, per port			UEP95	UEPVF	2.56						15.75			1.97	
	All Select Features Offered, per port			UEP95	UEPVS	0.00	404.98					15.75			1.97	
	All Centrex Control Features Offered, per port			UEP95	UEPVC	2.56						15.75			1.97	
NARS																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00								Ī
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00								1
	ous Terminations					0.00									-	+
2-Wire Trun															-	+
	Trunk Side Terminations, each			UEP95	CEND6	8.25	120.00	18.85	61.77	3.88		15.75			1.97	+
	al (1.544 Megabits)			02.00	OL. IDO	0.20	120.00	10.00	01	0.00		10.10		+	1.07	+
	DS1 Circuit Terminations, each			UEP95	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75			1.97	+
	DS0 Channels Activated, each			UEP95	M1HDO	0.00	14.56	00.20		2.01		10.10		+	1.07	+
	Channel Mileage - 2-Wire			OLI 30	WITTE	0.00	14.00							 		+
	Interoffice Channel Facilities Termination			UEP95	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75		 	1.97	+
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0098	40.77	21.51	17.20	7.11		13.73			1.57	+
Footure Act	tivations (DS0) Centrex Loops on Channelized DS1 Service			UEF93	IVIIGDIVI	0.0096								 	 	+
														 	 	+
	Bank Feature Activations			UEP95	1PQWS	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot					0.57										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP95	1PQW6	0.57								ļ	ļ	
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP95	1PQW7	0.57								<u> </u>		
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP95	1PQWP	0.57								<u> </u>		
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.57										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP95	1PQWQ	0.57										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.57										
Non-Recurr	ing Charges (NRC) Associated with UNE-P Centrex															
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port			UEP95	USAC2		0.10	0.10				15.75			1.97	
	Conversion of Existing Centrex Common Block, each			UEP95	USACN		37.97	16.68								
	New Centrex Standard Common Block			UEP95	M1ACS	0.00	666.32					15.75			1.97	
	New Centrex Customized Common Block			UEP95	M1ACC	0.00	666.32					15.75			1.97	
	NAR Establishment Charge, Per Occasion			UEP95	URECA	0.00	72.63					15.75			1.97	T T
	TREX - DMS100 (Valid in All States)															1
	.oop/2-Wire Voice Grade Port (Centrex) Combo													+	 	+
	pop Combination Rates (Non-Design)	1	1									1		 	 	†
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	†	1	UEP9D		12.22						1		 	 	+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	+	2	UEP9D	+	17.13						1		 	+	+
		+			-									 	 	+
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design	+	3	UEP9D	+	26.26					-				 	+
LINE S "	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design	1	4	UEP9D		44.91									 	+
	pop Combination Rates (Design)	-	-	LIEBAR		,						1			 	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	1	1	UEP9D		15.12										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9D		19.98										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP9D		28.78								L	<u> </u>	
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design	1	4	UEP9D		46.95									1	
	Rate	1	1	-	1			1			1	1	1	1	1	1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	vs. Electronic-	
										curring						
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	10.98									+	
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	15.91									+	
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	25.04									+	
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP9D	UECS1	43.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	13.89										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	18.75										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	27.55										
	2-Wire Voice Grade Loop (SL21) - Zone 4		4	UEP9D	UECS2	45.72										
UNE Port R																
ALL STATE					1 IEEE 14											
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area 2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local Area	-	-	UEP9D UEP9D	UEPYB UEPYC	1.23 1.23	40.31 40.31	19.84 19.84	24.90 24.90	6.58 6.58	 	15.75 15.75			1.97 1.97	
	2-Wire Voice Grade Port (Centrex / EBS-PSE1)3Basic Local Area	+		UEP9D	UEPYD	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local Area			UEP9D	UEPYE	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local Area			UEP9D	UEPYF	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local Area			UEP9D	UEPYG	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area			UEP9D	UEPYT	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local Area			UEP9D	UEPYU	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local Area			UEP9D	UEPYV	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local Area			UEP9D	UEPY3	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic															
	Local Area			UEP9D	UEPYW	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area 2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local			UEP9D UEP9D	UEPYJ	1.23 1.23	40.31 108.35	19.84 70.57	24.90 54.24	6.58		15.75 15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYO	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-8009)2, 3 Basic Local Area			UEP9D	UEPYQ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic Local Area			UEP9D	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
AL, KY, LA,	, MS, SC, & TN Only			UEP9D	LIEDOA	4.00	40.04	40.04	24.00	0.50		45.75			1.97	
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)			UEP9D	UEPQA UEPQB	1.23 1.23	40.31 40.31	19.84 19.84	24.90 24.90	6.58 6.58		15.75 15.75			1.97	
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3			UEP9D	UEPQC	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3			UEP9D	UEPQD	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQE	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5112)3			UEP9D	UEPQF	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3			UEP9D	UEPQG	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPQW	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3	1		UEP9D	UEPQJ	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2	-	-	UEP9D UEP9D	UEPQM UEPQO	1.23	108.35 108.35	70.57 70.57	54.24	11.70 11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3	1		UEP9D UEP9D	UEPQO	1.23	108.35	70.57	54.24 54.24	11.70		15.75 15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-N/5009)2, 3 2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3	+		UEP9D	UEPQP	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-0209)2, 3	1		UEP9D	UEPQR	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3	1		UEP9D	UEPQ6	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9D	UEPQ9	1.23	40.31	19.84	24.90	6.58	1	15.75		I	1.97	1

ATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Increment al Charge - Manual Svc Order vs. Electronic- Disc 1st	al Charg Manua Svc Ord vs. Electron
										curring						
						Rec	Nonred		Disco					ATES (\$)		
			-				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAI
			1													
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9D	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
Local Switch	hing															
	Centrex Intercom Funtionality, per port			UEP9D	URECS	0.7947										
	per Portability															
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Features																
	All Standard Features Offered, per port			UEP9D	UEPVF	2.56	404.98					15.75			1.97	
	All Select Features Offered, per port			UEP9D UEP9D	UEPVS	0.00 2.56	404.98					15.75			1.97 1.97	
NARS	All Centrex Control Features Offered, per port		1	UEP9D	UEPVC	2.56						15.75			1.97	
	Unbundled Network Access Register - Combination			UEP9D	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Inward			UEP9D	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00									
	ous Terminations															
2-Wire Trun	k Side															
	Trunk Side Terminations, each			UEP9D	CEND6	8.25	120.00	18.85	61.77	3.88		15.75			1.97	
	al (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75			1.97	
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.56									
	Channel Mileage - 2-Wire			LIEBAR												
	Interoffice Channel Facilities Termination Interoffice Channel mileage, per mile or fraction of mile			UEP9D UEP9D	MIGBC	22.52 0.0098	40.77	27.57	17.26	7.11		15.75			1.97	
	tivations (DS0) Centrex Loops on Channelized DS1 Service			DEFaD	IVIIGBIVI	0.0096										
	Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.57										
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.57										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9D	1PQW7	0.57										
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9D	1PQWP	0.57										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9D	1PQWV	0.57										
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9D	1PQWQ	0.57										
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9D	1PQWA	0.57										
	ing Charges (NRC) Associated with UNE-P Centrex			LIEBAR												
	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port		1	UEP9D	USAC2		0.10	0.10				15.75			1.97	-
	Conversion of existing Centrex Common Block, each New Centrex Standard Common Block			UEP9D UEP9D	USACN M1ACS	0.00	37.97 666.32	16.68				15.75			1.97	
	New Centrex Standard Common Block		1	UEP9D	M1ACC	0.00	666.32					15.75			1.97	
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.63					15.75			1.97	
UNE-P CENT	TREX - EWSD (Valid in AL, FL, KY, LA, MS & TN)			OLI SD	OILLON	0.00	72.00					10.70			1.07	
	.oop/2-Wire Voice Grade Port (Centrex) Combo															
	pop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP9E		12.22										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design			UEP9E		17.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design			UEP9E		26.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		4	UEP9E		44.91										
	pop Combination Rates (Design)		-	UEP9E		45.40										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		2	UEP9E UEP9E		15.12 19.98										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9E		28.78										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design			UEP9E		46.95										
UNE Loop R			_			.0.50										1
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9E	UECS1	10.98										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9E	UECS1	15.91										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9E	UECS1	25.04										
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP9E	UECS1	43.68										1
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9E	UECS2	13.89					1	1			1	-
	2-Wire Voice Grade Loop (SL 2) - Zone 2	-	2	UEP9E	UECS2	18.75						1				1
	2-Wire Voice Grade Loop (SL 2) - Zone 3 2-Wire Voice Grade Loop (SL21) - Zone 4	-	3	UEP9E UEP9E	UECS2	27.55					-	1			1	1
UNE Port Ra		1	4	UEPSE	UECS2	45.72					1	1			1	1-
	LA, MS, & TN only	1										 				-
, x=, r =, x r,	2-Wire Voice Grade Port (Centrex) Basic Local Area	1	1	UEP9E	UEPYA	1.23	40.31	19.84	24.90	6.58	1	15.75			1.97	1
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area			UEP9E	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP9E	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	İ
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local	1	1	UEP9E	UEPYM	1.23	108.35	70.57	54.24	11.70		15.75	1	l	1.97	1

MS

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			d Elec	Svc Order Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	Manual Svc Order vs.	al Charge Manual Svc Orde vs. Electronic
						ъ			Nonred				000 0	ATEO (A)		
						Rec	Nonred		Disco					ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic															
	Local Area			UEP9E	UEPYZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local			UEP9E	UEPY9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP9E	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
AL, KY, LA	A, MS, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP9E	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP9E	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP9E	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP9E	UEPQM	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9E	UEPQZ	1.23	108.35	70.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP9E	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP9E	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
Local Swit																
	Centrex Intercom Funtionality, per port			UEP9E	URECS	0.7947										
Local Nun	nber Portability															
	Local Number Portability (1 per port)			UEP9E	LNPCC	0.35										
Features																
	All Standard Features Offered, per port			UEP9E	UEPVF	2.56						15.75			1.97	
	All Select Features Offered, per port			UEP9E	UEPVS	0.00	404.98									
	All Centrex Control Features Offered, per port			UEP9E	UEPVC	2.56						15.75			1.97	
NARS																
	Unbundled Network Access Register - Combination			UEP9E	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP9E	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP9E	UAROX	0.00	0.00	0.00								
Miscellane	eous Terminations															
2-Wire Tru	unk Side															
	Trunk Side Terminations, each			UEP9E	CEND6	8.25	120.00	18.85	61.77	3.88		15.75			1.97	
4-Wire Dio	gital (1.544 Megabits)															

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge - Manual Svc Order vs.	Manual Svc Orde vs. Electronic
						_		_		curring						
						Rec	Nonred			nnect				ATES (\$)		T
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		ļ														
	DS1 Circuit Terminations, each			UEP9E	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75			1.97	+
	DS0 Channel Activated Per Channel	1		UEP9E	M1HDO	0.00	14.56	30.23	74.00	2.04		15.75			1.97	
Interoffice (Channel Mileage - 2-Wire			02.02		0.00	1 1.00					10.70				†
	Interoffice Channel Facilities Termination			UEP9E	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75			1.97	T
	Interoffice Channel mileage, per mile or fraction of mile			UEP9E	MIGBM	0.0098										
Feature Act	tivations (DS0) Centrex Loops on Channelized DS1 Service															
	I Bank Feature Activations															
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9E	1PQWS	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9E	1PQW6	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP9E	1PQW7	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP9E	1PQWP	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP9E	1PQWV	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot			UEP9E	1PQWQ	0.57						15.75			1.97	
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP9E	1PQWA	0.57						15.75			1.97	-
	ring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port	-		UEP9E	USAC2		0.10	0.10				15.75			1.97	+
	Conversion of Existing Centrex Common Block, each			UEP9E	USACN		37.97	16.68				15.75			1.97	
	New Centrex Standard Common Block			UEP9E	M1ACS		31.91	10.00				15.75			1.97	
	New Centrex Standard Common Block			UEP9E	M1ACC							15.75			1.97	
	NAR Establishment Charge, Per Occasion			UEP9E	URECA							15.75			1.97	
	ITREX - DCO - Valid in AL, KY, LA, MS, & TN)			02.02	OITEOIT							10.70				†
	Loop/2-Wire Voice Grade Port (Centrex) Combo	1														
	oop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		1	UEP93		12.22										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP93		17.13										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		3	UEP93		26.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Non-Design		4	UEP93		44.91										
	oop Combination Rates (Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		1	UEP93		15.12										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP93		19.98										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design			UEP93		28.78										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo - Design		4	UEP93		46.95										
UNE Loop F		-	1	UEP93	LIECC4	10.98										+
	2-Wire Voice Grade Loop (SL 1) - Zone 1	-	2	UEP93 UEP93	UECS1	15.91										+
	2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP93	UECS1	25.04										+
	2-Wire Voice Grade Loop (SL 1) - Zone 4		4	UEP93	UECS1	43.68										†
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP93	UECS2	13.89										+
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP93	UECS2	18.75										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP93	UECS2	27.55										1
	2-Wire Voice Grade Loop (SL21) - Zone 4		4	UEP93	UECS2	45.72										1
UNE Port R																
	, MS, & TN only															
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP93	UEPYA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local Area	1		UEP93	UEPYB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area	ļ		UEP93	UEPYH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local	-		UEP93	UEPYM	1.23	108.35	7.57	54.24	11.70		15.75			1.97	1
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic			UEP93	UEPYZ	1.23	100.05	7.57	E4.04	44 70		15.75			1.97	
	Local Area 2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local	1	<u> </u>	UEP93 UEP93	UEPY2	1.23	108.35 40.31	19.84	54.24 24.90	11.70 6.58		15.75	1		1.97	+
	2-Wire Voice Grade Port Terminated in on Wegain's or equivalent - Basic Local Area			UEP93	UEPY2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex)	1	!	UEP93	UEPQA	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex) 2-Wire Voice Grade Port (Centrex 800 termination)	1		UEP93	UEPQB	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP93	UEPQH	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP93	UEPQM	1.23	108.35	7.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP93	UEPQZ	1.23	108.35	7.57	54.24	11.70		15.75			1.97	
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP93	UEPQ9	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP93	UEPQ2	1.23	40.31	19.84	24.90	6.58		15.75			1.97	
Local Switc	hing															
		1														
	Centrex Intercom Funtionality, per port	1	i .	UEP93	URECS	0.7947		1	ı		1	1	1	1	1	0

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			d Elec	Svc Order Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Manual Svc Order vs.	al Charge Manual Svc Order vs.	al Charge Manua Svc Ord vs. Electror
						_			Nonred					(4)		
						Rec	Nonrec		Disco					ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
Local Num	ber Portability															
	Local Number Portability (1 per port)			UEP93	LNCCC	0.35										
Features																
	All Standard Features Offered, per port			UEP93	UEPVF	2.56						15.75			1.97	
	All Centrex Control Features Offered, per port			UEP93	UEPVC	2.56						15.75			1.97	
NARS																
	Unbundled Network Access Register - Combination			UEP93	UARCX	0.00	0.00	0.00								
	Unbundled Network Access Register - Indial			UEP93	UAR1X	0.00	0.00	0.00								
	Unbundled Network Access Register - Outdial			UEP93	UAROX	0.00	0.00	0.00								
Miscellane	ous Terminations					0.00		0.00								
2-Wire Tru																
	Trunk Side Terminations, each			UEP93	CEND6	8.25	120.00	18.85	61.77	3.88		15.75			1.97	
4-Wire Dig	ital (1.544 Megabits)			02.00	OL. IDO	0.20	120.00	10.00	0	0.00		10.10			1.07	
	DS1 Circuit Terminations, each			UEP93	M1HD1	58.41	203.19	96.25	74.86	2.54		15.75			1.97	
	DS0 Channels Activated, Per Channel			UEP93	M1HDO	0.00	14.56	00.20	7 1.00	2.01		15.75			1.97	
Interoffice	Channel Mileage - 2-Wire			02.00		0.00	11.00					10.10				
	Interoffice Channel Facilities Termination			UEP93	MIGBC	22.52	40.77	27.57	17.26	7.11		15.75			1.97	
	Interoffice Channel mileage, per mile or fraction of mile			UEP93	MIGBM	0.0098	10.11	21.01	20			10.10			1.07	
Feature Ac	ctivations (DS0) Centrex Loops on Channelized DS1 Service			OLI 30	IVIIODIVI	0.0000										
	el Bank Feature Activations															
D4 Onamic	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP93	1PQWS	0.57										1
	Feature Activation on D-4 Channel Bank FX Line Side Loop Slot			UEP93	1PQW6	0.57										
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop Slot			UEP93	1PQW7	0.57										1
	Feature Activation on D-4 Channel Bank Centrex Loop Slot - Different Wire Center			UEP93	1PQWP	0.57										
	Feature Activation on D-4 Channel Bank Private Line Loop Slot	1		UEP93	1PQWV	0.57						1			1	1
	Feature Activation on D-4 Channel Bank Tirvate Line Loop Slot	1		UEP93	1PQWQ	0.57						1			1	1
	Feature Activation on D-4 Channel Bank WATS Loop Slot	+		UEP93	1PQWA	0.57										+
Non-Recur	rring Charges (NRC) Associated with UNE-P Centrex	1		SE. 55	0,117	0.07						1			1	1
- Itoli-Kecui	NRC Conversion Currently Combined Switch-As-Is with allowed changes, per port	1		UEP93	USAC2		0.10	0.10				15.75			1.97	1
	Conversion of Existing Centrex Common Block, each	+		UEP93	USACN		37.97	16.68				13.73			1.97	+
	New Centrex Standard Common Block	1		UEP93	M1ACS	0.00	666.32	10.00				15.75			1.97	+
	New Centrex Standard Common Block	+		UEP93	M1ACC	0.00	666.32					15.75			1.97	
	NAR Establishment Charge, Per Occasion	1		UEP93	URECA	0.00	72.63					15.75			1.97	
Note 1 - Pr	equired Port for Centrex Control in 1AESS, 5ESS & EWSD	+		OL: 33	JILLOA	0.00	12.03					13.75			1.97	-
	equired Port for Centrex Control in TAESS, 3ESS & EWSD	1			+	-						1			1	+
	equires Interoffice Channel Mileage equires Specific Customer Premises Equipment	1	-		1							 			 	+

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual	vs. Electronic-	- al Charge Manual Svc Orde vs.
						Rec	Nonrec	urring	Nonred Disco				066 B	ATES (\$)		
						Rec	First	urring Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							FIISL	Auu i	FIISL	Auu	SOWIEC	SOWAN	SUMAN	SOWAN	SOWAN	SOWAN
																+
																1
LOCAL INTERCONNE	CTION (CALL TRANSPORT AND TERMINATION)															
NOTE: "bl	k" beside a rate indicates that the Parties have agreed to bill and keep for that e	element p	oursua	nt to the terms and	conditi	ons in Attachr	ment 3.									
	Per MOU Rate for Local and ISP-bound Traffic (1/1/01-12/31/01)					\$0.0017500										
	Per MOU Rate for Local and ISP-bound Traffic (1/1/02-12/31/02)					\$0.0015000										
TANE	OWITOURIO		ļ		ļ											
IANDEM	SWITCHING Tandem Switching Function Per MOU	1	-	OHD	1	0.0005270										+
	Multiple Tandem Switching, per MOU (applies to intial tandem only)			OHD		0.0005379										+
TRUNK CH				UND		0.0005379										+
I KUNK CI	Installation Trunk Side Service - per DS0			OHD	TPP++		334.11bk	56.98bk								+
	Dedicated End Office Trunk Port Service-per DS0**			OHD	TDE0P	0.00	334.11DK	30.90DK								+
	Dedicated End Office Trunk Port Service-per DS1**			0H1 OH1MS	TDE1P											+
	Dedicated Tandem Trunk Port Service-per DS0**			OHD	TDW0P											+
	Dedicated Tandem Trunk Port Service-per DS1**				TDW1P											1
** This rat	te element is recovered on a per MOU basis and is included in the End Office Sv	vitching a														+
	TRANSPORT (Shared)			J. 1												
	Common Transport - Per Mile, Per MOU			OHD		0.0000026										
	Common Transport - Facilities Termination Per MOU			OHD		0.0004541										
	CTION (TRANSPORT)															
INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE															
	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per			OHL, OHM	1L5NF	0.0098bk										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility															
WITEDOE	Termination per month			OHL, OHM	1L5NF	22.52bk	27.57bk		7.11bk							
INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT - 56/64 KBPS			OLIL OLIM	41.550/	0.0098bk										
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month			OHL, OHM OHL, OHM	1L5NK 1L5NK	0.0098bk	27.57bk		7.11bk							+
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month	1		OHL, OHM	1L5NK	0.0098bk	27.57DK		7.11DK							+
	Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month			OHL, OHM	1L5NK	15.68bk	27.57bk		7.11bk							+
INTEROE	FICE CHANNEL - DEDICATED TRANSPORT - DS1			Onl, Onivi	ILSINK	13.000K	27.37 DK		7.11DK							+
INTEROIT	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month			OH1, OH1MS	1L5NL	0.201bk										+
	Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month			OH1, OH1MS	1L5NL	57.33bk	82.28bk		14.9bk							+
INTEROFE	FICE CHANNEL - DEDICATED TRANSPORT- DS3			Orri, Orrimo	TEGITE	07.0001	OZ.ZODK		1 1.0 DK							1
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			OH3. OH3MS	1L5NM	4.76bk										+
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			OH3, OH3MS	1L5NM	641.9bk	163.7bk		60.29bk							
LOCAL CH	HANNEL - DEDICATED TRANSPORT															1
	Local Channel - Dedicated - 2-Wire Voice Grade per month			OHL, OHM	TEFV2	14.91bk	194.22bk	33.36bk	37.79bk	3.30bk						
	Local Channel - Dedicated - 4-Wire Voice Grade per month			OHL, OHM	TEFV4	15.99bk	194.66bk	33.8bk	38.27bk	3.78bk						
	Local Channel - Dedicated - DS1 per month			OH1	TEFHG	36.83bk	178.5bk	154.61bk	22.89bk	15.74bk						
	Local Channel - Dedicated - DS3 Facility Termination per month			OH3	TEFHJ	413.87bk	454.13bk	264.47bk	123.23bk	86.19bk						
	ITERCONNECTION MID-SPAN MEET	1	l													
	Access service ride Mid-Span Meet, one-half the tariffed service Local Channel	rate is ap	plicabl	e.												
MULTIPLE		1	1													
	Channelization - DS1 to DS0 Channel System			OH1, OH1MS	SATN1	102.85bk	91.57bk	62.94bk	10.87bk	10.10bk						+
	DS3 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) per month	1		OH3, OH3MS OH1, OH1MS	SATNS	170.63bk 12.96bk	179.17bk 6.62bk	94.52bk 4.74bk	34.3bk	32.82bk	1			l		+

CAT	EGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)			Submitte d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge - Manual Svc Order vs. Electronic-	Increment al Charge - Manual Svc Order vs. Electronic- Disc Add'l
							_		_		curring						
							Rec	Nonred			nnect				ATES (\$)		
				-				First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	-			1													
PHYSICAL	COLLOCAT	ION		1		1											
		Physical Collocation - Application Fee - Initial			CLO	PE1BA		1,890.38		\$0.05							
		Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		1,575.69		\$0.51							
		Physical Collocation - Application Fee - Subsequent for Co-Carrier Cross Connect			CLO	PE1DT		583.13	583.13								
		Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		604.19									
		Physical Collocation - Space Preparation - C.O. Modification per square ft.			CLO	PE1SK	2.30										
		Physical Collocation - Space Preparation - Common Systems Modification per square ft Cageless	Ι.		CLO	PE1SL	2.52										i .
	_	Physical Collocation - Space Preparation - Common Systems Modification per	<u>'</u>		CLO	FEISL	2.52										\vdash
		Cage	l ı		CLO	PE1SM	85.67										i .
	1	Physical Collocation - Cable Installation			CLO	PE1BD		926.27	926.27	\$22.62							
		Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	5.74										
		Physical Collocation - Cable Support Structure			CLO	PE1PM	17.42										
		Physical Collocation - Power per Fused Amp	- 1		CLO	PE1PL	7.33										
		Physical Collocation - 120V, Single Phase Standby Power Rate			CLO	PE1FB	5.29										
		Physical Collocation - 240V, Single Phase Standby Power Rate	1		CLO	PE1FD	10.58										ļ
		Physical Collocation - 120V, Three Phase Standby Power Rate	1		CLO	PE1FE	15.87										ļ
		Physical Collocation - 277V, Three Phase Standby Power Rate			CLO UEANL,UEA,UDN,	PE1FG	36.65										\longmapsto
					UDC,UAL,UHL,UC												i .
		Physical Collocation - 2-Wire Cross-Connects			L,UEQ	PE1P2	0.0288	12.37	11.87	\$6.04	5.45						i .
		Physical Collocation - 4-Wire Cross-Connects			CLO	PE1P4	0.0576	12.47	11.94	\$6.59	5.91						
	† · · · · · · · · · · · · · · · · · · ·				CLO,UEANL,UEQ,					*****							
		Physical Collocation - DS1 Cross-Connects			WDS1L,WDS1S	PE1P1	1.14	22.16	16.02	\$6.60	5.97						i .
		Physical Collocation - DS3 Cross-Connects			CLO	PE1P3	14.49	21.01	15.29	\$7.61	6.10						
		Physical Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.87	21.01	15.29	\$7.61	6.10						
		Physical Collocation - 4-Fiber Cross-Connect	-		CLO	PE1F4	5.10	25.70	19.97	\$10.01	8.50						ļ
		Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.			CLO	PE1BW	183.20										1
	-	Priysical Collocation - Welded Wife Cage - Filst 100 Sq. Ft.	-	-	CLO	PE1C	103.20					1					
		Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.			CLO	W	17.97										i .
	† · · · · · · · · · · · · · · · · · · ·	Physical Collocation - Security Access System - Security System per Central			CLO	PE1AX	75.23										
		Physical Collocation - Security Access System - New Access Card Activation, per															
		Card	1		CLO	PE1A1	0.0576	27.95	27.95								
		Physical Collocation-Security Access System-Administrative Change, existing															1
		Access Card, per Card			CLO	PE1AA		7.84	7.84								\longmapsto
		Physical Collocation - Security Access System - Replace Lost or Stolen Card, per Card			CLO	PE1AR		22.91	22.91								1
		Physical Collocation - Security Access - Initial Key, per Key			CLO	PE1AK		13.17	13.17								
		Physical Collocation - Security Access - Initial Rey, per Rey Physical Collocation - Security Access - Key, Replace Lost or Stolen Key, per Key			CLO	PE1AL		13.17	13.17			†					\vdash
		Physical Collocation - Space Availability Report per premises			CLO	PE1SR		1,081.40	1,081.40								
	1	Triyorda Gonocation Opaco / Wanabinty Ttoport por promised			UEANL, UEA, UDN,	. E.O.		1,001.10	1,001110								
					UDC,UAL,UHL,UC												1
		POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			L,UEQ,CLO	PE1PE	0.0867										
					UEANL, UÉA, UDN,												i .
		POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			UDC,UAL,UHL,UC L,UEQ,CLO	PE1PF	0.1734										i .
	_	POT Bay Arrangements prior to 6/1/99 - 4-wire Cross-Connect, per cross-connect			UEANL, UEA, UDN,	PETPF	0.1734										\longrightarrow
					UDC,UAL,UHL,UC												i .
					L,UEQ,CLO,WDS												i .
		POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			1L,WDS1S,	PE1PG	1.22										i .
					UEANL, UEA, UDN,												
					UDC,UAL,UHL,UC							1		1			
	1	POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect	-		L,UEQ,CLO	PE1PH	10.91				 	1					\sqcup
					UEANL, UEA, UDN, UDC, UAL, UHL, UC							1		1			
		POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect			L,UEQ,CLO	PE1B2	37.26					1		1			
	+	programming prior to 0/1/33 - 2-1 iber Gross-Connect, per Gross-Connect		 	UEANL,UEA,UDN,	I LIDZ	31.20										\vdash
					UDC,UAL,UHL,UC							1		1			
		POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect			L,UEQ,CLO	PE1B4	50.24										1
		Collocation Cable Records - per request			CLO	PE1CR		763.69	490.94	\$133.77	133.77						
		Collocation Cable Records - VG/DS0 Cable, per cable record			CLO	PR1CD		328.81	328.81	\$190.22	190.22						
		Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		4.84	4.84	\$5.93	5.93						
		Collocation Cable Records - DS1, per T1TIE	ļ	ļ	CLO	PE1C1		2.27	2.27	\$2.78	2.78	ļ					
		Collocation Cable Records - DS3, per T3TIE	-	1	CLO	PE1C3		7.92	7.92	\$9.72	9.72	1					\longleftarrow
	1	Collocation Cable Records - Fiber Cable, per 99 fiber records			CLO	PE1CB		84.98	84.98	\$77.58	77.58	1	L				1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	I Charge - Manual Svc Order vs.	al Charge - Manual Svc Order vs. Electronic-	Manual Svc Order vs.
						Rec	Monro	curring		curring onnect			OSS B	ATES (\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN		SOMAN	SOMAN
							11131	Auu i	11131	Auu	JOINLE	JOWAN	JOWAN	JOWAN	JONAN	JOWAN
															-	1
	Physical Collocation - Security Escort - Basic, per Half Hour			CLO.CLORS	PE1BT		17.02	10.79								1
	Physical Collocation - Security Escort - Overtime, per Half Hour			CLO,CLORS	PE10T		22.17	13.94								1
	Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		27.32	17.08								1
	Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,															
	per linear ft.			CLO	PE1ES	0.0025										
	Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support															
	Structure, per lin. ft.			CLO	PE1DS	0.0037					1			 	<u> </u>	1
																ļ
ADJACENT COLLO																ļ
	Adjacent Collocation - Space Charge per Sq. Ft.			CLO	PE1JA	0.07					<u> </u>				<u> </u>	
	Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLO	PE1JC	4.68										
	Adjacent Collocation - 2-Wire Cross-Connects			CLO	PE1P2	0.02	12.37	11.87	\$6.04	5.45						
				UEA,UHL,UDL,UC												
	Adjacent Collocation - 4-Wire Cross-Connects			L,CLO	PE1P4	0.04	12.47	11.94	\$6.59	5.91						
	Adjacent Collocation - DS1 Cross-Connects			USL,CLO	PE1P1	1.05	22.16	16.00	\$6.60	5.97						
	Adjacent Collocation - DS3 Cross-Connects			CLO	PE1P3	14.27	21.01	15.29	\$7.61	6.10						
	Adjacent Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.42	21.01	15.29	\$7.61	6.10					<u></u>	
	Adjacent Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	4.62	25.70	19.97	\$10.01	8.50					<u></u>	
	Adjacent Collocation - Application Fee			CLO	PE1JB		1585.83	1585.83	\$0.51	0.51						
	Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker			CLO	DE4ED	5.00										
	Amp Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker			CLO	PE1FB	5.29										↓
	Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amb			CLO	PE1FD	10.58										
	Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker			CLO	PEIFU	10.58					-				├	
	Amp			CLO	PE1FE	15.87										
	Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker			CLO	FLIIL	15.67										+
	Amp			CLO	PE1FG	36.65							l	l	1	1
	p.o.p					22.00										1
PHYSICAL COLLOC	ATION IN THE REMOTE SITE															1
	Physical Collocation in the Remote Site - Application Fee *			CLORS	PE1RA		309.48		168.63							1
	Cabinet Space in the Remote Site per Bay/ Rack *			CLORS	PE1RB	210.05					1				1	1
	Physical Collocation in the Remote Site - Security Access - Key *			CLORS	PE1RD		13.17	13.17								+
	Physical Collocation in the Remote Site - Space Availability Report per Premises															1
	Requested *			CLORS	PE1SR		116.54	116.54								
	Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per															
	CLLI Code Requested *			CLORS	PE1RE		37.77	37.77			1					
	Remote Site DLEC Data (BRSDD), per Compact Disk, per CO			CLORS	PE1RR		233.14									
PHYSICAL COLLOC	ATION IN THE REMOTE SITE - ADJACENT															
	Remote Site-Adjacent Collocation - AC Power, per breaker amp			CLORS	PE1RS	6.27										
	Remote Site-Adjacent Collocation - Real Estate, per square foot			CLORS	PE1RT	0.134										
	n rates which are subject to true-up.															
NOTE: If	Security Escort and/or Add'l Engineering Fees become necessary for remote sit	e colloca	tion, th	ne Parties will nego	tiate app	ropriate rate	s.				1	1	1	1	1	1

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	usoc			RATES(\$)				Svc Order Submitted Manually	Charge - Manual Svo Order vs.	I Charge - Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	Increment - al Charge - Manual - Svc Order - vs Electronic- Disc Add'l
						B				curring			000 0	ATEO (A)		
						Rec	Nonred			nnect				ATES (\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
INTERIM OFFICE PROVIDER	ANUMBED DODTADUITY DOE															
	R NUMBER PORTABILITY - RCF															
	per number ported (Business Line)				TNPBL	3.08	0.2596	0.2596	0.0282	0.0282	3.50	15.75				
	per number ported (Residence Line)				TNPRL	3.08	0.2596	0.2596	0.0282	0.0282	3.50	15.75				
	Per Additional Path					1.17										
	ent that can be ordered electronically will be billed according to t															
	onically at present per the BBR-LO, the listed SOMEC rate reflect	ts the charge that	at would b	pe billed to a CLI	EC once	electronic ord	ering capabil	lities come on	-line for tha	t element. (Otherwise,	the manual	ordering ch	arge, SOMA	N, will be ap	plied to a
	it submits an LSR to BellSouth.							,	,				,			
	R NUMBER PORTABILITY - DID															
	er number ported (Residence)				TNPDR		0.4335	0.4335	0.4701	0.4701	3.50	15.75				
	er number ported (Business)				TNPDB		0.4335	0.4335	0.4701	0.4701	3.50	15.75				
	per trunk termination, Initial				TNPT2	58.41	191.75	71.25	28.94	28.94	3.50	15.75				
SERVICE PROVIDER NUMBER	R PORTABILITY (RIPH)															
RIPH,	Functionality, Per Rearrangement						19.93	19.93			3.50	15.75			15.75	
	5 11 1 5 1 1					4.00	0.4070	0.4070	0.0214	0.0044	0.50	45.75			45.75	
RIPH,	Per Number Ported					1.96	0.1972	0.1972	0.0214	0.0214	3.50	15.75			15.75	

CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			d Elec	Svc Order Submitted	Charge - Manual Svo Order vs.	I Charge - Manual Svc Order vs.	al Charge Manual Svc Order vs. Electronic	Manual
						n				curring			000 0	ATEO (A)		
			-			Rec		curring		onnect	201150	001111		ATES (\$)	001111	201111
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
ODUF/ADUF/CMDS	8				1											
	SS DAILY USAGE FILE (ADUF)															
ACCE	ADUF: Message Processing, per message				N/A	0.008087										
	ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	#########										
OPTIO	NAL DAILY USAGE FILE (ODUF)															
0	ODUF: Recording, per message				N/A	0.0000063										
	ODUF: Message Processing, per message				N/A	0.004707										
	ODUF: Message Processing, per Magnetic Tape provisioned				N/A	49.04										
	ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	#########										
CENTE	RALIZED MESSAGE DISTRIBUTION SERVICE (CMDS)															
	CMDS: Message Processing, per message				N/A	0.004										
	CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	0.001										
Notes:	If no rate is identified in the contract, the rate for the specific service or functi	on will be as	s set fo	th in applicable B	ellSouth	tariff or as ne	notiated by th	e Parties upo	n request b	v either Par	tv.					

	NOTES	UNBUNDLED NETWORK ELEMENT Int	erim Zone	BCS	USOC	1	1	RATES (\$)	1		1	1	OSS R	ATES (\$)	1	
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremer Charge Manual S Order v Electronic Add'I
							Nonre	curring	Nonre	curring	,					
									Disc	onnect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
									J							
h	http://www.ii	shown in the sections for stand-alone loops or loops as part of a combination refers to G interconnection.bellsouth.com/become_a_clec/html/interconnection.htm GE ACCESS LOOP	sograpriic	ally Deaveraged ONE	Zones. 10 Vi	ew Geographic	cally Deaverag	ed ONE Zone I	Designations	by Central C	Jince, reier	to internet v	vebsite.			
NDLED	JEXCHANG	GE ACCESS LOOP														
2	2-WIRE AN	NALOG VOICE GRADE LOOP														
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1	1	UEANL	UEAL2	18.48	70.44	44.05					44.22	13.55		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2	2	UEANL	UEAL2	27.87	70.44	44.05					44.22	13.55		
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3	3	UEANL	UEAL2	36.91	70.44	44.05					44.22	13.55		
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 1		UEPSR, UEPSB	UEALS	18.48	70.44	44.05					44.22	13.55		
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-Zone 2	2		UEALS	27.87	70.44	44.05					44.22	13.55		
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-Zone 3	3		UEALS	36.91	70.44	44.05					44.22	13.55		
		Engineering Information Document (EI)		UEANL			28.82	28.82			1		·			
		Manual Order Coordination for UVL-SL1s (per loop)*		UEANL	UEAMC		62.1	62.1								
		Order Coordination for Specified Conversion Time for UVL-SL1 (per LSR) *		UEANL	OCOSL		45.43	45.43								
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start														
		Signaling - Zone 1	1	UEA	UEAL2	21.57	178.12	128.8			1		44.42	13.55		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start														
		Signaling - Zone 2	2	UEA	UEAL2	32.53	178.12	128.8					44.42	13.55		1
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start														
		Signaling - Zone 3	3	UEA	UEAL2	43.08	178.12	128.8					44.42	13.55		
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		45.43									
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -														
		Zone 1	1	UEA	UEAR2	21.57	178.12	128.8					44.42	13.55		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -														
		Zone 2	2	UEA	UEAR2	32.53	178.12	128.8					44.42	13.55		
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse Battery Signaling -														
		Zone 3	3	UEA	UEAR2	43.08	178.12	128.8					44.42	13.55		
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		45.43									
4		NALOG VOICE GRADE LOOP														
		4-Wire Analog Voice Grade Loop - Zone 1	1	UEA	UEAL4	29.47	383.39	286.77					44.06	13.55		
		4-Wire Analog Voice Grade Loop - Zone 2	2		UEAL4	44.44	383.39	286.77					44.06	13.55		
		4-Wire Analog Voice Grade Loop - Zone 3	3	UEA	UEAL4	58.85	383.39	286.77					44.06	13.55		
		Order Coordination for Specified Conversion Time (per LSR)		UEA	OCOSL		45.43									
2	2-WIRE ISD	DN DIGITAL GRADE LOOP														
		2-Wire ISDN Digital Grade Loop - Zone 1	1	UDN	U1L2X	26.68	423.04	301.75					44.42	13.55		
		2-Wire ISDN Digital Grade Loop - Zone 2	2	UDN	U1L2X	40.24	423.04	301.75					44.42	13.55		
		2-Wire ISDN Digital Grade Loop - Zone 3	3		U1L2X	53.85	423.04	301.75					44.42	13.55		
		Order Coordination For Specified Conversion Time (per LSR)		UDN	OCOSL		45.43									
		· · · · · · · · · · · · · · · · · · ·														
2	2-WIRE Uni	niversal Digital Channel (UDC) COMPATIBLE LOOP									1			1		
T,		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 1	1	UDC	UDC2X	31.51	235.15	160.05	106.09	21.21			44.42	13.55		t —
-+		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 2	2		UDC2X	40.95	235.15	160.05	106.09	21.21	1		44.42	13.55		
-		2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone 3	3		UDC2X	47.12	235.15	160.05	106.09	21.21	†		44.42	13.55		1
		2 *****C Shiversal Digital Charlier (ODG) Compatible Loop - Zone 3	3	000	UDUZA	47.12	200.10	100.03	100.03	21.21	+	 	44.42	13.33		
-	2-WIRE 40	SYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE LOOP				+	+				+	 	1	1		
	2-11 INE A3	2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE 2-WIRE ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMPATIBLE				1	+		1		1	1			-	-
		LOOP														
-		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation														
			1	UAL	UAL2X	17 1	600.61	507.33					44.42	13.55		
		- Zone 1 2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation	1	UAL	UALZX	17.1	10.000	507.33	1		-	-	44.42	13.55		-
- 1		- Zone 2	2	UAL	UAL2X	25.79	600.61	507 22					44.42	13 55		
		2 Wire Unbundled ADSL Loop including manual service inquiry & facility reservation	- 2	UAL	UMLZA	23.79	1 0.000	507.33	+		+		44.42	13.55		1
		L TING CHOCK AND LEGOP INCIDENTS HIGHER SCITTLE INQUITY & Idellity ICSCIVATION	_	UAL	UAL2X	34.15	600.61	507.33			1		44.42	13.55		
		- 7one 3				J4.10	45.43	301.33			+	 	77.42	10.00		
		- Zone 3	3		OCOSI	1	40.40		1		+	1	1		-	
		Order Coordination for Specified Conversion Time (per LSR)	3	UAL	OCOSL											1
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservator -		UAL		17.1	205 20	120.22	100.74	1F 0C			44.49	13 55		
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1	1		UAL2W	17.1	205.28	129.32	100.74	15.86			44.42	13.55		
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -		UAL	UAL2W	1 1111										
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 Zone 2		UAL		17.1 25.79	205.28	129.32 129.32	100.74	15.86 15.86			44.42 44.42	13.55 13.55		
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	1 2	UAL UAL UAL	UAL2W UAL2W	25.79	205.25	129.32	100.74	15.86			44.42	13.55		
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3		UAL UAL UAL	UAL2W UAL2W UAL2W	1 1111	205.25 205.28									
		Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton -	1 2	UAL UAL UAL	UAL2W UAL2W	25.79	205.25	129.32	100.74	15.86			44.42	13.55		
	2.WIDE PIG	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	1 2	UAL UAL UAL	UAL2W UAL2W UAL2W	25.79	205.25 205.28	129.32	100.74	15.86			44.42	13.55		
2	2-WIRE HIG	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3 Order Coordination for Specified Conversion Time (per LSR) GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	1 2	UAL UAL UAL	UAL2W UAL2W UAL2W	25.79	205.25 205.28	129.32	100.74	15.86			44.42	13.55		
2	2-WIRE HIG	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3 Order Coordination for Specified Conversion Time (per LSR) GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP 2-WIRE HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE	1 2	UAL UAL UAL	UAL2W UAL2W UAL2W	25.79	205.25 205.28	129.32	100.74	15.86			44.42	13.55		
2	2-WIRE HIG	Order Coordination for Specified Conversion Time (per LSR) 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 1 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry & facility reservaton - Zone 3 Order Coordination for Specified Conversion Time (per LSR) GH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP	1 2	UAL UAL UAL	UAL2W UAL2W UAL2W	25.79	205.25 205.28	129.32	100.74	15.86			44.42	13.55		

NOTES	S UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	RATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incren Char Manua Orde sc Electror Ad
			+		 		Nonre	curring		ecurring						
			+							onnect						so
	2 Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		+-+		+	Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	So
	- Zone 2		2	UHL	UHL2X	18.41	600.61	507.33					44.06	13.55		
	Wire Unbundled HDSL Loop including manual service inquiry & facility reservation		+ -	UNL	UNLZX	10.41	000.01	307.33					44.00	13.33	+	+
	- Zone 3		3	UHL	UHL2X	24.39	600.61	507.33					44.06	13.55		
	Order Coordination for Specified Conversion Time (per LSR)		- 3	UHL	OCOSL	24.55	45.43	307.33					44.00	13.33	+	+-
	Wire Unbundled HDSL Loop without manual service inquiry and facility		++	OFF	OCCOSE	+	45.45							+	+	+
	reservation - Zone 1		1	UHL	UHL2W	12.21	222.65	146.68	100.74	15.86			44.06	13.55		
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility		++	OFF	OFFICE	12.21	222.00	140.00	100.74	10.00			44.00	10.00	+	+
	reservation - Zone 2		2	UHL	UHL2W	18.41	222.65	146.68	100.74	15.86			44.06	13.55		
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility		+ - +												+	_
	reservation - Zone 3		3	UHL	UHL2W	24.39	222.65	146.68	100.74	15.86			44.06	13.55		
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		45.43							1	1	
														1	1	
4-WIRE	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPATIBLE LOOP														1	
	4 Wire Unbundled HDSL Loop including manual service inquiry and facility				1									1	1	
	reservation - Zone 1		1	UHL	UHL4X	16.21	625.11	532.78					44.06	13.55		
	4-Wire Unbundled HDSL Loop including manual service inquiry and facility									1	1					
	reservation - Zone 2		2	UHL	UHL4X	24.45	625.11	532.78					44.06	13.55		
1	4-Wire Unbundled HDSL Loop including manual service inquiry and facility		1 [1						
	reservation - Zone 3		3	UHL	UHL4X	32.38	625.11	532.78					44.06	13.55		
	Order Coordination for Specified Conversion Time (per LSR)		$\perp \perp \perp$	UHL	OCOSL		45.43									
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility															
	reservation - Zone 1		1	UHL	UHL4W	16.21	279.96	203.99	110.24	20.75			44.06	13.55		4
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility				1 11 11 4147	04.45	070.00	000.00	440.04	00.75			44.00	40.55		
	reservation - Zone 2		2	UHL	UHL4W	24.45	279.96	203.99	110.24	20.75			44.06	13.55		_
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility		3	UHL	UHL4W	32.38	270.00	202.00	440.04	20.75			44.06	13.55		
	reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		- 3	UHL	OCOSL	32.38	279.96 45.43	203.99	110.24	20.75			44.06	13.55	+	+
	Order Coordination for Specified Conversion fillie (per LSK)		+-+	UNL	UCUSL		40.40							+	+	+
4-WIRE	DS1 DIGITAL LOOP		+-+		+	-								+	+	+-
4 00 11 12	4-Wire DS1 Digital Loop - Zone 1		1	USL	USLXX	59.61	715.77	421.5					43.77	13.55	+	+-
	4-Wire DS1 Digital Loop - Zone 2		2	USL	USLXX	89.9	715.77	421.5					43.77	13.55	+	+
	4-Wire DS1 Digital Loop - Zone 3		3	USL	USLXX	119.06	715.77	421.5					43.77	13.55	1	\top
	Order Coordination for Specified Conversion Time (per LSR)			USL	OCOSL		48.47							1	1	\top
														1	-	
4-WIRE	19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP															T
	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	34.26	602.73	393.5					44.06	13.55		
	4 Wire Unbundled Digital 19.2 Kbps		2	UDL	UDL19	51.67	602.73	393.5					44.06	13.55	1	
	4 Wire Unbundled Digital 19.2 Kbps		3	UDL	UDL19	68.43	602.73	393.5					44.06	13.55		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1		1	UDL	UDL56	34.26	602.73	393.5					44.06	13.55		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2		2	UDL	UDL56	51.67	602.73	393.5					44.06	13.55		
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3		3	UDL	UDL56	68.43	602.73	393.5					44.06	13.55		
	Order Coordination for Specified Conversion Time (per LSR)		4.4	UDL	OCOSL		45.43									
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1	-	1	UDL	UDL64	34.26	602.73	393.5	1	1	-	1	44.06	13.55		4
1	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2	l	3	UDL	UDL64	51.67	602.73	393.5	+	1	 	-	44.06	13.55	+	+
1	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	1	13	UDL UDL	UDL64 OCOSL	68.47	602.73 45.43	393.5	1	1	1	-	44.06	13.55	+	+
1	Graci Goordination for openined Conversion filling (bet Fox)	l	+	UDL	UUUUL	 	40.40	1	1	1	 			+	+	+
			+		1						†			1	1	+
2-WIRE	Unbundled COPPER LOOP															T
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility		\Box		1											
	reservation - Zone 1		1	UCL	UCLPB	15.24	283.95	163.99	120.42	22.42			19.99	19.99	19.99	
	2-Wire Unbundled Copper Loop/Short including manual service inquiry & facility	1	1 T	·												
	reservation - Zone 2		2	UCL	UCLPB	17.14	283.95	163.99	120.42	22.42			19.99	19.99	19.99	
1	2 Wire Unbundled Copper Loop/Short including manual service inquiry & facility		1 [1						
	reservation - Zone 3		3	UCL	UCLPB	17.68	283.95	163.99	120.42	22.42			19.99	19.99	19.99	
1	Order Coordination for Unbundled Copper Loops (per loop)	<u> </u>	+	UCL	UCLMC		62.1	62.1		1	-			4		4
1	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility			LICI	LICI DV	45.04	202.40	407.45	400.74	45.00			10.00	40.00	40.00	Ι.
-	reservation - Zone 1	ļ	1	UCL	UCLPW	15.24	203.42	127.45	100.74	15.86			19.99	19.99	19.99	1
1	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility		2	UCL	UCLPW	17 14	203.42	127.45	100.74	15.00			10.00	10.00	10.00	1.
1	reservation - Zone 2 2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility	-	+-+	UCL	UCLPW	17.14	203.42	127.45	100.74	15.86	1		19.99	19.99	19.99	1
1			3	UCI	UCLPW	17.68	203.42	127.45	100.74	15.86			19.99	19.99	10.00	1
1	reservation - Zone 3	-	+3	UCL		17.00		62.1	100.74	13.86	1		19.99	19.99	19.99	+
1	Order Coordination for Unbundled Copper Loops (per loop) 2-Wire Unbundled Copper Loop/Long - includes manual srvc. inquiry and facility	1	+	UUL	UCLMC	 	62.1	υZ. I	+	1	1			+	+	+
1	reservation - Zone 1		1 4	UCL	UCL2L	47.77	270.89	150.93	120.42	22.42			19.99	19.99	19.99	
+	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	 	+	UUL	UULZL	71.//	210.03	150.83	120.42	22.42	†	-	13.33	13.33	13.33	+
	reservation - Zone 2		2	UCL	UCL2L	69.16	270.89	150.93	120.42	22.42			19.99	19.99	19.99	
+	2-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility	1	+		UOLEL	03.10	270.00	100.00	120.72	22.72	1		15.55	13.33	15.55	+
	reservation - Zone 3		3	UCL	UCL2L	84.94	270.89	150.93	120.42	22.42			19.99	19.99	19.99	1
		1	1	UCL	UCLMC		62.1	62.1	0		1					+-
	Order Coordination for Unbundled Copper Loops (per loop) 2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility		+		002.00											

CATEGORY NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
							Nonre	curring		ecurring						-
						Rec	First	Add'I	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility					Nec	11100	Auu	Tillac	Addi	JOHILL	SOMAN	SOMA	SOMA	JOHN	SOMAN
	reservation - Zone 2		2	UCL	UCL2W	69.16	190.36	114.39	100.74	15.86			19.99	19.99	19.99	19.99
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility															
	reservation - Zone 3		3	UCL	UCL2W	84.94	190.36	114.39	100.74	15.86			19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		62.1	62.1								
	2-Wire Unbundled Copper Loop - Non-Designed Zone 1	_ !	1	UEQ	UEQ2X	11.01	44.69	22.4	25.65	7.06			44.22	13.55		
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 2	- !	2	UEQ	UEQ2X	12.67	44.69	22.4	25.65	7.06		1	44.22	13.55		
	2 Wire Unbundled Copper Loop - Non-Designed - Zone 3		3	UEQ UEQ	UEQ2X USBMC	20.22	44.69 62.1	22.4 62.1	25.65	7.06			44.22	13.55		
	Order Coordination 2 Wire Unbundled Copper Loop - Non-Designed (per loop) Engineering Information Document			UEQ	USBIVIC		28.82	28.82								
	Loop Testing - Basic 1st Half Hour			UEQ	URET1		78.92	78.92						-		
	Loop Testing - Basic Additional Half Hour			UEQ	URETA		23.33	23.33								
	Esop rooting Suoto Additional Figure Figure			- OEG	ORLIN		20.00	20.00								
4-WIRE C	OPPER LOOP															
	4-Wire Copper Loop/Short - including manual service inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4S	24.55	332.47	212.51	130.98	27.68			19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility															
	reservation - Zone 2		2	UCL	UCL4S	26.13	332.47	212.51	130.98	27.68			19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - including manual service inquiry and facility															
	reservation - Zone 3		3	UCL	UCL4S	24.17	332.47	212.51	130.98	27.68			19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		62.1	62.1								-
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -		1	UCL		24.55	054.04	475.04	110.24	20.75			40.00	40.00	40.00	40.00
	Zone 1 4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -		1	UCL	UCL4W	24.55	251.94	175.94	110.24	20.75			19.99	19.99	19.99	19.99
	Zone 2		2	UCL	UCL4W	26.13	251.94	175.94	110.24	20.75			19.99	19.99	19.99	19.99
	4-Wire Copper Loop/Short - without manual service inquiry and facility reservation -			UCL	UCL4VV	20.13	231.94	175.94	110.24	20.75			19.99	19.99	19.99	19.99
	Zone 3		3	UCL	UCL4W	24.17	251.94	175.94	110.24	20.75			19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC	24.17	62.1	62.1	110.24	20.70			13.33	13.33	13.33	10.00
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4L	96.61	319.41	199.45	130.98	27.66			19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 2		2	UCL	UCL4L	148.48	319.41	199.45	130.98	27.66			19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - includes manual svc. inquiry and facility															
	reservation - Zone 3		3	UCL	UCL4L	180.12	319.41	199.45	130.98	27.66			19.99	19.99	19.99	19.99
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		62.1	62.1								
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility															
	reservation - Zone 1		1	UCL	UCL4O	96.61	238.87	162.9	110.24	20.75			19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility		_	1101	1101.40	440.40	000.07	400.0	440.04	00.75			40.00	40.00	40.00	40.00
	reservation - Zone 2		2	UCL	UCL4O	148.48	238.87	162.9	110.24	20.75			19.99	19.99	19.99	19.99
	4-Wire Unbundled Copper Loop/Long - without manual svc. inquiry and facility		3	UCI	UCL4O	400.40	238.87	400.0	440.04	20.75			19.99	40.00	19.99	40.00
	reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLMC	180.12	62.1	162.9 62.1	110.24	20.75			19.99	19.99	19.99	19.99
	Order Coordination for Oribunded Copper Loops (per loop)			UCL	UCLIVIC		02.1	02.1						-		
														-		
LOOP MODIFICATION																
	Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or			UAL, UHL, UCL,		1		1					1	1		
	equal to 18k ft			UEQ, ULS	ULM2L		65.32	65.32					1			
	Unbundled Loop Modification, Removal of Load Coils - 2 wire greater than 18k ft			UCL, ULS	ULM2G		342.29	342.29								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to															
	18K ft			UHL, UCL	ULM4L		65.32	65.32								
	Unbundled Loop Modification Removal of Load Coils - 4 Wire pair greater than 18k															
	ft			UCL	ULM4G		342.29	342.29								
	Hatara Halland Market Branch (Str. 17-5		1	UAL, UHL, UCL,			05.55	05.55					1			
	Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop			UEQ, UEF, ULS	ULMBT		65.37	65.37		1	1		1	-		
CUD I CODE			1						1	 	-	1	1	-		
SUB-LOOPS			1			1		1	1	+	-		1	-	1	├
6L 1	Distribution		1						1	 	-	1	1	-		
Sub-Loop			1	UEANL	LICECA	1	507.75	507.75	1	+	-		44.22	13.55	1	
	Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up			UEANL	USBSA USBSB		45.37	45.37	1	 	1	-	44.22 44.22	13.55	1	
	Sub-Loop - Per Cross Box Location - Per 25 Pair Pariel Set-Up Sub-Loop - Per Building Equipment Room - CLEC Feeder Facility Set-Up		1	UEANL	USBSC	1	380.6	380.6	1	+	1	-	44.22	13.55	1	H
	Sub-Loop - Per Building Equipment Room - Per 25 Pair Panel Set-Up		1	UFANL	USBSD		111.15	111.15	1	1	1		44.22	13.55	1	t
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 1	i	1	UEANL	USBN2	11.09	131.88	62.05	90.69	13.42			44.22	13.55		—
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 2	i	2	UEANL	USBN2	15.72	131.88	62.05	90.69	13.42			44.22	13.55		†
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	- 1	3	UEANL	USBN2	18.49	131.88	62.05	90.69	13.42			44.22	13.55		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		45.43	45.43								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN4	17.64	158.41	88.58	99.64	18.17			44.22	13.55		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	24.25	158.41	88.58	99.64	18.17			44.22	13.55		
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 3		3	UEANL	USBN4	23.63	158.41	88.58	99.64	18.17			44.22	13.55		
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		45.43	45.43	1		1	1	1			<u></u>

	UNBUNDLED NETWORK ELEMENT	Interim Zone BCS	USOC			RATES (\$)	1		-		OSS R	RATES (\$)		
									Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manua Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incren Char Manua Orde Electron Adi
					Nonre	curring		curring						+-
				_				onnect						+
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	I UEANL	USBR2	3.01	First 106.26	Add'I 36.42	First 90.69	Add'I 13.42	SOMEC	SOMAN	SOMAN 44.22	3.55	SOMAN	SOI
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	UEANL	USBMC	3.01	45.43	45.43	30.03	13.42			44.22	13.33		+
		I UEANL	USBR4	6.7	118.76	48.93	99.64	18.17			44.22	13.55		+-
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC) Order Coordination for Unbundled Sub-Loops, per sub-loop pair	UEANL	USBMC	0.7	45.43	45.43	99.04	10.17	+		44.22	13.33		+
				0.50			00.00	40.40	_	-	44.22	40.55		+
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	1 1 UEF 1 2 UEF	UCS2X UCS2X	8.59 12.29	131.88 131.88	62.05 62.05	90.69 90.69	13.42 13.42			44.22	13.55 13.55		+-
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2													+
	Wire Copper Unbundled Sub-Loop Distribution - Zone 3 Order Coordination for Unbundled Sub-Loops, per sub-loop pair		UCS2X USBMC	13.1	131.88 45.43	62.05 45.43	90.69	13.42			44.22	13.55		+
	Order Coordination for Oribundled Sub-Loops, per Sub-loop pair	UEF I 1 UEF		9.81			99.64	18.17	_	-	44.22	13.55		+
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1		UCS4X		158.41	88.58			_	-				+-
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	l 2 UEF	UCS4X	17.71	158.41	88.58	99.64	18.17			44.22	13.55		4-
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I 3 UEF	UCS4X	15.8	158.41	88.58	99.64	18.17			44.22	13.55		4
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair	UEF	USBMC		45.43	45.43								
Sub-Loop F	Feeder													
		UEA,												T
1		UDN,UCL,UDL,UD		1					1		1	1		
1	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up	C	USBFW		507.75				1		1			
1		UEA,					1				1	1		1
		UDN,UCL,UDL,UD												1
1	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up	C	USBFX		45.37	45.37			1		1			1
1	USL Feeder DS1 Set-up at DSX location, per DS1 termination	USL	USBFZ	1	523.87	11.34	1		1	1		1		+
1	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1	1 UEA	USBFA	11.16	186.56	113.37	109.36	27.48	1	1	19.99	19.99	19.99	a
+	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 1 Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice Grade - Zone 2	2 UEA	USBFA	14.67	186.56	113.37	109.36	27.48	+	+	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start, Voice Grade - Zone 3	3 UEA	USBFA	18.43	186.56	113.37	109.36	27.48			19.99	19.99	19.99	
	Order Coordination for Specified Conversion Time, per LSR	UEA	OCOSL	10.43	45.43	113.37	103.30	21.40	+		13.33	15.55	15.55	4
				44.40		440.07	400.00	07.40			19.99	19.99	19.99	-
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 1	1 UEA	USBFB	11.16	186.56	113.37	109.36	27.48						
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2	2 UEA 3 UEA	USBFB	14.67	186.56	113.37	109.36	27.48			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice Grade - Zone 3		USBFB	18.43	186.56	113.37	109.36	27.48			19.99	19.99	19.99	3
	Order Coordination for Specified Time Conversion, per LSR	UEA	OCOSL		45.43									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 1	1 UEA	USBFC	11.16	186.56	113.37	109.36	27.48			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2	2 UEA	USBFC	14.67	186.56	113.37	109.36	27.48			19.99	19.99	19.99	3
	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse Battery, Voice Grade -													Т
	Zone 3	3 UEA	USBFC	18.43	186.56	113.37	109.36	27.48			19.99	19.99	19.99	3
	Order Coordination For Specified Conversion Time, per LSR	UEA	OCOSL		45.43									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 1	1 UEA	USBFD	27.04	215.82	140.72	124.52	35.03			19.99	19.99	19.99	3
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2	2 UEA	USBFD	34.46	215.82	140.72	124.52	35.03			19.99	19.99	19.99	3
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice Grade - Zone 3	3 UEA	USBFD	32.55	215.82	140.72	124.52	35.03			19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LSR	UEA	OCOSL		45.43									1
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 1	1 UEA	USBFE	27.04	215.82	140.72	124.52	35.03			19.99	19.99	19.99	3
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 2	2 UEA	USBFE	34.46	215.82	140.72	124.52	35.03			19.99	19.99	19.99	3
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice Grade - Zone 3	3 UEA	USBFE	32.55	215.82	140.72	124.52	35.03			19.99	19.99	19.99	a
	Order Coordination For Specified Conversion Time, Per LSR	UEA	OCOSL		45.43		1							1
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1	1 UDN	USBFF	21.31	212.94	137.84	111.61	26.73			19.99	19.99	19.99	a
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2	2 UDN	USBFF	26.15	212.94	137.84	111.61	26.73			19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3	3 UDN	USBFF	29.36	212.94	137.84	111.61	26.73			19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, Per LSR	UDN	OCOSL	20.00	45.43	.57.04		20.70	1	1	.5.55	. 5.55	15.55	+
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	1 UDC	USBFS	21.31	212.94	137.84	111.61	26.73	1	1	19.99	19.99	19.99	a
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	2 UDC	USBFS	26.15	212.94	137.84	111.61	26.73	1	1	19.99	19.99	19.99	
t	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)	3 UDC	USBFS	29.36	212.94	137.84	111.61	26.73	+	1	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder, 2 Wire OBC (IDSL companie) Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1	1 USL	USBFG	79.79	204.38	129.38	124.52	35.03	1	1	19.99	19.99	19.99	
 	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1 Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2	2 USL	USBFG	155.94	204.38	129.38	124.52	35.03	+	+	19.99	19.99	19.99	
1	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3	3 USL	USBFG	290.5	204.38	129.38	124.52	35.03	1	1	19.99	19.99	19.99	
		USL	OCOSL	290.5	45.43	129.30	124.32	33.03	+		19.99	19.99	19.99	-
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1	1 UCL	USBFH	7.47	167.94	92.84	106.27	21.38	+		19.99	19.99	19.99	
			USBFH		167.94	92.84			_	-	19.99	19.99	19.99	
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 2	2 UCL 3 UCL	USBFH	5.74		92.84	106.27 106.27	21.38 21.38			19.99		19.99	
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone 3	3 UCL UCL	OCOSL	5.74	167.94 45.43	92.84	106.27	21.30			19.99	19.99	19.99	4
	Order Coordination For Specified Conversion Time, per LSR			10.51		407.00	110.00	00.57	_	-	40.00	40.00	40.00	_
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1	1 UCL	USBFJ	16.51	202.43	127.33	116.06	26.57			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2	2 UCL	USBFJ	10.35	202.43	127.33	116.06	26.57			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3	3 UCL	USBFJ	10.52	202.43	127.33	116.06	26.57	1	-	19.99	19.99	19.99	1
	Order Coordination For Specified Conversion Time, per LSR	UCL	OCOSL		45.43				1	-				_
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	1 UDL	USBFN	26.27	204.38	129.28	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	2 UDL	USBFN	26.62	204.38	129.29	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop	3 UDL	USBFN	25.21	204.38	129.28	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 1	1 UDL	USBFO	26.27	204.38	129.28	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 2	2 UDL	USBFO	26.62	204.38	129.29	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop - Zone 3	3 UDL	USBFO	25.21	204.38	129.28	124.52	35.03			19.99	19.99	19.99	3
	Order Coordination For Specified Time Conversion, per LSR	UDL	OCOSL		45.43		1				1			
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 1	1 UDL	USBFP	26.27	204.38	129.28	124.52	35.03			19.99	19.99	19.99	3
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 2	2 UDL	USBFP	26.62	204.38	129.29	124.52	35.03			19.99	19.99	19.99	
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3	3 UDL	USBFP	25.21	204.38	129.28	124.52	35.03			19.99	19.99	19.99	
	Order Coordination For Specified Conversion Time, per LSR	UDL	OCOSL		45.43									

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone BCS	USOC	T		RATES (\$)					OSS R	ATES (\$)		
OAI EGGIL!	110120						120 (4)					000.	1 20 (0)	Incremental	Incremental
										Svc Order	Svc Order	Incremental	Incremental	Charge - Manual Svc	Charge - Manual Svc
										Submitted	Submitted	Charge - Manua		Order vs.	Order vs.
										Elec	Manually per	Svc Order vs.	Svc Order vs.	Electronic-Disc	Electronic-Disc
						Nonro	ecurring	Monre	curring	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
						Nonre	curring		onnect						
					Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Unbundled Sub-Loop Modification - 2-W Copper Dist Load Coil/Equip Removal per													
		2-W PR	UEF	ULM2X		356.5	12.29					44.22	13.55		
		Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR	UEF	ULM4X		356.5	12.29					44.22	13.55		
		Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged Tap Removal, per	OEF	ULIVI4X		330.3	12.29					44.22	13.33		
		PR unloaded	UEF	ULM4T		561.8	14.33					44.22	13.55		
	Unbundled	Network Terminating Wire (UNTW)													
		Unbundled Network Terminating Wire (UNTW) per Pair	UENTW	UENPP	0.41	62.71	62.71					44.22	13.55		
	Motwork In	nterface Device (NID)						-							
	MELWOIK III	Network Interface Device (NID) - 1-2 lines	UENTW	UND12		87.36	57.58					44.22	13.55		
		Network Interface Device (NID) - 1-6 lines	UENTW	UND16		128.84	99.06					44.22	13.55		
		Network Interface Device Cross Connect - 2 W	UENTW	UNDC2		11.83	11.83					44.22	13.55		
		Network Interface Device Cross Connect - 4W	UENTW	UNDC4		11.83	11.83					44.22	13.55		
LINIDLINIDI T	D I OOD 5	ONOFNED A TION			+		1								
UNBUNDLE	D FOOL C	ONCENTRATION Unbundled Loop Concentration - System A (TR008)	ULC	UCT8A	398.41	652.26	652.26			-	-	19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - System A (TR008) Unbundled Loop Concentration - System B (TR008)	ULC	UCT8B	58.36	271.78	271.78					19.99	19.99	19.99	
		Unbundled Loop Concentration - System A (TR303)	ULC	UCT3A	439.73	652.26	652.26			1	1	19.99	19.99	19.99	
		Unbundled Loop Concentration - System B (TR303)	ULC	UCT3B	98.34	271.78	271.78					19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - DS1 Loop Interface Card	ULC	UCTCO	5.52	126.85	92.35	33.65	9.42			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - ISDN Loop Interface (Brite Card)	UDN UDC	ULCC1	8.77	21.11	21	10.81	10.74			19.99	19.99	19.99	
		Unbundled Loop Concentration - UDC Loop Interface (Brite Card) Unbundled Loop Concentration2 Wire Voice-Loop Start or Ground Start Loop	ODC	ULCCU	8.77	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Interface (POTS Card)	UEA	ULCC2	2.19	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery Loop Interface													
		(SPOTS Card)	UEA	ULCCR	13.03	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - 4 Wire Voice Loop Interface (Specials Card)	UEA	ULCC4	7.77	21.11	21	10.81	10.74			19.99	19.99	19.99	
		Unbundled Loop Concentration - TEST CIRCUIT Card Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop Interface	ULC UDL	UCTTC ULCC7	37.98 11.51	21.11 21.11	21 21	10.81	10.74 10.74			19.99 19.99	19.99 19.99	19.99 19.99	19.99 19.99
		Unbundled Loop Concentration - Digital 19.2 Rops Data Loop Interface Unbundled Loop Concentration - Digital 56 Kbps Data Loop Interface	UDL	ULCC5	11.51	21.11	21	10.81	10.74			19.99	19.99	19.99	19.99
		Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interface	UDL	ULCC6	11.51	21.11	21	10.81	10.74			19.99	19.99	19.99	
															
UNBUNDLE	D SOB-LOC	DP CONCENTRATION (OUTSIDE CO)						-							
															
UNE OTHER	. PROVISIO	ONING ONLY - NO RATE													
	,	NID - Dispatch and Service Order for NID installation	UENTW	UNDBX											
		UNTW Circuit Id Establishment, Provisioning Only - No Rate	UENTW	UENCE											
			UEANL,UEF,UEQ,												
		Unbundled Contract Name, Provisioning Only - No Rate	UENTW	UNECN											
			UAL,UCL,UDC,UD L,UDN,UEA,UHL,U												1
		Unbundled Contact Name, Provisioning Only - no rate	LC LC	UNECN	0	0									1
		Oribundled Contact Name, Provisioning Only - no rate	LC	UNLCIN	0	U									-
			UEA,UDN,UCL,UD												
		Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no rate	C	USBFQ	0	0									
			UEA,USL,UCL,UD												1
		Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no rate	L	USBFR	0	0									
		Unbundled DS1 Loop - Superframe Format Option - no rate Unbundled DS1 Loop - Expanded Superframe Format option - no rate	USL	CCOSF CCOEF	0	0									
		Orbandica DOT 2009 Expanded dapername Format option ino rate	302	OOOLI	Ü	Ü									
HIGH CAPA	CITY UNBU	INDLED LOCAL LOOP													
		onth minimum billing period													
		High Capacity Unbundled Local Loop - DS3 - Per Mile per month	UE3	1L5ND	15.33										
		High Capacity Unbundled Local Loop - DS3 - Facility Termination per month	UE3	UE3PX	382.95	905.04	529.05	239.5	167.53			31.38	31.38	3.94	3.94
		High Capacity Unbundled Local Loop - STS-1 - Per Mile per month High Capacity Unbundled Local Loop - STS-1 - Facility Termination per month	UDLSX UDLSX	1L5ND UDLS1	15.33 391.86	905.04	529.05	239.5	167.53	+	1	31.38	31.38	3.94	3.94
		ringin capacity criteriand accordance of the control of the contro	ODESA	ODLOI	551.00	303.04	525.05	200.0	101.00	1		01.00	01.00	5.34	5.34
LOOP MAKE	E-UP														
		Loop Makeup - Preordering Without Reservation, per working or spare facility													
		queried (Manual).	UMK	UMKLW	1	48.07	48.07								
	-	Loop Makeup - Preordering With Reservation, per spare facility queried (Manual).	UMK	UMKLP		50.97	50.97								
		Loop MakeupWith or Without Reservation, per working or spare facility queried	1.18.417	DCI IMAL	1	0.6070	0.0070								
		(Mechanized)	UMK	PSUMK	+	0.6873	0.6873	1		+	1				—
LINE SHARII	NG				+		+			+	1	1	1		—
LINE SHARII	110				+										—
		Line Sharing Splitter, per System 96 Line Capacity	I ULS	ULSDA	216.22	378.42	0	356.76	0		0				
		Line Sharing Splitter, per System 24 Line Capacity	I ULS	ULSDB	54.05	378.42	0	356.76	0		0				
		Line Sharing Splitte, Per System, 8 Line Capacity	I ULS	ULSD8	18.02	378.42	0	356.76	0	1	0	1	1	I	1

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC		I	RATES (\$)	1	ı			OSS R	ATES (\$)	ı	
										ecurring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manua Svc Order vs. Electronic-1st	Incremental I Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic-Dis Add'I
							Nonre	ecurring		onnect						ī
						Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Line Sharing - per Line Activation	1	ULS	ULSDC	0.61	37.09	21.24	20.07	9.85			44.22	13.55		-
		Line Sharing - per Subsequent Activity per Line Rearrangement	1	ULS	ULSDS		32.84	16.41					44.22	13.56		
																ı
		Line Sharing-CLEC/DLEC Owned Splitter in CO-per occurrence of each group of 8														1
		lines (16 pair)	1	ULS	ULSDG		57.83		11.41							
INRIINDI E	D TRANSP	PORT														
ONDONDEL																i
	COMMON	TRANSPORT (Shared)														
		Common Transport - Per Mile, Per MOU				0.0000121 0.0004672										-
		Common Transport - Facilities Termination Per MOU				0.0004672										i
	NOTE: INT	EROFFICE CHANNEL - DEDICATED TRANSPORT - minimum billing period: below DS	3 = one month, DS	3 and above for	ur months											
																-
	INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per		U1TVX	1L5XX	0.0167		+			1					
		Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Fer Mile per Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade - Facility		UTTVA	ILUAA	0.0107		1								
		Termination per month		U1TVX	U1TV2	24.3	81.25	54.94	33.54	13.82			31.38	31.38	9.8	9
		Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade Rev Bat Per		U1TVX	11.577	0.0167					1					İ
		Mile per month Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat Facility	 	UTTVX	1L5XX	0.0167					+					i
	<u></u>	Termination per month		U1TVX	U1TR2	24.3	81.25	54.94	33.54	13.82	<u> </u>		31.38	31.38	9.8	9
		Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per														1
		month Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade - Facility	U1	TVX	1L5XX	0.0167										
		Termination per month	LI1	TVX	U1TV4	21.29	81.25	54.94	33.54	13.82			31.38	31.38	3.94	3.9
		Tommation por morali					01.20	0 1.0 1	00.01	10.02			01.00	01.00	0.01	I
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month		U1TDX	1L5XX	0.0167										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month		U1TDX U1TDX	U1TD5 1L5XX	16.76 0.0282	81.26	54.94	33.54	13.82			31.38	31.38	3.94	3.9
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month		U1TDX	U1TD6	16.76	81.26	54.94	33.54	13.82			31.38	31.38	9.8	9
	INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT - DS1														
		Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month		U1TD1 U1TD1	1L5XX U1TF1	0.3415 77.14	178.93	163.98	32.77	28.95			31.38	31.38	3.94	3.9
		Interestince original Dedicated Hamport 201 1 dointy Termination per month		OTIDI	01111	77.14	170.55	100.00	02.11	20.00			01.00	01.00	0.04	I
	INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT- DS3														
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month		U1TD3	1L5XX	8.02	558.74	326.23	120.66	117.17			31.38	31.38	3.94	2.0
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month		U1TD3	U1TF3	880.65	558.74	320.23	120.00	117.17			31.30	31.30	3.94	3.9
	INTEROFF	FICE CHANNEL - DEDICATED TRANSPORT- STS-1														ī
		Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per month		U1TS1	1L5XX	8.02										<u> </u>
		Interoffice Channel - Dedicated Transport - STS-1 - Facility Termination per month		U1TS1	U1TFS	880.55	558.74	326.26	120.66	117.17			31.38	31.38	3.94	3.9
																i
	LOCAL CI	HANNEL - DEDICATED TRANSPORT														
		CAL CHANNEL DEDICATED TRANSPORT - minimum billing period - below DS3=one r	nonth, DS3 and ab	ove=four months												i
		Local Channel - Dedicated - 2-Wire Voice Grade Per Month		ULCVX	ULDV2	15.33	387.05	66.48	73.44	6.41			31.38	31.38	3.94	3.9
		Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per month		ULCVX	ULDR2	15.33	387.05	66.48	73.44	6.41			31.38	31.38	3.94	3.9
		Local Channel - Dedicated - 4-Wire Voice Grade per month Local Channel - Dedicated - DS1 per month - Zone 1	1	UNCVX ULDD1	ULDV4 ULDF1	16.54 42.62	387.93 355.73	67.35 308.11	74.38 44.48	7.35 30.59			31.38 31.38	31.38 31.38	3.94 3.94	
		Local Channel - Dedicated - DS1 per month - Zone 2	2	ULDD1	ULDF1	70.32	355.73	308.11	44.48	30.59			31.38	31.38	3.94	
		Local Channel - Dedicated - DS1 per month - Zone 3	3	ULDD1	ULDF1	190.68	355.73	308.11	44.48	30.59			31.38	31.38	3.94	
		Local Channel - Dedicated - DS3 - Per Mile per month		ULDD3	1L5NC	11.93 446	005.04	500.05	000.5	407.50			04.00	04.00	0.04	
		Local Channel - Dedicated - DS3 - Facility Termination per month Local Channel - Dedicated - STS-1- Per Mile per month		ULDD3 ULDS1	ULDF3 1L5NC	11.93	905.04	529.05	239.5	167.53	+		31.38	31.38	3.94	3.
		Local Channel - Dedicated - STS-1 - Fer Mile per Month Local Channel - Dedicated - STS-1 - Facility Termination per month		ULDS1	ULDFS	435.1	905.04	529.05	239.5	167.53			31.38	31.38	3.94	3.9
	<u> </u>			•				1		-						
MULTIPLE	KERS	Channelization - DS1 to DS0 Channel System		UXTD1	MQ1	134.46	182.48	125.42	21.12	19.62	+		31.38	31.38	3.947	3.9
		OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)		UDL	1D1DD	1.49	182.48	9.45	21.12	19.62			31.38	31.36	3.947	3.9
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per month		UDN	UC1CA	3.2	13.18	9.45								
		Voice Grade COCI - DS1 to DS0 Channel System - per month		UEA	1D1VG	0.7012	13.18	9.45								
		DS3 to DS1 Channel System per month STS1 to DS1 Channel System per month	 	UXTD3 UXTS1	MQ3 MQ3	180.03 180.03	357.07	188.36	66.66	63.79	1		31.38 31.38	31.38 31.38	3.94 3.94	
		DS3 Interface Unit (DS1 COCI) used with Loop per month		USL	UC1D1	10.8	13.18	9.45					31.30	31.30	5.94	
DARK FIBE	R	Dealt Fiber From Fiber Careeds Des Deadt Affile of Front Careed		·							1					-
		Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		UDF	1L5DC	97.65										I
		NRC Dark Fiber - Local Channel		UDF	UDFC4	31.00	1281.02	276.34	635.52	396.21	1		31.26	31.26	3.94	3.9
						1					1					

CATEGORY NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC			RATES (\$)			1	I	OSS R	ATES (\$)	la anaroni de la	Incrementa
										Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental I Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremen Charge Manual S Order v: Electronic-I Add'I
						Nonre	curring	Nonr	ecurring	p., 2011					
									onnect						
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -				Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Interoffice Channel		UDF	1L5DF	36.41										
	NRC Dark Fiber - Interoffice Channel		UDF	UDF14	30.41	1281.02	276.34	635.52	396.21			31.38	31.38	3.94	3.
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction Thereof per month -		05.	02		1201.02	270.01	000.02	000.21			01.00	01.00	0.01	0.
	Local Loop		UDF	1L5DL	97.65										
	NRC Dark Fiber - Local Loop		UDF	UDFL4		1281.02	276.34	635.52	396.21			31.38	31.38	3.94	3.
TRANSPORT OTHER															
Optional Fe	atures & Functions:														
optional i	Clear Channel Capability (B8ZS/ESF) Option - Subsequent - per DS1 Channel		UNC1X	CCOEF		185.26	23.86	1.99	0.78			29.33	3.93		
	Clear Channel Capability (B8ZS/SF) Option - Subsequent - per DS1 Channel		UNC1X	CCOSF		185.26	23.86	1.99	0.78			29.33	3.93		
BXX ACCESS TEN DIGI															
	8XX Access Ten Digit Screening, Per Call		OHD		0.0005227										
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX Number Reserved		OHD	N8R1X	+	6.38	0.9583			1	1	27.84	27.84		
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS 8XX Access Ten Digit Screening, Per 8XX No. Established With POTS		OHD OHD	N8FTX	+	22.63 22.63	2.73 2.73			1	1	27.84 27.84	27.84 27.84		l
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS 8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number		OHD	N8FCX	+	5.64	2.73			+	1	27.84	27.84		-
	8XX Access Ten Digit Screening, Customized Area of Service Fer 8XX Number		OI ID	1101 07	1	0.04	2.02					21.07	27.07		
	Requested Per 8XX No.		OHD	N8FMX	<u> </u>	6.6	3.78		<u> </u>		<u> </u>	27.84	27.84		<u></u>
	8XX Access Ten Digit Screening, Change Charge Per Request		OHD	N8FAX		7.34	0.9583					27.84	27.84		
	8XX Access Ten Digit Screening, Call Handling and Destination Features		OHD	N8FDX	1	5.64	1				ļ	27.84	27.84		
I INF INFORMATION DA	TA DAOE ACCECC (LIDD)														
	TA BASE ACCESS (LIDB) LIDB Common Transport Per Query		OQT		0.0000442										
	LIDB Validation Per Query		OQU		0.0145288										
	LIDB Originating Point Code Establishment or Change		OQT, OQU	NRPBX	0.0110200	61.62						27.84	27.84		
	W to W														
SIGNALING (CCS7)															
	CCS7 Signaling Termination, Per STP Port		1DB	PT8SX	156.33							19.99	19.99	19.99	19.9
	CCS7 Signaling Usage, Per TCAP Message		1DB	TPP++	0.0001108	277.07	277.07					19.99	19.99	19.99	40.0
	CCS7 Signaling Connection, Per link (A link) CCS7 Signaling Connection, Per link (B link) (also known as D link)		1DB 1DB	TPP++	21.79 21.79	277.07	277.07					19.99	19.99	19.99	19.9 19.9
	CCS7 Signaling Connection, if en link (B link) (also known as B link)		1DB	111177	0.0000452	211.01	211.01					15.55	15.55	15.55	13.3
	CCS7 Signaling Usage Surrogate, per link per LATA		1DB	STU56	396.55							19.99	19.99	19.99	19.9
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change,														
	per STP affected		1DB	CCAPO		40	40					19.99	19.99	19.99	19.9
	CCS7 Signaling Point Code, per Destination Point Code Establishment or Change,			00100			_								
	Per Stp Affected		1DB	CCAPD		8	8					19.99	19.99	19.99	19.9
E911 SERVICE															
L911 SERVICE															
CALLING NAME (CNAM) SERVICE														
	CNAM for DB Owners, Per Query		OQV		0.016										
	CNAM for Non DB Owners, Per Query		OQV		0.01										
	CNAM (Non-Databs Owner), NRC, applies when using the Character Based User														
	Interface (CHUI)		OQV	CDDCH		595	595					27.84	27.84		
	michael (Oriei)			OBBOIL		000	000					21.01	27.01		
LNP QUERY SERVICE				1			1					1			
			1	1	+		1	-		-	1				-
OPERATOR	R SERVICES AND DIRECTORY ASSISTANCE		+	1	+		1			+	1				-
OF ENATOR	SELLIGES AND DIRECTORY ADDIOTANCE		1	1	 										
OPERATOR CALL PRO	CESSING		1	1	1		1				†				
	Oper. Call Processing - Oper. Provided, Per Min Using BST LIDB			1	1.2										
	Oper. Call Processing - Oper. Provided, Per Min Using Foreign LIDB				1.24										
	Oper. Call Processing - Fully Automated, per Call - Using BST LIDB				0.2										
	Oper. Call Processing - Fully Automated, per Call - Using Foreign LIDB		1	1	0.2		1			1		1			
INWARD OPERATOR S	ED/I/CES		1	1	+		1	-		-	-	1	-		-
INVIARD OPERATOR S	Inward Operator Services - Verification, Per Minute		1	1	1.15		1			1	-				
	Inward Operator Services - Verification, Per Minute Inward Operator Services - Verification and Emergency Interrupt - Per Minute		1	1	1.15										
	To windle		1	1	0		1				†				
BRANDING - OPERATO	R CALL PROCESSING														
	Recording of Custom Branded OA Announcement			CBAOS		7000	7000					19.99	19.99	19.99	19.9
	Loading of Custom Branded OA Announcement per shelf/NAV			CBAOL		500	500					19.99	19.99		

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC	1		RATES (\$)			т —		OSS R	ATES (\$)		
									, , , , , , , , , , , , , , , , , , ,			Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs.
								Nonre	curring	Nonr	ecurring	per LSR	LSR	Electronic-1st	Electronic-Add'l	I 1st	Add'l
								Home	ouring		connect						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
DIRECTORY		ICE SERVICES												_			
		Y ASSISTANCE ACCESS SERVICE Directory Assistance Access Service Calls, Charge Per Call					0.275					+				+	+
		Directory Assistance Access Service Gails, Charge Fer Gail		1			0.273				+	+		-		+	+
	DIRECTOR	Y ASSISTANCE CALL COMPLETION ACCESS SERVICE (DACC)															
		Directory Assistance Call Completion Access Service (DACC), Per Call Attempt					0.1										
														_		4	
	UNBRANDI	NG Y TRANSPORT												-			+
		Directory Transport - Local Channel DS1					37.2	534.81	462.81			+		87.99	87.99	3.11	3.11
		Directory Transport - DS1 Level Interoffice Per Mile					0.7598	004.01	402.01		+	+		07.55	07.55	0.11	0.11
		Directory Transport - DS1 Level Interoffice Per Facility Termination					94.98	216.27	162.7					39.63	39.63	3.11	3.1
		Switched Common Transport Per DA Access Service Per Call					0.000327								1		<u> </u>
		Switched Common Transport Per DA Access Service Per Call Per Mile Access Tandem Switching Per DA Access Service Per Call		-			0.0000303 0.0024809							ļ	 		├
		Directory Transport - DA Interconnection Per DA Service Call					0.0024809				+	+				+	+
		Directory Transport - Installation NRC, Per Trunk or Signaling Connection					0.000200	407.81	11		1	-		87.99	87.99	3.11	1 3.1
	DIRECTOR	Y ASSISTANCE DATA BASE SERVICE (DADS)		1													4
		Directory Assistance Data Base Service Charge Per Listing				DBSOF	0.04 150								 		+
BRANDING		Directory Assistance Data Base Service, per month RY ASSISTANCE	†	+		DDSUF	130				+	+		 		+	+
BICANDING	DIRECTO	Custom Branding Announcement, per Recording to be used with the provision of DA			AMT	CBADA		3000	3000		+	+				+	+
		Loading of Custom Branded Announcement per DRAM Card/Switch	1	1	AMT	CBADC		690	690		†	+				1	
		·															
SELECTIVE	ROUTING																
		Selective Routing Per Unique Line Class Code Per Request Per Switch				USRCR		226.22	226.22					43.19	9.91	4	
VIDTUAL CO	OLLOCATIO	NI		-							+	+		_		+	+
VIKTUAL CO	OLLOCATIO	JN			ueanl,uea,udn,udc,						+	+				+	+
		Virtual Collocation - 2-wire Cross Connects (loop)			ual,uhl,ucl,ueq	UEAC2	0.3648	41.5	38.94					19.99	19.99	19.99	19.9
		Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.3648	41.5	38.94					19.99	19.99	19.99	
		Virtual Collocation - 2-wire Cross Connects (port)				VE1R2	0.3648	41.5	38.94					19.99	19.99	19.99	
		Virtual Collocation - 4-wire Cross Connects (loop)		-	uea,uhl,ucl,udl	UEAC4	0.7297	41.56	38.9 38.9					19.99	19.99	19.99	
		Virtual Collocation - 4-wire Cross Connects (port) Virtual Collocation - 2-Fiber Cross Connects			CLO	VE1R4 CNC2F	0.7297 15.06	41.56 69.28	48.89			+		19.99 19.99	19.99 19.99	19.99 19.99	
		Virtual Collocation - 4-Fiber Cross Connects			CLO	CNC4F	27.08	84.07	63.68			+		19.99	19.99	19.99	
		Virtual Collocatin - DS1 Cross Connects			USL,ULC,CLO	CNC1X	7.5	155	14								
															1		<u> </u>
AIN SELEC	TIVE CARRI	ER ROUTING			SRC	SRCEC		391788						10.00	40.00	10.00	10.0
		Regional Service Establishment End Office Establishment		-	SRC	SRCEO		320.53	320.53		+	+		19.99 19.99	19.99	19.99	
		Line/Port NRC, per end user		1	SRC	SRCLP		2.06	2.06		+	+		19.99	19.99	19.99	
		Query NRC, per query			SRC	O. COL.	0.000448	2.00	2.00			+		10.00	10.00	10.00	10.0
											+						
AIN - BELLS		SMS ACCESS SERVICE															
		AIN SMS Access Service - Service Establishment, Per State, Initial Setup				CAMSE		296.16	296.16					27.84	27.84		
		AIN SMS Access Service - Port Connection - Dial/Shared Access	1	-		CAMAR		87.29	87.29				1	27.84	27.84		
		AIN SMS Access Service - Port Connection - ISDN Access AIN SMS Access Service - User Identification Codes - Per User ID Code	-	+	 	CAM1P CAMAU	1	87.29 202.08	87.29 202.08		-	-		27.84 27.84	27.84	+	+
		AIN SMS Access Service - User Identification Codes - Per User ID Code AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement		1-	 	CAMAC	1	172.26	172.26			+		27.84	27.84 27.84	+	+
		AIN SMS Access Service - Security Card, Fer User 15 Code, militar of Replacement	1	1		5, 1,111 (5	0.0028							27.04	204		1
		AIN SMS Access Service - Session, Per Minute					0.0942966				+						
		AIN SMS Access Service - Company Performed Session, Per Minute					2.07										
				L													
AIN - BELLS		TOOLKIT SERVICE		_		B.15	1					4				4	4
		AIN Toolkit Service - Service Establishment Charge, Per State, Initial Setup	-	1		BAPSC BAPVX		291.41	291.41					27.84	27.84		
		AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt	 	1-	-	BAPTT	1	8333 73.02	8333 73.02		-	+	-	27.84 27.84	27.84 27.84	+	+
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Term. Attempt AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay	-	1		BAPTD	 	73.02	73.02		 	+	 	27.84	27.84	+	+
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook				D/II 1D		70.02	70.02		+	+		27.04	27.04	+	+
		Immediate	<u></u>	L	<u> </u>	BAPTM		73.02	73.02					27.84	27.84		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				BAPTO		150.25	150.25					27.84	27.84		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, CDP				BAPTC		150.25	150.25					27.84	27.84		
		AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Feature Code	ļ	1		BAPTF		150.25	150.25					27.84	27.84		
		AIN Toolkit Service - Query Charge, Per Query					0.0250662										
		AIN Toolkit Service - Type 1 Node Charge, Per AIN Toolkit Subscription, Per Node, Per Query					0.0062979										
		AIN Toolkit Service - SCP Storage Charge, Per SMS Access Account, Per 100	1	1-	 		0.0002979		+ +			+		+		+	+
1		Kilobytes					1.73										
			+	+					1		+	+	+	+		+	1
		AIN Toolkit Service - Monthly report - Per AIN Toolkit Service Subscription AIN Toolkit Service - Special Study - Per AIN Toolkit Service Subscription				BAPMS BAPLS	15.93 0.0872769	72.15 47.35	72.15					27.84 27.84	27.84 27.84		

ANT TOTAL SCANCE, CASE From Papers, For Arth Total Scance Securities Security Secu	CATEGORY	NOTES UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
## NACES ONLY SEATER PROPERTY PROPERTY SEATER SEATE	MILGORI	NOTES CHOCKELEMENT	III.CTIIII ESIIC	500	0000			KATEO (ψ)					00010	ATEO (#)	Incremental	Incremental
APPLICATION APPLICATION											Submitted	Submitted	Charge - Manual	Charge - Manual	Charge - Manual Svc Order vs. Electronic-Disc	Charge - Manual Svc Order vs. Electronic-Dis
WIT Total Service - Call Event Report - Far ANT Total Service Scientifican 1.00															1st	Add'I
1			+-+				Nonre	curring								
ACCESS DATE (ALL PROPERTY Spocial Busy. Per Air Total Service)											SOMEC	SOMAN			SOMAN	SOMAN
ACCESS NAT VISIGE FIX EADY																
ACCESS DAY VINANCE RELIANDY PARTIE Nesseas processing, per messease		AIN TOOIKIT Service - Call Event Special Study - Per AIN Toolkit Service	 		BAPES	0.0029092	47.35	47.35					27.84	27.84		í
ADE: Message Processing and reseases. Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional Design Processing or message. ODE: Message Processing or Message. O	JF/EDC	UF/ADUF/CMDS														
ADE: Message Processing and reseases. Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional, DAY VIANGE PER (GOUP) Optional Design Processing or message. ODE: Message Processing or Message. O																-
DANAIGN DEPTENAL DATA VIALOR FLE FOODS						0.004										
COTIONAL DATE PROCESSING Processing		ADUF: Data Transmission (CONNECT:DIRECT), per message														
CODIE Research Focusion F		ENHANCED ORTIONAL DAILY LICAGE FILE (FORLIE)	++													
OFFICIAL DIAX USAGE FIXE FORCE OCHE Missage Processing, per resease OCHE Missage Processing, per resease OCHE Missage Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per Research Processing, per R			+-+			0.004										
O.0002862 O.00																
ODUE: Message Processing or tesperating by tesperating port processing or this profit floor portionated (ODM CH. Message) ODM CH. Message) ODM CH. Message ODM CH. Mes			++			0.0003963										
ODUP Message Processing prof Magnetic Tags provisioned ODUP Seal Transmissor (OWNECT PROFEL), pur message ARKED EXTENDED LIME (ELL.) NOTE In Set List annibition in Situate of Georgia, denotity zone 1 of following SMAC Ciscon, FL Mann, FL, FL Luddricks, FLL Habrinia, Tags. NOTE In all Est annibition in Situate of Georgia, denotity zone 1 of following SMAC Ciscon, FL Mann, FL, FL Luddricks, FLL Habrinia, Tags. NOTE In all Est annibition in Situate of Georgia, denotity zone 1 of following SMAC Ciscon, FL Mann, FL, FL Luddricks, FLL Habrinia, Tags. NOTE In all States, ELL reviews demotes Sealersh (MC, Gramerosen-Wirenton Satern-High Pront, NC, Use all rates before except Swarth As in Change). NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all States, ELL reviews demotes states the Swarth As in Change. NOTE In all Swarth As in Change. NOTE		ODUF: Necording, per message ODUF: Message Processing, per message				0.0032344										i
NOTE: Nortice States and Section And Section (Section Control of Secti		ODUF: Message Processing, per Magnetic Tape provisioned														
NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL, Mann, FL, FL, Enuderdale, FL, Nashville, TN, New Orlando, LA, NOTE: In Central Accounts Accounts (Accounts A		ODUF: Data Transmission (CONNECT:DIRECT), per message	+-+			0.0000357										i
NOTE: In all states, ELL network demonstrate, Port Not. Use all states below except Switch As Is Charge, NOTE: In all states, ELL network demonstrates about health and accommentation. Use Tables, A Switch As Is Charge applies to currently combined confidence commend to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commend retwork elements sport the GR PSG cortect, Will Switch As Is Charge.) NOTE: In all states, ELL network demonstrates apply to ordinarily commend to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commend to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commend to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, ELL network demonstrates apply to ordinarily commended to UNEs. Non-necurring ordinarily commended to UNEs. Non-necurring value apply commended to UNEs. Non-necurring rates do not apply.) NOTE: In all states, EVERLAGE LOOP and In all states and the UNES. Non-necurring value apply commended to UNEs. Non-necurring value apply commended to UNEs. Non-necurring commended to UNEs. Non-necurring commended to UNEs. Non-necurring commended to UNEs. Non-necurring commended to UNEs. Non-nec	ANCE	D EXTENDED LINK (EELs)														
NOTE: In all states, ELL retwork demonstrate shown below and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly contributed and apply to correctly apply to the apply to correctly apply to the apply to correctly apply to the apply to correctly apply to the apply to the apply to correctly apply to the apply																—
NOTE: In all states, EEL network elements apt to currently combined facilities which are converted to INE rates. A Switch As Is Charge apples to currently combined facilities on extra poly.) NOTE: In Georgia, the EEL network elements apt to drawly combined retwork elements apt to the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control, the APS Control of the APS Control, the APS Control, the APS Control of the APS		NOTE: New EELs available in State of Georgia, density zone 1 of following SMAs: Orlando, FL;	Miami, FL; Ft. Lau	iderdale, FLI; Nas	shville, TN; New	Orleans, LA;										
NOTE In Georgia, the EEL network elements apply to ordinarily continend network elements apply to ordinarily continen						Is Charge app	lies to currently	v combined fac	ilities conve	rted to UNE:	s.(Non-recur	ring rates do	not apply.)			
First 2-Wire VG Grade Loop(SL2) in a DS1 Interrofficed Transport Combination - 2						J										
First 2-Wire V6 Capdical Loop(SL2) in a DS1 Interrofficed Transport Combination - 20		A WIDE VOICE CRADE EXTENDED LOOP WITH DEDICATED DOA INTEROFFICE TRANSPO	DT (EEL)													
First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination				UNCVX	UEAL2	21.57										
First 2-Wire VG Grade Loop(SL2) in DSI Interofficed Transport Combination - 2		First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed Transport Combination -														
Zone 3			2	UNCVX	UEAL2	32.53										
Interoffice Transport - Dedicated - DSI combination - Facility Termination per UNCIX UNCIX UTFT 77.14			3	UNCVX	UEAL2	43.08										ı
DS1 Channelization System Per Month Notice Grade Coci D-18st To 28th Interesting Fer Month Notice Grade Coci D-18st To 28th Interesting Fer Month Notice Grade Loop in a DS1 Interesting Fransport 1				UNC1X	1L5XX	0.3415										
Voice Grade COCI - DS1 To Ds0 Interface - Per Month LINCVX 1D1VG 0.7012																
Combination - Zone 1		Voice Grade COCI - DS1 To Ds0 Interface - Per Month														ı
Each Additional 2-Wire VS Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 2 Each Additional 2-Wire VS Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3 Voice Grade COCI - DS1 to DS0 Channel System combination - per month Visice Grade COCI - DS1 to DS0 Channel System combination - per month Nonrecurring Currently Combined Network Elements Switch - As-1s Charge Visice Grade COCI - DS1 to DS0 Channel System combination - per month Visice Grade COCI - DS1 to DS0 Channel System combination - per month Visice Grade COCI - DS1 to DS0 Channel System combination - per Month Visice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 Interoffice Transport - Dedicated - DS1 - Combination - Per Minth Visice Visice Transport - Dedicated - DS1 - Combination - Per Minth Visice Transport - Dedicated - DS1 - Combination - Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Facility Termination Per Month Visice Transport - Dedicated - DS1 - Visice Transport Combination - Visice Transport Combination - Visice Transport Combination - Visice Average Visice Transport Combination - Visice Transport Combination - Visice Transport Combination - Visice Average Visice Transport Combination - Visice Solvey Digital Grade L				LINCVO	LIEALO	24.57										ı
Combination - Zone 2		Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport	 	UNCVA	UEALZ	21.57										
Combination - Zone 3		Combination - Zone 2	2	UNCVX	UEAL2	32.53										
Work Wick Crade COCI - DS1 to DS0 Charnel System combination - per month UNCVX 101VG 0.7012			2	LINCVY	LIEAL2	43.08										ı
### VINCE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 1		Voice Grade COCI - DS1 to DS0 Channel System combination - per month	- 3			0.7012										
First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - 2		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.
First 4-Wire Analgy Voice Grade Loop in a DS1 Interoffice Transport Combination - 2		4-WIRE VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPO	RT (FFL)			+										í
First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - 2 UNCVX UEAL4 44.44 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - 2 UNCVX UEAL4 58.85 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month UNC1X 1L5XX 0.3415 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X 1L5XX 0.3415 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X U1TF1 77.14 Channelization - Channel System DS1 to DS0 combination Per Month UNC1X MQ1 134.46 Voice Grade COCI - DS1 to DS0 Channel System combination - per month UNC1X MQ1 134.46 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport 2 UNCVX UEAL4 29.47 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3 Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 4-Wire 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 UNCDX UDL56 34.26 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 UNCDX UDL56 51.67		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination	- 1													
Zone 2 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - 2 UNCVX UEAL4 44.44 First 4-Wire 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 Interoffice Transport Combination - 2 UNCVX UEAL4 44.44 First 4-Wire 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 Interoffice Transport Combination - 2 UNCX UDL56 51.67 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 58.85 UNCVX UEAL4 58.85 UNCVX UEAL4 44.44 UNCX 115XX 0.3415 UNCX UEAL4 58.85 UNCX UEAL4 44.46 UNCX 1DIVG 0.7012 UNCVX UEAL4 29.47 UNCVX UEAL4 29.47 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1 UNCVX UEAL4 44.44 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3 UNCVX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 44.44 UNCX UEAL4 58.85 UNC			1	UNCVX	UEAL4	29.47										
First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X UNC1X UNC1X UNC1X UNTF1 T7.14 UNC1X UNTF1 Channelization - Channel System DS1 to DS0 combination Per Month UNC1X Woice Grade COCI - DS1 to DS0 Channel System Combination - per month UNC1X Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3 Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 4-Wire 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 UNCX UDL56 UNCX UDL56 51.67			2	UNCVX	UFAL4	44.44										ı
Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month UNC1X 1L5XX 0.3415 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X U1TF1 77.14 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X U1TF1 77.14 Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month UNC1X UNC1X MQ1 134.46 Interoffice Transport UNC1X MQ1 134.46 Interoffice Transport UNC1X U		First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice Transport Combination														
Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month		Zone 3 Interoffice Transport - Dedicated - DS1, combination - Per Mile Per Month	3		UEAL4											
Channel Ization - Channel System DS1 to DS0 combination Per Month UNC1X MQ1 134.46																
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 1 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2 Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 2 UNCVX UEAL4 44.44 UNCVX UEAL4 44.44 UNCVX UEAL4 44.44 UNCVX UEAL4 58.85 UNCVX U		Channelization - Channel System DS1 to DS0 combination Per Month		UNC1X												-
Combination - Zone 1		Voice Grade COCI - DS1 to DS0 Channel System combination - per month Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport	+-+	UNCVX	1D1VG	0.7012										i
Combination - Zone 2		Combination - Zone 1	1	UNCVX	UEAL4	29.47										ı
Additional 4-Wire Analog Voice Grade Loop in same DS1 Interoffice Transport Combination - Zone 3 UNCVX UEAL4 58.85 Nonrecurring Currently Combined Network Elements Switch -As-Is Charge UNC1X UNCCC 11.21 11.21 13.99 13.99 31.38 31.38 4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 UNCDX UDL56 34.26 First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 UNCDX UDL56 51.67				1110101	115414											ı
Combination - Zone 3			2	UNCVX	UEAL4	44.44										i
4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPORT (EEL) First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination 1 UNCDX UDL56 34.26		Combination - Zone 3	3			58.85										l
First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination- Zone 2 UNCDX UDL56 34.26 2 UNCDX UDL56 51.67		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	+	UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.
First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 1 First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - 2 Zone 2 UNCDX UDL56 34.26 2 UNCDX UDL56 51.67		4-WIRE 56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANS	PORT (EEL)		+			1				 				
First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 2 2 UNCDX UDL56 51.67		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination														
Zone 2 2 UNCDX UDL56 51.67			1 1	UNCDX	UDL56	34.26							-			
		Zone 2		UNCDX	UDL56	51.67										İ
		First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination														
- Zone 3			3				-						-	-		i
Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per		Interoffice Transport - Dedicated - DS1 - combination Facility Termination Per														 I
Month UNC1X U1TF1 77.14 31.38 31.38		Month		UNC1X	U1TF1	77.14	L				1		31.38	31.38	3.94	3.

1	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC	1		RATES (\$)		1			OSS R	ATES (\$)	1	
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc I Order vs. Electronic-Disc	Incre Cha Manu Ord Electro
							Nonre	curring		ecurring						
						Rec	First	Add'I	Pirst	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	so
	Channelization - Channel System DS1 to DS0 combination Per Month			UNC1X	MQ1	134.46	FIISt	Add I	FIISt	Addi	SUMEC	SOMAN	SOMAN	SOMAN	SOMAN	30
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	1.49										
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport															
	Combination - Zone 1		1	UNCDX	UDL56	34.26							31.38	31.38	3.94	
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		_													
	Combination - Zone 2 Additional 4-Wire 56Kbps Digital Grade Loopin same DS1 Interoffice Transport		2	UNCDX	UDL56	51.67							31.38	31.38	3.94	
	Combination - Zone 3		3	UNCDX	UDL56	68.43							31.38	31.38	3.94	
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-		3	UNCDA	ODLOG	00.43							31.30	31.30	3.34	
	64kbs)			UNCDX	1D1DD	1.49										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE 64	KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1 INTEROFFICE TRANSI	PORT (EI	EL)													
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination															
	- Zone 1 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination		1	UNCDX	UDL64	34.26										-
	- Zone 2		2	UNCDX	UDL64	51.67										
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination			ONODA	ODLOT	31.07										1
	- Zone 3		3	UNCDX	UDL64	68.43										
	Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month			UNC1X	1L5XX	0.3415										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per															
	Month			UNC1X	U1TF1	77.14										ļ
	Channelization - Channel System DS1 to DS0 combination Per Month OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-			UNC1X	MQ1	134.46										
	64kbs)			UNCDX	1D1DD	1.49	0	0								
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport			OHODA	10100	11.10	-									
	Combination - Zone 1		1	UNCDX	UDL64	34.26										
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport															
	Combination - Zone 2		2	UNCDX	UDL64	51.67										
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1 Interoffice Transport		3	UNCDX	LIDLO4	00.40										
	Combination - Zone 3 OCU-DP COCI (data) - DS1 to DS0 Channel System combination - per month (2.4-		3	UNCDX	UDL64	68.43										
	64khs)			LINCIDX	1D1DD	1 49										
	64kbs) Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX UNC1X	1D1DD UNCCC	1.49	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X UNC1X	1D1DD UNCCC	1.49	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR	T (EEL)		UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1	T (EEL)	1	UNC1X UNC1X	UNCCC	59.61	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2	T (EEL)	2	UNC1X UNC1X UNC1X	UNCCC USLXX USLXX	59.61 89.9	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3	T (EEL)		UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX	59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month	T (EEL)	2	UNC1X UNC1X UNC1X	UNCCC USLXX USLXX	59.61 89.9	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month	T (EEL)	2	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	USLXX USLXX USLXX 1L5XX U1TF1	59.61 89.9 119.06			13.99	13.99					3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per	T (EEL)	2	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	USLXX USLXX USLXX 1L5XX	59.61 89.9 119.06 0.3415	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		2	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	USLXX USLXX USLXX 1L5XX U1TF1	59.61 89.9 119.06 0.3415										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 10 IGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	USLXX USLXX USLXX USLXX 1L5XX U1TF1 UNCCC	59.61 89.9 119.06 0.3415 77.14										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1		3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX 1L5XX U1TF1 UNCCC	59.61 89.9 119.06 0.3415 77.14										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonth Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2		1 2	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX IL5XX USLXX USLXY UNCCC USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge B1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge B1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month		3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX 1L5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Declicated - DS1 combination - Per Mile Per Month Interoffice Transport - Declicated - DS1 combination - Per Mile Per Month Interoffice Transport - Declicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month		1 2	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X	UNCCC USLXX USLXX USLXX 1L5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 1-Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per 1-Month 1-Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-1-		1 2	UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X	UNCCC USLXX USLXX 1L5XX 1L5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1 Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1 Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1 Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month		1 2 3	UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX USLXX IL5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 1 Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per 1 Month 1 Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 1 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month 1 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month 1 Interoffice Transport - Dedicated - DS3 - Facility Termination per month 1 DS3 to DS1 Channel System combination per month 1 DS3 Interoffice DS1 Interoffice Transport Combination - Zone 1 1 Additional DS1 Loop in DS3 Interoffice Transport Combination - Per Mile Per Month 1 DS3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month 1 DS3 Interoffice Transport Combination per month 1 DS3 Interoffice Transport Combination - Zone 1		1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX 1L5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UTF3 MQ3 UC1D1 USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month DS3 Interoffice Transport DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2		1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX USLXX IL5XX U1TF1 UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8										
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interoffice Transport Combination - Zone 3 DS3 Interoffice Transport Combination - Zone 3 DS3 Interoffice Transport Combination - Zone 3 DS3 Interoffice Transport Combination - Zone 3		1 2 3	UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX USLXX 1L5XX U1TF1 UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1 Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1 Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1 Loop in DS3 Interoffice Transport Combination - Zone 2		1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX USLXX IL5XX UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06										
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interace Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	T (EEL)	1 2 3	UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX USLXX 1L5XX U1TF1 UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge S1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge	T (EEL)	1 2 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X	UNCCC USLXX USLXX USLXX USLXX USLXX USLXY UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UC1D1 USLXX USLXX USLXX USLXX USLXX USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 SS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	T (EEL)	1 1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX IL5XX 1L5XX UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UTF3 MO3 UC1D1 USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UC1D1 UNCCC	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 1-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS1 combination - Zone 1 1-First DS1 Loop in DS3 Interoffice Transport Combination - Zone 2 1-First DS1 Loop in DS3 Interoffice Transport Combination - Zone 2 1-First DS1 Loop in DS3 Interoffice Transport Combination - Zone 3 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - DS3 - Per Mile Per Month 1-Interoffice Transport - Dedicated - Per Mile Per	T (EEL)	1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXY USLXX USLXX USLXY USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 SS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1	T (EEL)	1 1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X	UNCCC USLXX USLXX USLXX IL5XX 1L5XX UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UTF3 MO3 UC1D1 USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UC1D1 UNCCC	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interdace Unit (DS1 COC1) combination per month DS3 Interdace Unit (DS1 COC1) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 DS3 Interface Unit (DS1 COC1) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOR 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month	T (EEL)	1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX	UNCCC USLXX USLXX USLXX USLXX USLXX USLXX UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06 10.8 59.61 10.8	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 51 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interface Unit (DS1 COC) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interface Unit (DS1 COC) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DCCE GRADE EXTENDED LOOP 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 1 1-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 1-WireVG Loop used with 2-wire VG Interoffice T	T (EEL)	1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX	UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UC1D1 USLXX UC1D1 USLXX UC1D1 USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06 10.8 59.61 10.8	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interdace Unit (DS1 COC1) combination per month DS3 Interdace Unit (DS1 COC1) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 DS3 Interface Unit (DS1 COC1) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOR 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month	T (EEL)	1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX	UNCCC USLXX USLXX USLXX USLXX USLXX USLXX UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 19.06 10.8 59.61 19.06 10.8	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Sacility Termination per month DS3 to DS1 Channel System combination per month DS3 to DS1 Channel System combination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interdec Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge INCE GRADE EXTENDED LOOP/2 WIRE VOICE GRADE INTEROFFICE TRANSPOR 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedic	T (EEL)	1 2 3 3 1 2 3 3 1 1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX	UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UC1D1 USLXX UC1D1 USLXX UC1D1 USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 19.06 10.8 59.61 19.06 10.8	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge S1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combinat	T (EEL)	1 2 3 3 1 2 3 3 1 1 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC3X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX UNCYX	UNCCC USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06 10.8 21.57 32.53 43.08 0.0167	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG com	T (EEL)	1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 1 1 1 1	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UN	UNCCC USLXX USLXX USLXX USLXX USLXX UNCCC USLXX USLX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 10.8 59.61 10.8 21.57 32.53 43.08 0.0167 24.3	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 - Facility Termination per month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 to DS1 Channel System combination per month DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3 DS3 Interace Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2 2-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 2 4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 1 4-WireVG Loop used with 4-wir	T (EEL)	1 1 2 3 3 1 1 2 3 3 1 1 2 2 3 3	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UN	UNCCC USLXX USLXX USLXX USLXX IL5XX UNCCC USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX USLXX UT1F3 UC1D1 USLXX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 89.9 119.06 10.8 21.57 32.53 43.08 0.0167 24.3	11.21	11.21	13.99	13.99			31.38	31.38	3.94	
4-WIRE DS	Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTEROFFICE TRANSPOR 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 2 4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3 Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS1 combination - Facility Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge 31 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPOR First DS1Loop in DS3 Interoffice Transport Combination - Zone 1 First DS1Loop in DS3 Interoffice Transport Combination - Zone 2 First DS1Loop in DS3 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month Interoffice Transport - Dedicated - DS3 - Facility Termination per month DS3 Interface Unit (DS1 COCI) combination per month DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 1 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 2 DS3 Interface Unit (DS1 COCI) combination per month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge DICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE INTEROFFICE TRANSPOI 2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG combination - Per Mile Per Month Interoffice Transport - Dedicated - 2-wire VG com	T (EEL)	1 1 2 3 3 1 1 2 3 3 1 1 1 2 3 3 1 1 1 2 1 1 1 1	UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC1X UNC3X UNC3X UNC3X UNC3X UNC1X UN	UNCCC USLXX USLXX USLXX USLXX USLXX UNCCC USLXX USLX	59.61 89.9 119.06 0.3415 77.14 59.61 89.9 119.06 8.02 880.65 180.03 10.8 59.61 10.8 59.61 10.8 21.57 32.53 43.08 0.0167 24.3	11.21	11.21	13.99	13.99			31.38	31.38	3.94	

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc
								Nonre	curring	Nonr	ecurring	per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
											onnect						
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCVX	UNCCC	Rec	First 11.21	Add'I 11.21	First 13.99	Add'I 13.99	SOMEC	SOMAN	31.38	31.38	3.94	3.94
		Interior Controlled Notwork Elements Owner 7.5 to Orlange			ONOVA	011000		11.21	11.21	10.00	10.00			01.00	01.00	0.54	0.54
		AL EXTENDED LOOP WITH DEDICATED DS3 INTEROFFICE TRANSPORT (EEL)															
		High Capacity Unbundled Local Loop - DS3 combination - Per Mile per month High Capacity Unbundled Local Loop - DS3 combination - Facility Termination per			UNC3X	1L5ND	15.33										
		month			UNC3X	UE3PX	382.95										
		Interoffice Transport - Dedicated - DS3 - Per Mile per month			UNC3X	1L5XX	8.02										
		Interoffice Transport - Dedicated - DS3 combination - Facility Termination per per			10001												
		month Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC3X UNC3X	U1TF3 UNCCC	880.65	11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
		Interior Controlled Notwork Elements Owner 7.5 to Orlange			ONOOA	011000		11.21	11.21	10.00	10.00			01.00	01.00	0.54	0.54
	STS1 DIGIT	AL EXTENDED LOOP WITH DEDICATED STS1 INTEROFFICE TRANSPORT (EE	L)														
		High Capacity Unbundled Local Loop - STS1 combination - Per Mile per month			UNCSX	1L5ND	15.33										
		High Capacity Unbundled Local Loop - STS1 combination - Facility Termination per month			UNCSX	UDLS1	391.86										
		Interoffice Transport - Dedicated - STS1 combination - Per Mile per month			UNCSX	1L5XX	8.02										
		Interoffice Transport - Dedicated - STS1 combination - Facility Termination per															
		month			UNCSX	U1TFS	880.55	11.21	44.04	13.99	13.99			31.38	31.38	2.04	3.94
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		†	UNCSX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
		N EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPORT (EEL)															
		First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 1		1	UNCNX	U1L2X	26.68										
		First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 2 First 2-Wire ISDN Loop in a DS1 Interoffice Combination Transport - Zone 3		3	UNCNX	U1L2X U1L2X	40.24 53.85										
		Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNC1X	1L5XX	0.3415										
		Interoffice Transport - Dedicated - DS1 combintion - Facility Termination per month			UNC1X	U1TF1	77.14										
		Channelization - Channel System DS1 to DS0 combination - per month 2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combination - per month			UNC1X UNCNX	MQ1 UC1CA	134.46										
		Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone			UNCIX	UCTCA	3.2										-
		1		1	UNCNX	U1L2X	26.68										
		Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone															
		2 Additional 2-wire IDSN Loop in same DS1Interoffice Transport Combination - Zone		2	UNCNX	U1L2X	40.24										
		3		3	UNCNX	U1L2X	53.85										
		2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel System combintaion- per month			UNCNX	UC1CA	3.2										
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
	4-WIRE DS	1 DIGITAL EXTENDED LOOP WITH DEDICATED STS-1 INTEROFFICE TRANSPO	RT (EEL	.)													
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 1	,	1	UNC1X	USLXX	59.61										
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 2		3	UNC1X UNC1X	USLXX	89.9 119.06										
		First DS1 Loop in STS1 Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - STS1 combination - Per Mile Per Month		3	UNCSX	1L5XX	8.02										
		Interoffice Transport - Dedicated - STS1 combination - Facility Termination			UNCSX	U1TFS	880.55										
		STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	180.03										
		DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1		1	UNC1X UNC1X	UC1D1 USLXX	10.8 59.61										
		Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 2		2	UNC1X UNC1X	USLXX	89.9										
		Additional DS1Loop in STS1 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	119.06										
		DS3 Interface Unit (DS1 COCI) combination per month	-	-	UNC1X UNCSX	UC1D1 UNCCC	10.8	11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		†	UNCOA	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	5.94	3.94
	4-WIRE 56	KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTEROFFICE TRANSPORT (E	EL)														
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	34.26										
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 2 4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56 UDL56	51.67 68.43			+							
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Per Mile			UNCDX	1L5XX	0.0167										
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination - Facility			UNCDX	U1TD5	16.76										
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge		-	UNCDX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
	4-WIRE 64	I KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROFFICE TRANSPORT (E	EL)			1											
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 1	•	1	UNCDX	UDL64	34.26										
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	51.67										
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport Combination - Zone 3 Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Per Mile	-	3	UNCDX	UDL64 1L5XX	68.47 0.0167										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination - Facility			UNCDX	U1TD6	16.76										
		Nonrecurring Currently Combined Network Elements Switch -As-Is Charge			UNCDX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	3.94
ADDITIONA	I NETWOP	 K ELEMENTS		1													
DUITONA	W OR			<u> </u>													
		das a part of a currently combined facility, the non-recurrng charges do not a															
	When used	as ordinarilty combined network elements in Georgia, the non-recurring charge	ges apply	and 1	the Switch As Is Cl	harge does no	ot.										1
				-													1
	Node (Synd	chroNet)		 		 				+							
	Jus (Syric	····				I .	ı		I .	-1	I .	1	1	I.		I.	

GORY N	IOTES UNBUNDLED NETWO	RK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st		Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increment Charge Manual St Order vs Electronic-I Add'I
								Nonre	curring		curring						
											onnect						
	Node per month				UNCDX	UNCNT	Rec 14.55	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
	Node per month				UNCDA	CINCINI	14.55										-
Non	recurring Currently Combined Network Elements	"Switch As Is" Charge (One applies t	o each o	combir	nation)												
	2/4-Wire VG Interoffice Channel used in a C	OMBINATION - "Switch As Is"			,												
	Conversion Charge				UNCVX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	
	56/64 kbps Interoffice Channel used in a CC	MBINATION - "Switch As Is"															i i
	Conversion Charge				UNCDX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	. 3
	DS1 Interoffice Channel used in a COMBINA	TION - "Switch As Is" Conversion															
	Charge				UNC1X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	. :
	DS3 Interoffice Channel used in a COMBINA	ATION - "Switch As Is" Conversion															
	Charge				UNC3X	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	. ;
	STS1 Interoffice or Local Loop used in a CO	MBINATION - "Switch As Is"															
	Conversion Charge				UNCSX	UNCCC		11.21	11.21	13.99	13.99			31.38	31.38	3.94	. 3
NOT	FE: Local Channel - Dedicated Transport - minimu	m hilling period - Below DS3-one mo	nth DS	3 and a	hove-four mont	he											
	2. 200al Glamio. Dodloated Hanopert Illiania	Simily period 2010 it 200 one inc	, 20	1													
RATIONAL	SUPPORT SYSTEMS																
NOT	FE: (1) Electronic Service Order: CLEC-1 should con	act its contract negotiator if it prefers the	ne state s	specific	electronic service	e ordering charg	es as ordered	by the State C	ommissions								1
NOT	TE: (1) Continued: The electronic service ordering cha	arge currently contained in this rate exh	ibit is the	BellSo	outh regional electi	ronic service or	dering charge	•									1
NOT	TE: (1) Concluded: CLEC-1 may elect either the state	specific Commission ordered rates for	the elect	tronic s	ervice ordering ch	arges, or CLEC	-1 may elect th	e regional elec	tronic service	ordering cha	arge.						1
NOT	TE: (2) Manual Service Order charge: disconnect, in	the state of Florida, to be billed on a pe	r LSR ba	sis													
		•															
	Electronic OSS Charge, per LSR, submitted	via BST's OSS interactive interfaces															
	(Regional)					SOMEC		3.5									
	"Zone" shown in the sections for stand-alone loops o		o Geogra	aphical	ly Deaveraged UN	E Zones. To vie	ew Geographic	ally Deaverage	ed UNE Zone	Designations	by Central C	Office, refer t	to Internet V	ebsite:			
http:/	://www.interconnection.bellsouth.com/become_a_cled	/html/interconnection.htm															
							ı		r	т-	1	1	1	ı	1	ı	
JNDLED LC	OCAL EXCHANGE SWITCHING(PORTS)																
	Lawrence Barrier			+								1					
	hange Ports	and the same in CA 8 TN the dealer 15			. b	an makail USCC						1					-
NOI	TE: Although the Port Rate includes all available for	eatures in GA & TN, the desired featu	res Will I	neea to	pe oraerea usir	ig retail USOCS	1			1	l	1	-		-	1	
	UDE VOICE ORADE LINE DODT BATES (DES)											1					1

DLED LOCAL EXCHANGE SWITCHING(PORTS)												
Exchange Ports												
NOTE: Although the Port Rate includes all available features in GA & TN, the desired features	will need to be ordered using	rotail USOCs										
NOTE. Almough the Fort tate monates an available readines in OA a 11, the desired readines	will ficed to be ordered dailing	Tetali 00003										
2-WIRE VOICE GRADE LINE PORT RATES (RES)												
Exchange Ports - 2-Wire Analog Line Port- Res.	UEPSR	UEPRL	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.	UEPSR	UEPRC	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.	UEPSR	UEPRO	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with Caller ID - Res.	UEPSR	UEPAU	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled South Carolina Area Calling port with Caller ID - Res (LW8)	UEPSR	UEPAJ	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled res, low usage line port with Caller ID	UEPSR	UEPAP	2.35	24.98	24.98				44.42	14.63		
Subsequent Activity	UEPSR	USASC	0	0	0							
FEATURES												
All Available Vertical Features	UEPSR	UEPVF	6.29	0	0				44.42	14.63		
2-WIRE VOICE GRADE LINE PORT RATES (BUS)												
Exchange Ports - 2-Wire Analog Line Port without Caller ID - Bus	UEPSB	UEPBL	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled Line Port with unbundled port with Caller+E484 ID - Bus.	UEPSB	UEPBC	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.	UEPSB	UEPBO	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled SC extended local dialing parity Port with												
Caller ID - Bus.	UEPSB	UEPAZ	2.35	24.98	24.98				44.42	14.63		
Exhange Ports - 2-Wire VG unbundled incoming only port with Caller ID - Bus	UEPSB	UEPB1	2.35	24.98	24.98				44.42	14.63		
Exchange Ports - 2-Wire VG unbundled South Carolina Bus Area Calling Port with												
Caller ID - Bus (LMB)	UEPSB	UEPAB	2.35	24.98	24.98				44.42	14.63		
Subsequent Activity	UEPSB	USASC	0	0	0							
FEATURES												
All Available Vertical Features	UEPSB	UEPVF	6.29	0	0				44.42	14.63		
EXCHANGE PORT RATES (DID & PBX)												
Exchange Ports - 2-Wire DID Port	UEPEX	UEPP2	8.86	239.14	37.56	120.05	7.54		67.52	67.52		
Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID capability	UEPDD	UEPDD	73.62	404.94	191.8	145.5	4.93		19.99	19.99	19.99	
Exchange Ports - 2-Wire ISDN Port (See Notes below.)	UEPTX UEPSX	U1PMA	13.38	145.86	106.21	95.79	21.52		67.52	67.52		
All Features Offered	UEPTX UEPSX	UEPVF	6.29	0	0							
NOTE: Transmission/usage charges associated with POTS circuit switched usage will also apply to o												
NOTE: Access to B Channel or D Channel Packet capabilities will be available only through BFR/Nev	w Business Request Process.	Rates for the p	acket capabili	ities will be det	ermined via the	Bona Fide F	Request/New Bus	iness Request P	rocess.			
Exchange Ports - 2-Wire ISDN Port Channel Profiles	UEPTX UEPSX	U1UMA	0	0	0							
Exchange Ports - 4-Wire ISDN DS1 Port	UEPEX	UEPEX	107.44	408.53	203.56	158.7	21.52		65.48	65.48		
2-Wire VG Unbundled 2-Way PBX Trunk - Res	UEPSE	UEPRD	2.35	24.36	24.36				41.86	14.46		
2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus	UEPSP	UEPPC	2.35	24.36	24.36				41.86	14.46		
2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus	UEPSP	UEPPO	2.35	24.36	24.36				41.86	14.46		

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
CALLGORI	NOTES	ONDONDEED NET WORK ELEMENT	Internii Lone	500	0000			IXATEO (4)					0001	ΑΙ ΕΟ (ψ)	Incremental	Incremental
															Charge -	Charge -
											Svc Order Submitted	Svc Order Submitted	Incremental	Incremental	Manual Svc Order vs.	Manual Svc Order vs.
											Elec	Manually per	Svc Order vs.	Charge - Manual Svc Order vs.	Electronic-Disc	Electronic-Disc
											per LSR	LSR	Electronic-1st	Electronic-Add'l	1st	Add'l
							Nonre	curring	Nonre	curring						
									Disco	onnect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus		UEPSP	UEPP1	2.35	24.36	24.36					41.86	14.46		
		2-Wire Analog Long Distance Terminal PBX Trunk - Bus		UEPSP	UEPLD	2.35	24.36	24.36					41.86	14.46		
		2-Wire Voice Unbundled PBX LD Terminal Ports		UEPSP	UEPLD	2.35	24.36	24.36					41.86	14.46		
		2-Wire Vice Unbundled 2-Way PBX Usage Port		UEPSP UEPSP	UEPXA	2.35 2.35	24.36 24.36	24.36 24.36					41.86	14.46		
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports 2-Wire Voice Unbundled PBX LD DDD Terminals Port		UEPSP	UEPXB UEPXC	2.35	24.36	24.36					41.86 41.86	14.46 14.46		
		2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port		UEPSP	UEPXD		24.36	24.36					41.86	14.46		
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port		UEPSP	UEPXE	2.35 2.35	24.36	24.36					41.86	14.46		
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling		OLFOF	OLFAL	2.00	24.30	24.30					41.00	14.40		
		Port		UEPSP	UEPXL	2.35	24.36	24.36					41.86	14.46		ı
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port		UEPSP	UEPXM	2.35	24.36	24.36					41.86	14.46		
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room														
		Calling Port		UEPSP	UEPXO	2.35	24.36	24.36					41.86	14.46		ı
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port		UEPSP	UEPXS	2.35 2.35	24.36	24.36					41.86	14.46		
		2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus Calling Port		UEPSP	UEPXT	2.35	24.36	24.36					41.86	14.46		
		Subsequent Activity		UEPSP	USASC	0	0	0						1		
	FEATURES															
		All Available Vertical Features	UE	PSP UEPSE	UEPVF	6.29	0	0					41.86	14.46		
	EXCHANGE	PORT RATES (COIN)														
		Exchange Ports - Coin Port				2.77	24.75	24.75					43.48	14.57		
																_
		ching Features offered with Port						1								-
	NOTE: Trai	nsmission/usage charges associated with POTS circuit switched usage will also apply	y to circuit switched	voice and/or ci	rcuit switched	data transmiss	ion by B-Chan	nels associate	d with 2-wire	ISDN ports	i.					
	NOTE: Acc	ess to B Channel or D Channel Packet capabilities will be available only through BFR	/New Business Req	uest Process.	Rates for the p				Bona Fide F	Request/Ne	w Business I	Request Pro				
		Exchange port - 4-wire ISDN trunk port -all available features included			UEPEX	251	311.73	311.73					65.48	65.48		-
		Exchange Port - 2-wire ISDN digital line side port with three features included			U1PMA	36.01	70.32	70.32					67.52	67.52		
UNBUNDLE	D LOCAL S	WITCHING, PORT USAGE														
																
	End Office	Switching (Port Usage)														
		End Office Switching Function, Per MOU				0.0019295										
		End Office Trunk Port - Shared, Per MOU				0.0002581										
	Tandom Cu	interpretation (Port Usage) (Local or Access Tandem)														
	ranuem sv	Tandem Switching Function Per MOU				0.0006843										
		Tandem Trunk Port - Shared, Per MOU				0.0004034			-							i
		Tanachi Trank Tork Onarca, Fer Moo				0.0004004										i
	Common T	ransnort														ı
		Common Transport - Per Mile, Per MOU				0.0000121										í
		Common Transport - Facilities Termination Per MOU				0.0004672										
																i
UNBUNDLE	D PORT/LO	OP COMBINATIONS - COST BASED RATES														i
																i
	Cost Based	Rates are applied where BellSouth is required by FCC and/or State Commission rule	to provide Unbundle	ed Local Switch	ing or Switch I	Ports.										i
	Features sh	nall apply to the Unbundled Port/Loop Combination - Cost Based Rate section in the s	ame manner as they	are applied to	the Stand-Alor	ne Unbundled P	ort section of	this Rate Exhib	oit.							
	End Office a	and Tandem Switching Usage and Common Transport Usage rates in the Port section	of this rate exhibit s	shall apply to all	combinations	of loop/port ne	twork elements	s except for Ul	NE Coin Port	Loop Comb	oinations.					
	For Georgia	a and Tennessee, the recurring UNE Port and Loop charges listed apply to Currently C	Combined and Not C	urrently Combin	ed Combos ar	nd the first and	additional Port	t nonrecurring o	charges apply	to Not Cur	rently Combi	ned Combos	s. For Curren	tly Combined	Combos in G	A, TN and
	all other sta	tes, the nonrecurring charges shall be those identified in the Nonrecurring - Currently	Combined sections.													
	2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)														
	UNE Port/L	oop Combination Rates		-												
		2-Wire VG Loop/Port Combo - Zone 1	1	· ·		20.71										
		2-Wire VG Loop/Port Combo - Zone 2	2			29.35										
ļ		2-Wire VG Loop/Port Combo - Zone 3	3			37.68	ļ		1				-	1		
ļ		<u></u>				1	ļ		1				-	1		
	UNE Loop I			HEDDY	LIEDLY	47.00		-					-			
	-	2-Wire Voice Grade Loop (SL1) - Zone 1	1	UEPRX	UEPLX	17.02		 			-	1	-	-		
		2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3	3	UEPRX UEPRX	UEPLX UEPLX	25.66 33.99	 	+	_		+	-	-	1		
		z ***** ***** Oldo Glade Loop (OLT) - Zolle 3	3	JEFIX	OLFLA	55.88		+	1		+	+	1	1		i
	2-Wire Vein	ce Grade Line Port Rates (Res)						+	 			 	 	<u> </u>		
	T TAILE AOIC	2-Wire voice unbundled port - residence		UEPRX	UEPRL	3.69		+	 			 	43.19	9.91		
		2-Wire voice unbundled port with Caller ID - res		UEPRX	UEPRC	3.69		1			1	†	43.19	9.91		
		2-Wire voice unbundled port outgoing only - res		UEPRX	UEPRO	3.69		1					43.19	9.91		
		2-Wire voice Grade unbundled South Carolina extended local dialing parity port with			020	0.00								0.01		
		Caller ID - res		UEPRX	UEPAU	3.69							43.19	9.91		ı
		2-Wire voice unbundled South Carolina Area Calling port with Caller ID - res (LW8)		UEPRX	UEPAJ	3.69							43.19	9.91		
		2-Wire voice unbundles res, low usage line port with Caller ID (LUM)		UEPRX	UEPAP	3.69							43.19	9.91		
																 I
	FEATURES															
		All Features Offered		UEPRX	UEPVF	6.29	0	0					43.19	9.91		
'																

NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC	ļ		RATES (\$)		,		,	OSS R	ATES (\$)	,	
											Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manua Svc Order vs.	Incremental Charge - Manual Svc order vs. Electronic-Disc	In M (c) Ele
		\vdash	+				Nesse	curring	Name		per LSR	LSR	Electronic-1st	Electronic-Add'l	I 1st	+
			+-+		+		Nonre	curring		ecurring connect						+
			+		+	Rec	First	Add'I	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	٠,
					1											
LOCAL NU	JMBER PORTABILITY			 I												
	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										
			\perp													
NONRECU	JRRING CHARGES (NRCs) - CURRENTLY COMBINED	├	\bot	LIEDDY	110400		4.50	0.4					10.10	0.04		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with	-	+-+	UEPRX	USAC2		1.59	0.4					43.19	9.91		-
	change	1		UEPRX	USACC		1.59	0.4					43.19	9.91		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent		+	OLFIX	USACC		1.55	0.4					45.15	3.31		+
	Database Update	1		İ			0.71						8.91			
				i												
ADDITION				ļ												
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity		\perp	UEPRX	USAS2	0	0	0								
0 MIDE 1/6	DIOC OR ADE LOOP WITH A WIRE LINE BORT (DUO)	├	+-+		-											_
2-WIRE VO	DICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	├	+													+
LINE Do-4/I	Loop Combination Rates		+		+						 	 				+
JINL FUIT/I	2-Wire VG Loop/Port Combo - Zone 1	$\overline{}$	1		+	20.71				1	l	1	1		1	+
	2-Wire VG Loop/Port Combo - Zone 1		2		+	29.35				1	1	1				+
	2-Wire VG Loop/Port Combo - Zone 3		3		†	37.68										t
				<u> </u>												L
UNE Loop										1						
	2-Wire Voice Grade Loop (SL1) - Zone 1	<u> </u>	1	UEPBX	UEPLX	17.02										1
	2-Wire Voice Grade Loop (SL1) - Zone 2	├	2	UEPBX	UEPLX	25.66										
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	33.99										+
2-Wire Voi	ice Grade Line Port (Bus)		+		+											+
2-11110 101	2-Wire voice unbundled port without Caller ID - bus		+	UEPBX	UEPBL	3.69							43.19	9.91		+
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	3.69							43.19	9.91		
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	3.69							43.19	9.91		
	2-Wire voice Grade unbundled South Carolina extended local dialing parity port with	1		İ												
	Caller ID - bus		\perp	UEPBX	UEPAZ	3.69							43.19	9.91		
	2-Wire voice unbundled incoming only port with Caller ID - Bus	├	+-+	UEPBX UEPBX	UPEB1 UEPAB	3.69							43.19 43.19	9.91 9.91		_
	2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LMB)		+-+	UEPBA	UEPAB	3.69							43.19	9.91		+
LOCAL NU	JMBER PORTABILITY		+		1											1
	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATURE:				ļ												
	All Features Offered		\perp	UEPBX	UEPVF	6.29	0	0					43.19	9.91		
NONDECLI	JRRING CHARGES (NRCs) - CURRENTLY COMBINED		+-+		-											_
NUNKECU	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is		++	UEPBX	USAC2		1.59	0.4								+
													43.10	0.01		
	12-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with	l .		OLI DX				0.4					43.19	9.91		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with change			UEPBX	USACC		1.59	0.4					43.19	9.91		
	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent		\vdash		USACC		1.59							9.91		
	change				USACC		1.59 71						43.19 8.91	9.91		
ADDITIO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update				USACC									9.91		
ADDITION	change - Z-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent - Database Update - AL NRCs			UEPBX									8.91			
ADDITION	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update				USACC USAS2									9.91		
	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCs 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX									8.91			
	change - Z-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent - Database Update - AL NRCs			UEPBX									8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCs 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			UEPBX									8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1		1	UEPBX		20.71							8.91			
2-WIRE VO	change - Z-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent - Database Update - AL NRCS - Z-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity - DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) - Loop Combination Rates - Z-Wire VG Loop/Port Combo - Zone 1 - Z-Wire VG Loop/Port Combo - Zone 2		2	UEPBX		29.35							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates [2-Wire VG Loop/Port Combo - Zone 1			UEPBX									8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		2	UEPBX		29.35							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates		3	UEPBX	USAS2	29.35 37.68							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 3 2-Wire VG Loop/Port Combo - Zone 3 Rates [2-Wire Voice Grade Loop (SL 1) - Zone 1		2	UEPBX UEPBX UEPRG	USAS2	29.35 37.68							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates		3	UEPBX	USAS2	29.35 37.68							8.91			
2-WIRE VO	change - Z-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX	29.35 37.68 17.02 25.66							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 ice Grade Line Port Rates (RES - PBX)		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX	29.35 37.68 17.02 25.66 33.99							8.91	9.91		
2-WIRE VO	change - Z-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX	29.35 37.68 17.02 25.66							8.91			
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 ice Grade Line Port Rates (RES - PBX) 2-Wire Voi Unbundled Combination 2-Way PBX Trunk Port - Res		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX	29.35 37.68 17.02 25.66 33.99							8.91	9.91		
2-WIRE VO	change		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX UEPRD	29.35 37.68 17.02 25.66 33.99							8.91	9.91		
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 ice Grade Line Port Rates (RES - PBX) 2-Wire Voi Unbundled Combination 2-Way PBX Trunk Port - Res		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX	29.35 37.68 17.02 25.66 33.99							8.91	9.91		
2-WIRE VO UNE Port/I UNE Loop 2-Wire Voi	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 ice Grade Line Port Rates (RES - PBX) 2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res JMBER PORTABILITY Local Number Portability (1 per port)		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX UEPRD	29.35 37.68 17.02 25.66 33.99							8.91	9.91		
2-WIRE VO	change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent Database Update AL NRCS 2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity DICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX) Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL 1) - Zone 1 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 2 2-Wire Voice Grade Loop (SL 1) - Zone 3 ice Grade Line Port Rates (RES - PBX) 2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res JMBER PORTABILITY Local Number Portability (1 per port)		1 2	UEPBX UEPBX UEPBX UEPRG UEPRG UEPRG UEPRG	USAS2 UEPLX UEPLX UEPLX UEPLX UEPRD	29.35 37.68 17.02 25.66 33.99							8.91	9.91		

NOT	TES UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Svc Order vs.	Incremental Charge - Manual Svc al Order vs. Electronic-Disc	Incren Char Manua Orde Electron
+							Nonrec	urring	Nonr	ecurring	per LSR	LSR	Electronic-1st	Electronic-Add	'I 1st	Ad
1							, , , , ,	urring		onnect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SON
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As															
	ls			UEPRG	USAC2		1.59	0.4					43.19	9.91		
	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch wit	ו		UEPRG	USACC		1.59	0.4					43.19	9.91		
+	Change 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent			UEPRG	USACC		1.59	0.4					43.19	9.91		
	Database Update						0.71						8.91			
+	Database Opuate						0.71						0.51			
ADDI	TIONAL NRCs															
1	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPRG	USAS2	0	0	0								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					19.99	19.99	19.99	9
O VACIE	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
2-W IR	RE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															
LINE	Port/Loop Combination Rates															
ONE F	2-Wire VG Loop/Port Combo - Zone 1		1			20.71										
+	2-Wire VG Loop/Port Combo - Zone 2		2			29.35										
1	2-Wire VG Loop/Port Combo - Zone 3		3			37.68										
UNE L	Loop Rates															
$\perp =$	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	17.02										
+	2-Wire Voice Grade Loop (SL 1) - Zone 2		3	UEPPX	UEPLX	25.66										
+	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	33.99										
2-Wir	e Voice Grade Line Port Rates (BUS - PBX)															
2-99116	Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	3.69							43.19	9.91		
1	Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	3.69							43.19	9.91		
1	Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	3.69							43.19	9.91		
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	3.69							43.19	9.91		
	2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	3.69							43.19	9.91		
₩	2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	3.69							43.19	9.91		
_	2-Wire Voice Unbundled PBX LD DDD Terminals Port 2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX UEPPX	UEPXD	3.69 3.69							43.19 43.19	9.91 9.91		
+	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	3.69							43.19	9.91		
+	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Callin	1		OLITA	OLIAL	0.00							40.13	3.51		
	Port	^		UEPPX	UEPXL	3.69							43.19	9.91		
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port			UEPPX	UEPXM	3.69							43.19	9.91		
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room															
+	Calling Port			UEPPX	UEPXO	3.69							43.19	9.91		
 	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port 2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus Calling Port			UEPPX UEPPX	UEPXS	3.69							43.19 43.19	9.91 9.91		
	2-Wife Voice Oribunded 2-Way F BA South Carolina Area Flus Calling For			OLITA	OLIXI	3.03							45.15	3.31		
LOCA	L NUMBER PORTABILITY															
1	Local Number Portability (1 per port)			UEPPX	LNPCP	3.15										
FEAT																
	All Features Offered			UEPPX	UEPVF	6.29	0	0					43.19	9.91		
NONE	FOURDING OUADOES (AIDO-) OURDENEL V COMPINED															
NONK	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch-As															
	le			UEPPX	USAC2		1.59	0.4					43.19	9.91		
+	2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Conversion - Switch wit	า		OLITA	00/102		1.00	0.4					40.10	3.51		
	Change			UEPPX	USACC		1.59	0.4					43.19	9.91		
	2-Wire Voice Grade Loop / Line Port Combination - Conversion - Subsequent															
	Database Update						0.71						8.91			
4000	FIGURE NIDO															
ADDII	FIONAL NRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) - Subsequent Activity			UEPPX	USAS2	0	0	0								
	PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group			UEFFX	U3A32	U	14.64	14.64					19.99	19.99	19.99	9
+							14.04	14.04					10.00	13.33	13.30	,
 																
2-WIR	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															
	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															
	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															
	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT Port/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1					21.06										
	POT/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2					29.7										
UNE F	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT Port/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3															
UNE F	POT/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2					29.7										
UNE F	POT/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3			LIEPCO	LIEPLY	29.7 28.03										
UNE F	RE VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT Port/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3			UEPCO UEPCO	UEPLX UEPLX	29.7 28.03										
UNE F	POT/Loop Combination Rates 2-Wire VG Coin Port/Loop Combo – Zone 1 2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3			UEPCO UEPCO UEPCO	UEPLX UEPLX UEPLX	29.7 28.03										

SC

1	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC		F	RATES (\$)					OSS R	ATES (\$)		
									- (,,			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manua Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Increme Charg Manual Order Electronic Add'
								Nonrec	urring	Nonr	ecurring	percen	LOIX	Electronic lot	Electronic Add	100	Aud
										Disc	connect						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
+		2-Wire Coin 2-Way without Operator Screening and without Blocking (SC) 2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD			UEPCO	UEPSD	4.04							43.19	9.91		
		(SC)			UEPCO	UEPSA	4.04							43.19	9.91		
	į	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSH	4.04							43.19	9.91		
		2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity			LIEDOO	LIEBOO	4.04							40.40	0.04		
		(SC) 2-Wire Coin 2-Way with Operator Screening and: 900 Blocking: 900/976, 1+DDD,			UEPCO	UEPSC	4.04							43.19	9.91		
		011+, and Local (SC)			UEPCO	UEPCC	4.04							43.19	9.91		
+		2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local;			02. 00	02. 00	1.01							10.10	0.01		
		Enhanced Call OPT 3YV (SC)			UEPCO	UEPCE	4.04							43.19	9.91		
		2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local;															
		Enhanced Call OPT AP7 (SC)			UEPCO	UEPCF	4.04							43.19	9.91		
	į	2-Wire Coin Outward without Blocking and without Operator Screening (SC)			UEPCO	UEPSG	4.04							43.19	9.91		
+		2-Wire Coin Outward with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSF	4.04							43.19	9.91		
		2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	4.04							43.19	9.91		
+		2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD,			OLI OO	OLI OU	4.04							40.10	3.51		1
		011+, and Local (SC)			UEPCO	UEPCM	4.04							43.19	9.91		
T		2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, Local;															
		Enhanced Calling OPT 3YW (SC)			UEPCO	UEPCP	4.04							43.19	9.91		
		2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	4.04							43.19	9.91		
١		2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	4.04							43.19	9.91		
AL		L UNE COIN PORT/LOOP (RC) UNE Coin Port/Loop Combo Usage (Flat Rate)		-	UEPCO	URECU	4.05	0	0								
+		UNE COM PON/LOOP COMO Usage (Flat Rate)			UEPCO	URECU	4.05	U	- 0								
LC	OCAL NUM	MBER PORTABILITY															
1		Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
FE	EATURES																
4		All Features Offered			UEPCO	UEPVF	6.29	0	C)				41.86	14.46	5	
NIC	ONDECLID	RING CHARGES - CURRENTLY COMBINED															
NC		2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is			UEPCO	USAC2		1.59	0.4					43.19	9.91		
+		2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch-as-is 2-Wire Voice Grade Loop / Line Port Combination - Conversion - Switch with			UEPCO	USACZ		1.59	0.4					43.19	9.91		
		change			UEPCO	USACC		1.59	0.4					43.19	9.91		
									***						0.0.		
ΑE	DDITIONAL																
		2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPCO	USAS2		0	0					43.19	9.91		
2-1	WIRE VOI	CE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNK PORT															
UN		pop Combination Rates															
		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1		1			29.68										
-		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2		2			37.74										
+		2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		3			44.4										
UN	NE Loop R	Pates															
-		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX	UECD1	20.85							19.99	19.99	19.99	19
T		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX	UECD1	28.91							19.99	19.99	19.99	19
		2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX	UECD1	35.57							19.99	19.99	19.99	19
٠																	
UN	NE Port Ra				UEPPX	UEPD1	8.83							19.99	19.99	40.00	
		Exchange Ports - 2-Wire DID Port			UEPPX	UEPDI	8.83							19.99	19.99	19.99	,
	ONRECUR	RING CHARGES - CURRENTLY COMBINED															
NC				1	UEPPX	USAC1		14.62	3.73					19.99	19.99	19.99	19
NC		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination - Switch-as-is															
NC		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth					1 1	44.00	3.73	1	1	1		19.99	19.99	19.99	19
NC					UEPPX	USA1C		14.62	3.73							1	İ
		2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes			UEPPX	USA1C		14.02	3.73								
	DDITIONA	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs							3.73					10.00		10.00	10
	DDITIONA	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes			UEPPX	USA1C USAS1		53.68	5.75					19.99	19.99	19.99	19
ΑĽ	DDITIONAL	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk							3.73					19.99		19.99	19
ΑĽ	DDITIONAI	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port)			UEPPX	USAS1 NDT	0	53.68	0					19.99	19.99	19.99	19
ΑĽ	DDITIONAL elephone I	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCS 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPPX UEPPX UEPPX	USAS1 NDT NDZ		53.68	0 0					19.99 19.99	19.99 19.99 19.99	19.99 19.99	19
ΑĽ	DDITIONAL elephone I	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers			UEPPX UEPPX UEPPX UEPPX UEPPX	USAS1 NDT NDZ ND4	0	53.68 0 0	0 0 0					19.99	19.99	19.99	19
ΑĽ	DDITIONA elephone I	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers DID Numbers, Per Number			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	USAS1 NDT NDZ ND4 ND5	0	53.68 0 0 0 0	0 0 0				19.99	19.99 19.99	19.99 19.99 19.99	19.99 19.99	19
ΑŒ	DDITIONA elephone I	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers DID Numbers, Non- consecutive DID Numbers, Per Number Reserve Non-Consecutive DID numbers			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	NDT NDZ ND4 ND5 ND6	0	53.68 0 0 0 0 0	0 0 0 0				19.99	19.99 19.99	19.99 19.99 19.99	19.99 19.99	19
ΑĽ	DDITIONA elephone I	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers DID Numbers, Per Number			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	USAS1 NDT NDZ ND4 ND5	0	53.68 0 0 0 0	0 0 0					19.99 19.99	19.99 19.99 19.99	19.99 19.99	19
Te	DDITIONAl	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion with BellSouth Allowable Changes L NRCs 2-Wire DID Subsequent Activity - Add Trunks, Per Trunk Number/Trunk Group Establisment Charges DID Trunk Termination (One Per Port) DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers DID Numbers, Non- consecutive DID Numbers, Per Number Reserve Non-Consecutive DID numbers			UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX UEPPX	NDT NDZ ND4 ND5 ND6	0	53.68 0 0 0 0 0	0 0 0 0				19.99	19.99 19.99	19.99 19.99 19.99	19.99 19.99	19 19 19 19

Y I	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Add'I	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic-Di Add'I
								Nonre	curring		ecurring	per con	LOIT	Electronic lot	Licotronic Add 1	104	7001
							_				onnect						
2-V	VIRE ISDI	N DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LINE SIDE PORT					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
UNI	E PORt/LO	pop Combination Rates			UEPPB												
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPR		38.58										
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB UEPPR		48.25										
					UEPPB												
		2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 3		3	UEPPR		55.29										
UN	E Loop R	ates															
		2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB UEPPR	USL2X	27.38							19.99	19.99	19.99	19.99
				2	UEPPB												
		2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPR	USL2X								19.99	19.99	19.99	19.99
		2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB UEPPR	USL2X	44.09							19.99	19.99	19.99	19.99
UN	E Port Ra	ate															
		Exchange Port - 2-Wire ISDN Line Side Port			UEPPB UEPPR	UEPPB	11.2							19.99	19.99	19.99	19.99
NO	NRECUR	RING CHARGES - CURRENTLY COMBINED															
		2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port Combination -															
		Conversion			UEPPB UEPPR	USACB	0	77.18	54.15					19.99	19.99	19.99	19.99
AD	DITIONAL	L NRCs															
LO	CAL NUM	IBER PORTABILITY															
								_									
		Local Number Portability (1 per port)			UEPPB UEPPR	LNPCX	0.35	0	0								
B-C	CHANNEL	USER PROFILE ACCESS:															
		CVS/CSD (DMS/5ESS)			UEPPB UEPPR	U1UCA	0	0	0								
		CVS (EWSD)			UEPPB UEPPR	U1UCB	0	0	0								
-		CSD			UEPPB UEPPR	U1UCC	0	0	0								
B-C	CHANNEL	. AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS SC,MS, & TN)															
		CVS/CSD (DMS/5ESS)			UEPPB UEPPR	U1UCD	0	0	0								
		CVS (EWSD)			UEPPB UEPPR	U1UCE	0	0	0								
		CSD			UEPPB UEPPR	U1UCF	0	0	0								
		COD			OEFFB OEFFR	UTUCF	0	U	U								
US	ER TERM	IINAL PROFILE															
		User Terminal Profile (EWSD only)			UEPPB UEPPR	U1UMA	0	0	0								
VE	RIICAL	EATURES															
		All Vertical Features - One per Channel B User Profile			UEPPB UEPPR	UEPVF	6.29	0	0								
INT	EROFFIC	CE CHANNEL MILEAGE															
		Interoffice Channel mileage each, including first mile and facilities termination			UEPPB UEPPR	M1GNC	20.74	136.44	51.37					19.99	19.99	19.99	19
		Interoffice Channel mileage each, additional mile			UEPPB UEPPR	M1GNM	0.0373	0	0				0				
4-V	VIRE DS1	DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUNK PORT															
LIN	F Port/Lo	pop Combination Rates															
0.4		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 1		1	UEPPP		221.03										
		4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 2 4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE Zone 3		2	UEPPP UEPPP		301.73 434.8										
				J	02/11		.54.0										
UN	E Loop R	tates 4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	113.59							19.99	19.99	19.99	19.9
		4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPPP	USL4P	194.29							19.99	19.99	19.99	19.9
		4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	327.36		-			1		19.99	19.99	19.99	19.9
UN	E Port Ra																
_		Exchange Ports - 4-Wire ISDN DS1 Port			UEPPP	UEPPP	107.44							19.99	19.99	19.99	19.9
		RING CHARGES - CURRENTLY COMBINED	-			 	+	1	l	1	 	+	 	 	 	—	1

1	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incre Cha Manu Ord Electro
								Nonre	ecurring		ecurring	per con	LOIT	Electronic lot	Electronic Add	150	
											connect						
_		4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port Combination -					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	so
		4-Wife DST Digital Loop / 4-Wife ISDN DST Digital Trunk Port Combination - Conversion -Switch-as-is			UEPPP	USACP	0	238.67	157.46					19.99	19.99	19.99	19
+		CONVERSION - SWILCH-dS-15			UEFFF	USACE	0	230.07	137.46					19.99	19.99	19.99	- 13
AD	DITIONA	L NRCs															
		4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy- Inward/two way tel nos															
		within Std Allowance			UEPPP	PR7TF		0.9822						19.99	19.99	19.99	1
		4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port - Outward Tel Numbers (All			HEDDD	DDTTO		00.00	00.00					40.00	40.00	40.00	
		States except NC) 4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port - Subsequent Inward Tel Nos			UEPPP	PR7TO		23.02	23.02					19.99	19.99	19.99	1
		Above Std Allowance			UEPPP	PR7ZT		46.05	46.05					19.99	19.99	19.99	1
LO	CAL NU	MBER PORTABILITY				Lunali											
		Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
+		Voice/Data			UEPPP	PR71V	0	0	0								+
1		Digital Data			UEPPP	PR71D	0	0	0								1
T		Inward Data			UEPPP	PR71E	0	0	0							1	
┇																	L
Ne		itional "B" Channel			-					_							
↓		New or Additional - Voice/Data B Channel		ш	UEPPP	PR7BV	0	29.11						19.99	19.99	19.99	
-		New or Additional - Digital Data B Channel		\vdash	UEPPP	PR7BF	0	29.11	1					19.99	19.99	19.99	
-		New or Additional Inward Data B Channel			UEPPP	PR7BD	0	29.11						19.99	19.99	19.99	
+		New or Additional Useage Sensitive Voice Data B Channel New or Additional Useage Sensitive Digital Data B Channel			UEPPP	PR7BS PR7BU	0	29.11 29.11						19.99 19.99	19.99 19.99	19.99	
+		New of Additional Oseage Sensitive Digital Data B Channel			UEFFF	FR/BU	U	29.11						19.99	19.99	19.98	9
CA	ALL TYPE	S															+
Ţ.,		Inward			UEPPP	PR7C1	0	0	0								
		Outward			UEPPP	PR7C0	0	0	0								
		Two-way			UEPPP	PR7CC	0	0	0								
Inte	eroffice C	hannel Mileage Fixed Each Including First Mile			UEPPP	1LN1A	95.7398	216.27	162.7	0				19.99	19.99	19.99	
		Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.7598	210.27	102.7	- 0				15.55	13.33	19.98	9
4-1	WIRE DS'	I DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT															
UN		pop Combination Rates 4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		1	UEPDC		187.21							19.99	19.99	19.99	0
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1		2	UEPDC		267.91							19.99	19.99	19.99	
		4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3		3	UEPDC		400.98							19.99	19.99	19.99	
UN	IE Loop F	Rates			LIEBBO												
-		4-Wire DS1 Digital Loop - UNE Zone 1 4-Wire DS1 Digital Loop - UNE Zone 2		2	UEPDC UEPDC	USLDC	113.59				23.46			19.99 19.99	19.99 19.99	19.99 19.99	1
+		4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPDC	USLDC	327.36				23.40			19.99	19.99	19.99	
		4 WIIC DOT DIGITAL COOP ONE ZOILE O		Ü	OLI DO	COLDO	027.00							10.00	10.00	10.00	
UN	E Port R																
		4-Wire DDITS Digital Trunk Port			UEPDC	UDD1T	73.62							19.99	19.99	19.99	9
	NIDEOLID	DING OUADOES, OUDDENTLY COMPINED															
NC	NKECUK	RING CHARGES - CURRENTLY COMBINED 4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Switch-as-is			UEPDC	USAC4		259.56	134.33					19.99	19.99	19.99	0
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with			OLF DC	03AC4		239.30	134.33					15.55	15.55	19.95	9
		DS1 Changes			UEPDC	USAWA		259.56	134.33					19.99	19.99	19.99	9
		4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with															
1		Change - Trunk		\perp	UEPDC	USAWB	1	259.56	134.33					19.99	19.99	19.99	9
A F	DITIONA	I NDCe	-	-		1	1		+		-		 			-	+-
AL		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent Channel		+					+ +				 			<u> </u>	+
		Activation/Chan - 1-Way Outward Trunk			UEPDC	UDTTB		29.01	29.01			1		19.99	19.99	19.99	9
T		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Channel Activation/Chan															
4		Inward Trunk w/out DID			UEPDC	UDTTC		29.01	29.01					19.99	19.99	19.99	9
		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation Per Chan -			LIEBBO	LIDTTO		00.04	20.01			1		40.00	40.00		
+		Inward Trunk with DID 4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsqnt Chan Activation / Chan - 2-	ļ	+	UEPDC	UDTTD	1	29.01	29.01			1		19.99	19.99	19.99	9
		Way DID w User Trans			UEPDC	UDTTE		29.01	29.01					19.99	19.99	19.99	۵
BIF		ZERO SUBSTITUTION		\vdash	OLFDO	ODITE		23.01	23.01					10.00	10.00	19.98	9
-"		B8ZS -Superframe Format			UEPDC	CCOSF		0	605					19.99	19.99	19.99	1
		B8ZS - Extended Superframe Format			UEPDC	CCOEF		0	605					19.99	19.99	19.99	1
				Ш						_							
		ark Inversion															

															Incremental	Т
															Charge -	
											Svc Order	Svc Order	Incremental	Incremental	Manual Svc	
											Submitted Flec	Submitted Manually per	Charge - Manua Svc Order vs.	Charge - Manua Svc Order vs.	Order vs.	.
											per LSR	LSR	Electronic-1st		I 1st	٦
							No	onrecurring	None	ecurring						
										connect						
	AMI - Extended SuperFrame Format			UEPDC	MCOPO	Rec	First ()	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	+
	AMI - Extended SuperFrame Format		+	UEPDC	MCOPO		U	0								+
			+													+
Telenhone	Number/Trunk Group Establisment Charges															+
	Telephone Number for 2-Way Trunk Group			UEPDC	UDTGX	0										+
	Telephone Number for 1-Way Outward Trunk Group			UEPDC	UDTGY	0						19.99				+
			-						_	-				1	-	+
	Telephone Number for 1-Way Inward Trunk Group Without DID			UEPDC	UDTGZ	0						19.99				-
	DID Numbers, Establish Trunk Group and Provide First Group of 20 DID Numbers			UEPDC	NDZ	0	0	0				19.99				\perp
	DID Numbers for each Group of 20 DID Numbers			UEPDC	ND4	0						19.99				
	DID Numbers, Non- consecutive DID Numbers , Per Number			UEPDC	ND5	0						19.99				
	Reserve Non-Consecutive DID Nos.			UEPDC	ND6	0	0	0				19.99				
	Reserve DID Numbers			UEPDC	NDV	0	0	0				19.99				T
	DS1 (Interoffice Channel Mileage) - FX/FCO for 4-Wire DS1 Digital Loop with 4-V	Vire DDIT	TS Tru													
	Interoffice Channel Mileage - Fixed rate 0-8 miles (Facilities Termination)			UEPDC	1LNO1	94.98	216.27	162.7	0	0			19.99	19.99	19.99	┸
	Interoffice Channel Mileage - Additional rate per mile - 0-8 miles	l	1 -	UEPDC	1LNOA	0.7598	0	0								1
	Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination)	1		UEPDC	1LNO2	0	0	0		1	1	1	1	1	1	Т
	Interoffice Channel Mileage - Additional rate per mile - 9-25 miles			UEPDC	1LNOB	0.7598	0	0								1
	Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)		1	UEPDC	1LNO3	0.7000	0	0	0							\top
1	Interoffice Channel Mileage - Additional rate per mile - 25+ miles	 	1	UEPDC	1LNOC	0.7598	0	0		1	1	1	1	1	1	+
1	Local Number Portability, per DC0 Activated	-	+				0	0	0	1	1	1	 	1	1	+
 	Local Number Portability, per DS0 Activated	-	+	UEPDC	LNPCP	3.15	U	U	U	+	1	+	+	+	+	+
	Central Office Termininating Point		1	UEPDC	CTG	0	-			1			1	1	1	+
			+			1	+			1	1	1		1	1	+
4-WIRF DS1	1 LOOP WITH CHANNELIZATION WITH PORT		+			1	-					 		1		+
	1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations															+
	em can have up to 24 combinations of rates depending on type and number of	norts us	ed													+
			Ī													T
UNE DS1 Lo	OOD															T
	4-Wire DS1 Loop - UNE Zone 1		1	UEPMG	USLDC	113.59	0	0								1
	4-Wire DS1 Loop - UNE Zone 2		2	UEPMG	USLDC	194.29	0	0								T
	4-Wire DS1 Loop - UNE Zone 3		3	UEPMG	USLDC	327.36	0	0								1
			Ť	-				_								T
UNE DSO C	Channelization Capacities (D4 Channel Bank Configurations)		1													Т
	24 DSO Channel Capacity - 1 per DS1			UEPMG	VUM24	103.47	0	0								
	48 DSO Channel Capacity - 1 per 2 DS1s		1	UEPMG	VUM48	206.94	0	0		1				1	1	\top
		 	1	UEPMG	VUM96	413.88	lo lo	0		1	1	1	1	1	1	+
	96 DSO Channel Capacity -1per 4 DS1s	-	-				U			1	1	1	1	1	1	4
	144 DS0 Channel Capacity - 1 per 6 DS1s		1	UEPMG	VUM14	620.82	0	0		1	1	1	1		1	
	192 DS0 Channel Capacity -1 per 8 DS1s	l	1 -	UEPMG	VUM19	827.76	0	0								1
	240 DS0 Channel Capacity - 1 per 10 DS1s			UEPMG	VUM20	1034.7	0	0								T
		 	1	UEPMG	VUM28	1241.64	lo lo			1	1	1	1	1	1	+
+	288 DS0 Channel Capacity - 1 per 12 DS1s		-				U	0	_	1	1	1		1	1	+
	384 DS0 Channel Capacity - 1 per 16 DS1s		1	UEPMG	VUM38	1655.52	0	0			1	1				4
	480 DS0 Channel Capacity - 1 per 20 DS1s			UEPMG	VUM40	2069.4	0	0								
	576 DS0 Channel Capacity -1 per 24 DS1s			UEPMG	VUM57	2483.28	0	0	1	1		1	1		1	1
		-	+-	UEPMG	VUM67	2897.16	lo lo	0		1	1	1		1	1	+
	672 DS0 Channel Capacity - 1 per 28 DS1s	-	+-	UEFING	VUIVIO/	2097.10	U	U	_	1	-	-	-	1	1	+
Non-Poorer	Iring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with P	ort - Com	Nore:	on Charge Based	on a System	1	+			+				1	+	+
	ring Charges (NRC) Associated with 4-Wire DST Loop with Channeliztion with P n System configuration is One (1) DS1, One (1) D4 Channel Bank, and Up To 24						-					 		1		+
	f this configuration functioning as one are considered Add'I after the minimum															+
	NRC - Conversion (Currently Combined) with or without BellSouth Allowed Changes			UEPMG	USAC4	0	301.62	16.76			†		19.99	19.99	19.99	19
	ditions at End User Locations Where 4-Wire DS1 Loop with Channelization with															T
	currently Combined) In Georgia & Tennessee Only															T
	NRC - 1 DS1/D4 Channel Bank - Add NRC for each Port and Assoc Feature															
	Activation - New GA & TN Only	1		UEPMG	VUMD4	0	717.71	425.81	149.08	17.69	1		19.99	19.99	19.99	19
	ero Substitution															T
	Clear Channel Capability Format, superframe - Subsequent Activity Only			UEPMG	CCOSF	0	0	605					19.99	19.99	19.99	19
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only		†	UEPMG	CCOEF	o .	n n	605	_	+	1	+	19.99	19.99	19.99	19
			1	OLI IVIG	COUEF	U	U	000		-	 	 	13.33	13.33	13.33	19
	Mark Inversion (AMI)		-		L	1				1	1	1	-	1	1	+
	Superframe Format			UEPMG	MCOSF	0	0	0								1
	Extended Superframe Format			UEPMG	MCOPO	0	0	0								1
		l	1								1	1				_
	Ports Associated with 4-Wire DS1 Loop with Channelization with Port		1				-							1		_
Exchange F			-													4
	Line Side Combination Channelized PBX Trunk Port - Business		-	UEPPX	UEPCX	1.65	0	0	0	0		19.99				4
	Line Side Outward Channelized PBX Trunk Port - Business		1	UEPPX	UEPOX	1.65	0	0	0	0		19.99		1	1	1
	Line Side Inward Only Channelized PBX Trunk Port without DID		1	UEPPX	UEP1X	1.65	0	0	0	0		19.99				1
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	8.86	0	0	0	0		19.99		1	1	
Feature Act	tivations - Unbundled Loop Concentration		L													╚
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank		L	UEPPX	1PQWM	0.7	25.45	13.44	4.2	4.17			19.99	19.99	19.99	19.
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.7	78.31	18.46	59.37	11.6			19.99	19.99	19.99	19.

RY NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					U33 K	ATES (\$)		
										ecurring	Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Increme Charg Manual Order Electroni Add
							No	nrecurring		onnect						
						Rec	First	Add'I	First	Add'I	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOM
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0	riist	Addi	THOC	Addi	JOHILC	SOMAN	JOHNA	SOMA	SOMA	SOM
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0	0	0				19.99				
	DID Numbers - groups of 20 - Valid all States			UEPPX	ND4	0	0	0				19.99				
	Non-Consecutive DID Numbers - per number			UEPPX	ND5	0	0	0				19.99				
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0	0	0				10.00				
	Reserve DID Numbers			UEPPX	NDV	0	0	0								
Local Nun	mber Portability			OLI I X	1.01		ľ									
Localitali	Local Number Portability - 1 per port		1	UEPPX	LNPCP	3.15	n	0								
FEATURE	S - Vertical and Optional			OLI I X	2.4. 0.	0.10	ľ									
Local Swit	itching Features Offered with Line Side Ports Only		1													
Local Own	All Features Available		1	UEPPX	UEPVF	6.29	0	0				19.99				
	All Fediules Available			UEFFA	UEFVF	0.29	U	U				19.99				
			-													
DI ED DODE I	COR COMPINATIONS MARKET DATES		-													
JEED PORT LO	OOP COMBINATIONS - MARKET RATES		-			1	-			-	-	 			-	├
							-				-					
Martin	des abell and unberg DellOs ab is not as a fee for the control of the fee for the control of the fee for the control of the fee for the control of the fee for the	77.1		- FOC 1/ O/	Commission	la a	-			-	-	 			-	-
	ates shall apply where BellSouth is not required to provide unbundled local switchin	g or switch po	nts pe	er FCC and/or State	Commission ru	ies.	1			-			1		-	₩
	enarios include:		L	tadfar Canair	d Tanana :	1	1						1			₩-
1. Unbund	dled port/loop combinations that are Not Currently Combined in all of the BellSouth dled port/loop combinations that are Currently Combined or Not Currently Combine	din Zooo 1 -	the T	on a Meas in Ballo	outh's region fo	r and uppro ···	th 4 or mar-	DS0 oguirologt II-	100	-			 		-	!
	oled port/loop combinations that are Currently Combined or Not Currently Combine MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale, Miami); GA (Atlanta)												1			├
PollCouth	currently is developing the billing capability to mechanically bill the recurring and no	n requiring A	Aorkot	Potos in this section	n In the interior	Policouth oh	Ill bill the ret	na in the Cost Boo	ad acation	propoding in I	iou of the M	orket Beter	and recented	the right to t	ruo un the hill	lina d
	et Rate for unbundled ports includes all available features in all states.	on-recurring iv	iaikei	Rates III triis sectio	in the interim	Delisoutii Siid	dii bili trie rate	es ili ille Cost-Das	eu section	preceding in i	led of the ivi	arker Kales	and reserves	the right to th	ue-up trie bili	ling u
End Office	e and Tandem Switching Usage and Common Transport Usage rates in the Port se	ction of this r	ato ovi	hihit chall apply to a	Il combinations	of loop/port pe	twork olomo	nte except for LIN	E Coin Por	/Loop Combi	nations which	h have a fla	at rate usage	sharge (LISO)	2: LIDECLI)	
Ear Not Cu	urrently Combined scenarios where Market Rates apply, the Nonrecurring charges	ore listed in	the Eir	rot and Additional MI	C columns for	anah Bart I ISC	C For Cur	rantly Combined a	control th	o Nopropurri	na oborgoo	oro listed in	the NPC Cui	ropth.	J. UKLCOJ.	-
	section. Additional NRCs may apply also and are categorized accordingly.	are listed in	ine Fil	St and Additional IN	C COIUITIIS IOI	each Full 030	JC. FUI CUI	rentily Combined S	Jenanos, u	e Nonecum	ig criarges a	are iisteu iii	trie NRC - Cui	renuy		
Combined	section. Additional NRCs may apply also and are categorized accordingly.					1							1			
o MUDE :	OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)		-													
	OICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
2-WIRE V			-													_
	/Loop Combination Rates															
	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		1			31.02										
	//Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2		2			39.66										
	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1															
UNE Port/	// /Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3		2			39.66										
	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3 5 Rates 5 Rat		3			39.66 47.99										
UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 5 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		3	UEPRX	UEPLX	39.66 47.99										
UNE Port/	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX	UEPLX	39.66 47.99 17.02 25.66										
UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 5 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1		3	UEPRX UEPRX		39.66 47.99										
UNE Port/	Cloop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3		3 1 2	UEPRX UEPRX	UEPLX	39.66 47.99 17.02 25.66										
UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (S		3 1 2	UEPRX UEPRX UEPRX	UEPLX UEPLX	39.66 47.99 17.02 25.66 33.99										
UNE Port/	// Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3- Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3- Sone Substitution		3 1 2	UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL	39.66 47.99 17.02 25.66 33.99	90	90					43.19	9.91		
UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Unbundled port - residence 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	39.66 47.99 17.02 25.66 33.99	90	90					43.19	9.91		
UNE Port/	// // Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	39.66 47.99 17.02 25.66 33.99	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Unbundled port - residence 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC	39.66 47.99 17.02 25.66 33.99	90	90					43.19	9.91		
UNE Loop 2-Wire Vo	// // Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	39.66 47.99 17.02 25.66 33.99	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	// // Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	// // Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO	39.66 47.99 17.02 25.66 33.99	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled series, low usage line port with Caller ID (LUM) UMBER PORTABILITY Local Number Portability (1 per port)		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14	90 90	90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled series, low usage line port with Caller ID (LUM) UMBER PORTABILITY Local Number Portability (1 per port)		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI	Loop Combination Rates		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NI	Loop Combination Rates		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NI FEATURE	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE	// // // // // // // // // // // // //		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates		3 1 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates		1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 10 0.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3- Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3- Rates 2-Wire Voice Grade Loop (SL1) - Zone 3 3- Wire Voice Grade Loop (SL1) - Zone 3 3- Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled sers, low usage line port with Caller ID (LUM) 3- Wire Voice Unbundles 3- Wire Voice Wir		1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 0.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates		1 1 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 10.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3- Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3- Rates 2-Wire Voice Grade Loop (SL1) - Zone 3 3- Wire Voice Grade Loop (SL1) - Zone 3 3- Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled port outgoing only - res 2-Wire voice unbundled sers, low usage line port with Caller ID (LUM) 3- Wire Voice Unbundles 3- Wire Voice Wir		1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 0.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NU FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates		1 1 2 2	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 10.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID (LUM) 2-WIRE PORTABILITY Local Number Portability (1 per port) 2-Wire Voice Grade Loop/Line Port Combination - Subsequent 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-		1 1 2 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 0.35	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates		1 1 2 3 3 1 1 1 1 2 3 3 1 1 1 1 1 1 1 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF USAS2	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 10 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NI FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 2-Wire VG Loop/Port Combo - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire Voice Grade Loop (SL1) - Zone 3 2-Wire voice unbundled port - residence 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled port vith Caller ID - res 2-Wire voice unbundled sers, low usage line port with Caller ID (LUM) UMBER PORTABILITY Local Number Portability (1 per port) 2-Wire Voice Grade Loop/Line Port Combination - Subsequent UNITABLE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS) 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3-Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire VG Loop Grade Loop (SL1)		1 1 2 3 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE LOOP 2-Wire Vo LOCAL NU FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates		1 1 2 3 3 1 1 1 1 2 3 3 1 1 1 1 1 1 1 1	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF USAS2	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 10 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NO FEATURE ADDITION UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3-Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3-Rates 3-Wire Voice Unbundled port - residence 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire Voice Unbundled 3-Wire VG Loop/Port Combo - Zone 1 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone		1 1 2 3 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NO FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3-Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3-Rates 3-Wire Voice Unbundled port - residence 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire Voice Unbundled 3-Wire VG Loop/Port Combo - Zone 1 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone		1 1 2 3 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NO FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates		1 1 2 3 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPRL UEPRC UEPRO UEPAP LNPCX UEPVF UEPVF	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 0.35 0 31.02 39.66 47.99	90 90 90	90 90 90 90					43.19 43.19	9.91 9.91		
UNE Loop 2-Wire Vo LOCAL NO FEATURE ADDITION 2-WIRE VO UNE Port/	Loop Combination Rates 2-Wire VG Loop/Port Combo - Zone 1 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 2 2-Wire VG Loop/Port Combo - Zone 3 3-Rates 2-Wire Voice Grade Loop (SL1) - Zone 1 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 2 2-Wire Voice Grade Loop (SL1) - Zone 3 3-Rates 3-Wire Voice Unbundled port - residence 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire voice unbundled port vith Caller ID - res 3-Wire Voice Unbundled 3-Wire VG Loop/Port Combo - Zone 1 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop/Port Combo - Zone 3 3-Rates 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 2 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone 3 3-Wire VG Loop Carde Loop (SL1) - Zone		1 1 2 3 3	UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX UEPRX	UEPLX UEPLX UEPLX UEPRC UEPRC UEPRO UEPAP LNPCX UEPVF USAS2	39.66 47.99 17.02 25.66 33.99 14 14 14 14 14 10 0.35 0 31.02 39.66 47.99 47.99	90 90 90 90 90 90 90 90 90 90 90 90 90 9	90 90 90 0					43.19 43.19 43.19	9.91 9.91 9.91		

Page 114 of 120

Y N	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC		I	RATES (\$)		Т		Т	OSS R	ATES (\$)	ı	
												Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incre Chi Mani Ord c Electro
								Nonre	curring	Non	recurring	per LSR	LSR	Electronic-1st	Electronic-Add'	I 1st	-
											connect						
		O Wise seize Conde seb andred Conde Constitute and and least distinct and side					Rec	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	S
		2-Wire voice Grade unbundled South Carolina extended local dialing parity port with Caller ID - bus			UEPBX	UEPAZ	14	90	90					43.19	9.91		
		2-Wire voice unbundled South Carolina Bus Area Calling Port with Caller ID (LMB)			UEPBX	UEPAB	14	90	90					43.19	9.91		+
		2 WHO VOICE UIDUIGUE COURT CATOIITA DAS ARCA CAIIITY TOTE WITH CAILCE TO (END)			OLI DX	OLI AD	1.7	30	30					40.10	3.31		+-
LO	CAL NUN	MBER PORTABILITY															
		Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
	ATURES			-													_
FE#	ATURES																+
NOI	NRECUR	RING CHARGES - CURRENTLY COMBINED															+-
ADI	DITIONAL																
		NRC - 2-Wire Voice Grade Loop/Line Port Combination - Subsequent			UEPBX	USAS2		0	0								_
2-10	VIDE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)															+
2-44	VIIXE VOI	CE GRADE EOOF WITH 2-WIRE LINE FORT (RES-FBA)															+
UNE		pop Combination Rates								1							+
		2-Wire VG Loop/Port Combo - Zone 1		1			31.02										
		2-Wire VG Loop/Port Combo - Zone 2		2		1	39.66										╨
-		2-Wire VG Loop/Port Combo - Zone 3	-	3		1	47.99			-							+
UNF	E Loop R	Rates		1		+	+						1				+
5141		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRG	UEPLX	17.02										+
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRG	UEPLX	25.66										
		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRG	UEPLX	33.99										4
2-10	Niro Voice	e Grade Line Port Rates (RES - PBX)															+
- "		2-Wire VG Unbundled Combination 2-Way PBX Trunk Port - Res			UEPRG	UEPRD	14	90	90					43.19	9.91		+-
															0.0		
LO		MBER PORTABILITY															
		Local Number Portability (1 per port)			UEPRG	LNPCP	3.15										+-
FF/	ATURES																+
	ATORLO																_
NOI	NRECUR	RING CHARGES - CURRENTLY COMBINED															
ADI	DITIONAL																+-
		Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity- Nonrecurring						0	0								
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group		1				14.64	14.64					19.99	19.99	19.9	9
2-W	VIRE VOI	CE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)															4
LINE	F Port/Lo	pop Combination Rates															+
0.11		2-Wire VG Loop/Port Combo - Zone 1		1			31.02										+
		2-Wire VG Loop/Port Combo - Zone 2		2			39.66										
		2-Wire VG Loop/Port Combo - Zone 3		3			47.99										4
UNF	E Loop R	Rates		+		+	+						1				+
5141		2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPPX	UEPLX	17.02										+
		2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPPX	UEPLX	25.66										
-		2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPPX	UEPLX	33.99		1	-							+
2-14	Vire Voice	e Grade Line Port Rates (BUS - PBX)		+		+	+						1				+
2-44		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus		1	UEPPX	UEPPC	14	90	90				1	43.19	9.91		+
		Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	14	90	90					43.19	9.91		
		Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	14	90	90	ļ				43.19	9.91		—
-		2-Wire Voice Unbundled PBX LD Terminal Ports 2-Wire Voice Unbundled 2-Way Combination PBX Usage Port	-	1	UEPPX UEPPX	UEPLD UEPXA	14 14	90 90	90 90	-				43.19 43.19	9.91 9.91		+-
-		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port 2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports		1	UEPPX	UEPXA	14	90	90				1	43.19	9.91		+
		2-Wire Voice Unbundled PBX LD DDD Terminals Port		L	UEPPX	UEPXC	14	90	90	<u></u>				43.19	9.91		1
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	14	90	90					43.19	9.91		
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port	l	-	UEPPX	UEPXE	14	90	90				1	43.19	9.91		+
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port			UEPPX	UEPXL	14	90	90				1	43.19	9.91		
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Room Calling Port		1	UEPPX	UEPXM	14	90	90	1	1		1	43.19	9.91		+
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital Discount Room															1
		Calling Port			UEPPX	UEPXO	14	90	90					43.19	9.91		4
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	ļ	-	UEPPX	UEPXS	14	90	90	1				43.19	9.91		+
100	CAL NUM	MBER PORTABILITY		1													+
			l	1	UEPPX	LNPCP	3.15	l		1	1	+	1			+	+
	J	Local Number Portability (1 per port)															

		UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC	ļ		RATES (\$)					U33 K	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manua Svc Order vs. Electronic-Add'	Electronic-Disc	Charge Manual S Order v
\rightarrow				-				Nonre	curring		ecurring						+
\rightarrow				-							onnect						+
-	FEATURES						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMA
_																	
	NONRECU	RRING CHARGES - CURRENTLY COMBINED															
	ADDITIONA	AL NRCs															1
-		2-Wire Voice Grade Loop/ Line Port Combination - Subsequent 2 Wire Loop/Line Side Port Combination - Non feature - Subsequent Activity-			UEPPX	USAS2		0	0								1
\rightarrow		Nonrecurring Nonrecurring						0	0					40.00	40.00		
\rightarrow		PBX Subsequent Activity - Change/Rearrange Multiline Hunt Group						14.64	14.64					19.99	19.99	19.9	19 1
- ;	2-WIRE VO	ICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN PORT															1
!		oop Combination Rates															
\rightarrow		2-Wire VG Coin Port/Loop Combo – Zone 1					31.02										-
\rightarrow		2-Wire VG Coin Port/Loop Combo – Zone 2 2-Wire VG Coin Port/Loop Combo – Zone 3					39.66 47.99										+-
-+		2-vvire vG Coin Port/Loop Combo – Zone 3					47.99										+
,	UNE Loop I	Rates															+
		2-Wire Voice Grade Loop (SL1) - Zone 1			UEPCO	UEPLX	17.02										1
		2-Wire Voice Grade Loop (SL1) - Zone 2			UEPCO	UEPLX	25.66										
		2-Wire Voice Grade Loop (SL1) - Zone 3			UEPCO	UEPLX	33.99										
-	2 Miro Voic	ce Grade Line Port Rates (Coin)															+
		2-Wire Coin 2-Way without Operator Screening and without Blocking (SC)			UEPCO	UEPSD	14	90	90					43.19	9.91		+
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (AL, KY, LA, MS, SC)			UEPCO	UEPRA	14	90	90					43.19	9.91		
		2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD			UEPCO	UEPSA	14	90	90					43.19	9.91		
-+		(SC) 2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSH	14	90	90					43.19	9.91		+
-+		2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SC)			OLFCO	OLF SIT	14	30	30					45.15	3.31		+
		(SC)			UEPCO	UEPSC	14	90	90					43.19	9.91		
		2-Wire Coin 2-Way with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	14	90	90					43.19	9.91		
		2-Wire Coin 2-W Oper Screen & Blocking: 900/976, 1+DDD, 011+ & Local;			LIEBOO	LIEBOE		00	00					40.40	0.04		
		Enhanced Calling OPT 3YV (SC) 2-Wire Coin 2-W Oper Screen & Block: 900/976, 1+DDD, 011+, & Local;			UEPCO	UEPCE	14	90	90			1		43.19	9.91		+
		Enhanced Calling OPT AP7 (SC)			UEPCO	UEPCF	14	90	90			1		43.19	9.91		
\dashv		2-Wire Coin Outward without Blocking and without Operator Screening (SC)		1	UEPCO UEPCO	UEPSG UEPSF	14 14	90 90	90 90	-		+	-	43.19 43.19	9.91 9.91		+-
		2-Wire Coin Outward with Operator Screening and 011 Blocking (SC) 2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD			UEPCO	UEPSJ	14	90	90					43.19	9.91		+
		(SC) 2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD,															+
-		011+, and Local (SC) 2-Wire Coin Out Oper Screen & Block: 900/976, 1+DDD, 011+, & Local : w/		1	UEPCO	UEPCM	14	90	90					43.19	9.91	-	+-
-		Enhanced Call OPT 3YW (SC)			UEPCO	UEPCP	14	90	90					43.19	9.91		-
		MDED BODTADILITY															
!	LOCAL NUI	MBER PORTABILITY Local Number Portability (1 per port)			UEPCO	LNPCX	0.35								-	1	+
\dashv		Local Number Fortability (1 per port)			UEFCU	LINFOX	0.33						 				+
		RRING CHARGES - CURRENTLY COMBINED															+
_	ADDITIONA	AL NRCs															1

ATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
											Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-1st	Incremental Charge - Manual Svc Order vs. Electronic-Add'l	Incremental Charge - Manual Svc Order vs. Electronic-Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic-D Add'I
							Nonre	curring	Nonre	curring						
										onnect						
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	LOCALINIT	ERCONNECTION														
		TION (CALL TRANSPORT AND TERMINATION)														
CAL INTE	ENCOMMEC	TION (CALL TRANSPORT AND TERMINATION)														
		Per MOU Rate for Local and ISP-bound Traffic (1/1/01-12/31/01)				\$0.0017500										
		Per MOU Rate for Local and ISP-bound Traffic (1/1/02-12/31/01)				\$0.0017300										+
		r et MOO (tate foi Local and foi -bound frame (1/1/02-12/31/02)				\$0.0013000										+
	TANDEM S	WITCHING														
		Tandem Switching Function Per MOU		OHD		\$0.0014911										
		Multiple Tandem Switching, per MOU (applies to intial tandem only)		OHD		\$0.0014911										
	TRUNK CH	ARGE														
		Installation Trunk Side Service - per DS0		OHD	TPP++		335.14bk	57.16bk								
		Dedicated End Office Trunk Port Service-per DS0**		OHD	TDE0P	\$0.00										
		Dedicated End Office Trunk Port Service-per DS1**		OH1 OH1MS	TDE1P	\$0.00										
		Dedicated Tandem Trunk Port Service-per DS0**		OHD	TDW0P	\$0.00										
		Dedicated Tandem Trunk Port Service-per DS1**		DH1 OH1MS	TDW1P	\$0.00										
		element is recovered on a per MOU basis and is included in the End Office Switching	and Tandem Switc	ning, per MOU r	ate elements			ļ						1		1
CAL INTE	ERCONNEC	TION (TRANSPORT)														
	COMMON 1	RANSPORT (Shared)		0110												
		Common Transport - Per Mile, Per MOU		OHD		\$0.0000121										
		Common Transport - Facilities Termination Per MOU		UHD		\$0.0004672										
	INTERNEEL	CE CHANNEL - DEDICATED TRANSPORT - VOICE GRADE														+
	INTEROTT	Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade - Per Mile per		OHL. OHM	1L5NF	0.0167bk										+
		Interoffice Channel - Dedicated Transport 2- Wire Voice Grade - Facility		OTIL, OTIV	TEST	0.0107 BK										
		Termination per month		OHL, OHM	1L5NF	24.3bk	81.25bk	54.94bk	33.54bk	13.82bk						
	INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT - 56/64 KBPS														
		Interoffice Channel - Dedicated Transport - 56 kbps - per mile per month		OHL, OHM	1L5NK	0.0167bk										
		Interoffice Channel - Dedicated Transport - 56 kbps - Facility Termination per month		OHL, OHM	1L5NK	16.76bk	81.26bk	54.94bk	33.54bk	13.82bk						
		Interoffice Channel - Dedicated Transport - 64 kbps - per mile per month		OHL, OHM	1L5NK	0.0167bk	04.0051	E4.04bl.	22 5 4 5 1 5	40.0061						
		Interoffice Channel - Dedicated Transport - 64 kbps - Facility Termination per month		OHL, OHM	1L5NK	16.76bk	81.26bk	54.94bk	33.54bk	13.82bk						-
	INTERNEEL	CE CHANNEL - DEDICATED TRANSPORT - DS1														
	INTEROTT	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per month		DH1 OH1MS	1L5NL	0.3415bk										+
		Interoffice Channel - Dedicated Tranport - DS1 - Facility Termination per month		OH1 OH1MS	1L5NL	77.14bk	178.93bk	163.98bk	32.77bk	28.95bk						
		The office of the first state of		5111 G111111G	120112	77111DK	110.000.	100.0001	OZ.II I DIK	20.0001						
	INTEROFFI	CE CHANNEL - DEDICATED TRANSPORT- DS3														
		Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month	(DH3 OH3MS	1L5NM	8.02bk										
		Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month	(OH3 OH3MS	1L5NM	880.65bk	558.74bk	326.23bk	120.66bk	117.17bk						
	LOCAL CH	ANNEL - DEDICATED TRANSPORT														
		Local Channel - Dedicated - 2-Wire Voice Grade per month		OHL OHM	TEFV2	15.33bk	387.05bk	66.48bk	73.44bk	6.41bk						
		Local Channel - Dedicated - 4-Wire Voice Grade per month		OHL OHM	TEFV4	16.54bk	387.93bk	67.35bk	74.38bk	7.35bk						
		Local Channel - Dedicated - DS1 per month Local Channel - Dedicated - DS3 Facility Termination per month		OH1 OH3	TEFHG TEFHJ	42.62bk 446bk	355.73bk 905.04bk	308.11bk 529.05bk	44.48bk 239.5bk	30.59bk 167.53bk						-
		Local Charlier - Dedicated - DSS Facility Termination per month		UHS	IEFFIU	440DK	903.04bk	329.03DK	239.30K	107.33DK						
	LOCAL INT	ERCONNECTION MID-SPAN MEET														
		coess service ride Mid-Span Meet, one-half the tariffed service Local Channel r	ate is applicable			1										t
	A	Local Channel - Dedicated - DS1 per month		OH1MS	TEFHG	\$0.00	\$0.00									1
		Local Channel - Dedicated - DS3 per month		OH3MS	TEFHJ	\$0.00	\$0.00									<u> </u>
						*****	*****									
	MULTIPLE)	KERS														
		Channelization - DS1 to DS0 Channel System		OH1 OH1MS	SATN1	134.46bk	182.48bk	125.42bk	21.12bk	19.62bk						
		DS3 to DS1 Channel System per month		DH3 OH3MS	SATNS	180.03bk	357.07bk	188.36bk	66.66bk	63.79bk						
		DS3 Interface Unit (DS1 COCI) per month		DH1 OH1MS	SATCO	10.8bk	13.18bk	9.45bk								-
						+								1		+
	Notes: 1)If	no rate is identified in the contract, the rates, terms, and conditions for the specific s									1					
		ariff; 2) "bk" beside a rate indicates that the parties have agreed to bill and keep for th														

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS P	ATES (\$)		
CALLGORT	NOTES	ONSOTOPING THE TYPINT EXAMINET		20.10	250	5555			(V)			Svc Order Submitted Elec	Svc Order Submitted Manually per	Incremental Charge - Manual Svc Order vs.	Incremental	Incremental Charge - Manual Svc Order vs. Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc
												per LSR	LSR	Electronic-1st	Electronic-Add'l		Add'l
								Nonre	curring		curring						
				1			Rec	First	Add'l	Disc	onnect Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							Rec	riist	Audi	riiot	Addi	JOINEC	JOHNA	SOMP	SOMA	JOHNA	JOHNA
	COLLOCAT	TION															
PHYSICAL COL	LLOCATION			<u> </u>		PF1BA											<u> </u>
		Physical Collocation - Application Fee - Initial Physical Collocation - Application Fee - Subsequent		_	CLO CLO	PE1BA PE1CA		3,768.00 3,141.00	3,768.00 3,141.00								
		Physical Collocation - Application Fee - Subsequent for Co-Carrier Cross Connect		1	CLO	PE1DT		585.42	585.42					<u> </u>			
		Physical Collocation - Space Preparation - Firm Order Processing	I		CLO	PE1SJ		1,204.00	1,204.00								
		Physical Collocation - Space Preparation - C.O. Modification per square ft.	1	-	CLO	PE1SK	2.75										
		Physical Collocation - Space Preparation - Common Systems Modification per square ft Cageless Physical Collocation - Space Preparation - Common Systems Modification per Cage			CLO CLO	PE1SL PE1SM	3.24 110.17										
		Physical Collocation - Cable Installation			CLO	PE1BD	110.17	1,621.00	1,621.00			<u> </u>					
		Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	3.95										
		Physical Collocation - Cable Support Structure	-	1	CLO	PE1PM PE1PL	21.33 9.19					ļ		ļ			ļ
		Physical Collocation - Power per Fused Amp Physical Collocation - 120V, Single Phase Standby Power Rate	1	+	CLO CLO	PE1PL PE1FB	9.19 5.67										-
		Physical Collocation - 240V, Single Phase Standby Power Rate	i	1	CLO	PE1FD	11.36										
		Physical Collocation - 120V, Three Phase Standby Power Rate	I		CLO	PE1FE	17.03										
		Physical Collocation - 277V, Three Phase Standby Power Rate	1	1	CLO	PE1FG	39.33					1	<u></u>	<u> </u>			ļ
		Physical Collocation - 2-Wire Cross-Connects			UEANL, UEA, UDN, UDC, UAL, UHL, UCL, UEQ	PE1P2	0.034	33.75	31.86								
		Physical Collocation - 4-Wire Cross-Connects			CLO	PE1P4	0.068	33.71	31.75								
Т		Physical Collection - DS1 Cross-Connects		1	CLO,UEANL,UEQ,WDS 1L,WDS1S	PE1P1	1.12	53.05	39.96		1						1
		Physical Collocation - DS1 Cross-Connects Physical Collocation - DS3 Cross-Connects		 	1L,WDS1S CLO	PE1P1 PE1P3	1.12	53.05 52.11	39.96			-		<u> </u>			
		Physical Collocation - 2-Fiber Cross-Connect		1	CLO	PE1F2	2.82	52.11	38.69								
		Physical Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	5.01	64.69	51.26								
		Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.		┞	CLO	PE1BW	219.19										
		Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.		+	CLO	PE1CW	21.50 74.12										-
		Physical Collocation - Security Access System - Security System per Central Office Physical Collocation - Security Access System - New Access Card Activation, per Card		 	CLO	PE1AX PE1A1	0.06	55.70	55.70								†
				1			0.00										
		Physical Collocation-Security Access System-Administrative Change, existing Access Card, per Card	- 1	-	CLO	PE1AA		15.62	15.62								
		Physical Collocation - Security Access System - Replace Lost or Stolen Card, per Card Physical Collocation - Security Access - Initial Key, per Key		+	CLO CLO	PE1AR PE1AK		45.66 26.25	45.66 26.25								
		Physical Collocation - Security Access - Key, Replace Lost or Stolen Key, per Key		t	CLO	PE1AL		26.25	26.25					·			
		Physical Collocation - Space Availability Report per premises	- 1		CLO	PE1SR		2,155.00	2,155.00								
					UEANL, UEA, UDN, UDC, UAL, UHL, UCL, UEQ, CL												
		POT Bay Arrangements prior to 6/1/99 - 2-Wire Cross-Connect, per cross-connect			0	PE1PE	0.1091										
					UEANL,UEA,UDN,UDC,												
		POT Bay Arrangements prior to 6/1/99 - 4-Wire Cross-Connect, per cross-connect			UAL,UHL,UCL,UEQ,CL	PE1PF	0.2181										
					UEANL,UEA,UDN,UDC,		0.2.0										
		POT Bay Arrangements prior to 6/1/99 - DS1 Cross-Connect, per cross-connect			UAL,UHL,UCL,UEQ,CL O,WDS1L,WDS1S,	PE1PG	0.9004										
		POT Bay Arrangements prior to 6 1799 - DST Cross-Connect, per cross-connect		1	UEANL,UEA,UDN,UDC,	FEIFG	0.9004										
					UAL,UHL,UCL,UEQ,CL												
		POT Bay Arrangements prior to 6/1/99 - DS3 Cross-Connect, per cross-connect		 	O UEANL,UEA,UDN,UDC,	PE1PH	5.64					ļ		 			ļ
					UAL,UHL,UCL,UEQ,CL												
		POT Bay Arrangements prior to 6/1/99 - 2-Fiber Cross-Connect, per cross-connect		-	O UEANL,UEA,UDN,UDC,	PE1B2	37.36										-
					UAL,UHL,UCL,UEQ,CL										ĺ		
		POT Bay Arrangements prior to 6/1/99 - 4-Fiber Cross-Connect, per cross-connect		1	0	PE1B4	50.38							1			<u> </u>
		Collocation Cable Records - per request		 	CLO	PE1CR PR1CD	-	1,712.00 925.57	1,168.00 925.57		 	ļ	1	<u> </u>			
		Collocation Cable Records - VG/DS0 Cable, per cable record Collocation Cable Records - VG/DS0 Cable, per each 100 pair		1	CLO	PR1CD PE1CO		925.57 18.06	925.57 18.06		-	+		1			
		Collocation Cable Records - Verbas Cable, per each 100 pair			CLO	PE1C1		8.45	8.45			1					
		Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		29.59	29.59								
		Collocation Cable Records - Fiber Cable, per 99 fiber records		ļ	CLO	PE1CB		279.57	279.57								
-		Physical Collocation - Security Escort - Basic, per Half Hour	-	-	CLO,CLORS	PE1BT PF1OT		33.92	21.50 27.77			1	-				
		Physical Collocation - Security Escort - Overtime, per Half Hour Physical Collocation - Security Escort - Premium, per Half Hour		1	CLO,CLORS CLO,CLORS	PE10T PE1PT		44.19 54.45	34.04		 	1	1	1			
		Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear ft.		<u> </u>	CLO,CLORS	PE1ES	0.0022	54.45	34.04								
		Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per lin. ft.			CLO	PE1DS	0.0033										
				1								1	<u></u>	<u> </u>			ļ
ADJACENT CO		Adjacent Collection - Space Charge per Sq. Et	-	1	CLO	PE1JA	0.094					1	1	1	-		
		Adjacent Collocation - Space Charge per Sq. Ft. Adjacent Collocation - Electrical Facility Charge per Linear Ft.		 	CLO	PE1JA PE1JC	6.40						†				
		Adjacent Collocation - 2-Wire Cross-Connects		L	CLO	PE1P2	0.034	33.75	31.86								
					UEA,UHL,UDL,UCL,CL	PF1P4	0.000	00.74	0.175								
		Adjacent Collocation - 4-Wire Cross-Connects Adjacent Collocation - DS1 Cross-Connects		1	O USL,CLO	PE1P4 PE1P1	0.068 1.12	33.71 53.05	31.75 39.96		-	+		1			
		Adjacent Collocation - DS3 Cross-Connects		1	CLO	PE1P3	14.21	52.11	38.68			1	 				
		Adjacent Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.82	52.11	38.69								
		Adjacent Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	5.01	64.69	51.26								
		Adjacent Collocation - Application Fee		<u> </u>	CLO	PE1JB		3,161.00									ļ
		Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp		1	CLO	PE1FB	5.67		l		l		1	<u> </u>	l		<u> </u>

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zone	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR			Electronic-Disc	Incremental Charge - Manual Svc Order vs. Electronic-Disc Add'I
								Nonre	curring		curring						
							_				onnect						
		Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FD	Rec 11.36	First	Add'I	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
			_		CLO	PE1FD PF1FF	17.03										
		Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp	_			PE1FG	39.33										
		Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FG	39.33										
PHYSICAL CO	LLOCATION IN	I THE REMOTE SITE															
	1	Physical Collocation in the Remote Site - Application Fee *			CLO	PE1RA		\$871.12	\$871.12								
		Cabinet Space in the Remote Site per Bay/ Rack *			CLO	PE1RB	\$246.44									1	
		Physical Collocation in the Remote Site - Security Access - Key *			CLO	PE1RD		\$26.25	\$26.25								
		Physical Collocation in the Remote Site - Space Availability Report per Premises Requested *			CLO	PE1SR		\$232.25	\$232.25								
		Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI Code Requested *			CLO	PE1RE		\$75.27	\$75.27								
		tes which are subject to true-up.															
	NOTE: If Se	ecurity Escort and/or Add'l Engineering Fees become necessary for remote site collo-	cation, the	Parti	ies will negotiate app	ropriate rates.											

CATEGORY	NOTES	UNBUNDLED NETWORK ELEMENT	Interim Zor	ne	BCS	USOC			RATES (\$)					OSS R	ATES (\$)		
												Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Svc Order vs.	Incremental Charge - Manua Svc Order vs. Electronic-Add'	Electronic-Disc	Charge - Manual Sv Order vs.
								Nonre	ecurring	Nonre	curring						
										Disc	onnect						
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	NUMBER P	PORTABILITY	_	+													+
INTERIM SE		OVIDER NUMBER PORTABILITY - RCF								1							_
		RCF, per number ported (Business Line)				TNPBL	\$2.17	\$0.7046									1
		RCF, per number ported (Residence Line)				TNPRL	\$2.17	\$0.7046									+
		RCF, add'l capacity for simultaneous call forwarding, per additional path					\$0.3854	Q 00.10									+
		RCF, per service order, per location (Business)				TNPBD	ψ0.0001	\$1.37	\$1.37			\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
		RCF, per service order, per location (Residence)				TNPRD		\$1.37	\$1.37			\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
INTERIM SE		OVIDER NUMBER PORTABILITY - DID	+	_		THIRDS		#0.05									
		DID per number ported (Residence)				TNPDR TNPDB		\$2.25									
		DID per number ported (Business)						\$2.25	64.07	04470	644.70	00.50		040.00	040.00	040.00	640.00
		DID per service order, per location (Residence)				TNPRD		\$1.37	\$1.37	\$44.70	\$44.70	\$3.50		\$19.99	\$19.99		\$19.99
		DID per service order, per location (Business)				TNPBD	040.40	\$1.37	\$1.37	\$44.70	\$44.70	\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
		DID, per trunk termination, Initial				TNPT2	\$13.16	\$218.03				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
		DID, per trunk termination, Subsequent				TNPT2	\$13.16	\$73.63				\$3.50		\$19.99	\$19.99	\$19.99	\$19.99
SERVICE P	ROVIDER N	UMBER PORTABILITY (RIPH)															1
	Note: If no	rate is identified in the contract, the rate for the specific service or function will be	as set forth in a	applicab	le BellSouth	ariff or as											
																	+
	USAGE RA	TES															
ODUF/EDO	UF/ADUF/CI	MDS															
	400E00 B	AILY USAGE FILE (ADUF)															-
						N1/A	00.004										
		ADUF: Message Processing, per message		_		N/A N/A	\$0.004										
		ADUF: Data Transmission (CONNECT:DIRECT), per message				N/A	\$0.001										+
	OPTIONAL	DAILY USAGE FILE (ODUF)							+	1		+					+
		ODUF: Recording, per message				N/A	\$0.0002862										1
		ODUF: Message Processing, per message				N/A	\$0.0032344			1					1	1	1
		ODUF: Message Processing, per Magnetic Tape provisioned				N/A	\$54.72					1				1	t
		ODUF: Data Transmission (CONNECT:DIRECT), per message				N/A	\$0.0000357										1
	CENTRA!	ZED MESSAGE DISTRIBUTION SERVICE (CMDS)					1										_
						NI/A	\$0.00¢		1	1		1			-	1	+
		CMDS: Message Processing, per message	+	_		N/A N/A	\$0.004		1	+		1	1		 	 	+
		CMDS: Data Transmission (CONNECT:DIRECT), per message				N/A	\$0.001										+
	Notes: If no	Lorate is identified in the contract, the rate for the specific service or function will be	as set forth in	applica	ble BellSouth	tariff or as	1			1		1					+

Amendment to the Interconnection Agreement By and Between BellSouth Telecommunications, Inc. And ITC^DeltaCom Communications, Inc. Dated January 31, 2002

This Amendment ("Amendment") is made and entered into by and between ITC^DeltaCom Communications, Inc. ("ITC^DeltaCom") and BellSouth Telecommunications, Inc. ("BellSouth") to amend the Interconnection Agreement ("the Agreement") entered into by ITC^DeltaCom and BellSouth on January 31, 2002 for the states of Kentucky, Louisiana, Mississippi, and South Carolina.

WHEREAS, the Parties desire to amend that certain Interconnection Agreement between BellSouth and ITC^DeltaCom dated January 31, 2002 to incorporate rates established by the South Carolina Public Service Commission in Docket Number 2001-65-C-Order Number 2001-1089, November 30, 2001;

NOW THEREFORE, in consideration of the provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, ITC^DeltaCom and BellSouth hereby covenant and agree as follows:

- 1. The rate sheets for South Carolina in Attachment 11, Table 1 of the Agreement are hereby deleted and replaced with the Attachment 11, Table 1 rate sheets for South Carolina in Exhibit 1 attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Interconnection Agreement shall remain unchanged and in full force and effect.
- 3. Either or both of the Parties are authorized to submit this Amendment to the South Carolina Public Service Commission or other Regulatory Agency for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.
- 4. This Amendment is made effective thirty (30) calendar days following the last signature of both Parties.
- 5. On certain elements, BellSouth has required a Non Recurring Charge (NRC) Additional Fee that is not specified in the South Carolina Order Price List. ITC^DeltaCom reserves its right to seek review from the SC Commission as to whether it is appropriate to have a NRC Additional Fee for those certain elements. BellSouth agrees that ITC^DeltaCom's acquiescence to the inclusion of these NRC Additional Fees pending such review is not to be considered an admission or agreement of ITC^DeltaCom that it is appropriate to assess a NRC Additional Fee for these elements.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

ITC^DeltaCom Communications, Inc.	BellSouth Telecommunications Inc.
Signature	Signature
Name	Name
Title	Title
Date	Date

UNBU	INDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEG		RATE ELEMENTS	Interi m	Zone	BCS	usoc			TES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
							Rec		curring	Nonrecurring		001150	001111		Rates(\$)	001441	0011411
<u> </u>								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		The "Zone" shown in the sections for stand-alone loops or loops as part of a combination refers to Geographically Deaveraged UNE Zones. To view Geographically Deaveraged UNE Zone Designations by Central Office, refer to Internet Website: http://www.interconnection.bellsouth.com/become_a_clec/html/interconnection.htm as set forth by the Commission.															
OPERA	TIONAL	SUPPORT SYSTEMS															
		(1) Electronic Service Order: CLEC should contact its contract															is rate
		is the BellSouth regional electronic service ordering charge.															
		(2) Any element that can be ordered electronically will be bill															
		elements that cannot be ordered electronically at present per t ng charge, SOMAN, will be applied to a CLECs bill when it sub				e in this cated	gory reflects the	e cnarge that	would be billed	a to a CLEC on	ce electronic (ordering cap	adilities co	me on-line for	tnat element	. Otnerwise,	tne manual
	orderir	Manual Service Order Charge, per LSR, Disconnect Only (SC)	mus ar	LORT	o benoouth.	SOMAN				1.97							T
	1	Electronic OSS Charge, per LSR, submitted via BST's OSS								1.51							
		interactive interfaces (Regional)				SOMEC		3.50									
UNBUN		EXCHANGE ACCESS LOOP															
	2-WIRE	ANALOG VOICE GRADE LOOP		1	LIFANII	LIEALO	44.04	07.00	47.00	00.50	5.00		45.00				ļ
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 1 2-Wire Analog Voice Grade Loop - Service Level 1- Zone 2		2	UEANL UEANL	UEAL2 UEAL2	14.94 21.39	37.92 37.92	17.62 17.62	23.56 23.56	5.32 5.32		15.69 15.69				1
		2-Wire Analog Voice Grade Loop - Service Level 1- Zone 3		3	UEANL	UEAL2	26.72	37.92	17.62	23.56	5.32		15.69				
		Loop Testing - Basic 1st Half Hour			UEANL	URET1		34.23	34.23				15.69				1
		Loop Testing - Basic Additional Half Hour			UEANL	URETA		19.90	19.90				15.69				
		Engineering Information Document (EI)			UEANL	LIEANAO		13.47	13.47								ļ
		Manual Order Coordination for UVL-SL1s (per loop) Order Coordination for Specified Conversion Time for UVL-SL1			UEANL	UEAMC		8.17	8.17								1
		(per LSR)			UEANL	OCOSL		18.13	18.13								
	2-WIRE	Unbundled COPPER LOOP			-				-								
		2-Wire Unbundled Copper Loop - Non-Designed Zone 1		1	UEQ	UEQ2X	12.94	36.40	16.10	22.66	4.42		15.69				
	ļ	2 Wire Unbundled Copper Loop - Non-Designed - Zone 2			UEQ	UEQ2X	14.51 15.02	36.40 36.40	16.10 16.10	22.66 22.66	4.42 4.42		15.69 15.69				4
		2 Wire Unbundled Copper Loop - Non-Designed - Zone 3 Order Coordination 2 Wire Unbundled Copper Loop - Non-	-	3	UEQ	UEQ2X	15.02	36.40	16.10	22.00	4.42		15.69				
		Designed (per loop)			UEQ	USBMC		8.17	8.17				15.69				
		Engineering Information Document			UEQ			13.47	13.47				15.69				
		Loop Testing - Basic 1st Half Hour			UEQ	URET1		34.23	34.23				15.69				
LIMPLE	IDI ED 1	Loop Testing - Basic Additional Half Hour			UEQ	URETA		19.90	19.90	-			15.69				
UNBUN		EXCHANGE ACCESS LOOP E ANALOG VOICE GRADE LOOP		-													
	- *****	2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-								<u> </u>							
		Zone 1		1	UEPSR UEPSB	UEALS	14.94	37.92	17.62	23.56	5.32		15.69				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting-		l .	LIEDOD LIEGOS	LIEARS			.=								
	 	Zone 1 2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting-		1	UEPSR UEPSB	UEABS	14.94	37.92	17.62	23.56	5.32	1	15.69				
		Zone 2		2	UEPSR UEPSB	UEALS	21.39	37.92	17.62	23.56	5.32		15.69				
		2 Wire Analog Voice Grade Loop- Service Level 1-Line Splitting- Zone 2		2	UEPSR UEPSB	UEABS	21.39	37.92	17.62	23.56	5.32		15.69				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEALS	26.72	37.92	17.62	23.56	5.32		15.69				
		2 Wire Analog Voice Grade Loop-Service Level 1-Line Splitting- Zone 3		3	UEPSR UEPSB	UEABS	26.72	37.92	17.62	23.56	5.32		15.69				
UNBUN		EXCHANGE ACCESS LOOP															
	2-WIRE	E ANALOG VOICE GRADE LOOP								-							
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 1		1	UEA	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 2		2	UEA	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
		2-Wire Analog Voice Grade Loop - Service Level 2 w/Loop or Ground Start Signaling - Zone 3		3	UEA	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				

04/12/02 Page 1 of 30

IDUNDLE	D NETWORK ELEMENTS - South Carolina			ı									Attachment:		Table 1	
TEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	TES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'
						Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	•	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.13									
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 1		1	UEA	UEAR2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse															
	Battery Signaling - Zone 2		2	UEA	UEAR2	23.13	105.98	68.43	53.05	10.61		15.69				
	2-Wire Analog Voice Grade Loop - Service Level 2 w/Reverse		_													
	Battery Signaling - Zone 3		3	UEA	UEAR2	28.46	105.98	68.43	53.05	10.61		15.69				
4 14/100	Order Coordination for Specified Conversion Time (per LSR)			UEA	OCOSL		18.13									
	ANALOG VOICE GRADE LOOP 4-Wire Analog Voice Grade Loop - Zone 1		1	UEA	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	4-Wire Analog Voice Grade Loop - Zone 1 4-Wire Analog Voice Grade Loop - Zone 2		2	UEA	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	4-Wire Analog Voice Grade Loop - Zone 2	-	3	UEA	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		-	UEA	OCOSL	45.56	18.13	34.03	55.55	14.01		10.09				
2-WIRE	ISDN DIGITAL GRADE LOOP			OL/(OOOOL		10.10									
	2-Wire ISDN Digital Grade Loop - Zone 1		1	UDN	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
	2-Wire ISDN Digital Grade Loop - Zone 2			UDN	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
	2-Wire ISDN Digital Grade Loop - Zone 3		3	UDN	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
	Order Coordination For Specified Conversion Time (per LSR)			UDN	OCOSL		18.13									
2-WIRE	Universal Digital Channel (UDC) COMPATIBLE LOOP															
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone				LIB COV							4= 00				
	1 Wire Universal Digital Channel (UDC) Competible Lean Zone		1	UDC	UDC2X	25.21	117.58	80.03	53.05	10.61		15.69				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone		2	UDC	UDC2X	32.76	117.58	80.03	53.05	10.61		15.69				
	2-Wire Universal Digital Channel (UDC) Compatible Loop - Zone			ODC	ODOZX	32.70	117.50	00.03	33.03	10.01		13.03				
	3		3	UDC	UDC2X	37.70	117.58	80.03	53.05	10.61		15.69				
2-WIRE	ASYMMETRICAL DIGITAL SUBSCRIBER LINE (ADSL) COMP	ATIBLE	LOOF		OB OZA	01.10	111.00	00.00	00.00	10.01		10.00				
	2 Wire Unbundled ADSL Loop including manual service inquiry	Ī														
	& facility reservation - Zone 1		1	UAL	UAL2X	12.19	120.84	70.56	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UAL	UAL2X	13.71	120.84	70.56	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop including manual service inquiry															
	& facility reservation - Zone 3		3	UAL	UAL2X	14.14	120.84	70.56	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UAL	OCOSL		18.13									
	2 Wire Unbundled ADSL Loop without manual service inquiry &															
	facility reservaton - Zone 1		1	UAL	UAL2W	12.19	95.81	57.82	50.37	7.93		15.69				
	2 Wire Unbundled ADSL Loop without manual service inquiry &		2	UAL	UAL2W	13.71	95.81	57.00	50.37	7.00		45.00				
	facility reservaton - Zone 2 2 Wire Unbundled ADSL Loop without manual service inquiry &			UAL	UALZVV	13.71	95.81	57.82	50.37	7.93		15.69				
	facility reservation - Zone 3		3	UAL	UAL2W	14.14	95.81	57.82	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)		3	UAL	OCOSL	14.14	18.13	37.02	30.37	7.33		15.05				
	HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIBLE	LOOP	UAL	OCCOL		10.13									
	2 Wire Unbundled HDSL Loop including manual service inquiry	1	<u> </u>													
	& facility reservation - Zone 1		1	UHL	UHL2X	9.58	129.52	79.24	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 2		2	UHL	UHL2X	10.92	129.52	79.24	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop including manual service inquiry															
	& facility reservation - Zone 3		3	UHL	UHL2X	11.40	129.52	79.24	50.37	7.93		15.69				
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	2 Wire Unbundled HDSL Loop without manual service inquiry															
	and facility reservation - Zone 1		1	UHL	UHL2W	9.58	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry	l	_	l			,					,=				1
	and facility reservation - Zone 2	ļ	2	UHL	UHL2W	10.92	104.49	66.50	50.37	7.93		15.69				
	2 Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3	l	3	UHL	UHL2W	11.40	404.40	66.50	50.37	7.93		15.69				1
	and facility reservation - Zone 3 Order Coordination for Specified Conversion Time (per LSR)	 	3	UHL	OCOSL	11.40	104.49 18.13	66.50	50.37	7.93		15.69		-	-	-
	I DIGGET COORDINATION FOR SPECIFIED CONVERSION TIME (PER LSK) HIGH BIT RATE DIGITAL SUBSCRIBER LINE (HDSL) COMPA	TIRLE	LOCE	UTL	UCUSL		18.13									
	. IIIOII DII NATE DIGITAL GUDGUNIDEN LINE (NDGL) CUNFA	HOLE	LUUP	1	_				-							
4-WIRE	4 Wire Unbundled HDSL Loop including manual service inquiry				l l		I		1							

UNBUNDL	ED NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	ES(\$)			Submitted Elec	Svc Order Submitted Manually per LSR	Incremental Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	urring	Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	4-Wire Unbundled HDSL Loop including manual service inquiry		2	UHL	LILII AV	14.33	150 10	107.90	EE 10	10.20		15.60				ł
	and facility reservation - Zone 2 4-Wire Unbundled HDSL Loop including manual service inquiry			UHL	UHL4X	14.33	158.18	107.89	55.12	10.38		15.69				
	and facility reservation - Zone 3		3	UHL	UHL4X	16.84	158.18	107.89	55.12	10.38		15.69				ł
	Order Coordination for Specified Conversion Time (per LSR)			UHL	OCOSL		18.13									
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 1		1	UHL	UHL4W	16.02	133.14	95.16	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop without manual service inquiry															1
-	and facility reservation - Zone 2		2	UHL	UHL4W	14.33	133.14	95.16	55.12	10.38		15.69				
	4-Wire Unbundled HDSL Loop without manual service inquiry and facility reservation - Zone 3		3	UHL	UHL4W	16.84	133.14	95.16	55.12	10.38		15.69				ł
	Order Coordination for Specified Conversion Time (per LSR)		3	UHL	OCOSL	10.04	18.13	33.10	33.12	10.30		10.00				
4-WIF	RE DS1 DIGITAL LOOP															i
	4-Wire DS1 Digital Loop - Zone 1			USL	USLXX	79.51	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop - Zone 2			USL	USLXX	136.00	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	USL USL	USLXX	229.15	253.03 18.13	157.89	44.80	11.73		15.69				
4-WIF	RE 19.2, 56 OR 64 KBPS DIGITAL GRADE LOOP			USL	UCUSL		10.13									
1	4 Wire Unbundled Digital 19.2 Kbps		1	UDL	UDL19	29.93	126.66	89.12	59.35	14.61		15.69				1
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	33.99	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital 19.2 Kbps			UDL	UDL19	34.74	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 1			UDL	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 56 Kbps - Zone 2			UDL	UDL56 UDL56	33.99 34.74	126.66	89.12	59.35	14.61		15.69 15.69				
-	4 Wire Unbundled Digital Loop 56 Kbps - Zone 3 Order Coordination for Specified Conversion Time (per LSR)		3	UDL UDL	OCOSL	34.74	126.66 18.13	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 1		1	UDL	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 2			UDL	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	4 Wire Unbundled Digital Loop 64 Kbps - Zone 3			UDL	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				1
	Order Coordination for Specified Conversion Time (per LSR)			UDL	OCOSL		18.13									
2-WIF	RE Unbundled COPPER LOOP															
	Wire Unbundled Copper Loop/Short including manual service inquiry & facility reservation - Zone 1		1	UCL	UCLPB	12.19	119.91	69.62	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Short including manual service		2	UCL	LICI DD	40.74	440.04	69.62	50.07	7.00		45.00				ł
-	inquiry & facility reservation - Zone 2 2 Wire Unbundled Copper Loop/Short including manual service			UCL	UCLPB	13.71	119.91	69.62	50.37	7.93		15.69				
	inquiry & facility reservation - Zone 3		3	UCL	UCLPB	14.14	119.91	69.62	50.37	7.93		15.69				i
	Order Coordination for Unbundled Copper Loops (per loop)		Ť	UCL	UCLMC		8.17	8.17	00.07	7.00		10.00				
	2-Wire Unbundled Copper Loop/Short without manual service															
	inquiry and facility reservation - Zone 1		1	UCL	UCLPW	12.19	94.87	56.89	50.37	7.93		15.69				.
	2-Wire Unbundled Copper Loop/Short without manual service inquiry and facility reservation - Zone 2		2	UCL	UCLPW	13.71	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Short without manual service		_					=		= 00						ł
-	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL	UCLPW	14.14	94.87 8.17	56.89 8.17	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - includes manual srvc.			UCL	UCLIVIC		0.17	0.17								
	inquiry and facility reservation - Zone 1		1	UCL	UCL2L	38.22	119.91	69.62	50.37	7.93		15.69				l
	2-Wire Unbundled Copper Loop/Long - includes manual svc.								22.37	30						
	inquiry and facility reservation - Zone 2		2	UCL	UCL2L	55.33	119.91	69.62	50.37	7.93		15.69				1
	2-Wire Unbundled Copper Loop/Long - includes manual svc.															
	inquiry and facility reservation - Zone 3 Order Coordination for Unbundled Copper Loops (per loop)		3	UCL UCL	UCL2L UCLMC	67.95	119.91 8.17	69.62 8.17	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - without manual service		<u> </u>	UUL	UCLIVIC		8.17	8.17			 					ſ
	inquiry and facility reservation - Zone 1		1	UCL	UCL2W	38.22	94.87	56.89	50.37	7.93		15.69				l
	2-Wire Unbundled Copper Loop/Long - without manual service															Ī
	inquiry and facility reservation - Zone 2		2	UCL	UCL2W	55.33	94.87	56.89	50.37	7.93		15.69				
	2-Wire Unbundled Copper Loop/Long - without manual service inquiry and facility reservation - Zone 3		3	UCL	UCL2W	67.95	94.87	56.89	50.37	7.93		15.69				
	Order Coordination for Unbundled Copper Loops (per loop)			UCL	UCLMC		8.17	8.17		•				-		
4-WIF	RE COPPER LOOP		<u> </u>													

Part Part	UNBUNDLE	NETWORK ELEMENTS - South Carolina												Attachment:		Table 1	
March Marc	CATEGORY	RATE ELEMENTS		Zone	BCS	USOC						Submitted Elec	Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'I	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
### After Copyon and Incomposition According Procession 2. Dec. 1 CCL CCL 1964 144.17 39.86 55.12 19.36 15.66							Rec										
Month Early Vision Control Tricked in grown a service Parity								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
A-Wine Cooper Long-Point - reducing manual services requiry 2 UCL UCLVS 20.95 144.17 93.86 55.12 10.38 15.66				4	LICI	1101.46	10.64	144 17	02.00	EE 10	10.20		15.60				ĺ
Second Conference on Confere				1	UCL	UCL4S	19.64	144.17	93.88	55.12	10.38		15.69				
After Cooper Loco/Sibrar including manual service inquiry and 1 UCL UCLAW 1804 11013 1155 1156				2	UCI	UCL4S	20.90	144 17	93.88	55 12	10.38		15 69				ĺ
and facility reservation - Zeros 3 3 UCL UCLMS 19.34 144.17 93.88 55.12 10.38 15.69																	
A-Wile Copper LoopShirt - willhord manual service inquiry and learning remoders. Zenia 1.0CL UCLAW 18.64 119.13 81.15 56.12 10.38 15.69		and facility reservation - Zone 3		3			19.34			55.12	10.38		15.69				l
Settley reservation - Zone 1 1 UCL					UCL	UCLMC		8.17	8.17								
4-Wire Copper Loop Short - without manual service incigary and facility reservation 2-2 or 2 or 2 or 2 or 2 or 2 or 2 or 2																	ĺ
Facility reconstants - Zone 2				1	UCL	UCL4W	19.64	119.13	81.15	55.12	10.38		15.69				
### 4-Wire Copper Loop Stop in willout manual service inquiry and fairly reservation 12 or 10 or				2	LICI	LICL AW	20.00	110 12	01 15	EE 10	10.20		15.60				ĺ
Subtley reservation - Zone 3					OOL	UCL44V	20.90	119.13	01.15	55.12	10.38	 	15.69				<u> </u>
Order Coordination for Unburidinal Copper Loops (per loop)				3	UCL	UCL4W	19.34	119.13	81.15	55.12	10.38		15.69				ĺ
Inquiry and facility reservation - Zone 1					UCL	UCLMC		8.17	8.17								
### Hebundled Copper Logo/Logo - includes manual size. Hard Published Co																	
Inciguly and facility reservation - Zone 2				1	UCL	UCL4L	77.29	144.17	93.88	55.12	10.38		15.69				
4-Wire Unbunded Copper Loop (Long - Includes manual sec. Includes and any sec. Includes and any sec. Includes and any sec. Includes any					LICI	1101.41	440.70	444.47	02.00	55.40	40.00		45.00				ĺ
Indiguiry and facility reservation - Zone 3				2	UCL	UCL4L	118.78	144.17	93.88	55.12	10.38		15.69				
Order Coordination for Unburndied Copper Loops (per loop)				3	LICI	LICI 4I	144 10	144 17	93.88	55.12	10 38		15.69				ĺ
4-Wire Unbundled Copper Loop/Long - without manual svc. 1 UCL UCL4O 77.29 119.44 81.45 55.12 10.38 15.69							144.10			00.12	10.00		10.00				
4-Wife Unburdied Copper Loop Long - without manual svc. Inguity and facility reservation - Zone 2 2 UCL UCL40 118.78 119.44 81.45 55.12 10.38 15.69																	
Inquiry and facility reservation - Zone 2				1	UCL	UCL4O	77.29	119.44	81.45	55.12	10.38		15.69				<u> </u>
4-Wire Unbundled Coppler Logo/Long - without manual svc. In unjury and facility reservation - Zone 3																	ĺ
Inquiry and facility reservation - Zone 3				2	UCL	UCL4O	118.78	119.44	81.45	55.12	10.38		15.69				├
Order Coordination for Unbundled Copper Loops (per loop)				2	LICI	LICL 4O	144 10	110 44	91 15	55 12	10.20		15.60				ĺ
UAL, UHL, UCL, UEQ, ULS, UEA, UEAN, UDL, UDC, UDN, UDL, USL ULMZG 15.68 15.69 170.89				, J			144.10			33.12	10.36		13.03				
Unbundled Loop Modification, Removal of Load Coils - 2 Wire pair less than or equal to 18k ft UEANIL, UDL, UDC. UDN, UDL, USL ULMZL ULZ	LOOP MODIFIC																
Greater than 18k ft		pair less than or equal to 18k ft			UEQ, ULS, UEA, UEANL, UDL, UDC,	ULM2L		32.46	32.46				15.69				
Unbundled Loop Modification Removal of Load Coils - 4 Wire less than or equal to 18K ft UHL, UCL ULM4L 32.46 32.46 32.46 15.69								470.00	470.00				45.00				ĺ
Less than or equal to 18K ft			 	1	UUL, ULO	ULIVIZG		170.89	170.89	 			15.69				
Unbundled Loop Modification Removal of Load Coils - 4 Wire UCL ULM4G 170.89 170.89 170.89 15.69					UHL, UCL	ULM4L		32.46	32.46				15.69				İ
Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop USL UEQ, UEF, ULS, UEQ, UEF, ULS, UDC, UDN, UDL, UDC, UDC, UDC, UDC, UDC, UDC, UDC, UDC		Unbundled Loop Modification Removal of Load Coils - 4 Wire															
Unbundled Loop Modification Removal of Bridged Tap Removal, per unbundled loop		pair greater than 18k ft				ULM4G		170.89	170.89				15.69				!
Sub-Loop Distribution Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up					UEQ, UEF, ULS, UEA, UEANL, UDL, UDC, UDN, UDL,	ULMBT		32.48	32.48				15.69				
Sub-Loop - Per Cross Box Location - CLEC Feeder Facility Set-Up		au Diatuikusiau		-		-											
Up	Sub-L0			1							1	1					
Sub-Loop - Per Building Equipment Room - CLEC Feeder UEANL USBSC 177.84 177.84 177.84 15.69 15			I		UEANL	USBSA		241.42	241.42				15.69				<u> </u>
Sub-Loop - Per Building Equipment Room - CLEC Feeder UEANL USBSC 177.84 177.84 177.84 15.69 15.69																	1
Facility Set-Up		Sub-Loop - Per Cross Box Location - Per 25 Pair Panel Set-Up Sub-Loop - Per Building Equipment Room - CLEC Feeder			UEANL	USBSB		22.69	22.69				15.69				
Set-Up		Facility Set-Up	ı		UEANL	USBSC		177.84	177.84				15.69				<u> </u>
Zone 1		Set-Up	1		UEANL	USBSD		55.58	55.58				15.69				<u> </u>
Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -			ı	1	LIFANI	USBN2	8 87	65 94	31 03	45.35	6.71		15.60				
		Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop -	<u> </u>	2	UEANL	USBN2	12.58			45.35	6.71						

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RAT	TES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop Distribution Per 2-Wire Analog Voice Grade Loop - Zone 3	1	3	UEANL	USBN2	14.79	65.94	31.03	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 1		1	UEANL	USBN4	14.11	79.21	44.29	49.82	9.09		15.69				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop - Zone 2		2	UEANL	USBN4	19.40	79.21	44.29	49.82	9.09		15.69				
	Sub-Loop Distribution Per 4-Wire Analog Voice Grade Loop -															
	Zone 3		3	UEANL	USBN4	18.90	79.21	44.29	49.82	9.09		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop 2-Wire Intrabuilding Network Cable (INC)	ı	-	UEANL	USBR2	2.41	53.13	18.21	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	Sub-Loop 4-Wire Intrabuilding Network Cable (INC)	1		UEANL	USBR4	5.36	59.38	24.47	49.82	9.09		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEANL	USBMC		8.17	8.17								
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	I		UEF	UCS2X	7.11	65.94	31.03	45.35	6.71		15.69				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	- 1	2	UEF	UCS2X	9.83	65.94	31.03	45.35	6.71		15.69				
	2 Wire Copper Unbundled Sub-Loop Distribution - Zone 3	I	3	UEF	UCS2X	10.48	65.94	31.03	45.35	6.71		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.17	8.17								
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 1	- 1	1	UEF	UCS4X	7.85	79.21	44.29	49.82	9.09		15.69				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 2	ı	2	UEF	UCS4X	14.17	79.21	44.29	49.82	9.09		15.69				
	4 Wire Copper Unbundled Sub-Loop Distribution - Zone 3		3	UEF	UCS4X	12.64	79.21	44.29	49.82	9.09		15.69				
	Order Coordination for Unbundled Sub-Loops, per sub-loop pair			UEF	USBMC		8.17	8.17								
Unbun	dled Sub-Loop Modification						_	-								
	Unbundled Sub-Loop Modification - 2-W Copper Dist Load															
	Coil/Equip Removal per 2-W PR			UEF	ULM2X		176.17	5.11				15.69				
	Unbundled Sub-loop Modification - 4-W Copper Dist Load Coil/Equip Removal per 4-W PR			UEF	ULM4X		176.17	5.11				15.69				
	Unbundled Sub-loop Modification - 2-w/4-w Copper Dist Bridged															
Habira	Tap Removal, per PR unloaded dled Network Terminating Wire (UNTW)			UEF	ULM4T		278.82	6.13				15.69				
Unbun	Unbundled Network Terminating Wire (UNTW) per Pair			UENTW	UENPP	0.3303	30.20	30.20				15.69				
Netwo	rk Interface Device (NID)			OLIVIV	OLIVIT	0.3303	30.20	30.20				15.05				
	Network Interface Device (NID) - 1-2 lines			UENTW	UND12		43.68	28.79				15.69				
	Network Interface Device (NID) - 1-6 lines			UENTW	UND16		64.42	49.53				15.69				
	Network Interface Device Cross Connect - 2 W			UENTW	UNDC2		5.92	5.92				15.69				
	Network Interface Device Cross Connect - 4W			UENTW	UNDC4		5.92	5.92				15.69				
SUB-LOOPS																
Sub-Lo	pop Feeder			1154												
	USL-Feeder, DS0 Set-up per Cross Box location - CLEC Distribution Facility set-up			UEA, UDN,UCL,UDL,UDC	USBFW		241.42					15.69				
	USL Feeder - DS0 Set-up per Cross Box location - per 25 pair set-up			UEA, UDN,UCL,UDL,UDC	USBFX		22.69	22.69				15.69				<u>-</u>
-	USL Feeder DS1 Set-up at DSX location, per DS1 termination	 		USL	USBFZ		523.87	11.34				15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground Start, Voice Grade - Zone 1		1	UEA	USBFA	8.93	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Ground-Start, Voice															
	Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, Per 2 Wire Ground-Start,		2	UEA	USBFA	11.74	93.28	56.69	54.68	13.74		15.69				
	Voice Grade - Zone 3 Order Coordination for Specified Conversion Time, per LSR		3	UEA UEA	USBFA OCOSL	14.74	93.28	56.69	54.68	13.74		15.69				
	Unbundlde Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice		1	UEA	OCOSL		18.13									
	Grade - Zone 1	l	1	UEA	USBFB	8.93	93.28	56.69	54.68	13.74		15.69				

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	TES(\$)				Svc Order Submitted Manually per LSR	Incremental	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
															Disc 1st	Disc Add I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	When the LO Labora Fredrick and Oliver Long Oliver Market						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Unbundled Sub-Loop Feeder Loop, 2 Wire Loop-Start, Voice Grade - Zone 2		2	UEA	USBFB	11.74	93.28	56.69	54.68	13.74		15.69				
 	Unbundled Sub-Loop Feeder Loop, 2 Wire Start Loop, Voice			OLA	OSBI B	11.74	93.20	30.09	34.00	13.74		13.09				
	Grade - Zone 3		3	UEA	USBFB	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination for Specified Time Conversion, per LSR			UEA	OCOSL		18.13									
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery,															
	Voice Grade - Zone 1		1	UEA	USBFC	8.93	93.28	56.69	54.68	13.74		15.69				
	Unbundled Sub-Loop Feeder Loop, 2 Wire Reverse Battery, Voice Grade - Zone 2		2	UEA	USBFC	11.74	93.28	56.69	54.68	13.74		15.69				
 	Unbundled Sub-Loop Feeder Loop, 2 Wire Analog Reverse			OLA	USBI C	11.74	93.20	30.09	54.00	13.74		13.09				
	Battery, Voice Grade - Zone 3		3	UEA	USBFC	14.74	93.28	56.69	54.68	13.74		15.69				
	Order Coordination For Specified Conversion Time, per LSR			UEA	OCOSL		18.13									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice															
	Grade - Zone 1		1	UEA	USBFD	21.63	107.91	70.36	62.26	17.52		15.69				
1 1	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground-Start, Voice Grade - Zone 2		2	UEA	USBFD	27.57	107.91	70.36	62.26	17.52		15.69				
 	Unbundled Sub-Loop Feeder Loop, 4 Wire Ground Start, Voice	<u> </u>		ULA	USDFD	21.51	107.91	10.36	02.20	17.52		15.69		1		
	Grade - Zone 3		3	UEA	USBFD	26.04	107.91	70.36	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR			UEA	OCOSL		18.13									
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice															
	Grade - Zone 1		1	UEA	USBFE	21.63	107.91	70.36	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		2	UEA	LIODEE	07.57	407.04	70.00	00.00	47.50		45.00				
	Grade - Zone 2 Unbundled Sub-Loop Feeder Loop, 4 Wire Loop-Start, Voice		2	UEA	USBFE	27.57	107.91	70.36	62.26	17.52		15.69				
	Grade - Zone 3		3	UEA	USBFE	26.04	107.91	70.36	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR		Ŭ	UEA	OCOSL	20.04	18.13	70.00	02.20	17.02		10.00				
	Unbundled Sub-Loop Feeder Loop, 2 Wire ISDN BRI - Zone 1		1	UDN	USBFF	17.05	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 2		2	UDN	USBFF	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire ISDN BRI - Zone 3		3	UDN	USBFF	23.49	106.47	68.92	55.81	13.37		15.69				
	Order Coordination For Specified Conversion Time, Per LSR Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		-1	UDN UDC	OCOSL USBFS	17.05	18.13 106.47	68.92	55.81	13.37		15.69				
—	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)		2	UDC	USBFS	20.92	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder, 2 Wire UDC (IDSL compatible)			UDC	USBFS	23.49	106.47	68.92	55.81	13.37		15.69				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 1			USL	USBFG	55.85	102.19	64.64	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 2		2	USL	USBFG	109.16	102.19	64.64	62.26	17.52		15.69				
	Unbundled Sub-Loop Feeder Loop, 4-Wire DS1 - Zone 3		3	USL	USBFG	203.35	102.19	64.64	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, Per LSR		_	USL	OCOSL USBFH	5.98	18.13	40.40	50.44	40.00		45.00				
-	Unbundled Sub-Loop Feeder, 2-Wire Copper Loop - Zone 1 Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone		1	UCL	USBFH	5.98	83.97	46.42	53.14	10.69		15.69				
	2		2	UCL	USBFH	4.80	83.97	46.42	53.14	10.69		15.69				
	Unbundled Sub-Loop Feeder Loop, 2-Wire Copper Loop - Zone	1			1		55.57	10.72	55.14			.0.00				
	3		3	UCL	USBFH	4.59	83.97	46.42	53.14	10.69		15.69				
	Order Coordination For Specified Conversion Time, per LSR			UCL	OCOSL	40	18.13					45.00				
\vdash	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 1	<u> </u>		UCL	USBFJ	13.21	101.22	63.67	58.03	13.29		15.69				
\vdash	Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 2 Sub-Loop Feeder - Per 4-Wire Copper Loop - Zone 3	 		UCL	USBFJ USBFJ	8.28 8.42	101.22 101.22	63.67 63.67	58.03 58.03	13.29 13.29	 	15.69 15.69				
	Order Coordination For Specified Conversion Time, per LSR	 	3	UCL	OCOSL	0.42	18.13	03.07	30.03	13.29	 	15.09				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		1	UDL	USBFN	21.02	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		2	UDL	USBFN	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 19.2 Kbps Digital Grade Loop		3	UDL	USBFN	20.17	102.19	64.64	62.26	17.52		15.69				
1 1	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -		1	UDL	USBFO	24.22	400.40	04.04	60.00	47.50		45.00				
\vdash	Zone 1 Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -	 	1	UDL	OSBLO	21.02	102.19	64.64	62.26	17.52	 	15.69				
	Zone 2		2	UDL	USBFO	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 56 Kbps Digital Grade Loop -	t e			1 0	250	.020	004	52.20	52	l –	.0.00				
	Zone 3	<u> </u>	3	UDL	USBFO	20.17	102.19	64.64	62.26	17.52	<u></u>	15.69				
	Order Coordination For Specified Time Conversion, per LSR			UDL	OCOSL		18.13	•								
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_	LIDI	HODED	24.22	400.40	04.61	00.00	47.50		45.00				
	Zone 1	<u> </u>	1	UDL	USBFP	21.02	102.19	64.64	62.26	17.52	1	15.69	l	l		l

UNBUNDLI	ED NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop -		_													
	Zone 2		2	UDL	USBFP	21.30	102.19	64.64	62.26	17.52		15.69				
	Sub-Loop Feeder - Per 4-Wire 64 Kbps Digital Grade Loop - Zone 3		3	UDL	USBFP	20.17	102.19	64.64	62.26	17.52		15.69				
	Order Coordination For Specified Conversion Time, per LSR		3	UDL	OCOSL	20.17	18.13	04.04	02.20	17.52		15.69				
SUB-LOOPS				ODL	OCOGL		10.13									
	Loop Feeder															
0	Sub Loop Feeder - DS3 - Per Mile Per Month			UE3	1L5SL	20.44										
	Sub Loop Feeder - DS3 - Facility Termination Per Month	П		UE3	USBF1	348.12	3,392.00	407.90	160.83	91.17		15.69				
	Sub Loop Feeder – STS-1 – Per Mile Per Month	I		UDLSX	1L5SL	20.44	-,			-						
	Sub Loop Feeder - STS-1 - Facility Termination Per Month	I		UDLSX	USBF7	369.07	3,392.00	407.90	160.83	91.17		15.69				
	Sub Loop Feeder – OC-3 – Per Mile Per Month	- 1		UDLO3	1L5SL	15.51										
	Sub Loop Feeder - OC-3 - Facility Termination Protection Per															
	Month	I		UDLO3	USBF5	56.04										
	Sub Loop Feeder - OC-3 - Facility Termination Per Month	I		UDLO3	USBF2	565.50	3,392.00	407.90	160.83	91.17	<u> </u>	15.69				
	Sub Loop Feeder - OC-12 - Per Mile Per Month	I		UDL12	1L5SL	19.08										
	Sub Loop Feeder - OC-12 - Facility Termination Protection Per	١.														
	Month	<u> </u>		UDL12	USBF6	669.82		107.00	100.00			1= 00				
	Sub Loop Feeder - OC-12 - Facility Termination Per Month	<u> </u>		UDL12	USBF3	1,840.00	3,392.00	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-48 - Per Mile Per Month	l l		UDL48	1L5SL	62.60			-							
	Sub Loop Feeder - OC-48 - Facility Termination Protection Per Month	١.,		UDL48	USBF9	326.16										
	Sub Loop Feeder - OC-48 - Facility Termination Per Month	l i		UDL48	USBF4	1,560.00	3,578.00	407.90	160.83	91.17		15.69				
	Sub Loop Feeder - OC-12 Interface On OC-48	H		UDL48	USBF8	366.86	789.85	407.90	160.83	91.17		15.69				
IINBIINDI ED	LOOP CONCENTRATION	<u>'</u>		UDL46	USBI 6	300.00	709.00	407.50	100.03	91.17		13.09				
UNBUNDEED	Unbundled Loop Concentration - ISDN Loop Interface (Brite															
	Card)			UDN	ULCC1	7.02	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - UDC Loop Interface (Brite			05.1	02001	7.02	10.00	.0.00	0	0.01		10.00				
	Card)			UDC	ULCCU	7.02	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration2 Wire Voice-Loop Start or								-							
	Ground Start Loop Interface (POTS Card)			UEA	ULCC2	1.75	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - 2 Wire Voice - Reverse Battery															
	Loop Interface (SPOTS Card)			UEA	ULCCR	10.42	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - 4 Wire Voice Loop Interface															
	(Specials Card)			UEA	ULCC4	6.22	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - TEST CIRCUIT Card			ULC	UCTTC	30.38	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - Digital 19.2 Kbps Data Loop															
	Interface	ļ	<u> </u>	UDL	ULCC7	9.21	10.56	10.50	5.41	5.37		15.69				
	Unbundled Loop Concentration - Digital 56 Kbps Data Loop			LIDI		20.	40.50	10.50				45.00				
	Interface	<u> </u>	<u> </u>	UDL	ULCC5	9.21	10.56	10.50	5.41	5.37	1	15.69				
	Unbundled Loop Concentration - Digital 64 Kbps Data Loop Interface	1		UDL	ULCC6	9.21	10.56	10.50	5.41	5.37		15.69	1		1	
UNE OTHER	PROVISIONING ONLY - NO RATE			UDL	OLCOO	5.21	10.56	10.50	5.41	5.57		13.09		-	1	1
I I	NID - Dispatch and Service Order for NID installation	 	t	UENTW	UNDBX											
	UNTW Circuit Id Establishment, Provisioning Only - No Rate		<u> </u>	UENTW	UENCE											
	The factor of th		1	UEANL,UEF,UEQ,U					1				1		İ	Ì
	Unbundled Contract Name, Provisioning Only - No Rate			ENTW	UNECN											
UNE OTHER,	PROVISIONING ONLY - NO RATE			<u> </u>									İ	<u> </u>		
				UAL,UCL,UDC,UDL,												
	Unbundled Contact Name, Provisioning Only - no rate		<u> </u>	UDN,UEA,UHL,ULC	UNECN	0.00	0.00									
	Unbundled Sub-Loop Feeder-2 Wire Cross Box Jumper - no															
	rate		<u> </u>	UEA,UDN,UCL,UDC	USBFQ	0.00	0.00									
	Unbundled Sub-Loop Feeder-4 Wire Cross Box Jumper - no	1			LIODED	0.00	0.00						1		1	
	Inhundled DS1 Lean Superframe Format Option, no rate	 	 	UEA,USL,UCL,UDL	USBFR	0.00	0.00		 		1		 	-	 	1
	Unbundled DS1 Loop - Superframe Format Option - no rate Unbundled DS1 Loop - Expanded Superframe Format option -	 	1	USL	CCOSF	0.00	0.00									
	no rate	1		USL	CCOEF	0.00	0.00						1		1	
		•	1	OOL	JUULI	0.00	0.00		1		1		1	1	1	1

UNBUNDL	ED NETWORK ELEMENTS - South Carolina		_	I .		T							Attachment:		Table 1	<u> </u>
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Charge -
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	History Book Banking						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	High Capacity Unbundled Local Loop - DS3 - Per Mile per month			UE3	1L5ND	12.26										
	High Capacity Unbundled Local Loop - DS3 - Facility			ULS	TESIND	12.20									1	+
	Termination per month			UE3	UE3PX	306.36	452.52	264.53	119.75	83.77		15.69				
	High Capacity Unbundled Local Loop - STS-1 - Per Mile per						-									
	month			UDLSX	1L5ND	12.26						15.69				
	High Capacity Unbundled Local Loop - STS-1 - Facility															
LOOPMAKE	Termination per month			UDLSX	UDLS1	313.49	452.52	264.53	119.75	83.77		15.69				
LOOP MAKE	Loop Makeup - Preordering Without Reservation, per working or					-									-	+
	spare facility queried (Manual).			UMK	UMKLW		24.04	24.04								
	Loop Makeup - Preordering With Reservation, per spare facility			OWIN	OWINE		24.04	24.04								+
	queried (Manual).			UMK	UMKLP		25.49	25.49								
	Loop MakeupWith or Without Reservation, per working or															1
	spare facility queried (Mechanized)			UMK	PSUMK		0.34	0.34								
	DEDICATED TRANSPORT	L	<u> </u>		1											
	: INTEROFFICE CHANNEL DEDICATED TRANSPORT - minimu	m billin	g perio	od - below DS3=one	month, DS3/	STS-1=four mo	nths									
INTE	ROFFICE CHANNEL - DEDICATED TRANSPORT Interoffice Channel - Dedicated Transport - 2-Wire Voice Grade -															-
	Per Mile per month			U1TVX	1L5XX	0.0167										
	Interoffice Channel - Dedicated Transport- 2- Wire Voice Grade -			OTTVX	TESTON	0.0107										+
	Facility Termination per month			U1TVX	U1TV2	24.30	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel - Dedicated Transpor t- 2-Wire Voice Grade															1
	Rev Bat Per Mile per month			U1TVX	1L5XX	0.0167										
	Interoffice Channel - Dedicated Transport- 2- Wire VG Rev Bat.	1		l <u> </u>	l											
	Facility Termination per month			U1TVX	U1TR2	24.30	40.63	27.47	16.77	6.91		15.69				+
	Interoffice Channel - Dedicated Transport - 4-Wire Voice Grade - Per Mile per month	1		U1TVX	1L5XX	0.0167										
	Interoffice Channel - Dedicated Transport - 4- Wire Voice Grade			UTIVA	ILJAA	0.0107									1	+
	- Facility Termination per month			U1TVX	U1TV4	21.29	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel - Dedicated Transport - 56 kbps - per mile															
	per month			U1TDX	1L5XX	0.0167										
	Interoffice Channel - Dedicated Transport - 56 kbps - Facility			l	l											
	Termination per month			U1TDX	U1TD5	16.76	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel - Dedicated Transport - 64 kbps - per mile			U1TDX	1L5XX	0.0167										
	per month Interoffice Channel - Dedicated Transport - 64 kbps - Facility			UTIDX	ILSAA	0.0167										+
	Termination per month			U1TDX	U1TD6	16.76	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel - Dedicated Channel - DS1 - Per Mile per														İ	†
	month			U1TD1	1L5XX	0.3415										
	Interoffice Channel - Dedicated Tranport - DS1 - Facility															
	Termination per month			U1TD1	U1TF1	77.14	89.47	81.99	16.39	14.48		15.69				
	Interoffice Channel - Dedicated Transport - DS3 - Per Mile per month			U1TD3	1L5XX	8.02										
	Interoffice Channel - Dedicated Transport - DS3 - Facility			01103	ILSAA	0.02										+
	Termination per month			U1TD3	U1TF3	880.65	279.37	163.12	60.33	58.59		15.69				
	Interoffice Channel - Dedicated Transport - STS-1 - Per Mile per								55.55							
	month			U1TS1	1L5XX	8.02										
	Interoffice Channel - Dedicated Transport - STS-1 - Facility															
	Termination per month		<u> </u>	U1TS1	U1TFS	880.55	279.37	163.12	60.33	58.59		15.69				₩
	AL CHANNEL - DEDICATED TRANSPORT E: LOCAL CHANNEL DEDICATED TRANSPORT - minimum billin	a noric	d - bol-	W DS3-one menth	D\$3/\$T\$_4	four months									 	
INOTE	Local Channel - Dedicated - 2-Wire Voice Grade Per Month	y perio	u - pelo	ULDVX	ULDV2	15.33	193.53	33.24	36.72	3.21		15.69			+	+
	Local Channel - Dedicated - 2-Wire Voice Grade Fer Month Local Channel - Dedicated - 2-Wire Voice Grade Rev Bat per	-	l -	525 V.	JLD 12	10.00	190.00	33.24	30.72	5.21	1	10.08			†	
	month			ULDVX	ULDR2	15.33	193.53	33.24	36.72	3.21		15.69			1	
	Local Channel - Dedicated - 4-Wire Voice Grade per month			UNDVX	ULDV4	16.54	193.97	33.68	37.19	3.68		15.69				
	Local Channel - Dedicated - DS1 per month - Zone 1		1	ULDD1	ULDF1	42.62	177.87	154.06	22.24	15.30		15.69				
igwdown	Local Channel - Dedicated - DS1 per month - Zone 2	ļ	2	ULDD1	ULDF1	70.32	177.87	154.06	22.24	15.30		15.69			1	
	Local Channel - Dedicated - DS1 per month - Zone 3		3	ULDD1	ULDF1	190.68	177.87	154.06	22.24	15.30		15.69				

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	FES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - DS3 - Per Mile per month			ULDD3	1L5NC	11.93										
	Interoffice Channel - Dedicated Transport - DS3 - Facility Termination per month			U1TD3	U1TF3	446.00	452.52	264.53	119.75	83.77		15.69				
	Local Channel - Dedicated - STS-1- Per Mile per month			ULDS1	1L5NC	11.93	452.52	204.55	119.75	03.11		15.69			-	
	Local Channel - Dedicated - STS-1 - Facility Termination per			OLDOT	TESINO	11.95										
	month			ULDS1	ULDFS	435.10	452.52	264.53	119.75	83.77		15.69				
MULTIPLEXE	RS															
	Channelization - DS1 to DS0 Channel System			UXTD1	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UDL	1D1DD	1.19	6.59	4.73				15.69				
	2-wire ISDN COCI (BRITE) - DS1 to DS0 Channel Systsem - per															
	month			UDN	UC1CA	2.56	6.59	4.73				15.69				
	Voice Grade COCI - DS1 to DS0 Channel System - per month	1	<u> </u>	UEA	1D1VG	0.56	6.59	4.73				15.69				
	DS3 to DS1 Channel System per month			UXTD3	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
	STS1 to DS1 Channel System per month DS3 Interface Unit (DS1 COCI) used with Loop per month			UXTS1 USL	MQ3 UC1D1	144.02 8.64	178.54 6.59	94.18 4.73	33.33	31.90		15.69 15.69				
	DS3 Interface Unit (DS1 COCI) used with Local Channel per			USL	OCIDI	0.04	0.59	4.73				13.09				
	month DS3 Interface Unit (DS1 COCI) used with Interoffice Channel			ULDD1	UC1D1	8.64	6.59	4.73				15.69				
	per month			U1TD1	UC1D1	8.64	6.59	4.73				15.69				
DARK FIBER	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction				+											
	Thereof per month - Local Channel			UDF	1L5DC	97.65										
	NRC Dark Fiber - Local Channel			UDF	UDFC4	57.00	640.51	138.17	317.76	198.11		15.69				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Interoffice Channel			UDF	1L5DF	36.41										
	NRC Dark Fiber - Interoffice Channel			UDF	UDF14		640.51	138.17	317.76	198.11		15.69				
	Dark Fiber, Four Fiber Strands, Per Route Mile or Fraction															
	Thereof per month - Local Loop			UDF	1L5DL	97.65	0.10 = 1		0.4==0	100.11		4= 00				
TRANSPORT	NRC Dark Fiber - Local Loop			UDF	UDFL4		640.51	138.17	317.76	198.11		15.69			-	
	nal Features & Functions:				+										1	
	TEN DIGIT SCREENING															
	8XX Access Ten Digit Screening, Per Call			OHD	1	0.0006673									İ	
	8XX Access Ten Digit Screening, Reservation Charge Per 8XX															
	Number Reserved			OHD	N8R1X		2.59	0.44				15.69				
	8XX Access Ten Digit Screening, Per 8XX No. Established W/O POTS Translations			OHD			5.95	0.81	4.58	0.54		15.69				
	8XX Access Ten Digit Screening, Per 8XX No. Established With POTS Translations			OHD	N8FTX		5.95	0.81	4.58	0.54		15.69				
	8XX Access Ten Digit Screening, Customized Area of Service Per 8XX Number			OHD	N8FCX		2.59	1.30				15.69				
	8XX Access Ten Digit Screening, Multiple InterLATA CXR															
	Routing Per CXR Requested Per 8XX No.	ļ	ļ	OHD	N8FMX		3.03	1.74				15.69			1	
	8XX Access Ten Digit Screening, Change Charge Per Request			OHD	N8FAX		3.03	0.44				15.69				
	8XX Access Ten Digit Screening, Call Handling and Destination Features	1	1	OHD	N8FDX		2.59	2.59				15.69				
 	8XX Access Ten Digit Screening, w/ 8XX No. Delivery	1	1	OHD	INOLDV	0.0006673	2.59	2.59			1	15.69			+	1
 	8XX Access Ten Digit Screening, w/ 6XX No. Delivery	-	l	OHD	+	0.0006673									†	-
LINE INFORM	ATION DATA BASE ACCESS (LIDB)	1			1										1	
	LIDB Common Transport Per Query			OQT		0.0000246										
	LIDB Validation Per Query			OQU		0.0138158										
	LIDB Originating Point Code Establishment or Change			OQT, OQU	NRPBX		34.40		42.18			15.69			1	
SIGNALING (C		<u> </u>	!	LIDD	TDD	40.00	05.01	05.01	40.10	40 10						ļ
	CCS7 Signaling Connection, Per 56 Kbps Facility CCS7 Signaling Termination, Per STP Port	 	!	UDB UDB	TPP++ PT8SX	16.93 163.49	35.61	35.61	16.48	16.48					1	
		1	1	UUDB	PIBOX	163.49			1		1	i l			1	<u> </u>
	CCS7 Signaling Usage, Per TCAP Message			UDB		0.0000692										

UNBUNDLE	D NETWORK ELEMENTS - South Carolina				· · · · · · · · · · · · · · · · · · ·		· · · · · · · · · · · · · · · · · · ·			· · · · · · · · · · · · · · · · · · ·			Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			ΓES(\$)				Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge -	Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	CCS7 Signaling Connection, Per link (B link) (also known as D															
	link)			UDB	TPP++	16.93	35.61	35.61	16.48	16.48		15.69				
	CCS7 Signaling Usage, Per ISUP Message			UDB	CTUEC	0.0000173										-
	CCS7 Signaling Usage Surrogate, per link per LATA			UDB	STU56	791.37										
	CCS7 Signaling Point Code, per Originating Point Code Establishment or Change, per STP affected			UDB	CCAPO		29.08	29.08	35.65	35.65		15.69				İ
	CCS7 Signaling Point Code, per Destination Point Code			UDB	CCAPO		29.06	29.00	33.63	33.03		15.69				
	Establishment or Change, Per Stp Affected			UDB	CCAPD		29.08	29.08	35.65	35.65		15.69				
E911 SERVICE				ODD	00/11/2		20.00	20.00	00.00	00.00		10.00				
	Local Channel - Dedicated - 2-wr Voice Grade					15.33	193.53	33.24	36.72	3.21		15.69				
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Mile				İ	0.0167	.00.00		332	0.21			Ì	Ì	1	
	Interoffice Transport - Dedicated - 2-wr Voice Grade Per Facility		1													
1 1	Termination	ĺ				24.30	40.63	27.47	16.77	6.91		15.69			1	1
	Local Channel - Dedicated - DS1 - Zone 1					42.62	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated - DS1 - Zone 2					70.32	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated - DS1 - Zone 3					190.68	177.87	154.06	22.24	15.30		15.69				
	Interoffice Transport - Dedicated - DS1 Per Mile					0.3415										
	Interoffice Transport - Dedicated - DS1 Per Facility Termination					77.14	89.47	81.99	16.39	14.48		15.69				
CALLING NAM	E (CNAM) SERVICE															
	CNAM For DB Owners - Service Establishment			OQV			23.00	23.00	21.15	21.15		15.69				
	CNAM For Non DB Owners - Service Establishment			OQV			23.00	23.00	21.15	21.15		15.69				
	CNAM For DB Owners - Service Provisioning With Point Code			001/			002.00	704.47	200 52	400.40		45.00				
	Establishment			OQV	+		993.09	734.47	269.53	198.18		15.69				
	CNAM For Non DB Owners - Service Provisioning With Point Code Establishment			oqv			343.09	245.69	275.87	198.18		15.69				
	CNAM for DB Owners, Per Query			OQV	+	0.0010433	343.05	245.09	213.01	190.10		13.09				
	CNAM for Non DB Owners, Per Query			OQV	+	0.0010433										
LNP Query Se				04.		0.00.00.00										
	LNP Charge Per query					0.0008837										
	LNP Service Establishment Manual						25.09	25.09	23.07	23.07		15.69				
	LNP Service Provisioning with Point Code Establishment						594.82	303.88	269.53	198.18		15.69				
OPERATOR C	ALL PROCESSING															
	Oper. Call Processing - Oper. Provided, Per Min Using BST															
	LIDB					1.20										
1 1 -	Oper. Call Processing - Oper. Provided, Per Min Using	1	1]	1	_	1
	Foreign LIDB		<u> </u>			1.24										1
	Oper. Call Processing - Fully Automated, per Call - Using BST		1		1								1			1
\vdash	LIDB	<u> </u>	<u> </u>			0.20							 	ļ	-	
1 1	Oper. Call Processing - Fully Automated, per Call - Using	ĺ				0.00									1	1
INWARD ORE	Foreign LIDB	<u> </u>	<u> </u>		+	0.20									-	
INVVARD OPEN	Inward Operator Services - Verification, Per Minute		!		+	1.15							1	-		
 	Inward Operator Services - Verification, Per Minute Inward Operator Services - Verification and Emergency Interrupt	 	!	1	+	1.15			 				1	1	t	
1 1	- Per Minute	l	1		1	1.15							1	1	I	1
BRANDING - C	PERATOR CALL PROCESSING	1	†		1	1.13					<u> </u>		 	 	I	—
	Recording of Custom Branded OA Announcement	1	†		CBAOS		7,000.00	7,000.00				15.69	1	1	1	t
	Loading of Custom Branded OA Announcement per shelf/NAV		1		CBAOL		500.00	500.00				15.69	1	İ	1	
Unbrar	iding via OLNS for UNEP CLEC		i –												1	
	Loading of OA per OCN (Regional)						1,200.00	1,200.00				15.69				
	SSISTANCE SERVICES															
DIREC	TORY ASSISTANCE ACCESS SERVICE															
	Directory Assistance Access Service Calls, Charge Per Call					0.275										
DIREC	TORY ASSISTANCE CALL COMPLETION ACCESS SERVICE (D	DACC)														
	Directory Assistance Call Completion Access Service (DACC),														1	1
ļļ.	Per Call Attempt	ļ	<u> </u>			0.10									1	1
	TORY TRANSPORT	ļ	<u> </u>												1	└
	SSISTANCE SERVICES	ļ	<u> </u>										ļ	ļ	.	1
DIREC	TORY ASSISTANCE DATA BASE SERVICE (DADS)				1						1	l]]		

UNBL	JNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEC		RATE ELEMENTS	Interi m	Zone	BCS	usoc			TES(\$)			1	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I		Incremental Charge - Manual Svo Order vs. Electronic- Disc Add'l
							Rec	Nonred		Nonrecurring					Rates(\$)		
								First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Directory Assistance Data Base Service Charge Per Listing					0.04										
		Directory Assistance Data Base Service, per month				DBSOF	150.00										
BRANI		IRECTORY ASSISTANCE															
	Facility	Based CLEC															
		Recording and Provisioning of DA Custom Branded			***	00404		0.000.00	0.000.00								
		Announcement			AMT	CBADA		6,000.00	6,000.00								
		Loading of Custom Branded Announcement per DRAM Card/Switch			AMT	CBADC		1,170.00	1,170.00								
	UNEP				AIVII	CBADC		1,170.00	1,170.00								
	UNEF	Recording of DA Custom Branded Announcement						3,000.00	3,000.00								
		Loading of DA Custom Branded Announcement per DRAM						3,000.00	3,000.00			1					
l	1	Card/Switch per OCN		1				1,170.00	1,170.00				1	I		1	
-	Unbrar	nding via OLNS for UNEP CLEC		1				1,170.00	1,170.00					-			-
	3 .ui	Loading of DA per OCN (1 OCN per Order)				1		420.00	420.00					†			<u> </u>
	†	Loading of DA per Switch per OCN						16.00	16.00					t		1	1
SELEC	TIVE R	DUTING				İ			. 2.00					İ		İ	1
		Selective Routing Per Unique Line Class Code Per Request Per															
l	1	Switch				USRCR		84.89	84.89	14.14	14.14		15.69	1			
VIRTU	AL COL	LOCATION															
		Virtual Collocation - Application Cost			AMTFS	EAF		1,207.95	1,207.95	0.51	0.51						
		Virtual Collocation - Cable Installation Cost, per cable			AMTFS	ESPCX		794.22	794.22	22.54	22.54						
		Virtual Collocation - Floor Space, per sq. ft.			AMTFS	ESPVX	3.95										
		Virtual Collocation - Power, per breaker amp			AMTFS	ESPAX	9.19										
		Virtual Collocation - Cable Support Structure, per entrance															
		cable			AMTFS	ESPSX	18.66										
					UEANL,UEA,UDN,U DC,UAL,UHL,UCL,U EQ, AMTFS, UDL, UNCVX, UNCDX,												
		Virtual Collocation - 2-wire Cross Connects (loop)			UNCNX	UEAC2	0.0317	12.32	11.83	6.04	5.45		15.69				
		Virtual Collocation - 4-wire Cross Connects (loop)			UEA,UHL,UCL,UDL, AMTFS, UAL, UDN, UNCVX, UNCDX AMTFS,UDL12, UDLO3, U1T48,	UEAC4	0.0634	12.42	11.90	6.40	5.74		15.69				
					U1T12, U1T03,												
l	1				ULDO3, ULD12,									1			
	ļ	Virtual Collocation - 2-Fiber Cross Connects		<u> </u>	ULD48, UDF	CNC2F	2.86	20.94	15.23	7.40	5.93		15.69	1		ļ	
		Virtual Collocation - 4-Fiber Cross Connects			AMTFS,UDL12, UDLO3, U1T48, U1T12, U1T03, ULDO3, ULD12, ULD48, UDF	CNC4F	5.71	25.61	19.90	9.73	8.26		15.69				
		Virtual collocation - DS1 Cross Connects			USL,ULC,AMTFS, ULR, UXTD1, UNC1X, ULDD1, U1TD1, USLEL, UNLD1	CNC1X	1.12	22.08	15.96	6.42	5.80		15.69				
		Virtual collocation - DS1 Cross Connects			UNLDT USL,ULC,AMTFS,U E3, U1TD3, UXTS1, UXTD3, UNC3X, UNCSX, ULDD3, U1TS1, ULDS1, UDLSX, UNLD3	CND3X	14.21	20.94	15.23	7.39	5.80		15.69				
		Virtual Collocation - DSS Cross Connects Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure, per linear foot			AMTFS	VE1CB	0.0022	20.94	15.23	7.39	5.93		15.69				
		Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support Structure, per linear ft			AMTFS	VE1CD	0.0033										

UNBUNDL	ED NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	res(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st		Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
															2.00 .01	
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	Virtual Collocation - Co-Carrier Cross Connects - Fiber Cable		+		-		First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Support Structure, per cable			AMTFS	VE1CC		536.56									1
-	Virtual Collocation - Co-Carrier Cross Connects - Copper/Coax		1	AWIII-3	VLICC		330.30		1							
	Cable Support Structure, per cable			AMTFS	VE1CE		536.56									1
	Virtual collocation - Security Escort - Basic, per half hour			AMTFS	SPTBX		16.96	10.75								
	Virtual collocation - Security Escort - Overtime, per half hour			AMTFS	SPTOX		22.10	13.89								
	Virtual collocation - Security Escort - Premium, per half hour			AMTFS	SPTPX		27.23	17.02								
	Virtual collocation - Maintenance in CO - Basic, per half hour			AMTFS	CTRLX		27.99	10.75								
	Virtual collocation - Maintenance in CO - Overtime, per half hour			AMTFS	SPTOM		36.56	13.89								J
	Virtual collocation - Maintenance in CO - Premium per half hour			AMTFS	SPTPM		45.12	17.02								1
VIRTUAL CO	PLLOCATION	1	1	,	O1 11 1VI		70.12	17.02								
	Virtual Collocation - 2-wire Cross Connect, Exchange Port 2-															
	Wire Analog - Res			UEPSR	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2- Wire Line Side PBX Trunk - Bus			UEPSP	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire						40.00					4= 00				1
	Voice Grade PBX Trunk - Res Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire		1	UEPSE	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Analog Bus			UEPSB	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				<u> </u>
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPSX	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				<u>j</u>
	Virtual Collocation 2-Wire Cross Connect, Exchange Port 2-Wire ISDN			UEPTX	VE1R2	0.0317	12.32	11.83	6.04	5.45		15.69				
	Virtual Collocation 4-Wire Cross Connect, Exchange Port 4-Wire ISDN DS1			UEPEX	VE1R4	1.12	22.08	15.96	6.42	5.80		15.69]
VIRTUAL CO				-												
	Virtual Collocation-2 Wire Cross Connects (Loop) for Line Splitting			UEPSR, UEPSB	VE1LS	0.0317	12.32	11.83	6.04	5.45		15.69				
AIN SELECT	IVE CARRIER ROUTING															
	Regional Service Establishment			SRC	SRCEC		101,324.34	101,324.34	8,609.85	8,609.85		15.69				,
	End Office Establishment		<u> </u>	SRC	SRCEO	0.0025020	175.66	175.66	1.70	1.70		15.69				
AIN - DELL S	Query NRC, per query OUTH AIN SMS ACCESS SERVICE		1	SRC		0.0035036										
AIN - BEEES	AIN SMS Access Service - Service Establishment, Per State,		1													
	Initial Setup			A1N	CAMSE		39.53	39.53	40.78	40.78		15.69				<u> </u>
	AIN SMS Access Service - Port Connection - Dial/Shared Access			A1N	CAMDP		7.85	7.85	9.11	9.11		15.69				i
	AIN SMS Access Service - Port Connection - ISDN Access			A1N	CAM1P		7.85	7.85	9.11	9.11		15.69				
	AIN SMS Access Service - User Identification Codes - Per User															
\vdash	ID Code	ļ	ļ	A1N	CAMAU		35.08	35.08	27.12	27.12		15.69				
	AIN SMS Access Service - Security Card, Per User ID Code, Initial or Replacement			A1N	CAMRC		41.98	41.98	11.74	11.74		15.69				
	AIN SMS Access Service - Storage, Per Unit (100 Kilobytes)					0.0027		· · · · · ·		· · · · · ·						
	AIN SMS Access Service - Session, Per Minute					0.7121										
	AIN SMS Access Service - Company Performed Session, Per Minute					0.8364										ı
AIN - BELLS	OUTH AIN TOOLKIT SERVICE		1			0.0304			1							
	AIN Toolkit Service - Service Establishment Charge, Per State,				D.1065							,				
 	Initial Setup	ļ	1	CAM	BAPSC		39.53	39.53	40.78	40.78		15.69				
\vdash	AIN Toolkit Service - Training Session, Per Customer AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per	1			BAPVX		4,211.54	4,211.54	0.00	0.00		15.69				
	DN, Term. Attempt				BAPTT		7.85	7.85	9.11	9.11		15.69				<u>ı </u>
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Delay				BAPTD		7.85	7.85	9.11	9.11		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, Off-Hook Immediate				BAPTM		7.85	7.85	9.11	9.11		15.69				
	DIN, OIT-HOOK IMMediate	l			DAPTIVI		7.85	7.85	9.11	9.11	l	15.09		L		

UNBUNDLE	D NETWORK ELEMENTS - South Carolina			•									Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC			TES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
	AIN Tablis Carios Trianas Assau Charas Das Trianas Das						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per DN, 10-Digit PODP				BAPTO		34.54	34.54	14.39	14.39		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per				BAFTO		34.34	34.34	14.55	14.35		13.09			1	
	DN. CDP				BAPTC		34.54	34.54	14.39	14.39		15.69				
	AIN Toolkit Service - Trigger Access Charge, Per Trigger, Per															
	DN, Feature Code				BAPTF		34.54	34.54	14.39	14.39		15.69				
	AIN Toolkit Service - Query Charge, Per Query					0.0558238										
	AlN Toolkit Service - Type 1 Node Charge, Per AlN Toolkit					0.000004.4										
	Subscription, Per Node, Per Query AIN Toolkit Service - SCP Storage Charge, Per SMS Access					0.0069214										
	Account, Per 100 Kilobytes					0.07										
	AIN Toolkit Service - Monthly report - Per AIN Toolkit Service					0.07										
	Subscription			CAM	BAPMS	11.87	7.85	7.85	5.52	5.52		15.69				
	AIN Toolkit Service - Special Study - Per AIN Toolkit Service															
	Subscription			CAM	BAPLS	3.51	8.68	8.68				15.69				
	AIN Toolkit Service - Call Event Report - Per AIN Toolkit Service															
	Subscription			CAM	BAPDS	8.48	7.85	7.85	5.52	5.52		15.69				
	AIN Toolkit Service - Call Event Special Study - Per AIN Toolkit Service Subscription			CAM	BAPES	0.12	8.68	8.68				15.69				
NHANCED E	XTENDED LINK (EELs)			CAW	DAFLS	0.12	0.00	0.00				13.09			1	
	: New EELs available in GA, TN, KY, LA, MS, & SC and density	zone 1	of foll	lowing MSAs: Orlan	do. FL: Miam	i. FL: Ft. Laude	rdale. FL:									
NOTE	: Charlotte-Gastonia-Rockhill, NC; Greensboro-Winston Salem-	High P	oint, N	C. Use all rates belo	w except Sw	itch As Is Charg	ge.									
	: In all states, EEL network elements shown below also apply t							As Is Charge a	pplies to currer	ntly combined	facilities co	nverted to	UNEs.(Non-re	curring rates	do not apply	·.)
	: In GA, TN, KY, LA, MS & SC the EEL network elements apply				lements.(No	Switch As Is Ch	arge.)									
2-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	RANSPORT (EEL)												
	First 2-Wire VG Loop(SL2) in a DS1 Interofficed Transport Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed		<u> </u>	UNCVX	ULALZ	10.00	105.90	00.43	33.03	10.01		13.09			1	
	Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	First 2-Wire VG Grade Loop(SL2) in a DS1 Interofficed															
	Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
	per month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 combination - Facility Termination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	DS1 Channelization System Per Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	Voice Grade COCI - DS1 To Ds0 Interface - Per Month			UNCVX	1D1VG	0.56	6.59	4.73	10.00	0.01		15.69				
	Each Additional 2-Wire VG Loop(SL 2) in the same DS1															
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1															
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.69				
	Each Additional 2-Wire VG Loop(SL2) in the same DS1 Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Voice Grade COCI - DS1 to DS0 Channel System combination -		3	UNCVA	UEALZ	20.40	105.96	00.43	55.05	10.01		15.69			-	-
	per month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
1	Nonrecurring Currently Combined Network Elements Switch -As-			0.1017	15110	0.00	0.00	0				10.00			İ	
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIR	E VOICE GRADE EXTENDED LOOP WITH DEDICATED DS1 INT	EROFF	ICE TR	RANSPORT (EEL)												
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice	1	١				400		=0.5=			4.5.5				
	Transport Combination - Zone 1 First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69			-	-
	Transport Combination - Zone 2	1	2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	First 4-Wire Analog Voice Grade Loop in a DS1 Interoffice			J. NO VA	JLAL4	45.09	102.00	34.03	39.33	14.01		13.08			 	
	Transport Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Interoffice Transport - Dedicated - DS1 combination - Per Mile															
1	Per Month			UNC1X	1L5XX	0.27										
			_	1	1			·		· ·			·		1	1
	Interoffice Transport - Dedicated - DS1 - Facility Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				

NRONDLE	D NETWORK ELEMENTS - South Carolina			1							_	_	Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	ΓES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						1100	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Channelization - Channel System DS1 to DS0 combination Per			LINGAY		407.57	04.04	00.74	40.50	0.04		45.00				İ
	Month Voice Grade COCI - DS1 to DS0 Channel System combination -			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				—
	per month			UNCVX	1D1VG	0.56	6.59	4.73				15.69				
	Additional 4-Wire Analog Voice Grade Loop in same DS1			ONOVA	15170	0.00	0.00	4.70				10.00				
	Interoffice Transport Combination - Zone 1		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				
	Additional 4-Wire Analog Voice Grade Loop in same DS1															
	Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	Additional 4-Wire Analog Voice Grade Loop in same DS1		_			40.00	400.00					4= 00				
	Interoffice Transport Combination - Zone 3 Nonrecurring Currently Combined Network Elements Switch -As-		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				İ
4-WIR	56 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE		0.1000		0.01	0.01	7.00	1.00		10.00				
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice			<u> </u>												
	Transport Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
	First 4-wire 56Kbps Digital Grade Loop in a DS1 Interoffice		_													
	Transport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				-
	First 4-Wire 56Kbps Digital Grade Loop in a DS1 Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				İ
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	UNCDA	ODESO	34.74	120.00	09.12	39.33	14.01		13.09				
	Per Month			UNC1X	1L5XX	0.27										İ
	Interoffice Transport - Dedicated - DS1 - combination Facility					,										
	Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Channelization - Channel System DS1 to DS0 combination Per															
	Month			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1			UNCDA	טטוטו	1.19	6.59	4.73				15.69				—
	Interoffice Transport Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				İ
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1									_						
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
	Additional 4-Wire 56Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
	OCU-DP COCI (data) - DS1 to DS0 Channel System - combination per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.59	4.73				15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-			UNCDA	טטוטו	1.19	6.59	4.73				15.69				-
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIR	64 KBPS EXTENDED DIGITAL LOOP WITH DEDICATED DS1	INTERC	FFICE	TRANSPORT (EEL)												
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice															
	Transport Combination - Zone 1		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
	First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	LINODY	LIDI 04	00.00	400.00	00.40	50.05	44.04		45.00				İ
_	Transport Combination - Zone 2 First 4-Wire 64Kbps Digital Grade Loop in a DS1 Interoffice		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	Transport Combination - Zone 3		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				İ
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		_													
	Per Month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 combination - Facility															
	Termination Per Month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
	Channelization - Channel System DS1 to DS0 combination Per			LINCAY	MO1	407.57	04.04	00.74	40.50	0.01		45.00				
	Month OCU-DP COCI (data) - DS1 to DS0 Channel System			UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81	-	15.69				
	combination - per month (2.4-64kbs)	l		UNCDX	1D1DD	1.19	6.59	4.73				15.69				1
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1					1.19	0.00	4.73				10.00				
	Interoffice Transport Combination - Zone 1	<u> </u>	1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				<u> </u>
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1															
	Interoffice Transport Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
	Additional 4-Wire 64Kbps Digital Grade Loopin same DS1	1	^	LINCDY	LIDICA	0471	400.00	00.40	50.05	44.61	1	45.00				1
	Interoffice Transport Combination - Zone 3	l	3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61	l	15.69				<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RA ⁻	TES(\$)				Submitted	Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec			Disconnect				Rates(\$)		
	OCU-DP COCI (data) - DS1 to DS0 Channel System						First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	combination - per month (2.4-64kbs)			UNCDX	1D1DD	1.19	6.59	4.73				15.69				j '
	Nonrecurring Currently Combined Network Elements Switch -As-	-		0.1027		0	0.00					10.00				
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
4-WIRE	E DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS1 INTI	EROFFI	CE TRA	NSPORT (EEL)												
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				Ĭ
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice															
	Transport - Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	4-Wire DS1 Digital Loop in Combination with DS1 Interoffice Transport - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				Ĭ
	Interoffice Transport - Dedicated - DS1 combination - Per Mile		3	ONOTA	OOLXX	201.03	200.00	137.03	44.00	11.75		13.03				
	Per Month			UNC1X	1L5XX	0.27										
	Interoffice Transport - Dedicated - DS1 combination - Facility			LINGAY	114754	04 = 1	00.17	04.00	40.00	44.50		45.60				1
	Termination Per Month Nonrecurring Currently Combined Network Elements Switch -As-			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				-
	Is Charge			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				ĺ
4-WIRE	DS1 DIGITAL EXTENDED LOOP WITH DEDICATED DS3 INT	EROFFI	CE TRA	NSPORT (EEL)												
	First DS1Loop in DS3 Interoffice Transport Combination - Zone			LINIOAY	1101.307	00.07	050.00	457.00	44.00	44.70		45.00				Ĭ
	First DS1Loop in DS3 Interoffice Transport Combination - Zone		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				Ĭ
	First DS1Loop in DS3 Interoffice Transport Combination - Zone															
	3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
	Interoffice Transport - Dedicated - DS3 combination - Per Mile Per Month			UNC3X	1L5XX	6.42										Ĭ
	Interoffice Transport - Dedicated - DS3 - Facility Termination per			CHOOK	120701	0.42										
	month			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				
	DS3 to DS1 Channel System combination per month			UNC3X	MQ3 UC1D1	144.02	178.54	94.18	33.33	31.90		15.69				
	DS3 Interface Unit (DS1 COCI) combination per month Additional DS1Loop in DS3 Interoffice Transport Combination -			UNC1X	UCIDI	8.64	6.59	4.73				15.69				
	Zone 1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
	Additional DS1Loop in DS3 Interoffice Transport Combination -															
	Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				!
	Additional DS1Loop in DS3 Interoffice Transport Combination - Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				ĺ
	DS3 Interface Unit (DS1 COCI) combination per month		Ť	UNC1X	UC1D1	8.64	6.59	4.73	100			15.69				
	Nonrecurring Currently Combined Network Elements Switch -As-	-														
2.WIDE	Is Charge VOICE GRADE EXTENDED LOOP/ 2 WIRE VOICE GRADE IN	TERNER	ICE TE	UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
Z-WIKL	2-WireVG Loop used with 2-wire VG Interoffice Transport	LICOLI	ICE III	ANOI ON (LLL)	1											
	Combination - Zone 1		1	UNCVX	UEAL2	16.68	105.98	68.43	53.05	10.61		15.69				
	2-WireVG Loop used with 2-wire VG Interoffice Transport Combination - Zone 2		2	UNCVX	UEAL2	23.13	105.98	68.43	53.05	10.61		15.00				
 	2-WireVG Loop used with 2-wire VG Interoffice Transport			DINCVA	UEALZ	23.13	105.98	08.43	53.05	10.01		15.69				
	Combination - Zone 3		3	UNCVX	UEAL2	28.46	105.98	68.43	53.05	10.61		15.69				
	Interoffice Transport - Dedicated - 2-wire VG combination - Per															
	Mile Per Month Interoffice Transport - Dedicated - 2- Wire Voice Grade	1	-	UNCVX	1L5XX	0.0134										
	combination - Facility Termination per month			UNCVX	U1TV2	19.44	40.63	27.47	16.77	6.91		15.69				1
	Nonrecurring Currently Combined Network Elements Switch -As-	-														
4 14/15/5	Is Charge		105.75	UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				1
4-WIRE	VOICE GRADE EXTENDED LOOP/ 4 WIRE VOICE GRADE IN 4-WireVG Loop used with 4-wire VG Interoffice Transport	LEKUFF	ICE IR	ANSPURI (EEL)	+											
	Combination - Zone 1		1	UNCVX	UEAL4	32.59	132.38	94.83	59.35	14.61		15.69				1
	4-WireVG Loop used with 4-wire VG Interoffice Transport															
	Combination - Zone 2	1	2	UNCVX	UEAL4	43.89	132.38	94.83	59.35	14.61		15.69				
	4-WireVG Loop used with 4-wire VG Interoffice Transport Combination - Zone 3		3	UNCVX	UEAL4	43.38	132.38	94.83	59.35	14.61		15.69				1
	1	1			,, ,	.0.00	.02.00	200	00.00		·	.0.00	1	l		

CATEGORY	TWORK ELEMENTS - South Carolina												Attachment:		Table 1	
	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	TES(\$)			Svc Order Submitted Elec per LSR	Submitted	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
Mile Po	office Transport - Dedicated - 4-wire VG combination - Per Per Month			UNCVX	1L5XX	0.0134										
combin	office Transport - Dedicated - 4- Wire Voice Grade ination - Facility Termination per month			UNCVX	U1TV4	17.03	40.63	27.47	16.77	6.91		15.69				<u> </u>
ls Cha				UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				<u> </u>
	EXTENDED LOOP WITH DEDICATED DS3 INTEROFFIC	E TRAI	NSPOR	T (EEL)												
	Capacity Unbundled Local Loop - DS3 combination - Per per month			UNC3X	1L5ND	12.26										l
High C	Capacity Unbundled Local Loop - DS3 combination -															
	ty Termination per month			UNC3X	UE3PX	306.36	452.52	264.53	119.75	83.77		15.69				
	office Transport - Dedicated - DS3 - Per Mile per month office Transport - Dedicated - DS3 combination - Facility	1		UNC3X	1L5XX	6.42										
Termin	ination per per month			UNC3X	U1TF3	704.52	279.37	163.12	60.33	58.59		15.69				<u> </u>
ls Cha				UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				İ
	L EXTENDED LOOP WITH DEDICATED STS1 INTEROFF	FICE TR	ANSPO	ORT (EEL)												
	Capacity Unbundled Local Loop - STS1 combination - Per per month			UNCSX	1L5ND	12.26										
	Capacity Unbundled Local Loop - STS1 combination - ty Termination per month			UNCSX	UDLS1	313.49	452.52	264.53	119.75	83.77		15.69				1
	ffice Transport - Dedicated - STS1 combination - Per Mile			UNCSX	1L5XX	6.42	402.02	204.33	113.73	03.77		13.09				
Interof	office Transport - Dedicated - STS1 combination - Facility															
	ination per month ecurring Currently Combined Network Elements Switch -As-			UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				
ls Cha	arge			UNCSX	UNCCC		5.61	5.61	7.00	7.00		15.69				
	EXTENDED LOOP WITH DS1 INTEROFFICE TRANSPOR	RT (EEL)													
Transp	2-Wire ISDN Loop in a DS1 Interoffice Combination sport - Zone 1		1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
Transp	2-Wire ISDN Loop in a DS1 Interoffice Combination sport - Zone 2		2	UNCNX	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
	2-Wire ISDN Loop in a DS1 Interoffice Combination															1
	sport - Zone 3 office Transport - Dedicated - DS1 combination - Per Mile		3	UNCNX UNC1X	U1L2X 1L5XX	37.70 0.27	117.58	80.03	53.05	10.61		15.69				
	office Transport - Dedicated - DS1 combination - Facility			ONOTA	ILOXX	0.21										
Termin	ination per month			UNC1X	U1TF1	61.71	89.47	81.99	16.39	14.48		15.69				
per mo				UNC1X	MQ1	107.57	91.24	62.71	10.56	9.81		15.69				
combir	e ISDN COCI (BRITE) - DS1 to DS0 Channel System ination - per month			UNCNX	UC1CA	2.56	6.59	4.73				15.69				<u></u> _
	ional 2-wire ISDN Loop in same DS1Interoffice Transport		1	UNCNX	U1L2X	25.21	117.58	80.03	53.05	10.61		15.69				
Additio	ional 2-wire ISDN Loop in same DS1Interoffice Transport		2	UNCNX	U1L2X	32.76	117.58	80.03	53.05	10.61		15.69				
Additio	ional 2-wire ISDN Loop in same DS1Interoffice Transport		3	UNCNX	U1L2X	37.70	117.58	80.03	53.05	10.61		15.69				
2-wire	pination - Zone 3 e ISDN COCI (BRITE) - DS1 to DS0 Channel System		3						53.05	10.01						
	intaion- per month ecurring Currently Combined Network Elements Switch -As-			UNCNX	UC1CA	2.56	6.59	4.73				15.69				
ls Cha		TEROF	ICE T	UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
	DS1 Loop in STS1 Interoffice Transport Combination -	IEKUH	-IUE II	KANSPUKI (EEL)	1											
Zone 1	1		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
Zone 2			2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
First D Zone 3	DS1 Loop in STS1 Interoffice Transport Combination - 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				<u> </u>

IINRI	INDI F	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
UNDU	INDLE									1		Svc Order	Svc Order	Incremental			Incremental
												Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
												Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEG	ORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
			m						- (1)			per LSK	per LSK	Electronic-	Electronic-	Electronic-	Electronic-
														1st	Add'l	Disc 1st	Disc Add'l
																DISC 1St	DISC Add I
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Interoffice Transport - Dedicated - STS1 combination - Per Mile															
		Per Month			UNCSX	1L5XX	6.42										
		Interoffice Transport - Dedicated - STS1 combination - Facility															
		Termination			UNCSX	U1TFS	704.44	279.37	163.12	60.33	58.59		15.69				
		STS1 to DS1 Channel System conbination per month			UNCSX	MQ3	144.02	178.54	94.18	33.33	31.90		15.69				
	<u> </u>	DS3 Interface Unit (DS1 COCI) combination per month		<u> </u>	UNC1X	UC1D1	8.64	6.59	4.73				15.69				
		Additional DS1Loop in STS1 Interoffice Transport Combination -		1	LINIOAN	1101 207	00.07	050.00	457.00	44.00	44.70		45.00				
-		Zone 1 Additional DS1Loop in STS1 Interoffice Transport Combination -		1	UNC1X	USLXX	90.87	253.03	157.89	44.80	11.73		15.69				
		Zone 2		2	UNC1X	USLXX	155.43	253.03	157.89	44.80	11.73		15.69				
	1	Additional DS1Loop in STS1 Interoffice Transport Combination -			OI VO I A	JJLAA	100.43	203.03	137.69	44.00	11./3		15.69		1		
		Zone 3		3	UNC1X	USLXX	261.89	253.03	157.89	44.80	11.73		15.69				
-	†	DS3 Interface Unit (DS1 COCI) combination per month	1		UNC1X	UC1D1	8.64	6.59	4.73	44.50	11.75	 	15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-				1	2.01	2.00									
		Is Charge			UNCSX	UNCCC		5.61	5.61	7.00	7.00		15.69				
	4-WIRE	56 KBPS DIGITAL EXTENDED LOOP WITH 56 KBPS INTERO	FFICE 1	RANSI							,,						
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
		Combination - Zone 1		1	UNCDX	UDL56	29.93	126.66	89.12	59.35	14.61		15.69				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
		Combination - Zone 2		2	UNCDX	UDL56	33.99	126.66	89.12	59.35	14.61		15.69				
		4-wire 56 kbps Loop/4-wire 56 kbps Interoffice Transport															
		Combination - Zone 3		3	UNCDX	UDL56	34.74	126.66	89.12	59.35	14.61		15.69				
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination -															
		Per Mile			UNCDX	1L5XX	0.0134										
		Interoffice Transport - Dedicated - 4-wire 56 kbps combination -			LINODY	LIATOR	40.44	40.00	07.47	40.77	0.04		45.00				
		Facility Termination Nonrecurring Currently Combined Network Elements Switch -As-			UNCDX	U1TD5	13.41	40.63	27.47	16.77	6.91		15.69				
		Is Charge			UNCDX	UNCCC		5.61	5.61	7.00	7.00		15.69				
	4-WIRE	is charge 64 KBPS DIGITAL EXTENDED LOOP WITH 64 KBPS INTEROL	FFICE 1	RANSI		UNCCC		3.01	5.01	7.00	7.00		13.09				
	7 ******	4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport		I	OITT (LLL)	1											
		Combination - Zone 1		1	UNCDX	UDL64	29.93	126.66	89.12	59.35	14.61		15.69				
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport							****	00.00							
		Combination - Zone 2		2	UNCDX	UDL64	33.99	126.66	89.12	59.35	14.61		15.69				
		4-wire 64 kbps Loop/4-wire 64 kbps Interoffice Transport															
		Combination - Zone 3		3	UNCDX	UDL64	34.74	126.66	89.12	59.35	14.61		15.69				
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -									-	1					
		Per Mile			UNCDX	1L5XX	0.0134										
		Interoffice Transport - Dedicated - 4-wire 64 kbps combination -															
	<u> </u>	Facility Termination		<u> </u>	UNCDX	U1TD6	13.41	40.63	27.47	16.77	6.91		15.69		ļ		
		Nonrecurring Currently Combined Network Elements Switch -As-			LINCDY	LINICCO		F 04	F 04	7.00	7.00	1	45.00				
ADDIT	IONAL S	Is Charge IETWORK ELEMENTS		<u> </u>	UNCDX	UNCCC		5.61	5.61	7.00	7.00		15.69		-		
ADDITI		NETWORK ELEMENTS used as a part of a currently combined facility, the non-recurr	na cha	rnes do	not apply but a C	witch As Is a	harge does ann	alv				-	-		1		
-		used as a part of a currently combined facility, the non-recurr								 					1		
-		SynchroNet)		- 11011-1	Julian Stranges a			90 0063 1101	•			 	 				
		curring Currently Combined Network Elements "Switch As Is"	Charge	(One a	pplies to each com	bination)											
	1	Nonrecurring Currently Combined Network Elements Switch -As-		1		,									İ		
		Is Charge - 2 wire/4-Wire VG			UNCVX	UNCCC		5.61	5.61	7.00	7.00		15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-															
		ls Charge - 56/64 kbps			UNCDX	UNCCC		5.61	5.61	7.00	7.00		15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-									-	1					
		Is Charge - DS1			UNC1X	UNCCC		5.61	5.61	7.00	7.00		15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-	1									1					
	<u> </u>	ls Charge - DS3		<u> </u>	UNC3X	UNCCC		5.61	5.61	7.00	7.00		15.69				
		Nonrecurring Currently Combined Network Elements Switch -As-	•		LINCOV	LINICOS		5.01	F 6 :	7.00	7.00		45.00				
<u> </u>	NOTE	Is Charge - STS1	d D-1-	W D00	UNCSX	UNCCC	r month -	5.61	5.61	7.00	7.00		15.69				
-	NOTE:	Local Channel - Dedicated Transport - minimum billing period Local Channel - Dedicated - 2-Wire Voice Grade per month	u - Belo	w D23:	eone month, DS3 an UNCXV	IULDV2	r months 15.33	193.53	33.24	36.72	3.21		15.69		-		
-	}	Local Channel - Dedicated - 2-Wire Voice Grade per month	-		UNCXV	ULDV2	16.54	193.53	33.24	37.19	3.68	1	15.69		1		
ь	<u> </u>	Leoda Ghanner - Dedicated - 4-Wile Voice Grade per month	l	<u> </u>	0110/14	JLD V4	10.54	133.37	33.00	31.19	3.00	l	13.09		l		

ONBONDE	ED NETWORK ELEMENTS - South Carolina				•							,	Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	USOC		RA	ΓES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs.	Charge -	Incremental Charge - Manual Svo Order vs.
		m									per zerk	por Lore	Electronic- 1st	Electronic- Add'l	Electronic- Disc 1st	Electronic- Disc Add'l
						Boo	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC		SOMAN	SOMAN	SOMAN	SOMAN
	Local Channel - Dedicated - DS1 per month Zone 1			UNC1X	ULDF1	42.62	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated -DS1 Per Month Zone 2		2	UNC1X	ULDF1	70.32	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated - DS1- Per Month Zone 3		3	UNC1X	ULDF1	190.68	177.87	154.06	22.24	15.30		15.69				
	Local Channel - Dedicated - DS3 - Per Mile per month			UNC3X	1L5NC	11.93										
	Local Channel - Dedicated - DS3 - Facility Termination per				550							4= 00				
	month			UNC3X	ULDF3 1L5NC	446.00	452.52	264.53	119.75	83.77		15.69				
	Local Channel - Dedicated - STS-1- Per Mile per month Local Channel - Dedicated - STS-1 - Facility Termination per			UNCSX	1L5NC	11.93										
	month			UNCSX	ULDFS	435.10	452.52	264.53	119.75	83.77		15.69				
LINDIINDI EI	D LOCAL EXCHANGE SWITCHING(PORTS)			UNCOA	OLDI 3	433.10	432.32	204.33	119.73	03.77		13.09				
	lange Ports				+											
	E: Although the Port Rate includes all available features in GA,	KY. LA	L TN. t	he desired features	will need to h	e ordered usin	g retail USOCs	<u> </u>	†		 				1	1
	RE VOICE GRADE LINE PORT RATES (RES)	I .,, . ,	, .		1		J	-								
	Exchange Ports - 2-Wire Analog Line Port- Res.			UEPSR	UEPRL	1.65	2.38	2.28	1.42	1.33		15.69		İ		Ì
	3				1				i	,,,				İ		
	Exchange Ports - 2-Wire Analog Line Port with Caller ID - Res.			UEPSR	UEPRC	1.65	2.38	2.28	1.42	1.33		15.69				
<u> </u>	Exchange Ports - 2-Wire Analog Line Port outgoing only - Res.		<u> </u>	UEPSR	UEPRO	1.65	2.38	2.28	1.42	1.33	<u> </u>	15.69		<u> </u>		
	Exchange Ports - 2-Wire VG unbundled SC extended local															
	dialing parity Port with Caller ID - Res.			UEPSR	UEPAU	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled South Carolina Area															
	Calling port with Caller ID - Res (LW8)			UEPSR	UEPAJ	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled res, low usage line port															
	with Caller ID (LUM)			UEPSR	UEPAP	1.65	2.38	2.28	1.42	1.33		15.69				
	Subsequent Activity			UEPSR	USASC	0.00	0.00	0.00				15.69				1
FEA	TURES		<u> </u>	LIEDOD	LUED) =							7= 00			ļ	
	All Available Vertical Features		<u> </u>	UEPSR	UEPVF	3.04	0.00	0.00				15.69			ļ	
2-WI	RE VOICE GRADE LINE PORT RATES (BUS)		-		+									-	1	+
	Exchange Ports - 2-Wire Analog Line Port without Caller ID -			UEPSB	UEPBL	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled Line Port with	1	 	ULFOD	UEFBL	1.05	2.38	2.28	1.42	1.33		15.09			1	
	unbundled port with Caller+E484 ID - Bus.		1	UEPSB	UEPBC	1.65	2.38	2.28	1.42	1.33	1	15.69				
H	and and our with Caller + L + 04 ID - Dus.			021 00	OLI BO	1.00	2.30	2.20	1.42	1.33		13.08		1	1	1
	Exchange Ports - 2-Wire Analog Line Port outgoing only - Bus.			UEPSB	UEPBO	1.65	2.38	2.28	1.42	1.33		15.69				
	Exchange Ports - 2-Wire VG unbundled SC extended local				152.50	1.00	2.00	2.20	1.72	1.00	 	10.00			1	1
	dialing parity Port with Caller ID - Bus.		1	UEPSB	UEPAZ	1.65	2.38	2.28	1.42	1.33	1	15.69				
	Exhange Ports - 2-Wire VG unbundled incoming only port with			- ::	1		2.00	2.20	2	50		.0.00		İ		Ì
	Caller ID - Bus		l	UEPSB	UEPB1	1.65	2.38	2.28	1.42	1.33	1	15.69				
	Exchange Ports - 2-Wire VG unbundled South Carolina Bus															
	Area Calling Port with Caller ID - Bus (LMB)			UEPSB	UEPAB	1.65	2.38	2.28	1.42	1.33		15.69		<u> </u>		
	Subsequent Activity			UEPSB	USASC	0.00	0.00	0.00				15.69	_			
FEA	TURES															
	All Available Vertical Features			UEPSB	UEPVF	3.04	0.00	0.00				15.69				
	All Available Vertical Features				UEPVF	3.04	0.00	0.00				15.69				
EXC	HANGE PORT RATES (DID & PBX)		<u> </u>	LIEBOE	Luene -							,			ļ	
 	2-Wire VG Unbundled 2-Way PBX Trunk - Res		<u> </u>	UEPSE	UEPRD	1.65	31.34	14.88	13.97	0.90		15.69			ļ	ļ
 	2-Wire VG Line Side Unbundled 2-Way PBX Trunk - Bus		<u> </u>	UEPSP	UEPPC	1.65	31.34	14.88	13.97	0.90		15.69			ļ	ļ
	2-Wire VG Line Side Unbundled Outward PBX Trunk - Bus 2-Wire VG Line Side Unbundled Incoming PBX Trunk - Bus		 	UEPSP UEPSP	UEPPO UEPP1	1.65	31.34	14.88	13.97 13.97	0.90 0.90		15.69		-	1	1
 	2-Wire Analog Long Distance Terminal PBX Trunk - Bus	-	-	UEPSP	UEPLD	1.65 1.65	31.34 31.34	14.88 14.88	13.97	0.90	-	15.69 15.69		-	1	1
	2-Wire Voice Unbundled PBX LD Terminal Ports			UEPSP	UEPLD	1.65	31.34	14.88	13.97	0.90		15.69			1	1
 	2-Wire Vice Unbundled 2-Way PBX Usage Port		 	UEPSP	UEPXA	1.65	31.34	14.88	13.97	0.90		15.69			1	1
 	2-Wire Voice Unburidled 2-Way PBX Osage Port 2-Wire Voice Unburidled PBX Toll Terminal Hotel Ports			UEPSP	UEPXA	1.65	31.34	14.88	13.97	0.90		15.69		1	1	1
	2-Wire Voice Unbundled PBX LD DDD Terminals Port	-		UEPSP	UEPXC	1.65	31.34	14.88	13.97	0.90	 	15.69			1	1
	2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPSP	UEPXD	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD			UEPSP	UEPXE	1.65	31.34	14.88		0.90		15.69				
	Capable Port 2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy			UEPSP	UEPAE	1.65	31.34	14.88	13.97	0.90		15.69		 	ļ	1

UNBUNI	DLED NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
											Svc Order	Svc Order	Incremental			Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec	Manually	Manual Svc	Manual Svc		Manual Svc
CATEGOR	RY RATE ELEMENTS	Interi	Zone	BCS	USOC		RAT	TES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
		m						- ()			per LSK	per Lon	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						_	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	1	··
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy															
	Room Calling Port			UEPSP	UEPXM	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital															
	Discount Room Calling Port			UEPSP	UEPXO	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port			UEPSP	UEPXS	1.65	31.34	14.88	13.97	0.90		15.69				
	2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus															
	Calling Port			UEPSP	UEPXT	1.65	31.34	14.88	13.97	0.90		15.69				
	Subsequent Activity			UEPSP	USASC	0.00	0.00	0.00				15.69				
FE	EATURES															
	All Available Vertical Features			UEPSP UEPSE	UEPVF	3.04	0.00	0.00				15.69				
EX	XCHANGE PORT RATES (COIN)				1											
	Exchange Ports - Coin Port					1.65	2.38	2.28	1.42	1.33		15.69				
	ocal Switching Features offered with Port				1											
NO	IOTE: Transmission/usage charges associated with POTS circuit	switched	usage	will also apply to o	ircuit switche	ed voice and/or	circuit switche	ed data transm	ission by B-Ch	annels associ	ated with 2	wire ISDN p	orts.			
	IOTE: Access to B Channel or D Channel Packet capabilities will													s Request Pro	ocess.	
UNBUNDL	LED LOCAL EXCHANGE SWITCHING(PORTS)															
E)	XCHANGE PORT RATES (DID & PBX)															
	Exchange Ports - 2-Wire DID Port			UEPEX	UEPP2	8.86	119.57	18.78	60.03	3.77		15.69				
	Exchange Ports - DDITS Port - 4-Wire DS1 Port with DID															
	capability			UEPDD	UEPDD	73.62	202.47	95.90	72.75	2.47		15.69				
	Exchange Ports - 2-Wire ISDN Port (See Notes below.)			UEPTX UEPSX	U1PMA	13.38	72.93	53.11	47.90	10.76		15.69				
	All Features Offered			UEPTX UEPSX	UEPVF	3.04	0.00	0.00								
NO	IOTE: Transmission/usage charges associated with POTS circuit	switched	usage	will also apply to o	ircuit switche	ed voice and/or	circuit switche	ed data transm	ission by B-Ch	annels associ	iated with 2	wire ISDN p	orts.			
NO	IOTE: Access to B Channel or D Channel Packet capabilities will	be availa	ble onl	y through BFR/New	Business Re	quest Process.	Rates for the	packet capabi	lities will be de	termined via t	he Bona Fi	de Request/l	New Business	Request Pro	ocess.	
	Exchange Ports - 2-Wire ISDN Port Channel Profiles			UEPTX UEPSX	U1UMA	0.00	0.00	0.00								
	Exchange Ports - 4-Wire ISDN DS1 Port			UEPEX	UEPEX	107.44	204.27	101.78	79.35	20.10		15.69				
UNBUNDL	LED LOCAL SWITCHING, PORT USAGE															
Er	nd Office Switching (Port Usage)															
	End Office Switching Function, Per MOU					0.0010519										
	End Office Trunk Port - Shared, Per MOU					0.0002136										
Ta	andem Switching (Port Usage) (Local or Access Tandem)															
	Tandem Switching Function Per MOU					0.0001634										
	Tandem Trunk Port - Shared, Per MOU					0.0002863										
Co	Common Transport															
	Common Transport - Per Mile, Per MOU					0.0000045										
	Common Transport - Facilities Termination Per MOU					0.0004095										
	LED PORT/LOOP COMBINATIONS - COST BASED RATES															
	ost Based Rates are applied where BellSouth is required by FCC															
	eatures shall apply to the Unbundled Port/Loop Combination - C															
Er	nd Office and Tandem Switching Usage and Common Transport	Usage ra	tes in t	he Port section of the	nis rate exhib	it shall apply to	all combination	ons of loop/po	rt network elen	nents except	for UNE Co	n Port/Loop	Combination	ns.		
	or Georgia, Kentucky, Louisiana, MIssissippi, South Carolina and															
	Currently Combined Combos for all states. In GA, KY, LA, MS, SC								and NC these	nonrecurring	charges are	Market Rat	es and are als	so listed in th	ie Market Rate	e section.
	or Currently Combined Combos in all other states, the nonrecurr	ng charg	ges sha	III be those identifie	d in the Nonr	ecurring - Curre	ently Combined	d sections.								
	-WIRE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)															
UI	INE Port/Loop Combination Rates															
	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										1
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										1
UI	INE Loop Rates		1													1
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPRX	UEPLX	13.76										1
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPRX	UEPLX	20.38										1
	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPRX	UEPLX	26.04								ļ		
2-'	-Wire Voice Grade Line Port Rates (Res)															
2-1	-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence			UEPRX	UEPRL	1.13	37.93	16.72				15.69				
2-1	-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.13	37.93	16.72				15.69				
2-	-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res 2-Wire voice unbundled port outgoing only - res				-											
2-1	-Wire Voice Grade Line Port Rates (Res) 2-Wire voice unbundled port - residence 2-Wire voice unbundled port with Caller ID - res			UEPRX	UEPRC	1.13	37.93	16.72				15.69				

04/12/02 Page 19 of 30

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	ı
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RAT	TES(\$)						Incremental Charge -		Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						_ 1	Nonrec	urring	Nonrecurring	Disconnect		1	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	2-Wire voice unbundled South Carolina Area Calling port with															
	Caller ID - res (LW8)			UEPRX	UEPAJ	1.13	37.93	16.72				15.69				
	2-Wire voice unbundles res, low usage line port with Caller ID															
	(LUM)			UEPRX	UEPAP	1.13	37.93	16.72				15.69				 '
FEATU				LIEBBY .		2.24						4= 00				
LOCAL	All Features Offered NUMBER PORTABILITY			UEPRX	UEPVF	3.04	0.00	0.00				15.69				
LOCAL	Local Number Portability (1 per port)			UEPRX	LNPCX	0.35										\vdash
NONR	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLI IOX	LIVI OX	0.33										-
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPRX	USAC2		0.10	0.10				15.69				İ
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch with change		<u> </u>	UEPRX	USACC		0.10	0.10				15.69				1
ADDIT	IONAL NRCs	<u> </u>	<u> </u>		4											↓
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent			LIEDBY	LICACO	0.00	0.00	0.00				45.00				1
2 WIDE	Activity E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)			UEPRX	USAS2	0.00	0.00	0.00				15.69				
	ort/Loop Combination Rates															
ONLI	2-Wire VG Loop/Port Combo - Zone 1		1			14.89										—
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
UNE L	oop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPBX	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPBX	UEPLX	20.38										
0.140	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPBX	UEPLX	26.04										
2-Wire	Voice Grade Line Port (Bus) 2-Wire voice unbundled port without Caller ID - bus			UEPBX	UEPBL	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled port with Caller + E484 ID - bus			UEPBX	UEPBC	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled port outgoing only - bus			UEPBX	UEPBO	1.13	37.93	16.72				15.69				
	2-Wire voice Grade unbundled South Carolina extended local				-											
	dialing parity port with Caller ID - bus			UEPBX	UEPAZ	1.13	37.93	16.72				15.69				ĺ
	2-Wire voice unbundled incoming only port with Caller ID - Bus			UEPBX	UPEB1	1.13	37.93	16.72				15.69				
	2-Wire voice unbundled South Carolina Bus Area Calling Port															İ
	with Caller ID (LMB)			UEPBX	UEPAB	1.13	37.93	16.72				15.69				
LOCAL	NUMBER PORTABILITY			UEPBX	LNPCX	0.25										
FEATU	Local Number Portability (1 per port)			UEPBX	LNPCX	0.35										
FEATO	All Features Offered			UEPBX	UEPVF	3.04	0.00	0.00				15.69				
NONRE	ECURRING CHARGES (NRCs) - CURRENTLY COMBINED			OLI DA	OLI VI	0.04	0.00	0.00				10.00				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															
	Switch-as-is			UEPBX	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion -															1
ļ	Switch with change			UEPBX	USACC		0.10	0.10				15.69				
ADDIT	IONAL NRCs															
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent Activity			UEPBX	USAS2		0.00	0.00				15.69				İ
2-WIDE	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES - PBX)			UEPBX	USAS2		0.00	0.00				15.69				
	ort/Loop Combination Rates															-
15.42	2-Wire VG Loop/Port Combo - Zone 1		1		1	14.89									1	
	2-Wire VG Loop/Port Combo - Zone 2		2			21.52										
	2-Wire VG Loop/Port Combo - Zone 3		3			27.17		•								
UNE L	pop Rates															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPRG	UEPLX	13.76								ļ	ļ	<u> </u>
—	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPRG	UEPLX	20.38			-	-				1	 	
2/4:	2-Wire Voice Grade Loop (SL 1) - Zone 3 Voice Grade Line Port Rates (RES - PBX)		3	UEPRG	UEPLX	26.04									-	
Z-Wire	2-Wire VG Unbundled Combination 2-Way PBX Trunk Port -	l -			+									1		
	Res			UEPRG	UEPRD	1.13	37.93	16.72				15.69				1
LOCAL	NUMBER PORTABILITY		 			0	350	.5.72				.0.50				
			1	l	1	l l	l		I	I	1		1	1	1	

IINDII	וחו בי	NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 4	1
ONBUN	ADLEL	NET WORK ELEMENTS - SOUTH CAROLINA	1							1	1	Svc Order	Svc Order	Attachment: Incremental		Table 1 Incremental	Incremental
			1									Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
												Elec					Manual Svc
CATEGO	DRY	RATE ELEMENTS	Interi	Zone	BCS	usoc		RAT	TES(\$)					Manual Svc			
OAT LOC	,,,,	NATE ELEMENTO	m	20.10	500	0000		TOA!	- Ε Θ (ψ)			per LSR	per LSR	Order vs.	Order vs. Electronic-	Order vs.	Order vs. Electronic-
														Electronic-		Electronic-	
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonrec	urring	Nonrecurring	Disconnect			oss	Rates(\$)	•	
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Local Number Portability (1 per port)			UEPRG	LNPCP	3.15	0.00	0.00				15.69				
F	EATU																
		All Features Offered			UEPRG	UEPVF	3.04	0.00	0.00				15.69				
1		CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -															
		Conversion - Switch-As-Is			UEPRG	USAC2		7.93	1.91				15.69				
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			LIEDDO	110400		7.00	4.04				45.00				
<u> </u>		Conversion - Switch with Change			UEPRG	USACC		7.93	1.91				15.69				
		DNAL NRCs 2-Wire Voice Grade Loop/ Line Port Combination (PBX) -				+											
		Subsequent Activity			UEPRG	USAS2	0.00	0.00	0.00				15.69				
\vdash		PBX Subsequent Activity - Change/Rearrange Multiline Hunt	 		OLFING	UUAUZ	0.00	0.00	0.00	 			15.69		1	1	1
		Group						7.34	7.34				15.69				
	2-WIRF	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS - PBX)	1	1		+ +		7.04	7.04				10.00				
		rt/Loop Combination Rates	†			1 1							 		1	 	
	1	2-Wire VG Loop/Port Combo - Zone 1		1		1	14.89										
		2-Wire VG Loop/Port Combo - Zone 2		2		1	21.52										
		2-Wire VG Loop/Port Combo - Zone 3		3			27.17										
U		op Rates															
		2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEPPX	UEPLX	13.76										
		2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEPPX	UEPLX	20.38										
		2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEPPX	UEPLX	26.04										
2	2-Wire	/oice Grade Line Port Rates (BUS - PBX)															
		Line Side Unbundled Combination 2-Way PBX Trunk Port - Bus			UEPPX	UEPPC	1.13	37.93	16.72				15.69				
		Line Side Unbundled Outward PBX Trunk Port - Bus			UEPPX	UEPPO	1.13	37.93	16.72				15.69				
		Line Side Unbundled Incoming PBX Trunk Port - Bus			UEPPX	UEPP1	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled PBX LD Terminal Ports			UEPPX	UEPLD	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled 2-Way Combination PBX Usage Port			UEPPX	UEPXA	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled PBX Toll Terminal Hotel Ports			UEPPX	UEPXB	1.13	37.93	16.72				15.69				
-		2-Wire Voice Unbundled PBX LD DDD Terminals Port			UEPPX	UEPXC	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard Port			UEPPX	UEPXD	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled PBX LD Terminal Switchboard IDD Capable Port			UEPPX	UEPXE	1.13	37.93	16.72				15.69				
-					UEPPX	UEPAE	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy Administrative Calling Port	1		UEPPX	UEPXL	1.13	37.93	16.72				15.69			1	
\vdash		2-Wire Voice Unbundled 2-Way PBX Hotel/Hospital Economy	 		OLFFA	JLFAL	1.13	31.83	10.72	 			15.69		1	1	1
		Room Calling Port			UEPPX	UEPXM	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Hotel/Hospital	1	1	5 <u>-</u> . 1 /	CEI /WI	1.10	07.33	10.72				10.00				
		Discount Room Calling Port			UEPPX	UEPXO	1.13	37.93	16.72				15.69				
		2-Wire Voice Unbundled 1-Way Outgoing PBX Measured Port	<u> </u>		UEPPX	UEPXS	1.13	37.93	16.72				15.69			1	
		2-Wire Voice Unbundled 2-Way PBX South Carolina Area Plus				1	0	21.00								İ	
		Calling Port			UEPPX	UEPXT	1.13	37.93	16.72				15.69				
	OCAL	NUMBER PORTABILITY				1 1											
		Local Number Portability (1 per port)			UEPPX	LNPCP	3.15	0.00	0.00				15.69				
F	EATU																
		All Features Offered			UEPPX	UEPVF	3.04	0.00	0.00				15.69				
	NONRE	CURRING CHARGES (NRCs) - CURRENTLY COMBINED															
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -						_]	
		Conversion - Switch-As-Is			UEPPX	USAC2		7.93	1.91				15.69				
1 T		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -	1			1 7							1]	
		Conversion - Switch with Change	<u> </u>		UEPPX	USACC		7.93	1.91				15.69				
/	ADDITIO	ONAL NRCs				 									ļ		
		2-Wire Voice Grade Loop/ Line Port Combination (PBX) -			LIEDDY	110466							4				
\vdash		Subsequent Activity	<u> </u>	_	UEPPX	USAS2	0.00	0.00	0.00				15.69				
		PBX Subsequent Activity - Change/Rearrange Multiline Hunt	1					701	7.01				45.00			1	
 	\ \A/' D.	Group	Ļ			1		7.34	7.34				15.69		ļ	 	
		VOICE GRADE LOOP WITH 2-WIRE ANALOG LINE COIN POR	τ Ι	<u> </u>		1					ļ						1
	JINE PO	rt/Loop Combination Rates	<u> </u>			1				l	l .	1	l		l	L	l

ONBONDL	ED NETWORK ELEMENTS - South Carolina			1							_	_	Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RAT	res(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Charge -	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
													1st		DISC 1St	DISC Add I
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Coin Port/Loop Combo – Zone 1		1			14.89										
	2-Wire VG Coin Port/Loop Combo – Zone 2		2			21.52										
	2-Wire VG Coin Port/Loop Combo – Zone 3		3			27.17										
UNE	Loop Rates															
	2-Wire Voice Grade Loop (SL1) - Zone 1		1	UEPCO	UEPLX	13.76										
	2-Wire Voice Grade Loop (SL1) - Zone 2		2	UEPCO	UEPLX	20.38										
0.140	2-Wire Voice Grade Loop (SL1) - Zone 3		3	UEPCO	UEPLX	26.04										
2-Wir	e Voice Grade Line Ports (COIN)															
	2-Wire Coin 2-Way without Operator Screening and without Blocking (SC)			UEPCO	UEPSD	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSA	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSH	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and 011 Blocking; with Dialing Parity (SC)			UEPCO	UEPSC	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-Way with Operator Screening and: 900 Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCC	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced Call OPT 3YV (SC)			UEPCO	UEPCE	1.13	37.93	16.72				15.69				
	2-Wire Coin 2-W Operator Screen: 900 Block: 900/976, 1+DDD, 011+, Local; Enhanced Call OPT AP7 (SC)			UEPCO	UEPCF	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward without Blocking and without Operator Screening (SC)			UEPCO	UEPSG	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and 011 Blocking (SC)			UEPCO	UEPSF	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 011, 900/976, 1+DDD (SC)			UEPCO	UEPSJ	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward with Operator Screening and Blocking: 900/976, 1+DDD, 011+, and Local (SC)			UEPCO	UEPCM	1.13	37.93	16.72				15.69				
	2-Wire Coin Out Operator Screen & Block: 900/976, 1+DDD, 011+, Local; Enhanced Calling OPT 3YW (SC)			UEPCO	UEPCP	1.13	37.93	16.72				15.69				
	2-Wire 2-Way Smartline with 900/976 (all states except LA)			UEPCO	UEPCK	1.13	37.93	16.72				15.69				
	2-Wire Coin Outward Smartline with 900/976 (all states except LA)			UEPCO	UEPCR	1.13	37.93	16.72				15.69				
LOC/	AL NUMBER PORTABILITY	1			32. 3	0	300	.5.72				.0.00			1	1
	Local Number Portability (1 per port)			UEPCO	LNPCX	0.35										
NON	RECURRING CHARGES - CURRENTLY COMBINED	1													İ	1
	2-Wire Voice Grade Loop / Line Port Combination - Conversion Switch-as-is	-		UEPCO	USAC2		0.10	0.10				15.69				
	2-Wire Voice Grade Loop / Line Port Combination - Conversion Switch with change	-		UEPCO	USACC		0.10	0.10				15.69				
ADDI	TIONAL NRCs	1			122.30		50	5.70				.0.00			1	
	2-Wire Voice Grade Loop/Line Port Combination - Subsequent	1			1									1	t	1
1	Activity	1		UEPCO	USAS2		0.00	0.00				15.69			I	
UNB	JNDLED REMOTE CALL FORWARDING - RES	1			1		2.00	2.00								
Non-l	Recurring	1														
UNBI	JNDLED REMOTE CALL FORWARDING - Bus															
	Unbundled Remote Call Forwarding, InterState/Intra LATA-Bus			UEPVB	UEPVJ	1.65	2.38	2.28	1.42	1.33		15.69				
	Recurring															
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIR															
	RE VOICE LOOP/ 2WIRE VOICE GRADE IO TRANSPORT/ 2-WIR	E LINE I	PORT ((BUS)												
	PORT/LOOP COMBINATIONS - COST BASED RATES															
	RE VOICE GRADE LOOP- BUS ONLY - WITH 2-WIRE DID TRUNI	K PORT			4										ļ	
UNE	Port/Loop Combination Rates 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 1	 	<u> </u>													
			1	1	1	23.75			1		l	i l		i	1	1
		_		1					1							
	2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 2 2-Wire VG Loop/2-Wire DID Trunk Port Combo - UNE Zone 3		2			30.20 35.52										

ONBONDLE	D NETWORK ELEMENTS - South Carolina													Attachment:		Table 1	└
CATEGORY	RATE ELEMENTS	Interi m	Zone	E	cs	USOC		RAT	TES(\$)				Manually	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Charge -	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic- Disc Add'l
							_	Nonrec	urring	Nonrecurring	Disconnect			OSS	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 1		1	UEPPX		UECD1	16.68										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 2		2	UEPPX		UECD1	23.13										
	2-Wire Analog Voice Grade Loop - (SL2) - UNE Zone 3		3	UEPPX		UECD1	28.46										
UNE P	Port Rate					UEDD 4				440.00				4= 00			
NOND	Exchange Ports - 2-Wire DID Port ECURRING CHARGES - CURRENTLY COMBINED			UEPPX		UEPD1	7.06	225.55	87.21	113.08	14.38			15.69			
NONK	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Combination -					1											
	Switch-as-is			UEPPX		USAC1		7.32	1.87					15.69			
	2-Wire Voice Grade Loop / 2-Wire DID Trunk Port Conversion			OL. IX		00/101		7.02						10.00			
	with BellSouth Allowable Changes			UEPPX		USA1C		7.32	1.87					15.69			
ADDIT	TONAL NRCs																
	2-Wire DID Subsequent Activity - Add Trunks, Per Trunk			UEPPX		USAS1		26.84	•		•			15.69			
Telepi	none Number/Trunk Group Establisment Charges					L											
	DID Trunk Termination (One Per Port)			UEPPX		NDT	0.00	0.00	0.00					15.69			
1	DID Numbers, Establish Trunk Group and Provide First Group			HEDDY		ND7	0.00	0.00	0.00					45.00			1
	of 20 DID Numbers Additional DID Numbers for each Group of 20 DID Numbers	1		UEPPX		NDZ ND4	0.00	0.00	0.00	-				15.69 15.69			
	DID Numbers, Non- consecutive DID Numbers, Per Number			UEPPX		ND5	0.00	0.00	0.00					15.69			
	Reserve Non-Consecutive DID numbers			UEPPX		ND6	0.00	0.00	0.00					15.69			
	Reserve DID Numbers			UEPPX		NDV	0.00	0.00	0.00					15.69			
LOCA	L NUMBER PORTABILITY					1		0.00									
	Local Number Portability (1 per port)			UEPPX		LNPCP	3.15	0.00	0.00								
	E ISDN DIGITAL GRADE LOOP WITH 2-WIRE ISDN DIGITAL LI	NE SIDE	PORT														
UNE P	ort/Loop Combination Rates																
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 1		1	UEPPB	UEPPR		30.86										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port - UNE Zone 2		2	UEPPB	UEPPR		38.60										
	2W ISDN Digital Grade Loop/2W ISDN Digital Line Side Port -																
I NIE I	UNE Zone 3		3	UEPPB	UEPPR		44.23										
UNE L	.oop Rates 2-Wire ISDN Digital Grade Loop - UNE Zone 1		1	UEPPB	UEPPR	USL2X	21.90							15.69			
	2-Wife ISDN Digital Grade Loop - ONE Zorie I		-	UEPPB	UEPPR	USLZA	21.90							15.69			
	2-Wire ISDN Digital Grade Loop - UNE Zone 2		2	UEPPB	UEPPR	USL2X	29.64							15.69			
	2-Wire ISDN Digital Grade Loop - UNE Zone 3		3	UEPPB	UEPPR		35.27							15.69			
UNE P	Port Rate																
	Exchange Port - 2-Wire ISDN Line Side Port			UEPPB	UEPPR	UEPPB	8.96	190.51	133.14	100.95	21.37			15.69			
NONR	ECURRING CHARGES - CURRENTLY COMBINED																
	2-Wire ISDN Digital Grade Loop / 2-Wire ISDN Line Side Port					[]											1
	Combination - Conversion			UEPPB	UEPPR	USACB	0.00	38.59	27.08					15.69			
	TONAL NRCs																
LUCA	L NUMBER PORTABILITY Local Number Portability (1 per port)			UEPPB	UEPPR	LNPCX	0.35	0.00	0.00								
B-CH4	ANNEL USER PROFILE ACCESS:			UEFFB	UEPPR	LINPUX	0.35	0.00	0.00								
B 0117	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCA	0.00	0.00	0.00								
	CVS (EWSD)			UEPPB	UEPPR	U1UCB	0.00	0.00	0.00								
	CSD			UEPPB	UEPPR	U1UCC	0.00	0.00	0.00								
B-CHA	ANNEL AREA PLUS USER PROFILE ACCESS: (AL,KY,LA,MS S	C,MS, &	TN)				_										
	CVS/CSD (DMS/5ESS)			UEPPB	UEPPR	U1UCD	0.00	0.00	0.00		•						
	CVS (EWSD)			UEPPB	UEPPR	U1UCE	0.00	0.00	0.00								
	CSD TERMINAL PROFILE			UEPPB	UEPPR	U1UCF	0.00	0.00	0.00								
USER	TERMINAL PROFILE			HEDDD	UEPPR	1.141.1844	0.00	0.00	0.00						-	-	
VEDT	User Terminal Profile (EWSD only)	1		UEPPB	UEPPR	U1UMA	0.00	0.00	0.00								
VERII	All Vertical Features - One per Channel B User Profile	1		UEPPB	UEPPR	UEPVF	3.04	0.00	0.00			1	1	15.69	1	1	
INTER	OFFICE CHANNEL MILEAGE			OLI FD	OLFFR	JLI VI	3.04	0.00	0.00					13.09			
	Interoffice Channel mileage each, including first mile and						İ										
	facilities termination			UEPPB	UEPPR	M1GNC	24.30	40.63	27.47	16.77	6.91			15.69			1
	Interoffice Channel mileage each, additional mile			UEPPB	UEPPR	M1GNM	0.0167	0.00	0.00	1							

LINBLINDI	.ED NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
ONBONDE	LED NETWORK ELEMENTS - South Carolina	1		1	1					I	Svc Order		Incremental			Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec		Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interi	Zone	BCS	usoc		RAT	TES(\$)								
OATEGORT	NATE ELEMENTO	m		500	0000		TOA.	Δ (ψ)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
													Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
-						1	Nonrec	urring	Nonrecurring	n Disconnect		l .	220	Rates(\$)	l	
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
4-10//	RE DS1 DIGITAL LOOP WITH 4-WIRE ISDN DS1 DIGITAL TRUN	K DODT					FIISL	Auu i	FIISL	Auu i	SOWIEC	JOWAN	JOWAN	JOWAN	JOWAN	SOWAN
	Port/Loop Combination Rates	KFOKI														
ONL	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE				-	+										
	Zone 1		1	UEPPP		176.82										
\vdash	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE		-	ULFFF	-	170.02										
	Zone 2		2	UEPPP		241.38										
\vdash	4W DS1 Digital Loop/4W ISDN DS1 Digital Trunk Port - UNE			UEPPP	-	241.30										
	Zone 3		3	UEPPP		347.84										
LINE	Loop Rates		3	UEPPP		347.04										
UNE	4-Wire DS1 Digital Loop - UNE Zone 1		1	UEPPP	USL4P	90.87							15.69			
\vdash	4-Wire DS1 Digital Loop - UNE Zone 1		2	UEPPP	USL4P USL4P	155.43							15.69			
\vdash				UEPPP												
- I.I.	4-Wire DS1 Digital Loop - UNE Zone 3		3	UEPPP	USL4P	261.89							15.69			
UNE	Port Rate	 		LIEDDD	HEDES	05.05	457.00	050.67	101.1-	01.00			15.00			
h	Exchange Ports - 4-Wire ISDN DS1 Port	 		UEPPP	UEPPP	85.95	457.30	259.67	124.15	31.83			15.69			
NON	RECURRING CHARGES - CURRENTLY COMBINED	 	<u> </u>		+										1	
	4-Wire DS1 Digital Loop / 4-Wire ISDN DS1 Digital Trunk Port															
	Combination - Conversion -Switch-as-is			UEPPP	USACP	0.00	119.34	78.73					15.69			
ADD	ITIONAL NRCs															
	4-Wire DS1 Loop/4-W ISDN Digtl Trk Port - Subsqt Actvy-															
	Inward/two way tel nos within Std Allowance (except NC)			UEPPP	PR7TF		0.49	0.49					15.69			
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trunk Port -															
	Outward Tel Numbers (All States except NC)			UEPPP	PR7TO		11.54	11.54					15.69			
	4-Wire DS1 Loop / 4-Wire ISDN DS1 Digital Trk Port -															
	Subsequent Inward Tel Nos Above Std Allowance			UEPPP	PR7ZT		23.07	23.07					15.69			
LOC	AL NUMBER PORTABILITY															
	Local Number Portability (1 per port)			UEPPP	LNPCN	1.75										
	Voice/Data			UEPPP	PR71V	0.00	0.00	0.00								
	Digital Data			UEPPP	PR71D	0.00	0.00	0.00								
	Inward Data			UEPPP	PR71E	0.00	0.00	0.00								
New	or Additional "B" Channel															
	New or Additional - Voice/Data B Channel			UEPPP	PR7BV	0.00	14.56						15.69			
	New or Additional - Digital Data B Channel			UEPPP	PR7BF	0.00	14.56						15.69			
	New or Additional Inward Data B Channel			UEPPP	PR7BD	0.00	14.56						15.69			
CAL	L TYPES															
	Inward			UEPPP	PR7C1	0.00	0.00	0.00								
	Outward			UEPPP	PR7C0	0.00	0.00	0.00								
	Two-way			UEPPP	PR7CC	0.00	0.00	0.00								
Inter	office Channel Mileage															
	Fixed Each Including First Mile			UEPPP	1LN1A	77.4815	89.47	81.99	16.39	14.48			15.69			
	Each Airline-Fractional Additional Mile			UEPPP	1LN1B	0.3415										
4-WI	RE DS1 DIGITAL LOOP WITH 4-WIRE DDITS TRUNK PORT	1														
	Port/Loop Combination Rates	1														
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 1	1	1	UEPDC	1	149.77			İ					İ	İ	İ
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 2	1	2	UEPDC	1	214.33			İ		İ	İ		İ	İ	İ
	4W DS1 Digital Loop/4W DDITS Trunk Port - UNE Zone 3	1	3	UEPDC	1	320.78					İ			İ		İ
UNE	Loop Rates	1		-	1				İ		İ	İ		İ	İ	İ
	4-Wire DS1 Digital Loop - UNE Zone 1	1	1	UEPDC	USLDC	90.87			İ				15.69	İ	İ	İ
	4-Wire DS1 Digital Loop - UNE Zone 2	1	2	UEPDC	USLDC	155.43					i	1	15.69	1		1
	4-Wire DS1 Digital Loop - UNE Zone 3	1	3	UEPDC	USLDC	261.89					i		15.69	1		1
UNF	Port Rate	1	t -	-	1						i	1		1		1
	4-Wire DDITS Digital Trunk Port	1		UEPDC	UDD1T	58.90	455.50	253.79	117.55	14.20			15.69	1		1
NON	RECURRING CHARGES - CURRENTLY COMBINED	1				55.55	.00.00	2000	50	20			.0.50	1		1
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1			+	t					-	1		 		
	- Switch-as-is	1		UEPDC	USAC4	l	129.78	67.17				1	15.69	Ì		l
	4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination	1			20,10.	t	.20.70	017			-	1	.0.00	 		
	- Conversion with DS1 Changes			UEPDC	USAWA	l	129.78	67.17				1	15.69	Ì		l
																1
		1		OLI DO	00/11//	+	120.70	01111								
	- Conversion with DS1 Changes 4-Wire DS1 Digital Loop / 4-Wire DDITS Trunk Port Combination - Conversion with Change - Trunk	1		UEPDC	USAWB		129.78	67.17					15.69			

UNBUPLICED NETWORK ELECTRITS - South Carolina ARTE ELEMENTS ARTE	HINDHIND	ED NETWORK ELEMENTS South Carolina												Attack	44	Toble 4	
ARTECON PARTE ELEMENTS PRATE ELEMENTS PRATE PLANT South	UNDUNDL	ED INE I WORK ELEMIEN 19 - SOUTH CAPOLINA	1	ı		1				I		Svo Order	Sup Orde-			Table 1	Increment-1
ATTEMPT OF PART REMEMPS March 2						1											
AFFECRIVE STATE ELEMENTS																	
Name	CATEGORY	PATE ELEMENTS	Interi	Zone	BCS.	usoc		PΔ	r=9/\$)								
Part Part	CATEGORI	RATE ELEMENTS	m	Zone	603	0300		NA.	L3(\$)			per LSR	per LSR				
Proceedings																	
Proceedings Procedure (1987) Conference Content (198														1st	Add'l	Disc 1st	Disc Add'l
Proceedings Procedure (1987) Conference Content (198						+	1	Nonrec	urring	Nonrecurring	n Disconnect			oss	Rates(\$)		
With Cold Large / With COTTS Trans Part - Subsequent USPOC							Rec					SOMEC	SOMAN			SOMAN	SOMAN
Charmer Anderdor/Chart - TWING Octobal Tracks UEPOC UOTTO 14.51 1.51 1.51 1.51 1.50		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsequent						11131	Auu	11100	Addi	COME	COMPAN	COMPAR	COMPAR	COMPAN	COMPAN
### Advisor 1.500					UEPDC	UDTTB		14.51	14.51					15.69			i .
Accordance from two root Down DePoil Depoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoil DePoi					02. 50	00110								10.00			
Affect OR 1.00 of PAPE OF TRUE POTE Study OF Cabon 1.60 of Cabon 1.60					UEPDC	UDTTC		14.51	14.51					15.69			1
Activation For Colons Fearer Frank with DO UPPOC UPP																	
April Apri					UEPDC	UDTTD		14.51	14.51					15.69			i
BIPCLAR 2 ZERO SUBSTITUTION		4-Wire DS1 Loop / 4-Wire DDITS Trunk Port - Subsgnt Chan															
BIPCLAR 2 ZERO SUBSTITUTION		Activation / Chan - 2-Way DID w User Trans			UEPDC	UDTTE		14.51	14.51					15.69			i
BRGS - Extended Superframe Format	BIPC	LAR 8 ZERO SUBSTITUTION												15.69			
Alternate Marks Triversion		B8ZS -Superframe Format			UEPDC	CCOSF		0.00	605.00					15.69			
AM - Superfame Format					UEPDC	CCOEF		0.00	605.00					15.69			
Mail - Extended Support Format UEPPC MCOPO 0.00	Alter																
Telephone Number Trans Group Establishment Charges																	
Telephone Number for 17-Way Transf Group					UEPDC	МСОРО		0.00	0.00								
Telephore Number for 1-Very Cutumer for Fusy Cutumer fo	Tele																
Telephore Number for I-Wej Invasor Trusk Group Affords Frat Group DID Numbers Earlie Number Earlie Num								, and the second									
DID Numbers Establish Trunk Group and Provide First Group UEPDC NDZ 0.00 0.00 0.00 15.50 15.			1														
of 20 DD Numbers for and Group of 20 DD Numbers UEPC N02 0.00 0.00 0.00 15.69					UEPDC	UDTGZ	0.00							15.69			1
DID Numbers for each Group of a DID Numbers DEPDC NOS 0.00 0.00 15.69																	i
DID Numbers, Non-consecutive DID Numbers, Per Number UEPDC NDS 0.00 0.00 0.00 15.69								0.00	0.00								
Reserve Non-Consecutive DID Nos. UEPDC NO6 0.00 0.00 0.00 15.689																	
Decision Dot Numbers Decision St (Interoffice Channel Mileage) - FXFCO for 4-Wire DST Digital Loop with 4-Wire DST Struke Port																	
Desirated DSf (Interoffice Channel Mileage - Fixef rate 0-8 miles (Facilities UEPDC LINO1 77.14 89.47 81.99 16.39 14.48 15.69				<u> </u>													
Interoffice Channel Mieage - Fixed rate 0-8 miles Facilities UEPDC 1LNOA 0.3415 0.00 0	B		1 8''1-				0.00	0.00	0.00					15.69			+
Termination UEPDC	Deal		Digita	Loop	With 4-Wire DDITS	runk Port											
Interoffice Channel Mileage - Additional rate per mile - 0-8 miles UEPDC					LIEDDC	11 NO1	77 14	90.47	94.00	16 20	11 10			15.60			i
Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination) UEPDC 1LNO2 0.00 0.0		Termination)	1		OLFDC	ILINOT	77.14	09.47	01.99	10.35	14.40			13.09			-
Interoffice Channel Mileage - Fixed rate 9-25 miles (Facilities Termination) UEPDC 1LNO2 0.00 0.0		Interoffice Channel Mileage - Additional rate per mile - 0-8 miles			LIEPDC	11 NOA	0.3415	0.00	0.00								i
Termination UEPDC	 				OLI DO	TENTON	0.0410	0.00	0.00								—
Interoffice Channel Mileage - Additional rate per mile - 9-25 UEPDC					LIEPDC	1I NO2	0.00	0.00	0.00								i
Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities UEPDC 1LNOS 0.00 0.0					02. 50	12.102	0.00	0.00	0.00								
Interoffice Channel Mileage - Fixed rate 25+ miles (Facilities Termination)					UEPDC	1LNOB	0.3415	0.00	0.00								i
Interoffice Channel Mileage - Additional rate per mile - 25+ miles UEPDC LINOC 0.3415 0.00																	
Interoffice Channel Mileage - Additional rate per mile - 25+ miles UEPDC 1LNCC 0.3415 0.00					UEPDC	1LNO3	0.00	0.00	0.00								i
Local Number Portability, per DSD Activated UEPDC LNPCP 3.15 0.00 0.00		,			-												ſ
Local Number Portability, per DSD Activated UEPDC LNPCP 3.15 0.00 0.00		Interoffice Channel Mileage - Additional rate per mile - 25+ miles	1		UEPDC	1LNOC	0.3415	0.00	0.00					1	1	1	1
Central Office Termininating Point UEPDC CTG 0.00																	
System is 1 DS1 Loop, 1 D4 Channel Bank, and up to 24 Feature Activations		Central Office Termininating Point	<u> </u>		UEPDC	CTG	0.00										
Each System can have up to 24 combinations of rates depending on type and number of ports used																	
UNE D\$1 Loop - UNE Zone 1																	
4-Wire DS1 Loop - UNE Zone 1			type a	nd num	ber of ports used			, and the second second									
4-Wire DS1 Loop - UNE Zone 2 2 UEPMG USLDC 155.43 0.00 0.0	UNE		<u> </u>														<u> </u>
A-Wire DS1 Loop - UNE Zone 3 3 UEPMG USLDC 261.89 0.00																	
UNE DSO Channelization Capacities (D4 Channel Bank Configurations) UEPMG VUM24 82.78 0.00 0.00 15.69 15.69 24 DSO Channel Capacity - 1 per 2 DS1s UEPMG VUM48 165.56 0.00 0.00 15.69 15.69 96 DSO Channel Capacity - 1 per 4 DS1s UEPMG VUM96 331.12 0.00 0.00 15.69 15.69 144 DSO Channel Capacity - 1 per 6 DS1s UEPMG VUM14 496.68 0.00 0.00 15.69 15.69 192 DSO Channel Capacity - 1 per 8 DS1s UEPMG VUM14 496.68 0.00 0.00 15.69 15.69 240 DSO Channel Capacity - 1 per 8 DS1s UEPMG VUM19 662.24 0.00 0.00 15.69 15.69 288 DSO Channel Capacity - 1 per 10 DS1s UEPMG VUM20 827.80 0.00 0.00 15.69 15.69 384 DSO Channel Capacity - 1 per 12 DS1s UEPMG VUM28 993.36 0.00 0.00 15.69 15.69 480 DSO Channel Capacity - 1 per 20 DS1s UEPMG VUM38 1,324			ļ														
24 DSO Channel Capacity - 1 per DS1	 		Ц	3	UEPMG	USLDC	261.89	0.00	0.00								
48 DSO Channel Capacity - 1 per 2 DS1s	UNE		ns)	ļ	LIEDMO	\ (I I I I I I I I I I I I I I I I I I I	00.70	0.00	0.00					45.00			
96 DSO Channel Capacity -1 per 4 DS1s UEPMG VUM96 331.12 0.00 0.00 15.69 15.69 144 DS0 Channel Capacity -1 per 6 DS1s UEPMG VUM14 496.68 0.00 0.00 0.00 15.69 192 DS0 Channel Capacity -1 per 10 DS1s UEPMG VUM19 662.24 0.00 0.00 0.00 15.69 15.69 16	\vdash		<u> </u>	<u> </u>													+
144 DS0 Channel Capacity - 1 per 6 DS1s	 		1	1						ļ					1	1	
192 DS0 Channel Capacity -1 per 8 DS1s	\vdash		 	 													
240 DS0 Channel Capacity - 1 per 10 DS1s UEPMG VUM20 827.80 0.00 0.00 15.69			 	 											-	 	
288 DS0 Channel Capacity - 1 per 12 DS1s UEPMG VUM28 993.36 0.00 0.00 15.69 384 DS0 Channel Capacity - 1 per 16 DS1s UEPMG VUM38 1,324.48 0.00 0.00 15.69 480 DS0 Channel Capacity - 1 per 20 DS1s UEPMG VUM40 1,655.60 0.00 0.00 15.69 576 DS0 Channel Capacity - 1 per 24 DS1s UEPMG VUM57 1,986.72 0.00 0.00 15.69 672 DS0 Channel Capacity - 1 per 28 DS1s UEPMG VUM67 2,317.84 0.00 0.00 15.69	\vdash		1	 											-	-	
384 DS0 Channel Capacity - 1 per 16 DS1s UEPMG VUM38 1,324.48 0.00 0.00 15.69 15.69 480 DS0 Channel Capacity - 1 per 20 DS1s UEPMG VUM40 1,655.60 0.00 0.00 15.69 15.69 1672 DS0 Channel Capacity - 1 per 24 DS1s UEPMG VUM57 1,986.72 0.00 0.00 15.69 15.69 1672 DS0 Channel Capacity - 1 per 28 DS1s UEPMG VUM67 2,317.84 0.00 0.00 15.69 15.69			 	-											-	-	
480 DS0 Channel Capacity - 1 per 20 DS1s UEPMG VUM40 1,655.60 0.00 0.00 15.69 576 DS0 Channel Capacity - 1 per 24 DS1s UEPMG VUM57 1,986.72 0.00 0.00 15.69 672 DS0 Channel Capacity - 1 per 28 DS1s UEPMG VUM67 2,317.84 0.00 0.00 15.69			 	 											-	 	
576 DS0 Channel Capacity -1 per 24 DS1s UEPMG VUM57 1,986.72 0.00 0.00 15.69 15.69 672 DS0 Channel Capacity - 1 per 28 DS1s UEPMG VUM67 2,317.84 0.00 0.00 15.69 15.69	\vdash		1							1					1	1	
672 DS0 Channel Capacity - 1 per 28 DS1s UEPMG VUM67 2,317.84 0.00 0.00 15.69	 		1							 					1	1	
Non-Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with Channeliztion with Port - Conversion Charge Based on a System	 		1							 					1	1	
	Non-	Recurring Charges (NRC) Associated with 4-Wire DS1 Loop with	h Chani	neliztio					0.00				 	10.00		-	

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	ES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR		Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	imum System configuration is One (1) DS1, One (1) D4 Channe															
Multip	les of this configuration functioning as one are considered Ac	ia i arte	r tne m	inimum system con	ifiguration is	countea.										
	BellSouth Allowed Changes			UEPMG	USAC4	0.00	150.81	8.38					15.69			
Syster	m Additions at End User Locations Where 4-Wire DS1 Loop with	th Chan	nelizat				100.01	0.00					10.00			
	Not Currently Combined) In GA, KY, LA, MS & TN Only															
Bipola	r 8 Zero Substitution															
	Clear Channel Capability Format, superframe - Subsequent															
	Activity Only			UEPMG	CCOSF	0.00	0.00	605.00								
	Clear Channel Capability Format - Extended Superframe - Subsequent Activity Only			UEPMG	CCOEF	0.00	0.00	605.00								
Altern	ate Mark Inversion (AMI)			ULFIVIG	COUEF	0.00	0.00	00.600								
Aitem	Superframe Format			UEPMG	MCOSF	0.00	0.00	0.00								†
	Extended Superframe Format	1		UEPMG	MCOPO	0.00	0.00	0.00								1
Excha	nge Ports Associated with 4-Wire DS1 Loop with Channelization	on with	Port													
	nge Ports									•						
				l												
	Line Side Combination Channelized PBX Trunk Port - Business			UEPPX	UEPCX	1.13	0.00	0.00	0.00	0.00			15.69			
	Line Side Outward Channelized PBX Trunk Port - Business			UEPPX	UEPOX	1.13	0.00	0.00	0.00	0.00			15.69			
	Line Side Inward Only Channelized PBX Trunk Port without DID			UEPPX	UEP1X	1.13	0.00	0.00	0.00	0.00			15.69			
	2-Wire Trunk Side Unbundled Channelized DID Trunk Port			UEPPX	UEPDM	7.09	0.00	0.00	0.00	0.00			15.69			
Featur	re Activations - Unbundled Loop Concentration			OLITA	OLI DIVI	7.03	0.00	0.00	0.00	0.00			10.00			
	Feature (Service) Activation for each Line Side Port Terminated in D4 Bank			UEPPX	1PQWM	0.56	25.45	13.44	4.20	4.17			15.69			
	Feature (Service) Activation for each Trunk Side Port Terminated in D4 Bank			UEPPX	1PQWU	0.56	78.31	18.46	59.37	11.60			15.69			
Telepl	none Number/ Group Establishment Charges for DID Service															
	DID Trunk Termination (1 per Port)			UEPPX	NDT	0.00	0.00	0.00								
	Estab Trk Grp and Provide 1st 20 DID Nos. (FL,GA, NC,& SC)			UEPPX	NDZ	0.00	0.00	0.00								
	DID Numbers - groups of 20 - Valid all States Non-Consecutive DID Numbers - per number			UEPPX UEPPX	ND4 ND5	0.00	0.00	0.00								
	Reserve Non-Consecutive DID Numbers			UEPPX	ND6	0.00	0.00	0.00								
	Reserve DID Numbers			UEPPX	NDV	0.00	0.00	0.00								
Local	Number Portability			OLITA	INDV	0.00	0.00	0.00								
	Local Number Portability - 1 per port			UEPPX	LNPCP	3.15	0.00	0.00								
	JRES - Vertical and Optional															
Local	Switching Features Offered with Line Side Ports Only							•		•						
	All Features Available			UEPPX	UEPVF	3.04	0.00	0.00					15.69			<u> </u>
	CENTREX PORT/LOOP COMBINATIONS - COST BASED RATE:		C4=+-					itali Darit								<u> </u>
	t Based Rates are applied where BellSouth is required by FCC tures shall apply to the Unbundled Port/Loop Combination - C								dlad Part sastir	n of this Deta	Evhibit					-
	tures snall apply to the Unbundled Port/Loop Combination - C I Office and Tandem Switching Usage and Common Transport											oin Port/I o	on Combinat	ons	-	
For G	eorgia, Kentucky, Louisiana, MIssissippi and Tennessee, the re	ecurring	UNE	Port and Loop chard	ges listed an	oly to Currently	Combined and	Not Currently	y Combined Co	mbos. The th	e first and a	dditional P	ort nonrecurr	ing charges a	apply to Not C	urrently
	ined Combos for all states. In GA, KY, LA, MS and TN these no															
	ined Combos in all other states, the nonrecurring charges sha		•	·			,									•
	rket Rates for Unbundled Centrex Port/Loop Combination will	be nego	tiated	on an Individual Ca	se Basis, un	til further notice	э.									
	CENTREX - 5ESS (Valid in All States)									· · · · ·						
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo				1											
UNE P	Port/Loop Combination Rates (Non-Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo-		1	LIEDOE		44.00										
	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Non-Design		2	UEP95 UEP95		14.89 21.52										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -				1											
	Non-Design		3	HEP95		27 17										
UNF P	Non-Design		3	UEP95		27.17										
UNE P	Non-Design Nort/Loop Combination Rates (Design) 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo		3	UEP95		27.17										

	D NETWORK ELEMENTS - South Carolina			1	1 1								Attachment:		Table 1	ł. —
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RA ⁻	TES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic-	Incremental Charge - Manual Svc Order vs. Electronic-	Charge -	Charge - Manual Sv Order vs. Electronic
													1st	Add'l	Disc 1st	Disc Add
						Dan	Nonrec	curring	Nonrecurring	Disconnect			oss	Rates(\$)		1
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		2	UEP95		24.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design		3	UEP95		29.59										
UNE L	oop Rate															
	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP95	UECS1	13.76										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP95	UECS1	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP95	UECS1	26.04										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP95	UECS2	16.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP95	UECS2	23.13										
	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP95	UECS2	28.46										
	ort Rate			Į	1									ļ		
All Sta			<u> </u>	ļ. <u></u>												ļ
	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP95	UEPYA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPYB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID)1Basic Local Area			UEP95	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2 Basic Local Area			UEP95	UEPYM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term - Basic Local Area			UEP95	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent - Basic Local Area			UEP95	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port Terminated on 800 Service Term - Basic Local Area			UEP95	UEPY2	1.13	40.30	19.90	24.98	6.65		15.69				
AL, KY	, LA, MS, SC, & TN Only															
	2-Wire Voice Grade Port (Centrex)			UEP95	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex 800 termination)			UEP95	UEPQB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex with Caller ID)1			UEP95	UEPQH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center)2			UEP95	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP95	UEPQZ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent			UEP95	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port Terminated on 800 Service Term			UEP95	UEPQ2	1.13	40.30	19.90	24.98	6.65		15.69				
Local S	Switching															
Local	Number Portability															
	Local Number Portability (1 per port)			UEP95	LNPCC	0.35										
Feature																
	All Standard Features Offered, per port			UEP95	UEPVF	3.04						15.69				
NARS																
	Unbundled Network Access Register - Combination			UEP95	UARCX	0.00	0.00	0.00				15.69				
	Unbundled Network Access Register - Indial			UEP95	UAR1X	0.00	0.00	0.00				15.69				
	Unbundled Network Access Register - Outdial			UEP95	UAROX	0.00	0.00	0.00				15.69				
	laneous Terminations															
2-Wire	Trunk Side				1									ļ		
	Trunk Side Terminations, each		<u> </u>	UEP95	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				ļ
4-Wire	Digital (1.544 Megabits)		<u> </u>													<u> </u>
	DS1 Circuit Terminations, each			UEP95	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				ļ
	DS0 Channels Activated, each		<u> </u>	UEP95	M1HDO	0.00	14.51					15.69				
Interof	fice Channel Mileage - 2-Wire		<u> </u>	<u> </u>	1											ļ
	Interoffice Channel Facilities Termination		<u> </u>	UEP95	MIGBC	24.30	40.63	27.47	16.77	6.91		15.69				ļ
	Interoffice Channel mileage, per mile or fraction of mile			UEP95	MIGBM	0.0167										
Featur	e Activations (DS0) Centrex Loops on Channelized DS1 Servic	e		ļ	1 1											
	annel Bank Feature Activations	1								-						
	Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP95	1PQWS	0.56						15.69				

ONRONDEE	D NETWORK ELEMENTS - South Carolina			1									Attachment:		Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RAT	ΓES(\$)				Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Sv Order vs. Electronic Disc Add'l
						Rec	Nonrec		Nonrecurring					Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop											4= 00				l
	Slot Feature Activation on D-4 Channel Bank Centrex Loop Slot -			UEP95	1PQW7	0.56						15.69				
	Different Wire Center			UEP95	1PQWP	0.56						15.69				İ
	Billiotetic Wile Gerico			OL: 50		0.00						10.00				
	Feature Activation on D-4 Channel Bank Private Line Loop Slot			UEP95	1PQWV	0.56						15.69				İ
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop															
	Slot			UEP95	1PQWQ	0.56						15.69				
	Feature Activation on D-4 Channel Bank WATS Loop Slot			UEP95	1PQWA	0.56						15.69				
Non-R	ecurring Charges (NRC) Associated with UNE-P Centrex NRC Conversion Currently Combined Switch-As-Is with allowed															
	changes, per port			UEP95	USAC2		37.93	16.72				15.69				1
UNE-P	CENTREX - DMS100 (Valid in All States)	1		021 00	55,152		37.33	10.72				10.08				
	VG Loop/2-Wire Voice Grade Port (Centrex) Combo				1											
	ort/Loop Combination Rates (Non-Design)															
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -															
	Non-Design		1	UEP9D		14.89										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		_	LIEDOD		04.50										
-	Non-Design 2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -		2	UEP9D	+	21.52										
	Non-Design		3	UEP9D		27.17										
UNE P	ort/Loop Combination Rates (Design)			OLI OD		27.17										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex) Port Combo -				1											
	Design		1	UEP9D		17.81										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo -															
	Design Control of the		2	UEP9D		24.26										
	2-Wire VG Loop/2-Wire Voice Grade Port (Centrex)Port Combo - Design		3	UEP9D		29.59										
LINE	poesign oop Rate		3	UEP9D	-	29.59										
OIL E	2-Wire Voice Grade Loop (SL 1) - Zone 1		1	UEP9D	UECS1	13.76										
	2-Wire Voice Grade Loop (SL 1) - Zone 2		2	UEP9D	UECS1	20.38										
	2-Wire Voice Grade Loop (SL 1) - Zone 3		3	UEP9D	UECS1	26.04										
	2-Wire Voice Grade Loop (SL 2) - Zone 1		1	UEP9D	UECS2	16.68										
	2-Wire Voice Grade Loop (SL 2) - Zone 2		2	UEP9D	UECS2	23.13										
LINE	2-Wire Voice Grade Loop (SL 2) - Zone 3		3	UEP9D	UECS2	28.46										├
	ort Rate FATES				+						-					
ALL 3	2-Wire Voice Grade Port (Centrex) Basic Local Area			UEP9D	UEPYA	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex 800 termination)Basic Local				1											
	Area			UEP9D	UEPYB	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-PSET)3Basic Local															
	Area			UEP9D	UEPYC	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5009)3Basic Local Area			UEP9D	UEPYD	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5209))3 Basic Local			UEF9D	UEPTD	1.13	40.30	19.90	24.90	0.00		15.09				
	Area			UEP9D	UEPYE	1.13	40.30	19.90	24.98	6.65		15.69				İ
	2-Wire Voice Grade Port (Centrex / EBS-M5112))3 Basic Local				3											
	Area			UEP9D	UEPYF	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5312))3Basic Local															
	Area			UEP9D	UEPYG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5008))3 Basic Local Area	l		UEP9D	UEPYT	1.13	40.30	19.90	24.98	6.65		15.69				1
_	2-Wire Voice Grade Port (Centrex / EBS-M5208))3 Basic Local			OEFBD	UEFTI	1.13	40.30	19.90	24.98	0.05	-	15.09				
	Area	l		UEP9D	UEPYU	1.13	40.30	19.90	24.98	6.65		15.69				1
	2-Wire Voice Grade Port (Centrex / EBS-M5216))3 Basic Local					5			50	2.30						
	Area			UEP9D	UEPYV	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5316))3 Basic Local	l														
	Area			UEP9D	UEPY3	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>

UNBUNDLE	D NETWORK ELEMENTS - South Carolina												Attachment:	11	Table 1	
CATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	usoc		RAT	TES(\$)			Svc Order Submitted Elec per LSR	Submitted	Incremental			Incremental Charge - Manual Svc Order vs. Electronic- Disc Add'l
						Rec	Nonrec	curring	Nonrecurring	Disconnect				Rates(\$)		
						Nec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
	2-Wire Voice Grade Port (Centrex with Caller ID) Basic Local Area			UEP9D	UEPYH	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYW	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication))3 Basic Local Area			UEP9D	UEPYJ	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2 Basic Local Area			UEP9D	UEPYM	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3 Basic Local Area			UEP9D	UEPYO	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3 Basic Local Area			UEP9D	UEPYP	1.13	108.36	70.71	54.47	11.94		15.69				ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3 Basic Local Area			UEP9D	UEPYQ	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3 Basic Local Area			UEP9D	UEPYR	1.13	108.36	70.71	54.47	11.94		15.69				ļ
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3 Basic Local Area			UEP9D	UEPYS	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3 Basic Local Area			UEP9D	UEPY4	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3 Basic Local Area			UEP9D	UEPY5	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3 Basic Local Area			UEP9D	UEPY6	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3 Basic Local Area			UEP9D	UEPY7	1.13	108.36	70.71	54.47	11.94		15.69]
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service Term			UEP9D	UEPYZ	1.13	108.36	70.71	54.47	11.94		15.69				<u></u>
	2-Wire Voice Grade Port terminated in on Megalink or equivalent Basic Local Area			UEP9D	UEPY9	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>
	2-Wire Voice Grade Port Terminated on 800 Service Term Basic Local Area			UEP9D	UEPY2	1.13	40.30	19.90	24.98	6.65		15.69				
AL, KY	Y, LA, MS, SC, & TN Only			LIEBAB	LUEBO A		40	10	-							
	2-Wire Voice Grade Port (Centrex)	<u> </u>		UEP9D	UEPQA	1.13	40.30	19.90	24.98	6.65		15.69		ļ		
	2-Wire Voice Grade Port (Centrex 800 termination) 2-Wire Voice Grade Port (Centrex / EBS-PSET)3	ļ		UEP9D UEP9D	UEPQB UEPQC	1.13 1.13	40.30 40.30	19.90 19.90	24.98 24.98	6.65 6.65		15.69 15.69				
 	2-Wire Voice Grade Port (Centrex / EBS-PSE1)3 2-Wire Voice Grade Port (Centrex / EBS-M5009)3	1		UEP9D UEP9D	UEPQD	1.13	40.30	19.90	24.98	6.65	1	15.69		1		
 	2-Wire Voice Grade Port (Centrex / EBS-M5209)3	 		UEP9D	UEPQE	1.13	40.30	19.90	24.98	6.65		15.69		1		
 	2-Wire Voice Grade Port (Centrex / EBS-M5209)3			UEP9D	UEPQF	1.13	40.30	19.90	24.98	6.65		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex / EBS-M5312)3	1		UEP9D	UEPQG	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5008)3			UEP9D	UEPQT	1.13	40.30	19.90	24.98	6.65		15.69				 I
	2-Wire Voice Grade Port (Centrex / EBS-M5208)3			UEP9D	UEPQU	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5216)3			UEP9D	UEPQV	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex / EBS-M5316)3			UEP9D	UEPQ3	1.13	40.30	19.90	24.98	6.65		15.69				1
	2-Wire Voice Grade Port (Centrex with Caller ID)			UEP9D	UEPQH	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Caller ID/Msg Wtg Lamp Indication)3			UEP9D	UEPQW	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex/Msg Wtg Lamp Indication)3			UEP9D	UEPQJ	1.13	40.30	19.90	24.98	6.65		15.69				
	2-Wire Voice Grade Port (Centrex from diff Serving Wire Center) 2			UEP9D	UEPQM	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-PSET)2, 3			UEP9D	UEPQO	1.13	108.36	70.71	54.47	11.94		15.69				<u></u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5009)2, 3			UEP9D	UEPQP	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-5209)2, 3			UEP9D	UEPQQ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5112)2, 3			UEP9D	UEPQR	1.13	108.36	70.71	54.47	11.94		15.69				<u> </u>
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5312)2, 3			UEP9D	UEPQS	1.13	108.36	70.71	54.47	11.94		15.69				

NRONDLE	D NETWORK ELEMENTS - South Carolina			1							_		Attachment:		Table 1	
ATEGORY	RATE ELEMENTS	Interi m	Zone	BCS	USOC		RAT	ES(\$)			Svc Order Submitted Elec per LSR	Svc Order Submitted Manually per LSR	Incremental Charge - Manual Svc Order vs. Electronic- 1st	Incremental Charge - Manual Svc Order vs. Electronic- Add'I	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incrementa Charge - Manual Svo Order vs. Electronic Disc Add'l
					+		Nonrec	urring	Nonrecurring	Disconnect			220	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
					+		THOL	Addi	11130	Addi	CONILC	JONAN	JOMAN	JONAN	JONIAN	JONAN
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5008)2, 3			UEP9D	UEPQ4	1.13	108.36	70.71	54.47	11.94		15.69				
	2 11110 10100 01000 1 011 (001110) 01110 1 0110 1 220 1110000/2; 0			02. 02	02. Q.		100.00		0			10.00				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5208)2, 3			UEP9D	UEPQ5	1.13	108.36	70.71	54.47	11.94		15.69				
	· ·															
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5216)2, 3			UEP9D	UEPQ6	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port (Centrex/differ SWC /EBS-M5316)2, 3			UEP9D	UEPQ7	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port, Diff Serving Wire Center - 800 Service						400.00					4= 00				
	Term			UEP9D	UEPQZ	1.13	108.36	70.71	54.47	11.94		15.69				
	2-Wire Voice Grade Port terminated in on Megalink or equivalent	1		UEP9D	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69			1	
-	2-Wire Voice Grade Port terminated in on Megalink or equivalent 2-Wire Voice Grade Port Terminated on 800 Service Term	1		UEP9D	UEPQ9	1.13	40.30	19.90	24.98	6.65		15.69			1	1
Local	Switching	1		021 00	OL: 42	1.13	70.50	13.30	24.30	0.00		13.03				
	Centrex Intercom Funtionality, per port	1		UEP9D	URECS	0.7996						15.69			1	
Local	Number Portability					,										
	Local Number Portability (1 per port)			UEP9D	LNPCC	0.35										
Featu																
	All Standard Features Offered, per port			UEP9D	UEPVF	3.04						15.69				
	All Select Features Offered, per port			UEP9D	UEPVS	0.00	406.42					15.69				
	All Centrex Control Features Offered, per port			UEP9D	UEPVC	3.04						15.69				
												15.69				
NARS				LIEDOD	LIADOV	0.00	0.00	0.00				45.00				
	Unbundled Network Access Register - Combination			UEP9D UEP9D	UARCX UAR1X	0.00	0.00	0.00				15.69 15.69				
	Unbundled Network Access Register - Inward Unbundled Network Access Register - Outdial			UEP9D	UAROX	0.00	0.00	0.00				15.69				
Misce	laneous Terminations			OLF 9D	UAROX	0.00	0.00	0.00				15.05				
	Trunk Side				+											
	Trunk Side Terminations, each			UEP9D	CEND6	8.86	119.57	18.78	60.03	3.77		15.69				
4-Wire	Digital (1.544 Megabits)															
	DS1 Circuit Terminations, each			UEP9D	M1HD1	73.62	202.47	95.90	72.75	2.47		15.69				
	DS0 Channels Activiated per Channel			UEP9D	M1HDO	0.00	14.51					15.69				
Intero	fice Channel Mileage - 2-Wire															
	Interoffice Channel Facilities Termination			UEP9D	MIGBC	24.30	40.63	27.47	16.77	6.91		15.69				
	Interoffice Channel mileage, per mile or fraction of mile			UEP9D	MIGBM	0.0167										
	e Activations (DS0) Centrex Loops on Channelized DS1 Service	e			+											
D4 Ch	annel Bank Feature Activations Feature Activation on D-4 Channel Bank Centrex Loop Slot			UEP9D	1PQWS	0.56						15.69				
	realure Activation on 5-4 channel Bank Centrex Loop Glot			OLI 3D	11 QVV0	0.50						15.05				
	Feature Activation on D-4 Channel Bank FX line Side Loop Slot			UEP9D	1PQW6	0.56						15.69				
	Feature Activation on D-4 Channel Bank FX Trunk Side Loop					0.00						10.00				
	Slot			UEP9D	1PQW7	0.56						15.69				
	Feature Activation on D-4 Channel Bank Centrex Loop Slot -															
	Different Wire Center			UEP9D	1PQWP	0.56						15.69				
		1			1]	
	Feature Activation on D-4 Channel Bank Private Line Loop Slot	<u> </u>		UEP9D	1PQWV	0.56						15.69			ļ	
	Feature Activation on D-4 Channel Bank Tjie Line/Trunk Loop Slot	l		UEP9D	1PQWQ	0.56						15.69				
	Feature Activation on D-4 Channel Bank WATS Loop Slot	 		UEP9D UEP9D	1PQWQ 1PQWA	0.56						15.69			1	
Non-P	ecurring Charges (NRC) Associated with UNE-P Centrex	 		021 30	11 6444	0.50						13.09			 	1
14071-14	NRC Conversion Currently Combined Switch-As-Is with allowed				1											
	changes, per port	1		UEP9D	USAC2		37.93	16.72				15.69			1	
	New Centrex Standard Common Block			UEP9D	M1ACS	0.00	668.70					15.69				
	New Centrex Customized Common Block			UEP9D	M1ACC	0.00	668.70					15.69				
	NAR Establishment Charge, Per Occasion			UEP9D	URECA	0.00	72.89	•		•		15.69				
	- Required Port for Centrex Control in 1AESS, 5ESS & EWSD			1				·								
Note	2 - Requires Interoffice Channel Mileage - Requires Specific Customer Premises Equipment															

Amendment to the Interconnection Agreement By and Between BellSouth Telecommunications, Inc. And ITC^DeltaCom Communications, Inc. Dated January 31, 2002

This Amendment ("Amendment") is made and entered into by and between ITC^DeltaCom Communications, Inc. ("ITC^DeltaCom") and BellSouth Telecommunications, Inc. ("BellSouth") to amend the Interconnection Agreement ("the Agreement") entered into by ITC^DeltaCom and BellSouth on January 31, 2002 for the states of Mississippi and South Carolina.

WHEREAS, the Parties desire to amend that certain Interconnection Agreement between BellSouth and ITC^DeltaCom dated January 31, 2002 to delete and replace Attachment 10 in its entirety;

NOW THEREFORE, in consideration of the provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, ITC^DeltaCom and BellSouth hereby covenant and agree as follows:

- 1. Attachment 10 of the Agreement is hereby deleted and shall be replaced with the Attachment 10 in Exhibit 1 attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Interconnection Agreement shall remain unchanged and in full force and effect.
- 3. Either or both of the Parties are authorized to submit this Amendment to the appropriate Commission or other Regulatory Agency for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.
- 4. This Amendment is made effective following the last signature of both Parties. The term of this Agreement shall expire on December 31, 2002.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

ITC^DeltaCom Communications, Inc.	BellSouth Telecommunications, Inc.
Signature	Signature
Name	Name
Title	Title
Date	Date

EXHIBIT 1

Attachment 10 Page 1

Attachment 10

Performance Measurements

EXHIBIT 1

Attachment 10 Page 2

Performance Measurements

To the extent any state Commission has not ordered performance measures and/or remedies as of the date hereof, BellSouth will provide to ITC^DeltaCom the Georgia Service Quality Measures (SQMs) and associated remedies ordered by the Georgia Public Service Commission in GA Docket 7892-U. In the event the Georgia Public Service Commission adds, deletes or otherwise modifies any SQMs and/or associated remedies, such additions, deletions or modifications shall be deemed made to the SQMs and associated remedies applicable to ITC^DeltaCom hereunder. Upon the effective date of any order from a state Commission regarding performance measures and/or associated remedies, such ordered measures and remedies, as they may be amended or modified by the state Commission from time to time, shall become effective hereunder in lieu of the Georgia measures and remedies for the state in which such measures and remedies have been ordered.

BellSouth Service Quality Measurement Plan (SQM)

Georgia Performance Metrics

Measurement Descriptions
Version 1.01

Issue Date: April 6, 2001

This version of the Georgia SQM reflects the Order in GA Docket 7892-U. Some of the measures, business rules, disaggregations and/or exclusions are under development and will be reflected in the monthly reports in the near future. The other Georgia SQM posted on this site will be removed at that time.



Introduction

The BellSouth Service Quality Measurement Plan (SQM) describes in detail the measurements produced to evaluate the quality of service delivered to BellSouth's customers both wholesale and retail. The SQM was developed to respond to the requirements of the Communications Act of 1996 Section 251 (96 Act) which required BellSouth to provide non-discriminatory access to Competitive Local Exchange Carriers (CLEC)¹ and its Retail Customers. The reports produced by the SQM provide regulators, CLECs and BellSouth the information necessary to monitor the delivery of non-discriminatory access.

This plan results from the many divergent forces evolving from the 96 Act. The 96 Act, the Georgia Public Service Commission (GPSC) Order (Docket 7892-U 12/30/97), LCUG 1-7.0, the FCC's NPRM (CC Docket 98-56 RM9101 04/17/98), the Louisiana Public Service Commission (LPSC) Order (Docket U-22252 Subdocket C 04/19/98), numerous arbitration cases, LPSC sponsored collaborative workshops (10/98-02/00), and proceedings in Alabama, Mississippi, and North Carolina have and continue to influence the SQM. This version of the SQM reflects the Order of the Georgia Public Service Commission in Docket 7892-U dated January 12, 2001.

The SQM and the reports flowing from it must change to reflect the dynamic requirements of the industry. New measurements are added as new products, systems, and processes are developed and fielded. New products and services are added as the markets for them develop and the processes stabilize. The measurements are also changed to reflect changes in systems, correct errors, and respond to both 3rd Party audit requirements and the Georgia PSC.

This document is intended for use by someone with knowledge of telecommunications industry, information technologies and a functional knowledge of the subject areas covered by the BellSouth Performance Measurements and the reports that flow from them.

Once it is approved, the most current copy of this document can be found on the web at URL: https://pmap.bellsouth.com in the Help folder.

Report Publication Dates

Each month, preliminary SQM reports will be posted to BellSouth's SQM web site (https://www.pmap.bellsouth.com) by 8:00 A.M. EST on the 21st day of each month or the first business day after the 21st. Final validated SQM reports will be posted by 8:00 A.M. on the last day of the month. Reports not posted by this time will be considered late for SEEM payment purposes. Preliminary SEEM reports will be posted on the same day as the SQM validated reports. Validated SEEM reports will posted on the 15th of the following month. Payments due will also be paid on the 15th of the following month. For instance: May data will be posted in preliminary SQM reports on June 21. Final validated SQM reports and preliminary SEEM reports will be posted on the last day of June. Final validated SEEM reports will be posted and payments mailed on July 15th.

1. Alternative Local Exchange Companies (ALEC) and Competing Local Providers (CLP) are referred to as Competitive Local Exchange Carriers (CLEC) in this document.



Report Delivery Methods

CLEC SQM and SEEM reports will be considered delivered when posted to the web site. The Georgia Public Service Commission (GPSC) will be given access to the web site. In addition, a copy of the Monthly State Summary reports will be filed with the GPSC as soon as possible after the last day of each month.



Contents

Section 1:	S Operations Support Systems (OSS)	
OSS-1:	Average Response Time and Response Interval (Pre-Ordering/Ordering)	1-1
OSS-2:	Interface Availability (Pre-Ordering/Ordering)	1-6
OSS-3:	Interface Availability (Maintenance & Repair)	1-9
OSS-4:	Response Interval (Maintenance & Repair)	1-11
PO-1:	Loop Makeup - Response Time - Manual	1-13
PO-2:	Loop Make Up - Response Time - Electronic	1-15
Section 2:	Ordering	
O-1:	Acknowledgement Message Timeliness	2-1
O-2:	Acknowledgement Message Completeness	2-3
O-3:	Percent Flow-Through Service Requests (Summary)	
O-4:	Percent Flow-Through Service Requests (Detail)	2-8
O-5:	Flow-Through Error Analysis	
O-6:	CLEC LSR Information	
O-7:	Percent Rejected Service Requests	2-19
O-8:	Reject Interval	
O-9:	Firm Order Confirmation Timeliness	
O-10:	Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual	2-29
O-11:	Firm Order Confirmation and Reject Response Completeness	2-3
O-12:	Speed of Answer in Ordering Center	
O-13:	LNP-Percent Rejected Service Requests	2-35
O-14:	LNP-Reject Interval Distribution & Average Reject Interval	2-3
O-15:	LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval-	2-40
Section 3:	Provisioning	
P-1:	Mean Held Order Interval & Distribution Intervals	3-
P-2:	Average Jeopardy Notice Interval & Percentage of Orders Given Jeopardy Notices	3-4
P-3:	Percent Missed Installation Appointments	3-7
P-4:	Average Completion Interval (OCI) & Order Completion Interval Distribution	3-10
P-5:	Average Completion Notice Interval	3-13
P-6:	% Completions/Attempts without Notice or < 24 hours Notice	3-16
P-7:	Coordinated Customer Conversions Interval	
P-7A:	Coordinated Customer Conversions - Hot Cut Timeliness% Within Interval and Average Interval	3-20
P-7B:	Coordinated Customer Conversions – Average Recovery Time	
P-7C:	Hot Cut Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order	
P-8:	Cooperative Acceptance Testing - % of xDSL Loops Tested	3-20
P-9:	% Provisioning Troubles within 30 days of Service Order Completion	3-28
P-10:	Total Service Order Cycle Time (TSOCT)	3-3
P-11:	Service Order Accuracy	3-34
P-12:	LNP-Percent Missed Installation Appointments	
P-13:	LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution	3-38
P-14:	LNP-Total Service Order Cycle Time (TSOCT)	3-40



Section 4:	Maintenance & Repair	
	Missed Repair Appointments	4-1
M&R-2:	Customer Trouble Report Rate	4-3
M&R-3:	Maintenance Average Duration	4-5
	Percent Repeat Troubles within 30 Days	
	Out of Service (OOS) > 24 Hours	
M&R-6:	Average Answer Time – Repair Centers	4-11
M&R-7:	Mean Time To Notify CLEC of Network Outages	4-13
Section 5:	Billing	
B-1:	Invoice Accuracy	5-1
B2:	Mean Time to Deliver Invoices	
B3:	Usage Data Delivery Accuracy	5-5
B4:	Usage Data Delivery Completeness	
B5:	Usage Data Delivery Timeliness	
B6:	Mean Time to Deliver Usage	
B7:	Recurring Charge Completeness	
B8:	Non-Recurring Charge Completeness	5-15
Section 6:	Operator Services And Directory Assistance	
OS-1:	Speed to Answer Performance/Average Speed to Answer - Toll	6-1
OS-2:	Speed to Answer Performance/Percent Answered with "X" Seconds – Toll	6-3
DA-1:	Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA)	
DA-2:	Speed to Answer Performance/Percent Answered within "X" Seconds – Directory Assistance (DA)	6-6
Section 7:	Database Update Information	
D-1:	Average Database Update Interval	7-1
D-2:	Percent Database Update Accuracy	7-3
D-3:	Percent NXXs and LRNs Loaded by the LERG Effective Date	7-5
Section 8:	E911	
E-1:	Timeliness	8-1
E-1:	Accuracy	
E-3:	Mean Interval	
Section 0	Trunk Group Performance	
	Trunk Group Performance-Aggregate	0.1
TGP-1:	Trunk Group Performance-Aggregate	9-4
Section 10	: Collocation	
		10.1
C-1:	Collocation Average Response Time	
C-2:	Collocation Average Arrangement Time	
C-3:	Collocation Percent of Due Dates Missed	10-3
Section 11	: Change Management	
CM-1:	Timeliness of Change Management Notices	11-1
CM-2:	Change Management Notice Average Delay Days	
CM-3:	Timeliness of Documents Associated with Change	
CM-4:	Change Management Documentation Average Delay Days	11-7
CM-5:	Notification of CLEC Interface Outages	11-9
	: Bona Fide / New Business Request Process	
BFR-1:	Percentage of BFR/NBR Requests Processed Within 30 Business Days	12-1



BFR-2:	Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed Within	n X (10/30/60) Business Days 12-3
Appendix A-1: A-2:	X A: Reporting Scope Standard Service Groupings Standard Service Order Activities	
Appendix	x B: Glossary of Acronyms and Terms	B-1
Appendix	x C: BellSouth Audit Policy	C-1



Section 1: Operations Support Systems (OSS)

OSS-1: Average Response Time and Response Interval (Pre-Ordering/ Ordering)

Definition

Average response time and response intervals are the average times and number of requests responded to within certain intervals for accessing legacy data associated with appointment scheduling, service & feature availability, address verification, request for Telephone numbers (TNs), and Customer Service Records (CSRs).

Exclusions

None

Business Rules

The average response time for retrieving pre-order/order information from a given legacy system is determined by summing the response times for all requests submitted to the legacy systems during the reporting period and dividing by the total number of legacy system requests for that month.

The response interval starts when the client application (LENS or TAG for CLECs and RNS or ROS for BellSouth) submits a request to the legacy system and ends when the appropriate response is returned to the client application. The number of accesses to the legacy systems during the reporting period which take less than 2.3 seconds, the number of accesses which take more than 6 seconds, and the number which are less than or equal to 6.3 seconds are also captured.

Calculation

Response Time = (a - b)

- a = Date & Time of Legacy Response
- b = Date & Time of Legacy Request

Average Response Time = $c \div d$

- c = Sum of Response Times
- d = Number of Legacy Requests During the Reporting Period

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report MonthLegacy Contract (per reporting dimension)	Report MonthLegacy Contract (per reporting dimension)
Response Interval	Response Interval
Regional Scope	Regional Scope

BELLSOUTH®

OSS-1: Average Response Time and Response Interval (Pre-Ordering/Ordering)

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
 RSAG – Address (Regional Street Address Guide-Address) – stores street address information used to validate customer addresses. CLECs and BellSouth query this legacy system. RSAG – TN (Regional Street Address Guide-Telephone number) – contains information about facilities available and telephone numbers working at a given address. CLECs and BellSouth query this legacy system. ATLAS (Application for Telephone Number Load Administration and Selection) – acts as a warehouse for storing telephone numbers that are available for assignment by the system. It enables CLECs and BellSouth service reps to select and reserve telephone numbers. CLECs and BellSouth query this legacy system. COFFI (Central Office Feature File Interface) – stores information about product and service offerings and availability. CLECs query this legacy system. DSAP (DOE Support Application) – provides due date information. CLECs and BellSouth query this legacy system. HAL/CRIS (Hands-Off Assignment Logic/Customer Record Information System) – a system used to access the Business Office Customer Record Information System (BOCRIS). It allows BellSouth servers, including LENS, access to legacy systems. CLECs query this legacy system. P/SIMS (Product/Services Inventory Management system) – provides information on capacity, tariffs, inventory and service availability. CLECs query this legacy system. OASIS (Obtain Available Services Information Systems) – Information on feature and rate availability. BellSouth queries this legacy system. 	• Parity + 2 seconds

Table 1: Legacy System Access Times For RNS

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. Sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	X	X	х
RSAG	RSAG-ADDR	Address	x	X	X	х	х
ATLAS	ATLAS-TN	TN	X	X	X	X	X
DSAP	DSAP	Schedule	X	X	X	Х	Х
CRIS	CRSACCTS	CSR	X	X	X	Х	Х
OASIS	OASISCAR	Feature/Service	X	X	X	X	Х
OASIS	OASISLPC	Feature/Service	X	X	X	X	Х
OASIS	OASISMTN	Feature/Service	X	X	X	X	Х
OASIS	OASISBIG	Feature/Service	X	X	X	X	Х

Table 2: Legacy System Access Times For R0S

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	х	X	Х	х	Х
RSAG	RSAG-ADDR	Address	Х	X	Х	X	Х
ATLAS	ATLAS-TN	TN	X	X	X	X	X



Table 2: Legacy System Access Times For R0S

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
DSAP	DSAP	Schedule	x	X	X	x	X
CRIS	CRSOCSR	CSR	х	Х	Х	X	X
OASIS	OASISBIG	Feature/Service	X	x	X	X	X

Table 3: Legacy System Access Times For LENS

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	X	X	X
RSAG	RSAG-ADDR	Address	х	X	Х	X	X
ATLAS	ATLAS-TN	TN	х	X	Х	X	X
DSAP	DSAP	Schedule	X	X	х	X	X
HAL	HAL/CRIS	CSR	X	X	Х	X	X
COFFI	COFFI/USOC	Feature/Service	X	X	Х	X	X
P/SIMS	PSIMS/ORB	Feature/Service	X	X	X	X	X

Table 4: Legacy System Access Times For TAG

System	Contract	Data	< 2.3 sec.	> 6 sec.	<u><</u> 6.3 sec.	Avg. sec.	# of Calls
RSAG	RSAG-TN	Address	X	X	X	x	x
RSAG	RSAG-ADDR	Address	х	X	Х	х	X
ATLAS	ATLAS-TN	TN	X	X	X	X	X
ATLAS	ATLAS-MLH	TN	X	X	X	х	X
ATLAS	ATLAS-DID	TN	X	X	X	х	X
DSAP	DSAP	Schedule	X	X	X	х	X
CRIS	CRSECSRL	CSR	х	X	Х	х	X
CRIS	CRSECSR	CSR	X	X	X	Х	X

SEEM Measure

SEEM Measure				
	Tier I			
Yes	Tier II	X		
	Tier III			

Note: CLEC specific data is not available in this measure. Queries of this sort do not have company specific signatures.



SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
 RSAG – Address (Regional Street Address Guide-Address) – stores street address information used to validate customer addresses. CLECs and BellSouth query this legacy system. RSAG – TN (Regional Street Address Guide-Telephone number) – contains information about facilities available and telephone numbers working at a given address. CLECs and BellSouth query this legacy system. ATLAS (Application for Telephone Number Load Administration and Selection) – acts as a warehouse for storing telephone numbers that are available for assignment by the system. It enables CLECs and BellSouth service reps to select and reserve telephone numbers. CLECs and BellSouth query this legacy system. COFFI (Central Office Feature File Interface) – stores information about product and service offerings and availability. CLECs query this legacy system. DSAP (DOE Support Application) – provides due date information. CLECs and BellSouth query this legacy system. HAL/CRIS (Hands-Off Assignment Logic/Customer Record Information System) – a system used to access the Business Office Customer Record Information System (BOCRIS). It allows BellSouth servers, including LENS, access to legacy systems. CLECs query this legacy system. P/SIMS (Product/Services Inventory Management system) – provides information on capacity, tariffs, inventory and service availability. CLECs query this legacy system. OASIS (Obtain Available Services Information Systems) – Information on feature and rate availability. BellSouth queries this legacy system. 	 Percent Response Received within 6.3 seconds: > 95% Parity + 2 seconds



SEEM OSS Legacy Systems

System	BellSouth	CLEC		
	Telephone Number/Address			
RSAG-ADDR	RNS, ROS	TAG, LENS		
RSAG-TN	RNS, ROS	TAG, LENS		
ATLAS	RNS,ROS	TAG. LENS		
	Appointment Scheduling			
DSAP	RNS, ROS	TAG, LENS		
CSR Data				
CRSACCTS	RNS			
CRSOCSR	ROS			
HAL/CRIS		LENS		
CRSECSRL		TAG		
CRSECSR		TAG		
Service/Feature Availability				
OASISBIG	RNS, ROS			
PSIMS/ORB		LENS		



OSS-2: Interface Availability (Pre-Ordering/Ordering)

Definition

Percent of time applications are functionally available as compared to scheduled availability. Calculations are based upon availability of applications and interfacing applications utilized by CLECs for pre-ordering and ordering. "Functional Availability" is defined as the number of hours in the reporting period that the applications/interfaces are available to users. "Scheduled Availability" is defined as the number of hours in the reporting period that the applications/interfaces are scheduled to be available.

Scheduled availability is posted on the Interconnection web site: (www.interconnection.bellsouth.com/oss/oss_hour.html)

Exclusions

- CLEC-impacting troubles caused by factors outside of BellSouth's purview, e.g., troubles in customer equipment, troubles in networks owned by telecommunications companies other than BellSouth, etc.
- Degraded service, e.g., slow response time, loss of non-critical functionality, etc.

Business Rules

This measurement captures the functional availability of applications/interfaces as a percentage of scheduled availability for the same systems. Only full outages are included in the calculations for this measure. Full outages are defined as occurrences of either of the following:

- Application/interfacing application is down or totally inoperative.
- Application is totally inoperative for customers attempting to access or use the application. This includes transport outages when they
 may be directly associated with a specific application.

Comparison to an internal benchmark provides a vehicle for determining whether or not CLECs and retail BST entities are given comparable opportunities for use of pre-ordering and ordering systems.

Calculation

Interface Availability (Pre-Ordering/Ordering) = $(a \div b) \times 100$

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report MonthLegacy Contract Type (per reporting dimension)Regional Scope	Report MonthLegacy Contract Type (per reporting dimension)Regional Scope
Hours of Downtime	Hours of Downtime

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	SQM Analog/Benchmark
Regional Level	• ≥99.5%



OSS Interface Availability

Application	Applicable to	% Availability
EDI	CLEC	X
TAG	CLEC	X
LENS	CLEC	X
LEO	CLEC	X
LESOG	CLEC	X
LNP Gateway	CLEC	X
COG	CLEC	Under Development
SOG	CLEC	Under Development
DOM	CLEC	Under Development
DOE	CLEC/BST	X
SONGS	CLEC/BST	X
ATLAS/COFFI	CLEC/BST	X
BOCRIS	CLEC/BST	X
DSAP	CLEC/BST	X
RSAG	CLEC/BST	X
SOCS	CLEC/BST	X
CRIS	CLEC/BST	X

SEEM Measure

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• ≥ 99.5%



SEEM OSS Interface Availability

Application	Applicable to	% Availability
EDI	CLEC	X
HAL	CLEC	X
LENS	CLEC	X
LEO Mainframe	CLEC	X
LESOG	CLEC	X
PSIMS	CLEC	X
TAG	CLEC	X



OSS-3: Interface Availability (Maintenance & Repair)

Definition

Percent of time applications are functionally available as compared to scheduled availability. Calculations are based upon availability of applications and interfacing applications utilized by CLECs for maintenance and repair. "Functional Availability" is defined as the number of hours in the reporting period that the applications/interfaces are available to users. "Scheduled Availability" is defined as the number of hours in the reporting period that the applications/interfaces are scheduled to be available.

Scheduled availability is posted on the Interconnection web site: (www.interconnection.bellsouth.com/oss/oss_hour.html)

Exclusions

- CLEC-impacting troubles caused by factors outside of BellSouth's purview, e.g., troubles in customer equipment, troubles in networks owned by telecommunications companies other than BellSouth, etc.
- Degraded service, e.g., slow response time, loss of non-critical functionality, etc.

Business Rules

This measurement captures the functional availability of applications/interfaces as a percentage of scheduled availability for the same systems. Only full outages are included in the calculations for this measure. Full outages are defined as occurrences of either of the following:

- Application/interfacing application is down or totally inoperative.
- Application is totally inoperative for customers attempting to access or use the application. This includes transport outages when they
 may be directly associated with a specific application.

Comparison to an internal benchmark provides a vehicle for determining whether or not CLECs and retail BST entities are given comparable opportunities for use of maintenance and repair systems.

Calculation

OSS Interface Availability (a ÷ b) X 100

- a = Functional Availability
- b = Scheduled Availability

Report Structure

- Not CLEC Specific
- Not Product/Service Specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Availability of CLEC TAFI Availability of LMOS HOST, MARCH, SOCS, CRIS, PREDICTOR, LNP and OSPCM ECTA	Availability of BellSouth TAFI Availability of LMOS HOST, MARCH, SOCS, CRIS, PREDICTOR, LNP and OSPCM

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark	
Regional Level	• ≥99.5%	



OSS Interface Availability (M&R)

OSS Interface	% Availability
BST TAFI	x
CLEC TAFI	x
CLEC ECTA	x
BST & CLEC	X
CRIS	x
LMOS HOST	x
LNP	X
MARCH	x
OSPCM	x
PREDICTOR	X
SOCS	x

SEEM Measure

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Regional Level	• ≥99.5%

OSS Interface Availability (M&R)

OSS Interface	% Availability
CLEC TAFI	х
CLEC ECTA	x



OSS-4: Response Interval (Maintenance & Repair)

Definition

The response intervals are determined by subtracting the time a request is received on the BellSouth side of the interface from the time the response is received from the legacy system. Percentages of requests falling into each interval category are reported, along with the actual number of requests falling into those categories.

Exclusions

None

Business Rules

This measure is designed to monitor the time required for the CLEC and BellSouth interface system to obtain from BellSouth's legacy systems the information required to handle maintenance and repair functions. The clock starts on the date and time when the request is received on the BellSouth side of the interface and the clock stops when the response has been transmitted through that same point to the requester.

Note: The OSS Response Interval BellSouth Total Report is a combination of BellSouth Residence and Business Total.

Calculation

OSS Response Interval = (a - b)

- a = Query Response Date and Time
- b = Query Request Date and Time

Percent Response Interval (per category) = $(c \div d) \times 100$

- c = Number of Response Intervals in category "X"
- d = Number of Queries Submitted in the Reporting Period

where, "X" is
$$\leq 4$$
, $> 4 \leq 10$, ≥ 10 , or > 30 seconds.

Report Structure

- Not CLEC Specific
- Not product/service specific
- · Regional Level

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
CLEC Transaction Intervals	BellSouth Business and Residential Transactions Intervals

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
Regional Level	• Parity



Legacy System Access Times for M&R

System BellSouth & CLEC	Count					
	CLEC	<u><</u> 4	> 4 <u><</u> 10	<u><</u> 10	> 10	> 30
CRIS	x	Х	X	X	X	X
DLETH	x	X	X	X	X	X
DLR	x	X	X	X	X	X
LMOS	x	X	X	X	X	X
LMOSupd	x	X	X	X	X	X
LNP	X	X	X	X	X	x
MARCH	X	X	X	X	X	x
OSPCM	X	X	X	X	X	x
Predictor	X	X	X	X	X	X
SOCS	x	X	X	X	X	X
NIW	x	X	X	X	X	x

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



PO-1: Loop Makeup - Response Time - Manual

Definition

This report measures the average interval and percent within the interval from the submission of a Manual Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- Inquiries, which are submitted electronically.
- Designated Holidays are excluded from the interval calculation.
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation.
- · Canceled Inquiries.

Business Rules

The CLEC Manual Loop Makeup Service Inquiry (LMUSI) process includes inquiries submitted via mail or FAX to BellSouth's Complex Resale Support Group (CRSG).

This measurement combines three intervals:

- 1. From receipt of the Service Inquiry for Loop Makeup to hand off to the Service Advocacy Center (SAC) for "Look-up."
- 2. From SAC start date to SAC complete date.
- From SAC complete date to date the Complex Resale Support Group (CRSG) distributes loop makeup information back to the CLEC.

The "Receive Date" is defined as the date the Manual LMUSI is received by the CRSG. It is counted as day Zero. LMU "Return Date" is defined as the date the LMU information is sent back to the CLEC from BellSouth. The interval calculation is reset to Zero when a CLEC initiated change occurs on the Manual LMU request.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = $(c \div d)$

- c = Sum of all Response Intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = $(e \div f) \times 100$

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- · CLEC Aggregate
- CLEC Specific
- Geographic Scope
 - State
 - Region
- Interval for manual LMUs:
 - 0-1 day
- >1-2 days
- >2-3 days
- 0 < 3 days
- >3-6 days



- >6-10 days
- > 10 days
- · Average Interval in days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Total Number of Inquiries	
SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Loops	Benchmark • 95% in 3 Business Days

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation - Analog/Benchmark

SEEM Disaggregation	SEEM Analog/Benchmark
• Loops	Benchmark
	• 95% in 3 Business Days



Georgia Performance Metrics

PO-2: Loop Make Up - Response Time - Electronic

Definition

This report measures the average interval and the percent within the interval from the electronic submission of a Loop Makeup Service Inquiry (LMUSI) to the distribution of Loop Makeup information back to the CLEC.

Exclusions

- Manually submitted inquiries.
- Designated Holidays are excluded from the interval calculation.
- · Canceled Requests.
- · Scheduled OSS Maintenance.

Business Rules

The response interval starts when the CLEC's Mechanized Loop Makeup Service Inquiry (LMUSI) is submitted electronically through the Operational Support Systems interface, LENS, TAG or RoboTAG. It ends when BellSouth's Loop Facility Assignment and Control System (LFACS) responds electronically to the CLEC with the requested Loop Makeup data via LENS, TAG or RoboTAG Interfaces.

Note: The Loop Make Up Service Inquiry Form does not require the CLEC to furnish the type of Loop. The CLEC determines whether the loop makeup will support the type of service they wish to order or not and qualifies the loop. If the loop makeup will support the service, a firm order LSR is submitted by the CLEC. EDI is not a pre-ordering system, and, therefore, is not applicable in this measure.

Calculation

Response Interval = (a - b)

- a = Date and Time LMUSI returned to CLEC
- b = Date and Time the LMUSI is received

Average Interval = $(c \div d)$

- c = Sum of all response intervals
- d = Total Number of LMUSIs received within the reporting period

Percent within interval = $(e \div f) \times 100$

- e = Total LMUSIs received within the interval
- f = Total Number of LMUSIs processed within the reporting period

Report Structure

- CLEC Aggregate
- CLEC Specific
- · Geographic Scope
 - State
 - Region
- Interval for electronic LMUs:
 - 0-1 minute
- >1-5 minutes
- $0 \le 5$ minutes
- > 5 8 minutes
- > 8 15 minutes
- > 15 minutes
- · Average Interval in minutes



Georgia Performance Metrics

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Legacy Contract	
Response Interval	
Regional Scope	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• Loops	Benchmark • 90% in 5 Minutes (05/01/01) • 95% in 1 Minute (08/01/01)

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
• Loop	• 90% in 5 Minutes (05/01/01) • 95% in 1 Minute (08/01/01)



Section 2: Ordering

O-1: Acknowledgement Message Timeliness

Definition

This measurement provides the response interval from the time an LSR or transmission (may contain multiple LSRs from one or more CLECs in multiple states) is electronically submitted via EDI or TAG respectively until an acknowledgement notice is sent by the system.

Exclusions

· Scheduled OSS Maintenance

Business Rules

The process includes EDI & TAG system functional acknowledgements for all messages/Local Service Requests (LSRs) which are electronically submitted by the CLEC. Users of EDI may package many LSRs into one transmission which will receive the acknowledgement message. EDI users may place multiple LSRs in one "envelope" requesting service in one or more states which will mask the identity of the state and CLEC. The start time is the receipt time of the message at BellSouth's side of the interface (gateway). The end time is when the acknowledgement is transmitted by BellSouth at BellSouth's side of the interface (gateway). If more than one CLEC uses the same ordering center (aggregator), an Acknowledgement Message will be returned to the "Aggregator". However, BellSouth will not be able to determine which specific CLEC or state this message represented.

Calculation

Response Interval = (a - b)

- a = Date and Time Acknowledgement Notices returned to CLEC
- b = Date and Time messages/LSRs electronically submitted by the CLEC via EDI or TAG respectively

Average Response Interval = $(c \div d)$

- c = Sum of all Response Intervals
- d = Total number of electronically submitted messages/LSRs received, from CLECs via EDI or TAG respectively, in the Reporting Period.

Reporting Structure

- CLEC Aggregate
- · CLEC Specific/Aggregator
- Geographic Scope
 - Region
- Electronically Submitted LSRs
 - $0 \leq 10$ minutes
- $>10 \le 20$ minutes
- $>20 \le 30$ minutes
- $0 \leq 30$ minutes
- $>30 \le 45$ minutes
- >45 <u><</u>60 minutes
- $>60 \le 120$ minutes
- >120 minutes
- · Average interval for electronically submitted messages/LSRs in minutes

(A) BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report monthRecord of functional acknowledgements	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• EDI	• EDI - 90% within 30 minutes (05/01/01) - 95% within 30 minutes (08/01/01)
• TAG	• TAG – 95% within 30 minutes

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	• EDI - 90% within 30 minutes (05/01/01) - 95% within 30 minutes (08/01/01)
• TAG	• TAG – 95% within 30 minutes



O-2: Acknowledgement Message Completeness

Definition

This measurement provides the percent of transmissions/LSRs received via EDI or TAG respectively, which are acknowledged electronically.

Exclusions

- · Manually submitted LSRs
- · Scheduled OSS Maintenance

Business Rules

EDI and TAG send Functional Acknowledgements for all transmissions/LSRs, which are electronically submitted by a CLEC. Users of EDI may package many LSRs from multiple states in one transmission. If more than one CLEC uses the same ordering center, an Acknowledgement Message will be returned to the "Aggregator", however, BellSouth will not be able to determine which specific CLEC this message represented. The Acknowledgement Message is returned prior to the determination of whether the transmission/LSR will be partially mechanized or fully mechanized.

Calculation

Acknowledgement Completeness = $(a \div b) \times 100$

- a = Total number of Functional Acknowledgements returned in the reporting period for transmissions/LSRs electronically submitted by EDI or TAG respectively
- b = Total number of electronically submitted transmissions/LSRs received in the reporting period by EDI or TAG respectively

Report Structure

- · CLEC Aggregate
- · CLEC Specific/Aggregator
- Geographic Scope
 - Region

Note: The Order calls for Mechanized, Partially Mechanized, and Totally Mechanized, however, the Acknowledgement message is generated before the system recognizes whether this electronic transmission will be partially or fully mechanized.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month Record of Functional Acknowledgements	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• EDI	Benchmark: 100%
• TAG	

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
• EDI	Benchmark: 100%
• TAG	



O-3: Percent Flow-Through Service Requests (Summary)

Definition

The percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual intervention.

Exclusions

- · Fatal Rejects
- · Auto Clarification
- · Manual Fallout
- · CLEC System Fallout
- · Scheduled OSS Maintenance

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and two types of service: Resale, and Unbundled Network Elements (UNE). The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions:

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

- 1. Complex*
- 2. Special pricing plans
- 3. Some Partial migrations
- 4. New telephone number not yet posted to BOCRIS
- 5. Pending order review required
- CSR inaccuracies such as invalid or missing CSR data in CRIS
- 7. Expedites (requested by the CLEC)

- Denials-restore and conversion, or disconnect and conversion orders
- Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- 12. Transfer of calls option for the CLEC end users
- 13. Directory Listings (Indentions and Captions)

*See LSR Flow-Through Matrix following O-6 for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

Georgia Performance Metrics

Calculation

Percent Flow Through = $a \div [b - (c + d + e + f)] \times 100$

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c =the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status.

Percent Achieved Flow Through = $a \div [b-(c+d+e)] \times 100$

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued.
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

- · CLEC Aggregate
 - Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance:
Report Month	Report Month
Total Number of LSRs Received, by Interface, by CLEC	Total Number of Errors By Type
- TAG	- Bellsouth System Error
- EDI	
- LENS	
Total Number of Errors by Type, by CLEC	
- Fatal Rejects	
- Auto Clarification	
- CLEC Caused System Fallout	
Total Number of Errors by Error Code	
Total Fallout for Manual Processing	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	Benchmark: 90%
• UNE	Benchmark: 85%
• LNP	Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."

SEEM Measure

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	Benchmark: 90%
• UNE	Benchmark: 85%
• LNP	Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."



O-4: Percent Flow-Through Service Requests (Detail)

Definition

A detailed list, by CLEC, of the percentage of Local Service Requests (LSR) and LNP Local Service Requests (LNP LSRs) submitted electronically via the CLEC mechanized ordering process that flow through and reach a status for a FOC to be issued, without manual or human intervention.

Exclusions

- · Fatal Rejects
- · Auto Clarification
- · Manual Fallout
- CLEC System Fallout
- · Scheduled OSS Maintenance

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued, without manual intervention. These LSRs can be divided into two classes of service: Business and Residence, and three types of service: Resale, and Unbundled Network Elements (UNE). The CLEC mechanized ordering process does not include LSRs, which are submitted manually (for example, fax and courier) or are not designed to flow through (for example, Manual Fallout.)

Definitions:

Fatal Rejects: Errors that prevent an LSR, submitted electronically by the CLEC, from being processed further. When an LSR is submitted by a CLEC, LEO/LNP Gateway will perform edit checks to ensure the data received is correctly formatted and complete. For example, if the PON field contains an invalid character, LEO/LNP Gateway will reject the LSR and the CLEC will receive a Fatal Reject.

Auto-Clarification: Clarifications that occur due to invalid data within the LSR. LESOG/LAUTO will perform data validity checks to ensure the data within the LSR is correct and valid. For example, if the address on the LSR is not valid according to RSAG, or if the LNP is not available for the NPA NXXX requested, the CLEC will receive an Auto-Clarification.

Manual Fallout: Planned Fallout that occur by design. Certain LSRs are designed to fallout of the Mechanized Order Process due to their complexity. These LSRs are manually processed by the LCSC. When a CLEC submits an LSR, LESOG/LAUTO will determine if the LSR should be forwarded to LCSC for manual handling. Following are the categories for Manual Fallout:

- Complex*
- 2. Special pricing plans
- 3. Some Partial migrations
- 4. New telephone number not yet posted to BOCRIS
- 5. Pending order review required
- 6. CSR inaccuracies such as invalid or missing CSR data in
- 7. Expedites (requested by the CLEC)

- Denials-restore and conversion, or disconnect and conversion orders
- Class of service invalid in certain states with some types of service
- 10. Low volume such as activity type "T" (move)
- 11. More than 25 business lines, or more than 15 loops
- 12. Transfer of calls option for the CLEC end users
- 13. Directory Listings (Indentions and Captions)

*See LSR Flow-Through Matrix following O-6 for a list of services, including complex services, and whether LSRs issued for the services are eligible to flow through.

Total System Fallout: Errors that require manual review by the LSCS to determine if the error is caused by the CLEC, or is due to BellSouth system functionality. If it is determined the error is caused by the CLEC, the LSR will be sent back to the CLEC for clarification. If it is determined the error is BellSouth caused, the LCSC representative will correct the error, and the LSR will continue to be processed.

(A) **BELLSOUTH**®

Z Status: LSRs that receive a supplemental LSR submission prior to final disposition of the original LSR.

Calculation

Percent Flow Through = $a \div [b - (c + d + e + f)] \times 100$

- a = The total number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c =the number of LSRs that fall out for manual processing
- d = the number of LSRs that are returned to the CLEC for clarification
- e = the number of LSRs that contain errors made by CLECs
- f = the number of LSRs that receive a Z status.

Percent Achieved Flow Through = $a \div [b-(c+d+e)] \times 100$

- a = the number of LSRs that flow through LESOG/LAUTO and reach a status for a FOC to be issued.
- b = the number of LSRs passed from LEO/LNP Gateway to LESOG/LAUTO
- c = the number of LSRs that are returned to the CLEC for clarification
- d = the number of LSRs that contain errors made by CLECs
- e = the number of LSRs that receive Z status

Report Structure

Provides the flow through percentage for each CLEC (by alias designation) submitting LSRs through the CLEC mechanized ordering process. The report provides the following:

- CLEC (by alias designation)
- Number of fatal rejects
- · Mechanized interface used
- · Total mechanized LSRs
- · Total manual fallout
- Number of auto clarifications returned to CLEC
- · Number of validated LSRs
- · Number of BellSouth caused fallout
- · Number of CLEC caused fallout
- · Number of Service Orders Issued
- · Base calculation
- · CLEC error excluded calculation

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
Total Number of LSRs Received, by Interface, by CLEC	Total Number of Errors by Type
- TAG	- Bellsouth System Error
- EDI	·
- LENS	
Total Number of Errors by Type, by CLEC	
- Fatal Rejects	
- Auto Clarification	
- CLEC Errors	
Total Number of Errors by Error Code	
Total Fallout for Manual Processing	

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
Residence	Benchmark: 95%

(4) BELLSOUTH®

SQM Level of Disaggregation	Retail Analog/Benchmark ^a
Business	Benchmark: 90%
• UNE	• Benchmark: 85%
• LNP	Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark ^a
Residence	Benchmark: 95%
Business	Benchmark: 90%
• UNE	Benchmark: 85%
• LNP	Benchmark: 85%

a. Benchmarks do not apply to the "Percent Achieved Flow Through."

(A) **BELL**SOUTH

O-5: Flow-Through Error Analysis

Definition

An analysis of each error type (by error code) that was experienced by the LSRs that did not flow through or reached a status for a FOC to be issued.

Exclusions

Each Error Analysis is error code specific, therefore exclusions are not applicable.

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

Total for each error type.

Report Structure

Provides an analysis of each error type (by error code). The report is in descending order by count of each error code and provides the following:

- Error Type (by error code)
- · Count of each error type
- · Percent of each error type
- · Cumulative percent
- · Error Description
- · CLEC Caused Count of each error code
- · Percent of aggregate by CLEC caused count
- · Percent of CLEC caused count
- · BellSouth Caused Count of each error code
- · Percent of aggregate by BellSouth caused count
- · Percent of BellSouth by BellSouth caused count.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Total Number of LSRs Received Total Number of Errors by Type (by error code) CLEC Caused Error 	Report Month Total Number of Errors by Type (by error code) BellSouth System Error

SQM Level of Disaggregation	Retail Analog/Benchmark
Not Applicable	Not Applicable



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



(A) **BELLSOUTH** ®

O-6: CLEC LSR Information

Definition

A list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period.

Exclusions

- Fatal Rejects
- · LSRs submitted manually

Business Rules

The CLEC mechanized ordering process includes all LSRs, including supplements (subsequent versions) which are submitted through one of the three gateway interfaces (TAG, EDI, and LENS), that flow through and reach a status for a FOC to be issued. The CLEC mechanized ordering process does not include LSRs which are submitted manually (for example, fax and courier).

Calculation

NA

Report Structure

Provides a list with the flow through activity of LSRs by CC, PON and Ver, issued by each CLEC during the report period with an explanation of the of the columns and content. This report is available on a CLEC specific basis. The report provides the following for each LSR.

- CC
- PON
- Ver
- Timestamp
- Type
- Err #
- · Note or Error Description

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report Month Record of LSRs Received by CC, PON and Ver Record of Timestamp, Type, Err # and Note or Error Description for each LSR by CC, PON and Ver 	Not Applicable

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Not Applicable	Not Applicable

SEEM Measure

	SEEM Me	easure
	Tier I	
No	Tier II	
	Tier III	

(4) BELLSOUTH®

O-6: CLEC LSR Information

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

BELLS(

			LSK FIOW-I	LSR Flow-Inrough Matrix				
PRODUCT	F/T ³	COM PLEX SERVICE	COM PLEX ORDER	PLANNED FALLOUT FOR MANUAL HANDLING ¹	EDI	TAG ²	LENS ⁴	COMMENTS
2 wire analog DID trunk port	No	UNE	Yes	NA	z	z	z	
2 wire analog port	Yes	UNE	No	No	Y	Y	z	
2 wire ISDN digital line side port	No	UNE	Yes	NA	z	z	z	
2 wire ISDN digital loop	Yes	UNE	Yes	No	Y	Y	z	
3 Way Calling	Yes	No	No	No	Y	Y	Y	
4 wire analog voice grade loop	Yes	UNE	Yes	No	Y	Y	z	
4 wire DS0 & PRI digital loop	No	UNE	Yes	NA	z	z	z	
4 wire DS1 & PRI digital loop	No	UNE	Yes	NA	z	z	z	
4 wire ISDN DSI digital trunk ports	No	UNE	Yes	NA	z	z	z	
Accupulse	No	Yes	Yes	NA	z	z	z	
ADSL	Yes	UNE	No	No	Y	Y	z	
Area Plus	Yes	No	No	No	Y	Y	Y	
Basic Rate ISDN	No	Yes	Yes	Yes	Y	Y	z	
Call Block	Yes	No	No	No	Y	Y	Y	
Call Forwarding-Variable	Yes	No	No	No	Y	Y	Y	
Call Return	Yes	No	$N_{\rm O}$	No	Y	Y	Y	
Call Selector	Yes	No	No	No	Y	Y	Y	
Call Tracing	Yes	No	No	No	Y	Y	Y	
Call Waiting	Yes	No	No	No	Y	Y	Y	
Call Waiting Deluxe	Yes	No	No	No	Y	Y	Y	
Caller ID	Yes	No	No	No	Y	Y	Y	
CENTREX	No	Yes	Yes	NA	z	Z	z	
DID WITH PBX ACT W	No	Yes	Yes	Yes	Y	z	Y	
DID ACT W	oN	Yes	Yes	Yes	Y	Z	Y	
Digital Data Transport	No	UNE	Yes	NA	z	Z	z	
Directory Listing Indentions	No	No	No	Yes	Y	Y	Y	
Directory Listings Captions	No	No	Yes	Yes	Y	Y	Y	
Directory Listings (simple)	Yes	No	No	No	Y	Y	Y	
DS3	No	UNE	Yes	NA	z	Z	Z	
DSI Loop	Yes	UNE	Yes	No	Y	Y	z	

BELLS(

Matrix
Through
Flow-
LSR

			LON LIOW-I	LON FIOW-IIIIOUGII Maulia				
PRODUCT	F/T³	COM PLEX SERVICE	COM PLEX ORDER	PLANNED FALLOUT FOR MANUAL HANDLING ¹	EDI	TAG ²	LENS ⁴	COMMENTS
DSO Loop	Yes	UNE	Yes	No	Y	Y	z	
Enhanced Caller ID	Yes	No	No	No	Y	Y	Y	
ESSX	No	Yes	Yes	NA	Z	z	z	
Flat Rate/Business	Yes	No	No	No	Y	Y	Y	
Flat Rate/Residence	Yes	No	No	No	Y	Y	Y	
FLEXSERV	No	Yes	Yes	NA	Z	z	z	
Frame Relay	No	Yes	Yes	NA	z	z	z	
FX	No	Yes	Yes	NA	z	z	z	
Ga. Community Calling	Yes	No	No	No	Y	Y	Y	
HDSL	Yes	UNE	No	No	Y	Y	z	
Hunting MLH	No	C/S ⁴	C/S	Yes	Y	Y	z	
Hunting Series Completion	Yes	C/S	C/S	No	Y	Y	Y	
INP to LNP Conversions	No	UNE	Yes	Yes	Y	Y	z	
LightGate	No	Yes	Yes	NA	z	z	z	
Line Sharing	Yes	UNE	No	No	Y	Y	z	
Local Number Portability	Yes	UNE	Yes	No	Y	Y	z	
LNP with Complex Listing	No	UNE	Yes	Yes	Y	Y	z	
LNP with Partial Migration	No	UNE	Yes	Yes	Y	Y	z	
LNP with Complex Services	No	UNE	Yes	Yes	Y	Y	z	
Loop+INP	Yes	UNE	No	No	Y	Y	z	
Loop+LNP	Yes	UNE	No	No	Y	Y	z	
Measured Rate/Bus.	Yes	No	No	No	Y	Y	Y	
Measured Rate/Res.	Yes	No	No	No	Y	Y	Y	
Megalink	No	Yes	Yes	NA	Z	Z	z	
Megalink-T1	No	Yes	Yes	NA	Z	Z	Z	
Memory Call	Yes	No	No	No	Y	Y	Y	
Memory Call Ans. Svc.	Yes	No	No	No	Y	Y	Y	
Multiserv	No	Yes	Yes	NA	N	Z	Z	
Native Mode LAN Interconnection (NMLI)	No	Yes	Yes	NA	z	z	z	
Off-Prem Stations	No	Yes	Yes	NA	z	z	Z	
Optional Calling Plan	Yes	No	No	No	Y	Y	Y	

LSR Flow-Through Matrix

×
Ξ
¥
₩
_
_
0
Ž
5
ᡓ
F
٦,
₹
<u> </u>
ш
~
፟
٣í

PRODUCT	F/T ³	COM PLEX SERVICE	COM PLEX ORDER	PLANNED FALLOUT FOR MANUAL HANDLING ¹	EDI	TAG ²	LENS ⁴	COMMENTS
Package/Complete Choice and area plus	Yes	No	No	No	Y	Y	Y	
Pathlink Primary Rate ISDN	No	Yes	Yes	NA	z	z	z	
Pay Phone Provider	No	No	No	NA	Z	z	z	
PBX Standalone ACT A,C, D	No	Yes	Yes	Yes	Y	Y	z	
PBX Trunks	No	Yes	Yes	Yes	Y	Y	z	
Port/Loop Combo	Yes	UNE	No	No	Y	Y	Y	
Port/Loop PBX	No	No	No	Yes	Y	Y	z	
Preferred Call Forward	Yes	No	No	No	Y	Y	Y	
RCF Basic	Yes	No	No	No	Y	Y	Y	
Remote Access to CF	Yes	No	No	No	Y	Y	Y	
Repeat Dialing	Yes	No	No	No	Y	Y	Y	
Ringmaster	Yes	No	No	No	Y	Y	Y	
Smartpath	No	Yes	Yes	NA	z	z	z	
SmartRING	No	Yes	Yes	NA	Z	z	z	
Speed Calling	Yes	No	No	No	Y	Y	Y	
Synchronet	No	Yes	Yes	Yes	Y	Y	z	
Tie Lines	No	Yes	Yes	NA	Z	z	z	
Touchtone	Yes	No	No	No	Y	Y	Y	
Unbundled Loop-Analog 2W, SL1, SL2	Yes	UNE	No	No	Y	Y	Y	
WATS	No	Yes	Yes	NA	Z	Z	Z	
XDSL	Yes	UNE	No	No	Y	Y	z	
XDSL Extended LOOP	No	UNE	Yes	NA	z	z	z	
Collect Call Block	Yes	No	No	No	Y	Y	Y	
900 Call Block	Yes	No	No	No	Y	Y	Y	
3rd Party Call Block	Yes	No	No	No	Y	Y	Y	
Three Way Call Block	Yes	No	No	No	Y	Y	Y	
PIC/LPIC Change	Yes	No	No	No	Y	Y	Y	
PIC/LPIC Freeze	Yes	No	No	No	Y	Y	Y	

Note 1: Planned Fallout for Manual Handling denotes those services that are electronically submitted and are not intended to flow through due to the complexity of the service.

Note²: The TAG column includes those LSRs submitted via Robo TAG.

LSR Flow-Through Matrix

Note³: For all services that indicate 'No' for flow-through, the following reasons, in addition to errors or complex services, also prompt manual handling: Expedites from CLECs, special pricing plans, denials restore and conversion or disconnect and conversion both required, partial migrations (although conversions-as-is flow through for issue 9), class of service invalid in certain states with some TOS e.g. government, or cannot be changed when changing main TN on C activity, low volume e.g. activity type T=move, pending order review required, more than 25 business lines, CSR inaccuracies such as invalid or missing CSR data in CRIS, Directory listings – Indentions, Directory listings – Captions, transfer of calls option for CLEC end user – new TN not yet posted to BOCRIS. Many are unique to the CLEC environment.

Note⁴: Services with C/S in the Complex Service and/or the Complex Order columns can be either complex or simple.

Note⁵: EELs are manually ordered.



O-7: Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) received which are rejected due to error or omission. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by the CLEC prior to being rejected/clarified.
- · Scheduled OSS Maintenance

Business Rules

Fully Mechanized: An LSR is considered "rejected" when it is submitted electronically but does not pass LEO edit checks in the ordering systems (EDI, LENS, TAG, LEO, LESOG) and is returned to the CLEC without manual intervention. There are two types of "Rejects" in the Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR but required fields are either not populated or incorrectly populated and the request is returned to the CLEC before it is considered a valid LSR.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. Fatal rejects are excluded from the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An **Auto Clarification** occurs when a valid LSR is electronically submitted but rejected from LESOG because it does not pass further edit checks for order accuracy.

Partially Mechanized: A valid LSR, which is electronically submitted (via EDI, LENS, TAG) but cannot be processed electronically and "falls out" for manual handling. It is then put into "clarification" and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs electronically submitted by the CLEC.

Non-Mechanized: LSRs which are faxed or mailed to the LCSC for processing and "clarified" (rejected) back to the CLEC by the BellSouth service representative.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Interconnection Purchasing Center (IPC). Trunk data is reported separately.

Calculation

Percent Rejected Service Requests = $(a \div b) \times 100$

- a = Total Number of Rejected Service Requests in the Reporting Period
- b = Total Number of Service Requests Received in the Reporting Period

Report Structure

- · Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- · CLEC Aggregate
- · Geographic Scope
 - State
 - Region
- Product Specific Percent Rejected
- · Total Percent Rejected

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Total Number of LSRs	
Total Number of Rejects	
State and Region	
Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Mechanized, Partially Mechanized and Non-Mechanized	Diagnostic
Resale - Residence	
Resale - Business	
Resale – Design (Special)	
Resale PBX	
Resale Centrex	
Resale ISDN	
LNP Standalone	
INP Standalone	
2W Analog Loop Design	
2W Analog Loop Non-Design	
2W Analog Loop w/INP Design	
• 2W Analog Loop w/INP Non-Design	
• 2W Analog Loop w/LNP Design	
• 2W Analog Loop w/LNP Non-Design	
• UNE Loop + Port Combinations	
• Switch Ports	
• UNE Combination Other	
UNE xDSL (ADSL, HDSL, UCL) Line Sharing	
• Line Sharing	
UNE ISDN Loop UNE Other Design	
 UNE Other Design UNE Other Non-Design	
Local Interoffice Transport	
Local Interoffice Transport Local Interconnection Trunks	
- Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



O-8: Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is submitted by the CLEC and passes edit checks to insure the data received is correctly formatted and complete.

Exclusions

- Service Requests canceled by CLEC prior to being rejected/clarified.
- Designated Holidays are excluded from the interval calculation.
- LSRs which are identified and classified as "Projects"
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

· Scheduled OSS Maintenance

Business Rules

Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is rejected (date and time stamp or reject in EDI, TAG or LENS). Auto Clarifications are considered in the Fully Mechanized category.

Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until it falls out for manual handling. The stop time on partially mechanized LSRs is when the LCSC Service Representative clarifies the LSR back to the CLEC via LENS, EDI, or TAG.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.

Non-Mechanized: The elapsed time from receipt of a valid LSR (date and time stamp of FAX or date and time mailed LSR is received in the LCSC) until notice of the reject (clarification) is returned to the CLEC via LON.

Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Local Interconnection Service Center (LISC). Trunk data is reported separately. All interconnection trunks are counted in the non-mechanized category.

Calculation

Reject Interval = (a - b)

- a = Date and Time of Service Request Rejection
- b = Date and Time of Service Request Receipt

Average Reject Interval = $(c \div d)$

- c = Sum of all Reject Intervals
- d = Number of Service Requests Rejected in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate



- Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- · Geographic Scope
 - State
- Region
- · Mechanized:
- $0 \leq 4 \text{ minutes}$
- >4 ≤ 8 minutes
- >8 \leq 12 minutes
- >12 \leq 60 minutes
- $0 \leq 1 \text{ hour}$
- $>1 \leq 4 \text{ hours}$
- >4 ≤ 8 hours
- $> 8 \le 12 \text{ hours}$
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- >20 \leq 24 hours
- >24 hours
- · Partially Mechanized:
 - $0 \leq 1 \text{ hour}$
- $>1 \leq 4$ hours
- $>4 \leq 8 \text{ hours}$
- $> 8 \le 10 \text{ hours}$
- $0 \leq 10 \text{ hours}$
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- $> 18 \le 24 \text{ hours}$
- >24 hours
- · Non-mechanized:
- $0 \leq 1 \text{ hour}$
- $>1 \leq 4 \text{ hours}$
- $>4 \leq 8 \text{ hours}$
- >8 \leq 12 hours
- >12 \leq 16 hours
- >16 \leq 20 hours >20 - \leq 24 hours
- $0 \leq 24$ hours
- > 24 hours
- Trunks:
 - \leq 4 days
- $>4 \le 8 \text{ days}$
- $> 8 \le 12 \text{ days}$
- $>12 \le 14 \text{ days}$
- $> 14 \le 20 \text{ days}$
- >20 days

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Reject Interval	
Total Number of LSRs	
Total Number of Rejects	
State and Region	
Total Number of ASRs (Trunks)	

BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
 Resale – Residence Resale – Business Resale – Design (Special) Resale PBX Resale Centrex Resale ISDN LNP Standalone INP Standalone 2W Analog Loop Design 2W Analog Loop Non-Design 2W Analog Loop w/INP Design 2W Analog Loop w/INP Non-Design 2W Analog Loop w/LNP Design 2W Analog Loop w/LNP Non-Design 2W Analog Loop w/LNP Non-Design UNE Loop + Port Combinations Switch Ports UNE Combination Other UNE xDSL (ADSL, HDSL, UCL) Line Sharing UNE Other Non-Design Local Interoffice Transport UNE Other Design 	Mechanized: - 97% within I Hour Partially Mechanized: - 85% within 24 hours - 85% within 18 Hours (05/01/01) - 85% within 10 Hours (08/01/01) Non-Mechanized: - 85% within 24 hours
Local Interconnection Trunks	Trunks: - 85% within 4 Days

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 97% ≤ 1 hour
Partially Mechanized	 85% within 24 hours 85% within 18 hours (05/01/01) 85% within 10 hours (08/01/01)
Non-Mechanized	85% within 24 hours



O-9: Firm Order Confirmation Timeliness

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of valid LSR to distribution of a Firm Order Confirmation.

Exclusions

- · Rejected LSRs
- Designated Holidays are excluded from the interval calculation.
- · LSRs which are identified and classified as "Projects"
- · The following hours for Partially Mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group - Monday through Saturday 7:00PM until 7:00AM

From 7:00 PM Saturday until 7:00 AM Monday.

Business Resale, Complex, UNE Groups - Monday through Friday 6:00PM until 8:00AM

From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute

· Scheduled OSS Maintenance

Business Rules

- Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is processed, appropriate service orders are generated and a Firm Order Confirmation is returned to the CLEC via EDI, LENS or TAG.
- Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS, or TAG) which falls out for manual handling until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is returned to the CLEC via EDI, LENS, or TAG.
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.
- Non-Mechanized: The elapsed time from receipt of a valid paper LSR (date and time stamp of FAX or date and time paper LSRs received in LCSC) until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is sent to the CLEC via LON.
- Interconnection Trunks: Interconnection Trunks are ordered on Access Service Requests (ASRs). ASRs are submitted to and processed by the Local Interconnection Service Center (LISC). Trunk data is reported separately.

Calculation

Firm Order Confirmation Interval = (a - b)

- a = Date & Time of Firm Order Confirmation
- b = Date & Time of Service Request Receipt)

Average FOC Interval = $(c \div d)$

- c = Sum of all FOC Intervals
- d = Total Number of Service Requests Confirmed in Reporting Period

FOC Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Service Requests Confirmed in interval
- f = Total Service Requests Confirmed in the Reporting Period

O-9: Firm Order Confirmation Timeliness



Georgia Performance Metrics

Report Structure

- · Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- CLEC Aggregate
- · Geographic Scope
- State
- Region
- · Fully Mechanized:
 - $0 \leq 15 \text{ minutes}$
- $>15 \leq 30 \text{ minutes}$
- $>30 \leq 45 \text{ minutes}$
- >45 \leq 60 minutes
- $>60 \leq 90 \text{ minutes}$
- >90 \leq 120 minutes
- >120 \leq 180 minutes
- $0 \leq 3 \text{ hours}$
- $>3 \leq 6$ hours
- $>6 \le 12 \text{ hours}$
- $> 12 \le 24 \text{ hours}$
- $>24 \le 48$ hours
- >48 hours
- Partially Mechanized:
 - $0 \leq 4 \text{ hours}$
- $>4 \le 8 \text{ hours}$
- $> 8 \le 10 \text{ hours}$
- $0 \leq 10 \text{ hours}$
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- $> 18 \le 24 \text{ hours}$
- $0 \leq 24 \text{ hours}$
- >24 \leq 48 hours
- >48 hours
- · Non-Mechanized
- $0 \leq 4 \text{ hours}$
- $>4 \leq 8 \text{ hours}$
- >8 \leq 12 hours
- >12 \leq 16 hours
- >16 \leq 20 hours
- >20 \leq 24 hours
- >24 \leq 36 hours
- $0 \leq 36 \text{ hours}$
- >36 \leq 48 hours
- >48 hours
- Trunks:
 - $0 \leq 5 \text{ days}$
- >5 \leq 10 days
- $0 \leq 10 \text{ days}$
- $>10 \le 15 \text{ days}$
- >15 ≤ 20 days >20 days

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Not Applicable
Interval for FOC	
Total Number of LSRs	
State and Region	
Total Number of ASRs (Trunks)	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
 Resale – Residence Resale – Business Resale – Design (Special) Resale PBX Resale Centrex Resale ISDN LNP Standalone INP Standalone 2W Analog Loop Design 2W Analog Loop w/INP Design 2W Analog Loop w/INP Design 2W Analog Loop w/INP Non-Design 2W Analog Loop w/LNP Design 2W Analog Loop w/LNP Non-Design WINE Loop + Port Combinations Switch Ports UNE Combination Other UNE Combination Other UNE ISDN Loops UNE Other Design UNE Other Design Local Interoffice Transport 	Mechanized: - 95% within 3 Hours Partially Mechanized: 85% within 24 hours 85% within 18 Hours (05/01/01) 85% within 10 Hours (08/01/01) Non-Mechanized: - 85% within 36 hours
Local Interconnection Trunks	Trunks: - 95% within 10 days

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% within 3 hours
Partially Mechanized	• 85% within 24 hours
	85% within 18 Hours (05/01/01) 85% within 10 Hours (08/01/01)



SEEM Disaggregation	SEEM Analog/Benchmark
Non-Mechanized	85% within 36 hours
IC Trunks	95% within 10 days



O-10: Service Inquiry with LSR Firm Order Confirmation (FOC) Response Time Manual¹

Definition

This report measures the interval and the percent within the interval from the submission of a Service Inquiry (SI) with Firm Order LSR to the distribution of a Firm Order Confirmation (FOC).

Exclusions

- Designated Holidays are excluded from the interval calculation.
- Weekend hours from 5:00PM Friday until 8:00AM Monday are excluded from the interval calculation of the Service Inquiry.
- · Canceled Requests
- · Electronically Submitted Requests
- · Scheduled OSS Maintenance

Business Rules

This measurement combines four intervals:

- 1. From receipt of Service Inquiry with LSR to hand off to the Service Advocacy Center (SAC) for Loop 'Look-up'.
- 2. From SAC start date to SAC complete date.
- 3. From SAC complete date to the Complex Resale Support Group (CRSG) complete date with hand off to LCSC.
- 4. From receipt of SI/LSR in the LCSC to Firm Order Confirmation.

Calculation

FOC Timeliness Interval = (a - b)

- a = Date and Time Firm Order Confirmation (FOC) for SI with LSR returned to CLEC
- b = Date and Time SI with LSR received

Average Interval = $(c \div d)$

- c = Sum of all FOC Timeliness Intervals
- d = Total number of SIs with LSRs received in the reporting period

Percent Within Interval = $(e \div f) \times 100$

- e = Total number of Service Inquiries with LSRs received by the CRSG to distribution of FOC by the Local Carrier Service Center (LCSC)
- f = Total number of Service Inquiries with LSRs received in the reporting period

Report Structure

- · CLEC Aggregate
- CLEC Specific
- Geographic Scope
 - State
 - Region
- Intervals
 - $0 \leq 3 \text{ days}$
- $>3 \le 5$ days $0 \le 5$ days
- $>5-\leq 7$ days
- $>7-\leq 10 \text{ days}$
- $>10 \le 15 \text{ days}$
- >15 days
- Average Interval measured in days

1. See O-9 for FOC Timeliness

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month Total Number of Requests	Not Applicable
• SI Intervals	
State and Region	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
 xDSL (includes UNE unbundled ADSL, HDSL and UNE Unbundled Copper Loops) Unbundled Interoffice Transport 	95% Returned within 5 Business days

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELLSOUTH** *

O-11: Firm Order Confirmation and Reject Response Completeness

Definition

A response is expected from BellSouth for every Local Service Request transaction (version). More than one response or differing responses per transaction is not expected. Firm Order Confirmation and Reject Response Completeness is the corresponding number of Local Service Requests received to the combination of Firm Order Confirmation and Reject Responses.

Exclusions

- · Service Requests canceled by the CLEC prior to FOC or Rejected/Clarified
- Non-Mechanized LSRs
- · Scheduled OSS Maintenance

Business Rules

Mechanized – The number of FOCs or Auto Clarifications sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG).

Partially Mechanized – The number of FOCs or Rejects sent to the CLEC from LENS, EDI, TAG in response to electronically submitted LSRs (date and time stamp in LENS, EDI, TAG), which fall out for manual handling by the LCSC personnel.

Total Mechanized - The number of the combination of Fully Mechanized and Partially Mechanized LSRs

Non-Mechanized – The number of FOCs or Rejects sent to the CLEC via FAX Server in response to manually submitted LSRs (date and time stamp in FAX Server).

Note: Manual (Non-Mechanized) LSRs have no version control by the very nature of the manual process, therefore, non-mechanized LSRs are not captured by this report.

For CLEC Results:

Firm Order Confirmation and Reject Response Completeness is determined in two dimensions:

Percent responses is determined by computing the number of Firm Order Confirmations and Rejects transmitted by BellSouth and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Percent of multiple responses is determined by computing the number of Local Service Request unique versions receiving more than one Firm Order Confirmation, Reject or the combination of the two and dividing by the number of Local Service Requests (all versions) received in the reporting period.

Calculation

Single FOC/Reject Response Expected

Firm Order Confirmation / Reject Response Completeness = $(a \div b) \times 100$

- a = Total Number of Service Requests for which a Firm Order Confirmation or Reject is Sent
- b = Total Number of Service Requests Received in the Report Period

Multiple or Differing FOC / Reject Responses Not Expected

Response Completeness = $[(a + b) \div c] \times 100$

- a = Total Number of Firm Order Confirmations Per LSR Version
- b = Total Number of Reject Responses Per LSR Version
- c = Total Number of Service Requests (All Versions) Received in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- · State and Region
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Specific

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Reject Interval	
Total Number of LSRs	
Total Number of Rejects	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	95% Returned
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
LNP Standalone	
INP Standalone	
2W Analog Loop Design	
• 2W Analog Loop Non – Design	
2W Analog Loop w/ INP Design	
• 2W Analog Loop w/ INP Non – Design	
2W Analog Loop w/ LNP Design	
• 2W Analog Loop w/ LNP Non – Design	
UNE Loop and Port Combinations	
Switch Ports	
UNE Combination Other	
UNE xDSL (ADSL, HDSL, UCL)	
Line Sharing	
UNE ISDN Loops	
UNE Other Design	
UNE Other Non - Design	
Local Interoffice Transport	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Fully Mechanized	• 95% Returned

(A) **BELLSOUTH** *

O-12: Speed of Answer in Ordering Center

Definition

Measures the average time a customer is in queue.

Exclusions

None

Business Rules

The clock starts when the appropriate option is selected (i.e., 1 for Resale Consumer, 2 for Resale Multiline, and 3 for UNE-LNP, etc.) and the call enters the queue for that particular group in the LCSC. The clock stops when a BellSouth service representative in the LCSC answers the call. The speed of answer is determined by measuring and accumulating the elapsed time from the entry of a CLEC call into the BellSouth automatic call distributor (ACD) until a service representative in BellSouth's Local Carrier Service Center (LCSC) answers the CLEC call.

Calculation

Speed of Answer in Ordering Center = $(a \div b)$

- a = Total seconds in queue
- b = Total number of calls answered in the Reporting Period

Report Structure

Aggregate

- CLEC Local Carrier Service Center
- · BellSouth
- Business Service Center
- Residence Service Center

Note: Combination of Residence Service Center and Business Service Center data.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Mechanized tracking through LCSC Automatic Call Distributor	Mechanized tracking through BellSouth Retail center support system.

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Aggregate CLEC – Local Carrier Service Center BellSouth Business Service Center Residence Service Center	Parity with Retail

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



O-13: LNP-Percent Rejected Service Requests

Definition

Percent Rejected Service Request is the percent of total Local Service Requests (LSRs) which are rejected due to error or omission. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete, i.e., fatal rejects are never accepted and, therefore, are not included.

Exclusions

- · Service Requests canceled by the CLEC
- · Scheduled OSS Maintenance

Business Rules

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR (via EDI or TAG) but required fields are not populated correctly and the request is returned to the CLEC.

Fatal rejects are reported in a separate column, and for informational purposes ONLY. They are not considered in the calculation of the percent of total LSRs rejected or the total number of rejected LSRs.

An **Auto Clarification** is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which is electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back (rejected) to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

LNP-Percent Rejected Service Requests = $(a \div b) \times 100$

- a = Number of Service Requests Rejected in the Reporting Period
- b = Number of Service Requests Received in the Reporting Period

Report Structure

- · Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized
- CLEC Specific
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Not Applicable	Not Applicable

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
LNP UNE Loop w/LNP	Diagnostic



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



O-14: LNP-Reject Interval Distribution & Average Reject Interval

Definition

Reject Interval is the average reject time from receipt of an LSR to the distribution of a Reject. An LSR is considered valid when it is electronically submitted by the CLEC and passes LNP Gateway edit checks to insure the data received is correctly formatted and complete.

Exclusions

- · Service Requests canceled by the CLEC
- Designated Holidays are excluded from the interval calculation.
- · LSRs which are identified and classified as "Projects".
- The following hours for Partially mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group – Monday through Saturday 7:00PM until 7:00AM From 7:00 PM Saturday until 7:00 AM Monday

Business Resale, Complex, UNE Groups – Monday through Friday 6:00PM until 8:00AM From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

· Scheduled OSS Maintenance

Business Rules

The Reject interval is determined for each rejected LSR processed during the reporting period. The Reject interval is the elapsed time from when BellSouth receives LSR until that LSR is rejected back to the CLEC. Elapsed time for each LSR is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of rejected LSRs to produce the reject interval distribution.

An LSR is considered "rejected" when it is submitted electronically but does not pass edit checks in the ordering systems (EDI, TAG, LNP Gateway, LAUTO) and is returned to the CLEC without manual intervention.

Fully Mechanized: There are two types of "Rejects" in the Fully Mechanized category:

A **Fatal Reject** occurs when a CLEC attempts to electronically submit an LSR but required fields are not populated correctly and the request is returned to the CLEC.

An **Auto Clarification** is a valid LSR which is electronically submitted (via EDI or TAG), but is rejected from LAUTO because it does not pass further edit checks for order accuracy. Auto Clarifications are returned without manual intervention.

Partially Mechanized: A valid LSR which electronically submitted (via EDI or TAG), but cannot be processed electronically due to a CLEC error and "falls out" for manual handling. It is then put into "clarification", and sent back to the CLEC.

Total Mechanized: Combination of Fully Mechanized and Partially Mechanized rejects.

Non-Mechanized: A valid LSR which is faxed or mailed to the BellSouth LCSC.

Calculation

Reject Interval = (a - b)

- a = Date & Time of Service Request Rejection
- b = Date & Time of Service Request Receipt

Average Reject Interval = $(c \div d)$

- c = Sum of all Reject Intervals
- d = Total Number of Service Requests Rejected in Reporting Period

(A) **BELLSOUTH** *

Reject Interval Distribution = $(e \div f) \times 100$

- e = Service Requests Rejected in reported interval
- f = Total Number of Service Requests Rejected in Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- · CLEC Specific
- · CLEC Aggregate
- · State, Region
- · Fully Mechanized:
- $0 \leq 4 \text{ minutes}$
- $>4 \leq 8$ minutes
- >8 \leq 12 minutes
- >12 \leq 60 minutes
- $0 \leq 1 \text{ hour}$
- $>1 \leq 4$ hours
- $>4 \leq 8 \text{ hours}$
- >8 ≤ 12 hours
- $> 12 \le 16 \text{ hours}$
- $> 16 \le 20 \text{ hours}$
- >20 \leq 24 hours
- > 24 hours
- · Partially Mechanized:
- $0 \leq 1 \text{ hour}$
- >1 \leq 4 hours
- >4 ≤ 8 hours
- $> 8 \le 10 \text{ hours}$
- $0 \leq 10 \text{ hours}$
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- >18 \leq 24 hours
- > 24 hours
- · Non-Mechanized:
 - $0 \leq 1 \text{ hour}$
- $>1 \leq 4 \text{ hours}$
- $>4 \leq 8 \text{ hours}$
- >8 ≤ 12 hours
- $> 12 \le 16 \text{ hours}$ $>16 - \le 20 \text{ hours}$
- >20 \leq 24 hours
- $0 \leq 24 \text{ hours}$
- >24 hours
- · Average Interval in Days or Hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Reject Interval	
Total Number of LSRs	
Total number of Rejects	
State and Region	

BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
LNP UNE Loop with LNP	 Mechanized: 97% within I Hour Partially Mechanized: 85% within 24 Hours Partially Mechanized: 85% within 18 Hours (05/01/01) Partially Mechanized: 85% within 10 Hours (08/01/01) Non-Mechanized: 85% within 24 Hours

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



O-15: LNP-Firm Order Confirmation Timeliness Interval Distribution & Firm Order Confirmation Average Interval

Definition

Interval for Return of a Firm Order Confirmation (FOC Interval) is the average response time from receipt of a valid LSR to distribution of a firm order confirmation.

Exclusions

- · Rejected LSRs
- Designated Holidays are excluded from the interval calculation.
- LSRs which are identified and classified as "Projects".
- The following hours for Partially Mechanized and Non-mechanized LSRs are excluded from the interval calculation:

Residence Resale Group - Monday through Saturday 7:00PM until 7:00AM

From 7:00 PM Saturday until 7:00 AM Monday.

Business Resale, Complex, UNE Groups - Monday through Friday 6:00PM until 8:00AM

From 6:00 PM Friday until 8:00 AM Monday.

The hours excluded will be altered to reflect changes in the Center operating hours. The LCSC will accept faxed LSRs only during posted hours of operation.

The interval will be the amount of time accrued from receipt of the LSR until normal closing of the center if an LSR is worked using overtime hours.

In the case of a Partially Mechanized LSR received and worked after normal business hours, the interval will be set at one (1) minute.

• Scheduled OSS Maintenance.

Business Rules

- Fully Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS or TAG) until the LSR is processed, appropriate service orders are generated and a Firm Order Confirmation is returned to the CLEC via EDI, LENS or TAG.
- Partially Mechanized: The elapsed time from receipt of a valid electronically submitted LSR (date and time stamp in EDI, LENS, or TAG) which falls out for manual handling until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is returned to the CLEC via EDI, LENS, or TAG.
- Total Mechanized: Combination of Fully Mechanized and Partially Mechanized LSRs which are electronically submitted by the CLEC.
- Non-Mechanized: The elapsed time from receipt of a valid paper LSR (date and time stamp of FAX or date and time paper LSRs received in LCSC) until appropriate service orders are issued by a BellSouth service representative via Direct Order Entry (DOE) or Service Order Negotiation Generation System (SONGS) to SOCS and a Firm Order Confirmation is sent to the CLEC via LON.

Calculation

Firm Order Confirmation Interval = (a - b)

- a = Date & Time of Firm Order Confirmation
- b = Date & Time of Service Request Receipt)

Average FOC Interval = $(c \div d)$

- c = Sum of all FOC Intervals
- d = Total Number of Service Requests Confirmed in Reporting Period

FOC Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Service Requests Confirmed in interval
- f = Total Service Requests Confirmed in the Reporting Period

Report Structure

Fully Mechanized, Partially Mechanized, Total Mechanized, Non-Mechanized

- CLEC Specific
- · CLEC Aggregate
- · State and Region
- Fully Mechanized:
- $0 \leq 15$ minutes
- >15 \leq 30 minutes
- >30 \leq 45 minutes
- >45 \leq 60 minutes
- $>60 \le 90 \text{ minutes}$
- >90 \leq 120 minutes
- $> 120 \le 180 \text{ minutes}$
- $0 \leq 3$ hours
- >3 \leq 6 hours
- $>6 \le 12 \text{ hours}$
- $> 12 \le 24 \text{ hours}$
- >24 \leq 48 hours
- >48 hours
- Partially Mechanized:
 - $0 \leq 4$ hours
- $>4 \leq 8 \text{ hours}$
- $> 8 \le 10 \text{ hours}$
- $0 \leq 10 \text{ hours}$
- $> 10 \le 18 \text{ hours}$
- $0 \leq 18 \text{ hours}$
- $> 18 \le 24 \text{ hours}$
- $0 \leq 24 \text{ hours}$
- >24 \leq 48 hours
- > 48 hours
- · Non-Mechanized:
- $0 \leq 4 \text{ hours}$
- $>4 \le 8 \text{ hours}$
- >8 ≤ 12 hours
- >12 **-** ≤ 16 hours
- $> 16 \le 20 \text{ hours}$
- >20 \leq 24 hours
- >24 \leq 36 hours
- $0 \leq 36 \text{ hours}$
- >36 \leq 48 hours
- >48 hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Total Number of LSRs	
Total Number of FOCs	
State and Region	

BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
• LNP	Mechanized: 95% within 3 Hours
UNE Loop with LNP	Partially Mechanized: 85% within 24 Hours
	• Partially Mechanized: 85% within 18 Hours (05/01/01)
	• Partially Mechanized: 85% within 10 Hours (08/01/01)
	Non-Mechanized: 85% within 36 hours

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 3: Provisioning

P-1: Mean Held Order Interval & Distribution Intervals

Definition

When delays occur in completing CLEC orders, the average period that CLEC orders are held for BellSouth reasons, pending a delayed completion, should be no worse for the CLEC when compared to BellSouth delayed orders. Calculation of the interval is the total days orders are held and pending but not completed that have passed the currently committed due date; divided by the total number of held orders. This report is based on orders still pending, held and past their committed due date at the close of the reporting period. The distribution interval is based on the number of orders held and pending but not completed over 15 and 90 days. (Orders reported in the >90 day interval are also included in the >15 day interval.)

Exclusions

- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D) & From (F) orders
- · Orders with appointment code of 'A' for Rural orders.

Business Rules

Mean Held Order Interval: This metric is computed at the close of each report period. The held order interval is established by first identifying all orders, at the close of the reporting interval, that both have not been reported as completed in SOCS and have passed the currently committed due date for the order. For each such order, the number of calendar days between the earliest committed due date on which BellSouth had a company missed appointment and the close of the reporting period is established and represents the held order interval for that particular order. The held order interval is accumulated by the standard groupings, unless otherwise noted, and the reason for the order being held. The total number of days accumulated in a category is then divided by the number of held orders within the same category to produce the mean held order interval. The interval is by calendar days with no exclusions for Holidays or Sundays.

CLEC Specific reporting is by type of held order (facilities, equipment, other), total number of orders held, and the total and average days.

Held Order Distribution Interval: This measure provides data to report total days held and identifies these in categories of >15 days and >90 days. (Orders counted in >90 days are also included in >15 days).

Calculation

Mean Held Order Interval = $a \div b$

- a = Sum of held-over-days for all Past Due Orders Held for the reporting period
- b = Number of Past Due Orders Held and Pending But Not Completed and past the committed due date

Held Order Distribution Interval (for each interval) = $(c \div d) \times 100$

- c = # of Orders Held for ≥ 15 days or # of Orders Held for ≥ 90 days
- d = Total # of Past Due Orders Held and Pending But Not Completed)

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Circuit Breakout < 10, ≥ 10 (except trunks)

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
CLEC Order Number and PON (PON)	BellSouth Order Number
Order Submission Date (TICKET_ID)	Order Submission Date
Committed Due Date (DD)	Committed Due Date
Service Type (CLASS_SVC_DESC)	Service Type
Hold Reason	Hold Reason
Total line/circuit count	Total line/circuit count
Geographic Scope	Geographic Scope
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
2W Analog Loop-Non-Design	Retail Residence and Business - POTS Excluding Switch- Based Orders
2W Analog Loop w/LNP - Design	Retail Residence and Business Dispatch
2W Analog Loop w/LNP- Non-Design	Retail Residence and Business - POTS Excluding Switch- Based Orders
2W Analog Loop w/INP-Design	Retail Residence and Business Dispatch
2W Analog Loop w/INP-Non-Design	Retail Residence and Business - POTS Excluding Switch- Based Orders
UNE Digital Loop < DS1	Retail Digital Loop < DS1
UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations	Retail Residence and Business
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
UNE ISDN	Retail ISDN - BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non-Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

P-1: Mean Held Order Interval & Distribution Intervals



Georgia Performance Metrics

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-2: Average Jeopardy Notice Interval & Percentage of Orders Given **Jeopardy Notices**

Definition

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC.

The interval is from the date/time the notice is released to the CLEC/BellSouth systems until 5pm on the commitment date of the order. The Percent of Orders is the percentage of orders given jeopardy notices for facility delay in the count of orders confirmed in the report period.

Exclusions

- · Orders held for CLEC end user reasons
- Disconnect (D) & From (F) orders
- · Non-Dispatch Orders

Business Rules

When BellSouth can determine in advance that a committed due date is in jeopardy for facility delay, it will provide advance notice to the CLEC. The number of committed orders in a report period is the number of orders that have a due date in the reporting period. Jeopardy notices for interconnection trunks results are usually zero as these trunks seldom experience facility delays. The Committed due date is considered the Confirmed due date. This report measures dispatched orders only. If an order is originally sent as nondispatch and it is determined there is a facility delay, the order is converted to a dispatch code so the facility problem can be corrected. It will remain coded dispatched until completion.

Calculation

Jeopardy Interval = a - b

- a = Date and Time of Jeopardy Notice
- b = Date and Time of Scheduled Due Date on Service Order

Average Jeopardy Interval = $c \div d$

- c = Sum of all jeopardy intervals
- d = Number of Orders Notified of Jeopardy in Reporting Period

Percent of Orders Given Jeopardy Notice = (e ÷ f) X 100

- e = Number of Orders Given Jeopardy Notices in Reporting Period
- f = Number of Orders Confirmed (due) in Reporting Period)

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Dispatch Orders
- · Mechanized Orders
- · Non-Mechanized Orders

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number and PON Date and Time Jeopardy Notice Sent Committed Due Date Service Type 	Report Month BellSouth Order Number Date and Time Jeopardy Notice Sent Committed Due Date Service Type
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark:
% Orders Given Jeopardy Notice	
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
• INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
• 2W Analog Loop Non-Design	Retail Residence and Business - (POTS Excluding Switch Based Orders)
• 2W Analog Loop w/LNP Design	Retail Residence and Business Dispatch
• 2W Analog Loop w/LNP Non-Design	Retail Residence and Business - (POTS Excluding Switch Based Orders)
2W Analog Loop w/INP Design	Retail Residence and Business Dispatch
2W Analog Loop w/INP Non-Design	Retail Residence and Business (POTS Excluding Switch- Based Orders)
• UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	• Retail Digital Loop ≥ DS1
• UNE Loop + Port Combinations	Retail Business and Residence
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non -Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail
Average Jeopardy Notice Interval	• 95% ≥ 48 Hours

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



P-3: Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that the CLEC can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for Total misses and End User Misses.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders Test Orders, etc.)
- Disconnect (D) & From (F) orders
- End User Misses on Local Interconnection Trunks

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of orders with completion dates in the reporting period that are past the original committed due date. Missed Appointments caused by end-user reasons will be included and reported separately. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date. Which means there cannot be a cutoff time for commitments, as certain types of orders are requested to be worked after standard business hours. Also, during Daylight Savings Time, field technicians are scheduled until 9PM in some areas and the customer is offered a greater range of intervals from which to select.

Calculation

Percent Missed Installation Appointments = $(a \div b) \times 100$

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- · CLEC Specific
- CLEC Aggregate
- · BellSouth Aggregate
- Report in Categories of <10 lines/circuits ≥ 10 lines/circuits (except trunks)
- · Dispatch/No Dispatch

Report Explanation: The difference between End User MA and Total MA is the result of BellSouth caused misses. Here, Total MA is the total percent of orders missed either by BellSouth or CLEC end user. The End User MA represents the percentage of orders missed by the CLEC or their end user.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
CLEC Order Number and PON (PON)	BellSouth Order Number
Committed Due Date (DD)	Committed Due Date (DD)
Completion Date (CMPLTN DD)	Completion Date (CMPLTN DD)
Status Type	Status Type
Status Notice Date	Status Notice Date
Standard Order Activity	Standard Order Activity
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
 2W Analog Loop Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business - (POTS Excluding Switch- Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/LNP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/LNP Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/INP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/INP Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
 UNE Loop + Port Combinations Dispatch Out Non-Dispatch Dispatch In Switch-Based 	Retail Residence and Business Dispatch Out Non-Dispatch Dispatch In Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo OtherDispatchNon-Dispatch (Dispatch In)	Retail Residence, Business and Design Dispatch (Including Dispatch Out and Dispatch In) Dispatch Non-Dispatch (Dispatch In)
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
UNE ISDN	Retail ISDN - BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non - Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

BELLSOUTH®

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail



P-4: Average Completion Interval (OCI) & Order Completion Interval Distribution

Definition

The "average completion interval" measure monitors the interval of time it takes BellSouth to provide service for the CLEC or its own customers. The "Order Completion Interval Distribution" provides the percentages of orders completed within certain time periods. This report measures how well BellSouth meets the interval offered to customers on service orders.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Disconnect (D&F) orders (Except "D" orders associated with LNP Standalone)
- "L" Appointment coded orders (where the customer has requested a later than offered interval)

Business Rules

The actual completion interval is determined for each order processed during the reporting period. The completion interval is the elapsed time from when BellSouth issues a FOC or SOCS date time stamp receipt of an order from the CLEC to BellSouth's actual order completion date. This includes all delays for BellSouth's CLEC/End Users. The clock starts when a valid order number is assigned by SOCS and stops when the technician or system completes the order in SOCS. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33-day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on the same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

The interval breakout for UNE and Design is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, $\ge 30 = 30$ and greater.

Calculation

Completion Interval = (a - b)

- a = Completion Date
- b = Order Issue Date

Average Completion Interval = $(c \div d)$

- c = Sum of all Completion Intervals
- d = Count of Orders Completed in Reporting Period

Order Completion Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Service Orders Completed in "X" days
- f = Total Service Orders Completed in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Residence & Business reported in day intervals = 0,1,3,4,5,5+
- UNE and Design reported in day intervals =0-5,5-10,10-15,15-20,20-25,25-30,≥ 30
- All Levels are reported <10 line/circuits; ≥ 10 line/circuits (except trunks)
- ISDN Orders included in Non-Design

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month CLEC Company Name Order Number (PON) Application Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Geographic Scope	 Report Month BellSouth Order Number Application Date & Time Order Completion Date & Time Service Type Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
 2W Analog Loop Non-Design Dispatch Non-Dispatch (Dispatch In)	 Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/LNP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/LNP Non-Design Dispatch Non-Dispatch (Dispatch In) 	 Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/INP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/INP Non-Design Dispatch Non-Dispatch (Dispatch In) 	 Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	• Retail Digital Loop ≥ DS1
 UNE Loop + Port Combinations Dispatch Out Non-Dispatch Dispatch In Switch-Based 	 Retail Residence and Business Dispatch Out Non-Dispatch Dispatch In Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)



Georgia Performance Metrics

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
UNE Combo Other	Retail Residence, Business and Design Dispatch (Including Biggst 110 (111) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail Residence, Business and Design Dispatch (Including) Retail
- Dispatch	Dispatch Out and Dispatch In) - Dispatch
•	*
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
UNE xDSL (HDSL, ADSL and UCL) without conditioning	• 7 Days
UNE xDSL (HDSL, ADSL and UCL) with conditioning	• 14 Days
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non-Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
UNE xDSL without conditioning	• 7 Days
UNE xDSL with conditioning	• 14 Days
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

P-5: Average Completion Notice Interval

Definitions

The Completion Notice Interval is the elapsed time between the BellSouth reported completion of work and the issuance of a valid completion notice to the CLEC.

Exclusions

- · Cancelled Service Orders
- · Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D&F orders (Exception: "D" orders associated with LNP Standalone)

Business Rules

Measurement on interval of completion date and time entered by a field technician on dispatched orders, and 5PM start time on the due date for non-dispatched orders; to the release of a notice to the CLEC/BellSouth of the completion status. The field technician notifies the CLEC the work was complete and then he/she enters the completion time stamp information in his/her computer. This information switches through to the SOCS systems either completing the order or rejecting the order to the Work Management Center (WMC). If the completion is rejected, it is manually corrected and then completed by the WMC. The notice is returned on each individual order.

The start time for all orders is the completion stamp either by the field technician or the 5PM due date stamp; the end time for mechanized orders is the time stamp the notice was transmitted to the CLEC interface (LENS, EDI, OR TAG). For non-mechanized orders the end timestamp will be timestamp of order update to C-SOTS system.

Calculation

Completion Notice Interval = (a - b)

- a = Date and Time of Notice of Completion
- b = Date and Time of Work Completion

Average Completion Notice Interval = $c \div d$

- c = Sum of all Completion Notice Intervals
- d = Number of Orders with Notice of Completion in Reporting Period

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Mechanized Orders
- · Non-Mechanized Orders
- Reporting intervals in Hours; 0,1-2,2-4,4-8,8-12,12-24, ≥ 24 plus Overall Average Hour Interval (The categories are inclusive of these time intervals: 0-1 = 0.99; 1-2 = 1-1.99; 2-4 = 2-3.99, etc.)
- Reported in categories of <10 line / circuits; ≥ 10 line/circuits (except trunks)

BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report Month
CLEC Order Number (so_nbr)	BellSouth Order Number (so_nbr)
Work Completion Date (cmpltn_dt)	Work Completion Date (cmpltn_dt)
Work Completion Time	Work Completion Time
Completion Notice Availability Date	Completion Notice Availability Date
Completion Notice Availability Time	Completion Notice Availability Time
Service Type	Service Type
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	NOTE: Code in parentheses is the corresponding header found in the raw data file.

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone)	Retail Residence and Business (POTS)
INP (Standalone)	Retail Residence and Business (POTS)
2W Analog Loop Design	Retail Residence and Business Dispatch
 2W Analog Loop Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business - (POTS Excluding Switch- Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/LNP Design	Retail Residence and Business Dispatch
2W Analog Loop w/LNP Non-Design Dispatch Non-Dispatch (Dispatch In)	Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/INP Design	Retail Residence and Business Dispatch
2W Analog Loop w/INP Non-Design Dispatch Non-Dispatch (Dispatch In)	Retail Residence and Business (POTS Excluding Switch- Based Orders) Dispatch Non-Dispatch (Dispatch In)
UNE Digital Loop < DS1	Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE Loop + Port Combinations Dispatch Out Non-Dispatch Dispatch In Switch-Based UNE Switch Ports	Retail Residence and Business Dispatch Out Non-Dispatch Dispatch In Switch-Based Retail Residence and Business (BOTS)
UNE Switch Ports	Retail Residence and Business (POTS)



Georgia Performance Metrics

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
UNE Combo Other	Retail Residence, Business and Design Dispatch (Including
- Dispatch	Dispatch Out and Dispatch In) - Dispatch
- Non-Dispatch (Dispatch In)	- Non-Dispatch (Dispatch In)
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non-Design	Retail Residence and Business
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Georgia Performance Metrics

P-6: % Completions/Attempts without Notice or < 24 hours Notice

Definition

This Report measures the interval from the FOC end timestamp on the LSR until 5:00 P.M. on the original committed due date of a service order. The purpose of this measure is to report if BellSouth is returning a FOC to the CLEC in time for the CLEC to notify their customer of the scheduled date.

Exclusions

"0" dated orders or any request where the subscriber requested an earlier due date of < 24 hours prior to the original commitment date, or any LSR received < 24 hours prior to the original commitment date.

Business Rules

For CLEC Results:

Calculation would exclude any successful or unsuccessful service delivery where the CLEC was informed at least 24 hours in advance. BellSouth may also exclude from calculation any LSRs received from the requesting CLEC with less than 24 hour notice prior to the commitment date.

For BellSouth Results:

BellSouth does not provide a FOC to its retail customers.

Calculation

Percent Completions or Attempts without Notice or with Less Than 24 Hours Notice = $(a \div b) \times 100$

- a = Completion Dispatches (Successful and Unsuccessful) With No FOC or FOC Received < 24 Hours of original Committed Due Date
- b = All Completions

Report Structure

- CLEC Specific
- CLEC Aggregate
- Dispatch /Non-Dispatch
- Total Orders FOC < 24 Hours
- Total Completed Service Orders
- % FOC < 24 Hours

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Committed Due Date (DD)	Not Applicable
FOC End Timestamp	
Report Month	
CLEC Order Number and PON	
Geographic Scope	
- State / Region	

P-6: % Completions/Attempts without Notice or < 24 hours Notice

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	Diagnostic
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
2W Analog Loop Design	
2W Analog Loop-Non-Design	
2W Analog Loop w/LNP - Design	
• 2W Analog Loop w/LNP- Non-Design	
2W Analog Loop w/INP-Design	
2W Analog Loop w/INP-Non-Design	
• UNE Digital Loop < DS1	
• UNE Digital Loop >=DS1	
UNE Loop + Port Combinations	
UNE Switch ports	
UNE Combo Other	
UNE xDSL (HDSL, ADSL and UCL)	
UNE ISDN	
UNE Line Sharing	
UNE Other Design	
UNE Other Non -Design	
Local Transport (Unbundled Interoffice Transport)	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

P-7: Coordinated Customer Conversions Interval

Definition

This report measures the average time it takes BellSouth to disconnect an unbundled loop from the BellSouth switch and cross connect it to CLEC equipment. This measurement applies to service orders with INP and with LNP, and where the CLEC has requested BellSouth to provide a coordinated cut over.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement.
- Delays due to CLEC following disconnection of the unbundled loop
- Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested.

Business Rules

When the service order includes INP, the interval includes the total time for the cut over including the translation time to place the line back in service on the ported line. When the service order includes LNP, the interval only includes the total time for the cut over (the port of the number is controlled by the CLEC). The interval is calculated for the entire cut over time for the service order and then divided by items worked in that time to give the average per-item interval for each service order.

Calculation

Coordinated Customer Conversions Interval = (a - b)

- a = Completion Date and Time for Cross Connection of a Coordinated Unbundled Loop
- b = Disconnection Date and Time of an Coordinated Unbundled Loop

Percent Coordinated Customer Conversions (for each interval) = $(c \div d) \times 100$

- c = Total number of Coordinated Customer Conversions for each interval
- d = Total Number of Unbundled Loop with Coordinated Conversions (items) for the reporting period

Report Structure

- CLEC Specific
- · CLEC Aggregate
- The interval breakout is 0-5 = 0-4.99, 5-15 = 5-14.99, $\ge 15 = 15$ and greater, plus Overall Average Interval.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog Exists
CLEC Order Number	_
Committed Due Date (DD)	
Service Type (CLASS_SVC_DESC)	
Cut over Start Time	
Cut over Completion Time	
Portability Start and Completion Times (INP orders)	
Total Conversions (Items)	
Note: Code in parentheses is the corresponding header	
found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Unbundled Loops with INP/LNP	• 95% ≤ 15 minutes
Unbundled Loops without INP/LNP	

Georgia Performance Metrics

P-7: Coordinated Customer Conversions Interval

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Unbundled Loops	• 95% ≤ 15 minutes



P-7A: Coordinated Customer Conversions – Hot Cut Timeliness% Within Interval and Average Interval

Definition

This category measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. It measures the percentage of orders where the cut begins within 15 minutes of the requested start time of the order and the average interval.

Exclusions

- Any order canceled by the CLEC will be excluded from this measurement.
- Delays caused by the CLEC
- · Unbundled Loops where there is no existing subscriber loop and loops where coordination is not requested.
- All unbundled loops on multiple loop orders after the first loop.

Business Rules

This report measures whether BellSouth begins the cut over of an unbundled loop on a coordinated and/or a time specific order at the CLEC requested start time. The cut is considered on time if it starts 15 minutes before or after the requested start time. Using the scheduled time and the actual cut over start time, the measurement will calculate the percent within interval and the average interval. If a cut involves multiple lines, the cut will be considered "on time" if the first line is cut within the interval. \leq 15 minutes includes intervals that began 15:00 minutes or less before the scheduled cut time and cuts that began 15 minutes or less after the scheduled cut time; >15 minutes, \leq 30 minutes includes cuts within 15:00 – 30:00 minutes either prior to or after the scheduled cut time; >30 minutes includes cuts greater than 30:00 minutes either prior to or after the scheduled cut time.

Calculation

% within Interval = $(a \div b) \times 100$

- a = Total Number of Coordinated Unbundled Loop Orders for the interval
- b = Total Number of Coordinated Unbundled Loop Orders for the reporting period

Interval = (c - d)

- c = Scheduled Time for Cross Connection of a Coordinated Unbundled Loop Order
- d = Actual Start Date and Time of a Coordinated Unbundled Loop Order

Average Interval = $(e \div f)$

- · Sum of all Intervals
- Total Number of Coordinated Unbundled Loop Orders for the reporting period.

Report Structure

- · CLEC Specific
- · CLEC Aggregate

Reported in intervals of early, on time and late cuts %≤ 15 minutes; %>15 minutes, ≤30 minutes; %>30 minutes, plus Overall Average Interval

P-7A: Coordinated Customer Conversions – Hot Cut Timeliness% Within Interval and Average Interval

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Report Month CLEC Order Number (so_nbr) Committed Due Date (DD) Service Type (CLASS_SVC_DESC) Cut over Scheduled Start Time Cut over Actual Start Time Total Conversions Orders 	No BellSouth Analog exists
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
Product Reporting Level SL1 Time Specific SL1 Non-Time Specific SL2 Time Specific SL2 Non-Time Specific	95% Within + or – 15 minutes of Scheduled Start Time

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
- UNE Loops	• 95% Within + or – 15 minutes of Scheduled Start time

P-7B: Coordinated Customer Conversions – Average Recovery Time

Definition

Measures the time between notification and resolution by BellSouth of a service outage found that can be isolated to the BellSouth side of the network. The time between notification and resolution by BellSouth must be measured to ensure that CLEC customers do not experience unjustifiable lengthy service outages during a Coordinated Customer Conversion. This report measures outages associated with Coordinated Customer Conversions prior to service order completion.

Exclusions

- Cut overs where service outages are due to CLEC caused reasons
- · Cut overs where service outages are due to end-user caused reasons

Business Rules

Measures the outage duration time related to Coordinated Customer Conversions from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The duration time is defined as the time from the initial trouble notification until the trouble has been restored and the CLEC has been notified. The interval is calculated on the total outage time for the circuits divided by the total number of outages restored during the report period to give the average outage duration.

Calculation

Recovery Time = (a - b)

- a = Date & Time That Trouble is Closed by CLEC
- b = Date & Time Initial Trouble is Opened with BellSouth

Average Recovery Time = $(c \div d)$

- c = Sum of all the Recovery Times
- d = Number of Troubles Referred to the BellSouth

Report Structure

- · CLEC Specific
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	• None
CLEC Company Name	
CLEC Order Number (so_nbr)	
Committed Due Date (DD)	
Service Type (CLASS_SVC_DESC)	
CLEC Acceptance Conflict (CLEC_CONFLICT)	
CLEC Conflict Resolved (CLEC_RESOLVE)	
CLEC Conflict MFC (CLEC_CONFLICT_MFC)	
Total Conversion Orders	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Unbundled Loops with INP/LNP Unbundled Loops without INP/LNP	Diagnostic

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



P-7C: Hot Cut Conversions - % Provisioning Troubles Received Within 7 days of a completed Service Order

Definition

Percent Provisioning Troubles received within 7 days of a completed service order associated with a Coordinated and Non-Coordinated Customer Conversion. Measures the quality and accuracy of Hot Cut Conversion Activities.

Exclusions

- Any order canceled by the CLEC
- Troubles caused by Customer Provided Equipment

Business Rules

Measures the quality and accuracy of completed service orders associated with Coordinated and Non-Coordinated Hot Cut Conversions. The first trouble report received on a circuit ID within 7 days following a service order completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed Coordinated and Non-Coordinated Hot Cut Conversion service orders and following 7 days after the completion of the service order for a trouble report issue date.

Calculation

% Provisioning Troubles within 7 days of service order completion = $(a \div b) \times 100$

- a = The sum of all Hot Cut Circuits with a trouble within 7 days following service order(s) completion
- b = The total number of Hot Cut service order circuits completed in the previous report calendar month

Report Structure

- · CLEC Specific
- CLEC Aggregate
- · Dispatch/Non-Dispatch

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog exists
CLEC Order Number (so nbr)	
• PON	
Order Submission Date (TICKET_ID)	
Order Submission Time (TICKET_ID)	
Status Type	
Status Notice Date	
Standard Order Activity	
Geographic Scope	
Total Conversion Circuits	
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
UNE Loop Design UNE Loop Non-Design	• ≤ 5%

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
• UNE Loops	• ≤ 5%



P-8: Cooperative Acceptance Testing - % of xDSL Loops Tested

Definition

The loop will be considered cooperatively tested when the BellSouth technician places a call to the CLEC representative to initiate cooperative testing and jointly performs the tests with the CLEC.

Exclusions

- Testing failures due to CLEC (incorrect contact number, CLEC not ready, etc.)
- xDSL lines with no request for cooperative testing

Business Rules

When a BellSouth technician finishes delivering an order for an xDSL loop where the CLEC order calls for cooperative testing at the customer's premise, the BellSouth technician is to call a toll free number to the CLEC testing center. The BellSouth technician and the CLEC representative at the center then test the line. As an example of the type of testing performed, the testing center may ask the technician to put a short on the line so that the center can run a test to see if it can identify the short.

Calculation

Cooperative Acceptance Testing - % of xDSL Loops Tested = $(a \div b) \times 100$

- a = Total number of successful xDSL cooperative tests for xDSL lines where cooperative testing was requested in the reporting period
- b = Total Number of xDSL line tests requested by the CLEC and scheduled in the reporting period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- Type of Loop tested

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth analog exists
CLEC Company Name (OCN)	
 CLEC Order Number (so_nbr) and PON (PON) 	
Committed Due Date (DD)	
Service Type (CLASS_SVC_DESC)	
 Acceptance Testing Completed (ACCEPT_TESTING) 	
 Acceptance Testing Declined (ACCEPT_TESTING) 	
Total xDSL Orders	
Note : Code in parentheses is the corresponding header	
found in the raw data file.	

SQM LEVEL of Disaggregation:	Retail Analog/Benchmark:
• UNE xDSL - ADSL	95% of Lines Tested
- HDSL - UCL - OTHER	

P-8: Cooperative Acceptance Testing - % of xDSL Loops Tested

SEEM Measure

SEEM Measure		
Yes	Tier I	X
	Tier II	X
	Tier III	

SEEM Disaggregation:	SEEM Analog/Benchmark:
• UNE xDSL	• 95% of Lines Tested



Georgia Performance Metrics

P-9: % Provisioning Troubles within 30 days of Service Order Completion

Definition

Percent Provisioning Troubles within 30 days of Service Order Completion measures the quality and accuracy of Service order activities.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- Trouble reports caused and closed out to Customer Provided Equipment (CPE)

Business Rules

Measures the quality and accuracy of completed orders. The first trouble report from a service order after completion is counted in this measure. Subsequent trouble reports are measured in Repeat Report Rate. Reports are calculated searching in the prior report period for completed service orders and following 30 days after completion of the service order for a trouble report issue date.

D & F orders are excluded as there is no subsequent activity following a disconnect.

Note: Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

% Provisioning Troubles within 30 days of Service Order Activity = $(a \div b) \times 100$

- a = Trouble reports on all completed orders 30 days following service order(s) completion
- b = All Service Orders completed in the previous report calendar month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- BellSouth Aggregate
- Reported in categories of <10 line/circuits; ≥ 10 line/circuits (except trunks)
- Dispatch / No Dispatch (except trunks)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
CLEC Order Number and PON	BellSouth Order Number
Order Submission Date (TICKET_ID)	Order Submission Date
 Order Submission Time (TICKET_ID) 	Order Submission Time
Status Type	Status Type
Status Notice Date	Status Notice Date
Standard Order Activity	Standard Order Activity
Geographic Scope	Geographic Scope
Note: Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	Retail Residence



SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Business	Retail business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
2W Analog Loop Design	Retail Residence and Business Dispatch
 2W Analog Loop Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business - (POTS Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/LNP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/LNP Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business - (POTS Excluding Switch- Based Orders) Dispatch Non-Dispatch (Dispatch In)
2W Analog Loop w/INP Design	Retail Residence and Business Dispatch
 2W Analog Loop w/INP Non-Design Dispatch Non-Dispatch (Dispatch In) 	Retail Residence and Business (POTS - Excluding Switch-Based Orders) Dispatch Non-Dispatch (Dispatch In)
• UNE Digital Loop < DS1	• Retail Digital Loop < DS1
• UNE Digital Loop ≥ DS1	Retail Digital Loop ≥ DS1
UNE xDSL (HDSL, ADSL and UCL)	ADSL provided to Retail
• UNE ISDN	Retail ISDN BRI
UNE Line Sharing	ADSL Provided to Retail
INP (Standalone)	Retail Residence and Business (POTS)
• LNP (Standalone)	Retail Residence and Business (POTS)
 UNE Loop + Port Combinations Dispatch Out Non-Dispatch Dispatch In Switch-Based 	 Retail Residence and Business Dispatch Out Non-Dispatch Dispatch In Switch-Based
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo OtherDispatchNon-Dispatch (Dispatch In)	Retail Residence, Business and Design Dispatch (Including Dispatch Out and Dispatch In) Dispatch Non-Dispatch (Dispatch In)
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice
• UNE Other Non -Design	Retail Residence and Business
• UNE Other Design	Retail Design
Local Interconnection Trunks	Parity with Retail

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

(A) **BELLSOUTH**®

P-10: Total Service Order Cycle Time (TSOCT)

Definition

This report measures the total service order cycle time from receipt of a valid service order request to the return of a completion notice to the CLEC Interface.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D (Disconnect Except "D" orders associated with LNP Standalone.) and F (From) orders. (From is disconnect side of a move order when the customer moves to a new address).
- "L" Appointment coded orders (where the customer has requested a later than offered interval)
- Orders with CLEC/Subscriber caused delays or CLEC/Subscriber requested due date changes.

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval. For UNE XDSL Loop, this measurement combines Service Inquiry Interval (SI), FOC Timeliness, Average Completion Interval, and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI) and the BellSouth Legacy Systems. Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on same day. They can be either flow through orders (no field work-non-dispatched) or field orders (dispatched).

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = $(c \div d)$

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = (e ÷ f) X 100

- e = Total Number of Service Requests Completed in "X" minutes/hours
- f = Total Number of Service Requests Received in Reporting Period

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of <10 line/circuits; > 10 line/circuits (except trunks)
- Dispatch / No Dispatch categories applicable to all levels except trunks
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, \geq 30 Days. The interval breakout is: 0-5=0-4.99, 5-10=5-9.99, 10-15=10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, > 30 = 30 and greater.

(A) BELLSOUTH®

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month Interval for FOC CLEC Company Name (OCN) Order Number (PON) Submission Date & Time (TICKET_ID) Completion Date (CMPLTN_DT) Completion Notice Date and Time Service Type (CLASS_SVC_DESC) Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file	 Report Month BellSouth Order Number Order Submission Date & Time Order Completion Date & Time Service Type Geographic Scope

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	Retail Analog/Benchmark
Resale Residence	Diagnostic
Resale Business	
Resale Design	
Resale PBX	
Resale Centrex	
Resale ISDN	
• LNP (Standalone)	
• INP (Standalone)	
2W Analog Loop Design	
2W Analog Loop Non-Design	
2W Analog Loop w/LNP Design	
2W Analog Loop w/LNP Non-Design	
UNE Switch Ports	
UNE Loop + Port Combinations	
UNE Combo Other	
UNE xDSL (HDSL, ADSL and UCL)	
UNE ISDN	
UNE Line Sharing	
UNE Other Design	
UNE Other Non -Design	
• UNE Digital Loops < DS1	
• UNE Digital Loops ≥ DS1	
Local Transport (Unbundled Interoffice Trans port)	
Local Interconnection Trunks	

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



P-11: Service Order Accuracy

Definition

The "service order accuracy" measurement measures the accuracy and completeness of a sample of BellSouth service orders by comparing what was ordered and what was completed.

Exclusions

- · Cancelled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.)
- D & F orders

Business Rules

A statistically valid sample of service orders, completed during a monthly reporting period, is compared to the original account profile and the order that the CLEC sent to BellSouth. An order is "completed without error" if all service attributes and account detail changes (as determined by comparing the original order) completely and accurately reflect the activity specified on the original order and any supplemental CLEC order. For both small and large sample sizes, when a Service Request cannot be matched with a corresponding Service Order, it will not be counted. For small sample sizes an effort will be made to replace the service request.

Calculation

Percent Service Order Accuracy = $(a \div b) \times 100$

- a = Orders Completed without Error
- b = Orders Completed in Reporting Period

Report Structure

- · CLEC Aggregate
- Reported in categories of <10 line/circuits; > = 10 line/circuits
- · Dispatch / No Dispatch

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	No BellSouth Analog Exist
CLEC Order Number and PON	
Local Service Request (LSR)	
Order Submission Date	
Committed Due Date	
Service Type	
Standard Order Activity	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark:
Resale Residence	95% Accurate
Resale Business	
Resale Design (Specials)	
UNE Specials (Design)	
• UNE (Non-Design)	
Local Interconnection Trunks	



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation:	SEEM Analog/Benchmark:
Not Applicable	Not Applicable

(A) **BELLSOUTH**®

P-12: LNP-Percent Missed Installation Appointments

Definition

"Percent missed installation appointments" monitors the reliability of BellSouth commitments with respect to committed due dates to assure that CLECs can reliably quote expected due dates to their retail customer as compared to BellSouth. This measure is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates and reported for total misses and End User Misses.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable

Business Rules

Percent Missed Installation Appointments (PMI) is the percentage of total orders processed for which BellSouth is unable to complete the service orders on the committed due dates. Missed Appointments caused by end-user reasons will be included and reported in a separate category. The first commitment date on the service order that is a missed appointment is the missed appointment code used for calculation whether it is a BellSouth missed appointment or an End User missed appointment. The "due date" is any time on the confirmed due date, which means there cannot be a cutoff time for commitments as certain types of orders are requested to be worked after standard business hours.

Calculation

LNP Percent Missed Installation Appointments = (a ÷ b) X 100

- a = Number of Orders with Completion date in Reporting Period past the Original Committed Due Date
- b = Number of Orders Completed in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · Geographic Scope
 - State/Region
- Report in Categories of <10 lines/circuits > 10 lines/circuits (except trunks)

Report explanation: Total Missed Appointments is the total percent of orders missed either by BellSouth or the CLEC end user. End User MA represents the percentage of orders missed by the CLEC end user. The difference between End User Missed Appointments and Total Missed Appointments is the result of BellSouth caused misses.

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
CLEC Order Number and PON (PON)	
Committed Due Date (DD)	
Completion Date (CMPLTN DD)	
Status Type Status Notice Date	
Standard Order Activity	
Geographic Scope	
Note: Code in parentheses is the corresponding header	
found in the raw data file.	

P-12: LNP-Percent Missed Installation Appointments



Georgia Performance Metrics

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation	SQM Retail Analog/Benchmark
• LNP	Retail Residence and Business (POTS)

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
• LNP	• 95% Due Dates Met ^a

^aDue to data structure issues, BellSouth is using a benchmark comparison for SEEM rather than the Truncated Z as stated in the Order.



P-13: LNP-Average Disconnect Timeliness Interval & Disconnect Timeliness Interval Distribution

Definition

Disconnect Timeliness is defined as the interval between the time ESI Number Manager receives the valid 'Number Ported' message from NPAC (signifying the CLEC 'Activate') until the time the Disconnect is completed in the Central Office switch. This interval effectively measures BellSouth responsiveness by isolating it from impacts that are caused by CLEC related activities.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable.

Business Rules

The Disconnect Timeliness interval is determined for each telephone number ported associated with a disconnect service order processed on an LSR during the reporting period. The Disconnect Timeliness interval is the elapsed time from when BellSouth receives a valid 'Number Ported' message in ESI Number Manager (signifying the CLEC 'Activate') for each telephone number ported until each telephone number on the service order is disconnected in the Central Office switch. Elapsed time for each ported telephone number is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the total number of selected telephone numbers disconnected in the reporting period.

Calculation

Disconnect Timeliness Interval = (a - b)

- a = Completion Date and Time in Central Office switch for each number on disconnect order
- b = Valid 'Number Ported' message received date & time

Average Disconnect Timeliness Interval = $(c \div d)$

- c = Sum of all Disconnect Timeliness Intervals
- d = Total Number of disconnected numbers completed in reporting period

Disconnect Timeliness Interval Distribution (for each interval) = $(e \div f) \times 100$

- e = Disconnected numbers completed in "X" days
- f = Total disconnect numbers completed in reporting period

Report Structure

- · CLEC Specific
- CLEC Aggregate
- · Geographic Scope
- State, Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Order Number	Not Applicable
Telephone Number / Circuit Number	
Committed Due Date	
Receipt Date / Time (ESI Number Manager)	
Date/Time of Recent Change Notice	



SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation:	SQM Retail Analog/Benchmark:
• LNP	• 95% within 15 Minutes

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
LNP Standalone	• 95% within 15 Minutes

P-14: LNP-Total Service Order Cycle Time (TSOCT)

Georgia Performance Metrics

P-14: LNP-Total Service Order Cycle Time (TSOCT)

Definition

Total Service Order Cycle Time measures the interval from receipt of a valid service order request to the completion of the final service order associated with that service request.

Exclusions

- · Canceled Service Orders
- Order Activities of BellSouth or the CLEC associated with internal or administrative use of local services (Record Orders, Listing Orders, Test Orders, etc.) where identifiable
- "L" appointment coded orders (indicating the customer has requested a later than offered interval)
- "S" missed appointment coded orders (indicating subscriber missed appointments), except for "SP" codes (indicating subscriber prior due date requested). This would include "S" codes assigned to subsequent due date changes.

Business Rules

The interval is determined for each order processed during the reporting period. This measurement combines three reports: FOC Timeliness, Average Order Completion Interval and Average Completion Notice Interval.

This interval starts with the receipt of a valid service order request and stops when a completion notice is sent to the CLEC Interface (LENS, TAG OR EDI). Elapsed time for each order is accumulated for each reporting dimension. The accumulated time for each reporting dimension is then divided by the associated total number of orders completed. Orders that are worked on zero due dates are calculated with a .33 day interval (8 hours) in order to report a portion of a day interval. These orders are issued and worked/completed on the same day.

Reporting is by Fully Mechanized, Partially Mechanized and Non-Mechanized receipt of LSRs.

Calculation

Total Service Order Cycle Time = (a - b)

- a = Service Order Completion Notice Date
- b = Service Request Receipt Date

Average Total Service Order Cycle Time = $(c \div d)$

- c = Sum of all Total Service Order Cycle Times
- d = Total Number Service Orders Completed in Reporting Period

Total Service Order Cycle Time Interval Distribution (for each interval) = (e ÷ f) X 100

- e = Total Number of Service Orders Completed in "X" minutes/hours
- f = Total Number of Service Orders Received in Reporting Period

Report Structure

- CLEC Specific
- CLEC Aggregate
- · Fully Mechanized; Partially Mechanized; Non-Mechanized
- Report in categories of <10 lines/circuits; \(\geq \) lines/circuits (except trunks)
- Intervals 0-5, 5-10, 10-15, 15-20, 20-25, 25-30, \geq 30 Days. The interval breakout is: 0-5 = 0-4.99, 5-10 = 5-9.99, 10-15 = 10-14.99, 15-20 = 15-19.99, 20-25 = 20-24.99, 25-30 = 25-29.99, \geq 30 = 30 and greater.



Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Not Applicable
Interval for FOC	
CLEC Company Name (OCN)	
Order Number (PON)	
Submission Date & Time (TICKET_ID)	
Completion Date (CMPLTN_DT)	
Completion Notice Date and Time	
Service Type (CLASS_SVC_DESC)	
Geographic Scope	
Note: Code in parentheses is the corresponding header found in the raw data file	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggrega	tion Retail Analog/Benchmark
• LNP	• Diagnostic

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 4: Maintenance & Repair

M&R-1: Missed Repair Appointments

Definition

The percent of trouble reports not cleared by the committed date and time.

Exclusions

- · Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

The negotiated commitment date and time is established when the repair report is received. The cleared time is the date and time that BellSouth personnel clear the trouble and closes the trouble report in his/her Computer Access Terminal (CAT) or workstation. If this is after the Commitment time, the report is flagged as a "Missed Commitment" or a missed repair appointment. When the data for this measure is collected for BellSouth and a CLEC, it can be used to compare the percentage of the time repair appointments are missed due to BellSouth reasons. (No access reports are not part of this measure because they are not a missed appointment.)

Note: Appointment intervals vary with force availability in the POTS environment. Specials and Trunk intervals are standard interval appointments of no greater than 24 hours. Standalone LNP historical data is not available in the maintenance systems (LMOS or WFA).

Calculation

Percentage of Missed Repair Appointments = $(a \div b) \times 100$

- a = Count of Customer Troubles Not Cleared by the Quoted Commitment Date and Time
- b = Total Trouble reports closed in Reporting Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
CLEC Company Name	BellSouth Company Code
Submission Date & Time (TICKET_ID)	Submission Date & Time
Completion Date (CMPLTN_DT)	Completion Date
Service Type (CLASS_SVC_DESC)	Service Type
 Disposition and Cause (CAUSE_CD & CAUSE_DESC) 	Disposition and Cause (Non-Design /Non-Special Only)
Geographic Scope	Trouble Code (Design and Trunking Services)
Note : Code in parentheses is the corresponding header found in the raw data file.	Geographic Scope

M&R-1: Missed Repair Appointments

SQM Disaggregation - Retail Analog/Benchmark

SQM Level of Disaggregation	SQM Retail Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of Switch- Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non – Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-2: Customer Trouble Report Rate

Definition

Percent of initial and repeated customer direct or referred troubles reported within a calendar month per 100 lines/circuits in service.

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

Customer Trouble Report Rate is computed by accumulating the number of maintenance initial and repeated trouble reports during the reporting period. The resulting number of trouble reports are divided by the total "number of service" lines, ports or combination that exist for the CLECs and BellSouth respectively at the end of the report month.

Calculation

Customer Trouble Report Rate = $(a \div b) \times 100$

- a = Count of Initial and Repeated Trouble Reports closed in the Current Period
- b = Number of Service Access Lines in service at End of the Report Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Service Type (CLASS_SVC_DESC) Disposition and Cause (CAUSE_CD & CAUSE_DESC) # Service Access Lines in Service at the end of period Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file.	 Report Month BellSouth Company Code Ticket Submission Date & Time Ticket Completion Date Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) # Service Access Lines in Service at the end of period Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone) (Not Available in Maintenance)	Not Applicable



SQM Level of Disaggregation	SQM Analog/Benchmark
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of Switch- Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non – Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-3: Maintenance Average Duration

Definition

The Average duration of Customer Trouble Reports from the receipt of the Customer Trouble Report to the time the trouble report is cleared.

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

For Average Duration the clock starts on the date and time of the receipt of a correct repair request. The clock stops on the date and time the service is restored and the BellSouth or CLEC customer is notified (when the technician completes the trouble ticket on his/her CAT or work systems).

Calculation

Maintenance Duration = (a - b)

- a = Date and Time of Service Restoration
- b = Date and Time Trouble Ticket was Opened

Average Maintenance Duration = $(c \div d)$

- c = Total of all maintenance durations in the reporting period
- d = Total Closed Troubles in the reporting period

Report Structure

- Dispatch / Non-Dispatch
- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience:	Relating to BellSouth Performance:
Report Month	Report Month
Total Tickets (LINE_NBR)	Total Tickets
CLEC Company Name	BellSouth Company Code
Ticket Submission Date & Time (TICKET_ID)	Ticket Submission Date
Ticket Completion Date (CMPLTN_DT)	Ticket Submission Time
Service Type (CLASS_SVC_DESC)	Ticket Completion Date
Disposition and Cause (CAUSE_CD & CAUSE_DESC)	Ticket Completion Time
Geographic Scope	Total Duration Time
Note : Code in parentheses is the corresponding header	Service Type
	Disposition and Cause (Non-Design /Non-Special Only)
found in the raw data file.	Trouble Code (Design and Trunking Services)
	Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex
Resale ISDN	Retail ISDN
LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of Switch- Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non – Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-4: Percent Repeat Troubles within 30 Days

Definition

Closed trouble reports on the same line/circuit as a previous trouble report received within 30 calendar days as a percent of total troubles closed reported

Exclusions

- Trouble tickets canceled at the CLEC request.
- BellSouth trouble reports associated with internal or administrative service.
- Customer Provided Equipment (CPE) troubles or CLEC Equipment Trouble.

Business Rules

Includes Customer trouble reports received within 30 days of an original Customer trouble report

Calculation

Percent Repeat Troubles within 30 Days = $(a \div b) \times 100$

- a = Count of closed Customer Troubles where more than one trouble report was logged for the same service line within a continuous 30 days
- b = Total Trouble Reports Closed in Reporting Period

Report Structure

- · Dispatch / Non-Dispatch
- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Total Tickets (LINE_NBR) CLEC Company Name Ticket Submission Date & Time (TICKET_ID) Ticket Completion Date (CMPLTN_DT) Total and Percent Repeat Trouble Reports within 30 Days (TOT_REPEAT) Service Type	Report Month Total Tickets BellSouth Company Code Ticket Submission Date Ticket Submission Time Ticket Completion Date Ticket Completion Time Total and Percent Repeat Trouble Reports within 30 Days
 Disposition and Cause (CAUSE_CD & CAUSE_DESC) Geographic Scope Note: Code in parentheses is the corresponding header found in the raw data file. 	 Service Type Disposition and Cause (Non-Design /Non-Special Only) Trouble Code (Design and Trunking Services) Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale ISDN	Retail ISDN
LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of Switch- Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence and Business (POTS)
UNE Combo Other	Retail Residence, Business & Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
UNE Other Non – Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Resale POTS	Retail Residence and Business (POTS)
Resale Design	Retail Design
UNE Loop + Port Combinations	Retail Residence and Business
UNE Loops	Retail Residence and Business Dispatch
• UNE xDSL	ADSL Provided to Retail
UNE Line Sharing	ADSL Provided to Retail
Local Interconnection Trunks	Parity with Retail

M&R-5: Out of Service (OOS) > 24 Hours

Definition

For Out of Service Troubles (no dial tone, cannot be called or cannot call out) the percentage of Total OOS Troubles cleared in excess of 24 hours. (All design services are considered to be out of service).

Exclusions

- Trouble Reports canceled at the CLEC request
- BellSouth Trouble Reports associated with administrative service
- Customer Provided Equipment (CPE) Troubles or CLEC Equipment Troubles.

Business Rules

Customer Trouble reports that are out of service and cleared in excess of 24 hours. The clock begins when the trouble report is created in LMOS/WFA and the trouble is counted if the elapsed time exceeds 24 hours.

Calculation

Out of Service (OOS) > 24 hours = $(a \div b) \times 100$

- a = Total Cleared Troubles OOS > 24 Hours
- b = Total OOS Troubles in Reporting Period

Report Structure

- Dispatch / Non Dispatch
- CLEC Specific
- BellSouth Aggregate
- · CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Total Tickets	Total Tickets
CLEC Company Name	BellSouth Company Code
Ticket Submission Date & Time (TICKET_ID)	Ticket Submission Date
Ticket Completion Date (CMPLTN_DT	Ticket Submission time
Percentage of Customer Troubles out of	Ticket Completion Date
• Service > 24 Hours (OOS>24_FLAG)	Ticket Completion Time
Service type (CLASS_SVC_DESC)	Percent of Customer Troubles out of Service > 24 Hours
Disposition and Cause (CAUSE CD & CAUSE-DESC)	Service type
Geographic Scope	Disposition and Cause (Non-Design/Non-Special only)
Note: Code in parentheses is the corresponding header	Trouble Code (Design and Trunking Services)
found in the raw data file.	Geographic Scope

SQM Level of Disaggregation	SQM Analog/Benchmark
Resale Residence	Retail Residence
Resale Business	Retail Business
Resale Design	Retail Design
Resale PBX	Retail PBX
Resale Centrex	Retail Centrex



SQM Level of Disaggregation	SQM Analog/Benchmark
Resale ISDN	Retail ISDN
LNP (Standalone) (Not Available in Maintenance)	Not Applicable
2W Analog Loop Design	Retail Residence & Business Dispatch
2W Analog Loop Non – Design	Retail Residence & Business (POTS) (Exclusion of Switch- Based Feature Troubles)
UNE Loop + Port Combinations	Retail Residence & Business
UNE Switch Ports	Retail Residence & Business (POTS)
UNE Combo Other	Retail Residence, Business and Design Dispatch
UNE xDSL (HDSL, ADSL and UCL)	ADSL Provided to Retail
• UNE ISDN	Retail ISDN – BRI
UNE Line Sharing	ADSL Provided to Retail
UNE Other Design	Retail Design
• UNE Other Non – Design	Retail Residence & Business
Local Interconnection Trunks	Parity with Retail
Local Transport (Unbundled Interoffice Transport)	Retail DS1/DS3 Interoffice

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

${\bf SEEM\ Disaggregation\ -\ Analog/Benchmark}$

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-6: Average Answer Time - Repair Centers

Definition

This measures the average time a customer is in queue when calling a BellSouth Repair Center.

Exclusions

None

Business Rules

The clock starts when a CLEC Representative or BellSouth customer makes a choice on the Repair Center's menu and is put in queue for the next repair attendant. The clock stops when the repair attendant answers the call (abandoned calls are not included).

Note: The Total Column is a combined BellSouth Residence and Business number.

Calculation

Answer Time for BellSouth Repair Centers = (a - b)

- a = Time BellSouth Repair Attendant Answers Call
- b = Time of entry into queue after ACD Selection

Average Answer Time for BellSouth Repair Centers = $(c \div d)$

- c = Sum of all Answer Times
- d = Total number of calls by reporting period

Report Structure

- CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
CLEC Average Answer Time	BellSouth Average Answer Time

SQM Disaggregation - Analog / Benchmark

SQM Level of Disaggregation	Retail Analog / Benchmark
Region. CLEC/BellSouth Service Centers and BellSouth Repair Centers are regional.	For CLEC, Average Answer Times in UNE Center and BRMC are comparable to the Average Answer Times in the BellSouth Repair Centers.

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

M&R-7: Mean Time To Notify CLEC of Network Outages



Georgia Performance Metrics

M&R-7: Mean Time To Notify CLEC of Network Outages

Definition

This report measures the time it takes for the BellSouth Network Management Center (NMC) to notify the CLEC of major network outages.

Exclusions

None

Business Rules

BellSouth will inform the CLEC of any major network outages (key customer accounts) via a page or email. When the BellSouth NMC becomes aware of a network incident, the CLEC and BellSouth will be notified electronically. The notification time for each outage will be measured in minutes and divided by the number of outages for the reporting period. These are broadcast messages. It is up to those receiving the message to determine if they have customers affected by the incident.

The CLECs will be notified in accordance with the rules outlined in Appendix D of the CLEC "Customer Guide" which is published on the internet at: www.interconnection.bellsouth.com/guides/other_guides/html/gopue/indexf.htm.

Calculation

Time to Notify CLEC = (a - b)

- a = Date and Time BellSouth Notified CLEC
- b = Date and Time BellSouth Detected Network Incident

Mean Time to Notify CLEC = $(c \div d)$

- c = Sum of all Times to Notify CLEC
- d = Count of Network Incidents

Report Structure

- BellSouth Aggregate
- · CLEC Aggregate
- CLEC Specific

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Major Network Events	Major Network Events
Date/Time of Incident	Date/Time of Incident
Date/Time of Notification	Date/Time of Notification

SQM Level of Disaggregation	Retail Analog / Benchmark
BellSouth AggregateCLEC AggregateCLEC Specific	Parity by Design



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 5: Billing

B-1: Invoice Accuracy

Definition

This measure provides the percentage of accuracy of the billing invoices rendered to CLECs during the current month.

Exclusions

- Adjustments not related to billing errors (e.g., credits for service outage, special promotion credits, adjustments to satisfy the customer)
- · Test Accounts

Business Rules

The accuracy of billing invoices delivered by BellSouth to the CLEC must enable them to provide a degree of billing accuracy comparative to BellSouth bills rendered to retail customers of BellSouth. CLECs request adjustments on bills determined to be incorrect. The BellSouth Billing verification process includes manually analyzing a sample of local bills from each bill period. The bill verification process draws from a mix of different customer billing options and types of service. An end-to-end auditing process is performed for new products and services. Internal measurements and controls are maintained on all billing processes.

Calculation

Invoice Accuracy = $[(a - b) \div a] \times 100$

- a = Absolute Value of Total Billed Revenues during current month
- b = Absolute Value of Billing Related Adjustments during current month

Report Structure

- · CLEC Specific
- CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month	Report month
Invoice Type	Retail Type
- UNE	- CRIS
- Resale	- CABS
- Interconnection	Total Billed Revenue
Total Billed Revenue	Billing Related Adjustments
Billing Related Adjustments	

B-1: Invoice Accuracy



Georgia Performance Metrics

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product / Invoice Type Resale UNE	CLEC Invoice Accuracy is comparable to BellSouth Invoice Accuracy
- Interconnection	

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC StateBellSouth State	Parity with Retail



B2: Mean Time to Deliver Invoices

Definition

Bill Distribution is calculated as follows: CRIS BILLS-The number of workdays is reported for CRIS bills. This is calculated by counting the Bill Period date as the first work day. Weekends and holidays are excluded when counting workdays. J/N Bills are counted in the CRIS work day category for the purposes of the measurement since their billing account number (Q account) is provided from the CRIS system.

CABS BILLS-The number of calendar days is reported for CABS bills. This is calculated by counting the day following the Bill Period date as the first calendar day. Weekends and holidays are included when counting the calendar days.

Exclusions

Any invoices rejected due to formatting or content errors.

Business Rules

This report measures the mean interval for timeliness of billing records delivered to CLECs in an agreed upon format. CRIS-based invoices are measured in business days, and CABS-based invoices in calendar days.

Calculation

Invoice Timeliness = (a - b)

- a = Invoice Transmission Date
- b = Close Date of Scheduled Bill Cycle

Mean Time To Deliver Invoices = $(c \div d)$

- c = Sum of all Invoice Timeliness intervals
- d = Count of Invoices Transmitted in Reporting Period

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report month	Report month
Invoice Type	Invoice Type
- UNE	- CRIS
- Resale	- CABS
- Interconnection	Invoice Transmission Count
Invoice Transmission Count	Date of Scheduled Bill Close
Date of Scheduled Bill Close	

(4) BELLSOUTH®

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product / Invoice Type Resale UNE Interconnection	 CRIS-based invoices will be released for delivery within six (6) business days. CABS-based invoices will be released for delivery within eight (8) calendar days. CLEC Average Delivery Intervals for both CRIS and CABS Invoices are comparable to BellSouth Average delivery for both systems.

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
• CLEC State - CRIS	Parity with Retail
- CABS - BellSouth Region	



B3: Usage Data Delivery Accuracy

Definition

This measurement captures the percentage of recorded usage that is delivered error free and in an acceptable format to the appropriate Competitive Local Exchange Carrier (CLEC). These percentages will provide the necessary data for use as a comparative measurement for BellSouth performance. This measurement captures Data Delivery Accuracy rather than the accuracy of the individual usage recording.

Exclusions

None

Business Rules

The accuracy of the data delivery of usage records delivered by BellSouth to the CLEC must enable them to provide a degree of accuracy comparative to BellSouth bills rendered to their retail customers. If errors are detected in the delivery process, they are investigated, evaluated and documented. Errors are corrected and the data retransmitted to the CLEC.

Calculation

Usage Data Delivery Accuracy = $(a - b) \div a \times 100$

- a = Total number of usage data packs sent during current month
- b = Total number of usage data packs requiring retransmission during current month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- · Geographic Scope
 - Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Record Type BellSouth Recorded Non-BellSouth Recorded	Report month Record Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Region	CLEC Usage Data Delivery Accuracy is comparable to BellSouth Usage Data Delivery Accuracy

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	



${\bf SEEM\ Disaggregation\ -\ Analog/Benchmark}$

SEEM Disaggregation	SEEM Analog/Benchmark
CLEC State BellSouth Region	Parity with Retail

(A) **BELLSOUTH**®

B4: Usage Data Delivery Completeness

Definition

This measurement provides percentage of complete and accurately recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is processed and transmitted to the CLEC within thirty (30) days of the message recording date. A parity measure is also provided showing completeness of BellSouth messages processed and transmitted via CMDS. BellSouth delivers its own retail usage from recording location to billing location via CMDS as well as delivering billing data to other companies. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of these measurements is to demonstrate the level of quality of usage data delivered to the appropriate CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Completeness = $(a \div b) \times 100$

- a = Total number of Recorded usage records delivered during current month that are within thirty (30) days of the message recording
- b = Total number of Recorded usage records delivered during the current month

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month Record Type BellSouth Recorded Non-BellSouth Recorded 	Report month Record Type

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Region	CLEC Usage Data Delivery Completeness is comparable to BellSouth Usage Data Delivery Completeness

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELLSOUTH**®

B5: Usage Data Delivery Timeliness

Definition

This measurement provides a percentage of recorded usage data (usage recorded by BellSouth and usage recorded by other companies and sent to BellSouth for billing) that is delivered to the appropriate CLEC within six (6) calendar days from the receipt of the initial recording. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the level of timeliness for processing and transmission of usage data delivered to the appropriate CLEC. The usage data will be mechanically transmitted or mailed to the CLEC data processing center once daily. The Timeliness interval of usage recorded by other companies is measured from the date BellSouth receives the records to the date BellSouth distributes to the CLEC. Method of delivery is at the option of the CLEC.

Calculation

Usage Data Delivery Timeliness Current month = $(a \div b) \times 100$

- a = Total number of usage records sent within six (6) calendar days from initial recording/receipt
- b = Total number of usage records sent

Report Structure

- · CLEC Aggregate
- CLEC Specific
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Record Type BellSouth Recorded Non-BellSouth Recorded	Report Monthly Record Type

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	CLEC Usage Data Delivery Timeliness is comparable to BellSouth Usage Data Delivery Timeliness

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

BELLSOUTH®

${\bf SEEM\ Disaggregation\ -\ Analog/Benchmark}$

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

B6: Mean Time to Deliver Usage

Georgia Performance Metrics

B6: Mean Time to Deliver Usage

Definition

This measurement provides the average time it takes to deliver Usage Records to a CLEC. A parity measure is also provided showing timeliness of BellSouth messages processed and transmitted via CMDS. Timeliness, Completeness and Mean Time to Deliver Usage measures are reported on the same report.

Exclusions

None

Business Rules

The purpose of this measurement is to demonstrate the average number of days it takes BellSouth to deliver Usage data to the appropriate CLEC. Usage data is mechanically transmitted or mailed to the CLEC data processing center once daily. Method of delivery is at the option of the CLEC.

Calculation

Mean Time to Deliver Usage = $(a \times b) \div c$

- a = Volume of Records Delivered
- b = Estimated number of days to deliver
- c = Total Record Volume Delivered

Note: Any usage record falling in the 30+ day interval will be added using an average figure of 31.5 days.

Report Structure

- · CLEC Aggregate
- · CLEC Specific
- · BellSouth Aggregate
- Region

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
Report Month Record Type BellSouth Recorded	Report Monthly Record Type
- Non-BellSouth Recorded	

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Region	Mean Time to Deliver Usage to CLEC is comparable to Mean Time to Deliver Usage to BellSouth

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



B7: Recurring Charge Completeness

Definition

This measure captures percentage of fractional recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Recurring Charge Completeness = $(a \div b) \times 100$

- a = Count of fractional recurring charges that are on the correct bill¹
- b = Total count of fractional recurring charges that are on the correct bill

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	Report month
Invoice type	Retail Analog
Total recurring charges billed	Total recurring charges billed
Total billed on time	Total billed on time

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Product/Invoice Type	
Resale	• Parity
• UNE	Benchmark 90%
Interconnection	Benchmark 90%

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

¹Correct bill = next available bill



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELLSOUTH** *

B8: Non-Recurring Charge Completeness

Definition

This measure captures percentage of non-recurring charges appearing on the correct bill.

Exclusions

None

Business Rules

The effective date of the non-recurring charge must be within 30 days of the bill date for the charge to appear on the correct bill.

Calculation

Non-Recurring Charge Completeness = $(a \div b) \times 100$

- a = Count of non-recurring charges that are on the correct bill¹
- b = Total count of non-recurring charges that are on the correct bill

Report Structure

- CLEC Specific
- · CLEC Aggregate
- · BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report month	Report month
Invoice type	Retail Analog
Total non-recurring charges billed	Total non-recurring charges billed
Total billed on time	Total billed on time

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
Product/Invoice Type	
Resale	• Parity
• UNE	Benchmark 90%
Interconnection	Benchmark 90%

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

¹Correct bill = next available bill



BELLSOUTH®

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 6: Operator Services And Directory Assistance

OS-1: Speed to Answer Performance/Average Speed to Answer - Toll

Definition

Measurement of the average time in seconds calls wait before answered by a toll operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer - Toll = $a \div b$

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- · Average Speed of Answer

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



OS-2: Speed to Answer Performance/Percent Answered with "X" Seconds – Toll

Definition

Measurement of the percent of toll calls that are answered in less than ten seconds

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for toll is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- · Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (Toll)
- · Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	Retail Analog/Benchmark:
• None	Parity by Design

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



DA-1: Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA)

Definition

Measurement of the average time in seconds calls wait before answered by a DA operator.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

Speed to Answer Performance/Average Speed to Answer – Directory Assistance (DA) = $a \div b$

- a = Total queue time
- b = Total calls answered

Note: Total queue time includes time that answered calls wait in queue as well as time abandoned calls wait in queue prior to abandonment.

Report Structure

- Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP
- Month
- Call Type (DA)
- Average Speed of Answer

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



DA-2: Speed to Answer Performance/Percent Answered within "X" Seconds – Directory Assistance (DA)

Definition

Measurement of the percent of DA calls that are answered in less than twelve seconds.

Exclusions

None

Business Rules

The clock starts when the customer enters the queue and the clock stops when a BellSouth representative answers the call or the customer abandons the call. The length of each call is determined by measuring, using a scanning technique, and accumulating the elapsed time from the entry of a customer call into the BellSouth call management system queue until the customer call is abandoned or transferred to BellSouth personnel assigned to handle calls for assistance. The system makes no distinction between CLEC customers and BellSouth customers.

Calculation

The Percent Answered within "X" Seconds measurement for DA is derived by using the BellCore Statistical Answer Conversion Tables, to convert the Average Speed to Answer measure into a percent of calls answered within "X" seconds. The BellCore Conversion Tables are specific to the defined parameters of work time, number of operators, max queue size and call abandonment rates.

Report Structure

- · Reported for the aggregate of BellSouth and CLECs
 - State

Data Retained (on Aggregate Basis)

- For the items below, BellSouth's Performance Measurement Analysis Platform (PMAP) receives a final computation; therefore, no raw data file is available in PMAP.
- Month
- Call Type (DA)
- · Average Speed of Answer

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 7: Database Update Information

D-1: Average Database Update Interval

Definition

This report measures the interval from receipt of the database change request to the completion of the update to the database for Line Information Database (LIDB), Directory Assistance and Directory Listings. For E-911, see Section 8.

Exclusions

- Updates Canceled by the CLEC
- · Initial update when supplemented by CLEC
- BellSouth updates associated with internal or administrative use of local services.

Business Rules

The interval for this measure begins with the date and time stamp when a service order is completed and the completion notice is released to all systems to be updated with the order information including Directory Assistance, Directory Listings, and Line Information Database (LIDB). The end time stamp is the date and time of completion of updates to the system.

For BellSouth Results:

The BellSouth computation is identical to that for the CLEC with the clarifications noted below.

Other Clarifications and Qualification:

- For LIDB, the elapsed time for a BellSouth update is measured from the point in time when the BellSouth file maintenance process makes the LIDB update information available until the date and time reported by BellSouth that database updates are completed.
- Results for the CLECs are captured and reported at the update level by Reporting Dimension (see below).
- The Completion Date is the date upon which BellSouth issues the Update Completion Notice to the CLEC.
- If the CLEC initiates a supplement to the originally submitted update and the supplement reflects changes in customer requirements (rather than responding to BellSouth initiated changes), then the update submission date and time will be the date and time of BellSouth receipt of a syntactically correct update supplement. Update activities responding to BellSouth initiated changes will not result in changes to the update submission date and time used for the purposes of computing the update completion interval.
- · Elapsed time is measured in hours and hundredths of hours rounded to the nearest tenth of an hour.
- Because this should be a highly automated process, the accumulation of elapsed time continues through off-schedule, weekends and holidays; however, scheduled maintenance windows are excluded.

Calculation

Update Interval = (a - b)

- a = Completion Date & Time of Database Update
- b = Submission Date and Time of Database Change

Average Update Interval = $(c \div d)$

- c = Sum of all Update Intervals
- d = Total Number of Updates Completed During Reporting Period



Report Structure

- CLEC Specific (Under development)
- CLEC Aggregate
- BellSouth Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Database File Submission Time Database File Update Completion Time CLEC Number of Submissions Total Number of Updates 	 Database File Submission Time Database File Update Completion Time BellSouth Number of Submissions Total Number of Updates

SQM Disaggregation - Analog/Benchmark

SQM LEVEL of Disaggregation:	Retail Analog/Benchmark:
Database Type • LIDB	Parity by Design
Directory Listings	
Directory Assistance	

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



D-2: Percent Database Update Accuracy

Definition

This report measures the accuracy of database updates by BellSouth for Line Information Database (LIDB), Directory Assistance, and Directory Listings using a statistically valid sample of LSRs/Orders in a manual review. This manual review is not conducted on BellSouth Retail Orders.

Exclusions

- Updates canceled by the CLEC
- · Initial update when supplemented by CLEC
- · CLEC orders that had CLEC errors
- BellSouth updates associated with internal or administrative use of local services.

Business Rules

For each update completed during the reporting period, the original update that the CLEC sent to BellSouth is compared to the database following completion of the update by BellSouth. An update is "completed without error" if the database completely and accurately reflects the activity specified on the original and supplemental update (order) submitted by the CLEC. Each database (LIDB, Directory Assistance, and Directory Listings) should be separately tracked and reported.

A statistically valid sample of CLEC Orders are pulled each month. That sample will be used to test the accuracy of the database update process. This is a manual process.

Calculation

Percent Update Accuracy = $(a \div b) \times 100$

- a = Number of Updates Completed Without Error
- b = Number Updates Completed

Report Structure

- · CLEC Aggregate
- CLEC Specific (not available in this report)
- BellSouth Aggregate (not available in this report)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Performance
 Report Month CLEC Order Number (so_nbr) and PON (PON) Local Service Request (LSR) Order Submission Date Number of Orders Reviewed 	Not Applicable
Note : Code in parentheses is the corresponding header found in the raw data file.	

SQM LEVEL of Disaggregation	Retail Analog/Benchmark:
Database Type	95% Accurate
• LIDB	
Directory Assistance	
Directory Listings	



SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



D-3: Percent NXXs and LRNs Loaded by the LERG Effective Date

Definition

Measurement of the percent of NXX(s) and Location Routing Numbers LRN(s) loaded in end office and/or tandem switches by the Local Exchange Routing Guide (LERG) effective date when facilities are in place. BellSouth has a single provisioning process for both NXX(s) and LRN(s). In this measure, BellSouth will identify whether or not a particular NXX has been flagged as LNP capable (set triggers for dips) by the LERG effective date.

An LRN is assigned by the owner of the switch and is placed into the software translations for every switch to be used as an administrative pointer to route NXX(s) in LNP capable switches. The LRN is a result of Local Number Porting and is housed in a national database provided by the Number Portability Administration Center (NPAC). The switch owner is responsible for notifying NPAC and requesting the effective date that will be reflected in the LERG. The national database downloads routing tables into BellSouth Service Control Point (SCP) regional databases, which are queried by switches when routing ported numbers.

The basic NXX routing process includes the addition of all NXX(s) in the response translations. This addition to response translations is what supports LRN routing. Routing instructions for all NXX(s), including LRN(s), are received from the Advance Routing & Trunking System (ARTS) and all routing, including response, is established based on the information contained in the Translation Work Instructions (TWINs) document.

Exclusions

- · Activation requests where the CLEC's interconnection arrangements and facilities are not in place by the LERG effective date.
- · Expedite requests

Business Rules

Data for the initial NXX(s) and LRN(s) in a local calling area will be based on the LERG effective date or completion of the initial interconnection trunk group(s), whichever is longer. Data for additional NXX(s) in the local calling area will be based on the LERG effective date. The LERG effective date is loaded into the system at the request of the CLEC. It is contingent upon the CLEC to engineer, order, and install interconnection arrangements and facilities prior to that date.

The total Count of NXX(s) and LRN(s) that were scheduled to be loaded and those that were loaded by the LERG effective date in BellSouth switches will be captured in the Work Force Administration -Dispatch In database.

Calculation

Percent NXXs/LRNs Loaded and Tested Prior to the LERG Effective Date = (a ÷ b) X 100

- a = Count of NXXs and LRNs loaded by the LERG effective date
- b = Total NXXs and LRNs scheduled to be loaded by the LERG effective date

Report Structure

- · CLEC Specific
- · CLEC Aggregate
- BellSouth (Not Applicable)

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Company Name	Not Applicable
Company Code	
• NPA/NXX	
LERG Effective Date	
Loaded Date	



SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
Geographic scope Region	100% by LERG effective date

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 8: E911

E-1: Timeliness

Definition

Measures the percent of batch orders for E911 database updates (to CLEC resale and BellSouth retail records) processed successfully within a 24-hour period.

Exclusions

- Any resale order canceled by a CLEC
- · Facilities-based CLEC orders

Business Rules

The 24-hour processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing batch orders extracted from the BellSouth Service Order Control System (SOCS). Processing stops when SCC loads the individual records to the E911 database. The E911 database includes updates to the Automatic Location Identification (ALI) database. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Timeliness = $(a \div b) \times 100$

- a = Number of batch orders processed within 24 hours
- b = Total number of batch orders submitted

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



${\bf SEEM\ Disaggregation\ -\ Analog/Benchmark}$

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



E-2: Accuracy

Definition

Measures the percent of E911 telephone number (TN) record updates (to CLEC resale and BellSouth retail records) processed successfully for E911 (including the Automatic Location Identification (ALI) database).

Exclusions

- Any resale order canceled by a CLEC
- · Facilities-based CLEC orders

Business Rules

Accuracy is based on the number of records processed without error at the conclusion of the processing cycle. Mechanical processing starts when SCC (the BellSouth E911 vendor) receives E911 files containing telephone number (TN) records extracted from BellSouth's Service Order Control System (SOCS). The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Accuracy = $(a \div b) \times 100$

- a = Number of record individual updates processed with no errors
- b = Total number of individual record updates

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- · Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



E-3: Mean Interval

Definition

Measures the mean interval processing of E911 batch orders (to update CLEC resale and BellSouth retail records) including processing against the Automatic Location Identification (ALI) database.

Exclusions

- Any resale order canceled by a CLEC
- Facilities-based CLEC orders

Business Rules

The processing period is calculated based on the date and time processing starts on the batch orders and the date and time processing stops on the batch orders. Data is posted is 4-hour increments up to and beyond 24 hours. The system makes no distinction between CLEC resale records and BellSouth retail records.

Calculation

E911 Interval = (a - b)

- a = Date and time of batch order completion
- b = Date and time of batch order submission

E911 Mean Interval = $(c \div d)$

- c = Sum of all E911 Intervals
- d = Number of batch orders completed

Report Structure

Reported for the aggregate of CLEC resale updates and BellSouth retail updates

- State
- Region

Data Retained

- · Report month
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• None	Parity by Design

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 9: Trunk Group Performance

TGP-1: Trunk Group Performance-Aggregate

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk groups for which valid data is not available for an entire study period
- Duplicate trunk group information
- Trunk groups blocked due to CLEC network/equipment failure
- · Trunk groups blocked due to CLEC delayed or refused orders
- · Trunk groups blocked due to unanticipated significant increases in CLEC traffic
- · Final groups actually overflowing, not blocked

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches
- · Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

	Point A	Point B
Category 1:	BellSouth End Office	BellSouth Access Tandem
Category 3:	BellSouth End Office	CLEC Switch
Category 4:	BellSouth Local Tandem	CLEC Switch
Category 5:	BellSouth Access Tandem	CLEC Switch
Category 10:	BellSouth End Office	BellSouth Local Tandem
Category 16:	BellSouth Tandem	BellSouth Tandem



BellSouth Affecting Categories:

Point A Point B

Category 9: BellSouth End Office BellSouth End Office

Calculation

Monthly Average Blocking:

- For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.
- The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:

- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- · CLEC Aggregate
- · BellSouth Aggregate
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Total Trunk Groups	Total Trunk Groups
Number of Trunk Groups by CLEC	Aggregate Hourly blocking per trunk group
Hourly blocking per trunk group	Hourly usage per trunk group
Hourly usage per trunk group	Hourly call attempts per trunk group
Hourly call attempts per trunk group	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
CLEC aggregateBellSouth aggregate	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1,
	3, 4, 5, 10, 16 for CLECs and 9 for BellSouth

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	X



SEEM Disaggregation	SEEM Analog/Benchmark:
CLEC aggregate BellSouth aggregate	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1,3,4,5,10,16 for CLECs and 9 for BellSouth

TGP-2: Trunk Group Performance-CLEC Specific

TGP-2: Trunk Group Performance-CLEC Specific

Definition

The Trunk Group Performance report displays, over a reporting cycle, aggregate, average trunk group blocking data for each hour of each day of the reporting cycle, for both CLEC affecting and BellSouth affecting trunk groups.

Exclusions

- Trunk Groups for which valid data is not available for an entire study period
- Duplicate trunk group information
- Trunk groups blocked due to CLEC network/equipment failure
- Trunk groups blocked due to CLEC delayed or refused orders
- Trunk groups blocked due to unanticipated significant increases in CLEC traffic
- Final groups actually overflowing, not blocked

Business Rules

The purpose of the Trunk Group Performance Report is to provide trunk blocking measurements on CLEC and BellSouth trunk groups for comparison only. It is not the intent of the report that it be used for network management and/or engineering.

Monthly Average Blocking:

- The reporting cycle includes both business and non-business days in a calendar month.
- Monthly average blocking values are calculated for each trunk group for each of the 24 time consistent hours across a reporting cycle.

Aggregate Monthly Blocking:

- Used to compare aggregate blocking across trunk groups which terminate traffic at CLEC points of presence versus BellSouth switches.
- Aggregate monthly blocking data is calculated for each hour of the day across all trunk groups assigned to a category.

Trunk Categorization:

• This report displays, over a reporting cycle, aggregate, average blocking data for each hour of a day. Therefore, for each reporting cycle, 24 blocking data points are generated for two aggregate groups of selected trunk groups. These groups are CLEC affecting and BellSouth affecting trunk groups. In order to assign trunk groups to each aggregate group, all trunk groups are first assigned to a category. A trunk group's end points and the type of traffic that is transmitted on it define a category. Selected categories of trunk groups are assigned to the aggregate groups so that trunk reports can be generated. The categories to which trunk groups have been assigned for this report are as follows.

CLEC Affecting Categories:

	Point A	Point B
Category 1:	BellSouth End Office	BellSouth Access Tandem
Category 3:	BellSouth End Office	CLEC Switch
Category 4:	BellSouth Local Tandem	CLEC Switch
Category 5:	BellSouth Access Tandem	CLEC Switch
Category 10:	BellSouth End Office	BellSouth Local Tandem
Category 16:	BellSouth Tandem	BellSouth Tandem

BellSouth Affecting Categories:

	Point A	Point B
Category 9:	BellSouth End Office	BellSouth End Office

Calculation:

Monthly Average Blocking:



- For each hour of the day, each day's raw data are summed across all valid measurements days in a report cycle for blocked and attempted calls.
- The sum of the blocked calls is divided by the total number of calls attempted in a reporting period.

Aggregate Monthly Blocking:

- For each hour of the day, the monthly sums of the blocked and attempted calls from each trunk group are separately aggregated over all trunk groups within each assigned category.
- The total blocked calls is divided by the total call attempts within a group to calculate an aggregate monthly blocking for each assigned group.
- The result is an aggregate monthly average blocking value for each of the 24 hours by group.
- The difference between the CLEC and BellSouth affecting trunk groups are also calculated for each hour.

Report Structure

- CLEC Specific
 - State

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
Report Month	Report Month
Total Trunk Groups	Total Trunk Groups
Number of Trunk Groups by CLEC	Aggregate Hourly blocking per trunk group
Hourly blocking per trunk group	Hourly usage per trunk group
Hourly usage per trunk group	Hourly call attempts per trunk group
Hourly call attempts per trunk group	

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
CLEC trunk group	• Any 2 hour period in 24 hours where CLEC blockage exceeds
	BellSouth blockage by more than 0.5% using trunk groups 1,
	3, 4, 5, 10, 16 for CLECs and 9 for BellSouth

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark:
CLEC trunk group BellSouth trunk group	• Any 2 hour period in 24 hours where CLEC blockage exceeds BellSouth blockage by more than 0.5% using trunk groups 1, 3, 4, 5, 10, 16 for CLECs and 9 for BellSouth



Section 10: Collocation

C-1: Collocation Average Response Time

Definition

Measures the average time (counted in calendar days) from the receipt of a complete and accurate collocation application (including receipt of application fee if required) to the date BellSouth returns a response electronically or in writing. Within 10 calendar days after having received a bona fide application for physical collocation, BellSouth must respond as to whether space is available or not.

Exclusions

Any application canceled by the CLEC

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate collocation application accompanied by the appropriate application fee if required. The clock stops on the date that BellSouth returns a response. The clock will restart upon receipt of changes to the original application request.

Calculation

Response Time = (a - b)

- a = Request Response Date
- b = Request Submission Date

Average Response Time = $(c \div d)$

- c = Sum of all Response Times
- d = Count of Responses Returned within Reporting Period

Report Structure

- · Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

Level of Disaggregation	Retail Analog/Benchmark
• State	Virtual - 20 Calendar Days
Virtual-Initial	Physical Caged - 30 Calendar Days
Virtual-Augment	Physical Cageless - 30 Calendar Days
Physical Caged-Initial	
Physical Caged-Augment	
Physical-Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	

SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

(A) **BELLSOUTH**®

C-2: Collocation Average Arrangement Time

Definition

Measures the average time (counted in calendar days) from receipt of a complete and accurate Bona Fide firm order (including receipt of appropriate fee if required) to the date BellSouth completes the collocation arrangement and notifies the CLEC.

Exclusions

- Any Bona Fide firm order canceled by the CLEC
- Any Bona Fide firm order with a CLEC-negotiated interval longer than the benchmark interval.

Business Rules

The clock starts on the date that BellSouth receives a complete and accurate Bone Fide firm order accompanied by the appropriate fee. The clock stops on the date that BellSouth completes the collocation arrangement and notifies the CLEC.

Calculation

Arrangement Time = (a - b)

- a = Date Collocation Arrangement is Complete
- b = Date Order for Collocation Arrangement Submitted

Average Arrangement Time = $(c \div d)$

- c = Sum of all Arrangement Times
- d = Total Number of Collocation Arrangements Completed during Reporting Period.

Report Structure

- · Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

SQM Disaggregation - Retail Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• State	Virtual - 50 Calendar Days (Ordinary)
Virtual-Initial	Virtual - 75 Calendar Days (Extraordinary)
Virtual-Augment	Physical Caged - 90 Calendar Days
Physical Caged-Initial	Physical Cageless - 60 Calendar Days (Ordinary)
Physical Caged-Augment	Physical Cageless - 90 Calendar Days (Extraordinary)
Physical Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark:
Not Applicable	Not Applicable

(A) **BELLSOUTH** *

C-3: Collocation Percent of Due Dates Missed

Definition

Measures the percent of missed due dates for both virtual and physical collocation arrangements.

Exclusions

Any Bona Fide firm order canceled by the CLEC

Business Rules

Percent Due Dates Missed is the percent of total collocation arrangements which BellSouth is unable to complete by end of the BellSouth committed due date. The clock starts on the date that BellSouth receives a complete and accurate Bona Fide firm order accompanied by the appropriate fee if required. The arrangement is considered a missed due date if it is not completed on or before the committed due date.

Calculation

% of Due Dates Missed = $(a \div b) \times 100$

- a = Number of Completed Orders that were not completed within BellSouth Committed Due Date during Reporting Period
- b = Number of Orders Completed in Reporting Period

Report Structure

- Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- · Report period
- · Aggregate data

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• State	• \geq 95% on time
Virtual-Initial	
Virtual-Augment	
Physical Caged-Initial	
Physical Caged-Augment	
Physical Cageless-Initial	
Physical Cageless-Augment	

SEEM Measure

SEEM Measure		
	Tier I	X
Yes	Tier II	X
	Tier III	X

SEEM Disaggregation	SEEM Analog/Benchmark
All Collocation Arrangements	• $\geq 95\%$ on time.



Section 11: Change Management

CM-1: Timeliness of Change Management Notices

Definition

Measures whether CLECs receive required software release notices on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system software vendor changes. For example: a patch to fix a software problem.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process (CCP)

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Timeliness of Change Management Notices = $(a \div b) \times 100$

- a = Total number of Change Management Notifications Sent Within Required Time frames
- b = Total Number of Change Management Notifications Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
• Region	• 95% ≥ 30 days of Release



SEEM Measure

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	X

${\bf SEEM\ Disaggregation\ -\ Analog/Benchmark}$

SEEM Disaggregation	SEEM Analog/Benchmark
• Region	• 95% ≥ 30 days of Release



CM-2: Change Management Notice Average Delay Days

Definition

Measures the average delay days for change management system release notices sent outside the time frame set forth in the Change

Exclusions

- Changes to release dates for reasons outside BellSouth control, such as the system software vendor changes. For example: a patch to fix a software problem.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process

Business Rules

This metric is designed to measure the percent of change management notices sent to the CLECs according to notification standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the notification due date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. A revised notification would be required and the clock would restart. Based on release constraints for defects/expedites, notification may be less than the agreed upon interval in the CCP for new features.

Calculation

Change Management Notice Delay Days = (a - b)

- a = Date Notice Sent
- b = Date Notice Due

Change Management Notice Average Delay Days = $(c \div d)$

- c = Sum of all Change Management Notice Delay Days
- d = Total Number of Notices Sent Late

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- · Notice Date
- · Release Date

SQM Disaggregation - Analog/Benchmark

SQM Level of Disaggregation:	Retail Analog/Benchmark:
Region	• ≤8 Days

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

CM-3: Timeliness of Documents Associated with Change

CM-3: Timeliness of Documents Associated with Change

Definition

Measures whether CLECs received requirements or business rule documentation on time to prepare for BellSouth interface/system changes so CLEC interfaces are not impaired by change.

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process.

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Timeliness of Documents Associated with Change = $(a \div b) \times 100$

- a = Change Management Documentation Sent Within Required Time frames after Notices
- b = Total Number of Change Management Documentation Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	 95% ≥ 30 days if new features coding is required 95% ≥ 5 days for documentation defects, corrections or clarifications

SEEM Measure

SEEM Measure		
	Tier I	
Yes	Tier II	X
	Tier III	X



SEEM Disaggregation	SEEM Analog/Benchmark
• Region	• $95\% \ge 30$ days of the change

CM-4: Change Management Documentation Average Delay Days

(A) **BELLSOUTH**®

CM-4: Change Management Documentation Average Delay Days

Definition

Measures the average delay days for requirements or business rule documentation sent outside the time frames set forth in the Change

Exclusions

- Documentation for release dates that slip less than 30 days for reasons outside BellSouth control, such as changes due to Regulatory mandate or CLEC request.
- Type 6 Change Requests (Defects/Expedites), as defined by the Change Control Process.

Business Rules

This metric is designed to measure the percent of requirements or business rule documentation sent to the CLECs according to documentation standards and time frames set forth in the Change Control Process. The CCP is used by BellSouth and the CLECs to manage requested changes to the BellSouth Local Interfaces.

The clock starts on the business rule documentation release date. The clock stops on the software release date. When project events occur (scope changes, analysis information, etc.), the software release date may change. Revisions to documentation could be required and the clock would restart.

Calculation

Change Management Documentation Delay Days = (a - b)

- a = Date Documentation Provided
- b = Date Documentation Due

Change Management Documentation Average Delay Days = $(c \div d)$

- c = Sum of all CM Documentation Delay Days
- d = Total Change Management Documents Sent

Report Structure

· BellSouth Aggregate

Data Retained

- · Report Period
- · Notice Date
- · Release Date

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark:
• Region	• ≤8 Days

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable

CM-5: Notification of CLEC Interface Outages

Definition

Measures the time it takes BellSouth to notify the CLEC of an outage of an interface.

Exclusions

None

Business Rules

This measure is designed to notify the CLEC of interface outages within 15 minutes of BellSouth's verification that an outage has taken place. This metric will be expressed as a percentage.

Calculation

Notification of CLEC Interface Outages = $(a \div b) \times 100$

- a = Number of Interface Outages where CLECS are notified within 15 minutes
- b = Total Number of Interface Outages

Report Structure

· CLEC Aggregate

Data Retained

Relating to CLEC Experience	Relating to BellSouth Experience
 Number of Interface Outages Number of Notifications ≤ 15 minutes 	Not Applicable

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
By interface type for all interfaces accessed by CLECs	• 97% in 15 Minutes

Interface	Applicable to
EDI	CLEC
CSOTS	CLEC
LENS	CLEC
TAG	CLEC
ECTA	CLEC
TAFI	CLEC/BellSouth

SEEM Measure

SEEM Measure		
	Tier I	
No	Tier II	
	Tier III	



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Section 12: Bona Fide / New Business Request Process

BFR-1: Percentage of BFR/NBR Requests Processed Within 30 Business Days

Definition

Percentage of Bona Fide/New Business Requests processed within 30 business days for the development and purchases of network elements not currently offered.

Exclusions

Any application cancelled by the CLEC

Business Rules

The clock starts when BellSouth receives a complete and accurate application. The clock stops when BellSouth completes application processing for Network Elements that are not operational at the time of the request.

Calculation

Percentage of BFR/NBR Requests Processed Within 30 Business Days = $(a \div b) \times 100$

- a = Count of number of requests processed within 30 days
- b = Total number of requests

Report Structure

- Individual CLEC (alias) aggregate
- · Aggregate of all CLECs

Data Retained

- Report period
- · Aggregate data

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	• 90% ≤ 30 business days

SEEM Measure

SEEM Measure							
	Tier I						
No	Tier II						
	Tier III						



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



BFR-2: Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed Within X (10/30/60) Business Days

Definition

Percentage of quotes provided in response to Bona Fide/New Business Requests within X (10/30/60) business days for network elements not currently offered.

Exclusions

Requests that are subject to pending arbitration

Business Rules

The clock starts when BellSouth receives a complete and accurate application. The clock stops when BellSouth responds back to the application with a price quote.

Calculation

 $\textbf{Percentage of Quotes Provided for Authorized BFR/NBR Requests Processed Within X (10/30/60) Business \ Days = (a \div b) \ X \ 100 + (b)$

- a = Count of number of requests processed within "X" days
- b = Total number of requests where "X" = 10, 30, or 60 days

Report Structure

- New Network Elements that are operational at the time of the request.
- New Network Elements that are ordered by the FCC.
- New Network Elements that are not operational at the time of the request.

Data Retained

- · Report period
- · Aggregate data

SQM Level of Disaggregation - Analog/Benchmark

SQM Level of Disaggregation	Retail Analog/Benchmark
• Region	 90% ≤ 10/30/60 business days Network Elements that are operational at the time of the request – 10 days Network Elements that are Ordered by the FCC – 30 days New Network Elements – 90 days

SEEM Measure

SEEM Measure							
	Tier I						
No	Tier II						
	Tier III						



SEEM Disaggregation	SEEM Analog/Benchmark
Not Applicable	Not Applicable



Appendix A: Reporting Scope

A-1: Standard Service Groupings

See individual reports in the body of the SQM.

A-2: Standard Service Order Activities

These are the generic BellSouth/CLEC service order activities which are included in the Pre-Ordering, Ordering, and Provisioning sections of this document. It is not meant to indicate specific reporting categories.

Service Order Activity Types

- Service Migrations Without Changes
- · Service Migrations With Changes
- Move and Change Activities
- Service Disconnects (Unless noted otherwise)
- · New Service Installations

Pre-Ordering Query Types

- Address
- · Telephone Number
- · Appointment Scheduling
- Customer Service Record
- · Feature Availability
- · Service Inquiry

Maintenance Query Types:

TAFI - TAFI queries the systems below

- CRIS
- March
- · Predictor
- LMOS
 - DLR
 - DLETH
 - LMOSupd
- LNP
- NIW
- OSPCM
- SOCS

Report Levels

- CLEC RESH
- CLEC State
- · CLEC Region
- Aggregate CLEC State



- Aggregate CLEC Region
- BellSouth State
- BellSouth Region



Appendix B: Glossary of Acronyms and Terms

Symbols used in calculations

- Σ A mathematical symbol representing the sum of a series of values following the symbol.
- A mathematical operator representing subtraction.
- + A mathematical operator representing addition.
- ÷ A mathematical operator representing division.
- () Parentheses, used to group mathematical operations which are completed before operations outside the parentheses.

Α

ACD: Automatic Call Distributor - A service that provides status monitoring of agents in a call center and routes high volume incoming telephone calls to available agents while collecting management information on both callers and attendants.

Aggregate: Sum total of all items in like category, e.g. CLEC aggregate equals the sum total of all CLECs' data for a given reporting level.

ALEC: Alternative Local Exchange Company = FL CLEC

ADSL: Asymmetrical Digital Subscriber Line

ASR: Access Service Request - A request for access service terminating delivery of carrier traffic into a Local Exchange Carrier's network.

ATLAS: Application for Telephone Number Load Administration System - The BellSouth Operations System used to administer the pool of available telephone numbers and to reserve selected numbers from the pool for use on pending service requests/service orders.

ATLASTN: ATLAS software contract for Telephone Number.

Auto Clarification: The number of LSRs that were electronically rejected from LESOG and electronically returned to the CLEC for correction.

В

BFR: Bona Fide Request

BILLING: The process and functions by which billing data is collected and by which account information is processed in order to render accurate and timely billing.

BOCRIS: Business Office Customer Record Information System (Front-end to the CRIS database.)

BRI: Basic Rate ISDN



BRC: Business Repair Center - The BellSouth Business Systems trouble receipt center which serves business and CLEC customers.

BellSouth: BellSouth Telecommunications, Inc.

C

CABS: Carrier Access Billing System

CCC: Coordinated Customer Conversions

CCP: Change Control Process

Centrex: A business telephone service, offered by local exchange carriers, which is similar to a Private Branch Exchange (PBX) but the switching equipment is located in the telephone company Central Office (CO).

CKTID: A unique identifier for elements combined in a service configuration

CLEC: Competitive Local Exchange Carrier

CLP: Competitive Local Provider = NC CLEC

CM: Change Management

CMDS: Centralized Message Distribution System - Telcordia administered national system used to transfer specially formatted messages among companies.

COFFI: Central Office Feature File Interface - Provides information about USOCs and class of service. COFFI is a part of DOE/SONGS. It indicates all services available to a customer.

COG: Corporate Gateway - Telcordia product designed for the electronic submission of xDSL Local Service Requests.

CRIS: Customer Record Information System - The BellSouth proprietary corporate database and billing system for non-access customers and services.

CRSACCTS: CRIS software contract for CSR information

CRSG: Complex Resale Support Group

C-SOTS: CLEC Service Order Tracking System

CSR: Customer Service Record

CTTG: Common Transport Trunk Group - Final trunk groups between BellSouth & Independent end offices and the BellSouth access tandems.

D

DA: Directory Assistance

DESIGN: Design Service is defined as any Special or Plain Old Telephone Service Order which requires BellSouth Design Engineering Activities.

DISPOSITION & CAUSE: Types of trouble conditions, e.g. No Trouble Found, Central Office Equipment, Customer Premises Equipment, etc.



DLETH: Display Lengthy Trouble History - A history report that gives all activity on a line record for trouble reports in LMOS.

DLR: Detail Line Record - All the basic information maintained on a line record in LMOS, e.g. name, address, facilities, features etc.

DS-0: The worldwide standard speed for one digital voice signal (64000 bps).

DS-1: 24 DS-0s (1.544Mb/sec., i.e. carrier systems)

DOE: Direct Order Entry System - An internal BellSouth service order entry system used by BellSouth Service Representatives to input business service orders in BellSouth format.

DOM: Delivery Order Manager - Telcordia product designed for the electronic submission of xDSL Local Service Requests.

DSAP: DOE (Direct Order Entry) Support Application - The BellSouth Operations System which assists a Service Representative or similar carrier agent in negotiating service provisioning commitments for non-designed services and Unbundled Network Elements.

DSAPDDI: DSAP software contract for schedule information.

DSL: Digital Subscriber Line

DUI: Database Update Information

Ε

E911: Provides callers access to the applicable emergency services bureau by dialing a 3-digit universal telephone number.

EDI: Electronic Data Interchange - The computer-to-computer exchange of inter and/or intra-company business documents in a public standard format.

ESSX: BellSouth Centrex Service

F

Fatal Reject: LSRs electronically rejected from LEO, which checks to see of the LSR has all the required fields correctly populated.

Flow-Through: In the context of this document, LSRs submitted electronically via the CLEC mechanized ordering process that flow through to the BellSouth OSS without manual or human intervention.

FOC: Firm Order Confirmation - A notification returned to the CLEC confirming that the LSR has been received and accepted, including the specified commitment date.

FX: Foreign Exchange

G

Н

HAL: "Hands Off" Assignment Logic - Front end access and error resolution logic used in interfacing BellSouth Operations Systems such as ATLAS, BOCRIS, LMOS, PSIMS, RSAG and SOCS.

HALCRIS: HAL software contract for CSR information

HDSL: High Density Subscriber Loop/Line

I

ILEC: Incumbent Local Exchange Company

INP: Interim Number Portability

ISDN: Integrated Services Digital Network

IPC: Interconnection Purchasing Center

L

LAN: Local Area Network

LAUTO: The automatic processor in the LNP Gateway that validates LSRs and issues service orders.

LCSC: Local Carrier Service Center - The BellSouth center which is dedicated to handling CLEC LSRs, ASRs, and Preordering transactions along with associated expedite requests and escalations.

Legacy System: Term used to refer to BellSouth Operations Support Systems (see OSS)

LENS: Local Exchange Negotiation System - The BellSouth LAN/web server/OS application developed to provide both preordering and ordering electronic interface functions for CLECs.

LEO: Local Exchange Ordering - A BellSouth system which accepts the output of EDI, applies edit and formatting checks, and reformats the Local Service Requests in BellSouth Service Order format.

LERG: Local Exchange Routing Guide

LESOG: Local Exchange Service Order Generator - A BellSouth system which accepts the service order output of LEO and enters the Service Order into the Service Order Control System using terminal emulation technology.

LFACS: Loop Facilities Assessment and Control System

LIDB: Line Information Database

LISC: Local Interconnection Service Center - The center that issues trunk orders.

LMOS: Loop Maintenance Operations System - A BellSouth Operations System that stores the assignment and selected account information for use by downstream OSS and BellSouth personnel during provisioning and maintenance activities.



LMOS HOST: LMOS host computer

LMOSupd: LMOS updates

LMU: Loop Make-up

LMUS: Loop Make-up Service Inquiry

LNP: Local Number Portability - In the context of this document, the capability for a subscriber to retain his current telephone number as he transfers to a different local service provider.

LOOPS: Transmission paths from the central office to the customer premises.

LRN: Location Routing Number

LSR: Local Service Request – A request for local resale service or unbundled network elements from a CLEC.

M

Maintenance & Repair: The process and function by which trouble reports are passed to BellSouth and by which the related service problems are resolved.

MARCH: BellSouth Operations System which accepts service orders, interprets the coding contained in the service order image, and constructs the specific switching system Recent Change command messages for input into end office switches.

Ν

NBR: New Business Request

NC: "No Circuits" - All circuits busy announcement.

NIW: Network Information Warehouse

NMLI: Native Mode LAN Interconnection

NPA: Numbering Plan Area

NXX: The "exchange" portion of a telephone number.

0

OASIS: Obtain Availability Services Information System - A BellSouth front-end processor, which acts as an interface between COFFI and RNS. This system takes the USOCs in COFFI and translates them to English for display in RNS.

OASISBSN: OASIS software contract for feature/service

OASISCAR: OASIS software contract for feature/service

OASISLPC: OASIS software contract for feature/service

OASISMTN: OASIS software contract for feature/service

OASISNET: OASIS software contract for feature/service

OASISOCP: OASIS software contract for feature/service



ORDERING: The process and functions by which resale services or unbundled network elements are ordered from Bell-South as well as the process by which an LSR or ASR is placed with BellSouth.

OSPCM: Outside Plant Contract Management System - Provides Scheduling Information.

OSS: Operations Support System - A support system or database which is used to mechanize the flow or performance of work. The term is used to refer to the overall system consisting of hardware complex, computer operating system(s), and application which is used to provide the support functions.

OUT OF SERVICE: Customer has no dial tone and cannot call out.

Р

PMAP: Performance Measurement Analysis Platform

PMQAP: Performance Measurement Quality Assurance Plan

PON: Purchase Order Number

POTS: Plain Old Telephone Service

PREDICTOR: The BellSouth Operations system which is used to administer proactive maintenance and rehabilitation activities on outside plant facilities, provide access to selected work groups (e.g. RRC & BRC) to Mechanized Loop Testing and switching system I/O ports, and provide certain information regarding the attributes and capabilities of outside plant facilities.

Preordering: The process and functions by which vital information is obtained, verified, or validated prior to placing a service request.

PRI: Primary Rate ISDN

Provisioning: The process and functions by which necessary work is performed to activate a service requested via an LSR or ASR and to initiate the proper billing and accounting functions.

PSIMS: Product/Service Inventory Management System - A BellSouth database Operations System which contains availability information on switching system features and capabilities and on BellSouth service availability. This database is used to verify the availability of a feature or service in an NXX prior to making a commitment to the customer.

PSIMSORB: PSIMS software contract for feature/service.

Q

R

RNS: Regional Negotiation System - An internal BellSouth service order entry system used by BellSouth Consumer Services to input service orders in BellSouth format.

ROS: Regional Ordering System

RRC: Residence Repair Center - The BellSouth Consumer Services trouble receipt center which serves residential customers.

RSAG: Regional Street Address Guide - The BellSouth database, which contains street addresses validated to be accurate with state and local governments.



RSAGADDR: RSAG software contract for address search.

RSAGTN: RSAG software contract for telephone number search.

S

SAC: Service Advocacy Center

SEEM: Self Effectuating Enforcement Mechanism

SOCS: Service Order Control System - The BellSouth Operations System which routes service order images among Bell-South drop points and BellSouth Operations Systems during the service provisioning process.

SOG: Service Order Generator - Telcordia product designed to generate a service order for xDSL.

SOIR: Service Order Interface Record - any change effecting activity to a customer account by service order that impacts 911/E911

SONGS: Service Order Negotiation and Generation System.

T

TAFI: Trouble Analysis Facilitation Interface - The BellSouth Operations System that supports trouble receipt center personnel in taking and handling customer trouble reports.

TAG: Telecommunications Access Gateway – TAG was designed to provide an electronic interface, or machine-to-machine interface for the bi-directional flow of information between BellSouth's OSSs and participating CLECs.

TN: Telephone Number

Total Manual Fallout: The number of LSRs which are entered electronically but require manual entering into a service order generator.

U

UNE: Unbundled Network Element

UCL: Unbundled Copper Link

USOC: Universal Service Order Code

V

W

WATS: Wide Area Telephone Service

WFA: Work Force Administration

WMC: Work Management Center

WTN: Working Telephone Number.



X

Υ

Z



Appendix C: BellSouth Audit Policy

BellSouth currently provides many CLECs with certain audit rights as a part of their individual interconnection agreements. However, it is not reasonable for BellSouth to undergo an audit of the SQM for every CLEC with which it has a contract. BellSouth has developed a proposed Audit Plan for use by the parties to an audit. If requested by a Public Service Commission or by a CLEC exercising contractual audit rights, BellSouth will agree to undergo a comprehensive audit of the aggregate level reports for both BellSouth and the CLEC(s) each of the next five (5) years (2001-2005) to be conducted by an independent third party. The results of that audit will be made available to all the parties subject to proper safeguards to protect proprietary information. This aggregate level audit includes the following specifications:

- 1. The cost shall be borne 50% by BellSouth and 50% by the CLEC or CLECs.
- 2. The independent third party auditor shall be selected with input from BellSouth, the PSC, if applicable, and the CLEC(s).
- 3. BellSouth, the PSC and the CLEC(s) shall jointly determine the scope of the audit.

BellSouth reserves the right to make changes to this audit policy as growth and changes in the industry dictate.

Amendment to the Interconnection Agreement By and Between BellSouth Telecommunications, Inc. And ITC^DeltaCom Communications, Inc. Dated January 31, 2002

This Amendment ("Amendment") is made and entered into by and between ITC^DeltaCom Communications, Inc. ("ITC^DeltaCom") and BellSouth Telecommunications, Inc. ("BellSouth") to amend the Interconnection Agreement ("the Agreement") entered into by ITC^DeltaCom and BellSouth on January 31, 2002 for the state of South Carolina.

WHEREAS, the Parties desire to amend that certain Interconnection Agreement between BellSouth and ITC^DeltaCom dated January 31, 2002 to incorporate collocation rates established by the South Carolina Public Service Commission in Docket Number 2001-65-C-Order Number 2001-1089, November 30, 2001;

NOW THEREFORE, in consideration of the provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, ITC^DeltaCom and BellSouth hereby covenant and agree as follows:

- 1. The rate sheets for South Carolina in Attachment 11, Table 1 of the Agreement are hereby amended to add Collocation rates for South Carolina in Exhibit 1 attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Interconnection Agreement shall remain unchanged and in full force and effect.
- 3. Either or both of the Parties are authorized to submit this Amendment to the South Carolina Public Service Commission or other Regulatory Agency for approval subject to Section 252 (e) of the Federal Telecommunications Act of 1996.
- 4. This Amendment is made effective thirty (30) calendar days following the last signature of both Parties.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

ITC^DeltaCom Communications, Inc.	BellSouth Telecommunications Inc.
Signature	Signature
Name	Name
Title	Title
Date	Date

BellSouth/ITC^DeltaCom Rates South Carolina

CATEGOR		UNBUNDLED NETWORK ELEMENT	Interim	Zon	BCS	USOC						
Y	NOTES	ONDONDEED NETWORK ELEMENT	IIILEIIIII	е		0300	RATES (\$)					
ı												
ŀ												
ŀ												
ŀ												
ŀ												
ŀ												
				<u> </u>				Nonrec	urring		curring	
ļ!											nnect	
							Rec	First	Add'l	First	Add'l	
	COLLOCA	ATION										
	L											
PHYSICAL (COLLOCAT											
		Physical Collocation - Application Fee - Initial			CLO	PE1BA		1,883.67	1,883.67	0.51	0.51	
		Physical Collocation - Application Fee - Subsequent			CLO	PE1CA		1,570.10	1,570.10	0.51	0.51	
		Physical Collocation - Space Preparation - Firm Order Processing			CLO	PE1SJ		602.05	602.05			
ı		Physical Collocation - Space Preparation - C.O. Modification per square ft.	_		CLO	PE1SK	2.75					
		Physical Collocation - Space Preparation - Common Systems Modification per square										
l		ft Cageless	- 1		CLO	PE1SL	3.24					
ŀ		Physical Collocation - Space Preparation - Common Systems Modification per Cage	- 1		CLO	PE1SM	110.16					
ı		Physical Collocation - Cable Installation			CLO	PE1BD		794.22	794.22	22.54	22.54	
		Physical Collocation - Floor Space per Sq. Ft.			CLO	PE1PJ	3.95					
		Physical Collocation - Cable Support Structure			CLO	PE1PM	21.33					
		Physical Collocation - Power per Fused Amp	-		CLO	PE1PL	9.19					
		Physical Collocation - 120V, Single Phase Standby Power Rate	-		CLO	PE1FB	5.67					
		Physical Collocation - 240V, Single Phase Standby Power Rate			CLO	PE1FD	11.36					
		Physical Collocation - 120V, Three Phase Standby Power Rate			CLO	PE1FE	17.03					
		Physical Collocation - 277V, Three Phase Standby Power Rate	i		CLO	PE1FG	39.33					
		I nyolodi boliobalion. Errit, mico i nabo blanday i bilor nabo	·		UEANL,UEA,UDN,		00.00					
ŀ					UDC,UAL,UHL,UC							
ļ		Physical Collocation - 2-Wire Cross-Connects			L,UEQ	PE1P2	0.034	12.32	11.83	6.04	5.45	
		Physical Collocation - 4-Wire Cross-Connects		1	CLO	PE1P4	0.068	12.42	11.90	6.40	5.74	
	 	Triyologi Collocation - Trino Close Collinotte		1	CLO,UEANL,UEQ,	12.114	0.000	12.72	11.50	0.40	5.74	
ŀ		Physical Collocation - DS1 Cross-Connects			WDS1L.WDS1S	PE1P1	1.12	22.08	15.96	6.42	5.80	
		Physical Collocation - DS3 Cross-Connects		+	CLO	PE1P3	14.21	20.94	15.23	7.39	5.93	
		Physical Collocation - 2-Fiber Cross-Connect		1	CLO	PE1F2	2.82	20.94	15.23	7.40	5.93	
	\vdash	Physical Collocation - 4-Fiber Cross-Connect		1	CLO	PE1F4	5.01	25.61	19.90	9.73	8.26	
		Physical Collocation - Welded Wire Cage - First 100 Sq. Ft.		1	CLO	PE1BW	219.19	25.01	19.90	9.73	0.20	
	\vdash	Physical Collocation - Welded Wire Cage - Add'l 50 Sq. Ft.		1	CLO	PE1CW	21.50					
	\vdash	Physical Collocation - Weided Wife Cage - Add 130 Sq. 1 t. Physical Collocation - Security Access System - Security System per Central Office	-	1	CLO	PE1AX	74.72					
		Physical Collocation - Security Access System - Security System per Central Office Physical Collocation - Security Access System - New Access Card Activation, per	'		CLO	FEIAA	14.12					
ŀ		Card	1		CLO	PE1A1	0.06	27.85	27.85			
		Physical Collocation-Security Access System-Administrative Change, existing Access	'		CLO	FEIAI	0.00	21.00	21.00			
ŀ		Card, per Card			CLO	PE1AA		7.81	7.81			
		Physical Collocation - Security Access System - Replace Lost or Stolen Card, per			CLO	FEIAA		7.01	7.01			
ļ l	I	Card		1	CLO	PE1AR		22.83	22.83			
		Physical Collocation - Security Access - Initial Key, per Key	1	+	CLO	PE1AK PE1AK		13.13	13.13			
		Physical Collocation - Security Access - Initial Key, per Key Physical Collocation - Security Access - Key, Replace Lost or Stolen Key, per Key		 	CLO	PE1AL		13.13	13.13			
	├──	Physical Collocation - Space Availability Report per premises	-	 	CLO	PE1SR		1,077.57	1,077.57			
	 	r riyordar Comodation - Space Avaliability Neport per premises	<u> </u>	1	UEANL,UEA,UDN,	FEISK		1,077.57	1,077.57			
	I			1	UDC,UAL,UHL,UC							
ŀ	1	Physical Collegation 2 Wire POT Pay			L,UEQ,CLO	PE1PE	0.0050]			
		Physical Collocation - 2-Wire POT Bay		+	UEANL,UEA,UDN,	PEIPE	0.0850	-				
ŀ	I			1	UDC,UAL,UHL,UC]			
ŀ	1	Physical Callegation 4 Wire POT Pay				DE4DE	0.1704]			
		Physical Collocation - 4-Wire POT Bay			L,UEQ,CLO	PE1PF	0.1701					

BellSouth/ITC^DeltaCom Rates South Carolina

CATEGOR Y	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zon e	BCS	USOC	RATES (\$)						
'	NOILS			-	UEANL,UEA,UDN,		1	- N	Α, Ευ (ψ)	T			
					UDC,UAL,UHL,UC								
					L,UEQ,CLO,WDS1								
		Physical Collocation - DS1 POT Bay			L,WDS1S,	PE1PG	1 20						
		Physical Collocation - DST POT Bay			UEANL,UEA,UDN,	PEIPG	1.20						
					UDC,UAL,UHL,UC								
		Physical Callegation - DC2 DOT Day			L,UEQ,CLO	DEADLL	40.74						
		Physical Collocation - DS3 POT Bay			UEANL,UEA,UDN,	PE1PH	10.71						
					UDC,UAL,UHL,UC								
		Dissipation Cally and the Cally and DOT Day				DE 4 D 0	00.55						
		Physical Collocation - 2-Fiber POT Bay			L,UEQ,CLO UEANL,UEA,UDN,	PE1B2	36.55						
1													
		DI 1 10 II 11 15 15 15 15 15 15 15 15 15 15 15 15			UDC,UAL,UHL,UC	55.15.4							
		Physical Collocation - 4-Fiber POT Bay			L,UEQ,CLO	PE1B4	49.29						
		Collocation Cable Records - per request			CLO	PE1CR		760.98	489.20	133.29	133.29		
		Collocation Cable Records - VG/DS0 Cable, per cable record			CLO	PR1CD		327.65	327.65	189.54	189.54		
		Collocation Cable Records - VG/DS0 Cable, per each 100 pair			CLO	PE1CO		4.82	4.82	5.91	5.91		
		Collocation Cable Records - DS1, per T1TIE			CLO	PE1C1		2.26	2.26	2.77	2.77		
		Collocation Cable Records - DS3, per T3TIE			CLO	PE1C3		7.90	7.90	9.68	9.68		
		Collocation Cable Records - Fiber Cable, per 99 fiber records			CLO	PE1CB		84.68	84.68	77.30	77.30		
		Physical Collocation - Security Escort - Basic, per Half Hour			CLO,CLORS	PE1BT	1	16.96	10.75				
		Physical Collocation - Security Escort - Overtime, per Half Hour			CLO.CLORS	PE1OT		22.10	13.89				
		Physical Collocation - Security Escort - Premium, per Half Hour			CLO,CLORS	PE1PT		27.23	17.02				
		Physical Collocation - Co-Carrier Cross Connects - Fiber Cable Support Structure,			OLO, OLO IXO			21.25	17.02				
		per linear ft.			CLO	PE1ES	0.0010						
		Physical Collocation - Co-Carrier Cross Connects - Copper/Coax Cable Support			CLO	TEILS	0.0010						
		1 '			01.0	DEADO	0.0045						
		Structure, per lin. ft.			CLO	PE1DS	0.0015	504.40					
		Physical Collocation - Co-Carrier Cross Connects - Application Fee, Subsequent						584.42					
ASSEMBLY	DOINT						-						
ASSEMBLY	POINT	Accomply Deint 2 Win Cons Connects					60.70	£40.00	C44.00	0.04	F 45		
		Assembly Point - 2-Wire Cross Connects					\$0.72	\$12.32	\$11.83	6.04	5.45		
		Assembly Point - 4-Wire Cross Connects					\$1.45	\$12.42	\$11.90	6.40	5.74		
		Assembly Point - DS1 Cross Connects					\$11.64	\$22.08	\$15.96	6.42	5.80		
ADJACENT	COLLOCA	-											
		Adjacent Collocation - Space Charge per Sq. Ft.			CLO	PE1JA	0.094						
		Adjacent Collocation - Electrical Facility Charge per Linear Ft.			CLO	PE1JC	6.40						
		Adjacent Collocation - 2-Wire Cross-Connects			CLO	PE1P2	0.026	12.32	11.83	6.04	5.45		
					UEA,UHL,UDL,UC								
		Adjacent Collocation - 4-Wire Cross-Connects			L,CLO	PE1P4	0.053	12.42	11.90	6.40	5.74		
		Adjacent Collocation - DS1 Cross-Connects			USL,CLO	PE1P1	1.03	22.08	15.96	6.42	5.80		
		Adjacent Collocation - DS3 Cross-Connects			CLO	PE1P3	14.00	20.94	15.23	7.39	5.93		
		Adjacent Collocation - 2-Fiber Cross-Connect			CLO	PE1F2	2.37	20.94	15.23	7.40	5.93		
		Adjacent Collocation - 4-Fiber Cross-Connect			CLO	PE1F4	4.53	25.61	19.90	9.73	8.26		
		Adjacent Collocation - Application Fee			CLO	PE1JB		1,580.20	10.00	0.51	0.20		
		Adjusterit Comodulori Application 1 co			020	1 L TOD		1,000.20		0.01			
		Adjacent Collocation - 120V, Single Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FB	5.67						
		Adjacent Conocation - 1201, Single I hase Standby I ower trate per AC Breaker Amp			CLO	ILIID	3.07						
		Adjacent Collocation - 240V, Single Phase Standby Power Rate per AC Breaker Amp			CLO	PE1FD	11.36						
 		Aujacent Conocation - 240V, Single Phase Standby Power Kate per AC Breaker Amp		!	CLU	PEIFU	11.36						
		Adjacent Collegation 120V/ Three Phase Ctardley Payers Bata and AC Basel on A con-			CLO	DE4EE	47.00						
		Adjacent Collocation - 120V, Three Phase Standby Power Rate per AC Breaker Amp		<u> </u>	CLO	PE1FE	17.03						
		A live of Oally of the OTTM Three Phase Of the Property of the A			01.0	DE4E0	00.00						
1		Adjacent Collocation - 277V, Three Phase Standby Power Rate per AC Breaker Amp		<u> </u>	CLO	PE1FG	39.33						
PHYSICAL (COLLOCAT	ION IN THE REMOTE SITE											
PHYSICAL (COLLOCAT	Physical Collocation in the Remote Site - Application Fee			CLO	PE1RA		308.38	308.38	168.60	168.60		
PHYSICAL (COLLOCAT				CLO CLO CLO	PE1RA PE1RB PE1RD	246.44	308.38	308.38	168.60	168.60		

SC

BellSouth/ITC^DeltaCom Rates South Carolina

CATEGOR Y	NOTES	UNBUNDLED NETWORK ELEMENT	Interim	Zon e	BCS	usoc		RATES (\$)	
		Physical Collocation in the Remote Site - Space Availability Report per Premises							
		Requested			CLO	PE1SR	116.13	116.13	
		Physical Collocation in the Remote Site - Remote Site CLLI Code Request, per CLLI							
		Code Requested			CLO	PE1RE	37.64	37.64	
	NOTE: If Se	curity Escort and/or Add'l Engineering Fees become necessary for remote site collocati	ion, the Pa	arties v	will negotiate approp	riate rates.			

AMENDMENT TO THE

AGREEMENT BETWEEN ITC^DELTACOM COMMUNICATIONS, INC. d/b/a ITC^DELTACOM AND

BELLSOUTH TELECOMMUNICATIONS, INC. DATED JANUARY 31, 2002

Pursuant to this Amendment, (the "Amendment"), ITC^DeltaCom Communications, Inc. d/b/a ITC^DeltaCom, ("ITC^DeltaCom") and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties," hereby agree to amend that certain Interconnection Agreement between the Parties dated January 31, 2002 ("Agreement") to be effective on the date of the last signature executing the Amendment.

WHEREAS, BellSouth and ITC^DeltaCom entered into the Agreement on January 31, 2002, and;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Parties agree to add to the rates in Attachment 11, the rates set forth in Exhibit 1 of this Amendment, attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Agreement, dated January 31, 2002, shall remain in full force and effect.
- 3. Either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.
- 4. Neither party waives any right to seek clarification from the Commission regarding retroactive application of the rates contained in this Amendment.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

ITC^DeltaCom Communications, Inc. d/b/a ITC^DeltaCom	BellSouth Telecommunications, Inc.
By:	Ву:
Name:	Name:
Title:	Title:
Date:	Date:

INBUNDLE	D NETWORK ELEMENTS - South Carolina													nent: 11		
ATEGORY	RATE ELEMENTS	Interim	Zone	one BCS	usoc							Submitted	Manual Svc	Incremental Charge - Manual Svc Order vs. Electronic- Add'l	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Incremen Charge Manual S Order vs Electroni Disc Add
						_	Nonrec	urring	Nonrecurrin	g Disconnect		1	oss	Rates(\$)		
						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN		SOMAN	SOMAN	SOMAN
	one" shown in the sections for stand-alone loops or loops as part www.interconnection.bellsouth.com/become_a_clec/html/interconn			refers to Geographic	ally Deaverag	ed UNE Zones.	To view Geogr	aphically Deave	eraged UNE Zo	ne Designations	by Central	Office, refer	to Internet We	ebsite:		
NBUNDLE	D LOCAL EXCHANGE SWITCHING(PORTS)															
Excha	ange Ports															
NOTE:	Although the Port Rate includes all available features in GA,	KY. LA 8	TN. the	e desired features w	ill need to be	ordered using	retail USOCs									
							,									
2-WID	RE VOICE GRADE LINE PORT RATES (RES)				-											
2-7711	South Carolina Extended Local Dialing Port without Caller ID	1														
	capability			UEPSR	UEPWL	1.65	2.38	2.28	1.42	1.33		15.69				
	Capability			02. 0.0		1.00	2.00	2.20				10.00				
2-WIR	E VOICE GRADE LINE PORT RATES (BUS)															
	South Carolina Extended Local Dialing Port without Caller ID													İ		
	capability			UEPSB	UEPWM	1.65	2.38	2.28	1.42	1.33		15.69				
	South Carolina Business Area Calling Port without Caller ID				UEPBB											
	capbility			UEPSB		1.65	2.38	2.28	1.42	1.33		15.69				
	Incoming Only without Caller ID capability			UEPSB	UEPBE	1.65	2.38	2.28	1.42	1.33		15.69				
INDUNDUE	D DODT// OOD COMPINATIONS COST BASED DAT	FFC			-					 						
NBUNDLE	D PORT/LOOP COMBINATIONS - COST BASED RAT	IES														
	Based Rates are applied where BellSouth is required by FCC and															
	ures shall apply to the Unbundled Port/Loop Combination - Cost E															
	first and additional Port nonrecurring charges apply to Not Currer	ntly Comb	ined Co	mbos. For Currently	y Combined C	ombos, the non	recurring charg	es shall be tho	se identified in	the Nonrecurrin	g - Currently	/ Combined s	sections.			
Addition	nal NRCs may apply also and are categorized accordingly.									1	1	1	1			
2-WIR	LE VOICE GRADE LOOP WITH 2-WIRE LINE PORT (RES)														
2-Wire	Voice Grade Line Port Rates (Res)															
	South Carolina Extended Local Dialing Port without Caller ID				UEPWL											
	capability			UEPRX	UEPWL	1.13	37.93	16.72				15.69				
1			igspace											ļ		
	L VOIGE ORA DE LOOR WITH A WIRE LINE BOST (T. C.)	1	\vdash		<u> </u>					ļ						
0.14/15-		1	+		+					ļ	1	ļ		1		
2-WIRE	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)				1					1				+		
	Voice Grade Line Port (Bus)															
	Voice Grade Line Port (Bus) South Carolina Extended Local Dialing Port without Caller ID capability			UEPBX	UEPWM	1.13	37.93	16.72				15.69				
	Voice Grade Line Port (Bus) South Carolina Extended Local Dialing Port without Caller ID capability South Carolina Business Area Calling Port without Caller ID				UEPWM											
	Voice Grade Line Port (Bus) South Carolina Extended Local Dialing Port without Caller ID capability			UEPBX UEPBX UEPBX		1.13 1.13 1.13	37.93 37.93 37.93	16.72 16.72 16.72				15.69 15.69 15.69				

AMENDMENT TO THE

AGREEMENT BETWEEN ITC^DELTACOM COMMUNICATIONS, INC. d/b/a ITC^DELTACOM

AND BELLSOUTH TELECOMMUNICATIONS, INC. DATED JANUARY 31, 2002

Pursuant to this Amendment, (the "Amendment"), ITC^DeltaCom Communications, Inc. d/b/a ITC^DeltaCom, ("ITC^DeltaCom") and BellSouth Telecommunications, Inc. ("BellSouth"), hereinafter referred to collectively as the "Parties," hereby agree to amend that certain Interconnection Agreement between the Parties dated January 31, 2002 ("Agreement") to be effective on the date of the last signature executing the Amendment.

WHEREAS, BellSouth and ITC^DeltaCom entered into the Agreement on January 31, 2002, and;

NOW THEREFORE, in consideration of the mutual provisions contained herein and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties hereby covenant and agree as follows:

- 1. The Parties agree to add to the rates in Attachment 11, the rates set forth in Exhibit 1 of this Amendment, attached hereto and incorporated herein by this reference.
- 2. All of the other provisions of the Agreement, dated January 31, 2002, shall remain in full force and effect.
- 3. Either or both of the Parties is authorized to submit this Amendment to the respective state regulatory authorities for approval subject to Section 252(e) of the Federal Telecommunications Act of 1996.
- 4. Neither party waives any right to seek clarification from the Commission regarding retroactive application of the rates contained in this Amendment.

IN WITNESS WHEREOF, the Parties hereto have caused this Amendment to be executed by their respective duly authorized representatives on the date indicated below.

ITC^DeltaCom Communications, Inc.

BellSouth Telecommunications, Inc.

d/b/a ITC^DeltaCom

By: (Signature on File)

By: (Signature on File)

Name: Jerry Watts Name: Elizabeth R. A. Shiroishi

Title: Vice President Title: Assistant Director.

Interconnection Services

Date: 09/19/02 Date:09/20/02

UNBUNDLE	D NETWORK ELEMENTS - Kentucky												Attachr	ment: 11		
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec	Manually	Manual Svc Order vs. Clectronic-1st CAdd'l Disc 1st COS Rates(\$) SOMAN SOMAN SOMAN SOMAN SOMAN Sile:			
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR	Order vs.	Order vs.	Order vs.	Order vs.
											•	-	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
		ļ											l			
						Rec	Nonrecurring		Nonrecurring Disconnect							
							First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
The "7	Lone" shown in the sections for stand-alone loops or loops as part of a	ambinatia	n refere	to Coographically Doo	Larged LINE	Zanas Tavisuu	Coographically F	Nonversed LINE	Zone Designati	ana hu Cantral Of	fina rafartal	ntornot Woh	ita	l .	l .	1
	ww.interconnection.bellsouth.com/become a clec/html/interconnection		n reters	to Geographically Dea	iveraged UNE	Zones. To view	Jeographically L	eaveraged UNE	E Zone Designation	ons by Central Or	rice, refer to i	nternet webs	site:			
	D LOCAL EXCHANGE SWITCHING(PORTS)	1.11011	1		1	1					ı		1	ı	1	1
UNBUNDLE	D LOCAL EXCHANGE SWITCHING(FORTS)															+
	<u> </u>															
	ange Ports															
	: Although the Port Rate includes all available features in GA, K	Y, LA & T	N, the d	esired features will	need to be or	dered using ret	ail USOCs									
2-WIF	RE VOICE GRADE LINE PORT RATES (BUS)															
	Kentucky Extended Local Dialing Port without Caller ID capability			UEPSB	UEPWF	1.49	3.74	3.63	2.23	2.13		7.86				
	Incoming Only without Caller ID capability			UEPSB	UEPBE	1.49	3.74	3.63	2.23	2.13		7.86				
UNBUNDLE	L D PORT/LOOP COMBINATIONS - COST BASED RATE	S														
> Cost	Based Rates are applied where BellSouth is required by FCC and/or S	state Comm	nission r	ule to provide Unbundl	ed Local Switc	hing or Switch Po	rts.									
> Feat	ures shall apply to the Unbundled Port/Loop Combination - Cost Based	Rate secti	ion in the	same manner as they	are applied to	the Stand-Alone	Unbundled Port	section of this R	ate Exhibit.							
	Office and Tandem Switching Usage and Common Transport Usage ra									Port/Loop Comb	inations					
	first and additional Port nonrecurring charges apply to Not Currently Co											s Additional	NRCs may			1
	also and are categorized accordingly.		01110001	r or ourronay combin	04 00111000, 111	o nonecounting of	iai goo orian bo n	1000 1001 111100 111		g carrona, com		o. / idailioriai	· · · · co · · · · · · ·			
	E VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)	1	1		1	l					l .	1	1			-
	- VOICE ORABE ESST WITH E VIREE ENVET ORT (BOS)															+
2-Wire	voice Grade Line Port (Bus)															
	Kentucky Extended Local Dialing Port without Caller ID capability			HEDDY	UEPWF		24.00	45.40	0.05	0.07		7.00				
	Incominy Only without Caller ID capability		+	UEPBX UEPBX	UEPBE	1.15 1.15	21.29 21.29	15.49 15.49	2.85 2.85	2.67 2.67	ļ	7.86 7.86		 	<u> </u>	-

UNBL	INDLE	NETWORK ELEMENTS - Louisiana												Attachn	nent: 11		
CATEG	GORY	RATE ELEMENTS						T. Marranovasia	g Disconnect		Submitted Manually	Charge - Manual Svc Order vs. Electronic- 1st	Charge - Manual Svc Order vs.	Incremental Charge - Manual Svc Order vs. Electronic- Disc 1st	Charge -		
							Rec	First	curring Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
									7.00.		7144	0020		00111711	00	00	00
	T						LINE 7	T		11005.7			0000		b - 20 -		
		ne" shown in the sections for stand-alone loops or loops as part oww.interconnection.bellsouth.com/become_a_clec/html/interconnection.bellsouth.com/become_a_clec/html/interconnection.			refers to Geographica	ally Deaverage	ed UNE Zones.	To view Geogr	aphically Deave	eraged UNE Zo	ne Designations	by Central	Office, refer	to Internet We	bsite:		
UNBL		D LOCAL EXCHANGE SWITCHING(PORTS)		Ï													
	Excha	nge Ports															
	NOTE:	Although the Port Rate includes all available features in GA, I	KY, LA &	TN, th	e desired features wi	ill need to be	ordered using	retail USOCs									
	0.14/10	E VOICE OR ARE LINE BORT RATEO (RUO)															
	2-WIR	E VOICE GRADE LINE PORT RATES (BUS) Louisiana Extended Local Dialing Port without Caller ID									+						
		capability			UEPSB	UEPWH	1.52	2.31	2.21				15.20				
		Louisiana Business Area Calling Port without Caller ID capability				UEPBA											
		Incoming Only without Caller ID capability			UEPSB	UEPBE	1.52 1.52	2.31	2.21	ļ	1		15.20 15.20				
		incoming Only without Caller ID capability				OLFBL	1.32	2.31	2.21				13.20				
	FEAT	JRES															
		All Available Vertical Features			UEPSB	UEPVF	0.00	0.00	0.00				15.20				
UNRU	INDI FI	PORT/LOOP COMBINATIONS - COST BASED RAT	FS														
ONDO		TORTIZOGI GOMBINATIONO GOOT BAGED KAT															
	> Cost I	Based Rates are applied where BellSouth is required by FCC and	l/or State	Commi	ssion rule to provide U	Jnbundled Lo	cal Switching o	r Switch Ports.	ı	l	1						
		res shall apply to the Unbundled Port/Loop Combination - Cost B					-		bundled Port se	ection of this Ra	ate Exhibit.						
		office and Tandem Switching Usage and Common Transport Usage															
		irst and additional Port nonrecurring charges apply to Not Curren	tly Comb	ined Co	mbos. For Currently	Combined C	ombos, the non	recurring charg	es shall be tho	se identified in	the Nonrecurrin	g - Currently	Combined	sections.			
	Addition	al NRCs may apply also and are categorized accordingly.	1	1			1		ı		1						
			-														
	2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
	2-Wire	Voice Grade Line Port (Bus) Louisiana Extended Local Dialing Port without Caller ID		-							-						
		capability			UEPBX	UEPWH	1.36	38.85	19.08				15.20				
		Louisiana Business Area Calling Port without Caller ID capability				UEPBA											
		Incoming Only without Caller ID capability		1	UEPBX UEPBX	UEPBE	1.36 1.36	38.85 38.85	19.08 19.08			-	15.20 15.20				
		incoming only without caner ib capability			OLIBA	OLI DL	1.50	30.03	19.00				13.20				
	FEATU																
		All Features Offered		-	UEPBX	UEPVF	0.00	0.00	0.00		1		15.20				
											1						
UNBU	INDLE	PORT LOOP COMBINATIONS - MARKET RATES															
	> Mark	et Rates shall apply where BellSouth is not required to provide un	hundled	local si	witching or switch port	s ner ECC an	nd/or State Com	mission rules									
		ludes unbundled port/loop combinations that are Currently Comb							n for end users	with 4 or more	DS0 equivalent	lines.					
		op 8 MSAs in BellSouth's region are: FL (Orlando, Ft. Lauderdale															
		outh currently is developing the billing capability to mechanically larket Rates and reserves the right to true-up the billing difference		ecurring	and non-recurring Ma	arket Rates in	this section. Ir	n the interim wh	ere BellSouth o	cannot bill Mark	et Rates, BellSo	outh shall bil	I the rates in	the Cost-Base	ed section pre	ceding in lieu	
		larket Rates and reserves the right to true-up the billing difference. Market Rate for unbundled ports includes all available features in		S.													†
	> End (Office and Tandem Switching Usage and Common Transport Usa	age rates	in the F													
		lot Currently Combined scenarios where Market Rates apply, the ed section. Additional NRCs may apply also and are categorized			arges are listed in the	First and Add	ditional NRC col	umns for each	Port USOC. Fo	or Currently Co	mbined scenario	s, the Nonre	ecurring cha	ges are listed	in the NRC - 0	Currently	
			accordin	gry.													
	2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															

UNB	UNBUNDLED NETWORK ELEMENTS - Louisiana														nent: 11		
												Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
CATEGORY		RATE ELEMENTS												Charge -	Charge -		Charge -
				-		11000											Manual Svc
			Interim	Zone	BCS	USOC	RATES(\$)						per LSR				Order vs.
					i									Electronic-	Electronic-	Electronic-	
														1st	Add'l	Disc 1st	Disc Add'l
							Rec	Nonred	urring	ing Nonrecurring Disconnect				oss	Rates(\$)		
							Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
		Voice Grade Line Port (Bus)															
		Louisiana Extended Local Dialing Port without Caller ID capability			UEPBX	UEPWH	14.00	90.00	90.00					31.92	7.32		
		Louisiana Business Area Calling Port without Caller ID capability			UEPBX	UEPBA	14.00	90.00	90.00					31.92	7.32		
		Incoming Only without Caller ID capability			UEPBX	UEPBE	14.00	90.00	90.00					31.92	7.32		
	FEATURES															<u> </u>	
		All Features Offered			UEPBX	UEPVF	0.00	0.00	0.00					31.92	7.32	ļ	ļ
	1																

UNBUNDLE	NETWORK ELEMENTS - Mississippi												Attachr	nent: 11		
											Svc Order	Svc Order	Incremental	Incremental	Incremental	Incremental
											Submitted	Submitted	Charge -	Charge -	Charge -	Charge -
											Elec	Manually	Manual Svc	Manual Svc	Manual Svc	Manual Svc
CATEGORY	RATE ELEMENTS	Interim	Zone	BCS	USOC			RATES(\$)			per LSR	per LSR		Order vs.	Order vs.	Order vs.
											,	P	Electronic-	Electronic-	Electronic-	Electronic-
													1st	Add'l	Disc 1st	Disc Add'l
						1	Nonrec	urring	Nonrecurring	Disconnect			088	Rates(\$)		
 						Rec	First	Add'l	First	Add'l	SOMEC	SOMAN	SOMAN	SOMAN	SOMAN	SOMAN
							11130	Auu i	11131	Addi	JOINEC	JOINAN	JONAN	JOHIAN	JOHAN	JOHAN
The "Zo	ne" shown in the sections for stand-alone loops or loops as part	of a comb	ination i	refers to Geographica	Ily Deaverage	ed UNE Zones.	To view Geogra	aphically Deave	raged UNE Zor	ne Designation	s by Central	Office, refer	to Internet We	bsite:	L	-
http://w	ww.interconnection.bellsouth.com/become_a_clec/html/interconn	ection.htm	n	0 .	, ,		Ü	. ,	Ü	Ü	•					
UNBUNDLE	LOCAL EXCHANGE SWITCHING(PORTS)															
1											-					-
Eveha	nge Ports										1					
	•	<u> </u>									1					├
	Although the Port Rate includes all available features in GA,	KY, LA &	TN, the	desired features wi	Il need to be	ordered using	retail USOCs									
	E VOICE GRADE LINE PORT RATES (BUS)															
	Mississippi Extended Local Dialing Port without Caller ID				UEPWK											
	capability			UEPSB	_	1.41	2.39	2.29	1.42	1.33		15.75				
	Incoming Only without Caller ID capability			UEPSB	UEPBE	1.41	2.39	2.29	1.42	1.33		15.75				
ONBONDLEI	PORT/LOOP COMBINATIONS - COST BASED RAT	ES														
> Cost F	Based Rates are applied where BellSouth is required by FCC and	d/or State	Commis	ssion rule to provide U	Jnbundled Lo	cal Switching or	Switch Ports.									
> Featu	res shall apply to the Unbundled Port/Loop Combination - Cost E	Based Rat	e section	n in the same manne	r as they are	applied to the S	tand-Alone Unb	undled Port se	ction of this Ra	te Exhibit.						
> End C	Office and Tandem Switching Usage and Common Transport Usa	ne rates ir	the Po	ort section of this rate	exhibit shall a	apply to all com	hinations of loor	n/port network e	elements excen	t for UNE Coin	Port/Loop (Combinations				
	irst and additional Port nonrecurring charges apply to Not Curren															
	al NRCs may apply also and are categorized accordingly.	,			00111011104		.couring onarg	00 011411 20 11100			.g cac	, 00,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	3001.01.01			
2-WIRE	VOICE GRADE LOOP WITH 2-WIRE LINE PORT (BUS)															
	()															
			i i													
2-Wire	Voice Grade Line Port (Bus)															
	Mississippi Extended Local Dialing Port without Caller ID				UEPWK											
	capability			UEPBX		1.23	40.31	19.84	24.90	6.58		15.75				
	Incoming Only without Caller ID capability			UEPBX	UEPBE	1.23	40.31	19.84	24.90	6.58		15.75				